FUNDAMENTALS OF Commission DATABASE SYSTEMS

FUNDAMENTALS OF Fourth Edition DATABASE SYSTEMS



Ramez Elmasri

Department of Computer Science Engineering University of Texas at Arlington

Shamkant B. Navathe

Collescof Company Georgia Institute of Technology



Boston, Say Francisco, New Y H. Tornicu, Econo, Nyeney, Tokso, Singupoy, Midral Mesneot, ity, Marach, Pens, Crye, Jowr, Hong Koug, Marakal Spansoning Ecitor Project Editor: Senary Pusdaction Supervision Pusdaction Services: Cove: Designer: Markeoing Manager Senar Markening Coordinator Prim Bayer: Marte Snarez-Rivas Kalbernie Handtoman Junet Silvern Argosy Poldislang Beth Anderson Nathan Schultz Cas'y Hersliman Caroline Fell

Cover image () 2003 Digital Vision

Access the Satest information about Addison-Wesky infostional our World Wide Web site: http://www.uk.com/cs

Figure 12.14 is a logical data model diagram demition in Rational Rose¹. Figure 12.15 is a graphisub-late model diagram in Rational Rose¹. Figure 12.17 is the compton database class diagram drawn in Rational Rose¹. IBM¹ has accurred Rational Rose¹.

Alany minimenks-ignations used by normal characteristical selfers in distinguishment products are claiment as (galeryanks, Where those designations appear in this back hand Ackleson Wesley was avoide of a trademark charge the designations have been priored or initial captor will styrs.

The programs and applications pre-ented in this bank have been reclifted for their instructional value. They have been tested with care, but are not guaranteed for any particular pulpose. The pulpisher does not often any warranties or representations, nor does it acceptionly helpfiltes with respect to the programs or applications.

labrary of Congress Cataloging in Publication Data

Emissir, Ramez
 Fundamentals of detabase systems / Rumez Elmasti, Shankani B
 Scabe — 4th ed
 (a.s.0)
 tochtles http://opticul.org/geographical.iodex,
 ISBN 0-321-12226-7
 Database monagement, T. Navathe, Sham, H. Tuby,

QA06.9 [33237 2003 005.74 [de21

2003057731

ISBN 0-321-12226-7

Egent contation an obtaining permission for the use of material firm this work, please submit a written request to Pearson Education. The , Brebis and Contaters Department, 35 Arbograp St., Saar 300, Bretsan, MA, 021 Justic fax organization 617, 848, 2047

Copyright (12004 by Pearson Education, loc

All rights reserved. No part of this publication may be reproduced, stored in a removal system, so transmitted, many form or by my incass, excitomic, mechanical, photocrypying, recombing, or other wise, without the print written germission of the publisher. Printed in the United States of America,

0.2.3.4.5.6.7.8.9.10 HT+ 06050403

To Amalia with low R. E.

To my mother Vijaya and wife Artino for their love and support S. B. N.

Preface

This book numedates the fundamental concepts necessary for designing, asing and in plenenting database systems and applications. One presentations stresses the fundamentals of database moduling and design, the bingnases and facilities provided by the database monagement systems, and system implementation techniques. The book is meant to be used as a newthook for a one for two semister coarse in database systems of the junior sequence gradience level, and is a meaninee book. We assume that the readers are fundar with elementary programming, and data similarities concepts and that they have back some exposure to the bosts computer organization.

We start at Part 1 with an introduction and a presentation of the basic concepts and terminology, and database concepts of modeling protoples. We conclude the book at East-7 and 8 with an out-selection to concepting tech tologies, such as data mining, XML seconds, and Web databases. Along the war—in Parts 2 through 6—we provide on indepth treatment of the most important aspects of database functioned.

The following key fortutes are uncluded in the fourth ed time

- The entire book follows a self-contained, they ble organization that can be tailored to individual needs.
- Coverage of data modeling trow includes both the TR model and UST
- A new advanced set chapter with instead on set: programming techniques, such as (18) and softe(1).

- Evol examples summing throughout the book—called COMPANY and UNIVER-STEP—allow the reader co-compare different approaches that use the same aspheration.
- Coverage has been updated era security, mobile databases. Obs. and Genetice data management.
- A new shapter on XML and Internet databases
- A new chapter on data numming.
- A seguineant revision of the supplements to include a rebust set of materials for instructors and students, and an online case study.

Main Differences from the Third Edition

There are several organization dicharges in the fourth edition, as well as some important new chapters. The input charges are as follows:

- The chapters on the organizations and indexing (Chapters 5 and 6 in the third educion) have been moved to Part 4, and are now Chapters 15 and 14. Fart 4 also recludes Chapters 15 and 16 on query processing and eptimizer on, and physical database design and running (this corresponds to Chapter 18 and sections 16.3-16.4 of the third educion).
- The relational model coversee has been reorganized and updated in Part 2, whapter 5 covers relational model concepts and constraints. The material of relational alcohomaral calculus is now together in Chapter 0. Relational database design using ER-toorelational and EER-tropelational mapping is in Chapter 7. SQL is covered as Chapters 5 and 9, with the new antignal in SQL programming techniques in a choice 9.3 choogh 9.6.
- Part 3 covers dualwase design theory and methodology. Chapters 10 and 13 on normal matter theory correspond to Chapters 14 and 15 of the third edition. Chapter 12 on practical database design has been updated to include more UML coverage.
- The chapters or, transactions, concurrence control, and recovery (19, 30, 21 m, the duild dution) are now Chapters 17, 18, and 19 in Part 5.
- The duptors on observationted concerns, ODMCi (Mierr model, and objecture attend screens {01 = 17 = 13 in the three edition) are now 22, 21, and 22 an Part 6. Chapter 22 has been corresponded and opdated.
- Chapters 10 and 17 of the thad edition have been dr #ped. The material or clientserver inclutectures has been merged into Chapters 2 and 25.
- The chapters on secondy, enhanced models (active, temporal, spatial, malterredia), and distributed databases (Chapters 22, 23, 24 in the third edition) are new 23, 24, and 25 im Part 7. The secondy chapter has been updated. Chapter 25 of the third edition on distributed databases has been merged into Chapter 24, and is new section 24.5.

- Chapter 26 is a new chapter on XML (eXtended Markap Longuage), and how it is related to accessing relational databases over the 'nto met.
- five notenal conduct mining and data watchoosing (Chapter 26 of the third edition) has been separated into two chapters. Of optimized and attaining has been expanded and epideted.

Contents of This Edition

Part 1 describes the basic concepts necessary for a good inderstanding of diabase design and singlementation, as well as the conceptual modeling rechriques used in database systums. Chapters 1 and 2 miniculae databases, their typical users, and 00005 concepts, icrminology and architectures. In Chapter 3, the concepts of the Entity-Relationship (BR) insolid and BC diagrams are preserved and used to illustrate conceptual database design. Chapter 4 focuses on databasement and sent mite data modeling concepts and extends the Ek model to incorporate these ideas, leading to the enhanced (BC) (FEC) data model and EBC diagrams. The concepts presented include subclasses, specialitation, generalitation, and unical types (categories). The notation for the class diagrams of CMI are also introduced in Chapters 4 and 4.

Part 2 describes the relational data model and relational (PMS). Chapter 5 describes the basic relational model, its integrity constructs and update operations. Chapter 6 describes the operations of the relation if algebra and outroduces the relational calculus. Chapter 7 discusses relational database dusing using 38 and (TR-foredational mapping Chapter 8 grees a detailed overview of the soft language, covering the soft standard, which is implemented in most relational systems. Chapter 9 covers 520 programming top es in 5 is soft data, and soft 0.

Part i covers several region related to database design. Chapters 10 and 11 cover the tomolisms, theories, and algorith incloveloped for the relational database design benomination. This material and algorith incloveloped for the relational database design benominations of relations. Step-by-step intornet normalization is presented in Chapter 10, and teleponter design algorith instate given in Chapter 11, which also defines other types of dependencies of the transitional design algorith instate given in Chapter 11, which also defines other types of dependencies, such as multivalised and tom dependencies. Chapter 12 presents at overview of the different places of the database design process for mediom-stied and large ripheations, using Usita.

4. Part 4 scars with a description of the physical file structures and access methods used in database systems. On pter 13 describes primary methods of organizing files efficients on disk, methoding static and dynamic history. Unapter 14 describes indexing techniques for files and oding B free and B (circle data structures and grid files. Chapter 15 introduces the basics of query processing and optimization, and Chapter 16 discusses physical database design and uning.

Part 3 discusses non-action processing, concurrency control, and recovery rechsupport including discussion of how these concepts are need to set Part C gives a comptchensive anticulaction to object databases and object-relational systems. Chapter 22 interstices object oriented concrets. Chapter 21 gives a database object model and in associated oble and opt languages. Chapter 22 describes how relational databases are being extended to useful object-oriented concretion and prosents the features of object-relational systems, as well as giving an overview existing of the respectively induction of the teames of the set 3 standard, and the instead relational data model.

Firs 7 and 8 cover information addanced topes. Chapter 23 goes an overview of database sectors and conformation including the SQL commands to GRANT and EFVOSE provileges and expanded secondly on sectors concepts such as encorption roles, and flow control. Chapter 24 acrospheres sectors concepts such as encorption roles, and flow control. Chapter 24 acrospheres and tragges, temporal, spatial, malinimedia, and deductive databases. Chapter 25 gives an introduction to distributed datatimedia, and deductive databases. Chapter 25 gives an introduction to distributed datatimedia, and deductive databases. Chapter 25 gives an introduction to distributed datatimedia, and deductive databases. Chapter 25 gives an introduction to distributed datatimedia and deductive databases. Chapter 25 gives an introduction to distributed datatimedia, and the three-tier cheric-server inclinectory. Chapter 26 is a new chapter on XME (eXtended Markup Language). It has discusses the differences between structured, semistructured and instructored models, then presents XME concepts, and heally compares the XME model to induce a database. Chapter 25 on data mixing has been expanded and opdated. Chapter 28 introduces data wateboasing concepts. Finally, Chapter 29 gives pure databases to the topics of models databases, multimedia databases, GIS (Geographic Information Systems), and vienome data management in biomom toes.

Appendix A gives in number of chemistry collegitor matter notations for doploying a conceptual 50 or FER schemo. These new be satisficated for the notation we use of the instructor solutions as: Appendix C gives some important physical parameters of disks. Appendixes B, E, and Fore on the web site. Appendix B is a new rose study that follows the design and implementation of a book-cere's database. Appendixes E and F cover leaver database systems, based on the recoverk and hierarchical database models. These have been used for over thirry grass as a basis for a non-existing commercial database applications and transactions processing systems and will take decides to replace completely. We consider a important to every studens of database in increment to these long-standing approaches full chapterfrom the theory con-by found on the web site for this order of

Guidelines for Using This Book

Then are more different ways to reach a database course. The chapters in Parts 1 through 5 can be used in an introductory course on database systems in the order that they are given or in the protoned order of each individual instructor. Selected chapters and sections may be first out, and the instructor can odd other chapters from the rest of the book depending on the emphasis of the course. At the cod of each chapter's opening sections we list sections that are condicates for bone left out whenever a less detailed discussion of the topic in a particular chapter is desired. We suggest covering up to Chapter 14 in an introductory database course and use fielded as the desired covering. For an emphasis en system implementation or implicit chapter trein Farts 4 and 5 can be included.

Chapters 3 and 4 which cover conceptual nacdeling using the F8 and i +3 models, are important for a good conceptual understanding of databases. However, they may be parnative covered covered later insectorise, or even left out at the emphasis is on 19843 implementation. Chapters 13 and 14 an the organizations and indexing new also be covered early one later, or even left our if the emphasis is on database models and languages, for students who have already taken a course on the organization, parts of these chapters could be assigned as reading material or some exercises may be assigned to review the encopts.

A rotal lite-type database design and implementation project covers conceptual design (Chapters V and 4), data model impping (Chapter 7), nermalization (Chapter 15) and implementation in sqt (Chapter 9). Additional documentation on the specific statistic would be required.

The book has been written such a mis possible to cover topics in a variety of orders. The chair included here shows the major dependence is between chapters. As the diagram docarates, it is possible to start with several difference peaceful wing the first two introdactory chapters. Although the chair may seem complex, it is important to more that if the chapters are covered in order, the dependencies are not lost. The chart can be consolved by instructors realing rates on alternative order of traveniation.



Fer inside coseniester course hased on the book come do opters car, bu issignost as reading material, Daits 4, 7, and 8 c in the cose-identified for such an assignment. The book contails

N

be used for a two-seriester sequence. The first course, "Introduction to Database Denet/ Systems," at the sophian ore, janzan or sensor level, could cover most of Ultapters 4 to 74. The second course, "Database Desired and huplementation 'Techniques," at the sensor or first-year griduate level, canceover Chapters 15 to 28. Chapters from Parts 7 and 8 can be used selectively in either seniester, and material describing the DBMS invaliable to the strudents of the legisl institution can be covered in addition to the material in the book.

Supplemental Materials

The supplanents to this book have been significantly revised. With Addison-Wesley's Patabase Place there is a robust set of interactive reference materials to help anderes with their study of modeling, normalization, and SQL Each rutorial asks statents to solve problems (each as writing on SQL query, drinking on ER diagram or controlizing a relation), and then provides useful feedback based on the student's solution. Addison-Wesley's Database Place helps anderes material technologies are non-concepts of all database courses. For more information was invested hashes place.

In addition the following supplements are available to all readers of this book at assessment support.

- Additional conteral This includes a new Cose 5r dy on the design and implementation of a bookstore's database is well as exapters from previous editions that are that included in the foorth edition.
- A set of RowerPoint fecture notes

A solutions notical of also covalable to qualified instructors. Please contact year local Addison-Wesley sides representative, or send consul to awe-offer commutation on here to access it.

Acknowledgements

It is a great pleasure for us to beknowledge the assistance and contributions of a large musben of undradials to this effort. First, we would also to thank our editors, Marte Suarez Rivers, Katherine Hammunan, Daniel Rausch, and Juher Silven. In particular we would file to acknowledge the efforts and help of Katherine Hammunan, our primary contrast for the fourth edition. We would like to acknowledge also those persons who have contributed to the toarth edition. We would like to acknowledge also those persons who have contributed to the toarth edition. We appreciated the contributions of the following reviewers. Phil Bern hard, Flends Tech, Zhergyun Chen, Chenesery & Nelwashare Onaue; Jan Chengelei, Universery of Baffalo, Halon Ferbarosmanoglo, Oher, Sare University, Len Fisk, California Sing Chenesery, Chen, William Henkley, Karge State University, Alt R. Forsen, Peter Sare University, Jones View, Vipo, Kuman, University of Missian Kastas Cay, Peter, Sloval, Ben Granes University, Jones Vipo, Kuman, University of Missian Kastas Cay, Peter, Sloval, Ben Granes University, Jones Vipo, Kuman, University, Jenes Jone, Jenes Jenes, Jenes Davie, University, Alt R. Forsen, Peter Sare University, Jenes Teal, Peter Sare University, Jenes Jenes, Jenes Care State University, Jenes Jenes Care State University, Jenes Jen

Rainet Ebnash woold like to though his stude six listed Han. Babak Hajaba, Jack La, O'r eley Lo Ando Swath, and Seeven Wu who concubried to the raitenul in Chapter He would also like to acknowledge the support provided by the University of Texas at Arangian.

Shori Navashe would like to acknowledge Dur Forsethe and the following students or Georgia Tech. Weimm Fene, Argeshianan Gura, Abror 19, Haque, Bir Liu, Ying Liu, Weivia Xie and Wogen Yee.

We would like to repeat our thanks to those who have reviewed and courtibured to encirous editions of Fundamentals of Unabase Systems. For the first edition these and oduals metode Afan Apt tedror). Dan Baroya Sectt, Assoning Damis Heintbinger, Julia Hodees, Yannis Ioanodis, Jun Lason, Deenis McLeod, Per-Ake Larson, Rohal Parel, Nicholas Roussopoulos, David Stemple, Michael Stomebraker, Frank, Jonpa, and Kyu-Young Whang: for the second edition they include Pan Jordanstad (editor). Rafi Almied, Anaomo Albano, David Beech, Jose Blakeley, Pario- Chrysanthis, Seanne Dictrich, Vic Chorpadey, Coras Graele, Eric Hanson, Jurguk L. Kun, Roger King, Vian Kouramajian. V pe Komar, Juhn Lawrher, Sirapy Mancharda, Toshum Muroura, Inderpal Muruck, Ed Unnourska, Unrich Pathak, Raghu Ramakrishnan, Ed Robertson, Eugene Sheng, David Story, Attriange Wusderr, and Stan Zdonick. For the shind edition they include Steamer Diemch, Ed Omiccinsko, Rań Ahmed, Francias Barcolhon, Jose Blakeley, Rick Cattell, Any Chervenik, David W. Embley, Henry A. Ethnget, Contidos begans, Dan Forseth-Fashad Foroulu, Michae, Franklie, Sreenth Copinoila Coset, Craefe, Richord Hulb Sosh Laedia, Ramestek, Kame, Hansh Kotbaei, Meny Kumut, Totosio Lima, Ramon A Muta Joledo, Jack McCine, Dorma McLeod, Rokot Missioni, Mbidi Mersi, M. Natavom-wang Carlas Ordenez, Juan Peckhan, Betty Saliheng Mung-Chen Shan, Jungang Sur, Rajshekhar Sonderraman, Arnymdan Ngorasany, and Eanlia E. Villareal.

Last but not least, not grarefully acknowledge the support, encouragement, and primere of our families.

8 F N 8 N

Contents

PART 1 INTRODUCTION AND CONCEPTUAL MODELING

CHAPTER 1 Databases and Database Users 3

- 1.1 Introduction 4
- 1.2 An Example 6
- Characteristics of the Database Approach = -8.
- 1.4 Actors on the Scene 12
- 1.5 Workers behind the Scene 14
- Advantages of Using the DBMS Approach 15
- A Brief History of Database Applications 20
- 1.8 When Not to Use a DBMS = 23
- 1.9 Summary 23 Review Questions 23 Exercises 24 Selected Bibliography 24

CHAPTER 2 Database System Concepts and Architecture 25

- Data Models, Schemas, and Instances 26
- 2.2 Three Schema Architecture and Data Independence 29
- 2.3 Database Languages and Interfaces 32.
- The Database System Environment 35
- Centralized and Client/Server Architectures for DBMSs = 38
- Classification of Dutabase Management Systems 43
- 2.7 Summary 45 Review Questions 46 Exercises 46 Selected Bibliography 47

CHAPTER 3 Data Modeling Using the Entity-Relationship Model 49

- Using High-Level Conception Data Models for Database Design 50
- 3.2 An Example Dirabase Application 52
- 3.3 Entity Types, Entity Sets, Attributes, and Keys = 53
- 3.4 Relationship Types, Relationship Sets, Roles, and Structural Constraints 61
- Weak Entity Types 68
- 3.6 Refining the FK Design for the COMPANY Dat doase 69.
- 3.4 ER Diogramis, Norming Conventions, and Design Issues 50
- 3.8 Notation for UML Class Diagrams 74

3.9 Summary 77 Review Questions 78 Exercises 78 Selected Bibliography 83

CHAPTER 4 Enhanced Entity-Relationship and UML Modeling 85

- 4.1 Subclasses, Seperclasses, and Inheritance 86
- Specialization and Generalization 88
- Constraints and Characteristics of Specialization and Ocneralization 91
- 4.4 Modeling of UNION Types Using Categories 98
- 4.5 An Example UNIVERSITY EUR Schema and Formal Definitions for the RER Model 101

١

- 4.6 Representing Specialization/Generalization and Inheritance in USU Class Diagrams 104
- 4.7 Relationship Types of Degree Higher Than Two 105
- 4.8 Data Abstraction, Knowledge Representation, and Ontology Concepts 112
- 4.9 Summary 115
 Review Questions 116
 Exercises 117
 Selected Bibliography 121

PART 2 RELATIONAL MODEL: CONCEPTS, CONSTRAINTS, LANGUAGES, DESIGN, AND PROGRAMMING

CHAPTER 5 The Relational Data Model and Relational Database Constraints 125

- 5.1 Relational Model Concepts 126
- 5.2 Relational Model Constraints and Relational Database Schemas 132
- 5.3 Update Operations and Dealing with Constraint Violations 140
- 5.4 Summary 143
 Review Questions 144
 Exercises 144
 Selected Bibliography 147

CHAPTER 6 The Relational Algebra and Relational Calculus 149

- 6.1 Unary Relational Operations: SELECT and PROJECT 1151
- 6.2 Relation & Algebra Operations from Set Theory 155.
- 6.3 Binary Relational Operations, JOIN and DIVISION 158
- 6.4 Additional Relational Operations 165
- 6.5 Examples of Queries in Relational Algebra 171
- 6.6 The Taple Relational Calculus 173.
- 6.7 The Domain Relational Calculus 181
- 0.8 Summary 184 Review Questions 185 Exercises 186 Selected Bibliography 189

| СНАР 7.1 | TER 7 Relational Database Design by ER- and EER-to-Relational Mapping 191 Relational Extabase Design Using ER-to-Relational Maximum 192 |
|-------------|--|
| 7.2 | Marping 192 Marping EER Model Constructs to Relations 199 |
| 7.3 | Summary 203 |
| | Review Questions 204 |
| | hveraises 204 |
| | Schered Bibliography 205 |
| СНАР | TER 8 SQL-99: Schema Definition, |
| | Basic Constraints, and Queries 207 |
| 8.1 | SQL Data Definition and Data Types 209 |
| 8.2 | Specifying Basic Constraints in SQL = 213 |
| 8.3 | Schema Change Statements in SQL 217 |
| 8.4 | Basic Queries in SQL 218 |
| 8.5 | More Complex SQL Queries 229 |
| ກ.ບ ປີ 5 | A Horizond Researce of extrements in SQL 245 |
| 5.5 | Additional regiures of SQR 246 Summary 240 |
| 0.0 | Review Onestions 751 |
| | Exercises 251 |
| | Selected Bibliography 252 |
| | |
| CHAP | TER 9 More SQL: Assertions, Views, and Programming |
| | Techniques 255 |
| 9.1 | Specifying General Constraints as Assertions 250 |
| 9.Z | Views (Virtual Tables) in SQL 257 |
| 9.3 | Database Programming: Issues and Techniques 261 |
| 9.4 | Embedded SQL, Dynamic SQL, and SQLJ 264 |
| 9.5 | Database Programming with Function Calls: SQL/CU and name 275 |
| NA. | j. 19. – Kara Darabara Sranal Drazadana, wakisin 1985 – 1984 |
| N.7 | Summers 257 |
| | Review Questions 287 |
| | Exercises 287 |
| | Selected Bibliography 289 |

PART 3 DATABASE DESIGN THEORY AND METHODOLOGY

CHAPTER 10 Functional Dependencies and Normalization for Relational Databases 293 10.1 Informal Design Guidelines for Relation Schemas 2.25 10.2 Functional Dependencies 3,74 12.3 Normal Forms Based on Primary Keys 317 12.4 General Definitions of Second and Thurd Normal Forms 320 10.5 Boyce-Codd Normal Form 304 126 Summary 326 327Review Questions Exercises 328Selected Bibliography اذذ CHAPTER 11 Relational Database Design Algorithms and Further Dependencies 333 11.1 Properties of Relational Decompositions ή ή-11.2 Algorithms for Relational Darabase Schema Design 54C 11.3 Multivalued Dependencies and Fourth Normal Form 147 11.4 Juin Dependencies and Fifth Normal Form 15 i 11.5 Inclusion Dependencies 354 11.6 Other Dependencies and Normal Forms 355 11.7 Summary. 357 Review Questions 358 Exercises 558 Selected Bibliography 360 CHAPTER 12 Practical Database Design Methodology and Use of UMI. Diagrams 361 12.1. The Role of Information Systems in Organizations 362 12.2 The Database Design and Implementation Process 300 12.3 Use of COL Diagrams as an Aid to Database Design Specification: 355 12.4 Rational Rose, A UML Based Design Tool 395 12.5 Automated Database Design Tools 402 12.6 Summary. 404 Review Questions 405 Selected Bibliography 406

PART 4 DATA STORAGE, INDEXING, QUERY PROCESSING, AND PHYSICAL DESIGN

CHAPTER 13 Disk Storage, Basic File Structures, and Hashing 411

- 13.1 Introduction. 412 13.2 Secondary Storage Devices 415 13.3 Buttering of Blocks. 421 13.4 Placing File Records on Disk 422 13.5 Operations on Files 427 13.6 Files of Unordered Records (Heap Files). 430 13.7 Files of Ordered Records (Sorred Files). 431 13.8 Hashing Techniques. 434 13.9 Other Primary File Organizations 442 13.10 Paralleliting Disk Access Using RAD Technology. 443 [3,1] Storage Area Networks. 447 13.12 Summary 440 Review Questions 150 Exercises. 451 Selected Bibhography – 454 CHAPTER 14 Indexing Structures for Files 455 14.1 Types of Single-Level Ordered Indexes 456
- 14.2 Multilevel Indexes 464
- 14.3 Dynamic Multilevel Indexes Using B-Trees and BT-Trees 469
- 14.4 Indexes on Multiple Keys 483
- 14.5 Other Types of Indexes 485
- 14.6 Summary 486
 Review Questions 487
 Exercises 488
 Selected Bibliography 490

CHAPTER 15 Algorithms for Query Processing and Optimization 493

- Translaring SQL Queries into Relational Algebra 495
- 15.2 Algorithms for External Sorting 496
- 15.3 Algorithms for SELECT and JOLS Operations 498
- 15.4 Algorithms for PROJECT and SET Operations = 508

584

- 15.5 Implementing Aggregate Operations and Outer Joins 509
- 15.6 Combining Operations Using Pipelming 511
- 15.7 Using Heuristics in Query Optimization 512
- 15.8 Using Selectivity and Cost Estimates in Query Optimization = 523
- [5.9] Overview of Query Optimization in OSACLE 532
- 15.10 Semantic Query Optimization 533
- 15.11 Summary 534
 Review Questions 534
 Exercises 535
 Selected Bibliography 536

CHAPTER 16 Practical Database Design and Tuning 537

- 16.1 Physical Database Design in Relational Databases 537
- 16.7 An Overview of Database Tuning in Relational Systems 541
- 16.3 Summary 547 Review Questions 547 Selected Bibliography 548

PART 5 TRANSACTION PROCESSING CONCEPTS

CHAPTER 17 Introduction to Transaction Processing Concepts and Theory 551

- 17.1 Introduction to Transaction Processing 552
- 17.2 Transaction and System Concepts 559.
- 17.3 Desirable Properties of Transactions 562
- 17.4 Characterizing Schedules Based on Recoverability 563
- 17.5 Characterizing Schedules Based on Serializability 566.
- 17.6 Transaction Support in SQL = 576
- 15.7 Summary 578 Review Questions 579 Exercises 580 Selected Bibliography 581

CHAPTER 18 Concurrency Control Techniques 583

- 18.1 Two-Phase Locking Techniques for Concurrency Control
- 18.2 Concurrency Control Based on Timestomp Ordering 594.
- 18.3 Multiversion Concurrency Control Techniques 596
- 18.4 Validation (Optimistic) Concurrency Control Techniques 599

- [18.5] Granularity of Data Items and Midriple Granularity Locking 900
- 13.6 Using Locks for Concurrency Control in Indexes 605
- 18.7 Other Concurrency Control Issues 60%
- 13.8 Summary 607
 Review Questions 608
 Exercises 609
 Selected Bibliography 609

CHAPTER 19 Database Recovery Techniques 611

- 19.1 Recovery Concepts 612
- 19.2 Recovery Techniques Based on Deferred Update 618.
- 19.3 Recovery Techniques Based on Immediate Update 622
- D.4 Shadow Paging 624
- 19.5 The ARIES Recovery Algorithm 625
- 19.6 Recovery in Multidatabase Systems 629
- 19.7 Database Backup and Recovery from Catastrophic Failures 660
- 19.8 Simimary 631 Review Questions 632 Exercises 633 Selected Bibliography 635

PART 6 OBJECT AND OBJECT-RELATIONAL DATABASES

CHAPTER 20 Concepts for Object Databases 639

- 30.1 Overview of Object-Oriented Concepts 641
- 20.3 Object Identity, Object Structure, and Type Constructors 643

20.3 Encopsulation of Operations, Methods, and Persistence 649.

- 20.4 Type and Class Hierarchies and Inheritance 654
- 20.5 Complex Objects 657
- 20.6 Other Objected Oriented Concepts 59
- 20.7 Summary 662 Review Questions 663 Esercises 664 Selected Bibliography 664

CHAPTER 21 Object Database Standards, Languages, and Design 665

21.1 Overview of the Object Model of ODMG = 666

- 21.2 The Object Definition Language ODE 679.
- 21.3 The Object Query Language OQ. 684
- 21.4 Overview of the C++ Language Binding 693
- 21.5 Object Database Conceptual Design 694
- 21.6 Summary 697
 Review Questions 698
 Exercises 698
 Selected Bibliography 699

CHAPTER 22 Object-Relational and Extended-Relational Systems 701

- 22.1 Overview of SQL and Its Object-Relational Features 702
- 22.2 Evolution and Current Trends of Database Technology 709
- 21.3 The Informix Universal Server 711
- 22.4 Object-Relational Features of Oracle 8 721
- 23.5 Implementation and Related Issues for Extended Type Systems 724
- 22 b. The Nestod Relational Model 725
- 22.7 Summary 777 Selected Bibliography 728.

PART 7 FURTHER TOPICS

CHAPTER 23 Database Security and Authorization 731

- 25.1 (pproduction to Database Security Issues) 732
- 23.2 Discretionary Access Control Based on Granting and Recoking Privileges 735
- 23.3 Mandatory Access Control and Role-Rused Access Control for Multilevel Scenity — 740
- 23.4 Introduction to Statistical Database Security 746.
- 23.5 Introduction to Flow Control 747
- 23.6 Encryption and Public Key Infrastructures 749.
- 23.7 Sommary 751
 Review Questions 752
 Exercises 753
 Selected Bibliography 753

CHAPTER 24 Enhanced Data Models for Advanced Applications 755

- 24.1 Active Database Concepts and Triggers 757
- 24.2 Temporal Database Concepts 767
- 34.3 Multimedia Databases 780
- 24.4 Introduction to Deductive Databases 784
- 24.5 Summary 797 Review Questions 797 Exercises 798 Selected Bibliography 801

CHAPTER 25 Distributed Databases and Client-Server Architectures 803

- 25.1 Distributed Database Concepts 804
- 25.2 Data Fragmentation, Replication, and Allocation Techniques for Distributed Database Design — 610
- 25.3 Types of Distributed Database Systems 815
- 25.4 Query Processing in Distributed Databases 818
- 25.5 Overview of Concurrency Control and Recovery in Distributed Databases 824
- 25.6 An Overview of 3 Ther Client-Server Architecture 827
- 25.7 Distributed Databases in Oracle 830
- 25.8 Sommary 832 Review Questions 833 Exercises 834 Selected Bibliography 835

PART 8 EMERGING TECHNOLOGIES

CHAPTER 26 XML and Internet Databases 841

- 20.1 Structured, Semistructured, and Unstructured Data 842
- 20.7 XMI Hierarchic d (Tree) Data Model 846
- 20.3 XMI Documents, DTD, and XMI Schema 848
- 26.5 XML Querying 862
- 26.6 Summary 865 Review Questions 865 Exercises 866 Selected Bibliography 866

CHAPTER 27 Data Mining Concepts 867 27.1 Overview of Data Mining Technology 865 27.2 Association Rules 87 i 27.3 Ciassification. 882 27.4 Clustering 885 27.5 Approaches to Other Data Mining Problems 338 27.6 Applications of Data Mining 89 F 27.7 Commercial Data Mining Tools. 891 27.8 Summary 894 Review Onestions 394 Exercises. 895 Selected Bibliography 396 CHAPTER 28 Overview of Data Warehousing and OLAP 899

- 28.1 Introduction, Definitions, and Terminology 900.
- 28.2 Characteristics of Data Worehouses 901
- 28.3 Data Modeline for Data Watchouses 902
- 28.4 Building a Data Warehouse 907
- 25.5 Typical Functionality of a Data Warehouse 910
- 28.6 Data Warehouse Versus Views 911
- 28.7 Problems and Open Issues in Data Warehouses 912
- 28.8 Summary 913 Review Questions 914 Selected Bibliography 914

CHAPTER 29 Emerging Database Technologies and Applications 915

- 29.1 Mobile Databases 916
- 29.2 Multimedia Databases 923
- 29.3 Geographic Information Systems 930
- 29.4 Genome Data Managemenr 936

| APPENDIX A Alternative Diagrammatic Notations 947 |
|---|
| APPENDIX B Database Design and Application Implementation Case Study-located on the we |
| APPENDIX C Parameters of Disks 951 |
| APPENDIX D. Overview of the QBE Language 955 |
| APPENDIX E Hierarchical Data Model-located on the web |
| APPENDIX F Network Data Model—located on the web |
| Selected Bibliography 963 |
| Index 1009 |





CONCEPTUAL MODELING





Databases and Database Users



Databases and dotabase systems more become on essential component of evendas life in movem source. In the consecutive day, most of insensionitier several sectivities that its objects some interactions with indatabase. For example, if we go to the bank to deposit of withcitize finish, if we make a horefor arrhite reservation, if we access a computerized lift ary catabastic search for a brohographic in incomplete two bay some iterim—such as a book, toy, or computer—from an internet condor through its Web page, a since- no that on activatics will involve someone or some computer program increasing a database. Even purchasing recession a supermarker movindays in many cases involves an automatic update of the analise that keeps the ancentory of supermarket duras.

These printactions on examples of what we may call traditional database applications, in which most of the programmon that is sound and accessed is influe rescal or maneral. In the past few years, advances in its biology have been hinding to example interpret distributes systems. Multimedia databases can now some printes, video clyps, and sound messages. Gengraphic information systems (G(S) can some and malvie maps, weather data, and satellite images. Data warehouses and unline analytical processing 101 API systems are used in many comparises to extract and active database technology is used in controlling mainstrial and manufacturing processes. And database technology is used in controlling mainstrial and manufacturing processes. And database south rechniques are being upplied to the World Wide Web to improve the -, relign momentum that is needed by users to wave process.

To anderstand the toydament do of database technology, however, we no sustant from the basics of madimonal database applications, so an Section 1.1 of this charter we define what a database is and there we give dematrons of other basic terms. In Section 1.2, we procede a simple curvatative database example to allastrate our docussion. Section 1.3 describes some of the main characteristics of database systems, and Sections 1.4 and 1.3 categorize the types of personnel whose jobs modified using and interacting with database systems. Sections 1.6, 1.7, and 1.8 offer a more theorems for docussion of the various capabilities provided by database systems and discussions of the various capabilities provided by database systems and discuss some topical database applications. Section 1.9 summaries the clopper.

The made who desires indvia quick introduction to database systems can study Sections 1.1 almorgly 1.5, three-step or brow-webrough Sections 1.6 through 1.8 and go on on Chapter 2

1.1 INTRODUCTION

Databases accludatabase technology are howing a major in most on the growing use of computers, it is that to say that databases place occurs of role in object all areas where comparers are used, uncluding business, electronic confinence, erganeering, freed one, low education, and library science, to none a tew. The word database is in such common use that we must beem by defining what a database is. On the unit, det timo is quite general

A database is a collection of related cluto. By data, we mean known focts that can be recerded and that have implicit meaning. For example, consider the names, telephone manabers, and addresses of the prople year know. You new have recorded this shift in an indexed address book, or you may have stored at own it hard drive, using a personal computer and software such as Microsoft. Access or Excel. This is a collection of related data with an implicit meaning and hence is a database.

The proceeding derivation of database is quite generali for example, we may consider the collection of words that make up this page of text to be related data and hence to constrain a database. However, the common use of the term database is usually more respired. A database his the following implicit properties.

- A database represents some aspect of the real world, sometimes all of the minjworld on the universe of discourse (UoD). Up and she the miniworld are reflected in the shakes.
- A database is a logically coherent collection of data with some (pherent meaning) (and on assoringent of Articlamor correctly be released to as a database)
- A database is designed, built, and populated with data for a specific purpose. It cas an intended group of users and some preconce vid applications in which these users are interested.

4. We will one intervantioking as both singular and plot does us a monominal data self-term region test or doleter in provident or singular or plot doleter to data larghshow as used or ly the plot alwater is result of singular. In other words, a database has some source from which data is derived, some degrie of interaction with events in the real world, and an audience that is actively interested in the contents of the database.

A doublise can be of any site and of carving complexite. For example, the list of names and addresses referred to earlier may consist at only a few bundred records, each with a simple structure. On the other band, the computerized curdog of a large bloars may contain half to multical entries organized usalet different categories—by primary authors last name, by subject, by book title—with each category organized in alphabetic oder. A database of even greater site and complexity is maintained by the Internal Reve us Service tokkey track of the tax forms filed by U.S. taxpayers If we assume that there are 100 million taxpayers and it each taxpayer files an average of free forms build uppresentely 422 characters of information per form, we would get a database of 122 × 10° × 400 × 5 characters (bytes) of information. If the IRS keeps the past three returns for , whittay addition to the current return, we would get a database of 8 × 10⁻¹ beros (82) graphytes). This hage amount of information must be organized and managed so that uses can spin b (or minery, and update the data as predict).

A doublese may be generated and many-aned many-ally or many be comporented, for example, a library cord coraleg is a dirichose that may be created and monitoried manually. A computerized database may be created and maintained earbier by a group of operations programs written specifically for that task or two database menagement statem. Of course, we are only concerned with computerized databases in the book.

A database management system (DBMS) is a collection of programs that enables users to court and maintaine a database. The courts is hence a general-people ophrace system that tachtates the processes of defensigle contracting manipalianage and sharing databases among various users and applications. **Defining a** database involves specifying the data types, structures, and constraints for the data to be stored in the database. **Constructing** the database is the process of storing the database involves specifying reaching the database is the process of storing the database includes such timetions reaching the database is the process of storing the database includes such timetions is gotiving the slowly we to terrieve specific data, opdating the database includes such timetions is during the slowly we to terrieve specific data, opdating the database includes such timetions in the migroord, and generating reports from the data. **Sharing** a database allows moltiple asymptotic programs to agrees the database generation.

Other important functions provided by the DBMS includes proceeding the database and interactional in over a long period of time. Protection includes both system protection against hardware or software multimetion for crushest, and sectory protection, gainst unanthorized or mathematications access. A typical large database may have a life cycle of many versus of the DMS musc be able to maintain the database system by allowing the system to evolve as requirements change over time.

It is not necessary to use general pairose 1981s software to implement a computerized database. We could write out own set of polytanis to error and maintain the database, in other creating out own spagid-parison DBMs software. In other case whether we use a general purpose DBMS or not take usually have to deploy a considerable at ment of complex software. In fact, new DBMs sure very conclex software servers

To complete out in rial definitions, we will call the data use and 19918 software togethers database system. Figure 1.1 illustates some of the concepts we docubed softa-



FIGURE 1.1 A simplified database system case comment.

1.2 AN EXAMPLE

Let us consider a simple example that most readers may be further with a aktyFRSTTY database for monitoring maximum on concertaing subjects, courses, and grades in a anversity revironment, higher 11, shows the database structure and a tow sample data for such a database. The statistics is organized as five fires each of which stores data records of the same type for the statistics is organized as five fires, each of which stores data records of the same type for the statistics end of each of section of a course, the stores data on each course, the statistic fires data endeath section of a course, the useff data on each course, the statistic fires data endeath section of a course, the useff data on statistics the endes that students receive in the various sections they have completed, and the **PRURT** of the stores the procequises of each course.

To define this database, we mass specify the structure of the records of each tide by specifying the different types of data **elements** to be stored in each record on Figure 1.2, each Stripping incord includes data to represent the student's Name, StudentNumber, Class

^{3.} We use the term plantic on the base Atta commepting devely a pV scale dhate was responde that may optimate it be other of:

1.2 An Example 7

| STUDENT | Name ₁ | StudentNorman | Class | Мајол |
|---------|-------------------|---------------|-------|-------|
| | Scut | 17 | і I | 05 |
| | Brown | 8 | . × . | C9 |

| COUPSE | . Cour-alvane | CourseNumber | Cred #mains | - Department | ' |
|--------|---------------------------|--------------|-------------|-----------------|---|
| | Intro to Computer Science | CS1316 | ા | CS | : |
| | Data Directures | 050320 | | ÇŞ | |
| | Discrete Mathematics | MATH2419 | 2 | MATH | |
| | Dutabase | (\$3369 | . 2 | CS | |

| SECTION | Socionhilantilier | CourseMornber | Semistra | Yoar | 06000000 |
|---------|-------------------|---------------|----------|-----------|------------|
| | 95 | MA1H2410 | : Fal: | 06 | lung . |
| | 36 | C51310 | Fail | . 18 1 | Andeulos |
| | 102 | CS3320 | SENOL . | | Knall |
| | 0.2 | MATH2410 | , Fail | , az , | Chang . |
| | 119 | C51010 | , Fail | . 99 . | Anacison . |
| | 135 | 0602080 | . F 11 | . 92 . | Smrc |

| GRADE | REPORT | StudentNumber | SectionIder | bfier | | Grade |
|-------|--------|---------------|-------------|-------|---|-------|
| | | | | •• | • | |

..... . ..

| . 17 | 112 | C |
|------|-----|------|
| 17 | 119 | , C |
| Б | 65 | . P. |
| к | 92 | . A |
| Б | 102 | G |
| к | 135 | A |

| PREBEQUISITE CourseNur | mbor Preriou sceNumber | |
|------------------------|------------------------|--|
| 055360 | n <u>C6332</u> 3 | |
| 05336 | C MATH2410 | |
| C 5332 | M CS1510 | |

FIGURE 1.2 A database that stores student and course informations

its shourder 1, sophomore or 2.1.1.1. and Maior (numbered every norb, computer source or US, 1.1.1.2), each course record includes dott to represent the CourseName, CourseNamber, CrishtHeurs, and Department (the department that others the course), and so on. We must also specify a data type for each dota elerce a wobin a record. For example, we can specify that Name of statement is a string of diphobeus characters. Succin Number of statement is an integer and Oracle of Good (REORT is a stock characters pagine set {A, E, C, D, E, }. We may also use a estimated store to represent the values of pagine set {A, E, C, D, E, }. a data item. For example, in Figure 4.2 we represent the Class of a \$10060 as 1 terfreshman, 2 for september, 3 for primes 4 for senior, and 5 for graduate student.

To consist are the UKIVERSI IV database, we store data to represent each student, course, scatton, grade report, and prerequeste as a record to the appropriate the Notice that records in the various files may be related. For example, the record to "Smith" in the SULTENT file is related to two records in the GRADE_REPORT file that specify Sunce's grades in two sectors. Similarly, each record in the PEEREFULSTIE file relates two course records can representing the course, and the other representing the preventies include many reposed records and have usery relationships intoing the records.

Database manipulation involves querying and uppering, boundes at queries are treatieve the transcript—a list of all courses and grades— of Supple" "for the names of students who took the section of the Database course offered in fall 1999 and their gradeits that section," and "who are the presequence of the claribose course." Examples of opdates are febringe the class of Stimburg Suphemory," "where a new section on the Database course for this sequence," and tenter or grade of A for Spitch in the Database section of last semester." These information grades in the barabase section of last semester. "These information discuss and upbates in the barabase the query long age of the DBVS before they can be prevened."

1.3 CHARACTERISTICS OF THE DATABASE APPROACH

A number of characteristic distinguish the database opposed from the treational approach of programming with thes for treat and file processing, each user defines and implements the tiles meeted for a specific scheme application as part of programming the application. For example, one user, the gade repertog, *ffice*, may keep a the on students and their gades. Programs to prime a student's transcript and conner new gades into the application. A second care the schemes define these theory of programs to prime a student's transcript and conner new gades into the the or complemented as part of the application. A second care the schemes of the data application is tradents, the student's tensor of the application of the schemes to manage differ, may keep track of students' fees and their population. A second care the schemes of the data application is tracked to transcript and their population of the scheme data in about students, each user mattering second files, and programs to manipulate these files—because each requires some data new wasted stories space and in reducion efforts to maintain comment data op roches.

In the datative opproach, a single reporting of data is manualited that is defined once and then is accessed by votious users. The main characteristics of the database approach versus the me-processing approach are the following

- Self-describing nature of a database system
- Insulation between programs and data, and data distraction
- Support of multiple views of the data
- Sharing or later and multiments associate processing

We next describe each of these characteristics in a separate section. Additional characteristics of database systems are discussed in Sections 1.6 through 1.8.

1.3.1 Self-Describing Nature of a Database System

A fordimental characteristic of the database approach is that the database assertion cansing only the database uself but also be complete definition of description of the database concerns, and constraints. This definition is stored in the DESS catalog, which concerns information such as the structure of each file, the type and storage formation each data new and various constraints or the gluta. The information stored in the catalog is called **meta-data**, and it describes the structure of the primary database (Figure 1.1).

The caralog is used by the 1983's software and also by database users who need internation also it the database structure γ_1 general purpose DDS's software package is not written for a specific database application, and hence it must teter to the catalog to know the structure of the files in a specific database, such as the type and former of data it will access. The DBS's software must work equally well with new provide of database application—for example, a any east to database, a banking database, or a company ξ (abase—as long as the database denotion is stored in the catalog.

In mathional the processing, doce vertice on is typically part or the application programs themselves. Hence, these programs the constrained to work with only one specific database, whose structure is declared in the application programs. Insteadingle, an application program written as citet may have struct or class declarations, and a COROL program has Para Division structure its to denice us tiles. Whereas the processing software can access only specific databases, 1965, software can access diverse databases by estructing the database dematrons from the catalog and then using these definitions.

In the example shown as beene 1.2, the DWS cutalog well store the definitions of all the files shown. These definitions are specified by the database designed prior to creating the actual database and are stored in the ratalog. Whenever a request is finde to access, say, the Name of a SIODEC record, the DWS software refers to the ratalog to determine the structure of the SIODEC record, the DWS software refers to the ratalog to determine the structure of the SIODEC record, the DWS software refers to the ratalog to determine the structure of the SIODEC record, the position and size of the Name cuto iteration within a SODEC record. By convest, iteration of the processing opply ation, the file structure and, in the extreme case, the exter location of Name within a SIODERT record are already coded within each program that accesses this data trent

1.3.2 Insulation between Programs and Data, and Data Abstraction

| Starting Position in Record | Length in Characters (bytes) |
|-----------------------------|--|
| - I | 30 |
| .41 | 4 |
| 35 | 4 |
| (8) | 4 |
| | Slatting Position in Record 1 39 35 38 |

FIGURE 1.3 Internal storage formal for a student record

In some types or database systems, such as object oriented and object relational systems (see Chapters 12 to 22), mers can define operations on data as part at the database definitions. An operation (also called a *position or markall* is specified power tarks. The manager (or signal of) or an operation includes the operation name and the data types of us arguments (or potameters). The mathematication is prefited separately and can be changed without affecting the interface. User application programs can operate on the data by two king these operations through their names indiagometrics, regardless of how the operations are implemented. This may be termed program-operation independence.

The characteristic that allows programmats independence and program-aperation independence is called data abstraction. A 1985% provides uses with a conceptual representation of data do recession include non-youthe details of how the data is stored or how the operations are implemented. Into mally, a data model is a type of data obstruction that is used to provide this conceptual representation. The state model uses leared concepts, such as objects, their properties, and then interrelationshows, that may be easier for most users is understind than you puter storage concepts. Hence, the data model halo storage and implementation details that are not of interest transitional data wave users.

For example, consider significant Figure 1.2. The internal implementation of a the maximdefined by its record length. The minder of characters (leves) in each record clarad back data item may be specified by its starting with without record and us length in bytes. The **STOPENT** record would thus be represented as shown in Figure 1.3. Pour repred database user is not concerned with the location of each data item within a record or its length rather, the concerned with the location of each data item within a record or its length rather, the concerned with the location of each data item within a record or its length rather, the concerned is the reference is using as Name of **STLDFAT**, the correct value is returned. A conception representation of the **STLDENT** records without in Figure 1.2. Many other conduct the file so riggit organization, such is the accession rule specified on a file can be hidded from database users by the 1985; we discuss storing a dotabase in Chapters 15 and 14.

In the database approach, the dotated summure and organization of each file restored in the catalog. Database data and opplication programs reter to the conception representation of the files, and the 1998 extracts the details of file storige from the catalog when these are weeded by the D998 file access modules. More data models can be avail us provide this data abstruction to database users. A manual part of this bask is devoted to presenting various data models and the concepts they use to distinct the representation of data.

In object-oriented and object-relational databases, the abstractive groups metudes not only the data structure but, iso the operations on the data. These operations providin abstraction of numwerly activities commonly understood by the asses. For example,

| 151 | TO AND CHIEF | StudentName | Student Franschist | | | | |
|-----|--------------|-------------|--------------------|-------|----------|-------|-----------|
| | PANSCHIPT | | CourseNumber | Grade | Semester | Year | Sectionic |
| | | | CS1910 | с – | Lat | | . us . |
| | | Smillt | MATHSPAO | | | | · · |
| | | | MATH2-10 | A | - alı | 16 | |
| | | | 05/300 | | • | ••9 | · |
| | | Brown | CS7320 | . H | 50 mg | ່ 11 | 102 |
| | | | C#3350 | . 4 | =., | ໍ່ຕໍ່ | 125 |
| | | • • | | • | | • | - |

| ħ. | PRERECUISITES | CourseName | CourseNumber | Prorectoursetes |
|----|---------------|----------------|--------------|--------------------|
| | | Database | 0.519907 | CS (NO MALE:STR |
| | | Deta Stockares | L9 90 | 091310 |

FIGURE 1.4 Two views derived from the database in Figure 1.2. (a) the Stopest Teasserter view, the The Course Precedure is view.

an operation CAUCOLATE CPA can be applied to a STUDENT object to telefulate the grade point average. Such operations can be invoked by the user queries of application programs valued baying to know the details of how the operations are implemented. In that sense, in adstructions of the manivoid accessor is made to allable to the user of an albumact operation.

1.3.3 Support of Multiple Views of the Data

A darabase typically has many asers, each or when may require a different perspective or ejew of the database. A view may be a subset of the database or throng continuity intrual data that is derived from the database trees but is not explicitly stored. Some users in remotioned to be aware of whether the data they refer to its stored of derived. A multiple whose as is baye a variety of distinct applearions must provide focinities for deriving multiple views but example, one user of the catabase of higher of higher has be received only those as a student, the view for the variety of distinct applearions must provide focinities for deriving multiple views but example, one user of the catabase of higher has be received only those easily are and printing the transcript of catabase of higher has students have taken at lighter has received only included only in checking that students have taken all the presequentes of each course net which they receiver, may require the view shown in the presequentes of each course net which they receiver, may require the view shown in the receiver has the receiver of a student the view shown in the presequentes of each course net which they receiver, may require the view shown in the receiver here.

1.3.4 Sharing of Data and Multiuser Transaction Processing

A nultioser (1885-), as its name implies, must allow multiple users to necess the database at the same upper. This is assertial if data for most ple application is its be integrated and

maintained juilly single database. The revisions include **concurrency control** software to ensure that see rule users using to optime the same database in a controlled mariner so that the assoluted the updates is concernible example, when several reservation clerks by transign assert enorgy at highlight, the DWFs should ensure that each sear can be accessed by only one clerk at a time for assignment to a passenger. These types of opplications are generally called online transaction processing (OFOP) applicators. A fundamental rule of multituser DWP software is to ensure that concernent transactions operate correctly.

The concept of a **transaction** has become control to more database applications. A consultance is an executing to grow of process that includes one of more database accesses, such is reading or opdating of slatabase access it executes fact in its entities, without interference to us other transactions. The provision of process to execute in its entities, without interference to us other transactions. The provision appears to execute in realiant from other transactions for each transaction appears to execute in realiant from other transactions, even though hundreds of introductions may be executed in its interference. The **atomicity** property ensures that either all the database operations in a transaction are executed or near are. We docuss transactions in detail in Part V of the textbook.

The preciding characteristics are mest important an distinguisting a CBMs from traditional file-processing software. In Section 1.6 we discuss additional features that characterize a costs, bust, however, we categorize the different types of persons who work in a database second environment.

1.4 ACTORS ON THE SCENE

For a small personal durabase, such as the first of addresses discussed in Section 1.1, one person typically denoes constructs, and manipalities the duratose, and there is to a sharting. However, many gets ons are involved in the design, use, and maintenance of a large catabase with hundreds of users. In this section we identify the polyfle whose jobs model in the database with hundreds of users. In this section we identify the polyfle whose jobs model in Section 1.5 we consider people whose new below fled "workers behind the scene," in Section 1.5 we consider people whose sectem environment but who are not actively interested in the database itself.

1.4.1 Database Administrators

In any organization where many gers insuse the same resources, there is a need for a chief administrator or oversee and manage these resources. In a database confinemment, the primary resource is the disabase itself and the recondary resource is the OWS and related software. Administrating these resources is the re-generality of the **database** administratic (DRA). The OW is responsible for autoor increase coss to the catabase to coordinating and mornitoring more, and for acquiring software and hardware resources as needed. The 1920 is accommable for problems such as breach of security or poor system resources increase in large organizations, rise DRA is assisted by contribution to helps carry out these functions.

1.4.2 Database Designers

Database designers are responsible on identifying the data to be stored in the database and for choosing appropriate structure two represent and store this data. These tasks are noistly undertaken belong the database decay term to commonicate with all prospection database users in order to understand their requirements, and to come up with a decay root database users in order to understand their requirements, and to come up with a decay root that mean these requirements. In many stress, the decay eristic of the state of the PPV and may be along reductive static test-onsidentics, after the database design its completed. Database designers type: Its interact with each potential group of core and decelop views of the database type: Its interact with each potential group of core, and decelop views of the then in dynet and antigrated with the views of other user groups. The final database design must be capable of supporting the requirements of the segments.

1.4.3 End Users

End users are the people whose jobs to price access to the database for querying updating, and generating reports, the database printerial excists for them use. There are several encogories of end users.

- Casual end matrix accessionally access the database, but they may need different information each time. They use a supported tel database query forguage to specify their response and are typically and the for high-level managers or other exclosional managers.
- Naive or parametric end users make up instable portion of database end overs. Their man jub function revolves around constantly querying and apdaturg the database, using standard types of queries and update-—e alked canned transactions—that have been carefully programmed and cested. The to-ks that size users beform are varied;

Pank tellers direck account bilances and post withdravids and deposit-

Reservanton clerks for terlanes, hotels, and carriential comparities check availability for a giving request and make resorcations.

Ulerks at receiving stations for conner multicater package identifications via for codes, and descriptive, information through furthers to update a central database of received and in transactiologies.

- Suphisticated end users include anguages, scalar sis, business malysis, and others whathoroughly fundance there was work the fundance of the P-Ma scale to juple ment their opplications is once their complex requirements.
- Stand-atome users monetatic personal databases by using ready-made program packages that provide gas accoss meradoawd or graphic abased mitribees. An example is the user of cause package that stores a capter of personal financial stora for any purpose-

A typical CPMS provides nulliple to a cress to access as inclusive. Naive endosers need to term very linke about the facilities provided by the DBMS, they have to understand only the neurinterfaces of the standard mansactions designed and implementer for their use. Cavial neers fearn endy a tew fourlines that they now use repeatedly. Suphrstrated users try to learn most of the DBMS facilities in order to achieve their complex requirements. Stand alone users typically because very prolonant in using a specific servoir package.

1.4.4 System Analysts and Application Programmers (Software Engineers)

Seatem analysts derivative the resourcements or end users, especially noise and parametric end overs, and develop specifications for canned transactions that meet these requires ments. Application programmers implement these specifications as programs then they tests debug, downnent, and maintain these connect transactions. Each analysts and programmers—commody referred to as software engineers—should be foodlar with the full times of capitalities provided by the tests to accomplish their tasks.

1.5 WORKERS BEHIND THE SCENE

In addition to those who design, use, and adjumoter a database, others are used ated with the design, development, and operation of the DBMS softcare and system considerable. These persons the typically not interested in the database itself. We call them the "work ers behind the scene," and they include the following enterprise.

- DWIs system designers and implementers are persons who design and implement the PPUs medules and interfaces as a software package. A DWIS is a very complex software a step that very street party competence or modules, including reaches for implementing the condegs processing dreve language, processing the attentice, accessing and battering data controlling concurrency, and handling data receivery and security. The eMS must method with other system software, such as the operatacter system and conductors for concursing languages.
- Tool developers include periods who vesign and implement tools—We betware packages that tachnace database system design and use and that help improve performance. Tools are optional packages that are often pare based separately. They include packages for database design performance instituoning material language or graphical interfaces, processping, simulation, and test data generation, formany cases, independent software vendors decelop and market these tools.
- Operators and maintenance personnel are the system administration personnel who are regionsible for the actual minimum and maintenance of the hardware and some recoveronment for the dealast system.

Although these corregories of workers behind the sense are instrumental in a during the database system readable recend users, they typically do not use the database for their own purposes.
1.6 Advantages of Using the DBMS Approach

In this section we docases some of the advantages of using all these indition capabilities that a good testify should conserve. These capabilities the interdiducion to the four near characteristic docases. These capabilities the indition to the four near characteristic docases are section. I. A. The TBN must call the faces capabilities to account by a network spectry constrained to the design, adaptistic operation, and use of a large modified dotabase.

1.6.1 Controlling Redundancy

In tradicional software development of intring the processing, every user goors maintains us over this for handling its data processing applications. For every let consider the ustyfiest ty untriase example of Section 1.2: here, two groups of users much be the consumerent-trading proving and the accounting office. As the matrices of approach, each enoup independently keeps they ere students. The accounting office also keeps data on registration and related fulling information, whereas the registration of the class factor of student courses and gredes. Much of the data is stored twice; once in the class of each user group. Additional user groups may further data is stored twice; once in the class of each user group. Additional user groups may further data back courses while of the same data in these.

This related are year on oning the sum data multiple times hads to several problems. From the sensitive pertaining angle logical update is not as entering data on a new order to multiple times once for each file where student data is received. This leads to addication of effects. Second, see groups to there databases. Third, files that represent the same data may become occursorie to these databases. Third, files that represent the same data new to the second for the state databases. Third, files that represent the same data new become occursorie to the state databases. Third, files that represent the same data new become occursorie to the state databases. Third, files that represent the same data new become occursorie to the state where studies an order to applied to come or the files fut new southers, be each on applied to stude the studies are update to applied to come or all the appropriate files, the data concerning the student may still be inconsistent because (acting dates are applied independently by each over group, bor example, one user group and enter a student's built data concerning the student where is the other user group and enter a student's built data enteries only as (200-1934) where is the other user groups any enter the correct value of 198-c9-1984.

In the database approximative views of different user evolptions integrated during database design. Ideally, we should have a database design that stores each logical data item—such as a student's name or brob date. -to *orly one* (database of queries) for crashes remissioned, and it sayes scorage space. However, in produce, it is structured accessive to use **controlled redundancy** for improving the performance of queries. For evangle, we may store Student Nume and Course-Number recondently in a 0440- RE92RT the theorem 1.540 because whenever we remove a 0640- RE92RT the theory for a database student number, and some entropy a 0640- RE92RT record because whenever we remove a 0640- RE92RT the theory for a course whenever we remove a 0640- RE92RT the theory for a main and course number along bether we do not have to acan to remove the student mane and course number along bether we do not have to acard to only find the redundance is no problem reconsistent as an and the application of the redundance to acard be used to not have to acard to only the redundance is no problem reconsistent as analy the the application of the redundance to acard be uncontained by reducing the rate of problem reconsistent as an and the application of the reduction of the rate of the original to be uncontained by the set of problem reconsistent as an and the application of the reduction of the rate of the reduction of th

| · # i | GRADE | REPORT | StudeutNumber | StudentName | Section Doubling | GourseN, mber | Grade |
|-------|---------------|--------|---------------|-------------|--------------------------------|---------------|---------|
| | | | 17 | Smilt | 112 | MA1H2410 | . ĸ . |
| | | | 17 | South | 110 | 051510 | . Ç. |
| | | | . 8 | Hrowr. | ×!: | MA1H2410 | . A |
| | | | . 8 | Hrow | 92 | CS1310_ | . A |
| | | | . º | 3(041) | 192 | 053320 | E |
| | | | . e | Biown . | 135 | 260390 | A |
| .çı | GRA <u>DE</u> | AEPORT | Sluce Number | SlucaniName | Section/dentifier _, | CourseNumber | Grade j |
| | | | 17 | Biowh | 112 | MATH2410 | 8 |

HGURE 1.5 Reduction storage of StudentName and CourseNember in CARE REPORT to Consistent data the forom-istent extend.

CENDE REPORT can be checked against SCOTOM records. Such checks can be specified to the UPVPS during database design and automatically entraced by the DEVD whenever the CENDE REPORT file is applicated. Hence 1,55 shows a GRADE REPORT record that is precisistent with the STADENT file is engote 1,2, which may be entered enrandomly. If the reduid meet is not controlled.

1.6.2 Restricting Unauthorized Access

When multiple users share a large database, it is likely that news users will not be authorited to access all information in the database. For example, hence allowed data is often consultent of a orthonormal, and hence only authorited persons are allowed to access such data. In exhibition, some users may be perturbed only to nervice data, whereas others are allowed to the tree user of the reducer must also be controlled. Tepically users or user groups are given access to the database. A towns should provide a security and authorization subsystems which the taba is some respectively and authorization subsystems which the taba is some respectively and to apply a subsystems which the taba is some respective and authorization subsystems which the taba is some respective count restrictions. The Divis should ther enterior these restrictions autointically, only the "Max soft may be allowed to use certain privileged software, such as the software for the allowed to use certain privileged software, such as the software for the allowed to use certain privileged software, such as the software for the allowed to use certain privileged software, such as the software for the allowed to be certain privileged to relate the software such as the software for the allowed respectively access the allowed respectively access the database of the relative treations.

1.6.3 Providing Persistent Storage for Program Objects

Databases can be used to provide persistent storage for program objects and data storetures. This is one of the more reasons on object oriented database systems, Programming bigloages typically have complex data structures, such as record types in Provid or clasdominants are 0.00 at Jaca. The cohest of program variables are discarded once a program terrenaries, index the programmer explicitly stores throat a permanent riles, which atteninvolves converting these complex structures are a contral surable for the storage. When its need inject to read this data once more, the programmer must convert from the file formation the program variable structure. Object s neural database systems are compatible with programming longuages such as the and fact, and the three software automatic for a more programming longuages such as the analytical and the three software automatic afficients are necessary conversions. Elements complex abject in a t++ can be stored permanently an an object order of program ever (i) on object is stad to be persistent, since it survives the remunation of program ever (i) on and can have be directly nameved by mather (++ program

The persistent storage of program expects and data semicrones is an important function of database systems. Frachrisch, database severas open suffered from the socalled impedance mismatch problem, once the data structures travided by the resistory incompatible with the programming acquirages data structures. Objects mented arabise systems typically after data structure compatibility with one of more objects structure database of non-mining acquirate.

1.6.4 Providing Storage Structures for Efficient Query Processing

Details set systems must provide capabilities for endoerly excerbing queries distraphates because the database is typically stored on disk, the fewty must provide specialized data subctares to specific pulsk search for the desired records. Aux have these illed **indexes** are used for this purpose, indexes are typically based to structures or hash data structure, suitably modified for disk search, he order to process the database records needed by a group larger those records must be copied room disk to memory. Hence, the DBMs offers has a **buffering** module that minimum parts of the database in main increase buffers. In other cases, the DBMs may use the operating system to do the buffering et disk data.

The query processing and optimization module of the 92% is responsible for choosing an efficiency provide union when for each query based on the existing strenge structures. The choice of which indexes occurrent and maintain is part of physical database accurate range, which every of the respectivities of the 1265 staff.

1.6.5 Providing Backup and Recovery

A 1964's must provide tacilities for recovering from hardware or extware tailories. The backup and recovery subsystem of the 0% to is responsible for recovery, for example, if the compart responsible for making streather the database is restored to the care in was in base the transaction stated executing. Alternatively, the recovery absorbent could ensure that the transaction is resolved from the point of which it was an empired so that us follower the recorded by the database.

1.6.6 Providing Multiple User Interfaces

Because many types of users with varying levels of technical knowledge, og a darthase, a (2008 should provide the mery of user interfaces. These include quark integrages for availances, programmers, forms and command codes for parametric cosets, and nemo-deven interface and neutral forguage interfaces for stand-done users. Both forms scyle interfaces and memo-driven interfaces are commonly known as graphical user interfaces. (GUIR), Mark specialited languages and environments exist for specifying GUIS. Capabilities for providing Web GUI interfaces to a database—or Web-readiling a database—etc also quite common.

1.6.7 Representing Complex Relationships among Data

A database may include numerous variet es et data that are interrelated in minity ways. Cereader the example souver in Figure 1.2. The record for Bower in the SUDEX the is related to four records in the **ERDE** REFERENCE file. Sourbally, each section records related to one course record as well as near minister of GRADE REPERT records, some for each student who completed that section. A DBM's must have the capability to represent a verifice of complex relationships among the end as well as to retrieve and update related data easily and concernity.

1.6.8 Enforcing Integrity Constraints

Most database designed in the constraints of the terms of the maximum state of the database design of the type of the maximum state of the terms of the state of the class database design where the type of type of the type of the type of type of the type of type of the type of type of the type of type of type of the type of type

A data mean may be entered enorecously and will satisfy the specified integrity constraints. For example, it a sudent receives acquise of A but a grade of C is entered in the database, the DWIs change discover this error automatically, because C is a valid value for the C-fide data repet Such data entry errors can only be discovered manually (when the studewith receives the grade and complaine) and corrected later by updating the database. However, a grade of Z can be rejected automatically by the DWIs, because Z is not a valid value for the Oracle data type

1.6.9 Permitting Inferencing and Actions Using Rules

Some dividuals sorrers provide capabilities for defining dedect at sides for a ferrating new information from the strend database target. Such systems are called deductive database systems, horey caple, there may be complex tiles in the numworld apple monitor determining when a student is on probation. These can be specified deductive as rules, which when complex to a monitored by the DBMS on determined. If statements or probation the strend of events of the strend database to a specific database to a specific database to a probation. These can be specified deductive as rules, which when complex to a monitored by the DBMS on determined. If statements of probation to a probation of the specified provided by even of the work of the second provided have to be written to support on the phonorement. But it is not the mine with talks charge, it is generally more convention of a phonorement. So that the mine with talks charge, it is generally more convention to charge the destated deduction rules than to recode proceduat programs. More powerful functionable, is provided by active database systems, which provide as the plant actions when events and conditions occur.

1.6.10 Additional Implications of Using the Database Approach

This section discusses some additional implications of using the database approach that can beacht mass organizations.

Polential for Enforcing Standards. The dealase approach periors the triver define indealarde state indealing database events in a large argumentation. This facilitates communication and cooperation among ventus departments, projects, and assess of him the organization. Standards can be defined for names and formats of data elements, deploy formates report structures, terramology, and wrom. The 1933 can enforce standards are centralized database environments more cavity those to an environment where each user group has control of its own hies and software.

Reduced Apprication Development Time - A prime selfage tensor of the database approach is that development new apple non-such as the extremal discussion data too, the database for proting a new report—takes very little time. Despiting and implementing a new database form scratch may take more time to in writing a single performed file applications. For exactly, is up and time to go substantially less time is generally required to create new apple atoms using 100% tacilities. Development time using a 10% to standard to be one single or existent of the atoms using a 10% to estimated to be one single or existent of the atoms using a 10% to estimated to be one single or existent of the atoms using a 10% tacilities.

Forvibility. It movies necessary we change the structure of a dorable as requirements of the loc example, a new user group may emerge that needs thiormation not cut entry to the database. In response, it may be necessary to add a file to the database of the extend the data elements in an existing the. Modern 1958s allow certain types of evolutionary changes to the structure of the database without, flectime the stored data and the existing application programs.

Availability of Up to-Oate Information . A pixts makes the database would be to all uses. As soon as one esserving late is applied to the database, all other uses can muncharely see this opdate. This availability of up to due anformation is essential for many transaction-processing applications, such as to environmy stems of banking databases and at is made possible by the concurrence control and recovery subsestems of a 19945.

Economies of Scale. The OPDS approach permits consolidation of data and applications, thes reducing the amount of waterful overlap between ectivities of data processing performed to different projects of departments. This enables the whele organization is investin more powerful processors, storage devices, or conjunction performance that approach department, evolution argument purchase its own (weaker) equipment. This reduces overall costs of operation and natiogement.

1.7 A BRIEF HISTORY OF DATABASE APPLICATIONS

We now give a brief historical an interve of the opplications that use (1998), and how these opp reamons provided the imperas for new types of database systems.

1.7.1 Early Database Applications Using Hierarchical and Network Systems

Many or le database applications manufined records in large orientations, such as corporations, any cisities (aosoitals, and (sinks) In many of these applications, there were large minders of records of similar structure. For example, in a introductive application, similar information would be kept for each student, each course, each grade record, and so on. There were also more types of records and many interrelationships among them.

One of the mun problems with early database systems was the intermiseing of conceptual relationships with the physical storage and physically stored next to the example, the grade records of a particular student could be physically stored next to the student record. Although this provided vity efficient access for the original queries and transactions that the doublese was designed to buildle at did not provide enough therability in access records efficiently when new queries and transactions were identified in particular, new queries that required a different storing organization for efficient processing were quite difficult to undenteen efficiently. It was also quote difficult to reorganize the database often changes were finde to the region ments of the apply attort.

Another shortcoming of early systems was that they provided only programming language interfaces. This made it time-consuming and expensive to implement new quotes and transactions, since new programs had to be verticed tested, and debagg 2. Most of these database systems were implemented on large and expensive maniforme computers storting in the and 1960s and through the 1972s and 1980s. The main types of only systems were lossed on three main produgms; hierarchical systems, network model based systems, and inverted tile systems.

1.7.2 Providing Application Flexibility with Relational Databases

Relational databases were enginelly proposed to separate the physical storage of data from to conceptual representation and to provide a mathe natural foundation for databases. The relational data model also jum slocist high fixed query languages that provided an alternative to programming language interfaces, hence, it was a for querker to write new orenes. Relational representation of datason what resembles the example we presented in lagaret 2. Relations disystems were aroundly targeted to the same opplications as earlier asternes, but were mean to provide the orbital to quickly develop new queries, and to restging the database as requirements changed.

Early experimental relational sections developed in the late 1952s and the concernent ROPADS (relational database in neighborid systems) anneabled in the early 1952s were spirit, slow, since they did not use physical storage parates or record physical storage and later accords. With the development of new storage and addential feedballows and better query processing and optimization, their performance acrower liberitually, relational databases because the domitional type of stations, systems for final transfer applications. Relational databases row exist encloses all types of computers from small personal computers to large services.

1.7.3 Object-Oriented Applications and the Need for More Complex Databases

The emergence of object oriented programming languages in the 1982s and the need to seen and share complex structured objects had to the development of object oriented databases, mixally, they were considered incompetitor to relational databases, since they provided more general data structures. They also incorporated many of the rise of object elepted particle instead as a toronet data, types, such publication of operations and mutures, and object density. However, the considering of the model and the lick of a nearly sumdatabased in the detailed as a toronet data, types, such publication of operations and mutures, and object density. However, the considering of the model and the lick of a nearly sumdatabased in the data braned as a They are new painly used in specialized replace to object here y splayening design, making she publishing, and nonotocruping systems.

1.7.4 Interchanging Data on the Web for E-Commerce

Inc.World With With provided a large network of interconnected computers. Users currence documents using a Web publishing language, such as at 91. (HyperText Markup Language), and store these documents on Web servers where other users (chcurs) can access them. Documents can be larked together through hyperTinks, which are conners to other documents. In the 1990s, electronic commerce resconducted) emerged as a might application on the Web. It quickly because apparent that parts of the information on a commerce Web pages were often dynamically extracted data from 1994s. A variety of nebringness were deceloped to allow the merch rate of data on the Web: Cornently, XV: TeXrended Markup Language) is considered to be the gravity standard for intergranging data enjoing value is types of databases and Web pages. XVII genbares concepts from the paylels used in document systems with database modeling cornepts.

1.7.5 Extending Database Capabilities for New Applications

The success of data we systems in traditional applications executinged developers of other types of applications to attempt to use them. Such applications traditionally used their own specialized file and data structures. The following attemption of these applications:

- Scientific applications that store large amounts of dota resoluting from scientific experiments in mass such is high-chergo physics or the magging of the human genome.
- Storage and refree of or images, from scanned news or personal photographs to satebrate proceedings and images to an medical procedures such its X-rays or Ms. (magnetic resonance imaginal)
- Storing and intracy if of videose such as movies, or video clips from news or personal digital cameras.
- Data mining applications that analyte large amounts of data scalelung for the accuratences of systemic process or relationships.
- Spatial applications that store spatial locations of dota such as weather information or mays used in geographical information systems.
- Time series applications that solid automatical such as economic data at regular points in the for example, daily service monthly gross rational product ignes.

It was quickly apparent that have relational systems were not very sumble for many of these applications, as fully for one or more or the following gasons.

- More complex data structures were herded for modeling the application than the surple relational testesettration.
- New data repeations the result of modulition to the basic contend should be obtained in the second should be repeated by the second should be basic contend of the second s
- New operations and query hitzage constructs were occessive to tracapilate the new data types.
- · New storing, and indexing structures were needed.

This left DMs developers (codd functionality to their sevenis. Some functionality was general purpose, such as incomparating convepts from object oriented databases into relational systems. Other functionality was special purpose, in the form of epidemal modules that could be used for specific applications, for example, user, could have a rung series module to use with their relational DRMs for their time series application.

1.8 WHEN NOT TO USE A DBMS

In spire of the advantages of using a DBMS, there are a few structures in which such a systercurate tryofice onnecessary overhead cast of an would not be incluted in traditional file processing. The overhead casts of using a DBMs are day in the following.

- · Fightmuch measurement characterizes of wares, and unitaring
- The generality that a service provides for demongland processing, but
- Overheid for providing scorers concurrence control messary and megray functions

Additional problems may at sent the database deseners and (16), do not protein a desen-the database of if the database systems applications are not implemented properly Benediation of the more describle to use regular bies under the following superscribes.

- Eventiate and applications are on pleased dolined, and not expected to change
- There are struggent real-true requirements for some programs that not be toet require of 1989; overhead
- Milliple-user access to data is not required.

1.9 SUMMARY

In this chapter we defined it database as a collection of related data, where data means recoded facts. A typical database represents some aspect of the real world and is used for specific purposes by error or more groups of incress. A DPAD is a generalized software pack are for implementing and maintaining a computerized database. The database and software together form a database system. We identified several characteristics that distinguish the database approach from traditional file processing applications. We then discussed the main categories of database users, or the "actions or support personnel, or "workers belind the scene," in a database second categories or support personnel, or "workers belind the scene," in a database environment.

We then presented a fistor capabilities that should be treated by the ODDS settimetric the 1993, during or designers, and users to help them design administry, and use a dualase. Following this, we give a three fusioned perspective on the evolution of dualases applications. Finally, we discussed the overhead costs of using a pays and desired some structures of perspective or which a pay not be advantageous transport a DBMS.

Review Questions

- 1.1 Define the following terms shift database 2.0868, database system, database catalog programsdata in a portence - as no write 10% to call over control terms when a data terdatabase system, personal object - neurodatase transaction (socressing application).
- 1.2. What three main represent terrors involve databases? Briefly decress each,

- Discuss the main characteristics of the database approach and how it differs from inadimonal file systems.
- 1.4. What are the responsibilities of the 1503 and the database designers?
- What are the different types of datatype end users' Discuss the main activities of each.
- 1 to Discuss the capabilities that should be provided by a 10Ms.

Exercises

- 1.7 Identity some orderinal queries and opdate operations that you would expect to apple to the database shown in Pigme 1.2.
- 1.5. What is the difference between controlled and oncontrolled redundary? Illustrate with examples.
- 1.9. Name all the relationships matorg the records of the database shown in Figure 1.2.
- 1.10. Give some additional views that may be needed by other user groups for the database shown in Figure 1.2.
- Cure some examples of invegrity constraints that you think should hold on the database shown in Figure 1.2.

Selected Bibliography

The CVD ber 1991 issue of Contentinestons of the ACM and Kun (1995) include several anales describing maxing mention. Other many of the database features discussed in the former are now commencially available. The March 1976 issue of 5000 Compating Sava so offers an early incoduction in database systems and may provide a historical perspective for the mark studied reader.





Database System Concepts and Architecture



The architecture of trivial packages haves a ved to to the early monoidable systems, where the whole DDA's software package was one rightly integrated system to the moders (DDIs packages that are module or design, with we here 0-cryst system avalatecture. This cooldoch namers the trends at computing, where large controlled monotrante contraints statbeing reshared by hundleds of distributed workstations and personal computers conacted car communications networks to carious types of server machines. (Web servers, dot base servers, the servers, application servers, and works

In a basic chern's over 1945 an bitecture, the sestem functionality is distributed between two opes of modules.¹ A chern module is topically designed so that it will unor a user workstmon or personal computer. Topically, application programs and use micriales and access the database run in the chern module. Hence, the chern module handles as a micricación and provides the user oriendly interfaces such as tours, or manulesed toffly (Graphicel Oser Interfaces). The other kind of module, called a servier module, typically handles data storage access, sourch, and other functions. We discuss chernylerver as have tores in more detail in Section 2.5. First, we must study mere base excepts the well gave as a bence understanding of insidem dualities induce tures.

In this chapter we present the terminology and basic concepts that will be used throughout the book. We start in Scenon 24, by discussing data models and detruing the

^{1.} Some shall see in Section 2. Earliege network the south to satisfy the show on the section

concerts of schemas, and instances, which are investmental to the study of database systems. We idea discuss the three schema 1905s architecture and data independence in Section 2.2; this provides it user's perspective on what it is 905 its supposed to do. In Section 2.3, we describe the types of interfaces and lateriages that are typeally evolved by a lateria. Section 2.4 discusses the database system software convolution. Section 2.5 gives an overview of various cepts of chember contactions. Including the constraints are types of chember or an interface and here the software convolution. Section 2.5 gives an overview of various cepts of chember cent in interfaces. Including feeting to 2.6 presents it classification of the types of 1980s pickages. Section 2.7 summarizes obschipte to

The material an Sceriens 2.4 Cacugh 2.0 provides more detailed concepts that may be locked upon is a supplement to the basic introductory material.

2.1 DATA MODELS, SCHEMAS, AND INSTANCES

One fondagiental de macrostico table database aperiori has that in provide some level of databbase apon by finding derials of data storage that are not model by most database users. A data model—a deficiento of concerns transion be used to describe the structure of a database—provides the processary means to achieve the substructure. Possegie oc of a database, we mean the data report relationships, and contrating that should hald to the data. Most data models also are first or basic operations for specifying retray de to a opdition the database.

In oddition to the basic or endrous provided by the data to definit is becoming note common to such deconcepts in the data model to specify the domantic aspect or behavior of a chirabase approaches. This allows the database domance to specify the et of such userdefined operations that are allowed on the database of jets.¹ An example of a considerings operation could be concept, asy which can be applied to us to be object. On the other band, generic operations conserve delete, no objects or environ and which of jets are other making in the four data model operation as. Concepts to specify be assisted are included to object oriented data models operations as Concepts to specify be assisted are fundamental to object oriented data models they. Chapters 20 and 215 but are also being metoporated in more traditional data models for example, object orientation are *chapters* 22 econd the traditional data models to be example, object orientation and *chapters* 22 econd the traditional data models to example, object orientation and *chapter* 22 econd the traditional relational model to example, object orientation and *chapters* 22 econd the traditional relational model or methods such concepts, mong others

2.1.1 Categories of Data Models

Many data models have been proposed, which on can also go to a cording to the types of unice pix may use to describe the database structure. **High-level** or **conceptual data mode** els previde concepts that are close to the way many users percepte data, where, s**low-level** or **ploysical data models** provide concepts that describe the details of how data is stored in

^{2.} Sometimes the word model is need to denote a specific drabbed destription, or sciencial, for example, the arriver include, a crustle of We off not us, they receive prefation.

^{3.} The inclusion of converges to describe behavior in facts in trans whereby database resign and soft a residuation activities are inclusionally being a softmeed area datable written. Resht cardly, perioding behavior as the software being:

the computer. Concepts provided by low-level data models are contrally meant for computer specialists not for regionlend users. Baryeen these two extremes is a class of **repres sentational** (or **implementation**) data models, which provide concepts that may be inderstood by end users but that are not not to removed from the way data is organized ording the computer. Representational data models hade some details of data storage bucarbor optimized in a computer system in a data of way.

Conceptual data models use concepts such as entities, and manuships, for entity represents a real world object or concept, such as an employee or a project, have described in the database. An **attribute represence** some property or interest that further describes on entity, such as the employee's name or salary. A **relationship** imang woor more entities represents an association among two or more entities, for example, a works on relationship between an employee and a project, whepter 3 presents the entry relationship between an employee and a project, whepter 3 presents the entry relationship model to popular high-basel conceptual data model. Chapter 4 describeinditional conceptual data modeling concepts, such as generalization, specialization, and categories.

Representational or implementation data on dels one the monels used most trequently in traditional commencial (1995). These includes the widely used **relational data model**, as well is the so-celled legacy of a recedels-so as **network** and **bierarchical models**—that have been widely used in the post. Part II or this book is devoted to the relational data model, us over time and languages, and some of the rechanges for programming relational data base applications.¹ The SQL structure of the relation of data bases is described in Chopters 8 and 9. Representation of data models, represent data by using record structures and hence are sometrice, collict record-based data models.

We can regard object data models as a new function back to be describen the elementation data models that are closer to conceptual data models. We describe the general data models, we describe the general data sector stress of electric databases and the otives involved standard in Chapters 10 and 21. Object data models are also frequently utilited as high-level conceptual models, gate data in the software cognocering domain.

Physical data models describe how data is stored to ideo or the computer by retrearning information such as recent formats, becord enderings, and necess paths. An access path is a structure that makes the search to proticinal database records efficient. We discuss physical storage techniques and access structures in Chapters I 3 and 14.

2.1.2 Schemas, Instances, and Database State

In any data model in its important to distinguish between the description of the database and the database step. The description of a database is called the **database schema**, which is specified database descent and is now expected to change frequently.² Most data

^{4.} As upper us of the prevent or shiften exitical dependence includes provide X_{i} and z_{i} , \overline{z}_{i} and \overline{z}_{i} . The z_{i} depreses there the solution of \overline{z}_{i} backs are as near the training Weissite.

^{5.} Schema charges and us all cance of his chematican ments are be detected by his more charge. Near thanks a second particulations to obtain deviating science of magical disorger functional charge process is more encodered for simple directions op dates.

models have certain conventions for displaying schemas as diagrams? A displayed schema is called a schema diagram, figure 2.1 shows a schema slingram for the database shown in Figure 1.2; the diagram cusplays the structure of each record type but not the actual instances of records. We call each object to the schema-chema-such as proced or COME—schema-construct.

A schema diagram displays only some aspects of a schema, such as the names of record types and data atoms, and some types of constraints. Other aspects are not specified in the schema diagram for example, Figure 2.1 shows not her the data type of each data item nor the relationships among the various files. Many types of constraints are necessarily in scheme diagrams. A constraint such as students majorine in computer science must take visited before the end of their sophomore year its quite difficult to represent.

The actual date in reductive may change quite trequently. For example, the database shown on figure (i.) changes every time we add a student or enter a new grade for a student. The data in the database or a particular moment in time is called a **database state** or **snapshot**. It is clear alled the enter as a documenter in time is called a **database state** a group database state each scheme construct has its own concerts or instances in the database. In example, the subset construct will contain the series of individual student entries (records) is its instances. More subset, we add to a particular basis over constructs of instances, for example, the subset construct will contain the series individual student entries (records) is its instances. More subset, states can be constructed to consepond to a particular database schema. Every time we insert of delete a record or change the value of a database schema in record, we change out state of the catabase into another states.

The distinction between database schema and database state is very important. When we define a new database, we specify its database when it only to the 1986. At this

FIGURE 2.1 Schema chagram for the database in Figure 1.2

to forscense incrementations pertained to use vibrate or the photal for schematic ven through echemanis the proper pland torus. The word scheme is senier area used for a schema. Although as reconsiderable, the schema is not supposed to change frequently, it is not uncernition that changes need to be occasionally applied to the schema is the application requirements change. For example, we may decide that another data item needs to be stored for each tecord may file, such as adding the DatCOPField to the storeg schema in Figure 2.1. This is known as **schema evolution**. Must medern 15505 monde some approximis for schema evolution, that care be applied while the database is overarional.

2.2 THREE-SCHEMA ARCHITECTURE AND DATA INDEPENDENCE

This effort from important characteristics of the details selepproach, as set in Section US and US inset(Consolution of programs and data optage in data and programs performance and endence), (1) support of multiple user values, and (3) occurs a carding to store the data is sedescription (schema). In this section we specify on architecture for dataleties systems, colled the three-schema architecture? That was progressed to help achieve and variable these characteristics. We then for her data is some the concept of data independences.

2.2.1 The Three-Schema Architecture

The soal of the three-schema incluter (ite, illustrated in France 2.2, is to separate the user applications and the physical database. In this architecture, schemas can be defined at the tokoons three levels.

 The internal level has an internal schema, which describes the physical storage structure at the database. The internal schema uses a physical data model and describes the complete details of data storage and access paths for the database.

^{7.} They are a singly also called the access suggister the database

^{5.} The scalar known as the ANE state, architecture, site the commuter that stepasslap (Techtus and Klog 1978).



FIGURE 2.2 The three-schema architecture.

- 2 The conceptual level has a conceptual schema, which describes the sum mesof the whole duringse for a community of users. The conceptual schema hides the details of physical storage structures and concentrates on describing entities data types, relationships, user operations, and constraints. Usually, a representational deta model is used to describe the conceptual schema when a database system is implemented. This roptementation conceptual schema is often based or a conceptual schema design in a melo-a yel data model.
- 3 The external sylvew level includes a number of external schemas of user views, Each external schemer describes the part of the database that a particular user group is interested in and lindes tha sest of the database from that user group. As in the previous case, each external schema is typically implemented using a representational data model, possibly based on an external schema design in a nighlevel data model.

The three schema architectory is a convencent roal with which the user car resultive the schema levels in a doublase system. Must DBUss do not separate the three levels completely har support the three schema at bars rate to some event. Some DBUss note include physical level density in the correspondix formal dramost (2005) that support user views, external schemas are specified in the same data model that describes the conceptual developformation. Some 1 (2005) allow different data models to be used at the conceptual and external levels.

Nonce that the three schemas are only decognoss of data the only data that are addiexists is at the elaysteal level. In a 1990s based on the three schema areitteerure, each user goat, refers only to its own external schema, blency, the 1908 must transform a remost specified on an external schema nots a requesting not the conceptual schema, and dienticity requestion the internal schema for inscessing out the stored database. If the remost is a data see, retrieval, the data extractive from the stored database must be retornated to match the inserve schemal view. The processes of native immercipiests and results intoverally called mappings. These mappings may be time-concording, so some DWISS—especially those that are meant to support small databases—chemot support exernal views. Even in such systems, now year a certain amount of mapping is necessary rour anstorm requests between the conceptual and internal level-

2.2.2 Data Independence

The development reduced recent be used to purcher explain the conveption data indebendence, which can be defined as the capacity to change the schema at one level of a database system without having to change the schema at the next higher level. We can define two types of data independence.

- 1 Edgical data independence is the capacity to choice, the conceptual schema with our conceptual schema to expand the data/set the addatg a become type of data nem), to change constraints, or to reduce the data/set (by removing a teo)rd type of data nem). In the last case, external schemas that refer only to the behaving data should not be affected. For example, the external schema of Figure 145 should not be affected. For example, the external schema of Figure 145 should not be affected. For example, the external schema of Figure 145 should not be affected. For example, the external schema of Figure 145 should not be affected. For example, the external schema of Figure 145 should not be affected. For example, the external schema of Figure 145 should not be affected. For example, the external schema of Figure 145 should not be affected by changing the GMG [355687 file shown in Figure 1.2 into the one shows in Figure 1.5.1. Only the way centarion and the mappings need by changed in a topon's logical chan independence. After the conceptual schema is also also external schema without affective Changes to constraints can be applied to able conceptual schema without affective the conceptual schemas or applied to able conceptual schemas or application programs.
- 2 Physical data independence is the capacity to change the thremal schema withcorr view rochange the conception schema. Hence, the external schemas need not be changed as well. Changes to the internal schema new be needed because some physical tiles had to be reorganized, for example, by ergoing additional access structures—well improve the performance of retrieval or update. If the same data as before remains in the database, we should not have to change the conceptual schema for example, providing an access path to improve terrieval speed of section records (lingure 1.2) by Scinester and Year should not require a apery such as "first all sections offered an tell 1998" to be changed, through the create would be executed more efficiently by the results be unitarial below access pub-

Whenever we have a multiple-level posts its catalog must be expanded to include information on how to map requests and data among the various levels. The 1090s useadditional sufficient to accomplish these multiplings by reterring to the multipling information in the catalog. Data independence becaus because when the schemic is changed at some level, the schemic is the next higher level terrains inclustrated or let the indipling between the race levels is changed. Hence, application programs referring to the higher level schemic need not be changed.

The three-schend architecture consider rosister to achieve the data independence, both physical and legical. However, the two levels of mappings cruite an overhead drinner compilation or esconton et a query of congrane leading to methy eaches in the 1988-Because of the two DOSs have implemented the full three-schena architecture.

2.3 DATABASE LANGUAGES AND INTERFACES

In Section 1.4 we discussed the volume of users supported by a 2005. The tasks must provide appropriate languages and internacis to a carb caregory of users. In this section we discuss the types of longuages and internacis provided by a 1980s and the user categories rangeosily each interface.

2.3.1 DBMS Languages

Orace the design of a data/viscus completed and is DPOE is chosen to implement the database, the first order of the envisito specify conceptual and internal schemas for the database and any mappings between the two fit many DPOEs where no study separation of hive sits magnitude, one language, called the **data definition language** (DDD), is used be the 19 y and by database designers to define both schemas. The 1905 will have a DD compiler whose type on site process DDEs or majors in order to identify descriptions of the schema constructs and reasons the schema description in the DBOEs, and og.

In DBW-s where a clear separation to manipulated between the conceptual and internal levels, the DC, is used to specify the conceptual scheme only. Another language the storage definition language (SDU), is used to specify the internal scheme. The mappings between the two schemes only be specificd in either one of these language eyes for a time takes scheme architecture, we would need a third language (NDL), to specify user views and then mappings to the conceptual scheme, but in most (DM-s) the SPC is used in china but the mappings to the conceptual scheme, but in most (DM-s) the DV is used in china but there exceptual schemes.

Once the database schemas are compiled and the database is populated with datausers must have some means to manipulate the database. Typical manipulations are adretrieval, insertion, deferiant and modification of the data. The 1990s provides a serier operations or a language called the data manipulation language (DMII) for these jumposes.

In current 1983s, the proceding types of Enginees are usually not constant domalonginges; rather, a comprehensive integrated language is used that includes constructs for conceptual scheren determinen, view determinent, includes that includes constructs for as input labor separate, once a is used for Betrage physical stronge structures to fine time the performance of the database system, which is usually come by the 1497 staff. A teptial example of a comprehensive database language is the SQL relation it database language (see Chapters S and 9), which represents a combination of 150, 5 eV, and 1500, is well as scatements for convariant specification, schematevolution, and other tenouries. The SQL was a comparison in early versions of SQL but has been removed from the language to keep just rise conceptoral, and versions of SQL but has been removed from the language to keep just rise conceptoral, and version keeps.

Then are two near types of tMos. A high-level or **nonprocedural** (9) can be used on its own to specify complex database operations in a coorise transfer. Many 1004ss show high-level DMI statements either to be entered interactively form a display no interor terminal or to be embedded in a general-jurpose programming language. In the core cover, ML statements must be identified within the procent so that the care be extracted by sprecomption and processed by the COMES A low-level or procedural DME must be embedded in a general purpose programming language. This type of DMI typically a terves individual records or objects from the citabase and processes each separately. Hence, it finds to use programming language edistricts and processes each separately there, it merels to use programming language censtricts and its language to recreate and process each record from its er of records. Low-level DMI state also called records transfer process of this property. Eigh-level DMI state also called records transfer to extend to a single DMI statement and are hence called separately interpret or instructed to a single DMI statement and are hence called separately interprets at the spectra before that lay for the state of the state of the type of the formation are event, hence, such languages are also called declaratios.

Whenever CML commands, whether high level of low level, its embedded in a sciend-purpose programming language, that language is called the **host language** and the spin is called the **data sublanguage**? On the other hand, a bightevel, we used in a standarbare material increaseries manner is called a **query language**. In general, both retries all and update commands of a large level town may be used instructionaly and are hence considered part of the query language.¹

Canadiend users repeatly use a jugble w? quere language to specify their requests, whereas programmers use the CML or its epibelded form. For naive and par merric users, there usually are user-friendly interfaces to random truttering with the 2 radiuse, these can also users discound users of affairs whereas our water to form the details of a logb level que y values. We discuss these types of not interfaces rest.

2.3.2 DBMS Interfaces

User thendly interfaces precided here (990)s may include the following:

Menu-Based Interfaces for Web Clients or Browsing. These annuaces present thense with lists effortions colled memos, can lead the user targuigh the toraulation of

² In chieven durithases, the host fund data so by anguistics type effective one instrumental hinguage—for example, the mode some exercised envisiopport database tancer mount. Some relational systems of so through an equated comproges interfective sample of toold (11) with

⁽²⁾ According to the section provide work interpret indication should really be associated estribution acrossible non-npdates.

arequest. Means do away with the need to memorize the specific community and source of a query longuage rather, the query is composed step by step by picking options from a menu that is displayed by the system. Pull-down memorias are a very popular technique in Web-based user interfaces. They are also often used in browsing interfaces, which allow a user to look through the contents of a database in an exploratory and instructured manner.

Formis/Based Interfaces. A formy-based interface displays a form to each user-Use - callful out all of the form entries to disent new data, or they fill out only certain entries, to which case the DBMs will refere to instructing data for the remaining entries forms are usually designed and programmed for their energy is interfaces to estimed transactions. Many OSHS's have **forms specification languages**, which are special languages that help programmers specification languages, which are special languages that help programmers specification languages, which are special define a form by letting, the end user interactively construct a sample form on the screen.

Graphical User Interfaces. A graphical interface (%0) typical versplays a schemo to the user in diagrammatic local. The association specify a energy by manipulating the diagram. In many cases, (9) is influe both menois and loture. Most (9) is use a **pointing device**, such as a morist, to gick certain years of the dopt and scheme diagram.

Natoral Language Interfaces. "Duse interfaces occept requests written in Explose or some other language and attempt to "understand" them. A neural language interface issuelly has its own "scheme," which is similar to the database conceptual schema, as well as a dictionary of important words. The natural language interface refers to the words in its schema, as well is to the sector standard words in its dictionary, to interpret the request If the interpretation is successful, the interface generates a high level query corresponding to the natural language request and submits it to the PDMs for processing, otherwise, a dictionary stand with the user to clarify the request.

Interfaces for Parametric Users. Parametric users, such as book telles, often have a small set of operations that they must perform repeatedly. Systems and/ors and programmers dearn and implement a special interface for each known class of have users. Usually, a small set of arbiteviated commands is included, with the goal of manufacing the number of keystockes required for each request. For example, bing and keys in a ferminal can be programmed to initiate the various commands. This allows the parametric user to proceed with a minimula number of keystockes.

Interfaces for the DBA. Most database sectors contain produced commands that on be used only by the PRAN staff. These mellide commands for existing accounts, setting system polaneters, wanting account authoritation, champing a schemal and reorganizing the storage structures of a database.

2.4 THE DATABASE SYSTEM ENVIRONMENT

A 1048 is a complex setuxore system. In this section we discuss the types of software components that construct a 1980s, and the types of comparet system software with which the 1886s metric is.

2.4.1 DBMS Component Modules

Figure 2.5 affustrates, in a samplified form, the typical PBMs components. The standard and the texts caralog are usually statisfied disk. Access to the disk to controlled provinty by the operating systems (OS), which schedules cask impation part. A higher coel stored data manager module of the USMS controls access to DBMS internetion, that is stored on disk whether it is part of the database of the cardog. The dotted area indicated unrefer



FIGURE 2.3 Component modules or a DESS and their interactions

A. B. C. D. and E. n. logare § 4 illustrate increases that are under the control of this stored data manager. The stored data parager new use basic test services for curving our local level data transfer between the dask and contraret main storage. For in centrols other is peers of data transfer, such as bandhing batters in mean memory. Once the data is in main memory butters, in can be processed by other D80- modules, is well as by application programs. Some "30-base their own **buffer manager module**, while others as the OS for transfing the buffering of disk pages.

The DDL compiler processes schema definitions, succities an the DDL, and stores descriptions of the schemas given obtailing the DBMS catalog. The catalog includes antomiation such as the many and sizes of bles, names and data types of data news, storage details of each file, inopping information among schemas, non-constraints, in addition to many other types of information that are needed by the DBMS modules (2015) software modules then look up the catalog information as seeded.

The runtime database processor bondles database accesses at function it receives terreval or update operations and cattles them out on the entabase. Access to dok goes tracegle the store data manager, and the botter ontroger keeps track of the data size pages in memory. The query compiler bondles high-level query by creating database access code, and then generates calls to the runtime processor for executing the orde

The precompiler extracts that commands from in application program written in a host programming fanguage. These community are sent to the DML compiler for compilation into object code for database access. The rest of the program is sent to the host language compiler. The object codes for the DML community and the rest of the program are bricked minimum reasoned transaction whese executable code includes calls to the running database processor.

It is now common to have the **client program** that accesses the 1005 minuing on a separate comparer from the constance on which the database resides. The former is called the **client computer**, ond the litter is called the **database server**. In some cases, the client accesses a middle componer, called the **application server**, which in term consets the clinate server. We claborate on this region in Section 2.5.

Figure 2.3 is not operating describe a specific DWB; righer, it allustrates repeal 1.60% modules. The DPMS interacts with the operating system when disk accesses into the database or to the database on to the database served by an useded. If the computer system is should be non-closely, the V° will schedule 189% ask access respects and DWB; processing along with other processes. On the order hould, if the compare system is notify disficited to it into grade server, the 29% will control many memory buffering of disk pages. The 240% also interacts with complete for general-propose host programming languages and with upplication servers and chem programs running on separate mechanis through the water network interface.

2.4.2 Database System Utilities

In addition to possessing the software involutes just described, most 1984 as him database utilities that help the DDA in managing the database system. Common utilities have the following types of turn const

- Inadate: A loading attility is used to load existing data ides—such as text tills or scorenical tills initia the database. Usually, the current (source) format of the duration and the desired trager) analyses the subcrare are spectral to the utility, which the accomptional culture in stores the database with the profilers increase to the second culture of the spectral transfer in generation of the spectral transfer in the database. Some very source of the second culture is becoming comment in any organizations. Some very data are offering products that generate the appropriate isolated in programs, given the existing source and rarger database storage descriptions (uncertails charast). Such tools, transfer allocation tools.
- Backup: A backup unitity creates a backup copy of the database usually be dumping one or treatment dumpase on the upper like backup copy can be used to restore the database an case of catestraphic radiuse. Increment dirackups are also often used, where only changes once the previous backup are seconded. Incremental backup is more complex bot saves space.
- bit is regimentation. This utility can be used to the regime of doubles file onsets differentially equation is on prove performances.
- Performance many one. Such a order monitors durasse usige and provides statistics to the 15-X. The DPA uses the statistics in making decisions such as whether or not to recording the files to improve performance.

Other utilities may be available for sorting files, handling data comprision, remnomences, by iscrementations, with the network, and performance other functions.

2.4.3 Tools, Application Environments, and Communications Facilities

Other moly the effect available to database designers users, and COASE reads if apused in the design phase of database systems. Another nod that can be quite oseful in large organizations is an explanded data discipancy for data repositored system. It is done non-to-scoring catalog information about when us trail constraints, the date discrepance stores other information, such to design decisions, usage standards, application program descriptions, and user information. Such a system is also called an **information reposi**tory. This information can be accessed directly by users of the USA when needed. A data description of the similar to the DNAs can lag, but to accude a conder variet or reformation and successed match by the DNAs can lag, but to accude a conder variet or reformation and successed match by the DNAs can lag, but to accude a conder variet or reformation and successed match by the DNAs can lag.

Application development environments, such as the Power-Builder (Sylvest or [Builder (Borland) system, are becoming quite popular flows systems provide an sono-moving to developing database hyphesteops area include facilities that help in neuro facets of database systems, meltiding database design, out development, querying indupdament, and applications program development.

⁽a) All longlett Ast, struck for a complete data with energy frequency more COSE (coloring used prior registric database design.

The pBMs documents instruction with communications software, whose function is to adove users at locations reports from the doubtes system site to access the during through computer remainals, workstations, or that book personal component. These are connected to the darabase site through a removement arous bardware such as phone breach personal networks, local area networks, or strektic communication devices. Many commercial database sections have communication packages that work with the tradition mercial database sections have communication packages that work with the tradithe integrated tradition data communications system to called to 100/DC system. In addition, some distributed DBMs are physically distributed over multiple machines, by this converse traditions networks are needed to convect the machines. These are offer local area networks (CANs), but they can also be observed to reworks.

2.5 CENTRALIZED AND CLIENT/SERVER ARCHITECTURES FOR DBMSS

2.5.1 Centralized DBMSS Architecture

Archinectures for 0430s have followed frends suralize to these for general computer system inclinectures. Further reduces to estimation computers to provide the main processing for all functions of the system, including user application programs and user unerface programs, as well as of the towes to genoralize Tae reason was that mean user accessed such systems via computer retaineds that did note bare processing power and only provided displace capabilities. Second processing was performed remote be on the computer system, and only doplay information and controls were sensitive the computer to the displace remainds, which were connected to the computer via computer to the displace remainds, which were connected to the control computer via comote repeating communications betworks.

As mices of hordware declined, nost users replaced their remninds with personal computers (0.5) and workstations. At this, datatives assume used these computers in the same way as they had used display terminals, so that the 1989's methods will be centralized D984's in which all the clews functionality, application program execution, and use interface processing were canned out on one machine. Engine 2.4 illustrates the physical components in a centralized architecture. Under the 1989's systems shifted to explore the available processing precise time as or side, which led to chemise verifiests include the available processing precise time accurate which led to chemise verifiests include the available processing precise time accurate which led to chemise verifiests include to resplet the available processing precise time accurate which led to chemise verifiest include to resplet the available processing precise time accurate state.

2.5.2 Basic Client/Server Architectures

We first discuss characterizer architecture in control theo see how it is applied to 2008s. The characterizer architecture was developed to deal with computing controlments in which a large member of a scatterization the screars granters, disables screars. We'r servers, and other computer are connected via a nerveals. The idea is to derive specials insiliser servers with specific ton conclutives. For example, it is possible to connect a nonline of this or small workstrations as gluines for the server four materians the files of the shear



FIGURE 2.4 A physical centralized architecture

tool new Another machine could be desegnated as a printer server by being connected to y non-printers, thereafter, all prior requests by the chents are forwarded to this machine. Web servers or exmail servers also fall into the specialized server energies. In this way, the resources provided by specificed servers can be accessed by many elemmachines. The client machines provide the aser with the appropriate interfaces to unligtures servers as well as with least processing power to run least applications. This conequileting the carried over to software, with specificed software—such as a 1004s at a 1004 consider added design), pockage, theng started on specific server machines and heng made accessible to moltople clients. Figure 2.5, Host these hepperset exclude the schedage if level, and happens. Figure 2.5, Host these hepperset exclude the



FIGURE 2.5 Logical two-tier of iont/server as bitecture

urchirecture would look. Some unchares would be only chern area (for example, diskless works more or workstations/fais with disks that have only client satisfate installed). Other machines would be dedicated servers. Still other machines would have both client and server functionality.

The concern of client/server architectore assumes an underlying transwork that consists of namy is sond workstations as well as a smaller nurobe, of nonthane machine connected via local trea networks and other types of computer networks. A client in this transwork is typically a user trachine that prevides user interface capitalities and local processing. When a client resource access to additional functionality—such as database access—that does not exist at that undefine that connects to a server that provides the needed functionality. A server is a machine that can provide services to the client machines such as the access printing, inclusing, or database access. In the general case, some machines such as the access printing, inclusing, or database access. In the general case, some machines usuall only chemicality as illustrated in Figure 2.6. Henciver, in is more common that chemican and server servers usually run on separate machines. Two main represent loss to a substrated on this underlying chemicacity receiver and three tracks are accessed on the client and grows are availed on this underlying chemicacies framework receiver and three tracks are accessed on the other and server as a functional transmission of servers.



FIGURE 2.6 Physical two-ter client-server architecture.

(12) There are many other community of cherte server architectures. We arity discuss the two messbase error evide in Computer 15, we case as oddire, rule there server, and discributed architectures.

2.5.3 Two-Tier Client/Server Architectures for DBMSS

The criteri Server inclured are is increasingly being incorporated into commercial 1984S packages. In relation d (1995) (2008) straining of which storted as contralized systems, the estem condynemes that were first moved to the client side were the user interface and oppleaned programs. Pecause SQL (see Chopters S) and 95 provided a standard large rise for RESSES, this created a locical dividure point between client and server, lience, the acety and transaction turationshity remained on the server side. In such an architecture, the server is often called a query server or **transaction server**, because it provides these two turationalities. In REBWE so the server is also often called an SQL server, since most REBWE servers are based on the SQL lenguage and standard.

In such a cherd/server architecture, the user interface programs and application programs can run on the cherd side. When DBMs increases is rearright the program is transfer the chern program can communicate with the robots. A signification program is mared, the chern program can communicate with the robots. A signification **Database Connectivity** (ODDC) provides an **application programming interface** (API), which allows chern side programs to call the DBMP, is long as both client and server machines have the necessary softwore installed. Most robots vender quovi is ODDC drivers to their systems. Hence, a chern program on actually connect to several (DBMP's) and subject and program robots using the vOBC API, which are then processed at the structure systems. Hence, a chern program back to the chern program which can processed at the structure systems. Hence, a chern program back to the chern program which can process or display the robot as medded. A related spondard for the two programming language, called **JDBC**, how also been defined. This allows taxa sheat programs to access the 1001s through a structure therefore.

The second approach to chern/server and otecture was taken by some object-oriented 1986's flexanse many of these systems were developed in the era of chemil-erver acontecture, the approach taken was to divide the software involutes of the 1950s between them and server in a more integrated way. For example, the server level internalide the pair of the 1700 software responsible for handling data starting on disk pages local condurrency control and recovery bulkering and caching of disk pages, and other such turi cons. Meanwhile, the cleant level new handle the user interface, this dienonary tractions with interactions with programming language compilers global query epitimization, concurrency counted, and reservery a ross multiple servers, structuring of complex objects from the data in the burters, and other such functions. In this approach, the chemisterser nucerication is more rightly compled and is done internally by the USMinclules—some of which reside on the chieft and some on the server—rather thing by the users. The exact division of functionality carries from system to system. In such a cherry, sever oth regiate, the surver los by n called a data serves, because it groundes data in by page to the cleast. This dout can then by structured note objects for the clean in com- to the chemistide 1948 solution math

The orchitectores described here are collect two-tier architectores because the - more components are distributed over two systems, chent and server. The ideant get this architectore are its simplicity and seamless compatibility with existing systems. Incrementer of the World Wale Web changed the roles of clicits and server leading to the three-net architectore.

2.5.4 Three-Tier Client/Server Architectures for Web Applications

Many Web applications use on tochtrecture colled the **three-tice architecture**, which tails an intermediate layer between the chem and the carabise server, as illustrated in Figure 2.7. This intermediate layer or **middle fier** is sometimes called the **application** *server* and sometimes the Web server, depending on the application. This server plays an intermediary role by stirring histories rules (procedures or constraints) that are used to access data from the database server. If can also improve database scenary by checking a chem's credentials before forwarding a request to the database scenary by checking a chem's credentials before forwarding a request to the database scenary. Chemis contained to interfaces and some additional application specific business rules. The intermediate server accepts requests from the chemic processes the request and sends database commands to the database server and there acts as a conduct for possing (partially) processed data from the database server in the chemic where it may be processed farther and filtered to be presented to asserve in 60 (format. Thus, the axe interface, application rules) and dataas wall as the three tiers.

Advances in encryption and decryption reclinitiony make it safet to transfer a native data from server to chem in encryption form, where it will be decrypted. The latter can be done by the hardware of by advanced sufficient. This recliniality gives higher levels of data become, but the network socurity issues remain a projot concern. Various verbiologies for data content soon, we else helping at ransferring large trajonic of data non-server-travilients over wheel and wireless networks.



FIGURE 2.7. Logical three-tier cheatserver architecture

2.6 CLASSIFICATION OF DATABASE MANAGEMENT SYSTEMS

Several store of the more likeled to close ty DBM-s. The unsuls the **data model** on which the test subject. The more data model used or many correct commerced DBM-ss is the relational data model. The object data model was implemented on some commerce of systeris but has not had wide-preasing. Many leases to deta models. The relational databuse systems based on the **incrarchical** and **incrework data** models. The relational combination are evolving continuously, and, in particular, how been obsorptioning nons of the concepts that were developed in effect databases. This has led to a new class of DBMs called object-relational DFMss. We can hence categories DFMss based on the data models relational, effect, object relational incrarchical network, and other.

The second enternations of to classify 1989 is the number of users supported by the system. Single-user systems support only one user at a time and are mostly used with personal computers. Multiuser systems, which include the majority of 1980 is support naturb users concurrently.

A third or report is the number of sites over which the database is distributed. A 1865 is comparized if the data is stored at a single computer site. A computer database can support prolitigle users, but the USAS and the database themselves uside multiple at a single computer site. A distributed DBMS UPPOIST can have the actual database and UBAS structure distributed over its new stress connected by a computer movie. Homogeneous 1990's new the same DBMS software at multiple stress. A variant means develops a fixed to a federated result database stored under heterogeneous 1990's Third leads to a federated result (or multiple to develops) for software to increase several auto-computer product database system), so related to a federated result (or multipletabase system), so related to a partition of the database and how a degree of local auto-normal, Mark 1990's new to the structure at the base of the structure of local auto-normal structure.

A loath of renord - the cost of the DBMs. The moonly of 19MS packages cost between \$10,000 and \$100,000. Single-user low-end systems that work with microcomputers cost between \$100 and \$3000. At the other end of the scale, if two elaborate gackages cost more than \$100,000.

We can also classify a 1080s on the basis of the **types of access path** options for commutes. One well-known tamly of 0980s is based on inverted tile structures. Finally, a 1680s can be general purpose or special purpose. When performance is a primary consideration, a special purpose of 85 can be disregred and built for a special application; such systemic inner be used for other applications without major charges. Many airlane is second systemic inner be used for other applications without major charges. Many airlane is second indephone directory systems developed in the past are special purpose (805). These full into the category of online transaction processing (OLIP) systems which major support it large number of concurrent maisterious without imposing accessive delays.

For us builty etaborate on the main criterion for classifying towns the data model. The basic relational data model represents a detabase as a collection of tables, where each table can be stread as a separate the. The database in Figure 1.2 is shown in a number cory so plants a relation of representation. Most retained databases are the high-level party inguage called soft and support a finated tors of user views. We discuss the relational model, its languages and operations, and textstropies for programmate relational applications of Chapters 5 through 9

The object data model defines a database or terms of objects, their properties, and their operations. Objects with the same structure and behavior below to a class, and classes are ensured into hierarchies for acyclic graphs). The operations of each class are specified in terms of predefined procedures called **methods**. Relational (1981)s how been extending their models to meory orate object statutose concepts and other capabilities these systems on referred to as object-relational or extended relational systems. We discussible of unbases and object-relational systems in Chapters 20 to 22.

Evolution interactionally important data models, new known as leave, data models are the network and interactional models. The network model represents data is record repreand also represents a himlest type of 1 N relationship, called a set type. Figure 2.8 shows a network scheme diagram for the database of Figure 1.2, where record types are shown is recordered and set types are shown as infield data real arrows. The network model, data known as the scopy-of DNCG mode β^2 has an associated recordant caping language that must be embedded in a bost programming language. The hierarchical model represents data as hierarchical tree structures. Each literarchical model, on the hierarchical model in presents there is no standard language for the factor structure at the order of relatest recordtion of the record optication and an anguage. We give a structure of the network and hierarchical models in Appendices b and 1.4

The XML (eXtended Markup Longroge) model, now considered the standard for data interchange over the Internet, associes locate locate acad tree structures. It combines database concepts with concepts from document representation, models. Data is represented a elements, which can be used to create concles linearchical structures. This model



FIGURE 2.8 The schema or Ligure 2.1 an network model notation

(3) Coldes a 1980 (status) for Contenence on Data Systems Languages Interaction Law Group, which a the commutee that specifical the activistic in default its language.

1.1. The pull chapters on the network and have retrical enodel offer a the second solution of the basic are available even the last cost time the Web site. conceptionly resembles the object model, but uses different terminologic. We discuss VML and how it is related to databases in Chapter 25.

2.7 SUMMARY

In this chapter we introduced the main concepts used in database systems. We defined a data model, and we distinguished these main categories of data models:

- Elightlevel or conceptual data models (based on entities and to anonships)
- Low-level or physical data models
- Representational or in plementation data models (record-based, object-oriented)

We distinguished the schemator description of a database, from the database itself. The schema dessine champ very often, whereas the database state changes every tane data is inserted, deleted, or modified. We then described the three-schema DB0s architecture, which allows three schema levels.

- An internal schema dearnles the physical starage structure of the database.
- A conceptual schema is a high-facel description of the whole durabase
- External schemas describe the views of different aset groups.

A 15557 that cleanly separates the three levels must have inappings between the scheme to transform requests and results from one level to the next. Most 1955s do not square the three levels completely. We used the three scheme architecture to define the encept of logical and physical data independence.

We then docussed the main types of larguages and interfaces that 1980 is support. A data definition begauge (0.52) is used to define the database conceptual schemo. In most PDMs, the 40L absolutions user views and somethics, strategy and strategy, it other 40Ms, separate languages (VEL 501) may exist for specifying views and storage structures. The PDMs compiles all schemo definitions and scores them descriptions on the 1980 floated of A data in impulation. In accase (1982) is used for specifying database terms and updates that in impulation, have again (1982) is used for specifying database terms all updates that is beingt level. (Sector inted, morphy-educid) or level level. (record-one field, precedural). A high-level (we can be embedded in a host programming language, in it can be used we stand-above language; or the latter case it is often, cilled a query language.

We discussed different types of interfaces provided by PDRess, and the types of DDMs users with which each interface is associated. We three discussed the database system environment, typical 2008 software medules, and DDMS arbities for helping users and the PDV perform there i isks. We there never an overview of the Overfier and three tier inclutionies for database applications, which are new very community more modern applications, particularly Web database applications.

In the final scenery, we closefied 4500s according to several anterna data model, number of users, number of sites, cost, types of access paths, and grow data. The many elesateurian or 0001se is based on the data model. We build discussed the many data workels used in current commercial 0001se

Review Questions

- 2.1 Define the following certify dott boald, database second, database state internet schemet, conceptual schema, extendet schema, dott tradependence, (001, 10M), 517 510, query hungless, host hangings, data tablerightes, database multy, cutalog, therail schema include time.
- 2.2. Discuss the numeratogenes of data models.
- 2.3. What is the difference between a database schenen and a database state?
- 2.4. Describe the three-schema architecture. Why do see need inappings between schema lovels? How do different schema definition languages support the archirecture?
- 2.5. When is the difference between logical data independence and physical data independence?
- 2.6. When is the difference between procedural and nonprocedural (20.52
- 2.7. Discuss the different types of user transfit interfaces and the types of users who repeatly use each.
- 2.5. With what ather comparently strin software due to (20.05) interfered
- 2.9. What is the difference between the two usr and three usr client/server architectures?
- 24.5. Discussisone hyperor database artificies and randomly their functions,

Exercises

- 2.11. Thick of different asers for the database of ligan-1.2. What types of applications would each user need? To which user category would each belong, and what type of attenting would each need?
- 2.12. Choose a database application with which you are function. Design a schema and show a sample database for the application using the non-tion of Figures 2.1 and 1.2. What types of additional information and constraints would you like to represent to the scheme? Think of several overs for contributolase, and design a view for each.

Selected Bibliography

Many day, loss textbooks, including Date (2021), Silber charger of (2001). Romokrishnan an I Gebrke (2002), Garcia-Mohn eccal (1999, 2001), and Alarebeu eral. (1995), provide a discussion of the various database correspondence progented here. Tsichuras and Lochawsky (1982), son-earle resultook on care models. Tsichuras and Slug (1978) and Jardibe (1977) present the three scheme architectures, which was first sugge-tod in the CDTG (1999), report (1977) present the three scheme architectures, which was first sugge-tod in the CDTG (1999). An indepth coalesie of the relational data model and some of its possible extensions is given as (local (1992). The proposed standard for objects mented databases is described in Catell (1997). Many documents describing XML are available on the Web, such as XML (2003).

Examples of database authors are the Eff Extract Toolkit (www.eri.com) and the database administration 6 of DB Arman from Endwrendero Technologies (www.embiac.dor.com)



Conceptual modeling is a very important phase in designing a successful database aggle carea. Generally, the terro database application reters to a part copy database on hibs associated programs that implement the database queries and opdates. For example, a pair d thas application that keeps track or constrained are given to yould onclude programs that inclearent database and ites corresponding to costonices making deposits and withdriveos filese programs provide oscistriendis graphical accumentacis to 550 militarig bruns and also as to the end users of the application-the bark tellers, or this example, clenke, $_{c}$ of the database application with regard the design implementation, and testing of these application programs. To detoot also the design and testing of any heation programs has been considered. Observory in the reductof the software ensuremme domain than in the dotabase domain. As database design in the dologies mellide more of the concepts for spectrug operations on details is ensured as software engineering methodologies specty or more detail the structure of the databases that set were programs will use and access in is clear that these introducts are solvingly related. We brindly discuss some of the concepts for spreativing database operations in Chapter 4, and again when we discuss durihas design methodology with completarphenions in Chapter 12 or this in ele-

In this coapter, we follow the mathemath approach by concentrating on the database structures and constraints during database design. We present the modeling concepts of the **Futity-Relationship** (FR) model, which us a popular high-level conceptual data model. This model and its carations are troquently used for the conceptual design of database applications and many database design reads epiples up concepts. We describe

the basic duriest neturing species and constraints of the RB model and discuss their use in the disign of conceptual schemas for durings applications. We also present the diagrammine notation associated with the RS model. Known as **EU diagrams**.

Object fundeling methodologies such as CMI (Universal Modeling Language) are beyoning methodologies and provide design and engineering. These methodologies by beyond database design to specify detided design of software modules and their interactions using various types of diagons. As impositant part of these methodologies namely elses diagonastic are similar in many ways to the FF diagons. In class diagonas, genatives on objects are specified an addition to specify methodologies scheme sum time Operations can be used to specify the forchored sequecuests during database design, as discussed in Section 5.1. We present some of tha UM, inclution and concepts for class diagrams that are particularly relevant to database design in Section 5.8, and briefly compare these to F8 nontrino and concepts. Additional 001, notation and concepts are presented in Section 4.6 and in Chapter 12.

This chapter is organized as follows. Section 3.1 discusses the role of high-level conceptial data models in database design. We introduce the role report from the FR model, this example database in Section 3.2 no all istrate the role of concepts from the FR model. This example database is database and we gradially introduce the diagonituatic technique for displaying an ER schema. In Section 3.4 we introduce the diagonituatic technique for displaying an ER schema. In Section 3.4 we introduce the diagonituatic technique for displaying an ER schema. In Section 3.4 we introduce the concepts of binary televisity period them to be addimentative constraints. Section 3.5 introduces weak event types. Section 3.6 shows how a schema design is wined to include relationships. Section 3.7 to choose how to choose the modes for database while the role weak events the role with the notation for ER diagrams sub-nontex the role whether an schema design is wined to include relationships. Section 3.6 introduces how to choose the nontex for database whether the role weak events and discusses how to choose the nontex for database whether to ER model concepts, and applies there to the same database example. Section 5.4 subminances the choice the concepts, and applies there to the same database example. Section 5.4 subminances the choice of the concepts, and applies there to the same database example.

The material in Sections 3.8 may be left out of an introductive coursent desired. On the other hand, it mate thorough coverage of data modeling concepts and conceptual database design is desired, the reader should commus on or the material in Chapter 4 area concluding Chapter 3. Chapter 4 describes extensions to the sit model that lead to the Enhanced-ER (0683 model, which includes concepts such as specialization, generalization inheritance, and unities types (categories). We also introduce some additional CM, concepts and potation in Chapter 4.

3.1 USING HIGH-LEVEL CONCEPTUAL DATA MODELS FOR DATABASE DESIGN

fogure 3.1 shows a simplified description of the database design process. The first step shown is requirements collection and analysis. Foring this step, the database conducts interview prospective database users to undercool and document their data requirements. The result of this

A class is sometrized an entry type in tauty across



FIGURE 3.1 A simplified diagram to diagram the main phases of database design
step is a concisch witten set of users' requirements. These resputences should be specified in a detailed and complete a form as possible, in parallel with spectrum the data requirements, it is useful to specify the known functional requirements of the application. These except of the user defined operations to transactions) that will be applied to the database including both tetractional updates. To software design, it is common to use data first database, sequencing any of these feedbacks because they are usually decided in details. We will not discuss any of these feedbacks of some because they are usually decided in details of wave engineering texts. We give an overview of some of these feedbacking on Chapter 12.

Once all the requirements have been collected inclassified, the next step is to create a conceptual schema for the database, using a high-level conceptual data model. This step is called conceptual design. The conceptual schema is a concert description of the data requirements of the users and includes detailed descriptions of the contextpart data model. In the users and includes detailed descriptions of the contextpart of the users and includes detailed descriptions of the contextpart data model. Because these concepts do not using the concepts provided by the high-level data model. Because this is concepts do not using the concepts provided by the high-level data model. Because these concepts do not using implementation details, they are usually users to understand and can be used to communicate with nonreclained users. The high-level conceptual schema can also be used as a reference to ensure that all users' data represents are merianed dust the requirements do not conflict. This approach enables the dust have designers in concentrate on spectroing the properties of the data without remains on the angle to accept denois. Consequently, it is closer for their receiver approach with a good conceptual dustation design.

. Suring or other the exceeption schema design, the basic data model operations can be used to specify the high level over operations identified during functional analysis. This also serves to confirm that the correspond schema meets off the identified functional requirements. Modifications to the correspond schema can be introduced of some fonctional requirements cannot be specified using the gated schema.

The next step in database design is the actual implementation of the database, cancell complement (1998). Most content contributed (1998), are an applementation data model such as the relation during outperturbational database model—so the conception scheme is transformed from the high-level data model into the implementation. Due to exclude 1 has step is called **logical design** or **data model mapping** and its result is a database scheme to the applementation, data model to the *L*908.

The last step is the **physical design** chase during which the internal storage structures, indexes, access paths, and the organizations for the database files are specified in parallel with these activities, application programs are vesigned and implemented as database transactions corresponding to the ballelevel transaction specifications. We discuss the database design process in more detail in Chapter 12.

We present only the basic attended concepts for conception scheme desire in this chapter. Additional modeling concepts are discussed in Chapter 4, when we introduce the treated.

3.2 AN EXAMPLE DATABASE APPLICATION

In this section we describe an example database application, called convolution serves to illustrate the base 10 model conseque and their use in schema design. We list the data requirements for the database here, and there exists conceptual schema step by step as we introduce the modeling concepts of the 4K model. The General database keeps mark of a company's employees: departments, and projects, a typese that after the requirements orderion and analysis phase, the database designets projected the following description of the formy world—the part of the company to be represented in the database.

- 1. The company is organized into departments, Each department has a integer name, a origine number, and a particular enveloped who manages the department. We keep track of the start date when that employee began managing the department. A department new baye several locations.
- A deportment controls a trainfer of projects, each of which has a image name, a characteristic end a single proportion.
- 5. We store each employee's matter, social security numbers, address, snarv, sex, and high date. An employee is assigned to one department but new wark on several projects, which are not accessingly controlled by the same department. We keep track of the number of hours per week that an employee works increase project. We also keep track of the direct supervisit of each employee.
- 2 We want to keep track of the dependents of each employee for insurance purposes. We keep each dependent's first name, see, birth date, and relationship to the employee.

Figure 3.2 shows how the schema for this database application can be displayed by inclusion the graphical notation known as EB diagrams. We describe the step by step process of deriving this schema from the standaux propertients - and explain the Ka diagrammatic monotion. To we introduce the ER model concepts in the following section.

3.3 ENTITY TYPES, ENTITY SETS, ATTRIBUTES, AND KEYS

The estimated case has due as comes relation/Qs, and attributes the Section 3.3.1 we introduce the concepts of contribution and their attributes. We discuss control types and key turbates in Section 3.3.4. Then, in Section 3.3.5 we specify the mitrix conceptual design of the contry types for the toosaw database. Beliationships are described in Section 3.4.

3.3.1 Entities and Attributes

Entities and Their Attributes. The base object that the domodel represents is in entity whether of thing? In the real world with an independent existence. An entity in w to an object with a physical existence (few example, a particular ecronic car, bease, ec

^{2.} The social sector to mendous or 556, is a one-procurrently multimuted assigned by each individual on the Protech States to be environed of basis of two couples much benefits, and each stock to each mesh in w hay, sometimated theorem schemes, such as is sometime interaction coupling in sometimes.





FIGURE 3.2 An *ER* scheme diagram for the cycles database.

employed) or a may be an object with a conceptual existence (for example, a company, a job, or a university conise). Each entity has **attributes**—the particular properties that describe at For example, an employee entity may be described by the employee's same, age, address is dury, and job. A particular entity will have a value for each of as artribuyes. The annihum values that describe cach entity beforme a major part of the data stored in the databas.

Figure 3.3 shows two expressional the values of their numbries. The employed energy s₁ has four autributes Name, Address, Age, and HomePhone; then calles are "foler bundl," "2311 Kirby, "Janson, Texas 77001," "53.1 and "713-749-2630," respectively. The company entry glibas three multiples. Name, Headquarters, and President three values are "Since Od," (Hower, and "John Simth," respectively.



RECORE 3.3 Two entries, employee er and company or , and zoen attributes.

Second types of an ubgress occur in the FR model: angle values engine values angle radius versus india doub, and stoval versus defined. We first define these arribute represent disorate their one classical problem. We then an issue the concept of a publicative for an atmoster.

Finetposite versus Simple (Atomic) Altributes — Composite attributes can be divided into smaller subjorts, which recovery more base our bases with independent assumes for example, the Address attribute of the end, fore carry shown in Figure 3.3 cm is subjorded into StreetAddress. City, Store, and Zight soft the values (2011 Kirby) "Hustor," "Texas," and (7762)." Attributes that are not devisible are called simple or atomic attributes. Composite attributes can tarm a bicrarchy: the example, StreetAddress can be reflec salxhyided into three simple attributes. Number, Street, and ApartmentNumber, as down in Figure X4. The value of a composite attribute is the concatenation of the values of recomposite attributes.



DGURE 3.4 A hierarchy of composite attributes

3. The operate is the name used in the United prices right digit associated.

Composite a tributes are toeful to model situations in which causer sometimes refers to the composite autilitie as comit but at other times refers secondarly to its components. If the composite attribute is referenced only as a whole, there is no need to subdivide it into component attributes. For example, if there is no need to refer to the individual components of on eddless (approxies super, and so on), then the whole address can be designated as a outple attribute.

Single Valued versus Multivalued Altributes. More arritudes have a large value for a princolar many such attributes are called single-valued. For example, Age is a single-valued attribute of a person, in some cases an attribute can have a set of values for the some entity—for examples a Colors can have done at a CollegeDegree attribute for a person class with one color have a single value, whereas the colors have two values for Colors. Similarly, one color have a single value, whereas the construct have two values for Colors. Similarly, one person now not have a sollege degree, another persons that have case and a florid person more have two compose degrees, therefore, different persons can have different standard of robust for the CollegeDegrees of therein. Such attributes one called **multivalued.** A funditioned at ubore more have rower and upper bornels to construct the number of values allowed for each and value is the low example, the Colors attribute of a calmust have between one and three values, if we assume that a car can have at most three celors.

Stored versus Derived Attributes. In some case, two commonly informations are related—for example, the Age and BurbDate curchates of a person. For a particular cerson entry, the value of Age can be determined from the current (rodos's) dots and the value of that version's BurbDate. The Age attribute is hence called a derived **attribute** and its studies be detecable from the BurbDate attribute, which is called a stored **attribute**. Some attribute values can be derived from scotter example, on attribute. Some attribute values can be derived from scotter events, for example, on attribute NumberDEcoployees of a derived from scotter for events. We complete the in takes of employees related to two king for) that denotifient.

Null Values — In some cases a portrollar every may not have an applicable value for an architect for example, the Aroninis of Nucleer architect at a coldress opplies only to addresses that are to appriment buildings and not to other types of residences, such a single-family homes. Similarly, a CollegeDegrees arribate applies only to persons with college degrees. For such structions, a special value called null is created. An address of a structure to approximate buildings and not to other types of residences, such a single-family home would be structures a special value called null is created. An address of a structure degree degree would have null for the ApproximentNumber attribute, and a person with no college degree would have null for CollegeDegrees. Null can also be used if we do not know the value of an attribute as a particular entity. Her example, if we do not know the home gluore of "lohn Simili" or having 3.5. The meaning of the functiones (spectructure) to applicable, whereas the meaning of the latter is anformed. The "unknowe" category of null can be further classified anto two cases. The first case arises wardon is known that the attribute value exists but as massing. For example, if the Height attribute of a person is have a null. The second case prices when at is not know whether the attribute value exists—for example, if the HomePhone attribute of a person is call

Complex Attributes. Nonce that composite and multivalued attributes can be nested in an arbitrary way. We can represent arbitrary nesting by morphic convoluents of

(AddressPhone) (Phone(AreaCode(PhoneNumber)), Address(StreetApprimentNumber,StreetApprimentNumber), City(State,Zip(1))

FIGURE 3.5 A complex attribute: AddressPhone.

a composite attribute between parentheses () and separating the components with compast and by displaying multivalued attributes between braces {}. Such attributes are called complex attributes. For example, if a person can have more than one residence and each residence can have multiple phones, an attribute AsklressPhone for a person can be specified as shown in Figure 3.5.4

3.3.2 Entity Types, Entity Sets, Keys, and Value Sets

Entity Types and Entity Sets. A database usually centains groups of entries that an sundar For example, a company employing handreds of employees may ward to store similar information concerning each of the employees. These employee entries share the anal attributes, but each entry has its own rababast for each attribute. An **entry type** defines a sufficient or set1 of entries that have the same attributes. Each entry type in the database is described by its mane and attributes. Figure 5.0 shows two entry types, named attributes and entries of a littlibutes. Figure 5.0 shows two entry types, named attribute and entries of a littlibutes the enclo. A few individual entries of each type are also illustrated, along with the values of their attributes. The collection of all entries of a particular entry type in the database at any priming time is called an **entity set**; the entry set is usually referred to using the same name as the entry type. For example, second refers to both a type of entity as well as the current set of all employee courts in the database.

An entity rype is represented in FR diagrams? (see Figure 3.2) as a rectangolar loss enclosing the entity type name. Attribute names are enclosed in swals and are attached to dear entity type by straight lines. Composite attributes are attached to their component attributes by straight lines. Multivadued attributes are displayed in double so als

An entity type describes the scheeta or intension for a set of entites that share the same structure. The cohection of entities of a particular entity type are grouped into an entity set, which is also called the extension of the entity type.

Key ABGDutes of an Erifity Type. An important constraint on the entities of an entity type is the key or aniqueness constraint on attributes. An entity type usually has an attribute whose values are distinct for each risk idual entity in the entity set. Such an annuale is called a key attribute, and its values can be used to identify each entity.

⁴ See chose familiar with ASEL we should note here that complex autobates are smaller to complex aleganisms with SEE Chapter 201

^{3.} We are using a motion row 18 diagrams that is close to the original proposed notation (Cherristian Conternational Automation) and the robust are in use. We illustrate some of the other naturations in Appendix A and later in this chapter when we present 1.012 closs diagrams.



FIGURE 3.6 Two entity types, evenous and constant, and some member entities of each

ompachy for example, the Nome antibate is a key of the **CREWOV** entry type in Figure 3.6, because no two companies are allowed to have the same name. For the PERSOV entity type, a typical key attribute is SocialSecontyNumber. Sometriaes, several attributes together form a key, meaning that the combination of the attribute values must be distinct for each entry. If a set of attributes possesses this property, the proper way to represent this in the ER model that we describe here is to define a combinate attribute attribute it as a key attribute of the entry type. Notice that such a composite key must be obtained, that is all component attributes must be included in the composite key must be uniquely that is all comporter attributes must be included in the composite key must be uniquely inside the oval, as dissumption of the notion, such key attribute bis its name **underlined** inside the oval, as dissumed in Figure 3.2.

Specifying that an orthbute is a key of an erury type means that the preceding uniqueness property must hold for every early set of the entity type. Hence, it is a constrong that prohibits any two entities from having the same value for the key attribute of the same time. It is not the property of a particular extensions rather, it is a centrum on all everyspects of the entity type. This key constraint (and other constraints we decreas later) is derived from the constraints of the immoved that the database represents.

Some entry types have note don'ten, key attabute. For example, each of the VehicleID and Registration attributes of the entry type (ss. (Figure 3.7) is a key in its own right. The Registration attribute is an example of a composite key formed from two simple component attributes. RegistrationNumber and State, neither of which is a key on its own. An entry type may also have no key, in which case it is called a mask entry eye (see Segman 3.5).

6. Superfluczes strubutes must not be included in a key: however, a superflewing o include superfluczes attributes, as explained in Chapter 5.

CAR Registration/Cargistration/Number, Statej, VehicletD, Maite, Model, Year, (Color)



FIGURE 3.7. The cas entity type with two key attributes, Registration and VenicleID.

Value Sets (Dormains) of Attributes. Each simple attribute of an entry type is associated with a value set (in domain of values), which specifies the sec of values that may be asserted to that intribute for each individual entry. In Figure 3.6, all the range of ages illowed for employees is between 16 and 70, we can specify the take set of the Age infibure of secretuations between 16 and 70. Scotlarly, we can specify the value set for the Name artificate as being the set of strings of ophabetic doracters separated by black characters, and so on. Value sets are nor displayed in FR displayers. Value sets are typically specified using the basic data types available in most gramming languages, such as integer, string, beolean, float, enumerated type, subtange, and so on. Additional data types in represent date, time, and other concepts are also employed.

Mathematically, an attribute A of centry type E whose value set is V cut be defined as a function from E to the power set² $\mathcal{P}(V)$ of V:

 $A: F \rightarrow P(V)$

We refer to the value of antibury A for entity c as A(c). The previous decontrain every both single valued and individued antibures, as well as nulls. A null value is represented by the empty set. For single-valued antibures, A(c) is restricted to bring a angiene set for each entity c in Γ , whereas there is no restriction on multivalued antibures? For a composite antibure A, the value set V is the Cartonan product of $P(V_1)$.

^{7.} The power set PAU of a set V is the second H is bets of V

^{8.33} singleton set is a set with only one chement (vidue).

 $P(V_2)_1, \ldots, P(V_n)$, where V_1, V_2, \ldots, V_n are the value sets of the surple component attributes that form A:

 $V = P(V_1) \times P(V_2) \times \ldots \times P(V_n)$

3.3.3 Initial Conceptual Design of the COMPANY Database

We can now define the entry types for the consolv database, based on the requirements described in Section 3.2. After defining several entity types and their attributes here, we refine our design in Section 3.4 after we introduce the concept of a relationship. According to the requirements leaded in Section 3.2, we can identify four entity types—one contesponding to each of the four items in the specification (see Figure 3.8).

- An entity type terestiver with attributes Name. Number, Locations: Manager, and ManagerStartDate. Locations is the only multivalued attribute. We can specity that both Name and Number are (separate) key attributes, because each was specified to be unique.
- An entity type matter with attributes Name. Number, law atton, and ControlhugDepartment. Both Name and Number are (separate) key attributes.
- 3 An enrory type 668 (art) with articlutes Name, SSN (for social security number), Sex, Address, Solary, BirthDate, Department, and Supervisor. Both Name and Address may be composite articlutes; however, this was not specified in the sequiremeters. We must go hack to the users to see if any of them will refer to the individual components of Name—FarstName, MiddleInitial, L28(Name) set of Address.
- Art entry type controlst with articlates Employee, DependentName, Sex, Birth-Date, and Relationship (to the employee).

DEPARTMENT Name Number, (Locations), Manager, Manager, StarDate

PROJECT Name Number, Location ControllingDepartment

5MPI CYEF Name (FName, Mint, LName), SSN, Sex, Address, Salary, BithData, Department, Supervisor, (WorksOn (Project, Hours))

DEPENDENT

Employee: DecendentNama, Sex, BithQale, Relationsh.p. ...

FIGURE 3.8 Prohiminary design of entity types for the coward database

So far, we have not represented the fact that an employee can work on several projects not have we represented the number of hours per needs an employee works on each project. This characteristic is listed as part of requirement 3 in Section 3.2, and it can be represented by a multivalued composite armibute of userant called WorksOn with the simple components (Project, Heory). Alternatively, it can be represented as a incluvalued composite armibute of excited. Workers with the simple components (Lingdoyce, Heory). We choose the first alternative in Figure 3.8, which shows each of the entry types just described. The Name attribute of second is shown as a composite attribute, presumably after consultation with the users.

3.4 RELATIONSHIP TYPES, RELATIONSHIP SETS, ROLES, AND STRUCTURAL CONSTRAINTS

In figure Vix there are several anglical velocionships among the various entity types. In fact, whenever an attribute of one entity type refers to another entity type, some relationship exists for examples the attribute. Manager of **transition** refers to an employee who manages the department; the attribute ControllingDepartment of **secret** refers to the department; the attribute ControllingDepartment of **secret** refers to the department of **secret** refers to another entry to the attribute the attribute. Supervisor of **secret** refers to another entry to the attribute the attribute. Supervisor of **secret** refers to another entry to the attribute the employee works; and so is, in the LG model, these references should not be represented as attributes but as **relationships**, which are discussed in this section. The conserve database achieve will be refined in Section 3.6 to represented in the form of anyibures. As the design is refered, these attributes ger converted inverted-inverted up to the department of anyibures.

This section is organized as follows. Section 3.4.1 introduces the concepts of relationship types, relationship sets, and relationship instances. We then define the concepts of relationships degree, role names, and recursive relationships in Section 3.4.2, and theory structural constraints on relationships—such as cardinality ratios and existence dependencies —in Section 3.4.3. Section 3.4.4 shows how relationship types can also have attributes.

3.4.1 Relationship Types, Sets, and Instances

A relationship type K among neutry types E_1, E_2, \ldots, E_n defines a set of associations or a relationship set—among entries from these entries types. As for the case of entry opes and entity sets, a relationship type and its corresponding relationship set are costopically referred rates in relationship type and its corresponding relationship set R is a set of relationship instances t_0 where each t_0 associates a individual entries (e.e., \ldots, e, t_0 and eith entry e_1 in this a member of entry type $E_1 | 1 \leq i \leq n$. Hence, a relationship type is a mathematical relation on E_0, E_2, \ldots, E_n alternatively, it can be deneed as a subset of the Gatesian geoduci $E_1 \otimes E_2 \otimes \ldots \otimes E_n$. Each of the entry type E_1, E_2, \ldots, E_n is and to **participate** in the relationship type **R**; similarly, each of the individual entries e_1, e_2, \ldots, e_n is said to participate in the relationship instance $i_1 \cong \{e_1, e_2, \ldots, e_n\}$.

Internally, each relationship instance i, in R is an association of entities, where the association includes exactly one entity from each participating entity type. Each such relationship instance i, represents the fact that the entities participating as τ_i are related as some way in the corresponding commonld situation. For example, consider a relationship reprosenses to between the swarenity types encoder and introduces each employee with the department for which the employee works. Each relationship instance in the relationship ser wave car associates one employee entity and one department entity. Figure 3.9 fluorrates this example, where each relationship instance τ_i is shown connected to the employee and department entities that participate in τ_i . In the minimorial represented by Figure 3.9, employee c_0 , c_0 and c_0 work for department d_1 , c_2 and c_4 work for d_2 and c_6 and c_6 work for d_6 .

In the diagrams, relationship types are displayed as diamond-shiped boxes, which are connected by straight lines to the rectangular boxes representing the potnegating entity types. The relationship name is displayed in the diamond-shaped box (see Figure V2).



FIGURE 3.9 Some instances in the waxs real relationship set, which represents a relationship type waxs, so: between exercises and creatment.

3.4.2 Relationship Degree, Role Names, and Recursive Relationships

Object of a Relationship Type. The degree of a relationship used is the number of participating entity types. Hence, the wave_scale relationship is of degree two. A relationship type of degree row is called binary, and care of degree three is called **ternary**. An example of a returney relationship is subcry, shown in Figure 3.10, where each relationship instance classicates three entities—a supplier subcard a project i whenever i supplies part p to project i. Relationships can generally be of any degree, but the ones must common are binary relationships. Higher-degree relationships are generally independent then binary relationships: we characterize them bother in Section 4.7.

Relationships as Attributes. If is sometimes convenient to think of a relationship type in terms of attributes, as we discussed in Section 3.3.3. Consider the wave (we relationship type of Figure 3.9. One can think of an attribute called Department of the potent entity type whose value for each simplayce entity is (a reference to) the department only that the employee works for Thence, the value set for this Department attribute is the set of all creativity coddes, which is the opeartment printy set. This is what we dod in Figure 3.5 when we specified the united design of the entity type opposes for the toway database. However, when we think of a binary relationship as an attribute, we always have two



FIGURE 3.10. Some relationship instances in the super-iterative relationship set

contension the example, the alternative is to think of a multivalued attribute Employees of the entity type **DEPOSITEST** whose values for each department entity is the set of engloyee entities who work for that department. The value set of this Employees attribute is the power set of the DEPOSITE entity, set. Either of these two attributes — Department of DEPOSITE on Employees of DEPOSITEST — can represent the "DEPOSITEST. Figure 16 both are represented, they are constrained to be avverses of each other."

Role Names and Recursive Relationships. Each entry type that participates in a relationship type plays a particular role in the relationship. The role mane significant role in the relationship. The role name significant the interval of the role that a participating entry from the entry type plays in each relationship instance, and helps to explain what the relationship means. For example, in the second containership type, epstoyee plays the role of outployee or nonkey and become plays the role of department as employed.

Role names are not technically necessary in telationship types where all the participating entity types are distinct, since each participating entity type name can be used as the role name. However, in some cases the same entity type participates more than incern a relationship type in *different* roles. In such the sole name becomes essential for distinguishing the meaning of each participation. Such relationship types are called **recursive relationship**s. Figure 3.11 shows an example. The supervisor relationship type relationships to the sole relationship type relationships. Figure 3.11 shows an example. The supervisor relationship type relationships to a supervisor, where both employee and supervisor entities are members of the same two entity type. Hence, the ENP, off entity type farming the value for supervisor τ_i in Supervisor and the role of operator for sufficient to the role of supervisor (or box), and once in the role of supervisor for supervisor role, and the other the role of supervisor τ_i in Supervisor role, and the role of supervisor τ_i represent the supervisor role, and the other the role of supervisor τ_i in Supervisor role, and the role of supervisor τ_i is supervisor role, and the role of supervisor τ_i represent the supervisor role, and τ_i supervisor τ_i an

3.4.3 Constraints on Relationship Types

Relationship types usually have certain constraints that I nin the possible conditiations of entities that may participate at the corresponding relationship set. These constraints are determined from the minimorld stuation that the relationships represent. For example, in Figure 3.9, if the company has a rule that each employee must work for exactly escalepartment, then we would like to describe this constraint in the schema. We can distinguish two main types of relationship constraints; endinably ratio and paracepation.

^{9.} This concept of representing relationship (eposits intributes is used on a cass of dust mackets called functional data models. In object doublines used Unipret 200 relationships can be represented by reference armitaires content in one direction or in both directions as inverses. In relational databases (see Chapter 3), foreign keys are a type of reference armitoire used to represent relationships.



FIGURE 3.11 A recursive relationship survives to between tweater in the supervisor role 1) and survives in the subordinate role (2).

Cardinality Ratios for Binary Relationships. The cardinality ratio for a binary relationship spectrum the actorsion number of relationship instances that an entry comparticipate in. For example, in the waves_see binary relationship type, or seminar composes is of cardinality ratio 1:N, meaning that each department can be related to (that is, employs) any number of employees.²⁵ but an employee can be related to (work text only one department. The possible cardinality ratios for binary relationship types are 1:1, 1:N, N 1, and M:N.

An example of a 1-1 binary relationship is **usuas** (Figure 3.12), which relates a department entity to the employee who manages that department. This represents the minworld constraints thus, as any point in time – an employee can manage only one department and a department has only one manager. The relationship type wississes (Figure 3.13) is of cardinality ratio MON, because the minworld rule is that an employee can work on several projects and a project can have several employee.

Cardinality ratios for Empire relationships are represented on ER diagrams by Japlaying I. M. and N on the diagnonds as shown in Figure 3.2.

ECN status for any monitor of related entries (zero or incredi-



FIGURE 3.12 A 1:1 relationship, variants.



FIGURE 3.13 An MON relationship, score on

Participation Constraints and Existence Dependencies. The participation constraint specifies whether the existence of an upday depends on its being related to another entry to the relationship type. This constraint specifies the information databer of relationship instances that each entry can participate in, and its sometimes called the **minimum** *cardinality* constraints. There are two types of participation constraints—total and particiolach we illustrate by example. If a company policy states that every employee must work for adaption there an employee entry can exist only fit it participation at least one solestrain there an employee entry can exist only fit it participation at least one solestrain there an employee entry can exist only fit it participation at least one solestrain there an employee entry can exist only fit it participation at least one solestrain there are the entry of every entry in "the total set" of employee entries rule total participation, meaning that every entry in "the total set" of employee entries numbered to a department entry via soles (see, Total participation is also called **eastence dependency**. In Figure 3.12 we do not expert every employee to manage a dipatiment, so the participation of solester in the exolars relationship type is **particip** meaning that some on "part of the set of " employee entries are related to some department entry via bases, but not necessarily all. We will refer to the continuity ratio and gatarpetion constraints, taken togethe, as the structural constraints of a relationship type.

In fit diagrams, total participation: for exo-tence dependency) is displayed as a shuble are connecting the participating entity type to the relationship, whereas partial participation is represented by a angle late (see Figure 3.2).

3.4.4 Attributes of Relationship Types

Relationship types can also have attributes, similar to those of entry types. For example, to record the number of hours per week that an employee works on a particular project, we can include on autribute. Hours for the week or relationship type of Figure 3.13. Another example is to include the date on which a manager storted managing a deportment of an antibute Stort Date for the PARSER relationship type of lighte 3.12.

Notice that uttributes of 1.1 or UN relationship types duty be interated to one of the participating entity types, for example, the StartDate attribute for the **30368** relationship can be an attribute of either useroor) or prastry so, although conceptionly is before to some some user of the relationship, so every department or implove entity porticipates in a more one relationship instance. Hence, the value of the StartDate attribute can be determined separately, other by the participating deparament entity or by the participating employee finance being on by the participating employee finance being.

Kata EN relationship type, a telationship attribute can be inspired only to the entity type in the N-side of the relationship. For example, in Figure 3.9. If the votes Fos relationship also has an attribute StartDate that indicates when an employee started variation adaptition of this attribute on the included as meatiribute of FOS for Example, in Figure 3.9. If the votes for variation adaptition of the attribute on the included as meatiribute of FOS for this attribute on the included as meatiribute of FOS for the entry of the entry

For MN relationship types, some a trabutes may be determined by the combination of amountary changes in a relationship instance, not be any single entity. Such arrithmes anase be specified as indunouslige attributes. An example is the Heans attribute of the MeN relationship sates (or (Figure 3.13)) the combine of hours an engloying works on a project is determined by an employee-project combination and not separately by either entiry.

3.5 WEAK ENTITY TYPES

Entity types that do not have key arributes or their two are called weak entity types. In contrast, regular entity types that do have a key attribute – which include all the examples we discussed so for—are called strong entity types. Entities belonging to a weak entity type are identified by being related to specific entities from one-her entity type in combination with one of their arribbate values. We call this other entity type the identifying or owner entity type,¹ and we call the relationship type that relates a weak entity type to its owner the identifying relationship of the weak entity type. A weak entity type always has a total paracipation constant lexistence dependency) with respect to its identifying relationship, because 1 weak entity control by identified without on owner entity. However, not every existence dependency results in a weak entity type. For example, a restyre_ticesst entity carried exist onless it is related to a reactive entity even though it has its even key (Licen-eNumber) and hence is not a weak entity.

Consider the entity type relation, related to objects which is used to keep track of the dependents of each employee via a 1-N relationship (Figure 3.2). The attributes of referencest are Name (the first name of the dependent), PerthDate. Ses, and Relationship (to the employee). Here dependents of two district employees may, by chance, have the same values for Name, BirthDate. Ses, and Relationship, but they are still distinct emitties. They are identified as distinct entities only after determining the particular couployee entry to which each dependent is related. Each employee entry is soil to own the dependent entries that are related to it.

A weak entity type normally has a **partial key**, which is the set of actualities that can unsignely identify weak relates that are related to the same overer entity ¹¹ In our example, if we assume that no two dependents of the same employee over have the same first name the attribute Name of is resonance the partial key. In the worst case, a composite attribute of all the relative will be the partial key.

In EC diagrams, both a weak entry type and its identitying relationship are distinguished by surrounding their boxes and diamonds with double lines (see Figure 3.2). The partial key attribute is underlined with a dashed or dotted line.

Wrok entry types can sometimes be represented as complex (composite, multivalued) artificates. In the preceding example, we could specify a multivalued artificate Dependents for execute, which is a composite artificate with component artificates. Name, BirthDate,

12. The weak entries type is also some nucleated the child entity type or the subordinate entity type.

13. The partial key is sometimes called the discriminator-

The identifying cours age is also a octoors called the parent courty type or the dominant entity type.

Sec, and Relationship. The choice of which representation to use is made by the database designer. One criterion that may be used is to choose the weak entity type representation if dwie are many autributes. If the weak entity participates independently in relationship was other than its identifying relationship type, then it should not be modeled as a compass attribute.

In general, any mittaber of levels of weak entrity types of whe defined: an owner entry type has itself be a weak entry type. In addition, a weak entry type may have more than eacidentifying entry type and an electrifying relationship type of degree higher than two, as we illustrate in Section 4.7.

3.6 REFINING THE ER DESIGN FOR THE COMPANY DATABASE

We can now terms the database design of Figure 3.8 by changing the attributes that represent relationships into relationship types. The cardinality ratio and participation construct of each relationship type are determined from the requirements listed in Section 3.3. It some confinality ratio or dependence comporible determined from the requirements, the uses must be guestioned further to determine these structural constraints.

In our example, we specify the following relationship types:

- konsilions a UN relationship type between networker and inconer. Both participations are notal.
- 3. controls, a UN relationship type between passensen and radges. The participation of sector is total, whereas that of apparent is determined to be partial, after consubation with the users indicates that some departments may control to projects.
- 4. succession, a 15N relationship type between twiltering time supervisor role/ and assumed (in the supervisor role). Both participations are determined to be partial, after the users indicate that not every employee is a supervisor and not every employee has a supervisor.
- KORKS 100, determined to be an MEN relationship type with attribute Hours, after the users indicate that a project can have several employees working on it. Both participations are determined to be total.

14 The rules in the initial which has determined the constraints are sometimes called the business needsheet frequencies are determined by the "business" or organic from that will under the database.

6 GEPENDENTS_OT, a 1-N relationship type between SNP-OVEF and DEPENDENT which is also the identifying relationship for the weak contry type OSEENDENT. The participation of INP_COFF_IS partial, when as that of DEPENDENT is total.

After specifying the above six relationship types we service from the entity types in Figure 3.8 all autibutes that have been refined into relationships. These include Manager and ManagerStartDate from unservicest, ControllingDepartment from another Department. Supervisor, and WorksOn from unservices, and Employee from increases. It is important to have the least possible reductionsty when we design the conceptual scheme of a database. If some reductance is desired in the storage lovel or at the user view level, at can be introduced later, is discussed in Section 1.64.

3.7 ER DIAGRAMS, NAMING CONVENTIONS, AND DESIGN ISSUES

3.7.1 Summary of Notation for ER Diagrams

Figures 3.9 through 3.13 illustrate examples of the participation of court types in relationship types by displaying their extensions —the individual courty instances and relationship instances in the entity sets and relationship sets. In Fic diagrams the oupbasis is on representing the schemas rather than the instances. This is more oscillant database design because a database schema changes rately, whereas the contents of the entity sets change frequently. In addition, the schema is usually caster to slisplay than the extension of a database, because it is much smaller.

Equite 5.2 displays the CORMAN ER database scheme soon ER diagram. We now review the full UR diagram motimum. Entry types such as tercover, pression, and explicit are shown in rectangular bases. Relationship types such as versigned, evolues, contools, and voisiges are shown in diamendological bases attached to the participating entity types with straight lines. Attachets are shown in orable and each attachet is attached by a straight line to its entity type or relationship type. Component officients attached by a straight line to its entity type or relationship type. Component officients at a composite attribute are straighed to the oval representing the composite attribute, as illustrated by the Name attribute of recurse. Multimatical attributes are shown in double ovals, as illustrated by the Locations any thore of other bases in double ovals, as illustrated by the Locations any bases are shown in double ovals, as illustrated by the NumberOfEmployces attributes are shown in double ovals, as illustrated by the NumberOfEmployces attribute of period periods.

Weak entity types are destinguished by bring placed in double regrangles and by having their identitying relationship placed in double drametsta, as illustrated by the present entity type and the **dependence** or identitying relationship type. The partial key of the weak entity type is underlined with a dotted line.

In figure 5.2 the cardinality ratio of each boary relationship type is specified by attaching a J. M. (n. Norse cach positicipations edge. The conditionity ratio of conversion exercises in associate is 1.1, whereas it is 1.1N. For conversion constraints are specified, and M.N. Jur waves, co. The conception constraint is specified by a single line for partial participation and by double bies for total participation (existence dependency).

In Figure 3.2 we show the role names for the suffexience relationship type because the assume entry type plays both roles in that relationship. Notice that the cardinality is 1:N from supervisor to supervise because each entrylevee in the role of supervise bas at most one onest supervisor, whereas an employee in the role of supervisor can supervise tercort into onglogees.

Figure 3.14 summatizes the conventions for FR diagrams.

3.7.2 Proper Naming of Schema Constructs

When designing a database schema, the choice of cames hereining types, antributes, relationship types, and tparticularly) roles is not always wraightforward. One should choose names that convey, as much as possible, the meanings attached to the different constructs in the schema. We choose to use angular names for entity types, rather than plucal ones, becaus the entity type name applies to each undividual entity belonging to that entity type, he out ER diagrams, we will use the conversion that entity appe and relationship type names are in opportase letters, arreduce rounds are capitalized, and relations are in lowness thereas. We have already used this conversion in Figure 3.2.

As a general practice, given a narrative description of the database requirements, the scavapporting in the narrative tend to give use to entity type names, and the orby toud condition names of information tends, Arrinbute names generally area from additional neurs that describe the mouns corresponding to entity types.

Another matting consideration involves choosing binary relationship matter to make the Hildugram of the schema reachable from left to right and from top to born 40. We have senarally followed this guideline in Figure 3.2. To explain this naming convention onner, we have one exception to the convention in Figure 3.2—the pre-bless is relationship type, which reads from borrow is not 50 When we describe this relationship, we can say that the DE-black entries (borrow) entity type) are describe this relationship nericlian the DE-black entries (borrow) entity type) are describe this relationship nericlian the react (rop entries (borrow) entity type) are describe the follows: An resome the relationship type to have presented in an electron to the relationship domain entry type). Notice that this issue arises because each birrary relationship (an the described starting home other of the two participating entry) types as discussed in the beginning of Section 3.4.

3.7.3 Design Choices for ER Conceptual Design

It is recassionally difficult to decide whether a particular enterpt in the numworld should be noticed as an epitty type, an attribute, or a colationship type. In this section, we give some brief guide inter us to which construct should be chosen in paracellar situations.



FIGURE 3.14 Summary of the notation for Ht diagrams.

In general, the schema design process should be considered an iterative refinement process where an initial design is created and then iteratively refined multilithe next suitable design is reached. Some of the refinements that are often used include the following

- A concepting bettest modeled as an attribute and then refined into a relationship because in is determined that the attribute is a reference to another entity type. It is inten able case that a pair of such auribures that are ancerses of one another are mined jures a binary relationship. We discussed rbis type of refinement in decid in Section 3.6.
- Similarly, an artribute that exists in several entity repeatincy be elecated or promoted to an independent curry type. For example, suppose that several entity types in a university ideathase, such as shored, respective, and coast, each has an actribute Orbaniment in the annual designs the designer may then choose to create an entity type assessment with a single attribute Deprivate and reference to the three entity types for the discovered later. Other armbutes/ relationships of activation and coast) via appropriate relationships. Other armbutes/ relationships of activation may be discovered later.
- An inverse referement to the previous case may be applied—for example, if an entity type repayrent exists in the initial design with a single attribute DeprName and is related to only one other entity type, stutest. In this case, repayrent may be reduced or denoted to an attribute of student.
- in Chapter 4, we discuss other refinements concerning specialization/generalization and relationships of higher degree. Chapter 12 discusses additional top-down and lottom-up refinements that are common in large-scale conception schema design.

3.7.4 Alternative Notations for ER Diagrams

There are many differnative diagrammatic notations for displaying EC diagrams. Appendix A gives some of the more popular norations. In Section 3.8, we introduce the Universal Moleling Language (1941) meration for class diagrams, which has been proposed as a standard for conceptual object modeling.

In this section, we describe one alternative FK notation for specifying structural constraints on relationships. This notation involves associating a pair of integer numbers (non-nois) with each predictation of an entity type E in a relationship type R, where $0 \le n \le \infty$, and $\max \ge 1$. The numbers mean that for each entitle c in E, c must participate in at least mon and at most max relationship instances in R along point instances in R along point instances in R along point instances in a cluster of a matter of a participation.

by the block of the content of the section of the form the form the form the form in the form. The form the form the form of the form.

¹⁵ In some participation, particularly those used in object modeling, methods (oper such as 0.80, do (one, class) is placed on the opticate states to the object love shown. For example, for the object relationship in Figure 3.15, this (1.10) would be on the EDMSTREE side and the (4.83) would be on the EDMSTREE side. Here we used the engined contation from Algorith (1974).





max) normport. The (mm, pays) normbor is more precise, and we can use it easily to specify structural constraints for relationship types of one degree. However, it is nor sufficient for specifying some key constraints on higher-degree telepionships, as discussed in Section 4.7.

Figure 3.15 also displays all the role nones for the coseword at these schema

3.8 NOTATION FOR UML CLASS DIAGRAMS

The 054L methodology is being used extensively in software design and has many types of diagrams for various software design purposes. We only briefly present the basics of OML



RGURE 3.16 The convertional schema in UML class diagram potation.

class diagrams here, and compare them with EK diagrams. In some ways, class diagrams can be considered as an alternative izotation to FR diagrams. Additional 050: notation and concepts are presented in Section 4.6, and in Chapter 12. Figure 3.16 shows how the classes FR database schema of Figure 3.15 can be displayed using 050: class diagram notanon. The ender pypes in Figure 3.15 are modeled as closes in Figure 3.16. An entry of FR correspondence in *Date*.

In total class diagrams, a class is displayed as a low (see Figure 3.16) that includes into section. The rop section gives the class names the middle section includes the attributes for individual objects of the class and the last section includes operations that can reapplied to these objects. Operations are not specified in 68 diagrams. Consider the section (its in ligate 5.16) its arributes are Name. See, Bdare, Sex, Address, and Schry. The designer can optionally specify the domain of an entribute of desired, by placing a color (it) followed by the domain name of description, as illustrated by the Name. Sex, and Bdare arributes of section in Figure 5.16. A composite arribute is no clobel as a sinitured domain, as illustrated by the Name. Sex and Bdare arributes of section in Figure 5.16. A composite arribute is no clobel as a sinitured domain, as illustrated by the Name articles as separate class, as illustrated by the coverex class in Figure 4.16.

Relationship types are called associations in 0500 terromology and relationship instatics are called **links**. A **hinary association** (b part telationship type) is represented as a line connegring the participating classes (entity types), and may optionally have a name. A relationship orticities collect a link attribute, is placed in a box that is connected to the associations line by a dashed line. The time, max) notation described in Section 3.7.4 is used to specify relationship constraints, which are called **multiplicities** in OML terminology Multiplicities are specified in the form one, may and an asterisk (*) indicates no maximum function participation. However, the multiplicities are placed as the spinolate ends of the science are placed as the spinolate ends of the science are specified with the notation discussed in Section 3.7.4 formpare Figures 3.16 and 3.15). In UML a single asterisk indicates a right below of 0.2° , and a single 1 indicates a multiplicity of 1.1.1. A recursive relationship (see Section 3.4.2) is called a reflexive association in UML, and the role nomes—like the multiplicities are placed at the opposite ends of an association when compared with the placing of role nomes—like the place of role nomes—like the place of role nomes association when compared with the place of the science of 3.1.5.

In UML, there are two types of relationships association and appreation. Aggregation is meant to represent a relationship between a whole object and its component parts and it has a distinct diagrammatic notation. In Figure 5.16, we mindeled the locations of a department and the single location in a project as apprepriations. However, aggregation and association do not have different scatetural projection, and the choice as to which type of relationship to use is somewhat subjective. In the FR model, both are represented as relationships.

UVU also diaringuishes however unidirectional and hidirectional associations for aggregations). In the unidirectional case the line connecting the classes is displayed with an arrow to indicate that only one direction for accessing related objects is needed. If no arrow is displayed, the bidirectional case is associated which is the default. For example, if we always expect to access the manager of a department starting from a upwritten object, we would draw the insortation have representing the employee objects related to each to make the insortation have representing the employee objects related to each department through the waves to be could specify that the employee objects related to each department through the waves too could specify that the employee objects related to each department through the waves too could specify that the employee objects related to each department through the waves too could in a box anached with a dashed line to the line top counting the association/aggregation (see StartDate and Hours on Figure 3.16).

The operations given in each class are derived from the functional requirements of the application, is we discussed in Section 3.1. It is generally sufficient to specify the operation names initially for the logical operations that are expected to be applied to individual objects of a class, as shown in Figure 3.16. As the design is refined, more details are added, such as the exact argument types (parameters) for each operation, plug a functional description, of each operation. One has functional descriptions and sequence diagrams to specify some of the operation details, but these are beyond the scope of our discussion. Chapter 12 will introduce some of these diagrams.

Work entries can be modeled using the construct colled qualified association for qualified aggregation) in UVL, this can represent both the identifying relationship and the partial key, which is placed in a box attached to the owner class. This is illustrated by the obtained class and its qualified aggregation to E900000 in Figure 3.16. The partial key DependentName is called the **discriminator** in UML terminology, since its called distinguishes the objects associated with treated to) the same remarks. Qualified associations are not restricted to modeling weak entries, and they can be used to model other situations in UML.

3.9 SUMMARY

In this chapter, we presented the modeling concepts of a high-level conceptual data model, the Entry-Relationship (FR) model. We started by discussing the role that a highlevel data model plays in the database design process, and then we presented in example set of database requirements for the contain database, which is one of the examples that is real throughout this book. We then defined the base 60 model concepts of entries and their attributes. We discussed null values and presented the various types of attributes, mach can be nested arbitrarily to produce complex attributes.

- Sample or atomic
- Composite
- Multivalued

We also briefly discussed stored versus derived attributes. We then discussed the ER aidel concepts at the schema or "intension" levels

- Entry types and their corresponding entry sets
- Key astrohungs of entity types
- Value sets (domains) of attributes
- Relationship types and their corresponding relationship sets
- Participation roles of coriny types in relationship types

We presented two methods for specifying the structural constraints on relationship upps. The first method distinguished two types of structural constraints.

- Cardinality rands (1)1, 1/N, M/N for binary relationships).
- Europerion constraints (rotal, partial).

We noted that, alternatively, another method of specifying structural constraints is to specify minimum and moximum numbers (min, max) on the participation of each contry type in a relationship type. We discussed weak earlier types and the related concepts of issues entity types, identifying relationship types, and participles and participles.

Entity-Relationship -chemics can be represented diagrammatically as 68 diagrams. We showed how in design in ER schema for the C298we database by risk defining the entity types and then attributes and then refuting the design to include relationship types. We highlight the FK diagram for the C398we database schema. Finally, we discussed some of the basic concepts of CML class diagrams and how their relate to 28 model concepts.

The BE modeling concepts we have presented that fur-entity types, relationship types, and sometimal constraints in care model traditional hypress dataprocessing database applications. However, many newer, more complex applications such as engineering design, medical information systems, or relecommunications inplies additional concepts if we want to model them with grown accuracy. We discuss these advanced modeling concepts in Chapter 4. We also describe ternary and higherdegree relationship types in more detail in Chapter 4, and discuss the circumstances under which they no distinguished from binary relationships.

Review Questions

- 3.1. Discuss the role of a high-level data model in the database design process.
- 3.2. List the various cases where use of a reall value would be appropriate.
- 3.3. Define the following respectively, attribute, attribute value, relationship ensance, composite armitige, multivalued attribute, densed attribute, complex armitute, key applicate radius set (domain).
- 3.4 What is on entity type? What is an entity set? Explain the differences among an entity, an entity type, and on entity set.
- 3.5 Explain the difference between an attribute and a value set.
- 3.6 What is a relationship type? Explain the differences among a relationship instance, a relationship type, and a relationship set.
- 5.7 What is a participation role! When is it necessary to use role names in the description of relationship types?
- 3.8. Describe the two alternatives for specifying structural constraints on relationship types. What are the advantages and disadvantages of each?
- 3.9 Under what conditions can an attribute of a omary relationship type be sugrated to become an attribute of one of the participating entity types?
- 3.10. When we think of relationships as attributes, what are the value are of those attributes? What class of data models is bould on this concept?
- 3.11. What is means by a recursive relationship type? Over some examples of recursive relationship types.
- 3.12. When is the concept of a weak entity used in data modeling? Define the terms ounce entity type, weak entity type, identifying relationship type, and partial key.
- CB: Cars or identifying relationship of a weak entity type he of a degree greater than two? Give examples to illustrate your prover.
- 5.14 Discuss the conventions for displaying an ER schema as on Et diagram.
- § 15 Discuss the nationg conventions used for ER schema diagrams.

Exercises

- 5.15. Consider the following ser of requirements for a university database that is used to keep track of students' transcripts. This is similar but not identical to the database shown in Figure 1.2.
 - a. The university keeps track of each student's name, student number, social scentry number, current address and phone, permanent address and phone, buildate, sex, class (reshman, sophotnote, ..., graduate), major department, indice department (if any), and degree program (B.A., B.S., ..., Fh.D.). Some user applications need to refer to the city, state, and op code of the student's permanent address and to the student's last name. Both social security munifer and student number have unique values for each student.
 - Each department is described by a name, department ende, other number, other phone, and college. Both name and code have unique volues for each department.

- c. Each course has a coarso name, description, coarso number number of semester hours, level, and affering department. The value of the coarso number is arrighte for each coarso.
- d. Each section has an instructor, semester, year, course, and section number. The section number distinguishes sections of the same course that are taught during the same semester/year; its values are 1, 2, 3, ..., up to the number of sections ranght during each semester.
- A grade report has a student, section, letter grade, and nonneric grade (0, 1, 2, 3, or 41.

Design at ER schema for this application, and draw on UK diagram for that schema. Specify key attributes of each contry type, and structural constraints on each relationship type. Note any unspecified requirements, and make appropriate assumptions to make the specification complete.

- 5.17. Composite and mathy alued attributes can be nested to any minibut of levels. Suppose we want to design an attribute for a second entity type to keep track of previsions of lege education. Such an attribute will have one entity for each codege previously attended, and each such entry will be composed of codege name, start and end dates, degree entries (degrees awarded at that college, at any), and transmitting tentries (courses completed at that college, at any). Each degree entry contains the degree name and the month and year the degree was awarded, and each transmitted to transmitte unity contains a course name, semicate, year, and grade. Design an attribute to hold this information. Use the conventions of Figure 3.5.
- 3.18. Show an alternative design for the applicate described in Even ise 3.17 that uses only entry types (including weak entry types, it needed) and relationship types.
- 3.19. Consider the FR diagram of Figure 3.32, which shows a simplified schema for an arbite reservorions system. Extract from the FR diagram the requirements and constraints that produced this schema. Try to be as precise as possible in your requirements and constraints specification.
- 3.3. In Chapters 1 and 2, we discussed the database environment and database users. We can consider many entity types to describe such on environment, soch as PBMS, stored database, 1993, and catalog/data diction ity. Try to specify all the entity types that can fully describe a database system and its environment; then specify the relationship types among them, and data an BR diagram to describe such a general database environment.
- (21) Design are fit schema for keeping track of otformation about cores taken in the U.S. House of Representatives during the current two-year congressional session. The database needs to keep track of each U.S. start's Name (e.g., Texas, New York, California) and include the Region of the state (whose domain is (North-oast, Malwesi, Southcost, West)). Each costresspersor in the House of Representatives as described by his or her Name, plus the District represented, the StartDare when the congressperson reas first elected, and the political Party to which he at she belongs twhese domain is [Republican, Demograt, including the BillName, the DateOfVate on the bill, whether the bill PasedOrFailed (whose domain is (Ves, No)), and the Spensor (the congresspersor) (d. who sponsored—



FIGURE 3.17 An Exidiagram for an atmixed database schema

that is, proposed—the bill). The database keeps track of how each congressperson voted on each bill (domain of vote attribute is O'es. No. Abstain, Absent). Draw an FE schemes diagram for this application. State clearly any assumptions you make.

- 3.22. A database is being constructed to keep track of the reams and games of a sports league. A ream has a number of players, not all of whom participate in each game. It is desired to keep track of the players participating in each game for each ream, the positions they played in that game, and the result of the game. Design an E0 scheme diagram for this application, stating any assumptions you make. Choose your favorite sport (e.g., souce, baseful), tootbally.
- 3.23. Consider the for diagram shown in Figure 3.16 for part of a sear database. Each bank can have multiple branches, and each branch can have multiple accounts and learns.
 - c. List the (nonweak) entity (gas in the 98 diagram.
 - b. Is there a weak entity type? It so, give its name, partial key, and identitying relationship.
 - What constraints do the partial key and the identifying relationship of the weak entity type specify in this diagram?
 - d. List the names of all relationship types, and sporthy the (min, max) constraint on each participation of an entity type in a relationship type, [astity your choices.
 - e List concisely the user requirements that led to this E2 schema design-
 - Suppose that every customer must have of least one account but is restricted to at most two leans at a time, and that a bank branch cannot base more than 1000 leans. How does this show up on the limit, max1 constraints?



FIGURE 3.18 An FR diagram for a sus-database schema.

- 3.24. Consider the 5R ibagiant in Figure 5.19. Assume that an employée may work in up to two departments or may not be assigned to any department. Assume that each department must have one and may have up to three phone numbers. Supply (num, max) constraints on this diagram. *Star* clearly any additional assumptions you make. Under what conditions would the relationship assumption be redundant on this example?
- 3.25 Consider the S8 diagram in Figure 3.20. Assume that a course may or may not use a textbook, but that a text by definition is a book that is used in some course. A course more not use more than five books. Instructors reach from two to four courses. Supply think, max? constraints on this diagram. State clearly any additional assumptions you make. If we add the relationship way is between instructors and text, what (min, max) constraints would you put on it? Why?
- 3.26 Consider an entry type service in a terrestite database, which describes the section offerings of courses. The attributes of section are SectionNumber. Seniester, Year. CourseNumber. Instructor, RoomNo (where section is taught). Building (where section is taught). Weakdoys (domain is the possible combinations of weekdays in which a section can be offered (MWE, MW, TT, etc.)), and Homa (domain is all possible time periods during which sections are offered 19–9:50 A.M., 10–10:50 A.M., 10–10



FIGURE 3.19 Part of an ER diagram for a comparidatabase.



FIGURE 3.20 Part of an FR diagram for a consest database

Nember is unique for each course within a particular sensitivity (that is, if a course is offered multiple times during a particular sensitier, its section otherings are numbered 1, 2, 3, etc.). There are several composite keys for 5EC2038, and some arrifactes are components of more than one keys identify three compositive keys and show how they can be represented in an ER schetna diagram.

Selected Bibliography

The Entiry-Relationship model was introduced by Chen (1976), and related work wyests in Schmidt and Swenson (1975), Witslethold and Elmasti (1979), and Senko (1975) Since then, turnetous modifications to the E8 model have been suggested. We have the opportated some of these in our presentation. Substituted constituties on relationslops are discussed in Abrial (1974). Elmastriand Wiederhold (1980), and Lenterini and Samicer (1983). Multivalued and composite attributes are incorporated in the 58 model in Elinasitier al. (1985). Although we did not discuss languages for the entity relationship model and its extensions, there have been several proposals for such languages. Elinasri and Wiederheild (1981) proposed the took! AS query language for the 1-3 mindel. Another Expery language was proposed by Markowice and Rat (1983). Senko (1980) presented a quere language for Senkels (135) model. A formal set of operations called the ER algebra ws priserred by Parene and Spaceapietra (1985). Cogolla and Eubensieun (1991) prescored mother formal language for the FR model. Comphell et al. (1985) presented a ser of EB operations and showed that they are relation dly complete. A conference for the dissimilation of research results related to the ES model has neen held regularly surre 1979. The conference, now known as the International Conference on Conceptual Modeling. lusbeen held in Los Angeles (EA 1979, 80 1953, EX 1997), Wal-Fington, D.C. (ES 1981), Chatage (ER 1985), Dajon, France (ER 1986), New York City (ER 1985), Reme (FR 1988). Taroute (10, 1989). Lausarine, Switterland (10, 1995). Sin Mateo. California (ER 1991). Earlande, Cormany (Elt 1992). Arlungton, Texas (ER 1993), Manchester, England (ER 1954). Brisbane, Australia (Ed. 1995). Cottl us. Cermany 1-6 (1995), Singapon. (IB 1998). Sia Láce Cuy, Duah (13, 1999), Yokohama, Japan, (14, 2001), and Tangaere, Finland (16, 2020. The next conference is scheduled for Chicago in October 2003.





Enhanced Entity-Relationship and UML Modeling



The FK modeling concepts docursed in Chapter 3 are sufficient for representing namy database schemps for "traditional" database applications which mainly include down processing applications in business and industry. Since the late 1970s, however, designers it database applications have tried to design more incurate database schemas that reflect the data properties and constraints more precisely. This was particularly important for rew rapplications of database to brodogy, such as database for engineering design and manufacturing (CAOV/CAOT), reflections of database to brodogy, such as database for engineering design and manufacturing (CAOV/CAOT), reflections of the design more traditional splications. These rypes of database have more complex requirements than do the more traditional applications. This test to the development of additional semantic data models such as the 20 more studied and applications. This test to be opposed to the hierarchy. Many of these to tacque were also developed needs, each of the object modeling area in service elementing.

In this chapter, we describe features that have been proposed for semantic data nodels, and show how the ER model can be enhanced to include these concepts, leading to the enhanced ER, is: EER, model.⁷ We start in Section 4.1 by memporating the

^{1.} On 5. On wardy for computers acked design/computers aded instantic traing-

Clear has also been used to stripp for Exerned 56 model.

concepts of closofunction volutionships and cyto interfastice into the E8 model. Then, in Section 4.2, we add the concepts of specialization and generalization. Section 4.3 discusses the various types of constraints on specialization (generalization, and Section 4.4 shows how the UNION constract can be modeled by including the concept of category in the EE5 model. Section 4.5 gives an example UNIVESSITE database schema in the EE0 model and summarizes the EER model concepts by giving formal definitions.

We then present the UGL class diagram notation and concepts for representing specialization and generalization in Section 4.6, and briefly reinpart these with EER notation and concepts. This is a continuation of Section 3.8, which presented basic OM, class-diagram notation.

Section 4.7 documes some of the name complex issues involved in modeling of ternary and higher-degree relationships. In Section 4.8, we discuss the fundamental distractions that are used as the basis of many semantic data models. Section 4.9 summarizes the diapter

For a detailed introduction to conceptual nucleling, Chapter 4 should be considered a continuation of Chapter 5. However, it only a basic introduction to f.R modeling is desired, this chapter may be constead. Alternatively, the reader may choose to skip some or all of the facer seriions of this chapter (Sections 4.4 through 4.8).

4.1 SUBCLASSES, SUPERCLASSES, AND INHERITANCE

The FFK (Enhanced (K) model includes all the modeling concepts of the FK model that were presented in Chapter 3. In addition, it includes the concepts of subclass and superclass and the related concepts of specialization and generalization (see Sections 4.2 and 4.3). Another concept included in the TED model is that of a category or union type (see Section 4.4), which is used to regresent a collection of objects that is the above of objects of different entry types. Associated with these concepts is the inportant incluants in a tribute and relationship inheritance. Unfortunately, ne standard terminology exists for these concepts, so we use the most common terminology. Alternative terminology is given includes, We also describe a diagrammatic technique for displaying these concepts when they arise in an Edit schema. We call the resulting schema diagrams enhanced ER or EER diagrams.

The first EER model conceptive take up is that of a subclass of an entity type. As we discussed in Chapter 5, an entity type is used to represent both a type of easily and the entity set or collection of entities of that type that exist in the database. For example, the entity type metael doscribes the type (that is, the attributes and rebranships) of each employee entity, and also refers to the current set of entities of its entities in the conservation of entities to the entity type for entity of each doscribes the type that current set of entities and rebranships) of each employee entity, and also refers to the current set of entities of its entities in the conservational and need to be represented explicitly because of their significance in the database application. For example, the entities that are members of the entities in the database application. For example, the entities that are members of the entity type may be grouped for their interval explicitly example, indexes to the entities that are members of the entities of entity type may be grouped for their interval explicitly example, indexes to the entities that are members of the entities of entity type may be grouped for their interval explicitly example, indexes the entities in the entities into set of entities in each of the latter groupings is a subset of

the entries that before to the twelvest entity set, meaning that every entity that is a member of one of these subgroupings is also an employee. We call each of these subgroupings a subclass of the merover entity type, and the merover entity type is called the superclass for each of these subclasses. Figure 4.1 shows how to diagramatically represent these concepts in TER diagrams.

We call the relationship between a superclass and any one of its subclasses a superclass/subclass or simply class/subclass relationship.' In our previous example, evaluate/secretary and evaluates/relationship.' In our previous example, evaluates/secretary and evaluates/reconstruction are two class/subclass relationships. Notice that simember entity of the subclass represents the secret real-science add entity as some member of the superclass, for example, a secretary entity 'Joan Logano' is also the evaluates includes includes the same of the entity in the superclass. For example, a secretary entity 'Joan Logano' is also the evaluate assumption at the same of the entity in the superclass. But in a distinct specific cole. When we implement a superclass/subclass relationship in the



FIGURE 4.1 FER diagram motorion to represent subclusses and specialization.

¹ A circlewide has relationship is often called in US-A for IN-AND relationship because of the way you rest to the concept. We say "it SESSETARY is an EMPLOYED," To "LUMMS DM is on EMPLOYED," and show the set of the concept. We say "it SESSETARY is an EMPLOYED," To "LUMMS DM is on EMPLOYED," and show the set of the concept. We say "it SESSETARY is an EMPLOYED," To "LUMMS DM is on EMPLOYED," and show the set of database system: however, we may represent a member of the subclass as a distinct database object – say, a disriper record that is related via the key attribute to its superclass entry. In Section 7.2, we discuss various options for representing superclass/subclass relationships in relational databases.

An entrice comparies of in the database merely by being a member of a subclass at must also be a member of the superclass. Such an entrity can be included optionally as a member of any number of subclasses. For example, a subclassed employee who is also an engineer belongs to the two subclasses exports and subclasses, precent of the avecase entries type. However, it is not necessary that every entity in a superclass be a member of some subclass.

Arcomportant concept associated with subclasses is that of type inheritance. Recall that the type of an entity is defined by the attributes it possesses and the relationship types in which it participates. Because an entity in the subclass represents the same real-world entity from the superclass of should possess values for its specific attributes is a weighter of a subclass inherits all the attributes of the superclass. We can that an entity that is a member of a subclass inherits all the attributes of the superclass. We can that an entity that is a member of a subclass inherits all the attributes of the superclass. We can that an entity that is a member of a subclass inherits all the attributes of the superclass parts apares. Notice that a subclass, with its own specific (or local) artificates and relationships regether with all the orthobates and relationships in which the superclass, can be considered an entity rays an as own right.⁴

4.2 SPECIALIZATION AND GENERALIZATION

4.2.1 Specialization

Specialization is the process of defining a set of subclasses of an entity type; this entity type is called the **superclass** of the specialization. The set of subclasses that form a specialization is defined on the basis of some distinguishing characteristic of the entities in the superclass. For example, the set of subclasses [SCGETARS, ESCAETARS, ESCAETARS,

Freuze 4.1 shows how we represent a specialization diogrammatically in an EER diagram. The subclasses that donne a specialization are attached by times to a circle that represents the specialization, which is connected to the superclass. The subclass equivalent of the superclass in the circle indicates the direction of the superclass subclass relationship.⁵ Astro-Fines that apply only to entities of a particular subclass—such

^{4.} In some information programming languages, a common institution is that an entire for object) has adviced type. This is generally is a centrarize for conceptual distribution wedging.

There are more aberrated in a domain signal of the new present the EVR potation in Section 4.6 and other proposed notations in Appendix A.
as hypereSpeed of attributes (or local attributes) of the subclass. These are called specific attributes (or local attributes) of the subclass. Similarly, a subclass can participate in specific relationship types, such as the still estimate subclass participating in the second protocol of the subclass participating in the second protocol of the subclass participating in the second protocol of the subclass of the subclass is a subclass of the subclass participating in the second protocol of the subclass participating in the second protocol of the subclass participating in the second protocol of the subclass participation of the subclass participating in the second protocol of the subclass participating in the second protocol of the subclass participating in the second participation of the subclass participating in the second participating in the second participating in the second participation of the subclass participating in the second participation of the subclass participating in the second participation of the subclass participating in the second participating in the second participating in the second participating in the second participating is subclass. The second participating is subclass participating in the second participating is subclass. The second participating is subclass participating is subclass participating in the second participating is subclass. The second participating is subclass participating is subclass participating is subclass participating is subclass. The second participating is subclass participating is sub

Frame 4.2 shows a two entry instances that belong to soliclasses of the [SECRETARY FROM SER. TEORECIES] specialization. Again, notice that an entry that belongs to a subclass represents the same mathematik noticy as the entry connected to it in the EVPLOSE superclass, even though the same entry is shown twater for example, c_1 is shown in both EVPLOSEs and SECRETARY on Figure 4.2. As this figure suggests, a superclass/subclass relationship such as



fIGURE 4.2 Instances of a specialization.

EVEN OVER SECRETISES somewhat rescribbles a 1:1 relationship at the instance liver (see Figure 5.12). The many difference is that its a 1:1 relationship two district courses are related, whereas on a superclass/orbitas relationship the entity in the Subclass is the same real-world entity as the entity in the Subclass is the same real-world entity as the entity in the superclass/orbitas hut is playing a *Specialized role* – dor example, on Specialized an the role of SECRETUSE, or an EVECOMES specialized on the role of SECRETUSE, or an EVECOMES specialized on the role of SECRETUSE.

There are two main reasons for including class/subclass relationships and specializations in a data racial. The first is that certain attributes may apply to some but not all entities of the superclass. A subclass is defined in order to group the entities to which these attributes apply. The members of the subclass may still share the trajecting of theor attributes with the other members of the superclass. For example, in Figure 4.1 the stratebase shelass has the specific attribute hypergSpeed, whereas the *example*, whether has the specific attribute EngType, but setservey and extincts share their other inherited attributes from the execute entity type.

The second reason for using subclasses is that some relationship types may be participated in only by enuties that are members of the subclass hor example of only subclass secret belong to a made orient, we can represent that fact by creating the subclass secret ensures of environe and relating the subclass to an entity type trade_usion via the BECONS_TO relationship type, as illustrated in Heave 4.1.

In summary, the specialization process allows us to do the following:

- Define a set of subclasses of an entity type
- Establish additional specific attributes with each subclass
- Establish additional specific relationship types between each subclass and other enrity types or other subclasses

4.2.2 Generalization

We can think of a receive focess of abstraction in which we suppress the differences among several entity types, identify their common features, and generalize them into a single superclass of which the original entity types are special subclasses. For example, consider the entity types are special subclasses. For example, consider the entity types are special subclasses. For example, consider the entity types are special subclasses. For example, consider the entity types are special subclasses, for example, consider the entity types are special subclasses. For example, consider the entity type view, it is shown in Figure 4.35. Both over and more subclasses are new subclasses of the generalized superclass venters. We use the term generalization to refer to the process of defining a generalized entity type from the given entity types.

Notice that the generalization process can be viewed as being functionally the inverse of the specialization process. Hence, in Figure 4.3 we can view [GR, 1806] as a specialization of vertex, rather than viewing vertice as a generalization of Ge and Ouck. Similarly, in Figure 4.1 we can view memory vertice as a generalization of Ge and Ouck. Similarly, in Figure 4.1 we can view memory vertice as a generalization of specialization and specialization diagrammatic notation to distinguish between generalization and specialization is used in some design mechanologies. An arrow pointing to the generalized super-last represent a generalization, whereas arrows pointing to the specialized subclasses represent a specialization. We will out use this notation, because the decision as to which process is more appropriate in a particular situation is often subjective. Appendix A gives some of the suggested alternative diagrammatic polations for schema diagrams and class diagrams.



FIGURE 4.3 Generalization. (a) Evolentity types, call and teack. (b) Generalizing call and teack into the superclass science.

So far we have introduced the concepts of subclasses and superclass/subclass relationships, is will as the specialization and generalization processes. In general, a superclass or subclass represents a collection of entities of the same type and hence obsolds enters an entry type, that is oby superclasses and subclasses are shown in certaingles in EEE diagrams, like entry types. We next discuss in more detail the properties of specializations and generalizations.

4.3 CONSTRAINTS AND CHARACTERISTICS OF SPECIALIZATION AND GENERALIZATION

We hist discuss constraints that apply tool single specialization or a single generalization. For energie, our discussion refers only to specialization even though it applies to both specalization and generalization. We then discuss differences between specializations/generalization knows (undeple inheritation) and formershes (single inheritation), and claborate on the differences between the specialization and generalization processes during conceptual database selection.

4.3.1 Constraints on Specialization and Generalization

In general, we may have several operations defined on the same critity type for superclass), as shown in Figure 4.1. In such a case, contries may belong to subclasses in each of the specializations. However, a specialization may after consist of a single subclass only, such as the boost of specialization in Figure 4.1; in such a case, we do not use the circle notation.

In some specializations we can directione exactly the entities that will become members of each solutions by placing a condition on the value of some attribute of the superclass Such solutions are called **predicate-defined** () **c** condition-defined) subclasses. For example, the execute entity type has an archive Job Type, as shown in Figure 4.4, we can specify the condition of membership in the subclass by the condition (Job Type = "Secretary"), which we call the **defining predicate** of the subclass. This condition is a construant specifying that exactly those entities of the subclass. We display a predicate-defined subclass by writing the predicate condition next to the line that connects the subclass to the specialization encle.

If all subclasses in a specialitation have their membership condition on the same articluse of the supervises, the specialization itself is called an **attribute-defined specialization**, and the attribute is called the **defining attribute** of the specialization, and the attribute is called the **defining attribute** of the specialization.⁶ We display an orthogonalized specialization, by placing the defining orthogonalized many next to the arc from the circle to the superclass, as shown in Figure 4.4.



FIGURE 4.4 FER diagram notation for an attribute-defined specialization on JohType.

in. Such an attribute is called a discumption in 13(1) terrained away

When we do not have a condition for determining membership in a sub-lass, the subclass is called **user-defined**. Membership in such a subclass is determined by the operation to add an entity to the subclass, hence, member-hip is specified ordinationly for each only by the associated sub-statically be each only by the associated sub-statically.

two other constraints may apply to a specialization. The first is the disjointness constraint, which specifies that the subclasses of the specialization must be disjoint. This means that an entity can be a member of at most one of the subclasses of the specialization. A specialization that its attribute-defined implies the disjointness curvation if the attribute used to define the membership predicate is single valued. Figure 4.4 illustrates this case, where the **dua** observed stards for disjoint. We also use the disjoint, its illustrates the specialization (assesses of a specialization must be disjoint, its illustrated by the specialization (assesses) of a specialization must be disjoint, its illustrated by the specialization (asses) concorer, satisfies are ordered in Figure 4.1. If the subclasses are not constrained to be disjoint, their sets of entities may overlaps that is, the same iter-lowerfalle entity may be a member of more them one subclass of the specialization. This case, which is the default, is displayed by placing an entit the careles at shown in Figure 4.5.

The second constraint on specialization is called the completeness constraint, which not be rotal or partial. A total specialization constraint specifies that every entity in the super-loss must be a member of an least one subclass in the specialization. For example, it every neuroper constraint for nither an -convergencess or a statistical entropy, then the specialization broker_member, seconvergencess() of Figure 4.1 is a rotal specialization of convergences, shown in EER diagrams by using a double line to connect the superclass rotheoret. A single line is used to display a partial specialization, which alloss an entity not to belong to any of the subclasses. For example, if some matter entities do not belong



HOURE 4.5 ELE diagram notation for an overlapping (nondisjoint) specialization.

to any of the subclasses (secretary, evaluated (tomotion) of lighter 4.1 and 4.4, then that specialization is partial

Nonce that the disjunctness and completeness constraints are independent. Hence, we have the following four possible constraints on specialization:

- Disjourt total
- Disjonat, partial
- Overlapping, total
- Overlapping, partial

Of course, the correct constraint is determined from the real world meaning that applies to each specialization. In general, a superclass that was identified through the extended tios process usually is **total**, because the superclass is *deticed from* the subclasses and hence contains only the entities that are in the subclasses.

Certain insertion and deletion rules apply to specialization (and generalization) is a consequence of the constrainty spirulied earlier. Some of these rules are is follows:

- Deleting an entity from a supercloss implies that it is automatically deleted from all the subclasses to which it belong-
- Inverting an entity in a superclass nucleis that the entity is mandatorily inserted in all perdense-defined (in attribute-defined) subclasses for which the entity satisfies the detiring predicate.
- Inserting an entity in a superclass of a total specification implies that the entity is mandatorily inserted in ar least one of the subclasses of the specialization.

The reader is encouraged to make a complete list of rules for insertions and deletions for the various types of specializations.

4.3.2 Specialization and Generalization Hierarchies and Lattices

A subclass itself may have further subclasses specified on at forming a hierarchy or a furrice of specializations. For example, in Figure 4.6 managers a subclass of specializations also a superclass of specialization gamma is this tepresents the real-world ore-traint that every engineering non-nect is required to be an engineer. A specialization hierarchy has the constraint that every subclass participates in a subclass in only one classified as relationships that every subclass has only one parent, which results in a free structure. In contrast, for a specialization lattice, a subclass can be a subclass in noise data one classified as relationship. Hence, Figure 4.6 is a lattice.

Figure 4.7 shows another specialization lattice of more than one level. This may be part of a conceptual schema for a assistently database. Notice that this an angement would

The notation of using single or double lines is similar to that for partial or total participation of an entity type in a relationship type, as described in Christer 3.



EGURE 4.6 A specialization lattice with shared sub- Jays more serve, www.re-

have been a hierarchy except for the surgert assistent subclass, which is a subclass in two distinct class/subclass relationships. In Figure 4.7, all person entities represented in the datase are members of the sensor entity type, which is specialized into the subclasses (server). This specialization is overlapping; for example, an aluminus may also be an employee and may also be a student pursuing an advanced digree. The subclass (server) is the superclass for the specialization (deviced prime), mensionantic (control, and the subclass (later) is the superclass for the specialization (deviced prime), mensionantic (control, student, mensionantic (control, student, mensionantic (control, student, second class), statest, statest is also a subclass of storest. Finally, superclass for the specialization (storest, finally, superclass) is the superclass for the specialization (storest, finally, superclass) is the specialization and storest. Finally, superclass for the specialization (storest, finally, superclass) is the specialization (storest, finally, superclass) is the specialization may leave of storest, finally, superclass for the specialization (storest, finally, superclass) is the specialization may leave of storest, finally, superclass (statest) is the specialization (storest) (storest).

In such a specialization lattice or hierarchy, a subclass inherits the attributes nor only affix-liter target class but also of all its predecessor superclasses at the may to the root of the hierarchy or lattice, for example, an entry in **CAPTUATE_STUDENT** inherits all the attributes of the hierarchy, where a **Rearch**. Norice that an entry may exist as several leaf node of the hierarchy, where a **Rear node** is a class that has no sufe bases of as one. For example, anember of **Rearchy, where a Rear node** is a class that has no sufe bases of as one. For example, anember of **CRAPUTE_STUDENT** may also be a member of **EREARC_ASSISTENT**.

A subclass with more dependence superclass is called a shared subclass, such as excisionated to be concept known as multiple inheritance, where the dualed subclass entitieteened in the concept known as multiple inheritance, where the dualed subclass entitieteened is dualed subclass. Nonce that the existence of at least one shared subclass leads to a lattice (and hence to multiple inheritance); at no shared subclass existed, we would have a heruchy turber than a lattice. An important rule related to multiple inheritance can be to struct by the example of the shared subclass subclass in a lattice.



FIGURE 4.7 A specialization lattice with multiple inheritance for a interview database.

inhents apprihums from both concerns and stongst. Here, both searchest and stongst inheret the sum analysis from previou. The rule stones that of an antibure for relationship) originaring in the story, repeated (search) is inherited more than once you different paths (search) and to over an in the lattice, then at should be included only once in the should solve (second associated by another and story). Hence, the attributes of possible or an inherited code only once in the story (second associated by association on the should be included only once in the should be appreciated as (second associated by associated by an another the story) in the lattice of the attributes of possible or an an an an attribute or the story (second associated by a story).

It is also important to note that some inheritance mechanisms that allow include inheritance do not allow an entity to have included types, and hence an entity can be a transfer of only one class.² In such a model, it is also necessary to create additional shared subcloses as leaf nodes to cover all possible combinations of classes that may have some entry belong to all these classes simultaneously. Hence, we would require the same seven additioned to some seven

Although we have used specialization to illustrate our discussion, similar concepts apply qually to generalization, as we mentioned at the beginning of this section. Hence, we can also speak of generalization hierarchies and generalization lattices.

4.3.3 Utilizing Specialization and Generalization in Refining Conceptual Schemas

S hi some models, the class is former restricted to be a kofforste of the kner order or brince

tool schema deags. So far, we have a hierarchy; we then realize that STUGENT_ASSISTANT is a shared subclass, since it is also a subclass of STUGEST, leading to the lattice.

It is possible to prove at the same hierarchy or lattice from the other direction. In such a case, the process involves generalization rather than specialization and corresponds to a **bortom-up conceptual synthesis.** In this case, designers much first discover entity types such as a size in a context, and so one then this case, designers much first discover entity types such as a size in a context, and so one then they generalize [case(a)] into store(case(a)) into its into store(case(a)) into its into store(case(a)) into its int

In structural terms, hierarchies or lattices resulting from either process may be identical; the only difference relates to the manner or order in which the schema superclasses and sub-lasses were specified. In practice, it is likely that neither the generalization process nor the specialization process is tollowed structly, but that a combination of the two processes is employed. In this case, new classes are continually incorporated into a hierarchy or lattice is they become opparent to users and designets. Nonce that the notion of representing data and knowledge by using superclass/subclass methods and lattices is quite common in knowledge-based systems and expert systems which combine database technology with artificial intelligence techniques. For example trans-based knowledge representation schemes closely rescable class literarchies. Specialization is also common in software engineering design methodologies that are losed on the object oriented paradigm.

4.4 MODELING OF UNION TYPES USING CATEGORIES

Ail of the superclass/subclass relationships we have seen thus far have a single superclass. A shared subclass such as softwarsatve_wavess in the fattice of Figure 4.6 is the subclass in three dware? superclass/subclass relationships, where each of the three relationships have angle superclass. It is not uncorridon, however, that the need ansats for modeling a single superclass/subclass relation however, that the need ansats for modeling a single superclass/subclass relationships into a superclass/subclass relationships, where each of the used ansats for modeling a single superclass/subclass relationship with more down one superclass, where the superclasses represent different energy types. In this case, the subclass will represent a collection of objectivities a subset of the UNON of distinct entry, repose we call such a subclass o union type or a category.²

For example, suppose that we have three entity types: PEPSUP, EAXE and EUPPARY. In a database for vehicle registration, an owner of a vehicle can be a person, a bank thatding a here one a vehicle), or a company. We need to create a class (collection of entities) that includes entities of all three types to play the role of reliade owner. A category SNEX that is a variables of the UNION of the three entity sets of 2000xx. RANK, and GERSON as created for this purpose. We display categories in on FFE diagram as shown in Figure 4.8. The superclasses

9. Course of the tenu category is based on the fact (Entitie-Category-Relationship) model (Elusion grad = 585). rousiant associated FERSON are contracted to the circle with the U symbol, which stands for the spinian operation. An are with the subset symbol examents the circle to the (subclass) using category. If a defining predicate is needed, at is displayed next to the line from the



superclass to which the predictre applies. In Figure 4.8 we have two configures, not a which is a subclass of the union of Steppe news, and Operator, and a Costan a power, a which is a subclass of the union of cast and react.

A category has two or more systemilasses that may represent distinct coshs types, whereas other superclass/subclass relationships always have a single superclass. We can compare in category, such as done in Figure 4.8, with the sourcestro paragets shared subclass of Figure 4.6. The latter is a subclass of each of the three superclasses reactives, moveers, and superclasses reactives that is a member of systemetry mathematic paragets in an entity that is a member of systemetry paraget must be an excepted, a subclass, and a superclasses (sets of child excepted paragets is a subclass of each of the other hand, a category is a subclasse of the three subclasses (sets of child excepted paragets). On the other hand, a category is a subset of the amendement of its superclasses. Hence, an entity that is a member of over must be accepted of the amendement of its superclasses. Hence, an entity that is a member of over must be a subset of the amond of its superclasses. First, an entity that is a member of over must be a subset of the amendement of its superclasses. First, an entity that is a member of over must be a subset of the amendement of its superclasses. First, an entity that is a member of over must be a constraint of the amendement of the appreciases. First, an entity that is a member of over must be a constraint that an two of the appreciase in Figure 4.8.

Attribute inhoritance works more selectively in the case of conegories. For example, its Fagure 4.8 each Gave entry inherits the attributes of a corpose, a 2003ca, or a fam, depending on the superclass to which the entrity belongs. On the other band, a shared subclass such as transportation (secore 4.6) inherits all the attributes of as superclasses second activities, and associate

It is interesting to exite the difference between the category KONALRO (WHELL (Lighte 4.4) and the generalized superclass virua (Trigore 4.3b). In Figure 4.3b, every sar and every truck is a warrar) but it. Figure 4.5b, the REALSTREE_WHELL entercory includes some dues and some marks but not necessarily all of them (for example, some cars of marks may not be registered). In general, a special nation of generalization such as that in Figure 4.3b, if it were Semial, would not preclude viscous from containing other types of entities, such as motoreveles. However, a category such as realistered visitive of Figure 4.8 inglues that only cars and trucks. Sur not other types of entities can be members of accessing visual.

A caregory can be total or partial: A total category holds the many of all entries in its superclasses, whereas a partial category can hold a subset of the amon. A total caregory is represented by a double line connecting the category and the rinch, whereas partial categories are indicated by a single from.

The superclasses of a category may have different key attributes, as demonstrated by the basis category of Figure 4.6 or they now have the sum key attribute, as demonstrated by the keystesus, volume category. Nonce that of a category is total (nor partial), it may be represented alternatively as a total specialization for a total generalitation). In this case the choice of which representation to use is subjective. If the two classes represent the same type of entities and share numerous artificates, including the same key attributes, specialization/generalitation is preferred, otherwise, categorization (union type) is increappropriate.

4.5 AN EXAMPLE UNIVERSITY EER SCHEMA AND FORMAL DEFINITIONS FOR THE EER MODEL

In this section, we first give an example of a database schema in the FBR model to illustrate the use of the vorticus concepts discussed here and in Chapter 3. Then, we summarise the SPR model concepts and define them formally in the same manner in which we tograph defined the concepts of the basic ER model in Chapter 3.

4.5.1 The UNIVERSITY Database Example

For our example dirations: application, consider a usiversite database that keeps track of sudents and their majors, transcripts, and registration as well as of the aniversity's course offerings. The database also keeps track of the sponsored research projects of facility and gallate sudents. This scheme is shown in Figure 4.9. A discussion of the requirements that left of shows been a follows.

For each person, the database maintains information on the person's Name [Name], some security number [Ssn], address [Address], sex [Sex], and both date (PDate]. Two socialises of the eccent entity type were identified: socially and specific attribute-of social are triak (Rank) (assistant, associate, adjunct, research, visiting, etc.), office [E0fice], office phone [FPhone], and safary [Salary]. All faculty members are related to the academic department(s) with which they are attributed [Sreaks] (a faculty members can be academic department(s) with which they are attributed [Sreaks] (a faculty members are related to the academic department(s) with which they are attributed [Sreaks] (a faculty members can be academic department), so the relationship is M(N). A specific attribute of sizes is [Class] (freshman = 1, sophimming = $2, \ldots$, graduate student = 51. Each student is also related to his or her major and minor departments. If known ([ewiss] and [whick]), to decompt accords the or she is correctly arranding [960 (more)], and to the courses or pleted [newscore]. Each transcript instance includes the gride the student received [0rade] in the course section.

DREP_CONSET is a subclass of subset, with the defining predicate Class = 5. For each endance student, we keep a list of previous degrees in a composite, maltivatued withbate [Degrees]. We also relate the endoate student to a faculty advisor 'Aby1504] and to a thesis committee [COMPTTEE], if one exists.

An academic department has the attributes name [DName], telephone [DUhme], and ance number [Office] and is related to the teculty member who is no Avanperson [aut6], and to the college to which it belongs [50]. Each college has attributes college rune [CName], office number [COffice], and the name of its dean [Dean].

A course has arributes course number [Or], course name [Oname], and course decorption [ODesc]. Several sections of each course are offered, with each section basis the arributes section number [Sec=] and the year and quarter in which the section was detect [Peur] and [Orr[S¹² Section numbers unsquely identify each section. The section-beng offered during the current quarter are in a subclass current_section of sections.

^{17.} We issume that the specific vestion rather than the second system is used in this measurements



FIGURE 4.9 An EER conceptual schema for a invessme database.

the administration of the # CharlentQu and Year + administration. Each section is related to the instruction who toight or is reaching it ([104067], it that instructions in the database.

The category instruction of second students who are supported by teaching or and includes all faculty, as well as enduate students who are supported by teaching or teacch. Finally, the endity type GMM keeps thatk of research grants and contracts twided to the university. Each grant has attributes grant table [Title), grant number [Ne], the availing agency [Agency], and the starting date [StDate]. A grant is related to our general investigator [91] and to all researchers in supports (subtact). Each instance of support haves attributes the starting date of supports (subtact). Each instance of support haves attributes the starting date of support [Start], the ending date of the support of known [End], and the procentare of time being sprint on the project [Time] by the usarcher being supported.

4.5.2 Formal Definitions for the EER Model Concepts

We now summarize the fitter model concepts and give formal definitions. A class¹ is a set of collection of entities, this includes any of the E06 scheme constructs that group entities, such as entity types, subclasses, superclasses, and categories. A subclass *S* is a class whose entities must always be a subset of the entities in another class, called the superclass *S* of the superclass *S* and class that its of the entities we denote such a relationship. We denote such a relationship by US. But such a superclass below relationship, we must always have

 $S \subseteq C$

A specialisation $\mathbb{D} = \{S_1, S_2, \dots, S_n\}$ to a set of subclasses that have the some superclass G, that is, G(S₁) is a superclass/subclass relationship for $i = 1, 2, \dots, n$. G is called a generalized entity type (or the superclass of the specialization, or a generalization of the sub-less $\{S_1, S_2, \dots, S_n\}$). D is said to be total affive always (at any point matrix) have

 $\bigcup_{i=1}^{n}S_{i}^{i}=G$

Otherwise, I is such to be partial. I is said to be disjoint of we always have

 $S_i \cap S_i := i\mathbb{Z}$ (empty set) for $i \neq j$

Otherwise, 2 is said to be overlapping.

A solutions S of C is said to be **predicate-defined** if a predicate β on the articlater of C is used to specify which particles in C artimembers of S that is, $S = x \beta \beta$, where C $\beta \beta$ is the set of entries in C that lands β_0 . A subclass that is not defined by a predicate is called **pseudefined**.

The use of the work costs here differs from its more communication in object oriented programming trigings such as (198). In (199), a case is a structured type definition independent as applicable finature top metors).

A specialization \mathbb{Z} (or generalization \mathbb{O}) is said to be **attribute-defined** it is producer: (A = q), where A is an attribute of G and q is a constant value from the domain of A, is used to specify membership in each subclass \hat{S}_i in \mathbb{Z} . Nonce that if $q \neq c$ for $i \neq j$, and A is a single-vidual attribute, then the specialization will be disjoint.

A category T is a class that is a subset of the union of a defining superclasses $U_1 D_1, \dots, D_n$ a > 1, and 5 formally specified as follows:

 $T \subset (D_1 \cup D_2 \cup \cup D_1)$

A predicate p on the attributes of D can be used to specify the members of each D_i that are members of *i*, it is predicate is specified on every D_i we get

 $\Gamma = (D_1|p_1) \in D_2(p_2) \dots \cup D_p(p_n))$

We should now extend the definition of relationship type given in Chapter 3 by allowing any class—net rady nev entity type—to participate in a relationship. Hence, we should replace the words entity type with class in that definition. The graphical notation of FFR is consistent with SR because all classes are represented by rectangles.

4.6 REPRESENTING SPECIALIZATION/ GENERALIZATION AND INHERITANCE IN UML CLASS DIAGRAMS

We now discuss the UML extration for generalization/specialization and inheritance. We already presented basic UML class diagram monotion and terminology in Section 3.8. Figure 4.10 dimetrates a possible UML class diagram corresponding to the EPR diagram in Figure 4.7. The basic potonian for generalization is to connect the subclasses by vertical lines to a herizontal line, which has a triangle contracting the horizontal line through another section have to the superclass (see Figure 4.10). A blank triangle indicates a specialization/generalization with the displot constraint, and a filled triangle indicates an orerlaping constraint. The root superclass is called the base class, and leaf nodes are called leaf elasses. Both single and multiple inheritorice are permitted.

The above discussion and example (and Section 3.8) give a birel overview of UML class diagrams and terminology. There are many details that we have not discussed because they are noiside the scope of this book and are manify relevant to software engineering. For example, classes can be of various types:

- Abstract classes define antibutes and operations but do not have objects corresponding to those classes. These are manify used to specify a set of attributes and operations that can be inherited.
- Concrete classes can have objects (corrities) instantisted to belong to the class
- Template classes specify a reinplate that can be further used roadefine other classes.



EGURE 4.10 A DML class chagram corresponding to the Frk diagram in Figure 4.7. Illustrating UMU notation for specialization/generalization

In database design, we are mainly concerned with specifying concerne plasses whose callections of objects are permanently (or persistently) stored in the database. The bibliographic notes at the end of this chapter give some references to books that describe complete details of CMU. Additional multiplication to CMU is covered in Chapter 12, indonect modeling in general is further discussed in Chapter 20.

4.7 RELATIONSHIP TYPES OF DEGREE HIGHER THAN TWO

In Section 3.4.2 we defined the degree of a relationship type as the number of portrapating energy types and called a relationship type of degree two tonary and a relationship type of degree three terminy. In this section, we claborate on the differences between breaty and higher-degree relationships, when its chaose higher degree or binary relationships, and remetraints on higher degree relationships.

4.7.1 Choosing between Binary and Ternary (or Higher-Degree) Relationships

The ER diagram existion for a ternary relationship type is shown in Figure 4.11a, which displays the schema for the relationship set of supervise a set of relationship instances (x, j, p), where x is a softener who is currently supplying a soft p to a former j in general, p relationship type R of degree n will have n edges in an ER diagram, one connecting R to each participating entity type.

Figure 4 110 shows an 6% diagram for the three binary relation-buy types Can_30807. 2515, and SOFFETTS. In general, a ternary relationship type represents different information that do three binary relationship types. Consider the three binary relationship types 646 soperies used sopered. Suppose that the support between supported and ever, uncludes an instance (v, p) whenever supplier v can supply part p (to any protect): uses, between Pio(EC)and puttimulates an instance (j, j) whenever project integrat p_j and foth (16), however torsutes and two ect, includes an instance (s, j) whenever supplier symplicize some just to primet). The converte of three relationship instances (x, p), y, p), and (x, p) in the Subtry, of 55, and 5010 (15, respectively, does not necessarily imply that an instance (s, i, p) exists in the zernary gelationship same, because the normal is different. It is often tricky to decide scherher a particular relationship should be represented as a relationship type of degree a or should be broken down into several relationship types of smaller degrees. The designer must base this decision on the semantics on memory of the particular situation being represented. The typical solution is to include the remark relationship this one or more of the binary relationships, of they represent different meanings and if all are needed No the application

Some database design tools are based on variations of the ER model that permit only binary relationships. In this case, a ternary relationship such as same a most be represented as a weak entry type, with no pointak key and with three identitying relationships. The three participatine entry types separately, WKT, and PROFED are together the owner entry types (see Figure 4.11c). Hence, an entry to the weak entry type support of Figure 4.11c is identified by the combination of its three evener entries from supporter, PART, and PROFET.

Another example is shown in Figure 4.12. The terrary relationship type DEFERS represents information on instructors offering courses during particular semesters, hence it includes a relationship instructor (z, z, z) whenever two outcours zones is zones zone form sentences zones in the three binary relationship types shown in Figure 4.12 have the following includes a relationship instructor to the instructors who can easily the rooms, is out taking test tester relates a course to the instructors who can easily the rooms, is out taking relates a seniester to the instructors who caused of the rooms, is out taking relates a seniester to the instructors who course during that seniester, and attended taking relates a seniester to the courses offered during that seniester by any numeric. These is runary and binary relations/ups represent different intermation, for cert on constraints should hold among the relationships, for example, a relationship instance (i, k) exists in causes in cause only on instance (i, k) exists in taking precises.



EGURE 4.11 Ternary relationship types, (a) The subary relationship, doi Three binary relationships and equivalent to subary, the subary represented as a weak entity type.



FIGURE 4.12 Areabor example of tormary versus binary relationship types.

an instance (s, c) exists in effects bounds and an instance (s, c) exists in (w_resolutions), the reverse is nor absaus true, we may have instances (i, c), exists in (w_resolution) the trace boosty relationship types with no corresponding instance (i, s, c) in effects. Note that in this example, based on the meanings of the relationships, we can inter the instances of instances in press, but we can inter the instances of instances in press, but we can inter the instances of instances of the relationships, we can inter the instances of instances of the relationships, we can inter the instances of instances of the relationships, we can inter the instances of instances of the relation and or interview and or rest press and presses are related and an and can be left into the instances of the relationships.

Although in general three binary relationships cannot replace a turnary relationship, they may do so under certain additional constraints. In our example, if the Gaugestan relationship is left fan instructor can teach one course, and a course can be raught by only one instructor), then the terrary relationship CETAS can be left out because it can be inferred from the three binary relationships (SECTEACH, TROCH, COURT, and CETESCE SERVA, The scheme designer must analyze the meaning of earth specific situation to double which of the h pary and require relationship types are needed.

Notice that it is possible to have a weak entry type with a term ty (or nors) identify by relationship type. In this case, the weak entity type can have second owner entity types. An example is shown in Figure 4.13.

4.7.2 Constraints on Ternary (or Higher-Degree) Relationships

There are two notations for specifying structural constraints on o-aiv relationships, and they specify different constraints. They should thus both he used if it is important to fully specify the structural constraints on a ternary of higher-degree relationship. The first



FIGURE 4.13 A weak entries type tweeview with a ternary identifying relationship type.

unionen is based on the cardinably ratio netation of binary relationships displayed in Figine 3.2. Here, a.1. M, or N is specified on each participation are (both M and N symbols word for more or any analyse). ² Let us illustrate this constraint using the 0.650 relationship in Figure 4.00.

Becall that the relationship set of supervise set of relationship instances (s, a pl, where are a solution) is a regular, and pair a gain. Suppose that the constraint exists that for a particular project-pair conductation, only one supplier will be used (only one supplier supplies a particular pair to a particular project). In this case, we place 1 on the straint participation, and M. N on the security gain participations in Figure 4.11. This specifies the constraint that a particular (j, p) construction can appear at not once in the relationship set because each such (project, part) conductation anappear at not once in the relationship set because each such (project, part) conduction uniquely identified in the relationship set by its (j, p) conduction, which makes (j, p) is to grady identified in the relationship set by its (j, p) conduction, which makes (j, p) a key for the relationship set, in the participation that have a one specified on them are not required to begain of the identifying key for the relationship set.¹³

The scondenoration is based on the (num mark) notation displaced in Figure 3.15 for break relationships. A (mm, max) on a participation ben, specifies that each entry is cloud to an least out and at most mark relationship instances in the relationship set. This constraints have no bearing on determining the key of an usary relationship, where $a > 2^{16}$ but specify a different type of constraint, that places restrictions on how many relationship instances each entry can participate in.

18 This is also rice for cardinatary ratios of binary relationships.

19 The turn, max's constraints can determine the keys for binary relativiships, though

¹² Tais neuron of eways to determine the key of the relation by relation as we discuss in Chapter 7.

4.8 DATA ABSTRACTION, KNOWLEDGE REPRESENTATION, AND ONTOLOGY CONCEPTS

In this sector, we discuss in obstract terms some of the modeling concepts that we described quite specifically in our presentation of the FR and FFR models in Chapter i and earlier in the chapter. This terminology is used both in conceptual data modeling and in artificial intelligence (iteratore when discussing knowledge representation (libbles are discussing knowledge representation (libbles are discussing knowledge representation (libbles are discussing knowledge) representation (libbles are discussing knowledge) to take the concepts of the describes the concepts of the domain of knowledge by creating an ontology¹¹ that describes the concepts of the domain. This is then used to store and manipulate knowledge for drawing inferences, making decisions, or just answering questions. The grafs of KR are smaller to those of semantic data models, but there are some important similarities and differences between the two de-tighnes:

- Both disciplines use an abstraction process to identify countron properties and important aspects of objects in the minimodal (domain of discourse) while suppressing insegnition) differences and unimportant details.
- Both disciplines provide scalespis, constraints, operations, and larguages for defining data and representing knowledge.
- xR is generally broader in scope than semantic data models. Efferent forus of knowledge, such as rules fused in inference dedicerton, and search), incomplete and default knowledge, and reinportal and sparial knowledge, are represented in KR schemes. Database models are being expanded to include some of these concepts (see Chapter 24).
- S8 schenses include reasoning mechanisms that deduce additional facts from the facts stored in a database. Hence, whereas most current database systems are limited to answering direct queries, knowledge-based systems using K8 schemes can answer queries that involve inferences over the stored data. Database technology is being extended with interence mechanisms (see Section 24.4).
- Whereas most data models concentrate on the representation of database schemas, or meta-knowledge, Kx schemes often mix up the -chemas with the instances them selves in order to prioride flexibility in representing exceptions. This effect results in methanemics when these KR schemes are implemented, repreally when compared with databases and when a large origination data for facts) pends to be stored.

in the section we discuss thus **abstraction** concepts that are used in both sengence data models, such as the ELE model, and KR schernes (1)) classification and paramitation, (2) identification, (3) specialization and generalization, and (4) aggregation and association. The paired concepts of classification and instantiation are inverses of one another, as are generalization and speculitation. The concepts of aggregation and association are also related. We discuss these abstract concepts and their relation to the concept representations used in the EES model to clarify the data abstraction process and

^{15.} An our-dup is somewhat similar to a consepting schema, but with more anothelige, rules, and exceptions.

to improve our independing of the related process of concuptual schema design. We descahe section with a biref discussion of the term *antolog*, which is being used widely in recent knowledge representation research.

4.8.1 Classification and Instantiation

The process of classification involves systematically assigning similar objects/entiries to elsective classification (vertices). We can now describe (m. 108) or reason about (vertice) the classes rather than the individual objects. Collections of objects share the same types of attractes, relationships, and constraints, and by classifying objects we simplify the process of discovering rheir properties. **Instantiation** is the inverse of classification and refers rathe generation and specific examination of discovering objects of a classification and refers in the generation and specific examination of discovering objects of a classification and refers in the generation and specific examination of discovering of a class. Hence, an object observe the techned to incodure class by the **IS-AN-INSTANCE-OF** of **IS-A-MEMBER-DF** relationship. Although UMI diagrams do not display instances, the DAI diagrams allow a rom of instantiation by permitting the display of individual objects. We define describe this feature in our introduction to a 300.

In general, the objects of a class should have a similar type structure. However, some objects may display properties that differ in some respects from the other objects of the class, these exception objects also need to be modeled, and KR schemes allow more varied exteriors than do database models. In addition, certain properties upply to the class as a whole and not to the individual objects; KX schemes allow such class properties. USE digrams also allow specification of class properties.

In the EFR model, entities are classified into entity types according to their basic attractes and relationships. Entities are further classified into subclasses and categories basic on additional similarities and differences (exceptions) among them. Relationship instances are classified into relationship types. Hence, entry types, subclasses, categories, and relationship types are the different types of classes in the EER model. The EFR model days is a provide explicitly for class properties, but it may be extended to do so. In 1986, aboves are classified into classes, and it is possible to display, both class properties and industrial objects.

Knowledge representation models allow multiple clossification schemes in which one class is an observe of another class (called a meta-class). Notice that this cannot be inpresented directly in the EER model, because we have only two levels—classes and instance. The only relationship among closes in the EER model is a superclass/subclass relationship, whereas in some SR schemes an additional classificatione relationship can be inpresented directly in a class hierarchy. An instance may itself be another class, allowing multiple-level classification schemes.

4.8.2 Identification

Identification is the obstraction process whereby classes and objects are made anothely dentitable by means of some identifier. For example, a class name uniquely identifies a whole class. An additional ineclamation is necessary for telling defined object instances apart by means of object identifiers. Moreover, it is necessary to identify multiple manifestations in the database of the same real-world object. For example, we may have a tuple <Matthew Clarke, 612618, 376 9521 > in a reason relation and another tuple <301-54-0336, CS, 3.8 > in a storight relation that happen to represent the same real-world entity. There is no way to identify the fact that there two database objects (tuples) represent the same real-world entity on loss we make a provision of daugi tone for appropriate crowrelationing to supply this (dentification). Hence, identification is needed at two leves

- To distinguish among database objects and classes
- To identify doublose objects and to relate them to their real world counterparts

In the EF8 model identification of schema constructs is based on a system of unique names for the constructs. For example, every class in an EE6 schema—whether it is an entity type, a subclass, a category, or a relationship type—must have a distruct name. The names of attributes of a given class must also be distoret. Roles for unambiguously identifying attribute name references in a specialization or generalization battice or hierarchy are medically well.

At the object level, the values of key attributes are used to distinguish among entries of a particular entity type. For weak entity types, entries are identified by a combination of their own parcial key values and the entities they are related to in the owner entity type(s). Relationship instances are identified by some combination of the entities that they relate, depending on the cardinality nace specified.

4.8.3 Specialization and Generalization

Specialitation is the process of classifying a class of objects into more specialized subclasses. Generalitation is the inverse process of generalizing several classes into a higherlevel abstract class that includes the objects in all these classes. Specialitation is conceptual refinement, whereas generalitation is conceptual somhese. Subclasses are used in the EER model to represent specialitation and generalitation. We call the relationship between u subclass and its superclass an IS-A-SUBCLASS-OF relationship, or simply an IS-A relationship.

4.8.4 Aggregation and Association

Aggregation is an abstraction enocept for building composite objects from their component objects. There are three cases where this concept can be related to the FSR model. The first case is the situation in which we aggregate attribute values of an object to four the whole object. The second case is when we represent an aggregation relationship as an ordinary relationship. The chiral case, which the SFR model does not provide for explicitly involves the possibility of combining objects that are refined by a particular relationship instance into a higher-level aggregate absent. This is sometimes useful when the higher-level aggregate object is useful to be related to mother object. We call the relationkep between the primitive objects and their segregate object (S-A-PART -OF) the inverse is called (S-A-COMPONENT-OF, UML provides for all three types of aggregation.

The abstraction of **association** is used to associate objects from several *indefendent* classs, i lence, it is somewhat similar to the second use of aggregation. It is represented in the LER model by relationship types, and in UMI. By associations. This abstract is anomship is called **IS-ASSOCIATED-WITH**.

In order to understand the different uses of aggregation better, consider the EX storma shown in Engine 4.14a, which scores information about interviews by job applicants to various companies. The class corever is an aggregation of the attributes (or compariant objects) CName (company name) and CAddress (compare address), whereas the attributes (or angregate of Sau, Name, Address, and Phone. The relationship ambates Cantae Name and Canta (Flame represent the name and phone number of the person in the company who is responsible for the interview. Suppose that some nerviews requirem job offers, whereas others do nor. We would like to treat offeretware a day in associate it with the offer. The schema shown in Figure 4.14b is interview because a requires each interview relationship instance to have a job offer. The schema shown in Figure 4.34c is not allowed, because the FR model does not allow relationships among a schembigs (although 0)(1 does).

One way to represent this situation is to greater a higher level aggregate class composed in 92900, paraget, that, and the REP and to relate this class to the action as shown in from 14d. Although the FEP model as described in this book does not have this facility, which semiantic data models do allow in and call the resolution object a composite or malcolar object. Other models from entity types and relationship types and outly and have permit relationships among relationships, is illustrated in Figure 1.14c.

To represent this introtion concertly in the E0 model as described here, we need to chare a new weak entity type to represent these situations contextly in the E0 model by man flence, we can always represent these situations contextly in the E0 model by creating additional entity types, although it may be conceptually more desirable to allow liter representation of agenegation, as in Figure 4.14d, or to allow relationships among clation-lips, as in Figure 4.14d.

The main structural distinction between aggregation and association is that when an association instance is deleted, the participating objects may continue to exist. However, two support the notion of an aggregate object—tor example, a CAS that is made up of object—torsective, cassis, and tiszs—then deleting the apprepare CAR object amounts to deleting the apprepare CAR object amounts to deleting the is component objects.

4.8.5 Ontologies and the Semantic Web

In event years, the amount of comparement data and information available on the Web hospitaled out of control. Many different models and formats are used. In addition to the database models that we present in this book, much information is stored in the form of degments, which have considerable less superior than database information does. One result project that is attempting to allow unformation exchange among computers on the Web is called the Semantic Web, which attempts to create knowledge representation.



FIGURE 4.14 Aggregation (a) the relationship type transversion in hidding, or costs in a termary relationship type tincorrectly (c) Having the selects, is relationship participate in other relationships (generally not allowed in ER). (d) Using aggregation and a composite timoleculari object (generally not allowed in ER). (e) Correct representation in ER.

models that are spure general in order to to allow meaningful information on barge and south among nuclimes. The concept of our sligs is considered to be the most promising bait for achieving the goals of the Semantic Web, and is closely related to knowledge represtruction. In this section, we give a buff introduction to what an intrology is and how it can be used as a basis to a from the information understanding, search, and exchange

The study of our observes attempts to describe the structures and relationships that are cossible in reality through some common so-calculary, and so it can be considered as a way to describe the knowledge of a certain community about reality. Oncology originated at the fields of philosophy and metaphysics. One commonly need defaution of orthology is "o specification of a conceptualization." ¹⁰

In this definition, a conceptualization is the set of concepts that its used to represent the part of reality or knowledge that is of interest to a commutative of users. Specification refers to the language and vocabulary terms that are used to specify the conceptualization. The ontology includes both specification and conceptualization. For example, the same conceptualization may be specified in two different languages, giving two separate autologies. Based on this quite general definition, there is no consensor on what exactly on autology is Some possible techniques to describe ontrologies that have been mentioned are 2 follows:

- A thesaurus (or even a dictionary or a glossary of terms) describes the relationships between words (voca sidary) that represent various concepts.
- A taxonomy describes how concepts of a particular area of knineledge are related using structures similar to those used on a specialization or generalization.
- A detailed database schema is considered by some to be in ontology that describes the concepts (entities and attributes) and relationships of a turnitorial from reality.
- A logical theory uses concepts from mathematical logic to try to define concepts and disir interrelationships

Usually the concepts used to describe or tologies are quite similar to the concepts we docased in conceptual modeling, such as entities, attributes, relationships, secondations, and so or. The mean difference between an outpology and, are a database schema is that it e schema is usually limited to describing it small subset of o manisorial from reality in only to score and manage data. An ontology is usually cert-idered to be more general in that is should attempt to describe a part of reality as completely as possible.

4.9 SUMMARY

In this chapter we first discovered extensions to the ER model that improve its representatorol capabilities. We called the resoluting model the enhanced ER or FER model. The contept of a subclass and its superclass and the related mechanism of attribute/relationship inheritance were presented. We saw how or its sometimes necessary to create additional classes of enriptes, either because of additional specific attributes or because of specific retationship repex. We discussed two main processes for defining superclass/subclass hierarchies and larrices specialization and generalization.

We then showed how to display these new constructs in an FER diagram. We also discussed the various types of constraints that may apply to specialization or generalization. The two main constraints are total/partial and disjoint/overlapping. In addition, a defining predicate for a subclass of a defining attribute for a specialization may be specified. We discussed the differences between user-defined and predicate-defined subclasses and between aser-defined and arcthone-defined specializations. Finally, we docussed the concept of a category or ration type, which is a sobset of the enior of two or more classes and we gave fermal definitions of all the concepts presented.

We then introduced some of the notation and tertimology of UML for representing specialization and generalization. We also discussed some of the issues concerning the difference between binary and higher-degree relationships, inder which encourstances real should be used when designing a conceptual schema, and how different types of constraints on a arry relationships may be specified. In Section 4.8 we discussed briefly the discipline of knowledge representation and how it is related to semantic data modeling. We also give an overview and similarity of the types of ubstract data representation concepts, classification and instruction indemnification, specification and generalization, and aggregation and association. We saw how affit and UML concepts are related to each of these

Review Questions

- 4.1. What is a subclass? When is a subclass needed in data modeling?
- 4.2 Define the following terms superclass of a statellast superclass/subclass telafonship to a relationship, specialization, generalization, category, specific docatt attributes specific relationships.
- 4.3. Discuss the machinusm of notribute/relationship inheritanaci. Why is it useful?
- 4.4. Decreas user-defined and predicate-defined subclasses, and identify the differencebetween the two.
- 4.5. Diacous user-defined and attribute-defined specializations, and identify the differences between the two.
- 4.0. Discuss the two main types of constraints on specializations and generalizations.
- 4.7. What is the difference between a specialization hierarchy and a specialization lattice?
- 4.6. What is the difference is tween specialization and generolization? Why do we not display the difference in schemo diagrams?
- 4.9. How does a category differ from a regular shorts' subclass? What is a category and for? Illustrate your answer with examples.
- •Fig. For each of the following OSE terms (see Sections 3.8 and 6.6), discuss the corresponding term in the FEB model, if only object, class, as existing aggregation generalitation, reability, anothers, discrimination, link, link another, reflexive association, qualified association.
- 4.11. Discuss the main differences between the notation for LTR schema diagrams and USE class diagrams by comparing how common concepts are represented in each

- 5.0. Discuss the two notations for specifying constraints on neary relationships, and what such can be used for.
- 4.11 List the corious data obstraction concepts and the corresponding modeling concepts in the EPS model.
- 4.14 What aggregation feature is missing from the FER model: blow can the FER model, be further enhanced to support (C).
- 4.15 What are the main similarities and differences between conceptual database nockeline techniques and knowledge representation techniques?
- Discuss the similarities and differences between in ontology and a database schema.

Exercises

- 4.17. Design an EER schema for a database application that you are interested in Special and all constraints that should hold on the database. Make sure that the schema has at least five entity types, four relationship types, a weak entity type, a superclass/sure lass relationship, a category, and an assay for 21 feationship type.
- •18. Consider the ease ER schema of Figure 3-18 and suppose that it is necessary to keep mark of different types of accuracy (savies_acces, correspond_acces, i) and constructs, cass, output, iii). Suppose that it is also describle to keep track of each account's teases (case, cass, correspond), with drawing cheeks, i.e. if and each loan's respect to both of these include the annount, date, and time. Modely the some schema, using ER and EFR concepts of specialization and generalization. State are assumptions you make about the additional requirements.
- 4.19. The fullowing narrative describes a simplified version of the organization of Olympic facilities planned for the summer Olympics. Draw an sHe diagram that shows the entity types, arithmetics, relationships, and spougherings for this application. State any assumptions you make. The Olympic toribries are divided into sports complexes. Sports complexes are divided into one sport and pultigen types. Multisport complexes have ansas of the complex designated for each sport with a location indicator (e.g., comet, NE corner, etc.). A complex has a location, chief organizing individual, rotal occurried area, and so on, Each complex holds a series of events (e.g., the track studies) may hold many different faces). For each event there is a planued date, dupation, number of participaties, number of officials, and so on. A rester of all officials will be maintained together with the list of events each official will be involved in Different equipment is needed for the events (e.g. goul posts, poles, parallel (sits) as well as for maniferance. The two types of faultites tene-sport and multisport1 will have different types of information. For each type, the number of facilities needed is kept, together with an approximate Indect.
- 4.50 Identity all the important concepts represented in the ldway database case study described here. In particular, identity the abstractions of classification (emity ryps) and relationship types), aggregation, identification, and specialization/generalization. Specify (min. max) cardinality constraints, whenever possible, law.

details that will offeet the eventual design but have no bearing on the conceptual design. List the semantic constraints separately. Draw an EEK diagram of the library database

Case Study: The Georgia Tech Library (GTL) has approximately 16200 intenders, 100,000 titles and 250,000 volumes for an average of 2.5 copies perfocide). About 10 percent of the volumes are out on loos at any one time. The librarians ensure that the books that members want to borrow are available when the members want to borrow them. Also, the librarians must know flow many copies of each book are in the library or out on loan at any given time. A catalog of bracks is available online that lists books by author, title, and subject area. For each title ig the library or book description is kept in the catalog that ranges from one sentence to several pages. The reference librarians want to be able to access this description when members require interface about to be able to access the description when members require into history about a book Library stables developing when members require into history about a book Library stables developing when members require to history about a book Library access the description about a book Library stable developing when members require to history associate history about a book Library to head a library stable developing when members require to history associate history about a book Library stable developing on the bit provide the history assistance.

Broks can be checked currier 21 days. Members are allowed to have only five brocks out at a time. Members could return books within three to four weeks. Most members know that they have one week of grace before a notice is sent to them, so they try to get the book returned before the grace period ends. About 5 percent of the members have to be sent termindets to return a book. Most overdue books are returned within a mentils of the due date. Approximately 5 percent of the inverdue books are either kept or never returned. The most active members of the library are defined as these who borrow at least ters times during the year. The type 1 percent of membership does 15 percent of the forrowing, and the rep 10 percent of the membership does 40 percent of the borrowing. Almor 20 percent is members are totally marine in that they are members but rever borrow.

To become a member of the library, applicants fill out a form including their SoN, campus and home modine addresses, and phone numbers. The librarians their issue a manifered, machine-readible cand with the member's phono as a This card is good for four years. A month before a card is pites, a notice is sent to a member for renewal. Professors at the institute are considered automatic members. When a new faculty member, outs the institute, his er ben information is pulled from the conflexive records and a library circles mailed to has in her campus address. Professors are allowed to theck out books for three-month intervals and have a two-week price period. Renewal notices to professing up sent to the campus pasticidness.

The library does not lend scene books, such as interence books, rare books and maps. The librarians must differentiate between books that can be left and those that cannot be left. In addition, the librarians have a bay of some books they are interested it, negating but connot obtain, such as rare or out-of-print books and books that were lost or destrived but have not been replaced. The librarians most lence a system that keeps track of books that cannot be left as well as books that they are interested in ocquitite. Some books may have the same title: theratore, the title cannot be used as a means of identification. Every books adoutted by its externational Standard book. Number (ISBN), a active enternational code assigned to all books. Two books with the same rirle can have different. ISBN- if they are in different knowages of have different bindings (bask cover or soft cover). Editions of the same book have different ISBN-

The proposed dotabase section must be designed to keep track of the members, the peoks, the catalog, and the borrowing activity

- 4.11 Design a database to keep track of information for an air museous. Assame that the following requirements were callected:
 - The moseum has a collection of ART_OBJECTS. Each ART_OBJECT has a unique IdNet an Artest (if known), a Year (when it was cruited, if known), a Title, and a Description. The art objects are categorized in several ways, as discussed index.
 - ART (ADE) Is are consignized based on their type. There are three main types, estimate, set const, and statue, plus another type called ones to accommodate objects that do not fall title one of the three main types.
 - A symplex basic PointType (cd. warercolor, etc.), material on which it #TrownOn (parer, carves, wood, etc.), and Style (modern, abstract, etc.).
 - A structure or a status has a Material from which it was created (wood, stone, etc.). Height, Weight, and Style.
 - An art object in the criter category has a Type (print, photo, cic) Fand Style.
 - wit_DBCEC's are also entergoneed as PESPAKENT_cortECTION which are owned by the museum takes charge information on the DateAcquired, whether it is OrDisplay or stored, and Cost) or BOKINEE, which has information on the Collection (from which it was borrowed). Direformoved, and DateReturned.
 - aut_objects also have information describing their country/culture using information on country/culture of Origin (Irahaa, Egypture, American, Indian, etc.) ind hpicals (Renausance, Modern, Ancient, etc.).
 - The maseum keeps track of sources information, it known. Nome, DateBorn (if known). DateDied (if not living). CountryOfOrigin: Epsels, ManStyle, and Descriptions. The Name is assumed to be unique.
 - Different EXHIBITIONS occur, each howing a Name, StartDate, and EndDate synstrices are related to all the circobjects that were on display during the exhibition.
 - Information is kept on other carts taks with which the nuiseum interacts, including Name (intique), Type (inuseum, personal, cre.), Description, Address, Phone, and current ContactPerson.

Priny an EFR suberna diagram for this application. Discuss any assumptions you made, and that justify your EFR design choices.

4.22 Figure 4.15 shows an example of on EEK diagram for a small private apport database that is used to keep track of arplanes, their owners, apport employees, and pilots. From the requirements for this database, the following information was collected: Each strender has a registration number [Reg#], is of a particular plane type [or_type], and is stored in a particular hangar [storet_iw]. Each store two has a model number [Model], a capacity [Capacity], and a weight [Weight]. Each store has a number [Number], a capacity [Capacity], and a location [Location]. The database also keeps track of the Genere of each plane [store] and the ENPLOYEES who



FIGURE 4.15 FER schema for a search statistic database.

have maintained the plane [wirvista]. Each relationship instance in www.relates an airplane to an owner and metides the purchase date [Fdate]. Each relationship instance in wirvists relates an employee to a service record [SERVICF]. Each plane indergnes service many times; hence, it is related by [6486, STRETE] to a miniber of strene records. A service meand includes as attributes the date of mantepane [Date], the nature of hours spent on the work [Hears], and the type of work done [Workcode]. We use a weak entity type [SERVICF] to represent anglishe service.

because the airplane registration number is used to adentify a service record. An owner is either a person or a corporation. Hence, we use a upcon type (category) [code] that is a subset of the union of corporation [concert] are subset as person or a corporation. Hence, we use a upcon type (category) [code] that is a subset of the union of corporation [concert] are substated period [cutegory] [code] that is a subset of the union of corporation [concert] are substated and period [cutegory] [code] that is a subset of the union of corporation [concert] are substated period [cutegory] [code] that is a subset of the union of corporation [concert] are substated and period [cutegory] [code] that is a substate of the union of corporation [concert] are substated and period [cutegory] [code] and period [cutegory] [code] [code] [code] [code] and period [cutegory] [code] [code

•23 Show how the OVENESSIE EER scheme of Figure 4.9 may be represented in USU netation.

Selected Bibliography

Main papers have proposed conceptual or semantic data models. We give a representation list here. One group of popers, including Abrial (1974). Seako's 1(AM model (1975), the SIAM method (Verheijen and VanBekkom 1982), and Bracchi et al. (1976), presents smanne includes that are based on the concept of binary relationships. Another group of tark papers discusses methods for extending the relational model to enhance its modeling equilations. This methods for extending the relational model to enhance its modeling equilations. This methods the papers by Schnid and Swenson (1975), Navathe and Schlehnek (1978), Codd's (M/T model (1979), Furtado (1978), and the structural model of Wisleighd and Elmasti (1979).

The FR model was proposed originally by Chen (1976) and is formalized in Ng (1981). Since then, numerous extensions of its modeling capabilities have been proposed, as in Schenemann et al. (1979). This Samos et al. (1979), Torrey et al. (1986). Gegella and ridhenstein (1991), and the emity-caregory relationship (ECR) model of Elmism et al. (1985). Smoth and Smuth (1977) present the concepts of generalization and aggregation. The semantic data model of Hammer and McLeod (1981) introduced the concepts of the data set of Hammer and McLeod (1981) introduced the concepts of the data and large souther advanced modeling concepts.

A survey of sentactic data modeling appears in Hall and King (1987). Eack (1991) discusses design and transformations of conceptual schemas. An dysis of constraints for 4ary relationships is given in Souton (1998). USE is described in dutañ in Beech, Rudsingh, and Jacobson (1999). Fowler and Scotti (2000) and Stevens and Pooles (2000) give concise introductions to OSE concepts.

Fensel (2000) is a good reference on Semantic Web, Uschold and Gruninger (1990) and Upder (1995) discuss corrologies. A recent entity roue of Communications of the 30% is devised to ontology concepts and applications.





RELATIONAL MODEL: CONCEPTS, CONSTRAINTS, LANGUAGES, DESIGN, AND PROGRAMMING





The Relational Data Model and Relational Database Constraints



This chapter opens Part II or the book on relational databases. The relational model was this intextioned by Ted Codd of USV Research in 1972 in a coasie paper (Codd 1970), and arranged immediate attention due to us simplicity and mothematical fourieration. The model uses the concept of a mathematical relation—which looks somewhat like a table of values—as its base building Elock, and has its theoretical base of set theory and instance predicate legic to this chapter we discuss the twice characteristics of the model and the test theoretical base of set theory and instance predicate legic to this chapter we discuss the base characteristics of the model and the constraints.

The first commercial implementations of the relational model becaute available in the eroy 1980s, such as the Orocle 1980s and the SQLO's system on the 50x5 operating system by IRM Since them, the model has been implemented in a large number of commercial systems. Oncert popular relational 2080s (ROMPS) include 1982 and Informat Dynamic Server throug (35). Oracle and Relb (from Oracle), and SQL Server and Access (from Microsoft).

Because of the importance of the relational model, we have decoted all of Para II of this textbook to this model and the languages associated with it. Chapter 6 covers the operations of the relational algebra and introduces the relational calculus nonation for too types of calculus implementations and domain calculus. Chapter 7 relates the relational model data structures to the constructs of the EC and 656 models, and presents algorithms for designing a relational database is hered by mapping a conceptual schema in the EV or futurable (see Chapters 3 and 4) into a relational representation. These mappings are incorporated into many database design and CAEE usals. In Chapter 8, we describe the

^{1.0} Assistands for complicer orded software elegimenting

SQL query language, which is the standard for commercial relational OBMss. Chapter 9 discusses the programming rechniques used to access database systems, and presens additional topics concerning the SQL language—constraints, views, and the portor, of connecting to relational databases via OORC and 10900 standard protocols. Chapters E and U in Part III of the book present another aspect of the relational model, namely the formal constraints of functional and individual dependencies; these dependencies in used to develop a refutional database design theory based on the concept known as nonnellicition.

Data models that preceded the relational model include the hierarchical and network models. They were proposed in the 1966s and were implemented in early 1980s during the 1970s and 1980s. Because of their historical importance and the large existing user base for these onests, we have included a summary of the highlights of these models in appendices, which are available in the Web site for the book. These models and systems will be with us for many years and are now referred to as legacy database systems.

In this chapter, we concentrate on describing the basic principles of the relational model of data. We begin by defining the modeling concepts and notation of the relational model in Serman 5.1. Serman 5.2 is devited to a discussion of relational constraints that are now considered an important part of the relational model and are automatically unforced in most relational pixes. Section 5.3 defines the update operations of the relational model and discusses how violations of integrity constraints are handled.

5.1 RELATIONAL MODEL CONCEPTS

The relational model represents the database as a collection of solutions. Informally, each relation resembles a table of values or, to some extent, a "flat" file of records. For example, the database of files that was shown in Figure 1.2 is similar to the relational model representation. However, there are important differences between relations and tiles, as we shall scon see.

When a relation is thought of as a table of values, each row in the table represents a collection of related data values. We introduced entity types and relationship types as concepts for modeling real-world data in Chapter 3. In the telational model, each row in the table represents a fact that typically corresponds to a real-world entry or relationship. The table name and column names are used to help in interpreting the meaning of the values in each row. For example, the first table of Figure 1.2 is called stored because each row tegresents facts about a particular student entry. The column matters—Nome. StudentNumber, Class, and Major—specify how to interpret the data values in each row based on the column cach value is in. All values in a column are of the same data type.

In the formal relational model terminology, a row is called a tiple, a column header is called an *autilistic* and the table is called a velicion. The data type describing the types of values that can appear in each column is represented by a *dimitin* of possible values. We now define these terms—dominit, taple, *autilistic*, and relation—shore precisely.
5.1.1 Domains, Attributes, Tuples, and Relations

A demain D is a set of atomic values. By **atomic** we mean that each value in the domain is indivisible as far as the relational model is concerned. A common method of specifying i demain is to specify a data type from which this data values turning for domain are from d is also useful to specify a name for the domain, to help in interpreting its values. Since examples of domains follow:

- U-A_phone_nombers: The set of rendern phone numbers calld in the United States.
- Local_plane_transfers: The set of seven-digit plane numbers valid within a paracelatorea code in the United States.
- Scenty security numbers. The set of valid nime-digit social security numbers.
- Names: The set of character strings that represent names of persons.
- Grade_point_averages: Possible values of computed grade point averages; each must be a real (Rooting-point) number between 0 and 4.
- Employee_ages: Possible ages of employees of a company, each must be a value between 15 and 50 years old.
- Academic_department_to new The set of academic department names in a university such as Computer Science, Economics, and Physics
- Academic department_codes: The set of academic department codes, such as US, ECON, and PDVS.

The preceding ne called logical definitions of domains. A data type or format is also specified for each domain. For example, the data type for the domain USA_phone_numbers can be declared as a character string of the form tald/ddddatat, where each data remerie (declared) digit and the host three digits form a valid releptone are reade. The data type for Employee ages is an integer number between 15 and 80. For Academic_Jepartment_names, the data type is the set of all character strings that represent valid lepartment_names. A domain is thus given a name, data type, and format. Additional information for interpreting the values of a domain can also be given; for example, a numeric domain such as Person_weights should have the units of measurement, such as proofs or kilograms.

A relation schema? *R*, denoted by $R(A_1, A_2, ..., A_n)$, is and depert a relation come *B* and a list of attributes $A_1, A_2, ..., A_n$. Each **attribute** A_n is the name of a rule played by sime domain *D* in the relation schema *R*. *D* is called the **domain** of *A*, and is denoted by **dom**(*A*₁). A pelotion schema is used to describe a pelotion; *R* is called the **name** of this relation. The **degree** (or **arity**) of a relation is the number of attributes at of its relation schema is the number of attributes at of its relation schema.

2 A relation schempt is sometimes called a relation scheme.

An example of a relation schema for a relation of degree seven, which describes any endepris, is the following:

sought(Name, SSN, FlorizPhone, Address, OfficePhone, Apr. OPA).

Using the data type of each arribute, the definition is sometimes written as

stocon (Name: string, SSN: string, HomePhone: string, Address: string, OfficePhone: string, Age integer, GPA: really

For this relation schema, succes is the name of the relation, which has seven articlates. In the above definition, we showed assument of generic types such as similar integer to the attributes. More precisely, we can specify the following previously defined domines for some of the attributes of the specify the specify the following previously defined domines for some of the attributes of the specify mean domiName) = Names, dom(SSN) = Social security_numbers; dom(HennePhone) = Local_phone_numbers; and dom(GRA) = Grade_point_swerngs. It is also possible to refer to attributes of a relation schema by their position within the relation; thus, the second attribute of the storest relation is SSN, whereas the locath attribute is Address.

A relation (or relation state)⁴ *i* of the relation schema $R(A_1, A_2, ..., A_n)$, also denoted by i(R), is a set of retuples $\mathbf{r} = \{i, i_1, ..., i_n\}$. Each *n*-tuple *i* is an order d list of *n* values $\mathbf{t} = \{i_1, i_2, ..., i_n\}$, where each value $n_i + i_i = 5n_i$ is an element of doub A_i) or is a special null value. The i^4 value in tuple *i*, which corresponds to the attribute A_i is referred to as $\{A_i\}$ for i_1 if we use the positional notation). The terms relation intension for the schema R and relation extension for breakness relation schema R and relation extension for breakness relation schema R.

Figure 5.1 shows on example of a COGMI relation, which corresponds to the (COGMI schema just specified, Each rayle on the relation represents a particular student entity, We



FIGURE 5.1 The attributes and tuples of a relation success.

3. With the large improve to photic numbers during by the prohitegenest of neighbors, some metry obtain a payment have and optication radius so that several get head duding has been discuss transitioned by the case, we would use USA physical prophets as the definite.

4. This has also been called a relation instance. We will not use this term become using a used is refer to evaluate up to use.

display the relation as a rable, where each ruple is shown as a way and each attribute corresponds to a column feeder indicating a role of interpretation of the values in that column. Null values represent attributes whose values are unknown or do not exist for some individual studies to be.

The earlier definition of a relation can be restated more formally as follows. A relation (or relation state) r(R) is a mathematical relation of degree n on the domains dem(A₁), dam(A₂), dom(A_n), which is a subset of the **Cartesian product** of the domains that define R:

 $i(R) \subseteq \text{tdom}(A_1) \times \text{dom}(A_2) \times \dots \times \text{dom}(A_n)$

The Cortesian product specifies all possible combinations of colors from the macriying domains. Hence, if we denote the total number of values, or cardinality, in a daman D by (D1) (assuming that all domains are finite), the total number of tuples in the Cirtesian product is

 $\operatorname{dout}(A_1(1 \times \operatorname{dout}(A_2))) \geq \ldots \geq \operatorname{bdun}(A_n)$

Of all these possible combinations, a relation state at a given rune—the current relation state—reflects only the valid tuples that represent a transitulat state of the real world changes, so does the relation, by being transformed into another relation state. However, the schema *R* is relatively static and also not change exceptively influquently—for example, as a result or adding an arribute to represent new information that was not originally stored in the relation.

It is possible for several attributes to have the same domain. The attributes indicate different roles, or interpretations, for the domain, but example, in the storest relation, the same domain Local_phone munders plays the tole of HomePhone, referring to the home phone of a student," and the role of OfficePhone, referring to the "office phone of the student."

5.1.2 Characteristics of Relations

The early redemoning of relations implies contain characteristics that make a relation duttermit from a file of a rable. We now discuss some of these characteristics.

Ordering of Juples in a Relation. A relation is defined as a set of tiples. Mathenatically, elements of a set have no order among them; hence, tiples in a relation do not have any particular order. However, in a file, records the physically stored on disk (or in incurry), so there always is an order among the records. This indefining induction has a condord, ob, and have records in the file. Similarly, when we display a relation as a rable, the rows are displayed in a certain order.

Tuple ordering is not part of a relation definition, because a relation arigingts to represent theirs at a logical or abstract level. Many logical orders can be specified on a relation. For example, ruples in the stoppst relation in Figure 5.1 could be logically ordered by values of Name, or by SSN, or by Age, or by some other arribute. The definition of a relation does not specify my order. There is its preference for one logical ordering over another. Hence, the relation displayed in Figure 5.2 is considered ideoted to the one shown in Figure 5.1. When a relation is implemented as a file or displayed as a rable, a particular ordering may be specified on the records of the file or the rows of the table.

Ordering of Values within a Tuple, and an Alternative Definition of a Relation. According to the preceding definition of a relation, an astrople is on ordered lat of a values, so the ordering of values in a tuple – and hence of attributes in a relation schema—is important. However, an a logical level, the order of attributes and their values is not that important as long as the correspondence between autobutes and values is maniported.

An alternative definition of a relation can be given, making the ordering of values in a tuple antecessory. In this definition, a relation schema $R = [A_1, A_2, \dots, A_n]$ is a set of articlattic, and a relation state r(R) is a finite set of mapping $r = \{r_1, r_2, \dots, r_n\}$, where each tuple it is a mapping from R to D, and D is the anton of the attribute domains; that is, $D = \operatorname{dom}(A_1) \cup \operatorname{dom}(A_2) \cup \cdots \cup \bigcup \operatorname{dom}(A_n)$. In this definition, do not be implemented at $\operatorname{dom}(A_1)$ of a mapping r in r. Each mapping r is called a tuple.

According to this definition of tuple as a mapping, a tuple can be considered as a set of (<artificite>, <volue>) pairs, where each pair gives the value of the mapping from an autiplyte A, to a value v, from dom(A). The ordering of intributes is not important, because the attribute name appears with its value. By this definition, the neurophysis shown in Figure 5.5 are identical. This makes sense at an abstract or logic d level, since their really is no reason to prefer lawing one attribute value oppear before another in a tuple.

When a relation is implemented as a file, the attributes are physically ordered as fields within a record. We will generally use the **first definition** of relation, where the attributes and the values within tuples are ordered, because it simplifies much of the incriation. However, the alternative definition given here is more general.⁵

Values and Nulls in the Tuples. Each value is a tople is an atomic value; that is it is not divisible intra components within the tranework of the basic relational model. Hence, composite and multivalued attributes (see Chapter 3) are not allowed. This

| STUDENT | Nare | SSN | HomePhone | A13-365 | OfficePriore | Age | GFA |
|---------|-----------------------|-------------|------------------|------------------------|--------------|------------|------|
| | Oct Colorban. | 772-112-20 | o al | 3452 Exp: Forel | 74341,855 | 75 | 3.55 |
| | Валина Велого | 500891706 | ј қаннас | ANN Fordaria (Hori- | n d | -9 | 325 |
| | Crentes <u>Cooper</u> | 489-82-1101 | 575 <u>9</u> -21 | <u>,2651</u> wiki Jane | 49-6485 | i # | 382 |
| | Kallenne Aulty | 361-62-1845 | 075-9409 | 125 Kirlsy Road | n, 1 | 10 | 2 09 |
| | Deryamin Bayer | 006 61 2400 | 373-1016 | 2912 Usedonnel Lann | 6,4 | 19 | 3 ZI |

FIGURE 5.2 The relation scenar from Figure 5.1 with a different order of tuples

 As we shall see, the alternative definition of relation is useful when we decrease query processing in Chapters 15 and 16.

```
L+ <//i>

        L+ <//i>
        Address, 3452 Eign Road),

        (OfficePhone, 748-1254),
        Address, 3452 Eign Road),

        (OfficePhone, 748-1254),
        Address, 3452 Eign Road),
```

```
1 - < (Address, 3452 Figh Brad), (Name, Dick (Davideon) (sev. 422-11-2320), (Age, 25)
(OfficePhone,749-1253), 844, 3-53, (NomePhone, nulls)
```

EGURE 5.3 Two identical tuples when the order or attributes annivalues is not part is relation definition.

molel is sometimes called the flat relational model. Much of the theory behind the relational model was developed with this assumption in mind, which is called the first nornal form assumption." Hence, multivalued attributes must be represented by separate relations, and composite attributes are represented only by their simple component attributes in the Usic relational model.⁵

An important concept is that of asslits which are used to represent the Values of attributes that may be unknown or may not apply to a tuple. A special value, called null, is used for these cases. For example, in Figure 5.1, some student tuples have null for their affect phone because they do not have an other that is, office phone does not apply to does students). Another student has a null for home phone, presumably because either be does not apply to not have a home phone phone or he has one but we do not know it twalke is trakmach, in general, we can have a home phone or he has one but we do not know it twalke is trakmach, in general, we can have a home phone or he has one but we do not know it twalke is trakmach, in general, we can have a none phone does not apply to this tuple." An example of the last report of null will be out if we add injuttribute Values, such as 'value unknown," 'value exists but is not available," or "attribute does not apply to this tuple." An example of the last report for tuples that represent doing students. It is possible to devise different ordes for different meanings of null values. Incorporating different types of null values into the relational model, operations (see Chapter 6) has priven difficult and is conside the scope of our presentation.

interpretation (Moanring) of a Relation. The relation schema can be interpreted is a decharation of a type of assertion. For example, the schema of the Stimust relation of Figure 5.1 asserts that, in general to student entity has a Name, SSN. HomePhone, Address, OfficePhone, Age, and GPA. Each tuple in the relation can then be interpreted is a fact of a particular instance of the assertion. For example, the first rople in Figure 5.1 asserts the fact that there is a student scheme to me to Benjamin Bayer, SSN is 305-61-305. Age is 19, and 40 sm.

Nouce that some relations may represent locts about courses, whereas other relations may represent facts about sclarouslips. For example, a relation schema success (StudierrSSS, DepartmentCode) asserts that students major in academic departments. A tuple in this

We discuss this assumption in more detail in Chapter 1C.

I. Extensions of the relational model remove these testricitions, for example, opported anothal systems allow complexistriculated methods, as its the **non-first normal form** or **nested relational** reaches as we shall see in Chapter 22.

relation relates a student to his or her imagor department. Hence, the relational model represents facts about both entities and relationships rangingly as relations. This semetimes compromises understandability because one has to guess whether a relation represents as entity type or a relationships). The mapping procedures in Chapter 7 show how different constructs of the ER and EER models get converted to relations.

An alternative unceptication of a relation schements as a predicate; in fais case, the values in each tuple are interpreted as values that valiefy the predicate. This interpretation is quite institution the context of logic programming languages, such as Prolog, because it illows the relational model to be used within these languages (see Section 24.4).

5.1.3 Relational Model Notation

We will use the following notation, in our presentation.

- A relation schema & of degree is denoted by R(A₁, A₂, . . . , A_i).
- An accupie can a relation rtR) is denoted by r = <v₁, v₂, ..., v_n>, where v_n is the value corresponding contribute. A, The following normon refers to component values of ruples.
 - Both rlA [and a A, (and sometimes rld) refer to the value (, in clocartribute A).
 - Both dA₁₀ A₁₀, ..., A₂1 and (GA₁₀ A₁₀, ..., A₂) others A₁₀ A₁₀, ..., A₂ is a list of attributes from *B*, refer to the subsigile of values <*v*₁₀, *v*₂₀, ..., *v*₁> from *t* corresponding to the attributes specified in the list.
- The letters Q, B, S denote relation names
- The letters q. r. edenoise relation states
- The letters raise denote tuples.
- In general, the name of a relation schema such as studiate abo (adicates the current set of taples in that relation—the current relation state—whereas studiat(Name, SSN,) refers only to the relation schema.
- An attribute A can be qualified with the relation name R to which it belones by issue the database R A—for example, stagger.Name or staggy.Arc. This is because the same name may be used for two attributes in different relations. However, all attribute names in a particular relation must be distinct.

As an example, consider the mple t = < [Rarbins Perison], (533-6941238), (539-8461), (7384 Foutana Lanci, null, 19, 3.25) from the - (00-6) relation in Figure 5.1; we have t^{0} Name[= < [Rarbins Benson] >, and t[SSN, (3PA, Age] = <(553-69-1238), 3.25, 79>.

5.2 RELATIONAL MODEL CONSTRAINTS AND RELATIONAL DATABASE SCHEMAS

So far, we have discussed the characteristics of single relations. In a relational database, there will typically be many relations, but the tuples in these relations are usually related

in various ways. The state of the whole database will correspond to the states of all intelations at a particular prime in time. There are generally many restrictions of constraints on the actual values in a database state. These constraints are derived from the fales in the natiovorid that the database represents, as we discussed in Section 1.6.8.

In this section, we discuss the various restrictions on data that can be specified on a relational database in the form of constraints. Constraints on databases can generally be divided into three many categories:

- Constraints that are inherent in the data model. We call these inherent modelhased constraints.
- 2. Constraints that can be directly expressed in the schemas of the data model, typically by specifying them in the DDT (data definition language, see Section 2.3.1). We call these schema-based constraints.
- 5 Constraints that connor be directly expressed in the schemas of the data model, and hence must be expressed and enforced by the application programs. We call these application-based constraints.

The characteristics of relations that we discussed in Section 5.1.2 are the inherent constraints of the relational model and belong to the first category for example, the constraint that a relation cannot have diplicite tuples is an inherent constraint. The constraints we discuss in this section are of the second category, namely, constraints that clobe expressed in the scheme of the relational model via the OB. Constraints in the third category are more general and are difficult to express and enforce within the data node, so they are usually checked within application programs.

Another important category of constraints is detailed productes, which include protokial dependencies and analtisalities dependencies. They are used monthly for testing the "goodness" of the design of a relational database and are officed in a process called isomalization, which is discussed in Chapters 10 and 11.

We now discuss the main types of constraints that can be expressed in the relational model—the schema-based constraints from the second category. These include domain constraints, key constraints, constraints on nulls, entity indentity constraints, and information any relations.

5.2.1 Domain Constraints

Duratin constraints specify that within each ruple, the value of each articlute A must be an atomic value from the domain dom(A). We have alreade discussed the ways in which domains can be specified in Section 5.1.1. The data types accordated with domains type rails include standard numeric data types for integers (such as short integer, integer, and long integer) and real numbers (float and double-precision float). Characters, boo eans, read- ength strings, and carable-length strings are also readable, as are date, time, times string, and, in some cases, money data types. Other possible contains may be described by a striange of values from a data type or as an enumerated data type in which all possible values are explicitly listed. Rather than describe these in detail better we discuss the data types offered by the SQL-99 relational strange and in Section 8.1.

5.2.2 Key Constraints and Constraints on Null Values

A reference of defined as a set of taples. By definition, all elements of a set are distinct hence, all implies in a relation must also be distinct. This means that no two tuples can have the same combination of values for all their attributes. Usually, there are other subsets of **attributes** of a relation schema *K* with the property that no two tuples in any refarron state r of *R* should have the same combination of values by -k, then for any root distinct tuples t_j and j_j in a relation state of **R**, we have the constraint that

ણુકારી સંગોધની

Am such set of artributes et is called a **superkey** of the relation schema B. A superkey SK specifies a angueness constraint that **r** o two distinct taples in any state **r** of B can have the same value for SK. Every relation has at least one default superkeys, the set of all us outplutes. A superkey can have technicaria artificates, however, so a more useful concept is that of a key, which has no redundancy. A **key** K of a relation schema B is a superkey of K with the additional property that removing any autiliant A from N leaves a set of attributes K² that is not a superkey of K any more. Hence, a key satisfies two constraints

- Two distinct ruples in invision of the relation conner have isleptical values for Cult the artificities in the key.
- It is a matheal superlay that is, a superkey from which we cannot remove any attributes and still have the uniqueness constraint in condition 1 hold.

The first condition applies to both keys and superkeys. The second condition is required only for keys. For example, consider the studyor relation of Figure 5.1. The attribute set (SSN) is a key of studyt because no two student tuples can have the same value the SSN? Any set of attributes that includes SSN—for example, ISSN, Name, Age[—is a superkey. However, the superkey (SSN, Name, Age] is not a key of student, because removing Name or Age or both from the set still leaves as with a superkey. In general, any superkey formed from a single attribute is also a key. A key with multiple attributes must work or difficult to have the unqueness property hold.

The value of a key attribute can be used to identify muquely each tuple in the relation. For example, the SSN value 305.61.2435 identifies uniquely the ruple corresponding to Benjamin Bayer in the STMENT relation. Notice that a set of off-thetes reporting a key is a property of the relation schema; it is a constraint that should hold on every valid relation state of the schema. A key is determined from the meaning of the intributes, and the property is maximum valid. It must return the block we user new ruples in the relation. For example, we cannot and should not designate the Name new ruples in the relation. For example, we cannot and should not designate the Name intribute of the schema in Figure 5.1 as a key because it is possible that two students with identical rate will exist at some point in a valid state."

^{9.} Note that SSN is also a superkey.

Somes are somenices used as keys, but then some attribut—so choss sppendime or ordinal miniber must be used to distinguish between identical a times.

| CAR | License Number | EngneSerialNumber | Mahe | Model | Yaar |
|-----|---------------------------------|-------------------|-----------------|------------|------|
| | Ceras ABC 730 | 409352 | Fand | , Westing | 99, |
| | Hoots TVP 347 | HARGEG | Ortemotele | Castage | 40 |
| | New York MMO-22 | X83554 | Oktemobile | Deta | 95 |
| | <u>U</u> aitoma 4.2 (F <u>Y</u> | C23/a2 | <u>Mercedas</u> | | _93 |
| | Column R54 629 | V62036 | Τογοτά | Carrey | 96 |
| | To-22 R5K (20) | U025066 | | <u>xus</u> | |

FIGURE 5.4 The GR relation, with two capitidate keys LicenseNumber and EngineSerialNumber.

In general, a relation schema may have more than one key. In this case, each of the keys is called a **candidate** key. For example, the two relation in Figure 5.4 has two candidate keys ElicenseNumber and EngineSetialNumber. It is common to designate one of the candidate keys as the primary key of the relation. This is the candidate key whose values are used to identify topics in the relation. We use the convention that the attributes that form the primary key of a relation schema are underlined, as shown in Figure 5.4. Notice that when a relation schema has several candidate keys, the choice of one to become the primary key is arbitrary however, it is usually better to choose a primary key with a single attributes or a small number of attributes.

Another constraint on attributes specifies whether null values are or are not primitted. For example, if every states, tople must have a valid, nonnull value for the Nime arrithmet then Name of strong is constrained to be NIME NULL.

5.2.3 Relational Databases and Relational Database Schemas

The definitions and constraints we have docussed so tai apply to single relations and their attributes. A relational database usually contains many relations, with toyles in relations that are related in corrects ways, but this section we define a relational database and a relational database schema. A relational database schema S is a set of relation schemas $S = \{b_1, B_2, \ldots, B_n\}$ and a set of integrity constraints to the relational database state¹⁶. Of of S is user of relation states ($b^2 = (b_1, T_2, \ldots, T_n)$ such that each τ_i is a state of R_i and such that the relation states $(b^2 = (b_1, T_2, \ldots, T_n)$ such that each τ_i is a state of R_i and such that the relation states ($b^2 = (b_1, T_2, \ldots, T_n)$ such that each τ_i is a state of R_i and such that the relation states satisfy the integrity constraints specified in (C). Figure 5.5 shows a relational database schema that here call convert = here were conversed by the second states represent primary keys. Figure 5.6 shows a relational database state corresponding to the correspondence to the correspondence. We will use this schema and database state in this chapter and in Chapters 6 through θ for developing example queries in different relational languages. When we relation a relational database.

^{10.} A relational database state is sometimes called a relational database analysis. However, as we mere oned earlier, we will not use the term restance since it data applies to single tuples.



FIGURE 5.5 Schema diagram for the coverage relational database schema.

we implicitly include both its schemo and its correct store. A database store that does not obey all the integraty constraints is called an **invalid state**, and a state that satisfies all the constraints in this called a **valid state**.

In Figure 5.5, the proper operations in both operation and oper reactions stands for the same real-world concept—the number given to a department. That some concept is called awarn to state and over an exotent. Attributes that represent the same real-world concept may or have departed names in different relations. Attributes that represent different concepts in a have the same name in different relations. For example, we could have used the attribute name was for both every of succept and every of ossesses. We would have two attributes that share the same through and every of ossesses with this case, we would have two attributes that share the same three but represent different real-world concepts—project tennes and department realizes.

| EVICED | FRIAKI_ | MAN | DV9M2 | 52 7 5 | RUSIE | ADUALSS | SÐ | Selves. | SUPLESS | $_{\rm c}$ (erg. |
|---------------|----------------|----------|----------|---------------|--------------------|-------------------------|----------|---------|------------|------------------|
| | ي مالح | 'n | :en/h | 0.000406778 | 1.225-01-05 | 7.0 Easther Housen 1X | ы | 10000 | 301HH5575 | 15 |
| | وهن 6 | 1 | Waa | 300-466655 | 1265-12468 | 6 И фра Намер - 1X | 6ª | 10000 | 1444203775 | 5 |
| | ALC . | _, | ži teja | 2066-177 | T #86-01/16 T | TO Cale Spring 18 | 1 | 25000 | 997654221 | Ŀ. |
| | , classic | - | Sec.4 | Jo/564323 | 1347-06420 | 201 Beny Belanc T | _;+ T | | 8599003555 | 1 |
| | Range | ĸ | File and | 000504443 | 1 <u>962-08</u> 15 | 015 Frencies Hurden FX | ĺΝ. | 36000 | 33344.555 | 5 |
| | , me | 1 A 1 | E-george | CONCERCE | 1975 37 91 | 300 Proc. Acador TR. | <u> </u> | 2,000 | | Ŀ |
| | <u>erusz</u> [| <u> </u> | .0200 | | T <u>ree</u> use - | T_geo Laras, Focsión 12 | . * | 2900 | 99/05/221 | Ŀ. |
| | 10.000 | F | Boy | XXXXXXXXXXX | 1992 14 15 | all trans perpert for | ÷μ. | - 490.0 | 0.4 | ۱ |

| | Un 4 | ME diaor aileo | | MGF2 1346 46/65 100685 | Srv 1991 1991 | 14 <u>6</u> 89 14 14 | (HEPT_D)CATA BLACTDATE BLA | | UVUE - 01 (21 | |
|-----------|---|----------------------|-------------------------|---------------------------------|---------------------|-----------------------------------|--|-----------------|-------------------|----------|
| | 1(346e7e3 1(345e7e3 1(345e75e 1(1(66g1111) 1(346)345e | | 205 75 200 200 | _ | | | | | | |
| - | 40460430 | L² | 202 | 11 | NULECT _ | | =NAVE | PNU <u>B</u> EF | PLOCATION | DIALINE |
| | | ⊢: I | <u></u> | | - | - Pro | 0.00 | | Beter | |
| - | 1,2300 | 3 | - 16 7 | | | PN | | <u> </u> | Arcpaterd | 5 |
| - | | <u>n.</u> | 10.2 | | | Pto | unz 👘 | .1 | Hurdon | 5 |
| - | 305007777 | 20 | 20.2 | | | _ CM | (3.894 <u>54</u>) ; | <u> </u> | <u>CLANE</u> | <u> </u> |
| • | 200410777 | 10 | 10.2 | | | Here | pene2500 | 20 | Hustan | <u> </u> |
| - | | | 180 | | - | 444 | | • | <u>5.e</u> len: | <u> </u> |
| - | 2012/01/2012 | 30 | 50 | | | | | | | |
| | A.254.07 | 30 | 200 | | | | | | | |
| : | #75-1P | 20 | 15C | | | | | | | |
| - | 866842223 | :00 | La L | | | | | | | |
| | | | | | | | | | | |
| DEPENDENT | 6359 | <u>CE 36</u> | ENCENT_NAME | . ≞EX | BCATE | 1 | RELATIONSH | 1F | | |
| | | <u> </u> | A2 | 1 - 4 | 1080.020 | ٥ <u> </u> | CAUC <u>I-T</u> ER | . | | |
| | 3000-5556 | | Theorem | | 1480-0012 | 9 | 800 | | | |
| | - *11a=7375 - | - | *r | 1 - | 14/4/50 | <u>`</u> — | COLT OF | | | |
| | - 362/2135/ - | — | Allent | - М– | 1948-000 | я | 0P30 <u>0</u> F | | | |
| | <u>-20</u> 404754 | · | Weiger | | <u> </u> | | | - 1 | | |
| | 12.04/200 | | | 1. | 1948(4.2.5) | л | EAU/GOILE | | | |
| | 2010/202 | | 1 Carbon 11 | 1 P | 146/1518 | a | SPO4.65 | | | |

FIGURE 5.6. One possible database state for the conexy relational database schema

Each relational DEXIS must have a data definition language (2000) for defining a relational database schema. Correct relational DEVISS are mostly using SQL for this surplise. We present the SQL PTL in Socions 8.1 through 6.3.

Integrity constraints are specified on a database schema and are expected to held on every valid database space of that schema In Adirton to domain, key, and NOT NUU. constraints, two other rupes of constraints are considered part of the relational model: entry integrity and referential integrity.

5.2.4 Entity Integrity, Referential Integrity, and Foreign Keys

The entity integrity constraint states that no primary key volue can be null. This is because the primary key value is used to identify individual raples in a relation. Having call values for the primary key implies that we cannot identify some raples. For example, if two or more raples had notl for their primary keys, we might not be able to distibute them if we tried to reference them from other relations.

Key constraints and entity integrity constraints are specified on individual relations. The referential integrity constraint is specified between two relations and as used to maintain the consistency among tiples in the two relations. Informally, the referential integrity constraint states that a tiple in one relation that refers to another relation materefer to an existing tiple in that relation. For example, in Figure 5.6, the attribute fact of softwirfs gives the department number for which each employee works; hence, its value in every forces for the time match the cample value of some tiple in the provident relation.

To define referenced integrity more formally, we first define the concept of a foreign key. The conditions for a foreign key, given below, specify a reference all megrity constraint between the two relation schemas R_1 and R_2 . A set of autobates FK in relation schema R_1 is a foreign key of R_1 that references relation R_2 if it satisfies the following two rules:

- The attributes in FK have the same domain(s) as the primary key purphuses FK of K₂, the attributes FK are said to reference or refer to the relation R₂.
- 2. A value of % in a tuple r of the current state r₁(R₁) either occurs as a value of % for some tuble to in the current state r₂(R₂) or is sull. In the former case, we have c₁(rK) = r₂(rK)₁ and we say that the tuple t₁ references or refers to the tuple t₂.

In this definition, R_1 is called the referencing relation and R_1 is the referenced relation. If these two conditions hold, a referential integrity constraint from R_1 to R_2 is said to hold. In a database of many relations, there are usually many referential integrity constraints.

To specify these constraints we must first have a clear nucleistanding of the meaning or role that each set of attributes plays in the various relation schemas of the database. Referential integrity constraints typically arise from the valuouslips among the onlines represented by the relation is hence. For example, consider the database shown in Figure 5.6. In the Formaté relation, the arribute two refers to the department for which in simplicities works, hence, we design to find to be a torough key of Formate relating to the patient relation. This means that a value of loss in any topic t_1 of the second relation must match a value of the primary key of influence - the causes t_1 of the second relation t_2 of the generation, or the value of the value of the real of the regions to be loss t_2 of the generation, or the value of the value of the real of the regions not below t_2 of the relation, or the value of the value of the real of the regions not below to a department. In Figure 5.6 the tuple for employee 'John Smith' references the tuple for die 'Research' department, indicating that 'John Smith' works for this department.

Notion that a function key can refer to its provide and bot example, the articlete second in events to for apervisor of an employee; this is another employee, represented by a tople in the eventy relation. Hence, supervisor is a foreign key that references the events, relation itself. In Figure 5.6 the tuple for employee John Smith' references the typle for employee 'Franklin Wongs' indicating that 'Franklin Wong is the supervisor of John Smith'

We can diagrammateally doploy referenced integrity constraints by drawing a directed are from each loating key to the relation it references. For clarity, the attroched may point to the primary key of the referenced relation. Figure 5.7 shows the schema in Figure 5.5 with the referenced integrity constraints displayed in this manner.

All integrity constraints should be specified on the relational database schema if we want to entired, these constraints on the database states. Hence, the 100, includes provisions for specifying the various cypes of constraints so that the 10505 can enumerically infinite them. Most relational 198058 support key and entity integrity.



FIGURE 5.7 Referential integrity constraints displayed on the common relational database schema.

constraints, and make provisions to support referential integrity. These constraints are specified as a part of data defouries.

5.2.5 Other Types of Constraints

The proceeding integrity constraints do not include a large class of general constraints, sometimes called sensitive integrity constraints, that may have in he specified and enforced on a relational database. Examples of such constraints are "the salary of an employee should not exceed the salary of the employee's supervisor" and "the machinan number of hours an employee can work on all projects per week is 56." Such constraints can be specified and enforced and enforced work on all projects per week is 56." Such constraints can be specified and enforced sorthin the opplication programs that update the dirabase, or by using a general-purp we constraint specification language. Mechations called triggers and assertions can be used. In SQL-99, a UREATE ASSERT UN statement is used for this purpose (see Chapters 3 and 9). It is more constraint specification for these repes of constraints within the application programs that updates, because the latter are difficult and complex to use constraint specification languages, because the latter are difficult and complex to use constraint specification languages in Section 24.1.

Another type of constraint is the functional dependency constraint, which establishes a functional relationship among two sets of attributes X and Y. This constraint specifies that the value of X determines the value of T in till states of a relation; it is denoted as a functional dependency $X \rightarrow Y$. We use functional dependencies and other types of dependencies in C highers 10 and 11 as tools to analyze the quality of relational designs and to inormalize relations to improve their quality.

The types of constraints we discussed so far may be called state constraints, because they define the constraints that a calif state of the database must satisfy. Another type of constraint, called **transition constraints**, can be defined to deal with state changes in the database.¹⁵ An example of a transition constraint is: The salary of an employee can only increase.¹⁵ Such constraints are typically entired by the application programs or specified using former rules and traggers, as we discuss in Section 24.0

5.3 UPDATE OPERATIONS AND DEALING WITH CONSTRAINT VIOLATIONS

The operations of the relational model can be categorized into opticmls and ideates. The relational algebra operations, which can be used to specify **retrievals**, are discussed in detail in Chapter 6. A relational algebra expression forms a new relation after applying a number of algebraic operators room existing set of relations its manness is for querying a disclose. The user form dates a query that specifies the data of interest, and a new relation is formed by applying relationsal operators to retrieve this data. That relation

^{11.} Store constructions are sometimes called story constructly, and transics or constructions are sometimes valved dynamics instruction

becomes the answer to the user's query. Chapter 6 also introduces the largeoge called relational calculus, which is used to declaratively denne the new relation without giving a queric index of operations.

in this section, we concentrate on the database modification in update operations. There are three basic update operations on relations insert, delete, and modify. Insert is used or insert a new reple or tuples in a relation. Delete is used to delete tuples, and Update (or Modify) is used to charge the values of some attributes in existing tuples. Whenever, these operations are applied, the turgency constraints specified on the relational database schema should not be violated. In this section we discuss the types of constraints that may be violated by each update operation and the types of actives that any cettaken if an update does cause a violation. We use the database shown in Figure 5.6 for examples and discuss only key constraints, entity integrity constraints, and the referential integrity constraints shown in Figure 5.7. For each type of update, we give some example operations and discuss any constraints that each operation and violate.

5.3.1 The Insert Operation

The **Insert** operation provides a list of attribute values for a new uple r that is to be inserted into a relation R_1 lister can violate any of the four types of constraints discussed in the previous section. Formain constraints can be violated if an attribute value is given that does not appear in the corresponding domain. New constraints can be violated if a key value in the tens tuple 1 already exists in another tuple in the relation r(R). Every integrity can be violated if the primary key of the new tuple 1 is null. Referential integrity can be violated if the value of any foreign key in the term to a tuple that does not exist in the referenced relation. Here are some examples to fluotrate this discussion

- Insert <"Circilia", "F", "Kolonsky", pull, "1960-04-25", "6357 Windy Lang, Katy, TX", F. 28000, pull, 4 > into Exposice.
 - This unservice violates the energy integraty constraint (full for the primary key sto), so it is rejected.
- Insert < Alicuit (1), (Zelaya), (999887777), (1902)/04-05), (0357 Windy Lane, Katy, TX1, F, 25000, (987054321), 42 (more)/ecovec
 - This insertion violates the key constraint because another tuple with the same ssection and so it is rejected.
- Insert <"Certha", "F", "Kolonsky", "677678989", "1960-04-05", "6357. Windswept, Kary, TX", F, 28000, "987654321", 74 units PPROVES.
 - This insertion violates the referential integrity constraint specifical on the hermise name as before rapid evides with non-266 = 7.
- Insert Cocciliat, 'F', 'Kolonsky', '677675989', '1960-24-05', '6357 Windy Lane, Katy, TX', E 2800', null. 4> into response.
 - This invertion satisfies all constraints, so it is acceptable.

If an insertion violates one or more constraints, the details option is to reject the insertion. In this case, it would be useful if the OBAts could explain to the user only the insertion was tejected. Another option is to attempt to enter the word for meeting the insertion, but this is typically not used for violations caused by Insertion 1 above, the DBAts could ask the user to provide a value for seven did be useful in option of a value for seven the word 1 above, the DBAts could ask the user to provide a value for seven did be explained in option of a value seven to add the user to provide a value for seven did be explained ask the user to the value of the to add the user to multiply of a could ask the user to the user to multiply of a could ask the user to insert a preservent ruple with more add value (or set in to multiply or a could ask the user to insert a preservent ruple with more add value (or set in to multiply or a could ask the user to insert a preservent ruple with more add value (or set in to multiply or a could ask the user to insert a preservent ruple with more add value (or set in to multiply or a could ask the user to insert a preservent ruple with more add value (or set in the latter case the insertion only after such as operation, was accepted. Notice that in the latter case the insertion violatioe cau case back to the second relation of the user attempts to insert a tiple for department 7 with a value for \$25556 that does not exist in the FNPLDEE relation.

5.3.2 The Delete Operation

The **Delete** operation can violate only referenced integrity, if the tuple being deleted is referenced by the foreign keys from other tuples in the database. To specify deletion, a condition on the attributes of the relation selects the tuple (or tuples) to be deleted. Here are some examples.

- Delete the works_painple with itsss = '999867777' and two = 10.
 - This deletion is acceptable
- Delete the EMPLOYES Tuple with SSN = '999887777'.
 - This deletion is not acceptable. Tecaese tuples in wosks_ow refer to this tuple. Hence of the tuple is deleted, referential integrity stolations will result.
- Delote the EVECORE tuple with SEC = 15354400001
 - This deletion will result in even worse referential integrity violations, because the tuple involved is referenced by tuples from the SNR, OFFR, DEPARTORNT, WORKS_ON, and DEPENDENT relations.

Several options are available if a deletion operation causes a violation. The first option is to reject the deletion. The second option is to arrited to assarb the propagate) the deletion by deleting ruples that reference the ruple that is being deleted. For example, in option 2, the DBMS could automatically delete the offending ruples from 1466, for with USSR = 19998877771. A third option is to modify the reference gambute values that cause the violation, each such value is either set to null or changed to reference another valuation, each such value and on the that causes a violation is root of the jaments (applied by reference another valuation) is root of the jaments (apple). Notice that if a referencing articlute that causes a violation is root of the jaments key, it cannot be set to mall, otherwise, it could violate entry integrity.

Combinations of these three options are also possible. For example, to avoid backing operation 3 cause a violation, the 1980s may automatically delete all tuples from Kesks ov and CLERNERT with ESSN = 13334455557. Tuples in EMPLOYCE with ECRESSN = 13354455557 and the tuple in CERNERT with KESSN = 13334455557 can have theat subjects and MIKESSN values changed to other valid values or to mill. Afthough it may make sense to delete

antomatically the varks, an and DEPENDENT tuples that refer to an ENFLOYFF tuple, it toay not make sense to delete other ENPLOYEE tuples or a DEFARINENT tuple.

In general, when a refinential integrity constraint is specified in the DDL the DBSD will allow the user to specify which of the options applies in case of a violation of the constraint. We discuss have to specify these options in the SQL 99 DDL in Chapter 8.

5.3.3 The Update Operation

The Update (or Modify) operation is used to change the values of one or more attributes in a tuple (or tuples) of some relation *R*. It is necessary to specify a condition on the attributes of tise relation to select the tuple (or tuples) to be modified. Here are some examples.

- Update the second of the enclosed tuple with six = '999887777' to 28000.
 - Acceptable.
- Update the use of the two and tuple with size = '999857777' to 1.
 - Acceptable
- Update the two of the tageover rapie with ssa # '999887777' to 7.
 - Unacceptable, because it violates referential integrity.
- Update the saved the topicyte tuple with sav = '999887777' to '987654321'.
 - Ongeoeptable, because it violates primary key and represential integrity constraints.

Updating on attribute that is notifier a primary key nor a foreign key issuelly consists to problems, the DBMS need only check to confirm that the new value is of the correct data type and domain. Modifying a primary key value is similar to deleting one uple and inserting another mats place, because we use the primary key to identify toples. Hence, the issues discussed corlier in both Sections 5.3.1 (Insert) and 5.3.2 (Delete) come into placific atomics her arribute is modified, the DBMS most moke sure that the new value refers to an existing tople in the referenced relation (or is null). Similar options exist to deal with inferential integrity violations caused by Update as those options discussed for the DMHs will allow the user to choose separate options to deal with a violation match by Delete and a violation caused by Update (see Section S.2).

5.4 SUMMARY

In this chapter we presented the modeling concepts, data structures, and constructions prociced to the relational model of data. We started by introducing the concepts of domains, attributes, and tuples. We then defined a relation schema as a fist of attributes that doesn'te the structure of a relation. A relation, or relation state, is a set of tuples that contorns to the schema. Several characteristics differentiate relations from ordinary tables or files. The first is that tuples in a relation are not ordered. The second involves the ordering of attributes in a relation scheme and the entrosponding executing of values within a tuple. We gave an alternative definition of relation that does not require these two orderings, but we continued to use the first definition, which requires attributes and tuple values to be induced, for convenience. We then discussed values in tuples and intriduced null values to represent mesting or induced information.

We then classified database constraints into raherent incide/shard constraints, schema-based constraints and application based constraints. We then discussed the schema constraints performing to the relational model, starting with domain constraints, then key constraints, including the concepts of superkey, candidate key, and primary key, and the NOT CONSTRAINT on attributes. We then defined relational database schemas. Additional relational constraints include the contry integrational database schemas. Additional relational constraints include the contry integrations which prohibits primary key attributes from being null. The interrelation referenced integral, constraint was then described, which is used to maintain consistency of references among taples from different relations.

The modification operations on the relational model are insert. Delote, and Update Each operation may crotate certain types of constraints. These operations were discussed in Section 5.3. Whenever an operation is applied, the distribuse state after the operation is escented must be checked to ensure that no constraints have been coolated.

Review Questions

- Define the following remiss domain, argibace, nonable relation schema, relation state, degree of a relation, relational durations schemes relational database state.
- 5.2 Why are tuples in a relation not ordered?
- 5.5 Why are duplicate tuples not oflowed in a relation?
- 5.4 What is the difference between 5 key and a superkey?
- 5.5. Why done designate one of the candidate keys of a relation to be the primary key?
- Discuss the characteristics of relations that make their different from erdinary tables and files.
- 5.7 Discuss the various reasons that lead to the occurrence of null values in relations.
- 5.8. Discuss the entity morphity and informatial integrity constraints. Why is each considered important?
- 5.9. Donne Jonege face. What is this concept used for?

Exercises

- 5.10. Suppose that each of the following update operations is applied directly to the database state shown in Figure 5.6. Docuss all integrity constraints violated by each operation, it any, and the different ways of entoteing these constraints.
 - a. Jusert S'Robert', F', 'Scott', '943775543', '1952-06-21', '2365 Newcastle R2, Bollaire, TX', M, 58000, '885005555', 12 into EVPlanEE.
 - E. Insert < Product A., 4, Bellane', 28 into succett.

- Insert «"Dyodia rion", 4, 1943775543", (1995-10-01"> into department.
- d Insert <'677678989', pull, '40.8'> into wasks_ow.
- c. Insert <14534534534537 [Julin], M, (1970) 12(12), [St OD-F12 mto DEPEndent).
- Delarg the many oscipples with -sys = (133445555).
- Delete the second topic with son = '987654321'.
- Delete the score topic with more = 'ProductX'.
- Modely the subsystand substantiant of the office state ruple with assault = 5 to 11234567891 and 11999-104011, respectively.
- Moduly the success armbore of the posterie tuple with size = "999667777" ro-"94 \$775543".
- k. Modify the nones articlute of the wave_os ruple with issu = '\$\$\$\$57777 and exc = 10 to 5.01
- 5.11 Consider the APPING relational database schema shown in Figure 5.5, which describes a database for airline flight information. Each ectial is identified by a flight variate, and consists of one or more Figure (just with rec_909688.1, 2, 3, and \$1000). Each leg has scheduled arrival and departure times and outpoets and has many rec_tostances-one for each Date on which the flight travely investigate are kept for each leg instance, SEAT_RESEPARTORS are kept, as are the ALEPLANE investigate interface for and the actual arrival and departure times and papers. An standard on the leg and the actual arrival and copartine times and papers. An standard on the leg and the actual arrival and copartine times and papers. An standard on the leg and the actual arrival and so is particular attended, the conjugate relates at severe to the standard and so is particular attended. The conjugate relates at severe to the standard and point attended to out a the standard and point of a particular attended to the standard and actual arrival and a point of a particular attended to the standard and and the standard attended to ender the standard attended to the standard attended to the standard attended to ender the standard attended to the standard attended to ender the standard attended to the standard attended to ender the standard attended to the standard attended to ender the standard
 - a. Give the operations for this update.
 - F. What types of constraints would with expect to check?
 - Which of these constraints are key a providegray, and referential integray constraints and which are not?
 - Breedy all the referent al integrity constraints that hold on the scheme shown in Figure 5.8.
- 3.13. Consider the relation (1888)(Course* Univ_Section», InstructorName, Senester, BoldangCode, Roson#, TimePeriod, Weekdows, CreditHours). This represents classes taught in a university, with unique Univ_Section#, Identify what you think should be various candidate keys, and write in your own words the constraints under which each candidate key would be valid.
- 513 Consider the following six relations for an order-processing database application in a company.

custoure(<u>Clust#</u>/Channe, Clitv1

oscer(Order=: Odate, Cust=: Ord. Arat1.

ORDER_TTENCORDER#, HEAD#, QUY!

(198<u>(hene</u>, Unit_price)

shtekent(Order#, Warehouse#, Ship_date)

```
RNACHAOSE (Warehouse#, Chty)
```

| AIBPORT AIRPORT_CODENAMECITYSTATE |
|--|
| |
| NUMBER _ AIRLINE _ WEEKDAYS |
| Pught Leg |
| FUCHT_NUMEERLEG_NUMBERDEPARTURE_ARPORT_CODESCHEDULED_DEPARTURE_TIME |
| |
| LEG_INSTANCE |
| FIGHT_NUMBER LEG_NUMBER _ DATE NUMBER_OF_AVAILABLE_SEATS AIRPLANE_ID |
| DEMATURE ANAPOPT_CODE DEPARTURE_TIME APRIVAL_ANAPORT_CODE ARRIVAL_TIME |
| FARES |
| FLIGHT, NUMBER FARE CODE AMOUNT RESTRICTIONS |
| |
| NITCANE I TE |
| |
| |
| |
| |
| AMPLANE_ID TOTAL_NUMBER_CF_SEATS AIRPLANE_TYPE |
| SEAT RESERVATION |
| FLIGHT, NLMBER LEG_NLMBER DATE SEAT_NUMBER CUSTOMER_NAME CUSTOMER_PHONE |

HOURE 5.8 The statuse relational database schema.

Here, Ord-Anit refers to total dollar amount of an order; Odate is the date the order was placed; Ship date is the date an order is shipped from the warehouse. Assume that an order can be shipped from several warehouses. Specify the foreign keys for this scheme, stating any assumptions you make.

 Consider the following relations for a database that keeps truck of business trips of salespersors in a sales office:

to espension(SSN: Name, Start_Year, Dopr_No)

| | REP(SSN, From Cu ₁ , To_Cir ₁ , Departure_Date Return_Date, <u>Trip_10</u>) |
|------|---|
| | evense(Top_ID, Account#, Amount) |
| 5 IS | Specify the foreign keys for this schema, statute any assumptions you make Consider the following relations for a database that keeps ttack of student entroll- ment in courses and the bucks adopted for each course: |
| | steeser(<u>SSN</u> , Name, Mijor, Bdate) |
| | conser(<u>Course</u> , Cname, Dept) |
| | (sear) (SSN, <u>Coursey</u> , <u>Quarter</u> , Grade) |
| | stor adoption(<u>Courses</u> , <u>Quarter</u> , Pook_fSBN) |
| | [7x1(Book_ISEN], Book_Tirle, Publisher, Auchar) |
| 516. | Specify the foreign keys for this schema, stating any assumptions you make. Consider the following relations for a dotabase that keeps track of auto-sales in a cardealership (Option refers to some optional soupment installed on an auto) |
| | um(<u>Seria), No.</u> Model, Mortufocturer, Proce) |
| | certons(<u>Serial-No</u> , <u>Cortion-Name</u> , Price) |
| | pares(<u>Salespersonal</u> , <u>Senal-Ne</u> , Date, Salespree) |
| | MESPERSON(<u>Subspersonad</u> , Name, Phone) |
| | Pirst, specify the foreign keys for this schema, staring any assumptions yee make. Next, gopulate the relations with a few example tuples, and then give an example of an in-settion in the setes and setesserson relations that tiokates the referential integrity constraints and of another insertion that sloes not. |

Selected Bibliography

The relational model was introduced by Cs22 (1976) in a clossic paper. Codd also introduced relational algebra and faid the theoretical foundations for the relational model in a sities of papers (Codd 1971, 1972, 1972a, 1974); by was later given the Toring award, the highest honor of the ACM, for his work on the relational model. It is later paper Codd (1979) discussed extending the relational model to incorporate more meta-data and semantics above the relations: he also proposed a three-valued logic to deal with uncertanty in relations and incorporating SULEs in the relational algebra. The resulting model is known as RMT. Childs (1968) had earlier used set theory to model databases. Later, Cixla (1992) published a book examining over 303 features of the relational data model and database systems.

Since Codd's preserving work, much rewarch has been conducted on various repects at the relational model. Todd (1976) describes an experimental rOMS called PRTV that directly implements the relational algebra operations. Schundt and Swenson (1975) introduces additional scenarios into the relational model by classifying different types of relations. Cheri's (1976) minipyrolationship model, which is discussed in Chapter 1, is a means to communicate the real-world semarates of a relational database at the conceptual level. Wederbold and Flinasti (1979) introduces carnots types of connections between relations to enhance us constraints. Extensions of the relational model are discussed in Chapter 24. Additional bibliographic noises for other aspects of the relational model and its languages, systems, extensions, and theory are given in Chapters 6 to 11–15, 16, 17, and 22 to 25.



The Relational Algebra and Relational Calculus

In this chapter we discuss the two formal languages for the relational model, the relational algebra and the relational nalcitlist. As we discussed in Chapter 2, a data model must include a set of operations to manipulate the database, in addition to the data and discussers for defining database structure and constraints. The basic set of operations for the relational algebra. These operations enable a user to specify lease for one or there relations. The factor which may have been formed from one or there relations. The algebra operations thus produce new relations, which can be further manipulated using operations of the same algebra. A sequence of relational algebra operations forms a relational algebra expression, whose result will also be a relation that represents the result of a database of the same algebra. When each operations to manipulated using operations of the same algebra. A sequence of relational algebra operations forms a relational algebra expression, whose result will also be a relation that represents the result of a database query (or remeval request).

The relational algebra is very important for several massive, first, it provides a found foundation for relational model operations. Second, and perhaps more important, it is used as a basis for implementing and optimizing queries in relational database monogement science (2020SO), as we does so in Part CV of the book. Third, some of its concepts are incorporated into the SQ, standard goory language for 80280-5.

Whereas the algebra derives a set of operations for the relational model, the relational calculus provides a higher-level declarative contation for specifying relational queties. A relational calculus expression, creates a new relation, which is specified in terms of variables that range over tows of the stored database relations (in tuple calculus) or over relations of the stored relations (an domain calculus). In a calculus expression, there is no order of operations to specify how to retrieve the query result—a calculus.

expression specifies only what internation the result should contour. This is the mondistinguishing learner between relational algebra and relational calculus. The relational calculus is important because at has a firm basis in mathematical logic and because the EQL (standard query language) for 2068(88 has some of its foundations in the tuple relational calculus).

The relational algebra is often considered to be an integral part of the relational data model, and its operations can be divided unit two groups. One group includes set operations from mathematical set theory; these are applicable because each relation is defined to be a set of tiples in the formal relational model. Set operations include UNION UNTERSECTION, SET DIFFERENCE, and CARDESIAN DECORDET. The other group consists of operations developed specifically for relational databases these include SSIF(1), (ROJECT, and (OD), among others. We first describe the SELECT and (ROJECT) operations in Section 6.1, because they are unany operations that operations single relations. Then we discuss set operations, in Section 6.2. In Section 6.3, we discuss (OD) and other complex binary operations, which operation two tables. The consist relational database shown in Figure 5.6 is used for our examples.

Some common database requests caneso be performed with the original relational algebra operations, so additional operations were created to express these requests. These include aggregate functions, which are operations that can same effect data from the tables, as well as additional types of JOIN and OSAIN operations. These operations were added to the original relational algebra because at their importance to many database applications, and are described in Section 6.4. We give examples of specifying queries that use relationed operations in Section 6.5. Some of these queries are used in subsequent chapters to illustrate various languages.

In Sections 6.6 and 6.7 we describe the other pion formal Linguage for relational calculus. There are two cariarons of relational calculus. There are two cariarons of relational calculus. The ruple relational calculus is described in Section 6.6, and the domain relational calculus is described in Section 6.7. Some of the SQL constructs discussed in Chapter 8 are based on the topic relational calculus. The relational calculus is a formal barguage, based on the topic relational calculus. The relational calculus is a formal barguage, based on the topic relational calculus. The relational calculus is a formal barguage, based on the branch of mathematical legic colled predicate calculus? In tuple relational calculus, visibles range over tuples, whereas in domain relational calculus, variables range over the domains (values) of attributes. In Appendix D we give an overview of the QBE (Query-By-dixample) language, which is a graphical user-truendly relational language based on domain relational calculus. Section 6.6 summarizes the chapter.

For the reader who is interested in a less detailed introduction to formal relational languages. Sections 6.4, 6.6, and 6.2 may be skipted.

SQL is based on hiple to attain data los, but also incorporates sione of the operatory from the clanenal algebra and its extensions as we shall see ap Chapters S and 9.

^{2.} In this chapter to funderity with instances pixels are calorins—search deals with quantitied canadia and values— is searched.

6.1 UNARY RELATIONAL OPERATIONS: SELECT AND PROJECT 6.1.1 The SELECT Operation

The SELECT operation is used to select a taken of the topics from a relation that satisfy a selection condition. One can consider the SELECT operation to be a *jden* that keeps only those uples that satisfy a qualifying condition. The SELECT operation can also be visualited to a *lower and poet/dam* of the relation into two sets of tuples -- these tuples that satisfy the condition and are selected, and these tuples that do not satisfy the condition and are selected, and these tuples that do not satisfy the condition and are selected, and these tuples that do not satisfy the condition and are selected, and these tuples that do not satisfy the condition and are selected, and these tuples that do not satisfy the condition and are selected, and these tuples that do not satisfy the condition and are selected, and these tuples that do not satisfy the condition and are selected, and these tuples that do not satisfy the condition and are selected, and these tuples that do not satisfy the condition and are down of the selected tuples whose department is 4, or those aloos of my is greater than \$30,000, we can individually specify each of these two conditions of the selected tuples whose department is 4.

```
ODNOE4(CMMIDNEL)
ISSALARY>30000(LMMLOVLE)
```

In general, the S90 K 1 operation is denoted by

of the second se

where the symbol \mathbf{r} (signual) is used to denote the SELECT operator, and the selection condition is a Boolean expression specified on the attributes of relation R. Notice that R is generally a relationant algebra expression whose result is a relation—the simplest such expression is just the name of a database relation. The relation resulting from the SELECT operation has the varie attributes as R.

The Biolean expression specified in Socketion Condition2 is made up of a number of clauses of the form

subminute comparison op> <constant values,

a,

amburen mež <comparison op> < umbure namež

where something names is the plane of an artificity of R. scomparison opens mutually creatified operators [=, <, ≤, >, >, +], and sconstant value? Is a constant value from the stribute domain. Chauses can be obtained worked by the Boolean operators AN(+, R) and S(-) is to show a general selection condition. For example, to select the toples for all articles where either work in department 4 and make over \$25,000 per year, or work in department 5 and make over \$30,000, we can specify the following SE Fitt operation:

C. Brief AND SALES CONTRACTORING AND SUPER PARTY (LAND ON SEC)

The result is shown in Freure 6.1a.

Notice that the comparison operators in the set $\{-, \le, \le, \ge, \ge, 2\}$ apply to attributes whose demains are objected easies, such as numeric of date domoins. Domains of strings of characters are considered ordered insed on the collaring sequence of the characters. If the domain of an articline is a set of anowled calles, then only the comparison operators in the set $[-, \pi]$ can be used. An example of in unordered domain is the domain Color **-** (red.

| (8) | FNAME | MINIT | LNAME | SSN | equate – | ADDRESS | S€X | SALARY | SUPERSSN | DNO. |
|-----|----------|-------|------------------------|-----------|------------|----------------------|-----|--------|------------|------|
| | Franse - | 1 | Wang | 353445565 | 1955-12-08 | 636 Woss Housing TX | ζ | 4,1060 | 588663666 | 8 |
| | Jenn ler | | YANACE | 987934331 | 1941-06-30 | 29' BenyElefake.TX | F | 43000 | 6890605555 | t |
| | Rowan | | fatue _n er: | 9668~1444 | 1962-09-15 | 975 FeeCes,House: TX | M | 38000 | 283445565 | 5 |

| Ы | LNAME | FNAME | SALARY | (ci | SEX | SALARY |
|---|-----------------|---------|--------|-----|---------|--------|
| | Smith | John | 3000 | | _N | _33000 |
| | WVg | frentin | 10000 | | ΓN | 49300 |
| | Zelaya | Aise | 25500 | | F | 25000 |
| | 434300 | Jennier | 43000_ | | F | 43000 |
| | 1918/1911 | Famegi | 36300 | | N | 59000 |
| | Lengkan | Joyus | 25006 | | <u></u> | 25000 |
| | Je <u>boo</u> r | Abread | 25000 | | N | 55000 |
| | R.vic | .lamea | 55000 | | | |

FIGURE 6.1 Results of SELECT and PROJECT operations call (Figure and Selects Section Characteristics) and selected and sel

Flue, green, white, yellow, . . .} where two order is specified among the various relates. State domains, allow, additional, types, of recupation, operators: for example, in domain, at character strings may allow the comparison operators of 05(0560_06.

In general, the result of a SELCT operation can be determined as follows. The selection conditions is applied independently to each raple i in \mathcal{B} . This is done by substanting each occurrence of an arribute A, in the selection condition with its value in the tuple (A_i). If the condition evaluates to TRUE, then raple i is selected. All the selected raples appear in the result of the SELECT operation. The Basican conditions AND, OR, and NOT have their normal interpretation, as follows:

- (cond1.580 cond2) is JEUE if both (cond1) and (cond2) are TEUE; otherwise if is FALSE
- (cond1/08/cond25/s/TRUE if other (cond1) or (cond2) or both are TRUE; other wise, it is IALSE.
- (NorT could) is TRUE if could is FALSE; otherwise, it is FALSE.

The SELECT operator is many; that is it is applied to a single relation. Moreover, the selection operator is applied to each tigle individually; hence, selection conditions cannot involve more than our tuple. The degree of the relation resulting from a SELECT operators, its miniber of attributes—as the sum as the degree of R. The number of attributes—in the sum as the degree of R. The number of attributes—in the sum as the degree of R. The number of apples there is one equal to the number of tuples in R. There, $\sigma_{\rm e}$ (Ref. ≤ 1.81 for any condition C. The fraction of tuples selected by a selection.

(a) (a) 22 (b) for any condition (c) the microin of suples selected by a selection condition is referred to as the selectivity of the condition.

Notice that the SE(EC) operation is commutative: that is,

 $\sigma_{\text{construct}}(\sigma_{\text{construct}}(R)) = \sigma_{\text{construct}}(\sigma_{\text{construct}}(R))$

Honor, a sequence of SELECON can be applied in any index. In addition, we can always configure a **caseade** of SELECC operations into a single SELECC operation with a conjunct ase ISNUU conductors that is:

 $\sigma_{\mathrm{sparses}}(\sigma_{\mathrm{sparses}}(\alpha_{\mathrm{sparses}}(\beta_{1}), \beta_{1}) = \sigma_{\mathrm{sparses}}(\beta_{1})(\beta_{1}) = \delta_{\mathrm{N}(\mathrm{sparses})}(\beta_{1})(\beta_{1}) = \delta_{\mathrm{N}(\mathrm{sparses})}(\beta_{1})(\beta_{1})(\beta_{1}) = \delta_{\mathrm{N}(\mathrm{sparses})}(\beta_{1})(\beta_{1})(\beta_{1})(\beta_{1}) = \delta_{\mathrm{N}(\mathrm{sparses})}(\beta_{1})$

6.1.2 The PROJECT Operation

If withink of a relation as a table, the SELD T operation selects some of the wais from the table while discarding other rows. The **PROJECT** operation, on the other band, selects control moving firsts the table and decards the other columns. If we are interested in only terrin attributes of a relation, we use the PROJECT operation to project the relation over task attributes of a relation, we use the PROJECT operation to project the relation over task attributes of a relation, we use the PROJECT operation can hence be visualized as a terrary partition of the relation into two relations: one has the meeded columns bittibutes) and contains the testified the operation, and the other contains the discarded columns. For example, to list each employee's last and last name and scarty, we can use the PROJECT operation as follows.

T. INT. THAT, M. AR (EVELONEE)

The resulting relation is shown in Figure 6.1(19). The general form of the PROJO T operatorus

 $\pi_{\rm contraction}(R)$

where π (put is the symbol used to represent the "KOFCT operation, and Cottribute back is the desired list of artributes from the artributes of relation R. Again, notice that R is, in general a relational algebra expression whose result is a relation, which in the simplest case is just the nome of a durabase relation. The result of the PSOFCT operation has only the autions specified in correlation (1st> in the spectral or they opposite in the latter, its degree is equal to the number of attributes in https://www.number.hem.com

If the attribute fist includes only nonkey attributes of R, duplicate tuples are likely to contr. The (ROJE) T operation proposes any signicate tuples, so the result of the (ROJE) operation is a set of tuples, and hence a cold volution.¹ This is known to **duplicate** elimination, for example, consider the following (POJE) (operation

THE SHORE (FAMILIER)

The result is shown in Figure 6 for Norsee that the tuple <E 25000 appears only once in Figure 6 for even though this combination of values appears to see in the PPP, 2500 set to a set of the PPP appears to be appeared by the PPP appear.

For number of tuples in a relation resulting from \times PROES T operation is always less during equal to the number of tuples in K. If the projection list is a superley of K—that

³ If dicheates are not charinated, the result would be a **molecet** so bap sit toples other than a set. Although this is not allowed in the formal relation models it is periodice, is practice. We shall see in this term 5 that say, allowes the user to specify whether dight areas should be allown and a sour

is, it includes some key of R-, the resulting relation has the same number of tuples as R. Moreover.

 $\pi_{choose}(\pi_{choose}(R)) = \pi_{choose}(R)$

as long as $\leq \text{list}^2 \geq \text{contains the armbures in <math>\leq \text{list} \geq \infty$ otherwise, the lett-hand ode is an incorrect expression. It is also noteworthy that commutativity does not hold on $(\mathbb{N}O) \in \mathbb{T}$.

6.1.3 Sequences of Operations and the RENAME Operation

The relations shown in Figure 6.1 do not have any names, in general, we may want to apply several relational algebra operations one after the other. Either we can write the operations as a single **relational algebra expression** by nesting the operations, in we can apply one operation of a time and create intermediate result relations. In the latter case we findst give matter to the felational hold that hold the intermediate results. For example, inreference the first name, last name, and salary of all employees whe work in department rundlet 5, we must apply a Soft.CT and a PR (10) operation. We can write a single relational algebra expression us follows.

 $\pi_{10,000}$ (non- non- $(\sigma_{0,01,0})$ (even (FF))

Figure 0.2a shows the result of this relational algebra expression. Alternatively, so can explicitly show the sequence of operations, giving a name rocasch intermediate relation:

СЕРБ -КНА₩СТ_{ИХО+}К(ТИРСКО++) REALT — (П. ₁₂04) - ₁₂07, 32 ие. (ОТЕБ ТИРА)

| (ar | FNAME | . r. Ala | BALARY T |
|-----|---------|----------|----------------|
| | 100 | 50.00 | 300] |
| | Dense | Eweng E | _ ¥mr] |
| | Panerd | Чазрат | 11600 |
| | A spect | analshi. | 25000 |

| John G Strict 1294/97219 (905/11-19) Zur Beugrein Hoyson Ta M 00000 N0042000 S Annels 1 Viorig 2034/95555 (265/11-01) C12 virte Linuxan,1X M 10000 mmm22200 S Namest 4 Namest 555555 (265/11-01) 274 Nice Linuxan,1X M 10000 mmm22200 S | (b) TENA | Pri Alvic | KINT | LYPINE | <u>55N</u> | BCATE | ADDRESS | _ 9€× | 34LARY | SUPERSON | CHC |
|---|----------|-----------|------|----------|------------|------------|--------------------------|-------------|----------------|----------|-----|
| Anne al 1 Viong 2004/0005 (2001) C12 vint buccon_11X VI (0000) mmm22222 1 Namest 4 Fuence 565555444 1961/06-10 211/11-10 1X 1 30000 2002/20224 1 | | പറ | - 0 | Smith 1 | 129494209 | 1907-0-19 | Zan Routten Hoyston Tall | | 32 <u>22</u> - | 30342307 | : |
| Norment 4 Fuerger 565554444 1967-56-775 Print Like Humber 13, 51 Jobbil (ICE4-55555 5 | | Acre e | - | Yiong | 200445805 | 1055 17-01 | CVP was I puezo, "X | <u>[</u>]u | Maca | 00002866 | |
| | | Homeon | + | Sec. 1 | 585554444 | 1961-3415 | Photos DAX Humber 1X | - u ' | 10101 | 100-5554 | 5 |
| Januar 4 Engen 456453453 1992 A 21 5551 (ku Maashar (4 € 2000) 7774-2005 3 | | Jø-ca | 6 | Engera - | 450459459 | 1912 (A.21 | 5651 (K.) Hossion 14 | ÷ | 25000 | 777-2765 | 5 |

| F. | FIRSTNAVE | LASTNAME | SO ARY |
|----|-----------|-----------|--------|
| | 1.000 | <u>s</u> | 2000 |
| | 5 meto | 1000 | - 4000 |
| | Renewo | Network 1 | 37,000 |
| | L K SH | tieden _ | 20000 |

FIGURE 6.2. Results of a sequence of operations, fait π_{excell} (see second σ_{excell}) (second to being intermediate relations and repairing of attributes.

it is often simpler to break down a complex sequence of operations by specifying intermediate result relations than to write a single relational algebra expression. We can also us this technique to rename the attributes in the intermediate and revits inflations. This can be useful in connection with more complex operations such as t NGN and ION, is to shell see. To remain the attributes in a relation, we simply first the new attribute action is parenthesed, as in the following example:

```
TRAPH (F<sub>UMB</sub> - (UMAI CYFF)
R(I - KURANG - UNDINGNIS - NALANA) ← TR<sub>RANG</sub> - <sub>Man</sub>g - <sub>Malana</sub> (UMA)
```

These two operations are dissurated in Figure 6.2b.

If no renoming is applied, the names of the attributes in the resulting relation of a SETO Dependion are the same as those in the original relation and in the same order. For a SEARCH operation with no renomine, the resulting relation has the same attribute names as those in the projection list and in the same order in which they appear in the list.

We can also define a formal RENAME operation—which can rename either the efation name of the attribute names, of both—in a mannet similar to the way we defined SHOT and PROPERT. The general RENAME operation when applied to a relation R of lagor 6 is denoted by any of the following three forms.

 $\rho_{\alpha\beta}|_{\mathcal{B}_{1}} = \sum_{k \in \mathcal{A}_{1}} (\mathcal{R})$ or $\rho_{\alpha}(k)$ or $\rho_{\alpha\beta}|_{\mathcal{B}_{2}} = \sum_{k \in \mathcal{A}_{2}} (\mathcal{R})$

where the symbol p (tho) is used to denote the RENAME operator, S is the new relation name, and B_1, B_2, \ldots, B_n are the new attribute manos. The first expression remaines both the relation and its attributes, the second remaines the relation only, and the third sciames the articlous only. If the attributes of R are (A_1, A_2, \ldots, A_n) in that order, then each A_n is renarmed as B_n .

6.2 RELATIONAL ALGEBRA OPERATIONS FROM SET THEORY

6.2.1 The UNION, INTERSECTION, and MINUS Operations

The rest group of relational algebra operations are the standard mathematical operations on sets. For example, to nettiove the social security numbers of all employees who either work in department if or directly supervise an employee who works in department 5, we can use the UNION operation as follows:

```
DEFS_CHPS: \mathcal{A}_{360^{-1}}^{\bullet} (LARLOWED)

RESULT 1\leftarrow \pi_{155} (DEPS_SHAD)

RESULT 2 (SSM) \leftarrow \pi_{50^{+}(855^{+})} (DEPS_SHAD)

RESULT \leftarrow RESULT 1 \bigcup RESULT 2
```

The relation RESULTE has the social security numbers of all employees who work in department 5, whereas RESULTE has the social security mumbers of all employees who

directly supervise an employee who works in department 5. The UNION operation purduces the ruples that are in other a surface of e6.0 if or both (see Figure 0.3). Thus, the SSN value 333445555 appears only since in the result.

Soveral set theoretic operations are used to increase the elements of two sets in various ways, including UNION, INTERSECTION, and SET DIFFERENCE (also called MINUS). These are binary operations; that is, each is opplied to two sets (of tuples). When these operations are adapted to relational databases, the two relations on which any of these three operations are applied must have the same type of tuples; this condition has been called union compatibility. Two relations $R(A_1, A_2, \ldots, A_n)$ and $S(B_1, B_2, \ldots, B_n)$ are said to be union compatibile if they have the same degree *n* and if dom(A_1) = dom(B_1) for $1 \le i \le n$. This means that the two relations have the same number of attributes, and each varies point of attributes has the same domain.

We can define the three operations UNION, INTERSECTION, and SET DETERBINGE on two unconscompatible relations R and Y as follows:

- union. The result of this operation, denoted by R U S, is a relation that includes all tuples that are either in R or in S is in both R and S. Duplicate tuples are eliminated.
- intersection: The result of this operation, denoted by R ∩ S, is a relation that includes all tuple- that are in both R and S.
- set difference (or MINUS): The result of this operation, denoted by K = S, is a relation that includes all tuples that are in R but not in S.

We will adopt the convention that the insulting relation has the same attribute names as the give relation *R*. It is sloways possible to remark the attributes methe result using the remark operator.

Figure 6.4 illustrates that there operations. The relations of CONNE and INSTRUCTOR in Figure 6.4 are union compatible, and their roples represent the names of andents and instructors, respectively. The result of the UNION operation in Figure 6.45 shows the names of all students and instructors. Note that duplicate toples appear only once in the result. The result of the INTERSECTION operation (Figure 6.4c) includes only those who are both students and instructors.

Notice that both UNIGN and INTERSECTION are comparative operations, that is,

ROVESOR and ROSESOR





:

| | | | · | . – | | | | |
|-----|----------|---------|--------|-----|------|---------|---------|---------|
| ונו | STUDENT | FN | LN | | INST | RUCTOR | ENAME | LNAME |
| | | Susan | Yec | | | - | John | Smath |
| | | Ramesh | Steh | - | | | Ricardo | Growne |
| | | Johnny | Kohler | i | | | Susan | Yao |
| | | Banara | Jones | _ | | | Francis | Janoson |
| | | Arm | Ford | - | | | Ranash | Slian |
| | | Jammy | Aanc | | | | | |
| | | Fritad | Galbad | 1 | | | | |
| | | | | | | | | |
| | | | | | . – | | | |
| ß | j FN İ | έN | | (2) | Ľ | EN | LN | 1 |
| | Şusan | Yec | | | | Sasan | Yao | |
| | Ramesh | Shah | | | L | Remesh | Shah | J |
| | латлау 🛛 | Kohlet | | | | | | |
| | Bernara | Jares - | | | | | | |
| | Amy | Ford | | | | | | |
| | յ հայառ, | Wang | | | | | | |
| | Emasi | Giben | | | | | | |
| | Jelvi | Smt0 | | | | | | |
| | Aucardio | Brane | | | | | | |
| | Francis | Johnson | | | | | | |
| | | | | | | | | |
| | | | | | | | | |
| 10) | FN | LN | ! | (9 | ı | FNAME | LNAME | |
| | Jutney | Kohler | | | | Jaho | ொர | |
| | Barbara | Jones | | | _ | Reards | Bitwie | |
| | Amy | Ford | | | • | Francis | Johnson | |
| | Jumy | Wang | | | _ | | | |
| | Erresi | Groed | | | | | | |

#GURE 6.4 The set operations UNADM, INTERSECTION, and AINON (A) Two unmocompatible relations, thi success U association is taken O associate (d) source -145 RULER, (C) INSTRUCTOR - STUDENT,

Pode UNRON and INTERSECTION can be instead as many operations applicable to any number of relations because both are associative spectralons, that is,

 $\mathbb{R} \cup (S \cup T) = (\mathbb{R} \cup S) \cup T$ and $(\mathbb{R} \cap S) \cap T = \mathbb{R} \cap (S \cap T)$

The MINUS operation is tex conocidative, that is, in general,

5≠8. R. - 8

Figure 0.4d shows the manax of students who are not distinctors, and legare 6.4e shows the names of instructors who are not students.

6.2.2 The CARTESIAN PRODUCT (or CROSS PRODUCT) Operation

Next we discuss the **CARTESIAN PRODUCT** operation—also known as **CROSS PROPACT** or **CROSS JOIN**—which is denoted by ×. This is also a binary set operation, but the relations in which it is applied do not have to be much compatible. This operation is used to combine ouples from two relations in a combinatorial fashing. In general, the result of $R(A_1, A_2, \ldots, A_n) \times S(B_1, B_1, \ldots, B_n)$ is a relation (2 with degree $n \neq n$ attributes (2) A_1 , A_2, \ldots, A_n , B_1 , B_2, \ldots, B_n) is a relation (2 with degree $n \neq n$ attributes (2) A_1 , A_2, \ldots, A_n , B_1 , B_2, \ldots, B_n), in that order. The resulting relation (2 has one topic for each combination of ruples—crost from B and one from S. Hence, if B has n_g ruples, (denoted as $-R_n \neq n_g$), and S has n_g tuples, then $R \propto S$ will have $n_g \neq n_g$ (as ruples.)

The operation opplied by itself is generally meaningless. It is useful when followed by a selection that murches values of armborns coming from the component relations. For example, suppose that we want to retrieve a list of names of each female employees dependents. We can do this as follows.

$$\begin{split} &\mathsf{F}(\mathsf{MALE}_\mathsf{PPS} \leftarrow (\mathcal{T}_{\mathsf{N},\mathsf{S},\mathsf{R},\mathsf{F}}) \in \mathsf{EUP}(\mathsf{MCE}) \\ &\mathsf{EUP}(\mathsf{NALE}_\mathsf{PPS} \leftarrow \mathcal{T}_{\mathsf{NAME},\mathsf{T},\mathsf{NAME}}) \in \mathsf{EUP}(\mathsf{SPENDENTS} \leftarrow \mathsf{EUP}(\mathsf{SPENDENTS})) \\ &\mathsf{EUP}(\mathsf{OSPENDENTS} \leftarrow \mathsf{EUP}(\mathsf{ADENTS} \leftarrow \mathcal{T}_{\mathsf{NAME}}) \in \mathsf{DEPENDENTS}) \\ &\mathsf{A}_\mathsf{T}(\mathsf{DAL}_\mathsf{SEPENDENTS} \leftarrow \mathcal{T}_{\mathsf{NAME},\mathsf{T},\mathsf{NAME}}) \in \mathsf{ACT}(\mathsf{A}_\mathsf{SEPENDENTS})) \\ &\mathsf{RSSULT} \leftarrow \Pi_\mathsf{FNCME}(\mathsf{T}_\mathsf{ADME},\mathsf{T}_\mathsf{NAME}) \subset \mathsf{MPNDE}(\mathsf{ACT}(\mathsf{A}_\mathsf{SE}) \mathsf{SEPENDENTS})) \end{split}$$

The resulting relations from this sequence of operations are shown in Figure 0.5. The SNS_STPESDERTS relation is the result of upplying the CARTERIAN PROPE CT operation to transmits from Figure 6.5 with DEPTROENT from Figure 5.6. In CNP_GEREDOUS, every tuple from SNSARES is combined with every tuple from DEPERDENT, giving a result that is not very meaningful. We want to combine a female employee tuple only with her particular dependence, manch, the REPERDENT tuples whose USEs values march the SSS value of the SNS relation is a good example of the case when relational a fector of the correctly applied in yield results that make no sense at all, it is therefore the responsibility of the agent to make sure to apply only meaningful operations to relations.

The CARTESIAN PRODUCT creates toples with the combined attributes of two relations. We can then SELSCO only related tuples from the two relations by specifying an oppropriate selection condition, as we did in the precedime example. Because this sequence of CARTESIAN PRODUCT followed by SELECT is used gatter commonly to identify and select related tuples from two relations, a special operation, called (ON), was created to specify this sequence as a single operation. We discuss the IOEN operation next

6.3 BINARY RELATIONAL OPERATIONS: JOIN AND DIVISION

6.3.1 The JOIN Operation

The JOIN operation, denoted by the is used to combine related topics from two relations into single tuples. This experiation is very unportant for any relational database with more

| FORMALE_ ENPS | FNAME | MINIT | . LN444 | GSN | BCATE | ADDHESS | GFX | SALARY | SUPPOSEN | CNC |
|------------------|----------|----------|----------------|---|------------|---------------------|-----|--------|-----------|-----|
| | 349 | _ر | 7 -6 -7 | 200837777 | 19(447-19 | 3321 Qarda Spong TX | - = | 2502 | 987654021 | |
| | - ANNO - | <u> </u> | WARAND | ::::::::::::::::::::::::::::::::::::::: | 1047-06-00 | 251 BOWERANG D. | , F | 4300. | 566527976 | . L |
| | Joyas | | Engez- | 453-60458 | 10/2/07 31 | Sold Rec Houston TX | [F | 2500c | 389-47496 | 5 |

| EVPNAHES | PNAVE | LN/4ME | SSN 1 |
|----------|----------|----------|------------|
| | 600 | Z1407 | 244692 |
| | Jan Olar | V-allace | \$663435° |
| | Jonar | Engra | 153/965452 |

| EMP_DEPENDENTS | ENAME | INM/E | - 589 T | FSSN | CEPENDENT NAME | SD 🗌 | BTATE | ++- |
|----------------|---|---------------|-----------------|-------------|----------------|----------|-------------|----------------------|
| | A.R. M | Zarya | 923687777 | 300440350 | Anc | ; ; | 1997;424-55 | 1 |
| | -Age to | 7440 | 999947777 | 12975655 | Песатеч | 'w ' | 1983 17-37 | |
| | Aligne | Снира | 9990077 / | 15,647.05 | te | ł | 1956-05-03 | |
| | A 44 | 24.343 | <u>)9900777</u> | 357661321 | Abner | . n | 1940 0 25 | 1 |
| | A.646 | 200240 | 37.68/77 | 12,496780 | Merael . | W. | 1902(01-24 | • • • |
| | | Adays. | 4796877777 | 125690212 | Auro | ' י ר | 1968 12:30 | |
| | Álcia - | Zeleva | 995687777 | 123456750 | Exzabelly | F | 1967 1:5 65 | |
| | in the second | wata le | 28755-221 | 20.94-0655 | - Anc | - F | BIGHER | 1 |
| | Jerosles | 6768.A.A. | \$5755-021 | 00045505 | Treasing | ' M [| INCH 325 | |
| | | Wallete | 30757-621 | 336077 | .+ | · - I | (30e 0)47 | |
| | Je Ver | Weilage | 99376301 | 99/654121 | . from | <u>м</u> | 1942-02,28 | · · · · |
| | Je na fe n | VARA | 967664321 | 12 956760 | No. at | 7 | 1966-01-04 | , • • • |
| | Jenner | Wallson | 10705-001 | 125456749 | As a | ' - | INTE 12:00 | |
| | .8-10-51 | White | ₩7554331 | 12,456789 | E 236-4* | F | 1367 05-07 | |
| | .kyue | Bayah | 263260853 | 72 9440 955 | A-00 | ۶ | 1080 04 05 | •••• |
| | .)-2// ⁻ Н | Signality | 4634639655 | 303445555 | Paccore | - 'm 1 | 1.85 (1.65 | ····· |
| | - 10 M | cispel. | ALMARK T | Tayuu (6) ? | .tey | ' ' ' | 194-56-00 | |
| | .koyte | Fright | 101812603 | 19/06/4021 | 40004 | ι M | 1942-02-28 | · · • • · |
| | ,юу-эн | Bigeti | 263459255 | 12455759 | 14. Yel | 74 | 1:58 01-04 | ••• |
| | 5769 | Signal | 45 MS/051 | 123150729 | Akte | ΓF j | 28512.0 | · · · · ; |
| | . Kyce | togeli | - Samais | 12,9458785 | Date | <u> </u> | 12010-00 | |
| | | | | · | | | | |

| PEGUT | FNAME | LNAME | OEPENDEN/ NAME |
|-------|--------|--------|----------------|
| | an nên | Waters | 600g |

FIGURE 6.5 The CARDESIAN PRODUCT (CROSS PRODUCT) operation.

than a single relation, because it allows as to process relationships attracte relations. To illustrate JOBN, suppose that we want to retrieve the name of the manager of each department. To get the manager's name, we need to coarsine each department tuple with the employee tuple whole site value matches the KESSS value in the department tuple. We do this by using the JOIN operation, and then projecting the result over the necessary primbutes, as follows:

The first operation is illustrated in Figure 6.6. Note that seessy is a foreign key and that the referential integery constraint plays a role in having matching tuples in the referenced relation ESPLOYEE.

The VIN operation can be stated in terms of a CAUTOSIAN PRODUCT followed by a SULECT operation. However, JOIN is corporation because it is used very frequently when specifying database queries. Consider the example we gave earlier to illustrate CARTESIAN PRODUCT, which included the following sequence of operations.

```
Смещо из или иту к — меналик № роедно кт
астолощиеранисти — Формараку (influenceptation)
```

These two operations can be replaced with a single JOIN operation as follows:

ACTUAL_DEPENDENTS - EMPHANES OFFENDENT

The general form of a fOIN operation on two relations⁴ $\mathcal{R}(A_1, A_2, \dots, A_n)$ and $\mathcal{S}(\theta_1, \theta_2, \dots, \theta_n)$ is

KS

The result of the EON is a relation Q with n = m attributes $Q(A_1, A_2, \dots, A_n, B_n B_2, \dots, B_n)$ in that order; Q has one tople for each combination of tiples—one from R and one from δ —minutes: the combination eacifies the join condition. This is the main difference between CORTESION 120.2007 and JOIN. In JOIN, only combinations of tiples are tricked in the result, whereas in the CARTESION PROPERTIA the result, whereas in the CARTESION PROPERTIA the result. The join condition is specified on attributes from the two relations R and S and is evaluated for each combination of tiples. Each tople combination for which the join condition is included in the result. The join condition is specified on attributes from the two relations R and S and is evaluated for each combination of tiples. Each tople combination for which the join condition coalities to TRUE is included in the result.

A general join condition is of the form

scondition > AND < condition > AND < DND < condition >

| DEPT_MGR | DNAME | DNUMBER | MGRSSN | ••• | FNAME | MINIT | UNANE | SSN | |
|----------|----------------|----------|-----------|-----|-----------|-------|----------|------------|-------|
| | Research | <u> </u> | 367445585 | | Equality | Т | Vhary | 3-3445555 | 1 - 5 |
| | Astroneutricon | | 087634321 | | Jacob Pri | 5 | We let a | \$P70560P1 | |
| | Headquarters | · I | 585665565 | | bine | F | Brog | 05068-565 | |

FIGURE 6.6 Result of the IOIN operation at γ were department γ_{maximum} endower.

4. Again, notice that B and S on the any relations that result from general relational algebra expressions

6.3.2 The EQUIJOIN and NATURAL JOIN Variations of JOIN

The most common use of ION involves join conditions with equality comparisons only Such a LON, where the only comparison operator used is =, is called an EQUIJOIN. Both examples we have considered were EQUIDONS. Notice that in the result of an EQUIJOIN, Both examples we have considered were EQUIDONS. Notice that in the result of an EQUIJOIN, we always have one or more pairs of attributes that have identical values in every raple. For evenple, in Figure 0.6, the values of the attributes were were and some are identical in every ruple of 0007, we because of the equality just condition specified on these two articlastics. Because one of each pair of attributes with identical values is superfluxes, a new operation called NATORAL JOIN - denoted by = - was created to get rid of the second (superfluxes) arming in an EQUISEN condition. The standard definition of NATORAL JOIN requires the the two join attributes (or each pair of join artributes) have the same name in both relations. If this is not the case, a renaming operation is applied first.

In the following example, we first rename the counses arytibute of Sevenius policies, so its in has the same name as the page articlote in ecolor — and then apply NACCRALJOAS.

The same query can be done to two steps by creating an intermediate table course follows:

```
LEPT ← () <sub>DHAPELOW</sub>PLPCPLSSLWOPS WATHIN</sub> (DEPGATPENT)
NEOL_CEPT ← IRECECT * DEPT
```

The introduce two is called the join **attribute.** The resulting relation is illustrated in Figure 6.7., both 6801–6601 relation, each tuple conditions a 980/601 ingle with the **DEPARTMENT** tuple for the lagrangent that controls the project, but only one join attribute is kept.

If the introduces on which the mountal point is specified already have the varie names in terb educates, renaming is unnecessary, hor example, to apply a natural point on the OWABER arributes of its WEINEST and DEPT I CONTIONS, it is sufficient to write.

CEPT I KN & DIPAKTHENT " CEPT LOCATIONS

The resulting relation is shown in Figure 6.7b, which combines each department with its locarons and has one tuple for each location, on general, NATERNATION is performed by optating afternibite gains that have the same name in the two relations. There can be a list of juminthoses from each relation, and each corresponding pair must have the same name.

5.837.36.0 Olyris buneady in EQ. (1018) followed by ready diof the superfluois turnbures.

162 Chapter 6 The Relational Algebra and Relational Calculus

| ω Γ | PROJ DEPT | PNAME | | UMBER | PLOCATION | ONUM | DNAME | WGRSSN | MORSTARTDATE | |
|-----|-----------|--------------------|------------|-------|------------------|------------|----------------|-------------|--------------|---|
| - | | Hon.co. | <u> </u> | ı | Seriene | Ē 5 | Persoaucn | 333445555 | 1988-06.22 | |
| | | Product? | — | 2 | Sectors | 匚, 一 | PRIMA CO | 0.00445656 | 1988-36372 | |
| | | Protect? | | 3 | Handizi | <u>_;_</u> | Terment Co | 339415656 | 1988-0662 | _ |
| | | Children Provale 4 | <u> </u> | -0 | Styliga.) | 4 | Ashimishaken | 9-56-4341 | 1945-014-1 | 4 |
| | | _ Copingang-Inzo | _ <u>.</u> | 20 | H0097071 | ' | rised-posters | 84865516 | 1981-30-19 | |
| | | Newbonelts | | 30 | 912 Ho rd | . 1 | Administration | 99.45543.01 | 1995-01-01 | |

| 15+ | CEPT LOCS CNAME | ONUMBER | MGASSN | MGASTARTOATE | LOCATION |
|-----|--------------------|---------|-------------|---------------------|------------|
| | Here, by so that a | 1 | cited#botto | 1.411-06-19 | Hassian |
| | Advantation (1925) | | >872-121 | <u>1885-010 (01</u> | ()totoot |
| | Heseworth | | 303445255 | 1980-03-22 | Liebane |
| | Hesearch | 5 | 303245555 | 1988-06-28 | Sugartance |
| | R <u>search</u> | · 1 | 333445355 | 1998-06-22 | Houston |

FIGURE 6.7 Kessulls of two NATURAL IONS operations, (a) and juter superstation? CEPT (b) CEPT_DOD s CEPACIENT ? CEPT_LOGATIONS.

A more general har constendary definition for NATURALION is

 $Q \leftarrow R^{*}_{(r,referred equals)}$

In this case, which \geq specifies a list of cattributes from R, and whist2× specifies a list of cattributes from S. The lists are used to form equality comparison conditions between paraol corresponding staributes, and the conditions are then ASDed together. Only the here corresponding to autobases of the test relation R – which \geq —is kept in the result Q.

Notice that it no combination of tiples satisfies the join condition, the result of a jobs is an empty relation with zero raples. In general, if R has n_k tiples and S has a tuples, the resolution a KaN operation $R_{\rm excess marked}$ is will have between zero and $n_k \ge n_k$ implies. The expected size of the join result doubled by the maximum size $n_k \ge n_k$ leads to a ratio called join selectivity, which is a property of each join condition. If there is no join condition, all combinations of toples spatially and the KuN degeneration into a CAR (FS158) (RODC) T. also called CROSS (RODC) T or CROSS (CN).

As we can see, the 'GN operation's used to combate data from multiple telations are also that related information can be presented in a single table. These operations are also known as inner joins, to dottinguish them from a different variation of join called ever jobs (see Section 64.3). Note that sometimes a join may be specified to tween a relation and itself, as we shall illustrate its Section 6.4.2. The NATURAL ROLE of EQUICIN operation can also be specified among multiple tables, leading to an array pair. For example, consider the following three-way join:

[[FRUCECT + CALVERINGER LEWISING]] - MERSON SON END. GREET

This links each project to its controlling department, and then relates the department to us manager couplinger. The net result is a consolidated relation in which each topic contions this projects department manager information.
6.3.3 A Complete Set of Relational Algebra Operations

It has seen shown that the set of relational algebra operations $\{\sigma_{i}, \sigma_{i}, \bigcup_{i} = i\}$ is a complete set that is, any of the other original relational algebra operations can be expressed as inspecte of operations from this set. For example, the INTERSECT ON operation can be expressed by using UNION and MINOS as follows:

 $R \cap S = (R \cup S) + ((R - S)) \cup (S - R))$

Although, structly speaking, INTERNOTTION is not required, it is inconvenient to sweat this complex expression every time we wasn'to specify an intersection. As onother complet, a JCIN operation, can be specified as a CARTENIAN TRODUCT followed by a \$500°C operation, as we discussed:

 $S = \sum_{\alpha \in A} (h \times S) = \sigma_{\alpha \in A} (h \times S)$

Similarly, a NATURAL JON can be specified as a CARTESIAN (RODUCT preceded by REVAUE and followed by FELECT and PROFECT operations. Hence, the Various ION operations are also not variable necessary for the expression power of the relational ideebra. However, they are unportant to consider as separate operations because they are canonically uptied in database applications. Other operations have been included in the relational algebra for convenience rather than accessing. We discuss one of these—the TVENON operation — in the next section.

6.3.4 The DIVISION Operation

The DMISION operation, denoted by \pm , is useful for a special kind of query that sometimes occurs in database upp teations. An example is "Refrieve the names of employees who work on all the projects that 'John 'Smith' works end.' To express this query using the DUISION operation, proceed as follows: First, retrieve the list of project combets that 'John Smith' works on an the intermediate relation SMITH 2009.

```
\begin{aligned} (\mathsf{WETM} &\leftarrow \{ \overline{\mathbf{T}}_{1,\mathsf{N},\mathsf{M},\mathsf{R}_{1}}, \gamma_{1,\mathsf{M}^{*}}, \gamma_{\mathsf{M},\mathsf{R}_{1}}, \gamma_{\mathsf
```

Next, create a relation that includes a tuple KEND, ESSED whenever the coupleyer whose social scenary number is 2659 works on the project whose number is 660 in the project whose number is 660 in the promodulte relation (see 966).

 $_{551}$ MOS $\leftarrow \pi_{1550}$ PbD (works 10b)

Jurally, apply the DIVISION operation to the two relations, which gives the desired analoyees used become numbers.

```
2555(250) ( 200⊾0935 ≐ 50110_0965
(6311) ( Ⅲ66556 (15061(1500) ) 20010711)
```

The previous operations are shown in Figure 6 8a

SSN_PNOS

la) –

| Τ | CSSN | PNO |
|-----|------------|----------|
| Ŀ | 123456789 | <u> </u> |
| - T | 123456789 | <u> </u> |
| Ŀ | 200004444 | 3 |
| Ŀ | 453453453 | 1 |
| | 153453458 | 2 |
| Ŀ | 333445555 | . ° |
| | 333445555 | <u> </u> |
| Ŀ | 333445555 | 10 |
| 1: | 3334466555 | _ 20 |
| 13 | 299887777 | ່ວບ |
| | 999667777 | 10 |
| | £17967987 | 10 |
| | 387987987 | 30 |
| | 197664321 | 30 |
| 1 | 907664321 | 20 |
| Ŀ | 489385555 | 20 |
| | | |

| SMITH_PNOS | PNO |
|------------|-----|
| | 1 |
| | \$ |

| SSN5 | SSN | | |
|------|-----|-----------|--|
| | | 123456789 | |
| | | 453453453 | |



FIGURE 6.8 The DEVISION operation, (a) Dividing sequence by semigravity the $l \leftarrow k \neq S$.

In central, the DIVISION operation is applied to two relations $R(Z) \doteq S(X)$, where $X \subseteq Z$. Let $Y = Z \subseteq X$ (and hence $Z = X \cup Y$), that is, let Y be the set of attributes of *R* that we not attributes of *S*. The result of DIVISION is a relation T(Y) that includes a tuple fit tuples t_{Y} appear in *R* with $t_{X}[Y] = t$, and with $t_{Y}[X] = t$. For every tuple t_{Y} in *S*. This means that, for a tuple *t* to appear in the result *T* of the DIVISION, the values in *t* must appear in *R* is combination, with every tuple in *S*. Note that in the tornulation of the DIVISION operation, the tuples in the denominator relation, restrict the numerator relation by selecting their tuples in the result rhat must be all values present in the domination. It is not receively to be a present in the dominator. It is not receively to know what these values are.

Figure 5.8b illustrates a CONENDS operation where $X = \{A\}, Y = \{B\}, \text{ and } Z = \{A, B\}$. Notice that the tuples (values) b_1 and b_1 appear in R in combination with all three tuples in λ , that is why they appear in the resulting relation T. All other values of B in R do not appear with all the tuples in S and are not selected; b_2 does not appear with a_2 , and b_1 does not appear with a_1 .

The DEEION operation can be expressed as a sequence of π , x, and π operations as follows:

$$T_{i} \leftarrow \pi T(R)$$

$$T_{i} \leftarrow \pi T((S + T_{i}) = R)$$

$$T_{i} \leftarrow T_{i} = T_{i}$$

The 14V.2008 operation is defined for convenience for dealing with queries that involve "introtesal quantification" (see Section 6.6.6) or the all conditions Most RD6818 inplementations with SQL as the primary query language do not directly introlement decision. At las a foundabout way of dealing with the type of query illustrated above (see Section 8.5.4). Table 6.1 lists the vacuus basic relational algebra operations we have discussed.

6.4 ADDITIONAL RELATIONAL OPERATIONS

Same common database requests include are needed an commercial query languages for EPAISE cannot be performed with the original relational algebra operations described in Sections 6.1 changeb 6.3. In this section we define additional operations to express these trajects. These operations enhance the expressive power of the original relational algebra.

6.4.1 Aggregate Functions and Grouping

The first type of request that cannot be expressed in the bood relational algebra is to specify mathematical aggregate functions of collections of values from the database. Exomples of such functions include retrieving the average or total sitiary of all employees or the total mathematical englowee tuples. These functions are used in simple stoustical queries that summarize information from the database tubles. Common functions applied is collections of management values include SUM, as ERAGE, MAXISUM, and MIN MUM. The CONT function is used for counting tuples or values.

LABLE 6.1 OPERATIONS OF RELATIONAL ALGEBRA

| Operation | Purpose | Notation |
|---------------|---|---|
| SILLE: T | Selects all tuples that satisfy the selection condition- from a relation & | σ.,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,, |
| 9840 LUCI | Produces a new relation with only some of the attributes of \$, and removes dopheate ruples. | $\pi_{(n_1,n_2) \in \{1,2\}}(R)$ |
| der Aljún N | Produces all combinations of tuples from R_1 and R_2 that satisfy the poin condition. | $R_1^{(2)}$, α_1 , α_2 , β_2 |
| BOULION | Produces all the combinations of tuples from R ₂ and R ₂ that satisfy a join condition with only equality compar- terms | $\begin{array}{c} R_1 \overset{\Theta}{=} & _{(1,1)} \otimes_{(0,1)} \otimes_{(0,1)} R_1, \phi \\ R_1 \overset{\Theta}{=} (1, \dots, 1, 1, 0, \dots, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1,$ |
| NATE RA, JON | Same as EQUION except that the join autifutes of R_2 are not included an the resulting relation; if the join attributes have the same names, they do not have to be specified at all. | $\begin{array}{cccc} R_1 & & & \\ R_1 & & & \\ R_1 & C_{n+1} & & & \\ C_{n+1} & & & \\ C_{n+1} & & & \\ C_{n+1} & & \\ \end{array}$ |
| UNAN | Produces a velation that includes all the topics in R_1 or R_2 is [with R_1 and R_2 ; R_1 and R_2 must be union comparable. | <i>κ</i> ₁∪ <i>κ</i> ₂ |
| INTERSECTION. | Py-ances a relation that includes all the toples in both R_1 and R_2 , R_1 and R_2 must be written compatible. | $s_i \cap s_j$ |
| DIFFERENCE | Produces a relation that includes all the tables in R_1 that are not in R_2 ; R_1 and R_2 must be union compatible. | $R_1 = R_1$ |
| CARTENAN | Produces a relation that has the attributes of R ₁ and R ₂ | $R_1 \times R_2$ |
| nada e ch | and includes as imples all possible combinations of implest from R_1 and R_2 . | |
| DIVESTON | Produces a relation $R(X)$ that includes all tuples $d[X]$ in $R_1(Z)$ that appear in R_1 in combination with every tuple from $R_1(Y)$, where $Z = X \cup Y$. | $R_1(Z) \div R_2(Y)$ |

Another common type of request involves econput; the tuples in a relation by the value of some of their attributes and then applying an appreptic function independently to each group. An example would be to group employee tuples by 1880, so that each group includes the tuples for employees working in the same department. We can then first each DSO value along with, site, the overtipe salary of employees working to the department, or the number of employees who work in the department.

We can define an AGBRERATE FUNCTION operation, using the sembal \widetilde{X} (propounded "script F")," to specify these types of inquests as follows:

6. There is no surger agreed up an notation let specifying again whether their the solutions consistent wings All proved. where signifying artributes a is a hyporfact barries of the relation specified in *R*, and sfuncnon-lists is a list of (sfunctions south these) pairs. In each such pair, sfunctions is one of the allowed functions—such as SUM, AVERAGE, MAXIMUM, SUSTAUGH, COUNT, and south response attribute of the relation specified by *R*. The resoluting relation has the grouping attributes plus one attribute for each element in the function fist. For example, a termere each department number, the number of employees in the departments and then average salary, while remaining the resulting attributes as indicated below, we write:

The result of this operation is shown in Figure 6.95.

In the above example, we specified a list of attribute names—between parentheses in the RES AME operation—for the resulting relation $R_{\rm e}$ if no remaining is applied, then the attributes of the resulting relation that correspond to two function list will each be the orienteration of the function name with the attribute name in the form «function» (attribute».⁷ For example, Figure 6.2b shows the result of the following operation:

DKO IN CONTRACT AVERAGES (ALMAY CONTRACT)

I no grouping attributes are specified, the functions are applied to all the topks in the relation, so the resulting relation has a single topke only. For example, Equip 6.9c shows the result of the following operation:

| e. | B 040 | NO_OF_EMP | LOYEES | AVERAGE_ | SAL |
|----|-------------------|-----------|---------|----------|-----|
| | 5 | - 4 | | 33290 | |
| | . 4 | 3 | | 31000 | |
| | · · · | 1 | | 55000 | |
| | | | | | |
| Ŀı | _ DNG] 00 | UNT SSN | AVERAGE | SALARY | |
| | E. | 4 | | 1250 | |

| 5 | 4 | 33250 | |
|---|---|-------|---|
| 4 | 3 | 31050 | _ |
| , | 1 | 55000 | _ |
| | | | |

| 17 | COUNT \$\$N | | AVERAGE_\$ALARY | | | |
|----|-------------|---|-----------------|---|--|--|
| | | • | | _ | | |
| | 8 | 1 | 35125 | | | |

NGURE 6.9 The ADVICATE CONCTION operation for $\rho_{attain where expression equals a state of the transmission and the expression of the property of the state of the expression of the express$

7 Sole that this is in arbitrary isoration we are assysting. There is no standard notation

It is important to note that, in general, duplicates are not choistaned when an aggregate function is applied; this way, the normal interpretation or functions such as SUM and AVERAGE is computed." It is worth emphasizing that the testil of applying an aggregate function is a relation not a scalar number reven if it has a single value. This makes the relational algebra a closed system.

6.4.2 Recursive Closure Operations

Another type of optition that, in general, rannot be specified in the basic original relational algebra is recursive clusure. This operation is applied to a recursive relationship between toples of the same type, such as the relationship between an employee and a supervisor. This relationship is described by the foreign key supervisor of the two orter relationship is described by the foreign key supervisor of the two orter relationship is described by the foreign key supervisor of the two orter relationship is described by the foreign key supervisor of the two orter relationship is described by the foreign key supervisor of the two orter relationship is described by the foreign key supervisor of the two orter relationship is described by the foreign key supervisor of the two operation is to remove all supervisors of an employee e at all levels lefter is, all employees of directly supervisor by e, all employees e^{t} directly supervised by each employee e^{t} , and so an

Although at is straightforward in the relational algebra to specify all employees supercised by e at a specify first first difficult to specify all supervises or all levels, for example, to specify the sesse of all employees of directly supervised—at level me—by the employee c whose name is "Junes Borg" (see Figure 5.6), we can apply the following operation:

To retrieve all employees supervised by Dorg at level 2—that is, all employees $z^{\prime\prime}$ supervised by some employee $z^{\prime\prime}$ who is directly supervised by Borg–, we can apply another 1098 to the result of the first query as follows:

 $\operatorname{Result}2(\operatorname{SSN}) \leftarrow \pi_{\operatorname{SSN}}(\operatorname{SUPTRVESEDM} = \operatorname{SUPVERM}$ result1)

To get both sets of employees supervised at levels 1 and 2 by 'James Borg,' we can apply the UNION operation to the two results, as follows:

ocsult ← REBULT2 ∪ Result1

The results of these queries are illustrated in Figure 6.12. Although it is possible to retrieve employees at each level and then take their 0.9900, we cannot, in general, specify a query such as "retrieve the supervises of 'James Borg' at all levels' without utilizing a looping mechanism." An eventuon valled the consider closure of relations has been proposed to compare the recursive relationship as the recursion proceeds.

^{6.} In SQL, the option of eliminatine diplicates before applying the aggregate function is available by analoding the keyword DiscHNCT (see Section 8.5.4).

^{9.} The RQL3 standard includes syntax for recursive classifier

| | (SSN) | (SUPERSSN) |
|-------------|------------|------------|
| SUPERVISION | SSN1 | SSNR2 |
| | 23456701 | 333446655 |
| | 3334445556 | 888877655 |
| | 909887777 | 997654321 |
| | 967654321 | 83855555 |
| | 555884444 | 333445555 |
| | 453463453 | 3334455555 |
| | 967937967 | \$87654321 |
| | 688665555 | സി |

| (30)0's SSN < 58666 | 5555) |
|---------------------|-------|
|---------------------|-------|

| BESULT | SSN | | RESULT 2 | 55N | RESULT | |
|----------------------|------------|--|-------------------|-------------------|--------|--------------|
| | | | | | | |
| | 3334455861 | | | 13348.67.89 | | 123456789 |
| | 987654321 | | | 999667777 | | 999887777 |
| · | | | | 656884444 | | 6668884444 |
| (Supervised by Borg) | | | | 463453463 | | 4534534873 |
| | | | | 987967387 | | 5673879R7 |
| ١٤ | | | | | | 3334458855 1 |
| | | | Supervised by Bot | gia subordinates) | | 087654321 |

IRESULT1 (J REPUT2)

SIGURE 6.10 A two-level recursive query.

6.4.3 OUTER JOIN Operations

We real docuss some extensions to the JON operation that are necessary to specify centain types of queries. The JON operations described earlier match tuples that satisfy the join condition. For example, for a NATURAL (SON operation K * S, only tuples from K that have nonching ruples in S—and vice viewa—appear in the result. Hence, tuples without a matching (or related) tuple are charmated from the JON result. Tuples with call values in the join orthibures are also charmated. This amounts to loss of information of the result of JON is apposed to be used to generate a report based on all the information in the component of tuples.

A set of operations, colled **outer joins**, can be used when we want to keep all the tapes in *R*, or all those in *S*, or all those in both relations its the result of the jobs, regardless of whether or nor they have note hing tuples in the other relation. This satisfies the need of queries in which toples from two rables are to be combined by matching corresponding rows. For without losing any tuples for lack of matching values. The join aperaticus we described earlier in Section 6.3, where only matching ruples are kept in the result, are called inner joins.

For example, suppose that we want a list of all employee names and also the name of the departments they manage if diay hipper to namage a department; if they do not manage any, we can so indicate with a null value. We can apply an operation LEFT OUTER JOIN, denoted by (), to retrieve the result as follows:

```
\begin{array}{rcl} \text{IEM} & \leftarrow & (\text{FMPLOF} + \mathbb{P}_{(VV) \rightarrow \mathcal{K}(VVM)} \rightarrow \text{EFARIALERI}) \\ \text{KEVL}^* & \leftarrow & \overline{\mathbf{h}}_{(VAM)} \rightarrow \text{Emain} \rightarrow \text{Web}_{(main)} \rightarrow \text{KEV}(\text{TEMP}) \end{array}
```

The LEFT of TER JOIN operation keeps every tuple in the first or left, relation R in R (5) if no triatelying tuple is found in S, then the attributes of S in the join result are filled or "yaddecP with null values. The result of these operations is shown in Figure 5.11.

A similar operation, BIGHT OLTER JODN, denoted by -1, keeps every tuple in the second, or right, relation S in the tesult of R + -S. A third operation, FULL OUTER JODS, denoted by -1 keeps all tuples in both the left and the right relations when no matching tuples are found, pudding them with null values as needed. The three outer pan operations are part of the SQU2 standard (see Chapter 8).

6.4.4 The OUTER UNION Operation

The OUTER UNION operation was developed to take the union of tople- from two relations if the relations are to contracted provide. This operation will take the UNION of tuples in two relations R(X, Y) and S(X, Z) that are partially compatible, meaning that solutions of their attributes, say X, are union compatible. The attributes that are anion compatible are represented only once in the result, and those attributes that are not anion compatible from either relation are not anion compatible from either relation are also kept in the result relation T(X, Y, Z).

Two teples t_1 in *R* and t_2 in *S* are suid to match at $t_1[X] * t_2[X]$, and are considered to represent the same entity or relationship instance. These will be combined (unioned) into a single inple in *T*. Toples in either relation that have no matching tiple in the other relation are puckled with null values. For example, on OUTLR UNION can be applied to two relations where schemes are studied (Name, SSN, Department, Advisor) and instruction(Name, SSN, Department, Rank). Tuples from the two relations are matched based on having the same

| RESULT | FNAME | MINIT | LNAME | ONAME |
|--------|-----------|-------|---------|----------------|
| | .kohn | Ĥ | Smith | ritual |
| | Arankin ' | I | Woog | Hesearch |
| | Alca | | Zelaya | nul |
| | Jennier | 5 | Wallace | Administration |
| | Hamesh | ĸ | Narayan | nut |
| | Joyce | Α | English | nul |
| | Ahmad | V | Jabbar | <u>rul</u> |
| | James | Ε | 60mg | Headquarters |

HOURE 6.11 The result of a 121 OL TER JOIN operation.

combination of volues of the sleared attributes. Name, SSN, Department, The result relation, student_st_at_intime tot, will have the following attributes:

sticest_ck_tristouctor(Nume: SSN, Department: Advisor, Rank)

Ail the topics from both relations are included in the result, but topics with the same (Name, SSN, Department) combination will appear only once in the result. Topics approxing only in \$1.6 st will have a null for the Rank attribute, whereas ruples approxing toly in \$5.6 will have a null for the Advisor attribute. A topic that exists in both relations, such as a student who is also an upsupport, will have values for all is attribute. ¹⁶

Notice that the same person may still appear twice in the result. For example, we could have a gradiente student in the Mathematics department who is an instructor in the Gospater Science department. Although the two tuples representing that person in Science and as noted will not be matched. This is because Topartment has two separate meanings in states) the department where the person studest and instructor), it we wanted to union provide department where the person is employed as an instructor), it we wanted to union provide based on the same (Name, SSN) combination only, we should renove the Department attribute in each table to reflect that they have different meanings, and Lignate them is not being part of the union-compatible attributes.

Another capability that exists in most commercial languages (but not in the basic ielational algebra) is that of specifying operations on values after they are extracted from the database. For example, arithmetic operations such as $+_{1} = 1$ and + can be applied to numeric values that appear in the result of a query.

6.5 EXAMPLES OF QUERIES IN RELATIONAL ALGEBRA

We now give additional examples to illustrate the use of the relational algebra operations. All examples refer to the database of Figure 5.6. In general, the same query can be stated in numerous ways using the various operations. We will state each query in one way and leaving to the reader to commup with equivalent formulations.

QUERY 1

Remote the name and address of all employees choosark for the 'Research' department

```
\begin{split} & \mathrm{Er}_{\mathrm{OMP}} = \mathrm{Op}_{\mathrm{OMP}} = - \left( T_{\mathrm{OMP}}_{\mathrm{OMP}} + \mathrm{Rescale}_{\mathrm{OMP}} + \left( \mathrm{Op}_{\mathrm{OMP}} + \mathrm{P}_{\mathrm{OMP}} \right) \right) \\ & \mathrm{Er}_{\mathrm{OMP}} = - \left( \mathrm{Er}_{\mathrm{OMP}} + \mathrm{Op}_{\mathrm{OMP}} + \mathrm{Op}_{\mathrm{OMP}} + \mathrm{Op}_{\mathrm{OMP}} \right) \\ & \mathrm{e}_{\mathrm{OMP}} = - \mathrm{Op}_{\mathrm{OMP}} + \mathrm{Op}_{\mathrm{OMP}} + \mathrm{Op}_{\mathrm{OMP}} + \mathrm{Op}_{\mathrm{OMP}} + \mathrm{Op}_{\mathrm{OMP}} + \mathrm{Op}_{\mathrm{OMP}} + \mathrm{Op}_{\mathrm{OMP}} \right) \\ & \mathrm{e}_{\mathrm{OMP}} = - \mathrm{Op}_{\mathrm{OMP}} + \mathrm{Op}_{\mathrm{
```

10. Notice that (2), HER OR(2)STIF equivalent to a SOLL COTOR (COS) If the foir attributes are all the compare attributes of the two relations This query could be specified in other ways: for example, the order of the JOIN and SPLECT operations could be reversed, or the JOIN could be replaced by a NATURALION after renaming one of the join attributes.

QUERY 2

For every project located in 'Stofford', lost the project number, the controlling department number, and the department nanoger's fast name, address, and birth date

QUERY 3

Find the names of employees who work on all the projects controlled by department much a 5

```
\begin{split} & \mathsf{CEATS}_{PROTS}(PROT) \ \longleftrightarrow \ \pi_{\mathsf{SMPE}^{\mathsf{P}}}(\mathsf{CF}_{\mathsf{CR},\mathsf{V}^{\mathsf{S}}}(\mathsf{PROTS}(\mathsf{PROTS})) \\ & \mathsf{SW}_{PROT}(\mathsf{SSM}, \mathsf{PMC}) \ \Longleftrightarrow \ \pi_{\mathsf{SSM}^{\mathsf{S}}}(\mathsf{MOEES}_{\mathsf{SM}}) \\ & \mathsf{SESULT}_{\mathsf{F}}(\mathsf{POSSMS}) \ \longleftrightarrow \ \mathsf{EH}^{\mathsf{S}}(\mathsf{ROS}) \ \stackrel{\leq}{\sim} \ \mathsf{DFATS}_{\mathsf{PROTS}} \\ & \mathsf{SESULT} \ \hookleftarrow \ \pi_{\mathsf{SMM}^{\mathsf{S}}}(\mathsf{DFAM}^{\mathsf{S}}(\mathsf{PSM}^{\mathsf{S}},\mathsf{SMM}^{\mathsf{S}})) \\ & \mathsf{SESULT} \ \hookleftarrow \ \pi_{\mathsf{SMM}^{\mathsf{S}}}(\mathsf{DFAM}^{\mathsf{S}}(\mathsf{PSM}^{\mathsf{S}},\mathsf{SMM}^{\mathsf{S}})) \\ & \mathsf{SMM}^{\mathsf{S}}(\mathsf{DFAM}^{\mathsf{S}}(\mathsf{SMM}^{\mathsf{S}})) \\ & \mathsf{SMM}^{\mathsf{S}}(\mathsf{DFAM}^{\mathsf{S}}) \\ & \mathsf{SMM}^{\mathsf{S}}) \\ & \mathsf{SMM}^{\mathsf{S}}(\mathsf{DFAM}^{\mathsf{S}}) \\ & \mathsf{SMM}^{\mathsf{S}}(\mathsf{DFAM}^{\mathsf{S}}) \\ & \mathsf{SMM}^{\mathsf{S}}) \\ & \mathsf{SMM}^{\mathsf{S}} (\mathsf{DFAM}^{\mathsf{S}}) \\ & \mathsf{SMM}^{\mathsf{S}} (\mathsf{DFAM}^{\mathsf{S}}) \\ & \mathsf{SMM}^{\mathsf{S}}) \\ & \mathsf{SMM}^{\mathsf{S}} (\mathsf{DFAM}^{\mathsf{S}}) \\ & \mathsf{SMM}
```

QUERY 4

Make a list of project numbers for projects that involve an employee whose last nome is "Smith's either as a worker or as a manager of the department that controls the project

```
\begin{aligned} & \operatorname{Symplets}(\operatorname{Cost}) \leftarrow \pi_{\operatorname{Sym}}(\sigma_{\operatorname{Integer}}, \operatorname{Cost}(\operatorname{Symplets})) \\ & \operatorname{Symplets}(\operatorname{Cost}(\operatorname{Cost})) \\ & \operatorname{Symplets}(\operatorname{Cost}(\operatorname{Cost}), \operatorname{Cost}(\operatorname{Cost})) \\ & \operatorname{Symplets}(\operatorname{Cost}(\operatorname{Cost}), \operatorname{Cost}(\operatorname{Cost}), \operatorname{Cost}(\operatorname{Cost})) \\ & \operatorname{Symplets}(\operatorname{Cost}(\operatorname{Cost}), \operatorname{Cost}(\operatorname{Cost})) \\ & \operatorname{Symplets}(\operatorname{Cost}(\operatorname{Cost}), \operatorname{Cost}(\operatorname{Cost})) \\ & \operatorname{Symplets}(\operatorname{Cost}(\operatorname{Cost}), \operatorname{Cost}(\operatorname{Cost}), \operatorname{Cost}(\operatorname{Cost})) \\ & \operatorname{Symplets}(\operatorname{Cost}(\operatorname{Cost}), \operatorname{Cost}(\operatorname{Cost}), \operatorname{Cost}(\operatorname{Cost})) \\ & \operatorname{Symplets}(\operatorname{Cost}(\operatorname{Cost}), \operatorname{Cost}) \\ & \operatorname{Symplets}(\operatorname{Cost}(\operatorname{Cost}), \operatorname{Cost}) \\ & \operatorname{Symplets}(\operatorname{Cost}(\operatorname{Cost}), \operatorname{Cost}) \\ & \operatorname{Symplets}(\operatorname{Cost}(\operatorname{Cost}), \operatorname{Cost}(\operatorname{Cost}), \operatorname{Cost}) \\ & \operatorname{Symplets}(\operatorname{Cost}(\operatorname{Cost}), \operatorname{Cost}(\operatorname{Cost}), \operatorname{Cost}) \\ & \operatorname{Symplets}(\operatorname{Cost}(\operatorname{Cost}), \operatorname{Cost}(\operatorname{Cost}), \operatorname{Cost}(\operatorname{Cost}), \operatorname{Cost}(\operatorname{Cost})) \\ \\ & \operatorname{Symplets}(\operatorname{Cost}(\operatorname{Cost}), \operatorname{Cost}(\operatorname{Cost}), \operatorname{Cost}(\operatorname{Cost}), \operatorname{Cost}()) \\ & \operatorname{Cost}(\operatorname{Cost}(\operatorname{Cost}), \operatorname{Cost}) \\ \\ & \operatorname{Cost}(\operatorname{Cost}(\operatorname{Cost}), \operatorname{Cost}) \\ \\ & \operatorname{Cost}(\operatorname{Cost}(\operatorname{Cost}), \operatorname{Cost}(\operatorname{Cost}), \operatorname{Cost}()) \\ \\ & \operatorname{Cost}(\operatorname{Cost}(\operatorname{Cost}), \operatorname{Cost}) \\ \\ & \operatorname{Cost}(\operatorname{Cost}(\operatorname{Cost
```

QUERY S

List the names of all employees with two or more dependencial

Strictly speaking, this query consol be done in the fear (original) relational algebra. We have to use the AGGREGATER NOTION operation with the COUNT eggregate function. We assume that dependents of the work employee have distinct operating wave values.

```
\begin{aligned} & \Pi(\mathsf{SSW}, \mathsf{WO}_\mathsf{CPEPTS}) \leftarrow_{\mathsf{LSSW}} \widetilde{\mathcal{N}}_{\mathsf{COMP}} \underset{\mathsf{CPPTNDH}^{\mathsf{CPTNDH}^{\mathsf{C}}}{\mathsf{HAVE}} (\mathsf{DPPTNDENT}) \\ & \mathcal{T}_2 \leftarrow & \sigma_{\mathsf{WC}} \underset{\mathsf{CPPN}^{\mathsf{CPT}}}{\mathsf{CPPT}} (\mathcal{T}_1) \\ & \mathsf{RF}(\mathsf{LLT} \leftarrow & \pi_{\mathsf{LMVM}} \underset{\mathsf{CPLAW}}{\mathsf{CPLAW}} (\mathcal{T}_2 \land \mathsf{FMPT}) \\ \end{aligned}
```

QUERY 6

Retrieve the numes of employees who have no dependents.

This is an example of the type of query that uses the MINES (Sci. 10E2ERENGE) operation.

```
 \begin{array}{l} \textbf{x} := \mathsf{SVPS} \longleftrightarrow \ \boldsymbol{\pi}_{\mathsf{AVP}}(\mathsf{SMPLOYED}) \\ \mathsf{SVPS} \mid \textbf{x} := \mathsf{SDPP}(\mathsf{SOP}) \mid \textbf{x} = \ \boldsymbol{\pi}_{\mathsf{SSPP}}(\mathsf{RCPUMPLND}) \\ \mathsf{SVPS} \mid \textbf{x} := \mathsf{COP}(\mathsf{SOP}) \mid \textbf{x} = \ \boldsymbol{\pi}_{\mathsf{SSPP}}(\mathsf{RCPUMPLND}) \\ \mathsf{SVPS} \mid \textbf{x} := \mathsf{COP}(\mathsf{SOP}) \mid \mathsf{COPS} \mid \mathsf{SOP}(\mathsf{SOP}) \\ \mathsf{R(SOP} \mid \mathsf{SOP}) \mid \mathsf{COP}(\mathsf{SOP}) \mid \mathsf{COPS} \mid \mathsf{SOP}(\mathsf{SOP}) \\ \mathsf{R(SOP} \mid \mathsf{SOP}) \mid \mathsf{SOP}(\mathsf{SOP}) \\ \mathsf{R(SOP}) \mid \mathsf{SOP}(\mathsf{SOP}) \mid \mathsf{SOP}(\mathsf{SOP}) \\ \mathsf{R(SOP}) \mid \mathsf{SOP}(\mathsf{SOP}) \\ ```

#### QUERY 7

list the names of managers who have at least one dependent

As we uncutained earlier, the same query can in general be specified in many different way. For example, the operations can often be applied in various orders. In addition some operations can be used to replace others for example, the iNTERSECTION operation in Query 7 can be replaced by a NATOR 31 JON. As an exercise, try to do each of the those example queries using different operations.<sup>1</sup> In Chapter 8 and in Sections 6.6 and 6 (low shore how these queries are setteed in other relational languages.

# 6.6 THE TUPLE RELATIONAL CALCULUS

In the and the next section, we introduce another termal query language for the relational model called **relational calculus**, **is relational calculus**, we write one **declarative** expression to specify a refreeval request, and hence there is no description of how to evaluse a query. A colculus expression specifies what is to be refreeved rather than foat to refreevent. Therefore, the relational calculus is considered to be a **nonprocedural** languing. This differs from relational algebra, where we must write a separation of opening is to specify a refreeval request; hence, it can be considered as a **procedural** way of storing a correct it is possible to nest algebra operations to form a single expression however, a certain order among the operations is always explicitly specified in a relational algebra expression. This order also inducnees the strategy for evaluating the query. A calculus expression may be written its different ways, but the way of is written has no bearing on how a query should be evaluated.

When queries are optimized (see Chapter 13), the system will choose a particular sequence of optimizers that corresponds to an execution strategy that can be executed efficiently.

It has been shown that any retrieval that can be specified in the basic relational algebra can also be specified in relational calculus, and vice versa; in other words, the **expressive power** of the two languages is identical. This led to the definition or the concept of a relationally complete language. A relational query language *l*, is canadered **relationally complete** if we can express in L any query that can be expressed in relational calculus. Relational completences has because an important basis for comparing the expressive power of high-level query languages. However, as we saw in Section 6.4, certain frequently required queries in database applications cannot be expressed in basic relational algebra or calculus. Most relational algebra or relationally complete basic basic power of high-level queries in database applications cannot be expressed in basic relational algebra or calculus. Most relational algebra or relationality complete basic power that relational algebra or relationality complete basic basic power of an advantable applications cannot be expressed in basic relational algebra or calculus. Most relational algebra or relationality complete basic power of high-level power that relational algebra or relationality complete basic power of the basic power of the basic relational algebra or calculus. Most relational algebra or relationality complete basic power of the basic power that relational algebra or relationality completes and algebra or relational database applications cannot be expressed in basic relational algebra or relationality completes and algebra or relational calculus because algebra or relational calculus because algobra or relational operatives such as agrees functions, grouping, and ordering.

In this section and the new, all our examples again refer to the database shown in Figures 5.6 and 5.7. We will use the some queues that were used in Section 6.5. Sections 6.6.5 and 6.0.6 discuss dealing with universal quantitiers and may be skipped by students interested in a general introduction to tuple calculos.

## 6.6.1 Tuple Variables and Range Relations

The tuple relational calculus as based on specifying a number of tuple variables. Each tuple variable usually bases over a particulor database relation, meaning that the variable may take as its value any individual tuple from that relation. A simple tuple relational calculus query is of the form.

#### I/ LCOND(2)

where  $\epsilon$  is a tuple variable and  $\epsilon \gtrsim 8.2(c)$  we condutional expression involving t. The result of such a query is the set of all ruples t that satisfy C(38.010). For example, to find all employees whose solves is above \$50,200, we can write the following tuple calculas expression:

#### $\{t \in \mathsf{FAP}(\mathsf{cose}(t)| \mathsf{and}(t| \mathsf{sample}(s)))\}$

The condition SPECYEE(1) specifies that the range relation of tuple variable i is EPPLOVES. Each eventives impled that satisfies the condition assumption 50000 will be removed. Notice that USARAN references attribute SPEAR of tuple variable 1; this notation recentlies how artribute numes are condition with relation names or aliases in SQL, as we shall see in Unapper 8. In the notation of Chapter 5, USARAN is the same as writing (SPEAR)

The above query permoves all attribute values for each selected permore uploir. In remove only some of the amphores is say, the first and last names — we write

(CAMPLE FLAME | LOPLOY LEFT AND ESOLARY SCOOL)

Informally, we need to specify the following information in a tuple calculus expression

 For each tuple variable to the range relation K of a This value is specified by a condition of the form K(2)

- A condition to select particular combinations of ruples. As tuple variables lange over their respective range relations, the condition is evaluated for every possible combination of suples to identify the selected combinations for which the condition evaluates to TRUE.
- A set of attributes to be retrieved, the requested attributes. The values of these attributes are retrieved for each selected combination of toples.

Before we discuss the formal syntax of tuple relational calculus, consider another query.

#### QUERY 0

Refreee the birth date and address of the employee (or employees) whose teams is John B. Snoth'.

QO\_[1]RATE\_LACCRESS\_\_\_\_ENRIFOLT[r]\_AND\_LENABE="]ohit" AND\_LENAT="B\_AND LISOF="Smith"]

In tuple relational takulas, we first specify the responsed attributes takan and tausus for each selected tuple t. Then we specify the condition for selecting a tuple telekong the bot (1)—nonicly, that the prople of the optimum relation whose court, ensur, and court attribute values are 'John', 'B's and 'Statch', respectively.

## 6.6.2 Expressions and Formulas in Tuple Relational Calculus

A general expression of the tuple relational calculus is of the form

 $\{i, A_{\mu}\}, A_{\mu}, \dots, i, A_{\mu} \in COND(i_1, i_2, \dots, i_{\mu}, i_{\mu}, j_{\mu}, j_{\mu}, \dots, j_{\mu, \mu})\}$ 

where  $t_1, t_2, \ldots, t_m, t_{m+1}, \ldots, t_{m}$  are tuple variables, each  $A_i$  is an attribute of the relation or which c ranges, and vOND is a condition or formula<sup>17</sup> of the tuple relational calcidus. A femula is made up of predicate calcidus atoms, which can be one of the following:

- 1. An atom of the form R(t), where R is a relation name and  $t_i$  is a tuple variable. This atom identities the range of the tuple variable  $t_i$  as the relation whose name is  $R_i$ .
- 2 An atom of the form q<sub>0</sub>A op q<sub>0</sub>B, where op is one of the comparison operators in the set 1<sup>+</sup>, ≤, ≥, ≥, ≥, ≥, ≠}, q and q are tuple variables. A is an attribute of the relation on which q tanges, and B is an attribute of the relation on which q ranges.
- 3. An assumption of the form q A option z op q B, where options on of the comparison operators in the set  $1=z < z_1 > z_2 > z_3$ ,  $Z_1$  and that support unlables. A is an attribute of the relation on which through a structure of the relation on which through a structure of the relation on which through a structure of the relation on which the ranges.

Each of the preceding atoms evaluates to either TRUE or FALSE for a specific combination of ruples, this is called the truth value of an atom. In general, a tuple variable (ranges over all possible tuples "in the universe." For atoms of the turn R(1), it is assigned to a tuple that is a member of the specified relation R, the atom is TRUE (theraise, it is FAUSE. In atoms of types 2 and 3, if the tuple variables are assigned to tuples such that the values of the specified attributes of the tuples satisfy the condition, then the upon is TRUE.

A formula (condition) is inside up of one or more atoms ronneered via the logical operators AND, OR, and NOT and is defined recursively as follows:

- 1. Every arom is a formula
- If P<sub>1</sub> and P<sub>2</sub> are formulas, then so are (F<sub>1</sub> AND F<sub>2</sub>), (F<sub>2</sub> OR F<sub>2</sub>), NOT(F<sub>1</sub>), and NOT (F<sub>2</sub>). The multivalues of these formulas are slerived from their controllent formulas F<sub>1</sub> and F<sub>2</sub> as follows:
  - a. (F\_AND Fy) is TRUE if both  $\Gamma_1$  and  $P_2$  are TRUE, otherwise, it is FALSE.
  - F. (F<sub>1</sub> OR F<sub>2</sub>) is FAUST of Noth F and F<sub>2</sub> are FALST, otherwise, it is TRUE.
  - c. NOT  $(F_1)$  is true if  $F_1$  is present is false of  $\Gamma_1$  is true
  - d. NOT(F<sub>2</sub>) is TRUE if F<sub>2</sub> is FALSE if is FALSE if F<sub>2</sub> is TRUE

## 6.6.3 The Existential and Universal Quantifiers

In addition, two special symbols called quantifiers can appear in formulas; these are the universal quantifier  $(\nabla)$  and the existential quantifier  $(\exists)$ . Truth values for formulas with quantifiers are described in rates 3 and 4 below; first, however, we need to define the concepts of free and bound tople variables in a formula, homework we need to define the bound of it is quantified, meaning thou it appears in on  $(\exists z)$  or  $(\nabla z)$  clause; otherwise, it is free. Formally, we define a tuple variable in a formula as free or bound according to the following tales.

- An occurrence of a tople variable in a formula F that is an atom is free in F.
- An executioned of a tuple clarable it is free or bound at a formula made up of logical connectives—(F<sub>1</sub> ANd F<sub>1</sub>, (F<sub>1</sub> OR F<sub>2</sub>), NOT(F<sub>1</sub>), and NOT(F<sub>2</sub>)—depending on whether it is free or bound in F<sub>1</sub> or F<sub>2</sub> (if it receips in eitherk. Notice that in a formula of the form F = (F<sub>1</sub> AND F<sub>1</sub>) or F = (F<sub>1</sub> OR (F<sub>2</sub>)), a tuple variable may be free in F<sub>2</sub> and bound in F<sub>2</sub>, or vice version in this case, one occertance of the tuple variable is bound and the other is free in F
- All free occurrences of a typic variable t in F are bound in a formula F' of the form F' = (∃ i)(F) or F' = (∀ i)(F). The typic variable is bound to the quantitie: specified in F'. For example, consider rise following formulas:
  - F1 : 0. DNAME "RESEARCH"
  - $F_{0} \neq (\exists \uparrow) (o, \text{ONUMAER-T}, \text{DVO})$
  - $F_1 = (\forall e) (e, MGRSSN-13334455551)$

The topic variable d is free an burb  $F_1$  and  $F_2$  , whereas it is bound to the ( $\forall$ ) quantities in  $F_2$ .

We can now give rules 3 and 4 for the demotion of 3 formula we started earlier:

- 3 If F is a functular then so is (B r)(F), where t is a tuple variable. The formula (B r)(F) is 180% if the formula F evaluaris to TRUE for some (at least one) tuple assigned to free productness of run F; or herwise, (B C(F) is FA18).
- i. If F is a formula, there so is (∀ :)(F), where t is a unple variable. The formula (∀ :)(F) is TRUE if the formula f evaluates to TRUE for cavity node (in the universal assigned to free occurrences of ( on F) otherways. (∀ :)(F) is FAUSE.

The (**3**F quantitier is called an existential quantitier because a formula (**3**  $\partial(F)$  is TRUS if "there exists" some tuple that makes *F* TRUS. For the universal quartcher, **fV** r(F) is TRUE if every possible tuple that can be assigned to free occutrences of r in *F* is substituted for that *F* is TRUE for every such variation. It is called the universal or "for all quantitier because every tuple in "the universal of " ruples must make *F* TRUE to make the quantified formula TRUE.

## 6.6.4 Example Queries Using the Existential Quantifier

We will use some of the same queries from Section 6.5 to give a flavor of how the same quenes are specified are relational algebra and in relational calculus. Notice that some queries are easier to specify in the relational algebra than in the relational calculus, and vice versa.

#### QUERY 1

Remove the name and address of all employees who work for the 'Research' department.

Q1: {r.FNAME, c.LNAHE, r.AODRESS || EMPLOYEE(7) AND (Bd) (OEPAH MENT(d) AND d.ONAME="Research" AND d.ONLMBER=c.DNO(-} |

The one free table variables in a relational calculus expression should be those that separate the left of the bar (1.). In QL it is the only free variables it is then bound second to reach taple. If a taple statisfies the conditions specified in QL the artificates were conditions specified in QL the artificates were conditions and sources are remeved for each such taple. The conditions potential and transmitted specify the targe telations for a and d. The condition doesn't = Research' is a selection condition and corresponds to a SPECT operation in the relational algebra, whereas the condition of reactes = tract is a join condition and serves a smaller purpose to algebra (0.5).

#### QUERY 2

For every project located in "Stational", list the project number, the controlling department number, and the department manager's last name, birth state, and address

 $\begin{array}{l} Q2a \left( p \text{PNUMBER}, p, \text{DNUM}, m, \text{LNAME}, m, \text{BDATE}, m, \text{ADDRESS} \right) \left( \text{PROJECT}(p), \text{AND} \right) \\ \text{CMPLOYEE}(m), \text{AND} \left( p, \text{PLOCATION-'STAHLOND'}, \text{AND} \right) \\ (1 \text{Ed}) \left( \text{DEPARTMENT}(a), \text{AND} \left( p, \text{DNUM-J, CNUMBER}, \text{AND}, \text{J}, \text{mssn} \right) \right) \end{array}$ 

In Q2 there are two free tuple coriables, p and in. Tuple variable A is bound to the existential quantifier. The query condition is evaluated for every combination of tuples assigned to p and m, and our of all possible combinations of tuples to which p and m are bound, only the combinations that satisfy the condition are selected.

Several ruple variables in a query can range over the same relation. For example, to specify the query Q5—tor each employee retrieve the employee's first and last name and the first and last name of birs of her minorduite supervisor—we specify two tuple variables clinics that both range over the employee relation:

Q8: [J.FNAME, 2 LVANE, S.FNAME, S.LNAME, F EMPLOYEE(2) AND EMPLOYEE(3) AND 2 SUPERSSN=3 SSN[

#### QUERY 31

Find the more of each complete who works on some project controlled by department number 5. This is a variation of query 3 in which "all" is changed to "some." In this case we need two join conditions and two constantial quantities.

```
Q3': \{c, UNAME, c, FNAME \} EMPLOYEE(c) AND ((\exists x) (\exists w) (AND works_0) AND works_0 (a) AND works_0 AND we show the second and the secon
```

#### QUERY 4

Make a list of project numbers for projects that involve an employee whose last name as 'Smith', other as a worker or as manager of the controlling department for the project.

Q4: {p.pnumber 1 project(p) AND { (  $(\exists c)(\exists c)(\exists c)(employeetc) AND works_on(w) AND works_on(w) AND c.ssn=weessn) 1$ 

ıГ

```
((∃m)(∃d)(EMPLOYEE1m) AND DEPARTMENT(d) AND
p.DNLM=d.CNUMBER AND d.MCRSSN=0..SSN AND m.LNAME="5mmh")))
```

Compare this with the relational algebra version of this query in Section 6.5. The UNION operation in relational algebra can usually be substituted with an OR connective in relational calculus. In the next section we discuss the relationship between the universal and existential quantifier and show how one can be transformed into the other.

## 6.6.5 Transforming the Universal and Existential Quantifiers

We now introduce some well-known traveformations from mathematical logic that relate the universal and existential quantifiers. It is possible to transform a universal quantitier into an existential quantifier, and vice cersa, to get an equivalent expression. One general transformation can be described informally as follows: Transform one type of quantifier new two other with negation. (preceded by NOT); AND and OR replace one another a peared formula becomes ennegated, and an unnegated formula becomes negated. Some genul cases of this transformation can be stated as follows, where the = quibol stapsk for quivalent for

 $(\forall x) (P(x)) (\text{NOT} (\exists x) (\text{NOT} (P(x)))$  $(\exists x) (P(x)) (\text{NOT} (\forall x) (\text{NOT} (P(x)))$  $(\forall x) (P(x)) (\text{NOT} (\forall x) (\text{NOT} (P(x))) \text{OR} (\text{NOT} (Q(x)))$  $(\forall x) (P(x)) \text{OR} Q(x)) (\text{NOT} (\exists x) (\text{NOT} (P(x))) \text{OR} (\text{NOT} (Q(x)))$  $(\exists x) (P(x)) \text{OR} Q(x)) (\text{NOT} (\forall x) (\text{NOT} (P(x))) \text{AND} (\text{NOT} (Q(x)))$  $(\exists x) (P(x)) \text{OR} Q(x)) (\text{NOT} (\forall x) (\text{NOT} (P(x))) \text{AND} (\text{NOT} (Q(x)))$  $(\exists x) (P(x)) \text{AND} Q(x)) (\text{NOT} (\forall x) (\text{NOT} (P(x))) \text{OR} (Q(x)))$  $(\exists x) (P(x)) \text{AND} Q(x)) (\text{NOT} (\forall x) (\text{NOT} (P(x))) \text{OR} (Q(x)))$  $(\exists x) (P(x)) \text{AND} Q(x)) (\text{NOT} (\forall x) (\text{NOT} (P(x))) \text{OR} (Q(x)))$  $(\exists x) (P(x)) \text{AND} Q(x)) (\text{NOT} (\forall x) (\text{NOT} (P(x))) \text{OR} (Q(x)))$  $(\exists x) (P(x)) \text{AND} Q(x)) (\text{NOT} (\forall x) (\text{NOT} (P(x))) \text{OR} (Q(x)))$  $(\exists x) (P(x)) \text{OR} (Q(x))) (P(x)) (P(x)))$ 

NOT  $(\exists x) (P(x)) \Rightarrow NOT (\forall x) (P(x))$ 

## 6.6.6 Using the Universal Quantifier

Whenever we use a universal quantifier, it is quite judicious to follow a tew rules to ensure that our expression markes sense. We discuss these tales with respect to Query 3.

#### QUERY 3

Find the names of employers who work on all the projects controlled by department analysis 5. One way of specifying this query is by mang the universal quantifier as shown.

Q3: ( , , LNAME, , , , FNAME ) EMPLOYEE ( ) AND ( , ( $\forall$  x) (NOT (PROJECT(x)) OR NOT (x, DNUH-5)

OR ( ( $\exists$  a)(acression(a) AND appendent SSN AND (a pnimeer-approx) ) () ()

We can break up Q3 into its basic components as follows:

Q3: { $e_1$ UNAME,  $e_1$ ENAME ( EMPLOYEE( $e_1$ ) AND  $P_1$ )  $P_1 = ( (\forall x) (NOT(PRDJECT(x))) OR |F_1\rangle )$   $P_1 = NOT(e_0NUMES) OR |F_2$  $F_2 = ( (\exists x) (WORKS ON(<math>e_1$ ) AND  $e_2$ ESSN =  $e_2$ SSN AND  $x_2$ PNUMBER= $e_2$ PNO) )

We want to make sure that a selected couployee e works on all the projects controlled by department 5, but the definition of universal quartifier says that to make the quartified formula TRUE, the inner formula must be TRUE for all table or de non-cose. The trick is to exclude from the universal quartification all toples that we dee not metested in by making the condition TRUE for all such tiples. This is necessary because a interested in by making the condition TRUE for all such tiples. This is necessary because a nucleusally quantified tople variable, such as tom Q3, must evaluate to TRUE for every proble tiple assumed to at to make the quant fied formula TRUE. The first tiples to exclude (by making them evaluate automatically to TROE) are those that are not in the relation R of interest. In Q3, using the expression **NOT**(**PROJECT**(x)) inside the universally quantified formula evaluates to TROE all tuples x that are not in the PROTET relation. Then we exclude the tuples we are not interested in from R itself. In Q3, using the expression **NOT**(x.DM.M=5) evaluates to TROE all tuples x that are in the PROTET relation but are not controlled by department 5. Finally, we specify a condition  $F_1$  that must hold on all the remaining tuples in R. Hence, we can explain Q3 as follows:

- 1. For the formula  $E' = (\forall x)(E)$  in the TRUE, we must have the formula E by LUE for all rapidly for the processe dual can be assigned to x. However, in Q3 we are only interested in E being (TRUE for all tuples of the FROM: relation that are controlled by department 5. Hence, the formula E is of the form (**NOT**(FROM:(x)) **OR** [E]). The (**NOT**(FROM:(x)) **OR** [E] condition is TRUE for all tuples two ratio escore relation and has the effect of elaminating these tuples from consideration in the truth space of  $E_1$ . For every tuple in the PROFECT relation,  $E_1$  must be TRUE if E' is to be TRUE.
- 2. Using the same line of reasoning, we do not want to consider tuples in the expert relation that are not controlled by department number 5, since we are only interosted in PROJECT tuples where that = 5. We can therefore write:

```
IF LODNUM-59 THEN F
```

which is equivalent to

(NOT (x-DNUM=5) OR F<sub>2</sub>)

- Formula F<sub>1</sub>, hence, is of the form NOT(x0808-5) OR F<sub>1</sub>. In the context of Qill this means that, for a tuple x in the PROFE relation, either its in Re5 of at most satisfy F<sub>1</sub>.
- 4 Fittaffy, F<sub>1</sub> gives the condition that we want to hold for a selected instance tople that the employee works on every secret siple the has not been excluded yet. Such employee toples are selected by the query.

In English, Q3 gives the following condition for selecting an environmetuple of for every topic vanisher protect relation with x and = 5, there must exist a tuple to in views\_0% such that we set x = 0.5% and works = 0.5% (where 5.1)

Using the general transformation from universal to existential quantitiers given in Section 6.6.5, we can implicible the query in Q5 as shown in Q5A:

Q3A: (c.lname, c.fname ( employee(c) AND (NOT (  $\exists$  )) (project(v) AND ( v dnum=5) AND

(NOT (3 4)(MORKS\_ON(A) AND (4:ESSN=2:SSN AND (2:PNUM6ER=4: PNO)))))

We now give some additional examples of queries that use quantitiers.

#### QUERY 6

Find the names of employees who have no dependents.

Q6: (C FNAME: C.LNAME I EMPLOYEE(C) AND (NOT (B/)(DEPENDENT(d) AND ( SSN=d, ESSN)))

U-me the general transformation role, we can rephrase QV as follows:

QDA: (c.fname, c.lname) | employee(c) AND (( $\forall 3$ ) (NOT(dependent(d))) OR NOT(e.ssn=d.essn))))

#### QUERY 7

last the names of managers who have at least one dependent.

 $Q_{72}^{2}$  {c.fname, c.lname, pepployee(c) AND (( $\exists d$ ) ( $\exists p$ ) (department(d) AND dependent(p) AND c.ssn\_d.wgrssn AND (department(ssn)))

This query is handled by interpreting "managers who have at least our dependent" as managers for whom there exists some dependent."

## 6.6.7 Safe Expressions

Whenever we use universal quantifiers, existential quantifiers, or negation of predicates in a calculus expression, we must make sure that the resulting expression makes sense. A **site expression** in relational calculus is one that is guaranteed to yield a *finite number of* agic as its result; otherwise, the expression is called **unsafe**. For example, the expression

```
& UNOT (EMPLOYEE(-)))
```

is notify because it yields all toples in the universe that are not epicone ruples, which are annitely numerous. If we follow the tales for Q4 discussed earlier, we will get a safe expression when using universal quantifiers. We can define safe expressions more precisely by introducing the concept of the domain of a table relational calcidits expression. This is the set of all values that eacher appear as constant values in the expression of exist in any tuple in the relations referenced in the expression. The domain of [e + NOT(evpression)] is the set of all actuality values appearing its some tuple of the PROVEE relation (for any attribute). The domain of the expression (for any attribute). The domain of the expression (for any attribute), the domain of the expression (for any attribute), indices the interview of the expression (for any attribute).

Au expression is said to be safe if all values in its result are from the domain of the expression. Notice that the result of (i 1 NOT(F#ProveF(d))) is unsafe, since it will, in grand, include tuples (and hence values) from untails the FMProveF relation; such values are not in the domain of the expression. All of our other examples are safe expressions.

# 6.7 THE DOMAIN RELATIONAL CALCULUS

There is another type of relational culculus called the domain relational calculus, or situply, **domain calculus**. While 362, (see Chapter 8), a language based on tuple relational calculus, was being developed by 1931 Research at Son Jose, Cabifornia, another language called QRE (Query-Po-Example) that is related to domain calculus was being developed almost concurrently at 004 Research at Yorktown Heights. New York. The formal speciacution of the domain calculus was proposed after the development of the QPE system.

Domon calculus differs from uple colculus in the cyte of weakles used in formulas. Bother than having variables range over tuples, the considers range over single values from domains of attributes. To form a relation of degree a for a query result, we must have a of these domain variables—one for each attribute. An expression of the domain colculus with the turn.

```
\{x_1, x_2, \dots, x_n \in CSOND(x_1, x_2, \dots, x_n, x_{n+1}, x_{n+2}, \dots, x_{n+n})\}
```

where  $x_1, x_2, \dots, x_n, x_{n+1}, x_{n+2}, \dots, x_{n+n}$  are domain sumbles that range over domains (of attributes) and CCSD is a condition of formula of the domain relational takents

A formula is made up of atoms. The atoms of a formula are slightly different from those for the tuple initialis and can be one of the following:

1. An atom of the form  $\mathbf{R}(\mathbf{x}_1, \mathbf{x}_2, \dots, \mathbf{x}_n)$ , where  $\mathbf{R}$  is the time of a relation of degree, and each  $y_n + z_n + z_n + z_n$  domain variable. This atom states that it has of values at  $\langle \mathbf{x}_1, \mathbf{x}_2, \dots, \mathbf{x}_n \rangle$  must be a tuple on the relation whose name is  $\mathbf{R}$ , where  $\mathbf{x}_1$  is the color of the n barmbore value of the tuple. To make a domain variables, thus, we can write

 $x_1, x_2, \ldots, x_n \in \mathbb{R}(x_1, x_2, x_0) \text{ AND } \ldots$ 

ansread of

 $\{x_1, x_2, \dots, x_n \in R(x_1, x_2, x_1) | AND \in A$ 

- An area of the form x<sub>i</sub> op x<sub>i</sub>, where op is one of the comparison operators in the set [=, <, >, >, >, =, 7], and x<sub>i</sub> and y<sub>i</sub> are domain variables.
- An atom of the form x, ope one op x, where op is one of the comparison operators in the set (=, < ≤, >, ≥, ≥, A), y, and y, are shown outlighted, and c is a constant value.

As in tuple calculus, aron- evaluate to either TROS or FMSE for a specific set of values, called the teach values of the arons. In case 1, if the domain variables are assigned values corresponding to a tuple of the specified relation *B*, then the atom is TROE, in case 2 and 3, if the domain variables are assigned values that satisfy the condition, then the atom is TRUE.

In a similar way to the tuple relational calculus, tormulas are made up of atoms, variables, and quantitiers, so we will not repeat the specifications for formulas here. Since estimptes of queries specified in the domain calculus follow. We will use linear case letters L or,  $n, \ldots, n$ , s is for domain variables.

#### QUERY 0

Retrieve the birthdate and address of the couplayse whose name is John B. Snuth-

Q0: (iii + (3 y) (3 i) (3 ·) (3 i) (3 ii) (3 ii) (3 ii) (3 ii) (3 ii) (EMPLOYEE ((minamery)) AND g="tors" AND i=18" AND s=15x(3x1)) We reached variables for the potenti relations one to range over the domain of each introduction order. Of the ten variables  $q_1 q_2, \dots, q_n$  only united note free. We first specify the variables in order. Of the ten variables  $q_1 q_2, \dots, q_n$  only united note free. We first specify the variables in the non-base and eccess, by the free domain variables in the non-base and it to some T then we specify the condition for selecting a tuple following the for (1)-model, that the sequence of values assigned to the variables grown explicitly be a tuple of the construct relation and that the values for q (2000), q (2001), and y (cover) be 'john's 'B', and Smith' respectively. For convenience, we will quantify only those variables actually appearing in a conductor (these would be  $q_1 r_1$  and y in QO) in the test of our examples  $r_1^*$ 

An alternative shorthand notation, used in QBE, for writing this query is to assign the censuits 'John', 'B', and 'Smith' directly as shown in Q&A. Here, all variables not ippening to the left of the bar are implicitly existentially quantified <sup>14</sup>.

Q0As [ac: 1 EMPLOYEE; ']ohm', 'B', 'South', engage, s, st ]

#### QUERY 1

Retrieve the name and address of all employees who work for the 'Research' department.

Q1: {  $pv \in (\exists i) (\exists i) (\exists i) (\exists in) (EMPLOYEF (generally)) AND$ DEPARTMENT(<math>purple) AND [==TRESEARCH] AND purple)

A condition relating two domain variables that range over attributes from two relations, such as m = z in Q1, is a join condition; whereas a condition that relates a domain variable rule constant, such as l = (Research), is a selection condition.

#### QUERY 2

horevery project located in 'Stafford', list the project number, the controlling department number, and the department manager's last name, bitth date, and address.

QZ: (#qat = (B\_i) (B\_in)(B\_ia) (B\_i)(PROJECT(##) AND EMPLOYEE(gradinality)) AND DEPARTMENT(form) AND k-ni AND n-/ AND (-'S'x-POR'))

#### QUERY 6

Find the numes of employees who have no dependents.

Q6:  $(q_{i} + (\exists :) \in MPLCYEE(q_{i}) \cap (a_{i}) \in MOT(\exists :) \in DEPENDENT(booth) AND (=0)))$ 

Query 6 can be restated using universal quantifiers instead of the existential quantifiers as shown in Q6A:

Q6A:  $\{q \in (\exists r) \in (\mathsf{EMPLOYCE}(q; q; q; q; q; z) \in \mathsf{ND} : (\forall i) \in \mathsf{NOT}(\mathsf{CEPENDENT}(ln q; b)) \\ \mathsf{OR} : \mathsf{NOT}(r \in (1)))\}$ 

(1) Note that the neuron of quantifying entry the domain variables actually used or conditions and otdowing operations such as 600 with (generative) without separating domain variables with conmissional brevented notation used for conveniences in is not the convection ad notation (1) Again this is not in multy occurate notation).

#### QUERY 7

last the names of monogers who have at least one dependent

Q7:  $(x_1 + (\exists i) + (\exists i)) = (\exists i) (EMPLOYEE (quarked a xy2) - AND DEPARTMENT (byk) AND DEPENDENT (hopping) - AND (=) AND (=)$ 

As we mentioned earlier, it can be shown that any query that can be expressed in the relational algebra can also be expressed in the domain or tople relational calculus. Also, any sign expression in the domain or tople relational calculus can be expressed in the relational algebra.

The Query-By-Example (QBE) have use based on the domain relational calculus, although this was realized later, after the domain calculus was formalized. QBE was one of the first graphical query languages with minimum syntax developed for database systems if was developed at first Research and is available as an first commercial product as part of the QME (Query Management Eacility) interface option to 1962. It has been minimuked by several other commercial products. Because of its important place in the field of relational languages, we have included an overview in QNS in Appendix D.

## 6.8 SUMMARY

In dos chapter we proceed overformal languages for the relational model of data. They are used to manipulate relations and produce new relations as answers to queries. We discussed the relational idgebra and its operations, which are used to specify a sequence or operations to specify a query. Then we introduced two types of relational calculi called raple calculus and domains calculus they are declarative in that they specify the result of a query without specifying how to produce the query result.

In Sections 6.1 through 6.3, we introduced the base relational algebra operations and discroted the types of queries for which each is used. The title's relational operation SELFOT and FROJEUT, as well as the RENAME operation, were discussed first. Then we discussed binary set theoretic operations requiring that relations on which they are applied be union compatible, these include UNION, INTERSECTION, and SET INFERENCE. The CLARTESIAN PROPUCT operation is a set operation that can be used to combine taples from two relations, producing all possible combinations. It is tarely used in practice; however, we showed how CLARTESIAN PROPUCT operations and leads to the can operation. Different join epitations called THETA RON, EQUION, and NATEGOR, ICIN were introduced.

We then discussed some important types of queries that cannot be stated with the basic relational algebra operations but are important for proctical situations. We introduced the ACGREGATE PLANTED superation could with aggregate types of requests. We discussed recursive queries, for which there is no direct support in the algebra but which can be approached in a step-by-step approach, as we demonstrated. We demonstrated we demonstrated. We demonstrated we demonstrated we demonstrated we demonstrated with and process and allow all information in source relations to be preserved in the result.

The last two sections described the basic concepts behind relational calculus, which is basic, on the branch of mathematical logic nation predicate calculus. There are two types of relational calculus (1) the tuple relational calculus, which uses tuple variables that runs over myles (now) of relations, and (2) the domain relational calculus, which uses domain variables that range over domains (culturins of relations). In relational calculus, a given is specified in a single declarative statement, without specifying any order or method for remeeting the query result. Hence, relational calculus is often considered to be a light level language than the relational algebra because a relational calculus is presented to version the cause a relational calculus is presented.

We discussed the spreak of relational calculus spieries using both ruple and domain variables. We also discussed the existential quantitier  $(\exists)$  and the universal quantitier  $(\forall)$ . We saw that relational calculus variables are bound by these quantitiers. We described in detail how queries with universal quantification are written, and we discussed the problem of spectrum safe queries whose results are finite. We also docused rules for transforming inversal into existential quantifiers, and vice versa it is the quantifiers that pie existence is no analog to grouping and aggregation functions in basic relational declarational algebra. There is no analog to grouping and aggregation functions in basic relational calculus, although some extensions have been sugge-ted.

#### **Review Questions**

- Use the operations of relational algebra and the purpose of earls.
- 6.2 What is union compatibility? Why do the UNION, INTERSECTION, and DIFFER-ENCE operations require that the relations on which they are applied by union compatible?
- 63 Discuss some types of queries for which teranoing of artributes is necessary in order to specify the query unambiguously.
- 64. Discuss the various types of inner join operations. Why is there join (equired)
- 6.5 What role does the concept of *foreign* key play when specifying the most common types of meaningful join operations?
- 5.6 What is the (US/CTION operation? What is it used for '
- 5.5 How are the OUTER JOIN operations different from the 2NNER JOIN operations? How is the 2027ER UNION operation different from UNION?
- 5.5 In what sense does relational calculus differ from relational algebra, and in what sense are risey similar?
- 6.9. How does tuple relational valculus differ from domain relational calculus?
- 6.12. Discuss the meanings of the existential quantifier (∃) and the universal quantifier (♥).
- 6.11 Define the following terms with respect to the tuple calculus: table canabie, range relation, arom, journaly, and expression.
- 6.12. Define the following terms with respect to the domain color dust domain variable, suggregation, aroun, formula, and expression.
- 615. What is meant by a safe expression in relational calculus?
- 614. When is a query language called relationally complete?

## Exercises

- 6.15 Show the result of each of the example queries in Section 6.5 as it would apply to the database state of Figure 5.6.
- 6 In: Specify the following queries on the database schema shown in Figure 5.5, using the relational operators docused in this chapter. Also show the result of each query as it would apply to the database state of Figure 5.6.
  - Remove the names of all employees in department 5 who work more than 10 hours per week on the 'ProductX' project.
  - List the names of all employees who have a dependent with the same first name as themselves.
  - Find the names of all employees who are directly supervised by 'Franklin Wong'.
  - d. For each project, list the project name and the total hours per week (by all employees) spent on that project.
  - e. Retrieve the names of all employees who work on every project.
  - 1. Retrieve the names of all employees who do not work on any project.
  - g. For each department, retrieve the department name and the average value of all employees working in that department.
  - b. Betrieve the average salary of all female employees
  - Find the names and addresses of all employees who work on a least one project located in Floisten For whose department has no location in Floisten.
  - j. List the last nomes of all department managers who have no dependents.
- 6.17 Consider the 450006 relational database scheme shown in Figure 5.8, which was described in Exercise 5.11. Specify the following queries an relational algebra:
  - a. For each tlight, list the flight number, the departure argont for the first legist the flight, and the arrival argont for the last legist the flight.
  - b. Lot the fight numbers and workdays of all flights or flight legs that depositrant houseon intercontinental Airpurt (airpurt code 1544) and arrive in Lo-Angeles Intercontinental Airpurt (airpurt code 5.537).
  - c. List the flight number departure surport code, scheduled departure take, arrival airport code, scheduled arrival time, and weekdays of all flights or flight legs that depart from some surport in the city of Llouston and arrive at some airport in the city of Los Angeles.
  - d. List all the information for flight number (CO197).
  - er. Retrieve the number of available scats for hight number 'Co197' on '1999-10-09.
- 6.18. Consider the 1155497 relational database schema shown in Figure 6.12, which is used to keep track of books, herrowers, and buok loans. Referential integrity constraints are shown as directed area in Figure 6.12, as in the notation of Figure 5.7. Write down relational expressions for the following queries:
  - a. How many copies of the book ruled The Lost Tribe are owned by the library branch whose name is "Shorpstown"?
  - b. How many copies of the book titled The Loss Tribe are owned by each library branch?
  - co. Retrieve the names of all horrowers who do not have any books checked out

- d. For each book that is loaned out from the 'Shatpstown' branch and whose DucDate is today, remove the book title, the homower's name, and the borrower's address.
- For each library branch, retrieve the branch name and the total number of books loaned out from that branch.
- Retrieve the names, addresses, and number of books checked out for all Jerrovers who have more than two brooks checked out.
- g. For each book authored (or counthored) by 'Stephen King,' remove the title and the number of copies owned by the library branch whose name is 'Central.'
- 6.15 Spraify the following queries in relational algebra on the database schema given in Exercise 5.13:
  - List the Order# and Ship\_date for all orders shipped from Warehouse number (W2)
  - Is List the Warehouse information from which the Costomer named flow Lope: was supplied his orders. Produce a listing. Crearet. Warehouse=
  - Produce a frating cosmon, #receivers, wc\_osure\_or, where the middle column is the rotal number of orders by the costomer and the last column is the average order orminal for that costomer.
  - d. List the orders that were not shipped within 30 days of ordering.
  - List the Order+ for orders that were shapped from all watchouses that the compary has in New York.
- 640. Sportly the following queries in relational algebra on the database schema given in Exercise 5.14:
  - Give the details (all attributes of true relation) for trues that exceeded \$2002 in expenses.



FIGURE 6.12 A relational database schema for a crease database.

- Print the SSN of salesman who took trips to 'Honelulu'.
- Print the total tup expenses monited by the salesman with SSN = (254-56-7890).
- 6.21. Specify the following queries in relational algebra on the database schema given in Exercise 5 15:
  - List the number of courses taken by all students named 'John Snith' in Winter (999 (i.e., Quarter = 'W'99').
  - b. Produce a dist of textbooks finclude Course#, Book\_ISBN, Book\_ Tirle) for courses offered by the CS' department that have used more than two books.
  - c. Erst any department that has ad its adopted books published by 'AWL Publishing'.
- 6.22. Consider the two tables T1 and T2 shown in Figure 6.13. Show the results of the following operations.
  - a. The miners of T2
  - N. TL 11/0-12.0 T2
  - TUT <sub>HARMEN</sub> T2
  - $J_{\rm e}=T I_{\rm e} = \frac{1}{T I_{\rm e} \gamma_{\rm e}} + \frac{1}{T I_{\rm e}} T 2$
  - < ΤΙ U ΤΫ
  - T1 TENET/AGD ENTRY T2
- 6.23 Specify the following queries on relational algebra on the database schema of Exercise 5.16;
  - For the salesper-on-named 'Jane Doe', list the following information torial) the cars she sold: Senal#, Manufacturer, Sale-price
  - East the Senal# and Model of cars that have no options.
  - c. Consider the NATURAL IOIN operation between stressessor and sales. What is the meaning of a left OUTEX IOIN for these tables (do not charge the order of februars). Explain with an example.
  - d. Write a query in relational algebra involving selection and one set operation and say in words what the query does.
- 6.24. Specify queries a, b, c, e, f, it and j of Exercise 6.15 in both tuple and domain relatorical calculus.
- 6.25. Specify queries a, b, c, and d of Exercise 6.17 m both ruple and domain relational calculus.
- Specify queries c, d. i, and g of Exercise 6.18 in both tiple and domain relational calculus.

| Table T1 |      |   | Table T2 |     |  |  |
|----------|------|---|----------|-----|--|--|
| _r       | [ ၁] | R | A , B    | c ] |  |  |
| 10       | а    | 5 | 10 b     | Б   |  |  |
| 15       | b    | Б | 25 c     | з   |  |  |
| 25       | a    | б | 10 p     | 5   |  |  |

FIGURE 6.13 A database state for the relations T1 and 12

- 6.27 In a ciple relational calculus query with a tuple variables, what would be the typical mamman number of pure conditions? Why? What is the effect of having a smaller number of join conditions?
- 6.28 Resource the domain relational calculus queries that followed QC in Section 6.7 in the aigle of the albreviated notation of QCA, where the objective is romanimum the number of domain variables by writing constants in place of variables when ever possible.
- n.19 Censiller this query: Retrieve the swist of employees who work on at least those perfects on which the employee with swi = 12.5456789 works. This may be stated as yEORA 1, v) (15.2, THEN Q), where
  - A is a tople variable that ranges over the protect relation.
  - P = employee with ssi = 123456769 works on project x.
  - Q = employee e works on project a.

Express the query in tuple relational calculus, using the tubes

- $(\forall x)(P(x)) = NOT(\exists x)(NOT(P(x))).$
- (IF P THEN  $\hat{Q}$ ) = (NOT(P) OR  $\hat{Q}$ ).
- 3.57 Show how you may specify the following relational algebra operations in both tuple and domain relational calculus.
  - $a = \sigma_{A_{n}, M}(R(A, B, C))$
  - $h_{1}(\pi_{\lambda,A_{1}|B_{1}}(\mathsf{R}(A,B,C)))$
  - $\epsilon \in \mathcal{B}(A, B, C) \cap S(C, D, E)$
  - $\mathfrak{g}: R(A, B, C) \cup S(A, B, C)$
  - $e \in B(A, B, C) \cap S(A, B, C)$
  - (-R(A,B,C) = S(A,B,C)
  - $_{\Sigma}$  R(A, B, C)  $\times$  S(D, E, F)
  - h.  $R(A, B) \div \delta(A)$
- (19) Suggest extensions to the relational calculus so that 0 may express the following types of operations that were docussed in Section 6.4. (a) aggregate functions and grouping: (b) CCTER ION operations; (c) recursive closure queries.

## Selected Bibliography

Cod2 (1970) defined the basic relational algebras. Dure (1983a) discusses outer joins. Work en extending relational operations is discussed by Carlos (1986) and Oppoyelu et al (1985). Catanaanaa et al. (1989) extends the relational model integrity constraints and joins.

Gold (1971) introduced the language Alpha, which is based on concepts of tuple relational calculus. Alpha also includes the notion of aggregate functions, which poes beyond relational calculus. The original formal definition of relational calculus was given by Gold (1972), which also provided on algorithm that transforms any tuple relational radius expression to relational algebra. The QCLI (Stonebroker et al. 1976) is based on rople relational calculus, with implicative strength quantifiers but no universal quartifiers, this was implemented in the linguist system as a commercially available language. Code denied relational completeness of a query language formean at least as powerful as relational calculus. Ullman (1988) describes a tornal proof of the equivalence of relational algebra with the sate expressions of tuple and domain relational calculus. Abitghoul relational (1995) and Atomic and JeAntonellis (1997) give a detailed treatment of formal relational bargrages.

Although ideas of domain relational calculus were mutually proposed in the QNianguage (Zloot 1975), the concept was formally defined by Laurux and Protte (1977). The oppinmental version of the Query By Example system is described in Zloof (1977). The d.t. (Lacroix and Provite 1977a) is based on domain relational calculus. Whanglet al. (1993) extends QRE with universal quantifiers. Visual query languages, of which QRE is an example, are being proposed as a means of querying databases; conferences such as the Visual Database Systems Workshop (e.g., Arisava and Cararet (2002) or Zhou and Pa (2002) have a number of proposals for such languages.



# Relational Database Design by ER- and EER-to-Relational Mapping

We now toous on how to design a relational database schema based on a correction chemiclosign. This corresponds to the logical database design or data model mapping step discussed in Section 3.1 (see Figure 3.1). We present the procedures to create a relational schemic from an emity-relationship (68) or an enhanced ER (EE8) schema. Our discussion relates the constructs of the FR and EER models, presented in Chapters 3 and 4, to the constructs of the relational model, presented in Chapters 5 and 6. Many COVE (computer-rated scheme engineering) tools are based on the LR of EER models, or other similar models, as we have discussed in Chapters 3 and 4. These computerized tools are used interactively by datahave designers to develop an EEO fills schema for a database application. Many cools use FR or EFS diagrams or variations to develop the schema conductal, and then automatically convention in into a relational database schema in the TXT of a specific relational (2005 by engloying algorithms similar to the ones presented in this chapter.

We outline a seven-step algorithm in Section 7.1 to convert the basic 68 model constructs control types (strong and weak), binary relationships (with various structural constraints), *n*-ary relationships, and attributes (simple, composite, and mult valued)—into relations. Then, in Section 7.2, we continue the mapping algorithm by describing how to the EER model constructs—specialization/generalization and utility types (categories)—into charges.

# 7.1 RELATIONAL DATABASE DESIGN USING ER-TO-RELATIONAL MAPPING

## 7.1.1 ER-to-Relational Mapping Algorithm

We now describe the steps of an algorithm for ER-no-relational mapping. We will use the cosme database example to diestrate the mapping procedure. The conserve ER schema is shown again in Factor 7.1, and the corresponding cosmet relational database schema is shown in Figure 7.2 to illustrate the mapping step-



FIGURE 7.1 The ER conceptual schema diagram for the costan database.



**FIGURE 7.2** Result of mapping the corresponds schema into a relational database schema.

Step 1: Mapping of Regular Entity Types. For each regular tamese) entry type E a fle ER schemal create a relation *R* that includes off the simple artributes of *E*. Include only the simple component attributes of a composite attribute. Choose one of the key rindutes of *E* as primary key for *R*. If the chosen key of *E* is composite, the set of simple automates that form it will together form the primary key of *R*.

It multiple keys were identified for E during the conceptual descent, the information describing the attributes that form each additional key is kept in order to specify accordary (unique) keys of relation R. Knowledge about keys is also kept for indexing purpose and other types of analyses.

respectively. Knowledge that travel of nERAS REST and ERANE of EROFET are severably keys a kept for possible are later in the design

The relations that are created from the mapping of entity types are sometimes called **entity relations** because each tuple (now) represents an entity instance.

Step 2: Mapping of Weak Entity Types. For each weak entity type W in the F8 schema with owner entity type E, create a relation R and include all simple attributes on simple components of composite armbarcal of W as attributes of R. In addition, include as foreign key attributes of R the primary key attributets) of the relation(a) that correspond to the owner entity type(s), this takes care of the identifying relationship type f. W. The primary key of R is the combination of the primary key (s) of the owner(s) and the partial key of the weak entity type W if any

If there is a weak entity type  $E_2$  whose owner is also a weak entity type  $E_1$ , then  $E_1$  should be magned before  $E_2$  so determine its primary key first

In our example, we create the relation depicted in this step to correspond to the weak entity type depicted. We include the primary key six of the sourcent relation—which corresponds to the owner entity type—as a foreign key attribute of operatives: we recained it essay although this is not recessary. The primary key of the depicted relation is the combination [assa, depicted\_investor depicted pressure depicted power of depicted key of dependent

It is common to closure the propagate (CASCADE) option for the relational triggened action (see Section 8.2) on the foreign key in the relation corresponding to the weak entity type, since a weak entity has an existence dependency on its owner entity. This can be used for both ON CEDATE and CN OPLICIE.

Step 3: Mapping of Binary 1:1 Relationship Types. For each binary I I relationship type R in the E8 schema, identify the relations 5 and T that correspond to the entity types participating in R. These are three possible approaches: (1) the tonigh key approach, (2) the merged relationship approach, and (3) the cross-reference or relationship relation approach. Approach L is the most useful and should be followed unless special conditions exist, as we discuss below.

 Foreign key approach: Choose one of the relations -S, say—and archivle as a traeign key in S the primary key of T. It is better to choose an entity rope with total paraciparion in R in the rain of S. Include all the simple artributes (or simple components of composite attributes) of the 1-1 relationship type R as artributes of S.

In our example, we map the 1.1 relationship type wssess from Figure 7.1 by choosing the participating energy type operative, to serve in the role of S, because its participation in the wavess relationship type is word (every department has a manager). We include the primary key of the inscrutine drive as tore an key or the department relation and remains it waves. We also include the simple arribute standard of the wavess relationship type in the department and remains it relation and remains it waves.

Note that it is possible to include the primary key of S as a foreign key in T instead. In our example, this amounts to having a foreign key attribute, say assume \_source in the typeste relation, but it will have a null value for

employee roples whendo not manage a department. If only 10 percent of employces manage a department, then 90 percent of the foreign keys would be null in this case. Another possibility is to have foreign keys in both relations 5 and T redandantly, but this income a penalty for consistency maintenance.

- 2 Merged relation option: An alternative mapping of a 1-1 relationship type is possible by merging the two contry types on 1 the relationship into a single relation. This may be appropriate when block persicipations are total.
- 3 Consistence or relationship relation option: The third obternative is to set up a third relation R for the purpose of cross-referencing the primore keys of the tworelations S and T representing the entity types. As we shall see, this approach is acquired for binary M/N relationships. The relation R is called a relationship relation, (or sometimes a lookup table), because each tuple in R represents a relationsship instance that relates me tuple from N with one tuple of  $T_1$ .

Step 4: Mapping of Binary 1:N Relationship Types. For each regular binary 150 relationship type R, identify the relation S that represents the participating entity type it the Nonle of the relationship type. Include as foreign key in S the primary key of the relation T that represents the other entity type participating on R; this is done because solution stance on the Nonle is related to at most one entity instance on the Usade is related to at most one entity instance on the Usade attactive termination to a most one entity instance on the Usade attacted to at most one entity instance on the Usade attacted to at most one entity instance on the Usade attacted to attact on the Usade attacted to entitle the single components of compositive single relationship type. Include any simple attributes (or single components of compositive single components of the UN relationship type as products of S.

In our example, we now map the E.N relationship types west\_FOR contracts, and superability from Figure 7.1. For week was we include the primary key converse of the measurement relation as foreign, key in the two or unrelations and call in two. For supercriston we include the primary key of the Februari relation is foreign key in the februarization attest relation as the relationship is recursive—and call it 30.568539. The foreign key means the support to the foreign key attribute population exclusion schick references the primary key must solve the relationship is recursive was of the primary key must be foreign key attribute population schick references the primary key must solve the relationship is recursive.

An alternative approach we can use here is agon the relationship relation (crossinvence) option as in the case of binary 1.1 relationships. We create a separate relation Rwios, attributes are the keys of S and T, and whose primary key is the some as the key of S. This option can be used if few tuples in S participate in the relationship to avoid excessive null values in the tonigh key.

Step 5: Mapping of Binary MiN Relationship Types. For each binary MiN valuouslup type R, create a new relation S to represent 8. Include as foreign key attributes in 2 the primary keys of the relations that represent the patheipsting entity types; their audination will form the primare key of S. Also include any simple attributes of the MiN valuarship type for simple components of composite attributes) as attributes of S. Notzer this variants represent an MiN relationship type by a single torcen key attribute in one af the participating relations (as we did for 1:1 or 1:1) relationship types) because of the MiN valuarship relations (as we did for 1:1 or 1:1) relationship types) because of the MiN mathriality ratio we must counter a separate relationship relation S.

In our example, we may the McN relationship type waks on from Figure 7.1 by county the relation works or in Figure 7.2. We include the pureacy keys of the source

and svelorEE relations as longing logs in MORALING and rename them PAG and ESSA, respectively. We also include an attribute -0.65 or 20345\_06 to represent the BORA attribute of the relationship type. The primary key of the BORAS ON relation is the combination of the foreign key attributes (FSSA, Pao).

The propagate (CASCAOF) option for the referential triggered action (see Section A.2) should be specified on the foreign keys in the relation corresponding to the relationship  $R_{\rm c}$  and existence dependence on each of the entries it relates. This can be used for both ON UPDATE and ON DELETY.

Nonce that we can always map 1.1 or 3.N relationships in a maner similar to MN selationships by using the cross-televance (relationship) relation) approach, as we discussed eather. This alternative is particularly useful when few relationship instances exist, in order to avoid null values to begap keys to this case, the primary key of the relationship relation will be only one of the foreign keys that reference the particularly relationship re

Step 6: Mapping of Multivalued Attributes. For call multivalued attribute A, create a new relation R. This relation 8 with include an attribute corresponding to A, plus the primary key attribute K—as a fore go key to  $R_{\pm}$  of the relation that represents the entity type or relationship type that has A as an attribute is composite, we include its simple composite, we include its simple components.

In our example, we create a relation PEPULIC ATTEN. The articlote point training represents the analytical attribute constrained of department, while therefore as foreign  $ke_1 \rightarrow$  represents the primary key of the department relation. The primary key of deputies the combination of (primary key of the department, to generic tuple will exist in DETT constraines the combination of (primary has

The propagate (CV-CV, E) option for the referential triggerial action (see Section 3.2) should be specified on the foreign key in the relation *R* corresponding to the moltivalued attribute for both ON CPDATE and ON DELETE. We should also note that the key of *R* when magging a controsite, multivalued attribute requires some at all sis of the meaning of the component attributes. In some cases when a multivalued attribute is composite, only some of the component attributes are required to be part of the key of *R*, these attributes are similar to a part of key of a weak entity type that corresponds to the multivalued attribute (see Section 3.5).

Figure 7.2 shows the coverse relational database schema obtained through steps 1 to 5, and Figure 5.6 shows a sample database state. Notice that we did not yet discuss the mapping of narry relationship types ( $n \ge 2$ ), because none exist in Figure 7.1: these are mapped in a similar way to M N relationship types by including the following additional step in the impping algorithm

Step 7: Mapping of N-ary Relationship Types. For each acare relationship type R, where is > 2, create a new relation S to represent R. Include as foreign key

attributes in *S* the primary keys of the relations that represent the participating critity type. Also include any simple attributes of the neary relationship type (or simple composition of all the foreign keys that reference the relations representing the participating entity by es. However, if the cardinality constraints int any of the entity types *E* participating and *S* is **1**, then the primary key of S should not include the foreign key attribute theories the relations for any of the relation E constraints attributes the relations of the entity types *E* participating the participating in *S* is **1**, then the primary key of S should not include the foreign key attribute theories the relation *E* corresponding to E (see Section 4.7).

For example, consider the relationship type (GREE) of Figure 4.11a. This can be regard to the relation supervisioner in jugar. 7.3, whose primary key is the combination of the three foreign keys (SORE, 2000), (ROBER).

## 7.1.2 Discussion and Summary of Mapping for Model Constructs

Table 7.1 sumparizes the correspondences between ER and relational model constraints independents relations.

One of the main points to note an intelational schema in contrast room F3 schema is that relationship 1976s are not represented explicitly: instead, they are represented by being two ortributes A and B, one a primary key and the other a foreign key (over the sine domain) included in two relations X and T. Two toples in S and T are related when two have the some value for A and D. By using the FQCJOCS operation for NATCRAL DN if the two join attributes have the some name) over S A and T B, we can combine all parts included roples from S and T and materialize the relationship. When a binary 1-1 so



FIGURE 7.3 Mopping the *p*-ary relationship type super from Figure 4.11 a

| ER MODEL                    | RELATIONAL MODEL                             |
|-----------------------------|----------------------------------------------|
| Ешној туре                  | "Entity" relation                            |
| 15 or 158 relationship type | lorenen key (or "relationship" relation)     |
| MIN relationship type       | "Relationship" relation and two loreign keys |
| waav relationship type      | "Relationship" relation and in foreign keys  |
| Sumple attribute            | Attubute                                     |
| Composite attribute         | Set of simple component attributes           |
| Multivalues actribute       | Relation and foreign key                     |
| V-fue ser                   | Danam                                        |
| Key attribure               | Primary (or secondary) key                   |

TABLE 7.1 CORRESPONDENCE BETWEEN ER AND RELATIONAL MODELS.

I:N relationship type is involved, a single join operation is usually needed. For a binary M:N relationship type, two join operations are needed, whereas for a any relationship types, in joins are needed to fully materialize the relationship instances.

For example, to turn a relation that includes the employee name, project name, and hours that the employee works on each project, we need to connect each BRE SER tuple to the related vector tuples that the works or relation of Figure 7.2. Hence, we must apply the EQUION operation to the project and works on relations with the join condition on a tion, and they apply another EQUIONS operations to the resulting relation and the resourrelation with join condition  $\infty$  = enough in general, when multiple relationships need to be traversed, contended by articles in order to use them correctly in combining relation why join two or more relations. This is sometimes considered to be a drawball of the relational data model because the foreign key/primary key correspondences are roualways obvious upon inspection of relational schemas. If an equipon is performed another always obvious upon inspection of relational schemas. If an equipon is performed another the result can after be meaningles and may lead to sporte they are be conduced as the result can after be meaningles and may lead to sporte the routies for such that. For example the result can after be meaningles and may lead to sporte the routies for the condition in No. 100 = workston and example the result (see also Chapter 10).

Another point to note up the relational otherma is that we create a separate relation to ouch indirectined artificity. For a particular entity with a set of volues for the multivalued artificate. For a particular entity with a set of volues for the multivalued artificate in a separate tuple. This is because the basic relational model does on allow antificite values (a list or a set of volues) for an attribute in a single tuple. For exactly, we apply EQUIPOR to because department 5 has three locations, three to ples exist in the run\_lectrose relation of figure 5.6, each tuple specifies one of the locations. In our example, we apply EQUIPOR to testications and the basic relations of all locations along with other repeated in separate tuples for every location, the values of all locations are repeated in separate tuples for every location that a department has
For basic relational algebra does not have a NFST or COMTRESS operation that would predice from the  $0001_{10}$  controls relation of Figure 5.0 a set of toples of the turns [<1 Heistone, <4. Statforde, <5. {Bellaire, Sugarland, Heustonde). This is a serious drawback of the basic normalized or "flat" version of the relational model. On the score, the objectoriented model and the legacy hierarchical and network models have before facilities that local model. The relational model of object relational model. The relational model only object relational vacuum (see Chapter 22) arrange to remody this.

# 7.2 MAPPING EER MODEL CONSTRUCTS TO RELATIONS

We now discuss the mapping of EER model constructs to relations by extending the Fiercy relational mapping algorithm that was presented in Section 7.1.1.

# 7.2.1 Mapping of Specialization or Generalization

There are several options for mapping a number of subclasses that rogether form a specialconon (or alternatively, that are generalized into a superclass), such as the [stratuum, utwars), neurotical subclasses of an error in Figure 4.4. We can add a further step to our (Sto-relational mapping algorithm from Section 7.1.1, which has seven steps, to handle the mapping of specialization. Step 8, which follows gives the most continuou opticity, other mappings are also possible. We then discuss the conditions under which each option should be used. We use Aurs(R) to denote the activates of relation R, and PS(R) to datate the primary key of R.

Sep 8: Options for Mapping Specialization or Generalization. Us over each second once with an subclasses  $\{S_1, S_2, \ldots, S_n\}$  and (generalized) superclass  $C_i$  where the uniform of G are  $\{k, a_1, \ldots, a_n\}$  and k is the (primary) key. model at or schemos using one or the four following options:

- Option 8A: Multiple relations—Supercluss and subclasses. Create a relation L for *C* with entributes Attributes  $A(tristL) = \{k, a_1, \ldots, a_k\}$  and FK(L) = k. Create a relation L for each subclass  $N_i$ ,  $1 \ge i \ge m$ , with the attributes  $A(tristL_i) = \{k\} \cup \{a(tributes of S)\}$  and FK(L) = k. This option works for any specialization (total or partial disjoint or occularprop).
- Optim 8B: Multiple relations—Subclass relations only. Create a relation *k*, for each subclass S<sub>p</sub> 1 ≥ i ≥ in, with the antibures Artrs(L) = (attributes of S<sub>k</sub>) ⊂ [k, a<sub>1</sub>, . . ., a<sub>i</sub>] and Pa(L) = k. This option only works for a specialization where subclasses are notal (every entity up the superclass point belong to (at least) unset the subclasses).
- Option 8C: Single relation with one type attribute. Create a single relation l with arrebutes  $Arres[l] = \{k, a_1, \ldots, a_n\} \cup [attributes of S_1\} \cup \ldots \cup \{attributes of S_n\} \cup [1] and <math>ps(L) = k$ . The arribute i is called a type for discriminating) arribute that

inducates the subclass to which each cuple belongs, if any. This option works only for a specialization whose subclasses are dojoint, and has the potential for generating using null values if many specific attributes exist in the subclasses.

• Option 8D: Single relation with multiple type attributes. Uncate a single relation scheme L with attributes Attrifue =  $\{k_1, a_1, \dots, a_n\} \cup \{\text{trivitures of } S_1\} \cup \dots \cup \{a_{n+1}, a_n\} \cup \{r_1, r_2, \dots, r_n\}$  and  $\text{PR}(D = k | \text{Each } r_i | 1 \le i \le in \le n$  Boolean type attribute inducating whether a tople belowgs to subclass S. This option works for a specialization whose subclasses are coertaiguing that will also work for a distornt specialization).

Options  $\delta A$  and  $\delta B$  can be called the **multiple-relation options**, whereas options  $\delta C$ and  $\delta D$  can be called the **single-relation options**. Option  $\delta A$  creates a relation *L* for the superclass C and its attributes, plus a relation *L*, for each subclass *S* : each *L*, uncludes the specific (or local) attributes of *S*, plus the primary key of the superclass *C*, which is propagated to *L*, and becomes its primary key. An 55000008 operation on the primary key between any *L*, and *L* produces all the specific and inherited attributes of the critics in *S*. This option is illustrated in Figure 7.4a for the EPC schema in Figure 4.4. Option  $\delta A$ 



**FIGURE 7.4** Options for mapping specialization or generalization. (a) Mapping the FIR schema in Figure 4.4 using option 8A. (b) Mapping the FIR schema in Figure 4.3b using option 6B. (c) Mapping the EEC schema in Figure 4.4 using option 6C. (d) Mapping Figure 4.5 using option 8D with Boolean type tields MFIag and PFIag.

vide for any constraints on the specialization: disjoint or overlanging, total or partial. Noice itiat the constraint

 $\pi_{+,+}(L_i)\subseteq\pi_{+,+,i}(L)$ 

must had to react L. This specifies a foreign key from each L, to L, as well as an inclusion decidery L,  $k \in Lk$  (see Section 11.5).

In option 8B, the 2QUIGNN operation is *built* into the schemal and the relation L is during way with, is illustrated in Figure 7.4b for the EUS specialization in Figure 4.3b. This uption works well, only when both the disjoint and total constraints hold. If the gravitalization is not total, an entity that does not belong to any of the succlasses  $S_1$  is lost. If the specialization is not disjoint, an entity belonging to more than one subclass will have its inherited attributes from the superclass C stored redundantly in more than one 1. Woh option 5B, no relation builds all the cathles in the superclass C) represequently, we may apply on 0.011EE UNION for P.0.1 (3, TEK [5000] operation, to the L relations to relieve all the entries in C. The result of the uniter union will be similar to the relations of decorrer union will be similar to the relations of entry in the type totals will be missing. Whenever we see that an arbitrary entity in C, we must served all the missing L.

Options 50 and 8D create a single relation to represent the superclass C and all its subclasss. An entry that does not belong to some of the subclasses will have null values of the specific attributes of these subclasses. These options are hence not recommended if may specific attributes of defined for the subclasses. If few specific subclass attributes are preferable to options 3A and 8B because they do away with the need to specify EQUIVAN and 10 TER UNION operations and hence can yield a time efficient implementation.

Option 6C is used to handle disjoint subclasses by including a single type (or image indecombinating) attribute (to indicate the subclass to which each tuple belongs, hence, the contain of ( could be 11, 2, ..., m). If the specialization is partial, ( can have null vises in tuples that do not belong to any subclass. If the specialization is attributelation, that attribute serves the purpose of t and t is not needed; this option is dlu-trated in Figure 7.4c for the FER specialization in Figure 4.4.

Option 5D is designed to handle overlapping subclasses by including Pr Boacas type telds one for each subclass. It can also be used for disjoint subclasses. Each type field t can lave a domain dyes, not, where a value of yes indicates that the tuple is a member of subclass 5, 10 we use this option for the 2DS spacialization in Figure 4.4, we would include drive types attributes—IsASecretary. IsAEngineer, and IsATechpreian—instead of the JeFType attribute in Figure 7.4c. Notice that it is also possible to create a single type angine of m 50 m stead of the surge fields.

When we have a multilevel specialization (or generalization) hierarchy or lattice, we denot have refollow the same mapping option for all the specializations. Instead, we can use one mapping option for part of the hierorchy or lattice and other options for other parts. Egure 7.5 shows one possible mapping into relations for the EGR lattice of Figure 4.5. Here we used option 8A for PLAS M/PARTARE, ALEMON, STOCK 1, SPECIAL STOCK



FIGURE 7.5 Mapping the FER specialization lattice in Figure 4th using multiple options.

# 7.2.2 Mapping of Shared Subclasses (Multiple Inheritance)

A shared solution, such as EXIMPERING MARGER of Freure 4.6, is a subclass of several superclasses, indicating multiple inheritance. These classes finistial have the same key attribute, otherwise, the shared subclass would be modeled as a category. We can apply any of the options discussed in step 6 to a shared subclass, subject to the restrictions discussed in step 5 of the mapping algorithm. In Figure 7.5, both options 5C and 8D are used for the share subclass structure, Uption SC is used in the Figure Figure field on (EmployeeType attribute) and option SD is used in the subclass of Studies attribute).

# 7.2.3 Mapping of Categories (Union Types)

We now add another step to the mapping procedure—step 9—to handle categories. A category (or union type) is a subclass of the *tanket* of two of more superclasses that can have different keys because they can be of different entry types. An example is the task category shown in Figure 4.7, which is a subset of the union of three entry types **2655**, **1056**, and **106966**. The other category in that figure, **800**(15)(15)(1, 00000), has two superclasses that have the same key attribute.

Step 9: Mapping of Union Types (Lategories). For mapping a category alway defining superclasses have different keys at is ensembly to specify a new key attribute called a surrogate key, when creating a relation in correspond to the category. This is because the keys of the defining classes are different, so we cannot use any one of them exclusively to identify all enricies in the category. In our example of Figure 4.7, we can create a relation twee to correspond to the issue category as ill istrated in Figure 7.6, and include any attributes of the category in this relation. The primary key of the issue relation

| FERSO         | N .             |           |            |             |
|---------------|-----------------|-----------|------------|-------------|
| <u>89</u>     | (inverticenseNo | Neme      | Address    | Ownerid     |
| BANK          |                 |           |            |             |
| <u>6Name</u>  | RAckiness ()    | bhanw     |            |             |
|               | NY              |           |            |             |
| Chane         | CAddress C      | hwneidal  |            |             |
| OWNER         | R               |           |            |             |
| <u>Ownerk</u> | 1               |           |            |             |
| REGIST        | FEAED_VEHICLE   | E         |            |             |
| Venidal       | d UcensoPlato   | Number    |            |             |
| CAR           |                 |           |            |             |
| Venetel       | n CShile Ch     | nake 17   | accel Cu   | Ra <b>r</b> |
| TRUCK         |                 |           |            |             |
| Vehidəl       | d Tiylake 1N    | Nocie) To | nrage ¦ Tነ | 'ear        |
| owns          |                 |           |            |             |
| Ownerk        | J Vehideld P    | uncheseDe | ale Leerk  | ) Pagular   |

**REURE 7.6** Mapping the CCR categories funion types) in Figure 4.7 to relations.

in the surrogate key, which we called Ownerld. We also include the surrogate key attribute Ownerld as foreign key at each relation corresponding to a superclass of the coreport to specify the correspondence in callees between the surrogate key and the key of each preclass. Notice that it a particular encode (or uses or concore) entiry is not a member of ownerld attribute anall value for its Ownerld attribute in its corresponding tople in the area to toward or concord) relation, and it would not have a tople in the owner relation.

For a congory whose superclasses have the same key such as voicer in Figure 4.7, there is no need for a surrogate key. The mapping of the statistic protocol category, which destroys that case, is also shown in Figure 7.6.

# 7.3 SUMMARY

Effection 7.1, we showed how a conceptual schema design in the ER model can be reapped to calibronial database schema. An algorithm for ER-to-relational mapping was given and illustened by exotoples from the 1092087 database. Table 7.1 seminarized the correspondences between the ER and relational model constructs and constraints. We then added addinional sepsio the algorithm in Section 7.2 for mapping the constructs from the ER model model into the relational model. Similar algorithms are uncorporated into graphical database design tools in automatically create a relational schema from a conceptual solution design.

### Review Questions

- 7.1. Discuss the correspondences between the ZR model constructs and the relational model constructs. Show how each KR model construct can be mapped to the relational model, and docuss any alternative mappings.
- Discuss the options for mapping 640 model constructs to relations.

### Exercises

- 7.3. Try tuning) the relational schema of Figure 6.12 actions 58 schema. This is part at a progress known as receive engineering, where a conceptual schema is created for an existing rapidemented database. State any assumptions you make.
- 7.4. Figure 7.7 shows an Eleschener for a database that may be used to keep track at transport ships and their locations for maritime authorities. Map this schema for a relational schema, and specify all primary keys and foreign keys.
- 7.5 Map the used FR schema of Exercise 3.23 (shown in Figure 3.47) into a relational schema. Specify all primary keys and foreign keys. Repeat for the snouw schema.



FIGURE 7.7 An ER schema for a side\_reactive database

(Figure 3.10) of fixercise 3.19 and for the other schemas for Exercises 5.16 through 3.24.

 Map the FFR diagrams in Figures 4.10 and 4.17 into relational schemas, Justify your choice of mapping optices.

# Selected Bibliography

The original (R-no-relational imapping algorithm was described in Chervs classic paper (claim 1976) that presented the original FR model





# sQL-99: Schema Definition, Basic Constraints, and Queries



The styl language may be considered one of the major reasons for the success of relational databases in the contributed dworld. Because it because a standard for relational durbases users were less concerned about migrating their database applications from energypes of database systems—tor example, network or hierarchital systems—to relational systems. The reason is that every if users because disartished with the particular relational systems. The reason is that every if users because disartished with the particular relational particle relations in use, converting to another relational 99808 product would follow the same fanguage standards. In practice, of course, there are nearly differences between various commercial relational 16838 packages. However, if the user is this generic using only those fortunes that are part of the standard, and it both relational systems faithfully support the standard, then conversion betweets the two systems should be tool is simplified. Another advantage of having such a standard is that users may write strements in a database application program that can access data stored on record more relational 268058 without having to change the database sublonguage (SqL) if both relational 198058 support standard SQL.

Inschapter presents the main features of the SQL standard for commental relational faves, whereas Chapter 5 presented the most minoritant concepts underlying the favoral infiminal data model. In Chapter 6 (Sections 6.1 through 6.5) we decreased the relational algebra operations, which are very important for understanding the types of requests that raybe specified on a relational database. They are also important for query processing and opmication in a relational Dists, as we shall see in Chapters 15 and 16. However, the relational algebra operations are considered to be too technical for most commercial 0588 issue because a query in relational objecta is written as a sequence of operations that, whit executed, produces the required result. Hence, the user must specify how othat is, written order to execute the query operations. On the other hand, the SQL language provides a lagher-level scalarshe language interface, so the user only specifies reliat the result is to be leaving the actual optimization and decisions on how to execute the query to the DSMS. Although SQL includes some features from relational algebra, it is based to a greater extent on the tiple relational algebra, which we described in Section 6.6. However, the SQL syntax is more user-threadly than eather of the two formal languages.

The name SQL is derived from Structured Query Language. Originally, SQL was called sEQOEL (for Structured English QC/Ery Language) and was designed and implemented at BM Research as the information for an experimental relational database system called S(S)(EM R. SQL is now the standard language for commercial relational DBMS). A join effort by ANSI (the American National Standards Institute) and (SO) (the International Standards Institute) and (SO) (the International Standards Organization) has led to a standard cersion of SQL (DS) (485) (referred to as SQL-3) was subsequently developed. The next version of the standard base originally called SQL-30 was subsequently developed. The next version of the standard base originally called SQL-30 but is now called SQL-00. We will the to extend the latest version of SQL as much is possible.

Because the specification of the SQL standard is expanding, with more features in such version of the standard, the latest SQL-99 standard is divided into a core specific gion plus optional specialized packages. The cure is supposed to be implemented by all RDRV- vendors that are SQL-99 compliant. The packages can be implemented a optional modules to be travelased independently for specific database applications such a data mixing, spatial data, temporal data, data warehousing, on late analytical processing (18.51), instrumedia data, and so on. We give a suburgery of some of these packages—and where they are discussed in the book—at the end of this chapter.

Because SQL is very important (and quite large) we devote two chapters to its bise teatures. In this chapter, Section 8.1 describes the SQL 10.9 commonds for creating schemas and tables, and eves an overview of the basic data types in SQL Section 8.2 presents how basic constraints such as key and referential integraty are specified. Section 8.4 discusses statements for modifying schemas, tables, and constraints. Section 84 discusses the basic SQL constraints for specifying referenced queries, and Section 8.5 gas over more complex features of SQL queries, such as aggregate functions and grouping. Section 8.6 describes the SQL commands for insertion, deletion, and updating of data.

<sup>1.</sup> Decembly, 500 had statements for entering and deepping in lesses in the files that represent relations, bet these have been dropped from the SQL standard for some time.

becom 5.7 lists some 564 features that are presented in other chapters of the book; these include transaction control in Chapter 17, security/authoritation in Chapter 33, active (authors unggers) in Chapter 24, object-oriented features in Chapter 22, and 0, AP (Online Analytical Processing) features in Chapter 28. Section 8 & sourcards the chapter.

In the next chapter, we discuss the concept of views (virtual tables), and then fearthe how more general constraints may be specified as assertions of checks. This is follower by a description of the various database programming techniques for programming with SQL.

For the reader who desires a less comprehenvive introduction to SQL pairs of Section 85 may be skipped.

# 8.1 SQL DATA DEFINITION AND DATA TYPES

S0, uses the terms tables row, and column for the termal relational model terms (obtained tople, and attrahute) respectively. We will use the corresponding terms interchangeably like main 80t command for data definition is the UREATE statement, which can be used to reate achieves, tables (relations), and domains (as well as other constructs such as seeverasettions, and triggers). Before we describe the relevant CREATE statements, such as seeverasettions, and triggers). Before we describe the relevant CREATE statements, we facus scheme and catalog concepts in Section 8.1.1 to place our discussion in perspective. Section 8.1.2 describes how tables are created, and Section 8.1.3 describes the must important data types available for attrabute specification. Because the SQL specification is one large, are give a description of the most important features. Forfber details can be found in the various 8.0, standards dominants (see bibliographic rares).

### 8.1.1 Schema and Catalog Concepts in squ

Each versions of sQL did nor include the concept of a relationel database schemo oll obles to lorients) were considered part of the same schema. The concept of an SQL schema assume opportated starting with sQL2 in order to group rogether tables and other anstrons that belong to the some database opplication. An SQL schema is identified by a schema name, and includes on nutborisation identifier to indicate the user or account above the schema, as well as descriptors for each detreast in the schema. Schema elenems include tables, constraints, views, domains, and other constructs (such as authoration goart-) that describe the schema. A schema is created via the CREATE SCHEVA assumpt, which can include all the schema elements' definitions. Alternatively, the schema can be assigned a mame and authorization identifier, and the elements can be atimed later. For example, the following statement creates a schema called Contour, owned by like user with authorization identifier.

#### CREATE SCHEMA COMPANY AUTHORIZATION JSMITH;

In general, not all users are authentized to create schemas and schema elements. The pivileg, to create schemas, tables, and other constructs must be explicitly granted to the relevant user accounts by the system administrator or DDA.

In addition to the concept of a schemal SQL2 uses the concept of a catalog—a based collection of schemas in an EQL covariancem. An EQL **environment** is boseally an installation of an EQL compliant RDBAS on a comparez system.<sup>2</sup> A catalog about contrains a special schema called (NECONATION\_SCHEM) which provides information on all the schemas in the catalog and all the element descriptors in these schemas. Integrity constraints such as referential integrity can be defined between relations only if they ever in schemas within the same catalog. Schemas within the same catalog cat also share certain elements, such as domain definitions.

# 8.1.2 The CREATE TABLE Command in SQL

The CREATE TABLE command is used to specify a new relation by giving it a name and specifying its annihures and minicil constraints. The attributes are specified first, and each attribute is given a name, a data (gpc to specific its domain of values, and any attribute constraints, such as NOT NULL. The kny entity integrite and referential integrity role straints can be specified within the CREATE TABLE (APUE command (see Section 8.4). Figure 5.1 shows sample data detroit on statements in SQL for the relational database scheme shown in 5 give 5.7.

Typically, the SQL-chemican which the relations are declored is implicitly specified in the environment at which the UREATE TABLE statements one executed. Alternatively, we can explicitly attach the schema name to the relation name, separated by a period. For example, by writing

#### GREATE TABLE COMPANY. EMPLOYEE ....

rather than

#### CREATE TABLE EMPLOYEE ....

as myligure 5.1, we can explicitly (rather than implicitly) make the evence table part of the crease scheme.

The relations declared through CREATE USBLE statements are called **base tables** for loss (elations), this means that the relation and its toples are actually created and stored as a file by the 199Ms. Base relations are distinguished from vietual relations, created through the CREATE VIEW statement (see Section 9.2), which have or may not correspond to an actual physical file. Its SQL the attributes in a base table are considered to be indexe in the sequence or which they are specified in the CREATE TARGE statement. However, rows (tuples) are not considered to be indexed within a relation.

soft also includes the concept of a choise of catalogs within an environment, but it is not viriclear at sometime levels of pestine are required in most applications.

| (a)                      |                         |                          |
|--------------------------|-------------------------|--------------------------|
| CREATE TABLE EMPLOYEE    |                         |                          |
| ( FNAME                  | VARCHAR(15)             | NOT NULL ,               |
| MINIT                    | CHAR,                   |                          |
| LNAME                    | VARCHAR(15)             | NOT NULL,                |
| SSN                      | CHAR(9)                 | NOT NULL,                |
| BLATE                    | LNATE.                  |                          |
| ADDRESS                  | VARCHAR(30)             |                          |
| \$EX .                   | ÇHAR 、                  |                          |
| SALARY                   | DECIMAL (10,2)          |                          |
| SUPERSSN                 | CHAR(9)                 |                          |
| DNO                      | INT                     | NOT NULL,                |
| PRIMARY KEY (SSN) .      |                         |                          |
| FOREIGN KEY (SUPERSSN    | ) REFERENCES EMPI       | LOYEE(SSNI               |
| FOREIGN KEY (CNO) REFE   | RENCES DEPARTMEN        | <pre>vT(DNUMBER));</pre> |
| CREATE TABLE DEPARTMENT  | -                       |                          |
| DNAME                    | VARCHAR/15)             | NOT NULL .               |
| DHUMBER                  | INT                     | NOT NULL .               |
| MGBSSN                   | CHAB(9)                 | NOT NULL .               |
| MGBSTARTDATE             | DATE.                   |                          |
| PRIMARY KEY (DNJIMB      | EB).                    |                          |
| UNDUE (DNAME)            |                         |                          |
| FOREIGN KEY (MGRSS       | N) REFERENCES EMP       | PLOYEE(SSN) ) :          |
| OPENTETABLE DEPT 1 OC AT | hows                    |                          |
| / ON IMPER               | INT                     | NOT NO D                 |
| DIOCATION                | ₩N<br>VARY°HAR(15)      | NOT NULL,                |
| DOUGHTON CON             | IER DU OCATIONE         | NOT NOLL,                |
| FOREKIN KEY (DINU MIR    | KERLEFFERENCES D        |                          |
|                          |                         |                          |
|                          | UK DOLLA LIZADO         |                          |
|                          | VANCHAR(15)<br>INT      | NOT NULL,                |
|                          | INTE<br>MARKÓ LIA DA AC | MOT MOLL,                |
| PLOCATION                | VAUKÇITAH(15),          |                          |
|                          |                         | NUT NULL,                |
|                          | ic.m) ,                 |                          |
| CODENCIA KEV (DAME)      |                         |                          |
| FUHERAR REF (DIVON)      |                         | ALMENT (CRACKREERL).     |
| CREATE TABLE WORKS_ON    |                         |                          |
| I FSSN                   | CHAR(9)                 | NOT NULL ,               |
| AND                      | INT                     | NOT NULL .               |
| HOURS                    | DECIMAL(3,1)            | NOT NULL,                |
| PHIMAHY KEY (ESSN, I     | PNO).                   |                          |
| FOHEIGN KEY (ESSN) I     | REFERENCES EMPLO        | YEE(SSN).                |
| FOREIGN KEY (PNC) R      | EFERENCIES PHOJEC       | T(PNUMBEH):              |
| CREATE TABLE DEPENDENT   |                         |                          |
| ( ESSN                   | CHAR(9)                 | NOT NULL ;               |
| DEPENDENT NAME           | VARCHAR(15)             | NOT NULL ,               |
| SEX                      | CHAR,                   |                          |
| BDATE                    | DATE,                   |                          |
| RELATIONSHIP             | VARCHAR(8),             |                          |
| PRIMARY KEY (ESSN        | DEPENDENT NAME)         |                          |
| FOREIGN KEY (ESSN)       | REFERENCES EMPLI        | DYEE(SSN));              |

**FIGURE 8.1** SERIE CREATE TABLE data definition statements for defining the consersherm from Figure 5.7

# 8.1.3 Attribute Data Types and Domains in SQL

The basic **data types** available for attributes include maneric, character oring, bit sting, biolean, date, and time.

- Numeric dark types include integer numbers of various sites (INTEGER or INT) and BRALLINT) and float hig-point treall numbers of various precision (ELOAD or REAL and EXAMPLE FRUCTSHOR). Formatted numbers can be declated by using EXAMPL(*j*)—or FEC(*i*) or NUMERIC(*i*)—where *i* the precision is the voral number of decimal desire and *i*, the scale, is the number of decimal desire the decimal point. The default for precision is implementation-defined.
- Character-string data types the either fixed length— CHAR(6) of CERRECTING(7), where K is the number of characters—or varyate length—VARCHAR(6) or CHAR VARCHAR(6) or CHAR VARCHAR(7) or CHAR VARCHAR(7) or CHAR VARCHAR(7) or CHAR VARCHAR(7) or CHAR VARCHAR(7), where *n* is the maximum number of characters. When specifying a literal string value, it is placed between single quotation ciarks (opnotrophes), and it is call constitute to distinction is made between uppercase and lowercase).<sup>5</sup> For record-length strings it shorter string is padded with black characters to the right. For example, it the value 'Smith' is for an artifiate of type CFAR(10), it is padded with two black characters to become 'Smith' is for an artifiate of type CFAR(10), it is padded with two black characters to become 'Smith' is for example, it appears to the generally ignored when strings are compared. For comparison purposes, strings are considered ordered in alphabetic order, there stril is considered to be less than size. There is also a concarenation operator denoted by 11 (daube vertical bart that can concatenate two strings to /Qt. For example 'abc' 11 'XiZ results in a single string 'abcXYZ'.
- Bit-string data types are either of fixed length n—0.7(n)—or volving length—0.7 vARYING(n), where n is the maximum toucher of bits. The detailt for n, the length of a characteristring or bit string, is 1. Literal bit strings are placed between single quotes but preceded by a B to distinguish them from characteristrings; for example, B1010117.
- A hoolean data type has the traditional values of TRUE or EADE. In SQL because of the presence of NUEL values, a three valued logic to used, so a third possible value set a boole in data type is UNKNOWN. We discuss the need for UNKNOWN and the three valued logic in Section 8.5 1.
- New data types for date and time were added in -QL2. The DATE data type has leave positions, and its components are YEVR, WONTH, and DAY in the form YYY-MM-DD The TIME data type has at least eight positions, with the components HODR, MINUTE, and SWOND in the form EEEMMSS. Only valid dates and times should be allowed by

5. But strings whose length is a maniple of 4 can also be specified on 'k suderout' notation, when the lateral string is preceded by X and each hex alconomic character represents 4 bits.

<sup>3.</sup> This is put the case with "QL keasards, such as IREACT or CHAR. With keynends, "QL is do not step, meaning that QL nexts upper or and lewer as determine prediction for keywords."

<sup>4.</sup> For non-alphabetic cherscience mercus a federed seder

the SQL implementation. The < (less than) comparison can be used with dates or times—an earlier date is considered to be smaller than a later date, and similarly with time. Eiteral values are represented by single-quoted strings preceded by the keyword PAT: or TIME for example, CATE 2002-09-27 or T.ME 50912:47. In addition, a data type TIME for example, CATE 2002-09-27 or T.ME 50912:47. In additional type TIME (i), where *i* is called *time forefored second* seconds precision, specifies (\*1) additional posteria for TIME—one positions for an additional separation character, and i positionfor specifying decimal fractness of a second. A TIME W.TO TIME 70KE data type radiales an additional six positions for specifying the displacement form the standard inversal time force, which is in the range \*10:00 to -12:59 in units of two teXMINE THE. II WITH TIME 20NE is next included, the default is the local time time for the SQL system.

- A timestamp data type (TIMESTAKAP) includes both the LOATE and COME fields, plus a noninnum of visipositions for decimal fractions at seconds and an optional Will'9 TIME 2005 qualities. Lateral values are represented by single-quoted strings precided by the kericoid TIMESTAME, with a blank space between data and sime; for example, TIMEsTAMP12002-08-27 09:12:47 e483021
- Another data reperiod realized to CATEC HATE, and TIMESTAMP is the INTERVAL data type. This specifies an interval to vehicle table that can be used to increment or decrement an absolute value of a date, time, or timestamp. Intervals are qualified to be when YE SRMEWTH intervals or DAT/TIME intervals.
- The formation DATE, 700E, and 100EE1 With can be considered as a special repetor string. Hence, they can generally be used in string comparisons by being east (or oriented or converted) into the equivalent strings.

It is possible to specify the data type of each attribute directly, as in Figure 8.1, alternatively, a domain can be declined, and the domain name used with the attribute quaterner. This makes it easier to change the data type for a domain that is used by nucleus attributes in a schema, and improves schema readability. For example, we can user, domain sequence by the following statement:

#### CREATE DOMAIN SSN\_TYPE AS CHAR(9);

We can use SSE\_TYPE in place of viteAR(9) in Figure 3.1 for the attributes SS2 and sufficient of precises, version of department, ESSE of meters\_ov, and ESSE of dependent. A domain can also have an optional default spon dication via a DEFAOUT clause, as we discuss later for attributes.

# **8.2 SPECIFYING BASIC CONSTRAINTS IN SQL**

We now describe the basic constraints that can be specified in SQL as part of table crosatom. These include key and referential integrity constraints, as well as restrictions on attribute domains and No. 1s, and constraints or, individual topies within a relation. We decise the specification of more general constraints, called assertions, in Section 9.1

# 8.2.1 Specifying Attribute Constraints and Attribute Defaults

Because SQL affords NULLS as introduce values, a constraint NOT N.911 may be specified it N.911 is not permitted for a particular articlute. This is always implicitly specified for the articlutes that are part of the prenary key of each relation, but it can be specified for any other articlutes whose values are required not to be NULL as shown in Figure 8.1.

It is also possible to define a definit value for an articlate by appending the class **DFFAULT** (value) to an attribute definition. The default value is included in any new tuple if an explicit value is not provided for that articlate. Figure 8.2 illustrates an example of specifying a default manager for a new department and a default department for a new employee. If no default classe is specified, the default default rate, is NULL for articlates do not have the NOT NULL constraint.

Another type of constructional entries of induces or domain values using the CHECK classes following an artribute or domain definition.<sup>2</sup> For example, suppose the department numbers are restricted to integer matches between 1 and 20, then, we can change the attribute declaration of transfers in the **DEFARTMENT** rable (see Figure 8.1) to the following:

#### DNUMBER INT NOT NULL CHECK (DNUMBER > 0 AND DNUMBER < 21).

The CHBCK clarse can also be used in conjunction with the CREATE LOWAIN statement. For example, we can write the following statement.

#### CREATE DOMAIN D\_NUM AS INTEGER CHECK (O\_NUM > 0 AND D\_NUM < 21);

We can then use the created domain to works the attribute type for all attributes that refer to department runneers in Figure (i.f., such as **CNUMPER** of **DEP**STREAT, **DOP** of **PROFE**), and of **SPETURE**, and show,

# 8.2.2 Specifying Key and Referential Integrity Constraints

Because keys and referential integrity constraints and very important, show are special charges within the CREVEE A09 - statement to sportly them. Someley implex to illustrate the specification of keys and referencial integrity are shown in figure 3.1.<sup>5</sup> The **PRIMAR**<sup>7</sup> KEY classe specifics one or more attributes that moke up the primary key of a relation by primary key has a single attribute, the clause can follow the attribute ducerly for example.

The CHRCR clause cut also be used for other purposes, as we shall see.

To Key and reference? (integrate constraint) or remain his likely in carry versions of SQU in some unfer anglementations, keys ovire specified anotherity at the internal ascel via the CBF VTF ISO-X communi-

CREATE TABLE EMPLOYEE I ... DNO INT NOT NULL DEFAULT 1. CONSTRAINT EMPPK PRIMARY KEY ISSN), CONSTRAINT EMPSUPERFK FOREIGN KEY (SUPERSSN) REFERENCES EMPLOYEE(SSN) ON DELETE SET NULL ON UPDATE CASCADE. CONSTRAINT EMPDEPTFK FOREIGN KEY (DNO) REFERENCES DEPARTMENT(D/NUMBER) ON DELETE SET DEFAULT ON UPDATE CASCADE ); CREATE TABLE DEPARTMENT(D/

CREATE TABLE DEPARTMENT

( ..., MGRSSN CHAR(9) NOT NULL DEFAULT 363655555', ... CONSTRAINT DEPTPK PRIMARY KEY (DNUMBER) , CONSTRAINT DEPTSK UNIQUE (DNAME). CONSTRAINT DEPTMOREK FOREIGN KEY (MCRSSN) REFERENCES EMPLOYEE(S\$N) ON DELETE SET DEFAULT ON UPDATE CASCADE ).

#### CREATE TABLE OFFT LOCATIONS

(... PRMARY KEY (DNUMBER DLOCATION), FOREIGN KEY (DNUMBER) REFERENCES DEPARTMENT(DNUMBER) ON DELETE CASCADE ON UPDATE CASCADE ):

**EQURE 8.2** Example illustrating how default attribute values and referential triggenerations are specifical in seg-

the partiant likes of paragraphics can be specified as follows (instead of the way in is specified in figure 8.1)

#### ONUMBER INT PRIMARY KEY;

The UNIQUE clause specifies alternate (secondary) keys, as illustrated in the DUMPNEX), newspectrable declarations in Figure 8.0.

Referential integrity is specified via the FOREIGN KEY clause, as showen a Figure 8.1. As we discussed in Section 5.2.4, a referential integrity constraint can be violated when agles are inserted or deleted on when a foreign key or primary key attribute value is induced. The default action that SQL takes for an integrity violation is to reject the aphroperation that will cause a violation. However, the schema designer can specify an alignative action to be taken it a referential integrity constraint is violated, by attracting a referential triggered action clause to any foreign key constraint. The options include SET NULL CREATED, and SET DEFAULT. An option must be qualified with either ON DELETE or CN\_UCDATE. We illustrate this with the examples shown in Figure 3.2. Here the database designer chooses SET NOLL ON DELETE and CREATE ON UCDATE for the foreign key subscent of supersister automotically set to NULL for all employee tuples that were referencing the deleted employee tuple. On the other hand, if the sist value for a supervising employee is ignated (say, because it was entered incorrectly), the new value is *cusceded* to supersist or all employee tuples.

In potenti, the action taken by the DBMS for 5ET NULL in SET PERACULT is the same to both ON DEFATE or ON UPDATE. The value of the affected referencing attributes is changed to NULL for SET NULL, and to the specified default value for SET DEFAULT. The action for CASCINDE ON CELETE is to delete all the referencing tuples, whereas the action for CASCINDE ON UPDATE is to change the value of the foreign key to the updated trees, primary key value for all referencing tuples. It is the topposition whereas the action to choose the appropriate action and to specify in in the database schema. As a general rule, the CASCINDE option is suitable for "relationship" relations (see Section 7.1), such as **100**, 200, 500, for relations that represent incluvialited artributes, such is **10**, 1000,000; and for relations that represent weak entity types, such is **10**,000.

# 8.2.3 Giving Names to Constraints

Figure 8.2 also illustrates how a constraint may be given a **constraint name**, following the keyword **CONSTRAINT**. The names of all constraints within a particular schema must be unaque. A constraint many is used to identify a particular constraint or case the constraint must be dropped later and replaced with another constraint, as we discuss in Section 83. Giving names to constraints is optimal.

# 8.2.4 Specifying Constraints on Tuples Using CHECK

In addition to key and referential integrity constraints, which are specified by special keywords, other table constraints can be specified through additional CHECK clauses at the end of a CREATS TABLE statement. These can be called **taple-based** constraints because they apply to each tuple *babealadly* and are checked whenever a tuple is inserted or minibled. For example, suppose that the **EEFARTSEE** table in Figure 3.1 had an additional attribute user essays with, which stores the date when the department was created. Then we could add the following CHECK clause at the end of the CREATE 1 ABLE statement for the previoust table to make sure that a manager's start date is later than the department creation date:

#### CHECK (DEPT\_CREATE\_DATE < MGRSTARTDATE);

The CHECK clause can also be used to specify more general constraints using the CREATE ASSERTION statement of SQL. We discuss this in Section 9.1 because in requires the full power of queries, which are discussed in Sections 8.4 and 8.5.

# 8.3 SCHEMA CHANGE STATEMENTS IN SQL

In this section, we give an overview of the schema evolution commands available in SQ, which can be used to alter a schema by adding or dropping tables, attributes, constraints, and other schema elements.

# 8.3.1 The DROP Command

The LGOP command can be used to drop noned schema elements, such as tables, domains, or constraints. One can also drop a schema Tor example, it a whole schema is not cooled any more, the LGOP SCHEMA command can be used. There are two dop island options: CASUADE and RESTRICT. For example, to remove the correspondations schema and all its tables, domains, and other elements, the viscoADE option is used as rollow:

#### DROP SCHEMA COMPANY CASCADE;

If the RESTRUCT option is chosen in place of CASUAPE, the schema is dropped only it is to above it in it; otherwise, the 1:800 command will cut be executed.

It a base relation within a schema is not needed any longer, the relation and its definition can be deleted by using the DOOP TAPLE command. For example, it we no opper wish to keep track of dependents of employees in the *Seisen* durabase of Engine 8.1, we can get rid of the *Stefanist* relation by using the following command.

#### DROP TABLE DEPENDENT CASCADE;

It the RESTRUTE option is chosen instead of CASUAUE, a cuble is drapped order that is surveyed an any constraints (for example, by foreign key definitions in another relation) or news (see Section 9.7). With the CA-CAUE option, all such constraints and sides that reference the table are dropped automatically from the schema, along with the table uselt.

The DPOP command can also be used re-drop other types of named schemu elements, sub-us constraints or domoins.

# 8.3.2 The ALTER Command

The definition of a base table of of other named schema elements can be changed by using the ALTER command. For base tables, the possible alternable actions include adding propping a culturar (attribute), changing a column definition, and adding or dropping role constraints. For example, to add an attribute for keeping track of jobs of engloyees with encore base relations in the conversions schema, we can use the command

#### ALTER TABLE COMPANY.EMPLOYEE ADD JOB VARCHAR(12):

We note will emer a value for the new arrithme number each individual -results tuple. This can be done either by specifying a default clause or by using the UTDATs command see Section 8.65. If no default clause is specified, the new articlaute will have SPULS in all the tuples of the relation intrieducely after the command is executed; hence, the 801 NUL constraint is recalled ed in this case.

To drop a colorum, we must choose either CAS, COP or RESTRICT for strop behavior lt CAS, COP is chosen, all consistents and clews that reference the colorum are dropped automatically from the schema, along with the colorum. If RESTRICT is chosen, the command is successful only if no views or constraints for other elements) reference the colorum. For example, the following command removes the attribute **autress from** the ENRICE has table:

#### ALTER TABLE COMPANY. EMPLOYEE DROP ADDRESS CASCADE;

It is also possible to alter a column definition by dropping an existing default class in by defining a new default class. The following examples distrate this clause:

#### ALTER TABLE COMPANY. DEPARTMENT ALTER MGRSSN DROP DEFAULT;

#### ALTER TABLE COMPANY.DEPARTMENT ALTER MGRSSN SET DEFAULT "333445555":

One can also change the construction specified on a table by adding or dropping a constraint. To be dropped, a construction and have been given a name when it mas specified. For example, to drop the constraint mental fixesory from Figure 3.2 from the (95 for fixelation, we write:

### ALTER TABLE COMPANY EMPLOYEE DROP CONSTRAINT EMPSUPERFK CASCADE;

Once this is done, we can redefine a replacement constraint by adding a new constraint to the relation, if needed. This is specified by using the ADD keyword in the ALTYR TABLE statement followed by the new constraint, which can be caused or annothed and can be af any of the table constraint types discussed.

The precedute subsections gave an everyiew of the schemin evolution commands of \$3.3. There are mony other details and options, and we refer the interested reader to the \$3.3. documents listed in the bibliographical notes. The next two sections discuss the querying capabilities of \$QL.

# 8.4 BASIC QUERIES IN SQL

SQL has one basic statement for retriening information from a database: the SELECT statement. The  $\beta$ FFF T superneut for retriening information from a database: the SELECT statement. The  $\beta$ FFF T superneut has no relationship to the  $\beta$ FFT operation of relational algebra, which was discussed in Chapter 6. There are many opticits and flavors to the SELET statement in SQL, so we will introduce its features gradually. We will use example queue specified on the schema of Figure 5.5 and will refer to the schema basis state shown of Figure 5.6 to show the results of some of the example queues.

Before proceeding, we must point out an important distinction between SQL and the timul relational model discussed in Chapter 5: SQL allows a table (relation) to have two activity ruples that are identical in all their attribute values. Hence, in general, an SQL take is not a set of tables, because a set does not allow two identical members, in their in its predicted processes a law constraint has been declared or because the DISTUNCT option has been used out the SUBTINCT option we discuss the examples.

# 8.4.1 The SELECT-FROM-WHERE Structure of Basic SQL Queries

Queres in SQ, can be very complex. We will start with simple queries, and then progress onnow, original unles in a step-by-step manner. The basic turns of the SELECT statement, sometimes called a mapping or a **select-trom-where** block, is formed of the three clauses \$450.7, ROM, and WHERT and has the following form:

| SELECT | <attribute list=""></attribute> |
|--------|---------------------------------|
| FROM   |                                 |
| WHERE  | <condition>;</condition>        |

u here

- <arritrate list> is a list of artribute names whose values are to be retrieved by the query.
- stable list > is a list of the relation names required to process the energy
- <condution 2 is a conditional (Boolean) expression that identifies the rubles to be removed by the query.

In SQL the basic logic d comparison operators for comparing attribute values with organother and with literal constants are =, <, <=, >>=, and <>. These correspond to the solutional algebra operators  $=, <, <=, >, <=, and \neq$ , respectively, and to the *c*/*i*++ more main algebra operators  $=, <, <=, >, <=, and \neq$ . The main difference is the owe operator. SQL has many additional comparison operators that we shall present gradially as needed.

We now dilustrate the basic SELECT statement in SQL with some example queues. The queues are labeled here with the same query numbers that uppear in C hapter 6 for easy assignment.

#### QUERY 0

Remeve the birthdate and address of the employee(-) whose name is John B. Smith'

```
Q0: SELECT BDATE, ADDRESS
FROM EMPLOYEE
WHERE FNAME='John' AND MINIT='B' AND LNAME='Smith';
```

This query involves only the PERIODE rolation listed in the HRAM clause. The query selects the DECOME tuples that satisfy the condition of the WEERE clause, then projects for result on the ROM, and REERS attributes listed at the SELECT clause. Q2 is similar to the following relational algebra expression, except that duplicates. If any, would not be clausinatesh

#### $\pi_{\text{slate quarks}}(\sigma_{\text{super}}^{-1})$ obtain AND MINITH'B' AND I NAMEHISMITH' (uncount))

Hence, a sample 501 query with a single robution name in the 30 W charse is similar to a SELFOT-PROJECT pair of relational algebra operations. The SELECT charse of SQL specifies the projection stockness, and the WHERE clause specifies the solution obtained The only difference is that in the 50L query we may get displicate toples in the result because the constraint that a relation is a set is not enforced. Figure 8.3a shows the result of query Q2 on the database of Figure 5.6.

The query Q0 is also similar to the following tuple relational calculus expression, except that duplicates, it any, would again not be eliminated in the SQL query:

# QU: (LEDATE, 1 ADDRESS \_ FMPLOYEE(I) AND 1.FNAME='John' AND 1.MINIT='B' AND 1.MANE='Smith'}

Hence, we can think of an implicit raple variable in the SQL query langing over each tople in the tweater table and evaluating the conditions in the WHERE clause. Only flyw toples that satisfy the condition - that is, these toples for which the condition evaluates to TRUE after substructing their corresponding attribute values—are selected.

#### QUERY 1

Retrieve the name and address of all employees who week for the 'Research' department

### Q1: SELECT FNAME, LNAME, ADDRESS FROM EMPLOYEE, DEPARTMENT WHERE DNAME='Research' AND DNUMBER=DNO;

Query Q1 is similar to a SELECT HOJECT JOIN sequence of relational algebra operations. Such queries are often called select-project-join queries. In the wHERE clausest Q1, the condition mass = 'Research' is a selection condition and corresponds to a 9FECT operation in the relational algebra. The condition noises = noises a join conditions which corresponds to a HOIN condition in the relational algebra. The result of query Q1 is shown in Freque 8.318. In general, any comber of select and point conditions may be specified in a single SQL query. The next example is a select-project-join query with two join conditions.

#### QUERY 2

For every project located in Stafford', list the project number, the controlling department number, and the department manager's last name, address, and birthdate.

Q2: SELECT PNUMBER, DNUM, LNAME, ADDRESS, BDATE FROM PROJECT, DEPARTMENT EMPLOYEE

| а   | BEATE                                                                           | _ ADDRS                                                    | iss _                                                           |                                                        | (D)                                         | FNAME                                | LNAME                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                          | ADDRESS                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|-----|---------------------------------------------------------------------------------|------------------------------------------------------------|-----------------------------------------------------------------|--------------------------------------------------------|---------------------------------------------|--------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|     | 1965 (* 195                                                                     | 73) Foldan' H                                              | wstan TX                                                        |                                                        |                                             | John<br>Franklin<br>Ramesh<br>Ramesh | Smar<br>Manag<br>Nanagan<br>Lingson                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 791 Ad<br>606 yr<br>979 Af<br>9831 P                                                                                                                                                                                                                     | nchar, Housion, Fx<br>Houghter, YX<br>Coak, Humbler, TX<br>Hue, Housion, TX                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| 1.) | <u>РМСМВ</u> ЯВ <u>-</u><br>10<br>91                                            | BNUM H<br>4 M<br>4 M                                       | NAME AF<br>Valaco 241 Aen<br>Valace 241 Ben                     | 1076555<br>y Oelaro, 'Y<br>y Refair TX                 | В <u>Э</u> АТЕ,<br>1960-46-20<br>1941-68-20 |                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
| ъ   | 5.4NAME                                                                         |                                                            | SPNAME                                                          | SLNAME                                                 |                                             |                                      | Ø) 35                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | N                                                                                                                                                                                                                                                        | ONAME                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
| 19  | 300<br>Fanton<br>300<br>Fanton<br>Fanter<br>Fanter<br>2000<br>Actual<br>SSN<br> | Smith<br>Werky<br>Doktyp<br>Marayên<br>Endesth<br>Jalixee: | Fighter<br>Ustrafia<br>Ustrafia<br>Lennes<br>Fighter<br>Fighter | Wong<br>Bang<br>Walice<br>Long<br>Wong<br>Wong<br>Wong |                                             |                                      | 12343<br>(03)44<br>(05)44<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05)45<br>(05 | 6755<br>5757<br>7777<br>4444<br>3452<br>6731<br>4444<br>3452<br>6731<br>6731<br>6731<br>6732<br>1444<br>444<br>444<br>444<br>446<br>6730<br>6730<br>6730<br>6730<br>6730<br>6730<br>7727<br>4431<br>4444<br>4444<br>4464<br>4464<br>4464<br>4464<br>4464 | Rescard<br>Heseard<br>Researd<br>Researd<br>Researd<br>Researd<br>Researd<br>Researd<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Asseard<br>Assear |

| 151 | FNAME    | <b>MINT</b> | LNAVE   | SSN_         | BOATE              | ADDRESS                  | SEX | SALARY | SUPERSON   | DNO |
|-----|----------|-------------|---------|--------------|--------------------|--------------------------|-----|--------|------------|-----|
|     | John     | в           | Broth   | 12:34G6 /3ar | 1965-34-31         | JI Hondrah, Houston, TX  | N.  | 18.00  | 3.03446676 | -   |
|     | Frankler | т           | Wong    | 3334455575   | 1955-12-0 <b>8</b> | 616 yeas, Housian TX     | - W | 47(34) | A296655555 | é.  |
|     | Ganeso   | к           | WEDGER  | 846184-144   | 1960 - 619 - 15    | 9/5 Fre Crik Horitati TX | - V | 30000  | 333447555  | - 7 |
|     | JANJOR   |             | 6 जुले। | 430493450    | 1972-07-31         | 6631 Reel Housen TX      | =   | 251.00 | 030445656  | 0   |

**BGURE 8.3** Results of SQL queries when applied to the cosmo database state shown in Figure 5.6. (a) (0. (b) Q) (c) Q2 (d) Q8. (c) Q9. (c) Q10, (g) Q1C

### WHERE DNUM=DNUMBER AND MGRSSN=SSN AND PLOCATION='Statford':

The join condition 0x8 - 0x04558 relates a project to its controlling department, shereas the join condition 958589 - 588 relates the controlling department to the suploved who manages that department. The result of query Q2 is shown in Figure 8.3c

# 8.4.2 Ambiguous Attribute Names, Aliasing, and Tuple Variables

In SQL the same name can be used for two (or more) products as long as the arributes are in different relations. If this is the case, and a query refers to two or time attributes with the same name, we must qualify the arribute name with the relation name to prevent ambiguuy. This is done by projecting the relation name to the attribute name and separating the two by a period. To illustrate this, suppose that in Figures 5.5 and 5.6 the two and cost arributes of the sectors: relation were called courses and ones, and the noise attribute of nessares transitions also called waves then, to prevent ambiguity, query Q1 would be replicated to show an Q1A. We must prefix the attributes one and courses in Q1A to specify which ones we are referring to, because the attribute names are used in both relations:

```
QIA: SELECT FNAME, EMPLOYEE NAME, ADDRESS
FROM EMPLOYEE DEPARTMENT
WHERE DEPARTMENT NAME='Research' AND
DEPARTMENT.DNUMBER=EMPLOYEE.DNUMBER:
```

Ambiguity also arises in the case of queries that refer to the same relation twice, as in the following example.

#### QUERY 8

For each employee, retrieve the employee's first and last name and the first and last name of his or her numediate supervisor-

### Q8: SELECT E.FNAME, E.LNAME, S FNAME, S LNAME FROM EMPLOYEE AS E. EMPLOYEE AS S WHERE E.SUPERSSN=S.SSN.

In this case, we are allowed to declare alternative relation homes  $\epsilon$  and s, called **aliases** or tuple variables, for the EMPLOYEE relation. An alias can follow the keyword AS, as shown in QS, or it can directly follow the relation manne—for example, by writing BMCMEE  $\epsilon$  EMPLOYEE is in the FROM clouse of QS. It is also possible to acroane the relation attributes within the query in SQ, by giving them aliases. For example, if we write

#### EMPLOYEE AS E(FN, MI, LN, SSN, BD, ADDR, SEX, SAL, \$SSN, DNO)

in the FROM clause, in becomes an alias for sear, at for statil, in for case, and so on-

In Qe, we can think of a and sites two different optes of the exercise relations the first the represents employees in the role of supervisors, the second, supervisors, unployees in the role of supervisors, the second, supervisors. We can now poin the two copies. Of course, its reality there is only one supervisors. We can now poin the two copies. Of course, its reality there is only one supervisors, and the joint combining its means to nom the relation, with itself by matching the toples that satisfy the joint conditioner, spreasor = 5.555. Notice that this is an example of a one-level recursive query, as we discussed in Section 54.2. In earlier versions of SQL, as in relational algebra, it was not possible to specify a general recursive query, with

as auknown number of levels, in a single 901 statement. A construct for specifying accusive quotics has been incorporated into 501-99, as described in Chapter 22.

The result of query Q5 is shown in Figure 8.5d. Whenever one or more abases are given to a relation, we can use these manes to represent different references to that relation. This optimits multiple references to the some relation within a query. Notice that if we want to, we can use this abasemaning mechanism in any SQL query to specify type variables for every table in the WHERG clause, whether or not the same relation meds to be referenced more than order. In fact, this practice is recommended since in reclusin queries that are easier to comprehend. For example, we could specify query QLA as in QLB:

### QIB: SELECT E.FNAME, E.NAME, E.ADDRESS FROM EMPLOYEE E, DEPARTMENT D WHERE DINAME='Research' AND D.DNUMBER=E.DNUMBER

If we specify imple variables for every table in the WIFRE chase, a select-project-join quity in AQL closely resembles the corresponding tople relational calculus expression (except for duplicate elimination). For example, compare Q1B with the following tuple relational calculus expression:

Q1: {e.fnwre. e.lname, e.aloress | lnflores(e) AND (Ed) (department(d) AND d.(name. Research' AND d.(name. Bosearch' Bosearch' AND d.(name. Bosearch' 
Nonce that the main difference—other their syntax—is that in the SQL query, the existential quantifier is not specified explicitly.

# 8.4.3 Unspecified where Clause and Use of the Asterisk

We discuss two more features of SQL here. A maximity WHERE clause indicates an condition on tuple selections lacroce, all tuples of the relation specthed in the FROM clause quality and are selected for the query result. If more than one relation is specified in the FROM clause and there is no WHERE clause, then the CRSOSS DRODUCT—all possible tuble combinations—or these relations is solected. For example, Query 9 selects all sectors say (Figure 8.36), and Query 10 selects all combinations of an FROM type 10 selects all acodes use (Figure 8.36).

#### QUERIES 9 AND 10

Select all independences sets (Q9), and all combinations of second sets and dependent root (Q10) in the database.

Q9: SELECT SSN FROM EMPLOYEE. Q10: SELECT SSN. DNAME FROM EMPLOYEE. DEPARTMENT: It is extremely important to specify every selection and join condition in the % HBE classe; if any such condition is overlooked, incorrect and very large relations may result. Notice that Q12 is similar to a COOPS (NODECT) operation followed by a PRODUCT operation in relational algebra. If we specify all the attributes of SPREVEL and ECONTRACT in Q10, we get the CRESS-PRODUCT (except for duplicate elemenation, it any).

To intrieve all the artribute values is the sclotted tuples, we do not have to list the attribute names explicitly in SQL; we just specify an owned (\*), which stands for elliptic articles. For example, duery QLC removes all the artribute values of any iserosu who works in certement number 5 (ligure 8.3g), query QLD remieves all the attributes of an inscrete and the attributes of the presentent in which be or she works for every employee of the 'Research' department, and QECA specifies the CB-385 PROPUSITIE of the twelvert and certements.

| $Q(C_{t})$ | SELECT | •                                 |
|------------|--------|-----------------------------------|
|            | FROM   | EMPLOYEE                          |
|            | WHERE  | DNO=5;                            |
| Q1D:       | SELECT |                                   |
|            | FROM   | EMPLOYEE, DEPARTMENT              |
|            | WHERE  | DNAME='Research' AND DNO=DNUMBER; |
| QIOA:      | SELECT | •                                 |
|            | FROM   | EMPLOYEE, DEPARTMENT.             |

# 8.4.4 Tables as Sets in SQL

As we mentioned garlier, soft usually treats a table not as a set bar rather as a **multiset** duplicate inplex can appear more flux once in a table, and in the result of a energy SQ does no automatically eliminate duplicate tuples in the results of queries, for the following reasons:

- Duplicate elimination is an exprisive operation. One way to implement it is to surthe ruples first and then elimitate duplicates.
- The user may want to see duplicate tuples in the result of a query
- When an aggregate function (see Section 8-5.7) is applied to tuples, in most cases we deriver want to eliminate duplicates.

An sightable with a key is restricted to being a set, since the key value must be detimer in each tuple.<sup>5</sup> If we do near to eliminate duplicate tuples from the result of an SQ query, we use the keyword **DISTINCT** in the SELSCT clause, meaning that only distinct tuples should remain in the result. In general, a query with SM SC (0STINCT eliminate duplicates, whereas a query with SELSCT ALL does not. Specifying SE FCT with nealer AU, not DISTINCT—is in our previous examples —is equivalent to SELSCT ALL For

<sup>5.</sup> In general, an 563, rathers not required to brow a key, although in most cases there will be one

exiciple, Query 11 retrieves the salary of every employee, if several employees have the same salary, that salary value will appear as many times in the result of the query, is shown in Figure 8.4a. If we are interested only in distinct salary values, we want each value to appear only once, regardless of how many employees each that salary. By using the knowl DISTINCT as in Q11A, we accomplish this, as shown in Figure 8.4b.

#### QUERY 11

Enricyc the salary of every couployre (Q11) and all distinct salary values (Q11A).

Q11: SELECT ALL SALARY FROM EMPLOYEE; Q11A: SELECT DISTINCT SALARY FROM EMPLOYEE;

s.) has directly incorporated some of the set operations of relational algebra. There are set or ion, (UNION), set difference (EACEPT), and set intersection (INTERSECT) operations. The relations resulting from these set operations are sets of tuples that is liplicate tiples are alignmented from the result. Because these set operations apply only us which we opply the operation have the some attributes and that the attributes appear on the same order in Sofi relations. The next example diastrates the use of ONICN.

#### QUERY 4

Make a list of all project numbers for projects that involve arcong loyce whose last name is 'Sunity', other as a worker or as a manager of the department that controls the project.

```
Q+: (SELECT DISTINCT PNUMBER
FROM PROJECT, DEPARTMENT, EMPLOYEE
```

| .al | SALARY |       | ipl  | SALARY |      |
|-----|--------|-------|------|--------|------|
|     | 30.0   |       |      | #X00.0 |      |
|     | 20000  |       |      | 100000 |      |
|     | 25000  |       |      | 25300  |      |
|     | 43000  |       |      | 433.0  |      |
|     | MARCO  |       |      | 363372 |      |
|     | 250.0  |       |      | 56000  |      |
|     | 25000  |       |      |        |      |
|     | 55000  |       |      |        |      |
| r;  | FNAUE  | LNAME | [t]ı | FNAME  |      |
|     |        |       |      | lanes. | Open |

**EQURE 8.4** Results of additional scal queries when applied to for commutatabase varies own in Figure 5.6. (a) Q11. (b) Q11A, (c) Q16, (d) Q18

```
 WHERE
 DNUMEDNUMBER AND MGRSSNESSN AND LNAME=Smith')

 UNION
 (SELECT DISTINCT PNUMBER

 FROM
 PROJECT, WORKS_ON, EMPLOYEE

 WHERE
 PNUMBER=PNO AND ESSN=SSN AND LNAME='Smith').
```

The first SELECT query removes the projects that involve a 'Smith' as manager of the department that controls the project, and the second retrieves the projects that involve a 'Smith' as a worker on the project. Notice that if several employees have the lost name 'Smith', the project names involving any of them will be removed. Applying the UNICK operation to the two SELECT queries eives the desired result.

SQL also has corresponding multiset operations, which are followed by the keywoid ALL (UNITAM ATT, SQLEPT ACL, INTERATOR ATT). Their results are multisets (duplicates are not eliminated). The behavior of these operations is dilatored by the examples in Figure 8.5. Basically, each tople — whether it is a duplicate or not—is considered as a different tople when applying these operations.

# 8.4.5 Substring Pattern Matching and Arithmetic Operators

In this section we discuss several more features of SQL. The first feature allows comparison conditions on only parts of a character string, using the LIKE comparison operator. This



FIGURE 8.5 The results of solimultiset operations (a) Two tables. R(A) and S(A), (b) R(A) UNION ALL SIAT OCTR(A) EXCEPTIALL S(A), (d) R(A, INTERSECTIAL S(A), iso he used for string pattern matching. Partial strings are specified using two reserved diarteters: ") replaces an arbitrary number of zero or more characters, and the inderscore [[Steplaces a single r number. For example, consider the following query.

#### QUERY 12

Retrieve all employees whose address is in Houston, Texas,

```
Q12: SELECT FNAME, LNAME
FROM EMPLOYEE
WHERE ADDRESS LIKE '%Houston,TX%';
```

To remove all employees who were born during the 1950s, we can use Query 12A. Fore, 5 must be the fluid character of the string foccarding mount format for date), so we use the value  $\frac{1}{2} = \frac{5}{2} = \frac{1}{2} = \frac{1}{2}$ , with each underscore serving as a placeholder for an ath may character.

#### QUERY 12A

Find all employees who seere born during the 1950s

```
Q12A: SELECT FNAME, LNAME
FROM EMPLOYEE
WHERE BOATE LIKE '__ 5 _____':
```

It in anderscore of % is needed as a literal character on the string, the character bould be preceded by an escape character, which is specified after the string using the key ad VSCAPE. For example, 'AB'\_CDV56EF' ESCAPE 'V represents the factor string AF CD9.5EF' because V is specified as the escape character. Any character not used in the string can be chesen as the escape character. Also, we need a rule to specify gostophes or single quotation marks (") if they are to be included in a string. Because the needed to begin and end strings, **k** an apostrophe () is needed, it is represented as or consecutive apost sphes (") so that it will not be interpreted as ending the string.

Arother teacher allows the use of attitudent in queries. The standard authmetic openers for addition (+), subtraction (+), multiplication (+), and decision (/) can be applied to conneric values or attributes with mummer dimesics. For example, suppose that we want to set be effect of giving all employees who work on the PreduciX' project a 10 percent case; we cruise Query [3] to see what their salures would become. This example also shows how we can rename an armbute in the query result using  $z_2$  in the selfect of chuse.

#### QUERY 13

Show the resulting salaries if every employee working on the ProductX1 project is given a 10 percent raise.

QD: SELECT FNAME, UNAME, 1.1"SALARY AS INCREASED\_SAL FROM EMPLOYEE, WORKS\_ON, PROJECT

### WHERE SSN-ESSN AND PNO=PNUMBER AND PNAME='ProductX';

beil-tring data types, the concurrente operator ill can be used in a query to appete two string values. For date, time, timestamp, and interval data types, operators include incrementing (+) or decrementing (+) a date, time, or timestamp by an interval liaddition, an interval value is the result of the difference between two date, time, or timestamp values. Another comparison operator that can be used for convenience is **DETWEEN**, which is illustrated in Query 14.

#### QUERY 14

Remove all employees in department 5 whose salary is between  $3\,30,000$  and 340,000.

```
Q14: SELECT *
FROM EMPLOYEE
WHERE (SALARY BETWEEN 30000 AND 40000) AND DNO ≥ 5.
```

The condition (SALARY BETWEEN 30000 AND 400000 in Q14 is equivalent to the condition ((SALARY >= 30000) AND (SALARY <= 40000))

# 8.4.6 Ordering of Query Results

sQU allows the user to order the ruples in the result of a query by the values of one or more artributes, using the ORDER BY clause. This is illustrated by Query 15.

#### QUERY 15

Retrieve a first of employees and the projects they are working on, ordered by deputment and, within each dejoirtment, ordered alphaberically by last name, first name,

Q15: SELECT DNAME, LNAME, FNAME, PNAME FROM DEPARTMENT, EMPLOYEE, WORKS ON, PROJECT WHERE DNUMBER=DNO AND SSN=ESSN AND PNO=PNUMBER ORDER BY DNAME, UNAME, FNAME;

The default order is in ascending order of values. We can specify the keyword DESCit we want to see the insult in a descending order of values. The keyword ASC can be used to specify according order explicitly. For evaluate, if we want descending order on towe and ascending order on towe, swell the ORDER to chause of Q15 can be written as

### ORDER BY DNAME DESC, LNAME ASC, FNAME ASC

# 8.5 MORE COMPLEX SQL QUERIES

In the previous section, we described some basic types of queries in  $\partial (\Sigma)$ . Because of the generality and expressive power of the longuage, there are many additional features that  $\rho^{0}(\alpha)$  users to specify more complex queries. We discuss several of these features in this general

# 8.5.1 Comparisons Involving NULL and Three-Valued Logic

90 his various rules for dealing with NGLE values. Recall from Section 5-1.2 that NGLE is used to represent a massing value, but that it usually has one of three different interpretation—value unknown (exists but is not known), value not available (exists but is porpody with/yeld), or attribute not applicable (andefined for this tugle). Consider the following examples to illustrate each of the three meanings of NPRE.

- 1.3 Skinsten cable: A part cular person has a date of birth burning conknown, so it is represented by NPTE on the database.
- Createdable to tethtickl take. A person has a home phone but does not want it to be listed, so it is withheld and represented as NULL in the database.
- Net applicable attrabate: An attrabute LastCollegeDegree would be NULL for a person who has no college degrees, because it does not apply to that person.

It is often not possible to determine which at the three meanings is interided; for example, a Nett. for the home phone of a person can have any of the three meanings. Hence SX, does not distance style=tween the different meanings of SO(1).

In general, earb NG U is considered to be different from every other NUL in the different, earb NG U is involved in a comparison operation, the result is considered to be UNKNOWN (it muc be TRUE or it may be FALAE). Hence, sQL uses a three-valued logic with values TRUE, FALSE, and UNKNOWN instead of the standard two-valued logic with values TRUE or FALSE. It is therefore necessary to define the results of three-valued logical connectives ANCO (08, and NOT are used. Table 3.) shows the NP for Gulues.

In select-projection queries, the general rule is that only those combinations of tiples that evaluate the logical expression of the query to TRUP are selected. Taple only nations that evaluate to EAUSE or UNKNOWN are not selected. However, there are exeptions to that rule for certain operations, such as outer joins, as we shall see.

•QUallows queries that clock whether an articlute value is NULL. Rather three only or <> to compare an attribute value to NULL, SQL uses 15 or 15 NOT. This is because SQL remates each NULL value as being distinct from every other NULL value, so equality emparison is not appropriate. It follows that when a join condition is specified, tuples only NULL values for the join attributes are not included in the result (inverse it is in a tribute state of \$5.6). Query 15 illustrates this; its result is shown in Figure 5.4d.

| AND     | TRUE    | EA] SE  | UNKNOWN |
|---------|---------|---------|---------|
| TRUE    | TRUE    | FALSE   | UNKNOWN |
| FALSE   | FALSE   | EALSE   | False   |
| UNKNOWN | UNKNOWN | FALSE   | Unknown |
| OR      | TRUE    | FALSE   | UNKNOWN |
| TRUE    | TRUE    | TRUE    | TRUE    |
| FALSE   | TRUE    | FALSE   | UNKNOWN |
| UNKNOWN | TRUE    | UNKNOWN | UNKNOWN |

TABLE 8.1 LOGICAL CONNECTIVES IN THREE-VALUED LOGIC

NOT

| TRUE    | FALSE   |
|---------|---------|
| FALSE   | TRUE    |
| UNKNOWN | UNKNOWN |

T

#### QUERY 18

Retrieve the names of all employees who do not have supervisor-

```
Q19: SELECT FNAME, LNAME
FROM EMPLOYEE
WHERE SUPERSSN IS NULL:
```

# 8.5.2 Nested Querics, Tuples, and Set/Multiset Comparisons

Some queries require that existing values in the database be forched and then used its comparison condition. Such queries can be conveniently formulated by long pestrel queries, which are complete select from where blocks within the WHERE clause of an the query. That other query is called the outer queey. Query 4 is formulated in Q4 without nested query, but it can be rephrased to use nested queries is shown in Q44. Q4A, into dues the comparison operator (N, which compares ) colies a with a set for midrated of values V and evaluates to TRUE if a some of the elements of V.

| Q4A: | SELECT | DISTINCT PNUMBER |         |                                  |
|------|--------|------------------|---------|----------------------------------|
|      | FROM   | PROJECT          |         |                                  |
|      | WHERE  | PNUMBER IN       | (SELECT | PNUMBER                          |
|      |        |                  | FROM    | PROJECT. DEPARTMENT,<br>EMPLOYEE |
|      |        |                  | WHERE   | DNUM=DNUMBER AND                 |

```
MGRSSN=$$N AND
LNAME='Smith')
OR
PNUMBER IN (SELECT PNO
FROM WORKS_ON, EMPLOYEE
WHERE ESSN=$$N AND
LNAME-'Smith)'
```

The first neared query selects the project numbers of projects that have a 'Smith moleculus manager, while the second selects the project numbers of projects that have a 'smith' nearest, while the second selects the project numbers of projects that have a 'smith' nearest, while the cuter cuter, we use the OB logical connective to retrieve arout imple if the suggest value of that tuple is in the result of either nested query.

Earnested query terms o smele artribute and a single taple, the query result will be a single (scalar) value. In such cases, it is permissible to use = instead of IN for the comparison operator, or general, the nested query will return a table (relation), which is a set of milliser of taples.

equallows the use of **suples** of values in comparisons by placing them within prettheses. To illustrate this, consider the following query:

### SELECT DISTINCT ESSN FROM WORKS\_ON WHERE (PNO, HOURS) IN (SELECT PNO, HOURS FROM WORKS ON WHERE SSN="123456789");

This query will select the second -econity numbers of all employees who week the same (gaper) hears) combinations on some project that employee 'John Smith' (whose 384 = 323456780') works on the flux example, the its operator compares the subtriple of values (piperortheses from money) for each tuple in work\_DV with the set of amone compatible in produced by the mested query.

In addition to the 19 operator, in number of other comparison operators can be used to compare a single value  $\pi$  (typically an introduce name) to a set or multiser V (typically a point query). The  $\pi$  ANT (or  $\pi$  is 00F) operator returns TRUE if the value  $\pi$  is equal to end user in the set V and is hence optivalent to 18. The keywords ANT and 6.00F have the space measure. Other operators that can be combined with NNT (or SONE) include  $\pi$ ,  $\geq \pi$  <, <-, and  $s \geq 1$  the keyword AUT can also be combined with each of these operators. Include  $\geq$ ,  $\geq -1$ , <-, and  $s \geq 1$  the keyword AUT can also be combined with each of these operators. Include  $\geq$ ,  $\geq -1$ , <-, and  $s \geq 1$  the keyword AUT can also be combined with each of these operators. Include  $\geq$  0.1. C) returns TRUE if the value  $\pi$  is greater than if the values in the set (or multiset) V. An example, is the following query, which ratios the names of employees believe salary is greater that the salary of all the employees independent  $\pi$ :

```
SELECT LNAME, FNAME

FROM EMPLOYEE

WHERE SALARY > ALL (SELECT SALARY FROM EMPLOYEE

WHERE DNO=5);
```

In general, we can have several levels of nested queries. We can once again be faced with possible ambiguing among anribute names of arributes of the same name exist—one is a relation in the FROM classe of the outer query, and another in a relation in the FROM classe of the incode query. The role is that a reference to an *unqualistic* distribute refers to the relation declated in the innermost nested query. For example, in the FROM classe and WHERE classe of the first mosted query of QAA, a reference to any unqualified attribute of the protective relation refers to the protect relation specified in the FROM classe of the nested query. To refer to an attribute of the protective relation specified in the sourquery, we can specify and refer to an attribute of the protective relation. These relevant similar to scope rules for program variables in most programming languages that allow nested querys and functions. To illustrate the potential ambiguity of attribute names in nested queries, consider Query 16, whose result is shown in Figure 8.4c.

#### QUERY 16

Retrieve the name of each employee who has a dependent with the same first name and same sex as the employee

```
Q16: SELECT E.FNAME, E.LNAME

FROM EMPLOYEE AS E

WHERE E.SSN IN (SELECT ESSN

FROM DEPENDENT

WHERE E.FNAME=DEPENDENT_NAME

AND E.SEX=SEX);
```

In the neared query of Q16, we must qualify 6.55x because it reters to the Six armibiate of DECOVE from the carter query, and DEPERTENT also has an autobury called Six. All unqualified references to SEX in the neared query reter to SEX of DEFERDENT. However, we do not have to qualify these and sex because the DEPERTENT relation does not have outplates called ENAL and SEX, to there is no ambiguity.

In is generally advisable to cosine rople variables (aliases) for all its referenced at an 9gL query to avoid potential environd ambiguities.

# 8.5.3 Correlated Nested Queries

Whenever a condition in the wHERE clause of a rested query references some ittribute of a relation declared in the outer query, the two queries are said to be correlated. We can architectand a correlated query better by considering that the rested query is coduated area *precision optical active precision optical query*. For example, we can think at Q16 is follows: For each entries with optic optic the nested query, which retrieves the asso values for all 057600 strategies with the same sex and name as that ENP. Order tuple; if the 588 values of the second care schemes are solved at the rested query. The same of the same sex and name as that ENP. Order tuple; if the 588 values of the second care, there select that ENPLOSE tuple.

In general, a query written with nested select from where blocks and using the flar in comparison operators can abrays be expressed as a single block query. For example, Q16 may be written as in Q16A

## Q16A: SELECT E.FNAME. E.LNAME FROM EMPLOYEE AS E, DEPENDENT AS D WHERE E.SSN=D.ESSN AND E.SEX=D.SEX AND E.FNAME=D.DEPENDENT\_NAME.

The original SQL implementation on SYSTEM 8 also had a CONTAINS comparison operation which was used to compare two sets or multisets. This operator was subsequently drapped from the language, possibly because of the difficulty of implementing it efficiently. Most commercial implementations of 500 do not have this operator. The UNIAINS operator compares we sets of values and returns TRUE if one set continues all takes in the other set. Quere Fulliseties the use of the CONTAINS operator.

### QUERY 3

Remove the nome of each employee who works on all the projects controlled by department number 5.

```
Q :: SELECT FNAME, LNAME
```

FROM EMPLOYEE

| WHERE | 1 | (SELECT  | PNO        |
|-------|---|----------|------------|
|       |   | FROM     | WORKS_ON   |
|       |   | WHERE    | SSN=ESSN)  |
|       |   | CONTAINS |            |
|       |   | (SELECT  | PNUMBER    |
|       |   | FROM     | PROJECT    |
|       |   | WHERE    | DNUM=5) ); |

In Q3, the second nested query (which is not correlated with the outer query) refrees the project numbers of all propers controlled by department 5. For each employee mple, the first method query (which is correlated) refrees the project numbers in which the employee works if these control all projects controlled by department 5, for each analysis tuple is selected and the name of that employee is refreeved. Notice that the CONTAINS companion operator has a similar function to the TAMSION operation of the relational dgebra (see Section 6.3.4) and to universal query in relational calculus see Section 6.6.6). Because the CONTAINS operation is not part of SQL we have to us other techniques, but for the EXISTS function, to specify these types of queries, as described in Section 6.5.4.

# 8.5.4 The exists and UNIQUE Functions in SQL

The MSTS function in SQL is asid to check whether the result of a correlated rested queries empty (contains no toyles) or not. We illustrate the use of EXISTS—and NOT

EXISTS—with some examples. First, we forundate Query 10 in an alternative form that uses EXISTS. This is shown as Q16B:

```
Q16D: SELECT E.FNAME, E.LNAME

FROM EMPLOYEE AS E

WHERE EXISTS (SELECT ·

FROM DEPENDENT

WHERE E.SSN=ESSN AND E.SEX=SEX

AND E.FNAME=DEPENDENT NAME);
```

EXISTS and NOT FXISTS(Q) recents TRCE if there are its taples in the result of network of the result query independent of the second query references the SSS, UNARE, and SCX attributes of the period relation from the octor query. We can think of Q16B as follows. For each period relation relation from the octor query, which retrieves all a second raples with the same social second romber, sex, and name as the second rople; if at least one rople FXISTS in the result of the nested query, then select that second raples in general, EX ETS(Q) returns TRUE if there is at least one taples in the result of the nested query Q, and it returns FALSE otherwise. On the other hand, NoT FXISTS(Q) returns TRUE if there are no suples in the result of nested query Q, and it returns FALSE exherition. Note that an extension, we allost are not suples in the result of nested query Q, and it returns FALSE exherition.

#### QUERY 6

Retrieve the names of employees who have no dependents.

```
Q6: SELECT FNAME, LNAME

FROM EMPLOYEE

WHERE NOT EXISTS (SELECT ·

FROM DEPENDENT

WHERE SSN=ESSN);
```

In Qb, the correlated nested query retrieves ad **DEPENDENT** tuples related to a particular entriciple fit noise error, the **EEECOPE** tuple is solicited. We can explain Q6 as follows For each exercise, tuple, the correlated nested query sclears all dependents are related to the value matches the operate sol, if the result is employ, no dependents are related to the employee, so we select that DEEDED tople and retrieve or **FAMO**, and DAPE.

#### QUERY 7

List the names of managers who have at least one dependent

| Q7: | SELECT | FNAME, LNA | ME      |           |
|-----|--------|------------|---------|-----------|
|     | FROM   | EMPLOYEE   |         |           |
|     | WHERE  | EXIST\$    | (SELECT | •         |
|     |        |            | FROM    | DEPENDENT |
|     |        |            | WHERE   | SSN=ESSN) |

### AND EXISTS (SELECT ' FROM DEPARTMENT WHERE SSN=MGRSSN);

One way to write this query is shown in Q7, where we specify two nested correlated queries the first selects all consistent tuples related to an twittyte, and the second selects all newswer tuples managed by the process. If at least one of the first and at least one of the second entry of the second selects all send on twitter tuples managed by the process. If at least one of the first and at least one of the second selects all send on the second entry of the first and the second selects and the second entry of the first and the second entry of the second entry of the second entry of the second entry of the first one of the second entry of the first entry of the second entry of the first entry of the first entry of the second entry of the first 
Query 3 ("Retrieve the name of each employee who works on all the projects actualled by department runnible 5," see Section 5.5.33 can be stated using 2.515Ts and 3.511As7s in SQL systems. There are two options. The first is to use the well-known set theory transformation that (S13200 FAISS S2) is logically equivalent to (S2 eXCFFT S7) is eagily. This option is shown as Q3A.

| Q3A; | SELECT | FNAME, LNAME<br>EMPLOYEE<br>NOT EXISTS |           |
|------|--------|----------------------------------------|-----------|
|      | FROM   |                                        |           |
|      | WHERE  |                                        |           |
|      | (      | (SELECT                                | PNUMBER   |
|      |        | FROM                                   | PROJECT   |
|      |        | WHERE                                  | DNUM-5)   |
|      | EXCEPT |                                        |           |
|      |        | (SELECT                                | PNO       |
|      |        | FROM                                   | WORKS_ON  |
|      |        | WHERE                                  | SSN=ESSN) |
|      |        |                                        |           |

In Q3A, the first subquery (which is not enrichted) selects all projects controlled by department 5, and the second subquery (which is correlated) selects all projects that the purculat employee being considered works on the the set difference of the first subquery USUS (EV) IPT1 the second subquery is empty in means that the employee works on all the projects and is hence selected.

j;

The second optima is shown as Q3P. Notice that we need two-level resting in Q3D and that this formulation is quite a bit more complex than Q3D which used the CONTAINS comparison operator, and Q3A, which uses NOT EVISTS and COUPT. However, CONTAINS a not put of SQL and not all relational systems have the EXCEPT operator even though it spart of SQL 90.

```
Q3B: SELECT LNAME, FNAME
FROM EMPLOYEE
```

Recoll that EXCEPT is the set difference operator
```
WHERE NOT EXISTS
(SELECT
FROM
 WORKS ON B
WHERE
 (B PNO IN
 (SELECT PNUMBER
 FROM
 PROJECT
 WHERE
 DNUM=5()
 AND
 NOT EXISTS (SELECT)
 FROM
 WORKS ON C
 WHERE
 C,ESSN=SSN
 AND
 C.PNO=B.PNO));
```

In Q3B, the outer nexted query selects any wood\_ow (B) toples whose we is of a project controlled by department S, if there is not a wood\_ow (C) rople with the same we and the same sets as that of the everyte uple under consideration in the outer query link such tuple exists, we select the two out ruple. The form of Q3B matches the follow or rephrasing of Query 4: Select each employee such that there does not exist a project controlled by department 5 that the employee does not work or 1t corresponds to the way we wrote this query in tuple terration calculus as Section 6.6.

There is another SQL function, CNIQUE(Q), which returns TRUE if there are no doplicate tuples in the result of query Q; intervise, if returns FALSE. This can be used to resc whether the result of a nested query is a set or a multiset.

## 8.5.5 Explicit Sets and Renaming of Attributes in SQL

We have seen several queries with a nested query in the WHRRE clause. It is also possible to use an **explicit set of values** in the WHERE clause, rather than a nested query. Such a st is enclosed in parentheses in SQL.

#### QUERY 17

Betrieve the social security numbers of all employees who work on project numbers 1, 2, or 3

```
Q17: SELECT DISTINCTESSN
FROM WORKS_ON
WHERE PNO IN (1, 2, 3);
```

In SQL, it is possible to remain any attribute that appears in the result of a query by adding the quadment AS followed by the desired new mime. Hence, the AS construct carbs used to show both attribute and relation mannes, and it can be used in both the SELCT and IFCAL danses. For example, QSA shows have query QS can be slightly changed to remeve the last manne at each employee and his or her supervisor, while remaining the resulting

archere nonice as species \_kave and successive\_kave. The new manage will oppose as reform headers in the query result.

Q6A: SELECT ELLNAME AS EMPLOYEE\_NAME, S.LNAME AS SUPERVISOR\_NAME FROM EMPLOYEE AS E, EMPLOYEE AS S WHERE E.SUPERSSN=S SSN;

## 8.5.6 Joined Tables in SQL

The concept of a joined table (or joined relation) was incorporated into  $\leq \leq$ , to permit uses to specify a table resulting from a join operation in the *FR* of classe of a query. This construct may be easier to comprehend than intering together all the select and join conditions in the Withfeld clause. For example, consider query Q1, which retrieves the name and iddress of every employee who works for the 'Research' department. It may be easier first to specify the join of the ENCORE and DEPARTMENT relations, and then to select the Joined toples and uttributes. This can be written in SQL as in Q1A:

#### QLA: SELECT FNAME, LNAME. ADDRESS FROM (EMPLOYEE JOIN DEPARTMENT ON DNO=DNUMBER) WHERE DNAME='Research';

The HOM clouse in Q1A contains a single *j*-field table. The artributes of such a table and the artributes of the first table, use over, followed by all the artributes of the second table, a surger. The concept of a joined table also allows the user to specify different types of join, such as NA1.08A1.00N and correct types of O12 at jOIN. In a NA108A1, join are detected an implicit equiper coredition for two relations *R* and *S*, no pair condition to specify do not be allowed by all the architecture of an implicit equiper coredition for two relations *R* and *S*, no pair condition to specified an implicit equiper coredition for two of carefustes with the solid none from *R* and *S* is created. Each such pair of attributes is included only once in the resulting relation (see Section 6.4.3).

If the cames of the ion attributes are not the same to the base relations, it is possible to remain the attributes so that they match, and then to apply NATURAL JOIN. In this case, the AS construct can be used to rename a relation and all its attributes in the FROM date. This is illustrated in QTB, where the **DEPATION** relation is relation as DEPT and its attributes are renamed as DEPT and the implicit join constitution for this NATURAL TORAL TOR

#### Q18: SELECT FNAME, LNAME, ADDRESS

```
FROM (EMPLOYEE NATURAL JOIN
(DEPARTMENT AS DEPT (DNAMÉ, DNO. MSSN, MSDATE)))
WHERE DNAME='Research;
```

The detault type of your in a journed table is in inner join, where a topic is included in the reals only of a matching topic exists in the other relation. For example, in query

QSA: only employees that have a supervisor are included in the result; an exercise tuple whose value for supervisor supervisor are included. If the user requires that all employees be included, an OUTER JOIN must be used explicitly (see Section 6.4.3 for the definition of OUTER CON). In SQL, this is handled by explicitly specifying the OUTER JOIN in a joined table, as illustrated in Q8B:

#### Q8B: SELECT E.LNAME AS EMPLOYEE\_NAME, S.LNAME AS SUPERVISOR NAME FROM (EMPLOYEE AS E LEFT OUTER JOIN EMPLOYEE AS S ON E.SUPERSSN=S.SSN):

The options available for specifying joined tables in SQU reliade (WNEUTON (same as JOINI, LEFT OCTERTION, RIGHT OCTERTION, and FCE, OUTERTION. In the latter three options, the keyword OOTER may be omitted. If the join attributes have the same name, one may also specify the natural join variation of outer joins by usine the keyword NATURAL before the operation (for example, NATURAL LEFT OCTERTION). The keyword DROSS JOIN is used to specify the Cartesian product operation (see Section 6.2.2), although this should be used only with the otness care because it generates all possible tuple combinations.

It is also possible to new join specifications; that is, one of the tables in a join may result be a joined table. This is illustrated by QPA, which is a different way of specifying query Q2, using the concept of a joined table:

Q2A: SELECT PNUMBER, DNUM, LNAME, ADDRESS, BDATE FROM ((PROJECT JOIN DEPARTMENT ON DNUM=DNUMBEA) JOIN EMPLOYEE ON MGRSSN=SSN) WHERE PLOCATION-'Stafford':

# 8.5.7 Aggregate Functions in SQL

In Section 6.4.1, we introduced the current of an aggregate function as a relational opertion. Because grouping and aggregation are inquired in many database applications, sqt has features that incorporate these concepts. A number of built in functions exots COUNI. SUM, MAN, MIN, and AVG-<sup>16</sup> The CSENT fonction returns the number of tuples of values is specified in a queue. The functions SUM, MAN, MIN, and AVG are applied to a serior mutti-er of numeric values and return, respectively, the same maximum value, number of tuples of values and average (mean) of those values. These functions can be used in the SETE clause cells a DAVSNO clause (which we antroduce later). The functions KAN and MIN can also be used with attributes that have manumentic domaids at the domain values have a total orderage annuagione another.<sup>11</sup> We illustrate the use of those functions with example queries.

10 Addaminal approprie functions for more solvanized statistic dicatalation have been added ar sQL99. 11. Torol order means that for any two values in the domain, () can be determined that one appears before the other in the domined order, for example, 19(16) (1991) and (1991) \$78500 domains lower total ordering: on their values as do alphabetic strates.

#### QUERY 19

find the sum of the salaries of all conployees, the maximum salary, the minimum salary and the exchage salary.

#### Q19: SELECT SUM (SALARY) MAX (SALARY), MIN (SALARY), AVG (SALARY) FROM EMPLOYEE;

It we want to get the preceding function values for coupleyees of a specific generation—say, the Research' deportment—we can write Query 20, where the suproves uples the restricted by the WHERE clause to those employees who work for the "Research" deportment.

#### QUERY 20

Find the sum of the salaries of all employees of the 'Research' department, as well as the maximum solary, the imminium salary, and the average salary in this department.

```
Q20: SELECT SUM (SALARY), MAX (SALARY), MIN (SALARY),
AVG (SALARY)
FROM (EMPLOYEE JOIN DEPARTMENT ON DNO=DNUMBER)
WHERE DNAME='Research';
```

#### QUERIES 21 AND 22

Serieve the rotal number of employees in the company (Q21) and the number of engloyees in the 'Besearch' department (Q22).

| Q21: | SELECT | COUNT (*)                         |
|------|--------|-----------------------------------|
|      | FROM   | EMPLOYEE.                         |
| Q22: | SELECT | COUNT (*)                         |
|      | FROM   | EMPLOYEE, DEPARTMENT              |
|      | WHERE  | DNO=DNUMBER AND DNAME='Research': |

Here the asterisk (\*) refers to the wave (toples), so  $G(3, S)^{(*)}$  returns the number of raw in the result of the query. We may also use the c3.00ST function to count values in a tumm rather than ruples, as in the next example.

#### QUERY 23

Count the number of distinct solary values in the database.

```
(24: SELECT COUNT (DISTINCT SALARY)
FROM EMPLOYEE;
```

If we write CODNT(SACARD) instead of CODMT(DIST NCT SALASYT in Q23, then duplicate values will not be claminated. However, any raples with NCT for SALASY and nor be counted. In general, NCC values are discarded when aggregate functions at applied to a particular colorm (arribute).

The preceding examples summarize a whole values: (Q19, Q21, Q22) or a selected subset of toples (Q20, Q22), and hence all produce single toples or single values. They illustrate how functions are applied to retrieve a summary value or summary heple from the database. These functions can also be used in selection conditions involving nested queries. We can specify a correlated nested query with an aggregate function, and then as the rested query in the wHERE choice of an outer query. For example, to retrieve the restes of all employees who have two or more dependents (Query 5), we can write the following of all employees who have two or more dependents (Query 5), we can write the following the following the set of a correlated previous of an outer query 5).

```
Q5: SELECT LNAME, FNAME

FROM EMPLOYEE

WHERE (SELECT COUNT (*)

FROM DEPENDENT

WHERE SSN=ESSN) >- 2;
```

The correlated nested query counts the number of dependents that each employee has **r** this is greater than or equal to two, the employee taple is selected.

## 8.5.8 Grouping: The GROUP BY and HAVING Clauses

In many cases we want to apply the aggregate functions as adaptites of caples in a relation where the subgroups are based on some arminute values. For example, we may wanted find the average salary of employees in each department or the number of employees who work on each project. In these cases we need to **partition** the relation into nonoverlapping subsets (or groups) of tuples. Each group (partition) will consist of the tuples that have the same value of some attribute(s), valled the grouping attribute(s). We can then upply the function to each such group independently SQL has a GROUP BY clause for this parpose. The GROUP BY clause specifies the grouping attribute(s), which should also appear of 2 SULUET clause, so that the value resulting bort applying each aggregate function to group at inples appears along with the value of the grouping attribute(s).

#### QUERY 24

For each department, retrieve the department number, the number of employees in the department, and their overage salary.

```
Q24: SELECT ONO. COUNT (1) AVG (SALARY)
FROM EMPLOYEE
GROUP BY ONO;
```

In Q24, the CPLOSE: uples are partitioned into groups-s each group having the same value for the grouping attribute 590. The COUNT and AVG functions are applied to each

scalings up of tuples. Notice that the SEERT clause includes only the grouping attribute and the functions to be applied on each group of tuples. Figure 6.0a illustrates here grouping works on Q24: it also shows the result of Q24.

If NOLLS exist in the grouping attribute, then a separate group is created for all tuples with a NULL value in the grouping attribute. For example, if the CPROSE table had some replet that had NULL for the grouping attribute 360, there would be a separate group for fixed tuples in the result of Q24.

#### QUERY 25

For each project, remove the project number, the project name, and the number of employees who work on that project.

| Q25: | SELECT   | PNUMBER, PNAME, COUNT (*) |
|------|----------|---------------------------|
|      | FROM     | PROJECT, WOHKS_ON         |
|      | WHERE    | PNUMBER=PNO               |
|      | GROUP BY | PNUMBER, PNAME.           |

Q25 shows how we can use a jour condition in corporation with GROUP Fit In this case, the grouping and functions are applied ages, the puting of the two relations. Sometimes we want to retrieve the values of these functions call for groups that satisfy orthocordinate. For example, suppose that we want to modify Query 25 so that only populations which can appear in complexies appear in the result. SU, provides a HAVING class, which can appear in complexies appear in the result. SU, provides a HAVING class, which can appear in complexies appear in the result. SU, provides a HAVING class, which can appear in complexies appear in the result. SU, provides a HAVING class, which can appear in complexies appear in the result of the solution of the groups of ruples associated with each value of the groups due satisfy the condition are retrieved in the result of the solery. This is illustrated by Query 26

#### QUERY 26

for each project on which more than two employees work, retrieve the project number, the project number of employees who work on the project.

| Q26: | SELECT   | PNUMBÉR, PNAME, COUNT (') |
|------|----------|---------------------------|
|      | FROM     | PROJECT, WORKS ON         |
|      | WHERE    | PNUMBER=PNO               |
|      | GROUP BY | PNUMBER, PNAME            |
|      | HAVING   | COUNT (") > 2:            |

Notice that, while selection conditions in the WHECs clause limit the tights to which functions are applied, the HAVING clause serves to choose while groups. Figure 6.05 illustrates do use of HAVING and displays the result of Q26. (8)

| FNAME        | MINIT    | LNAME    | <u>957</u> | · · · | SALARY | SUPERSSA                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 0.0 |               |        |           |                |
|--------------|----------|----------|------------|-------|--------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----|---------------|--------|-----------|----------------|
|              | - » -    | S-P      | 103/66789  |       | 30000  | 3293145935                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | . s | •             |        |           |                |
| huke         |          | 7/213    | 12000555   |       | 4.000  | 5280005500                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | 5   |               | المبيح | Constra   | aWC ( Set &D.4 |
| Runett       | <        | NAME AND | 100344-004 | _     | .WHIC  | 30040575                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 1.5 |               | L      | (Anit)(1  | ·              |
| C LARGE      | -4       | E geal   | 460450453  | · · • | 25000  | 300-449535                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | 5   | .! ·⊷         | 5      | د         | 1988-0         |
| · *** o      | _ ·      | 7etaja 🗌 | 2092/777   | -     | 294820 | ALC: NO. OF ALC: N | 1.0 | ) _~ <b>F</b> | b      |           | <u></u>        |
| Jan Hereiter | з        | Name     | .67e54321  |       | 43200  | 66 <b>86</b> 566938                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 4   | > 17 <b>=</b> | 1      | ٦         | 85000          |
| Abriad       | ¥        | Jaxer    | 967977847  | _     | 2500C  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 4   |               |        | 63aa      | 1/194          |
| A1755        | <u> </u> | 3019     | 389307535  |       | 7.000  | <u></u>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 1   | 13.4          |        | -result o | (LL24,         |

Grouping surveyor tuples by the value of over

ðυ

| FNAME                 | C <u>NUM</u> BEP |   | 658N        | PN0  | HOURS     | ]      |                         |
|-----------------------|------------------|---|-------------|------|-----------|--------|-------------------------|
| Prouction             | — , <u> </u>     |   | 129450709   | •    | 101       | 1      |                         |
| (headed)              |                  |   | 453453453   | ı .  | 204       |        |                         |
| PR& OF                | 5                |   | 12:466759   | 2    | 75        | i N    |                         |
| Aca.c//               | 2                |   | 43252450    | 2    | 204       |        |                         |
| Challet*              | j <u> </u>       |   | 1334495555  | 2    | 10.0      |        | These overpsions in the |
| Pros.c2               |                  |   | 077834424   | 1    | - Co-     |        | eelected op heelekstwa  |
| Phak242               | 1 2              |   | 11:04-95555 | 1 3  | 1250      |        | ann dhian (d' 1926      |
| Concurrence           | 10               | • | 3034455755  | 10   | 109       | ľ`     |                         |
| Concidentation        | 1U               |   | 995587777   | 10   | 104       | ۰<br>۱ |                         |
| Compatibility and the | 10               |   | 91/201/257  | 10   | 5<        |        |                         |
| Ассеранием            | :0               |   | 784-885     | - 20 | 100       | ' :    |                         |
| Hearphizeron          | 0                |   | 94/92-4-91  | 20   | 150       |        |                         |
| Beckpartakon          | · 7              |   | 585667755   | - 22 | 2.0       |        |                         |
| Newtoenable           | v                | 1 | 987987367   | æ    | <u>60</u> | 1.     |                         |
| Newpeneliz            | 30               |   | 90205-021   | -30  | 204       | 7      |                         |
| Fember (#5            |                  |   | 99968/227   | 30   | 5.00      | ]      |                         |

After applying the weak clause but heline applying reverse



After apprying the www.so clause cond-box.

FIGURE 8.6. Results of GROUP RY and HAVING Fail Q24, (b) Q25

#### QUERY 27

For each project, retrieve the project number, the project name, and the number of employees from department 5 who work on the project.

```
 Q27: SELECT
 PNUMBER. PNAME, COUNT (*)

 FROM
 PROJECT, WORKS_ON, EMPLOYEE

 WHERE
 PNUMBER=PNO AND SSN=ESSN AND DNO=5

 GROUP BY
 PNUMBER
```

Here we restrict the topics in the relation (and hence the tuples in each group) to those this study the condition specified in the WHERE clouse—trainely, that they work in department number 5. Nonce that we must be extra chieful when two different conditions apply the to the function in the SHLUCT clause and mother to the function in the HAVENG datas). For example, suppose that we want to count the total number of emphyses whose starts are department, but only for departments where their law species that we want to count the total number of emphyses whose starts are exceed \$40,000 in each department, but only for departments where their law emphyses were then here emphyses were the condition (\$5, 59 > 40000) applies only to the Opt (\$6, 50 ) function if the select of clause. Suppose that we write the following mearing near squares

```
SELECTDNAME, COUNT (*)FROMDEPARTMENT, EMPLOYEEWHEREDNUMBER=DNO AND SALARY>40000GROUP BYDNAMEHAVINGCOUNT (*) > 5:
```

This is incorrect because it will select only departments that have more than five exployees the each only nove than \$40,000. The rule is that the WHERE close is executed this resolution individual tuples; the HAVING close is applied later, to select individual tuples; the HAVING close is applied later, to select individual tuples; the ruples are already restricted to comployees who can more than \$40,000, before the function in the HAVING clause is applied. One way to write this cars correctly is to use a nested query, as shown in Query 28.

#### QUERY 28

For each department that has more than live employees, retrieve the department number and the number of its employees which are making more than \$40,020.

| Q48: | SELECT | DNUMBER | t, COUNT (*) |                |
|------|--------|---------|--------------|----------------|
|      | FROM   | DEPARTM | ENT, EMPLOYE | E              |
|      | WHERE  | DNUMBER | I=DNO AND SA | LARY>40000 AND |
|      |        | DNO IN  | (SELECT      | DNO            |
|      |        |         | FROM         | EMPLOYEE       |
|      |        |         | GROUP BY     | DNO            |
|      |        |         | HAVING       | COUNT (") > 5) |
|      |        |         |              |                |

GROUP BY DNUMBER;

## 8.5.9 Discussion and Summary of SQL Queries

A query in SQL can consist of up to aix choises, but only the first ravi—sELECT and EEOV —are mandatory. The clauses are specified in the following order, with the clauses between square brackets [111,1] being optimist:

```
SELECT SATISMUTE AND FUNCTION ESTS

FROM STABLE LISTS

[WHERE SCONDITIONS]

[GROUP BY SCROUTING ATTRIBUTE[S]S]

[HAVING SCROUP CONDITIONS]

[ORDER BY SATISMUCTE LISTS]:
```

The 960 CC clause lists the arcribures or functions to be introved. The FROM clause specifies all relations (rables) medded in the energy mehoding joined relations, but not those in netod queries. The W-0FRE clause specifies the conditions for selections of tight from these relations, including join conditions if needed. G0.201 BY specifies grouping attributes, whereas HAVING specifies a condition on the group period selected rather that on the individual taples. The built-in aggregate functions (CCCN1, SCM, MIN, MAX, and WCG are used in computerion with grouping, but they can also be applied to all the selected tuples in a query without a -080CC PY clause. Finally, ORGUR PV specifies an order for displaying the result of a query.

A query is evoluated conceptually<sup>15</sup> by first applying the ECM clause (to identify ill tables involved in the query or to insterialize any joined tables), followed by the WHESE clause, and then by (ECCC) PD and HAVESC. Conceptually, OROR BL is applied at the end to sore the doerv result. If none of the last three clauses (CROOT BL, HAVESC, and CROEP) are specified, we can chick e inceptually of a query as being executed as follows. For each combination of tuples—one from each of the relations specified in the filest clause. In TROF, place the values of the attributes specified in the EECCT clause from this tople combination in the result of the query O course, this is not an efficient way to implement the query in a real system, and each CROES has special query optimization matrices to decide on an execution plan that is efficient. We discuss query processing and optimizations in Chaptery 15 and 26.

In general, there are numerous ways to specify the same query in SQL. This flexibility in specifying queries has advantages and disadvantages. The main advantage is that uses can choose the technique with which they are most countertable when specifying a genbut example, many queries may be specified with joint conditions in the WHEIE clause, at by using joined relations in the -3-OM clause, or with some form of nested queries and the IN comparison operator. Some users may be more comfortable with one approach, whereas others may be more comfortable with unsider. From the programmer's and the

<sup>12.</sup> The receipt order or query synthation is implementation dependent, this is just a way to conceptically view a query invite to contractly for an late it.

ssients point of view regording query optimization, it is generally preferable to write a query with as butle resumg and implied ordering as possible.

The disadvantage of borong numerous ways of specifying the same query is that this matching the user, who may not know which rechnique to use to specify part cular apes of queries. Another problem is that it muy be more efficient to execute a query specified in one way than the same query specified in an abecauty wat Ideally, this doubt not be the case. The DBMS should process the same query in the same way tigniles of how the query is specified. But this is quite difficult in practice, since each tRMs last different methods for processing queries specified in different ways. Thus, as idenoital barden on the user is to determine which of the alternative specifications is the rest efficient. Ideally, the user should worry only also a specifying the query correctly, hi is the responsibility of the DMMS to execute the query efficiently. In practice, however, it leps it the user is availe of which types of constituets in a query are more expensive to process than others (see Chapter 16).

# 8.6 INSERT, DELETE, AND UPDATE STATEMENTS IN SQL

 $|p|<_{\rm T}$  , three communds can be used to modify the database: INSERT, DELETE, and UPO (F. We discuss each of these or turn.

## 8.6.1 The INSERT Command

Inus simplest form, to FRT is used to add a single tuple to intribution. We must specify the relation name and a lost of values for the tuple. The values should be listed in the same edge in which the corresponding outphores were specified in the URE VEE TAINE conread for example, to add a new tuple to the operate relation shown in Figure 5.5 and specified in the SREATE UNBLE operation - command in figure 8.1, we can use U1:

U1: INSERT INTO EMPLOYEE
 VALUES ('Richard , 'K', 'Marini', '653298653', '1962-12-30', '98
 Oak Forest,Katy,TX', 'M', 37000, '987654321', 4);

A second form of the INSERT statement allows the user to specify explicit attribute nines if at correspond to the values provided in the INSERT command. This is useful it a iddimension many attributes but only in tew of those attributes are assugated values to the new ople. However, the values must include all attributes with NOT NULL specification and include the value. Autributes with NOT allowed or DEFAULT values into the ones that an belieft on the second to the rate of the toget for a new EMPLORE for whom we know only the toget for a new EMPLORE for whom we know only the toget, toget, toget, toget, the attributes, we can use ULA:

```
 VALUES ('Fichard', 'Marini', 4, 653298653');
```

Attributes not specified in ULA are set to their 1995/02.7 or to 90011, and the colors are letted in the same order as the attributes are listed in the INSERT command itself. It is also possible to insert into a relation initiaple toples separated by commas in a single INSERT command. The outplote values forming each tiple are enclosed in gatentheses.

A DBMs that fully onglements SQL-99 should support and enforce all the integrity constraints that can be specified in the PEL However, some DPMss do not incorporate all the constraints in order to maintain the efficiency of the 1901s and because of the complexity of entersting all constraints. If a system does not support some constraint—are referential integrity—the users or programmers must enforce the constraint. For example, if we essue the command in U2 on the database shown in Figure 5.6, a DBMs not supporting referential integrity will do the insertion even though no MPASTMAT tuple exists in the database with DOMER = 2. It is the responsibility of the user to check that any such constraints arises check one not anglemented by the DBMs are not violated. However, the DFMs must implement checks to enforce all the SQL integrity constraints is support. A DFMs enforcing NOT NULL will reject an 1985BCT command in which an arched declared in the NOT NULL will reject a volue; for example, U2A would be special because no SN value is provided.

UZ: INSERT INTO EMPLOYEE (FNAME, LNAME, SSN, DNO)
 VALUES ('Robert', 'Hatcher', '980760540', 2);
 (5.13) a seise stad if as face of a bistory for above in accurate business.

(\* U2 is rejected if referential integrity checking is provided by dbms \*)

U2A: INSERT INTO FMPLOYEE (FNAME, LNAME, DNO) VALUES ('Robert', Hatcher', 5);

If U2A is rejected if not null checking is provided by dbms \*)

A catation of the INSERT contrand inserts multiple toples into a relation in conjunction with creating the relation and loading it with the relation of a query. Enexample, to create a temporary table that has the name, number of engloyees, and total sidenes for each department, we can write the statements in USA and USPs

#### USA: CREATE TABLE DEPTS\_INFO

|      | (DEPT NAME  | VARCHAR(15).                       |
|------|-------------|------------------------------------|
|      | NO_OF EMPS  | INTEGER,                           |
|      | TOTAL SAL   | INTEGER);                          |
| UBB: | INSERT INTO | DEPTS_INFO (DEPT_NAME, NO_OF_EMPS, |
|      |             | TOTAL SAL)                         |
|      | SELECT      | DNAME. COUNT (*), SUM (SALARY)     |
|      | FROM        | (DEPARTMENT JOIN EMPLOYEE ON       |
|      |             | DNUMBER-DNO)                       |
|      | GROUP BY    | DNAME:                             |

A table CEPTS\_UNET - creatice by URA and is loaded with the summary information retrieved from the database by the query in URB. We can now query reprisitive as we

word any other relation, when we do not need it any note, we can remove it by using the 0-00 TABLE command. Notice that the tEPTS that table may not be up to date; that is, if we update either the DEWETNESS of the EVENTREE relations after using 113B, the intermedium in DEE s\_then becomes outdated. We have to create a view (see Section 9.23 to keep such a table up to date.

## 8.6.2 The DELETE Command

The PHE E-command removes ruples from a relation, it includes a WHEE-clause, similar to that used in an SQL query, to select the tuples to be deleted. Tuples are explicitly detect from only one table at a time. However, the deletion may propagate to tuples in whet relations if defended engened actions are specified in the referential integrity conmants of the DOF (see Section 8.2.2).<sup>11</sup> Depending on the number of tuples selected by the condition in the WHERE clause, tero, one, or several tuples can be deleted by a single (EESE command: A missing WHERE clause specifies that all turbles in the relation are to be deleted, however, the table remains in the database as an empty table.<sup>11</sup> The DELETE commands in U4A to U4D, if applied independently to the database of Figure 5.6, will delete tero one, four, and all tuples, respectively, from the ENECEFF relation:

| U4A: DELETE FROM | EMPLOY  | rEE        |                    |
|------------------|---------|------------|--------------------|
| WHERE            | LNAME:  | ='Brown';  |                    |
| U4B: DELETE FROM | EMPLO   | 'EE        |                    |
| WHERE            | SSN='12 | 23456789"; |                    |
| U4C: DELETE FROM | EMPLO   | 'EE        |                    |
| WHERE            | DNO IN  | (SELECT    | DNUMBER            |
|                  |         | FROM       | DEPARTMENT         |
|                  |         | WHERE      | DNAME='Research'); |

U4D: DELETE FROM EMPLOYEE;

## 8.6.3 The UPDATE Command

The CPDATE command is used to modify attribute values of one or more selected ruples. As in the DFI ETE command, a WHFUE clause in the DFP VTE command selects the tuples incernodified from a single relation. However, updating a primary key value may prepatate to the tenergickey values of tuples in other relations if such a selection to general oction aspectied in the referential integrity constraints of the DFP (see Section 5.2.2). As addinard SET clause in the OPPATE command specifies the attributes to be medified and

<sup>(2)</sup> Other actions can be sufficient calls applied through (figures (see Section 24.1) and other realizing).

<sup>4</sup> We nast use the 1980 (1780). Compared to represente table delipition (see Section 5.5.1)

then new values. For example, to change the location and controlline department number of project number 10 to 'Bellane' and b, respectively, we use US:

```
US: UPDATE PROJECT
SET PLOCATION = 'Bollaire', DNUM = 5
WHERE PNUMBER=10;
```

Several tuples can be moduled with a single UPDATE command. An example is to give all employees in the 'Research' department a 10 percent rule in salary, as shown in US. In this request, the modified salaer value depends on the original solarit value in each ruple, so two references to the (water attribute are needed. In the SFI clause, the relevance to the salary antribute on the right refers to the old salaer value *before multipletania*, and the one on the left refers to the per Service value offer modification.

| U6: | UPDATE | EMPLOYEE          |                    |
|-----|--------|-------------------|--------------------|
|     | SET    | SALARY = SALARY * | 1.1                |
|     | WHERE  | ONO IN (SELECT    | DNUMBER            |
|     |        | FROM              | DEPARTMENT         |
|     |        | WHERE             | DNAME='Research'): |

It is also possible to specify NUIE or DEFAULT as the new attribute value. Nonce that each UPI-ATE command explicitly refers to a single relation only. To moduly multiple relations, we must usue several UPI-ATE commands.

# 8.7 ADDITIONAL FEATURES OF SQL

SQ: has a monther of addigional features that we have not described in this chapter of discuss clowberg in the book. These are as follows:

- SQ: has rho (apability to specify more general constraints, called assistants, using dx V3EATE ASSECTION spirature). This is described in Section 9.1.
- sett has language constructs for specifying views, also known as virtual tables, using the CREATE VIEW statement. Views are derived from the base tables declared through the CREATE TABLE statement, and are discussed in Section 9.2.
- SQL has several different rechniques for writing programs in various programming languages that can include SQL statisticity to access one or more databases. They include embedded (and dynamic) SQL SQL(0.2) (Coll Language Interface) and reprodecessor Of BC (Open Data Base Connectivity), and SQL(SMI (Proetam Stered Modules). We discuss the differences among these techniques in Section 9.3, then discus each technique in Sections 9.4 through 9.6. We also discuss how to access 5.0, databases through the Java programming language using 1090 and SQL.
- Each commercial RYONIS will have, in addition to the SQL commands, a set of commands for specifying physical database design parameters, file structures for relatence and access paths such by indexes. We called these commands a storage definition les-

gate (357) in Chapter 2. Earlier versions of SQL had commands for creating indexes, but these were removed from the language because they were not at the conceptual schema level (see Chapter 2).

- Synthis transaction control commands. These are used to specify units of database processing for concurrency control and recovery purposes. We discuss these commands in Chapter 17 after we discuss the concept of transactions in more detail.
- SQL has language constructs for specifying the gamming and recoking of provleges to users. Freedeges replicable correspond to the right to use certain SQL commands to users certain volations. Each gelation is assigned an owner, and either the owner of the 103 sufficient grant to selected users the privilege to use an SQL statement such as SELFCT INSERT, DELSTR, or CIP WIGHT to decess the relation. In addition, the OPA staff can grant the privileges to create schemas, tables, or views to certain users. These SQL commands—called GRANT and REVORE safe discussed in Chapter 23 where we discuss database security and authoritation.
- set has language constructs for creating triggers. These are generally referred to as active database techniques, since they specify actions that are automatically triggered by events such as database updates. We discuss these features in Section 24.1, where we discuss active database concepts.
- Set has incorporated many features from object-oriented models to have more powering capabilities, leading to enhanced relational systems known as object-relational. Capabilities such as creating complex-structured attributes (also called nested relations), specifying abstract data repes (called UDTs or user-defined types) for attributes and tables, creating object identifiers for referencing tuples, and specifying operations on types are discussed in Chapter 22.
- opti and relational databases can interact with new technologies such as X01, (rXtended Markup Language) see Chapter 26) and CLAP (Chi Line Analytical Proossing for Data Warehouses, see Chapter 28).

# 8.8 SUMMARY

In this chapter we presented the SQL database language. This language or cariations of it have been implemented as interfaces to many commercial relational 1983(s), including Grade, 1983 062 and SQL/98. Microsoft's SQL Server and ACCESS, 18630(s), and 553 952. The original version of SQL was implemented in the experimental 19838 called \$55750(R) which was developed at 498 Research (SQL)) designed to be a comprehensive language that includes attachments for data definition, queries, opdotes, view definition, and constraint specification. We discussed many of these in separate sections of the data terms that are described elsewhere in the bask. On emphasis was on the SQL 599 standard

Table 8.2 annuments the syntax (centracture) of various 5QL statements. This among is not mean to be comprehensive ner to describe every possible SQL constructs when, it is means to serve as a quick reference to the major types of constructs available.

#### TABLE B.2. SUMMARY OF SQL SYNTAX

```
CREATE TABLE < (able names (< column names < column type> [< attribute constraint>]
 1. <column_same> <column_syne> [<attribute_constraint>] }
 |<rable constraint> {.}||
INCOTABLE <1able name>
ALTER TABLE A(to <column name> <column; type>
season loismision) matterbute list >
FROM ((<alias>) (<joined table>) ((<alias>) (<)rable name> (<alias>) (<)rable>) (
WHERE <condition>1
[ORCUP BY <grouping attributes>[UAVING <group selection condition> []]
[ORDER BY <column name> [<order>] {; <column name> [<order>] }];
<a tribute http://www.commoname>...sharetion.e(()pisting() <column name>!*))}
 ((<coloren symme> | <function>(([USTANCT] <coloren name>1*)) |))
<grouping artributes>::= <column name>(, <column name>)
<order>c= (ASC | DESC);
INSERT INTO Krable name > [] Krolinon name >], Krolumn name >}]]
(VALUES { <constant value> , [<constant value>])[,{<constant value>],<constant value>])]
l<seiecr scarement>);
DELETE FROM Scable name >
WHERE <selection condition>|
UPDATE Stable name?
SET < column name>=< velue expression > [, < column nemy >=< value expression >]
WHERE ASSERTION CONDITION >1
UNEARE [UNIQUE] INDEX I Adsided name A
ON stable name> (scolumn mane> [sorder>] f . scolumn name> [sorder>]]).
ICLUSTER!
DECREMENTER Sindler name?
CREATE VIEW | Sprice come > [] | Scolumn name > [] | Scolumn name > []]
AS Select statement?
DRCEV.EW Kynew name?
```

<sup>1</sup> The List (wire compared) are not part of standard SQL2.

n eq. We use 4NF notation, where non-terminal sembols are shown in angled brackets < ... > optional parts are shown in square brackets |...|, repetitions are shown in braces  $y \dots |... |$ , and alternatives are shown in parentheses  $\{..., 1, ..., 1, ... \}$ .

## **Review Questions**

- 54. How do the relations (tables) in %21 differ from the relations defined formally in Chapter 57 Discuss the other differences in terminology. Why does \$00 allow Juplicate tuples in a table of in a guery result?
- 8.2. List the data types that are allowed for SQL attributes.
- 8.3. Hew does SQL allow implementation of the entity integrity and referential integrity constraints described in Chapter 5? What about referential imggered actions?
- 54 Describe the six clauses in the syntax of an SQL query, and show what type of constructs can be specified in each of the six clauses. Which of the six clauses are required and which are optimized?
- 53 Describe conceptually how an SQL query will be executed by specifying the conceptual order of executing each of the six clauses.
- 56 Discuss how NULLS are recared in companion operators in SQL flow are NULLS maned when aggregate functions are applied in an SQL query? How are NULLS beated if they exist in grouping attributes?

## Exercises

- 2.7. Consider the dotabase shown in Figure 1.2, whose schemans shown in Figure 2.1. What use the referential integrity constraints that should hold on the schema? Write appropriate SQL000, statements to define the database.
- 3.3. Repeat Exercise 6.7, but use the ATELINE database schema of Figure 5.8.
- 5.9. Consider the criticate relational database is being of Figure 6.1.2. Choose the appropriate action (reject, cascade, ser to null, series default) for each referential integrity constraint, both for default of a inferenced tuple, and for ignare of a primary key attribute value in a referenced tuple. Justify your choices.
- \$10 Write appropriate 5021000 statements for declaring the consist relational database schema of Figure 6-12. Specify appropriate keys and referential triggered actions.
- Write SQL queries for the cleases database queries given in Exercise 6-18.
- 8.12. How can the key and foreign key constraints be enforced by the D9005 Is the enforcement technique you suggest difficult to implement? Can the constraint shocks be executed efficiently when updates are applied to the database?
- 5.13. Specify the queries of Exercise 6.16 in SQL Show the result of each query if it is applied to the consider database of Figure 5.6.
- 8.14. Specify the following additional queries on the database of Figure 3.5 in sQL. Show the query results if each query is applied to the database of Figure 5.6.

Efficiently entrated setting and setting the setting and the setting of the setti

- For each department whose average employee salary is more than \$32,200, retrieve the department name and the number of employees working for that department.
- b. Suppose that we want the number of nucle employees in each department rather than all employees (as in Exercise 8.14a). Can we specify this query in SQL'Why is why nucl.
- 8.15. Specify the updates of Exercise 5.10, using the SQL update commands
- 8.16. Specify the following quoties in SQU on the database schema of Figure 2.2.
  - a. Remove the names of all senior sudents majoring in '05' (comparet science)
  - Retracte the names of all courses raught by Professor King in 1998 and 1999.
  - For each section raught by Professor King, retrieve the course number, sensiton year and number of students who nook the section.
  - d. Remove the name and transcript of each senior student (Class = 5) majoring in CS. A transcript includes course name, course number, credit hours, senster, year, and goods for each course completed by the student.
  - Retrieve the nomes and mojor departments of all straight-A students (students who have a grade of A in all their coarses).
  - Retrieve the names and major departments of all students who do not have a grade of A in any of their courses.
- 8.17. Write SQL update statements to est the following on the database schema shown in figure 1.2.
  - Insert a new student, <",christon', 25, 1, 'MATH >, in the database.
  - b. Change the class of student 'Smith' to 2,
  - Insert a new course, <'Knowledge Engineering', 'Cs4390', 3, 'Cs'>.
  - d. Delete the record for the student whose name is 'Smith' and whose student number is 17.
- Sperify the queries and updates of Exercises 6.17 and 5.11, which refer to the station database (see Figure 5.8), in sqt.
- 6.19. a. Using a relational database schema for your database application.
  - Declary your relations, using the 800-000 a
  - Specify a number of queries in 2(0) that are needed by your durabase application
  - d. Based on your expected use of the database, choose some artributes that should have indexes specified on them.
  - Implement your database of your have a 1994S that supports \$23.
- 8.30 Specify the answers to Exercises 0.19 through 6.21 and Exercise 6.23 in Sci-

## Selected Bibliography

The SQL language, originally named SEQUEL was based on the language SQUARE (Spenfying Queries as Relational Expressions), described by Boyce et al. (1975). The syntaxial SQUARE was incided into SEQUEL (Chamberline and Royce 1974) and then into SEQUEL 2 (Chamberline et al. 1976), on which SQUES based. The original implementation at SEQUEL was detee at time Research. San Jose, California.

Reisner (1977) describes a human factors evaluation of SEQ.91 in which she found that users have some difficulty with specifying join conditions and grouping correctly. Due (1984b) contains a critique of the SQL language that points out its sueneths and someonings. Date and Darwen (1993) describes SQL2, ANS (1996) ontlines the original SQL standard, and ANSI (1992) describes the SQL2 standard. Various vendor monoils describe the characteristics of SQL as implemented on D02, SQL28. Oracle, (SORES, ISOOSEX, and other commercial COMS products. Meitors and Sunon (1993) is a comprelative treatment of SQL2. Horison (1992) discusses some of the problems related to relrection integrity and propagation of updates in SQL2.

The question of view updates is addressed by Dayal and Bernstein (1978), Keller (1962), and Langerak (1990), among others. View nuplementation is discussed in Blake leveral. (1989). Negri et al. (1991) describes formal semantics of SQL querties.





# More SQL: Assertions, Views, and Programming Techniques

In the previous chapter, we described several aspects of the SQL language, the standard for relational databases. We described the SQL statements for data deterition, schema (positisue)of queries, and updates. We also described how common constraints such as key and referential integrity are specified. In this chapter, we present several additional aspects of SQL We start in Section 9.1 by describing the CREATE ASSERT with statement, which alkes the specification of more general constraints on the database. Then, it, Section S.L.we describe the SQL facilities for defining views on the database. Then, it, Section school deviced tables because they present the user with what appear to be tables; however, the reformation in those tables is derived from previously defined tables.

The next several sections of this chapter docust various techniques for accessing database fem programs. Most database access in practical situations is through software payrains that implement database applications. This software is usually developed in a general purpose programming language such as (AVA, COBO), or C/C+-. Recall from Sector 2.3.1 that when database statements are uncluded in a program, the general-timple programming language is culted the now language, whereas the database linguage (soft in our case) is called the language. In some cases, special database linguage (soft in our case) is called the database and analysis language, and developed specificable for verticing database applications. Although many of these were developed as research prototypes, some notable database trendspiration prototypes, some notable database trendspiration planguages have ordespread use, such as ODACLES (1)/SOE (Drogramming Language)sol.).

We send our presentation of database programming in Section 9.3 with an overview of the different rechtyppes developed for accessing a database from programs. Then, n Serijon 9.4, we discuss the rules for embedding SQL statements into a general-purple programming language, generally known as embedded \$3.5. This section also high discusses dynamic pott, in which queries can be dynamically constructed at minime, ad presents the basics of the inquirmation of mahedded oct, that was developed specifically for the programming language IAVA. In Section 9.5, we discuss the rechnique known a NGL/CityCall Level Intectace), in which a library of procedures and functions is provide to accessing the database. Various sets of library functions have been proposed. The  $S_{\rm e}S_{\rm e}$ set of functions is the one given in the SQL standard. Another library of functions is 0000 (Open Data Base Connection). We do not describe (0090 because it is considered to be the produces or to SQLA LL A third library of functions—which we do describe—to 300, this was developed specifically for accessing databases from JAVA. Finally, in Section 94. we discuss S(2) (M. (Persistent Stored Medules), which is a pair of the SOI standard the allows program modules, procedures and functions—to be stored by the Pasts as accessed through SQL Section 9.7 summarizes the elsiptet.

# 9.1 SPECIFYING GENERAL CONSTRAINTS AS ASSERTIONS

In SQL overs can specify growth constraints—three that do not tall rate any of the categories described in Section 8.2-1 via declarative assertions, come the CREATE ASSERTION statement of the 100. Each assertion is given a constraint name and is specified via a cadevier similar to the WHERE clause of an SQL query. For example, to specify the constraint du "the value, of an employee must not be greater than the salary of the monoger of the depresent that the employee works for "in SQL we can write the following assertion:

```
CREATE ASSERTION SALARY_CONSTRAINT
CHECK { NOT EXISTS
(SELECT :
FROM EMPLOYEE E, EMPLOYEE M. DEPARTMENT D
WHERE E.SALARY>M.SALARY AND
E.DNO=D.DNUMBER AND
D.MGRSSN=M.SSN) }:
```

The constraint name server constraint is followed by the keyword of PECK, which a followed by a condition in parentheses that must hold true on every doublise state for the assertion to be satisfied. The constraint name can be used later to refer to the constrain or to modify or drop to. The DBSS is responsible for ensuing that the condition is no violated. Any WHERE clause condition can be used, but many constraints can be specified using the EXISTs and NOT EXISTS style of EQL conditions. Whenever some tuples in the database cause the condition of an ASSERTION statement to evaluate to FAESE, the constraint is violated. The constraint is satisfied by a database state it no combination? Spley in that database state violates the constraint. The basic technique for writing such assertions is to specify a query that selects and reple-the rotate the densed condition. By encluding this query inside a NOTEX STS clause, in assertion will specify that the result of this query must be empty. Thus, the assertion is violated if the result of the query is not empty. In our example, the query selects all employees whose columns are greater than the salary as the manager of their department. If the result of the query is not empty, the assertion is violated.

Note that the CHECK clause and constraint condition can also be used to specify tensions on attributes and domains (see Section 8.2.1) and on tuples (see Section 8.4.4). A major difference between CREATE ASSENTION and the other two is that the CHECK clauses on attributes domains, and tuples are checked in 9.2, only other explosion explained. Hence, constraint checking can be implemented more efficiently by the PWP in these cases. The schema designer should use CHECK on intributes, domains, and tuples and the only be rioked by domains, and sugles only when he or she is sure that the constraint can only be rioked by distance and sugles only when he or she is sure that the constraint can only be rioked by distance and sugles only when he or she is sure that the constraint can only be rioked by distance and sugles only when he or she is sure that the constraint can only be rioked by distance assessing of tubles. On the other hand, the schema designer should use CHECK on attributes, domains, or uples, so that checks are implemented more claused by the DBMS.

Another statement related to CREATE ASSIGNOS' in SQL is CREATE TRIGGER, but trigges are used in a different way. In many cases it is convenient to specify the type of attain to be taken when certain events occur and when certain conditions are satisfied. Rulia than offering users only the option of aborting on operation that causes a valacon—is with CREATE ASSERTION—the DRVF should noke other options available. For example, it may be useful to specify a condition that, if violated, causes some user to lenderined of the violation. A manager may want to be informed if an employee's travel reprise exceed a certain hunt by receiving a message whenever this occurs. The action that the PRVS must take in this case is to send an appropriate message to that user. The innerity a specific stored procedure or triggering other updates. The CREATE TRIGGER travel to used to implement such actions in SQL A trigger specifies an event (such as a parentle database optime or triggering other updates. The cREATE TRIGGER travel a certain such actions in SQL A trigger specifies an event (such as a parentle database optime or reignering), and an action. The action is to be received allocatedly if the condition is sensified when the event occurs. We discuss larger in dual in Section, 24.1 when we describe active database.

# 9.2 VIEWS (VIRTUAL TABLES) IN SQL

In his section we introduce the concept of a view in SQL. We then show how views are gened, and we discuss the problem of updating a view, and how a view can be implemented by the DSMS.

## 9.2.1 Concept of a View in SQL

Aview in SQU terminology is a single table that is derived from other tables <sup>1</sup> These other ntk-sould be base tables of previously defined views. A view does not necessarily exist in

<sup>1</sup> A new new proventies the terror receasing or lumited than the terror as two waters don't hapters hand a successed on whether the water relations

physical form, it is considered a virtual table, in contrast to have tables, whose tuples are actually stored in the database. This limits the possible update operations that can be applied to views, but it does not provide any limitations on querying a view.

We can think of a view as a way of specifying a table that we need to belentee frequently, even though it may not exact physically. For example, in Figure 5.5 we now frequently issue queries that retrieve the employee nome and the project names that the employee works on. Rather than having to specify the join of the regional work 66 and evolutionable-every time we issue that goest, we can define a view that is a result of date joins. We can then usue queries on the view, which are specified as single-table retrieval tables then us retrieval- involving two joins on three tables. We call the function, movelow and explore tables the defining tables of the view.

## 9.2.2 Specification of Views in SQL

In SQL the command to specify a crew is **CREATE VIEW**. The crew is go en a (virial) table ratio (or view name), a list of attribute names, and a query to specify the contents of the view. If none of the view attributes results from applying functions or attributes operations, we do not have to specify attribute names for the view, since they would be the same as the names of the attributes of the detering tables in the detailt case. The view in VI and V2 create varial tables whose schemas are illustrated in Figure 9.1 when applied to the database schema of Figure 5.5.

| V1: CRI | EATE VIEW | WORKS ON1                                 |
|---------|-----------|-------------------------------------------|
| AS      | SELECT    | FNAME, LNAME, PNAME, HOURS                |
|         | FROM      | EMPLOYEE, PROJECT, WORKS, ON              |
|         | WHERE     | SSN=ESSN AND PNO=PNUMBER;                 |
| V2: CRI | EATE VIEW | DEPT_INFC(DEPT_NAME.NO_OF_EMPS,TOTAL_SAL) |
| AS      | SELECT    | DNAME, COUNT (*), SUM (SALARY)            |
|         | FROM      | DEPARTMENT, EMPLOYEE                      |
|         | WHERE     | DNUMBER=DNO                               |
|         | GROUP BY  | ONAME:                                    |

| WORKS_( | DIN1  |        |       |
|---------|-------|--------|-------|
|         | LNAME | PNIAME | HOUAS |

DEPT\_INFO

| DEPT NAME     | NO OF EMPS | TUTAL SAL  |
|---------------|------------|------------|
| COLL 1 TARAGE |            | i ione,ane |

FIGURE 9.1 Two views specified on the database schema of Figure 5.5.

In V., we dol not specify any new attribute names to the view many out faithough we could have) in this case, without interpret the particle of the view attributes from the driving tables exercised, entities to Wiew V2 explicitly specified new attributes new attribute range for the view 6601–1806, using a one-to-one correspondence between the attributes get fad in the CREATE VIEW clouse and those specified in the SELECT clause of the query particles.

We can now specify SQL queries on a view—or partial table—in the same way we bent carries involving base tables. For example, to retrieve the last name and first name efail employees who work on 'ProjectX', we can ottlice the works' on? view and specify the oper as in QVI:

```
QV1: SELECT FNAME, LNAME
FROM WORKS_ON1
WHERE PNAME-'ProjectX';
```

The tone query would require the specification of two joins if specified on the base relation role of the mon-advantages of a view is to samplify the specification of certain quere. Vers are also used as a security and authorization mechanism. See Chapter 2-9.

As owns supposed to be always ignor date; if we modify the tuples in the lose tables in when the view is defined, the view must automatically reflect these changes. Elence, the view is not realized at the time of tien, definition but rather of the time we specify a gray on the view. It is the responsibility of the 1000's and not the user to make sure that the view is up to date.

It we do not need a view any more, we can use the **DROP VIEW** command to dispose out For example, to get tid of the view V1, we can use the SQL statement in V1A:

```
VIA: DROP VIEW WORKS ONT:
```

## 9.2.3 View Implementation and View Update

The problem of chickently implementing a view for querying is complex. Two mean appear bechave been suggested. One strategy called query modification, modves made forgate view query initial query on the underlying base follows. For example, the query QVI would be our anativally modified to the following query by the DWE:

```
 SELECT
 FNAME. LNAME

 FROM
 EMPLOYEE, PROJECT, WORKS ON

 WHERE
 SSN=ESSN AND PNO-PNUMBER

 AND PNAME='ProjectX';
```

The disadvartage of this approach is that it is motificant for view defined vio couples queries that are time-consuming to execute, especially it multiple queries are replied to the view without a short period of time. The other strategy called view interalization, involves physically creating a temporary view table when the view efficit queries and keeping that table on the assumption that other dortes on the view wilfullo. In this case, an efficient strategy for automatically updating the view table when the base tables are updated must be developed in order to keep the view up to fate. Techniques using the concept of incremental update base been developed for the purpose, where it is determined what new tuples must be inserted, deleted, comodified a constant direction with table when a change is applied to one of the detuning base tables. The view is generally kept as long as it is being queried. If the view is not queried for a certain period of time, the system may then automatically remove the physical view table and recompute it from scratch when lating queries reference the view.

Updating of views is complicated and can be ambiguous. In general, an update or a view defined in a single table without any aggregate functions can be mapped to an update so the underlying base table under certain conditions. For a view involving joins, an update operation may be mapped to update operations on the underlying base relations in makek avies. To illustrate potential problems with updating a view defined on multiple table consider the mass out view and suppose that we issue the commond to update the use attribute of 'John Suppo' from 'ProductX' to 'ProductY'. This view opdate is shown in Update (N.)

```
UV1: UPDATE WORKS_ON1
SET PNAME = ProductY'
WHERE LNAME='Smith' AND FNAME=John AND
PNAME='ProductX';
```

This query can be mapped into several updates on the base relations to give the desired update effection the view. Two possible updates: (a) and (b), on the base relation corresponding to UV), are shown here

| (8)  | UPDATE | WORKS_ON                         |         |                                 |
|------|--------|----------------------------------|---------|---------------------------------|
|      | SET    | PNO ≈                            | (SELECT | PNUMBER                         |
|      |        |                                  | FROM    | PROJECT                         |
|      |        |                                  | WHERE   | PNAME='ProductY')               |
|      | WHERE  | ESSN IN                          | (SELECT | SSN                             |
|      |        |                                  | FROM    | EMPLOYEE                        |
|      |        | AND                              | WHERE   | LNAME='Smith' AND FNAME='John') |
|      |        | PNÓ =                            | (SELECT | PNUMBER                         |
|      |        |                                  | FROM    | PROJECT                         |
|      |        |                                  | WHERE   | PNAME='ProductX');              |
| сња- |        | DECISION RET DIMANE - (BraduetV) |         |                                 |

(b): UPDATE PROJECT SET PNAME = Productly

WHERE PNAME = 'ProductX';

Update (a) relates 'John Smith' to the 'Fraduct'' related tuple in place of the 'Product'' related update. However, (b) would also give the desired update effect on the view but it accomplishes this by changing the name of the 'Product'' tuple in the evolution to 'Product''. It is quite onlikely that the

as who specified the view update UV1 wants the quality to be interpreted as in (b), since it also has the side effect of changing all the view tuples with energy = . Product X'.

Some view updates may nor make much sense: for example, modifying the monotoamicae of the DEM\_ING view does not make sense because 1014.\_54c is defined to be the arrow the individual couplinger solaries. This request is shown as UV2:

```
UV2: UPDATE DEPT_INFO
SET TOTAL_SAL=100000
WHERE DNAME='Research':
```

A large number of applaces on the underlying base relations can satisfy this view opdate.

A view update is feasible when only one possible sprint on the base relations can wroup ish the desired update effect on the view. Whenever on update on the view can be mapped to more than one update on the underlying base relations, we must have a sensitive procedure for choosing the desired update. Some researchers have developed include for choosing the most likely update, while other researchers prefer to have the ustrolesses the desired update morphing curring view determines.

Insumory, we can make the following observations:

- A view with a single defining table is updatable if the view attributes contain the priimagikey of the base relation, as well as all attributes with the NPT NOUT constraint shird out have default values specified.
- Viewsitefined on multiple tables using joins are generally not updatible
- Views defined using enoughing and aggregate functions are not inplatable

In SQL the clause WITH CHECK OPTION must be added at the end of the view dominant is view is to be igakited. This allows the system to check for view updatability addo plan in execution strategy for view updates.

# 9.3 DATABASE PROGRAMMING: ISSUES AND TECHNIQUES

We now turn for attention to the techniques that have been developed for accessing drabots from programs and, in particular to the issue of how to access bou database for application programs. Our presentation of SQL so far has to used on the binguage manners for particular to generations of SQL so far has to used on the binguage manners for particular to generations of SQL so far has to used on the binguage manners for particular to generation of SQL so far has to used on the binguage manners for particulation programs. Our presentation of SQL so far has to used on the binguage manners for particular barrance from schema definition and conversing specification of views. More database systems have an interactive interface where these SQL commands can be typed directly into a nonitor or lumpit to the database system. For example, in a computer system where the  $(65010, 6000 \times is installed, the command SQL p. 115 will start the interface interface. Theorem on type SQL commands or queries directly over several lines, ended by a semi-source of the three key (that is "<math>|seres"$ ). A termitively, a **file of normands can be** created and executed through the interactive interface by typing **@stilenames**. The system of execute the commands written in the net and display the results, it any.

The interactive interface is quite convenient for schema and constraint creations for occasional ad hoc queries. However, the majority of database interactions in process are executed through programs that have been catchilly designed and tested. They programs are generally known as **application programs** or **database applications**, and ar used as caused convectors by the end neers, as the ussed in Section 1.4.3. Another sequences use of database programming is to access a database through an application program that implements a Web interface, for example, for making arritice reservations department score purchases. In fact, the vost majority of Web electronic commenapplications include some database access communds.

In this section, we first give an overview of the main approaches to dealer programming. Then we discuss some of the problems that occur when trying to access database from a general purpose programming longuage, and discuss the typical sequence of commands for interacting with a database from a software program.

## 9.3.1 Approaches to Database Programming

Several rechniques exist for including database interactions in application programs. The main approaches for database programming are the following:

- 1. Embedding database statements on a general propose programming language. In the approach, database statements are embedded into the host programming language, but they are identified by a special prefix. For example, the prefix is embedded FQL is the string EVE (SQL), which proceeds all SQL commands in shot language program.<sup>2</sup> A precompiler or preprocessor first consiste source gragan code to identify database statements and extract them for processing by the FRE They are replaced in the program by forterion calls to the DBMS-generated code.
- 2. Using a library of database factories. A library of functions is made available in the host programming language for database calls. For example, there could a functions to connect to indicabase, execute a query, execute an ipdate, and som. The outrial database query and update commands, and any other necessary into matical, are included as parameters in the function calls. This approach process what is known as an Application Programming Interface (API) for accessing database from application programs.
- 3. Designing a frand-new torgaage. A database programming fanguage is designed from scratch to be compatible with the database model and query language. Addtornal programming structures such as loops and conditional statements are adde to the database language to convert it must a full-fieldged programming language.

In practice, the first two approaches are note common, since more application at already written in general-purgose programming languages but require some database access. The drift approache is more appropriate for applications that have intense database interaction. One of the main problems with the first two approaches is more problems.

Other prefixes are concurries used, but this is the most connected are

## 9.3.2 Impedance Mismatch

Impedate mismatch is the term used to refer to the problems that occur because of difforeics between the database model and the programming language model. For example, the product relational model has three more constructs, attributes and their data types, toples fixeoids), and tables (sets or nulliters of records). The first problem that may occur is due the data types of the programming language differ from the attribute data types in the data model. Hence, it is necessary to have a binding for each host programming language that specifies for each attribute type the compatible programming language types. It is necessary to have a binding for each programming language because different language lave different data types; for example, the data types, it will data types, it and table in the soft data types.

Another problem occurs because the results of most queries are sets or multisets of tipls, and each upple is formed of a sequence of attribute values. In the program, it is iterationary to access the individual data values within individual tuples for printing or precising. Hence, a binding is needed to map the giveny result data structure, which is a table, in an appropriate data structure in the programming language. A nuccharastic is needed to be power the tuples in a query result in order to access a single tuple at a time and to extract individual values (rome the ruple. A cursor or iterator variable is used to be power the tuples in a query result in order to access a single tuple at a time and to extract individual values (rome the ruple, A cursor or iterator variable is used to be power the tuples in a query result. Individual values within each tuple are rypically excited into distince program variables of the appropriate type.

(updance instructly is less of a problem when a special database programming bigging a designed that uses the same data model and data types as the database model. One estimple of such a language is ORACLE's PJ/SQL. For object databases, the object data model (we Chapter 20) is quite similar to the data model of the 14XA programming big age, so the impedance mismatch is grattly reduced when CVXA is used as the host language for accessing a JAX 4-compatible object database. Several database programming languages have been implemented as research polytotypes (see bibliographic notes).

## 9.3.3 Typical Sequence of Interaction in Database Programming

When a programmer or settivare engineer writes a program that requires objects to a database, it sente common for the program to be mining on our computer system while the database a ratalled on another. Recall from Section 2.5 that a common architectore for database aces is the chent/server model, where a **client program** handles the logic of a software appliration, but includes some calls to one or more **database servers** to access or update the data.<sup>1</sup> When writing such a program, a gamman sequence of interaction is the fullowing:

 When the client program requires access to a particular database, the program must first evablish or open a connection to the database server. Typically, this

<sup>3.</sup> Sour discussed of Section 2.5. There are constructional firstering interimentations to keep and a use grow plot on will assume a two oppression as prior in procedure betw. We discuss additional contanor of these as late, trues in Chapter 25.

involves specifying the Internet address (Diot) of the machine where the database server is located, plos providing a login acrosmy name and password for database access.

- Once the connection is established, the program can interact with the database by submitting queries, updates, and other database commands. In general, nontypes of SQL statements can be included in an application program.
- 3 When the program no longer needs access to a particular database, it should ierroiniti or close the connection to the database.

A program can access multiple databases in needoc. In some database programming approaches, only one connection can be active at a time, whereas it other approaches multiple connections can be established at the same time.

In the next three spectrons, we discuss examples of each of the three approaches to database programming. Section 9.4 describes how SQL is embedded into a programming language. Section 9.5 discusses how function calls are used to necess the database, and Section 9.6 discusses an extension to SQL called SQL/PSM that allows general-purper programming constructs for detuning modules (procedures and functions) that are used within the database system.<sup>9</sup>

# 9.4 EMBEDDED SQL, DYNAMIC SQL, AND SQLJ

## 9.4.1 Retrieving Single Tuples with Embedded SQL

In this section, we give an overview of how SQL statements can be embedded in a general purpose programming language such as *P*, ADA, ODBOL, or CASUAL. The programming language such as *P*, ADA, ODBOL, or CASUAL. The programming language such as *P*, ADA, ODBOL, or CASUAL. The programming language such as *P*, ADA, ODBOL, or CASUAL. The programming language such as *P*, statements—including data or constant definitions, queries, updates, or crew definitions—can be embedded in a bust language program. An embedded SQL statement is distinguished from programming language sute ments by prefixing it with the keywords ENFC SQL so that a preprocessor (or precompiler) can separate embedded SQL statements from the law language code. The SQL statement is distinguished for a preprocessor (or precompiler) can be reminated by a semicology (1) or a matching (NESEXC).

To illustrate the concepts of embedded SQL we will use it as the has programming language. Within an embedded SQL command, we may refer to specially declared C program variables. These are called **shared variables** because they are used in both the C program and the embedded SQL statements. Shared variables are prefixed by a cision (a) when they appear in an SQL statement. This distinguishes program variable manes from the names of database schema constructs such is arterbutes and relations. It also allows program variables to have the same manes is attribute names, since they are distinguishable by the "" prefix in the SQL statement.

Although SQL'PSST is non-considered to be a fold-fledged programming long age, it of a-match a reproducement-purpose programming constructs—such as loops and conditional setuctures—carbs incorportied into SQ.

Names of database schema constructs — such as attributes and relations —can only be used within the SQL commands, but shared program variables can be used elsewhere in the C program variout the ":" prefix.

Suppose that we want to write C programs to process the Gausse database of Equip 55. We need to declare program variables to motch the types of the database attributes fracthe program will process. The programmer can choose the names of the program variables, they may or may not have names that are identical to their corresponding antibutes. We will use the C program variables declared in Figure 9.2 for all our examples, as the of 1 show C program segments surhout variable declarations. Shared variables are declared within a declare section in the program, as shown in Figure 9.2 fluxes 1 through 7)<sup>3</sup> A lew of the common bindings of C types to FQE types are as follows. The SQE types 1816-98, SMALLINT, REAC, and COOREE are imapped to the C types long, short, fileat, and pool it, respectively. Fixed-length and varying- ength strongs (CHAR)[]. VARCHAR[\$] wide can be mapped to attays of characters (char [3+1], variater [1+1]) in C that are one concert longer than the SQE type, because strings in C are terminated by a "AC" (coll) departed which is not part of the character string itself."

Nonce that the only embedded equ commands in Figure 9.2 are lines I and 7. which tell the precomption to take note of the C voriable names between BLON DECLARE and ENDITEDANT because they can be included in embedded SQL statements—as long as they as preceded by a color (3. Lines 2 through 5 are regular C program declarations. The C pregram similaries declated in lines 2 through 5 correspond to the attributes of the ENDIER and DEMARK AND

```
int loop ;
63
D.
 EXEC SOL BEGIN DECLARE SECTION ;
2\rangle
 varchar dname [16], tname [16], lname [16], address [31] ;
£,
 char ssn [10], bdate [11], sex [2], minit [2] ;
e
 float salary, raise ;
52
 int dno, dnumber ;
 ing SQLCODE ; char SQLSTATE [6] ;
€;
\mathbf{n}
 EXEC SOL END DECLARE SECTION :
```

**05086** 9.2 ic program variables used in the embedded sou examples F1 and E2.

 $\lambda$  We are law memory in our code segments for case reference, these numbers are not part of the smallede.

K901 strings can also be mapped to chare? Types mul-

Connecting to the Database. The sol command for establishing connection to a database has the following form:

#### CONNECT TO <server name> AS <connection name> AUTHORIZATION <user account name and password> ,

In general, since a user or **progr**am can access several database servers, several connections can be established, but only one connection can be acrive at any point in time. This programmer or user can use the <connection name > to change from the connective connection to a different one by using the following command.

#### SET CONNECTION <connection name> ;

Once a connetion is no longer needed, it can be terminated by the following command:

#### DISCONNECT <connection name> ;

In the examples in this chapter, we assume that the appropriate connection haalready been established to the cosmo database, and that it is the committy active connection

Communicating between the Program and the DEAN Using SQLEDDE and SQLSTATE. The row special communication variables that are used by the DEES to communicate exception or error conditions to the program are SQLODE and PQLSTATE. The SQLCODE variable shown in Figure 9.2 is an integer variable. After each shifting command is executed, the DEES returns a value in SQLODE. A value of 0 indicates that the statement was executed successfully by the DEESS. If SQLCODE > C (or, more specifically, it SQLCODE < 1.20), this indicates that no in the data (records) are available in a query result. If SQLCODE < 0, this indicates some error has occurred, in some systems—for example, in the DEESS = SQLCODE is a field of a record structure called SQLCA (SQL communication profiles) is of its referenced as SQLCA (SQL communication profiles) is referenced as SQLCA (SQL communication profiles) is referenced as SQLCA (SQL communication profiles) is referenced as SQLCA (SQL communication profiles).

EXEC SQL include SQLCA :

In later versions of the SQL standard, a communication variable colled SQUSTATE was added, which is a statue of five characters. A value of "WWW" in SQL TATE indicates no error or exceptions other values indicate corours errors or exceptions. For example, "WWW" indicates 'no more data? when using SQLSTATE. Currently, both SQLSTATE and SQLCCPE are available in the SQL standard. Many of the error and exception, order Artianed # SQLSTATE are supposed to be standard. Many of the error and exception, order Artianed # SQLSTATE are supposed to be standard. Many of the error and exception, order Artianed # SQLSTATE are supposed to be standardized for all SQL versions and platforms,<sup>1</sup> whereas ducedes returned in SQLCOPE are not -conductized but are defined by the TIMS version Hence, in a generally before to use SQLSTATE, because this makes error framiling in the application programs independent of a particular DSMS. As an exercise, the reader should revenue the examples given later up this chapter using SQLSTATE instead of SQLCOF.

 In periodiar, A3.81AIT coversistance with the characters Caliborgh 4 or A through H are up posed to be standardized, whereas their values can be impleated and in a defined. Example of Embedded 501 Programming. Our first example in: illustrate orbidded 503 programming is a repeating program segment (loop) that reads a social vertice number of an employee and prints can some information from the corresponding intro- record in the database. The C program code is shown as program segment E1 in figure 9.3. The program reads (inputs) a social security number value and then remeves its record in the total social security number value and then remeves its record in the UNTO clause (line 5) specifies the program veriables in the INTO clause ariticity values from the database are remeved <) in gram veriables in the INTO clause as projects with a colon (1), as we discussed earlier.

Lue 7 in EU flustrates the continuities on between the database and the program dwoch the special caroble SQL(MDE) If the value returned by the 18845 in SQL(MDE) is 0, the previous statement was executed without errors or exception conditions. Line 7 checks the and assumes that of an error occurred, it was because no 199, over tuple existed with the gven social seconts number, in therefore outputs a message to that effect (line 8).

In EL a single riple is selected by the encodded SQL query, that is why we are able to asen its numbure values directly to vi program variables in the INTO clause in line 5. In great, an SQL query can retrieve many tuples. In that case, the vi program will typically gothoegh the retrieved tuples and process them one at a rinte. A classifier used to allow indent-a-time processing by the host language program. We describe cursors next

## 9.4.2 Retrieving Multiple Tuples with Embedded SQL Using Cursors

We can think of a **cursor** as a printer that points to a single tople (now) from the result of a greathat retrieves multiple toples. The consortis declared when the SQL query command schedard on the program. Later in the program, an OPEN CORSOR command te ches the greatesile from the database and sets the cursor to a position fegore the first row in the

```
//Program Segment E1;
0)
 logg = 1 ;
 while (loop) [
n
 prompt("Enter a Social Security Number: ", ssm) ;
2)
Ð
 EXEC SOL
 select FNAME, MINIT, LNAME, ADDRESS, SALARY
4)
5)
 into :fname, :minit, .lname, :address, :salary
9
 from EMPLOYEE where SSN = :ssn :
 if (SQLCODE == 0) printf(fname, minit, lname, address, salary)
23
 else printf("Social Security Number does not exist: ", ssn) ;
9
 prompt("More Social Security Numbers (enter 1 for Yes. 0 for No): ", loop) ;
3)
1Ú)
 2
```

**nGult** 9.3. Program segment F1, all, program segment with embedded sQL.

result of the query. This becomes the current row for the cursor. Subsequently, FEUH communds are i-sued in the program each FEICH moves the cursor to the next row in the result of the query making it dis current row and copyrig its attribute values into the *i* (bost language) program variables specified in the FEICH command by an INTO clouse. The cursor variable is basically on **iterator** that iterates (loops) over the tuple in the query result—one tuple at 0 time. This is similar to traditional record-attaining file processing.

To determine when all the tuples in the result of the query have been processed the communication variable SQL ODE (or, alternatively, SQLSTATE) is checked. If a 1.TCH command is issued that results in theoring the cursor past the last tuple in the result of the query, a positive value 40000005. > 31 is returned in SQLCODE, indicating that no data (tuple) was bound (or the wring "C2OCO" is returned in SQLSTATE). The programmer uses this to remaindre a loop over the tuples in the query result. In general, numerous cursos can be opened at the same time. A CLOSE CORSOR command is issued to indicate that we are done with programmer the query associated with that runsor.

An example of using cursors is shown to Figure 9.4, where a cursor talled YMF is declared in line 4. We assume that appropriate C program variables have been declared as in Figure 9.2. The gragium segment in E2 reads (inputs) a department name (line 0), retrieves its department number (lines 1 to 3), and then retrieves the employees who

```
//Program Segment 22:
 prompt("Enter the Department Name: ". dname) ;
0)
1)
 EXEC SOL
2)
 select DNLMBER into :dnumber
 from DEPARTMENT where DNAME - ;dname ;
33
43
 EXEC SQL DECLARE EMP CURSOR FOR
52
 select SSN, FNAME, MINIT, LNAME, SALARY
6)
 from EMPLOYEE where DNO = tdnumber
7}
 FOR UPDATE OF SALARY ;
8)
 EXEC SQL OPEN EMP ;
9)
 EXEC SQL FETCH from EMP into (ssn. ;fname, ;minit, :lname, ;salary ;
10)
 while (SQLCODE == 0) [
 printf("Employee name is:", fname, minit, lname)
11)
12)
 prompt("Enter the raise amount: ", raise) ;
13)
 EXEC SQL
14)
 update EMPLOYEE
15)
 set SALARY = SALARY + Cranke
16)
 where CURRENT OF EMP :
 EXEC SOL FETCH from EMP into tosh, "fname, iminit, "name, isalary ;
17)
18)
 EXEC SOL CLOSE EMP ;
190
```

FIGURE 9.4 Program segment L2, a C program segment that uses cursors with embedded set for update purposes vick in that department via a curvet. A loop (lines 10 to 18) then iterates over each improves record, one at a time, and prints the employee name. The program then reack a time mount for that employee (line 12) and apdates the employee's salary in the datase by the ruse amount three 14 to 16).

When a cursor is defined for rows that are in be modified (updated), we must add the class FOR UPDATE OF in the cursor declaration and list the barnes of any attributes that will a aplitudly the program. This is illustrated in line 7 of code segment E2. If rows are to be defeed, the keywords EOR UPDATE must be added without specifying ony an safes. In the embedded UPDATE for DELETER command, the condition WHERE CUBENT OF Secursor name? Specifies that the cursor table of E2.

Notice that declaring a cursor and associating it with a query (lines 4 through 7 in E2) does not execute the query, the query is executed only when the CPEN <cursor name> commond (line 6) is executed. Also name that there is no need to include the KR UPDATE OF clouse in line 7 of E2 if the results of the query are to be used for remeat propose only (no update or delete).

Several options can be specified when declarang a cursor. The general form of a cursor decirition is as follows:

#### DECLARE <cursor name> [ INSENSITIVE ] [ SCROLL ] CURSOR

[WITH HOLO ] FOR <query specification>

ORDER BY <ordering specification> ]

[FOR READ ONLY | FOR UPDATE [ OF <attribute list> ] ] ;

We already briefly discussed the regions based in the last fine. The default is that the pervision retrieval purposes (FOR KEAUCONTED) if some of the tuples in the query result at no be updated, we need to specify FOR 0.000 ATE OF satisfactor basis and by the amburs that may be updated. If some roples are no be deleted, we need to specify SOR 0.000 specify some of deleted, we need to specify SOR 0.000 specify some to be deleted, we need to specify SOR 0.000 specify some to be deleted, we need to specify SOR 0.000 specify specify some to be deleted.

When the optional keyword SCROLL is specified in a cursor declaration, it is possible argument the cursor in other ways than for parely sequential across. A fetch orientation (a) by added in the FEULI command, whose using cursor of NEXT, DROR, FIGST, DST, APSOLL FL, and REEATTELL In the latter two commands, a must evaluate roam megnicular that specifies an absolute tayle position of a tuple position relative to the current cursor position, respectively. The default derive commands, a most evaluate roam examples, is NEXT. The lefth orientation allows the programmer to move the cursor second the tuples on the spectroely testih with greater flexibility providing random occess by System a access in reverse order. When SCROLL is specified on the cursor, the cursor, the cursor and of TFH command is as follows, with the parts in square brockets being option is

#### FETCH [ [ <letch orientation> ] FROM ] <cursor name> INTO <fetch target list> :

The ORDER FY charge orders the tuples to that the SEOTH command will fetch them in the specified order. It is specified in a similar mathematic to the corresponding clause for \$1 gueries (see Section 8.4.6). The last two options when declaring a cursor (SEDSTIVE and W Tri Hell)) refer to transaction characteristics of database programs, which we discuss in Chapter 17.

## 9.4.3 Specifying Queries at Runtime Using Dynamic SQL

In the previous examples, the embedded SQ, cherics were written as part of the host pargram source code. Hence, any time we want to write a different query, we must arise a new program, and go through all the steps involved (compiling, debugging, resting, and to orth. In some cases, it is convenient to write a program that can exercise different sQ, queries or updates (or other operations) dynamically as mounts. For example, we may want to write a program that accepts an SQL query typed from the monitor, executes it, and displays its result, such as the interactive interface available for most relational OFMS. Another example is when a user-finendly interface generates SQL queries dynamically for the user based on point-and-click operations on a graphical -chema (for example, a Q32like interface; see Appendix ID). In this section, we give a basef overview of **dynamic SQL** which is one technique for writing these type of database program, by pacing a simple example to illustrate how dynamic SQL can work.

Program segment **E**3 in Figure 9.5 reads a string that is input by the user (that string should be an SQL opdate command) into the acting variable sqlupdatestring in line 3.1; then prepares this as an SQL command in line 4 by associating it with the SQL variable sqloomand. Line 5 then executes the command. Notice that in this case no spitos check or other types of checks on the command are possible as couple and, since the command is not available useril function. This contrasts with our previous examples of embedded SQL, where the query could be checked at compile time because its text was in the program source code.

Although including a dynamic update command is relatively straightforward in dynamic sQL is dynamic query is much more complicated. This is because in the general case we do not know the type of the number of attributes to be intraeved by the sQL querwhen, we are writing the program. A complex data structure is sometimes needed to allow for different numbers and types of attributes in the query result if no pror information is known about the dynamic query. Techniques similar to those that we discuss in Section 9.5 can be used to assign query results (and query parameters) to host program carables

In E3, the reason for separating fREPARL and EXECUTE is that if the command is to be executed multiple times in a program, it can be prepared only encourage the command generally more was syntax and other types of checks by the system, so well a

FIGURE 9.5 Program segment E3, a C program segment that uses dynamic styll for updating a table

generating the code for execution it. It is possible to combine the PREPARE and EXECUTE reasonable (lines 4 and 5 in E3) intro a single statement by writing

EXEC SQL EXECUTE IMMEDIATE :sqlupdatestring .

Taxis useful if the command is to be executed only once. Alternatively, one can separate the two to ratch any errors after the IREDAR, statement, if any

## 9.4.4 SQLJ: Embedding SQL Commands in JAVA

In the provious sections, we give an overview of how SQL commands can be unbelled in a traditional programming language, using the C language in our examples. We now turn all statistical boost  $SQ_1$  can be embedded in an object oriented programming language," in parcolar, the (AVA language, SQL) is a standard that has been adopted by several versions for embedding SQL on (AVA). Historic ally, SQL) was developed after (1980, which is used for eccessing SQL databases from (AVA) using function calls. We diarous (DRC) in Sector, 9 of the our discussion, we focus on SQL) is it is used in the CRACHER (2000). An SQL mashed will generally convert SQL statements into (CVA), which can then be executed therefore the (DRC) interface. Hence, it is necessary to install a 2018, doine when using SQL\* In this section, we focus on box to use SQL) concepts to write ethosided SQL in a [4VA program.

Before being tible to process SQL) with [AVA in ORATE, it is necessary to import second class libratics, shown in Figure 9.6. These include the JDPC and IO classes (lines 1 and 2), plus the additional classes listed includes 3, 4, and 5. In addition, the program must histoprotect to the desired database using the function of ligetConnection, which is one of the methods of the oracite class in line 5 of Figure 9 of The format of this function call, which returns an object of type default context.<sup>6</sup> is as follows:

```
public static DefaultContext
getConnection(String or), String user, String password, Boolean
autoCommit)
throws SQLException ,
```

For example, we can write the warements in lines 6 through S in Figure 9.6 to connect to an ORACTE database located at the URL wirl name> using the logic of waser name> and small <pr

<sup>5</sup> This section assumes transforming with object oriented concepts and basic (as a concepts. If in de in tekning funderous, they should people are the section of other reading Chapter 12

v Weidisco - 1000 drivers in Section 3.5.2

<sup>2</sup> Adging other, when set applies to subsequent compositions the program until it is charged.

<sup>1)</sup> Anamore computation renghb means that each constant is supposed to the database attent is regarded. The observative is that the programmer words to execute several related database comrank and their commit them together. We discuss contrast concepts in Chapter 17 when we mer realizables transforms.

```
1)
 import java.sql.* ;
 import java.io.* :
2)
30
 import sqlj.runtime.4 ;
 import sqlj.runtime.ref.* ;
4) -
5)
 import oracle.sqlj.runtime.* ;
6)
 DefaultionText cnfxt =
7)
 oracle.getConnection("kurl name>", "kuser name>", "kpassword>", true);
8).
 DefaultContext.setDefaultContext(cntxt) :
```

FIGURE 9.6 Importing classes needed for including SQL in IAVA programs in GRACIE, and ustab-Tshing a connection and ifetault context.

> In the following examples, we will not show complete JAVA classes or programs anteit is not our intention to teach JAVA. Bather, we will show program segments that illustrate the use of SOLI. Figure 9-7 shows the JAVA program variables used in our examples. Program segment JL in Figure 9-8 reads an employee's secial security number and prints anne of the employee's information from the database.

> Notice that because IAVA idready uses the concept of exceptions for error handling a special exception colled 501 Exception is used to return errors or exception conditions also exception and QU behave command. This plays a similar role to SQUADE and SQUATE is embedded 502. (3373) has many types of predefined exceptions. Each 1373 (correction) must specify the exceptions that can be thrown -that is, the exception conditions that now occur while executing the JAVA code of that operation. If a defined exception occurs the system transfers control to the JAVA code specified for exception handling. In **W**, exception handling for an SQLException is specified for exception handling. In **W**, exception handling for an SQLException is specified for exception by the code in a particular operation should be specified as part of the operation declaration of other and particular operation should be specified as part of the operation declaration of other parts.

«operation return type» «operation mame»(«parameters») throws SQLE«ception. LOE»ception ;

In SQL, the multi-diled SQL commands within a JAVA program are proceeded by easily as illustrated in [1] line 3, so that they can be identified by the preprocessor. SQL uses as 90000 closed -similar to their used in embedded SQL to return the artificte value returned from the database by an SQL query into JAVA program catables. The program variables are preceded by colors (c) in the SQL statement, as in embedded SQL.

```
 string dname, ssn., fname, fn, lname, ln, bdate, address ;
 char Sex, minit, mi ;
 double salary, sal ;
 integer dno, dnumber ;
```

**EXCURE 9.7** two program variables used in soci examples [1 and [2].
```
//Program Segment 11:
n
 ssn = readEntry("Enter a Social Security Number: ") ;
21
 την {
 #Spl{select FNAME, MINIT, LNAME, ADDRESS, SALARY
31
4)
 into :fnume, :minit, :lname, :uddress, :salary
 from EMPLOYEE where SSN = :ssn} ;
51
61
 } catch (SQLException se) {
71
 System.put.println("Social Security Number does not exist: " + ssn) ;
8)
 Return :
9)
 1
 System.out.println(fname + " " + mirit + " " + lname + " " - address + " " +
101
 salary)
```

FIGURE 9.8 Program segment J1, a AVA program segment with SQLL

In [1 a single ruph: is selected by the embedded SOFT query, that is why we are able to assign to attribute values directly to AVA program variables in the INTO clause in line 4. For queries that retrieve many ruples, SOTT uses the concept of an atomas, which is surewhat similar to a cursor in embedded SOE.

#### 9.4.5 Retrieving Multiple Tuples in SQLI Using Iterators

In (Q), an iterator is a type of object associated with a collection (set or multiset) of topas in a query result <sup>17</sup>. The iterator is associated with the toples and attributes that appendix query result. There are two types of iterators:

- A named iterator is associated with a query result by listing the artribure names and types that appear in the query result.
- 2. A positional iterator lists only the attribute (gas that appear in the query result.

In both cases, the last should be in the same order as the attributes that are listed in the stills 1 clayse of the query. However, looping over a query result is different for the two bres of perators, as we shall see. First, we show an exaciple of using a network reprort in figure 9.4 program segment J2A. Line 9 an Figure 9.9 shows how a named iterator type Employed and the arms of the attributes in a parted iterator type Employed for the attributes in the SQL query result. The 10 shows how an iterator object e of type for its created in the program and then associated with a query fluxes 11 and 120.

When the instance object is associated with a query (lines 11 and 12 in Eigne 9.8), deprogram terches the query result from the database and sets the instance to a position when ne first own the result of the query. This becomes the **current** row for the iterator. Subsequently, next operations are usual on the ateration, each moves the iterator to the tor real in the test budy the query making in the correct row. If the row exists, the

<sup>17</sup> Wedsens, tenders in none detail in Chapter 21 when we discussed just charlases

```
//Program Segment 12A:
(0)
 dname = readEntry("Enter the Department Name: ") ;
1)
 TƘy {
22
 #sgl{select DNUMBER into :drumber
35
 From DEPARTMENT where DNAME = :dname} ;
4)
 }___
 catch (MULException se) {
5)
 System.out.println("Department does not exist: " + dwame) ;
6)
 Return ;
7)
 l
 System.out.printline("Employee information for Department: " + dname) ;
8)
 #sql iterator Emp(String ssn, String fname, String minit, String Iname,
9)
 double salary) ;
 Emp e = null ;
10)
 #sql e = {select ssn, fname, minit, lname, salary
11)
 from EMPLOYEE where DNO = :dnumber) ;
12)
132
 while (e.next()) {
 System.out.printline(e.ssn + " " + e.fname + " ' + e.Winit + " " +
14)
 c.lname + " " + c.salary) ;
15)
 };
16)
 e.close();
```

FIGURE 9.9 Program segment J2A, a IAVA program segment that uses a named iterator to print, employee information in a particular department.

> operation retrieves the attribute values for that row into the corresponding program variables if no more rows exist, the next operation returns null, and can thus be used to control the looping.

> In Figure 9.9, the command (elinext()) is line 13 performs two functions: It gets the next tuple its the query result and controls the while loop. Once we are done with the query result, the command elittlese() fline 16) closes the iterator.

Next, consider the some example using possional iterators is shown to Engure 9.0 (program segment 12B). Line 9 in Fratre 9.10 shows how a positional iterator type Engage is declared. The main difference between this and the named iterator is that there are reattribute names in the positional iterator—only attribute types. They still must be compatible with the attribute types in the SQL query result and in the same order. Line 10 shows have a positional iterator variable e of type Emptors is created in the program and then associated with a query (lines 11 and 12).

The pointional iterator behavior in a manuar that is more similar to embedded SQ. (see Section 9.4.2). A fetch <iterator variable> into <program variables> command is needed to get the next tuple and query result. The first rune forch is exerured, it grashe first tuple (line 1) in Figure 9.10). Line 16 gets the next tuple until no more tuples exist in the query result. To control the loop, a positional iterator function niemfFetch() is used. This function is set to a value of 1806 other, the iterator is initially cascilated with an SQL query fame 11), and is set to FALSE each tune a forch command returns a valid tople from the query result. It is set to TRUE again when a fetch command does not find only more tuples. Line 14 shows how the loop og is controlled by negation.

```
//Program Segment 028:
 dname - readEntry("Enter the Department Name: ") ;
D)
1)
 tty ł
21
 #sul{select DNUMBER into :dnumber
3)
 from OEPARTMENT where DNAME = :dname} :
4)
 } catch (SQLException se) {
 System.out.println("Orpartment does not exist: " + dname) ;
5)
6)
 Return :
7)
 1
 System.out.printline("Employee information for Department: " + drame) ;
8)
 #sql iterator Emppos(String, String, String, String, double) ;
9)
!a) Empgos e = null ;
 fsql e ⇒(select ssn. fname. minit, iname. salary
(1)
:2)
 from EMPLOYEE where DNO = :dnumber} .
3) #sql {fetch de into dassn, dfn, dmi, dn, dsal};
i4) while (!e.endFetch()) {
 System.out.printline(ssn + " " + fn + " " + mi + " ' + ln + " " + sal) ;
:5)
16)
 #sql (fetch te into tssn, tfn, twi, tln, tsal);
£7) -
);
(K) e.close() :
```

**ECORE 9.10** Program segment [28] a two program segment that uses a positional iterator to performation in a particular department.

# 9.5 DATABASE PROGRAMMING WITH FUNCTION CALLS: SQL/C11 AND JDBC

Inhedded SQ. (See Section 9.4) is semictimes referred to as a static database program ning approach because the query text is written within the program and cannot be fonged without recompiling or reprocessing the source code. The use of function calls is any odynamic approach for database programming than embedded SQL. We already site as database programming technique - dynamic SQL - in Section 9.4.3. The refragues discussed here provide another approach to dynamic database programming. A library of functions, also known as an **application programming interface** (API), is bedreakees the database. Although this provides more flexibility because no preprocesan second, one drawback is that syntax and other checks on -QL commonds have to be live to ontone. Another drawback is that it sometimes requires note complex programing to access query results because the types and numbers of attributes in a query result ment be known in advance.

arthas section, we give an overview of two function call interfaces. We first discussion of the SQL (Gall Level Interface), which is part of the SQL standard. This was developed as a follow up to the earlier technique know as OD60 (Open Data Base Connectivity). We use the host language in our SQLATI examples. Then we give an everywey of **JDBC**, which is the call function interface for accessing databases from (AVA). Although it is commonly assumed that (2000) stands for Java Data Base Connectivity, 1000 as institution interface of Sun Microsystems, not an activities.

The main advantage of using a function call interface is that it makes it easier to access multiple databases within the same application program, even if they are stored under different DWS packages. We discuss this further in Section 9.5.2 when we discuss (AVA database programming with DWC, although this edvantage also applies to database programming with SOUCH and ODWC (see Section 9.5.1).

## 9.5.1 Database Programming with SQL/CLL Using C as the Host Language

Perfore using the function calls in SQACU, it is necessary to install the appropriate librar packages on the database server. These packages are obtained from the cender of the DPMS being used. We now give an overview of how SQACU can be used in a C program. We shall allostrate our presentation with the example program segment CLH shown (a Figure W11).

When using SQ(0,0), the SQL-t itements are dynamically created and passed as string parameters in the function calls. Hence, it is necessary to keep track of the information about host programmateriotions with the database examining data structures, because the database commands are processed at runtime. The information is kept in few types it

```
//Program CLIL:
 #include_sqlcli.h :
0) -
 void print5al() {
1.)
\langle \rangle
 SUBSIME STHEET ;
30-
 SQUHDBC con1 :
4)
 SQLHENV env1 :
50.
 SOURETURN mot1, met2, met3, met4 ;
 ret1 = SQLAllocHandBe(SQL_MAMPLE_ENV, SQL_NULL_MANDLE, &env1) ;
65
7)
 if ('netl) ret2 = SQL411okHardle(SQL_HAMDLE_DBC, envl. &coml) else exit ;
 if (Iret2) ret3 = SQLConnect(con1, "dbs", SQL NTS, "js", SQL NTS, "xyz", SQL NTS)
8)
else exit :
 if (!ret3) ret4 - SQL4llocHandle(SQL_HANDLE_STMT, con1, &stmt1) else exit ;
90
 SQLPrepare(stmt1, "select LNAME, SALARY from EMPLOYEE where SSN = ?", SQL_NTS);
10)
11)
 prompt("Enter a Social Security Number: ", ssn) ;
 SQLBindParameter(somt1, 1, SQL CHAR, &ssn. 9, &ferchlen1) ;
12)
T3J
 ret1 = SQLExecute(simt1) ;
 if (!ret1) (
14)
15)
 SQLBindCol(stmt1, 1. SQL_CHAR, &lname, 15, &fetchlen1) ;
16)
 SQLBindEol(stmt1, 2, SQL_FLOAF, &salary, 4, &fetchlen2) ;
173
 ret2 = SQLFetch(stat1) ;
18)
 if (!ret2) prin(f(ssn, lname, salary)
 else printf("Social Security Number does not exist. ", ssr) ;
19)
2∏).
 }
21)
 }
```

records, represented as structs in C data types. An environment record is used as a consistent to keep track of one or more database connections and to set environment anomation. A connection record keeps track of the information needed for a particular disfose connection. A statement record keeps track of the information needed for one SC statement. A description record keeps track of the information about tuples of parameters—for example, the mutber of attributes and their types in a tuple, or the innuber and types of parameters in a function call.

Each record is accessible to the program through 5 st pointer variable—colled a handle to the record. The hundle is zetterned when a record is first created. To create a record and zeturn its handle, the following sQL/CT function is used:

```
SQLATTacHandle:<handle_type>. <handle_t>. <handle_t>.
```

In this function, the parameters are as follows:

- chandle\_types indicates the type of neurod being created. The possible values for this parameter are the keywords sql\_sAMDLE\_ars, Sql\_sAMDLE\_DEC, sql\_sAMDLE\_STAT. of Sql\_ seques\_LEM, for an environment, connecterin, statement, or description tecord, respectively.
- chandle its indicates the container within which the new handle is being created.
   Far example, for a connection record this would be the environment within which the connection is being created, and for a statement record this would be the connection for fair statement.
- drandle 2s is the painter (bandle) to the newly created record of type shanely, types.

When wrating a C program ther will include database calls through SCAAD, whe following are the typical steps that an taken. We illustrate the steps by referring to the example CLM on Figure 9.13, which reads a social accurity nonifer of an employee and print the engloyee's last name and salary:

- The Brany of functions comprising SQL/CLI must be included in the C program. This is called sq1c1n\_b, and is included using line C in Figure 9.11.
- 2 Declare fandle consider of repressions sections, source, and sources for the stare ments, connections, environments, and descriptions needed at the program respectively times 2 to 4).<sup>13</sup> Also declare variables of type squarment (line 5) to sold the return codes from the SQLATE function calls. A return code of 0 (ters) indicates successful elsections of the function call.
- 3 An excisionneck record must be set up in the program using SQLA1365Handle. The function to do this is shown in line or Persuase an environment record is not constanted in any other record, the parameter shandle\_1s is the null handle SQL\_NLLE\_RANDLE (null pointer) when creating an environment. The handle (pointer) to the newly created construment record is returned in variable environment in the 6.
- 4. A connection record is set up in the program using SQLA11actiand1e. In line 7, the connection record created has the handle con1 and is contained in the environ-

<sup>&</sup>quot;I. We call not show description records here, to keep our presentation simple

ment envir. A connection is then established in conflict a particular server durative using the SQLConnect function of SQL(0.1) (line 8). In our example, the database server name we are connecting to  $(s/200^{-3})$  and the second name and password for login are " $\mu$ " and " $\kappa z$ ", respectively.

- 5 A statement record is set up in the program using SQLATTOCHandDe. In line 9, the statement record created has the handle stat1 and uses the connection con1.
- 6. The statement is prepared using the SQLOTE function SQLPrepare, in line 10 this assigns the SQL statement string (the query in our example) to the statement handle start. The question mark (f) symbol acline 10 represents a statement parameter, which is a value to be determined at maintine—typically be building it to a C program smable. In general, there could be several parameters. They are distinguished by the order of appearance of the question marks in the statement (the list 2 represents parameter 1, the second 2 represents parameter 2, and so on). The lose parameter in SQLPrepare, should give the length of the SQL statement string in bytes, but if we enter the keyword SQL of a this indicates that the string holding the query is a molecomment string so that SQL can calculate the string length automatically. This also applies to other string parameters in the formation calls.
- 7. Before executing the query any parameters should be bound to program variable using the SQL/CLL function SQLBindParameter. In Figure 9.11, the parameter (radicated by 2) to the prepared query referenced by stmt1 is bound to the C program variable son in line 12. If there are a parameters in the SQL statement, we should have a SQLBindParameter function calls, each with a different parameter position (1, 2, ..., a).
- Following these preparations, we can now execute the R3D statement referenced by the handle stmt1 using the function SQ Facture (func 13). Notice the although the query will be executed in fine 13, the other results have not yet been assigned to oncomposition wattables.
- <sup>30</sup> In order to determine where the result of the query is termined, one common technique is the bound columns approach, there each columns in a query result is bound to a 0° program variable using the SQLB (ndCo° function. The columns are distinguished So there order of appearance in the SQL query in Figure 9.11 lines 15 and 16, the two columns to the query (cose, and sectors) are bound to the coprogram variables brane and satisfy respectively.<sup>4</sup>
- 10. Finally, invoidents remeve the column values into the C program variables, the function SQLFetCleis used three 17). This function is similar to the FFTCH command of embedded SQL. If a query result has a collection of tuples, each SQLFetCh call gets the next tuple and returns its column values into the bound.

<sup>14.</sup> An alternative technologic known as unbroughtendutures as eading out 5.000 () from these gamely SOLGETCOL > SQLGETDATA, to retrieve columns- from the query result without previously building these are upplied, due the SQLFETCH compared or explain.

program variables. SQLFetch returns an excuption (noncerts) code it there are not inore ruples.<sup>15</sup>

As we can see, using dynamic function calls requires a lot of preparation to set up the ~2, statements and to bind parameters and query readts to the operapriate program variables.

In CLU, a angle rapid is selected by the SQL query. Figure 4.12 shows an example of rationing multiple tuples. We assume that appropriate if program variables have been deduced as in Figure 9.12. The program segment in CUU reads (inputs) a department number and then retrieves the employees whice work on that department. A loop then accases over each employee means, one or a time, and prints the employee's last name and solars.

#### 9.5.2 JDBC: SQL Function Calls for JAVA Programming

We now turn out attention to how SQL can be called from the (AVA object-oriented programing language)<sup>16</sup> The function libraries for this access are known as **JDB**C.<sup>17</sup> The BVA programming language was descended to be platform independent—that is, a programshould be oble to run on any type of computer system that loss of AVA interpreter installed. Because of this portability, many RDBMS vendors provide 00bc driver is basically an implesentiation of the function calls specified in the 1040. Alt (Application Programming Installed for a particular vendor's RDBMS. Hence, a IAVA program with JDBC function ralls on access any RDBMS that has a 'DBC driver regulable.

Boouse (AVA is object-oriented, its function libraries are implemented as classes. Beforeboing able to process (LBC function calls with (AVA) it is increasing to import the JOE, class libraries, which are called java.sq1.\*. These can be downloaded and gaulled via the Web <sup>18</sup>

1910 is designed to allow a single (AVA) program to connect to several different datases. These are sometimes called the **data sources** accessed by the TAVA program. These data sources civild be stored using 800 attaction different vendors and calducide on different machines. Hence, different data source recesses within the source) (AVA) program have require (AVA) drivers from different vendors. To achieve this leading at special 1090 class called the driver manager class is employed, which keep track of the instituted drivers. A driver should be agriced with the driver

<sup>15.</sup> Contours of program can able the ased. SQLFEECh returns the upple case to reimporary program and ball subsequent SQLSeeColl (or SQLGetOata) returns to combate value as other.

In this section assumes (and an investigation object) oriented concepts and basic PACA copeople. If teaches lick instantian or other stocked postpone this section and after coding villapter 20.

<sup>[5]</sup> As as invarianted gather, IOBC is a registered in alcound of Son Microsystems, although it is someonly the optimized as a conserving for Jacka Data Base Conference by.

<sup>[3]</sup> These up available from several Web sites—for gaugele, through the Web size at the USU http:// monitopic.com/presh//regilbe/drocets.

manager before in is used. The operations (merbidds) of the driver manager class include getDriver, registerDriver, and deregisterDriver. These can be used to add and remove drivers dynamically. Other functions set up and close connections to data sources as we shall see.

To load a JDBC driver explicitly, the generic JAVA futurion for loading a class can be used. For example, to load the JDBC driver for the ORACLE RDBM-, the following command can be used:

(lass.forName('orarle\_jdbr.driver.OracleDriver")

This will recister the driver with the driver manager and make it available to the program It is also possible to load and register the driver(s) needed in the command line that runthe program, for example, by including the following its the command line:

-Djdbc.drivers = oracle.jdbc.driver

The following are typical steps that are taken when writing a JAVA application program with database access through (DSC function calls. We illustrate the steps by referring to the example JDBCI in Figure 9.13, which reads a social sciently number of an employee and prints the employee's last name and solary.

```
//Program Segment CLI2:
 #include sqlcli.h ;
0)
1)
 void printOepartmentEmps() [
2)
 SQLHSTWT stmtl ;
31
 SOLHDBC con1 :
4)
 SOLHENV env1 :
 SQLRETURN ret1, ret2, ret3, ret4;
50
 ret1 = SQLAFTocHandle(SQL HANDLE_ENV, SQL NULL_HANDLE, &env1) ;
6)
 if (:ret1) ret2 = SQL4llocHandle(SQL_HANDLE D8C, env1, &con1) else exit ;
7).
 if (!rer2) ret3 = SQLConnect(con1, "dbs", SQL NTS, "js", SQL NTS, "xyz", SQL NTS)
B)
else exit :
 if (!ret3) ret4 = SQLATTOCHandle(SQL_HANDLE STMT, con1, &stmt1) else exit ;
91
 SQLPrepare(stmt1, "select _NAME, SALARY from EMPLOYEE where DNO = ?", SQL NTS) :
10\rangle
 prompt("Enter the Department Number: ", dro) ;
11)
12)
 SQLB:ndParameter(stmt1, 1, SQL INTEGER, &dno, 4, &fetchlen1) ;
 ret1 = SQ_Execute(stmt1) ;
13)
 if ('retl) {
14)
15)
 SQLBind(ol(stm+1, 1, S0L_CHAR, &lname, 15, &fetchlen1) ;
 SQLBindLol(stmr1, 2, SUL_FLOAT, &salary, 4, Sfetchlen2);
16)
17.5
 ret2 = SQLFetch(stmt1) ;
 while (!ret2) {
181
 printf(lname. salary) :
19\rangle
 ret2 = SQLFetch(sumt1) ;
20)
21)
 3
22)
 }
23)
 }
```

FIGURE 9.12 Program segment CLI2 a C program segment that uses 5Qc/Cu for a query with a collection of toples in its result.

- The (D0, hbury of classes must be imported into the (AVA program. These classes are called pavalised 1.8, and can be imported using line 1 in Figure 9.13. Any additional (35.5 class libraries meriled by the program must also be imported.
- Load the JU00 driver as discussed previously (lines 4 to 7). The (AVA exception in line 5 openrs if the driver is not loaded successfully.
- 3. Creare appropriate variables as needed in the JAVA program ()mes 8 and 91.
- 4. A connection object is created using the getConnection function of the DriverManager class of JUDG in lines 12 and 13, the connection object is created by using the function call getConnection (unlist ring), where unlistering has the form.

```
jdbc:oracle:kdriverTypex:kdbaccounts/kpassword>
```

An alternative form is

getConnection(unl, dbaccount, password)

Various projecties can be set for a connection object, but they are mainly related to transactional properties, which we discuss it: Chapter 17.

- 5 A statement object is created in the program. In 1190, there is a basic statement class, Statement, with two specialized subclasses: PreparedStatement and CallableStatement. This example illustrates how PreparedStatement objects are created and used. The next example (Figure 9.14) illustrates the other type of Statement objects. In time 14, is query string with a single parameter—indicated by the "" symbol—is created in the variable statt). In blie 15, an object a soft type PreparedStatement is created on the query string in statt1 and using the connection object connut in general, the programmer should use PreparedStatement objects of a query is to be executed multiple times, since it would be prepared, chicked, and compiled only once, this saving this cust for the additional executions of the query.
- f. The question mark (7) symbol in line 14 represents a statement parameter, which is a value to be determined at rontime, typically by binding at to a JAV's program carable. In general, there could be several parameters, distinguished by the order of appearance of the question marks (first ? represents parameter 1, second ? represents parameter 2, and so only in the statement, as discussed previously.
- 3 Before executive to PreparedStatement query may parameters should be bound to program vortables. Depending on the type of the parameter, foretions such as setString, setInteger, setBooble, and so on are applied to the PreparedStatement object to set its parameters. In Figure 9.13, the parameter fundicated by () in object p is bound to the IAVA program variable sist in line 18. If there are a parameters in the SQL statement, we should have it Set functions, such with a different parameter position (1, 2, ..., s). Constally, it is advisable to chear ill parameters before setting any new values (line 17).
- 5. Following these preparations, we can now execute the SQL statement referenced by the object plusary the function: executeQuery three 19). There is a general function, execute in (1983), plus two specialized functions: executeUpdate and execute()upry, executeUpdate is used for SQL unset, define, or update statements.

```
//Program JDBC1:
 import java.io.* ;
D)
 import java.sql.1
1)
2)
 class getEmpInfo {
 public static yord wain (String args []) throws SULException, IOException {
3)
 try { Class.forName("oracle.jdbc.driver.BracleDriver")
4)
 } catch (ClassNotFoundException x) {
5)
 System.out.println ("Driver could not be loaded") ;
6)
72
63
 String dbacet, passwrd, ssn. lname :
9)
 Double salary :
100
 dbacct = readentry("Enter database Account:") ;
11)
 passwrd = readentry("Enter pasword:") ;
12)
 Connection comm = DriverManager.getConnection
 ("idbc:oracle:oci8:" + dbacct + "/" + passwrd) :
13)
14)
 String stat1 = "select LNAME, SALARY from EMPLOYEE where SSN = ?" :
15)
 PreparedStatement p = conn.prepareStatement(stat1) ;
16)
 ssn = readentry("Enter a Social Security Number: ") ;
17)
 p.clearParameters() ;
 p.setString(1, ssn) ;
18)
19)
 ResultSet r = p.executeQuery() :
20)
 while (c.next()) {
21)
 Iname = r.gerString(1) ;
22)
 salary = r.getDouble(2) ;;
 system.out.printline(lname + salary) :
23)
24)
 } }
25)
 ł
```



and remark on integer value indicating the number of ceptes that were offected, executeQuery is used for NQL retrieval statements, and returns an object of type ResultSet, which we docute next.

9. In late 19, the result of the query is retorned in an object r of type ResultSet. This resembles a two-dimensional array or a table, where the tuples are the two and the attributes retained one the columns. A ResultSet object is similar to a cursor in enfected solt and an iterator in eQU. In our example, when the query is executed, it refers to a tuple before the first tuple in the query result. The *r*.next() function (line 20) moves to the next ruple (row) in the ResultSet object and retains no0° if there are no more objects. This is used to control the looping. The programmer can refer to the attributes in the current tuple using various get i functions that depend on the type of each attribute for example, getStrong, getInteger, getBouble, and so on). The programmer can either use the attribute names ("LNAME", "SALARY").

with the get - tunctions. In our examples, we used the positional nutotian inlines 21 and 22.

In general, the programmer can check for SQLPSCeptions ofter each JOSC function call. Notice that JOPC does not distinguish between queries that return single tuples and those that return multiple ruples, inslike some of the other techniques. This is just fiable because a single tuple result set is just a special case.

In example JOBCE, a single tople is selected by the SQE query, so the loop in lines 20 in 24 is executed at most incle. The next example, shown in Figure 9.14, illustrates the reneval of multiple toples. The program segment in JDBC2 reads (upputs) a department manher and their recreases the conflusees who work in that department. A loop then traces over each employee record, one of a type, and prints the uniployee's last name and salare. This example also theatraces how we can execute a query directly, without having to prepare it us in the previous example. This accluding is preferred for queries.

```
//Program Segment JDBC2:
 import java.ja.* ;
()
 import java.sgl.*
D.
2)
 class printDepartmentEmps {
31
 public static void main (String args []) throws SQLException, JOException {
4)
 try { Class.forName("onable.jdbC.dr1ver.OracleOniver");
5)
] catch (ClassNotFoundException x) {
6)
 System.out.printld ("Oriver could not be loaded") ;
7)
B)
 String dbacct, passwird, lname ;
9)
 Double salary ;
10)
 integer dno ;
 dbacct = readentry("Enter database account:") ;
1.5
12)
 passwrd = readentry("Enter pasword:") :
 Connection corn - DriverManager.getConnection
10)
 ("(dbc:oracle:oci8:" + dbacct + `/" + passwid) ;
14)
 dno = readentry("Enter a Department Number: ") ;
19)
 String q = "select LNAME, SALARY from EMPLOYEE where DND = " +
16)
 dmo.tastring() ;
17
 Statement s = conn.createStatement() ;
LD.
 ResultSet r = s.executeQuery(q) ;
1^{(1)}
 while (c.next()) {
20)
 Iname = r.getString(1) ;
21)
 salary = r.getDouble(2) ;
221
 system.out.printline(lname + salary) :
23)
 } }
24) }
```

NGURE 9.14 Program segment IDBC2, a JAVA program segment that uses JDBC for a query with a collection of tuples in its result. that will be executed only once, since it is simpler to program. In first 17 of Figure 9.14, the programmer creates a Statement object substead of PropagedStatement, as in the previous example? without associating it with a particular query string. The query strings is possed to the statement object's when it is executed in large 18.

This concludes out blief introduction to JOBC The interested reader is referred to the Web (site, http://jawa.sun.com/docs/books/turonal/jdbc//\_which\_contains\_many\_further details on JDBC

# 9.6 DATABASE STORED PROCEDURES AND SQL/PSM

We conclude this chapter with two additional topics related to database programming. In Section 9.6-1, we discuss the exacept of stored procedures, which are program modules that are stored by the PBNs of the database server. Then in Section 9.6.2, we discuss the even sions to SQL that are specified in the standard to include general purpose programming constructs in SQL. These extensions are known as SQL/DN1 (SQL/Persistent Stored Modules) and can be used to write stored procedures, SQL/DN1 (SQL/Persistent Stored Modules) and can be used to write stored procedures, SQL/DN1 also serves as an example of a database programming language that extends a database model and language—maniely, SQL—with some programming constructs, such as conditional statements and large.

## 9.6.1 Database Stored Procedures and Functions

In our presentation of database programming techniques so for, there was an implicit assumption that the database application program was turning on a client machine that a different from the machine on which the database server—and the name pair of the DMIs software package—is located. Although this is solitable for many applications, it is some times useful to create database program modules—procedures of functions—that are solited and executed by the DMIs(a) the database server. These are instortically known as database stered procedures, although they can be functions or procedures. The term used in the 93 standard for stored procedures is persistent stored modules, because these programs are stored persistently by the DMIs(), contarly to the persistent clata stored by the DMIs.

Stored procedures are useful in the following pircumstances:

- If a database program is medical by several applications, it can be stored at the series and invoked by any of the application programs. This reduces daplication of efferand improves software modulative.
- Executing a program of the server can reduce duta transfer and bonce contumneation cost between the chemicand server in certain attuations.
- These procedures can enhance the modeline povert provided by clews by allowing more complex ropes of derived data to be made available to the database users in addition, they can be used to check for complex constraints that are beyond the specilicatives power of assertices and triggers.

In general, many commercial LOMs- allow stered procedures and functions to 5e written in a general-purpose programming language. Alternative 3, 5 stored procedure can be made of sample SQL commands such as retrievids and updates. The general form of defenge a trated procedures is as follows:

#### CREATE PROCEDURE <procedure name> ( <parameters> ) <local declarations> <procedure body>

The parameters and local declarations are optional, and are specified only it needed for declaring a function, a terrain type is necessary, so the declaration form is

CREATE FUNCTION </unclion name> ( <parameters> ) RETURNS <return type> <local declarations> <lunction body> :

If the procedure (or function) is written in a general-purpose programming language, it is typical to specify the language, as well as a file name where the program code is stand, for example, the following format can be need:

```
CREATE PROCEDURE «procedure namé» («párameters»)
LANGUAGE «prógramming language name»
EXTERNAL NAME «tile path name» ;
```

In general, each parameter should have a **parameter type** that is one of the SQL data types Each parameter should also have a **parameter mude**, which is one of (S, O, T, or two). These correspond to parameters whose values are impuriedly, corput frequenced any or both input and compar, respectively.

Because the primedures and functions are strictly providently by the DIME, it should be possible to call them from the various SQL (procedures and exogramming rechniques. The CALL statement in the SQL standard can be used to invoke a strong procedure either from an interactive interface on from embedded SQL or SQL). The format of the strength is as follows:

CALL <procedure or function name> ( <argument list> ) ;

othe statement is called from (PBC, it should be assumed to a statement object of type GPapleStatement (see Section 9.5.2).

## 9.6.2 SQL/PSM: Extending SQL for Specifying Persistent Stored Modules

80(15)) is the part of the 5QL standard that specifies how to voite persistent stored modules it includes the statements to are the functions and procedures that we described in the previauszeria of thatso includes additional programming constructs to enhance the power of SQL 6) die purpose of writing the code for bode? at stored procedures and functions.

In this section, we discuss the SOL/PSM constructs for conditional (branching) sciences and for looping science into Tracs, will give a flavor of the type of constructs

that SQMPSM has incorporated  $^{18}$  Then we give an example ob-illustrate how these constructs can be used

The conditional branching statement in SQLPSM has the following form

IF <condition> THEN <statement list>

ELSEIF <condition> THEN <statement list> ... ELSEIF <condition> THEN <statement list> ELSE <statement list> END IF :

Consider the example in Figure 9.15, which discretes how the conditional brack structure can be used in an SQL/SM binction. The function returns a string value (fire f) describing the site of a department based on the number of employees. There is one is integer parameter, departs, which gives a department number. A forcal canable NoOfFigs is declared in line 2. The query in lines 3 and 4 returns the number of employees in the department, and the condition of branch in lines 5 to 8 them returns one of the value (FRUCET, TLARGET, TMEDIUMT, TSMALLT) based on the number of employees.

Sci/28M has several constructs for looping. There are star-ducl while and repeat looping structures, which have the following forms

WHILE <condition> DO <statement list> END WHILE :

//Function PSML:

- CREATE FUNCTION DeptSize(IN deptno INTEGER)
- RETURNS VARCHAR [7]
- DECLARE NoOfEmps INTEGER ;
- SELECT COUNT(\*) INTO NoOfEmps
- FROM EMPLOYEE WHERE DND = deptnb :
- IF NoOfEmps > 100 THEN RETURN "HUGE"
- ELSEIF NoOFEmps > 25 THEN RETURN "LARGE".
- ELSEIF NoOFEmps > 10 THEN RETURN "MEDIUM"
- ELSE RETURN "SMALL"
- 9) END IF ;

FIGURE 9.15 Declaring a function in SQL/ISM.

 We contribute a function conduction to SQL/086 here. There are many other features of the SQL PSR standard. PEPEAT

```
<statement list>
UNTIL <condition>
ENO REPEAT .
```

There is also a conservised bopping structure. The statement list at such a loop is excludion to reach tuple in the query result. This has the following form:

```
FOR <loop name> AS <cursor name> CURSOR FOR <query> DO
<statement list>
```

END FOR ;

loops conduce names, and there is a LEAVE is loop in anose statement to break a loop where condition is satisfied. SQL/156 has many other features, but they are outside the sope of our presentation.

# 9.7 SUMMARY

In this chapter we presented additional features of the SQL database linguage. In particular, we presented an overview of the most important techniques for database programming. We stand in Section 9.1 by presenting the features for specifying general constraints as assertions. Then we discussed the concept of a view in SQL. We then discussed the various approaches to database application programming in Sections 9.5 to 9.6.

## **Review Questions**

- 91. How does set allow implementation of general integrity constraints?
- 9.2 What is a view in sign and how is it defined? Discuss the problems that may arise when one attempts to update a view. How ore views typically implemented?
- 3.5 Lot the three more opproaches to database programming. What are the advanages and disadvantages of each opproach?
- 9.4 What is the impedance manatch problem? Which of the three programming approaches innumizes this problem?
- 45 Describe the concept of a cursor and how it is used in embedded SQL.
- 96 What is SQL used for "Describe the two types of iterators available in SQL)

## Exercises

- 97. Consider the database shown in Figure 1.2, whose schema is shown in Figure 2.0. Write a program segment to read a student's name and point bis or her grade point average, assuming that A=4. B=3. C=2, and D=1 goints. Use embedded sQL with C as the bast language.
- 98. Repeat Exercise 9.7. Fur use 7QU with JAVA as the host language

- 9.9. Consider the consist relational database schema of Figure 6.12. Write a program segment that removes the list of books that became overdue yesterday and the prioris the book table and borrower name for each. Use embedded SQL with 0 is the host language.
- 9.10 Repeat Exercise 9.9, but use 8QU with JAVA as the bost language.
- 9.11 Repeat Exercises 9.7 and 9.9, but use SQL/CD with Clifs the basic language
- 9.12 Repeat Exercises 9.7 and 9.9, but use (1990) with IAVA as the host language.
- Repeat Exercise 9.7, but write a function in SQL/ISM.
- 9.14 Specify the following move in SQL on the C29069 database schema shown in Figure 5.5.
  - A view that has the department name, manager name, and manager salary to overy department.
  - A view that has the employee name, supervisor name, and employee sataryter each employee who works in the 'Besearch' department.
  - c. A view that has the project name, controlling department name, number of employees, and total hours worked per week on the project for each project.
  - d. A view that has the project name, controlling department name, number d employees, and total hours worked per week on the project for each projecsoft more than one employee a erking on it.
- 9.15 Consider the following view new source defined on the connex datatese of figure 5.6.

| CREATE VIEW | DEPT_SUMMARY (D, C, TOTAL_S, AVERAGE_S)    |
|-------------|--------------------------------------------|
| AS SELECT   | DNO: COUNT (*), SUM (SALARY), AVG (SALARY) |
| FROM        | EMPLOYEE                                   |
| GROUP BY    | DNÓ.                                       |

State which at the following queries and updates would be allowed on the environment of query or update would be allowed, show what the corresponding query or update on the base relations would look like and give its result when applied it the database of Figure 5.6.

... SELECT FROM DEPT\_SUMMARY; K SELECT D.C. FROM DEPT\_SUMMARY TOTAL\_S > 100000; WHERE SELECT D. AVERAGE\_S Ľ. FROM DEPT\_SUMMARY C > (SELECT C FROM DEPT\_SUMMARY WHERE D+4). WHERE : UPDATE DEP/\_SUMMARY SET D=3WHERE D=4: FROM DEPI\_SUMMARY DELETE 2 WHERE C > 4:

# Selected Bibliography

The question of view updates is addressed by Dayal and Bernstein (1978). Keller (1982), and Langrak (1990), among others. View implementation is discussed in Blakeley et al. (1999). Negri et al. (1991) describes formal semantics of sql queries.







# Functional Dependencies and Normalization for Relational Databases

In Chapters 5 through 9, we presented vortious aspects of the relational incide, and the brances associated with its Each relation schema consists of a number of attributes, and the selectional database schema consists of a number of relation schemus. So far, we have assued that attributes are grouped to form a relation schema be using the common sense or Le catabu-e designer or by marmine a dotabase scheren Besten from a concernal data radel such as the LR OF enhanced TR (1680) of some other conceptual data model. These midel make the designer identify entity types and relationship types and their respective attibutes, which leads to a material and logical genering of the attributes into relations ther the mapping procedures in Chapter 7 are followed. However, we still need some netal measure of why one grouping of attributes into a relation schema may be better that method. So far in our discussion of conceptifal doorgo in Chapters 3 and 4 and ats impirg into the relational model in Chapter 7, we have not developed any measure of gropriateness or "goodness" to measure the quality of the design, other them the intronemotified designer. In this chapter we discuss some of the theory that has been develrest with the goal of evaluating relational schemas for design quality — that is, to measure fimally why one set of groupings of attributes are relation schemas is better than nethers

There are two levels at which we can obscurs the "goodness" of relation schemas. The fixes the logical for conceptual) level—how users interpret the relation schemas and the noning of their attributes. Having good relation schemas at this level enables users to messand clearly the meaning of the data in the relations, and hence to formulate their

queries correctly. The second is the implementation (or storage) level—how the ruples in a base relation are strined and optiated. This level applies only to schemas of bas relations—which will be physically stored as files—whereas at the logical level we are interested in schemas of both base relations and views (virtual relations). The relational database design theory developed in this chapter applies mainly to how relations, although some concerns or appropriateness also apply to views, as shown in Section 10.1.

As with many design problems, database design may be performed using we approaches borrow up or rop down. A bottom-up design methodology (also called doing by certificial) considers the basic relationships among individual arributies as the statust point and uses these to construct relation schemas. This approach is not very populatin practice<sup>1</sup> because it suffers from the problem of having to collect a large number of having telationships among individual arributies as the statust relationships among the collect a large number of having telationships among attributes as the statust point. In contrast, a top-down design methodology (also called design by analysis) starts with a number of group logs of attributes into relations that exist regether naturally, for example, on at involuce a form, or a report. The relations are thear and yield individually and collectively, leading to retho decomposition until all desirable properties are inter. The theory described in this chapter is applicable, to both the top-down and histom-up design approaches, but as maginarities with the top-down approach.

We start this chapter by informally discussing some criteria for good and bad relation schemas in Section 10.1. Then in Section 10.2 we define the concept of function? dependency, a semicilit constraint among attributes that is the main tool for fourilla une saming the appropriateness or arribbite groupings into relation schemas. Properties of functional dependencies are also seed on group articlates into relation schemas that as in a rounal dependencies are also seed to group articlates into relation schemas that as in a rounal form. A relation schema is in a normal form when it satisfies certain destable properties. The process of normaligation consists of analyting relations to main increasingly more stringent normal forms leading to progressively better geographs of attributes. Normal forms are specified in terms of functional dependencies is which are identified by the database designer—and key attributes of relation schemas. In Section 10.4 we discuss more general continuous of normal forms that can be ducify applied or any given design and do not require sup-by step multiply and normalization.

Unspirer 11 contained the development of the theory related to the design of god is binard schemas. Whereas in Chapter 12 we concentrate on the gorinal topps for single relation schemas, in Chapter 11 we will discuss measures of appropriateness for a while set of relation schemas that regether form a relational datafies, whereas We specify resuch properties the nonadditive closeless) join property and the dependence preservation property—and discuss bottom-op design algorithms for relational datafies design this start off with a given set of function d dependence and achieve certain nerical forms while materianing the atorementioned properties. A general according the tests whether or not a decomposition has the Jossless join property (Algorithms 111) to

An exception resolution approach is reaching practice is based on a model called the have relational model. An example is the MAM northodology (Webenpric and WebBekker) 1982).

ils, presented. In Chapter 11 we also define additional types of dependencies and idoaced round forms that forther enhance the "goodness" of relation schemas.

Southe reader interested in only an informal introduction to incrimalization. Sections 1975; 1023, and 1025; may be slopped of Chapter 91 is not covered in a course, we reconcerd a quick introduction to the de-arable properties of decomposition from Section 111 and a discussion of Property L11 in addition to Chapter 10.

## 10.1 INFORMAL DESIGN GUIDELINES FOR RELATION SCHEMAS

We discussion insportant measures of quality for relation schema design in this section:

- Senamics of the attributes
- Reducing the reducidant values in tuples
- Reducing the null volues in tuples
- Disallowing the possibility of generating spurious tuples

These issues are not always independent of one another, as we shall see,

#### 10.1.1 Semantics of the Relation Attributes

Whenever we group articlates to form a relation schema, we assume that attributes weiging to one relation have certain real-world diversing and a proper interpretation associated with them. In Uhapter 5 we discussed how cach relation can be interpreted as astrofflars or statements. This meaning, or semantics, specifies how to interpreted as subme calces world in imple of the relation—in other words, how the intribute values as toget the to one possible. If the conceptual design is done carefully, followed by a granula implying into relation—most of the genantics will have been accounted for additional to the relation of the genantics will have been accounted for addition reaching design should have a clear meaning.

In eached, the event it is no explain the semantics of the relation, the better the infino schema design will be. Its illustrate thes, consider figure 10.1, a simplated version after the theory relation 1 database schema of Figure 5.5, and Figure 10.2, which presents an rangle et populated relation states of this schema. The meaning of the two with relation schema squite simple. Each tople represents an employee, with colors for the employee's and cover(), social security munder (sec), both date (stort), and address (MOSTSS), and the rember of the department that the employee works for (covere). The because and sources are also straightforward schema to a foreign key that represents an implicit relationship between every factories and for also straightforward Educations to the coveres and entitier schemas are also straightforward Educations to the coveres of partment cours, and each except to be employed there is a indicate control to be account relates a project to its controlling department, behavior munager, while two of exact relates a project to its controlling department, behavior replands to movie access with which the meaning of a relation's arributes and expland to main advance of how well the relation to the sensition of a relation of a relations.



**FIGURE 10.1** A simplified corrange relational database schema.

The semantics of the other rate relation schemas in Figure 10.1 are slightly more complex. Each ruple in 0.01\_1003 (0.03 gives a department number (0.03011) and one of the locations of the department (0.03010). Each tuple in 90885\_09 gives an employee works of security momber (553), the project number of the of the projects that the employee works or (0.03559), and the number of hears per week that the employee works on that project (-0.051 However, both schemas have a well-defined and unanalognous interpretation. The schema term totation shows a real-defined attribute of DEPATIENT, whereas 6685, or represents a real-methoded attribute of DEPATIENT, whereas 6685, or represents an MIN relationship between ENPLOSEF and 680067. Hence, all the relation schemas in Figure 10.1 and be relation at heaving during the analytes at the explain and hence good from the standpoint of having during the analytes. We can thus formulate the callowing interpretation and heaving and the standpoint of having during the analytes. We can thus formulate the callowing interpretation and heaving and the standpoint of having during the relation at the standpoint of having during the method design guideline.

**GUIDELINE 1.** Design a relation schema so that it is easy to explain its meaning. Its not conduct at informs from multiple centry types and relationship types into a single relation. Intuitively, it a relation schema corresponds to one entity type or one relation.

i

| ENAM                   | HE S            | SSN        | BDATE      | ADD            | RESS         | CNUM   | ØEA   |        |
|------------------------|-----------------|------------|------------|----------------|--------------|--------|-------|--------|
|                        | · ·             |            | 1966-01-09 | 201 Ecologica  | Housson TX   | ' s    |       |        |
| Wane Fornik            |                 |            | 1976 12:09 | ESP Mass Ha    | ston TX      | Ē      |       |        |
| 7olana deepa           |                 | 0897777    | 1068-07-10 | - 2021 Carde 2 | States TX    |        |       |        |
| Vblace Jere            | nders 98        | 065432     | 1941-06-20 | 201 Rami Ral   | laura T¥     | 3      |       |        |
| kare to Be             | meshK m≥        | RoBaaau    | 1968-CN-15 | H/MEre Cok     | Filming Fig. | 5      |       |        |
| Froish source          | nA 49           | 3453453    | 972-07-31  | 5531 Roe Ho    | esion TX     | 5      |       |        |
| lettra Alem            | alV 9€          | 2907907    | 964-03-29  | 390 Dalas He   |              | d      |       |        |
| Rore James             | E 85            | 9565555    | 987-11-10  | 450 Since Ho   | NECOD TX     | 1      |       |        |
|                        |                 |            |            |                |              | DEPT L | OCATI | ONS    |
| r                      | EPARTMENT       | ·          |            |                |              |        |       |        |
| DNAME                  | <u>DNUMBE</u> R | DWGRSSI    | N          |                | Ľ <u>¤</u>   | NUMBER | CLO   | CATION |
| I                      |                 | ·          |            |                |              | 1      | Hou   | 19000  |
| Research               | 5               | 383445555  |            |                |              | 4      | Sal   | tod    |
| Attravelyation         | i 4.            | 987034321  |            |                |              | 5      | Bel   | bire   |
| Headquartere           | . 1             | 688665555  |            |                |              | 5      | - 5ug | arland |
|                        |                 |            |            |                |              | 5      | Hou   | 15120  |
| WC                     | RKS_ON          |            |            | PE             | NMECT        |        |       |        |
| SSN                    | PNUMBER         | HOUR\$     | -<br>.     |                | PNUMBER      | PLOCA  |       | DNUM   |
| 125626789              | 1               | 32.6       | - I        |                | •            | 1      |       |        |
| 25456739               | 2               | 7.5        |            | ProductX       | 1            | Belar  | ne –  | 5      |
| 666684444              | 3               | 45.5       |            | ProductY       | 5            | Sugar  | tand  | 5      |
| 455453460              | 1               | 20.0       |            | ProductZ       |              | Houst  | 00    | 5      |
| 453453453              | 2               | 25.0       |            | Computenzatio  | n 10         | Staffe | rđ    | 4      |
| 33445565               | 2               | 10.0       |            | Reorganization | 20           | Houst  | 00    | 1      |
| 323445265              | 3               | 100        |            | Newboriefts    | 30           | Satio  | Æ     | 4      |
| 333445565              | 10              | 10.0       |            |                |              |        |       |        |
| 2334435555             | 20              | 10.0       |            |                |              |        |       |        |
| 999927777              | 30              | 30.3       |            |                |              |        |       |        |
| 99857777               | 10              | 10.0       |            |                |              |        |       |        |
| 927337987              | טו              | 35 0       |            |                |              |        |       |        |
| 987987987              | 30              | 6.0        |            |                |              |        |       |        |
|                        | ~ ~             |            |            |                |              |        |       |        |
| \$87654321             | 30              | 200        |            |                |              |        |       |        |
| 987654321<br>987554321 | 30<br>20        | 200<br>150 |            |                |              |        |       |        |

#### EMPLOYEE

**RCORE 10.2** Example database state for the relational database schema of Figure 10.1.

stas type in a straightforward to explore its meaning. Otherwise, if the relation correspeak to a nuistore of inditiple entries and relationships, scinantic and ignities will result ad the relation cannot be easily explained.

The relation schemas in Figures 10.3a and 10.3b also have clear smiantics. (The naio should ignore the lines index the relations for pow, they are used to illustrate forminal dependency purchase, discussed in Section 10.2.) A ruple in the project



FIGURE 10.3 Two relation schemas suffering from update anomalies.

relation schema of Figure 10.35 represents a single employee but includes additional information—namely, the name (1999) of the department for which the employee work and the social security number (1999) of the department for which the employee work relation of Loure 10.36, each ruple relates an employee to a project but also includes the employee name (1999), project name (1999), and project location (1999). Although there is nothing wrong logically with these two relations, they are considered poor designs because they could be found to be unployees and departments, and employee mode employees attributes of employees and departments, and employee increasing the use of employees and departments, and employee mode employees attributes of employees and departments, and employees attributes of employees and departments, and employees when used as base relations, as we discuss in the following section.

### 10.1.2 Redundant Information in Tuples and Update Anomalies

One goal of schema design is to minimize the storage space used by the base relations (and hence the corresponding files). Grouping arithmets into relation schemas has a significant effect on storage space. For example, compare the space used by the two has relations interval and a satisfiest in Figure 12.2 with that for an twe\_are base relation in Figure 10.4, which is the result of applying the NATOR MULTAR operation to two set and termstead in ter\_are, the attribute values performing to a part color department (2009), there is descend an expected for every employed to take for dual department (2009), there is department is information appears only once in the pressure relation in Figure 10.2 Only the department number (2009) the reported in the pressure relation for such employee who works in that department Similar comments apply to the twe\_set relation (Figure 10.4), which augments the source optimizer or path additional attributes from PROFE and PROFE and PROFE to the source for the two apply to the two sources for (Figure 10.4), which augments the source of relation with additional attributes from PROFE and PROFE and PROFE

Houston

Housian



**FROME 10.4** Example states for the cert and two reat resulting from applying NATURAL OIN to the majors in Figure 10.2. These may be stored as base relations for performance reasons.

Recruition

Boorgen@ation

Another senious problem with using the relations in Figure 10.4 as base relations is the problem of update anomalies. These can be classified into insertion anomalies, deteor anomalies, and modification anomalies.<sup>3</sup>

Wallace Jernier G

Borgulanies E

lisertion Anomalies. Insertion anomoles can be differentiated into two types distantly the following examples based on the searcest relation.

 To essent a new employee tople into SMP DSPT, we must include either the attribute values for the dependent that the employee works for, or nulls (if the employee does not work not ad partment as yet). For example, to insert a new tuple for an employee who works in department number 5, we must enter the artribute values of department 5 correctly so

37654721

100006666

20

20

15.0

nu)

These monorhabes were identified by Godd v1972 (0 to justify the need to) formalization of relanors, is we shall discuss in Section 10.3.

that they are consistent with values for department 5 in other ruples on EMP\_000T. In dedesign of Figure 10.2, we do not have to avoiry about this consistency problem because we enter only the department mumber in the employee cuple; all other attribute values of department 5 are recorded only once in the database, as a single ruple in the network relation

• It is difficult to insert a new department that has no employees as yer in the secon relation. The only way to do this is to place null values in the attributes for employee. This causes a problem because sign is the primary key of SP\_DEPT, and each topk is supposed to represent an employee entity — not a department entity. Moreover what the first employee is assigned to that department, we do not need this topic with rall values any more. This problem does not occur in the design of Engine 10.2, because a department is concreted in the NEORTEEST relation whether or not any employees and for ir, and whenever an employee is assigned to shat department, a corresponding implet is inserted in process.

Deletiton Automatics. The problem or deletion anomales is related to the score insertion anomaly situation discussed earlier. If we delete from EAP (62) on employee tighthat happens to represent the last employee working for a particular department, its information concerning that department is lost from the database. This problem does no occur in the database of Figure 10.2 because newspect topics are stored separately.

Mortification Anomalies. In use\_upp, if we change the value of one or the arribust or a particular detaitment—say, the manager of department 5—we must update the tapes of all employees who work in that department, otherwise, the database will become inconsistent. If we full to update some tuples, the same department will be showly to pay two different values for manager in different employee tuples, which would be wreng <sup>5</sup>

flosed on the preceding three anomalies, we can state the guideline that follows

**GUIDELINE 2.** Design the base relation schemes so that no insertion, deletion a module are present in the relations. If any aroundles are present, note that clearly and make sure that the programs that update the database will operate concern;

The second guideline is consistent with and, in a way, a restatement of the first guideline. We can also see the recel for a more formal approach in evaluating whether a design meets these guidelines. Sections 10.2 through 10.4 provide these needed formal concepts. It is important to note that these guidelines may sometimes have to be oblated in order to interval. The performance of certain queries. For example, it an important gas terrieves information concerning the department of an entrilowee along with employee artributes, the two dots schema may be used as a base relation. However, the anomates in eveloper must be noted and accounted for (for example, by using tragets or stars) procedures that would make automatic epdates) so that, whenever the base relation is updated, we do not easilop with inconsistencies in general, it is advisable to use anomale free base relations and to specify verses that include the poins for placing freeiber the

5. This is not as services is the other problems, by a per all trades can be applied by a single SQL and

inductes frequently referenced in important queries. This reduces the burnber of (00% tents spectred in the query, making it simpler to write the query correctly, and or many uses it improves the performance.<sup>1</sup>

### 10.1.3 Null Values in Tuples

insume schema designs we may group many attributes regerier into a "ter" relation. It many of the armbares do not apply to all tuples in the relation, we end up with many nolls in dosingles. This can waste space at the storage level and may also lead to problems with indexarding the meaning of the attributes and with spontying 6.0% operations of the lag calleeel." Another problem with bulls is how to account for their when suggregate operations of a COUNT or 40.50 are applied. Moreover, nulls can have multiple interpretations, while the following:

- The orrobate does not optify to this tuple.
- The artrobute value for this tuple is suknoice.
- The value is known but absence that is, it has not been recorded yet

Home the same representation for all nulls compromises the different meaningbeen as nave. Therefore, we may state another guideline.

**GUIDELINE 3.** As for as possible, avoid placing attributes as a base relation whose wars now frequently be pull. It null- are universable, make sure that they apply in original cases only and do not apply to a majority of tuples in the relation.

Usual space efficiently and avoiding joins are the two overriding enteria that demone whether to include the columns that may have ranks in a adation or to have a squate relation for these columns (with the appropriate key columns). For example, it only 0 percent of employees have individual offices, there is little pistification for including analytic articles (rate, a column for including analytic articles (rate, a column for including charter) controls of the events of relation of the engloyees with individual offices.

## 10.1.4 Generation of Spurious Tuples

Consider the two relation is beings (we lock and ever 9001) in Figure 10 50, which can be used instead of the single -we real relation of Figure 10.3b. A tuple in the cock means that the inplote whose name is more works on some paget whose location is place, for A tuple

<sup>4.</sup> The performance of a query specified over view that is the print of some all cose relatings depends where the DEV samplements the Wew Marry ROPARSS numericative a respective used were so that the doubt have to perform the prints from The DRVs terminations of oppiphly for opsiloting the materiatized on the manusclastely of periodically behavior of the base relations as upstated.

<sup>5.</sup> Bus also, one operated outer protections produce different results when mults are recorded in pares. The service of the last aware of the different involutions of the vertices types of parts. Although the stars calle for suphrstreaded users at most heid incoding others.



| Smith Jone B         | Belare    |
|----------------------|-----------|
| Smith John B         | Sugarland |
| Narayan, Ramesh K    | Houston   |
| English Jayee A      | Belave    |
| Engish Jayce A       | Sucarland |
| Wong, Franklin T     | Sugartand |
| Wong, Frankliv T     | Houston   |
| Wong, Franklin T     | Staford   |
| Zelava, Alipa J      | Siafore   |
| Jebbar, Ahmad V      | Siaflort  |
| Wallace, Jennitar S. | Station   |
| Walace, Jermfer S.   | Houston   |
| Som James E.         | Houston   |

#### EMP PROJ1

| SSN        | PNUMBER    | HOURS | PNAME           | PLOCATION |
|------------|------------|-------|-----------------|-----------|
| 123456789  | - <u> </u> | 32.5  | Product X       | Belare    |
| 12.4466789 | 2          | 75    | Product Y       | Supadand  |
| 6662454444 | 3          | 40.0  | Product Z       | Houston   |
| 453453453  | 1          | 20.0  | Product X       | Betare    |
| 153453463  | 2          | 20.0  | Product Y       | Sucadano  |
| 333445555  | 2 E        | 100   | Product Y       | Sugarland |
| 333445555  | ŝ          | 10.0  | Product Z       | Housion   |
| 333445555  | IG         | 101.0 | Computenzation  | Staford   |
| 333445555  | 20         | 10.0  | Reorganization  | Housing   |
| !XXXXE7777 | 30         | 30.0  | Newbenefits     | Stafoct   |
| 999687777  | 10         | 10.0  | Computerization | Stational |
| 96796/967  | 10         | 35.0  | Computerization | Stations  |
| 987987687  | 30         | 50    | Newbanehis      | Stational |
| 987654321  | 30         | 23.0  | Nowbenetits     | Slaford   |
| 597654321  | 20         | 15.0  | Reorganization  | Houston   |
| 6999665665 | 20         | nul   | Reorganization  | Housion   |

**FIGURE 10.5** Particularly poor design for the eveletor relation of Figure 10.3b. (a) The two relation schemas eveletists and eveleton. (b) The result of projecting the extension of telleron from Figure 10.4 onto the relations eveletors and eveleton.

in an exact means that the employee whose social security number is 558 works mass perweekon the project whose name, number, and location are 2660, 2609000, and ecocycles. Bigore 10 bb shows relation states of exe\_tool and exe\_proof corresponding to the exe\_proof relational Figure 1.2.4, which are obtained by applying the appropriate PROJECT (75) operations sock\_sec: (ignore the dotted lines in Figure 10 56 for now).

Suppose that we used we\_exc21 and eve\_tues as the base tellations instead of see\_evol. This produces a particularly bad schema design, because we cannot recover the excansion that was originally in eve\_evol from eve\_evol1 and eve\_tots. If we attempt a NATCHALION operation on eve\_evol1 and eve\_tots, the result produces many more toples that the original set of toples in twe\_evol1 and eve\_tots. The result of applying the part to only the toples above the dotted lines in Figure 10.5b is shown two reduces the size of the usulting relation). Additional toples that were not in eve\_evol are called spurious toples toples above they represent spurious or noting information that is not vulid. The spurious toples are marked by asterisks (\*) in Figure 10.6.

Decomposing top\_PRC1 into EVP\_CCCS and top\_PRC1 is undesirable because, when we puscher back using NATE/TOAL WIN, we do not get the correct original information. This elevance in this used "ECONTION is the attribute that relates CVP\_CCCS and EVP\_CVD\_ and elevans is nearbar a primary key not a foreign key in either CVP\_CVCS or top\_EVPLOT. We can pw\_pt/traally stars another design guideling.

| SSN                     | PNUMBER | HOURS | PNAME          | PLOCATION    | ENAME              |
|-------------------------|---------|-------|----------------|--------------|--------------------|
| 23458709                | 1       | 32.5  | ProductX       | -<br>Bellare | Smithubhn B.       |
| 123456763               | 1       | 32.5  | ProductX       | Bellare      | English, Joyce A.  |
| 123456769               | 2       | 7.5   | ProductY       | Sugarland    | Smith John B.      |
| 125466789               | 2       | 7.5   | ProductY       | Sugarland    | English, Joyce A.  |
| 123456789               | 2       | 7.6   | ProductY       | Sugarland    | Word Franklin T.   |
| 69 <del>6/1</del> 84444 | 3       | 40.0  | ProductZ       | HOLSKHI      | Narayan, Ramesh K. |
| 003384444               | а       | •C.O  | ProductZ       | Hauston      | Wong Franklin T.   |
| 463453453               | 1       | 20.0  | ProductX       | Beitare      | Sminuchn B         |
| 100400463               | 1       | 20.0  | ProductX       | Bellaro      | English, Jayoe A.  |
| 4534534853              | 2       | 20.0  | ProductY       | Sugarland    | Smith, John B      |
| 453463453               | 2       | 20.0  | Productin      | Sugarland    | Enclish, Jryce A   |
| 453463453               | 2       | 20.0  | Productr?      | Sugartand    | Wang Hankin F      |
| 2334455555              | 2       | 10.0  | Product?'      | Sugartand    | Sreth John B       |
| 133445685               | 2       | 10.0  | ProductY       | Sugarland    | English, Joyce A   |
| 1334455555              | 2       | 10.0  | ProductY       | Sugartand    | Word Frankin T     |
| 111445665               | 3       | 10.0  | ProntivitZ     | Haviston     | Naravar, Ramssh K  |
| 1134456855              | 3       | 16.0  | ProductZ       | Houston      | Wong Franklin T    |
| 113445655               | 10      | 100   | Computenzation | Stafford     | Wong Franklin T    |
| 3134455555              | 20      | 10.0  | Representation | Houston      | Narayan, Ramash K. |
| 183445655               | 20      | 100   | Reorganization | Houston      | Wong, Frenklin T   |
|                         |         |       |                | -            |                    |

**RGURE 10.6** Result of applying NATURAL ICHN to the tuples above the dotted lines in EVE\_SHOLL and SPLACS of Figure 10-5. Generated spurious tuples are marked by asterisks. **GUIDEUNE 4.** Design relation schemas so that they can be joined with equival conditions on attributes that are either grimary keys of foreign keys in a way that guarantees that no spurious tuples are guaranteed. Avoid relations that contain matching attributes that are not (foreign key, primary key) combinations, because joining on sea attributes tain psoduce spurious tagles.

This informal guideline obviously needs to be stated more formally. In Chapter 11 we obvious a formal condition, called the nonaddrive (or hissless) join property, that guarantee that certain joins do not produce sporious tagles.

#### 10.1.5 Summary and Discussion of Design Guidelines

In Sections 10.1.4 through 10.1.4, we informally docessed structions that lead to prolematic relation schemas, and we proposed informal guidebres for a good relatoral design. The problems we pointed out, which can be detected without sedimonal task a analysis, are us follows:

- Amomorales that cause redundant work to be done during insertion into and meditarion of a relation, and that may cause accidental loss of information during a deleter from a relation.
- Waste of storage space due to nulls and the difficulty of performing aggregation over atoms and provide ro null values.
- Generation of invalid and sparious data during joins on improperly related for relations.

In the rest of this chapter we present formal concepts and theory that may be used a define the "goodness" and "badness" of individual relation schemas more precisely. We fix discuss functional dependency as a tool for analysis. Then we specify the three nergitorins and Boyce-Codd normal form (BCW) for relation schemas. In Chapter 11, we define additional normal forms that which are based on additional types of data dependences ralled multivalued dependencies and jum dependencies.

# **10.2 FUNCTIONAL DEPENDENCIES**

The single most important concept in relational schema design theory is that of a brational dependency. In this section we form ally denne the concept, and in Section 103 we see how it can be used to define normal forms for relation schemas.

## 10.2.1 Definition of Functional Dependency

A functional dependency is a constraint between two sets of attributes from the database Suppose that car relational database schema has a attributes  $A_1, A_2, \ldots, A_n$ ; let us fail of the whole database as being described by a single **universal** relation schema  $\delta \in \mathbb{N}_{0}$ .  $A_{g_1} = A_{g_1}$ . We do not imply that we will actually store the database as a single universitially we use this concept only in developing the formal theory of data dependencies<sup>2</sup>

**Definition.** A functional dependency, denoted by  $\lambda \to Y$  between two sets of ambures X and Y that are subsets of *R* specifies a conservation the possible raples that can four a relation state *i* of *R*. The constraint is that, for any two tuples  $r_1$  and  $r_2$  in *i* that face:  $[X] = r_2[X]$ , they must also have  $r_1[Y] = r_2[Y]$ .

This means that the values of the Y-component of a tople in Y-depend on, or are *exercised by*, the values of the X-component alternatively, the values of the X-component example inquely for functionally) determine the values of the Y-component. We also say that there is a functional dependence from X-to Y, or that Y is **functionally dependence** or X. The althousant for functional dependency is three for fully dependence X is called the signature of although of the FU, and Y is called the right-hand side.

Thus, X functionally determines Y in a relation scheme R it, and only if, whenever iso taples of i(R) agree on their X-value, they must necessarily agree on their Y-value. Not the todosenge

- Ta constraint on R states that there cannot be more than one tuple with A given Xvalue in any relation instance v(R)—that is, X is a candidate key of R—this implies dot X → Y for any subset of attributes Y of R (because the key constraint implies that no two tuples in any legal state i(R) will have the same value of X3.
- If  $X \to Y$  in R, this does not say whether or not  $Y \to X$  in R.

Admittional dependency is a property of the semantics or meaning of the attributes. The database designees will use their understanding of the semantics of the attributes of  $B_{\pm}$  that is how they relate to one another — to specify the functional dependencies that which hold on all relation states (extensions)  $\tau$  of  $B_{\pm}$ . Whenever the semantics of two sets afornibutes in R inductes that a functional dependence should hold, we specify the dependency as a constraint. Relation extensions  $\tau(R)$  that varisfy the functional dependency as a constraint. Relation extensions  $\tau(R)$  that varisfy the functional dependency constraints are called legal relation states for legal extensions) of R. Hence, the run use of functional dependencies is to describe further a relation schema E by people constraints on its artification that us a property of those attributes. For earthe, (state, provide that certain functional dependences may done in the United faces it is also possible that certain functional dependences may cease to exist in the run value of the relation-functional functional dependences may cease to exist in the run value for the relation-functional dependences may cease to exist in the run value for the relation-functional dependences may cease to exist in the run value for the relation-functional dependences may cease to exist in the run value for the relation-functional codes and telephone runniber codes in the United facts by whethe proliferation of telephone area codes it is no longer (me

f The suncept of a number-al relation is only start when we discuss the algorithms to trelational know dauge in Obseter 11

<sup>&</sup>lt;sup>1</sup> The assumption implies that every arribute in the database should have a database for Origin has protoed a tribute names by relation names reaching constrained whereas whereas a strubutes adams income had the same name.

Consider the relation schema (w) was in Figure 10.3b, from the semantics of the arrobants, we know that the following functional dependences should hold:

```
4. 559 --- FRAME
```

```
by ensuring \to (inside , length rule)
```

 $\gtrsim$  (55%, 250MKEK)  $\rightarrow$  HOURS

These functional detendencies specify that (a) the value of an employee's social security number (sss) uniquely determines the employee name (suss), (b) the value of a project's number (suss) uniquely determines the project name (suss)) and location (subsystes), and (c) o combination of six and subsets values uniquely determines the number of hours the employee currently works on the project per week (suss). Alternatively, we say that susse is functionally determined by (or functionally dependent on) sixe or "seven a value of six, we know the value of sixes" and so en.

A functional dependency is a property of the velation is beaut R, not of a particular legal relation state if of R. Hence, and D causar be informal antimatically from a given relation extension is but must be defined explicitly by some meridation who knows the semantics of the attributes of R. For example, Figure 10.7 shows a particular state of the relation schema. Although a first glaner we may think that  $r(x) \rightarrow rates, we cannot confirm this anless we know that it is true for all possible legal singles of respectively. For example, there is disprove a functional dependency. For example, because Smith' teaches both 'Data Structures' and 'Data Management', we can conclude that it was slow to x functionally determine coust.$ 

Figure 10.3 introduces a diagrammatic notation for displaying FDs. Each 30 is displayed as a horizontal line. The left-hand-side attrabutes of the FD are connected to vertical lines to the line representing the FD, while the right-hand-side attrabutes are connected by proves pointing toward the attrabutes, as -town in Figure-10.3a and 10.3k

#### 10.2.2 Inference Rules for Functional Dependencies

We denote by F the set of functional dependencies that are specified on relation schema R. Typically, the schema designer specifies the functional dependencies that are senare only obvious; usually, however, numerous other functional dependencies hold in all legal relation instances that unitly the dependencies in  $\Gamma$ . Those other dependencies can be informed or deduced from the  $\pi$  is in E.

| TEACH   |                 |            |
|---------|-----------------|------------|
| TEACHER | COURSE          | TEXT       |
| Smath   | Dala Siructures | Barnein    |
| Smph    | Cala Management | AFNOUL     |
| Hşli    | Complex         | Holinian   |
| Brown   | Data Structures | Augenhaler |

**FIGURE 10.7** A relation state of usor with a possible functional dependency as — coase. However, respective coase is ruled out

In real ate, it is improvable to specify all possible fonctional dependencies for a given subarsed. For example, if each dependencies has one managed so that  $\text{DEPT}_{k0}$  uniquely determines  $\text{MARGES}_{k0}$  (DEPT,  $\text{NO} \rightarrow \text{NOE}_{k0}$ ), and a Mattager has a unique phone number called  $\text{MAE}_{k0}$  ( $\text{ME}_{k0}$ ),  $\text{ME}_{k0}$  ( $\text{ME}_{k0}$ ), then these two dependencies together might that  $\text{MAE}_{k0}$  ( $\text{ME}_{k0}$ ). This is an inferred  $\text{ME}_{k0}$  and lead to be explicitly stated in addition to the two given site. Therefore, formally it is useful in define a concept called closure that induces all possible dependencies that can be inferred from the given set H.

**Definition.** Formally, the set of all dependencies that include *F* as well as all kpondencies that can be inferred from *F* is called the closure of F; it is denoted by *F*?

For example, suppose that we specify the following set F of obvious functional approximates on the relation schema of France 12.3.5.

```
F = \{ss_{2} \rightarrow (ename, scate, address, orderer), \\ delineer \rightarrow (delane, ordersyn)\}
```

Sene of the additional functional dependencies that we contride from Plate the following:

```
ESH → {DNAME, DMCRSSN}
554 → SSN
D404BER → DNAME
```

An H (  $X \to Y$  is inferred from a set of dependencies F-positive developments in  $X \to Y$  holds in any liquindation state  $\tau$  of  $\mathcal{E}$  that is, whenever insatisfies all the dependencies in  $F, X \to Y$ ds holds in  $\tau$ . The closure F' of F is the set of all functional dependencies that can be introd from F. To determine a systematic way to infer dependencies, we must discover a set of inference index that can be used to infer new dependencies from a given set of dependencies. We consider some of these inference rules next. We use the notation F = X $\rightarrow h$  rodencies that the functional dependency  $X \to Y$  is inferred from the set of functional dependencies F.

In the following discussion, we use an oblycenated notation when discussing functional dependencies. We concurrence attribute variables and drop the communities for ameniance. Hence, the FD  $\{X,Y\} \rightarrow \downarrow$  is abbreviated to  $XY \rightarrow Z$ , and the FD  $\{X,Y,\downarrow\} \rightarrow \downarrow$  is abbreviated to  $XY \rightarrow Z$ , and the FD  $\{X,Y,\downarrow\} \rightarrow \downarrow$ ,  $X_i$  is abbreviated to  $XYZ \rightarrow XZ$ . The following six rules is 1 through attribute wells known inference rules for functional dependencies.

s) theflexive rade<sup>3</sup>). If X ⊇ Y, over X → Y, ev2 togenerization rule<sup>3</sup> is  $\{X \to Y\} \vdash XZ \to YZ$ (3) (transitive rule)  $\{X \to Y, Y \to Z\} \vdash X \to Z$ . (3) (decomposition, or projective, rule)  $\{X \to YZ\} \vdash X \to Y$ .

S The prior case take can also be stated as  $X \mapsto X_i$  that is, give set of attributes bineticnally generative t

A Die regeneration of solar data be stated as  $\{X \to Y\} = X\Sigma \to Y$ , that is bugging using the leftfamilial spin brites of a Coperdiance method solar. (R5 (armon, or additive, rule)  $\{X \rightarrow Y, X \rightarrow Z\} \models X \rightarrow YZ$ . (R6 (pseudotransitive rule)  $\{X \rightarrow Y, WY \rightarrow Z\} \models WX \rightarrow Z$ .

The reflexive role (R11) states that a set of attributes obviow determines itself or any efficiencies, which is obvious. Because IRI generates dependencies that are always frue such dependencies are called *wreal*. Formally, orfunctional dependency  $X \to Y$  is trivial if  $X \supseteq Y$  otherwise, it is nontrivial. The argumentation rule (R2) says that adding the same set of attributes to both the latis and right-hand sides of a dependency results in another valid dependency. According in (R3), functional dependencies are transitive. The decomposition rule (R4) says that we can remove attributes from the right-hand side of a dependency results in another valid dependency. According in (R3), functional dependencies are transitive. The decomposition rule (R4) says that we can remove attributes from the right-hand side of a dependency; applying this rule repeatedly can decompose due fto  $X \to \{\Delta_1, \Delta_2, \ldots, A_n\}$  into the set of dependencies  $[X \to A_1, X \to A_2, \ldots, X \to A_n]$ . The anisomule (R5) allows us to do the exposite) we can remove a dependencies  $[X \to A_1, X \to A_2]$  into the singly FD  $X \to \{\Delta_1, \Delta_2, \ldots, A_n\}$ .

Our cautionary note regarding the use of these rules. Although  $X \to A$  and  $X \to B$  amplies  $X \to AB$  by the mann rule stated above,  $X \to A$ , and  $Y \to B$  does not imply that  $XY \to AB$ . Also,  $XY \to A$  does not necessarily imply either  $X \to A$  or  $Y \to A$ .

Each of the preceding inference onles can be proved from the definition of functional dependency, other by direct proof or **by contradiction**. A proof by contradiction assume that the rule does not build and shows that this is not possible. We now prove that the first three rules (R1 through (R3) are valid. The second proof is by copyradiction.

#### PROOF OF 181

Suppose that  $X \supseteq Y$  and that two toples  $i_1$  and  $i_2$  must in wome relation instance for R such that  $i_1[X] \models i_2[X]$ . Then  $i_2[Y] \models i_2[Y]$  because  $X \supseteq Y$ , hence,  $X \to Y$  must head in  $i_2$ 

#### PROOF OF 1K2 (BY CONTRADICTION)

Assume that  $\lambda \to Y$  holds in a relation instance  $i \in R$  but that  $XZ \to YZ$  does not hold. Then there must exist two toples  $t_i$  and  $t_j$  is z such that  $\{1\} \in_i [X] = z [X], (Z) \in_i [YZ], (Y) \in_i [XZ] = t_j [XZ], and <math>(4) \in_i [YZ] \neq t_j [YZ]$ . This is not possible because turn (1) and (3) we deduce (5)  $t_j [Z] = t_j [Z]$ , and from (2) and (5) we deduce (6)  $t_j$  $[YZ] = t_j [YZ], contradicting (4).$ 

#### PROOF OF IR3

Assume that (1)  $X \to Y$  and (2)  $Y \to Z$  both hold in a teleform r. Then for any two suples  $t_i$  and  $t_j$  in a such that  $i : |X| = t_j |X|$ , we must have  $(3) |t_j||Y| = t_j |Y|$ , from assumption (1); hence we must also have  $(3) |t_j||Z| = t_j |Z|$ , from (3) and assumption (2); hence  $X \to Z$  must hold in (.)

Using similar priori organizates, we can prove the interence rules (R4 to IR6 and ony additional valid interence rules. However, a simpler way to prove that an interence rule for functional dependencies is valid is to prove it by using inference rules that have alende been shown to be valid. For example, we can prove 184 through 186 by using 164 through 186 by using 164 through 183 as follows.

#### PROOF OF IR4 (USING IR1 THROUGH IR3)

- 1,  $\mathbb{X} \to \mathbb{C}$  (given).
- 1. YZ  $\rightarrow$  Y (using (R1 and knowing that YZ  $\supseteq$  Y).
- $\mathbb{V} X \to \mathbb{V}$  (using 0.3 or 1 and 2)

#### PROOF OF IR5 (USING IR1 THROUGH IR3)

- $i: X \rightarrow Y (green).$
- $\mathbb{P}(X \to \mathbb{D} \text{(piven)})$
- $i: X \to XY$  (using 182 cm 1 by asymmetring with X: notice that XX = X).
- 4. N  $\rightarrow$  YZ (using 182 on 2 by augmenting with Y).
- $\mathbb{E}[\lambda] \rightarrow \mathbb{V}\mathbb{E}$  (using its  $\mathbb{E}$  on  $\mathbb{E}$  and  $\mathbb{A}$ ).

#### PROOF OF IR6 (USING IR1 THROUGH IR3)

- N + eY (given).
- $\downarrow WY \rightarrow Z$  (given).
- i  $\mathfrak{WX} \to \mathfrak{WY}$  (using 182 on 1 by augmenting with  $\mathfrak{W}$ ).
- 4 @X -> Z (using US) on 3 and 25.

It has been shown by Armstrong (1974) that inference rules (3) through (R) are stand and complete, by **sound,** we mean that given a set of functional dependencies F specified on a relation schema R, any dependency that we can infer from f by using (R) though R5 holds in every relation state  $\tau$  of R (for satisfies the dependencies in f). By complete, we mean that using (0) through (R3 repeatedly to infer dependencies until no incomplete, we mean that using (0) through (R3 repeatedly to infer dependencies until no incomplete, we mean that using (0) through (R3 repeatedly to infer dependencies until no incomplete, we mean that using (0) through (R3 repeatedly to infer dependencies until no incomplete, we mean that using (0) through (R3 repeatedly to infer dependencies until no incomplete, we mean that using (0) through (R3 repeatedly to infer dependencies until no incomplete, we mean that using (0) through (R3 repeatedly to infer dependencies until no incomplete, we mean that using (0) through (R3 repeatedly to infer dependencies f, which we called the closure of F, can be determined from F by using only inference rules (R1 through (R3 inference rules (0) through (R3 are known as **Armstrong's inference rules**).<sup>17</sup>

Typically, database designers first specify the set of functional dependencies F that can easily be determined from the semantics of the attributes of R; then .00, 002, and 065 are used inner additional functional dependencies that will also hold on R. A systematic way in demane these additional functional dependencies is first to determine each set of attributes X(a) appears as a left duard side of some functional dependency in F and then in determine the set of all attributes that are dependent on X. Thus, for each set of attributes X, we determine the set  $X^*$  or attributes that are functionally determined by X based on F:  $X^*$  is called the closure of X under F. Algorithm 10,1 can be used to calculate  $X^*$ .

<sup>1)</sup> They are actually known as **Armstrong's axioms**. In the struct northernatical sense the anomalizations the functional dependencies in F, since we assume that they are correct, whereas F, this gives in the end of the angle of the structure of the transmission of the presidencies (new tests).

```
Algorithm 10.1: Determining X^*, the Closury of X under F(X^*) = X;
repeat
old X^* := X^*:
for each functional dependency Y \to Z in F do
if X^* \supseteq Y then X^* := X^* \cup Z_s
angil (X^* = ald X^*);
```

Algorithm 10.1 starts by setting X<sup>+</sup> to all the attributes in X. By (R1, we know third, these attributes are functionally dependent on X. Using interactor rules (R3, and (R4, we add attributes to X<sup>+</sup>, using each functional dependency in E. We keep going abrough all the dependencies in F. (the repeat loop) until no more attributes are added to X<sup>+</sup> damy: consider cycle (of the for loop) transphate dependencies in F. For example, consider the relation schema EPP\_PRO, as Figure 10.50, from the sensitives of the attributes, we specific following set F of functional dependencies that should bold on  $CP_PRO$ :

```
F = \{ssn \rightarrow ehave, \\ Fhankfk \rightarrow \{rhankf, P, (cation), \\ \{sch, Pn(4880) \rightarrow hour()\}
```

Using Algorithm 10.1, we calculate the following closure sets with respect to F:

Into invelve the set of artributes in the right-hand side of each line represents all thus artributes that are functionally dependent on the set of attributes in the left-hand side based on the given set F.

#### 10.2.3 Equivalence of Sets of Functional Dependencies

In this section we discuss the equivalence of two sets of functional dependencies. First, at give some preliminary definitions.

**Definition.** A set of functional dependencies E is such to cover another set d functional dependencies E if every TP in E is also in  $\Gamma^{n}$ ; that is, if every dependency in Ecan be inferred from F; alternatively, we can say that E is covered by E.

**Definition.** Theo sets of turn normal dependencies E and F are equivalent if  $E^* = F$ . Hence, equivalence means that every  $E^*$  in E can be interact from E and every  $E^*$  in E can be interact from E and every  $E^*$  in E can be interact from E and every  $E^*$  in E can be interact from E overs F and F covers E hold.

We can determine whether F covers E by calculating  $X^*$  with respect to F for each  $F^*$  $X \rightarrow Y$  in E, and then checking whether due  $X^*$  includes the articlasus p(Y). If this is re-
costos energias in El shen El covers El We determine whether E and E are controlent by checking that E covers E and E covers E.

### 10.2.4 Minimal Sets of Functional Dependencies

Internally, a **minimal cover** of a set of functional dependencies E is a set of functional dependencies F that satisfies the property that every dependence in E is in the closure  $F^{+}$  of F in addition, this property is lost if any dependency from the set F is removed; F must have no redundancies in  $\pi$ , and the dependencies in E are in a standard form. To satisfy these properties, we can formally define a set of functional dependencies F to be **minimal** if a subject to the following conditions:

- 1. Every dependency in F has a single attribute for its nehr-hand side.
- 2. We cannot replace any dependency  $X \rightarrow A$  in U with a dependency  $Y \rightarrow A$ , where Y is a proper subset of X, and still have a set of dependencies that is equivalent to b
- We cannot remove any dependency from F and still have a set of dependencies that is equivalent to I.

We can think of a contained set of dependencies as being a set of dependencies in a statished organomial joint and with no relandances. Condition 1 just represents every dependency in arounded form with a single statished on the right-hand side,<sup>10</sup> Conditions 2 and 3 cleane hardene are no redundancies in the dependences of thirdy. Eaving redundant attributes or the fetch and side of a dependences. (Condition 2) or by having a dependency that can be infined from the remaining (it is in F (Condition 2)) or by having a dependency that can be infined from the remaining (it is in F (Condition 3)). A minimal cover of a set of functional dependencies F is a minimal set of dependencies F that is repredent to F. There can be sevral parental cover F for a set of functional dependencies. We concludely find of least use minimal cover F for any set of dependencies *E* using Algorithm 10.2.

If several server's His qualify as minimal covers of **E** by the definition above, it is a sumary to use additional criteria for "minimality." For example, we can choose the number set with the smallest number of dependencies or with the smallest risely length (the total length of a set of dependencies is calculated by concaverating the dependencies and number them as one long character string).

Algorithm 10-2). Finding a Minimal Cover F for a Ser of Functional Dependencies E

- $1:Se(F) \to U$
- 2. Replace each functional dependency  $X \to \{A_1, A_2, \dots, A_n\}$  in F by the n-functional dependencies  $X \to A_1, X \to A_2, \dots, X \to A_n$
- U For each functional dependency  $X \rightarrow A$  in F

II. The is a standard torus to complify the conductors and algorithms that ensure no redundance exists if 1.56 using the interference rule 184, we can convert a single dependency with multiple arribures on design fund side into a set of dependencies when single arritorities on the right band side. for each attribute B that is an element of X

 $||||f|\{I' - IX \rightarrow A_i^*|| \cup \{(X - IB_i) \rightarrow A_i^*|| is equivalent to E.$ 

then replace  $X \to A$  with  $(X - \{B\}) \to A$  in F.

4. For each remaining functional dependence  $X \to A$  in F

if  $||F - |X \rightarrow A||$  is equivalent to E

then remove  $X \to A$  from b

In Chapter 11 we will see how relations can be synthesized from a given set of dependencies E by first finding the national cover F to  $\Gamma$ 

## **10.3 NORMAL FORMS BASED ON PRIMARY KEYS**

Having studied functional dependencies and some of their properties, we are now ready to use them to specify some aspects of the sequenties of relation schemas. We assume that a set of functional dependencies is given for each relation, and that each relation has a deignated primary key; this information combined with the rows (conditions) for normal forms drives the nonradigation process for relational schema design. Most gradient relational design projects take one of the following two approaches:

- Sits perform a conceptual schema design using a conceptual model such as FR or 158 ind then map the conceptual design into a set of relations.
- Cenen the relations based on external knowledge derived from an existing implementation of files or forms or reports.

Following either of these approaches, it is then useful to evaluate the relations for goodness and decompose them turther as needed to achieve higher normal forms, using the normalization theory presented in this chapter and the next. We focus in this section on the first three normal forms for relation schemas and the intuition behind them, and discuss how they were developed historically. More general definitions of these normal torms, which take two account all candidate keys of a relation tarbet than just the primary key, are deterred to Section 10.4.

We sort by informally discussing normal focus and the matrixation behind then development, as well as reviewing some definitions from Chapter 5 that are needed here. We then discuss first normal form (TNF) in Section 10.3.4, and present the definitions of second normal form (2NF) and third formal form (4NF), which are based on primary keys in Sections 10.3.5 and 10.3.6 respectively.

### 10.3.1 Normalization of Relations

The normalization process, as first proposed by Codd (1972a), takes a relation schema through a series of tests to "certify" whether it studies a certion normal form. The process, which proceeds in a rop-down fashion by evaluating each relation against the crustria for normal forms and decomposing relations as pecessary, can thus be considered as

relational design by analysis. Initially, Codd grouposed three normal forms, which he called first second, and third normal form. A stronger definition of 500—colled Boyce-Codd totaal form (8000)—was proposed later by Boyce and Codd. All these normal forms are beed on the functional dependencies among the articlates of a relation. Later, a fourth rounal form (4000 and a fifth normal form (5000) were proposed based on the concepts of multiplicated dependencies and point dependencies, respectively, these are discussed in Chapter 11. At the beginning of Chapter 11, we also discuss how 300 relations may be sortheated from a given set of FDs. This approach is called relational design by confests.

Normalization of data can be looked upon as a process of analyzing the given relation schemas based on their 51s and primary keys to achieve the desirable properties of (1) minimizing redundancy and (2) minimizing the insertion, deletion, and update assumilies discussed in Section 10.1.2. Unsatisfactory rolution schemas that do not meet tertain conditions—the **normal form tests**—are decomposed into smaller relation schemas that the rosts and hence possess the desirable projecties. Thus, the normalization procedure provides database designers with the following:

- A formal formework for unargeng relation schemas haved on their keys and on the functional dependencies among their attributes.
- A series of normal form tests that can be carried out on midwidnal telation schemas so that the relational database can be normalized to any desired degree.

The normal form of a relation refers to the highest normal form condition that it arets, and hence indicates the degree to which it has been normalized. Normal forms, after considered in robition from other factors, do not guarantee a guid database design, busymerally not sufficient to check separately that each relation schema in the database case in PCNE in 501. Bother, the process of normalitation through decomposition must association the existence of additional properties that the relational schemas, taken regelier, double process. These would include rise properties:

- The Inssless join or nonadditive join property, which guarantees that the spurous ruple generation problem discussed in Section 10.1.4 does not occur with respect to the relation schemas created after decomposition.
- The dependency preservation property, which costness that each functional dependency is represented in some individual relation resoluting after decomposition.

The nonaddrive join property is extremely cricical and must be achieved at any cost, whereas the dependency preservation property, although desirable, is sometimes senticed, as we discuss in Section 11.1.2. We deter the presentation of the formal concepts and techniques that guarantee the above two properties to: Chapter 11.

### 10.3.2 Practical Use of Normal Forms

Mest practical design projects acquire execting designs of databases from previous designs, designs in legacy models, or from existing files. Normalization is carried out in practice so our the resulting designs are of high quality and meet the designble properties stated processly. Although several higher normal terms have been defined, such as the 480 and 5N° that we discuss in Chapter 11, the practical onliny of these normal forms becomes questionable when the constraints on Wach they are based are 2003 to understand or to detect by the database designess and users who must discover these constraints. Thus, database design as practiced in inclusity roday pays particular attention to normalization only up to SNE PONE or 4N?

Another point worth noting is that the database designers need not normalize to the highest possible normal form. Relations may be left in a lower normalization status, such as 2NT, for performance reasons, such as these discussed at the end of Section 10.1.2. The process of storing the joins of ligher normal form relations as a base relation—which is in a lower normal form—which is in a lower normal form—is known as **denormalization**.

### 10.3.3 Definitions of Keys and Attributes Participating in Keys

Before proceeding further, let us look again at the definitions of keys of a relation schema from Chapter 5.

**Definition.** A superkey of a relation schema  $R = \{A_1, A_2, \dots, A_n\}$  is a set of attrabutes  $S \subseteq R$  with the preperty that is, two tuples  $i_1$  and  $i_2$  in any legil relation state i of R will have  $i_1(S) = i_1(S)$ . A key K is a superkey with the additional property that removal of any attribute from K will cause K not to be a superkey any nine.

The difference between a key and a superkey is that a key has to be minimul; that is, if we have a key  $N = \{A_1, A_2, \dots, A_k\}$  of R, then  $K = \{A_k\}$  is nor a key of R for any  $A_k | l \leq i$ i, k. In Figure 10.1, [SS4] is a key for theoret, where as  $\{ss4\}$ ,  $\{ss4, track\}$   $\{ss4, track\}$ , and i, where k shows k and k we constrain a superkeys.

If a relation schema has more than one key such a called a **candidate** key. One of the candidate keys is arbitrarily disagnated to be the primary key, and the others are called secondare keys. Each relation schema must have a primary key. It. Figure 10.1, [soil is the only candidate key for private can be relative to primary key.

**Definition.** An artificute of relation schema R is called a **prime attribute** of R if it is a number of scale goodday law of R. An artificute is called **comptime** if it is not a prime artificute—that is, if it is not a periper of any candidate key.

In Figure 10.4 both (s) and (9,000) are prime artributes of works on whereas other artributes of works on are nonprime.

We now present the first three normal torus. (ed. 2N), and 4N). These were proposed by Codd (1992a) as a sequence to achieve the describle state of 5N) relations by progressing through the intermediate states of 1NP and 2N) if needed. As we shall see, 2NP and 3NF attack different problems. However, for historical reasons, it is castomory to follow them in that sequences hence we will assume that a 5NF relation already satisfies 26r.

### 10.3.4 First Normal Form

First normal form (TNP) is more considered to be part of the formal definition of a relation in the basic (flar) relational model.<sup>24</sup> historically, it was defined to deallow multivalcollarmbutes, composite attributes, and their combinations. It states that the domain of a simbute must include only at one (simple, indivisible) eakes and that the value of any arribute in a ruple must be a single relation the domain of that attribute. Hence, INF disallows having a set of values, a tople of collies, or a combination of both as an attribute calle for a ongle right. In organ words, INF disallows "relations within relations" or "relations as attribute values within taples." The only attribute values permitted by INF are single atomic (or indivisible) values.

Consider the measurement relation schema shown in logare 10.1, whose primary key is tracter and suppose that we extend it by including the measurement orthogon scheme is shown in logare 10.8.2. We assume that each department can have a menter of locations. The tracterious scheme and an example relation state are shown in Figure 10.5. As we can see,

| jai | CEPARTMEN      | ศ              |            |                                       |
|-----|----------------|----------------|------------|---------------------------------------|
|     | ONAME          | ONUMBER        | DMGRSSN    | DECCATIONS                            |
|     | ŧ              |                | <u>+</u>   | · · · · · · · · · · · · · · · · · · · |
| þ   | CEPARTMEN      | ៣              |            |                                       |
|     | CIVAMC         | <u>ONUMBER</u> | DMGRSSN    | DLOCATIONS                            |
|     | Deseauch       | s              | 333440006  | (Belare, Sugarland, Houston)          |
|     | Arministration | 4              | 387654321  | (Startund)                            |
|     | Headquarters   | I              | 839003655  | (Houston)                             |
| η   |                | ,т             |            |                                       |
|     | DNAME          | DNUMBER        | DAAGREESIN |                                       |
|     | Research       | 5              | 333445555  | Betare                                |
|     | Aesearch:      | 5              | 333445666  | Sugarland                             |
|     | Research       | 5              | 333445555  | Housion                               |
|     |                |                | AAA 4.444. | C + 8 - 1                             |
|     | Administration | +              | 967064331  | SUBRORU                               |

**FIGURE 10.8** Normalization into TNF. (a) A relation scheme that is not in TNF. BEExample state of relation DEPARTMENT. (c) TNP version of same relation with relations.

 This condition its transport in the neared relational worder and in object-telational systems (2002) - 0, both of relation discriminational relations (see Unipper 12). this is not an ENF because **DECATIONS** is not an atomic attribute, as illustrated by the first ruple in Figure 10.8b. There are two ways we can look at the DECATIONS attribute

- The domain of terrestions contains atomic values, but some tuples can have a set of these values. In this case, protettens is the functionally dependent on the primary key terrests.
- The domain of accentions contains sets of values and hence is miniatoane. In this case, manners → monstrons, because cash set is considered a single member of the attribute domain.<sup>11</sup>

In either case, the **DEEXTING** relation of Figure 10.5 is not in 1NF; in fact, it does not even qualify as a relation according to our definition of relation of School 5.1. There are three main rechroques to achieve test normal form for such a relation:

- Bemove the artribute constraints that variates UNE and place it in a separate relation of a constraint along with the primary key occurs of terms (see. The primary key of this relation is the combination houses, a according as shown in Figure 101. A distinct tuple in maximum exists for each factors of a department. This decomposes the non-UNE relation into two UNE relations.
- 2. Expand the key so that there will be a separate ruple in the original newspace relation for each location of a processist, as shown in Figure 10.5c. in this case, the primary key becomes the combination longests, prototool. This solution has the disadvantage of introducing refindency in the relation.
- 1. If a maximum number of values is known for the autibute—for example, due is known, that as most direct bearbars can exist for a department—replace the provisions armitude by three atomic articlates provided (a), provides and provides this solution has the disadvantage of introducing null values if most departments have fewer than three locations. It further introduces a spiritous semantics about the ordering among the location values that is not originally intended. Querying on this attribute becomes more difficult; for example, consider how you would write the energy "List the departments that have "Pellanc" as one of their locations" in this design.

Of the three solutions above, the first is generally considered best because it does not suffer from redundancy and it is completely general, howing no huri: placed on a maximum number of values. In fact, it we choose the second solution, it will be decomposed further during subsequent normalization steps into the first solution.

First mitting form also disallows michovalued attributes that are themselves composite. These are called **nexted relations** because each imple can have a relation within in Figure 10.9 shows how the ENP 200 relation could appear if nearing is allowed. Each imple represents an employee entity, and a relation from (formore, wors) within each

<sup>14.</sup> In this case we can consider the domain of 00.00411085 to be the power set of the solutional locations, that is, the domain is made up of all possible subsets of rac set of single locations.

P PROJ

|     |       | PR      | 215   |
|-----|-------|---------|-------|
| 55N | FNAME | PNLMBER | HOURS |

EMP\_PROJ

кî

3SN

| SSN        | ENAME           | PNUMBER | HOURS  |
|------------|-----------------|---------|--------|
| 123458759  | Siver.kton B    | - ·     | 325    |
|            |                 | 2       | . 75   |
| 606684444  | Narayan Farrest | K 3     | 40.0   |
| 463453453  | English Joyce A | 1       | 20.0   |
|            |                 | 2       | 20.0   |
| 333445566  | Woog Frankin T  | 2       | 10.0   |
|            |                 | 2       | T 10.0 |
|            |                 | 13      | 100    |
|            |                 |         | 10.9   |
| 293687777  | Zetaya, Aicia J | 30      | 30.0   |
|            |                 | 10      | 1/10   |
| 987987987  | Jabbar/Winau V. | 10      | 350    |
|            |                 | 30      | 50     |
| 987654321  | Walabeulennie S | 8 30    | 200    |
|            |                 | 20 .    | 15.0   |
| 88666655   | Borg Janves E.  | 20      | 101    |
| EMP_F      | -ROJ1           |         |        |
| <u>SSN</u> |                 |         |        |
| EMP F      | MO.12           |         |        |
|            |                 |         |        |

PNUMBER INCURS

**FIGURE 10.9** Normalizing nested relations into 15%, (a) Schema of the program relation with a "nested relation" attribute reads. (b) Example extension of the regress relation showing nested relations within each tople. (c) Decomposition wavegress into relations, size result, and evenesso? by propagating the primary key.

një terrevona the employed's projects and the bours per week that employee works on scheroleer. The scheme of the 982-980 relation can be represented as follows:

THE REAL (SAR, THANE, {PRODS(PROMBER, HEARS)})

The set braces  $\{ \}$  identity the artificing autos as multivalued, and we list the outpoint attrabutes that form agais between parentheses (). Interestingly recent trends frispporting complex objects (see Chapter 20) and SML data (see Chapter 26) using the relational model attempt to allow and formalize nested relations within relational diabase systems, which were disablewed each on by INF.

Notice that 250 is the primary key of the D9\_5800 relation in Figures 10.95 and b, while 60.9815 is the partial key of the nested relation, that is, within each tuple, the nested relation must have unique values of 60.0000. To normalize this into 1NT, we remove the nested relation attributes into a new relation and propagate the primary key of the new velation will confirm the partial key with the primary key of the original relation. Decomposition and primary key propagation yield the schemas E9\_ P0011 and E90 [9012 shown in Figure 10.96.

This procedure can be applied recursively to a relation with multiple-recel testing to unnest the relation into a set of INF relations. This is useful or converting an unmunutalized relation achieves with many levels of nesting into TNF relations. The existence of more than one multivalued attribute in one relation must be handled carefully. As an example, consider the following non-TNF relation:

PERSON (55#, {CAR\_LIC#], {P+ONE#})

This relation represents the fact that a person has multiple cass and multiple phones. If a strategy like the second option above is followed, it results in an all-key relation

PERSON\_THE INF (SS#1 CAS LITE#, PACHE#)

To avoid introducing any extrineous relationship between Cas (0:4 and Point\*, all possible combinations of values are represented for event 55\*, giving use to redundance. This leads to the problems handled by multivalued dependencies and 4NF, which we discuss in 3 hapter 11. The right way to deal with the two multivalued attributes in Presentovics to decompose it national void separate relations, using strategy 1 discussed above. P1(55#, Cas\_uit\*) and P2( 55#, Prove\*).

### 10.3.5 Second Normal Form

**Definition.** A relation schema R is in 2NF if every nonprime attribute A in R is fully functionally dependent on the primary key of R.

The row for 2N- involves testing for functional dependencies whose left-hand side attributes are part of the primary key. If the primary key contains a single arribute, the test need not be applied at all. The 399,000 relation in Figure 10.30 is in kNF but is not in 2NF. The nonprime attribute usawit violates 2NF because of 762, as do the nonprime attributes (soof, and endorrios because of FIG. The functional dependencies FIQ and FIG make usawit, enough and endorrios particulty dependence on the primary key (SSS, 6000)ce) of the endorrior the structure (source endored be) at the 2NF rest.

### 10.3.6 Third Normal Form

**Third normal form** (BISF) is based on the connept of transitive dependency. A functional dependence  $X \to Y$  in a relation schema R is a transitive dependency if there is a set or



FIGURE 10.10 Normalizing into 2NF and 3NF. (a) Normalizing engineer joto 2NF relations (b) Normalizing Engineer into 3NF relations

attributes 2 that is neither a candidate key not a subset of any key of  $R_1^{14}$  and both  $X \to 0$ and  $\Sigma \to Y$  hold. The dependency  $258 \to 085558$  is mansitive through twivite in  $198_{\pm}067$  of Figure 10.3a because both the dependencies  $558 \to 085858$  and  $1849868 \to 085858$  hold on 085888 is neither a key itself nor a subset of the key of  $288_{\pm}0877$ . In minimply, we can see that the dependency of Georgess on Demene is undestrable in  $298_{\pm}0877$  since DH08588 is not a key of  $897_{\pm}0677$ .

**Definition.** According to Could's original definition, a relation schema *R* is in 3NE if it satisfies 284 and no morphisme at robute of *R* as transitionely dependent on the primary key.

The relation schema (##\_0FF7 in Figure 10.5a is in 28.F. since no partial dependencies on a key exist. However, FMP (00.7 is new in 30.F because of the transitive dependence of 0008566 (and also 0000F) on 556 yea 660.00FR. We can normalize FMP (9F27) by decomposing it into the two 30.F relation schemas F01 and 562 shown to Figure 10.10Fr. Inturnely, we see that (01 and 102) represent independent entity form about employees and departments. A NATURAL IOIN operation on (01 and 502) will recover the original relation (2002) willow generating sparsed tuples.

Interfectly, we can see that any functional dependency in which the left-hand side is part (proper subset) of the primary key, or any functional dependency in which the lefthand side is a nonkey attribute is a "problematic" The 2NE and 5NE non-natization remove these problem FDs by decomposing the original relation into new relations. In terms of the transition process, it is not necessary to remove the pathal dependencies before the transitive dependencies, but battoneally, 3NE has been defined with the assumption that a relation is rested for 2NE first before it is used for 3NE. Table 4FAL informally summarizes the three normal forms based on primary keys, the tests used in each case, and the corresponding "remody" or norm definition performed to achieve the normal form

## 10.4 GENERAL DEFINITIONS OF SECOND AND THIRD NORMAL FORMS

In general, we want to design our relation schemas so that they have norther partial nertransitive dependencies, because these types of dependencies cause the update auonalies discussed in Section 10-1.2. The steps for normalization into 3X- relations that we have discussed so far disallow partial and transitive dependencies on the gravaris key. These definitions, however, do not take other candidate keys of a relation, if any, into account in this section we give the more general definitions of 2NE and 3NE that take all candidate keys of a relation into account. Notice that this does not affect the definitions of 1NE, since it is independent of keys and 6 networal dependencies. As a general definition of prime attribute, on otherhere that is part of any cardidate keys will be considered os prime

i4. This is the general definition of transitive dependencies. Because we are concerned on a writeprimary keys in this section, we adow transitive dependencies where X is the protony key but Z inside carsolicer of a condidate key.

| NORMAL FORM             | Test                                                                                                                                                                                                                                        | REMEDY (NORMALIZATION)                                                                                                                                                                                                               |
|-------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| hrs: (180)              | Relation should lisse no nenationic attributes or nested relations.                                                                                                                                                                         | Form new relations to reach nonatomic attreasters nested relation.                                                                                                                                                                   |
| Second (288)            | For relations where primary key contains<br>multiple auributes, no menkey attribute<br>should be functionally dependent on a part<br>of the primary key.                                                                                    | Decompose and set up a new relation for<br>each partial key with its dependent<br>attribute(s). Make size to keep a relation<br>with the original primary key and any<br>attributes that are fully for ctionally<br>dependent on it. |
| ∏rac <sup>(</sup> (384) | Relation should not have a nonkey autibute<br>functionally determined by another nonkey<br>autibute (or by a set of nonkey autibutes).<br>That is there should be no transitive depen-<br>dency of a nonkey autibute on the pairing<br>hey. | Decompose and set up a relation that<br>meludes the nonkey attribute(s) that<br>functionally determine(s) other nonkey<br>attribute(s).                                                                                              |

TABLE 10.1 SUMMARY OF NORMAL FORMS BASED ON PRIMARY KEYS AND CORRESPONDING NORMALIZATION

Panal and full functionful dependencies and transitive dependencies will now be considered with respect to all catchdate keys of a robution.

### 10.4.1 General Definition of Second Normal Form

**Definition.** A relation scheme *K* is in second normal form (ZNF) if every nonprime archite *A* in *R* is not partially dependent on any key of  $R_{\pm}^{-14}$ .

The cost for 2NF involves testing for functional dependencies whose left-band side induces are port of the primary key. If the primary key contains a single attribute, that test need not be applied at all. Consider the relation schemal on shown in Figure 12.11a, which describes parcels of land for sale in corrious counties of a state. Suppose that there are two condidate keys: essence out and (course same, corr); that is, lot numbers are imple only within each nontry, but essence at numbers, its unique across counties for the court state.

Based on the two candidate keys no-cervitate and toomor none, cere?, we know that the functional dependencies Hill and Hill of Figure 10.11a hold. We choose negative toe a the parameter key, so it is underlined in Figure 10.11a, but no special consideration will

If This definition can be restated as follows. A relation scheme B is in 2NE if every nonprime an interval  $M_{12}$  fully found smallly dependent schemes  $ke_{2} \neq B$ .

#### 322 Chapter 10 Functional Dependencies and Normalization for Relational Databases





be given to this key over the other candidate key. Suppose that the following two additional functional dependencies hold in (01%)

```
\label{eq:figure} \begin{array}{l} {}^{\rm 5DQCCOUNTY_NAME} \rightarrow {}^{\rm CAX}, {}^{\rm SATE} \\ {}^{\rm FDAT} {}^{\rm SAEX} \rightarrow {}^{\rm FDAT} {}^{\rm SAEX} \end{array}
```

In words, the dependency FD3 size that the tax time is fixed for a given county (does not cary for by for within the same county), while FD4 size that the price of a for is detended by its area regardless of which county it is in "(Assume that this is the price of the for for tax purposes).

The fors relation schema violates the general definition of 2NF because (w2.990) is prinally dependent on the condidate key (2000) [6991, 1004], due to FLS. To contradice (0)s into 2NS, we decompose in into the two relations (0)s1 and (2)s2, shown in bigate 10.11b. We renarrow tors1 by removing the attribute (w2.893) that violates 2NF front (0)s and plong at with 50000 9086 (the left-hand side of 110) that causes the partial dependence) into another relation (0)s2. Both (0)s1 and (0)s2 are in 2NF. Nonce that FLA does not violate 2NF and is fartied over to (0)s1.

### 10.4.2 General Definition of Third Normal Form

**Definition.** A relation schema R is in **third normal form (3NF)** its whenever a normal formational dependence  $X \to A$  holds in R either its) X is a superkey of R or (2). As a prime attribute of R.

According to this definition, 1952 (Figure 10.1995) is in SNF. However, 554 in 1952 (Figure 10.1995) is in SNF. However, 554 in 1952 (Figure 10.1995) and enter the prime attribute in 1952. To remain equation (SNF, we decompose at into the relation schemes votel) and 19529 soon in Figure 10.111. We construct total by removing the attribute (stat that stolate-SNF from rotal and placing it with were the left-hand on 6.95 that causes the residue dependence) into another relation rutate. Both cutsis and resi3a are in SNF.

Two points are worth netting about this example and the general definition of 3NF:

- raist violains 36t because Recens transitively dependent on early of the candidate keys of (0751 yearthe nonprime attribute 065).
- This general dentation can be applied directly to rest whether a tell tion schema is in 3N1; it does not have no go through 2N+ first. If we apply the above 3N+ definition to 10% with the dependencies 5D1 through 40%, we find that fork 40% and 40% violate 3N5. We could hence decompose rous into 10%1% rous10, and 10%2 directly. Hence the transitive and partial dependencies that violate 3N+ can be removed in any order.

### 10.4.3 Interpreting the General Definition of Third Normal Form

Arelation schema R violates the general definition of ANP if a functional dependency  $\lambda \rightarrow A$  holds in R char violates  $b_1 a_1$  conditions (a) and (b) of 38.8. Violating (b) means that

A is a nonprime attribute. Violating (a) mores that X is not a superset of any key of B hence, X could be nonprime or it could be a proper subset of a key of K. If X is nonprime, we typically have a transitive dependency that violates SNL, whereas if X is a proper subset of a key of R, we have a partial dependency that violates SNL (and also 2NT). Benesive can state a general alternative definition of SNF as follows: A relation schema R > r SNF if every nonprime antibute of R meets both of the following conditions:

- It is fully functionally dependent on every key of R.
- It is nontransitively dependent on every key of R.

## **10.5 BOYCE-CODD NORMAL FORM**

**Boyce-Codd normal form (BCNF)** was proposed as a simpler form of 5NF, but it was round to be structer than 5NF. That is, every relation in BCNF is also in 3NF, however, a relation in 5NF is not necessarily in PCNF. Intuitively, we can see the need for a stronger normal form than 3NF by going back to the cors relation schema of Figure 10.11a with its for function if dependencies F.51 through FC4. Suppose that we have theorem do lors in the relation for the lors are from only two counties. Dokalb and Ficheric Suppose also that for cless in Dekalfi Country are only 0.5, 2.6, 0.7, 0.8, 0.9, and 1.0 acres, whereas for one in Follow Country are restricted to 1.1, 0.2, ..., 1.9, and 2.2 acres. In such a struction would have the additional functional dependency FC5 areas  $\rightarrow$  (court page. If we add this to the other dependencies the relation schema 20(51) stats  $\rightarrow$  (court page. If we add this to the other dependencies the relation schema 20(51) stats in 3NF because courts **1** we is a prime attribute.

The atea of a lot that determines the county, as specified by EDS, can be represented by 16 tuples in a separate relation *Ristex*, (costy\_susc), since there are only 16 possible sets values. This representation reduces the reducdancy of repeating the sine miormation in the thousands of cortsis tuples. ECNT is a stronger normal from that would disaflow ratisfs and suggest the need for decomposing etc.

**Definition.** A relation scheme R is in **BCNF** if whenever a monoral functional dependency  $X \to A$  holds in R, then X is a superkey of R.

The formal definition of FCNF differ-slightly from the definition of PNF. The only difference between the definitions of PCNF and PNF is that condition (b) of PNF, which allows A to be prime, is absent from DCNF. In our example, EU5 violates (CNF in offs) because asta is not a superkey of CD151a. Note that 1.05 satisfies because cases were a prime antibute (condition b), but this condition does not reasonable definition of PCNF. We can decompose to tyle into two PCNF relations (1952a FD7 because its attributes inclusion bases the functional dependency FD7 because its attributes no longer exercising the same relation after decomposition.

In practice, most relation schemas that are in 3NE are also in 60NE Only if  $X \rightarrow A$ holds in a relation schema R with X not being a supervey and A being a prime attracte will R be in 3NE but not in 60NE. The relation schema R shown in Figure R RN illustrates the general case of such a relation. Ideally, relational database design should strive to achieve ECNE or 3NE for every relation schema. Achieving the normalization



RGURE 10.12 Boyce-Codd normal form. (a) BCNP controlization of (a1/1a with the functional dependency FD2 being lost in the decomposition, thi A schematic relation with FDS; it is in BNP, but not or BCNP.

name of just INF or CNF is not considered adequate, since they were developed intercally as stepping stones to BNF and HONF

As another example, consider Figure 10.13, which shows a relation tools with the following dependencies

 $\mathbb{P}[1] \mid \{|\mathsf{student}|,\mathsf{course}\} \rightarrow \mathsf{instructor}$ 

```
102.10 inverses 104 \rightarrow 100850
```

Note that [studget, conset] is a condidate key for this relation and that the dependencies shown follow the pattern in Figure 10.126, with student as A, constens B, and issuegies as C. Hence this relation is in ANF but not PCNF. Decomposition of this relation is not straightforward because it may be decomposed insome of the three following possible parts:

1 (stocent, instruction) and (stocess), caused

```
2 Iceases, assi<u>nguing</u>el and l<u>ocaese</u>, <u>support</u>}.
```

3. Instructory, course | und | instructory, studied

10 This dependency means that "each instruction teaches one course" suscentstande for this application.

| TEACH   |                   |            |
|---------|-------------------|------------|
| STUDENT | COURSE            | INSTRUCTOR |
| Narayan | Detebane          | Mark       |
| Smith   | Catabase          | Navaine    |
| Smith   | Operating Systems | Armmar     |
| Smith   | Theory            | Schulman   |
| Waltace | Detahase          | Mark       |
| Wallace | Operating Systems | Ahamad     |
| Wong    | Dalabase          | Omeansla   |
| Zelaya  | Dalacese          | Nevelhe    |
|         |                   |            |

FIGURE 10.13 A relation mark that is in 3NF but not BCNF.

All three decompositions "lose" the functional dependency FPI. The description of these just showers 5, because it will not generate sportious tuples their a joint.

A test to determine whether a decomposition is non-additive (los-less) is discussed in Section 11.1.4 under Property 1.1. In general, a relation net in BCNF should be decomposed so as to meet this property, whele possibly forgoing the preservation of all functional dependencies in the decomposed relations, as is the case in this example. Algorithm 11.3 does that and could be ased above to give decomposition 3 for these

### 10.6 SUMMARY

In this chapter we first discussed several virtally in relational database design using intutive argaments. We identified informable some of the measures for indicating whether a relation schema is "good" or "bad," and provided informal guidelines for a good design. We then presented some formal concepts that allow us to do relational design in a rapdown tashion by analyting relations individually. We defined this process of design be analysis and decomposition by introducing the process of portwortation.

We discussed the problems of nightre anomalies that occur when rish indencies are present in relations. Informal measures of good relation schemas include simple and clear attribute semantics and two nulls in the exceptions (states) of relations. A good decomposition should also avoid the problem of generation of spiniors tuples as a residual the join operation.

We defined the concept of functional dependency and decreased since of its properties. Functional dependencies specify semantic constraints among the attributes of a relation schema. We showed how from a piven set of functional dependencies additional dependencies can be interest using a set of interesting. We defined the concepts of closure and recer related to functional dependencies. We then defined

tannial cover of a set or dependencies, and provided an algorithm to compute a minimal cover. We also showed how to check whether two sets of functional dependencies are spin-zlein.

We then described the normalization process for achieving good designs by resurg relative for undesirable types of "problematic" functional dependencies. We provided a round number of successive normalization based on a prodefined primary key in each relation, then related this requirement and provided more general definitions of second normal form (2NF) and third normal form (3NF) that take all conditions for relation into account. We presented examples to illustrate how by using the general definition of 3NF is given relation into be analyzed and decomposed to eventually weld a set of relations in 3NF.

Finally, we presented Boxes Codd normal form (90%-) and discassed how it is a storget form of 3NF. We also illustrated how the decomposition of a non-BCNF relation may be done by considering the nonadditive decomposition requirement.

Chapter 11 presents synthesis as well as decomposition algorithms for relational database design based on functional dependencies. Related to decomposition, we discuss the concepts of lossless (normalizing) jum and dependency jectovorion, which are enforced by some of these algorithms. Other topics its Chapter 11 include multivalue, dependencies, join dependencies, and fourth and fifth normal forms, which take these dependencies into account.

### Review Questions

- 124 Discuss attribute sensations as an informal measure of goodness for a relation schemo.
- 12.2 Discussionsert on deletion, and modification anomalics. Why are they considered both Illustrate with examples.
- 10.3 Why should notly in a relation be avoided as for as possible? Docuss the problem of spurous tuples and how we may prevent it.
- 124. State the intormal guidelines for relation schema design that we discussed. Illustrate how studietion of these guidelines may be harmful.
- 12.1 What is a functional dependency? What are the possible sources of the information that defines the functional dependencies that hold among the attributes of a relation schema?
- 10.6 Why can we not order a functional dependency automatically from a particular relation state?
- 15.5 What role do Armstrong's interence rules the three inference rules IR1 through IR3--- play in the development of the theory of relational design?
- 12.5. What is meant by the completeness and soundness of Armstroug's inference rules?
- 12.9. What is meaning by the closure of a set of functional dependencies? Illustrate with an example.
- (212) When are two serveof functional dependencies equivalent? How can we determine their equivalence?
- 2.11. What is a minimal set of functional dependencies? Does every set of dependencies have a minimal equivalent set? as it always imagin?

- 10.12. What does the term motorizalized relation refer to? How did the normal forms develop historically frian first minual form up to Bocce-Codd minual form?
- 10.13. Define first, second, and third normal forms when only primary keys are reasidcred. How do the general definitions of 2NF and 3NF, which consider all keys it a relation, differ from those that consider only primary keys?
- 10.14. What applesizable dependencies are availed when a relation is in 2NF?
- 10.15. What endesirable dependencies are avoided when a relation is in HNP?
- 10.16. Define Boyce Codd normal form. How does in differ from 3NF? Why is in considened a stronger form of 3NF?

### Exercises

- 10.17. Suppose that we have the following requirements for a university database that is used to keep track of students' transcripts.
  - a. The university keeps track of each student's name (Suse), student number (Sus), social scority number (Su), current address (Second) and phone (Strucus), burb date (Doublesk (Second) and phone (Strucus), burb date (Doublesk (Second) and phone (Strucus), burb date (Doublesk (Second), cophomore, ..., graduate), major department (Strucus), and degree program (Second) (B. 47, 6.5..., Second). Both SSSS and student tumber have unique values for each student.
  - b Each department is described by a name (BRAPE), department code (DECDE), office number (EGFFEEE), office phone (BRIDAE), and college (DECDE). Rota name and code have unique values for each department.
  - c. Each course has a course mane (cosef), description (cofse), course number (course in an accurate hours (CREDIT), level (cover), and othering department (CDECT). The course number is unique for each course.
  - d Each section has an instructor (1889), semister (SENESUR), year (YEAR), course (SECRARS)), and section miniber (SECRAR). The section number distinguishes different sections of the same course that are rought doring the same semister? years its values are 1, 2, 3, ..., op to the total number of sections tought during each semister.

c. A grade record refers to a student (SSR), a particular section, and a grade (6843) Design a relational database schema for this database application. First show all the functional dependencies that should hold among the attributes. Then design relation schemas his the Gatabase that are each in 3NF or JORG. Specify the key attributes of each relation. Note any unspecified requirements, and make appropriate assumptions to render the specification complete.

10.18. Prove or disprove the following interence rules for timemonal dependencies. A proof can be made either by a proof argument or by using interence rules (R) rhomeh. R3. A dispross should be performed by demonstrating a relation instance that satisfies the conditions and functional dependencies in the left-band side of the inference rule but does not satisfy the dependencies in the left-band side.

 $\mathbb{A}_{\mathbb{A}} \{ \{ X \to Y, X \to \mathbb{C} \} \vdash \{ \{ X X \to Y \} \}$ 

 $\mathbb{P} = \{X \rightarrow Y\} \text{ and } Y \supseteq 2 \models \{X \rightarrow 2\}$ 

### Exercises 329

- - $1, \quad (XY \to \mathbb{Z}, \mathbb{Z} \to W) \models \{X \to W\}.$
- 12.19 Consider the following two sets of functional dependencies: F = [A → C, AC → D, E → AD, E → H] and C = [A → CD, E → AS]. Check whether they are equivalent.
- 1011. Is the set of functional dependencies Grap Exercise 10.20 normal? If not, try to find a minimal set of functional dependencies that is equivalent to G. Prove that your set is equivalent to G.
- 12.22 What update anomalies occur in the two way and two\_ate relations of Figures 1CB and 10.47
- 10.33 In what normal form is the cars relation schema in Figure 10.11a with respect to the restrictive interpretations of normal form that take ods the primary key smoaccount? Would it be in the same normal form if the general definitions of normal form sets used?
- 1224. Prove that any relation schema with two autobutes is in R/NR-
- 1235 Why do spurious ruples occur in the result of iomme the aw 19001 and ewg, cocs relations of ligare 10.5 (result shown in Figure 10.6.).
- 10.20 Consider the universal relation K = [A, B, C, D, E, F, G, H, L,] and the set of functional dependencies F = {[A, B] → [C], [A] → [D, L], [B] → {F}, [F] → (G, H], (D) → [J, i]]. What is the key for B? Decompose R into 2N3 and then 3N5 relations.
- 12.27. Repeat Exercise 10.26 for the following different set of functional dependencies (a = {(A, B) → {(C), (B, D) → {(E), F}, (A, D) → {(G, H), (A) → {(F, H) → {(P)}.
- 1218. Consider the following relation:

| <u> </u> | <u> </u> | . <u> </u> | TUPLER     |
|----------|----------|------------|------------|
| IC .     | ы        | : ]        | <i>=</i> 1 |
| 10       | F2       | c2         | #2         |
| 11       | t-4      | cl         | # i        |
| 12       | 63       | .4         |            |
| 15       | ы        | 21         | 25         |
| 14       | Ŀэ       | c4         | #6         |

a. Civen the previous extension (state), wheels of the following dependencies may lock in the abuve relation? If the dependency cannot hold, explain why by specifying the tights that cause the violation.

 $u \land A \to B, \ u, B \to C, \ u, C \to B, \ v \land B \to A, \ v \land C \to A$ 

- b. Does the above relation have a potential candidate key? If it does, what is it?If it deck not, why not?
- 10.29. Consider a relation R(A, B, C, D, E) with the following dependencies:

 $AB \rightarrow C_{+}CD \rightarrow E_{-}DE \rightarrow H$ 

Is AB a candidate key of this relation? If not, is ABD? Explain your answer.

10.30. Consider the relation R, which has attributed that hald schedules of courses and sections at a environmentary R = [CourseNot SecNot Offering] Opt, Credit Hours, CourseLevel, InstructorissN, Schnester, Year, Days\_Hours, RecomNot, NoOfSridents]. Suppose that the following functional dependencies hold on fit

{CourseNo} +> {OfferingDept, CreditHours, CourseLevel}

(CourseNo, SeeNo, Semester, Year)  $\rightarrow$  {Days\_Hears, RoomNo, NoOfStudents InstructorssN)

[RoomNo, Days\_Hours, Semester, Year] → [Instructorism, CourseNo, SecNo].

Try to determine which sets of attributes form keys of R. How would you normalize this relation?

 Consider the following relations for an order-processing application database at ABC, Inc.

ORDER 104, Odare, Custe, Total\_amount)

ORDER-ITEM(Or, Ir. Qty\_ordensi, Total price, Documenta)

Assume that each item has a different discourd. The Tetra 2010 refers to one item, 00000 is the date on which the order was placed, and the Tetra 200000 is the amount of the order. If we apply a natural join on the relations ORDER-ITEM and ORDER in this database, what does the resulting relation schema look like? What will be its key? Show the sits in this resulting relation. Is it in 2013 is it in 3665. Why or soly pool (State assumptions) if you make any 1

10.32. Consider the following relation-

CAR\_SALE(Car#, Date\_sold, Sol#smon#, Commission%, Discount\_amt)

Assume that a car may be sold by mustiple salesment and hence (CAR#, SALESWAP) is the primary key. Additional dependencies are

Date\_sold → Discount [am]

and

Salesman\* → Commission's

Based on the given grunning key, is this relation in INF. 2NF. or 3NF? Why or why not? How would you successively normalize it completely?

10.13. Consider the following relation for published books:

BOOK (Book\_title: Authomatic: Book\_type: Listprice: Author\_afhl, Publisher)

Author\_affil refers to the affiliation of worker. Suppose the following dependencies exists

Book\_rate → Pablisher, Book\_type

Basik (type  $\rightarrow$  Listprice

Authorisms -> Author-and

- a. What normal form is the relation in "Explain your answer,
- Apply normalization unril you cannot decompose the relations further. State the reasons behind each decomposition.

### Selected Bibliography

Enjoymal dependencies were originally autochoed by Codd (1970). The original definitions of first, second, and third normal form were also defined in Codd (1972a), where a discussion on update anomalies can be found. Boyce Codd normal form was defined in Codd (1974). The Aternative definition of third normal form is given in Udman (1988), is is the denartion of 1978; that we give here. Ullman (1988), Mater (1983), and Attern and De Antonellus (1993) contain many of the theorems and priofs concerning funcanal dependencies.

Armstring (1974) shows the soundness and completeness of the inference rules IB1 drough (8). Additional references to relational design theory are given in Chapter 11



# Relational Database Design Algorithms and Further Dependencies

In this chapter, we describe some of the relation a durabase design abourbars that maker increased of Chapter 12, we introduced the two main approaches for relational database design. The first approach utilities a tup-down design technique, and is currently used not exertistically an commercial database application design. This involves designing a correptual schema in a high-level data model, such as the AFR model, and then mapping the conceptual schema into a set of relations using mapping procedures such as the ones doosed in Chapter 7. Following this, each of the relations is analyzed based on the funcuenal dependencies and assigned primary keys. By applying the normalization procedure in Section 10.5, we can remove any remaining partial and transitive dependencies from the features in some design methodologies, this analysis is applied directly during conteptial design to the attributes of the entiry types and telationship types. In this case undesirable dependencies are discovered during conceptial design, and the relation schemas resulting from the mapping procedures would automatically be in higher normal tune, so there would be no need for additional normalization.

The accord approach orders a Bottom-up design technique, and is a more point approach that views relational durables schema design strictly in terms of functional and othertypes of dependencies specified on the dotabase attributes. It is also known as relational without After the database designer specifies the dependencies a normalization algorithm is uplied to synthesize the relation schemas. Each individual relation schema should posses the measure of geochesic associated with DNF or its NF or with some higher round form. In this chapter, we describe some at these normalization algorithms as well as the other types of dependencies. We also describe the two describle properties of nonadditive (lossless) joins and dependency preservation in more devol. The normalization algorithms repically some by symbolizing one gians relation schema, called the universal relation, which is a theoretical relation, that includes all the database artificates. We then perform decomposition —breaking opporto smaller relation schemas—ontil it is no longer teasible or no longer desirable, based on the functions, and other dependencies specified by the database designer.

We first describe in Section 11.1 the two desirable properties of decompositions, trainely, the dependency preservation property and the Jussless (or nonadditive) join property, which are both used by the design algorithms to achieve desirable decompositions. It is important to note that it is insightent to test the relation schemes adopendently of our souther for compliance with higher normal forms like 2NF, 3NF, and 60NF. The resulting relations must collectively straty these two additional properties to qualify as a good design. Section 11.2 presents several normalization algorithms based on functional dependences clong that can be used to design 3NF and 40NF schemas.

We then introduce other types of data dependencies, including multivalued dependencies and join dependencies, that specify consistants that cannot be expressed by functional dependencies. Presence of these dependencies reads to the definition of tourish normal form (4NF) and fifth normal form (5NF), respectively. We also define inclusion dependencies and template dependencies (which have not led to any new normal ferms so tar). We then briefly discuss domain-key normal ferm (DKNF), which is considered the most general normal form.

It is pusable to skip some or all of Sections (1.4, 11.5, and 11.6 a) an oppoductory database course.

## 11.1 PROPERTIES OF RELATIONAL DECOMPOSITIONS

In Section 11.1.1 we give examples to show that looking at an indudual relation to test whether it is in a higher normal torm does not, on its own, guarantee a erod design; rather, a set of seletions that trajecther turn the relational database schema must preserve turn additional properties to ensure a good design. In Sections 11.1.2 and 11.1.3 we discuss two of these properties: the dependency preservation property and the loadess at nurradditive join property. Section 11.1.4 discusses binary deecompositions, and Section 11.1.5 discusses successive normadditive join decompositions.

### 11.1.1 Relation Decomposition and Insufficiency of Normal Forms

The relational database design algorithms that we present in Section 11.2 start from a single universal relation schema  $R = \{N_1, N_2, \dots, N_n\}$  that includes of the artificities of the

durbase. We implicitly make the **universal relation assumption**, which states that every analytic name is unique. The set **F** of functional dependencies that should hold on the analytics of **R** is specified by the database designers and is made available to the design agains. Using the functional dependencies, the algorithms decompose the universal febrior schema **R** into a set of relation schemas  $D = \{R_1, R_2, \dots, R_n\}$  that will become iderelational database schema, **D** is called a **decomposition** of **R**.

We must make size that each attribute in R will appear in at least one relation when a R in the decomposition so that its attributes are "lost"; formally, we have

$$\bigcup_{i \in \mathcal{R}_{i}} R_{i} = R$$

This is called the attribute preservation condition of a decomposition.

Another goal is in have each individual relation **R**, in the decomposition **D** by in [CNF or 3NF. However, this combinantis not sufficient to guarantee a goal database design at its own. We must consider the decomposition of the universal relation as a whole, in addition to looking at the individual relations. To illustrate this point, consider the Orglocs (not, electrice) relation of Figure 10.5, which is in 3NF and also in FONE in fact, an relation schema with only two attributes is automatically in 20NF 1 Although (ne\_ lots is in BONE at still gives rise to spurious topics when joined with tegework (SNE, ennestration), energy explores topics when joined with tegework (SNE, ennestration), energy explores topics when joined with tegework (SNE, ennestration), energy explores to particularly bad relation schema because of ats consolited semantics by which excertible gives the location of one of the phylics on which in employee works, Joining DEF\_cos with PROJECT (PROME, PROHEES, Protection, DAM) of head for other criteria that, together with the conditions of WE of PONE. This underscores the anal for other criteria that, together with the conditions of WE of PONE, prevent such loc designs. In the next three subsections we discuss such additional conditions that should head in the decomposition D as a whole.

### 11.1.2 Dependency Preservation Property of a Decomposition

It would be useful if each functional dependency  $X \to Y$  specified in F other appeared directly in one of the relation schemas  $R_1$  in the decomposition D or could be interred for the dependencies that appear in some  $R_2$ . Informally, this is the dependence preserveto exclass. We want to preserve the dependencies because each dependency in F represents a constraint on the database. If one of the dependencies is not represented in some edo sheal relation  $R_1$  of the decomposition, we cannot enforce this constraint by dealing situation included relation, anstead, we have to your two or more of the relations in the datappear to you two or more of the relations in the datappear to you two or more of the relations in the datappear of the functional dependency holds in the result of the functional dependency holds in the result of the job operation. This is clearly an inefficient and impractical procedure.

I As an exercise, the reader should prove that this statement is true.

It is not necessary that the exact dependencies specified in  $\Gamma$  appear then elves in adaptidual relations of the decomposition D. It is sufficient that the union of the dependencies that hold on the individual relations in D be equivalent to F. We new define these concepts more formally.

**Definition.** Given a set of dependencies F on R, the projection of F on R, denoted by  $\pi_R(F)$  where  $R_i$  is a subset of  $R_i$  is the set of dependencies  $X \to Y$  in F' such that the artificates in  $X \cup Y$  are all contained in  $R_i$ . Hence, the projection of F or each relation schema  $R_i$  in the decomposition D is the set of functional dependencies in F', the closure of  $F_i$  such that all their lefts and right-hand-scale attributes are ins  $R_i$ . We say that a decomposition  $D = \{R_i, R_2, \ldots, R_n\}$  of R is **dependency-preserving** with respect to F of the union of the projections of F on each  $R_i$  in D is equivalent to F; that is,

 $R(\pi_{w_{i}}(F)) \cup \ldots \cup (\pi_{w_{i}}(F))F = F'$ 

If a decomposition is not dependency-preserving, some dependency is last in the decomposition. As we mentioned earlier, to check that a lost dependency holds, we must take the JOIN of two or more relations in the decomposition to get a relation that an hide all lefts and rightsbund-side arribures of the lost dependency, and then check that the dependency holds on the result of the JOIN - an option that is not practical.

An example of a decomposition that does not preserve dependencies is shown in Figure 10.12a, in which the functional dependency FD2 is lost when (ets) is a decomposition in Figure 10.11, however, are dependency preserving. Similarly, for the example in Figure 10.13, no matter what decomposition is chosen for the telation (180 (strain), cossect, using)(oc)) from the three provided in the rest, one or both of the dependencies originally present are lost. We state a claim below related to this property without providing any proof.

#### CLAIM J

It is always possible to find a dependency-preserving decomposition D with respective *P* such that each relation R<sub>1</sub> in D is in 3500

In Section **U** 2.1. We describe Algorithm 11.2, which creates a dependencepreserving determinisation  $D = \{R_i, R_j, ..., R_n\}$  of a universal relation R based on a set of functional dependencies E such that each R on D is in SN7.

### 11.1.3 Lossless (Nonadditive) Join Property of a Decomposition

Another property that a decomposition D should possess is the lossless jour or nonable time join property, which coveres that no sperious ruples are generated when a XATORAL UNS operation is applied to the relations in the decomposition. We already illustrated the problem in Section 10.1.4 with the example of Figures 10.5 and 10.6. Second this is a property of a decomposition of relation whence, the condition of polyptics ruples sland) hold on every legal relation state—that is, every relation state that satisfies the hardneral dependencies in *F.* Elence: the lossless poor property is always defined with respect rowspecific set F of dependencies.

**Definition.** Formally, a decomposition  $D = \{R_1, R_2, \dots, R_n\}$  of *R* has the lossless houndditively join property with respect to the set of dependencies *E* on *B* if, for every relation states of *R* that satisfies *F*, the following holds, where \* is the NATURAL WIN of all the relations in *D*:

 $^{*}(\pi_{0}, t_{0}), \ldots, \pi_{0}, (t_{0}) = t_{0}$ 

The word loss in *losskip* refers to *kess of information*, but to *loss* of tuples. It a decomposition does not have the lessless join property, we may get additional spunces applies after the 1980 FCT ( $\pi$ ) and NATURAL JOIN (\*) operations are applied; these additional tuples represent enumeration more accurately. If the property builds on a decomposition, we are guaranteed that no spurious tuples bearing wrong information are added on the property builds on a decomposition, we are guaranteed that no spurious tuples bearing wrong information are added on the property for the project and natural join operations are applied.

The decomposition of the peopless, provide, noted, every number of carton) from Igne 10.3 into the procession, proceeding and expendition, noted, notes, provide the figure 10.6 the integration of the procedure for resting whether any decomposition D of a relation into a relations is lossless (nonaddrive) with respect to a set of even tractonal dependencies 1 in the relation; it is presented as Algorithm 11.1 below it is possible to apply a simpler rest to check of the decomposition is nonaddrive for hubble to apply a simpler rest to check of the decomposition is nonaddrive for buary decompositions to the test is described by Section 11.4.

Algorithm 11.1). Festing for Lessless (non-additive) for Property

**Input**: A environal relation  $R_i$  a decomposition  $D = \{R_i, R_j, \dots, R_n\}$  of  $R_i$  and a set  $F_i$  or functional dependencies.

- Create an unual matrix S with one row i for each relation R in D, and one coltion / for each strabute A<sub>1</sub> in R.
- Set S(i,j):= b<sub>j</sub> for all matrix entries.
   C each b<sub>j</sub> is a distinct symbol associated with induces (i, j) \*)
- For each row trepresenting relation scheme R.

(for each column) representing attribute A,

(if trelation  $R_i$  includes autibute A.) then set  $S(r_i) := a_i \beta k$ 

. (Yeach a, is a distinct symbol associated with  $\operatorname{adex}(\hat{\beta}, \gamma)$ 

 Repeat the following hop until a couplete fospie wataon results in no changes to N for each functional dependency X → Y in F

 $\mathfrak{h}_X$  (1 pass in S that have the senie symbols in the columns corresponding to surplates in X

{make the sembols in each column that correspond to an attribute in Y be the some in all these rows as follows: If any of the rows has an 'u' sembol for the

column, set the other rows to that same "a" symbols in the column off roota' sembol roots for the attribute many of the rows, choose one of the "b" symbols that appears in one of the rows for the attribute and set the other rows in that same "b" symbol in the column  $d_{2}d_{2}$ :

 If a now is made up entirely of "a" symbols, then the decomposition has the loss loss jum property, otherwise, it does not.

Given a relation *R* that is decomposed into a number of relations  $R_1, R_2, \ldots, R_n$ . Algorithm 11.1 begins the matrix *S* that we consider to be some relation state r of *R*, Rov *i* in *S* represents a tople *i*, (corresponding to relation *R*) that has "a" symbols in the columns that correspond to the attributes of *R*, and "b" symbols in the remaining columns. The algorithm then transforms the rows of this matrix couring the loop of step 4) so that they represent tuples that satisfy all the functional dependencies in *F*. At the end of step 4, any two rows in *S*—which represent two tuples its r—that agree in their values for the feit-barst-side attributes *N* of a functional dependency  $X \to Y$  in *F* will also agree in their values for the right-band-side attributes *Y*. In cast be shown that after applying the loop step 4, it any tow in *S* ends up with all "b" symbols, then the decomposition *D* has the lossless jum property with respect to *F*.

It, on the other hand, no row ends up being all "n" semilois. D does not satisfy the leastess join property. In this case, the relation state  $\gamma$  represented by S at the end of the algorithm will be on example of a relation state  $\gamma$  of R that satisfies the dependencies in F but does not satisfy the loosless pair condition. Thus, this relation serves as a **counterexample** that proves that D does not have the leastess pair property with respect to *E*. Note that the " $\rho$ " and "b" symbols have an aprecial incarring of the end of the algorithm.

Figure 11.1.4 shows have we apply Algorithm 11.1 to the decomposition of the *tex*relation schema from Figure 10.3b into the tota relation schemas *two* total and *tex*cas of Figure 10.5a. The loop in step 4 of the algorithm cannot change any "b" symbols in "a" symbols, hence, the resulting matrix 5 does not have a row with all "a" symbols, are so the decomposition does not have the lossless join property.

### 11.1.4 Testing Binary Decompositions for the Nonadditive Join Property

Algorithm 11.1 allows us to test whether a particular decomposition D into a relations obeys the lossless jum property with respect to a set of functional dependencies E. There is a special case of a decomposition called a binary decomposition —decomposition of a relation K into two relations. We give an easier test to apply than Algorithm 11.1, but while it is very handy to use, it is lought to binary decompositions only.

(a) (b)

#### 31 A-/SSN ENAME PNUMDER PNAME PLOCATION, HOURST ALEEMP LOOS-FENAME, PLOCATION R2=EMF PROJ1-(SSN, PNUMBER, HOURS, PNAME, PLOCATION).

A-VISN-HENAMELPHUMSER-(PNAME, PLOCATION) (SSN, PNUMBER)-HOURS)

|       | _55N             | ENAME    | PNUMBER         | PHAME | PLOCATION | HQURS           |  |
|-------|------------------|----------|-----------------|-------|-----------|-----------------|--|
| Ŷı    | ъ <sub>1</sub> . | 8.2<br>8 | ь <sup>13</sup> | 114   | a 5       | <sup>0</sup> 16 |  |
| $s_2$ | ٦ı               | b 22     | a 3             | 34    | ° 5       | ٩e              |  |

(no changes la maire allér applying lunctional dependencies)

άł,

| EWP          |       | PROJECT |       |           | WORK | S_ON    |        |
|--------------|-------|---------|-------|-----------|------|---------|--------|
| <u>इन्</u> य | ENAME | PNUMBER | MNAME | FLOCATION | SSN  | HNUMBER | HCRIHS |

P+(SSN, ENAME, PNUMBER, PNAME, PLOCADON, HUUHS)
 P(+EMP+(SSN, ENAME)
 P<sub>2</sub>=PF0.1=(PNLMBEH, PNAME, PLOCATION)
 P<sub>3</sub>=WORKS\_ON+(SSN, PNLMBER, HOURS)

F (55N---(ENAME, PNUMBER-+(FNAME, PLOCATION), (55N, PNUMBER)-+HOURS)

|    | 55N             | ENAME           | PNUMBER         | Phase           | PLOCATIÓN        | HOURS           |  |
|----|-----------------|-----------------|-----------------|-----------------|------------------|-----------------|--|
| e, | а,              | *°              | ь <sub>13</sub> | t 14            | D 15             | <sup>р</sup> (с |  |
| Pg | <sup>b</sup> 21 | <sup>6</sup> 22 | ē 2             | <sup>0</sup> 4  | * <u>-</u>       | <sup>6</sup> 26 |  |
| Чэ | ינ              | °%              | <sup>9</sup> З  | <sup>0</sup> 34 | <sup>b</sup> .75 | °6              |  |

torginal matrix 5 at start of algorithmic

|    | SSN             | ENAME           | PMMBEP     | PNAME               | PLOCATION            | HOURS          |  |
|----|-----------------|-----------------|------------|---------------------|----------------------|----------------|--|
| п. | a <sub>i</sub>  | °2              | E 13       | n 14                | <sup>0</sup> 15      | <sup>ь</sup> ю |  |
| 82 | ь <sub>21</sub> | <sup>0</sup> 22 | <b>*</b> 3 | <sup>a</sup> 4      | "5                   | t 26           |  |
| Pg | ¥ 1             | D 32<br>55 2    | а з        | ⊳ <mark>34</mark> * | ۵ <sub>:35</sub> × 5 | ۰ <sub>6</sub> |  |

metrix 5 alter applying the List two functional dependencies tast row is all fail symbols, so we stop)

FIGURE 11.1 Loss/ess (nonaddiated) join test for n-ary decompositions. (a) Case 1: Decomposition of certerio into the tree: 1 and are toos fails test, (b) A decomposition of ereterio that has the lossless form property. (c) Case 2: Decomposition of eretrees into evel estour, and were to satisfies test.

#### PROPERTY LI1 (LOSSLESS JOIN TEST FOR BINARY DECOMPOSITIONS)

A decomposition  $D = (R_1, R_1)$  of R has the lossless (nonaddirive) pair property with respect to a set of constional dependencies F on R if and only if either

- The  $\operatorname{Fl}_{ij}(R_1 \cap R_j) \to (R_1 = R_j \cap \operatorname{sub} F)$ , or
- The Heig( $R_1 \cap R_2$ ) = ( $R_2 = R_1$ )) is in F'

You should verify that this property holds with respect to our informal successive norm distance examples in Sections 10.3 and 10.4.

### 11.1.5 Successive Lossless (Nonadditive) Join Decompositions

We saw the successive decomposition of relations during the process of second and third normalization in Sections 10.3 and 10.4. To verify that these decompositions are nonditive, we need to ensure another property as set forth in Clonic 2.

#### CLAIM 2 (Preservation of Nonadditivity in Successive Decompositions)

If a decomposition  $D = [R_1, R_2, \dots, R_n]$  of R has the nonadditive (losslos) jump property with respect to a set of functional dependencies F on R, and if a decomposition  $D_i = [Q_0, Q_0, \dots, Q_n]$  of R has the nonadditive purproperte order respect to the projection of F on R, then the decomposition  $D_2 = [R_1, R_2, \dots, R_n]$  $Q_1, Q_2, \dots, Q_n, R_{-1}, \dots, R_n$  of R has the randomize join property with respect to h.

## 11.2 ALGORITHMS FOR RELATIONAL DATABASE SCHEMA DESIGN

We now give three algorithms for creating a relational decomposition. Each algorithm has specific properties, as we discuss below.

### 11.2.1 Dependency-Preserving Decomposition into 3NF Schemas

Algorithm 11.2 creates a dependency-preserving decomposition  $D = R_1, R_2, \dots, R_n$  (or universal relation K based on a set of functional dependencies F, such that each K in D is in 55.2.3 togrammers only the dependency-preserving property of does not guarantee du lossless pain property. The first step of Algorithm 11.2 is to find a manimal cover G for S Algorithm 10.2 can be used for this step.

Algorithm, 11.2) Relational Synthesis into 3NF with Dependency Preservation Imput: A innversal relation R and a set of functional dependencies F on the attributes of R

- End a communication of Grow F fuse Algorithm 10.25;
- For each left-hand-side X of a functional dependency that appears in G, create a relation schema in D with attributes {X ∪ [A<sub>1</sub>] ∪ [A<sub>2</sub>] ∪ [A<sub>3</sub>], where X → A<sub>1</sub>, X → A<sub>2</sub>,..., X → A<sub>4</sub> are the only dependencies in G with X as the left hand side (X is the key of this relation);
- Show any remaining attributes (that have not hear placed in any relation) in a single relation sub-market ensure the artitibute preservation property.

#### CLAIM 3

Every relation schema created by Algorithm 11.2 is in 38% (We will not provide a formal proof hered the proof depends on G being a minimal set of dependencies.)

It is obvious that all the dependencies in G are preserved by the algorithm because each dependency appears in one of the relations R, in the decomposition D. Since G is equivalent to 5 all the detendencies in *F* are either preserved directly in the decomposition or are detable using the dependency preservation (10.2.2 from those in the resulting relations, this ensuing the dependency preservation (resperts). Algorithm 11.2 is called the relational arthesis algorithm, because each relation schema R, in the decomposition is sentice-sent sentences) from the sentence of functional dependences in G with the same left-band-side X.

### 11.2.2 Lossless (Nonadditive) Join Decomposition into BCNF Schemas

The next algorithm decomposes a universal relation schema  $\mathbb{R} = \{X_1, X_2, \dots, X_n\}$  into a decomposition  $D = \{R_1, R_1, \dots, R_n\}$  such that each  $R_i$  is in BCNF and the decomposition D has the loadess jum property with respect to F. Algorithm 11.3 uniform Property LU and Caure 3 (preservation of monadditivity in an eccusive decompositions) to create a nonadditive jum decomposition  $D = \{\mathbb{R}_1, \mathbb{R}_2, \dots, \mathbb{R}_n\}$  of a universitive factor  $\mathbb{R}$  based on a set of instantal dependencies  $\mathbb{R}$  such that each  $\mathcal{R}_i$  in D is in BCNF.

Algorithm 11.35. Relational Decomposition into BUNE with Nonaddiover Join Property Input: A universal relation R and a set of materior al dependencies F on the attributes of R.

```
1. Set D .= (#);
```

```
). While there is a relation schema Q in D that is not in 000 do
```

```
 \begin{array}{l} 1 \\ \text{shows a relation schema Q in O that is not in 10 w;} \\ \text{find a functional dependency X <math>\rightarrow Y in Q that violates use,
 triplace Q in D by two relation schemas (Q - Y) and (X \cup Y),
 }.
```

No Muer (1986) or Udman (1982) for a prost.

Each time through the loop in Algorithm 11.5, we decompose one relation schemaQ that is not in 30NF into two relation schemas. According to Property 1,1 for binar decompositions and Chara 2, the decomposition D has the nonadditive purp propert. At the end of the algorithm, all relation schemas in D will be in 90NF. The reader can check that the normalization example in Figures 10.11 and 10.12 hesteally follows this algorithm. The functional dependencies (D3, F04, and later 705 violate 90NT, so the 10Trelation is decomposition dependencies (D3, F04, and later 705 violate 90NT, so the 10Trelation is decomposition than satisfies the miniadditive joint property. Similarly, it we apply the algorithm to the field relation schema from Figure 12.13, it is decomposed into 16001 (<u>Instituction</u>, <u>stopen</u>) and 18400 (<u>instituctor</u>, const.) because the dependency FD2: (<u>instituctor</u>, const.)

In step 2 of Algorithm 12.3, it is necessary to determine whether a relation schema Q is in g N- or not. One method for during this is to test, for each first total dependency  $\lambda \to 0$  in Q, whether  $\lambda^*$  (link to us lide all the autibates in Q, thereby determining whether err not  $\lambda^*$  is a (superfixer in Q. Another rechnique is based on an observation that whetever a volation schema Q violates BONE, there exists a pair of combates A and B in Q such that  $\{Q = 1A, B\}\} \to A_1$  by computing the closure  $\{Q = 1A, B\}\}^*$  for each pair of autibates [A, B] of Q and checking whether the closure includes A for B), we can determ ne whether Q is in  $\mathbb{R}$  NE.

### 11.2.3 Dependency-Preserving and Nonadditive (Lossless) Join Decomposition into 3NF Schemas

If we want a decomposition to have the nonadditive join property and to proverve dependencies, we have to be satisfied with relation schemas in 3NF rather than BCNF. A simple module ition to Algorithm 11.2, shown as Algorithm 11.4, yields a decomposition D at 8 that does the following:

- Preserves dependencies
- Has the nonadditive join property.
- lasuch that each resulting relation schema in the decomposition is in TNF

Algorithm 11.4: Relational Synthesis into 5NF with Dependency Preservation and Non-addrive (Lossless) Join Diopetty

Input: A universal relation R and a set of functional dependencies F on the attributes of R.

- Find a normali cover G for E (use Algorithm 10.2).
- For each left hand-side X of a functional dependency (but appears in C create a relation schema in D with attributes [X ∪ [A<sub>1</sub>] ∪ {A<sub>2</sub>}....∪ [A<sub>i</sub>], wher X → A<sub>1</sub>, X → A<sub>2</sub>, ..., X → A<sub>i</sub> are the only dependencies in G with X as left hand-side (X is the key of this relation).
- 3. If none of the relation schemas in D contains a key of R, then create one more relation schema in D that contains attributes that form a key of R.

It can be shown that the decomposition formed from the set of relation schemas usae? by the preceding algorithm is dependency preserving and has the nonadditice propoperty. In addition, each relation schema in the decomposition is in 3NF. This algorithm is an improvement over Algorithm 11.2 in that the former guaranteed only espendency preservation."

Step 3 et Algorithm 13.4 involves identifying a key K of R. Algorithm 11.4a can be belief dentify a key K of R based on the set of given functional dependencies F. We start by setting K to all the arributes of R, we then remove one antibute at a time and check whether the remaining attributes still form a superkey. Notice that the set of functional dependencies used to determine a key in Algorithm 11.4a could be either F or G, since the are equivalent. Notice, not, that Algorithm 11.4a determines only one for our of the pasible condidate keys for R, the key reformed depends on the order in which attributes are removed from R in step 2.

Algorithm 11.48: Finding a Key K for R Given a set F of Functional Dependencies Input: A universal relation R and a set of functional dependencies F on the intributes of R.

- Set K := R.
- 2. For each attribute A in K

konspare (K - A)\* with respect to Fe

 $F(K - A)^*$  contains all the attributes in  $R_i$  then set  $K := K - \{A\}$ ;

it is important to note that the rheory of nervalditive turn decompositions is based on the assumption that wound values are allowed *for* the *you* attributes. The next section discuss some of the problems that nulls may cause in relational decompositions.

### 11.2.4 Problems with Null Values and Dangling Tuples

We user carefully consider the problems associated with rulls when designing a rule ional database schema. There is no fully soristocrory relational design theory as yer that incide null values. One problem occurs when some ruples have rull values for attributes that will be used to join individual relations in the decomposition. To the state this, conaler the database shown in Figure 11.2a, where two relations preterior and more state this, conaler the database shown in Figure 11.2a, where two relations preterior and more states the data the database shown in Figure 11.2a, where two relations preterior and more states the database who have not yet been assigned to a department to sume that this does not ordate any integrity constraints). New suppose that we want to retrieve a list of (2009), over values for all the employees. If we apply the NATORALION operation on EMPLOYEE indepretent (Figure 11.2b), the over aforementioned topics will out opped in the result

<sup>3.86 3</sup> or Algorithm 12.2 is not needed in Algorithm 12.2 to preserve attributes because the key all colore asy applaced attributes: these are the attributes that do not participate in any forceteria desculation.

(a) EMPLOYEE

|                    |             | — — —      | · <u>-</u> ··            | <u> </u> |
|--------------------|-------------|------------|--------------------------|----------|
| FNAVE              | 55 <u>4</u> | BOATE      | i ADOFESS                | DN JM    |
|                    | <u> </u>    | _          |                          | · · _    |
| She's Join B       | 12545878.2  | 1965-01-09 | 781 Server Hereiten TK   | 5        |
| Wong, Frankin T.   | 333445555   | 1953-12-00 | 638 yess Housian TX      | 5        |
| Zeleva, Alicia J   | 999067777   | 1964-07-10 | 335 Could, Spring TX     | +        |
| Wallace Jennier S  | 967654321   | 1941-06-20 | 291 Borry, Bellave, 1X   | 1        |
| Narayan, Hartesh K | CICHERTON   | 196-09415  | 975 File Cak, Humble, FX | 5        |
| English any + A    | 45.3453453  | 1972-07-01 | 562: File Hendikas TD    | 5        |
| Jabbar Ahmad V     | 65/38/987   | 1959-03-25 | 950 Callas, Pousion, TX  | -1       |
| Rosp Janes F       | 89665555    | 1997-11-10 | 450 58.0+, Haiston, TX   |          |
| Berger Anders C    | 996775555   | 1963-04-25 | 6530 Brows, Bellare, TX  | nu       |
| Eventez, Cerkas M  | N808544-14  | 1965-01-09 | 7654 Вники, тахалик ТХ   | Decili   |
|                    |             |            |                          |          |

| DECKERTICKER |
|--------------|
|              |
|              |
|              |

| UNAME         | <u></u> | DVOB555   |  |  |
|---------------|---------|-----------|--|--|
| Hesewort      | 5       | 313445558 |  |  |
| Annasaasa     | 4       | 927564321 |  |  |
| Headq: arters |         | 10666555  |  |  |

| -   |            |
|-----|------------|
| п   | <b>S</b> 1 |
|     |            |
| ••• |            |

|                   | <u>995</u> | PDATE      | 4004634                 | LWOV | DATANE           | Divigenças. |
|-------------------|------------|------------|-------------------------|------|------------------|-------------|
| Smithuchn B.      | 125465789  | 1065-01-09 | Zer Fondren Houston TX  | 5    | Pasearch         | 333445355   |
| Wong Wonlow T     | 333145556  | 1995-12-18 | 638 ybas Morelon TX     |      | Cesearch         | 0000015555  |
| Zolaya Alaza .    | 9-0-67177  | 1985-07-19 | . Д21 Снањ: Брлоз, TX   | +    | Accessed and a   | 587654321   |
| Valacci Jernier S | 987654121  | 1941 06-30 | 2ut Berry Solare 1X     | 4    | Adversariation   | 997154321   |
| ындан Альник К    | 606084434  | 196-409-15 | 975 File Oak Hombel TX  | - 5  | Fuce-ends        | 3334415656  |
| English Joyce A   | 453463453  | 1972 67 21 | 5631 Pice, Hausten TX   |      | Research         | 333445655   |
| Jebka, Alarac y   | 957967907  | 1969-03-29 | 900 (talas Historian TX | -    | As homes release | 967654321   |
| Bonji James E     | 323665656  | 1937-11-10 | 460 Stare, Houston, TX  | •    | Headquarters     | 859665505   |

I¢I.

| CRAWE               | <u>35N</u> | 00/01       | ADODESS                   | DNUM | ONAME -                         | CMC355N           |
|---------------------|------------|-------------|---------------------------|------|---------------------------------|-------------------|
| Smith, John J       | 123456769  | /903-0-02   | 731 Foodern Houston, TK   |      | Hestath                         | 737443535         |
| Vicinic: Frenken T  | 33 446565  | : 956-12-CH | 100° у жы таказий ТХ      | 5    | Fleationards                    | 310446565         |
| Zelava Alme J       | 999867777  | 1908/07-15  | 3321 Gastic Spring 1X     | ≤ 1  | Administration                  | 967654321         |
| Whiteon descales (c | 987654321  | 1941-08-20  | 291 Overty, Redware, TX   | 4    | Allmonedighters                 | <b>4</b> 8.9543-1 |
| Nalayan Romosti K   | 660894444  | 962-00-15   | \$75 File Day, Humple, TX | 5    | Riscart                         | 353446565         |
| Emploi Jame A       | 455453453  | 1972-07-31  | G531 Face, Houseker, 191  | ,    | (leseart)                       | 333440305         |
| Jacob Ahrran V      | 087987957  | 959-03-24   | 99 Jans Folson IX         | ź.   | Acronshaloo                     | 367654321         |
| Borg James L        | 603005555  | 1027-11-10  | 450 Stone, Houston, 1N    | 1    | <ul> <li>Macquarters</li> </ul> | 988665050         |
| Feisper Arkleis 2   | 907/755555 | 1966-04     | (243) States Betwee TX    | nuli | I HU                            | nult              |
| Bennez Carlos V     | \$8506-244 | 1963-01-00  | 7664 Boeth, Hauslah, UK   | ruli | null                            | nuli              |

**FIGURE 11.2** Issues with multivalue joins. (a) Some average tuples have null for the join attribute invite the Result of applying NV198A constants electored and preventient relations. (c) Result of applying 1-21 v0.168 (ON to overvice and occarrient.

The COTER JOIN operation, discussed in Chapter 6, can deal with this problem. Recall torn we take the LEFT OUTER (CIN of EMPLOYES with LEMARYEM), tuples in EMPLOYES that have null for the point attribute will still appear in the result, joined with an "imagencary" uplem DEMARYEMT that has nulls for all its attribute values. Figure 11.2c shows the result.

In general, whenever a relational database schema is designed in which two of more relations are interrelated via forcing keys, particular cure must be devoted to watching for priorital null values to forcing keys. This can cause unexpected loss of information in queres that unerflye priors on that forcing keys. Moreover, if builts occur in other attributes, such as such, their effect on built-in functions such as so, and average must be carefully evaluated.

A related problem is that of danglag tapler, which may occur at we carry a decorposition for far. Suppose that we decompose the eventses relation of Figure 11.2a independent 1 and eventses 2, shown in Figure 11.3a and 11.3H.<sup>1</sup> If we apply the NPTER'S 100K operation to eventses 1 AND eventses 2, we get the original eventses relation. However, we may use the alternative representation, shown in Figure 11.3a, where we do not online a caple in eventses 1 if the employee has not been assigned a devotent (instead of metudes) and apply a NVD realist on tectores\_2). If we use perfects instead of recordse\_2 and apply a NVD realists on tectores\_1. In devote 3, the representation in tectores 3, the representation in tectores 3, and apply a NVD realists on tectores\_1. In devote 3, the representation in the result these are called **dangling** inples because they are represented in only one of the two relations that represent imployees and hence are lost of we apply an (INNER) (AND event).

### 11.2.5 Discussion of Normalization Algorithms

One of the problems with the normalization algorithms we described is that the datahas designed must first specify all the relevant functional dependencies among the catabase antributes. This is not a simple task for a large database with hundreds of autisutes. Earline to specify one or two important dependencies may result in an indeable design. Another problem is that these algorithms are not deterministic in general, fin example, the synthesis algorithms (Algorithms 11.2 and 11.4) require the specification of a mountal cover G for the set of functional dependencies A Bicarise flure may bein general many mountal covers dottesponding to E the algorithm can give different issues depending on the particular mountal cover used. Some of these designs may not be describle. The documponium algorithm (Algorithm 11.3) depends on the order in which the functional dependencies are supplied to the algorithm to check for texts the ans set of functional dependencies, depending on the order in which such dependencies to the ansist of functional dependencies, depending on the order in which such dependencies are supplied to the algorithm to check for texts the ansist of functional dependencies, depending on the order in which such dependence (c) an considered for violation of PLNF. Some of the designs may be quite superior, three others may be undestrable.

A flas sometimes happens when we apply vertical tragmentation to a relation in the context of a anihured duribuse (see Chapter 25)

| ENAME                                            | : I                    | 2594      | BDATE      | AUCPE                                            | ISS         |
|--------------------------------------------------|------------------------|-----------|------------|--------------------------------------------------|-------------|
| Smith, John D                                    | l.                     | 123456789 | 1965-01-09 | 731 Fandron                                      | Houston T   |
| Wong Frankli                                     | гT                     | 333445555 | 1955-12-06 | 638 V9sa, Ho                                     | uelon, TX   |
| Zolaya, Akcia,                                   | <b>-</b> .             | 993887777 | 1968-07-19 | 3321 Castle,                                     | Spring, TX  |
| Warace, Jeon                                     | wer S.                 | 967654321 | 1941-06-20 | 291 Berry, Be                                    | stare. Di   |
| Narayan Ban                                      | nesh K                 | 656984a44 | 1962-09-15 | 975 Eve (Jak                                     | , humble, D |
| English, Javoa                                   | :A                     | 053453453 | 1072-07-01 | 9631 Pice, II                                    | custon, TX  |
| Jabber, Ahma                                     | e V.                   | 387387387 | 1969-03-28 | 980 Dallee H                                     | iousion, 77 |
| Gorg, James R                                    |                        | 808865555 | 1937-11-10 | 450 Sione, H                                     | kusion "X i |
| Bargar Andar                                     | кĆ                     | 999775555 | 1965-04-26 | 6630 Brees I                                     | Bellare, TX |
| Bentez, Carlo                                    | sМ                     | 088554444 | 1063-01-09 | 7654 Beech                                       | Houston D   |
|                                                  | DNUM                   |           | (M         | SSN                                              | ONUM        |
| 123456789                                        | Ŀ                      |           |            | 123456789                                        | 5           |
| 330445566                                        | 5                      |           |            | 333445555                                        | 5           |
| 305887777                                        | 4                      |           |            | 999887777                                        | 4           |
| 987654321                                        | 4                      |           |            | 987654321                                        | 4           |
| KEE994444                                        |                        |           |            | 655534444                                        | ÷           |
|                                                  | ••                     |           |            |                                                  |             |
| 452453453                                        | 5                      |           |            | 453453453                                        | 5           |
| 452453453<br>387(387987                          | <br>5<br>4             |           |            | 453453453<br>987987987                           | 5<br>+      |
| 452453453<br>387087987<br>889865555              | <br>5<br>4<br>1        |           |            | 453453453<br>987987987<br>88 <del>966</del> 5665 | 5<br>+<br>1 |
| 452453453<br>387087987<br>889965555<br>899775555 | <br>5<br>4<br>1<br>Ayl |           |            | 453453453<br>987987987<br>988995065              | 5<br>+<br>1 |

(a) EMPLOYEE\_1

FIGURE 11.3 The "daughting tuple" problem (a) The relation (waver\_1) (includes all attributes of evelowe from Figure 11.2a except own, that The relation evelopes 2 (includes ever attribute with null values). (c) The relation terms: 3 (includes own attribute but does not include tuples for which usin has null values)

Table 11.1 summarizes the properties of the algorithms discussed in this thingter schu-

| Algorithm | INPUT                                                                  | OUTPUT                                                       | PROPERTIES/PURPOSE                                                | REMARKS                                                              |
|-----------|------------------------------------------------------------------------|--------------------------------------------------------------|-------------------------------------------------------------------|----------------------------------------------------------------------|
| 11.1      | A decomposition<br>D of R and a set F<br>of functional<br>dependencies | Boolean result: yes<br>or no for nonaddi<br>Gve jum property | Tusting for<br>nursadditive jum<br>decomposition                  | See a simpler test in<br>Section 11.1.4 for<br>binary decompositions |
| P 2       | Set of functional<br>dependencies F                                    | A set of relations in<br>SNF                                 | Dependency<br>preservation                                        | No guarance of<br>secisfying lossless join<br>property               |
| 1.1       | Set of filnetional<br>dependencies ₽                                   | A set of relations in<br>BCNF                                | Noradditive join<br>decomposition                                 | No goarantee of<br>dependency<br>preservation                        |
| 1.4       | Set of Janetional<br>dependencies F                                    | A set of relations in<br>3NE                                 | Nonadditive join<br>AND dependency<br>preserving<br>decomposition | May not achieve<br>PCNF                                              |
| II.Ar     | Relation schema<br>R with a set of func-<br>tional Jupsenderigues F    | Key K of R                                                   | To find a key $K$<br>(that is a subset of $R$ )                   | The entry relation R is<br>abvays a default<br>superkey              |

TABLE 11.1 SUMMARY OF THE ALGORITHMS DISCUSSED IN SECTIONS 11.1 AND 13.2

## 11.3 MULTIVALUED DEPENDENCIES AND FOURTH NORMAL FORM

Sofer we have discussed only functional dependency, which is by far the most important type of dependency in relational database design theory. However, in many cases relations have constraints that connect be specified as functional dependencies. In this section, we datase the concept of multitudied dependencies (SPDI) and define fourth meaned form, which abased on this dependency. Multivalued dependencies are a consequence of first normal brits (DSF) (see Section 10-3.4), which disallows an artificite it a tuple to have a set of abase. If we have two or more multivalued independencies antifures in the same relation abase. If we have two or more multivalued independencies arrithmes in the same relation abase. If we have two or more multivalued independencies arrithmes in the same relation abases are generated as problem of having to repeat every value of one of the attributes onlinevery value of the other attribute to keep the relation state consistent and to minitan dat independence among the attributes involved. This constraint is specified by a ultivalued dependency.

for example, consider the relation EPP shown in Figure 11.4.1. A tuple in this EPP inhibit represents the fact that an employee whose name is E94PE works on the project slow name is E94PE and has a dependent whose name is E94PE. An employee may work in avail projects and may have several dependents, and the employee's projects and
(al EAPP

| EN4ME  | PNAAE | DNAME |
|--------|-------|-------|
| Smith  | х     | Jchn  |
| Smith  | Y     | Arria |
| Smith  | ×     | Anna  |
| Sirith | Y     | JC141 |

#### (b) EMP\_PROJECTS

#### EMP\_DEPENDENTS

| ENAME          | PNAM5        | ENAME | [:NAME   |
|----------------|--------------|-------|----------|
| Sinti<br>Senti | ' <u>*</u> ' | Sreta | <br>John |

#### (c) SUPPLY

| SNAME   | PAPTNAME | PROJNAME |
|---------|----------|----------|
| Smith   | Bolt     | Projk    |
| Smith   | Nuç      | Proy     |
| Adamsky | Bek      | Picjy    |
| Wahoo   | N.g      | ProjZ    |
| Ademsky | Nat      | ProjX    |
| Adamsky | Hot      | Нітух    |
| Smith   | Bat      | ProjV    |

| (d) | F1      |          | R2       |          | R3       | R3      |  |
|-----|---------|----------|----------|----------|----------|---------|--|
|     | SNAME   | PAPENAME | SNAME    | PROJNAME | PAHTMAME | PHUMANE |  |
|     | Smith   | Bott     | Smith    | PraX     | Batt     | Prox    |  |
|     | Smin    | Not      | 5mil-    | Proje    | Nut      | ProjY   |  |
|     | Adamsky | Boh      | Adamsky  | ProjY    | Boe      | Pro Y   |  |
|     | Wellow  | Nin      | Welton   | PosZ     | Not      | ProjZ   |  |
|     | Adamsky | Nat      | Adametey | Pig/X    | Ned      | BojX    |  |

**FIGURE 11.4** Fourth and fifth normal forms, ta: The periodation with two MVLS: 1996. — every and two we clatter with two MVLS: 1996. — every and two models and two models and two models and two models and two models. The relation stream with no MVDs is in 4NF but not on 5NF of this the IDIRT. B2, B3: 30 (Decomposing the relation stream into the 5NF relations B1, B2, B3.

dependents are independent of one mother.<sup>5</sup> To keep the relation state consistent or must have a separate tuple to represent every combination of an employee's dependent and on employee's project. This constraint is specified as a multivabled dependence at the ess relation. Informally, whenever two independencies's relationships A:B and A Clart mixed in the same relation, an MyD raty arise.

 In an FR diagram, each acceld be represented to a multi-called autobate on is a weak entry type (see Cooper 5).

## 11.3.1 Formal Definition of Multivalued Dependency

**Definition.** A multivalued dependency  $X \rightarrow X$  specified on relation schema R, where  $\lambda$  and Y are both subsets of R, specifies the following constraint on any relation variable R friction inples  $\eta$  and  $\eta$  exists in r such that  $r_{1}[X] = r_{2}[X]$ , then two ruples  $\eta$  and  $\eta$  exists in r such that  $r_{2}[X] = r_{2}[X]$ , then two ruples  $\eta$  and  $\eta$  exists in r such that  $r_{2}[X] = r_{2}[X]$ , then two ruples  $\eta$  and  $\eta \in \mathbb{R}$ . The following properties  $\mathbb{C}$  where we use  $\mathbb{Z}$  to denote 4R = 0.0530

- $\{i_i[X]+i_i[X]=r_i[X]+r_i(X)$
- $(|Y| i_1|Y| \text{ and } i_2|Y| i_3|Y|$ .
- (3) = (32) and (42) = (42).

Whenever  $X \rightarrow Y$  holds, we say that X multidetermines Y Because of the symmetry **n** the definition, whenever  $X \rightarrow Y$  holds in R, so does  $X \rightarrow G$ . Hence,  $X \rightarrow Y$  inplus  $X \rightarrow Z$ , and therefore it is sometimes written as  $X \rightarrow YLZ$ .

The formal definition specifies that given a particular value of X, the set of values of Y freemod by this value of X is completely determined by X alone and due not depend on the values of the termaining attributes  $\mathbb{Q}$  of R. Hence, whenever two tuples exist that have  $\operatorname{ipt}_{X'}$  values of Y but the same value of X, these values of Y must be repeared in separate tiples with overy domain value of  $\mathbb{Q}$  that occurs with that same value of X. This informally corresponds to Y bring a matrixable 3 and but of the cutities represented by tuples in R.

In Figure 11.4a the SW(S EVAN  $\rightarrow a$  1995, and (SANE  $\rightarrow a$  1997) for EVANE  $\rightarrow a$  1997 (and the every relation. The employee with EVANE (SANE) works on projects with 1999 (X) and (Y) and has reprodents with 1999 (John Land (Anna)). If we considerly the first two option and two dependents with 1999 (John Land (Anna)). If we considerly the first two option and two dependents with 1999 (John) and (Anna). If we considerly the first two option and (Anna), if we considerly the first two option and (X) an

An SVD X  $\rightarrow$  Y in R is called intrivial KVI of 40 V is a subset of X, or  $\phi F X \cup Y = R$  for example, the relation FSP\_PROFETS in Figure 11.4b has the trivial MVD FRAME  $\rightarrow$  FRAME. An SVD that satisfies neither (a) nor (b) is called a **nontrivial** MVD. A trivial OC collided in any relation state r of *R*: it is called the all recause it does not specify any again and or meaningful constraint on *R*.

If we have a nontrivial SWF in a relation, we not have to repeat values related antly in the toples. In the EFF relation of Figure 11.4a, the values "X" and "Y" of Front are optical with each value of DOFF (ar, by symmetry, the values "Juhn" and "Anna" of Soot an repeated with each value of DOFF. This redupding its clearly undesirable. However, there schema is in 2006 because no functional dependencies hold at 199. Therefore, we

when getex present, and appearing increasion to distinct.

<sup>1,2</sup> is shorthand for the antibules remaining in K otter the attributes in  $(X \cup Y)$  are removed for X

need to define a fourth normal form that is stronger than FCNF and disallows relation schemas such as 199. We first discuss some of the properties of MVDs and consider how rhey are related to functional dependencies. Notice that relations containing nontrivial MVDs rend to be all-key relations – that is, their key is all their attributes taken rogether.

# 11.3.2 Inference Rules for Functional and Multivalued Dependencies

As with functional dependencies (EOs), inference rules for multivalued dependences (NVUs) have been developed. It is herein though, to develop a unified framework that includes both EOs and NVOs so that both types of constraints can be considered rogener. The following interence rules IOC through R8 form a sound and complete set for inderving functional and multivalued dependencies from a given set of dependencies. Assume that all attributes are included in a "ontwiscid" relation scheme  $R = \{A_1, A_2, \ldots, A_n\}$  and that X, Y, Z, and W are subsets of R.

- IF I frothering rule for FDS): If  $X \supseteq Y$ , then  $X \to Y$ .
- (82) faagmentation rule for  $G(S): \{X \rightarrow Y\} \vdash XZ \rightarrow YZ$ .
- (B3 (transitive rule for E(s):  $(X \to Y, Y \to Z) \models X \to Z$ .
- R4 (complementation rule for SVDs):  $|X \rightarrow Y| = |X \rightarrow (R = (X \cup Y))$ .
- (c) (sugnegration rule for WyDs):  $0.\% \rightarrow 0.\%$  and  $W \supseteq \mathbb{Z}_0$  then  $WX \rightarrow 0.\%$ .
- af (manarize pule for MyDe):  $(X \rightarrow Y, Y \rightarrow Z) \models X \rightarrow \mu^{-}(Z = Y)$ .
- 6.7 (replication rule for ED to MVD):  $|X| \rightarrow Y\} \models X \implies |Y|$
- R8 (condescence rule for (1) and (0)(3). If  $X \to Y$  and there exists W with the properties that (a)  $W \cap Y$  is empty (b)  $W \to Z$ , and (c)  $Y \supseteq Z$ , then  $X \to Z$ .

III through IR3 are Armstrong's inference roles for FDs alone. IR4 through IR6 are inference rules pertaining to SWTs only III7 and IRS relate TCS and KWTs. In particula IR7 says that a functional dependence is a special case of a mality along dependency that is every ID is also an SWD because it satisfies the formal definition of an KWD. However, this equivalence has a catch: An ID  $X \to Y$  is an MVD  $X \to -Y$  with the additional infect restriction that at most one value of Y is associated with each value of  $X^*$ . Given a set Fof functional and multivalued dependencies specified on R = {A<sub>1</sub>, A<sub>2</sub>, ..., A<sub>n</sub>}, we can ge IR1, through IRS to enter the (complete) set of all dependencies (functional in multivalued) F\* that will hold in every relation state **r** of K that satisfies E. We again cul-F\* the closure of *F*.

<sup>5.</sup> That is, the series values of Y determined by a value of X is restricted to being a soughts, set out only one value. Hence, its practice, we never your an FD as an MVD.

# 11.3.3 Fourth Normal Form

We now present the definition of **fearth normal form** (486), which is violated when a alaton has unde-mable multivalued dependencies, and hence can be used to identify and Jucompose such relations.

**Definition.** A relation schema R is in **4NF** with respect to a set of dependencies P (nar includes functional dependencies and multivalued dependencies) it. for every national multivalued dependency  $X \rightarrow Y$  in F', X is a superkey for R

To illustrate the importance of  $\{N\}$ , Figure 11.56 shows the secretation with an additional employed, 'Brown', who has three dependents ('Jun', 'Juan', and 'Bob') and ackson four different projects ('W', 'X', 'Y', and 'Z'). There are its ruples in two in Figure 11.56, if we decompose each into two seconds and each projects, as shown in Figure 11.56, we need to store a total of only 11 tuples in both relations. Not only would the decomposition save on storage, but the apdate anomalies associated with inultivalued decomposition save on storage, but the apdate anomalies associated with inultivalued decomposition save on storage. For example, if Brown stars working on a new

| 12. | EMP   |       |       | ;Ľ | \$        | ICTS             |
|-----|-------|-------|-------|----|-----------|------------------|
|     | ELAME | PNAME | DNAME |    | ENAME     | PNAME            |
|     | Smith | x     | John  |    | ടന്നർ     | x                |
|     | Smith | Y     | Appa  |    | Smith     | Y                |
|     | Smith | ×     | Anna  |    | Brown     | W.               |
|     | Smith | Y     | John  |    | Brown     | x                |
|     | Brown | W     | Jo    |    | Brown     | Y                |
|     | Bicwn | x     | Jan   |    | Brown     | z                |
|     | Biown | Ŷ     | Jun   |    |           |                  |
|     | Biowo | z     | Jint  |    | EMP DEPEN | DENTS            |
|     | Brown | w     | Joan  |    |           |                  |
|     | Brown | х     | Joan  |    | ENAME     | DNAME            |
|     | Brown | Y     | Joan  |    |           | · · · · <u>-</u> |
|     | Brown | z     | Juan  |    | Smth      | Anna             |
|     | Brown | w     | Bob   |    | Smith     | John             |
|     | Brown | х     | Bob   |    | Brown     | Je               |
|     | Brewn | Y     | Bob   |    | Brown     | Jcan             |
|     | Brown | z     | Bob   |    | Brown     | Boh              |

**NGURE 11.5** Desimpliosing a relation state of the that is not in **4NE** ration relation with additional topics. (b) Two corresponding **4NE** relations on <u>\_\_\_\_\_</u>recuts and end operations.

project P, we must insert abree tuples in seconder for each dependent. If we lorger to insert any one of those, the relation violates the MNU and becomes inconsistent in their t incorrectly implies a relationship between project and dependent.

If the relation has nontrivial VADs, then insert delete, and applate operations on single ruples may cause additional ruples besides the one in question to be modified. If the update is bondled incorrectly, the meaning of the relation may change. However, after normalization, into ANE, these update anomalies despical. For example, to solve the information that Provide will be assigned to project P only a single tuple need be inserted in the 4NE relation set projects.

The two relation in engine 11.4a is not in 4NF because it represents two independent inN relationships—one between employees and the projects they work on and the other between employees and their dependents. We sometimes have a relationship imming their entries that depends on all three participating entities, such as the super-relation show in Figure 11.4c. (Consider only the topics in Figure 11.4c above the dotted line for new) in this case a topic represents a supplier supplying a specific part of a *justicular* project so there are no nontrivial MVCs. The SUPPLY relation is already on 4NF and should not be decomposed.

# 11.3.4 Lossless (Nonadditive) Join Decomposition into 4NF Relations

Whenever we decompose a relation schema *R* into  $R_1 = (X \cup Y)$  and  $R_2 = (R - Y)$  basil on an SIVD  $X \rightarrow Y$  that holds in *R*, the decomposition has the nonadditive join groperry. It can be shown that this is a necessary and sufficient condition for decomposites schema into two schemas that have the non-skillarity trun property as given by proper QV' which is a turber generalitation of Property (1) given or her. Property (1) dealt with FDS only, whereas (1) 'deals with both FDS and MVDS (rec of ther an Hers also an SVD)

#### PROPERTY LJ1'

The relation schemas  $R_1$  and  $R_2$  form a nonadditive join decomposition of k with respect to a set F of functional and multivature, dependensies it and only if

 $(R_1 \cap R_2) \rightarrow (R_2 - R_2)$ 

or, by symmetry, it and only if

 $(\mathcal{K}_1 \cap \mathcal{R}_2) \rightarrow (\mathcal{R}_2 - \mathcal{R}_1).$ 

We can use a slight modification of Algorithm 11.3 to develop Algorithm 115, which creates a nemadditive join decomposition into relation schemas that are in 487 yrather than in TCNT). As with Algorithm 11.5, Algorithm 11.5 does not necessarily produce a decomposition that preserves (1)8. Algorithm 11.5: Relational Decomposition into 484 Relations with Noradditive Ion Dieperty

Input: A universal relation R and a set of functional and multivalued dependencies (?

1 Set D := { R }:

2. While there is a relation schema Q in D that is not in 40%. do (choose a relation schema Q in O that is not an 40% field a nontrivial wit X → - Y in Q that violates 46%, replace Q in O by two relation schemas (Q = Y) and (X ∪ Y), is

# 11.4 JOIN DEPENDENCIES AND FIFTH NORMAL FORM

We on that UI and UI<sup>T</sup> give the condition for a relation schema R to be decomposed no two schemas R<sup>T</sup> and R<sub>2</sub>, where the decomposition has the nonadditive jum propent. However, in some cases there may be no nonadditive jum decomposition of R into no relation schemas, but there may be a remaddirive (lossless) jum decomposition into more dan no relation schemas. Moreover, there may be no functional dependency in R rva violates any normal form up to PCNs, and there may be no functional dependency in R rva violates any normal form up to PCNs, and there may be no functional dependency in R rva violates any normal form up to PCNs, and there may be no functional dependency in R rva violates any normal form up to PCNs, and there may be no functional dependency in R rva violates any normal form up to PCNs, and there may be no functional dependency and gament with a relates ANF. We then resom to another dependency called the jum gaments and at it is present, carry out a inductive decomposition into fifth normal form (SE). It is important to note that such a dependency is a very penalizity scientific curvations want that is very difficult to denser in practice, therefore, normalization into SNF is very andly dorum practice.

**Definition.** A join dependency (JD), denoted by  $\{0, R_1, R_2, \dots, R_n\}$ , securized on identical schema  $R_1$  specifies a constraint on the states Y of  $R_1$ . The constraint states that every legal state r of R should have a non-additive join decomposition into  $R_1, R_2, \dots, R_n$ ; do not or every such 1 we have

 $^{*}(\pi_{\gamma_{1}}(\tau),\pi_{K_{1}}(\tau),...,\pi_{K_{n}}(\tau))$  - r

Notice that an  $MV \supset$  is a special case of a  $\mu$  where  $\mu = 2$ . That is, a  $\mu$  denoted as  $\mu(R_1, R_2)$  implies on  $MVD(R_1 \cap R_2) \longrightarrow (R_1 = R_2)$  (or, by symmetry,  $(R_1 \cap R_2) \longrightarrow (R_2 = R_3)$ ). A join dependency  $(D(R_1, R_2, \dots, R_n)$ , specifies for relation schema  $R_1$  is a trivial  $(0, \mu, \mu)$  is called invial because it has the nonadditive join property for any relation size  $\mu$  of R and hence does not specify any constraint on R. We can now define with normalities, which is also called project-join morphism.

**Definition.** A relation scheme R is in fifth normal form (5NF) (or project-join normal form [P]NF]) with respect to a set F of functional, multivalued, and join dependencies at for every nontrivial join dependency  $\mathcal{D}(R_1, R_2, \ldots, R_n)$  in  $F^*$  (that is implied by F), every  $R_1$  is a superkey of  $R_1$ .

For an example of a [0] consider on elegan the same all-key relation of Eigne 11.4c. Suppose that the following additional constraint always holds: Whenever a supplier v supplies part [0] and a project [1] and part [0] and the supplier v supplies at *least one* part to project [1] show supplier v will also be supplying part [0] in project [1]. This constraint can be restored in other ways and specifies a join dependency U(R1, R2, R3) among the theprojections R1(Super), Ratheore(), R2(Super), Ratheore(), md R3(Constraint), ratheore(), ratheore(), ratheore(), ratheore(), R2(Super), Ratheore(), md R3(Constraint) for the projections R1(Super), ratheore(), R2(Super), Ratheore(), md R3(Constraint), ratheore(), ratheore(), ratheore(), ratheore(), R2(Super), Ratheore(), ratheore(), ratheore(), ratheore(), ratheore(), ratheore(), R2(Super), Ratheore(), ratheore(), R3(Super), ratheore(), ratheore(), ratheore(), ratheore(), R2(Super), R2(Super), ratheore(), ratheore(), R3(Super), ratheore(), ratheore

Discovering (0) in practical databases with bundreds of attributes is these to impossible It can be done only with a great degree of intuition about the data on the part of the designer. Hence, the correct practice of database design pays soon tattention to them.

# **11.5** INCLUSION DEPENDENCIES

Inclusion dependencies were defined in order to formalize two types of interrelational constraints.

- The foreign key (or referential integrity) constraint caunce be specified as a face nonal or multivalued dependency because it relates attributes across relations.
- The constraint between two relations that represent a class/subclass relationship (see Chapter 4 and Section 7.21 also has no formal definition in terms of the function), multivalued, and join dependencies.

**Definition.** An inclusion dependency  $K_i X \le S_i^*$  between two sets of attributes -3 of relation schema  $R_i$  and  $3^\circ$  or relation schema  $S_i^\circ$  specifies the constraint that, at an specific time when  $r_i$  is a relation state of R and s a relation state of  $S_i$  we must have

 $\pi_{\chi}(i(k)) \subseteq \pi_{1}(i(k))$ 

The  $\subseteq$  (subset) relationship does not necessarily have to be a proper subset. Obviously, the sets of attributes ere which the inclusion dependency is specified—X of Rand Y of S—must have the same number of attributes. In addition, the domains for each pair of corresponding attributes should be compatible. For example, if  $X = \{X_1, X_2, ..., X_n\}$  and  $Y = \{B \in B_1, \dots, M_n\}$  one possible correspondence is to have don( $A_n$ ). Comparise With  $A_n(B)$  for  $1 \le i \le n$ . In this case, we say that  $A_i$  corresponds to  $B_i$ .

for example, we can specify the following inclusion dependencies on the relational scale in Figure 12-1

```
LERAKIVENT (DACESSA < LMPCOVEE, SSA
VOKKSLOV, VVA < CAPLOYEC, SSA
LARLONEE, DAUMSER < DEPARTAENT, CARLABER
RECOLCT, DAUM < DEPARTAENT, DAUMECR
VORS DN (PAUMEER < PROJECT, PAUMEER
COTTLUCATIONS, DAUMEER < DEPARTAENT, DRAMECC
```

All the procedurg melosion dependencies represent referential integrity constraints. We can also use inclusion dependencies to represent class/subclass relationships. For evalue, in the relational schema of Figure 7.5, we can specify the following inclusion dependencies:

```
196 (WELLSON SEPERED LINN)
Allwuf, ASN SEPERED ISSN
Vithen Linda SEPERED ISSN
```

As with other types of dependencies, there are exclusion dependency extension rules (1965) The following are three examples:

1981 (reflexivity): R.X < R.X.</p>

to Q (intribute correspondence). FRX < SY, where  $X = \{A_1, A_2, \ldots, A_n\}$  and  $Y = \{B_1, B_2, \ldots, B_n\}$  and  $A_n$  Correspondence  $B_n$  then  $B_nA_n < SB$ , for  $1 \le i \le n$ . (5.8) (gravenergy). If B | X < SY and  $S \upharpoonright < TC$ , then  $B_nN < TC$ .

The precoding interence roles were shown to be sound and complete for inclusion dependencies. So far, no normal torms have been developed based on inclusion dependencies.

# 11.6 OTHER DEPENDENCIES AND NORMAL FORMS 11.6.1 Template Dependencies

Tour ate dependencies provide a fuch mape for representing constraints in relations that type efficience casy and formal definitions. No matter how many types of dependencies we godg, some predian constraint may come up based on the semantics of attributes within tarions that connect to represented by any of them. The idea behind remplate dependencies an specify a template—or example—that defines each constraint of dependences

There are two types of templates implementing templates and constraint-generating templates. A template consists of a number of **hypothesis tuples** that are meant to show an songle of the tuples that may appear in one or more relations. The other part of the implate s the template conclusion. For tuple generating templates the conclusion is a set

of rights that must also exist in the relations of the hypothesis tuples are there. For constraint generating templates, the template conclusion is a conductor that must hold or the hypothesis tuples.

Figure 11.6 shows how we may define functional, multivalued, and inclusion dependencies by templates. Figure 11.7 shows how we may specify the constraint that "in

| <b> </b> 2) |            | R≓¦A B.C.D                                                  |                     |         |
|-------------|------------|-------------------------------------------------------------|---------------------|---------|
|             |            | a, 7, °-                                                    | X=(A.B)             |         |
|             | nypomesis  | a <sub>1</sub> ° <sub>1</sub> ° <sub>2</sub>                | Y=[0.0]             |         |
|             | onndusion  | $c_1 = c_0 \text{ and } d_1 = d_0$                          |                     |         |
| (7)         |            | H= ( 4 . 8 . C . D .                                        |                     |         |
|             | h          | a <sub>1</sub> b <sub>1</sub> c <sub>1</sub> d <sub>1</sub> | $X = \{A, B\}$      |         |
|             | nypomese   | a1 61 62 d2                                                 | Y= ( C )            |         |
|             | conclusion | a <sub>1</sub> u <sub>1</sub> u <sub>2</sub> d.             |                     |         |
|             |            | <sup>a</sup> 1 <sup>b</sup> 1 <sup>c</sup> 1 <sup>d</sup> 2 |                     |         |
| (c)         |            | R= ( A , B , C , C )                                        | S= E,F,G            | X=/C.CI |
|             | hypothesis | ə, b, °, °,                                                 |                     | Y= E,F} |
|             | conclusion |                                                             | °₁ <sup>d</sup> ₁ g |         |

**FIGURE 11.6** Templates for some common type of dependencies. (a) Template for functional dependency  $X \rightarrow Y$ , the Template for the multivalued dependency  $X \rightarrow Y$  (c) Template for the inclusion dependency RX < SY).

| EMPLOYEE = { NAME |   | SSN , | , SALARY , SUP | FERVISORSSN |  |
|-------------------|---|-------|----------------|-------------|--|
|                   | а | ь     | c              | d           |  |
| hypothesis        | ə | d     | 1              | g           |  |
| conclusion        |   |       | c < 1          |             |  |

**FIGURE 11.7** Templates for the constraint that an employee's salary must be less than the supervisor's salary.

exploree's salary connor be higher than the solary of his or her direct supervisor" on the relation whema territric in Figure 5.5.

# 11.6.2 Domain-Key Normal Form

Three was back and fast rule about defining normal forms only up to 585. Historically, the process of normalization and the process of discovering undestable dependencies was carried through 584, but it has been possible to define structer normal forms that take into account additional types of dependencies and constructs. The idea behind **domain-key normal form** (DENE) is to specify theoretically, at least? the "ultimore normal form" that take into account all possible types of dependencies and constructes. A relative scheme is add to be in DENE if all constructs and dependencies that should hold on the valid idation states can be enforced simply by enforcing the domain constructs and key construction the relation. For a relation in DENE, it becomes very straightforward to enteric all drabase constructs by simply checking that each attribute value in a tuple is of the appropriate domain and that every key constructions is enforced.

However, because of the difficulty of meltiding complex constraints in a LENF relation, to partical attility is limited, since it may be quite difficult to specify general integrity onstraints, but example, consider a relation  $\cos(\cos \theta_{\rm e}, \sin \theta_{\rm e})$  (where  $\sin \theta_{\rm e}$  is the volucle doubteation number) and another relation  $\cos(\cos \theta_{\rm e}, \sin \theta_{\rm e})$  (where  $\sin \theta_{\rm e}$  is the volucle doubteation number) and another relation  $\cos(\sin \theta_{\rm e}, \sin \theta_{\rm e})$  (where  $\sin \sin \theta_{\rm e}$  is the ountry of matodoctine). A general constraint may be of the following form. "If the sole is other Toyota or Lexus, then the first character of the  $\sin \theta_{\rm e}$  is a "?" if the commy of instantiation is Japan; if the store is Houda or Actura, the second character of the  $\sin \theta_{\rm e} = \pi^2 P$ if the country of manufacture is Japan." There is no simplified way to represent such orstants short of writing a procedure (or general assertions) to test them.

# 11.7 SUMMARY

In this thapter, we presented several normalization algorithms. The televanel workess details create 3NF relations from a universal relation schema based on a given set of functional dependencies that has been specified by the database designer. The relational knowposition algorithms create (WNF for 4NF) relations by successive normadilities learnposition of unnormalized relations into two component relations at a time. We first discussed two unportant properties of decompositions, the lossless (nonaddirive) join maprix, and the dependency-preserving property. An algorithm to test for lossless defines station, and a simpler test for checking the losslesses of binary decompositions, was described. We saw that it is possible to synthesize 3NF relation schemas that meet but of the above properties, however, in the case of 0.05% it is possible to ann only for the minimum for one of these two, the romaddirize join condition is an absolute must. affect has to aim for one of these two, the romaddirize join condition is an absolute must.

We then defined additional types of dependencies and some additional normal forms. Multivalued dependencies, which arise from an improper combination of two or more sequendencinglify alged attributes in the same relation, are used to define fourth normal form (4NF). Join dependencies, which indicate a lossless multiway decomposition of a relation, lead to the definition of 6fth normal form (5×1), which is also known as projectjoin normal form (1\*NF). We also discussed inclusion dependencies, which are used a specify referenced integrity and class/subclass constraints, and template dependencies which can be used to specify arbitrary types of constraints. We concluded with a but discussion of the domain key normal form (DKNS).

### Review Questions

- What is mean by the autibute preservation condition on a decomposition?
- 11.2. Why an normal forms alone insufficient as a condition for a good schema design.
- 11.3. What is the dependency preservation property for a decomposition? Why is it important?
- 11.4. Why can we per graphine that PCNF relation schemas will be produced by dependency-preserving decompositions of non-PCNF relation arbitras? Give a counter-sample to illustrate this point.
- 11.5 What is the lossless (or novailditive) join property of a decomposition? Why an important?
- 11.0 Between the properties of dependency preservation and losslessness, which or mast demnitch be satisfied? Why?
- 11.7 Discuss the null value and Jongling tople problems.
- 11.8. What is a multivalued dependency? What type of constraint does it specif. When does it arise?
- 11.9. Illustrate have the process of the time first normalization be done properly statistical dependencies. How should the first normalization be done properly statistical vectors are avoided?
- 11.10. Denne fourth normal form. When is a violated? Why is a useful?
- [611] Denne jour dependencies and tith normal form: Why is 5NE also called project norm normal form (CJNE).
- 11.12. What types of constraints are inclusion dependencies meant to represent?
- 11.1.1 How do reinplate dependencies differ from the other types of dependencies we discussed?
- 11.14. Why is the domain-key notical form (DRNF) known as the ultimate normal total

### Exercises

- 11:15. Show that the relation solutions produced by Algorithm 11.2 are in 384
- 10.16 Show that, if the motion S result og from Algorithm 11.1 does not have a rowthat as all "a" synchols, projecting S on the decomposition, and journing at back sull absays predate at least one sportous tuple.
- 11.17. Show that the relation schemas produced by Algorithm 11.3 are in BCSU
- 11.13. Show that the relation schemes produced by Algorithm 11 4 are in 3NF.
- E.19. Specify a template dependency for join dependencies.
- 11.20. Specify all the inclusion dependencies for the relational schema of Figure 5.5.

- B.21. Prove that a functional dependency satisfies the formal definition of multivalueal dependency.
- 1122 Consider the example of normalicitie the Lors relation in Section 10.4 Electricitie whether the decomposition of rors into [cors1ax, cors1ax, ors10, rors2] has the lossless join property by applying Algorithm 11.1 and also by using the test under Property 13.
- 11.23 Show how the MVOS 1996 ---- appendix and 1996 ----- couple in legare 11.4 a may area during contradication into 184 of a relation, where the attributes page and page are multivalued.
- 11.24 Apply Algorithm 11.4a to the telation in Exercise 10.26 to determine a key for R. Greate a minimal set of dependencies G that is equivalent to E and apply the synthesis algorithm (Algorithm 11.4) to decempose R into 350 relations.
- (1.25) Repeat Exercise 11.24 for the functional dependencies in Exercise 10.27.
- (1.2a) Apply the decomposition algorithm (Algorithm 11.3) to the relation R and the set of dependencies F in Exercise 10 20, Repeat for the dependencies C in Exercise 10.27.
- [127] Apply Algorithm 11.4a to the relations in Exercises 10.29 and 12.30 to determine a key for R. Apply the synthesis algorithm (Algorithm 11.4) to decompose R into 381 relations and the docomposition algorithm (Algorithm 11.3) to docompose R into 2008 relations.
- B.5. Write programs that implement Algorithms 11.3 and 13.4.
- [1] 29. Consider the following decompositions for the relation schema R of Exercise 10.26. Determine whether each decomposition has 10 the dependence preservation property, and (a) the lossless join property with respect to E. Also determine which notical turn each relation in the decomposition is in.
  - a. D<sub>1</sub> = [R<sub>1</sub>, R<sub>2</sub>, R<sub>3</sub>, R<sub>4</sub>, R<sub>4</sub>; R<sub>1</sub> = [A, B, C], R<sub>2</sub> = {A, D, E<sub>0</sub>, R<sub>1</sub> = [B, F], R<sub>2</sub> = [F, G, H], R<sub>3</sub> = {D, C, J]
  - $b_{1}(t) = \{R_{1}, R_{2}, R_{3}, R_{4} = \{A, B, C, D, R_{4}, R_{4} = \{B, C, C\}, R_{4} = \{B, L, B, C, C\}, R_{4} = \{B, C$
  - $\begin{array}{l} & \langle O_1 = \{R_1, R_2, R_3, R_4, R_3\}, R_4 = \{A, B, C, O\}, R_2 = \{D, E\}, R_3 = \{B, F\}, R_4 = \{F, G, (B, R_4 = \{D, I, J\}\} \end{array}$
- if id idensider the relation FLFEC (MOTLER, YLAR, PRICE, MANUE\_PLANE, Locur), which is abbreviated as RELEIS (M. Y. P. MP. C), and the following set F = A functional dependencies  $F = (M + MP, \{N, Y\} \rightarrow P, MP \rightarrow C)$ 
  - Evaluate could of the following as a candidate key for 80300, giving reasons way it can of counor be a keys (M), (M, Y), (M, C).
  - Based on the above key determination, state whether the relation scratc is in 381 and in FCNL giving proper reasons.
  - c. Consider the decomposition of series into D = {R1(M, Y, P), R2(M, MP, C)] is this decomposition lossless? Show why, (You may consult the test under Property (3) in Section 11-1.4.).

# Selected Bibliography

The books re-Moret (1953) and Aronit and De Anropellis (1992) include a comprehensive discussion of relational dependency theory. The documposition algorithm (Algorithm 11-3) is doe to Bergerein (1950). Algorithm 11.4 is based on the normalization algorithm presented in Biskup et al. (1979). Tool and Fischer (1982) give a polynomialtime algorithm for 19 NF decomposition.

The theory of dependency preservation and assless ions is given in Ullman (1988), where proofs of some of the algorithms discussed here appear. The lossless join properties analyted at Abo et al. (1979). Algorithms to determine the keys of a relation from functional dependencies are given in Osborn (1950), re-rise for BONE is discussed in Osborn (1979). Testing for BNE is discussed in discussed in fisca and Fischer (1982). Algorithms for descenting bCNT relations are given in Wang (1990) and Hermandez and Chan (1991).

Multivatued dependencies and fourth cornial form are defined in Zaniolu (1976) and Nicolas (1978). Many of the advanced normal forms are due to Eagin, the fourth normal form in Eagin (1977). Open in Fagin (1979), and (WOF in Fagin (1981). The set of sound and complete rules for functional and multivalued dependencies was given by Buers et al. (1977). Join dependencies are discussed by Rissanen (1977) and Ahao et al. (1979). Interference rules for join dependencies are given by Science (1982), luchistical dependencies are discussed by Casarova et al. (1981) and analyzed further in Cosmadakis et al. (1992). Their use in optimizing relational schemas is discussed in Casarova et al. (1989). Template dependencies are discussed by Sadir and Ullinaa (1982). Other dependencies are discussed in Nicolas (1978), Furrado (1978), and Mendelson and Maior (1979). Abireboid et al. (1995) provides a theoretical relation of many of the ideas presented in this chapter and Chapter 10.





# Practical Database Design Methodology and Use of UML Diagrams

In this chapter, we make from the theory to the peacific of database design. We have alreaded corribed in several chapters material, that is relevant to the design of actual datafases for practical real-world applications. This material includes Chapters 5 and 4 on dislose correspond modeling: Chapters 5 through 9 car the relational model, the Soft lagrage, relational algebra and calculus, mapping a high-level conceptual ER or fast should introduce the schema and programming in relational systems (E199486); and Uppters 10 and 11 on data dependency theory and relational normalization algorithms.

The overall database design activity has to undergo a systematic process called the design methodology, whether the target database is managed by an ROWOS, object database management systems (ODROS), or object relational database management systems (ODROS). Various design methodologies are implicit in the database design around of a Posigner 2000 by Oracle, ERWin, ROWO, and Paradigny allas by Pariroum Technologies; and System Architect by Poplan Solars ER Studies by Embarcadero Technologies; and System Architect by Poplan activation of the database design area breader context, as it is undertaken in large againstons for the design and implementation of applications catering to hundreds or database design and implementation of applications catering to hundreds or database design and implementation of applications catering to hundreds or database design and implementations of applications catering to hundreds or database.

Contending the dusing of small databases with perhaps up to 20 users need not be very weight ated. But for incident sped or large databases that server several deverse application groups, each with tens or bundbeds of asers, a systematic approach to the

everall database design activity becomes necessary. The sheer site of a populated database dees not relieut the complexity of the design, it is the schema that is more important. An database with a schema that includes more than 30 or 40 entity types and a similar number of relationship types requires a careful design methodology.

Using the term large database for databases with several tens of gigabytes of data and a schema with more than 3C or 40 distinct entity types, we can cover a wide array of databases in geveniment, industry and financial and communical institutions. Service sector industries, including bailong, hotels, airlines, insurance, utdrites, and communications, use databases for their day-to-day operations 24 hours a day, 7 days a weeks. Anown in industry as 24 by 7 operations. Application systems for these databases are called inclusation processing systems due to the large transaction volumes and rates that are required. In this chapter we will be concentrating on the database design for such mechanys and large scale databases where transaction processing dominates.

This chapter has a variety of objectives. Section 12.1 discusses the information system his cycle within organizations with a particular emphasis on the database system. Section 12.2 high ights the phases of a database design methodology in the organizational context. Section 12.3 introduces UML diagrams and gives details on the norations of some of them that are particularly helpful in collecting requirements, and performing computed and logical design of databases. An illustrative partial example of designing a university database is presented. Section 32.4 introduces the pupplar software development tool called Rationa Rose which has UML diagrams as its main specification technique. Fratures or Rational Rose that are specific tool database requirements, modeling and schema design are highlighted. Section 17.5 mielly discusses automated database design cools.

# 12.1 THE ROLE OF INFORMATION SYSTEMS IN ORGANIZATIONS

# 12.1.1 The Organizational Context for Using Database Systems

Database systems have become a part of the information systems of many organizations ha the 1960s information systems were dominated by file systems, but since the early 1970s organizations have gradually moved to database systems. To accommodate such sytems, namy organizations have created the position of database administrator (984) or even database administration departments to oversee and control database life-cyck a revities. Similarly, information technology (77), and information resource magnetized (1884) have been recognized by large organizations to be a key to successful manacurant of the business. There are several masons for the:

- Data is regarded as a corporate resource, and its management and control is considered, entral to the effective working of the organization.
- More functions in organizations are computerized, increasing the need to keep large volumes of data available in an upsto-the-annuate current state.

- As the complexity of the data and applications provide complex relationships among the data need to be modeled and maintained.
- There is a rendericy roward consolidation of information resources in many organizations.
- Many organizations are reducing their personnel costs by leading the end user perform business transactions. This is evident in the form of travel services, financial services, online retail goads outlet and custometro-business electronic commerce examples such as anaton comfort. Ebuy, In these instances, a publicly accessible and updatable operational database must be designed and made available for these monscripts.

Exitabilities systems satisfy the preceding requirements in large measure. Two additional characteristics of durables systems are also very valuable in this environment.

- Data addrendence protects application programs from charges in the underlying logical organization and in the physical access paths and storage structures.
- Eucoid referes (views) allow the same data to be used for moltiple opplications, with each application having its own view of the data.

New capabilities provided by database systems and the following key features that the offer have made them integral computerits in computer-based information systems:

- Enegration of data across multiple applications into a single database
- Simplicity of developing new applications using high-level longuages like sQL
- Fossibility of supporting casual access for browsite and querying by managers while supporting major production-level transaction processing.

From the early 1970s through the rand 1980s, the move was toward creating large tentral red reportences of data managed by a single centralited cPMS. Over the last 12 to 15 years, this trend has been reversed because of the following developments

- 1. Personal computers and database system-like software products, such as EXCEL EX280, ACCESS (all of Microsoft) for SQL Anowhere (of Sybase), and public domain products such as MYSQL are being hereally attitued by users who previously belonged to the category of easail and occasional database users. Many administrators, secretarics, engineers, scientifies, and hereally attitue bloc belong to this category. As a result, the practice of creating personal databases is gaining popularity. It is now possible to check our a copy of part of a large database from a majnificance computer or a database secret, work on it from a personal workstation, and ther, system the mainframe. Similarly, users can design and create their copy databases and then merge them micro larger one.
- 2. The advent of distributed and chent-server DWEss (see Chapter 39) is opening up the extrem of distributing the database over multiple computer systems for herei local control and fister local processing. At the same time, local users can access remote data using the facilities provided by the PDWEs as a cherr, or through the Web. Application development tools such as PowerBudder at Developer 2000 (by Oracle) are being used beauty with built-archaelities to link applications to multiple back-end database servers.

- 5. Many organizations now use data dictionary systems in information reposituries, which are mini (1963) that manage metadata—that is, data that describes the database structure, constraints, applications, authorizations, and so on. These are often used as an integral roal for information resource indiogenerat. A ostful data dictionary system should store and manage the following types of information:
  - a. Descriptions of the schemas of the database system.
  - Detailed information on physical database design, such as storage structures, access paths, and bloand record sites.
  - Descriptions of the database users, their responsibilities, and then access rights.
  - High-level descriptions of the database transactions and opplications and of the relationships of users to transactions.
  - The relationship between slatabase transactions and the data items referenced by them. This is as ful in determining which transactions are infected when cortain data definitions are changed.
  - Usage sortistics such as frequencies of queries and transactions and access counts to difference portions of the database.

This metadata is available to DDAs, designers, and authorized discislas colline system documentation. This improves the control of DDAs over the information system and the asset? indepartuling and use of the system. The advent of data warehousing recording has highlighted the importance of metadata.

When designing high-performance transaction processing systems, which require around-the-clack nonstrop operation, performance becomes critical. These databases are oben accessed by handreds of transactions per primare from remote and local terminals. Transaction performance, in terms of the average countret of transactions per minute and the average and maximum transaction response time is critical. A coreful physical database design that meets the organization's transaction processing needs is a must in such systems.

Some organizations have continuted their information resource management to certain DROS and data dictionary products. Their investment in the design and implementation of large and complex systems makes it difficult for them to change to newer DROS spinlings which means that the expandations become locked in to their current DROS system. With regard to such large and complex databases, we cannot overemphasize the importance of a careful design that takes into account the need for possible system modifications. Called tuning i to respond to changing requirements. We will decuse runing in conjunction with spory optimizations in Chapter 16. The cost can be very high it a large and complex system carefor evolve, and it becomes necessary to move to other DROS yp shorts.

## 12.1.2 The Information System Life Cycle

In a large organization, the database system is typically part of the information system, which includes all resources that are involved in the collection, numagement, use, and descrimination of the information resources of the organization. In a computation exerronment, these resources include the data itself, the D805 software, the computer system hardware and storage media, the personnel who use and manage the data (D80, end overs, parametric users, and so eral, the applications software that accesses and opdates the data, ind the application programmer: who develop these applications. Thus the database systum is pair of a much larger organizational information system.

In this section we examine the typical life cycle of an information system and how the database system fits into this life cycle. The information system life cycle is often called the **macro life cycle**, whereas the database system life cycle is reterred to as the micro life cycle. The distinction between these two is becoming bary for information setens where databases are a major integral component. The macro life cycle republy acludes the following phases:

- Foundation analyses: This phase is concerned with analysing potential application areas, identifying the economics of information gathering and dissemination, pertorning gathminary cost-benefit studies, determining the complexity of data and processes, and setting up procedies acaeve applications.
- 2 Kanazeneous collection and analysis. Detailed requirements are collectes? by interacting with potential users and user groups to identify their particular problems and needs. Interapplication dependencies, communication, and reporting procedures are identified.
- Design: This phase has two aspects: the design of the database system, and the design of the application systems (programs) that use and process the database
- 4 Implementation. The information system is implemented, the database is leaded, and the database transactions are implemented and tested.
- 5 Validation and acception cuttures: The acceptibility of the system in incering users' inquiringents and performance uniteria is validated. The system is tested against performance criteria and la basico specifications.
- 6 Deployment, operation and transcentings. This may be preceded by conversion of users from its older system as well as by user training. The operational phase starts when all system functions are operational and have been calibrated. As new requirements or applications crop up, they gass through all the previous phases and they are validated and accomposited into the system. Monitoring of system performance and system maintenance are important activities during the operational phase.

# 12.1.3 The Database Application System Life Cycle

Amories related to the database application system (micro) life cycle include the followings

- System definition: The scope of the database system, its users, and its applications are defined. The interfaces for various categories of users, the response time constraints, and storage and processing needs are identified.
- Database design: At the end of this phase, a complete logical and physical design of the database system on the chosen DBSS is ready.

- Database implementation: This computes the process of specifying the conceptual external, and internal database definitions, creating entry database files, and implementing the software applications.
- Londons or data conversion: The database is populated either by loading the data directly or by conversing existing files into the database system format.
- Application correction: Any software applications from a previous system are converted to the new system.
- Testing and calidation. The new system is tested and validated.
- Operation: The database system and its applications are put intri operation. Usually, the old and the new systems are operated in parallel for some time.
- Monoroung and maintagence: During the optranional phase, the system is constantly monored and maintained. Growth and expansion can securify both data conteat and software applications. Major modifications and reorganizations may be needed from time to time.

Activities 2.3, and 4 together ore part of the design and implementation phases of the larger minimation system life cycle. Our emphasis in Section 12.2 is on activities 2 and 3, which cover the database design and implementation phases. Most databases in organizations indergn all of the preceding life-cycle activities. The conversion activities (4 and 5) are not applicable when both the database and the applications are new. When an organization indexes from an established system to a new one, activities 4 and 5 tend to be the most time-consuming and the offort to accumplish them is often underestimated. In general, there is often feedback among the various steps because new requirements frequently, arise an every stage. Figure 12.1 shows the feedback loop, affecting the conceptual and logical design phases as a result of system implementation and tomag

# 12.2 THE DATABASE DESIGN AND IMPLEMENTATION PROCESS

We now no us on activities 2 and 3 of the database application system lite cycle, which are database design and intelementation. The problem of database design can be stated as follows:

The goals of database design the multiple-

- Satisfy the informations content requirements of the specified users and applications
- Provide a natural and easy-to-understand structuring of the information
- Support processing requirements and any performance objectives, such as response time, processing time, and storage space.

These goals are very hard to accomplish and measure and they involve an inherent traceofficit one attempts to achieve more "naturalness" and "understandability" of the model, it may be at the cost of performance. The problem is approvated because the database design process often begins with informal and pourly defined requirements. In out is, the result of the design activity is a rigidly defined database schema that cannot cable be model by addited once the database is unplemented. We can identify some near phases of the overall database design and implementation process.

- Requirements collection and analysis.
- Conceptual database design.
- 3. Choice of a DIMS.
- Data model mapping (also called logical database design).
- Physical database design
- 6. Database system implementation and runing.

The design process consists of two parallel activities as off istrated in freure 12.1. The first action prior design of the data content and structure of the database, the scould relates to the design of database applications. To keep the figure simple, we have avoided showing most of the interactions among these two sides, but the two activities are closely intertunized. For example, by analyting database applications, we can identify due tends that will be stored in the database. In addition, the physical database design place, during which we choose the storage structures and access paths in database files, depends on the applications that will use these files. On the other hand, we usually speak the design of database applications by referring to the database schema constructs, which are specified during the first activity. Clearly, these two introduces strongly influence one another. Traditionally, database design methodologies have prinority for and on the use of these activities software design for strongly being recognized by database database activities whereas software design. It is rapidly being recognized by database design and software engineers that the two activities should proceed hand in hand, and design ands are increasingly comburing them.

Thes x phases mentioned previously do not have to proceed strictly in sequence, in non-cases we may have to makify the design from an earlier phase during a later phase. These feedback loops among phases cand also writtin phases—are common. We show only couple of feedback loops in Figure 12.1, but many more exist betweent various pors of phases. We have also shown some interaction between the data and the process sides of the figure; many more unteractions exist in realistic. Phase 1 in Figure 12.1, involves collecting information also in the interaction between the database. The figure 12.1 involves collecting information also in the interaction for the database, and Fluise 6 concerns dualises implementation and redesign. The heart of the database design process comprises Phases 2, 4, and 5, we briefly summarize these phases.

 Conceptual dealbase design (Phose 2). The goal of this phase is to produce a conceptual schema for the database that is independent of a specific DBMS. We often use a lacklevel data model such as the ER of EER model (see Chapters 3 and 4) during the phase. In addition, we specify as many of the known database applications is transacture as possible, using a notation that is independent of any specific DBMS. Often,



FIGURE 12.1 Phases of database design and implementation for large databases.

the 1984s choice is already made for the organization the interval conceptual deseris still to keep it as free as possible from implementation considerations.

Data model mapping (Phase 4) During this phase, which is also called logical database design, we map (or transform) the conceptual schemo from the high-level durinoidel used in Phase 2 into the data model of the chosen (1988). We can start the phase after choosing a specific type of (1986) – for example, if we decide to use some relational (1986) but have not yet decided on which particular one. We call the later system-independent (but data model-dependent) logical design. In terms of the fluce-

level DBMs architecture discussed in Chapter 2, the result of this phase is a conceptiol scheme in the chosen data model. In addition, the design of external schemes (views) for specific applications is esten done during this phase.

- Physical database design (Plass 5). During this phase, we design the specific tions for the stored database in terms of physical storage structures, record placement, and indexes. This corresponds to designing the internal scheme in the terminology of the three-level UBMs architecture.
- Database system implementation and kandt, (Piewe 67) During this phase, the database and application programs are implemented, tested, and eventually deployed to service. Vatious transactions and applications are tested individually and their in conjunction with each other. This typically reveals opportunities for physical design changes, data undexing, reorgamissium, and different placement of data—an activity referred to as database turning. Turing is an ongoing activity—a part of system mani-terance that comments for the life cycle of a database is long as the database and applications are problems are directed.

In the following subsections we discuss each of the six phases of database design in note details

# 12.2.1 Phase 1: Requirements Collection and Analysis<sup>1</sup>

Brow we can effectively design a database, we must know and analyte the expectations of the users and the intended uses of the database in as much detail as possible. This protests called **requirements collection and analysis**. To specify the requirements, we must not dearify the other parts of the information system that will interact with the database specify these include new and existing users and applications, whose requirements are then collected and analyzed. Typically, the following activities are part of this phase:

- The major application areas and user groups that will use the database or whose work will be affected by it are identified. Kno individuals and committees within each group are chosen to party out subarquent steps of requirements collection and specification.
- baising documentation concerning the applications is studied and analyzed. Other documentation—policy manuals, forms, reports, and organization charts— is reviewed to determine whether it has any influence on the requirements collec-uon and specification process.
- 5. The current operating environment and planned use of the information is studied. This includes analysis of the types of transactions and their frequencies as well as of the flow of extornation within the system. Geographic characteristics regarding users, origin of transactions, destination of reports, and so forth, are stedied. The input and output data for the transactions are specified.

A part of this action has been centrel intellety Colory Ports.

4. Written responses to sets of questions are sometimes collected from the potential database users of user groups. These questions meebled the users' priorities and the importance they place on various applications. Key individuals may be interviewed to help in assessing the worth of information and in setting up priorities.

Requirement analysis is carried out for the final isers, or "costonicis," of the database system by a team of analysis or requirement exparts. The initial requirements are likely to be informal, incomplete, incomestent, and partially incorrect. Much work therefore needs to be done to transform these only requirements into a specification or the application that can be used by developers and testers as the starting point for writing the implementation and test cases. Because the requirements reflect the initial understanding of a system that does not verseast, they will incortably charge. It is therefore important to use techniques that help customers converge quickly on the implementation requirements.

There is a lot of evidence that customer participation in the development process increases customer satisfaction with the delivered system. For this reason, many practitioners new me neutropy and workshops involving all stakeholdets. One such methodology of returning initial system requirements is called Joint Application Design (140). More memory to infining initial system requirements is called Joint Application Design (140). More memory to enough anomersed in the workplace in which the application is to be used. To help ensumer representatives herer upderstand the proposed system, is is common to walk through workflow or transaction scenarios or to create a more upprotype of the application.

The preceding modes help structure and refine requirements but have them still in an informal state. To transform requirements into a better structured form, requirements specification techniques are used. These include OCA (object-sciented analysis), 348 (data flow diagrams), and the refinement of application goals. These methods use diagramming techniques for organising and presenting information-processing requirements. Additional documentation in the form of cext, tables, charts, and decision requirements usually accompanies the diagrams. There are techniques that produce a formal specification that can be checked mathematically for consistency and "what if with the analyses. These methods are builty used now but may become arandard in the former for those parts of information systems that serve mosion-critical functions and which therefore must work as planned. The model based formal specification methods of which the 2-notation and methodology is the most prominent, can be thought of a extensions of the 16 model and are therefore the most applicable to information system design.

Some computer-orded techniques—called "Upper 0.456" tools—how been proposal to help check the consistency and completeness of specifications, which are usually stored in a single reprisitory and can be displayed and updated as the design progresses. Other tools are used to trace the links between requirements and other design entities, such as code mudules and rest cases. Such travaibley databases are especially important in conjunction with enforced change-management procedures for systems where the requirements change frequently. They are also used in contractual projects where the development organization must provide documentory evidence to the customer that althe requirements have been implemented. The recomments collection and analysis phase can be quite time-constrainty, but in scrucial to the success of the information system. Correcting a requirements error is nach note expensive than correcting an error mode during implementation, because the effects of a requirements error are usually perensive, and much more desensation work has to be resimplemented as a result. Not correcting the error means that the system will not outly the customer and may not even be used at all. Requirements gathering analysis have been the subject of entire books.

# 12.2.2 Phase 2: Conceptual Database Design

The excited phase of database design involves two parallel activities.<sup>1</sup> The first activity, conceptual schema design, examines the data requirements to a ring from Plase 1 and produces a conceptual database schema. The second activity, transaction and application design, examines the database activities analyted in Plase 1 and produces high-level specifications.

Phase 2at Conceptual Schema Design. The conceptual schema produced by the phase is usually contraned in a DBM-sindependent high-level data model for the following reasonse

- 7 The goal of conceptual schema design (a) complete understanding of the database structure, meaning (senior rocs), interrelationships, and constraints. This is lost achieved independently of a specific DIMS because each DIMS tepically has idiosyncrasies and restrictions that should not be allowed to influence the conceptual schema design.
- The conceptual schema is involuable as a stable descuption of the database contents. The chance of PPMs and later design decisions may change without changing the P908-mdependent conceptual schema.
- 3 A good understanding of the conceptual schema is circuial for database users and application designers, lose of a high-level data model that is more expressive and general than the data models of individual DBMSs is hence quite important.
- 4. The diagrammatic description of the conceptual schema can serve as an excellent vehicle of communication among duralose users, disigners, and study sta Because high-level data models usually rely on concepts that are casier to understand than lower-level (160% specific data multils, or syntactic definitions of data, any communication concerning the schema dosign becomes more exact and more straightforward.

In this phase of database design, it is important to use a conceptual high-level data rodel with the following characteristics:

) The chase of design is discussed in great detail in the first sever chapters of Paterick (1992), we uncome that encoded betw

- Expressiveness. The data model should be expressive enough to distinguish different types of data, relationships, and constraints.
- Singueas and independentiality. The model should be simple enough for typical nonspecialist users to understand and use its concepts.
- Manurenty: The model should have a small number of basic concepts that are distinet and non-verlapping in meaning.
- Diagnormala representation. The model should have a diagrammatic notation for displaying a conceptual scheme that is easy to interpret.
- Formatity: A conceptual schema expressed in the data model must represent a tarmate anambiguous specification of the data. Hence, the model concepts must be defined accurately and anambiguously.

Many of these requirements - the first one in particular- sometimes conflict web other requirements. Many high-level conceptual models have been proposed for database design (see the selected hibbliography for Chapter 4). In the following discussion, we will use the terminology of the Enhanced Entrice Relationship (FER) model presented in Chapter 4, and we will assume that it is bring used in this phase. Conceptual schema design including data modeling, is been thing an integral part of object conceptual schema and design insthiddologies. The UNIT has class diagrams that are largely based on extensions of the FER model.

Approaches to Conceptual Scheme Design. For conceptual schema design, we use identify the basic components of the schemic the entry types, relationship types, and attributes. We should also specify key attributes, conducative and participation constraints on relationships, weak entry types, and specialization/generalization hierarchies/lattice. There are two approaches to designing the conceptual schema, which is derived from the requirements culler red during. Phase 1.

The tirst approach is the centralized (or one-shot) achema design approach in which the requirements of the different applications and user groups from Phase I are narroad into a single ser of negoticities before schema design begins. A single schema corresponding to the negred set of negoticities is then designed. When nearly users and applications exist, merging all the requirements can be an ardiaxis and time-consuming task. The assumption is that a centralized authority, the 10(A) is responsible for deciding how to merge the requirements and for designing the conceptual schema for the while database. Once the conceptual scheme is designed and finalized, external schemes for the various user groups and apple ations can be specified by the D(A).

The second approach is the view integration approach, in which the requirements are normerged. Rathet a schema (or view) is designed for each user group or application based only on its own requirements. Thus we develop one high-level schema (view) ac each such user group or application. During a subsequent view integration phase, these schemas are merged or integrated into a global conceptual schema for the ennie database. The individual views can be reconstructed as esternal schemas after view integration. The main difference between the two approaches hes in the manner and stage in which multiple views or requirements of the many users and applications are reconciles? and merged. In the controlled approach, the reconciliation is done manually by the DBA's sufficient to designing any schemas and is applied directly to the requirements collected in these 1. This places the borden to reconcilie the differences and conflicts among user groups on the DBA's staff. The problem has been repically don't with by using external calculation/design experts to bring in their own ways of resolving these conflicts. Because is the difficulties of managing this task, the view integration approach is now guiting to exceptible.

In the view integration approach, each user group or opplication actually designs in own conception. (HK) schema from its requirements. Then an integrated schema applied to these schemas (views) by the OBA to form the global integrated schema Although view integration can be done manually, its application to a large catabase involving tens of user groups requires a methodology and the use of automated tools to help in carrying out the integration. The correspondences among the attributes, entity types, and relationship types in various views must be specified before the integration can be applied to an addition, problems such as integrating confidences among the attributes, entity is applied to a distribute problems such as integrating confidences and sendering the consistency of the specified orients from correspondences must be dealt with.

Strategies for Schema Design. Oreen a set of requirements, whether for a single user or for a large user community, we must create a conceptual schema that satisfies these requirements. There are various strategies for designing such a schema. Must strategies following materimental approach — that is, they start with some schema constructs derived fourthe requirements and then they incrementally modify, refine, or build on them. We now losues some of these strategies:

- Topolosis strategy. We start with a schema containing high-level obstractions and then apply successive row-down reinterments. For example, we may specify only a tew high-level entity types and then, as we specify their autointies, split them into lower-level entity types and relationships. The process of specialitation to refine an entity type into subclasses that we illustrated in Sections 3.2 and 4.3 (see Figures 4.1, 4.4, and 4.5) is another example of a top-down design strategy.
- 2. Borton-up strategy. Start with a schema containing basic obstractions and then combine or add to these abstractions. For example, we may start with the attaibutes and group these into entity types and relationships. We may add new relationships among entity types as the design progresses. The process of general-ting entity types into higher-level generalized superclasses (see Sections 4.2 and 4.3, Figure 4.3) is another example of a bottom-up design strategy.
- 3 fusile out strategy: This is a special case of a bettom op-strategy, where attention is focused on a central set of concepts that are most evident. Modeling then special sacurand by considering new concepts in the vicinity of existing ones. We could specify a trivial cardiac centraty types in the schema and continue by adding other entity types and relationships that are related to car b.

4. Mored stronger: Instead of following any particular strongy throughout the design, the requirements are partitioned according to a top-down strategy, and part of the schema is designed for each partition according to a bottom-up strategy. The varous schema parts are then combined.

Figures 12.2 and 12.3 illustrate rop-down and bottom-up refinement, respectively. An example of a rop down refinement primary as decomposition of an entity type inteseveral entity types. Figure 12.2(a) shows a constraints and accurs and arraws and the vacuus relationship is correspondingly split into matters. Figure 12.2(b) shows a constraints are the maximum types (access and ration use) and a relationship between them. Refinement typically forces a designer to ask increasing ending the completence of example, the (min-must cordinality ratios between constraints and details) for example, the (min-must cordinality ratios between constraints are obtained during refinement Figure 12.3(a) shows the bottom-up terment of primitive of generating new relationships around 12.3(a) shows the bottom-up terment of primitive of generating new relationships around the relationship solution of the terminative of generating new relationships around the relationship service of the termination of the relationships around the relationship between th



**FIGURE 12.2** Examples of top-flown refinement, (a) Generating a new entity type, (b) Occomposing an entity type into two ontity types and a relationship type.



**AGURE 12.3** Examples of bottom up refinement: (a) Discovering and adding new relation slips, the Discovering a new category tonion type) and relating it.

anny types. The botton up refinement using categorization (union type) is illustrated in Egen 12.3(b), where the new concept of discuss only is closer even the construction of electric control discussion of electric scattering a category and the related dynamic notation follows when we introduced in Section 4.4.

**Scheme (View) Integration.** For large databases with many expected users and generators, the view integration systemeth of designing individual schemes and then means them can be used. Because the individual views can be kept relatively small, dogo of the schemas is simplified. However, a methodology for integrating the viewinto a global database schema is needed. Schema integration can be divided into the klowing subtasks:

- Identifying correspondences and conflues aroing the schemas: Because the schemas are designed individually, it is necessary to specify constructs in the schemas that represent the same real-world concept. These correspondences must be identified before integration can proceed. During this process, several types of conflues among the schemas may be discovered:
  - a Notiong couplies. These are of two repest serionynis and inonionyms. A synonymic occurs when two schemes use different names to describe the same concept; for example, as entity type (0.500-8 in one scheme) may describe the same concept as an entity type (0.500-8 in another schema. A homenym occur when two schemas are the same frome to describe different concepts, for example, an entity type (0.500 computer parts in one schema and furniture parts in another schema and furniture parts in another schema.
  - b. Type confidence The same concept may be represented in two schemes by different modeling constructs. For example, the concept of a prover scheme we be mertially type in one scheme and on attribute in another.
  - c. Domain (eadle set) conflues. An attribute may have different domains in two schemas. For example, issuing the declared as an integer in one schema and as a characteristing in the other. A conflict of the unit of measure could occur d one schema regresented where in pronds and the other used kilograms.
  - 3. Conflicts above constraints: Two schemes may impose different constraints, for example, the key of an entity type may be different in each scheme. Another example, involves different structural constraints on a relationship such as theorem; one schema into represent it as LAN to course has one instructed, while the other schema represents it as M/N to course may have more that one instructor).
- Mod/ying occus to conform to one another. Some schemas are moduled so that they conform to other schemas more closely. Some of the conflicts identified in the first sobrask are resolved during this step.
- 3 Merging of views. The global schema is created by incruing the individual schemas. Corresponding concepts are represented only once in the global schema, and mappings between the clews and the global schema are specified. This is the most difficult step to achieve in real-life databases involving hundreds of contract and relationships. It involves a considerable and out of human intervention and negtration to resolve coefficies ond to settle on the most tensorable and acceptable solutions for a global schema.
- Restructuring: As a final optional step, the global schema may be analyzed and restructured in remove any redundancies or unnecessary complexity.

Some of those ideas are illustrated by the rather simple example presented in Equita- 12.4 and 12.5. In Figure 12.4, two views are immediate meater a hibblingrapher databas. During identification of correspondences between the two views, we discover that resonance and subseries synchronic database is concerned), as are controlling shown in Figure 12.4, to contorn to view 2. Figure 12.5 shows the result of merging wrother view 1 with view 2. We generalize the entry types source and subsection



FIGURE 12.4 Modifying views to conform before integration.



FIGURE 12.5 Integrated schema after merging views 1 and 2.

the entry type resultation, with their common attribute fittle. The relationships contributed around actives ar are increased, as are the entry types sessarches and acteus. The attribute Publicisher applies only to the entry type edds, whereas the attribute Size and the relationship type Fusicises. Is apply only to articles

The above example illustrates the complexity of the merging process and how the meaning of the various concepts must be accounted for in simplifying the resultant schema design. For real-life designs, the process of schema integration requires a more disciplined and systematic approach. Several strategies have been proposed for the view integration process (Figure 12.6).

- 1 Binary lodder integration. Two schemas that are quite similar are integrated first. The resulting schema is then integrated with another schema, and the process is repeated unrul all schemas are integrated. The ordering of schemas for integration can be based on some measure of schema similarity. This strategy is suitable for manual integration because of its step-by-step approach.
- 2. Nervy megration: All the views are integrated in one procedure after an analyse and specification of their correspondences. This strategy requires computerized roots for large design problems. Such tools have been built as research prototypes but are not yet commercially available.
- 3 Heavy balanced strategy: Parts of schemas are integrated first: then the resulting schemas are parted for further integration; the procedure is repeated initial a final global schema results.
- 4 Mixed valuege: humally, the schemas are participated into groups based on their similating, and each group is integrated separately. The intermediate schemas are grouped again and integrated, and so on.

Phase 25: Transaction Design. The purpose of Phase 26, which proceeds in parallel with Phase 2a, is to design the characteristics of kninwn database transmitions (applications) in a DR05-independent way. When a database system is being designed.



**FIGURE 12.6** Different strategies for the view integration process

the designers are incure of many known applications (or transactions) that will run on the cambase once in is implemented. An important part of database design is to specify the informational characteristics of these transactions early on in the design process. This ensures that the database schema will include all the information required by these innoactions. In addition, knowing the relative importance of the various transactions and the expected rates of their invocation plays a chiefd part in physical database design (Pase 5). Usually, only some of the database transactions are contrational database design (Pase 5). Usually, only some of the database transactions are contrational design time; after the database system is implemented, new transactions are often known in advance of specified at the worklead is represented by 20 present of the most frequently used transactions, which govern the design in applications that are of the act-box querying of batch processing variety, queries and applications that are of the act-box querying of batch processing variety, queries and applications that are of the act-box querying of batch processing variety, queries and applications that process a substantial amount of data must be identified.

A common technique for specifying transactions at a conceptual level is to identify their input/output and functional behavior. By specifying the input and output parameters (arguments), and meertal functional flew of control, desceners can specify a transaction in a conceptual and system-independent way. Transactions usually can be grouped into three coregones: (1) retrieval transactions, which are used to retrieve data for display on a species or for production of a report; (2) update transactions, which are used to enter new data or cormedity existing data in the database; (3) mixed transactions, which are used for more complex applications that do some correctal and some apdate. For example, consider an arrhog reservations database. A retrievel transaction could be to book a sear on oparticular flight. A mixed transaction may first display since data backs such as showing a constituent reservation on some flight, and then opdate the database, such as concelling the reservation by deleting it, or by adding a flight sciencent to an existing inservation. Transactions (applications) may originate in a fight science to a soft as ProverBuilder 9.0 (from Sybase) or Developer 2000 (from Oracle), which collect parameters on line and then send a transaction to the Disys as a backend <sup>3</sup>

Several techniques for requirements specification include notation for specifying processes, which in this context are more complex operations that can remist of several transactions. Process modeling roals like BPWin as well as workflow modeling roals are becoming popular to identify information flows in organizations. The UVL language which provides for data modeling via class and object diagrams, has a variety of process modeling transition diagrams, activity, diagrams, equiner diagrams, and collaboration diagrams. All of these refer to activity, diagrams, and operations within the information system, the inputs and outputs of the processes, are the sequencing or synchronization requirements, and other conditions. It is possible to refine these specifications and exercit individual transactions from them. Other proposals for specifying transactions include TAMS, tovatiect, and 0.080455 (see the selected bibliography at the end of this chapter). Semiclot these have been implemented interproteipe systems and cools. Process modeling still remains an arrive an of research.

Transaction design is just as important as schema design. For it is often considered ta be part of software engineering rather than database design. Many current design methodologies emphasize one over the other. One should go through Phases 2a and 2b in parallel, using feedback loops for tempement, until a stable design of scheme and transactions is reached.<sup>4</sup>

# 12.2.3 Phase 3: Choice of a DBMS

The choice of a 106Ms is governed by a number of factors—some technical, others connamic, and still others concerned with the politics of the organization. The rectinited factors are concerned with the suitability of the 06Ms for the rack at hand, issues to consider

This plub sophy has been followed to cover 20 years args to be products like the which serves as a tock to neutrate transactions for leady approxible field.

High-level transaction modeling is covered in Paora et al. (1992) chaps. 9-9, and 111. The tone functional and data aways philosophy is advocated throughout that book.

here are the type of OBMS frelational, object relational, object, other1, the storage structures and access paths that the D908 supports, the user and programmer interfaces available, the types of high-level query languages, the availability of development tools, ability to interface with other DBMSs via standard interfaces, architectural options related to chent-server operation, and solon. Nontechnical factors include the financial states and the support organization of the vendor. In this section we concentrate on discussing the ecoronic and organizational factors that affect the choice of DBMs. The following costs must be considered:

- 1. Software degabation cont: This is the "up-freed" cost of buying the software including language options, different interface options such as forms, mergi, and Webbased graphic user interface (GCV) tools, recovery@vallarp options, special accesmethods, and documentation. The correct (1686) version for a specific operating system must be selected. Typically, the development trafts, design tools, and additional language support are not included in basic priering.
- Maximum exist. This is the recurring cost of receiving standard monitonance service from the vendor and for kcoping the 1980s version up to date.
- Hawkwas acquisition core: New hardware may be needed, such as additional area ony remainals, disk drives and controllers, or specialized pixel storage and archivalistorage.
- 4 Octabase octation and concersion cost: This is the cost of either creating the database system from scratch or converting an existing system to the new OBMS software. In the latter case or is costomary to operate the existing system in parallel with the new system ontil all the new applications are fully implemented and rested. This cost is hard to project and is often onderestimated.
- Personnel cost: Acquisition of 10808 software for the first time by an organization is often accompanied by a reorganization of the data-piocessing department. Positions of 1084, and staff exast in most comparises that have adopted 108485.
- 6 Training cost: Because DBMSs are often complex systems, personnel must often be trained to use and program the DBMS. Framma is required at all levels, including programming, application development, and database administration.
- Operating cost: The cost of continued operation of the database system is (queuily nor worked into an evaluation of alternatives because it is incurred regardless of the 1988 selected.

The benefits of acquiring a office are not solvery to increase and quartity. A OR48 has everal intengible advantages over traditional tile systems, such as case of use, consolidation of company wide information, wider availability of data, and faster access transformation. With Web based access, certain parts of the data car, be made globally accessible to employees as well as external users. More tangible benefits include reduced agriculture development cost, reduced reducdancy of data, and better control and security. Although databases have been firmly entrenched in most organizations, the decision of whether to move an application from a file-based to a database-centered apposch comes up frequently. This move is generally driven by the following factors.

- Data complexity. As data relationships become more complex, the need for a DBME is felt more strongly.
- Sharing arrang applications: The greater the sharing among applications, the maxe the redundancy among bles, and hence the greater the need for a DPMs.
- Dynamically evolving or growing data. If the data changes constantly, it is easier to cope with these changes using a DBMS than using a file system.
- Frequency of addisor requests for data. File systems are not at all suitable for addisor removal of data.
- Data column and need for control. The sheer volume of data and the need to control it sometimes demands a basis.

It is difficult to develop a generic set of guidelines for adopting a single approach to data management within an organization s-whether relational, object-oriented, or object-relational lift the data to be stored in the database has a high level of complexity and deals with multiple data types, the typical approach may be to consider an object relational. DBMF<sup>3</sup> Also, the benefits of unbertrance among classes and the corresponding advantage of reuse favor whese approaches. Finally, several economic and organizational factors after the choice of one DFMS over another.

- 1 Orgentperior-ticke adoption of a contain pickosiphy. This is often a dominant factor attenting the acceptability of a certain data model (for example, relational vesus object), a certain vendor, or a certain development methodology and tools (for example, use of an object) ortented analysis and design tool and methodology may be required of all new applicationed.
- Foundarity of personnal nucli the system: If the programming staff within the organization is finalitar with a particular 16895, it may be factored to reduce training cost and learning time.
- 3 Availability of cender services: The availability of vendor assistance in solving problend with the system is unportant, since moving from a non-DBMS to a DEMS environment is generally a nojor undertaking and requires pouch vendor solvnance at the start.

Another factor to consider is the USV's portability among different types of hardwark-Many commercial DBSSs new have versions that run on many hardware/software configurations (or **platforms**). The need of applications for backup, recovery performance, integrity, and secondy must also be considered. Mony DBSs are current, being designed as rotat solutions to the information-processing and information resource management needs within organizations. Most DBSS vendors are combining their priducts with the following options or built-in features:

- Text editors and browsers.
- Report generators and listing utilities.
- Communication software (often called teleprocessing monitors).
- 5. See the docuses of in Chapter 22 concerning this issue

- Data entry and display features such as forms, screens, and menus with automatic editing features.
- aspury and access tools that can be used on the World Wide Web (Web enabling tools).
- Oraphical database design tools.

A large attount of "third-party" software is available that provides odded functionality to a 0.0048 in each of the phove areas. In thre cases it may be preferable to develop in-house software rather than use a 0.0005— for example, if the applications are very well defined and are off known beforehold. Under such circumstances, an an-house custom-designed system that be appropriate to implement the known applications in the next efficient way. In most cases, however, new applications that were not foreseen at design time come up after system implementation. This is precisely why DRMSs have become very popular. They facilitate the incorporation of new applications with only momential modifications to the existing design of a database. Such design evolution—is a feature present to various degrees in commercial DBMSs.

# 12.2.4 Phase 4: Data Model Mapping (Logical Database Design)

The next phase of database design is to create a conceptual schema and external changes in the data model of the selected UGMs by mapping these schemas produced in Phase 20. The mapping can proceed in two stages:

- System-independent matging. In this states, the matping does not consider any specific characteristics of special cases that apply to the DBMS implementation of the data model. We already discussed DBMS independent mapping of an FR schema to a relational schema in Section 7.1 and of EER schemas to relational schemas in Section 7.2.
- 2 Initiating the schemas to a specific DPMS: Dufferent DBMSs implement a data model by using specific modeling features and constraints. We may have to adjust the schemas obtained in Step I to conform to the specific implementation features of a data model as used in the selected DPMS.

The result of this phase abundable DOL statements in the language of the chosen (1958) that specify the conceptual and external level schemas of the database system. But if the DOL statements include some physical design parameters, a complete (DDL specification must react until after the physical database doing phase is completed. Many anomated (CASE (computer-assisted software cogniceting) design tools (see Section 12.5) on generate DDL for commercial systems from a conceptual schema design.

# 12.2.5 Phase 5: Physical Database Design

Theseal database design is the process of choosing specific storage structures and access paths for the database files to achieve good performance for the various database.
applications. Each DBOS offers a variety of options for ble inganitation and access paths. These usually include various types of indexing, clustering of related records on disk blocks, linking related records via pomers, and various types of hashing Drife a specific traffic is chosen, the physical database design process is restricted to choosing the most appropriate structures for the database files from among the options observed by that DBMs. In this section we give generic guide must for physical design decisions they hold for any type of DBMs. The following criteria are often used to guide the choice of physical database design options:

- 1 Redouse tone. This is the clapsed time between submitting a database transaction for execution and receiving a response. A mojor influence on response time that is under the control of the OPMS is the database access time for data items referenced by the transaction. Response time is also radiuenced by factors not under DBMS control, such as system load, operating system scheduling, or communication delays.
- Space unlegation. This is the amount of storage space used by the database files and their access path structures on disk, including indexes and other access paths.
- Transaction chronophysic: This is the average murdler of transactions that can be preensed per minute, it is a critical parameter of transaction systems such as those used for airline posetvations or banking. Transaction throughput must be measated under peak conditions on the system.

Typically, average and worst-case limits on the preceding parameters are specified as part of the system performance requirements. Analytical or experimental techniques which can include processpring and simulation, are used to estimate the average and worst-case values under different physical dealers decisions, to determine whether the meet the specified performance requirements.

Performance depends on record size and number of records in the file. Hence, we must estimate these parameters for each life. In addition, we should estimate the update and remeval partents for the file contributively from all the manacrinos. Attributes and for selecting recease should have primary access paths and secondary address constructed for them. Estimates of file growth, either in the record size because of new attributes or in the number of records, should also be taken into account during physical durabase design

I

:

The result of the physical database design phase is an provide termination of storage structures and access paths for the database files. It is almost always necessary to modify the design on the basis of its observed performance after the database system is implemented. We include this activity of **database turning** in the next phase and cover a mithe unitext of query optimization in Chapter 16.

## 12.2.6 Phase 6: Database System Implementation and Tuning

After the logical and physical designs are completed, we can implement the database system. This is repically the responsibility of the DDA and is carried out in conjunction with the database designers. Language statements in the DFC (data definition language) and ding the SFE (atorage definition language) of the selected DSEs are completed and used to create the database schemas and (empty) database files. The database can then be loaded (populated) with the data. If data is to be converted from an earlier computation, system, conversion contines may be needed to reform at the data for loading into the new database.

Database transactions must be implemented by the application transammers by referring to the conceptual specifications of transactions, and then writing and usuing program code with embedded OSE commands. Once the transactions are ready and the data is loaded into the database, the design and implementation phase is over and the certainnal phase of the database system begins.

Most systems and lide a maniforming utility to collect verformance statistics, which are lept in the system catalog of data dictionary for later analysis. These include statistics on the number of invocations of predelined transactions of queries, input/output activity quart files, counts of file pages or index records, and frequency of tudes usage. As the dataset system requirements changes in often becomes necessate to add or remove existing tables and to reorganize some files by changing primary access methods or by dropping old indexes and constructing new ones. Some queries or transactions may be rewritten for better performance. Database turing continues as long as the database is in evidence, as long as performance problems are discovered, and while the requirements kep changing.

## 12.3 Use of UML DIAGRAMS AS AN AID TO DATABASE DESIGN SPECIFICATION<sup>6</sup>

## 12.3.1 UML As a Design Specification Standard

in the first section of this chopter, we discussed in detail how expanii trans work outh intermition systems and obborated the carious activities in the information system life code. Onlywe- are an integral part of information systems in most expand trans. The phases of Jurbase design starting with requirements analysis op to system implementation and runagivere introduced at the end of Section 12.1 and discussed in detail in Section 12.2 Industries always in the need of Section 12.1 and discussed in detail in Section 12.2 Industries always in the need of section, implementation and deployment. The approach systements analysis, incohere, desired, implementation and deployment. The approach that is receiving a wide attention and accept bility and that is also proposed as a standard by the OMG (Object) Management Oroup) is the **Unified Modeling**. **Language** (UML) approach, it provides a mechanism in the form of diagrammatic notation and associated language synatic tensions the entire heavile. Presently UML is used by sufficient developers, dimendelory data designers, database architects, etc. receipting of software developers, databased and associated to specify the environment consisting of software, commuminions and bardware to implement and deploy the application.

The contribution of Almo Udd logic to the USM, and Rational Rose sections is much appreciated

UML combines commonly accerted concerts from many OO methods and methodologies (see Subhographic notes for the contributing methodologies that led to UML). It is upplicable in any domain, and is language, and platform-independent, ssoftware achitects can model any type of application, running on any operating system, programming language or network in UML. That has made the approach very widely applicable. Tools like Rammal Rose are cutrently popular for drawing UML diagrams – they – enable software developers to develop clear and ensy-to-nucleostand models for specifying, muching, constructing and documenting components of software systems. Since the scope of UML extinds to software and application development at large, we will not cover all aspects of UML here. Our goal is to show some relevant UML entations that are commonly used in the requirements collection and analysis as well as the conceptual design physics (planes 1 and 2 in Figure 12.1). A detailed application development methodology using UML is outside the scope of rbits look and may be found in various resthools deviced to object scienced design, software engineering, and UML (see hibliographic notes).

Class diagrams, which are the end result of conceptual database design have already been discussed in Sections 3.5 and 4.0. To arrive at the class diagrams, the information may be gathered and specified using use case diagrams, sequence diagrams and state chan diagrams. In the rest of this section we will first introduce the different types of UML diagrams binely to give the result an idea of the scope of UML. Then we will present a shall sample application to illustrate the use of use case, sequence and statechan diagrams and shaw how they lead to the eventual class diagram as the final conceptual desired. The diagrams presented in this section performs to the statefield UML notation and have been drawn using the coll Rational Rose. Section 12.4 will be devoted to a gateful discussion of the use of Rational Rose in database application design.

#### 12.3.2 UML for Database Application Design

The database community has started cubricing UML and now many database designers and developers are using UML for data moduling as well as for subacquent phases of database design. The advantage of UML is that it over rhough its concepts are based on object-oriented rechniques, the resulting models of structure and behavior can be used to design both relational, object-oriented and object-relational databases (see Ubipres 22 to 22 for detiration of object databases and object-relational databases). We alreade numduced UML Class Diagrams, which are similar to the FR and FFR diagrams in Section 3.6 and 4.6, respectively. They give a structural specification of the database schemas in an object-oriented sense by showing the name, are dures and operations of each class Their normal use is to describe the collections of data objects and their inter-relationships which is consistent with the goal of conceptual database design.

One of the major contributions of the UML approach has been to bring the traditional database modulers, analysis and designers - together with the software application developers. In Figure 12.1 we showed the phases of database design and implementation and how they apply to these two groups. UML has been able to garginera communities and container notation or a meta-medel that can be indepted by both of these communities and

ailared to their rouch. Whereas we dwely solely on the structural aspect of modeling up Chapters 3 and 4, UML also allows us to do behavioral or/and dynamic modeling by introducing -various types of diagrams. This results in a more complete specificarnosi description of the overall database application. In the next sections we will first summance the different UML diagrams and their give an example of the rose case, repence and statechair diagrams in a sample application. A complete case study of a database application development is presented in Appendix B.

### 12.3.3 Different Diagrams in UML

UML defines time types of diagrams divided into two corregones.

Structural Diagrams. These describe the structural or static relationships among components. They include Class Diagram, Object Diagram, Component Diagram, and Deplocment Diagram.

Rehavioral Diagrams. Their purpose is to describe the behavioral or denomic relativiships among components. They include Use Case Diagram, Sequence Diagram, Colaboration Diagram, Statesburt Diagram, and Activity Diagram.

We introduce the nine types briefly below. The structural diagrams include:

#### A. Class Diagrams

Class diagrams capture the static structure of the system and act as foundation for other insidely. They show Classes, Interfaces, Collaborations, Dependencies, Ocneralizations, Association and other relationships. Class diagrams are a very method way to model the conceptual database schema. We showed examples of class diagrams for the company database schema in Figure 3.16, and for a generalization hierarchy in Figure 4.10.

Package Diograms. Package diagrams are a subset of class diagrams. They organize elements of the system area related groups called packages. A package may be a collection of related classes and the relationships between them. Package diagrams help minimize dependencies in a system.

#### B Object Discerns

Chier diagrams show a set of objects and their relationships. They correspond to what we called instance diagrams in chapters 3 and 4. They give a static view of a system at a gancular time and are normally used to test class diagrams for accuracy.

#### C. Camponent Diagrains

Congeneral diagrams illustrate the organizations and dependencies among software components. A component diagram typically consists of components, interfaces and dependency relationships. A component may be a source code component, a construccomponent or an executable component. It is a physical building block in the system and is represented us a rectangle with two small rectangles or table overlaid on its left side. An interface is a group of operations used of created by a component and is usually represented by a small circle. Dependency relationship is used to model the relationship between two components is represented by a dotted arrow printing from a component to the component if depends on. For databases, component diagrams stand for stored data such as tablespaces or partitions. Interfaces refer to applications that use the stored data.

#### D. Deployment Diagrams

Deployment diagrams represent the distribution of components (executables, libraries, tables, files) across the hardware topology. They depict the physical resources in 5 system, including nodes, components and connections, and are basically used to show the configuration of turn-time processing elements (the nodes) and the software processes that reside on them (the threads).

Now we will describe the behavioral diagrams and expand on these that are of particular interest.

#### E. Use Case Diagrams

Use case diagrams are used to model the functional interactions between users and the system. A scenario is a sequence of steps describing an interaction between a user and a system. A use case is a set of scenarios that have a common goal. The use case diagram was appropriate for bacabian? to visculize use cases. The use case diagram shows actors interacting with use cases and can be understood easily without the knowledge of any notation. An individual use case is shown as an oval and stands for a specific task performed by the system. An actor, shown with a stick person symbol, regresents an esternal user, which may be a human user, a relatesextative group of users, a certain role of a person in the organization, or anything external to the system. The use case diagram shows possible interactions of the system (in our case, a database system kand describes a use cases the specific tasks the system pollorms. Since they do not specify any implementation detail and are very easy to indepatanti, they are a good vehicle for communicating between the end users and developuts and help in easur user validation at an early stage. Test plans can also be easily generated using use cases diagrams. Figure 12.7 shows the use case diagram monition. The **include relationship** is used to factor our some common behavior from two or more of the original use cases - it is a form of rease For example, in a university environment shown in Figure 12.8, the use cases "register for courses" and "enter godes" in which actors student and poliescollate involved, include a common use case called "validate user." If it use case incorporates two or more streation the different scenarios, based on enconstances or varyous conditions, the extend relationship is used to show the subcases attached to the base case (see Figure 12.7)

Interaction diagrams - Interaction diagrams are used to model the dynamic aspects of a system. They basically consist of a set of messages exchanged between a set of Objects. There are two types of interaction diagrams. Sequence and Collaboration

<sup>7.</sup> Soul tools on et al. (1997)



**NCURE 12.7** The use-case diagram notation.

#### R Sequence Diagrams

Squence diagrams describe the interactions between various objects over time. They inscally give a dynamic view of the system by showing the flow of messages between objects. Within the sequence diagram, an object or an actor is shown as a bix at the top stadashed vertical line, which is called the object's lifeline. For a database, this object is upcally something physical (like a book in the warehouse) that would be contained in the database, an external document or form such as an order form, or an external visual view which may be part of a user interface. The fifeline inpresents the existence of chief over time. Activation, which indicates when an object is performing an action, is represented as a performance of a diffeline. Each message is represented as an action, is represented as a performance of two objects. A message hears o name and not have arguments indicates indicates the objection. The order of messages is safficen top to bottom. A sequence diagram also gives the option of self-call, which is safficen top to bottom. A sequence diagram also gives the option of self-call, which is



FIGURE 12.8 An example use case diagram for a University Database.

busically just a message from an object to uself. **Condition and Iteration markers** can also be shown in sequence diagrams to specify when the message should be ann and to specify the condition to send multiple nurkers. A return dashed line shows a return from the message and is optional, indese it carries a special meaning. Object deletion is shown with a large X. Facure 12.9 explains the notation of the sequence diagram.

#### G. Collaboration Diagrams

Collaboration diagrams represent interactions between objects as a series of sequenced messages, by Collaboration Diagrams the emphasis is on the structural organization of the ubjects that send and receive messages whereas in Sequence Diagrams the emphasis is on the interaction of the messages. Collaboration diagrams show objects as icons and mumber the messages; mumbered messages represent an externing. The sparial layout electrony diagrams allows becages among objects that show their amornial relationships. Use of collaboration and sequence diagrams to represent interactions is a matter of choice; we will be easily use only sequence diagrams.

#### H. Statechart Diagram

Statechart dugrants describe how an object's state changes in response to external events

To describe the behavior of an object, it is commott in most object-oriented techniques to draw a state diagram to show all the possible states an object can get inter-



**FIGURE 12.9** The sequence diagram notation.

as lifetime - the UML storecharts are based on their Harris<sup>3</sup> storecharts. They basicalle down state neichine consisting of stores, transitions, events and actions and or very useful in the conceptual design of the application that works against the database of stored objects.

The important elements of a statechair diagram shown in Figure 12-12, are as follows

- States: shown as boxes with nonded content, represent signations in the lifetime of an object.
- Transitions: shown as solid arrows between the states, they represent the paths between different states at an object. They are labeled by the eventmatte [quard] /actions the event traggers the transitions and the action results from 0. The guard is an additional and optimical condition that specifies a condition under which the change of state may nor even.
- Starghur al State: shown by a solid circle with in outgoing arrow to a state.
- Stop/Final State: shown as a double-lined filled cardle with an arrive pointing incost from a state.

See Fluid (19875)



FIGURE 12.10 The statechart diagram potation

Statechart diagrams are useful in specifying how an object's reaction to a message depends on its state. An event is something done to an object such as being sent a message an action is semething that on object does such as sending a message.

#### L Activity Diagrams

As trenty diagrams present a dynamic view of the system by modeling the flow of control from activity to agricuity. They can be considered as flow charts with states. An activity is a starg of doing surgebing, which could be a real-world process or an open non-on-some class in the datafasse. Typically, activity diagrams are used to model worldlow and mornal business operations for an application.

## 12.3.4 A Modeling and Design Example: University Database

In this section, we will briefly illustrate the use of the UML diagrams we presented above to design a sample relational database in a university setting. A large number of details are left on, to conserve space; only a stepwise use of these diagrams that leads towards a conceptual design and the design of program components is illustrated. As we undicated hence, the eventual DBMS on which this database jets implemented may be relational, objectoriented or objectorelational. That will not change the stepwise analysis and mode ingreate application using the UML diagrams.

Imagine a scenario with students enrolling in courses which are offered by protosors. The registrar's office is in charge of maintraining a schedule of courses in a nourse catalog. They have the authority railed and delete courses and to do schedule changes. They also secondiment limits on courses. The financial and office is in charge of processing order's and applications for which the students have to apply. Assume that we have in keight database that maintrains the data about students, professors, courses, and, etc. We also one to design the application that enables us to do the course registration, financial adapplication professing, and maintraining of the university wide course could by the regenaric office. The above requirements more the deputed by a screes of UML diagrams is shown below.

As mentioned previously one of the first steps involved in designing a database is to giver customer requirements and the best way to do this is by using use case diagrams. Suppose one of the requirements in the University Database is to allow the professors to energisdes for the courses they are reaching and for the students to be able to register for corresponding to these use cases diagram corresponding to these use cases on be drawn as shown in Figure 12.5.

Another helpful thing while designing a system is to graphically represent some of the states the system can be in This helps in viscoliting the various states the system can be an during the course of the application. For example, in our university database the various states which the system goes torough when the registration for a course with 52 sets is operied can be represented by the statechart drigram in bigure 12-11. Note that i shows the states of a course while enrollment is in process. During the enrollme state, the "Enroff Student" transition continues as long as the court of enrolled submits when 50.

Now having made the use case and state chait diagtum we can make a sequence dugain to visualize the execution of the use cases. For the university database, the separate diagram corresponding to the use cases student requests to register and selects a functular course to register is shown in Figure 12.12. The principulates and course area by are then checked and the course is then added to the student's schedule if the principulates are met and there is space in the course.

The above UML diagrams are not the complete specification of the University diabase. There will be other use cases with the Registria as the actor of the student



FIGURE 12.11 An example state bart diagram for the University Database

appearing for a test for a course and receiving a grade an the course, etc. A complete methodology for how to arrow at the class diagrams from the various diagrams we illustrated above is outside our scope betc. It is explained further in the case study (Appendix B). Design methodologies termine is matter of judgement, personal preferences, etc. However, we can make sure that the class diagram will account for all the specifications that have been given in the form of the use cases, state that and sequence diagrams. The class diagram in Figure 12.13 shows the classes with the structural relationships and the operations within the classes that are derived from these diagrams. These classes will need to be implemented to develop the University Database and regerier with the operations, it will implement the complete day schedole/enrofinem/ford opplication. For clear understanding only some of the important attributes are shown in classes with certain methods that origin the form the shown diagrams. It is conceivable that these class diagrams can be constantly upgeded as more details get specified and more functions evolve to the University Application.



**RGURE 12.12** A sequence chagrant for the University Database.

# 12.4 RATIONAL ROSE, A UML BASED DESIGN TOOL

## 12.4.1 Rational Rose for Database Design

Exicted Rose is one of the most important modeling tools used in the ordestry to develop information systems. As we pointed out in the first two sections of this chapter, database is scentral component of most information systems, and hence. Rational Rose privides the initial specification in UML that eventually leads to the database development. Many extensions have been made in the latest versions of Rose for data modeling and mow Rational Rose provides support for conceptual, logical and physical database modeling additional solution.

## 12.4.2 Rational Rose Data Modeler

Rational Rose Data Modeler is a visual modeling tool for designing databases. Our of the issues for its popularity is that wildle other data modeling tools it is CML based, it



FIGURE 12.13 A graphical data model diagram in Rational Rose.

provides a common real and language to bridge the communication gap between datahase designers and application developers. It makes it possible for database designers developers and analysis to yourk together, coprore and share basiness requirements and track them as they change throughout the process. Also, by allowing the designer, so model and design all specifications on the some platform using the same notation it improves the design process and reduces the tisk of errors.

Another major advantage of Rose is its process modeling capabilities that allow the modeling of the behavior of database as we specie, the short example above in the form of use cases, sequence diagrams, and starechart diagrams. There is the additional machinery of oliab ration diagrams to show interactions between objects and cerivity diagrams to nodel the flow of control which we did not elaborate upon. The eventual goal is to gravate the database specification and approxime code as moch as possible. With the Sose Data Modeler we can exprise triggers, stead procedures etc. (see Chapter 24 where database contain these featores) explicitly on the dogram rather than representing them with hicklen tagged values behind the scenes. The Data Wodeler also provides the capability to forward engineer a database in terms of constantly changing requirements ad overse engineer an existing implemented database into its conceptial design.

#### 12.4.3 Data Modeling Using Rational Rose Data Modeler

There are many tools and options available in Rose Data Modeler for data anothering. Ratenal Rose Data Modeler allows creating a data model based on the database structure or creating a database based on the data model.

Reverse Engineering — Reverse Engineering of the database allows the user to orate a data model based on the database arractine. If we have an existing DBMS database or DDL life we care use the reverse implacencing whard in Rational Rose Data Modeler to generate a conceptual data model. The reverse engineering whard life where listenally eras the schema in the database or DDL life, and recreates it in a data model. While doing so, it also includes the names of all quoted identifier entities.

forward Engineering and DDE Generation. We can uses create a data model' mostly from scratch in Rational Rose. Having created the data model we can also use it to provide the DDE to a specific DBMS from the data model. There is a Forward Engineering Wroad in Modeler, which reads the schema in the data model or reads bith descherge in the data model and the tablespaces justice data strange multi-land generates the appropriate DDE code in a DDE file. The would also provides the option of guerging a database by executing the generated DDE file.

Conceptual Design in UML NOtation. As mentioned orders one of the major admitages of Rose is that it allows modeling or databases using UML motation. ER

v The verse data model used by Ratisonal Rose Modeline comes and to our outron of averyptic transmisti

diagrams must often used in the conceptual design of datatuses can be easily built using the UML mutation as class diagrams in Rational Rose, e.g. the ER schema of our company example in Chapter 3 can be rednawn in Rational Rose using UML notation as follows.

This can then be converted into a graphical form by using the data model diagram, option in Rose.

The above diagrams correspond parity to a relational (logical) schema although they are at a conceptual level. They show the relationships among tables on the primary key (PK) foreign key (FK) relationships. **Identifying relationships** specify that a child rable common exist outhout the primer rable. Dependent tables), whereas non-identifying relationships specify a regular association between two independent tables. For botter and clear understanding, foreign keys ouromatically appear as one of the attributes in the child entities. It is possible to update the schemas diperty in their rest or graphical form for example, the relationship between the FMPI OYEF and PROJECT called WORKS-ON gray be deleted and Rose atomatically takes care of all the integin keys ere in the table.

Supported Databases. Some of the DBMSs that are currently supported by Rational Ruse include the following:

- IBM DB2 versions MVS and UDB 5.8, 6.8, and 7.0.
- Oracle UBMS versions 7-8 and 5-8.
- SQL Server QL Server DBMS cersions 6.5, 7.0 & 2032.
- Schose Adaptive Server version 12 x.

The SQL 92 Data Modeler does not reverse engineer ANSI SQL 92 DDLs, however t can forward engineer SQL 92 data models to DDLs.

Converting Englical Data Model to Object Model and Vice Versa. Rational Rose Data Modeler also provides the option of converting a log calibrations design to an object model design and vice versa. For example the logical data model shown in Figure 12.14 can be converted to an object model. This sort of mapping allows a deep understanding of the relationships between the logical model and database and helps in keeping them both up to dote with changes made during the development process. Figure 12.16 shows the Employee table after converting it to a class in an object model. The various table in the window can then be used to enter/display different types of information. They include operations, attendents and relationships to that class.

Synchronization Between the Conceptual Design and the Actual Database, Rose Data Modeler allows keeping the data model and database synchronized. It allows visualizing both the data model and the database and then, based on the differences, it gives the option to update the model or change the database.

Extensive Domain Support - The Data Modeler allows database designers in create a standard set of aser-denired data ryges and assign them to any column in the data

F.

ţ

ľ



RECRE 12.14 A logical data model diagram definition in Rational Rose.

rodel. Properties of the domain are then cascaded to assigned columns. These domains an tien be maintained by a standard group and deployed to all modelers when they agricitating new models by using the Rational Rose Framework.

Easy Communication Among Design Teams. As mentioned earlier, using a communication Among Design Teams. In Data Modeler an application decloper can access both the object and data models and see how they are related and the make informed and better choices afour how to build data access methods. There is alothe option of using **Rational Rose Web Publisher** to allow the models and the madata beneath these models to be available to everyone on the team.



FIGURE 12.15 The design of the university database as a class diagram.

What we have described above is a partial description of the capabilities of the final as it related to the conceptual and legical design phases in Figure 12.1. The entire range of UML diagrams we described in Section 12.3 can be developed and manytained in Res. For further details the reader is referred to the product literature. Appendix B developed full case study with the help of UML diagrams and shows the progression of design through different phases. Figure 12.17 gives a version of the class diagrams in Figure 3.18 drawn using Rational Rese.

| 🗄 jev če opraleni po ovri 1964 (na 2011) 👘 🖓 👘 👔 🕅                                   |
|--------------------------------------------------------------------------------------|
| - Filmen   Compression   Named   Filme  <br>Bornet   Datal   Completions   Alabatase |
| Been Bratting and Person Case, (204999)                                              |
| Lett IX. M                                                                           |
| Service 1                                                                            |
|                                                                                      |
| - mar - martine                                                                      |
|                                                                                      |
|                                                                                      |
|                                                                                      |
|                                                                                      |
|                                                                                      |
| ك ا                                                                                  |
| De Caror 2000 braves - Bob                                                           |

**BOURE 12.16** The class OM\_LMPI OYEE convesponding to the table triployee in Figure 12.14.

# 12.5 AUTOMATED DATABASE DESIGN TOOLS

The database design activity predominantly spins Phase 2 (conceptual design). Phase 4 this model mapping, or logical design) and Phase 5 (physical database design) in the discussed in Section 12.2. Discussion of classe 5 is deferred to Gapter 16 in the context of query optimization. We discussed Phases 2 and 4 in detail intractive use of the UML notation in Section 12.3 and pointed out the textures of the two discussed responses. As we pointed out the textures of the two database design tool. It is a software development tool and does database radeling and scheme design in the form of class diagrams as part of its overall object-related application development in activity of the task base transmitted application development in another of class diagrams as part of its overall object-related application development in another of class diagrams as part of its overall object-related application development in another of class diagrams as part of its relation transmitted on automating the transmitted by the text of the section of the text of text of the text of the text of text

When database technology was dost introduced, must database design was carried out neucally by expert designers, who used their experience and knowledge in the design press. However, or least reac factors indicated that some torm of automation had to be spheric if possible.

1. As an application involves mon, and more courdexity of data in terms of relationships and constraints, the number of options or different designs to model the same information keeps meteosing rapidly. It becomes difficult to deal with this complexity and the conception design alternatives manually.





FIGURE 12.17 The Company Database Class Diagram (Fig.3.16) drawn in Rational Rose.

2. The sheer site of some databases runs into foundeds of entity types and relation ship types making the task of in initially managing these disagree almost impossible. The meta information related to the design process we described in Section 12.2 yields another database that must be created, maintained, and queried as a database in its own right.

The above factors have green rise to many rook on the marker that come under the general category of CASE (Computer-Arded Software Engineering) tools for database design. Rational Rose is a grind evaluable of a number CASE tool. Typically these took consist of a combination of the following facilities:

 Diagnoming: This allows the designer to show a conceptual schema diagram in some road-specific metazon. Most roat/tors include entry, types, relationship type that are shown either as separate bases or snuply as directed or undirected lines, or durable constructs shown alongside the lates or to series of the difference press? attrochends or minimum constraints, attributes, keys, and so on <sup>12</sup> Some roots display inheritance hierarchies and use additional notation for showing the partial versus total and disport cersus overlapping nature of the generalizations. The diagrams are internally stored as concentrical designs and are available for modification as well as generation of reports, cross reference listings, and other uses.

- 2 Model mapping. This implements mapping algorithms similar to the ones us presourced in Sections 9.1 and 9.2. The mapping is system specific — most tools generate schemas in 5.3. DDC for Oracle, D52, Internets, Sybase, and other KD8viss. This part of the tool is most amenable to automation. The designer can edit the preduced (opt files if needed).
- 3 Diarge normalization: This milities a set of final tonal dispendencies that are supplied at the conceptual design or after the relational schemes are produced during logical design. The design decomposition algor think from Chapter 15 are applied to decompose existing relations into higher normal form relations. Typically, tools lack the approach of generating alternative (NF or R/NF designs and allowing the designer to select among them based on some current like the minimum number of relations or least amount of storage.

Mest tools incorporate some form of physical design including the choice of indexes. Autole range of separate tools exists for performance monitoring and measurement. The gobion of terring a design of the database implementation is still mostly handled as a human decision malering activity. Out of the phases of design described in this chapter, on available there is hardly any commercial root support is view integration (see Serier 12.2.2).

We will not survey database design roals here, but only mention the following the according to a solution of the product design tool should process:

- An easy-mask obsplace. This is cruical because in enables designers to focus on the task at hand, not on understanding the tool. Chaphical and point and click interfaces are curranoolly used. A few tools like the 528 St nod from brance use natural language input. Different interfaces may be tailored to beginners or to expert designers.
- Analyza al components: Tools should provide analytical components for tasks that are difficult to purform manually, such as evaluating physical design alternatives or detecting conducting constraints among views. This area is weak in most cartern roots.
- Heavily compositions. Aspects of the design that current be precisely quantified can be automated by entrying branstic rules in the design real to evaluate doogn alternatives.

<sup>10</sup> We do used the size as K. and S.M. class diagram retrations on Chapters 1 and 4. See Appendix A. Encoder of the different special diagrammatic not more used.

- 4 Trade-off-analysis: A trul should present the designer with adequate comparative analysis whenever it presents multiple alternatives to chause from. Tools should ideally incorporate an analysis of a design rhange in the conceptual design level design to physical design. Because of the many alternatives possible for physical design in a given system, such tradeoff analysis is difficult to carry our and most correct rook avoid it.
- 5. Doplay of design readry: 1X sign results, such as schemas, are obery displayed in diagrammatic form. Aesthetically pleasing and will faid out diagrams are not easily generate automatically. Multipage design because that are easy to read are another challenge. Other types of results of design may be shown as tables, lists of reports that can be easily interpreted.
- 6 Design configuration. This is a highly desirable feature. Its purpose is no verify that the re-ulting design sousifies the itstical requirements. Unless the requirements are coprozed and internally represented in some analyzable form, the verification connot be attempted.

Currently there is increasing awareness of the value of design tools, and they are becoming a must for dealing with large database design problems. There is also an increasing awareness that schema design and application design should go hand in land, and the current trend among CWSE tools is to address both areas. The peptianty of Rational Rose is due to the fact that it approaches the too prins of the design process shown in Figure 12.1 concurrently, approaching database design and application design is a unified activity. Some verifiers like Platiania provide a tool for data modeling and schema design (ERWin) and another for process modeling and functional design is design process by including design expertises in the form of rules largert system technology is also useful in the requirements collection and analysis phase, which is repositiones and doing tools to achieve better designs for complex databases. Without a clamp of being explorative, Table 12.1 has some popular database design and applications clamp of being explorative, Table 12.1 has some popular database design and application modeling tools. Companies in the table are hered in alphabeneal order.

## 12.6 SUMMARY

We started this chapter by discussing the role of information systems in organizations database systems are looked upon as a part of information systems in large-scale applications. We discussed how databases fit within on information system for information system for information systems in large-scale applications. We discussed how databases fit within on information system for information system for information system for information system for information excurse management in an organization and the life cycle they go through. We then discussed the six phases of the design process. The three phases commonly included as a part of database design are conceptual design, logical design (data model in pping), and physical design. We also discussed the initial phase of requirements collection and ara enwhich is often conceptual to be a prefering phase. In addition, at some point during the design, a specific DWIS package most be chosen. We discussed some of the organizational

| COMPANY                  | Τοοι                                                               | FUNCTIONALITY                                                                             |
|--------------------------|--------------------------------------------------------------------|-------------------------------------------------------------------------------------------|
|                          |                                                                    |                                                                                           |
| Embarcadoro Technologies | 2K Studio                                                          | Database Modeling in 10 and<br>anta 18                                                    |
|                          | .v: Artisu                                                         | Database administration and<br>space and security manage<br>ment                          |
| Úrado                    | Developer 2020 and Designer 2022                                   | Extrabase modeling, application<br>development                                            |
| Roșin Scriware           | Sestem Architeer 2001                                              | Data niskleling, object nisklel<br>ing, process modeling, struc-<br>tured analysis/design |
| Potranii Tixhnology      | Plannim Enterprise Modeling<br>Some ERWID, BPWin, Paradiem<br>Plus | Data, process, and business cont-<br>ponent modeling                                      |
| Personnel Inc.           | Powerful                                                           | Mapping from 0.0 to relational<br>model                                                   |
| Reconct                  | Rational Rose                                                      | Modeling in USL and applace<br>rion generation in C++ and<br>JAVA                         |
| Resic Ware               | RW Metro                                                           | Magning from Ost to relational<br>totskel                                                 |
| Resolution Ltd.          | XCase                                                              | Conceptual modeling up to code<br>moniterance                                             |
| Show                     | Enterprise Application Suite                                       | Datums deling, business logic<br>modeling                                                 |
| Yiat.i                   | Visio Enterprise                                                   | Data modeling, design and<br>recongreeering Visual Basic<br>and Visual U++                |

TABLE 12.1 SOME OF THE CURRENTLY AVAILABLE AUTOMATED DATABASE DESIGN TOOLS

ontria that come into play in selecting a 1980s. As performance publicus are detected, and as new applications are added, designs have to be modified. The importance of desping both the schema and the applications (or transactions) was highlighted. We derived different approaches to conceptual schema design and the difference between certralitiest schema design and the view integration approach.

We impoliced UML diagrams as in did to the specification of database models and deams. We introduced the entire range of structural and behavioral diagrams and then deat bed the notational detail about the following 0, ies of diagrams, are case, sequence, satisfact. Class diagrams have already been discussed in Sections 3.6 and 4.6, rejectively. We showed how requirements for a university database are spectfied using their diagrams and can be used to develop the conceptual design of the database. Only

illustrative details and not the complete specification were supplied. Appendix B develops a complete case study of the design and implementation of a database. Then we discussed the currently popular software development tool—Rational Rose and the Rose Data Medeler—that provides support for the conceptual design and legical design phase of database design. Rose is a nucleobroader tool for design of infermation systems at large finally, we buildly discussed the functionship and the functionship and legical design to design to design to database design tools that are more locussed or database design as upposed at Rose. A tabular summary of leatures was presented.

#### Review Questions

- 12.1 What are the six phases of database design? Discuss each phases
- 12.2. Which of the as phases are considered the main activities of the database design process itself. Why?
- 12.5. Why is at important to design the schemas and applications in parallel?
- 12.4. Why is it important to use an implementation-independent data model during conceptual schema design? What incidels are used in current design tools? Why?
- 12.5. Discuss the importance of Requitements Collection and Analysis.
- 12.6. Consider an armal application of a database system of interest. Define the requirements of the different levels of users in terms of data meeded, types of queries, and transactions to be processed.
- 12.7. Discuss the characteristics that a data model for conceptiol schema design shoul powers.
- 12.8. Compare and compast the two main approaches to conceptual schema design.
- 12.9. Diacuss the strategies for designing a single conceptual schema from its importements.
- (2.12) What are the steps of the view integration approach to conceptual scheme design? What are the difficulties during each step?
- 12.94 How worklip view integration and work? Design a simple resolutar architecture for such a tool.
- 12.77. What are the different strategies for view integration,
- 12.15 Discuss the factors that influence the choice of a DBMS package for the robumation system of an organization.
- 12.14 What is system-independent data model suppling! How is it different for system-dependent data model mapping?
- 12.15. What are the nuportant foctors that influence physical database design?
- 12.16. Discuss the decisions made during physical database design.
- 12.17. Discuss the macrifund metro-life cycles of in information system.
- 12.1%. Disease the guidelines for physical database design in 8068685
- 12.19. Descuss the types of modulications that may be applied to the logical database design of a relational database.
- 12.20 What functions do the typical database design cools provide?
- 12.21. What type of functionality would be desirable in automated roots to support optimal design of large databases?

## Selected Bibliography

Here is a vast amount of hteratore on database design. We first list some of the books dotakliess database design. Pottra et al. (1992) is a comprehensive treatment of concepical ord legical database design. Wrecerhold (1986) covers all physics of database design, with an emphasis on physical design. O'Neil (1994) here a detailed discussion of physical design and transaction issues in reference to commercial RP80bs. A large hidy of work on wraspital modeling and design was done in the eightics. Broche et al. (1984) gives a colknor of chapters on conceptual modeling: constraint specification and analysis, and misaction design. Yao (1985) is a collection of works ranging from requirements specification techniques to scheme restructuring. Teoroy (1995) emphasizes F2R modeling and discusses various aspects of conceptual and logical database design. McEadlen and Hoffer (1997) is a good markeduction to the business applications issues of database management.

Navathe and Kerschberg (1986) discuss all phases of database design and point out the role of data dictionaries. Goldine and Korng (1988) and AASE (1989) discuss the role of data dictionaries in database design. Roten and Shasho (1991) and Chris and March (1994) present different models for the problem of physical database design. Objectoneprod database design is discussed in Schlaer and Mellor (1988), Rombaugh et al. (1992). Marrin and Cdeli (1991), and Jacobson (1992). Recent basks by Blaha and Primerlani (1998) and Rumbaugh et al. (1999) consolidate the existing techniques in objectorients? design. Envice and Score (1997) is a quick introduction to C01.

Reprirements collection and analysis is a heavily researched topic. Charcoglu et al. (1997) and Lobors et al. (1993) present surveys of current practices in requirements acoust modeling and analysis. Cartoll (1996) provides a set of readings on the ast of scattion for requirements gathering in early stages of system development. Wood and Sker (1989) gives a good overcees of the official Joint Application Design (JAD) poces. Porter et al. (1991) describes the Z neutrins and methodology for formal spectration of software. Zoo: (1997) has classified the research efforts in requirements inputsions.

A trige body or work hos been produced on the problems of schema and view integration, which is becoming particularly relevant now because of the meed to integrate avarety of existing databases. Novathe and Gangit (1982) densed approaches to view interation. Schema integration methodologies are compared in Patim et al. (1986). Bruild work out many view integration can be found in Navathe et al. (1986). Elmastrier a (1986), and harson et al. (1989). An artegration tool based on Elmastrier et al. (1986) is Escaled in Sherh or al. (1988). Another view integration system is discussed in Havine ad Ran (1992). Chemicyli et al. (1991) describes a trot for mudular database design Myre (1987) discusses integration with respect to preexisting databases. The binary blaced strategy to view integration is discussed in Teorey and Fry (1982). A formal appreh to very programme, which uses inclusion dependencies, is given in Ubsanitya zs Vidal (1962). Rapiesh and Ram (1997) describe a methodology for integration of gluarships at schemas artiting the knowledge of integrity constraints; this extends the process work of Nacathe et al. (1984a). Sheeh at al. (1994) describe the issues of hidug global schemas by reasoning about archburg relationships and entity tagethness. Navathe and Savas re (1996) describe a product approach to building global schemas based on operators opplied to schema components. Santucci (1998) provides a detailed treatment of refinement of FER schemas for integration. Cascano et al. (1999) present a comprehensive sorvey of conceptual schema analysis techniques.

Transaction design is a relatively less thoroughly researched topic. Mylopoulos et al. (1980) proposed the CAXIS language and Albano et al. (1987) developed the CALIED system, both of which are comprehensive systems for specifying transactions. The CORISAS language for the ECR model (Elmasti et al. (1985) contains a transaction specification copability. Naviathe and Polaroman (1991) and Nga (1991) discuss transaction modeling at general for semantic data models. Elmagornica (1992) discusses transaction modeling at general for semantic data models. Elmagornica (1992) discusses transaction models for advanced applications. Potion et al. (1992) chaps. 8, 9, and 100 discuss bigh level transaction design and joint podysis of data and functions. Shasha (1992) is an excellent source on database tuning.

Information about some well-known commerzial database design tools can be found at the Web attes of the venders (see company names in Table 12.1). Principles behind automated design tools are discussed in Batrin et al. (1992, chap. 15). The SD-9 tool from France is described in Metrics et al. (1998). DKL (1997) is a special issue on natura language issues in databases.





# DATA STORAGE, INDEXING, QUERY PROCESSING, AND PHYSICAL DESIGN



# Disk Storage, Basic File Structures, and Hashing

Datibates are stored physically as files of meanls, which are typically stored or magnetic lisks. This chapter and the reset deal with the orginization of databases in storage and the softmapes for accessing them efficiently using various algorithms, some of which require auchary data structures called indexes. We start in Section 13.1 by unmiducing the conserver of computer storage hierarchies and how they are used in database systems. Section 13.2 is devoted to a description of magnetic disk storage devices and their characteristics, and we also briefly describe imagnetic disk storage devices. Howing discussed different sprage technologies, we then turn our attention to the methods for organizing data on disks. Section 14.3 covers the technique of double buffering, which is used to speed stacked of multiple disk blocks. In Section 13.5 docuses the various ways of formatting and stating records of a file on disk. Section 13.5 docuses the various types of operations that are typically opplied to records of a file. We then present these primary methods for againing records of a file on disk, unonleted records, docused in Section 13.6 ordered used to speed used to a speed to records of a file. We then present these primary methods for againing records of a file on disk, unonleted records, docused in Section 13.6 ordered used in Section 13.6 and bashed records, in Section 13.8.

Section 13.9 very briefly discusses files of inixed records and other printics methods to opartiting records, such as B-trees. These are printicularly relevant for storage of idject-strented databases, which we discuss later in Chapter- 22 and 21. Section 13.9 describes RAID (Redundant Arrays (4 Inexpensive (or Independent) Disk-)—a data arrays (ostern architectore that is used commonly in large organizations for better islability and performance. Finally, in Section 13.12 we describe storage arca networks, a max recent approach for managing stored data on networks. In Chapter 14 we discustechniques for creating anothary data structures, called indexes, that speed up the search for and retrieval of records. These techniques involve storage of anothary data, called index files, in addition to the file records themselves.

Chapters 13 and 14 may be browsed through or even original by readers who have already studied file organizations. The material covered here is necessary for and estanding Chapters 15 and 16 that deal with query processing and query optimization.

# **13.1** INTRODUCTION

The collection of data that makes up a comparentized database must be stored **physically** on some computer storage medium. The DBMS software can then retrieves updates and process this data as acceded. Computer storage media form a *sorage bioteceky* that includes two main categories:

- Primary storage. This category includes storage media that can be operated an directly by the computer central processing rang (CPC), such as the computer main memory and smaller but faster cache memories. Primary storage usually provides fast access to data but is of limited storage capacity.
- Secondary storage. This category includes magnetic disks, optical disks, and tapo. These slevices usually have a larger capacity, cost less, and provide slower access to data than do primary storage devices. Data in secondary storage cannot be pricesiad directly by the CPD it must last be capaed into primary storage.

We will first give an interview of the various storage devices (see for primary and secondary storage an Section 13.1.1 and will then discuss how databases are typically handled in the storage barranche in Section 13.1.2.

## 13.1.1 Memory Hierarchies and Storage Devices

In a modern componen system data resides and is transported throughout a hierarchy of storage modua. The highest-speed memory is the most expensive and is therefore available with the least capacity. The lowest-speed memory is offline type storage, which is essentially available in indefinite storage capacity.

At the primary storage ferel, the memory hierarchy includes at the most expensive end cache memory, which is a static RAM (Random Access Memory). Cache memory is typically used by the UPC to speed up execution of programs. The next level of emiry storage is 160AV (Dynamic RAM), which provides the main work area for the 450 to keeping programs and data and is popularly called main memory. The advantage of 168AV is its low cost, which continues to decrease; the drawback is its volatility<sup>1</sup> and lower speed compared with static RAW. At the secondary storage ferel, the lucratedy recludes activities disks, as well as mass storage in the totic of 80M (Compact Disk-Read-Oub

Volathe memory twistally loses its content on case of a power ordage, where who work of any does not.

Memory) devices, and finally tapes at the least expensive end of the hierarchy. The starage capacity is measured in kilolytes (Kbyte or 1000 bytes), megalytes (Mbyte or 1 million bytes), gigapytes (Gbyte or 1 billion bytes), and even teraberes (1000 Gbytes).

Programs reside and execute in DRAM. Generally, large permanent databases reade or secondary storage, and portions of the database are read into and peritten from bullers in main memory as needed. Now that personal computers and workstations have bulleds of megabytes of data in DRAM, it is becoming possible to load a large fraction of the database into main memory. Eight to 16 gigabytes of 50% or, a stogle server are recoming commonplace. In some cases, entire databases can be kept in mean memory auth a backup copy on mognetic disk), leading to **main memory databases;** these are pationality useful in real-time applications, which store databases that contour routing and have recommended, applications, which store databases that contour routing and have recommended.

Between DRAM and magnetic disk storage, another form of memory, flash memory, is beening common, particularly because it is nonvolatile. Each memories are highdensity high-performance vacancies using ECRO94 (Electrically knowled Programmable Read-Only Memory) to hipology. The advantage of flash memory is the fast access speed; the high-year ingrass that an entire block must be enaced and written over at a time.<sup>5</sup> Flash nemory cards are appearing as the data storage medium in appliances with capacities arging from a law megabytes to a few gigabytes. These are appearing in camerus, MP3 theory, USB storage accessories, etc.

COROLI disks store dara optically and are read by a laser COROLI contain precorded data that cannot be occurrenteen, workin (Write-Once-Read Many) disks are a form of optical starge used for archiving data; they allow data to be written once and read any number of areas without the possibility of erasing. They hold about half algebras of data per disk and Isumach langer than magnetic disks. Optical juke box memories are an area of COROM data in the bundreds of graphytes, then interacted unres are in the bundreds of graphytes, then interacted unres are in the bundreds of graphytes, then interacted unres are in the bundreds of milleconds quite a bit slever than magnetic disks. This type of storage is contribuing to data the receive in cost and increase in capacitoes of magnetic disks. The COROM storage is contribuing to define because of the rapid decrease in cost and increase in capacitoes of magnetic disks. The COROM storage is a recent standard for optical disks. Clowing 4.5 to 15 graphytes of storage per disk. Most personal computer disk drives new ready ToPOO and TOP disks.

Finally, magnetic tapes are used for archiving and backup storage of data. Tape jakehozes—schich contain a bank of tapes that are intologice and can be automatically kided onto tape drives—are becoming popular as tertiary storage to held terabytes of data. For example, NASAN 60% (Earth Observation Satellate) system stores are breed database- in this tashion.

Many large organizations are already finding it normal to have terabyte-sized databases. The term very large database connor by defined precisely any more because

<sup>2</sup> For example, the INTELED281C 928A is a 32-increasing capacity flush memory with 78 nonconvendaccessions, and 450 killses and write transfer rate.

There is to conclude the lower carcunal 430 rpm], ground higher latence delays and how transfer maximum 2020 to 220 km/second).

disk storage capacities are on the tise and costs are declining. It may very some be reserved for databases containing tens of terabytes.

#### 13.1.2 Storage of Databases

Databases typically store large amounts of data than must period over long periods of time. The data is accessed and processed repeatedly during this period. This contrasts with the notion of transient data structures that perior for only a lumited time during program execution. Most databases are stoled permanently for persistantly on magnetic disk secondary storage, for the following reasons:

- Generally, databases are too large to fit entirely in main memory
- The circumstances that cause permanent less of stored data arise less frequently for disk sciendary storage than for primity storage. Hence, we refer to disk—and other secondary storage devices—as nonvolatile storage, whereas main memory is often called volatile storage.
- The cost of storage per unit of data is an order of neighbould less for disk than for puinary storage.

Some of the newer technologies (such as optical disks, PVDs, and rape jukebows are likely to provide valide alternatives to the use of inographic disks. Databases in the future may therefore reside at different levels of the memory hierarchy from thost descrabed in Section 13.1.1. However, it is anticipated that magnetic disks will continue to be the racdium of primary choice for large databases for years to come. Hence, it is important to study and understand the properties and character stics of magnetic disks and the way data files can be creatified on disk in order to design effective databases with acceptable performance.

Magnetic tipes are herperitly used us a storage medium for backing up the database because storage on tape ensists even less than storage en disk. However, access to data vertage is quite show. Data stead on tapes is offline; that is some intervention by an operate—or an antennatic leading device—to load a tape is needed before this data becomes available. In contrast, disks are online devices that can be accessed directly at any time.

The techniques used in store large amounts of structured data on disk are important for database designers, the 1984, and implementers of a 19845. Database designers and the 1984 must know the advantages and distdyantages of each storage technique when they design, implement, and operate a database on a specific 19878. Usually, the 1988 has several options available for organizing the data, and the process of physical database design involves choosing from among the options, the particular data organization recliniques that best surpling group application requirements. DBC 5 system unplementer must study data organization techniques so that they can implement them efficiently and thus provide the 1964 and users of the ORIS with sufficient options.

Typical database applications need only a small portion of the database at a time to processing. Whenever a certain portion of the data is needed, is must be located on disk, copied to mean methods for processing, and then rewritten to the data it the data is changed. The data stored on disk is organized as **files** of **eccords**, **b**ach record is a callection of data values that can be interpreted as facts about existings, their attributes, and their relationships. Recards should be stored on disk in a manner that makes it possible to anote them efficiently whenever they are needed.

There are several primary file organizations, which determine how the records of a fillure physically placed on the disk, and hence how did records can be accessed. A heap jde (or seached jdg) places the means on disk or no particular order by appending new means a diversition the fill, whereas a certed jde (or sequential jde) keeps the records ordered by the value of a particular field (called the sorr key). A *hashed* file uses a hash function applied to a particular field (called the sorr key). A *hashed* file uses a hash function applied to a particular field (called the hash key) to determine a record's placement on add. Other primary file organizations, such as *Batters*, use tree structures. We docuse parary file organizations on Sections US 6 through 13.9. A secondary organization or auxiliary access structure allows efficient access to the records of a file based or chemise jobs than these than have been used for the primary file organization. Most of these exist is indexes and will be discussed in Chapter 14.

# 13.2 SECONDARY STORAGE DEVICES

In first section, we describe some characteristics of magnetic disk and magnetic tape storiged vices. Readers who have studied these devices already may just browse through this symm.

## 13.2.1 Hardware Description of Disk Devices

More tild dyks are used for storing large arrounts of data. The most basic unit of data on the disk is a single bit of information. By mognetizing an area on disk in certain ways, one commake (the present a bit value of either 0 (tero) or 1 (one). To code information, bits ite coupled into bytes (or **characters**). Byte sites are typically 4 to 6 bits, depending on the computer and the device. We assume that one character is stored in a single byte, and so use the terms type and character interchangeably. The **capacity** of a disk is the number of bytes it can store, which is usually very large. Small floggy disks used with an erocomputers typically hold from 400 KEytes to 1.5 Mbytes, hand disks for morios typically hold from several hundred Mbytes up to a few Obytes; and large disk packs used with serverand maniformes have capacities that range up to a tew tens or hundreds of Obytes. Disk aspantices continue to grow as technology improves.

Whatever their capacity, disks are all made of magnetic material shaped as a thin aredar disk (Figure 13.1a) and protected by a plastic or acrylic cover. A disk is **single-side**d jt it stores intermation on only one of its sorfaces and **double-sided** if both surfaces are used. To increase storage capacity, disks are assembled into a disk pack (Figure 14.1b), which may include many disks and hence many surfaces. Information is stored on a disk soface in concerning modes of small and physical active fractional and stores.

<sup>4</sup> in some disks, the circles are now connected into a kind of continuous spiral



**FIGURE 13.1** (a) A single-sided disk with read/write hardware. 'b) A disk pack with read/write hardware.

citled a track. For disk packs, the tracks with the same diameter on the various surfaces are called a **cylinder be**cause of the shape they would form it connected in space. The concept of a cylinder is important because data stored on one cylinder can be removed much favor than if invert distributed among different cylinders.

The number of tracks on a disk target item a low banded to a few thousand, and the capacity of each track typically ranges from tens of Klyres to 150 Klyres. Because a track two-divided generating a large oncourt of information, it is divided generated blocks or sectors. The division of a track into sectors is hard-coded on the disk surface and cannot be changed. One type of sector organization colls a portion of a track that subtends a track angle at the center is a sector (Figure 33.25). Several other so for an adjustications are possible, one of which is to have the sectors subtend shaller angles at the center is a sector (Figure 33.25). Several other so for any interdistion are possible, one of which is to have the sectors subtend shaller angles at the center of other the sector density of recording (Figure 13.25). A technique called 283 (Zone Bit Recording) allows a range of callinders to have the same number of



**EXCERF 13.2** Different sector organizations on disk. (a) Sectors subtending a fixed angle (b) Sectors maintaining a uniform recording density.

scors per arc. For example, cylinders 0, 99 may have one sector per mack, 170–199 may law own per track, etc. Not all disks have their tracks divided into sectors.

The division of a mack into equal-sized disk blocks (or pages) is set by the operating even dorong disk formatting (or initialization). Block site is fixed during mitralication and cannot be changed dynamically. Typical disk block sites range from 512 to 4096 Sites A disk with hard-coded sectors often has the sectors subdivided into blocks during initialization. Blocks are separated by fixed-size interblock gaps, which include specially usled control information sentrem during disk initialization. This information is used to determine which block on the track follows each interblock gap. Table 13.1 impresents speakeations of a typical disk.

There is continuous improvement in the storage capacity and transfer rates associated with disks; they are also progressively gerring chapter — currently coving only a fraction of a dellar per inegabyte of disk violage. Casts are going down so rapidly that casts as low 0.1 couMB which transfers to \$16.16 and \$1KD? are not for away.

A disk is a random access addressable device. Transfer of data between munimemory and disk takes place in units of disk blocks. The hardware address of a block—is combination of a cylinder number, task number (soffske number within the cylinder of which the track is located), and block number (within the track) is supplied to the disk (Chardware, hi muny modern disk drives, a single number carled DA (Logical Block Address) which is a number between C and n (assuming the total capacity of the disk is C-1 blocks), is mapped automatically to the right block by the disk drive controller. The akliess of a huffer—a contingious reserved area in main storage that holds are block—is its provided. For a read command, the block from disk is copied into the buffer, whereas ke is write command, the contents of the buffer are copied into the disk block.

## TABLE 13.1 SPECIFICATIONS OF TYPICAL HIGH-END CHEETAIL DISKS FROM SEAGATE

| Description                        | Cheetah X15 36LP        | Cheetah 10K-6       |  |
|------------------------------------|-------------------------|---------------------|--|
| Model Number                       | ST335732LC              | ST3146827LC         |  |
| Lorm Factor (width)                | 3.5 mch                 | 5 Sunch             |  |
| t leight                           | 25 4 ann)               | 25.4 mm             |  |
| Walth                              | 101.6 mm                | 101 D mm            |  |
| Weight                             | $2.68~{ m K_{E}}$       | $0.73~\mathrm{Kp}$  |  |
| Capacity/Interface                 |                         |                     |  |
| Formarred Capacity                 | 36.7 Obytes             | 146.8 Obyres        |  |
| Interface Type                     | 82 pin                  | ՏԲ բiդ              |  |
| Configuration                      |                         |                     |  |
| Number of disks (physical)         | 4                       | 4                   |  |
| Number of heads (physical)         | 8                       | х                   |  |
| Number of Cylindets                | 18.47%                  | 49 654              |  |
| Bytes per Sector                   | 512                     | 512                 |  |
| Areal Densoy                       | N/A                     | 30.000 MInt /sq.mch |  |
| Track Density                      | N/A                     | 64.000 Trackshirich |  |
| Recording Density                  | N/A                     | 572,000 bits/inch.  |  |
| Performance                        |                         |                     |  |
| Transfer Rates                     |                         |                     |  |
| Internal Transfer Rate (min)       | 522 Mbns/sec            | 475 Mhits/sec       |  |
| Internal Transfer Rate (mars)      | 709 MPits/acc           | 840 Mburs/sec       |  |
| Formatted Int. Transfer Rate (mm)  | 51 MIlytes/sec          | 43 MBytes/ee        |  |
| Formatted Int. Transfer Baro (max) | 69 MByresher            | 75 MByresl-ec       |  |
| External IA3 Transfer Rate (max)   | 320 MByteshee           | 320 MBytushee       |  |
| Seek Times                         |                         |                     |  |
| Area Soul Trees (Road)             | 1.6 more turnicalt      | 4.7 mars (communit) |  |
| And Sach Theory (Weight)           | d 2 manual formatica st | 5.2 mana /manad)    |  |
| Track, constant, Saula Rand        | Distance from a shi     | O A mare (respect)  |  |
| Track out of Seek Winne            | 0.8 marc (uppeak)       | 0 Survey (republic  |  |
| Average Linency                    | 2 msec                  | 2 99 mage           |  |
| Other                              |                         |                     |  |
| Default Beffer (eache) site        | 8,192 Klytes            | 8.300 Khyros        |  |
| Spinale Spead                      | 15K gan                 | ICK (pun            |  |
|                                    | -                       | -                   |  |

#### TABLE 13.1 SPECIFICATIONS OF TYPICAL HIGH-END COFFETALI DISKS FROM SEAGATE (continued)

| Re | ы | hil | iεγ |  |
|----|---|-----|-----|--|
|    |   |     |     |  |

| Mean Time Bersees n Failure (MTBF) | (,200.000 Hours              | 1.239.000 Hours             |
|------------------------------------|------------------------------|-----------------------------|
| Recoverable Read Errors            | 10 per 10 <sup>1</sup> ° Hus | 10 per 1012 bits            |
| Nonecocerable Read Errors          | Fper 10 <sup>15</sup> Fats   | 1 per - C <sup>D</sup> birs |
| Seek Errors                        | 12 per 10° bits              | 10 per 10% has              |
| termore building for buildings     |                              |                             |

Somet thes secural confidences blocks, called a cluster, may be transferred as a unit. In this cise the buffer size is adjusted to match the maniferrot lostes in the cluster.

The actual bardware nuclearation that relate of whites a block is the disk read/write **beak** which is part of a system called a **disk drive**. A disk or disk pack is committed in the disk linke, which includes a motor that notates the disks. A read/write head uncludes an electronic component attached to a **mechanical arm**. Disk packs with multiple suffaces in controlled by several or allowing localism for each sustaine (see Figure 13.15). All missing connected to an actuator attached in another electrical motor, which more she walk-one for each sustaine type 13.15). All missing connected to an actuator attached in another electrical motor, which more she walk-more heads in musicin and positions them precisely over the cylinder of tracks systemed in a block address.

Disk drives for hard disks rotate the disk pack continuousle at a constant speed oppically ranging between 5400 and 15,000 rpm). For a floggy disk, the disk drive begins to notice the disk whenever a particular read or write reducest is indiated and reases rotation score after the data massler is completed. Once the read/write head is positioned or the right mark and the block specified in the block address moves under the read/write head the electronic component of the read/write head is activated to manster the data. Some disk units have fixed read/error bracks with as many heads as there are tracks. These are called fixed-head disks, a track or cylinder is selected by electronically switching to the appropriate read/order head in the line action of the work disk up to work action of the work brack head disks. For inved-head disks, a track or cylinder is selected by electronically switching to the appropriate read/order head in the cost of the additional read/write heads is pare high so fixed-head disks or end commonly used.

V disk controller, repically embedded in the disk drive, controls the disk drive and anotheces it to the computer system. One of the starshold methods used today for disk drives on 5% and workstations is called SCSI (Small Computer Storage Interface). The controller accepts high-level (je) commands and takes appreciate action to position the num and cases the read/write action to take place. To transfer a disk block, given its address, the disk to transfer action to do this is called the seek time. Typical seek times are 7 to 10 meet or disk operations and 3 to 8 meets on servers. Following that, there is another delay—called the rotational delay or latency—while the beginning of the desired block rotates into position under the rotation is 4 meet and the average rotational delay is the time per rotation is 4 meet and the average rotational delay is the time per light rotation is 4 meet and the average rotational delay is the date; this is called the start time is needed to transfer time, the rotational delay is the time. Hence, the tart from the time readed to locate and transfer time. Hence, the tart time needed to locate and transfer areal another regional delay and block indices is the same of the seek time, interfaced to locate and transfer time.

transfer time. The seek time and torariumal delay are estably much larger than the block transfer time. To make the transfer of multiple blocks more efficients, it is common to transfer several consecutive blocks on the same track or exlinder. This elaminates the seek time and torational delay for all but the first block and can result in a substantial saving of units when transfer rate for calculating the time required to transfer consecutive blocks. Appendix B contains a discussion of these and other disk parameters.

The time needed to locate and transfer a dok block is in the order of induseconds usually ranging from 12 to 60 mass. For configuous blocks, locating the first block takes from 12 to 60 most, but transferring subsequent blocks may take only 1 to 2 most each Many search techniques take advantage of consecutive removal of blocks whet searching for data on disk. In any case, a transfer time its the order of induseconds is considered quite high compared with the time required to process data in main memory by current v10s. Heads, linearmig data on disk is a major bondoteck in database applications. The file structures we discuss here and its Chapter 14 attempt to initiate the number of block ingusters needed to locate and transfer the required data from disk to main memory.

#### 13.2.2 Magnetic Tape Storage Devices

Dasks are random access secondary storage devices, because an arbitrary disk block may be accessed "at random" once we specify its address. Magnetic tapes are sequential access devices; recarcess the  $n^{th}$  block on tape, we must first scan over the preceding  $n^{-1}$ . blocks, Data is stored on geds of high-capacity magnetic tape, somewhar similar rotache or videotopes. A tape drive is required to read the data from er to write the data to a tape real. Cloudly, each group of bits that forms a byte is stored across the tape, and the lyre themselves are stored consecutively on the tape.

A readjurite head is used to read or write data on supe. Data records on tage are also stored in blocks-solthough the blocks may be substantially larger than those for disks and interblock gaps are also quite large. With typical tapit densities of 1600 to 6250 bytes ther inclusive upreal interblock gap1 of 0.0 itselfus corresponds to 960 to 5750 bytes or austed storage space. For better space utilization it is costonary to group many records together in one block.

The manucharacteristic of a tabe is us requirement that we access the data block on sequential order. To get to a block of the middle of a real of tape, the tape is mounted and then scatted until the required block gets under the read/write head. For this reason, tipe access can be slow and tapes are not used to store online data, except for some specialized applications. However, tapes serve a very important function—that of backing up the database. One reason for blockup is to keep copies of disk files in case the data is lost because of mechanical multionerion. For this reason, disk files are copied suffice. For tape, for many online critical applications such as achine reservation.

<sup>5.</sup> Called inwriterial gips in tape tempology.
systems, to avoid only downtune, numeral systems are used keeping three sets of uluntical disks -- rays in ordine operation and one as backup. Here, attland disks become a backup -knice. The three are rorated so that they can be switched in case there is a failure on one to the just disk drives. Topos can also be used to store excessively large database files. Finally, database files that are seldern used or are outdated but are required for historical neord keeping can be archived on tape. Recently, smaller Simus magnetic tapes (surplar io those used in comcorders) that can store up to 50 Obstess as well as 4-min helical scan dita cartifidges and writable (108 and 1830s have become popular media for backing up day files from workstations and personal computers. They are also used for storing images and system libraries. Packing up enterptise databases so that no transaction information is est is a major undertaking. Corrently tupe libraries with slots for several hurdred cumdees are used with Digital and Superdigitat Linear Tapes (DLIs and SULTS) having caucines in handleds of gigabsies that record data on linear tracks. Robotic arms are used to write on multiple cattindges in parallel using multiple type drives with automatic labeling software to identify the backup cartineges. An example of a gant library is the Life2 model of Storage Technology that can scale up to 13.2 Petabytes (Petabyte - 3000) TB' with a thrup of rate of 55 TB/hour. We deter the discussion of disk storage technology called SAID, and of storage area networks, in the end of the chapter.

## 13.3 BUFFERING OF BLOCKS

When several blocks need to be transferred from disk to main memory and ell the block sidnesses are known, several buffers can be reserved in main memory to speed up the cansfer. While one buffer is being read or written, the CP, Can process data in the other buffer. This is possible because an independent disk (A) processor (controller) exists that, once started, can proceed to transfer a data block botwern memory and disk independent at and in purallel to CPD processing.

Figure 13.3 illustrates have two processes can proceed in parallel. Policesses A and B are ranning concurrently in an interleaved fashion, whereas processes C and D are tuning concurrently in a parallel fashion. Where a single (T2, controls multiple processes, parallel execution, is not possible. However, the processes can still run concurrently in an interleaved way. Buffering it most in-etol when processes can tual concurrently in a parallel fashion, either because a separate disk (O processor is available in because multiple CPO processors exist.

Figure 13.4 illustrates how reading and processing can proceed in parallel when the trainequired to process a disk block in memory is less than the time required to read the near Elock and fills buffer. The CO can start processing a block once its transfer to main memory is completely at the same rime the disk b/c processor can be reading and consterring the next block into a different buffer. This technique is called double huffering indicate also be used to entitle a continuous stream of blocks from memory to the disk. Double buffering permits continuous reading or writing of data in consecutive disk block which eliminates the scele rine and rotational deby, for all but the first block into the reading or writing of data in consecutive disk blocks, which eliminates the scele rine and rotational deby, for all but the first block into the programs.



FIGURE 13.3 Interleaved concurrency versus parallel execution





# **13.4 PLACING FILE RECORDS ON DISK**

In this section we define the concepts of records, record types, and files. We then discustechniques for placing file records on disk.

### 13.4.1 Records and Record Types

Data is usually stored in the form of records. Each record consists of a collection of related data values of items, where couly value is formed of one or more lytes and correspands to a particular field of the record. Records assually describe entities and their autibutes. For example, an BOLGOU record represents an employee entity, and each field value in the record specifies some attribute of that employee, such as NAPE, RECORD F, SU-PR, or SUPERVISE. A collection of field names and their corresponding data types consulrices a record type or record format definition. A data type, associated with each field, getfies the types of values a field care take.

The data type of a field is usually one of the standard data types used in programming. Hase include numeric (integer, lone integer, or faving root), string of datacters incellength or varying), Poolean (having 2 and 1 or filler and FALSE values only), and smetures specially coded date and time data types. The combet of bytes required for each data type is based for a given computer system. An integer may require 4 bytes, a long integer 6 bytes, a real number 4 bytes, a Postean 1 byte, a date 10 bytes (assuming a ternat of YYYAMMADD), and a faxed-denyth string of k clearatters in each field value. For example, an E00(our record type may bytes as there are claracters in each field value. For example, an E00(our record type may by datand in using the C programming language narrow i as the following structure:

```
struct employee{
 char name[30];
 cbar ssn[9];
 int salary;
 int jobcode;
 char department[20];
}.
```

In recent database applications, the need trug arise for storing data items that consist of lage mathematical objects, which represent images, digitized video of audio atteams, or free set. These are referred to as BLOBS yParary Large Objects). A 910B data item is spirally stored separately from its record in a pool of disk blocks, and a geniter to the 30P concluded of the record.

### 13.4.2 Files, Fixed-Length Records, and Variable-Length Records

Affle is a sequence of records, for many coses, addresseds in a file are of the same record rgs. If every meand in the file has exactly the same size (in bytes), the file is soid to be male up of **fised-length records**. If different records in the file have different sizes the file is said to be made up of **variable-length records**. A file may have variable-length records for scenal reasons.

- The file records are of the same record type, but one of more of the fields are of varying size (variable-length fields). For example, the sum field of encored can be a variable length field.
- The trie records are of the same record type, but one or more of the fields may have multiple values for individual records, such a field is called a **repeating field** and a group of values for the field is often called a **repeating group**.

- The file records are of the same record type, but one or more of the fields are optional: that is, they move have values for some but not all of the file records (optional fields).
- The file contains records of differencies order types and hence of variang size (mixed file). This would occur it related records of different types were characted (placed rogether) on disk blocks for example, the second seconds of a particular student may be placed following that student's record.

The fixed-length PPPOPF records in Figure 13.5a have a record size of 71 bytes. Every record has the some fields, and held lengths are fixed, so the system can identify the starting byte position of each field relative to the starting position of the record. This facilitates locating field clues by programs that access such files. Notice that it is possible to represent a file that legalally should have carrable-fingth records us a fixed-length records life. For example, in the case of optional fields we could have every field included in every file record but store a special null value it no value exists for that field. For a repeating field, we could allocate as many spaces meach record as



**FIGURE 1.3.5** Three record storage formats: (a: A foed-length record with six fields and size of 21 bytes, (b) A record with two variable-length fields and three fixed-length fields in ( A variapte-field record with three types of separator characters. the maintain manher of traines that the field can take. In other case, space is wasted when certain records do not have values for all the physical spaces provided in each record. We now consider other optimis for formations records of a file of variablelength records.

For catable-length fields, each record has a value for each field, but we do not know the exact length of some field values. To determine the laytes within a particular record that represent each field, we can use special **separator** characters tank as 2 m % or \$)—which do not appear in any field value—to ferromate variablelength fields (Figure 14.3b), or we can store the length in laytes of the field in the record, preceding the field value.

A file of records with sprioral fields can be terminited an different ways. If the total number of fields for the record type is large but the number of fields that actually appear in a typical record is small, we can include in each record a sequence of sheld-name, field values pairs rather than just the field values. Three types of separator characteristic used in Figure 13.7c, although we could use the field values. Three types of separator character for the field name than the field value and separator character for the field name from the field value and separating one field for the next field. A more practical option is to using a short field type code—say, at mager number to each field and include in each record a sequence of sheld-type, field-values pairs rather than sheld-manae, field-values pairs.

A repeating field needs one separator character to separator the repeating values of the teld and another separator character to indicate termination of the field. Finally, for a file that includes (crowls of different types, each record is preceded by a record type indicator bridestandably, programs that process files of variable length records—which are usually failed the file system and hence hidden from the typical programmers—need to be more complex than those for fixed-length records, where the starting position and size of each tell are known and fixed.<sup>5</sup>

### 13.4.3 Record Blocking and Spanned Versus Unspanned Records

The records of a file most be allocated to disk blocks because a block is the one of dota introjal between disk and memory. When the block site is larger than the record site, each Nek will contain numerous records, although some files may have an usually large records that connot fit in one block. Suppose that the block site is *B* bytes. For a file of facilities that connot fit in one block. Suppose that the block site is *B* bytes. For a file of facilities that connot fit in one block, suppose that the block site is *B* bytes. For a file of facilities that connot fit in one block, such  $B \cong R$ , we can fit  $\delta f = ||B/R||$  records per block, where the ||(x)|| (flow function) means down the bundler *x* to an integer. The value *b* fit is called the **blocking factor** for the file. In general, *R* may not divide *B* exactly, so we have sime moved space in each. Slock equal to

6 = (46 + R) by test

<sup>6</sup> Ober schemes are also ressible for representing wireabledength records.

To utilize this unused space, we can store part of a record on one block and the test on another. A **pointer** at the end of the **first** block points to the block containing the remainder of the record in case it is not the text consecutive block on disk. This organization is called **spanned**, because records can spon more than one block. Whenever a record is larger than a block, we mass use a spatned organization. If records are not allowed to cross block boundaries, the organization is called **unspanned**. This is used with fixed-length records having  $0 \ge R$  because it makes each record start at a known location in the block, simplifying record processing. For variable-length records, either a spanned, if its average record is larger it is advantagerous to use spanning to reduce the lost space in each block. Figure 13.6 illustrates spanned versas unspanned organization.

For variable-length records using spanned organization, each block may store a different number of records. In this case, the blocking factor fity represents the areago number of records per block for the rile. We can use tip to calculate the number of blocks b needed for a file of a factoride.

 $b = |\langle \tau/b/\tau \rangle$  blocks

where the  $\left[ \left( \mathbf{x} \right) \right]$  (onlog function) rounds the value x up to the next integer.

#### 13.4.4 Allocating File Blocks on Disk

There are several standard rechniques for allocating the blocks of a file on disk. In configuous allocation the file blocks are allocated to consecutive disk blocks. This makes reading the whole file very fast using diouble buffering, but it makes expanding the file difficult. In **linked allocation** each file block contains a pointer to the next file block. This makes it easy to expand the file bit makes it slow to read the whole file. A combination of the two allocates elusters of consecutive disk blocks, and the clusters are linked. Clusters



FIGURE 13.6 Types of record organization, (a) Unspanned, (b) Spanned.

are sometimes called file segments or extents. Another possibility is to use indexed allocation, where one or more index blocks contain pointers to the actual file blocks. It is also common to use combinations of these techniques.

#### 13.4.5 File Headers

A file header or file descriptor contains information about a file that is needed by the systemprograms that access the file records. The header includes information to determine the disk addresses of the file blocks as well as to record format descriptions, which may include field lengths and order of fields within a record for fixed-length unspanned records and field type codes, separator characters, and record type codes for variable-length records.

To search for a record on disk, one or more blocks are copied into many memory bulkes. Programs then search for the desired record is records within the buffers, using the information of the file header. If the address of the block that contains the desired avoid is not known, the search programs must do a linear search through the the blocks. Each the block is copied into a buffer and searched other until the record is located or all de file blocks have been searched unsuccessfully. This can be very time consuming for a largefile. The goal of a good file organization is to locate the block that contains a desired word with a minimal number of block transfers.

# 13.5 OPERATIONS ON FILES

Operations on files are usually grouped into retrieval operations and update operations. The former durine charge any data in the file, but only locate certain records so that their fold values can be examined and processed. The ilitter change the file by insertion of deletion of records or by modification of field values, file either case, we may have to select one or more records for retrieval, deletion, or modification resold on a selection condition (or filtering condition), which specifies criteria that the desired record or records must satisfy

Consider an instant file with fields KAME SSR. SALARS, IDECCE, and ECFARTHERT. A simple selection condition may involve an equality comparison on some field value—for example, (SSR = 1234567897) or (GEFARTHERT = 'Research'). More complex conditions can make other types of comparison operators, such  $m \ge m \exp(mple)$  is (Second SWO). The general case is to have an arbitrary Boolean expression on the fields of the file so the selection condition.

Search operations on thest are generally based on simple selection conditions. A complex condition must be decomposed by the D805 (or the programmer) is extract a single condition that can be used to locate the records on disk. Each located record is then there discontinue whether it satisfies the full selection condition. For example, we may extract the simple condition (measurem = 'tiesconch') from the complex condition (too see 23000) AND (necessary n = 'tiesconch') is each record condition n = 'tiesconch') is exceed and then tested to see if it also satisfies (second  $\approx 30000$ ).

When several file records satisfy a search condition, the first record—with respect to the pressual sequence of file records—withinfully located and designated the **current**  record. Subsequent search operations commence from this record and locate the new record in the file that satisfies the condition.

At tual operations for locating and accessing file records vary from system to system. Below, we present a set of representative operations. Typically, high-fevel programs, such as 6800s software programs, necess the records by n-me these commands, so we sometimes refer to program variables in the following descriptions:

- Open. Propares the file for reading or writing. Allocates appropriate butters (typically at lease typic) to hold file blocks from disk, and refrieves the file header. Sets the file pointer to the beginning of the file.
- Reserv Sets the tile pointer of an open-life to the begaming of the life.
- Foid (or Liceoc). Searches for the first record that satisfies a search condition. Transfers the block containing that record into a main memory buffer (if it is not already there). The file geometry points to the record in the buffer and it becomes the current record. Supermass, different corbs are used to indicate behavior the located zoord is to be retrieved or opdated.
- Read (or Gar): Corpositive correct record from the buffer role program variable in the user program. This command may also advance the current record pointer to the next record in the file, which may necessative reading the next file block from disk.
- FordNext: Searches for the next record in the file that satisfies the search condition. Transfer: the block containing that record into a main memory inffer (if it is nat already there). The record is located in the butter and becomes the current record.
- Detec: Poletes the current record and (eventually) updates the file on disk to reflect the deteriority
- Modify: Medities some field values for the current record and (eventually) update the file on disk to reflect the modulication.
- Insert. Inserts a new proved in the full by locating the block where the record is take inserted, transferring that block into a main memory butter (if it is not already there) writing the record arrow the buffer, and (eventually) writing the better to disk roreflect the insertion.
- Closer Completes the file occess by releasing the bottles and performing any other meeded cleanup operations.

The preceding (except for Open and Close) an called record-at-a-time operations because each operation applies to a single record. It is possible to structuling the operations Find, buildNext, and Read into a single operation, Scan, whose description is as follows:

 Scan. If the file has just been opened or reset. Some entries the first record; otherwar it returns the next record. If a condition is specified with the operation, the returns record is the next or next record sometying the condition.

In database systems, additional sen-at-sectime higher-level operations into be applied to a file. Examples of these are as follows:

- FordAlf, Locares of the records in the file that satisfy a search condition.
- food for Locate) in Searches for the first woord that sorvices a search condition and then continues to locate the next n = 1 records satisfying the some condition. Transfers the blocks containing the n records to the num momenty failfer (if not already transf.)
- FudOnlocer, Remeves all the records in the file in some specified order.
- Rooganize: Starts the noorganization process. As we shall see, some his organizations require periodic rearginization. An example is to reorder the his records by sorting them on a specified field.

At this pent, it is worthwhile to note the difference between the terms ple organization and access method. A **file organization** refers to the organization of the data of a file into records, blocks, and access structures: this includes the way records and blocks are placed on the storage medium and interfunced. An **access method**, on the other hand, provides a group of operations—such as these listed earlier—that can be applied to a file ingeneral, it is possible to apply several access methods to a file organization. Some access methods, though, can be applied only to files organized in centars ways. For example, we omitt apply an indexed access method to a file without an index (see Chapter 6).

Usually, we expect to use some search conditions more than others. Some files may be static, meaning that update operations are ratch performed; other, more **dynamic** files may change frequently, so update operations are constantly applied to them. A successful fileorganization should perform as efficiently as possible the operations we expect to opply to goode to the file. For example, consider the owners file (Figure 14.5a), which stores the records for current employees in a company. We expect to insert records (when apployees are hitted), delete records (when employees leave the company), and mudify records for, when an employee's solary or job is changed). Deleting or modifying a record reputes a selection condition to identify a particular record for set of records. Retrieving one or more records also from the a selection condition.

If users expect mainly to apply a search condition based on sos, the designer must doose a file engametrion that facilitates locating a record given its sociale. This may molye physically ordering the records by lock value or defining an index on social (hagter 6). Suppose that a second appletation uses the file to generate employees products and requires that psychecks be grouped by department. For this appletation, it is test to store all employee records having the same department value contiguously, distering them into blocks and perhaps ordering them by name within each department. However, this arrangement conflicts with ordering the records by six values. If both upbeations are important, the designer should choose an organization that allows both operations to be done efficiently. Unfortunately, in many cases there may not be an organization that allows all needed operations on a file to be implemented efficiently. In subcases a compromise must be chosen that takes into account the expected importance and mix of remeval and update operations.

In the following sections and in Chopter 6, we discuss methods for organizing records of a file on disk. So coral general feelin quest such as ordering, hashing, and indexing, are ord to ensure access methods. In addition, various general techniques for handling periods and delenous work with many file arganizations.

## 13.6 FILES OF UNORDERED RECORDS (HEAP FILES)

In this simplest and most basic type of organization, records are placed in the rise in the order in which they are inserted, so new records are inserted at the end of the bloc Such on organization is called a heap or pile file.<sup>4</sup> This organization is often used with oldational access paths, such as the secondary indexes discussed in Chapter 6. It is also used to collect and store data network for future use

Inserting a new record is very efficient, the last disk block of the file is copied into a bottler, the new record is added, and the block is then rewritten back to disk. The address of the last file block is kept in the file block is then rewritten back to disk. The address of the last file block is kept in the file block is the order block is then rewritten back to disk. The address of condition involves a **linear search** through the file block by block—an expensive procedure. If only one record satisfies the search condition, then on the average, a program will read into memory and search ball the file blocks before it linds the record. For a file 0 blocks, this requires searching (l/2) blocks, intervenge. If not records in several records satisfies the program must read and wards all *b* blocks in the file.

To delete a second, a program must test find its black, copy the Block into a buffer, then delete the moord from the buffer, and finally **rewrite the black** back to the disk. This haves mused space in the disk block. Deleting a large number of records in this way reads in wasted strange space. Another rechrique used for record deletion is to have an exert lype or laticalled a **deletion marker**, stored with each record. A record is deleted by setting the deletion marker to a certain value. A different value of the marker indicates a valid (nor deleted) record. Search programs consider only valid records in a block when conducting their search. Both of these deletion rechrisples require periodic **reorganization** of the file to reclaim the unused space of deleted records. During reorganization, the file blocks are accessed consocutively, and records are packed by removing deleted records. After such a reorganization, the blocks are filled to copocity once more. Another possibility is to use the space of deleted records when inserting new records, although this requires extra bookk cepting to keep track of empty locations.

We can use either spinned or unspanned organization for on-unordered file, and it may be used with either fixed-length or variable-length records. Modifying a variable-length record may require detering the old record and inserting a modified record, because the modified record may not fit in its old space on disk.

To read all records in order of the values of some field, we create a sorted copy of the file. Sorting is an expensive operation for a large disk file, and special techniques to external sorting stoused (see Chapter 15).

For a file of unoncered fixed-longith records using anspended blocks and configuraallocation, it is straightforward to access any record by its position in the file. If the file records are numbered 0, 1, 2, ..., i = 1 and the records in each block are numbered 0, 1, ..., 1/r = 1, where bb is the blocking forcior, then the  $i^{20}$  record of the file is located in block  $\lfloor (q^{20}/r) \rfloor$  and is the 6 mod bb  $l^{20}$  record in that block. Such a file is often called a relative or direct file because records can easily be accessed threefy by their relative

<sup>7.</sup> Sometimes this organization is called a sequential file.

portions. Accessing a record by its position does not help locate a record based on a seach conditions however, it facilitares the construction of access paths on the file, such asibe indexes does seed to Chapter 5.

## 13.7 FILES OF ORDERED RECORDS (SORTED FILES)

We can physically order the records of a file on disk based on the values of one of their fields—called the **ordering field.** This leads to an **ordered** or sequential file.<sup>5</sup> If the ordering field is also a key field of the file—is field guaranteed to have a unique value in carfs is card—then the field is called the **ordering** key for the file. Figure 13.7 shows an ordered file with we as the ordering key field (assuming that employees have distinct names).

Ordered records have some advantages over unordered files. First, reading the records in order of the ordering key values becomes extremely efficient, because no sorting is recursed. Second, finding the next record from the current one in order of the ordering los availy requires no additional block accesses, because the next record is in the same block as the current one (unless the current record is the last one in the block). Third, using a watch condition based on the value of an ordering key field results in faster access when the binary search rechnique is used, which consultates an inspiratement over linear suches, although it is not often used for disk files.

A binary search for disk files can be done on the blocks rather than on the records. Soppose that the file has b blocks numbered 1, 2, ..., b; the records are ordered by according value of the rordenne key field, and we are searching for a record whose ordering key field value is K. Assuming that disk addresses or the file blocks are available rathe file header, the binary search can be described by Algor than 13 T. A binary search usefile accesses  $\log_2(b)$ . Blocks, whether the record is found or not—an improvement over litear searches, where, on the average, (*b*(2) blocks are accessed when the record is found, and *b* blocks are accessed when the record is found.

Algorithm 13.1) Binary search on an ordering key of a disk file.

```
1 ← 1; v ← b; (+ b is the number of file blocks-)
while (x $ 1) do
begin 1 ← (1 + v) div Z,
read block 1 of the file into the buffer,
if X < (ordering key field value of the first record in block ?)
ther v ← i 2 1
else if X > (ordering key field value of the fast record in block ?)
then l ← i + 1
else if the record with ordering key field value = K is in the buffer
them goto found
else goto notfound;
end;
yoto motfound:
```

#### Silbensteinseinen auf file bas also been used to refer to coordered tiles.

|                | NAME                                  | SŚN     | BIRTHDATE  | J <b>O</b> 8 | SALARY | SEX                                     |
|----------------|---------------------------------------|---------|------------|--------------|--------|-----------------------------------------|
| block 1        | Aaron, Ed                             |         |            |              |        | 1                                       |
|                | Apbott Dane                           |         |            |              |        | 7 I I                                   |
|                | · · · · · · · · · · · · · · · · · · · |         | <u>ŧ</u>   | _            |        |                                         |
|                | Acosta_Marc                           | L       |            | ]            |        | <u>:</u> )                              |
|                |                                       |         | <u> </u>   |              |        |                                         |
| block 2        | Adams, Juhn                           |         |            | L            |        | L                                       |
|                | Adams, Robin                          |         |            |              |        |                                         |
|                |                                       |         | <u> </u>   |              |        |                                         |
|                | Akers, Jan                            |         |            |              | i      |                                         |
|                |                                       |         |            |              |        |                                         |
| Diock 3        | Alexander, Ed                         |         | i          |              | 1      | ļļ                                      |
|                | Alfred Bob                            |         |            |              |        |                                         |
|                |                                       |         | :          |              |        |                                         |
|                | Allon, Sam                            |         |            |              |        | <u> </u>                                |
| hirek 4        | Alexa Terra                           |         |            |              |        | ——————————————————————————————————————— |
| 01040 H        | Andorn Koth                           | +       |            |              |        | ; I                                     |
|                |                                       |         | ⊑<br>•     | L J          |        | L                                       |
|                | Anderson Dob                          | Γ       | r <b>-</b> | г <b>~</b> – |        |                                         |
|                | <u> </u>                              | L ·     | <u> </u>   | L            |        |                                         |
| block 5        | Anderson, Zach                        | ī — · · | l. ——• -—  | $\square$    |        |                                         |
|                | Angoi, Joe                            |         |            |              |        |                                         |
|                | <b>_</b>                              |         | ŧ          |              |        |                                         |
|                | Archer: Sue                           |         |            |              |        |                                         |
|                |                                       |         |            |              |        |                                         |
| block 6        | Arrold Mack                           |         | I          | i i          |        |                                         |
|                | Anold Steven                          |         | <u> </u>   | <u>ا</u>     |        |                                         |
|                | l ,                                   |         | <u></u>    |              |        |                                         |
|                | Abine, Timothy                        |         |            |              |        |                                         |
|                |                                       |         |            |              |        |                                         |
|                |                                       |         | :          |              |        |                                         |
|                |                                       |         | · ·        |              |        |                                         |
| block n –1     | Wong, James                           |         |            | L            |        | L                                       |
|                | Wood, Donald                          | 1       |            |              |        |                                         |
|                |                                       |         | 1          |              |        |                                         |
|                | Woods Manny                           | I       | '          |              |        | I                                       |
| <b>b b b b</b> |                                       |         |            |              | ·      | F· — ]                                  |
| Diock A        | Winght Parn                           |         | i          |              |        | $\vdash$                                |
|                | Wysti, Charles                        |         | · ·        |              |        |                                         |
|                |                                       |         | t          |              |        | r                                       |
|                | Commer, Byron                         |         | · ·        |              | •      | L                                       |

FIGURE 13.7. Some blocks of an ordered (sequential) tile of record records with new as the ordering key field.

A search currentian involving the conditions >, <, =, and = on the ordering field is quite efficient, since the physical ordering of records means that all records satisfying the condition are contiguous in the file. For example, referring to higher 13.9, if the search currentian +16000 < 600—where simeans objectively before—the records satisfying the search cuterion are those from the beginning of the file up to the first record that has a set value storing with the letter G

Ordering does not provide any advantages for random or ordered access of the records based on values of the other nonordering pekis of the file. In these cases we do a breat search for random access. To access the records in order based on a nonocceting feld, it is necessary to create arother sorted copy—in a different order—of the file.

Inserting and deleting records are expensive operations for an ordered file because twirecords must teniam physically ordered. To insert a record, we must find its correct portion in the file, based on its indering field value, and then make spice in the file to must the record in that position. For a large file this can be very time consuming because, in the average, half the records of the file must be moved to make space for the record record. This means that half the file blocks must be read and rewritter after records are moved among them. For record deletion, the problem is less severe if deletion markers and periodic programmation are used.

One option for making insertion more efficient is to keep some unused space in each look for new records. However, once this space is used up, the original problem contracts. Another frequency used method is to create a temporary mondered the called at overflow or transaction (i.e. With this technique, the actual ordered the is called the main or master file. New records are inserted at the end or the coefflow file rather that in their correct position on the main file. Periodically, the overflow file is serted and merged with the master file during file reorganization. Insertion becomes very efficient, but at the cost of increased complexity in the search algorithm. The overflow file must be searched using a linear search of, after the binary search, the record is not found in the main file. For applications that do not require the most up-to-date information, overflow records can be grated during a search.

Modifying a field value of a record desends on two factors: (1) the search and non-to-locate the record and (2) the field to be modified. If the search condition incolors the ordering key field, we can locate the record using a binary search, when we must do a linear search. A monordering field can be modified by Conging the record and rewriting at in the same physical location of atsk—assuming heighteric ords. Modifying the rathering field means that the record can change it position in the life, which requires delation of the old record followed by insertion in the modified record.

Evading the file records in order of the ordering field is quite efficient divergence the records in overflow, since the blocks can be read consecutively using double futuring. To include the records in overflow, we must marge them in their correct perions; in this case, we can first reorganize the file, and then read its blocks separately. To reorganize the file, first som the records in the overflow file, and then increation using them with the master file. The records marked for deletion are removed during the reorganization.

| TYPE OF ORGANIZATION | ACCESS/SEARCH METHOD             | AVERAGE TIME TO ACCEM |  |  |
|----------------------|----------------------------------|-----------------------|--|--|
| Heap (Unordered)     | Sequential Sean (Linear          | A SPECIFIC RECORD     |  |  |
| Ordered<br>Ordered   | Sequential scan<br>Butary Search | b/Z<br>kog p          |  |  |

TABLE 13.2 AVERAGE ACCESS TIMES FOR BASIC FILE ORGANIZATIONS

Table 13.2 summarizes the average access time in block accesses to find a specific record in a file with *b* blocks.

Ordered files are rarely used in database applications onless an additional access path, called a primary index, is used; this results in an indexed-sequential file. This further improves the random access triae on the ordering key field. We discuss indexes in Chapter 14.

# 13.8 HASHING TECHNIQUES

Another type of primary file organization is based on hashing, which provides very fast access to records on corona search conditions. This organization is asially called a hash file." The search condition must be on equality condition on a single held, called the hash field of the file. In most cases, the hash held is also a key field of the file, in which case it is called the **bash** key. The idea behind hashing is to provide a function *b*, called *b* hash function or **randomizing functions** that is applied to the hash field value of a record and yields the address of the disk block in which the record is stored. A search for the most within the block can be carried out in a main memory buffer. For most records, we need only a single-block access to retrieve that record.

Hashing is also used as an internal search structure within a program whenever a group of records is accessed exclusively by using the value of one field. We describe the use of hushing for internal files in Section 13.9.1, then we show how it is modified to state external files on disk in Section 13.9.2. In Section 13.9.3, we discuss techniques to extending hashing to dynamically growing files.

### 13.8.1 Internal Hashing

For internal files, hashing is typically implemented as a **hash table** through the use of a array of records. Suppose that the array index range is from  $\delta$  to M = 1 (Figure 13.3.), then we have M slots whose addresses correspond to the array indexes. We choose a high function that transforms the bash field value into an integer between 0 and M = 1. One compare hash function is the  $\delta(N) = N$  mod M function, which returns the remainder of

A hash file has a co-been called a down file.



**RCURE 13.B** Internal hashing data structures that Array of Miposition of or use in internal bashing, the Collision resolution by charging records

an integer hash field value K after division by M, this calue is then used for the record address.

Noncoreger hash held values can be transformed into integers before the most firenen is applied. For character strings, the numeric (ASCII) codes associated with characters can be used in the transformation—for example, by notrophying those code takes. For a hash field whose data type is a string of 20 characters, Algorithm 13.26 can be used to calculate the hash address. We assume that the code function returns the

numeric code of a character and that we are given a bash field value if of type K laray [1-2016] that Go PASCAD or that K(2010m C).

Algorithm 13.2 Two snaple hosting algorithms: (a) Applying the most host time tion to a character string K. (b) Collision resolution by open addressing

```
(a) temp ← 1;
for i ← 1 to 20 do temp ← temp < code(X[i]) mod W;
hash address ← temp mod W;
(b) i ← hash_address(X); a ← i;
if location i is occupied
then begin i ← (i ← 1) mod M;
while (i f a) and location ; is occupied
do i < (i ← 1) mod M;
if (i = a) then all positions are full
else new_hash_audress ← i.
end:
```

Other hashing tonerions can be used. Une technique, called folding, involuapplying an arithmetic function such as addition or a logical function such as exclusive to different portions of the hash field value to calculate the hash address. Another technique involves picking some digits of the bash field value—tor example, the fluid, fifth, and eighth digits—to form the hash address. The problem with most hashing functions is that they do not guar intere that distinct values will hash realisance addresses, because the hash field space—the number of possible values a hash field can take—it usually much larger than the **address space**—the number of available addresses to records. The hashing function maps the hash field space to the address space.

A collision occurs when the hash held ratae of a record that is being inverted hashs to an address that already contains a different record. In this situation, we must meet the new record in some other position, since its hash address is occupied. The process finding another position is called collision resolution. There are numerous methods to collision resolution, including the following:

- Open addressing. Proceeding from the occupied position specified by the hash address the program checks the sub-equest positions in order until an unused (empty) postion is found. Algorithm 13.2b may be used for this purpose.
- Channest: For this method, various meriflow locations are kept, usually by extending the array with a number of overflow positions. In addition, a pointer field is addition each record location. A collision is resolved by placing the new record in an usual overflow location and sering the point resolved by placing the new record in an usual address of that overflow location. A bricked has of overflow records for each has address is thus maniformed, as shown in Figure U.35b.
- Madque hashing: The program applies a second bash function if the first resolts in a collision. If mother collision results, the program uses open addressing or apples a abod bash function and then uses open addressing if necessary.
- 10. A cetakel docussion of hashing functions is outside the score of our presentation

Each collision resolution method requires its own algorithms for insertion, retrieval, and deletion of records. The algorithms for chaining are the simplest. Deletion algorithms for open addressing are rather tricky. Data structures reschooks discuss internal bosting algorithms in more detail.

The goal of a coord hashing function is to distribute the records uniformly over the address space so us to minimize collisions while not leaving many unused focations. Similation and analysis studies have shown that it is usually best to keep a hash table between 70 and 90 percent full so that the number of collisions tertains low and we do not cost to much space. Hence, if we expect to have *r* records to store in the table, we should choose M locations for the address space such that (*ilM*) is betweere 0.7 and 0.9. It may also be useful to choose a prime number for M, since it has been demonstrated that this distributes the bash addresses before over the address space when the rund hashing function is useful. Other hash interficient may require M to be a power of 2.

## 13.8.2 External Hashing for Disk Files

Hading for disk files is called **external hashing**. To said the characteristics of disk storage, the target address space is made of **huckets**, each of which holds multiple records. A hold is either one disk block or a cluster of corrugaous blocks. The bashing function mays a key into a relative blocket number, rather than assign an absolute block address to the blocket. A rable maintained in the file header converts the blocket number into the corresponding disk block address, as illustrated in heater 13.9.

The collision problem is less severe with buckets, because as nonvinceords as will fir in a nucket can hash to the same bucket without causing problems. However, we must make provisions for the case where a bucket is filled to capacity and a new record being recred hashes to that bucket. We can use a variation of charmers in which a pointer is maintained in each bucket to a linked list of overflow records for the bucket, as shown in



FIGURE 13.9 Matching bucket numbers to disk block addresses.

Figure 3.5 10. The pointers in the linked list should be record pointers, which include isoth a block address and a relative record position within the block.

Howing provides the fastest possible access for retrieving on arbitrary record given the value of its hash acid. Although most good hash functions do not magnitum records in order of hash field values, some functions -called **order preserving**—do. A simple example of an order preserving bash function is to take the leftmost three digits of an anyone number field as the hash soldress and keep the records sorted by invoice number within each backet. Another example is to use an integer bash key directly as an index to a relative file, if the bash key values fill up a particular interval, for example, if employees we can use the identity bash function that incurtains order. Unfortunately, this only works it keys are generated in order by some application.

The backing scheme described is called static hashing because a fixed number of buckets M is allocated. This can be a series drawback for dynamic files. Suppose that we illocate M backets for the iddress space and let in be the maximum number of records the can fit in one backet, then an insist (at 4 M) records will fit in the allocated space. If the



FIGURE 13.10 Handling overflow for buckets by chaining.

evolve of records turns out to be substantially tower than ( $\sigma t \in MS$ , we are left with a lot of noised space. On the other hand, if the number of records increases to substantially more than to  $t \in M$ ), numerous collisions will result and retrieval will be slowed down because of the long lists of overflow records. In either case, we may have to change the number of Hecks M allocated and then use a new basing function that and not the new collision of M) to redstrated in the records. These reorganizations can be quite time consuming for large these to vary dynamic file organizations based on hisbing allow the number of brekets to vary dynamically with only localized morganization (see Section 13.8, 3).

When using external hashing, searching for a record given a value of some field other for the hash field is as expensive as in the case of an unordered file. Record deletion can be implemented by removing the record from its bucket. If the bucket has an overflow door, we can move one of the overflow records into the bucket to replace the deleted record. If the period to be deleted is already in overflow, we samply remove in from the linked for. Nonce that removing an overflow record implies that we should keep track of energy portrons in overflow. This is done easily by maintaining a briked first of unased averflow locations.

Modifying a record's field value depends or two factors (1) the south condition to loate the record and (2) the field to be modified. If the search condition is an equility conjustion on the hash field, we can locate the record efficiently by using the he-bing records; otherwise, we must do a linear search. A nonbash field can be modified by cations the record and rew firing it in the same backer. Modifying the high field means that the record can move to another backet, which requires deletion of the old record relowed by insertion of the modified record.

### 13.8.3 Hashing Techniques That Allow Dynamic File Expansion

A import drawback of the state hashing scheme just discussed is that the hash address spaces fixed. Hence, it is difficult to expand or shrink the file dynamically. The schemes described in this section attempt to remody this structure. The first scheme—extendible histing—stores an access structure in addition to the file, and hence is somewhat similar numbering (Chapter 6). The main difference is that the access structure is based on the calls that result after anytheation of the hash function to the search field. In indexing, the access structure is based on the values of the search field inself. The second rechmance, cifed linear bashing, dues not require additional access structures.

These hashing schemes take advantage of the fact that the result of applying a history function is a nontaegative integer and hence, can be represented as a binary number. The access structure is built on the **binary representation** of the bashing foretion result, which is a struct of bits. We call this the **bash** value of a record. Records red-bitboured among buckets based on the values of the *koding* bus in their hash values.

Extendible Hashang. In extendible hashing, a type of **directory** (an orray of 2<sup>7</sup>) Niker addresses—is maintained, where d is called the global **depth** of the directory. The mean value corresponding to the first thigh-order) of birs of a bash value is used as an



FIGURE 13.17 Structure of the extendible bashing scheme.

index to the array in determine a directory intry, and the address in that entry determines the bucket in which the corresponding records are stored. However, their does not have to be a distinct bucket for each of the 2<sup>st</sup> directory locations. Several directory locations with the same first  $d^s$  bits for their hash values may compare the same bucket address it all the records that bash to these locations in the a single bucket. A **local depth d^s** -stored with each bucket—specifies the number of bits on which the bucket contents are basis. Figure 13.13 shows a directory with global depth  $d^s = 3$ .

The value of d can be increased or decreased by one at a time, thus doubling or halving the number of contres on the directory array. Doubling  $\omega$  records if a backet, whose local depth d'us equal to the global depth d'eventions. Halving occurs if d > d for all the buckets after some deletions occur. Most record retrievals require two block accesses—one to the directory and the other to the bucket.

To illustrate backer splitting, suppose that a new inverted second causes overflow a the backet whose hash calues start with 21 , the third backet in Figure 13.D. Th records will be distributed between two buckets: the first contains all records whose hash values start with 310, and the second all those whose hash values start with 011. Now the readirectory locations for 310° and 011 point to the two new distinct buckets. Before the split, they pointed to the same bucket. The local depth d' of the two new buckets is 3 which since than the local depth of the two new buckets is 3.

If a backet that overflows and is split used to have a local depth d' equal to the global depth d of the directory, then the size of the directory must now be dealsted so that we can us an extra bit to distinguish the two new backets. For example, if the backet for records alose hash values start with 111 in Figure 13.11 overflows, the two new backets need a dectory with global depth d' = 4, because the two backets are now labeled 1110 and 1111, and hence their local depths are both 4. The directory size is hence doubled, and each of the other migmal locations in the directory is also split into two locations, both of which have the same pointer value as did the original location.

The main advantage of extendship hashing that makes it artitative is that the prioritance of the file does not degrade as the nle grows, as upposed to static external behing where cubisions increase and the corresponding chaining naises additional access. In addition, no space is allocated in extendeble hashing for future growth, but additional buckets can be allocated dynamically as model. The space eventional for the directory table is negligible. The maximum directory size is 25, where k is the norther of bosin the hash calue. Another advantage is that splitting causes minor reorganization in non-bucket are redistributed to the two new buckets. To only time a reorganization is more expensive is when the directory has to be doubled to below 0. A disadvantage is that the directory most be searched before accessing the bases themselves, resoluting or two block accesses instead of one in static hebring. This patematice penalty is considered manor and hence the scheme is considered upine devale.

Ungat Hashing. The idea behind linear bashing is to allow a hash file to expand and similars number of Nuckers dynamically sociate resoling a directory. Suppose that the forsities with M Euckers numbered 0, 1, ..., M + 1 and uses the model hash function  $h(0) = K \mod M$ , this hash function is called the initial hash function  $h_0$ . Overflow tecares of collisions is still needed and can be bandled by maintaining mainished or discretion and the file backet, the trace backet. However, when a collision leads to an overflow record in  $a_0$  file backet, the trace backet in the file—backet 0—is split into two backets, the tagend backet 0 and a new backet M at the end of the file. The records originally in baket 0 are distributed between the two backets based on a different hashing function  $k_{ab}KI = K \mod 2M$ . A key property of the two hash functions  $k_{a}$  and  $h_{a,1}$  is that any indexes that hashed to backet 0 based on  $h_0$  will hash to either backet 0 in backet M based on the start backet 0 in backet M based on the time tool backet M based on the trace of the backet 0 back

As further collisions lead to overflow records, additional backets are split in the linear index 1, 2, 3, ..., 11 enough overflows occur, all the original file backets 0, 1, ..., M = 1 will have been split, so the file now has 2M instead of M backets, and all backets use the lash function  $E_{0,1}$ . Hence, the records in overflow are eventually redistributed into right backets, using the function  $u_0$ , was a delever split of their backets. There is no directory only a value *n*—which is initially set to 0 and is incremented by 1 where you a

split sectors wis needed to determine which backets have been split. To retrieve a recert with Eash key value *K*, first apple the function *h*, to *K*: if *h*,(*K*) < *a*, then apple the function  $\hat{h}_{i+1}$  on *K* because the backet is already split. For all, n = 0, indicating that the function *h*, applies to all backets already shockets are split.

When n = M after being incremented, this signifies that all the original backets have been split and the hash function  $h_{i+1}$  applies to all records in the file. At this point, it is teset to 0 (zero), and any new collisions that cause overflow lead to the use of a new reashing function  $h_{i+1}(K) = K \mod 4M$ . In general, a sequence of hashing functions  $h_{i+1}(K) = K \mod 4M$ . In general, a sequence of hashing functions  $h_{i+1}(K) = K \mod 4M$ . In general, a sequence of hashing functions  $h_{i+1}(K) = K \mod 4M$ . In general, a sequence of hashing functions  $h_{i+1}(K) = K \mod 4M$ . In general, a sequence of hashing functions  $h_{i+1}(K) = K \mod 4M$ . In general, a sequence of hashing function  $h_{i+1}(K) = K \mod 4M$ . In general, a sequence of hashing function  $h_{i+1}(K) = K \mod 4M$ . The sequence of the set of a new here i = 0, 1, 2, ..., i move hashing function  $h_{i+1}(K) = K$  modes whenever all the buckets 2, 1, ..., (2M) = 1 have been split and n is reset to 2. The search for a record with hash key value K is given by Algorithm 13.3.

Splitting can be convolled by monitoring the file head factor instead of by splitting whenever an overflow occurs. In general, the file head factor t can be detived as  $i = \pi/(\theta) + N$ , where  $\tau$  is the current number of tile records  $2\theta$  is the maximum number of monitor that file the transmout number of monitor that for a bocket, and N is the current number of the falls below a certain rbreshold. Blocks at combined linearly and N is decremented appropriately. The net load can be used to trigger both splits and combinations; in this monner the file load can be keep within a desired integer splits can be traggered when the load exceeds a certain threshold—say, 0.5.

Algorithm 13.3: The search procedure for liteau lashing,

```
If n \in O

then m = h_n(K) (- m is the bash value of record with bash key K ·)

else begin

m \in h_n(K);

if w < n then m \leftarrow h_{n-1}(K)

end;
```

search the bucket whose bask value is m (and its overflow, if any):

# **13.9 OTHER PRIMARY FILE ORGANIZATIONS**

#### 13.9.1 Files of Mixed Records

The tile organizations see have studied so far assume that all records of a particular file are of the same mean type. The records could be of EARLOWERS, FROLENS, STOCENTS, OF DEPARTMENTS, but each file pentains records of only one type. In most database applications, we can one for subgroups in which memories types of entries are interclosed in various whys, as a saw in Chapter 3. Relationships among records in corrors thes can be represented by connecting fields.<sup>11</sup> For example, a variative record can have a connecting field worker i whose

The concept of fere grokeys in the relativitial model (Chapter 70 and references among objects in object-oriented models (Chapter 201 are examples of connecting fields).

value gives the name of the DEPARTMENT in which the student is majoring. This BACERDEPT fell afters to a DEPARTMENT entrity, which should be represented by a record of its own in the DEPARTMENT of a DEPARTMENT entrity, which should be represented by a record of its own in the DEPARTMENT of a DEPARTMENT for retained held values from two related records, we must retrieve one of the records first. Then we can use its connecting field value to retrieve the related record in the other file. Hence, relationships are implemented by logical field references among the records in distinct files.

hile organizations in object D00458, as well as legacy systems such as hierarchical indinetwork. D66685, often implement relationships omong records as physical relationships realized by physical contiguity (or clustering) of related records or by theored pointers. These file organizations typically assign an area of the disk to hold records of more than one type so that records of different types can be physically clustered on disk. If a particular relationship is expected to be used very hequently, implementing the relationship physically can increase the system's efficiency at removing related records. For example, if the query to retrieve a present record and all records for structure majoring in that department is very frequent, it would be describle to place each https://peetrocord.and.its.cluster of statist records contiguously on lisk in a mixed file. The concept of physical clustering of object types is used in object D09458 to store related objects together in a mixed file.

To duringuish the records in a cuxed file, each record has — in addition to its held values — a record type held, which specifies the type of record. This is type-ally the first field in each record and is used by the system software to determine the type of record it is about to process. Using the catalog information, the DSMF can determine the fields of that record type and their sizes, in order to interpret the data values in the record.

## 13.9.2 B-Trees and Other Data Structures as Primary Organization

Other data structures can be used for primary file organizations. For example, if both the iscard size and the number of records in a file are small, some 10.81% offer the option of a 6-nee data structure as the primary file organization. We will describe B-trees in Section 14.3.1, when we discuss the use of the P-tree data structure for indexing. In general, any due structure that can be adapted to the characteristics of disk devices can be used as a primary file organization for record placement on disk.

# 13.10 PARALLELIZING DISK ACCESS USING RAID TECHNOLOGY

With the exponential growth in the performance and caparity of semiconductor devices and memories, distort microprocessors with larger and larger primary memories are continuially becoming available. To match this growth, in is narrial to espect that secondary storage technology must also take steps to keep up in performance and reliability will processor technology.

A major advance in secondary storage technology is represented by the development of RAID, which originally storal for **Redundant Arrays of Inexpensive Disks**. Lately, the "I" on RAID is said to stand for **Independent**. The RAID idea received a very positive endoscenent by industry and has been developed into an elaborate set of alternative RAD architectures (RAIP levels 0 through 6). We highlight the main features of the technology below.

The name goal of 3 AO is to even our the widely different rates of performance improvement of disks against three in memory and interoprocessors.<sup>12</sup> While 8.85 capacities have quadrayled every two to three years, disk acrossomes are improving at losthan 10 percent per year, and disk mander name are improving at roughly 20 process per year. Disk equation are indeed improving at more than 50 percent per year, but the speed and access time improvements are of a much smaller magnitude. Table 13-3 shows trends in disk technology in terms of 1993 parameter values and rates of improvement, as well as where these parameters are in 2003.

A second qualitative disparity exists between the ability of special nucroprocessos than cater to new applications involving processing of video, audio, image, and spiral data (see Chapters 24 and 29 for details of tisese applications), with corresponding lack at for access to large, shared data sets.

The natural solution is a large array of small independent disks acring as a single higher-performance logical disk. A concept called **data striping** is used, which atalass parallelism to improve disk gentermance. Data striping distributes data transparently our multiple disks to make them appear as a single large, fast disk. Ergore 13.12 shows after distributed or stopped over time disks. Simpling improves overall, the performance by

|                                | 1993 PARAMETER VALUES   | HISTORICAL RATE OF<br>IMPROVEMENT PER<br>YEAR (%) | CURRENT (2003)<br>VALUES |
|--------------------------------|-------------------------|---------------------------------------------------|--------------------------|
| Areal density                  | 52-150 Mbits/sq. mak    | 27                                                | 36 Gbits/sg. mda         |
| Linear density                 | 42 000-00,000 bits/m.th | 13                                                | 570 Kbirshinch           |
| Inter-track density            | 1500-3000 (nicks/mch    | 10                                                | 64,000 tracks/meh        |
| Capacity<br>(3.5° form factor) | 100-2000 MB             | 27                                                | 14e GB                   |
| Fran-fer nate                  | 3-4 MB/s                | 22                                                | 43-78 MIVax              |
| Soek time                      | 7-22 aus                | \$                                                | 3.5 formers              |

#### **TABLE 13.3 TRENDS IN DISK TECHNOLOGY**

\*Shown, Francishara, Lee, Cubsing Kath, and Platersing 1, 894). At 20 Coordinating Surveys, Vol. 26, Nov. 2 (June 1964). Reprinted by prior solution.

\*"Source: IBM Curristan 36XP and 182X hard disk drives.

12. This was predicted by Gardian Bell to be about 40 percent every year between 1974 and 1894 and 1894 and supposed to escreed 50 percent per year.



**HGURE 13-72** Litata striping. File A is stripsed across four disks

alsoing multiple (A to be serviced in parallel, rhus providing high overall transfer rates. Data striping also accomplishes load balancing among disks. Moreover, by storing radialist unformation on disks using parity or some other error correction code, reliability can be improved. In Sections 13.3.1 and 13.3.2, we discuss how RAID achieves the non-important objectives of improved reliability and higher performance. Section 1933 discusses RAID organization-

#### 13.10.1 Improving Reliability with RAID

recar attay of 6 dicks, the likelihood of failure is a times as much as that for one disk done, if the NTTY (Mean Time To Failure) of a daw drive is assumed to be 200,000 hours or ibout 22.8 years (typical times range up to 1 million hours), that of a bank of 100 disk drive becomes only 2000 hours or 83.5 days. Keeping a single cost of data in such an arm of disks will cluse a significant loss of reliability. An obvious solution is to employ reliadancy of data so that disk tailores can be tributed. The disadvantages are minimialiformal 0/0 operations for write, extra computation to maintain redurdancy and to do relovery from errors, and additional disk capacity to store redundant information.

Our technique for introducing residualancy is called **mirroring** or **shadowing**. Data is vertical redundantly to two identical physical disks that are treated as one logical disk. When data is read, in can be retrieved from the disk with shorter queuing, werk, and rearing a delays. If a disk talk, the other disk is used used the tax is repaired. Suppose the mean time to repair is **24** hours, then the mean time to data loss of a mirroried disk vector any 100 disks with vetter of 200,000 boars each is (200,000)/(2 < 24) = 6.13 × 10° loars, which is 95.028 years?<sup>11</sup> Disk mirroring also doubles the rate at which read focuses are houlded, since a read can go to either disk. The transfer time of each read backets that for a single disk.

Another solution to the problem of rebuilding to iso store extra information that is net inervally needed but that can be used to reconstruct the lost information in case of disk falure. The incorporation of redundance must consider two problems: (1) selecting a redundance for computing the redundant information, and (2) selecting is method of distributive the redundant information and (2) selecting is method of distributive the redundant information and (2) selecting is method of distributive the redundant information across the disk array. The first problem is addressed by using error conjecting codes modeling pointe ints, or specialized codes such as

Hamming codes. Under the parity scheme, a redundant disk may be considered as having the sum of all the data in the other disks. When a disk fails, the missing information can be constructed by a process socilar to subtraction.

For the second problem, the two major opproaches are either to store the redundant information on a small number of disks or to distribute it uniformly across all disks. The latter results in better load balancing. The different levels of RAD choose a combination of these options to implement redundancy and hence to improve rehability.

#### 13.10.2 Improving Performance with RAID

The disk arrays coupling the technique of data striping to achieve higher transfer rates. Note that data can be read or written only one block at a time, so a typical transfer contains 512 bytes. Disk striping may be applied at a time granularity by breaking up a byte of data into fits and spreading the bits to different disks. Thus, bit-level data striping consists of splitting a hyre of data and writing but (no the  $p^{th}$  disk. With 6 bit bytes, high physical disks may be considered as one logical disk with an ingotfold increase in the data transfer rate. Each disk participates in such  $\frac{1}{2}0^{3}$  request and the total amount of data read per request is either times as such. But level striping can be generalized to a number of disks that is either time tiple or a factor of eight. Thus, in a four disk array, bit is greet on the disk which is to mod 4).

The grandance of dotal interleaving can be higher than a bit; for example, blocks of a file can be supped across disks, giving rise to block-level striping. Figure 13.12 shows blocklevel data striping assuming the data file contrained four blocks. With block-level strip in multiple independent requests that access single blocks (small requests) can be servicely in parallel by separate disks, thus decreasing the queuing time of 1/0 requests. Requests that access multiple blocks (large request-) can be parallelized, thus reducing their response time is general, the more the number of disks in an array, the larger the potential performance beacht. However, assuming independent tabates, the disk array of 1/20 disks collectively has a 1/100<sup>th</sup>, the reliability of a single disk. Thus, redundancy was error-correcting codes and disk mirrining is necessary to provide reliability along with high performance.

### 13.10.3 RAID Organizations and Levels

Elifferent RAID organizations were denned based on different combinations of the two fators of granularity of dota interfeaving (attituing) and pattern used to compare reduction information. In the initial proposal, levels 1 through 5 of 8 viewere proposed, and we additioned levels 1-0 and 6—were added force.

IGAD level 0 uses data stripting, has no redundarit data and hence has the best acte performance since updates do not have to be duplicated. However, its read performance a not as good as RAD level 1, which uses mitrored disks. In the latter, performance improvement is possible by schedoling a read request to the disk with shortest expected seek and rotational delay. RAD level 2 uses memory-style redundancy by using Hamming codes, which contain parity bits for distinct sweedapping sub-ets of components. Thus it one particular version of this level, three redundancy disks affice for four original leak whereas, with minoring—as in level 1—four would be required. Level 2 meludes (with erar detection and correction, although detection is generally not required because moleculasks identify themselves.

8A.D level 3 uses a single parity disk relying on the disk controller to hence our which disk has failed. Levels 4 and 5 use block-level data striping, with level 5 distributing data and parity information across all disks. Finally, 3 AD level 6 applies the so-called P + Q relandancy scheme using Reed Soloman codes to protect against up to two disk follows by using just two redundant disks. The seven RMD levels (Q through 6) are illustrated just for 13.13 schematically.

Bebuilding in case of dok foilure is easiest for 8500 level 1. Other levels require the reconstruction of a failed disk by reaching multiple disks. Level 1 is used for cortical applications such as storing logs of transactions. Levels 3 and 5 are preferred for large orlane storage, with level 3 providing higher transfer rates. Most populat use of 8A01 reducing organized to the storage, with level 3 providing higher transfer rates. Most populat use of 8A01 reducing organized transfer rates. Most populat use of 8A01 reducing organized transfer rates. Most populat use of 8A01 reducing organized rate of a given application mits have to contour many deaget decisions such as the level of 8A05 the number of disks, the choice of yanty schemes, and grouping of disks we block-level stripting. Detailed performance statics in small teads and writes frefering in 00 requests for one stripting onth and large reas and writes frefering to 00 requests for one stripting disk in an error interction group) have been performed.

# 13.11 STORAGE AREA NETWORKS

With the rapid growth of electronic commerce, Epterprise Resource Planning (MC) as a envitar integrate application data across organizations, and data warehouses that keep invortical aggregate information (see Charter 27), the domainal for storage has gone up alguingially. For today's internet-driven organizations it has become necessary to move nere a static fixed data center oriented operation to a more dexable and dynamic infrastructure for their information processing reatmemeters. The total case of managing all but 5 go using so rapidly that in many instances the cost of managing server attached storge exceeds the cost of the server itself. Furthermore, the procursinent cost of storage simily a small fraction—typically only 12 to 15 percent of the overall cost of storage narragement. Many users of RAIP systems channel use the cupacity effectively because it his to be attached in a fixed manner to one or more servers. Therefore, large organizatensiare mesure to a concept called Storage Area Networks (SANs). In a BAN, on the aceage peripherals are configured as mades on a high-speed tietwork and can be attached aid detached from servers in a very flexible manner. Several comparities have emerged as SN providers and supply their own proprietary repologies. They allow storage systems to beplaced at longer distances from the servers and provide different performance and conneitivity options. Existing storage management applications can be potted into SAN conignations using Fiber Channel networks that encapsulate the legaty setsi protocol. As a issift, the SAN-attached devices appear as Stor, devices

Current architectural alternatives for SAN include the following point-topping analyticities between servers and storage systems via fiber channel, use of a liber channel.





Block-Interleaved Distribution Parity (RAID Level 5)



P+D Redundancy (RAID Level 6)

FIGURE 13.13 Multiple levels of KAID. From Chen. Lev. Gibson, Katz, and Patterson (1994). AC M Computing Sarvey, Vol. 26, No. 2 (how 1994). Reprinted with permission.

satech to connect multiple RAID systems, type libraries, errors asservers, use of fiber channel hubs and switches to connect servers and storage systems in different configurations. Digenturions can slowly mave up from simpler repologies to more complex ones by adding server- and storage devices as needed. We do not provide further details here because they say among vendors of SANs. The main advantages claimed are the following.

- Floxible many-to-many connectivity among servers and storage devices using fiber channel hubs and switches
- Units 10 km separation between a server and a storage system is ing appropriate fiber opin cables.
- Better isolation capabilities allowing inordisruptive addition of new peripherals and wivers

Was are growing very rapidly, but are still faced with numy problems such as combining amage options from multiple vendors and dealing with evolving standards of storage naragement software and hardware. Most major comparises are evolutione 5 NN as a star-Fe option for database storage.

## 13.12 SUMMARY

We began this chapter by discussing the characteristics of memory incratches and then concentrated on secondary storage devices. In particular, we focused on magnetic disks because they are used most effect to store online database files.

Data an disk is stored in blocks, accessing a disk block is expensive because of the stek time orational delay, and block transfer time. Double bottening can be used when accessing oracterize disk blocks, to reduce the average block access time. Other disk parameters are disorded in Appendix B. We presented different ways of storing records of a tile on disk beard-of a file are grouped into disk blocks and can be of fixed length or variable lengths sunce or unspanned, and of the same record type or mixed types. We discussed the file bade, which describes the record formats and keeps track of the disk addresses of the file stake. Information in the file header is used by system software accessing the file records.

We then presented a set of typical commands for occessing individual file records and necs-ed the concept of the current record of a file. We discussed how complex record such conditions are transformed into simple search conditions that are used to locate neods in the file.

Three principy file organizations were then discussed unordered, ordered, and holed. Unordered files require a linear search to locate records, but record insertion is any imple. We discussed the deletion problem and the use of deletion markers.

Ordered hies sharten the time required to read records in order of the ordering field. The time required to search for an arbitrary meord, given the value of its ordering key relats also reduced if a binary search is used. However, maintaining the records in order taks insertion virty repensive: thus the rechnique of using an unordered overflow file to while the rost of record insertion was discussed. Overflow records are morged with the mean rile periodically during file reorganization. klashing provides very fast access to an arbitrary record of a file, given the value of its hash key. The most satisfile method for exteroid bashing is the backet recharging, well one of more configured blocks corresponding to each backet. Collisions causing backet overflow are familied by channing. Access on any nonlinsh field is slow, and so is ordered access of the records on any field. We then discussed two hashing techniques for the shut grow and shunk in the number of records dynamically—namely, extendible and linear bashing.

We briefly discussed other possibilities for primary file organizations, such as B-tress, and files of mixed records, which implement relationships among records of different types physically as part of the storage structure. Finally, we reviewed the recent advances in disk technology represented by ICAID (Redundant Arcins of Inexpensive [Independent] Disks).

#### Review Questions

- [3.1] What is the difference between primary and secondary storage?
- 14.2 Why are disks, not tapes, used to store onl ne database files?
- 13.5 Define the following terms disk disk pack, took, block, cyloider sector, metšek gop read/urate facel.
- 13.4 Discuss the process of disk mutahration.
- 13.5 Discuss the mechanism used to read data from or write data to the disk
- 13.6. What are the components of a disk block address?
- 13.7. Why is accessing a disk block expensive? Docuse the time components involved intercessing a disk block.
- 13.8. Describe the inismatch between processor and disk technologies.
- 13.9. What are the main goals of the RA D technology? Have does it achieve them?
- 13.10. How does disk nurroring help improve relacibity? Give a quantitative example,
- 13.11. What are the techniques used to improve performance of doks in RA13
- 13.12. What characterizes the levels in SAID organization?
- 13-13. How does double buffering improve block necess time?
- 13.14. What are the reasons for human variable length records? What types of separate characters are needed for each?
- 13.15. Discuss the sechniques for allocating file blocks on disk.
- 13-16. What is the difference between a file organization and on access method?
- 13.17. What is the difference between static and dynamic riles:
- B.15. What are the typical record at a time operations for accessing a file? Which if these depend on the current record of a fle?
- 13-19. Discuss the techniques for record deletion.
- 13.20. Discuss the advantages and disact outages of using (a) an involved file. (b) an ordered file, and (c) a static hash file with backets and channels. Which operations can be performed efficiently on each of these organizations, and which specificients are expensive?
- 13.21 Discuss the techniques for allowing a hash file to expand and shrink dynamically What are the advantages and disadvantages of each?
- 13.22 What are mixed files used for? What are other types of primary tile organizations'

#### Exercises

- (5.3) Consider a disk with the following characteristics (these are not parameters of any particular disk unit): block star H = 512 bytes, interblock gap size G = 128 bytes, number of blocks per track = 20; number of tracks per variace = 400. A disk pack consists of 15 double sided disks.
  - a. What is the total capacity of a track, and when it is useful capacity (excluding methlock gops)?
  - b. How many evaluates are there?
  - c. What are the rotal capacity and the aseful capacity of a colluder?
  - d. What are the rotal capacity and the useful capacity of a disk pack?
  - 5. Suppose that the disk drive rotates the disk pack at a speed of 24d2 rpm (zevolutions per numute): what are the transfer rate (tr) in Sytes/Insec and the block transfer time (/ar) in insec? What is the average controlled delay (ral) in insec? What is the bulk transfer rate? (See Appendix B.)
  - <sup>2</sup> Suppose that the average seek time is 50 inser. How much time does it take (on the average) in insec to locate and transfer a single block, given its block address?
  - g. Calculate the average time it would take to transfer 10 random blocks, and compare this with the time it would take to transfer 20 consecutive blocks using double bufft mig to save seek time and rotational delay.
- D.34. A file basin = 20,000 station records of *fixet length*. Each record has the following fields have (30 bytes), see (9 bytes), account (9 bytes), records of (40 bytes), records (9 bytes), statebare (8 bytes), see (1 bytes), second each (4 bytes), records of (4 bytes), constraints (4 bytes), and accurate (3 bytes). An additional byte is used as indeletion marker. The blo is stored on the disk where parameters are given in Exercise 13.23.
  - a. Calculate the record are R in bytes.
  - Cubulate the blocking factor (in and the number of file blocks b, issuming an anspanned organization.
  - c. Colculate the average time is takes to find a record by doing a linear search on the file it (a) the file blocks are stored contiguously, and double boffering is used, (a) the file blocks are not stored contiguously.
  - Assume that the file is ordered by soy calculate the time tetakes to search for a record given its say value, by doing a binary search.
- - a. Calculate the average record length R in bytes.
  - e. Calculate the number of blocks needed for the tile

- 13.26. Suppose that a disk unit has the following potentieters seek time s = 20 msec: nontrough delay ref = 10 misec; block transfer time bit = 1 misec: block size 0 = 2430 bytes; interfalsek gap size G = 500 bytes. An EPROPER tile has the following fields (SV, 9 bytes; isstants, 20 betes: EIRSTanss, 20 bytes; within the hollowing fields betes; anderss, 35 bytes: more, 12 bytes; supervisionsse, 9 bytes; tEPARTHERT, 4 bytes toucher, 4 bytes: ifederationative: 1 byte. The EPROPER file has n = 30,000 microlic fixed-length format, and misparated blocking. Write appropriate formalis called culate the following values for the above EPROPER file:
  - a. The record size is functioning the deterion marker), the blocking factor by, and the number of disk blocks b.
  - Calculate the warred space in each disk block because of the important opatigation.
  - Coordare the transfer ture tr and the bulk transfer rate by for this disk one (see Appendix B for definitions of r and firs).
  - Clicculare the increase number of block accesses needed to search for an arbitrar record in the file using linear search.
  - c. Colculate in rased the recorder and needed to search for an arbitrary record in the file, using linear search, if the file blocks are stored on consecutive displocks and double buffering is used.
  - Calculate in insec the overage root needed to search for an arbitrary recording the ble, using linear search of the blocks are not stored on consecutive disblocks.
  - g. Assume that the records one ordered consome key tield. Calculate the average number of block accesses and the sumage time needed to search for an arbitrar record as the file, using binary search.
- 13.27. A PARTS file with Part# as bosh key includes records with the following Part# osues: 2569, 3760, 4692, 4871, 5659, 1821, 1074, 7115, 1620, 2428, 3945, 4750, 6975, 4981, 9208, The file uses eight backets, numbered 0 to 7. Each backets in one disk block and holds two records. Load these records into the file in the green order, using the bash function 9(K) = K mod 8. Calculate the overage minifer of block accesses for a random refueval on Part#.
- 13.25. Lead the records of Exercise 33.27 into expandable bash bles based on extendity hashing. Show the structure of the directory at each step, and the global and had depths. Use the hash function htK) = K mod 128.
- 13.29. Lond the records of Exercise 13.27 into an expandable bash the, using linear habing. Sourt with a single disk block, using the bash function by = K mod 2, and show how the file grows and how the bash functions change as the records are inserted. Assume that blocks are split whenever an overflow occurs, and show the value of a reach stree.
- (3.50) Compare the file commands listed in Section 13.6 to those available on a file accessmethod you are familiar with.
- [3.34] Suppose that we have an unordered file of fixed length records that uses an unspanned record organization. Outline algorithms for insertions delet so and modification of a file record. Store any assumptions you make.

- 10.12. Suppose that we have an indered tile of fixed-length records and an innuclered overflow file to bandle insertion. Both files use unspanned records. Online algorithms for insertion, deletion, and modification of a file record and for reorganizing the life. State any assumptions you make.
- (iii) Can you think of techniques other than an unordered overflow the that can be used to make invertions or an ordered file more efficient?
- 3.34. Suppose that we have a bash the of fixed-length records, and suppose that over flow is bandled by chaining. Outline algorithms for cusortion, deletion, and molification of a file record. Stars any assumptions you make.
- 13.35. Can you think of techniques other than chaining to handle bucker overflow in exernal hashing?
- Write pseudocode for the insertion algorithms for binear hashing and for extendible hashing.
- (6.3.) Write program code to access indicadual fields of records under each of the following circonnaturces. For each case, state the assumptions you make concerning pointers, separator characters, and so forth. Determine the type of information needed in the file header in order for your code to be general in each case.
  - a Fixed-length records with unspanned blocking-
  - b. Excellength records with spanned blocking.
  - Variable-length records with variable-length fields and spanned blocking
  - Variable-length records with repeating groups and spanned blocking.
  - e. Variable-length records with optional fields and spanned blocking.
  - Variable-length records that allow all three cases in parts cad, and ca
- 1535. Suppose that a file annually contains ( = 120,000 records of & = 200 bytes each in an unsorted sheap) file. The block size B = 2400 bytes, the average seek trace s = 16 ms, the average rotational latency bd = 8.3 ms and the block transfer trace by = 0.8 ms. Assume that 1 record is deleted for every 2 records added antil the total number of active seconds is 240,000.
  - a. How many block transfers are needed to teorgature the file?
  - E. How long does it take to find a record right before reorganization?
  - c. How long does it take to find a record sight after reorganization?
- 13.9. Suppose we have a sequential (ordered) ide of 100.000 records where each record is 240 bytes. Assume that B = 2402 bytes x = 16 ms, xd = 8.3 ms, and  $b\pi = 0.8$  ms. Suppose we want to make X independent tandom technic reads from the life. We could make X random bluck reads or we could perform one exhaustive read of the entire tile looking for these X records. The question is to decide when it would be more efficient to perform one exhaustive read of the entire tile than to perform X independent is reads. That is, what is the value for X when an exhaustive read of the file is more efficient than random X reads? Develop this as a function of X.
- 15.40. Suppose that a starte hash file anitially has 600 buckets in the primary area and that records are inserted that create an overflow area of 600 buckets. If we reorganite the hash file, we can assume that the overflow is eliminated. If the cost of reorganizing the file is the cost of the bucket transfers treading and verticing all of the buckets) and the only periodic file operation is the forch operation, then how

many times would we have to perform a fetch (successfully) to make the reagannation cost-effective? That is, the reorganization cost and subsequent search cost are less than the search cost before reorganization. Support your answer, Assume s = 16 ms, al = 8.3 ms, bt = 1 ms.

- 13.41. Suppose we want to create a linear hash file with a file lead factor of 0.7 and a block ing factor of 23 records per backet, which is to contain 112,000 records initially
  - a. How many buckets should we allocate in the primary area?
  - b. What should be the number of bits used for bucket addresses?

## Selected Bibliography

Wiederhold (1983) has a detailed discussion and analysis of secondary storage device and file organizations. Optical disks are described in Berg and Roth (1988) and analysed in Ford and Christodoulakis 1991. Hash memory is discussed by Diopert and Less (1987) Raemmier and Walkes (1994) present a survey of the magnetic-disk technology Slot rextbooks on databases include discussions of the material presented here. Most data structures restlysses, including Krinth (1973), discuss static hashing in musc leial. Knoth has a complete discussion of bash functions and collision resolution fechnicus, a well as of their performance companion. Knuth also otters a detailed discussion of takinquestion sorting external tiles. Textbooks on tile structures include Ukebreek (1981) Smith and Barnes (1987), and Subberg (1988); they discuss additional file organization including tree structured files, and have defailed algorithms for operations on files. Additrunal textbooks on the organizations include Miller (1987), and Lavadas (1989). Sabler et al. (1990) describina distributed external sorting algorithms. File organizations arriva high Jegree of fault tolerance are described by Bitten and Gray (1988) and by Gray et al (1992). Disk stripping is proposed in Saleni and Garria Mohua (1980). The first paperor redundant arrays of inexpensive disks (83.11) is by Parterson et al. (1988). Then as Petreson (1990) and the excellent survey of textD by Chem et al. (1994) are addinard references. Cirachowski and Hoyr (1996) discuss future trends in disk drives. Variau tre pillos for the 8 MD urder course appear in Chen et al. (1994).

Morris (1998) is statearly paper on history. Extendible history is described in Egn et al. (1979). Linear historie is described by Litwin (1962). Dynamic history, which we did not discuss in details was proposed by Larson (1976). There are many proposed sattions for extensible and linear hadoing for examples, see Cesarian and Soda (1991), D and Tong (1991), and cluchent and Berra (1992).

Details of disk stotage devices can be futeral at manufacturer sites egwww.seagate.com, wownlon.com, www.storagetek.com, IBM has a storage technologi research center at IBM Almaden (www.almadem.ibm.cem/sst/).

# Indexing Structures for Files

In this chapter, we assume that a file already exists with some primary organization such as the unordered, ordered, or bashed organizations that were described in Chapter 14. We will describe additional auxiliary access structures called indexes, which are used to specify the retrieval of records in response to certain search conditions. The index strucmestry is ally provide secondary access paths, which provide alternative ways of access us the incords without affecting the physical placement of records on disk. They enable affective access so records based on the indexing fields that are used to construct the rides. Basically, any field of the file can be used to create an index and bashifte indexes on bidrent fields can be constructed on the same file. A variety of indexes are possible: each other fields can be constructed on the same file. A variety of indexes are possible: each other fields can be constructed on the same file. A variety of indexes are possible: each other fields can be constructed on the same file. A variety of indexes are possible: each other fields can be constructed on the same file. A variety of indexes are possible: each other fields can be constructed on the same file access in diving field, one has to annually access the index, which primes to one or more blocks in the file where the required words are located. The most prevident opposite indexes are based on ordered files (anglelevel indexes) and tree flats structures (multilevel indexes are based on ordered files (anglelevel indexes) and tree flats structures (multilevel indexes are based on ordered files (anglelevel indexes) and tree flats structures (multilevel indexes. B'strees), indexes can also be avariated located on hashing or other search data analytics.

We dearable different types of single-level ordered indexes—primary secondary, and charming in in Section 14.1. By viewing a single-level index as an ordered file, one can levelop additional indexes for it, giving rise to the concept of multilevel indexes. A boolar indexing scheme called ISAM (Indexed Sequential Access Method) is based on the idea. We discuss multilevel indexes in Section 14.2. In Section 14.3 we describe Btice and B1-rices, which are data structures that are commonly used to DFMSs to implement dynamically changing multilevel indexes. B\*-mess have become a commonly accepted default structure for generating indexes on demand in most relational (1988). Section 14.4 is devoted to the alternative ways of accessing data based on a combination of multiple keys. In Section 14.5, we discuss how other data structure—such is hashing—can be used to construct indexes. We also briefly introduce the conceptor basical indexes, which give an additional level of indirection from physical indexes allowing for the physical index to be flexible and extensible in its organitation. Section 14.6 summatices the chapter.

## 14.1 TYPES OF SINGLE-LEVEL ORDERED INDEXES

The idea behind in ardiered index access structure is structure to that behind the index net in a textbook, which lists important terms at the end of the book, we can search a index to find a list of gage numbers where the term appears in the book. We can search a index to find a list of addresses spage numbers in this case—and use these addresses a locate a term in the rectbook by searching the specified pages. The alternative, it mothe guidance is given, would be to sitt alowly through the whole textbook word by word to find the term we are interested in: this corresponds to doing a linear search on a ble. Of course, most books de furce additional information, such as chapter and vector tide that can help as trud a term without having to search through the whole book. However, the index is the only exact guidentiation of where each form occurs on the book.

For a file with a given record structure consisting of soveral fields (or errobuse), an index access structure is usually defined on a single field of a file, called ar, **indexing fiel** (or **indexing attributy**). The index typically stores each value of the index field along with a list of pointers to all disk blocks that contain records with that field caller. The values in the index are ordered so that we can do a binary search on the index. The **adv** file is much smaller than the data file, so searching the index using a binary search is reasonably efficient. Multilevel indexing (see Section 14.2) does away with the read to a binary search at the expert-e of creating indexes to the index itself.

There are several types of ordered indexes. A primary index is specified on the ordering key field of an ordered file of records. Recall from Section 13.7 that an ordered two determines are ordered on the file or devices on disk, and every record has a super calse for that field. If the ordered field is not a key field—that is, if numerous records a the file can have the same value for the ordering field—another type of index, catcher physical ordering field. So it can have at most one primary index or one clastering index, fail to the order primary index or one clastering index, induce the secondary index, can be specified at the two records a third type of index, catcher primary index or one clastering index, induction to tab. A third type of index, catcher a secondary index, can be specified enough each of the order of the true types of supervised and the file on have several secondary index or multition to its primary access including in the rest three subsections we discuss these three types of single-level acloses.
#### 14.1.1 Primary Indexes

A primary index is an ordered ble whose records are of tixed length with two fields. The first field is of the same data type is the ordering key field—colled the **primary** key—of the data file, and the second field is a pointer to a disk block (a block address). There is not index entry (or index record) in the index file for each block in the data file. Each index entry has the values of the primary key hold for the first record in a block and a pointer to that block as its two field values. We will retur to the two field values of index entry it as  $\langle K_i \rangle$ , F(i) >.

To create a primitry index on the ordered his shown in Figure 13.7, we use the vore tables primary key, because that is the ordering key field of the file (assuming flut each ende of wore is unique). Each entry in the index has a sore value and a pointer. The first frice index entries are as follows.

<K(1) = (Aaron,Ed), P(1) = address of block 1><K(2) ~ (Adams,Juhn), P(2) = address of block 2><K(3) = (Alexander,Ed), P(3) = address of block 3>

Figure 14.1 dlustrates this primary index. The total number of entries in the index is the same as the monter of dok blocks in the ordered data file. The first record in each block of the data file is called the **anchor** record of the block, or amply the block anchor.<sup>2</sup>

Indexes can also be characterized as dense or sparse. A **dense index** has an index entry for every sizefikey rulise (and hence even) record) in the data file. A **sparse** (or **nondense) index**, or the other hand, has index entries for only some of the search values. A primary index is hence a randomse (sparse) index, since it includes an entry for each disk block of the data file and the keys of its encourt record rather than for every search value (onevery recerd).

The index file for a primary index needs sub-tannally fewer blocks than does the data fiel for two reasons. First, there are fewer block entries than there are records in the data free Second, each index entry is (spinally smalles at the than a data record because it has only two fields; consequently, more index entries than data records can fit in one block. A binary search init the index file hence requires fewer block accesses than a binary search relate the data file. Reference back to Table 13.2, note that the binary search for an ordered data file required leggly block accesses. But if the primary index file contains by blocks, heatto locate or record with a search key value requires a binary search of that index and access to the block containing that records in total of log/by +1 accesses.

A record whose primary key value is K host in the block whose address is P(t), where  $K(t) \leq K \leq K(t+1)$ . The toblack in the data the contains all such records because of the physical ordering of the file records on the primary key field. To intruve a record, given the value K of its primary key field, we do a briary search on the index file to find the appropriate index entry  $t_0$  and then remove the data file block whose address is P(t).

<sup>2.</sup> We can not a scheme similar to the one described have, with the last record in each block (rather dat the first ) as the block muchar. This sightly, improves the efficiency of the sensitial approxim.

<sup>&</sup>lt;sup>1</sup> horize that the above formula would get be correct. (the state file were ordered on a backet plut in fact case the same index value of the block and on goald be repeated in the last records of the periods block.)



FIGURE 14.1 Primary index on the ordering key field of the file shown in Figure 13.7

beaugle Lillustrates the saving in block accesses that is attainable when a primary index is used to search for a record.

**EXAMPLE** (a Suppose that we have an ordered file with 1 = 30.000 records stored on a disk with block size B = 1024 bares. Edd records are of fixed size and are unsparated with record length R = 100 bares. The blocking factor for the file would be bit =  $\frac{1}{2}(B/R)^2 = (1004(100))^2 = 10^2$  records per block. The number of blocks needed for the file is b =  $\frac{1}{2}(b/R)^2 = 1(30.0007(10))^2 = 30000$  blocks. A binary search on the data file would need approximately flog b [ =  $\frac{1}{2}(\log 3000)^2 = -12$  block accesses.

Now suppose that the ordering key field of the file is V = 9 bytes long, a block pointer sf(=6) bytes long, and we have constructed a primary index for the file. The size of each index entry is  $R_1 = (9 - 6) = 15$  bytes, so the blocking factor for the index is  $bf_0 = (10)R_1 [] = [10024/15) [] = 68$  entries per block. The total number of undex entries  $r_0$  is ecal to the number of blocks in the data tile, which is 3000. The number of index blocks is hence  $b_0 = [(r_0/br_0)] = [(3002068)^2 = 45$  blocks. To perform a binary search on the index file would need  $[(3002068)^2 = ((log_0/455) + 6) block accesses to search for a record number of clock, we need one additional file k access to the data file four total of <math>6 - 1 = 7$  block accesses—an improvement over binary search on the data file, which required 12 block accesses.

A major problem with a primary index—as with any ordered life—is insertion and deletion of records. With a primary index, the problem is compounded because, if we attempt to insert a record in its correct position in the data tile, we have to not only move pearls to make space for the new record but also change some index entries, since noting records will change the unchor records of some blocks. Using an unordered outfearfile, as discussed in Section 13.7, can reduce this problem. Another possibility is pose winked list of overflow records for each block mathedata the. This is somith to the mained of dealing with overflow records described with hashing in Section 13.8.2, desords within each block and its overflow linked list can be sorted to improve retrieval time. Record deletion is handled using deletion markers.

#### 14.1.2 Clustering Indexes

Irrecords of a file are physically ordered on a runkey field—which downor have a dotinet take for each record — that field is called the **clustering field**. We can create a different ups of index, called a **clustering index**, to speed up removal of records that have the size value for the clustering field. This datters from a primary index, which requires that ilesidening field of the data file have a distinct take for each record.

A clustering index is use an ordered file with two fields; the first field is of the same type as the clustering field of the data file, and the second field is a block pointer. There is its entry in the clustering index for each distort while of the clustering field, containing the value and a pointer to the *first* block in the data file that has a record with that value to its clustering field. Figure 14.2 shows an example. Notice that record insertion and deletan still cause problems, because the data records are physically ordered. To alley are the problem of insertion, it is common to reserve a whole block for a cluster of contiguous block) for each edge of the clustering field, all records with that value are placed in the



DATA HILE

FIGURE 14.2 A clustering index on the prometer ordering bookey field of an record file.

block (or block cluster). This makes insertion and deletion relatively straightforward Figure 14.3 shows this scheme.

A clustering index is another example of a nonderise index, because it has an enter for every distinct radie in the indexing held which is a routkey by definition and hence has displicate values rather than the every record in the file. There is some similarity between Figures 14.1 to 14.3, on the one hand, and Figure 13.11, on the other. An index is somewhat similar to the directory structures used for extendible hashing, described in Section 13.8.3. Both are searched to find a pointer to the data Flock corraining the



**FIGURE 14.3** Clustering index with a separate bluck cluster for each group or records that state the same value for the clustering tield.

desired record. A main difference is that an index search uses the values of the search field itself, whereas a hash directory search uses the bash value that is calculated by applying the bash function to the search field.

#### 14.1.3 Secondary Indexes

A secondary index convides a secondary means of accessing a file for which some primaraccess already exists. The secondary index may be on a field which is a candidate key and has a unique value in every record, or a number with dopheate values. The index is an ordered file with two fields. The first field is of the same data type as some moundaring field of the data file that is an indexing field. The second field is of their index in reveal pointer. There can be using second any indexes (and hence, indexing fields) for the same file pointer. There can be using second any indexes (and hence, indexing fields) for the same file

We first consider a secondary index access structure on a key field that has a dwarg rathe for every record. Such a field is sometimes called a secondary key. In this case there is one index entry for each record in the data file, which contains the color of the secondary key for the record and a pointer either to the block in which the record is stored or to the record itself. Hence, such an index is dense.

We again refer to the two field values of index entry if as  $\langle K(i), F(i) \rangle$ . The entries as ordered by value of K(i) so we can perform a binary search. Because the records of the data file are not physically ordered by values of the secondary key field, we connectuse block and new That is why an index entry is created for each record in the data file, tather than for each block as in the vase of a primary index. Figure 14.4 illustrates a secondary index in which the primare P(i) in the index entries are block pointery, not record pointers. Once the appropriate block is many entry a search for the desired record within the block carbox.

A secondary index usually needs more storage space and longer south time than 3 year primary index, because of its larger number of entries. However, the improvement in search time for an arbitrary record is much greater for a secondary index than for a primary index, since we would have to do a interface of the data file if the secondary index did not easily. For a primary index, we could still use a binary search on the main file, even if the index did not exist. Example 2 illustrates the improvement in number of blocks accessed.

**EXAMPLE 2:** Consider the file of Example 1 with  $\tau = 30,000$  fixed-length records of size R = 100 lottes stored on a disk with Flock size B = 1024 bytes. The tile has b = 300 blocks, as calculated in Example 1. To do a hieror sourch on the file, we would require b? = 3000/2 = 1500 block accesses on the average. Suppose that we construct a secondary index on a nonordering key field of the file that is V = 9 bytes long. As in Example 1, a block pointer to T = 0 bytes long, so each index entry is  $R_1 = (9 + 6) + 15$  block in the blocking factor for the index is block in the file that is V = 9 bytes long. As in Example 1, a block pointer to T = 0 bytes long, so each index entry is  $R_1 = (9 + 6) + 15$  block in the blocking factor for the index is block is  $E_{10}(B/R_1) = 1(1024/15) = 55$  entries for block in a dense secondary index such as this, the total momber of index entries  $r_1$  is equal to the nonline of process in the dota hie, which is 50,000. The number of blocks needed for the index is block by  $E_1(R_1) = 1(1024/15) = 10000$  blocks needed for the index is block by  $E_2(B/R_2) = 10000$ .

A bittary search on this secondary index needs  $[(\log_2 b_1)^2 = ((\log_2 442)^2] + 3 block accesses. To search for a record using the index, we need an additional block access to the data life for a total of <math>9 + 1 - 10$  block accesses—a vest improvement over the 1500 block accesses needed on the average for a linear search, but slightly worse than the secon block accesses required for the primary index.



DATAFILE

FIGURE 14.4 A dense secondary index twith block opinities) on a nonordering key field of a file

We can also create a secondary index on a number field of a file. In this case, numerous avoids in the data file can have the same value for the indexing field. There are several groups for implementing soft on index.

 Option 1 is to include several index entries with the same K(i) value – one for each record. This would be a dense index.

- Option 2 is to have viriable length records for the index entries, with a repeating field for the pointer. We keep a list of pointers <P(i,1), . . . . P(i,k)> is the index entry for X(i)—one pointer to each block that contains a record whose indexing field value equals K(i). In either option 1 or option 2, the binary search algorithm entries index must be modified appropriately.
- Option 3: which is more commonly used, is to keep the index entries there elves any fixed length and have a single entry for each index field odd, but to create an exital level of indirection to handle the multiple pointers. In this nondecise scheme, the pointer PCF in index entry <KGE PGS> points to a block of record pointers, each record pointers in that block points to one of the data file records with value KGF for the indexing field. If some value K(i) occurs in too many records, so that their record pointers cannot fit in a single disk block, a cluster or block requires one or more additional block accesses because of the extra level, but the adjoint has to searching the index and finance importantly for inserting of new records in the data file are straightfreeward. In addition, retrievals on complex selection conditions may be handled by referring to the record pointers, without having to retrieve trany university.

Notice that a secondary index provides a **logical ordering** on the records by the indexing field. If we across the records in order at the entries in the secondary index, we get them in order of the indexing hold.

#### 14.1.4 Summary

To conclude this section, we summarize the discussion on index types in two tables. Table 14.1 shows the index field characteristics of each type of ordered single-level itidex discussed—primary, clustering, and secondary. Table 14.2 summarizes the properties of each type of index by comparing the number of index entries and specifying which indexes are dense and which use block anchors of the data file.

### **14.2 MULTILEVEL INDEXES**

The indexing schemes we have described thus tar involve an ordered index file. A binary search is applied in the index to lucate pointers to a disk block or to a record (at records) in the file having a specific index field value. A binary search inquires approximately (log b) block accesses for an order with b, blocks, because each stop of the algorithm reduces the part of the index file (but we continue to search by a factor of 2. This is whe we take the log function to the base 2. The idea behand a **multilevel** index is to reduce the part of the index due to search by block accesses that as economic to storth by blr, the blocking locar for the index which is larger than 2. Hence, the search space is reduced much faster. The value blr, is called the **fan-out** of the multilevel index, and we will sefer to 0 by the synthelito bearching a multilevel index (equites approximately (log, b) block accesses, which is a smaller number than for binary search of the fan-sout is larger than 2.



DATA FILE

FIGURE 14.5 A secondary index (with record printers) on a nonkey field implemented using one level of indirection so that index entries are of fixed length and have unique field values.

|                          | INDEX FIELD USED FOR<br>Ordening, the File | INDEX FIELD NOT USED FOR<br>ORDERING THE FILE |
|--------------------------|--------------------------------------------|-----------------------------------------------|
| Indexing field is key    | Primary index                              | Secondary index (Key)                         |
| Indexing field is nonkey | Clustering index                           | Secondary index (NonKey)                      |

A multilevel index considers the index (ik, which we will now refer to as the first (or **base**) level of a multilevel index, as an indexed file with a distort whe for each K(i). Hence we can create a primary index for the first level, this index to the first level is

| NUMBER OF (FIRST-LEVER)<br>IN DEX ENTRIES                            | DENSE OR<br>NONDEASE                                                                                                                                                                                                 | BLOCK ANCHORING ON<br>THE DATA FILE                                                                                                                                                                                                                                                                                                                                             |
|----------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Number of Nocks of<br>data ide                                       | Nonderisc                                                                                                                                                                                                            | Yes                                                                                                                                                                                                                                                                                                                                                                             |
| Number of distinct index<br>field values                             | Nordense                                                                                                                                                                                                             | Yeshoʻ                                                                                                                                                                                                                                                                                                                                                                          |
| Number of recordant.<br>Jara file                                    | Dense                                                                                                                                                                                                                | No                                                                                                                                                                                                                                                                                                                                                                              |
| Namber of records for<br>Namber of <u>distorct index held</u> values | Depse or<br><u>Nondens</u> e                                                                                                                                                                                         | No                                                                                                                                                                                                                                                                                                                                                                              |
|                                                                      | NUMBER OF (FIRST-LEVER)<br>IN DEX ENTRIES<br>Number of blocks on<br>data file<br>Number of distinct index<br>field values<br>Number of records for<br>Number of records for<br>Number of distinct index field values | NUMBER OF (FIRST-LEVEL)     DENSE OR<br>NONDER-SE       IN DEX ENTRIES     NONDER-SE       Number of blocks on<br>data file     Nonder-se       Number of distinct index     Nonder-se       Eold values     Dense       Jata file     Dense       Number of records in:     Dense       Jata file     Dense       Number of records for     Dense       Jata file     Nonderse |

#### TABLE 14.2 PROPERTIES OF INDEX TYPES

every distanct value of the ordering field starts a new block, passive to

For prion L

Recognitions 2 and 3.

called the second level of the multilevel index. Eccause the second level is a primary index, we can use block include so that the second level has one entry for each block of the first level. The blocking factor bfr, for the second level- and for all subsequent levels-is the same as that for the first-level index, because all index curres are the same size: each has one field value and one black address. It the first level has  $\tau_{\rm f}$  entries, and the blocking factor—which is also the tan-out—for the index is bit, = fo, then the first level needs  $(r_0/b_0)$  - blocks, which is therefore the number of entries  $r_0$  needed at the second level of the index

We can repeat this process for the second level. The **third level**, which is a primaindex for the second level, has an entry for each second-leve, block, so the number of third-level entries is  $a_i = -(r_i b_i a_i)$ . Notice that we require a second level only if the two level needs more than one block of disk storage, and, similarly, we require a third level only if the second level needs more than one block. We can report the preceding process until all the entries of some index level tint in a single block. This block at the P-levens called the top index level 4 Each level reduces the number of entries at the previous level by a factor of to—the index tan-out—to we can use the formula  $1 \leq (r_0(Rn)^2)$  to calculate ). Hence, a multilevel index with refinite level entries will have approximately ( levels, where  $t = -\langle \log_{10}(r_1|0) \rangle$ .

The multilevel scheme described here can be used on any type of index, whether it is primary, clustering, or secondary - is long as the first devel index has distinct values for K(i) and fixed-tength comes. Figure 14.6 shows a multilevel index bath over a primary index Example 3 illustrates the improvement in number of blocks accessed when a multilevel index is used to search for a record.

<sup>4.</sup> The northlyping scheme for order levels used here is the reservent the way evels are second denied for the data structures by the data structures, this telepish in as level ("itzero), t = i is level l. ett



**FIGURE 14.6** A two-level primary index resembling isom (Indexed Sequential Access Method) ogsnization.

**EXAMPLE** 37 Suggroup that the dense secondary index of Example 2 is converted into a multilevel index. We calculated the index blocking factor block = 68 index entries per block, which is also the harmour to for the multilevel index; the number of fast level blocks  $b_1 = 442$  blocks was also calculated. The number of second-level blocks will be  $b_2 = \frac{1}{2}(b_1 \hbar t^2) = 1$  (442/58)  $\frac{1}{2} = 7$  blocks, and the number of third-level blocks will be  $b_2 = \frac{1}{2}(b_2/5)^2 = 1$  blocks. Hence, the fit is the top becket of the index and (-3) To access a record by searching the multilevel index, we must access one block at (-3).

each level plus one block from the data file, so we need t = 1 = 3 + 1 = 4 block accesse. Compare this to Example 2, where 10 block accesses were needed when a single-level index and binary search were used.

Notice that we could also have a multilevel primary index, which would be nondense. Exercise 14.14(c) illustrates this case, where we wast access the data fled from the file before we can determine whether the record being searched for is in the file For a dense index, this can be determined by accessing the first index level (without having to access a data block), since there is an index entry for energy record in the file.

A common file organization used in biscuess data processing is an ordered file with a nultilevel primary index on its ordering key field. Such an organization is called as **indexed sequential** file and was used in a large number of early 1981 systems. In services bandled by some form of overflow file that is juerged periodically with the data file. The index is recreated during file reorganization. (BMS) ISAM organization incorporates a two level index that is closely related to the organization of she disk. The first level is a cylinder index, which has the key value of an one-hor record for each cybrider of adds pock and a painter to the track index for the cylinder. The track index that the key cate of an anchor record for each track in the cylinder and a painter to the track. The index can then be searched sequentially for the desired record or block.

Algorithm 14.1 outfines the search procedure for a record in a data file that uses a nondense multi-evel primary index with thevels. We refer to entry if at level j of the index as  $\langle \mathbf{K}_j(t), \mathbf{P}_j(t) \rangle_{ij}$  and we search for a second whose primary key value is **K**. We assume that any overflow records are ignored, if the record is in the file, there must be some entry at level 1 with  $\mathbf{K}_j(t) \cong \mathbf{K} \leq \mathbf{K}_j(t+1)$  and the record will be in the block of the data de whose address is  $\mathbf{P}_j(t)$ . Even use 14-19 discusses modifying the search algorithm for othe types of indexes.

Algorithm 14.1: Searching a nondense nudulevel primary index with r levels

```
p + address of too level block of index;
for j <= r step = 1 to 1 do
 tegin
 read the index block (at jth index [rvel]) whose address is p;
 smarch block p for entry i such that \kappa_i(i) \neq \kappa, \kappa_j(i + 1) (if \kappa_i(i)
 is the last entry in the block, it is sufficient to satisfy \kappa_i(i) \neq \kappa);
 p \leftarrow P_1(i) (* procks appropriate pointer at p^{(1)} index level *)
 end.
 read the data file block whose address is p.
 search block p for record with key = K;
```

As we have seen, a multilevel index reduces the number of blocks accessed what searching for a record, given its indexing field value. We are still faced with the problem of dealing with index insertions and deletions, because all index levels are physical ordered bloc. To recom the benefits of using nulliflevel indexing while reducing index insertion and deletion problems, designer sudopted a multilevel index that leaves sing space in each of its blocks for inserting new entries. This is called a dynamic multilevelindex and is often implemented by using data structures called B trees and B1-trees, when we describe its the next section.

# 14.3 DYNAMIC MULTILEVEL INDEXES USING B-TREES AND B\*-TREES

Betrees and B'strees are special cases of the well-known tree data structure. We introduce vary birely the terminology used in discussing rice data structures. A tree is formed of nodes. Each node in the tree, except for a special node called the **root**, has one parent node and several—zero or more—child nodes. The root node has no parent. A node that does not have any child nodes is called a leaf male, a nortical node is called an **internal** node. The level of a node is always one more than the level of its parent, with the level of the toot node being zero.<sup>5</sup> A subtree of a code consists of that node and all its descendant index—its child nodes, the child nodes in its child nodes, and so on the precise recurate kfinition of a subtree is that it consists of a node in and the subtrees of all the child nodes of it. Figure 14.7 illustrates a tree data structure. In this figure the root male is A, and its chilf nodes are B. C, and D. Nodes E. J. C, G, H, and K are leaf nodes.

Usually, we display a tree with the root node at the top, as shown in Figure 14.7. One way to implement in the risk to have as many pointets in tack much as there are child nodes



(notes EUIC/3/-4, and K are leaf notes of the stee,

FIGURE 14.7 A tree data structure that shows an unbalanced tree

 This standard definition of the level of a tree node, which we are throughout Section 14.3, which we at test the surface gave to any different indexes in Section 14.2. ut that mule. In some cases, a parent pointer is also stored in each nude. In addition ta pointers, a node usually contains some kind of stored information. When a multilevel index is implemented as a tree structure, this information includes the values of the lide's indexing held that are used to guide the search for a particular record.

In Section 14.3.1, we introduce search rises and then discuss B-trees, which can be used as dynamic multilevel indexes to goade the search for records in a data file. Forces nodes are kept between 50 and 100 percentifull, and penners in the data blocks are stored in both internal nodes and leaf nodes of the B-inter-structure. In Section 14.3.2 we discuss  $B^{+}$ -trees, a variation of B-inters in which penners to the data blocks of a file are stored only in leaf nodes; this can lead to lever levels and hagher-capacity indexes.

#### 14.3.1 Search Trees and B-Trees

A search tree is a special type of tree that is used to guide the search for a record, given the value of one of the record's fields. The multilevel indexes discussed in Section 14.2 can be thought of as a variation of a search tree, each node in the multilevel index can have as many as following and folkey values, where it is the index for-out. The index field values in each node rande us to the next node, until we reach the data file block that contains the required records. By following a pointer, we restrict our search at each level to a subtree of the search tree and ignore all nodes not in this subtree.

Search Trees. A search rise is slightly different from a underlevel index. A search tree of order p is a tree such that each node contains a most p = 1 search values and p pointers in the order  $\leq P_1$ ,  $K_1$ ,  $P_2$ ,  $K_3 = \dots$ ,  $P_{p+1}$ ,  $K_{p+1}$ ,  $P_1 > \infty$  where  $q \leq p$ , each  $P_1$  is a pointer to a child node for a null pointer), and each  $K_1$  is a search value from some ordered set of colors. All search values are assumed to be unique." Figure 14-8 illustrates a node in a search true. Two constraints must hold of all times on the search tree.

1. Within each node,  $K_{j} \leq K_{s} \leq ... \leq K_{a+1}$ 



FIGURE 14.8 A node in a scarch free with pnimers to subtrees below a:

6. This restriction can be reloved, if the tildex is on a nonkey field dopheate search values now exist and the mode stutentize and the navigation rules for the tree may be machined.

For all values X in the subtree pointed at by P<sub>0</sub> we have K<sub>1,1</sub> ≤ X ≤ K for 1 ≤ i ≤ s<sub>0</sub> X ≤ K for 1 ≤ 1; and K<sub>1</sub> ≤ X for i = q fsee Figure 14.6).

Whenever we search for a value  $\lambda$  ove follow the appropriate pointer P, according to the formulas in condition 2 above. Figure 14.9 illustrates a search tree of order  $\tau = 3$  and integer search values. Notice that some of the pointers P, in a node may be null pointers.

We can use a search tree as a mechanism to search for seconds stored in a disk file. The values in the mee can be the values of one of the fields of the file, called the **search** field (which is the same as the index field of a multilevel index guides the search). Each set value in the receipt associated with a pointer to the record in the data file having that used. Attem invelve the pointer could be to the disk block containing that record. The search tree itself can be stored on disk by asterning each tree node to a disk block. When a new record is inserted, we must update the search tree by inserting an entry in the tree optimizer of the new record.

Algorithms are necessary for insuring and deleting search values into and from the south the while maintaining the proceeding two constraints. In general, these algorithms do not guarantee that is south tree is **balanced**, meaning that all of its leaf nedes are at the some level.<sup>7</sup> The tree in Figure 14.7 is not balanced because it has leaf nedes at levels 1, 2, and 3. Keeping a search tree balanced is important because it guarantees that no nodes will be at very high levels and hence require many block increases during a tree search. Keeping die tree balanced yields a uniform search speed regardless of the value of the search key. Another problem with search trees is that record deletion may leave some moles in the tree rearly empty, thus wasting stenage space and increasing the number of Levels. The B-tace address both of these problems by specifying additional constraints on the search tree address both of these problems by specifying additional constraints on the search tree address both of these problems by specifying additional constraints and the search tree is addressed by a specifying additional constraints and the search tree is both of these problems by specifying additional constraints and the search tree is addressed by a specifying additional constraints and the search tree is addressed by a specifying additional constraints and the search tree is addressed by a specifying additional constraints and the search tree is a specifying additional constraints and the search tree is a specifying additional constraints and the search tree is a specifying additional constraints and the search tree is addressed by a specifying additional constraints and the search tree is a specifying additional constraints and the specifying aditional constraints and the spe

8-BOOS. The Bittee has additional constraints that ensire that the tree is always balanced and that the space wasted by deletion, if any, never becomes excessive. The



**RGDRE 14.9** A search free of order p = 3.

The definition of Againcrobic pattern for binary trees. Balances Honory investories for each statistics.

olgorithms for insertion and deletion, though, become more complex in order to maintain these constraints. Nonetheless, most invertions and deletions are simple processes, they become complicated only index special circumstances—namely, whenever we attempt at insertion into a node that is already full or a deletion from a node that makes it less than half full. More formally, n **B-tree** of **order p**, when used as an access structure on a key field to search for records in a data file, can be defined as follows:

1. Each internal node in the B-free (Figure 14.10a) is of the term

$$<\mathbb{P}_1,<\mathbb{N}_1,\mathbb{P}_2>,\mathbb{P}_2,<\mathbb{N}_2,\mathbb{P}_2>,\ldots,<\mathbb{N}_{n+1},\mathbb{P}_{n+1}>,\mathbb{P}_{n+1}>$$

where  $q \simeq p$ . Each P, as a **tree pointer**-to a pointer to another node in the Batree Each Pr, is a **data pointer**<sup>8</sup>—a pointer to the receptionlose search key field value is equal to K<sub>i</sub> (or to the data the block containing that recept).

- Within each node, K<sub>1</sub> ≤ K<sub>1</sub> ≤ ∞ ≤ K<sub>n-1</sub>.
- For all search key field values X in the subtree pointed at by P<sub>1</sub> (the i<sup>th</sup> subtree, see Figure 14 10a), we have:

$$K_{i,i} \le X \le K_i$$
 for  $1 \le i \le q$ ,  $X \le K$  for  $i = 1$ ; and  $K_{i+i} \le X$  for  $i = 0$ .

- 4. Each nucle has at most price pointers
- Each roado, except the root and leaf nodes, has at least U(p/23), tree pointers. The mut node has at least two tree pointers index it is the only node in the tree.
- A mode with q true pointers, q ≈ p, has q = 1 search key field values (and hence have = 1 data pointers).
- All leaf nodes are of the same level. Leaf nodes have the same structure as internol nodes except that all of the name jointees P<sub>i</sub> are null.

Figure 14 13b dlustrates a B-most of order p = 3. Notice that all search values K in the B-most are unique because we assumed that the tree is used as an access structure on a key field. If we use a B-tree in a *number* field, we must change the dematrion of the file pointers  $P_1$  to point to a block—or cluster of blocks—diat contain the pointers to the file records. This extra level of indirection is similar to Option 4, discussed in Section 14.13, for secondary indexes.

A Borec states with a single root node (which is also a leaf node) at level 2 (contribute the neutronode is full with p = 1 search, key values and we attempt to insert another entry in the tree, the root node splits into two nodes at level 1. Usily the middle values kept in the root node, and the test of the values are split evenly between the other methods. When a nonroot node is full and a new entry is inserted into (1) that node is split into two nodes at the values are split even when node, and the rest of the values are split evenly between the other methods. When a nonroot node is full and a new entry is inserted into (1) that node is split into two nodes at the same level, and the middle entry is inserted into (1) that node along with two pointers to the new split nodes. If the parent node is full, it is also split. Splitting can propagate all the way to the not node, creatize a new level of the tool is split. We do not discuss algorithms for Perces in detail, here, tather, we earline search and insertion procedures for Perces in the next section.

 A data pointer is either a block address so a record address the letter is essentially a block address and a record offser within the block.



**EXAMPLE 14.10** B-tree structures (a) A node m a B-tree with q = 1 search values, b/A B-tree of order p = 3. The values were inserted in the order 8, 5, 1, 7, 3, 12, 9, 6.

If deletion of a value causes a node to be less than balf full, it is combined with its needboring nodes, and this can also propagate all the way to the noor. Hence, deletion can reduce the number of tree levels, it has been shown by analysis and simulation that iter numerous random insertions and deletions on a B-tree, the nodes, or approximately whereast full when the number of values in the tree subdites. This is also true of B'ales. If this happens, node splitting and combining will occur only tarely so insertion and deletion become quite efficient. If the number of values may occur, so some insertions will take inter time, the number of nodes may occur, so some insertions will take inter time. Example 4 illustrates how we calculate the order p of a B-tree stored on disk.

**EXAMPLE 4:** Suppose the search field is V = 9 bytes long, the disk block securis B = 512 eves, a record (data) pairments  $P_1 = 7$  bytes, and a block pointer is P = 6 bytes. Each Each enders, and can have at most pittee pointers, p = 1 data pointers, and p = 7 search key field values (see Figure 14.(2a). These point of into a single disk block it each B-tree male is to arrespond to a disk block. Hence, we must have:

 $\begin{array}{l} (p^*(P) + ((p + 1)^*(P_1 + V)) \cong B \\ (p^*(e) + ((p + 1)^*(7 + 9)) \cong 512 \\ (22^*(e) \cong 526 \end{array}$ 

We can choose p to be a large value that satisfies the above inequality, which gives p = 23, p = 24 is not chosen because of the reasons either next).

In general, a Barree node may contain additional information needed by the algotithms that manipulate the tree, such as the number of entries of in the node and a parateto the parent node. Hence, before we do the preceding calculation for p, we shall reduce the block size by the amount of space needed for all such information. Next, we dilustrate how to calculate the number of blocks and levels for a B-tree.

**EXAMPLE 5:** Suppose that the search field of Example 4 is a nonordering key field and we construct a B-tree on this field. Assume that each node of the B-tree is 69 percent field and hence, on the average, will have  $p \approx C.69 = 23 \pm 0.69$  or approximately 16 pointer, and hence, 15 search key field values. The **average fan-out** to =10. We can start at the recet and see hervinanty values and pointers can exist, on the average, at each subsequent level:

| Runt:    | L node     | 15 entries     | 16 pointers   |
|----------|------------|----------------|---------------|
| Level 1: | Do noch s  | 240 entries    | 256 pointers  |
| Level 2: | 256 works  | 3840 entries   | 4090 pointers |
| Lend 3:  | 4096 nodes | 61,440 entries |               |

At each level, we calculated the number of entries by multiplying the rotal number of pointers at the previous level by 15, the average number of entries in each node. Hence, for the given block size, pointer size, and search key field size, a rota-level Patree hold 3540 - 240 + 45 = 4095 entries on the average, a three-level Battee boals 65.535 entries on the average.

B-trees are sometimes used as primary file organizations. In this case, whole records are stored within the B-tree nodes rather than just the </search key, record pointer </search This works well for files with a relatively solali manuar of records, and a small record key. Otherwise, the fan-out and the number of levels become too great to gettrat efficient access

In summary, B-trees provide a multilevel access structure that is a balanced trestructure in which each node is at least half full. Each node in a B-tree of order p can have at most p-1 search values.

#### 14.3.2 B\*-Trees

Most implementations of a dynamic multilevel index use a variation of the Barree data structure called a **B**'stree. In a Barree, every value of the search field appears once at sare level in the tree, along with a data pointer. In a B'atree, data pointers are stored only ante leaf mades of the tree: hence, the structure of leaf modes differs from the structure of incrnal nodes. The leaf modes have an entry for every value of the search field, along with a data pointer to the record for to the block that contains this zecord? at the search field is a capation field. For a network search field, the pointer points to a block containing pointers in the data file records, creating an extra level of inducation.

The leaf nodes of the B-stree are usually larked rogenher to provide ordered accesson the search field to the records. These leat nodes are similar to the first (base) level of an index. Internal nodes of the B'stree correspond to the other levels of a multilevel index. Some search field values from the hat nodes are reported in the internal nodes of the B's receive ender the search. The structure of the machatropies of a BT-mee of order p (Figure 14.11a) is as follows:

1. Each internal node is of the form

$$\leq P_{11} |\mathbf{K}|$$
,  $P_{21} |\mathbf{K}_{22}|$ ,  $|||, P_{a+1} |\mathbf{K}_{a+1}|$ ,  $P_{1} \geq$ 

where  $q \cong p$  and each  $P_i$  is a tree pointer

- 2. Within each internal node,  $K_1 \le K_2 \le ... \le K_{g,0}$ .
- 5 For all search field values X in the subtrice pointed at by P<sub>µ</sub> we have K<sub>µ+1</sub> ≤ X ≤ K for 1 ≤ i ≤ q; X ∼ K<sub>1</sub> for i − 1; and K<sub>µ+1</sub> ≤ X for i ≥ q (see Figure 14.11a).<sup>2</sup>
- Euch internal node has at most p tree pointers.
- Each internal node, except the nur, has at least [ (p/2) ] tree pointers. The nurnode has at least two tree painters if it is an internal node.
- An internal node with q pointers, q ≃ p, has q = 1 search field values.

The situation of the leginodes of a B\*-cree of order p (Figure 14-12b) is as follows

1. Each leaf needs is of the form

 $<<\!\!K_1,P_1\!\!>\!<\!\!K_2,P_1\!\!>\!<\!\!K_2,P_1\!\!>\!\!P_{1,0}\!\!>$ 



**PGURE 14.11** The nodes of a B\*-tree, (a) Internal node of a B\*-tree with q = 1 search values, (b) Leaf node of a B\*-tree with q = 1 search values and q = 1 data pointers.

2 One definition follows Knoth (1973). One can define a B<sup>\*</sup>-tree deterently by each arging the s<sup>\*</sup> and ≤ semicols (K<sub>1</sub>) ≤ X ≤ K<sub>2</sub> X ≤ K<sub>1</sub>, K<sub>1</sub> =  $\leq$  X), for the principles tension the same. where  $q \approx p_{\rm each}$  for a matrix pointer, and  $P_{\rm max}$  points to the next key sock of the B1-mee.

- 2. Within each leaf node  $|K_1 \sim K_2 \sim \pi | \leq K_{p+1} < \epsilon$
- Each Pr, is a **data pointer** that points to the record whose search held value is K or in a the block concenting the record (or to a block of record pointies that point is records whose scarch field value is K, if the scarch field is not a key).
- Each leaf node bas at least [ (p/2) ] values.
- 5. All leaf nodes are at the same level.

The pointers in internal nodes are tree powers to blocks that are tree nodes, whereas the pointers in leaf nodes are data pointers to the data file remarks or blocks. Except for the  $P_{max}$  pointer, which is a new pointer to the next leaf node. By starting at the leftmost leaf node, it is possible to traverse leaf nodes as a linked but using the  $P_{max}$  pointers. The provides ordered access to the data records on the indexing field. A  $P_{max}$  pointer can also be included. For a B-street on a nonkey field, an extra level of indirection is received smaller to the one shown in Figure 14.5, so the Pripointers are block pointers to block that contain a set of record pointers to the actual records in the data file, as discussed in Option 3 of Section 14.1.3.

Recause entries in the internal index of a  $R^2$ -free include search values and free primiters without any data primiters, more entries can be packed into an internal node of a  $R^2$ -free than for a antilar B-area. Thus, for the same block (node) size, the order p will be larger for the  $R^2$ -free than for the B-area, as we abastrate in Essangle 6. This can lead to fewer  $R^2$ -free levels, improving search true. Because the structures for internal and for leaf fractes of a  $R^2$ -free are different, the order p can be different. We will use p to denote the order for internal meters at the order  $R^2$  and  $R^2$  model, which we denote a being the maximum number of data pointers in a leaf node.

**EXAMPLE** 6: To colculate the order p of a B<sup>+</sup> tree, suppose that the search key field in V = 9 bytes long, the block one is B = 512 bytes, a record pointer is  $P_i = 7$  bytes, and a block pointer is P = 6 bytes, as in Example 4. An internal node of the B<sup>+</sup> tree can have up to p tree pointers and p = 1 search field values these number fit into a single block, bency we have:

 $(p * P) + ((p - 1) * V) \le P$   $(p * 6) - ((p - 1) * 0) \le 512$  $(15 * p) \le 521$ 

We can choose p to be the largest value satisfying the above inequality, which evolute p = 34. This is larger than the value of 23 for the B-free, resulting in a larger throcut and more carries in each internal node of a B<sup>\*</sup>-free than in the corresponding B-free. The lat nodes of the B<sup>\*</sup>-free will have the same number of values and pointers, except that de pointers are data pointers and a next pointer. Hence, the order  $p_{\rm b}$ , for the leaf rocks day be calculated as follows:

$$(\mathbb{P}_{v,d} * (\mathbb{P} + V)) + \mathbb{P} \leq \mathbb{R}$$

 $(p_{c,i} + (7 + 9)) + 6 \le 512$  $(16 + p_{c,i}) \le 506$ 

It follows that each leaf node carehold up to  $p_{\mu\nu} = 31$  key value/data pointer combinators, assuming that the data pointers are record pointers.

As with the B-tree, we may need additional information—to implement the insection and deletion algorithms —in each node. This information can include the type stands (internal or list), the number of current entries q in the node, and pointers to the paint and sibling nodes. Hence, before we do the above calculations for p and  $p_{h,a}$ , we should reduce the block size by the amount of space neisled for all such information. The rest completillustrates how we calculate the number of entries up a B\*-tree.

**EXAMPLE** 7: Suppose that we construct a B'-tree on the field of Example 6. To calculate the approximate number of entries of the B'-tree, we obside that each node is 69 percent fell. On the average, each internal node will have  $34 \times 1.69$  or approximately 23 gointers, incluence 22 values. Each leat node, on the average, will hold  $0.69 \times p_{lost} = 3.69 \times 11$  or approximately 21 data record pointers. A B\*-tree will have the following average number of entries at each level:

| Reege        | Uncide        | 22 entries       | 23 pointers     |
|--------------|---------------|------------------|-----------------|
| Level 1.     | 23 nodes      | 506 entrus       | 529 pointers    |
| Level 2      | 529 nodes     | 11.638 entries   | 12.167 pointers |
| I out levels | 12/167 nodes. | 255.507 record p | onters          |

For the Nock size, pointer size, and search field size given above, a three-level B'-tree holds up to 255,507 record pointers, on the average. Compare this to the 65,535 entries for the corresponding B-cree in Example 5.

Search, Insertion, and Deletion with B\*-Trees. Algorithm 44.2 southes the pocedure using the B\*-tree as access structure to south for a record. Algorithm 14.3 slustrates the procedure for inserting a record in a file with a B\*-tree access structure. These algorithms assume the existence of a key search field, and they must be modified appropriately for the case of a B\* tree on a carriery field. We now illustrate insertion and demon with an example.

Algorithm 14.2: Searching for a record with search key nold value K. using a B\*-tree.

```
n ... block containing root node of B'-tree;
reat block n;
while (n is not a leaf mode of the 8' tree) dc
begin
q ← mumber of tree pointers in node n;
if K # p.K, (*n.K, refers to the n⁰ search field value in rode n[*])
then n ← p.P_L (*n.P. refers to the ith tree pointer in node n[*])
else if K > n K_{q L}
then n ← n.P_L
```

```
else begin
 search node n for an entry i such that n_{i}K_{in} < K \neq n_i K_i;
 a ← p.P.
 end:
 read block n
 end:
search block in for entry (K_1, Pr_1) with K = K_1. (* search leaf node *)
if found
 then read data file block with address Pr. and retrieve record
 else record with search field value K is not in the data file;
Algorithm 14.3: Inserting a record with wards key field, volue K in a B'-treest
order p
in < block containing root node of Bi-tree;</p>
read block ni set stack 5 to empty:
while (a is not a leaf node of the B'-tree) to
 bearn
 push address of a constack 5:
 ("stack 5 holds parent nodes that are needed in case of split")
 c - number of tree pointers in while n:
 if K # n.K, (*m.K, refers to the 1th search field value in node n*)
 then n \leftarrow n.P, ("n.P, refers to the i" tree pointer in node n")
 Plse if K > n_1 K_{a,c}
 then a < a.P.
 clse begin
 search mode p for an entry f such that n, K_{i,1} \in K \neq n, K_{i,2}
 ∩ ⊢ ∩, P₄
 end:
 read block n
 end.
search block in for entry (K_i, Fr_i) with K = K_i, ("search "ea" node n")
If found
 then record already an file-cannot ansert
 else ("insert entry in B'-tree to point to record")
 begrin.
 create entry (K,Pr) where Pr points to the new record;
 if leaf node n is not full
 then insert enery (K, Fr) in correct position in leaf note n
 olse
 begin ("leaf node n is full with prove record painters is split")
 copy n to temp ("temp is an aversize leaf node to hold extra
entry^).
 insert entry (K, Pr) in temp in correct position;
 ("temp now holds p_{ini} + 1 entries of the form (K_i, \nabla i_i)^2)
 new \leftarrow a new empty leaf node for the tree, here, P_{max} \leftarrow n_{1}P_{max};
 j ilp_{ina} .u2 a
 n < first j entries in temp (up to entry (K_,Pr_)); n.P., - new:</pre>
 new \epsilon remaining entries in terp; K \leftarrow K_{i};
```

```
(fnow we must move (K, new) and insert in parent internal mode
 however, if parent is full, split may propagate-)
 finished - false:
 repear
 if stack S is emoty
 then ("no parent node-new root node is created for the tree")
 beann
 root \leftarrow a mem empty internal node for the tree:
 root \leftarrow root, K, news: finished \leftarrow true;
 end
 else
 beg: n
 n - pup stack S:
 if internal mode r is not full
 then
 begin ("parent node not full-no split")
 insert (K. new) in correct position in internal models;
 finished - true
 end
 else
 begin (tinternal pode n is full with p tree pointers-is splits)
 copy n to femp ("temp is an oversize internal node");
 insert (K,rew) in temp in correct position;
 ("temp now has p+] tree pointers")
 new + a new empty internal mode for the tree;
 j — "((p + 1)/2)";
 n — entries up to tree pointer P. in temp;
 ('m rontains <P, k₁, P₂, K₂, ..., P_{1,1}, k₁, P >')
 (*new contains < P₁., K₁₁..., K₂., P₀. K₀., P₀₋₁.»*)
 к — <-
 (*now we must move (K,new) and insert in parent internal node*)
 end
 end.
 unt 1 fiлished
 end;
end:
```

Figure 14.12 illustrates insertion of records in a B'-ince of order  $\mathbf{r} = 3$  and  $\mathbf{p}_{b,0} = 2$ . First, we observe that the more is the only node in the tree, so it is also a leaf rude. As som as more than one level is created, the tree is divided into internal nodes and leaf rodes. Notice that every key value must case or the top level, because a lidata pointers the at the leaf level. However, only some values exist in internal nodes to guide the sorth. Notice also that every value appearing in an internal node also oppears as the reference when in the leaf level of the subtree pointed at by the tree pointer to the left of the value.

When a (caf node is full and o new entry  $\alpha$  inserted there, the node overflows and must be split. The first j = 1 ( $(p_{poil} + 1)/2$ ) | entries in the original node are keep there.



FIGURE 14,12. An example of insertion in a B\*-tree with p=3 and  $\rho_{test}=2$ 

and the remaining cutties are moved to a new leaf node. The  $j^{th}$  scareb value is replicated in the parent interpal node, and an extra pointer to the new node is created in the parent. These must be inserted in the parent node in their correct sequence. If the parent normal node is full, the new value will cause it to overflow also, so it must be split. The entries in the internal node up to  $P_i$  — the  $j^{th}$  tree pointer after inserting the new value and particle where j = j ((p + 1)/2) j — are kept, while the  $p^{th}$  search value is moved to the parent not replicated. A new internal node will hold the entries from  $P_{in}$  to the end of the entries in the node two Algorithm 14.33. This splitting can propagate all the way op is denote a new recent node and hence a new level for the  $R^{1}$ -tree.

beare 14-15 illustrates deletion from a B<sup>2</sup>-tree. When an entry is deleted, it is always removed from the loaf level of it happens to soccur is an internal node, it must also be tensived from there. In the latter case, the value to its left in the leaf node must replace it in the internal node, because that value is now the rightmost entry in the subtree Deletion may cause underflow by reducing the number of entries in the leaf node most replace it left is the minimum required. In this case we try to find a sibling leaf node—a leaf node lated to the left or to the right of the number of leaf nodes is reduced. A common method is to try redstributing entries with the left sibling of this is not possible, an archip to redistributing entries with the left sibling of this is not possible, an archip in redistributing entries with the left sibling of this is not possible, an archip in redistributing on the right abling is made. If this is not possible other, the internal nodes because one lever from pointer and search value are incided. This can provide the tree levels.

Notice that implementing the insertion and deletion algorithms may require parent and sibling pointers for each node, or the use of a stack as in Algorithm 14.3. Each node should do couclude the nonder of entries in it and its type (leaf or miternal). Acother alignative is to implement insertion and deletion as recursive procedures.

Variations of B-Trees and B\*-Trees. To conclude this section, we briefly controls one carations of B-Trees and B\*-trees, to some cases, constraint 5 on the B-tree to: B\*-treet, which requires each node to be at least half full, can be charged to require each node to be at least half full, can be charged to require each node to be at least half full, can be charged to require each node to be at least that full, can be charged to require each node to be at least that full, can be charged to require each node to be at least that full, can be charged to require each node to be at least the former basebeen called a B\*-tree. In general, some systems doe the user to choose a fill factor between 0.5 and 1.0, where the latter means that the B-tree (index) index are to be completely full. It is also possible to specify two fill factors for a B\*-tree (index) nodes are to be completely full. It is also possible to specify two fill factors for a B\*-tree cases for the feat least of the tree. When the index is first obstracted reach node is filled up to approximately the fill tactors specified. Recently resting some cases have anode to become completely enjoy before merging, to simplify the deletion algorithm. Similation -to does show that this does not waste too much additional space under randomly databased insertions and obsertions.

CELETION SEQUENCE: 5, 12, 9



FIGURE 14:13 An example of deletion from a B1-tree.

# 14.4 INDEXES ON MULTIPLE KEYS

In our diagnesion, so far, we assumed that the primary or secondary keys on which files see accessed were single attributes (fields). In many tetrioval and update requests, mulaple attributes are involved. If a certain combination of attributes is a ed very frequently, a is advancing out to set up an access structure to provide efficient access by a key value also a combination of those attributes.

For example, consider an incorce file containing attributes **100** (department number), we start, kits, zerood, salary, and selected with the key of **500** (social security number). Consider the query: "List the employees in department number 4 where age is 50." Note that below and **450** are nonkey attributes, which means then a search value for either of these will pain to number be receeds. The following alternative search strategies may be considered:

- Assuming the has an index, but are does not, access the records having the 4 using the index then select from among them those records that satisfy 459 59.
- Alternately, if AGE is indexed but DBM is not, access the records having AGE = 59
  using the index then select from among them those records that satisfy PAO = 4
- 3. If indexes have been created on both 360 and 355, both indexes pay be used, each gives a set of meands or a set of primers (in blocks or records). An intersection of these sets of records or pointers could those records that satisfy both conditions, those records that satisfy both conditions, or the blocks in which records satisfying both conditions are located.

All of these alternatives eventually give the correct result. However, if the set of records that meet each condition ( $w_0 = 4$  or  $w_0 = 595$  individeally are large, yet only a few records satisfy the combined condition, there none of the above is a very efficient technique for the every search request. A number of possibilities exist that would near the combination ( $w_0$ ,  $w_0$ ,

#### 14.4.1 Ordered Index on Multiple Attributes

All the discussion in this chapter so for still applies if we create an index on a search key field that is a combination of < 595, acc>. The search key is a pair of values < 4, 59> m detabove example. In general, if an index is created on attributes < A , A<sub>1</sub>, . . . , A<sub>n</sub>>, detaench key values are rapidly with n values < <  $v_1, v_2, \dots, v_n$ >.

A lexicographic ordering of these tuple colors coublishes an order on this composite sarca key. For our example, all of department keys for department number 3 precede these for department 4. Thus <3,  $n \geq$  precedes <4,  $n \geq$  for any values of m and m. The averaging key order for keys with ew = 4 would be <4,  $18 \geq <4$ ,  $19 \geq <4$ ,  $20 \geq$ , and so or flexicographic ordering works similarly to ordering of characteristings. An index on a composite key of n articlates works similarly to ordering of characteristings. An index on a composite key of n articlates works similarly to any index discussed in the chapter so far.

#### 14.4.2 Partitioned Hashing

Parationed hashing is an extension of static external bashing (Section 14.3.2) that allows access on multiple keys. It is suitable only for equality comparisons, range species are not supported. In participated hashing, for a key consisting of a components, the hash function is designed to produce a result with a separate hash addresses. The backet address is a contratemation of these n addresses. It is then possible to search for the required composite search key by looking up the appropriate backets that match the parts of the address in which we are interested.

For example, consider the composite search key <500,  $u_0 \ge 11000$  and  $u_0$  has basis of 5 bit address respectively, we get an 6-bit backet address. Suppose that  $u_0 = 4$  has a hash address "100" and  $u_0 = 59$  has bash address "10101". Then to search for the combined search volue, 50 = 4 and  $u_0 = 59$ , one goes to bucket address 120 bit 121 just to search for all employees with  $u_0 = 59$ , all backets (right of them) will be searched whose addresses are "000" 10101", "001" 10101", "100" etc. An advantage of partitioned hashing is that it can be easily extended to any number of attributes. The backet addresses can be desired so that high order bits in the addresses correspond to more frequently accessed attributes. The mean drawback of partitioned hashing is that it butes. Additionally, no separate access structure needs to be maintained for the individual attributes. The main drawback of partitioned hashing is that it can be readily or separate access structure needs to be maintained for the individual attributes. The main drawback of partitioned hashing is that it can be readily or separate access structure needs to be maintained for the individual attributes.

#### 14.4.3 Grid Files

Another alternative is to organize the encoded file as a grid file. If we want to access a file on two keys, say two and x67 as in our example, we can construct a grid attac with one lincar scale (or datension) for each of the search detributes. Figure 14.14 shows a end ortay for the concert, file with one linear scale for two and another for the set attribute. The scales are made in a way as to achieve a sumform distribution of that attribute. Thus, in our example, we show that the linear scale for two has two = 1/2 combined as one value 0 on the scale, while that = 5 contesponds to the value 2 on that scale. Similarly, are is divided into its scale of 0 to 5 by grouping ages so as to distribute the employees ontionally by age. The grid array shown for the file has a total of 30 cells. Each cell points to some



FIGURE 14.14 Example of a grid prray on two and we attributes

bucket address where the records corresponding to that cell are stored. Figure 14.14 also slows assignment of cells to buckets (only partially)

This our request for two = 4 and xzz = 50 maps into the cell (1, 5) corresponding to the grid array. The records for this combination will be found in the corresponding backet. This method is particularly useful for range queries that would map into a set of cells corresponding to a group of values along the linear scales. Conceptually, the grid the integri may be applied to any number of scarch keys. For a search keys, the grid array would have in dimensions. The grid array thos allows a partitioning of the the along the dimensions of the search key arrithmes and gravides an access by combinations of values along those dimensions. Grid files perform well in terms of reduction in time for multiple keyaccess. However, they represent a space overhead in terms of the grid array structure. Marcover, with dynamic files, a frequent programation of the file odds to the maintenance gost.<sup>10</sup>

# 14.5 OTHER TYPES OF INDEXES

#### 14.5.1 Using Hashing and Other Data Structures as Indexes

It is also possible to create access structures similar to indexes that are based on hisbling. The index entries  $\langle K, P \rangle$  (or  $\langle K, P \rangle$ ) can be organized as a dynamically expandable hashride, using one of the techniques described in Section (B.8.5) soutching for an entry cost the hash search algorithm on K. Once an entry is lound, the pointer Pr (or P) is used to locate the corresponding record in the data file. Other search structures can also be used as indexes.

#### 14.5.2 Logical versus Physical Indexes

So far, we have assemed that the index entries  $\langle K, F \rangle (\alpha \langle K, F \rangle)$  always include a physical pointer Pr (or P) that specifies the physical record address on disk as a block number and offset. This is sumerimes called a physical index, and it has the disadvantage dot the pointer must be rhanged if the mental is moved to another disk location. For example, suppose that a primary file organization is based on linear bashing or extendible bashing; then, each time a backet is split, some records are allocated to new buckets and hence have new physical addresses. If there was a secondary index on the life, the pointers to thus records would have us by found and updated—a difficult task.

To remedy this situation, we can use a structure called a logical index, whose index entries are of the form  $\langle N, K_p \rangle$ . Each entry has one value K for the secondary indexing field unsched with the value  $K_p$  of the field used for the primary file organization. By

<sup>12.</sup> Insertion/Jelerien algorithms for grid files may be found in Nievergelt (1984)

searching the secondary index on the value of **K**, a program can locate the corresponding value of  $\mathbb{K}_{p}$  and use this to access the tecord through the primary file organization, logical indexes thus introduce an additional level of indirection between the access structure and the data. They are used when physical record addresses are expected to change treatments. The cost of this indirection is the extra search based on the primary file organization.

#### 14.5.3 Discussion

In many systems, an index is not an integral part of the data file but can be created and discarded dynamically. That is selve it is often called an access struction. Whenever we expect to access a file frequently based on some search condition involving a particular field, we can request the DBMs in create an index on that field. Usually, a secondary index is created to avoid physical ordering of the records in the data file or disk.

The main advantage of secondary indexes is that - theoremailly, at least—they arbe created in conjunction with torusally any principlicated organization. Hence, a secondary index could be used to complement other principlicates methods such iordering or bashing, or of could level be used with mixed files. To create p R<sup>2</sup> mersecondary index on some field of p life, we must go through all records in the file increate the entries at the leaf level of the tree. These entries are then would and filled according to the specified fill factor, simultaneously, the other index levels are created. It is note expensive and much harder to create primary indexes and clustering indexes dynamically because the seconds of the data file must be physically sorted on disk in order of the indexing field. However, some systems allow users to create these indexes dynamically or their tiles by sorting the tile dorme index creation.

It is common to use an index to enforce a key constrant on an attribute. When searching the index to insert a new record, it is straightforward to check at the same trawhether anather record in the file—and hence in the index trac—has the same key attribute value as the new record. If so, the insertion can be rejected.

A file that has a secondary index on every one of its field, is often called a fully inverted file. Because off indexes are secondary, new records are inverted or the end of the file, therefore, the data file itself is an unordered (brap) file. The indexestore usually implemented as **B**<sup>+</sup>-trees, so they are updated denomically to reflect insertion or deletion of records. Sine commercial DPMes, such as ADABAS of Software Asia are this method extensively.

We referred to the popular 38M file organization called ISAM in Section 142 Another 10M method, the virtual storage access method (VSAM), is somewhat similar to the B1Ance access structure.

## 14.6 SUMMARY

In this chapter we presented file organizations that involve additional access structures called indexes, to improve the efficiency of terrieval of records from a data file. These access structures may be used in conjunction with the primitivitle organizations discussed in Chapter 13, which are used to organize the file records themselves on disk. Three types of ordered single-level indexes were introduced: (1) primary 425 clustering and (3) secondary Each index is specified on a field of the file. Primary and distering indexes are constructed on the physical ordering field of a file, whereas woodary indexes are specified on nonoidering fields. The field for a primary index must darks is a workey to the file, whereas it is a conkey field for a clustering index. A single-level index is an ordered file and is corrected using a binarc search. We showed how multifevel indexes can be constructed to improve the efficiency of som larg an index.

We then showed how multilevel indexes can be implemented to Barrees and B<sup>4</sup> (neeshich are dynamic structures that allow an index to expand and shrink dynamically. The roles (blocks) of these index structures are kept between half full and completely full by its insertion and deletion algorithms. Nodes eventually stabilize as an overage occupancy of 69 percent full, allowing space for insertions performing reorganization of the index for the majority of insertions. B<sup>4</sup> trend can generally hold more entries in their internal nodes that role B trend, so they may have fewer levels or hold more entries than does a corresponding B inter-

We gave an overview of inultiple key access methods, and showed how an index can be constructed based on bash data structures. We then introduced the concept of a logical index, and compared it with the physical indexes we described before. Finally, we decoved how combinations of the above organizations can be used, for example, workday, indexes are often used with nuxed files, as well as write unordered and ordered files. Secondary indexes can also be created for hash files, and dynamic hash files.

#### Review Questions

- 19.1. Define the fullowing terms: indexag field, Januars key field, clustering field, scoundary key field, block cracher, dense rades, and renderse (sparse) rades.
- 14.2 What are the differences among primary, secondary, and clustering indexes? How do these differences affort the ways in which these indexes are implemented? Which is the indexes are dense, and which are not?
- 14.5. Why can we have at most one primary or clustering index on a file, but several secondary indexes?
- 114. How does multilevel indexing improve the efficiency of searching an index the?
- (4.5) What is the order p of a B tree? Describe the structure of B-tree pades.
- 14.6. What is the order p of a B\*-rice? Describe the -tructore of both internal and leaf nodes of a B\*-tree.
- 14.7. How does a B-mee differ from a B'-tree! Whens a B'-tree u-ually prefetted as an access structure to a data file?
- 14.5 Explain what alternative choices exist for accessing a file based on multiple search keys.
- 14.9. What is partitioned hashing? How does it work? What are its luminition-?
- 14-10. What is a grid file? What are its advantages and disadvantages?
- 14.11. Show an example of constructing a god array on two attributes on some file.
- 14.12. What is a fully inverted tile? What is an indexed sequential file?
- i4.13. How can hashing be used in construct as index? What is the difference between a logical index and a physical index?

#### Exercises

- 14.14. Consider a disk with block size B = 512 bytes. A block pointer is P = 6 bytes long, and a record portion is  $P_{ij} = 7$  bytes long. A the has r = 30,000 EMFLO) fit records of fixed length. Each record has the following helds: size (30 bytes), SSN (9 bytes), in metric (or (9 bytes), 4000533 (40 bytes), Prime 19 bytes), BERTHORE (S bytes), See (1 byte), integer (4 bytes) eacher (4 bytes) real number). An additional byte is used as a deletion market.
  - a. Calculate the record size R in byres.
  - Calculate the blocking factor bit and the number of the blocks b, assuming an unspanned organization.
  - c. Suppose that the file is onlined by the key field sis and we want to construct a primary index on say. Calculate (i) the index blocking factor bit, (which is also the index fan-out to); (ii) the number of first-level index entries and the number of first-level index entries and the number of hist-level index blocks; (iii) the number of blocks required by the malulevel index fan-out to); (ii) the rotal number of blocks required by the malulevel index fance (iv) the rotal number of blocks required by the malulevel index; and (iv) the rotal number of blocks required by the malulevel index; and (iv) the rotal number of blocks required by the malulevel index; and (iv) the rotal number of block accesses needed to search for and retrievels record from the hie—given its yet vide—i sing the primary index.
  - d. Suppose that the file is not ordered by the key field sscaral we want to construct a secondary index on ssc. Repeat the previous exercise (part c) for the secondary index and compare with the primeity index.
  - c. Suppose that the blc is not indexed by the nonlext field reservences and we want to construct a Grandkry Jusica on representation, using cytom 3 of Section 14.1.3, with an extra level of tradirection that stores actors granders. Assume there are 1020 distinct values of preprinted and that the stores actors betweenly distributed among these values. Calculate (1) the index blocking betweenly distributed among these values. Calculate (1) the index blocking betweenly distributed among these values. Calculate (1) the index blocking betweenly distributed among these values. Calculate (1) the index blocking betweenly distributed among these values. Calculate (1) the index blocks needed by the level at indirection that stores incord pointers; (iii) the number of line level index epiries and the number of test devel index (1) the number of line levels needed if we make it into a multilevel index (1) the total number of blocks required by the approximate number of block accesses reoded (1) stored for any for and rection 1 and rections) is included in the number of lock accesses reoded (1) stored for any recting and (1) the approximate number of block accesses reoded (1) stored for any recting and rections) is included by the indirection of the approximate number of block accesses reoded (1) stored for any recting the rest of indirection).
  - t Suppose that the file is ardened by the nonkey field in some accounted and we want to construct a clustering index on measurements, that uses block anchors very new value of pressence on starts at the beginning of a new block). Assume there are 1300 distinct values of measurements, and that the index block is Assume evenly distributed among these values. Calculate (0) the number of first-level index blocks; (a) the number of first-level index entries and the number of first-level index blocks; (a) the number of blocks required by the multilevel index; and (b) the number of block accesses medel to starts in the file that have a specific beyony structs in the file that have a specific beyony structs in the file that have a specific beyony structs (assume that multiple blocks in a cluster are contiguous).

- 8. Suppose that the file (short ordered by the key field say and we want to construct a B<sup>+</sup>-tree access structure (index) on say. Calculate (a) the orders b and  $p_{i,ij}$  of the B<sup>+</sup>-tree. (ii) the number of leaf-level blocks needed if blocks are approximately 69 percent full (rounded up for convenience): (iii) the number of levels needed if internal nodes are also 69 percent full (rounded up for convenience): (iv) the total number of blocks required by the B<sup>+</sup>-tree; and (v) the number of block accesses needed to search for and retrieve a record from the file—given its say value—using the B<sup>+</sup>-tree.
- Repeat partig, but for a B-free rather than for a B'sfree. Compare your results for the B-free and for the 8'sfree.
- 14.15 A totil15 file with Part# as key field includes records with the following Part# values: 23, o5, 37, 60, 46, 92, 45, 71, 56, 59, 18, 21, 10, 74, 78, 15, 16, 20, 24, 28, 39, 43, 47, 50, 69, 75, 5, 49, 34, 58, Suppose that the search field values are interfed in the given order in a Pi-tree of order p = 4 and p<sub>we</sub> = 3; show how the free will expand and what the final free will look like.
- (4.16) Repeat Exercise 14.15, but use a B-tree et onler p = 4 instead of a B\*-tree.
- (4.17. Suppose that the following scarch field values are deleted, in the given order, from the B1-tree of Exercise 14.15; show how the tree will shrink and show the final order. The deleted values are 65, 75, 43, 18, 20, 92, 59, 37.
- 14.14 Repeat Exercise 14.17, but for the Batter, of Exercise 14.16.
- [414] Algorithm [4,1 outlines the procedure for som hing a non-lease multilevel primory index to remove a ble record. Adapt the algorithm for each of the following cases:
  - A multilevel secondary index on a nonlosy transdering field of a file. Assume that option 3 of Section 14.1.3 is used, where an extra level of indirection stores pointers to the individual records with the corresponding index field value.
  - b. A multilevel secondary index on a nonordering key field of a file
  - (c) A multilevel clustering molecton a nonkey ordering field of a file.
- 14.20. Suppose that several secondary indexes exist on nonkey fields of a file implemented using option 3 of Section 14.1.3, for example, we could have secondary indexes on the fields reproverseers, record, and sector of the support file of Exercise 14.1.4. Describe an efficient way to search for and remeve records satisfying a complex selection condition on these fields, such as (resonance) used as a second sector = 12 ANT-sector = 52,000°), using the record pointers in the indirection level.
- 14.21. Adapt Algorithms 14.2 and 14-4, which outline search and discrimin procedures for a B\*-tree to a B-tree.
- 14.22 It is possible to modify the IV-tree insertion algorithm to delay the case when a new level is produced by checking for a possible redworkation of values arrange the leat nodes. Figure 14,15 affectives how this could be dure for our example in Figure 14,12; rather than splitting the leftmost leat node when 12 is inserted, we do a *left* region/dumo by moving 7 to the leaf under to its left (if there is space in this mode). Figure 14,15 shows how the tree would hole when its build to be subscribed on all provides considered. It is also possible to consider right redworkation. Try to modify the B1 area insertion algorithm to take redworkation.
- 14.23. O uline an algorithm for deletion from a B1 trees
- 14/24. Repeat Exercise 14/23 for a Barree.



FIGURE 14.15 (B)-tree insertion with left redistribution

#### Selected Bibliography

Baye: and McCreight (1972) membered B-trees and associated algorithms. Coner (1978) provides in excellent survey of B-trees and their history, and can more of B-trees. Knoth (1973) provides detailed enalysis of juary search rechangles, and dama for B-trees and surve of their variations. Nevergelt (1974) discusses the use of binary search trees for ble organization. Tevrbooks on the structures including Wirth (1977), Claybrook (1989). Simth and Barties (1987), Miller (1987), and Subberg (1988) discuss andexing in detail and may be consulted for search, insertion, and deletion of gerithms for B-trees and B-trees. Larson (1981) analytes index-sequential files, and Held and Stonebraket (1978).

compare static multilevel indexes with B-free dynamic indexes. Lebruari and Yao (1981) and Srinicasan and Carey (1991) did further analysis of concurrent access to B-frees. The backs'ry Wiederhold (1983). Smith and Parnes i 1987), and Sabberg (1988), among others, discuss many of the search techniques described in this chapter. Ord tiles are introduced in Nievergelt (1984). Partial match retrieval, which uses parritioned Easlong, is decised in Burkhaid (1976, 1979).

New techniques and applications of indexes and B<sup>4</sup>-trees are discussed in Lanka and Says (1991). Zobel et al. (1992), and Faloutsos and Jagadish (1992). Mehan and Narang (1992) discuss index creation. The performance of various B-tree and B<sup>4</sup>-tree algorithms is assessed in Baeta-Yates and Laison (1989) and Johnson and Shasha (1993). Butter nanagement for indexes is discussed in Chan et al. (1992).



# 15

# Algorithms for Query Processing and Optimization

In this chapter we discuss the rechniques used by a PBMS to process, optimite, and execute lightlevel queries. A query expressed in a high-level query language such as SQL must first re-scanned, parsed, and validated. The scanner identifies the language rokens—such as SQL keywords, attribute names, and relation names—in the text of the query, whereas the parer checks the query syntax to determine whether it is formulated according to the syntax roles (roles of grammar) of the query language. The query must also be validated, by enclong it is all articlate and relation names are valid and semantically meaningful names in the schema of the particular database being queries. An internal representation of the query is then created, usually as a tree data structure called a query tree. It is also possible to represent the query using a graph data structure called a query graph. The 1996 most then do use an execution strategy for retrieving the result of the query from the database files. A query typically has many possible execution strategies, and the process of choosine a suitable one for processing a query is known as query optimization.

Figure 15.1 shows the different steps of processing a high-level query. The query optimizer models has the task of producing an execution plan, and the code generator generator the code intersecute that plan. The random database processor has the task of running the query code,

<sup>.</sup> We will use discuss the parsing and sout its checking phase of chery processing here; this material

edisensed in compiler textbooks.




whether in compiled or interpreted mode, to produce the query result. If a random error results, on encompositive is generated by the random database processor

The series optimization is a crually a mismomer because in some cases the chosen execution plan is not the optimal (best) strategy in its just a tengonably efficient strategy for executing the query. Finding the optimal strategy is usually roo time-convening exceptor the simplest of queries and may require information on how the files are implemented and even on the contexts of the files—information that may not be fully available in the Devis carding. Hence, planning of an execution sporegy may be a more account description than query optimization.

For lower-level reoraeational database languages in legacy systems—such as the network 10% or the hierarchical ritios. Gee Appendixes E and F)—the programmer must

cases the query execution strategy while string is doublese program. If a DPMS provides only a navogational language, there is found need of opportunity for extensive query opportunition by the DBSS: instead, the programmer is given the capability to choose the "rptima," execution strategy. On the other hand, a high-level query language—such as soft or relational testiss (RDIssiss) in CC, (see Chapter 21) for object DBSSs (CDISSs) is note declarative in nature because at specifies what the intended results of the query are, after than identifying the identify of four the result should be obtained. Query optimization is this increasing for queries that are specified in a high-level query language.

We will concentrate on describing jointy optimization on the context of an RDWF because many of the techniques we describe have been adapted for ODPMSS.<sup>7</sup> A relational 55% most systematically evaluate alternative query execution strategies and choose a brainably efficient or optimal strategy. Each DWF typically lus o number of general diabase access algorithms that implement relational optimations such as SE FCT or ION or outbinations of these operations. Only execution strategies that can be implemented by the 26WF access algorithms and that apply to the particular query and particular physical diabase design can be considered by the query optimization module.

We start in Section 15.1 with a general discussion of how 3Q, queries are typically translated into relational algebra queries and then optimized. We then discuss disordaries is implementing relational optimization strategies. There are two main techniques to implementing query optimization strategies. There are two main techniques to implementing query optimization. The first technique is based on heuristic rules for ordering the operations in a query execution strategies. There are two main techniques to indening the operations in a query execution strategies. There are two main techniques to indening the operations in a query execution strategies and choosing the execution strategies are bounded on the technique involves systematically estimating the cost of different execution strategies and choosing the execution glan with the lowest cost estimate. The two techniques are usually combined in a query optimizer. We discuss heuristic optimization in Section 15.7 and cost estimation in Section 15.8. We then provide a brief operations in Section 15.0. Section 15.10 introduces the topic of semantic query optimization, in which known constraints are used to devise efficient speed execution strategies.

### 15.1 TRANSLATING SQL QUERIES INTO RELATIONAL ALGEBRA

In practice, SCL is the query language that is used in most commercial RDBMSs. An SQL query is first translated into an equivalent extended relational algebra expression – represented as a query tree data structure -- that is their optimized. Typically, SQL queries are decomposed into query blocks, which turn the base units that can be translated into the

? There are some query optimization problems and techniques that are performent only to a OBMAS. However, we do not discuss classe here as we can give color an introduction to query estimation. algebraic operators and optimized. A query block contoins a single SELECT-TROM-WHERE expression, as well as GRO-IP BY and HAVING clouses if these are part of the block. Hence, nested queries within a query are identified as separate query blocks. Because SOF includes aggregate operators—anch as MAX, MIN, SOM, and COUNT—these operators must also be included in the extended algebra, as we discussed in Section 6.4.

Consider the following sign query on the assume relation in Figure 5.5.

| SELECT | LNANE, FNAME |         |              |
|--------|--------------|---------|--------------|
| FROM   | EMP_OVEE     |         |              |
| WHERE  | SALARY >     | (SELECT | MAX (SALARY) |
|        |              | FRON    | ENPLOYEE     |
|        |              | WHERE   | ON0=51:      |

This greey includes a nested subquery and hence would be decomposed into two blocks. The inner block i-

```
(SELECT MAX (SALARY)
FROM EMPLOYEE
WHERE DNO-5)
```

and the outer block is

```
SELECT INAME, FNAME
FROM EMPLOYEE
WHERE SALARY > C
```

where threpresents the result returned from the other block. The inner block could be translated into the extended relational olechia expression.

Num ( Lage ( ( Dug ... ( SHA CYFF ) )

and the outer block into the expression

```
\pi_{1,\text{MARE}}(c_{\text{MARE}}(0)_{(0,1),(\text{MARE})})
```

The areasy operates would then chouse an execution plan for each Hock. We should note that in the above example, the inner block needs to be evolvated only once to produce the maximum salary, which is then used—as the constant o—by the outer block. We called this an incomplicit iterated points in Chapter 5. It is much border to optimize the more complex correlated nested games free Section 6.55, where a tuple variable from the outer block appears in the WIERE-clause of the inner block.

### **15.2 ALGORITHMS FOR EXTERNAL SORTING**

Sorting is one of the primary algorithms used in query processing. For example, whenever an OQL query specifies an OROBR PS clause, the query result must be sorted. Sorting is also key component its sort merge algorithms used for JOIN and other operations (soch as UNICN and NTERSECTION), and in duplicate chimatotic algorithms for the PROJECT operation (when an SQL query specifies the DiSTINCT option in the SELECT clause). We will discussions of these algorithms in the section. Note that sorting may be avoided if an appropriate index exists to allow ordered access to the records.

External sorting refers to serting algorithms that are solvable for large nice of accords good on disk that do not fit enrirely in main memory, such as most database files.<sup>1</sup> The speal external sorting algorithm uses a sort-merge strategy, which starts by surring small subiles – colled runs—of the main file and then merges the sorted runs, creating larger sets! subiles that are merged in torn. The sort-merge algorithm, like other database docubris, requires higher space in main memory, where the second surring and merging of mercines performed. The basic algorithm, outlined in Figure 15.2, consists of two phases: (1) the sering phase and (2) the merging phase.

In the sorting phases turns (perturns or pieces) of the file that can fit of the available hifer space are read into more memory, sorted using an internal sorting algorithm, and minimum back to disk as temperary sorted subfiles (or turns). The size of a turn and number of minimum  $(n_R)$  is dictated by the number of file blocks (h) and the available buffer

```
sei -
 14 1;
 (⊷ 0;
 (size of the file in blocks)
 k⊷.8₂
 (size of buller in blocks)
 क्तर मिं/फॉ
!Soft Phase>
ոինց (, Հա ա)
 da (
 read next A blocks of the file mig the buffer or if there are less than K blocks remaining.
 then read in the remaining blocks;
 son the records in the buller and write as a temporary sublice;
 t⊷ t + 10
 ķ
(Marge Phase: marga sublices unit only 1 remains)
581
 i⊷1,
 p — [[og__ ni], {p is the number of passes for the merging phase}
 i – mi
while (/ <= D)
 dolí
 ov 1:
 a \leftarrow \overline{i}(i/(k-\tau)); introduce of subNee to write in this base.
 while (n <= at
 dol
 read next A+1 sublies or remaining sublites (from previous pass) one block at a time;
 merge and write as new sublike one block at a time,
 n = n + 1;
 ì
 1⊷ ř÷ 1:
 1
```

FIGURE 15.2 Outline of the sort-merge algorithm for external surting

Unarreal scrong algorithms are suitable for sorting slata structures that a mitt not relating accord

**space**  $(n_0)$ . For example, if  $n_0 = 5$  blocks and the one of the nie b = 1024 blocks, then  $n_0 = \mathbb{C}\{b/n_b\}$ , or 205 mitral runs each of size 5 blocks (except the last run which will have 4 blocks). Hence, after the sort phase, 205 sorted runs are stored as temporary subtles on disk.

In the merging phase, the surved runs are merged during one or more passes. The degree of merging  $(d_M)$  is the number of runs that can be merged regether in each pass. In each pass one buffer block is needed to build one block from each of the runs being merged, and surver block is needed for containing one block of the merge result. Hence,  $d_M$  is the smaller of  $(n_e - 1)$  and  $n_{ee}$  and the number of passes is  $-k/(g_{MV}(n_{10}))$  [1]. In our example,  $d_M = 4$  (four way merging), so the 205 should writed runs would be merged into 52 at the end of the first pass, which are then merged into 13, then 4, thes 1 run, which means that four power are decided. The momentum  $d_M$  of 2 gives the west-case performance of the algorithm which is

(2 + b) + (2 + (b + (leg, b)))

The first term represents the number of block accesses for the sort phase, since each the block is accessed twice—sonte for reading into memory and once for writing the records block to disk after sorting. The second term represents the number of block accesses for the merge phase, assuming the worst-case  $d_M$  of 2. In general, the log is taken to the base  $d_M$  and the expression for number of block accesses becomes

 $(2 + \hat{n}) + (2 + (b + (\log_{10} n_k)))$ 

# 15.3 ALGORITHMS FOR SELECT AND JOIN OPERATIONS

### 15.3.1 Implementing the SELECT Operation

There are many options for executing a SFL% T operation; come depend on the file having specific access paths and may apply only to certain types of selection conditions. We docus some of the algorithms for implementing SFL% T in this section. We will use the following operations, specifical on the relational database of Figure 5.5, to illustrate out discussion.

(641): «Γ<sub>ΛΑΛ-1214</sub>τηταρη (ΕΝΡΙΟΝΕΕ) (142): «Γ<sub>ΟΜΑΝΤΕ Α</sub> (ΟΡΙΟΜΕΡΕΝΙ) (143): «Γ<sub>ΟΜΑΝΤΕ Α</sub> (ΕΝΡΙΟΝΕΡ) (144): «Γομαια Αλή Αληθητικής «Καταίλατη (ΕΜΡΙΔΥΓΕ) (645): «Γραφια Σροστάζου (Δημιρότικο (ΜΟλΑΚ ΟΒ)

Search Methods for Simple Selection. A number of souch algorithms are possible for selecting records from a file. These are also known as file sears, because they some the records of a file to worth for and retrieve records that satisfy a selection condition.

<sup>4.</sup> A selection operation is achieved called a filter, since a filters or table records in the file that do not sursty the selection condition.

If the search algorithm involves the use of an index, the index sourch is called an **index** way. The following search methods (S1 through S6) are examples of some of the search depithms that can be used to implement a select operation:

- S1. Linear search (frate face). Betrieve every zeroid in the file, and test whether its attribute values satisfy the selection condition.
- S2. Beauxy south: If the selection conductor involves on equality comparison on a key article to which the file is ordered, butary search—which is more efficient than line it search—can be used. An example is OPL if south the ordering articlet for the owned file <sup>5</sup>
- S3. Using a primary index (or bash key). If the relection condition involves an equility comparison on a key attribute with a primary reales (or bash key)—for example, say = '123456789' in OPD—use the primary realex (or bash key) to retrieve the record. Note that this condition retrieves a single record (at most).
- S4. Using a privacy index to teach evolutible seconds: If the comparison condition is >:
   >=, <, is << on a key field with a primary index—tor usample, tausses > 5 in OP2—use the radex to find the record satisfying the corresponding equality condition (causes = 5), then retrieve all subsequent records in the teadered) file. For the condition by user < 5, retrieve all the preceding records.</li>
- 55. Using a clustering ratex to retrieve multiple records: If the selection condition involves an equality companion on a non-key attribute with a clustering index. Screwample, 00 = 5 at OP3 use the index to retrieve all the records satisfying the condition.
- Sc. L'ong a scondery (B\* croc) axies on an equilativ compension. This search method can be used to retrieve a single record of the indexing field is a key (bus analytivalues) or to retrieve multiple records if the indexing field is not a key. This can also be used for companyons involving (2, 27, 5, or 5.7).

In Section 15.5, we discuss how to develop formulas that estimate the access cost of these search methods in terms of number of block accesses and access time. Method S1 applies to any file, but all the other methods depend on having the appropriate access path on the attribute used in the selection condition. Methods 54 and 50 can be used to active records in a certain tanze--for example. 30000×=sausex=35000. Queries involving such conditions are called **range queries**-

Search Methods for Complex Selection. If a condition of a SELECT operation is a conjunctive condition other is of it is made up of several simple condutions connected with the AND logical connective such as OPE above—the DIMS concise the following aldinonal methods to implement the operation:

 ST. Conjunctive referring an industrial index. If an attrabute involved in any single simple condition in the componence condition has an access path that

5 Generally, humary search is not used in durthase search because ordered bles are not used index dwy also hume a corresponding primary index. permuts the use of one of the Methods S2 to S6, use that condition to rearise the records and then check whether each ceteleved record satisfies the reasoning simple conditions in the conjunctive condition.

- 58. Comparative velocitions using a composite index: If two or more attributes are involved in equality conditions in the comparative condition and a composite index (or bash structure) exists on the combined fields. For example, if an index has been created on the composite key (1555, 662) of the sake- on file for OP5 - we can use the index directly.
- S9. Comparisons releases by anymenton of morel participating it secondary indexes or other access paths) are available on more than one of the fields involved in simple conditions in the conjunctive condition, and if the indexes include record pointers (rather than block pointers), then each index can be used to remove the set of record pointers that satisfy the individual condition. The intersection of these sets of record pointers gives the record pointers that satisfy the conductive their satisfy the condition. The intersection of these sets of record pointers gives the record pointers that satisfy the conductive records directly. It only some of the conditions have secondary indexes, each removed record record is forther to set to determine whether it satisfies the remaining conditions.

Whenever a single condition specifies the selection — such as O(1), O(2), or O(3-4), contoully check whether an access path exists on the attribute involved in that condition if an access path exists, the method corresponding to that access path is used, atherwise, the brate force linear search approach of method S1 can be used. Query optimization for a FEHA.T operation, is needed mostly for conjunctive select conditions whenever more than one of the attributes involved in the conditions have an access path. The optimization bould choose the access path that retrieves the fearer teeroids in the most efficient way by estimating the different costs (see Section 15.8) and choosing the method with the least estimated cost.

When the optimizer is choosing between multiple simple conditions in a conjunctive select condition, at typically crusiders the selectivity of each condition. The selectivity (s) is defined as the ratio of the number of records (tiples) that satisfy the condition of the total number of records stuples) in the file (selation), and thus is a number between zero) and 1—zero is electivity means the records satisfy the conditions and 1 means all the records satisfy the condition. Although exact selectivities of all conditions may not be available, **estimates of selectivities** are often kept in the DBMs catalog and are used by the optimizer. For example, for an equality condition on a key arribate of relation  $\tau(R)$  is the number of tuples in relation  $\tau(R)$ . For an equality condition on an attribute with *educate tubes*, years be estimated by (1)(R)1/0/1/0/1) or

<sup>6.</sup> A record pointer ortiquely identifies a record and provides the oldress of the record on disk before, it is also called the **record** identifier or record id-

The technique can have many variations—for example, if the indexes inclosed indexes rise store primary key values margad of record pointers.

16 assuming that the records are evenly distributed among the distinct values? Under this assumption, 16301/i records will satisfy an equality condition on this attribute. In grand, the number of records satisfying a selection combtion with selectivity values estimated to be  $15(R)1.8 \times$  The smaller this estimate is, the higher the dosirability of using that conditions first to retrieve records.

Compared to a comparence selection condition, a disjunctive condition (where sinple conditions are connected by the CR legical connective rather than by AND) is much harder to process and optimize. For example, consider  $0.04^{\circ}$ :

(CP49): DIENT VIDE SALKEMATING OF MEXAMIN (EMPLIATEE)

With such a condition, little optimization can be done, because the records sansying the disjunctive condition are the moon of the records sanstying the individual conditions. Hence, if any one of the conditions does not have an access path, we are compelled to use the brain force linear search approach. Only if an access path exists on every condition on we continize the solection by retrieving the records sourching each conditions—or their record ids— and then applying the union operation to climinate doplicates.

A 19846 will have available many of the methods discussed above, and equivally many additional methods. The query optimizer must choose the appropriate one for concuring each SELECE operation in a query. This optimization uses formulas that commute the costs for each available access method, as we shall discuss in Section 15.8. The optimizer chooses the access method with the lowest estimated cost.

### 15.3.2 Implementing the JOIN Operation

The 50% operation is one of the most time-consuming operations in query processing. Many of the join operations encountered in queries are of the EQUIOIN and NATURAL IOIN vineties, so we consider only these two here. For the remainder of this chapter, the term point refers to in EQUIORS (or NATURAL JOIN). There are many possible ways to implement a two-way juin, which is a join on two files. Jeths involving more than two files are called multiway joins. The number of possible ways to execute influences yours eroses very rapidly. In this section, we descute techniques for implementing only two-way joins. To influence or discussion, we refer to the relational schema of Figure 5.5 ence more—specifically, to the FME net. OFFACTMENT, and BADECT relations. The algorithms we consider are for join operations of the form.

8 M ... 5

where A and B are domain-compatible attributes of R and S, respectively. The methods we discuss can be extended to more general turns of join. We illustrate four of the most common techniques for performing such a join, using the following example operations:

(CPG); CM/LCYLL <sup>DC</sup> <sub>MCCONTREP</sub> CHEVERINES (CP7): CR /ORIVINE <sup>DC</sup> <sub>MCCONT</sub>SCH TMPT0-11

8 In more sophisticated optimizers thistograms representing the distribution of the records around the difference attribute values care by kept in the carolog.

Methods for Implementing Joins

- [1] Nester-loop y-in (*limite force*). For each record 1 in R (outer loop), remove even record s from S (unner loop) and rest whether the two records satisfy the juni condition ([A] = 3[H]?
- Single-hop part (mag on access structure to remete the matching occords): If an index (or hash key) exists for one of the two join attributes—say. B of an introve each record can R, one at a time (single loop), and then use the access structure to remeter directly all matching records s from S that satisfy [9] = (A)
- J3. Sost merge joint If the records of R and S are physically sorted (ordered) by value of the join attributes A and B, respectively, we can implement the join in the meter efficient way possible. Both files are scanned concurrently in order of the join attributes, matching the records that have the same values for A and B. If dirithes are not sorted, they may be sorted test by using external sorting (see Secon 15.2). In this merilied, pairs of the blocks are copied into memory buffers in order at the other the records of each the are nonkey attributes, in which case the method needs to be predicted slightly. A sketch or the sortenerge join algorithm is given in Figure 15.3a. We use R(d) to refer to the 3<sup>th</sup> record in *R*. A variation of the sort-merge join dattributes, the indexes provide the ability to access (scan) the records in order of the join attributes, but the records the method when secondary indexes exist on both join armbutes. The indexes provide the ability to access (scan) the records in order of the join attributes, but the records the method when secondary indexes of the poly order decord allower the join attributes, but the records the method when secondary indexes other mode an order of the join attributes, but the records the method when secondary indexes other does an other when the join attributes at other and the records the method of the join attributes at the records the method when we access any the records access may involve accessing a different disk block.
- J4 Hasimon The records of files *B* and S are both bashed to the same bashite using the same bashite function on the join artifibres A of R and B of S as his keys. First, a sizele pass through the file with featur records (say, R) hashes is records to the bash file buckets, this is called the partitioning phase, since the records of *R* are partitioned into the bash buckets. In the second phase, called the probing phase, a single pass through the other file (S) then bashes each of *s* records to probe the appropriate backet, and that record is combined with all matching records from *B* in their bocket. This simplified description of hashesn's phase, that the smaller of the two files fits entriely not thereory backets after the test phase. We will discuss variations of bash-join that do not require the assumption below.

In practice, techniques [1] to j4 one implemented by accessive rehabilities blocks of a idea after than individual records. Depending on the available butter space in memory the number of blocks read in from the file can be adjusted.

For disk files, it is obvious that the loops will be over disk blocks so this reclinique has also been called nested-block pro-

```
(8) surfille tuples in Pion affricule A. Caseume Rihas hituties (records) 15
 sort the tuples in 5 on altribute 6: infassume 5 has in tuplos (records) ()
 set in the set of
 while (15 m) and (15 m).
 ac_i = f A[A]A[> S[A]B]
 then
 set /4 /1*
 eise* RahlAl < SahBl
 56176-041
 theu
 (* P(i)[A] = S(i)(B), so we output a matched tuple').
 obe i
 subut the combined highle \langle F(t) \rangle = S(t) > I \cap T.
 ('oulput other hiples what metch (R/) of enviro
 set /. . /+1;
 while ((< m) and (Ph)(A) = 5(1)(B)).
 do.
 Т
 culpd like combined table «Fiti», S(h> to 7;
 seri. Jen
 L
 ("conput other topics that match 5(%, if any to
 867 Kull 2+1)
 wore (k \in \mathbb{N}) and (B[k][A] = S[i][B]).
 do
 Ĺ
 output me combined topic «R(A), S(J)» to 7:
 981 8 . 4 -1
 ł
 901 A 11 1. j. (+1
 Ł
 ı
(b) create a tupla ((kattribute lists) in 7) for each tuple (in F);
 ("T' contains the projection result before duplicate alimination")
 il valtribute lists includes a key of R
 then T + T'
 alsa ;
 solutibe types in T':
 set (11) - 2:
 while 75 m

 cotact the tuple 7 (i) to 7

 da í
 while T[i] = T[j] and j \in \sigma do j \in j+1, ("altrainate dependents").
 16 a (S. 1971)
 ı
 I
 [17 contains the projection result after suplicate domination.")
```

**FIGURE 15.3** Implementing  $\mathcal{O} \times$ , "POIFCT, UNION, INTERSECTION, and SET OFFERENCE by using sort-morge, where *R* has naturally and *S* has matuples: (a) implementing the operation  $T \leftarrow R \bowtie_{n=0} S$ . (b) implementing the operation  $T \leftarrow \pi_{n, \text{production}}(R)$ 

Effects of Available Butter Space and Joan Selection Factor on Join Performance. The bufter space available has an important effect on the various join algorithms. First, let us consider the rested-loop approach (JTE Looking again at the inpranon GPS above, assume that the number of butters available in manimercory for implementing the join is  $n_0 = 7$  block. (butters), for shortavien, assume that the **DEPAT**, we file consists of  $n_0 = 50$  records stored in  $n_0 = 70$  blocks and that the **DEPAT**.

ı

```
(c) som the luples in H and S using the same unique exit abilities.
 6017-17-1
 while (i similand () simil
 ii £iğ ≥ S(j)
 240 (
 then |
 purput S(j) to T
 1011-04
 ı
 भेरती तीन य इंग्रेन
 then ;
 pulpul A) (arc T;
 381 I--1+*
 L
 size yet (+ /+1 (16)) - S()) so we say one of the duployee tupley".
 L
 if (i \le n) than edd upples E(i) to F(n) to T_i.
 If l \leq m then add tuples S(l) to S(m) to T.
(d) earline typics in R and S using the same unique part attributes:
 990 to 1, to 1,
 while the related to and
 f(\theta|y) > S(y)
 de (
 ther
 5857-141
 planed H(t) = S(t)
 then.
 58⁴ - 1+1
 else (
 numple \Pi(t) to \mathcal{P} \in \Pi(A \circ S(t), so we conjuditionaple <math>\Omega
 560 (- 1+1, j. 7+1
 э
)
(a) – so the tuples in R and S using the same unique sof althoutes:
 6817-1174-1
 while (/ S of and (; S or)
 do 🌾
 f H(i) > S(i)
 Ker.
 58174-741
 EVALUATE: B(i) = B(j)
 surpul Regio T: (1.61) has no makting S(/), so oviput P(a1)
 Iher I
 NULL INT
 1
 el se
 BHL1- .-1 .+ J+1
 5
 4.1 <= 20 if en add tuples R(1) to R(2) ifs D</p>
```

**FIGURE 15.3(CONTINUED)** Implementing  $x \ge x$ , incuted, onlines in these non, and set therefore,  $z \ge x$  and  $z \ge x$ , there R has a suples and S has in tuples, i.e. implementing the operation  $T \leftarrow R \cup S$ . (if) implementing the operation  $T \leftarrow R \cap S$ , its implementing the operation  $T \leftarrow R \cap S$ .

consist- of  $y_1 = 6000$  records stored at  $h_1 = 2000$  disk. Nocks this advantageous in reading many blocks as possible at a time into moment from the file whose records are used for the outer loop (that is,  $n_0 = 2$  blocks). The algorithm can then read one block at a time in the inner-loop file and use its records to probe (that is, search) the outer loop blocks in innerview for matching regards. This reduces the total number of block accesses. An exity buffer block is reselfed to contain the resulting records after they are joined, and the row tents of this buffer black are appended to the result file- the disk ide that contains the pro-result - whenever it is filled. This buffer block is then is reused to hold additional result regards.

In the nested-loop join, it makes a difference which file is chosen for the outer loop and which for the inner loop. If general is used for the outer loop, each block of renoval is read once, and the entire presence file (each of its blocks) is read once for each one we read in  $(n_s - 2)$  blocks of the uncovar file. We get the following:

Total number of blocks accessed for outer file =  $b_{\mu}$ 

Number of times  $(n_B - \beta)$  blocks of outer file are loaded =  $\lceil b_{\rm E} / (n_B - \beta) \rceil$ 

Total number of blocks accessed for inner file  $= b_{\rm B} * ||| b_{\rm F} f(a_{\rm B} - 2) \frac{1}{2}$ 

Hence, we get the following total number of block processes.

 $b_0 + 4 \left[ b_{\rm F} / (a_{\rm B} + 2) \right] + b_0 0 + 2000 + (|(2000/3)|] + 10) = 0000$  block accesses

On the other hand, if we use the **DOWTHEN** records in the outer loop, by symmetry we get die following rotal number of block occesses:

 $b_0 + ([b_0](n_0 - 2)^2 + b_0) - 10 + (!_1(0/5)] + 2000) = 4010$  block accesses

The join algorithm uses a butter to hold the jointed records of the tesult file. Once the butter is filled, it is written to disk and reused.<sup>15</sup> If the result file of the join operation has  $b_0$ , disk blocks, each block is written once, so an additional  $\hat{b}_{DS}$  block accesses should be added to the preceding formulas in order to estimate the total cost of the join operation. The same holds for the tonnulas developed later for other joint algorithms. As this example shows, it is advantageous to use the file ust force blocks as the outer-loop file in the next form.

Another factor that affects the performance of a join, particularly the single-loop method J2, is the percentage of records in a tile that will be joined with records in the other blc. We call this the join selection factor<sup>11</sup> of a file with respect to an equijoin indition with another tile. This factor depends on the particular equijoin condition between the two files. To allow not this, consider the operation of 7, which joins each 2000 the two files. To allow no this, consider the operation of 7, which joins each 2000 the two files are 50 such records in our example) is expected to be joined with the performing upperformance of the 5952 of them that do not upmage a department) will not be joined.

Suppose this secondary indexes exist on both the attributes 550 of EP2,0766 and 90.6766 of EP2,0766 and 90.6766 of EP2,0766 and 90.6766 of EP2,0766 and 90.6766 a

R. If we reserve two buffers for the result file, double bufferoug can be used to speed the algorithm see Section 13.35.

This is different from the join schemary, which we shall discuss in Section 15.8.

marching record will be found for employees who do not manage a department. The number of block accesses for this case is approximately

$$b_{0} + (r_{\mu} + (r_{\mu,m,0} + 1)) = 2000 + (6000 + 3) = 30,000$$
 block accesses

The second option actrieves each presences record and then uses the index on sworf precett to find a matching manager correct record. In this case, every presences record will have one matching regions, meanl. The number of block accesses for this case is approximately.

 $b_{21} + (a_{12} + (b_{23} + 1)) = 10 + (50 + 5) = 260$  block accesses

The second option is more efficient because the som selection factor of **DENTRES** with respect to the join condition SN = **MENOR** is 1. Whereas the join selection factor of **ENELOYEE** with respect to the same join condition is 450/60000, or 00000. For method [2, either the smaller file or the file that has a match for every record (that is, the file with the fact, join selection factor) should be used in the (outer) join loop. It is also possible to create an index specifically for performing the join operational one does not already exist

The sent-merge join [5 is quite efficient if both file- are already softed by then join attribute. Only a single pass is finide through each file. Hence, the number of blocks increased is equal to the sum of the numbers of blocks to both files. For this method, both ODS and CP7 would need  $b_1 = b_0 = 2000 \pm 10 = 2010$  slock accesses. However, both files are required to be ordered by the join attributes. Force or both are not, they may be sorted spectraally for performing the join operation. If we estimate the rest of softeng an external file by  $(b \log b)$  blick accesses, and it both files need to be softed, the total cost of a sufficiency juin can be estimated by  $(b_1 + b_1 + b_1) = b_0 + b_1 + b_1 \log b_0 + b_0 + b_0 \log b_0 b^{-1}$ .

Partition Hash join and Hybrid Hash Join. The hish-join method [4] is also unterefficient. In this case only a single possis minderbrough each file, whether or not the files are ordered. If the hash table for the smaller of the two files can be kept entriely in main memory after hashing (partitioning) on its join attribute, the implementation is straightforward. If, however, parts of the hash file failst be stored on disk, the method becomes more complex, and a number of variations to improve the efficiency have been proposed. We discuss two technologies (partition hash join and a variation called hybrid hash join, which has been shown to be quite efficient.

In the partition hash join algorithm, cach file is instipartitioned into M partitions using a partitioning hash function on the join attributes. Then, each pair of partitions is pared. For example, suppose we are joining relations & and S on the join attributes *R*.A and S B:

Richard

In the partitioning phase, *K* is partitioned into the M partitions  $R_1, R_2, \ldots, R_M$ , and S into the M partitions  $S_1, S_2, \ldots, S_M$ . The property of each part of corresponding partitions  $R_1, S_2$  is that records an  $R_1$  only used to be pured with records in  $S_2$  and vice versa. This projectly is epsilted by using the same hash function to partition both files on their

<sup>12.</sup> We can use the epine ray mare formulas from Systems 15.7 it as known the number of non-lable butters for setting

join auch ores—attrabute A for R and outrobute B for S. The intrimum number of mmemory buffers needed for the partitioning phase is  $M \approx 1$ . Each of the files R and S are partitioned separately For each of the partitions, a single in-memory buffer—whose size is one disk block—is allocated to store the records that hash to this partition. Whenever the m-memory buffer for a partition gets filled at contents are appended to a disk subfile that stores this partition. The partitioning phase has neo iterations. After the first instation, the first file R is partitioned into the subfiles  $R_1, R_2, \ldots, R_M$ , where all the isotes that hashed to the same buffer are its the some partition. After the second instation, the second file S is similarly partitioned.

In the second phase, colled the joining or probing phase, M detailors are needed. Daring iteration is the two partitions  $R_i$  and  $S_i$  are joined. The minimum number of biffers needed for iteration is the number of blocks in the smaller of the two partitions, as  $R_i$  plus two additional buffers. It we use a nested loop join during iteration is the acords from the smaller of the two partitions  $R_i$  are copied into memory buffers: then all books from the smaller of the two partitions  $R_i$  are copied into memory buffers: then all blocks from the other partition  $S_i$  are read—one at a time —and each record is used to probe (that is, search) partition  $R_i$  for matching record(s). Any matching records are poined and written into the result file. To improve the efficiency of momentary probing, it is common to use in *in-memory hash table* for storing the records in partition  $R_i$  by using a different hash function from the garmoning hash turction.<sup>1</sup>

We can approximate the cost of this partition hash join as  $3 + (b_p + b_q) + b_{gl,N}$  for our example, since each record is read once and written back to disk once during the partitioning phase. During the joining (probing) phase, each record is read a second time to beform the join. The mean *difficulty* of this algorithm is to ensure that the partitioning hash function is **uniform** that is, the partition sizes are nearly equal in size. If the partitioning function is **uniform** that is, the partition sizes are nearly equal in size. If the partitioning function is **skewed** from only items, then some partition is not begin to be used on the available memory space for the arcoid joining phase.

Nonce that if the available inducency butter space  $u_0 \geq (b_0 + 2)$ , where  $b_0$  is the number of blocks for the stellar of the two files being joined, say  $B_i$  then there is no reason to do commonize since in this case the ion car be performed enough in memory using some variation of the nested-loop join based on hashing and probing. For illustration, assume we are performing the join operation (06), repeated below

#### (395) : AMALOYEE M CRIMINEER DUPAN MANT

In this example, the smaller file is the cussions of file; hence, if the number or available nemory butters  $n_0 \ge (b_0 + 2)$ , the whole cussions file can be read into main memory and significal into a hash table on the join attribute. Each express block is then read into a bitter and each executive record in the botter is hashed on its join attribute induction prior attribute induction of prior butters proved in the records are joined, and the result record(s) are written to the result file on each executive to the result file on each economic of the result file on each second is the records are joined, and the result record(s) are written to the result butter and eventually to the result file on each. The cost in terms of block accesses is hence  $(b_0 + 1_0)$ , plus  $b_{0.05}$ —the cost of writting the result rile.

<sup>1)</sup> If the leash function used for partitioning is used again, all records in a partition will hash to the size bucker again.

The hybrid hash-join algorithm is a variation of partition hash join, where the poloing phase for one of the formany is included in the territoring phase. To illustrate this, let as assume that the size of a memory buffer is one disk block, that  $n_{\rm B}$  such buffers are available, and that the lash function used o h(K) = K and M so that M partities are being created, where  $M \leq n_{\rm B}$ . For illustration, assume we are performing the pure operation. OPs, in the first pass of the partitioning phase, when the first pass of the partitioning phase, when the first pass of the partition in order (browsteet in OP6), the algorithm divides the buffer space analytic of the M partitions such that all the blocks of the join partitions of creation of creations (completely reside in main memory. For each of the other partitions, only a single memory buffer—whole size is one disk block —is allocated; the remainder of the partition is written in disk as in the regular partition blocks of the jum. Hence, at the end of the first pass of partitions each of the regular partition blocks of the jum. Hence, at the end of the first pass of phase and partitions of the partition of creations is allocated; the remainder of the partition is written in disk as in the regular partition bash jum. Hence, at the end of the first pass of the buffer space or other buffer, whereas each of the other partitions of creations is becaused.

For the second pass of the partitioning phase, the records of the second file being pointed—the larger file, **see even** in  $\partial t^{6}$ —are being partitioned. If a record hashes to the *fixe* partition, it is joined with the matching record in tension and the joined records are written to the result buffer (and eventually to disk). If an second record hashes to a partition rather than the first, it is partitioned normally Hence, at the end of the second pass of the partitioning phase, all records that bash to the first partition have been joined. Now there are M = 1 partitions on disk. Therefore, during the second joining or probing phase, M = 1 detailed are norded instead of M. The goal is to put as many records during the partitioning phase are so as to save the enst of suring those records back to disk and perioding them a second rune during the joining phase.

### 15.4 ALGORITHMS FOR PROJECT AND SET OPERATIONS

A fig01001 operation  $\pi_{-instance}$  (*RO*) is straightforward to implement it s attribute lists includes a key of relation *R*, because in this case the result of the operation will have the same number of implex as *R*, but with only the value- for the attributes in statistical will have the same number of implex as *R*, but with only the value- for the attributes in statistical will have the same number of implex as *R*, but with only the value- for the attributes in statistical will have the same number of implex as *R*, but with only the value- for the attributes in statistical will have the same number of implex as *R*, but with only the value- for the attributes in statistical will have the same number of implex as *R*, but with only the value- for the attributes in statistical attributes in statistical will have the same number of implex is *R*, but with only the result of the operation and then charmaning displace tuples, which appear consecutively after storing. A sketch of the information given in Figure 15.3b. Hashing can also be used to charmate displace as each record is hashed and anserted into a bucket of the bash file in memory in is checked against these already in the backet, after is a duplicate, it is not inserted. It is useful to recail here that its SQU queries, the default is not to eliminate duplicates from the query result, only if the keyword DISTINCT is included are duplicates eliminateed from the query result.

Set operations – UNION, INTERFECTION, SET DIFFERENCE, and CORTESIAN FROMULT – are conclumed expensive to implement. In particular, the CARTESIAN PRODUCT operation  $R \propto S$  is quite expensive. Excluse its result includes a record for each combination of

records from R and S. In addition, the attributes of the result metade all attributes of R and S. If R has n records and j attributes and S has a records and k attributes, the result relation will have  $n \approx m$  records and  $j \neq k$  attributes. Hence, it is important to avoid the CARTESTAN TROPT OF operation and to substitute other equivalent operations during query optimization (see Section 13.7).

The other three set operations: -UNCN, UNTER-EUTION, and SET DIMERESCE<sup>15</sup> – apply only to amon-compatible relations, which have the same number of attributes and the same attribute domains. The customary way to implement these operations is to use estimizes of the **sort-merge technique**: the two relations are sorted on the source attributes, and, after strong, a single ston through each relation is sufficient to produce the result. For example, we can implement the SO(OS operation,  $B \cup S$ , by scanning and merging both, sorted inles concurrently, and whenever the same tuple exists in both telations, only one is kept in the merged result. For the INTERSECTION operation,  $R \cup S$ , we keep in the merged result. For the INTERSECTION operation,  $R \cup S$ , we keep in the merged result for the INTERSECTION operation,  $R \cup S$ , we keep in the merged result of these uples that appoint in both relations. Figure 15.3, to (c) sketches the implementation of these operations by sorting and merging. Some of the details are not included in these algorithms.

**Hashing** can also be used to implement CNION. NTERSECTION, and SET DIFFERENCE. One table is partitioned and the other is used to probe the appropriate partition. For example, to emplement  $R \cup S$ , first both (partition) the records of R; then, hash (probe) the records of S. But do not insert depletate records in the backets. To implement  $R \cap S$ , not partition the records of R to the hash file. Then, while hashing each record of S, probe to check it an identical record from R is round in the backet, and it so add the record to the result file. To implement R = S, first hash the records of R to the hash file backets. While bashing (probably each record of S, it an identical proof is found in the backet, remove that record from the backet.

## 15.5 IMPLEMENTING AGGREGATE OPERATIONS AND OUTER JOINS

#### 15.5.1 Implementing Aggregate Operations

The aggregate operators (MIN, MAN, COUNT, WER WE, PUM), when applied to an entire take, can be computed by a table scan or by using an appropriate index, if available. For example, consider the following SQL query.

```
SELECT NAX(SALARY)
FROM EMPLOYEE:
```

If an (ascending) index on 50000 exists for the 090.0000 relation, then the optimizer can decade on using the index to search for the largest value by following the optimist pointer in each index uside from the tost to the rightmost leaf. That node would include

```
IA ST PPERSESS So all developments of a
```

the largest \$4.007 value as its last entry. In most cases, this would be more efficient than a full table scan of environe, since no actual records need to be retrieved. The SUN aggregate can be bandled on a similar manner, except that the *leftmost* pointer is followed from the root to Jeffmost leaf. That node would include the smallest score value as its *first* entry.

The index could also be used for the COMNT, AVERAGE, and SUM aggregates, but only if it is a **dense index**—that is, if there is an index entry for every record in the main file. In this case, the associated computation would be applied to the values in the index. For a **nondense index**, the actual number of records associated with each index entry must be used for a correct computation (except for COUNT DISTINCT, where the number of distinct values can be counted from the index itself).

When a GROOP BY clause is used in a query, the aggregate operator must be upplied separately to each group of tuples. Hence, the table must first be partitioned into subsets of tuples, where each partition (group) has the same value for the grouping attributes. In this case, the computation is more complex. Consider the following query:

```
SELECT DNO, AVGUSALARY)
HROM EMPLOYEE
GROUP BY DNO:
```

The usual technique for such queries is to first use either, sorting or hashing on the grouping attributes to partition the file into the appropriate groups. Then the agorithm computes the aggregate function for the tuples in each group, which have the same grouping attribute(s) value. In the example query, the set of tuples for each department number would be grouped together on a gratition and the accrage salary computed for each group.

Notice that if a clustering index (see Chapter 13) exasts on the grouping attribute(s), then the records are already paralalated (grouped) just the appropriate subsets. In this case, it is only necessary to apply the computation schedule group.

#### 15.5.2 Implementing Outer Join

In Section 6.4, the order job operation was introduced, with its three variations, left eater join, right outer join, and full outer join. We also discussed in Chapter 8 how these operations can be specified in SQL. The following is an example of a left eater join operation in SQL:

```
SELECT LNAME, FNAME, DHAME
FROM (EMPLOYEE LEFT DUTER DOIN (PEPARIMENT ON (POEDNUMBER);
```

The result of this query is a table of employee names and their associated departments. It is similar to a regular (inner) pair result, with the exception that it an exception to a figuration does not have an associated department, the employee's name will coll appear in the resulting table, but the department name would be righter such toples in the query result.

Outer join can be compared by modifying one of the join algorithms, such as nestedloop join or single-loop join. For example, to compare a left outer join, we use the left relation as the outer loop or single-loop because every tuple in the left relation must wpractar the result. If there are noticling tuples in the atten relation, the prince tuples in produced and sived in the result. However, it no matching tuple is found, the tuple is will included in the result but is padded with rull value(s). The sort anerge and bash-join algorithms can also be extended to compute outer joins.

Alternatively, outer join can be computed by executing a combination of relational algebra operators. For example, the left outer join operation shows above as equivalent to the following sequence of relational operations:

Compute the (inner) IOIN of the presence and reported tables.

Find the EPELINEE tuples that do not appear in the (miter) KHN result.

```
(1402 \leftarrow \pi_{(KAME, KAME} (1402)) = \pi_{(MAME, MAME} (11403)
```

3. Pad each ruple in tene2 with a null take held.

TEMP2 ← TEMP2 × 'NULL'

4. Apply the UNION operation to 1994, 1992 to produce the UFPT OUTER (OIN result.

The cost of the inner join as computed above would be the sum of the costs of the associated steps (miner join, projections, and union). However, note that step 3 can be done as the temporary relation is being constructed in step 2; that is we can simply pad ach reaching tuple with a null. In addition, in step 4, we know that the two operands of the option are disjoint (no common tuples), so there is no need to duplicate elimination.

# 15.6 COMBINING OPERATIONS USING PIPELINING

A query specified in SQL will typically be translated into a relational algebra expression that is a sequence of relational operations of we execute a single operation at a time, we must generate temperary ides on disk to toold the results of these temporary operations, creating excessive overhead. Clenerating and storing large temporary files on disk is timeconsuming and can be unnecessary in many cases, since these files will connectiately by used as input to the next operation. To reduce the number of temporary files, it is complea to generate query execution code that correspond to algorithms for combinations of operations in a specty.

For example, rather than being implemented separately, a JOIN can be combined with two 50 ECT operations on the input files and a final PSO(ECT operation on the resolving iderall this is implemented by one department with two input files and a single parput file Rather than creating four reinportant files, we apply the algorithm directly and get just one result file. In Section 15.7.2 we discuss how heuristic relational algebra optimization can group operations together for execution. This is called pipelining or stream-based processing. It is common to create the overy execution code dynamically to implement multiple operations. The generated code for producing the query combines several algorithms that correspond to individual operations. As the result roples from one operation are produced, they are provided as input for subsequent operations. For example, if a join operation follows two select operations on base relations, the tuples resulting from each select are provided as input for the join algorithm in a stream or pipeling as they are produced.

## 15.7 USING HEURISTICS IN QUERY OPTIMIZATION

In this section we discuss optimization techniques that apply bein stie rules to mobily the internal representation of a query —who has usually an the form of a query free or a query graph data structure—to improve its expected performance. The parset of a high-level query first generates an easilit internal representation, which is then optimized according a bearistic rules. Following that, a every execution plan is generated to execute groups of operations based on the access paths available on the files moders!

One of the man heuristic rules is to apply SEECT and DOJ-CT operations before applying the DDN or other binary operations. This is because the size of the file resulting from a binary operation—such as [ODN] as usually a multiplicative function of the sizes of the apparticles. The SEECT and DROJECT operations reduce the size of a file and hence should be applied force a join or other binary operation.

We start in Section 15.7.1 by introducing the query tree and query graph normations. These can be used as the basis for the data structures that are used for internal representation of queries. A query tree is used to represent a relational inference and calculus expression. We then show in Section 15.7.2 how heatistic optimization tales are applied to convert a query tree into an equivalent query tree, which represents a different relational algebra expression that is more officient to exercise but gives the same result as the original one. We also discuss the equivalence of various relational algebra expressions. Finally Section 15.7.3 discusses the generation of query execution plans.

### 15.7.1 Notation for Query Trees and Query Graphs

A query tree is a tree data structure that corresponds to a relational algebra expression. It represents the input relations of the query as log-index of the tree, and represents the relational algebra operations as internal nodes. An execution of the query tree consists of executing an internal node operation whenever its operation are available and then replacing that internal node by the relation that results from executing the operation. The execution remains the relation that results from executing the operation. The execution remains the relation that results from executing the result relation that resoluted and produces the result relation for the query.

Figure 15.4a shows a query tree for query Q2 of Chapters 5 to 8. For every project located us 'Stafford', retrieve the project number, the controlling department number.



FIGURE 15.4 Five query trees for the query Q2 (a) Query tree corresponding to the relational algebra expression for Q2, (b) Initial transmittable query tree for sQ1 query Q2.

and the department manager's last name, address, and berhdate. This query is specified on the relational schema of lague 5.5 and corresponds to the following relational algebra expression

```
 \begin{array}{l} \hline & \displaystyle \mbox{Prove}_{k,k,m} ((((\tau_{k}, \mathfrak{s}_{k+1}, \mathfrak{s
```



FIGURE 15.4(CONTINUED) (c) Query graph for Q2.

This corresponds to the following OU query:

```
Q2: SELECT P.PNUMBER, P.DWLM, E.LMANE, E.AOGRESS, E.BUATE
FROM PROJECT AS P. DEPARTMENT AS D. EMPLOYEE AS E
WHERE P.DNUM-O.DNUMBER AND O.MCRSSN-E.SSN AND
P.PLOCATION-'SIA-'CRP';
```

In Figure 15.4a the three relations isotric, treasmost, and twitter are represented by leaf nodes P. D. and E. Schile the relational algebra operations of the expression are represented by internal tree nodes. When this query free is executed, the node marked (1) in Figure 15.4a must begin execution before node (2) because some resulting tuples of operation (1) must be available before we can begin executing operation (2). Similarly, node (2) must begin executing and producing results before node (3) can start execution, and so on.

As we can see, the query tree represents a specific order of operations for executing a query. A more neutral representation of a query is the query graph notation. Figure 15.4, shows the query graph for query Q2. Belations in the query are represented by relation nodes, which are displayed as single circles. Constant values, typically form the query selection conditions, are represented by constant nodes, which are displayed as double circles or evals. Selection and join conditions are represented by relations are represented by relations are represented by relation and solver in Figure 15.4c. Finally, the articlores to be remeved from each relation are displayed in square brackers above each relation.

The query eraph representation does not indicate an order on which operations to parterni first. There is only a single graph corresponding to each query.<sup>15</sup> Although some optimization techniques were based on query graphs, it is now generally accepted that query trees are preferable because. In practice, the query optimizer needs to show the order of operations for query execution, which is not possible in query graphs.

15. Hence, a quere graph corresponde to a relational calculus expression (see Chapter 6)

### 15.7.2 Heuristic Optimization of Query Trees

In general, many different relational algebra expressions , and hence many different query trees --con be equivalent; that is, they can correspond to the same query.<sup>16</sup> The query parter will typically generate a standard initial query tree to correspond to an SQ query, without doing any optimization. For example, for a select-project-ion query such as Q2, the initial tree is shown in Figure 15.45. The CARTES AN PRODUCT of the relations specified in the FROM clause is first applied; then the selection and join conditions of the WHERE clause are applied, followed by the projection on the SELECT clause attributes. Such a canonical query tree represents a relational algebra expression that is cory *mellitent glaces* and contained 120, 20, and 5020 cuples, respectively, the result of the CARTESIAN FROUGET of the CARTESIAN PRODUCT (\*) operations. For example, if the PROJECT, DEPARTMENT, and EVENTEE relations had record sizes of 100, 50, and 150 bytes and contained 120, 20, and 5220 cuples, respectively, the result of the CARTESIAN PRODUCT would contain 10 million tuples of record size 300 bytes each. However, the query tree in Figure 15.46 is in a simple standard form that can be easily meated, it is row the job of the heuristic query optimizer to transform this unitial query tree into a final query tree that is efficient to execute.

The optimizer must include rules for occuvalence among relational algebra expressions that can be applied to the minglitree. The heuristic query optimization rules then utilize these equivalence expressions to transform the initial tree into the final, optimized query tree. We first discuss informally how a query tree is transformed by using hearistics. Then we discuss general transformation to les and show how they may be used in an algebraic heuristic optimizer.

Example of Transforming a Query. Consider the following query Q on the Jatabase of Figure 5.5. 'Find the last names of employees been after 1957 who work on a project named 'Aquarius'.' This query can be specified in SQL as follows:

```
Q: SELECT LNAME
FROM EMPLOYEE WORKS_ON, PROPECT
WHERE PNAME=TAQUERIUS' AND PROPEREPRO AND ESSN=SSN
AND BOATE > '1957-12-31';
```

The nurval query tree for Q is shown in Figure 15.55. Executing this tree directly first creates a very large file containing the CARTESIAN PRODUCT of the entire environce across to, and protect files. However, this query needs only one record from the sporter relation for the 'Aquarus' project—and easy the spectrum records for those whose date of birth is shere '1957-12-31'. Figure 15.55 shows an improved query tree that first applies the SUECT operations to reduce the number of tig-les that appear inside CARTESIAN PRODUCT.

A further suprovement is achieved by switching the positions of the EPEDYEE and ENGLECT relations in the tree, as shown in Figure 15.5c. This uses the information that ENGLECE is a key attribute of the project relation, and hence the SelECT operation on the

16. A query may also be acated in various ways in a linch-level query language such as 8(3) (see Chapter 8).



**EVGURE 15.5** Steps in converting a query tree during heuristic optimization. (a) Initial manonicall query tree for SQL query  $Q_1$  the Moving SERT operations down the query tree.



**FIGURE 15.5(CONTINUED)** Steps in converting a query free doring heuristic upsimization. (c) Applying the more restrictive SEGCT operation first, di Replacing 4 ART(SIAN PRODUCT) and strECT with KON operations



HGURE 15.5(CONTINUED) Steps in converting a query tree during heuristic optimization. (c) Moving PROJECT operations down the query tree

**SOULT** relation will settice a single received only. We can further improve the query tree by replacing any CARTESIAN (RODUCT operation that is followed by a join condition with a RON operation, as shown in Figure 15.5d. Another top revenent is to keep only the attributes meshed by subsequent operations in the intermedicite relations, so including source ( $\pi$ ) operations as early as possible in the query free, as shown in Figure 15 fe. This reduces the attributes (coloring) of the intermediate relations, whereas the SED f operations reduce the number of tuples (records).

As the preceding example denonstrates, sequery free can be transformed step by sepinto another every new that is more efficient to execute. However, we must make sue that the transformation steps always lead to an equivalent query tree. To do this, de query optimiter must know which transformation rules preserve this controllence. We discuss some of these transformation rules next.

Conteral Transformation Rules for Relational Algebra Operations. There are many rules for transforming relational algebra operations into equivalent ones. Here we are interested in the meaning of the operations and the resulting relations. Hence, it facrelations have the same set of armituites in a different order but the two relations represent the same information, we consider the relations equivalent. In Section 5.1.2 we give an exemptive definition of teleane that makes order of attributes unimportants we will use this definition here. We now state some transformation rules that are useful in query optimization, without proving them:

 Cascade of 0: A conjunctive selection conditions can be broken up into a cascade (that is, a sequence) of individual o operations;

$$\sigma_{i_1,2NECCANE} = _{ANECC} \{R\} = \sigma_{i_1} \{\sigma_{i_2} \} = \langle \sigma_{i_2}(R) \rangle = i \}$$

2. Commutativity of it: The 0 operation is commutative

 $\sigma_{c1}\left(\sigma_{c2}(R)\right)=\sigma_{c2}\left(\sigma_{c1}(R)\right)$ 

 Chocade et #: In measurable (sequence) of # operations, all put the last one can be ignored;

 $\pi_{\text{Locl}}\left(\pi_{\text{Locl}}\left(1,\ldots,\left(\pi_{\text{Locl}}(R)\right),\ldots\right)\right)=\pi_{\text{Locl}}(R)$ 

 Commuting σ with π. If the selection condition c involves only three attributes A1, ... As in the projection list, the two operations can be commuted:

 $\pi_{A_{1},A_{2}},\ldots,A_{n}\left(\sigma_{k}\left(\beta\right)\right)=\sigma_{k}\left(\pi_{A_{2}},\gamma_{2},\ldots,\gamma_{n}\left(\beta\right)\right)$ 

5. Commutativity of P4 (and X3: The P4 operation is commutative, as is the X operation.

8×3=3×8 8×3=3×8

Nonce that, although the order of attributes may not be the same or the relations resulting from the two joins (or two carrescue products), the "meating" is the same because order or attributes is not important in the alternative definition of relation.

δ. Commutane σ with M (or ×): If all the attributes in the selection condition a involve only the attributes of one of the relations being joined – say, R— the two operations can be commuted as fellows:

 $\sigma_{i} \left( R \bowtie S \right) = \left( \sigma_{i} \left( R \right) \right) \bowtie S$ 

Alternatively, if the selection condition c can be written as (c1 AN9 c2), where condition c1 involves only the arributes of R and condition c2 involves only the arributes of S, the operations commute as follows:

 $a_{i}(\mathbb{R} \bowtie S) = (\sigma_{i1}(\mathbb{R})) \bowtie \{a_{i2}(S)\}$ 

The superioles apply if the # is replaced by a × operation.

Commuting π with ≫ (or × 5. Suppose that the projection list is L = {A<sub>1</sub>, ..., A<sub>n</sub>, B<sub>1</sub>, ..., B<sub>n</sub>}, where A<sub>1</sub>, ..., A<sub>n</sub> are antibutes of R and B<sub>1</sub>, ..., B<sub>n</sub> are involutes of S. If the join condition c involves only attributes in *L*, the two operations can be commuted as follows:

 $\pi_{1}\left(\mathbb{R}\bowtie[S]\right) = \left(\pi_{A} - \left[\Delta_{0}\left(\mathbb{R}\right)\right)\bowtie\left(\pi_{P1} - \left[B_{0}\left(S\right)\right)\right)$ 

If the join condition *c* contains additional attributes not in L. these must be added to the projection list, and a final  $\pi$  operation is needed. For example, if attributes  $B_{n+1} = -A_{n+2}$  or *R* and  $B_{n+1} = -B_{n+2}$  of *S* are involved in the join condition *c* but are not in the projection list 2, the operations commute as follows:

 $\pi_{\mathbb{C}}(R \boxtimes S) = \pi_{\mathbb{C}}\left((\pi_{\mathbb{R}_{1},\dots,\mathbb{R}_{n},\mathbb{C}^{n+1},\dots,\mathbb{R}^{n+1}}(R))\boxtimes_{\mathbb{C}}(\pi_{\mathbb{R}_{1},\dots,\mathbb{R}^{n+1},\mathbb{R}^{n+1},\dots,\mathbb{R}^{n+1}}(S))\right)$ 

For  $\times$ , there is no condition c, so the first transformation tide always applies by replacing  $\mathbf{M}_i$  with  $\times$ .

- Commutativity of ser operations: The set operations Q and G are commutative but — is not.
- Associativity of №, X. U; and O: These four operations are individually associative; that is, if 0 stands for any one of these four operations (throughout the expression), we have:

 $(R \oplus S) \oplus T = R \oplus (N \oplus T)$ 

10. Commuting a with set operations: The inoperation commutes with U. O, and if 0 stands for any one of these three operations (throughout the expression), we have

 $\sigma_{13}((0.5) + (\sigma_{13}(0))) \oplus (\sigma_{13}(5))$ 

The π operation commutes with U:

 $\pi^-(\mathcal{R}\cup\mathcal{S}) \vdash (\pi,(\mathcal{R})) \cup (\pi_{\mathsf{L}}(\mathcal{S}))$ 

 Converting a (σ, ×) sequence into ∞. If the condition c of a σ that follows a × corresponds to a join condition, convert the (σ, ×) sequence into a Mas follows.

 $(o, (\mathcal{R} \times S)) = (\mathcal{R} \bowtie S)$ 

There are other possible transformations. For example, a selection or join condition 2 can be converted into an equivalent condition by using the following rules (DeMorgan's laws):

NOT (c1 AND c2) = (NOT c1) OR (NOT c2) NOT (c1 OR c2) = (NOT c1) AND (NOT c2)

Additional transformations discussed in Chapters 5 and 6 are not repeated here. We discuss near how transformations can be used in hemistic optimization.

Outline of a Heuristic Algebraic Optimization Algorithm — We can new oncore the steps of an algerithm that utilizes some of the above inlex to transform an entry query tree into an optimized tree that is more efficient to execute (in most close). The algorithm will lead to transformations similar to those discussed in our example of Equi-15.5. The steps of the algorithm are as follows:

 Using Bide 1, break up any SLECT operations with conjunctive conditions into a cascide of set 6.2 operations. This permits a greater degree of freedom in moving SETECT operations down different branches of the true.

- Using Rules 2. 4, 6, and 10 concerning the commutativity of SELECT with other operations, move each SELECT operation as far down the query tree as is permated by the autibutes involved in the select condition.
- 5. Using Roles 5 and 9 concerning commutativity and associativity of binary operations, rearrange the leaf nodes of the tree using the following criteria. Fust, pusition the leaf node relations with the most costructive SELFCT operations to they are executed first in the query tree representation. The definition of most extractive SELFCT can mean either the ones that produce a relation with the fewest toples or with the smallest absolute size.<sup>17</sup> Another possibility is to define the most restrictive SELFCT as the one with the smallest selectority; this is more practical because estimates of sclentivities are often available in the 0-805 catalog. Second, make sure that the ordering of had nodes does not cause CAR (ESEAN 198-001.2) operations; for example, if the two relations with the most restrictive SELFCT, do not have a direct join conditions between them, it may be desirable to change the order of leaf nodes to avoid Carresian products.<sup>10</sup>
- Using Rule 12: combine a CARTES AN PRODUCT operation with a subsequent SELSCT operation in the tree into a JOIN operation, if the cond-tion represents a join condition.
- 5. Using Rules 3, 4, 7, and 11 concerning the cascading of CROBECT and the constituting of PROBECT with other operations, break down and move lists of projection attributes down the tree as far as possible by creating new TROBECT operations as needed. Only those attributes needed in the query result and in subsequent operations in the query tree should be kept after each PROBECT operation.
- Identify subtrees that represent groups of operations that can be executed by a single algorithm.

In our example. Figure 15.5(b) shows the tree of Figure 15.5(a) after applying steps 1 and 2 of the algorithm. Figure 15.5(c) shows the tree after step 5: Figure 15.5(d) after step 4: and Figure 15.5(e) after step 5. In step 6 we may group regerber the operations in the abbree whose root is the operation  $\pi_{0,\infty}$  into a single algorithm. We may also group the according operations into another subtree, where the toples resulting from the first algorithm replace the softree whose root is the operation  $\pi_{0,\infty}$  because the first grouping means that this subtree is executed first.

Summary of Heuristics for Algebraic Optimization. We now summarize the base heuristics for algebraic optimization. The main beginstic is to apply first the operations that reduce the size of intermediate results. This includes performing is early as possible SELECE operations to reduce the number of tuples and PSC(EC) operations to reduce the number of attributes. This is done by moving SELECE and PSO(EC) operations.

<sup>17.</sup> Either definition can be used, since these jules are become.

<sup>18.</sup> Note that a Correston product is acceptable up which uses—for example, preach relation has only a single tuple because each badia previous select conditions on a key teld.

as for down the tree as possible. In addition, the FED-CL and JONS operations that are most restrictive—that is, result in relations with the fewerst tuples or with the smallest absolute size—should be executed before other similar operations. This is done by reordering the leaf nodes of the tree among themselves while avoiding Correston products, and adjusting the test of the tree appropriately.

### 15.7.3 Converting Query Trees into Query Execution Plans

An execution plan for a relational algebra expression represented as a query free methods information about the agents righthods available for each relation is well as the algorithms to be used in computing the relational operators represented in the tice. As a simple example, consider query QU from Chapter 5, whose corresponding relational algebra expression is

```
TENDER, THOSE , & DERIG (OF LAND & ARGENER) (USE ARDENET) MET HER AND A FAR WITH)
```

The query tree is shown in Figure 15.6. To convert this into an execution plan, the optimizer might choose an ordex search for the 95.50° operation (associating one existe), a table scatt as access method for the provide, a respectively algorithm for the join, and a scatt of the RON result for the PRODOT operators for additions the approach taken for executing the query may specify a materialized or a pipelined evaluation.

With materialized evaluation, the result of an operation is stored as a temporary relation (that is, the result is *physically materialized*). For instances the price operation can be computed and the entire result stored as a temporary relation, which is their read as input by the algorithm that computes the DICOPOT operation, which would produce the query result table. On the other hand, with **pipelined evaluation**, as the resulting tiples of an operation are produced, they are forwarded directly to the next operation in the quere sequence. For maniple, as the solected tuples from DEPARTMENT are produced by the SECOT operation, they are placed in a buffer; the ICON operation algorithm would then consume



I DEPARIMENT

FIGURE 15.6 A query tree for query Q1

one implies from the buffer, and those topics that result from the IOIs operation are specified to the projection operation algorithm. The advantage of pipelining is the cost swings in our having to write the intermediate results to disk and not having to read then back for the next operation.

# 15.8 Using Selectivity and Cost Estimates in Query Optimization

A query optimizer should cut depend solely on heimstic rules; it should also ostimate and compare the coars of executing a query using different execution strategies and should choose the strategy with the forest cast estimate. For this approach to work, accurate coar commers are required so that different strategies are compared fairly and realistically. In addition, we must limit the number of execution strategies to be considered, otherwise, too much time will be spent making cost estimates for the many possible execution strategies. Hence, this approach is more suitable for **compiled queries** where the optimization of lead at compile time and the resulting execution strategy code is stored and executed directly at function. For interpreted queries, where the entry proces shown in Figure 15.1 eccuts at runtime, a dufficale optimization may slow down the response time. A more elaborate optimization works best for interpreted queries, whereas a partial, less time-computed optimization works best for interpreted queries.

We call this approach **cost-based query optimization**,<sup>19</sup> and it uses traditional optimization rechtiques that search the volume space to a problem for a solution that minimizes an objective (cost) function. The cost functions used in query optimization are estimates and not experimedione. In Section 15.5 Live discuss the components of query execution root. Its Section 15.52 we discuss the type of information needed in cost functions. This information is kept in the DBMS catalog. In Section 15.84 we give examples of cust functions the SELECT operations, and in Section 15.8.4 we discuss cost functions for two-way ION operations. Section 15.8.5 discusses multivary pure, and Section 15.8.6 gives an example.

### 15.8.1 Cost Components for Query Execution

The cost of executing a query includes the following components.

Access cost to secondary storage. This is the cost of searchine for, reading, and writing data blocks that reside on secondary storage, mainly on disk. The cost of standing for records in a file depends on the type of access structures on that file, such as ordering, hashing, and primary or secondary indexes. In addition, factors

<sup>19.</sup> The approach was first used on the option for the SOSTEP Recommendal DBMS developed at DMF

such as whether the file blocks are allocated contignously on the same disk cylindet or scattered on the disk affect the access cost.

- Strange cost. This is the cost of storing any intermediate files that are generated by an execution strategy for the query.
- Comparation cost: This is the cost of performing in-memory operations on the data buffets during query execution. Such operations (nch descarching for and sorting records, merging records for a join, and performing computations on held values.
- Morazy mage cost: This is the cost pertaining to the number of memory buffers needed during query execution.
- Communication cast. This is the cost of shipping the query and its results from the database site to the site or terminal where the query originated.

For large databases, the main emphasis is on immuniting the access cost to secondary storage. Simple nost immuons ignore other factors and compare different query execution strategies in terms of the number of block transfers between disk and main memory for smaller databases, where most of the data in the files involved in the query can be completely stored in memory the emphasis is on minimizing comparation cost. In distributed databases, where many sites are involved (see Chapter 25), communication cost must be minimized also. It is difficult to include all the cost components in a (weighted) cost function because of the difficulty of assigning sintable weights to the cost components. That is why some cost functions consider a single factor only----disk access. In the next section we discuss some of the information that is needed for formulating coafunctions.

### 15.8.2 Catalog Information Used in Cost Functions

To estimate the costs of various execution strategies, we must keep track of any internation that is needed to the cost functions. This information may be stored in the 1580 natalog, where it is accessed by the query optimizer. First, we must know the size of each file. For a file whose records are all of the same type, the number of records (tuples) (r), the (average) record size (**R**), and the number of blocks (b) (or close estimates of then) are needed. The blocking factor (bfr) for the file may also be needed. We must also keep track of the primary axeep method and the painteen access analytics for each file. The file records may be unordered, ordered by an attribute with or without a truntary or clustering index, or hished on a key attribute. Information is kept on all secondary indexes and indexing attributes. The number of levels (z) of each multilevel index (primary, second are, or clustering) is needed for cost functions that estimate the number of block becease that occur during usery executions in some cost functions the number of block becease that occur during usery executions in some cost functions the number of first-level index blocks ( $b_{11}$ ) is needed.

Another important parameter is the number of distinct values (d) of an attribute and its selectivity (s0), which is the fraction of records satisfying an equality condition on the antribute. This allows estimation of the selection cardinality (s = st + r) of an attribute, which is the attrage number of records that will satisfy an equality selection condition on that autibute. For a key starbute, d = r, of = 1h and s = 1. For a markey synhiles by making an assumption that the distinct values are uniformly distributed arong the records, we estimate  $s^2 = (1/d)$  and  $so(s = (1/d))^{1/2}$ .

Information such as the number of index levels is easy to maintain because it does not charge very often. However, other information may change frequently for example, the number of records it in a file changes every time a record is inserted or deleted. The query opterizer will need reasonably close but not necessarily completely op-trothemante values of these parameters for use in estimating the cost of Various execution arategies. In this react two sections we examine how some of these parameters are used in cost functions for a cost-based query optimizer.

### 15.8.3 Examples of Cost Functions for SELECT

We now give cost functions for the selection algorithms S1 to S8 discussed in Section 15.3.1 in terms of number of block transfers between memory and disk. These cost functions are estimates that ignore computation time, storage cost, and other foctors. The cost for method S1 is referred to as  $C_{in}$  block accesses.

- S1. Linear search (brace force) approach: We search all the file blacks to retrieve all records satisfying the selection condition; hence, C<sub>s14</sub> = 0. For an equality condition on a key, only half the file blocks are searched on the average before finding the record, so C<sub>s13</sub> = (b/2) if the record is found; if no record satisfies the conditions C<sub>s10</sub> = k.
- S2. Binary search The search accesses approximately  $\mathbb{O}_{S2} = \log_2 b + \lceil (s/b/r) \rceil = 1$ file blocks. This reduces to  $\log_2 b$  if the equility condition is on a unique (key) attribute, because s = 1 in this case.
- S3. Using a process redex (S3a) or tash key (S3b) to retriere a single second. For a primary nidex, retrieve one more block than the number of index levels, hence, C<sub>NAI</sub> = x = 1. For hadring, the cost tenetion is approximately C<sub>NAI</sub> = 1 for static hashing or linear hashing, and it is 2 for extendible husbing (see Chapter 13).
- S4. Using an ordering index to recorder induple records. If the comparison condition is

   >=, <<. or <= on a key field with an ordering index, roughly half the file records will satisfy the condition. This gives a root function of C<sub>S4</sub> = v = (b/2). This is a very rough estimate, and although it may be correct on the average, it may be quite inaccurate in individual cases
- S5 Using a clustering index to verifies our inple records. Given an equility condition, since will satisfy the condition, where sits the selection cardinality of the indexney attribute. This means that [(s/bjrs]] file blocks will be accessed, giving C<sub>85</sub> = x + [(s/bjr)].
- S6: Using a secondary (B\*-tree) radex. On an equality comparison, viccords will satisfy the condition, where s is the selection cardinality of the indexing attribute.

<sup>20.</sup> As we mentioned caller, more accurate spiniaters may store histograms of the distribution of records over the data visities for so attribute.

However, because the index is nonclustering, each of the records may reside on a different block, so the (worst case) cost estimate is  $C_{yea} = x + s$ . This reduces that if for a key indexing attribute. If the comparison condition is  $\geq_{x} \geq_{y} <_{z}$  or < z and ball the file records are assumed to satisfy the condition, then (very roughly) half the first-level index blocks are accessed, plots half the file records via the index. The cost estimate for this case, approximately, is  $C_{wea} = x - (b_{H}R) + (i/2)$ . The objector can be refined if hence selectivity estimates are available.

- S7 Conjunctive selection. We can use either S1 or one of the methods S2 to S6 discussed above. In the latter case, we use one condition to retrieve the records are then check in the metaory buffer whether each retrieved record satisfies the remaining conditions in the conjunction.
- S6: Conjunctive selection using a composite index. Same as S3a, S5, or S6a, depending on the ope of index.

**Example of Using the Cost Functions.** This a query optimizer, it is common to enumerate the various possible strategies for executing a query and to extinate the case for different strategies. An optimization technique, such as dynamic programming, may be used to find the optimization (least) cost estimate efficiently, without having to consider all possible execution strategies. We do not discuss optimization algorithms here: rather, we use a simple example to illustrate how cost estimates may be used. Suppose that the second blocking factor bits  $\tau_{\rm F} = 10,000$  records sorted in  $b_{\rm F} = 2000$  disk blocks with blocking factor bits = 5 records block and the following access paths.

- A clustering index on seasy, with levels quase = 3 and overage selection cardinality sease = 20.
- 2. A secondary index on the key attribute ssn, with  $x_{eg} = 4 (y_{eg} = 1)$ .
- 5. A secondary index on the nonkey attribute fact with  $x_{y,u} = 2$  and fast-level index blocks  $b_{1,w} = 4$ . There are  $d_{w,u} = 125$  distinct values for Eq. so the selection radiative to  $0.0018 a_{y,u} = (r_1 k l_{w,u}) = 80$ .
- A secondary rodex on sex, with x<sub>a</sub>, = 1. There are d<sub>a</sub>, = 2 values for the sex attribute, so the average selectron cardinality is y<sub>a</sub> = (τ<sub>k</sub>/t<sub>a</sub>) = 5000.

We dlustrate the use of cost functions with the following examples.

$$\begin{split} & (0^{9}1): \sigma_{35^{10}-1/145(769)}(t^{\rm op},0^{\rm o}te) \\ & (0^{9}2): \sigma_{149(15)}(t^{\rm op}to^{\rm o}te) \\ & (0^{9}3): \sigma_{149(15)}(t^{\rm op}to^{\rm o}te) \\ & (0^{9}4): \sigma_{149(15)}(t^{\rm op}to^{\rm o}te) \\ & (0^{9}4): \sigma_{149(15)}(t^{\rm op}to^{\rm o}te) \\ & (0^{9}4): \sigma_{149(15)}(t^{\rm op}to^{\rm o}te) \\ \end{split}$$

The cost of the brate face (linear scare b) option SI will be estimated as  $C_{n,n} = b_1 = 2000$  (for a selection on a neakey attribute) or  $C_{n,n} = (b_p/2) = 1000$  (average cost for a selection on a key attribute). For of 1 we can use other method S1 or method S6; the case estimate for S6a is  $C_{n,n} = x_{n,n} + 1 - 4 + 1 - 5$ , and it is chosen over Method S2, whose average cost is  $C_{n,n} = 1200$ . For O(2 we can use other method S1 (with estimated cost  $C_{n,n} = 2000$ ) or method S6b (with estimated cost  $C_{n,n} = x_{n,n} + (b_{1,n,n}/2) + (c_{1,n}/2) = 2$ 

+14(2) + (10,000/2) = 5004), so we choose the brace force approach for 012. For 015 we can be either method S1 (with estimated cost  $C_{sta}$  = 2000) or method S6a (with estimated cost  $C_{sta}$  = 2000) or method S6a (with estimated cost  $C_{sta}$  = 2,  $x_{\mu\nu}$  +  $x_{\mu\nu}$  = 2 + 80 = 82), so we choose method S6a.

Finally, consider OP4, which has a comparative selection condition. We need to tsumate the cost of using any one of the three comparators of the selection condition to remove the records, plus the brute force approach. The latter gives cust estimate  $C_{sta} = 200$ . Using the condition ( $\omega_0 = 5$ ) first gives the row estimate  $C_{sta} = 82$ . Using the condition ( $\omega_0 = 5$ ) first gives a cost estimate  $C_{sta} = x_{sta} + (b_F/2) = 3 - (2200/2)$ = 500 Using the condition ( $\omega_0 = 5$ ) first gives a cost estimate  $C_{sta} = x_{sta} + (b_F/2) = 3 - (2200/2)$ = 500 Using the condition ( $\omega_0 = 5$ ) first gives a cost estimate  $C_{sta} = x_{sta} + s_{sta} - 1 + 500 = 5001$ . The optimizer would then choose method S6a on the secondary index or 20 occurse it has the lowest cost estimate. The condition ( $\omega_0 = 5$ ) is used to retrieve the records and the remaining part of the comparative condition ( $\omega_0 = 5$ ) is used to retrieve the records and the remaining part of the comparative condition ( $\omega_0 = 5$ ) as  $\omega_0 > 30000$ .

#### 15.8.4 Examples of Cost Functions for JOIN

To develop reasonably accurate cost functions for  $J(\lambda N)$  operations, we need to have an emistic for the size frammer of tuples) of the file that results ofter the  $J(\lambda N)$  operation. This is usually kept as a ratio of the size (number of tuples) of the resulting join file to the size of the Carrestan product file, if both are applied to the same input files, and it is called the join selectivity (js). If we denote the number of tuples of a relation R by J(R), we have

$$\mu = I(R \bowtie S) / I / (R \times S)) = I(R \bowtie S) I / (IR) \simeq SIE$$

If there is no join condition  $c_i$ , then p = 1 and the join is the same as the CARTESIAN **PERRON**. If no tuples from the relations satisfy the join condition, then p = 2. In general,  $\delta \leq p \leq 1$ . For a join where the condition c is an equality comparison R, A = S, B, we get the following two special cases:

1. If A is a key of B, then  $|1(\mathbf{R} \bowtie S)| = |1S1|$ , so is  $\simeq (1/181)$ 2. If B is a key of S, then  $|1(\mathbf{R} \bowtie S)| = |1\mathbf{R}|$ , so is  $\approx (1/181)$ .

Having an estimate of the join selectivity for commonly occurring join conditions mables the query optimizer to estimate the size of the resulting file after the join queation, given the sizes of the two input files, by using the fottoular  $(E \bowtie S)1 = j_{0} \approx 1R1 - -S1$ . We can now give some sample approximate cost functions for estimating the cost of some of the puth algorithms given in Section 15–3.2. The join operations are of the kine.

RMAN S

where A and B, ire domain-compatible attributes of R and S, respectively. Assume that R hosts blocks and that S has be blocks:

 II. Nexted-loop join. Suppose that we use R for the outer loop; then we get the toblowing cost function to estimate the number of block accesses for this method. assuantig three memory buffets. We assume that the blockning factor for the resulting file is prigg and that the join selections is known.

 $C_{21} = b_{12} = \{n_{12} \times n_{23}\} + (i_{12} \times 1R1 \times 1R1)/b(n_{123})$ 

The last past of the framaliens the cost of writing the resulting file to disk. This cost formula can be modified to take into account different numbers of memory butters, as discussed in Section 15.3.2.

• J2. Single-hole pair (avoig on access structure to network the matching second(s)). If an index expositor the point attribute B of S with index levels x<sub>D</sub>, we can intrive each record s in B and then use the index in records on the matching records a from 5 that society r[5] = s[A]. The case depends on the type of index. For a secondary index where s<sub>B</sub> is the selection cardinality for the pair antibute B of 5.<sup>2</sup> we get

$$C_{(2)} = (y_R + (1)Q) + (y_R + y_R)V + ((y_R + -R - 2)1)V (y_R + y_R)$$

For a clustering index where s<sub>0</sub> is the selection cardinality of b, we get

$$C_{2,0} = h_0 + (1R1 + (s_0 - (s_0 b \hat{n}_0))) + R_1 s_1 - R_1 + 1S_1) b f_{N_0}$$

For 5 primary index, we get

$$C_{\mu\nu} = b_{\mu} * (1R_{\mu} * (x_{\mu} - 1)) * ((j/*1R1 * 1S1)jb)_{\mu\nu}$$

If a hash key exists to row of the two join othebutes—say, B of 5—we get

 $C_{iNi} = b_N + (1R) + h(1 + ((j_i + 1R) + 1S1)) f_{iRis})$ 

where  $h \simeq 1$  is the average number of block accesses to retrieve a record, given its bash key value.

 J3. Sectore ye gain. If the files are already sorted on the join attributes, the cosi function for this method is.

 $C_{cb} = b_{g} + b_{s} + H_{1s} + 1R_{1s} + S1)/b(r_{gs})$ 

If we must som the files, the cost of setting must be added. We can use the formalastrom Section 15.2 to estimate the sorting cost.

Example of Using the Cost Functions. Suppose that we have for inclusional described in the example of the previous section, and assume that the discussion file of Figure 5.5 consists of  $\eta_5 = 125$  records stored in  $\theta_5 = 13$  disk blacks. Consider the join operations

```
(046): SNP. CYSE 04_{100-000761 #} DEPARTMENT
(047): DEPARTMENT 04_{0027000000 #} DEPARTMENT
```

<sup>31.</sup> Selection conducting was defined as the interage number of records that satisfy an equality condition on an intribute, schich as the average number of records that have the same value to the available and hence only by conditions single record to these ther the.

Suppose that we have a primary index on concern of department with  $|z_{max}| = 1$  level and a secondary index on wasse of department with selection conditiality  $|z_{max}| = 1$  subjectly  $z_{max} = 2$ . Assume that the non-selectivity for OP6 is  $|z_{1,1,2}| = (1)$  because concerns a key of department. Also assume that the blocking factor for the resulting jum file  $bf_{0,1,2} = 4$  accords per block. We can estimate the worst case costs for the DIS operation OP6 is just a follows:

1. Using Method J1 with events as outer loop:

$$\begin{split} \mathbf{C}_{11} &= b_{1} + (b_{2} \leq b_{1}) + (t_{2NO} \leq t_{1} \leq t_{1})/(9^{2} t_{1}) \\ &= 2000 - (2000 \leq 1.3) + (((1/125) < 10,000 < 125)/4) = 30,500 \end{split}$$

Using Method J1 with Greetnest as oncer loop.

 $\begin{array}{l} C_{11} = b_{11} + (b_{11} * b_{12}) + ((p_{12}p_{11} * p_{11})bb_{11}) \\ = 13 + (13 * 2000) + (((175) * 10.000 * 125)4) = 28.513 \end{array}$ 

Using Method J2 with evecate as occur loop.

$$\begin{split} \mathbf{C}_{12} &= h_{\mathbf{k}} + (\tau_{i} + (y_{\text{construct}} + 1)) + ((y_{\text{construct}} + \tau_{i})/bh_{\mathbf{k}}) \\ &= 2000 + (10,000 + 2) + ((11/125) + 10,000 + 125/4) = 24,500 \end{split}$$

4. Using Method 12 with occasing as ourer loop

$$\begin{split} C_{12,i} = & k_{13} + (r_1 + (x_{23} - s_{16})) + (0) k_{10} + r_2 + r_1 \beta (1/(-y_1)) \\ &= 13 + (125 + (2 + 30)) + \alpha ((1/125) + 10.000 + 125/4) = 12.763. \end{split}$$

Case 4 has the lowest cost estimate and will be chosen. Nonce that if 15 memory latters (or more) were available for executing the join instead of jost three. If of them could be used to hold the entire **represent** relation in memory one could be used as buffer to the result, and the cost for Case 2 could be driveneally reduced to just  $b_1 + b_2 + ((j_{2,25}, v_{1,25}, j_{2})))/(j_{1,2})$  or 4513 as ciscussed in Section 15.3.2. As an exercise, the reself should perform a similar analysis for OP7.

#### 15.8.5 Multiple Relation Queries and Join Ordering

The algebraic transformation rules in Section 15.7.2 include a commutative rule and an associative role for the join operation. With these rules, many equivalent join expressions on be produced. As a result, the number of alremative query crees grows very tapially as the number of joins in a query increases. In general, a query that joins a relations will have n = 1 join operations, and hence can have a large number of different join orders. Fromating the cost of every possible join true for a query with a large number of joins will require a substantial amount of time by the query optimizer. Hence, since privacy of the passible guery trees is needed. Query optimizers typically limit the structure of a (join) query trees is needed. Query optimizers typically limit the structure of a (join) query tree to that of left-deep for right-aleep) trees. A left-deep tree is a binory tree when the right child of each trouble trade is always a have relation. The optimizer would choose ine particular left deep tree with the lowest estimated rust. Two examples of left-deep trees are shown in Figure 15.7. (Note that the trees in Figure 15.5 are also befind ep trees.)


FIGURE 15.7 Two left-beep fjoin query trees.

With left-deep trees, the right child is considered to be the inner relation when executing a nested-loop join. One advantage of left-deep for right-deep) trees is that they are amenable to pipelining, as discussed in Soction 15.5. For instance, consider the first left-deep tree in Figure 15.7 and assume that the join algorithm is the single-loop methods in this case, a disk page of tuples of the corer relation is used to probe the maet relation for matching tuples. As a resulting block of tuples is produced from the join of R1 and R2, it could be used to probe R3. Likewise, as a resulting page of tuples is produced from this join, at could be used to probe R4. Another advantage of left-deep (or right-deep) trees is that having a base relation as one of the arputs of each join allows the upfunction to indice any access paths on that relation that may be used in exercising the join.

If materialization is used in-read of pipelining (see Section 15.6), the join results could be materialized and stored as temporary relations. The key idea from the optimized standpoint with respect to join orderine is to find an ordering that will reduce the scenit the reopenary results, since the temporary results (pipelined or materialized) are used by subsequent operators and hence affect the execution cust of these operators.

#### 15.8.6 Example to Illustrate Cost-Based Query Optimization

We will consider query Q2 and its query tree shown in Figure 15.40 to illustrate costbased query optimization.

```
Q2: SELECT PHUMPER, DNHM, LNAME, ADDRESS, BDATE
FROM PROJECT, DEPARTMENT, EMPLOYEE
WHERE DNHM-UNUMPER AND MERSSN-SSN AND PLOCATION-"STATIONS";
```

Suppose we have the statistical information about the relations shown in Eccure 15.8. The low\_volut and orcs\_volue statistics have been normalized for clarity. The tree in Figure 15.4a is assumed to represent the result of the algebraic heuristic optimization process and the stati of cost-based optimization (in this example, we assume that the bearistic continuer does not push the projection operations down the tree).

| TABLE_NAME        | COL.IMN_NA                                                                                                                                                                 | ME NUM                                                                                                                                                                                                                                                                                                                                                                                                                            | DISTINGT                                                                                                                                                                                                                                                                                                                                                                                                               | LOW_VAL                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | UE                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | HIGH_VALUE                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|-------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| PROJECT           |                                                                                                                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                   | 200                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 200                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
| PROJECT           | PNUM9EB                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                   | 2001                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 2000                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
| PROJECT           | <b>CNUM</b>                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                   | 50                                                                                                                                                                                                                                                                                                                                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 90                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
| OFPARTMENT        | CNUMBER                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                   | 50                                                                                                                                                                                                                                                                                                                                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 50                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
| <b>DEPARTMENT</b> | MGRSSN                                                                                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                   | 50                                                                                                                                                                                                                                                                                                                                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | i                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 50                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
| EMPLOYES          | \$5N                                                                                                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                   | 13030                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | i                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 10505                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| EMPLOYEE          | DNC                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                   | 50                                                                                                                                                                                                                                                                                                                                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 50                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
| EMPLOYES          | SALARY                                                                                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                   | 500                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 500                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
| TABLE NAME        | NUM HOWS                                                                                                                                                                   | BLOGKS                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
|                   | 3030                                                                                                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
| DEPADTMENT        | 2010                                                                                                                                                                       | 100                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
| EMPLOYEE          | 10000                                                                                                                                                                      | 2000                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
|                   |                                                                                                                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                   | LEAF                                                                                                                                                                                                                                                                                                                                                                                                                   | OISTINCI                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
| INCEX_NAME        | DNILIĢENES                                                                                                                                                                 | BLEVAL                                                                                                                                                                                                                                                                                                                                                                                                                            | BLOCKS                                                                                                                                                                                                                                                                                                                                                                                                                 | _KEYS                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
| DEC L DLOC        | WORK N ICH IE                                                                                                                                                              | <u> </u>                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
|                   |                                                                                                                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                   | 4                                                                                                                                                                                                                                                                                                                                                                                                                      | 206                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
| CMC_00N           | ONNAUE<br>MANUALITUR                                                                                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                   | 50                                                                                                                                                                                                                                                                                                                                                                                                                     | 10000                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
|                   | TABLE_NAME<br>PROJECT<br>PROJECT<br>PROJECT<br>OFPARTMENT<br>PMPLOYEE<br>EMPLOYEE<br>IAGLE NAME<br>PHOLECT<br>DEPARTMENT<br>EMPLOYEE<br>INDEX_NAME<br>PROJ_PLOC<br>FMP_SSN | TABLE_YAME     COL.IMN_NA       PROJECT     PLOCATION       PROJECT     PNUM9FR       PROJECT     PNUM9FR       PROJECT     DNUM       OFPARTMENT     ONUM8ER       OEPARTMENT     ONG       EMPLOYEE     DNG       EMPLOYEE     ONG       PHOLECT     Z000       DEPARTMENT     SG       PHOLECT     Z000       INDEX_NAME     UNIQUENES       INDEX_NAME     UNIQUENES       PROJ_PLOC     NONUNIQUE       PM_SSN     WONUNIQUE | TABLE_NAME     COL.IMN_NAME     NUM       PROJECT     PLOCATION       PROJECT     PNUM9FR       PROJECT     DNUM       OFPARTMENT     ONUMBER       OEPARTMENT     ONO       FMPLOVEF     SSN       EMPLOVEE     ONO       EMPLOVEE     SALARY       IAGLE_NAME     NUM HOWS       PHOLECT     2000       INCEX_NAME     UNIQUENES       PROJ_PLOC     NONUNKIUE       INCEX_NAME     UNKUE       INCEX_NAME     UNKUE | TABLE_YAME     COL.IMN_NAME     NUM_DISTINCT       PROJECT     PLOCATION     200       PROJECT     PNUM9FR     200       PROJECT     DNUM9FR     200       PROJECT     DNUM9FR     200       PROJECT     DNUM9FR     200       PROJECT     ONUM     50       OFPARTMENT     ONUMBER     50       DEPARTMENT     MGRSSN     50       FMPLOVEF     SSN     13000       EMPLOVEE     ONG     50       EMPLOVEE     SALARY     500       IAGLE NAME     NUW HOWS     BLOGKS       PHOLECT     2000     100       DEPARTMENT     50     5       EMPLOYEE     10000     2000       INDEX_NAME     UNIQUENES     BLEVEL       PROJ_PLOC     NONUNKIUE     1       FMP_SSN     UNKIUE     1       CMP_SKA     MONUNKIUE     1 | TABLE_NAME     COL.IMN_NAME     NUM_DISTINCT     LOW_VAL       PROJECT     PLOCATION     200       PROJECT     PNUM9FR     2001       PROJECT     PNUM9FR     2001       PROJECT     ONUM     501       OFPARTMENT     ONUMBER     501       OEPARTMENT     ONUMBER     501       OEPARTMENT     MORSSN     500       FMPLOVEF     SSN     13000       EMPLOVEE     ONC     500       IAGLE NAME     NUM HOWS     BLOCKS       PHOLECT     Z000     100       DEPARTMENT     50     5       EMPLOYEE     10000     2000       INCEX_NAME     UNIQUENES     BLEVEL     LEAF     OISTINC1       INCEX_NAME     UNIQUENES     BLEVEL     BLOCKS | TABLE_YAME     COL.IMN_NAME     NUM_DISTINCT     LOW_VALUE       PROJECT     PLOCATION     200     1       PROJECT     PNUM9ER     2001     1       PROJECT     DNUM     50     1       OFPARTMENT     ONUMBER     50     1       OEPARTMENT     ONUMBER     50     1       OEPARTMENT     ONUMBER     50     1       OEPARTMENT     MORSSN     50     1       FMPLOVEF     SSN     19000     1       EMPLOVEE     ONG     50     1       IAGLE NAME     NUM HOWS     BLOGKS       PHOLECT     2000     100       DEPARTMENT     30     5       EMPLOYEE     10000     2000       INDEX_NAME     UNIQUENES     BLOCKS       PROJ_PLOC     NONUNIQUE     1     4       PROJ_PLOC     NONUNIQUE     1     4       PMP_SSN     UNIQUE     1     50     10003 |

"BLEVFL is the number of levels without the leaf level

**EGURE 15.8** Sample statistical information for relations in Q2 (a) Column information, the table information, (c) Index information.

The first cost-based optimitation to consider is join ordering. As previously mentioned, we assume the optimizer considers only (choose prees, so the potential form orders) without Cartesian product – are

1.9POJCE NO OCHARIALAT DA UNPLOYER 2 DEPARTMENT DA EMPLOYER DA UNPLOYER 3.DEFARTMENT DA EMPLOYER DA PROJECT 4.EMPLOYER DA DEPARTMENT DA PROJECT

A same that the selection operation has dready been applied to the 50.600 relation have assume a materialized approach, then a new temporary relation is created after each peroperation. To examine the cost of ioin order (1), the first pointies between excision and testement. Both the join method and the access methods for the laplot relations must be determined. Since examine this isoundes according to higher 15.3, the only we lable access method is a table scan (that is, a linear search). The ensurement relations will have the selection operation genomed before the join, so two options exist table scan (linear search) or other nexts P80.9\_ (10), tracks, so the optimizer must compare their estimated costs. The souther all momention on the 2004 Eax's index (see Figure 15.5) shows the number of index levels x = 2 (not plus lat levels). The index is remaining the also processes in the state of key of protect), so the optimizer assumes a uniform data distribution and estimates the number of record pointers for each Pocotton value to be 10. This is computed from the tables in Figure 15.8 by multiplying SELR TROPT + REALER, 2014, where SELR TATES is estimated by 1780 Re\_D STINCT. So the cost of using the index and accessing the records is estimated by 1280 Re\_D STINCT. So the cost index and 10 for the data colorest. The cost of a rable scan is estimated to be 100 block accesses, so the index access is more efficient as expected.

In the moternalited approach, a temporary file Next of site 1 block is created to bold the result of the selection operation. The file site is calculated by determining the blocking factor using the formula NUM\_ROMS/BLOCKS which gives 2000/100 or 20 nows get block. Hence, the 1.0 accords selected from the BROWT relation will be into a since block. Now we can compute the estimated dest of the first join. We will consider only the instead-loop join method, where the outer relation is the temporary file, 19951, and the most relation is PERSINGER. Since the estimated dest of the first join. We will consider only the most relation is PERSINGER. Since the estimated dust of the first in the available buffer space, we need to read each of the DESERMENT table's two blocks only once, so the join dust is so block acrosses plus the cost of writing the temporary result file. TENEZ. The optimizer would have to determine the oct of TENE2. Since the join attribute the RESER is the key for DESERMENT, any tens value from TEVP1 will join with at most one second from DESERMENT, so the number of rows in TENE2 will be equal to the number of rows in TENE1, which is 10. The optimizer would determine the record size for TENE2 and the number of blocks meded to store these 10 most. For browing assume that the blocking factor for TENE2 is five rows per Flock, so a rural of two blocks are needed to store tended.

Finally, the cost of the last join needs to be estimated. We can use a single/cop join on 1642 since in this case the index EMP SSN (see Figure 15.8) can be used to probe and locate matching records from an iterative. Hence, the join method would involve reading in each block of 1642 and locking up each of the five wassavelues using the EMP SSN index. Each index lookup would require a root access, a leaf access, and a data block access 4x-1, where the number of levels x is 2). So, 10 lookups require 30 block accesses. Adding the two block accesses for tease a total of 32 block accesses for this join.

For the final projection, assume pipeliting is used to produce the final result, which does not require additional block accesses so the total cost for join order (i) is estimated as the sum of the previous costs. The optimizer would then estimate costs in a source manager for the other three join orders and choose the one with the lowest estimate. We leave this as an exercise for the reader.

# 15.9 OVERVIEW OF QUERY OPTIMIZATION IN ORACLE

The ORACLE DBMS (Version 7) provides two different approaches to query optimization rule-based and cost-based. With the rule based approach, the optimizer chooses execution plans based on heorytically ranked operations. CEACLE mannains a table of 15 ranked access paths, where a lower ranking implies a more choicent approach. The access paths range from table access by ROWID (most efficient)—where ROWID specifies the record) physical address that includes the data tile, data block, and row offset within the

block—to a full table scan (least efficient)—where all rows in the table are searched by doing multiblock reads. However, the role-based approach is being photed out index or of the cost-based approach, where the optimizer examines alternative access paths and operitor algorithms and chooses the execution plan with lowest estimated cost. The estimated query cost is proportional to the expected clapsed time needed to execute the query with the given execution plan. The COCACLE optimizer calculates this cost based on the estimized image of resources, such as l(0, CPC) (ince, and memory needed. The goal of custbased optimization in CRACLE is to minimize the clapsed time to process the entire query.

An interesting addition to the OCACLE query optimizer is the capability for an application developer to specify larits to the optimizer.<sup>1</sup> The idea is that on application developer might know more adormation about the data than the optimizer. For example, casiler the ENELOSE table shows in Figure 5.5. The SE column of that table has only two data or values. If there are 12,000 employees, then the optimizer would estimate that half are node and half are female, assuming a conform data distribution. If a secondary index ensist, it would more than likely not be used. However, if the apple; then developer knows that there are only 100 male employees, a birst could be specified in an  $<_{12}$  query whose MEER-clause condition is SEx = M' so that the associated index would be used in processing the query. Various hirts curb expectivel, such as:

- The optimization approach for an SQL statement.
- The access plath for a table accessed by the statement.
- The join order for a join starement.
- A particular ion operation in a join statement.

The cost-based optimization of ORACLE S is a good example of the sophisticated approach taken to optimize SQL queries in considerated ROBMES.

# 15.10 SEMANTIC QUERY OPTIMIZATION

3 different approach to query optimization, called semantic query optimization, has been suggested. This technique, which may be used in combination with the techniques discussed previously, uses constraints specified on the database schedula—such as imagine attributes and other more complex constraints—in order to modify one query auto another query that as more efficient to execute. We will not discuss this approach in detail but redy illustrate it with a simple example. Consider the SQL query:

```
SELECT E.LNAME, M.LNAME
FROM EMPLOYEE AS E. EMPLOYEE AS M
MMERE E.SUPERSSNUM.SSN AAD E.SALARY > M.SALARY
```

This query setucces the names or employees who carn more than their supervisors. Suppose that we had a constraint on the database schema that stated that no employee

```
12. Such hards have also been called query availables.
```

can earn more than his or her direct supervisor. If the semantic query optimizer checks for the existence of this constraint, it need not execute the query at all because it knows that the result of the query will be empty. This may save considerable time if the constraint checking can be done efficiently. However, searching through many constraints to find those that are applicable to a given query and that may seminarically optimize it can also be quite time-constrainty. With the inclusive of active rules in database systems (see Chapter 24), semantic query optimization rechtriques thay eventually be fully incorporated into the Disters of the future.

#### 15.11 SUMMARY

in this chapter we give an everyow of the techniques used by 00008 in proceeding and optimizing high-level queries. We that discussed how SQL queries are mushared into relational algebra and then how caroos relationed algebra operations may be executed by a OBDS. We such at some operations, particularly SELSCT and JOIN, may have many given tion options. We also discussed how operations can be combined during query processing to create pipelined or stream based execution instead of materiolized execution.

Following there we described broatistic approaches to query optimization, which us hearistic rules and algebraic techniques to improve the efficiency of query execution. We showed have a query tree that represents a relational algebra expression can be bearistically optimized by reorganizing the tree nodes and transforming is into another equivalent query tree that is more efficient to execute. We also give equivalence-preserving transformation tules that may be applied to a query tree. Then we introduced query execution plans for SQL energies, which add method execution plans to the query tree operations.

We then discussed the cost-based approach to query optimization. We showed how cost functions are developed to: some database access algorithms and how these cost functions are used to estimate the costs of different execution strategies. We presented as overview of the cRACH, query optimizer, and we mentioned the rechrique of semantic query optimization.

#### Review Questions

- 15.1. Discuss the reasons for converting SQL queries into relational algebra queries before optimization is done.
- 15.2. Discuss the different algorithms for implementing each of the following relational operators and the encountrinces under which each algorithm can be used induct, norm project, UNION, INTERSECT, SET DIFFERENCE, CARTESIAN PROPERT.
- 15.5. What is a quere execution plan?
- 15.4. What is meant by the term termine optimization? Discuss the main hemistics that are applied during query optimization.

- 15.5. How does a query free represent a relational algebra expression? What is meant by an execution of a query free? Discuss the rules for transformation, of query frees, and identify when each rule should be applie? during optimization.
- 15.6. How many different join orders are chere for a query that prins 10 relations?
- 15.7. What is me at by crossbaard query optimization?
- 15.8. What is the difference between pip long and mmenologitant
- 15.9. Discuss the cost components for a cost forguon that is used to estimate query execution cost. Which cost components are used most often as the basis for cost functions?
- 15.10. Discuss the different types of parameters that are used in cost functions. Where is this reformation kept?
- 15.11 List the cost functions for the SELECT and JOES methods discussed in Section 15.8.
- 15.12 What is meaningly sensitive query optimization? How does it differ from other query optimization techniques?

#### Exercises

- 15.13. Consider SQL queries Q1, Q8, Q1B, Q8, and Q22 from Chapter 8.
  - Drow at least two query trees that can represent each of these queries. Under what encamstances would you use each of your energy trees?
  - b) Drive the initial query free for each of these queries, then show how the query free is optimized by the algorithm outlined on Section 153.
  - For each query, compare your own query trees of part (a) and the initial and final query trees of part (b).
- 15.14. A rule of 40% blocks is to be seried with an available butter space of 64 blocks. How many passes will be needed in the merge phase of the external sort merge algorithm?
- 15.15. Develop cost functions for the PROJECT, UNION. INTER-FUTION. SET THEFERENCE, and CARTERSIAN DODD, 07. algorithmic discussed in Section 15.4.
- 15.16 Develop cost functions for an algorithm that consists of two SERCES, a IOES, and a final PROJECT, interms of the cost functions for the individual operations.
- 15.17 Can a nonderse index be used in the implementation of an aggregate operator. Why enaby not?
- Colculate the cost functions for different options of executing the JOIN operation OP7 discussed in Section 15,3.2.
- 15.19 Develop formulas for the hybrid hash join algorithm for calculating the site of the batter for the first backet. Develop from accurate cest estimation formulas for the algorithm.
- 15.20 Estimate the case of operations (016 and (017) using the formulas developed in Exercise 15.9.
- [52]. Extend the sort merge ion algorithm to implement the left outer join operation.
- (5.32) Compare the cost of two different query plans for the tollowing query:

(TSPLARS - 2010) (TPPEORTE MISS - JAPAN (SPARPERT)

Use the database statistics in Figure 15.8.

#### Selected Bibliography

A survey by Granfe (1993) discusses query execution in database systems and includes an extrastive bibliography. A survey paper by Jarke and Kach (1984) gives a taxonomy of query optimization and includes a Fibbography of work in this area. A detailed algorithm for relational algebra optimization is given by Smith and Chang (1975). The Ph D, these of Knot (1980) provides a foundation for query processing techniques.

Whang (1986) discusses query optimization in 1984 (v2000)-By-Frequiple), which is a system based on Q6E. Co-robused optimization was introduced in the SYSTEM R expenmerral DBVF and is discussed in Astraham et al. (1976). Selinger et al. (1979) discuss the optimization of raditions join- in SYSTEV R Join algorithms are discussed in Gorlah (1975). Blasgen and Eswaran (1976), and Whong et al. (1982). Hashing algorithms for implementing joins are described and analyzed in (DeWitt et al. (1964), Bratheresinger (1984). Shapiny (1980). Suburgawa et al. (1989), and Blakeley and Marin, (1984). among others. Approaches to finding a good jush order are pre-ented in loannide are Kang (1990) and in Swami and Gapta (1989). A docussion of the implications of leftdeep and bushy join trees is presented in loanradis and Kang (1991). Kine (1982) discusses transformations of nested SQL queries into canonical representations. Optimization of aggregate functions is discussed in Klug (1982) and Murshkrishna (1992). Salderg at al. (1990) describe a tast external sorting algorithm. Estimating the size of temporary relanons is crucial for query optimization. Sampling-based estimation schemes are presented in Hass et al. (1995) and in Hass and Swann (1995). Lipton et al. (1997) riso discuss selectionly estimation. Elaving the database system store and use more detailed statistics in the form of histograms is the topic of Murahki, shing and DeWart (1988) and Possible t al. (1996).

Kan et al. (1985) discuss advanced tippes in query optimization. Seminitic query optimization is discussed in King (1981) and Malley and Zdonick (1986). More recenwork on seminitic query optimization is reported in Chakravorthy et al. (1990), Sherae and Obsiyoglu (1989), and Siegel et al. (1992).





# Practical Database Design and Tuning

In this chapter, we first discuss the osnes that arise in physical database design in Section 161. Then, we discuss how to improve database performance through database turing in Section 16.2.

# 16.1 PHYSICAL DATABASE DESIGN IN RELATIONAL DATABASES

in this action, we trist discuss the physical design factors that affect the pettormagne of applications and transportions; we then comment on the specific goad lines for ROBACSS.

# 16.1.1 Factors That Influence Physical Database Design

Elected design is an activity where the goal is not only to come up with the appropriate structuring of data in storage but to do so in a way that guar intersigned performance. For a given conceptual schema, there are nonce physical design alternatives in a given possible to make meaningfol physical design decisions and performance analyses call we know the queries, transactions, and applications that are expected to run on the database. We must analyze these applications, their expected frequences of invocement.

ing time constraints on diebt execution, and the expected frequency of update operanons. We discusses the of these factors next.

A. Analyzing the Database Queries and Transactions. Before indertaking physicle database design, we must have a goad idea of the intended rise of the database by defining the species and transactions that we expect to run on the database in a high-level form. For each query, we should specify the following:

- 1. The files that will be accessed by the query.<sup>3</sup>
- 2. The attributes on which any selection conditions for the query are specified.
- The attributes on which ony join conditions or conditions to hink multiple tables or objects for the query are specified.
- 4. The attributes whose values will be retrieved by the query.

The attributes listed in items 2 and 3 above are candidates for definition of access structures. For each update transaction or operation, we should specify the following

- 1. The files that will be updated.
- 2. The type of operation on each fite tinsert, update, or deletel
- 3. The attributes on which selection conditions for a defete of update are specified
- 4. The attributes whose values will be chonzed by an update operation.

Again, the attributes listed previously in irem 5 are cardidates for access structures. On the other hand, the attributes listed in item 4 are candidates for avoiding an access structure, since multipling them will require updature the access structures.

8 Analyzing the Expected Frequency of Invocation of Queries and Iransactions. Besides identifying the characteristics of expected queries and consider their expected rates of invocation. This frequency information along with the attribute internation concreted in each query and transactions is used to complet a cumulative list of expected frequency of using each attribute in each file as a selection attribute or a join attribute, over all the queries and transactions. Generally, for large volumes of processing, the informat "80–20 mle" applies, which states that approximately operations. Therefore, in practical stuations of is rarely increasive to collect exhibition states that approximately stransactions. Therefore, in practical stuations of is rarely increasive to collect exhibition stransactions and transactions, the saturation stransactions and transactions of its sufficient to doername the 20 present or an used in protein or an internal stuations of its rarely increasive to collect exhibition in the stransaction of the processing is accounted to be only 20 present of the queries stransactions.

C. Analyzing the Time Constraints of Queries and Transactions. Seme spaces and transactions may have arringent performance constraints. For example, a transaction may have the constraint that it should terminate within 5 seconds on 95 percent of the

1. For simplicity we use the term (Rey This can be substituted by tables) is lasses or objects

consons when it is involved and that it should never take more than 20 seconds. Such performance constraints place further priorities on the attributes that are candidates for access paths. The selection orthology used by queries and transactions with time constraints become higher-priority candidates for primary access structures.

 Analyzing the Expected Frequencies of Update Operations — A manutum number of necess paths should be specified for a file that is updated frequently, because updating the access paths themselves slows down the update operation-

E Analyzing the Uniqueness Constraints on Attributes. Accessing the should respecified on all candidate key attributes—or sets of attributes—that are entire the pritary key or constrained to be unique. The existence of an index (or other access path) takes it sufficient to search only the index where checking this constraint, since all calues of the attribute will exist in the leaf node- of the index.

Once we have compiled the preceding information, we can address the physical database design decisions, which consist mainly of deciding on the storage structures and access paths for the database files.

#### 16.1.2 Physical Database Design Decisions

Met relational systems representing the law relation as a physical database ide. The actest path options include specifying the type of file for each relation and the attributes on which indexes should be defined. At most one of the indexes on each file may be a primary or clustering index. Any number of idditional secondary indexes can be related.<sup>5</sup>

Design Decusions about Indexing. The surfaces whose values are required in quality or range conditions (selection operation) and these that are keys of that principate in join conditions from operation) require access paths.

The performance of queries largely depends upon what indexes or bashing schemes exist to expedite the processing of selections and joins. On the other hand, during inserdelete, or update operations, existence of indexes adds to the overhead. This overhead must be justified in terms of the gain in efficiency by expediting queries and transactions

The physical design decisions for indexing fall into the following caregories:

 Whether to index an antibute. The attribute must be a key, or there must be some query that uses that attribute either in a selection condition (equality or range of values) or in a join. One factor in favor of setting up many indexes is that some queries can be processed by just scatting the indexes without returning any data.

<sup>2.</sup> The reader should review the various types of indexes described in Section 1911. For a device undextunding of this discussion, it is dominantly to be funder with the algorithms to query process inglishings of the Oppier 15.

- 2. What attribute or attributes to index on: An index can be constructed on one or multiple attributes. It multiple attributes from one relation are involved together in several queries, (for example, (garment\_style\_#, color) in a garment inventory database), a multiple index is warranted. The ordering of attribute-within a multiple findex must correspond to the queries. For example, the above index assumes that queries would be based on an ordering of colors within a garment style \* justice view.
- 3. Whether to service element orders: At most one index per table can be a primary or clustering index, because this implies that the file be ploysically ordered on dot attributes. In most R109858, this is specified by the keyword v1005E8. (If the attribute is a key, a primary index is created, whereas a clusterine index is created if the attribute is not a key.) If a table requires several indexes, the decision about whach one should be a clustered index depends upon whether keeping the table ordered on that attribute is needed. Range queries benefit a great deat from clustering. If several attributes require range queries, relative benefits must be evaluated before deciding which attribute to cluster on, if a query is to be answered by doing an index search only (wadant turneving data records), the corresponding index should nor be clustered, since the main benefit of clustering is achieved when retracting the records themselves.
- 4 Whether to use a hash index over a free index. In general, RDDMSs use B1-trees far indexing. However, is AN and hash indexes are also provided in some systems (see Chapter 144. B1-moss apport both equality and range queries on the attribute used as the search, key. Hash indexes work well with equality conditions, particularly during joins to find a matching record(s).
- 5 Whether to use dynamic hashing for the file. For files that are very colutile that a, these that grow and shinek continuously one of the dynamic hashing schemes discussed in Section 13.9 would be suitable. Currently, they are not offered by most continuously.

Denormalization as a Design Decision for Speeding Up Queries. The ultimate evol during normalization (see Chapters 10 and 11) was to separate the logically related articlutes into tables to minimize redundancy, and thereby avoid the update anomalies that lead to an estimplicacies goverhead to maintain consistency in the database.

The above ideals are sometime? sacrificed in fover of faster execution of frequentioccurring queries and transactions. This process of storing the logical database design (which may be in FCKT or 4NT) in a weaker normal form, see CNT or 188; is called **denormalization**. Typically, the designer adds to a table attributes that are needed for answering queries or producing reports so that a join with another table, which contains the newly added attribute, is avoided. This teintischoles a partial functional dependency or a transitive dependency into the table, thereby creating the associated reduiching problems (see Chapter 10).

Other forms of denormalization consist of storing extra tables to maintain original functional dependencies that are lest during a 90%F decomposition. For example, Figure 10.13 showed the "famil(studies), cosse, instructar,) relation with the functional dependencies {{studest, subset}  $\rightarrow$  instruction, instruction  $\rightarrow$  tooset). A lossless decomposition of teachanto f1(studest, instruction) and t2(instruction, course) does not allow quoties at the term "what course did student Smith take from Instructor Navathe" to be answered without jumming 11 and t2. Therefore, storing f1, f2, and teach may be a possible solution, which reduces the design from FGNF to 3NF. Here, teach is a materialized jum of the other two tables, representing an extreme scherelamey. Any updates to t1 and t2 would have to be applied to 1940. An alternate stratege is to consider 11 and 12 as updatable base tables whereas teach can be created as a view.

# 16.2 AN OVERVIEW OF DATABASE TUNING IN Relational Systems

After a database is deployed and is in operation, actual use of the applications, transactions, queries, and views reveals factors and problem are is that may not have been accounted for during the initial physical design. The impute to physical design listed in Section 16.1.1 can be revised by gathering actual statistics about usage patterns. Resource indication as well as internal DBMS processing—such as query optimization—can be monnored to reveal bottlenecks, such as contention for the same data or devices. Volumes of a river and sizes of data can be better estimated. It is therefore becessary to monitor and noise the physical database design constantly. The gashs of tursing are as follows:

- Trimake applications run taster.
- To lower the response time of queries/mansactions.
- To improve the overall throughput of transactions.

The dividing his between physical design and turning is very thin. The same design decisions that we discussed in Section 10.2.3 are revisited during the turning phase, which is a continued adjustment of design. We give only a brief overview of the turning process below.<sup>4</sup> The inputs to the turning process include statistics related to the forms mentioned in Section 10.1.4. In particular, DBass can internally collect the following statistics.

- Sees of and vidual tables
- Number of distinct values in a column
- The number of times a particular query or transaction is submitted/escented in an interval of time
- The times required for different phases of query and transaction processing (for a given set of queries or transactions).

Character readers should consult Shoshov (1997) for a detailed discussion of taxing

These and other statistics create a profile of the contents and use of the database. Other information obtained from monitoring the database system activities and processes includes the following:

- Storage statistics: Data about allocation of storage into tablespaces, indexspaces, and buffer ports.
- iff and dorse performance statistics. Total read/write activity (paging) on disk extension and disk hot spirits.
- Query/transation processing variance: Execution times of queries and transactions, optimization times during query optimization.
- Looking/logging valued stations: Rates of using different types of locks, transaction throughput rates, and log records activity.<sup>1</sup>
- bidevisionsness: Number of levels in an indexcounder of noncorriguous leaf pages, erc.

Many of the above statistics relate to transactions, concurrency control, and recovery, which are to be discussed in Chapters (7) through 19. Turning a database involves dealing with the following types of problems.

- How to avoid excessive lock contention, thereby increasing concurrency anone transactions
- How to minimize overheard of logging and intraversary dunping of data
- How to optimize butter size and scheduling of processes.
- How to allocate resources such as disks, RAM, and processes for an st efficient infinition.

Most of the previously mentioned problems can be solved by sorting appropriate physical 1906s parameters, changing configurations of devices, changing operating system parameters, and other similar activities. The solutions rend to be aboutly tred to specific systems. The DRAs are typically trained to build thus problems at turning for the specific DRAs. We briefly discuss the turning of various physical database design decisions below.

#### 16.2.1 Tuning Indexes

The initial choice of indexes may have to be revised on the following reasons:

- · Certain queries noy take too long to run for lack of an index
- · Certain indexes may not get ordired of all,
- Certain indexes may be cousing excessive overhead because the index is on an attribute rbat undergoes frequent changes.

Most 198454 have a command or trace facility, which can be used by the PPA to as the system to show how a guery was executed—what operations were performed in what order and what secondary access structures were used. By analyting these execution plane.

4. The reader will need to look alread and revices thapters 17, 19 for explanation of these terms,

it is possible to diagnose the croses of the above problems. Some indexes may be dropped and write new indexes may be created based on the tuning analysis.

The goal of turner is to dynamically evaluate the requirements, which sometimes illustrate sensenally or during different times of the month or week, and to rearganize the indexes to yield the best overall performance. Dropping and building new indexes is an averaged that can be justified in terms of performance emprovements. Updating of a table is generally suspended while an index is dropped or created; this loss of service must be acounted for. Besides dispping of creating indexes and changing from a neighbor of service many deficitors on the index key, index pages matematic wasted space, which can be changed during a rebuild operation. Similarly, too is insertions in a clustered index that affect performance Rebuilding a clustered index amounts to reorganizing the online table index that affect performance is described index and clustered index that affect performance is a clustered index another to reorganize the index that affect performance is descent.

The available opposes for indexing and the way they are defined, created, and toigandoi varies from system to system. Just for illustration, consider the sparse and dense indexes of Chapter 14. Sparse indexes have one index painter for each page (disk block) in the data files dense indexes have an index painter for each usual. Sybase provides chartering indexes as sparse indexes in the form of B'strees whereas INGRES provides sparse clattering indexes as ISAM files, and dense clustering indexes as B'strees. In some versions (4 Gracle and G82, the option of setting up to clustering index is limited to be dense index (with more index entries), and the PAA has powerly with this limit, for

#### 16.2.2 Tuning the Database Design

We already discussed in Section 16.12 the need for a possible denormalization, which 000 depoture from keeping off tables as PCINF relations. If a given physical database design doe not ment the expected objectives, we may revert to the foreigal database design, make aljustments to the logical solution, and nemap if to a new set of physical tables and rulexes.

As we pointed out the entrie database design has to be direct by the processing reportances as much as by data requirements. If the processing reportance)s are drawningly changing, the design needs to respond by making chances to the conceptual schema of necessary and to reflect those changes into the logical schema and physical dyard. These changes may be of the following nature:

- Existing rables may be printed (demonstrated) because corrang attributes from two enmore tables are frequently needed together. This reduces the normalization level from PCNF ro BN-, 2NF, or LNF.<sup>5</sup>
- For the green set of tables, there may be idternative design choices, all of which achieve 3NF or RONE. One may be replaced by the other.

3. Note that 498- and 700 a lduess dutered, bries of problem dependencies which are independent. Stark other hence the normalization (or den simplification) order benyon: them is arburary.

- A relation of the ferm R(E, A, B, C, D, ...,)—with K as a set of key attributes—that is in R(NF can be solied into moltiple tables that are also in ECNF—for example, R1(E, A, B), R2(G, C, D, ), R3(E, ...,1—by replicating the key K in each table Each table groups sets of an ributes that are accessed together. For example, the table Each table groups sets of an ributes that are accessed together. For example, the table Each table groups sets of an ributes that are accessed together. For example, the table Each table groups sets of an ributes that are accessed together. For example, the table Each table groups sets of an ributes that are accessed together. For example, the table Each table groups the end of the example, the table environce (SSN, Name, Phone) and EM-Z(SSN, Grade, SaNary). If the oriental table hada very latge number of tows (say 160,000) and queries about phone combens and salar information are totally distinct, this separation of tables may work better. This is also called vertical partitioning.
- Attrabute(s) from one table may be repeated as another over though this creates reducted dancy and a potential anomaly for example. Partnane may be replicated in tables when over the Part# appears (as foreign key), but there may be one master table called rate wster(Part#, Partnare, ....) where the Partnare is guaranteed to be up rochus.
- Just as vertical partitioning splits a table vertically into multiple tables, horizontal partitioning takes horizontal shoes of a table and states them as detined tables. For example, product sales data may be separated into ten tables based on ren product hree. Each table has the same set of columns (attributes) but contains a distinct set of products (tuples). If a query or transaction applies to all product data, it may have to run against all the tables and the tesults may have to be combined.

These types of adjustments designed to meet the high volume queries or transactions with or without sacrificing the normal forms, are commonplace in practices

#### 16.2.3 Tuning Queries

We already discussed how query performance is dependent upon appropriate selection of indexes and how indexes may have to be funed after analyzing queries that give poor performance by using the commands in the RAMMS that show the execution plan of the query. There are mainly two indications that suggest that query turing may be needed:

- A query issues for many disk necesses (for example, an exact match query stars on entire table).
- The query plan shows that relevant indexes are not bring used.

Some repital instances of structions prompting query running include the following:

- Mony query optimizers do not use indexes in the presence of arithmetic espressions (wich as socaes/365.5.10, 50), numerical comparisons of artributes of different sites and precision (wich as way = 60.5 where any is of type isolation and excision (wich as way = 60.5 where any is of type isolations), will comparisons (such as eovice its volu), and substring comparisons (such as usage itself "grass").
- 2. Indexes are often not used for nested queries using 16 for example, the query

```
SELECT SSN FROM EMPLOYEE
MHERE OND IN (SELECT ONUMBER FROM DEPARTMENT
WHERE MGRSSN = "333445555");
```

may not use the undex on D60 m E80.0466, whereas using D60 = D606866 in the 66666. Clause with a single block query may cause the molex to be used.

- Some 915TOCTS may be redundant and can be avoided without changing the result. A 015TEGT often causes a surraperation and must be avoided as far as possible.
- Upnecessary use of remporing result jubles can be avoided by collapsing multiple queries into a single query index the temporing relation is necked for some interinediate processing.
- In some objections involving use of correlated queries, temporates are useful. Consider the query.

```
SELECT SSN
FROM FMPLOYEF F
MHERE SALARY - SELECT NAX (SALARY)
FROM EMPLOYEE AS M
MNERE M.DND - E.DND;
```

This has the puteritual danger of scarching all of the inner everyte table M tor such taple more the outer everyte table §. To make it more efficient, it can be broken into two queries where the first query just computes the moximum salary in each department as follows:

```
SELECT MAX (SALARY) AS HICHSALARY, ONO INTO TEMP
FROM EMFLOYEE
CROUP BY OND:
```

```
SELECT SSN
FROM FMPL(IYEF, TEMP
WHERE SALARY = HIGHSALARY AND EMPLOYEE.OND = TEMP.DND;
```

- 7. One ideoxinerary with query optimizers is that the order of tables in the conscious error affect the join processing of that is the case, end may have to switch this order so that the smaller of the two relations is scanned and the larger relation is used with an appropriate index.
- Some query optimizers perform worse on nested queries compared to their equivalent annested counterparts. These are loar types of nested queries:
  - Uncorrelated subquences with aggregates in timer query.
  - Uncorrelated subspectics without aggregates.
  - Correlated subspecies with aggregates in ormer energy
  - Correlated subqueries without aggregates.

Out of the above four types, the first one typically presents no problem, since most query optimizers evoluate the inner query once. However, for a query of the

second type, such as the example in  $(\lambda)$  above, most quirty optimiters may not tree on index on use in two creds. The same optimiters may do so if the query is written as an ennested query. Transformaticity of correlated subgriefies may involve setting temporary tables. Detailed examples are outside our -cope here.<sup>6</sup>

9. Finally, many applications are based on views that define the data of interest to those applications. Sometimes, these views become an inverkall, because a query may be possibleneetly against a base table, rather than going through a view that is defined by a join.

#### 16.2.4 Additional Query Tuning Guidelines

Additional rechniques for improving queries apply in certain situations:

 A query with multiple selection conditions that are connected via or may not be prompting the query optimizer to use any index. Such a query may be split up and expressed as a union of queries, each with a condition on an attribute that cause an index to be used. For example,

```
SELECT FNAME, LNAME, SALARY, AGE
FROM EMPLOYEE
WHERE AGE - AS OR SALARY - SCOOD;
```

may be executed using sequential scan giving poor performance. Splitting if up a

```
SELECT FNAME, LNAME, SALARY, AGE
FROM ENPLOYEE
NHERE AGE 1: 45
```

```
UNION
Select Fname, Lname, Salary, Age
From Employed
Where Salary < S0000;
```

may othlice indexes on war as well as on socary

- To herp in expedicing a query, the following transformations may be med.
  - Are condition may be transformed into a positive expression.
  - Embedded select blocks using 16, all, Ave, and Sour may be replaced by joints.
  - If an equality join is set up between two tables, the range predicate (selection condition) on the joining attribute set up in one table may be repeated for the other table.

<sup>6.</sup> For further der tils, der Shashn (1992).

We involuted the scheme and used set in CPPC arET indiced of RUCTE.

 were conditions may be receiven to otilize the radixes or multiple columns. For symple.

```
SELECT REGIONA, PROD_TYPE, WONTH, SALES

FROM SALES STATISTICS

WHERE REGIONA = 3 AND ((PRODUCT TYPE BETWEEN 1 AND 3) OR (PRODUCT

TYPE BETWEEN 8 AND 10));

may use on index only on sector# undecords through all leat pages of the ordex for

a match on Paceoul_iver. Instead. using

SELECT REGIONA, PROD TYPE, MONTH, SALES

FROM SALES_STATISTICS

WHERE (REGIONA = 3 AND (PRODUCT TYPE BETWEEN 1 AND 3)) OR (REGIONA =

3 AND (PRODUCT TYPE BETWEEN 8 AND 10));
```

can use a composite index on (action#, somer\_rsi) and work much more enciently

We have covered in this section most of the common opportunities where in therency of a query may be concerted by some simple concertive action such as using a emporing avoiding certain types of constructs, or avoiding use of crows. The problems and the remothes will depend upon the workings of a query optimizer within an SDEMS. Boarded licenstone ensurements of individual manuals on database tuning guidelines for Juliase administration by the RDEMS vendors.

# 16.3 SUMMARY

In this chapter we discussed the factors that affect physical database design decisions and provided guidelines for choosing among physical design alternatives. We discussed damges to logical design, modifications of indexing, and changes to queries as a part of database mining.

#### Review Questions

- (6.1. What are the important factors that influence physical database design?
- (6.2) Discuss the decisions made during physical database docum-
- 16.3. Discuss the gri delines for physical database design in R. 2008s.
- 164 Discuss the types of modifications that may be applied to the logical database design of a relational database.
- [6.5] Under what struarrows would denormalization of a database scheme bruosci? Owe examples of denormalitation.
- 10€. Discuss the tuning of indexes for relational databases
- 16.7 Discuss the considerations for recyclusting and modifying 5.2 (queues
- 10.5 Illustrate the types of changes to SQL queries that must be worth considering for improving the performance during database juning.
- 168. What dute tions do the typical database design tools provide?

#### Selected Bibliography

Wiederhold (1986) covers all phases of database design, with an emphasis on physical design. O'Neil (1994) has a detioled discussion of physical design and transaction issues in reference in commercial R05Wiss.

Nuvathe and Kerschberg (1986) discuss all phases of database deager and point out the role of data dictionaries. Roten and Shasha (1991) and Carlis and March (1984) present different models for the proslem of physical database design. .



# Introduction to Transaction Processing

The new subsequent chapters continue with more details on the techniques used to support transaction processing. Chapter 18 describes the basic concurrency control techniques, and Chapter 19 presents an overview of recovery techniques.

# 17.1 INTRODUCTION TO TRANSACTION PROCESSING

In this section we informally introduce the concepts of concurrent execution of transactions and recovery from transaction failures. Section 17.1.1 compares single-user manulpinear database systems and demonstrates how concurrent execution of transactions can take place in multiuser systems. Section 17.1.2 defines the concept of transaction and presence sample model of transaction execution, based on real and write database operations, that is used to formalize concurrency control and recovery concepts. Section 17.1.3 shows by informal examples why concurrency control rechtaques are needed in multiuser systems. Finally, Section 17.1.4 discusses why techniques are needed in multiuser systems. Finally, Section 17.1.4 discusses why techniques are needed to permit provery from factors by discussing the different ways in which transactions can tail while executing.

#### 17,1.1 Single-User Versus Multiuser Systems

One criterion for ele-situing a dutabase system is according to the number of osers who can use the system concurrently—that is of the same time. A DIWF is single-user that most one user at a time can use the system, and it is **multius**er of name users can use the system—and hence access the database—concurrently. Single-user (1708s) are mostly restricted to personal computer actors, most other DIWFs are multiuser, for example, an author reservations system is used by hundred- of trave, agents and reservations clerks concurrently. Systems in banks, insurance agencies, stock exchanges, apprimatives, and the like are also operated on by many users who submit transactions concurrently to the system.

Multiple users can necess database—and use computer systems—simultations; because of the concept of multiprogramming, which allows the computer to execute multiple programs—or processes—of the same time. If only a single central processing multiple programming operating systems execute some commands from one process at a time. However, multiprogramming operating systems execute some commands from the next process, and so on A process is resumed at the point where it was suggended wherever it gets its turn to use the CO again. Hence, concurrent execution of processes is actually interleaved, as illustrated in Figure 17.1, which shows rise processes A and B executing concurrently in an interleaved fashion. Interleaving keeps the COD busy when a process imputes an input of operation, such as reading a block from disk. The yTO is witched to execute another process rather than remaining alle during b() time. Interleaving also prevents a long process from delaying other processes.



**FIGURE 17.1** Interleaved processing versus parallel processing of concurrent transactions.

If the composer system has multiple bardward processors (CPUs), parallel processing cfinaltiple processes is possible, as illustrated by processes Claud D in Figure 17.1. Most of the theory concerning concurrency control on databases is developed in terms of interfaved concurrency, so fist the remainder of this chapter we assume this model. In a multiple DBMS, the stored data items are the primary resources that may be accessed concurrently by interactive a-citor of photonex programs, which are constantly retrieving momentum from and modifying the database.

#### 17.1.2 Transactions, Read and Write Operations, and DBMS Buffers

A transaction is an executing program that turns a logical unit of database processing. A transaction includes one or nume database across operations—these can include memory deletion, modification, or territeval operations. The database operations that form a transaction can either be embedded by third an application program or they can be specified interactively volus high-level query language such as SQL. One way of specifying the transaction boundaries is by specifying explicit begin transaction and end transaction statements in an application program in the case, all database access operations between the two are considered as forming one transaction. A single application program may contain note than one transaction of the contains several transaction boundaries. If the database operations in a transaction do not update the database but orderet retrieve data, the transaction is called a **read-only transaction**.

The model of a database that is used to explain injusacion processing concepts is much simplified. A **database** is basically represented as a collection of **named data items**. The size of a data item is called us **granularity**, and use in be a field of some record in the database, or it may be a larger that such as a record or even a whole disk black, but the concepts we discuss are independent of the data item granularity. Using this simplified database model, the basic database access operations that is transaction can include areas follows:

- read\_iten(X): Reads a database trent names! X area a program variable. To simplify
  our rotation, we assume that the program activitie is also named X.
- arite (tea(X): Writes the value of program variable X into the database nemnamed X.

As we discussed an Chapter 13, the basic unit of data transfer from disk to main memory is one block. Executing a read, them(%) communitationalitates the following step-

- Lind the address of the disc block that controns usin X
- Copy that dok block into a buffer in monony (it that disk block is not already in some manememory buffer).
- 3. Copyritem X from the buffer to the program variable conied X-

Executing a wryte\_iten(X) command includes the following steps:

- 1. Find the address of the disk block that contours item X
- Copy that disk block into a butter in main memory (if that disk block is not already in some main memory butter).
- Copy trent X from the program vanishe named X into its correct location in the buffer.
- Struct the applaced block from the buffer back to disk (either mimediately or it some brief point in time).

Step 4 is the one that actually updates the database on disk. In some cases the batter is not immediately stored to disk, in case additional changes are to be made to the batter ithanily, the decision about when to store back a modified disk block that is in a manmemory buffer is implied by the receivery manager of the D0018 in cooperation with the inderlying operating system. The D001s will generally maintain a number of **buffers** in major memory that bold database disk blocks containing the database trans being processed. When these buffers are all occupied, and additional database Blocks mast be copied into memory, some *buffer* replacement policy is used to choose which of the current buffers is to be replaced. If the chosen buffer has been modified, it must be written back to disk before it is reused.

A masserion includes read\_item and arise litem operations to access and opdate the database. Figure 17.2 shows examples of two cere simple transactions. The read-set of a transaction is the set of all items that the transaction reads, and the write-set is the set of all items that the transaction writes. For example, the read set of 7 item figure 17.2 is [X, Y] and its series et or also [X, Y].

Concurrency control and recovery merborisins are morely concerned with the database access commands in a transaction. Transactions submitted by the various users into

We will not discuss botter reglectoment policies acreate these are reparally decreased in operating systems revelsables.

| 리 | <u> </u>                     | φ; | τ,                               |
|---|------------------------------|----|----------------------------------|
|   | read_item $\{X\}$ .          |    | reac_item $(X)$ .                |
|   | $X = X \cdot N;$             |    | X -X+65;                         |
|   | wrie $\operatorname{dem}(X)$ |    | write_item $\langle X \rangle$ ; |
|   | read dam (Y):                |    |                                  |
|   | 10=Y+9;                      |    |                                  |
|   | wrie dam (Y)                 |    |                                  |

FIGURE 17.2 Two sample transactions, (a) Transaction  $T_{12}$  (b) Transaction  $T_{22}$ 

execute concurrently and may access and update the same database items. If this concurrent eventum is uncontrolled, it may lead to problems, such as an inconsistent database. In the next section we intermally introduce some of the problems that may occur.

#### 17.1.3 Why Concurrency Control Is Needed

Several problems can occur when concurrent transactions execute in an unconstelled namer. We illustrate some of these problems by referring to a much simplified online reservations database in which a record is stored for each at line flight. Each record includes the number of reserved seats on that flight as a noned data (contraining) other informations figure 17.2a shows a transaction  $T_1$  that theorems N reservations from one flight whose number of reserved seats is stored in the database item named X to another flight whose number of reserved seats is stored in the database item named X to another flight whose number of reserved seats is stored in the database item named X. Figure 17.2b shows a simpler transaction  $T_1$  that protineer of M seats on the first flight (X) referenced in transacnon  $T_1$ . To simplify our example, we do not show addimined portions of the transactions, such as checking whether a flight has enough scars available before reserving additional core.

When a database access program is written, it has the flight numbers, their dates, and the number of sours to be booked as parameters; hence, the same program can be used to execute more transactions, each with different flights and numbers of sours to be booked for concorrency control purposes, a transaction is a formulate execution of a program on a specific date, flight, and number of sours. In Figure 17.2a and b, the transactions  $T_1$  and  $T_2$  are specific execution of the programs that refer to the specific flights whose transfers of sears are stored in data items X and Y in the database. We now discuss the types of problems we may encounter with these two transactions if they run concurrently.

The Lost Update Problem. This problem occurs when reactions data access the same database items have then operations interference in a way that makes the value of since database items inconect. Suppose that transactions  $T_{1}$  and  $T_{2}$  are submitted at approximately the same time, and suppose that their operations are interferenced as shown

L A smilling more comments used example assumes a bonk database, only one transaction doing moder of tands from account Y to account V and the observation doing a deposit resection X In Figure 17.34, then the trial value of trein X is incorrect, because T<sub>1</sub> reads the value of X before T<sub>1</sub> changes in in the database, and hence the opdated value resulting from T<sub>1</sub> is low. For example, if X = 30 of the start foriginally there were 80 restrictions on the flight). N =  $3 (T_1 )$  transfers is start reservoirions from the flight corresponding to X to the flight extresponding to Y), and M = 4 (T<sub>1</sub> reserves 4 scars on X), the final result should be X = 36 b or in the interleaving of operations shown in Figure 17.35, it is X = 84 because the update in T<sub>2</sub> that results of the five scars from X way law.

The Temporary Update (or Dirty Ruad) Problem. This problem access when one transaction opdates a database iron and then the transaction toils for some traser (or Section 174.4). The updated atem is accessed by another transaction before travecharged



FIGURE 17.3 Some problems that occur when concurrent execution is uncostrolled. (a) The lost update problem. (b) The temporary update problem.



**EGURE 17.3(CONTINUED)** Some problems that occur when concurrent execution is incontrolled. (c) the incorrect summary problem.

lack to its original value. Egure 17.3b shows an example where  $T_1$  updates item 3 and then fails before completing, so the system must change X back to its original value. Before it can do so, however, transaction  $T_2$  reads the "temporary" value of X, which will not be recorded permanently in the database because of the failure of  $T_1$ . The value of item X that is read by  $T_1$  is called doity data, because it has been created by a transaction that has not completed and committed yet; hence, this problem is also known as the ifing read perform

The Informed Summary Problem. If one transaction is calculating an aggregate summary function on a number of records while other transactions are updating some or dese records, the aggregate function may calculate some values before they are updated and others after they are updated. For example, suppose that a transaction  $T_{x}$  is calculating the rotal number of reservations on all the flights; meanwhile, transaction  $T_{y}$  is calculating if the interleaving of operations shown in Figure 17.3c occurs, the result of  $T_{y}$  calculated from it but reads the value of Y feibre those N seats have been added to it.

Another problem that may occur is called **unrepeatable read**, where a transaction T reads an item index and the item is changed by another transaction T' between the two reads. Hence, T receives different eables for its two reads of the same item. This in is occur in example, if during an airline reservation transaction is custofied with the vestories is suggring about sea availability on several dights. When the customer decides on a particular flight, the transaction ther reads the number of scats on that dight a second true before completing discussions.

#### 17.1.4 Why Recovery Is Needed

Whenever a transaction is submitted to a 6280 for execution, the system is responsible for making size that other (1) all the opticitums in the transaction are completed size costfully and their effect is recorded permanently in the database, or (2) the transaction has the effect where every the database or on any other transactions. The DBMs mucnor permit some operations of a transaction T to be applied to the database while other operations of T are not. This may happen if a transaction fails after exercising some due operations by before executing all of them.

**Dypes of Eachines**. Engloyes are generally classified as transaction, system, and mode follows: **There** are several possible reasons for a transaction rootal on the modèle of execution:

- A computer fallow (system cosh). A hardware, software, or network error occurs in the computer system during transaction essention. If industre crushes are estably media tailore—for example, main memory failure.
- 2. A transaction or system error. Some operation in the transaction may couse it to find, such as integer overflow or division by zeros. Transaction failure may also occur because of enrolieous parameter values or tecause of a logical programming error. In addition, the user may intertup: the transaction during its excention.
- 3. Local errors in conditions detected by the transaction. During transaction execution, certain conditions may occur that necessitate cancellation of the transaction. For example, data for the transaction may not be found. Notice that in exception condition.<sup>4</sup> such as insufficient account balance in a binking data base, may cause a transaction, such as a fund withdrivial, to be canceled. The exception should be programmed in the transaction itself, and hence would not be considered if fully.
- 4 Concurrences channel enforcement: The concurrence, control method (see Chapter 18) may decide to about the transaction, to be restarted later, because it makate serialitability (see Section 17.5) or because several transactions are in a state of deadleck.
- 5. Disk failure. Some disk blacks may lose their data because of a read or write redfurction or because of a disk mod/write bead crash. This may happen during a read or a write operation of the transaction.
- o. Physical problems and catastoticher. This refers to an endless list of problems that includes power or air-conditioning folure. Inc. theft, substage, overwriting doss or tapes by anistake, and mounting of a wrong type by the operator.

In general, a transie non-should be thoroughly rested to ensure that it has no long thereal ongroupping energy.

<sup>4.</sup> Exception conditions, it programmed correctly, do not spirit (ore framos tion follares

Eadures of types 1, 2, 3, and 4 are more common than those of types 5 or 6. Whenever a failure of type 1 through 4 occurs, the system must keep sufficient information to recover from the finiture. Disk failure or other catastrophic failures of type 5 or 6 do not happen frequently; if they do occur, recovery is a major task. We discuss recovery fram tailore in Chapter 19.

:

:

The concept of transaction is fundamental to non-y rechargies for concurrency anticoland recovery from failures.

# 17.2 TRANSACTION AND SYSTEM CONCEPTS

in this sector, we discuss additional concepts (elevant to transaction processing. Sector, (7.2.1) describes the various states a transaction can be in, and discusses additional reledat operations needed in transaction processing. Section 17.2.2 describes the system log, which keeps information model for mergery. Section 17.2.3 describes the concept of communications of transactions, and why they are important in transaction processing.

#### 17.2.1 Transaction States and Additional Operations

A transaction is an atomic unit of work that is either completed in its entirety or not done at all the tree overy purposes, the system needs to keep track of when the transaction stats, terminates, and commute or aborts (see Section 37.2.3). Hence, the recovery manzer keeps track of the following operations:

- BEED HOWSELLERS This marks the beginning of transaction execution.
- Ref. or write: These sporty real or write operations on the database items that are executed as part of a transaction.
- one\_consections: This specifies that non-indiance transaction operations have ended and marks the end or transaction execution. However, at this point it may be necessary to check whether the changes introduced by the transaction can be permanently applied to the database (committed) of whether the transaction has to be aborted because it violates senalizability (see Section 17.5) or for some other reason.
- rossit\_tasssection: This signals a successfid oid of the transierion so that any changes (updates) executed by the transaction can be safely committed to the database and will not be underse.
- ACLESAGE (OR ABORD): This signals that the transaction has ended towaccessfully, so that any changes or effects that the transaction may have applied to the database must be include.

Figure 17.4 shows a state transition diagram that describes how a transaction moves through its execution states. A transaction goes into an **active state** immediately after it state execution, where it can issue as cand with operations. When the transaction ends, it moves to the **partially committed state**. At this **point**, some recovery proceeds need to ensure that a system to lore will not result in an mability to record the changes of the



FIGURE 17.4 State transition dragtam dilustrating the states for transaction execution

transaction permanently (usually by recording changes in the system log, discussed in the next section).<sup>5</sup> Once this check is soccessful, the transaction is said to have reached in communication points and others the committed states Communicate discussed in more detail in Section 17.2.3. Once a transaction is committed at hus concluded its escatter successfully and all its changes must be recorded permanently in the database.

However, a transaction can go to the faded state it one of the effects fails of a the transaction is aborted during its active state. The transaction may then have to be folled back to undo the effect of its write operations on the database. The transaction information that is non-transaction backs while the transaction has been running is removed when the transaction trummates. Failed or aborted transactions may be estated fater—cities information being resubmitted by the user—as brand new transactions.

#### 17.2.2 The System Log

To be able to recover from futures that affect transactions, the system maintains a log to keep track of all transaction operations that affect the values of dualastic ateas. This information may be needed to permit recovery from failures. The log is kept on disk, so it is not affected by any type of future except for disk or catastrophic failure. In addition, the log is periodic dly backed up to archivel a torage (rape) to guard against such catastrophic failures. We now hat the types of entries—colled log records—that are performed to perform a these entries. The log is not account the log onder the action each performs. In these entries, Thefers to a unique transaction d that is generated account of by by the spreat and is used to identify each transaction.

4. [start [transaction,?]] Indicates that transaction 2 has started execution.

Optimized concorrect control (see Section 18.4) decomposes that certain class is be made at this point to prove that the transaction del not optimized on with other executing transactions.

<sup>6.</sup> The Sig has sometimes been called the 1950 querical.

- 'write\_item,7,X.old calacture rule is indicates that transaction T has charged the value of database etem X from adjustice to rate\_calac.
- 3. (read\_) ten,7,X): Indicates that transaction 7 has read the value of database noneX
- 4 (contribut): Inductes that transaction T has completed successfully, and attrans that its effort car, be communed (recorded permanently) to the database.
- 5. [abort,7]: Inducates that transaction, 7 has been aborted.

Protocols for recovery that avoid cosciding follbacks (see Section 17.4.2)—which include nearly all practical protocols—do not require that READ operations be written to the system log. However, if the log is also used for other pargoses—such as indiring Reeping track of all database operations)—then such entries can be included. Its addition, some recovery protocols require simpler with entries that do not include new , value (see Section 17.4.2).

Notice that we assume here that all permanent changes to the database occur within tanactions, so the notion of receivery from a transaction failure and/ons to either value or sedestic transaction operations individually from the log. If the system crashes, we can recover to a consistent database state by examining the log and using one of the techniques descarbed in Chapter 19. Because the log contains a record of every skills operations that changes the value of some database itera, it is possible to undo the effect of these skills operations of a transaction T by ending backward through the log and resting all items changed by a some operation of T to their **old values. Redning** the operations of a transaction fail if all its updates are recorded in the log but a failure occurs before we can be sore that all these **new\_values** have been written permanently in the actual database on disk.<sup>7</sup> Redning the operations of transaction T is applied by trooping forward through the log and setting all items changed by a sore operation of a transaction T is updates are recorded in the log but a failure occurs before we can be sore that all these **new\_values** have been written permanently in the actual database on disk.<sup>7</sup> Redning all items changed by a sore operation of T to their solution of T to their new\_values.

#### 17.2.3 Commit Point of a Transaction

A transaction of teaches its commit point when all its operations that access the datable assorber executed successfully and the effect of all the transaction operations on the database have been received in the log. Beyond the commit point, the transaction result to be committed, and its effect is assumed in the permanently recorded in the datalese. The transaction then writes a commit record (commit, T) into the log. It a system (shift occurs, we search back in the log for all transactions T that have written a 'start transaction,T] record into the log for all transactions T that have written a 'start transaction,T] record into the log for all transactions to their (commit,T] record yet; these transactions may have to be rolled back to undo them effect on the database during the recovery process. Transactions that have written their commit word in the log must also have recorded all their with coperations in the log, so ther effect on the database during the log must also have recorded all their with coperations in the log, so ther effect on the database during the log form the log records.

<sup>7</sup> Undo and redo are docussed more ruly in Chapter 19,

Nonce that the log file must be kept or slick. As decreased in Chapter 13, updating a disk file involves depeng the appropriate block of the file from disk to a buffer to man memory, updating the buffer in more intervery, and copping the buffer to disk (r) is common to keep one or more blocks of the log file in main memory buffers until they are filled with log entries and them to write them back to disk only once, rather than writing to disk every time a log entry is added. This saves the overhead of multiple disk writes of the same log file block. At the time of a system crush, only the log entries that have been urated back to disk one considered in the recovery process because the contents of man memory may be lost. Hence, *before* a transaction reaches at contact point, any portion at the log that has not been written to the disk yet main new be written to the disk. The process is called force-writing the log file before committing a transaction.

### 17.3 DESIRABLE PROPERTIES OF TRANSACTIONS

Transactions should possess several properties. These are often called the ACID properties, and they should be enforced by the concattency control and recovery methods of the DBMS. The following are the 3000 properties.

- Atomicity: A transaction is an atomic unit of processing, it is either performed in its entirety or nor performed at all.
- Consistency preservation: A transaction as consistency preserving if its complete execution take(s) the database from one consistent state to another.
- 3 Isolation: A transaction should appear as though it is being executed in isolation from other transactions. That is, the execution of a transaction should not be interfered with by any other transactions executing concurrently.
- Durability or permanency: The changes applied to the database by a compatied transaction must persist in the database. These changes must not be lost because of any failure.

The atomicing property requires that we execute a transaction to completion, his die responsibility of the transaction recovery subsystem of a DBas to ensure arometers has transaction task to complete for some reason, such as a system crisik on the multi-oftransaction execution, the recovery technique must undo any effects of the transaction on the dutabase.

The preservation of consistency is generally considered to be the responsibility of the programmers who write the database programs or of the DBMS module that enforces integring constraints. Recall that a **database state** is a collection of all the stored database (value-) in the database of a given point in time. A **consistent state** of the database satisfies the constraints specified in the schema as well as any other constraints that should hold on the database of a doubtiet program should be written in a way that guatantees that, if the database of its consistent state before executing the transactional will be in a consistent state after the complice execution of the transaction, assuming that so interference with edge transactions occuts.

Isolation is entered by the concurrency control subsystem of the 1988. If every transaction does not make its cyclutes inside to other transactions until it is committed, the form of volation is entered that solves the temporary update problem and chromates taxading colliques (see Chapter 19). There have been arranges to define the level of volume of a transaction. A transaction is and to have level 0 terrol isolation that does not overwrite the dirty reads of higher level transactions. Eved 1 terrol isolation has no lost cyclute; and level 2 isolation bas no level updates and no dirty reads. Finally, level 3 octation takes called the isolation) has, in addition to degree 2 properties, reportable teals.

Finally, the durability property is the responsibility of the recovery subsystem of the area. We will discuss how recovery protocols enforce durability and atomicity in Chapter 19.

# 17.4 CHARACTERIZING SCHEDULES BASED ON RECOVERABILITY

When transactions are executive concurrently in an interferies' historic then the order of execution of operations from the various transactions is known as a schedule (or hisiony). In this section, we first define the verycept of schedule, and then we characterize the types of schedules that docurrent recovery where failures occur. In Section 17.5, we characterize schedules materials of the interference of patterpating transactions, leading to the concepts of schedules that and scratterable schedules.

#### 17.4.1 Schedules (Histories) of Transactions

A schedule (or history) S of a transactions  $T_0, T_1, \ldots, T_n$  is on ordering of the operations of the reinsactions subject to the constraint that, for each transaction T that participates in S, the operations of T, in S must appear in the same order in which they occur in T. Note, however, that operations from other transactions T, can be interfluxed with the operations of T<sub>i</sub> in S. For now, consider the order of operations in S to be a rotal calcular, differing it is possible theorem colly to deal with schedules whose operations form facility (as we discuss later).

For the purpose of recovery and concountered control we are mainly interested in the read inter and write\_item operations of the transactions, as well as the commit and short operations. A shorthand notation for describing a schedule use- the symbols (.a.c. and a for the operations read\_item write\_item, commit, and abort, respectively, and appends as subscript the transaction of (transaction nomber) to each operation in the shedule. In this notation, the database item X that is read erorither to how the transaction.

operations in parentheses. For example, the schedule of Figure 17.3(a), which we shall call  $S_{ab}$  can be written as follows in this notation:

 $S_2(z_1(X), \pi_2(X), w_1(X), z_1(Y), w_2(X), w_1(Y),$ 

Similarly, the schedule for Former 17.3(b), which we call  $S_{\mu\nu}$  can be written as follows: if we assume that transactions  $T_{\mu}$  about other its read\_iten(3) operation:

 $S_{\mathcal{R}}(r_{1}(X), s_{1}(X), s_{2}(X), m_{2}(X), r_{1}(Y), s_{1}$ 

Two operations in a schedule are sold to conflict if they sotisfy all three of the following conditions: (1) they belong to different transactions; (2) they access the same item  $X_i$  and (3) at least one of the operations is a write\_item(X). For example, in schedule  $S_i$ , the operations  $v_i(X)$  and  $w_i(X)$  conthit, as do the operations  $v_i(X)$  and  $w_i(X)$  and the operations  $v_i(X)$  and  $w_i(X)$  theorem the operations  $v_i(X)$  and  $w_i(X)$  and  $w_i(X)$  and the operations  $v_i(X)$  and  $w_i(X)$  thowever, the operations  $v_i(X)$  and i(X) do not conthet, since they are both read operations: the operations  $w_i(X)$  and  $w_i(Y)$  do not conthet, because they operate on distinct data items X and Y and the operations  $v_i(X)$  and  $w_i(X)$  and  $w_i(X)$  do not conthet, because they because they belong to the same transaction.

A schedule S of a transactions  $\Gamma_0, \Gamma_2, \dots, T_n$  is said to be a complete schedule if the following conditions fold:

- 1. The operations in S are exactly those operations in  $T_1, T_2, \ldots, T_n$  including a committer above operation as the last operation for each transaction in the schedule.
- For any port of operations from the some transaction T<sub>1</sub>, their order of appearance in S<sub>1</sub> is the some as their order of appearance in T<sub>2</sub>.
- For any two conflicting operations, one of the two most occur before the other in the schedule."

The preceding condition (3) allows for two nonconflicting obtained to secur in the schedule without defining which occurs first, thus leading to the definition of a schedule as a **partial order** of the operations in the a transaction a<sup>16</sup> However, a total order must be specified in the schedule for my gau of conflicting operations (condition 4) and for an part of operations from the same transaction (condition 2). Condition 1 simply states that all operations in the transactions must appear in the complete schedule. Since even transaction has either committeel or aborted, a complete schedule will not contain an active transactions at the end of the schedule.

In general, it is difficult to encounter complete schedules in a transaction processing system, because new transactions are continually being submitted to the system. Hence it is includ to define the concept of the committed projection C(S) of a schedule S, which includes only the operations in S that belong to committed transactions—that is, transactions T, where committ operations  $c_1$  is in S.

<sup>9.</sup> Theoretically of a not necessary to determine an order between pairs of an only bug dystations of the practice mass set values have a total order of operations. It product processing is conformation to theoretically possible to have achieved with particily ordered noncombeting operations.

#### 17.4.2 Characterizing Schedules Based on Recoverability

For some schedules in is easy to receiver from transaction failures, whereas for other schedides the recovery process can be quite involved. Elerce, it is important to characterite the ripes of schedules for which receivery is possible, as well as those for which recovery is relinvely simple. These characterizations do not actually provide the receivery algorithm but ristead only attempt to theoretically characterize the different types of schedules.

First, we would like to ensure that, once a transaction T is committed, it should necocenceessary tonoll luck T. The schedules that theoretically meet this enterior are called score the schedules and those that do not are called **numeeoverable**, and hence should not be permitted. A schedule S is accoverable if no transaction T in S commits until all transactions  $T^*$  that have written an item that T reads have committed. A transaction Treals from transaction  $T^*$  in a schedule S it semication X is first written by  $T^*$  and face real by T. In addition,  $T^*$  should not have been about of before T reads item X, and there is all by T. In addition,  $T^*$  should not have been about of before T reads item X, and there is and be no transactions that write X about  $T^*$  writes it and before T reads v (upless these parameters, it any, have abouted before T reads X).

Recoverable schedules require a complex recovery process as we shall see, but if sufficient information is kept (in the log), a recovery algorithm can be devised. The (partial) schedules  $\delta_{\mu}$  and  $\delta_{\mu}$  from the preceding section, are both recoverable, since they such the above definition. Consider the schedule  $\delta_{\mu}^{-1}$  given below, which is the same as schedule  $\delta_{\mu}$  except that two communications have been added to  $\delta_{\mu}$ .

 $S_{i}^{i}(x_{1}(\Sigma), x_{1}(\Sigma), x_{2}(\Sigma), x_{2}(\Sigma), x_{2}(\Sigma), x_{3}(\Sigma), x_{4}(\Sigma), x_{4}(\Sigma), x_{5}(\Sigma), x_{5$ 

5.7 is recoverable, even though it soffers from the lost update problem. However, consider the two (partial) schedules S, and S<sub>1</sub> that follows

$$\begin{split} & S : q(\mathbf{X}); w_1(\mathbf{X}); v_2(\mathbf{X}); q(\mathbf{Y}); w_1(\mathbf{X}); c_2, w_3, \\ & S_2 : q(\mathbf{X}); w_1(\mathbf{X}); v_2(\mathbf{X}); v_1(\mathbf{Y}); w_2(\mathbf{X}); w_1(\mathbf{Y}); z_1, c_2; \\ & S_2 : q(\mathbf{X}); w_1(\mathbf{X}); v_1(\mathbf{X}); v_1(\mathbf{Y}); w_2(\mathbf{X}); w_1(\mathbf{Y}); z_1, c_2; \end{split}$$

Spin not recoverable, because  $T_j$  reads item X from  $T_j$ , and then  $T_j$  commits before  $T_j$  commits. If T inderts after the  $c_j$  operation in S<sub>j</sub>, then the value of X that  $T_j$  read is no longer valid and  $T_j$  must be abouted after it had been committed deading to a schedule that is not recoverable. For the schedule to be recoverable, the  $c_j$  operation in S<sub>j</sub> must be proposed until after  $T_j$  commits, as shown in S<sub>j</sub> if  $T_j$  aborts instead of committing, then  $T_j$  decide abort as shown in S<sub>j</sub> if  $T_j$  aborts instead of committing, then  $T_j$  decide abort as shown in S<sub>j</sub> because the value of X it read is no longer valid.

In a recoverable schedule, no committed transaction ever needs to be rolled lock However, it is possible for a phenomenon known as **cascading rollback** (or **cascading abort**) to occur, where an encounceted transaction has to be rolled back because it read an nemfrom a transaction that follod. This is illustrated in schedule S<sub>1</sub>, where transaction T<sub>1</sub> his to be pulled back because it read irom X from T<sub>2</sub> and T<sub>3</sub> then aborted.

Because case ding rollback can be quite time-constraing—time primerous transactions can be rolled back (see Chapter 19). It is important to characterize the schedules where this phenomenon is guitanteed nearty occur. A schedule is said to be **caseadeless**, or to avoid **caseading rollback**, st every transportion in the schedule reads only items that were written by committed transactions. In this case, all items read will not be discarded, so to cascading tollback will occur. To satisfy this criterion, the  $i_2(X)$  commond in schedules  $\delta_2$  and  $\delta_1$  must be postgoned until after  $T_1$  has committed (or aborted), thus delaying  $T_2$  but ensuing no cascading tollback of  $T_1$  aborts.

Initially, there is a third, more restrictive ryte of schedule, called a strict schedule, or which transactions can be the read normalic an item X until the last transaction that write X has committed (or aborted). Strict schedules simplify the recovery process in a strict schedule, the process of undoing a write\_ntentX) operation of an absited transaction is simply to restore the before image ("Al\_value or PE(M) of data item X. This simple procedure always works correctly for strict schedules, but it may not work for recoverable or usefulcies schedules, but example, consider schedule S<sub>1</sub>.

 $S_{C}(w_1(X, 5), u_2(X, 8); v_1)$ 

Suppose that the value of X was originally 9, which is the before image stoted in the system log along with the n (X, D) operation (W, T) abeets as it  $S_{2}$ , the receivery proceduc that re-tores the value of X (X) operation (W, T) abeets as it  $S_{2}$ , the receivery proceduc that re-tores the value of X (0.6, even though it has already been changed to 8 by then on tors  $W_{1}$  thus leading to potentially incorrect results. Although schedule  $S_{1}$  is cascadeless, it is not a street schedule, since 2 permits  $T_{2}$  to write item X even though the transaction  $T_{1}$  that last wrote X had out we committed (in aborted). A strip is schedule does not have this problem.

We have now characterized schedules according to the following terms: (i) recoverability (2) avoid new of cascading follback, and (3) strictness. We have thus size that thus properties of schedules are successively more stimpent conditions. This condition (2) implies condition (i) k and condition (3) implies both (2) and (1). This of strict schedules are engadeless, and all cascadeless schedules are recoverable.

## 17.5 CHARACTERIZING SCHEDULES BASED ON SERIALIZABILITY

In the previous section, we characterized schedules based on their recover bility preparties. We now characterize the types of schedules that are considered correct when concortent transactions are escenting. Suppose that two users intervaling cost with evolve  $k_{12}$  subnit to the PROS transactions  $T_{12}$  and  $T_{22}$  of Figure 17.2 st opportomately the same time If no interleaving of operations is permuted, there are only two pre-like outcomes

- Execute all the operations of transaction T can sequence) followed by all the operations of transaction T<sub>2</sub> (in sequence).
- Execute all the operations of transaction T<sub>2</sub> (in sequence) followed by all the operations of transaction T<sub>1</sub> (in sequence).

These alternatives are shown in Figure 17.5a and b, respectively. It interleaving it egenations is allowed, there will be namy possible orders in which the system on execute the individual operations of the transactions. Every possible schedules are shown


**FIGURE 17.5** Examples of serial and nonserial schedules involving transactions  $T_0$  and  $T_0$  (a) Serial schedule A:  $T_1$  followed by  $T_{P_1}$  (b) Serial schedule B:  $T_1$  followed by  $T_{P_2}$  (c) Two nonserial schedules C and D with interleaving of operations.

in Figure 17.5c. The concept of serializability of schedules is used to identify which whedules are correct when measoring executions have interleaving of their operations in the schedules. This section defines serializability and discusses how it may be assumption product.

# 17.5.1 Serial, Nonserial, and Conflict-Serializable Schedules

Schedules A and B in Figure 17.5 a and b are colled verify because the operations of each transaction are executed consecutively, without any unceleaved operations from the other repractions for a secial schedule, entire transactions are performed in secial order:  $T_1$  and then  $T_2$  in Figure 17.5a, and  $T_2$  and then  $T_1$  in Figure 17.5b. Schedules C and D

in Figure 17.5c are called nonserfal because each sequence interfeases operations from the two transactions.

Formally, a schedule S is serial if, for every transaction 2 participating in the schouldle, all the operations of T are executed consecutively in the schedule; otherwise, the schedule is called nonserial. Hence, in a serial schedule, only one transaction at a time is acreed whe compare for abard) of the active transaction initiates execution of the next transaction. Ne interfeating occurs in a senal schedule. One reasonable assumption we can make, if we consider the transactions to be independent, is that every senal schedule is considered correct. We can assume this because every transaction is assumed to be correct if executed on its own (according to the consistency preservation property of Section 17.3) Hence, it does not matter which transaction is executed first. As long is every transaction is executed from beginning to end without any interference from the operations of other transactions, we get a correct end result on the database. The problem with serial schololes is that they limit concorrency or interleaving of operations in a serial schedule, the transiention waits for the Weleperation to complete, we conner switch the CEU processor to another starsaction, they wasting viduable CEU processing time in addition, if some transaction T is quite long, the other transactions must war for T in complete all its operations before commencing. Hence, servel schedules are generally considered unacceptable in practice

To influstrate out discussion, consider the schedules in Freure 17.5, and assume that the initial values of database items are X = 90 and Y = 90 and that N = 5 and M = 2. After executing transactions  $T_1$  and  $T_2$  we would expect the database values to be  $\lambda = 59$  and Y = 93, according to the meaning of the transactions. Sure enough, executing either of the social schedules A or B proce the correct results. Now consider the non-serial schedules C and D. Schedule C (which is the same as Figure 17.5a) gives the results X = 92 and Y = 93, in which the X value is erron out, whereas schedule D gives the correct results

Schedulo C gives an erromeous result because of the lost opdate problem discussed in Section 17.1.3; transaction  $T_1$  reads the volue of X before it is changed by transaction  $T_1$ , so only due effect of  $T_1$  on X is reflected in the database. The effect of  $T_2$  on X is be, overwritten by  $T_2$ , leading to the incorrect result for item X. However, some non-send schedules give the correct expected result, such as schedule D. We would like to determine which of the non-senal schedules always give a currect result and which may give erromeous results. Dar concept used to obsole terize schedules in this manner is that of senalizability of a schedule.

A selectate 5 of n transactions as serializable of n is canonical to some serial schedules of the same of nansactions. We will define the concept of equivalence of schedules shoth. Nonce that there are all possible serial schedules of nansactions and many more possible nonserial schedules. We can form two disjoint groups of the nonserial schedules, these that are equivalent to one (or more) of the serial schedules, and hence are serialitable and those that are convequivalent to one some some serial schedule and hence are not serialitable and those that are convequivalent to one some serial schedule and hence are not serialitable.

Saying that a nonserial schedule 5 is serializable is equivalent to saying that it is considered schedule. Which is considered consect. The remaining question is: When are two schedules considered "equivalent". They are several ways to define equivalence of schedules. The simplest, but hast satisfation, definition of schedule equivalence involves comparing the checks of the schedules on the

database. Two schedules are called result equivalent if they produce the same final state at the database. However, two different schedules may accidentally produce the same final state. For example, in Figure 17.0, schedules  $S_1$  and  $S_2$  will produce the same final database state if they execute over database with an initial value of X = 100; but for other netial values of  $X_1$  the schedules rite not result equivalent. In addition, these two schedules execute different transactions, so they definitely should not be considered equivalent. Hence, result equivalence alone control by used to define equivalence of schedules. The satisfier and most peneral opproach to defining schedule equivalence as not to make any issumption about the types of operations included in the transactions. For owischedules should be applied to that item in both schedules in the same order. Two definitions of equivalence of schedules are generally used; conflict equivalence and indicates to be equivalent, the operations upplied to each data item influence of schedules should be applied to that item in both schedules in the same order. Two definitions of equivalence of schedules are generally used; conflict equivalence and condeministry of equivalence of schedules are generally used; conflict equivalence and condeministry of equivalence of schedules are generally used; conflict equivalence and condeministry of equivalence of schedules are generally used; conflict equivalence and condeministry.

Evolutions are sold to be **conflict equivalent** if the order of any two conflicting operations is the same in both schedules. Recall from Section 17.4.1 that two operations in a schedule are and to conflict if they belong to different transactions, access the same database item, and at least one of the two operations is a write\_item operation. If two conflicting operations are applied in different orders in two schedules, the effect can be different ort, the database of on other transactions in the schedule, and hence the sladdles are not conflict equivalent. For example, if a read and write (genation occur in the order  $x_1(X)$ ,  $u_2(X)$  in schedule 5, and in the teverse order  $u_2(X)$ ,  $u_1(X)$  in schedule 5, and in the teverse order  $u_2(X)$ ,  $u_2(X)$  in schedule 5, and in the teverse order  $u_2(X)$ ,  $u_3(X)$  in schedule 5, and in the teverse order  $u_2(X)$ ,  $u_3(X)$  in schedule 5, and in the teverse order  $u_2(X)$ ,  $u_3(X)$  in schedule 5, and in the teverse order  $u_3(X)$ ,  $u_4(X)$  in schedule 5, and in the teverse order  $u_3(X)$ ,  $u_4(X)$  in  $u_3(X)$ ,  $u_3(X)$  in  $S_1$ , and in the teverse order  $u_3(X)$ ,  $u_4(X)$  in  $u_3(X)$ ,  $u_4(X)$  in  $u_4(X)$ ,  $u_5(X)$  in  $u_5$ , and in the teverse order  $u_5(X)$ ,  $u_5(X)$ ,  $u_4(X)$  in  $u_5(X)$ ,  $u_5(X)$    $u_5(X)$  in  $u_5(X)$ ,  $u_5(X)$ ,  $u_5(X)$  in  $u_5(X)$ ,  $u_5(X)$ ,  $u_5(X)$  in  $u_5(X)$ ,  $u_5(X)$ ,  $u_5(X)$  in  $u_5(X)$ ,  $u_5(X)$ ,  $u_5(X)$ ,  $u_5(X)$ ,  $u_5(X)$ ,  $u_5$ 

Using the notion of conflict equivalence, we define a schedule S to be **conflict** serializable<sup>11</sup> if it is (conflict) equivalent in some serial schedule S'. In such a case, we can model the monorflicting operations in S until we from the equivalent serial schedule S'. According to this definition, schedule D of Figure 17.5c is equivalent to the serial

| _ <sup>s</sup> , . | $s_{\underline{v}}$ |
|--------------------|---------------------|
| read nem (X).      | read item(X).       |
| X-X-10.            | XX-1.1              |
| ania_tern(X).      | wite_0em(X);.       |

**FIGURE 17.6** Two schedules that are result equivalent for the initial value of  $\lambda = 100$  hot are not result equivalent to general.

<sup>11.</sup> We will use serializable to more conflict scrializable. Another definition of serializable used to practice (see Section 17.20) is no have repeatible reads, no dury reads, and ne phannin records (see Section 18.7.1 for a docussion on phannoms).

schedule A of Figure 17.5a. In both schedules, the **read** [n tentX] of  $T_1$  reads the value of X written by  $T_1$ , while the other **read** [n tentx] on **read** the database values from the initial database state. In addition,  $T_1$  is the last transaction to write Y, and  $T_2$  is the last transaction to write X in both schedules. Because A is a settal schedule and schedule 15 is equivalent to A. D is a seculizable schedule. Notice that the operations  $\tau_1(Y)$  and  $w_2(Y)$  of  $w_3(Y)$  since they access different data news. Hence, we can move  $r_1(Y)$ ,  $w_3(Y)$  before  $r_2(X)$ ,  $w_2(X)$ , leading to the equivalent secul schedule  $T_1$ .

Schedule C of Figure 17.5c is not equivalent to either of the two possible senal schedules  $\Delta$  and B, and hence is tax senaligable. Thying to reorder the operations of schedule C to tend an equivalent senal schedule table because  $\gamma_2(X)$  and  $u_2(X)$  conduct, which means that we cannot move  $r_1(X)$  down to get the equivalent senal schedule 1. T<sub>1</sub>. Similarly, because  $r_1(X)$  and  $u_2(X)$  conduct, we cannot move  $u_1(X)$  down to get the equivalent senal schedule T<sub>1</sub>.

Another, more complex detunition of convolunce- collect convergencies which leads to the concept of more structurability—is discussed in Section 17.5.4.

#### 17.5.2 Testing for Conflict Serializability of a Schedule

There is a simple algorithm for determining the condict serialitability of a schedule. Mosconcurrency control methods do not actually rest for senalizability. If their protocols or rules, are developed that gracantee that a schedule will be serializable. We discuss the algorithm for te-tong conflict serializability of schedules here to gain a better understording of these constituency control protocols, which are discussed in Chapter 18

Algorithm 17.1 can be used to test a schedule for conflict condical dity. The algorithm locks at only the read, item and **witte\_item** operations in a schedule to construct a precedence graph (or serialization graph), which is a directed graph  $G = \{N, E\}$  that consists of a set of needes  $N = \{T_i, T_j, \ldots, T_i\}$  and a set of directed edges E = k.  $e_1, \ldots, e_n\}$ . There is one node in the graph for each transaction T in the schedule. Each edge  $e_i$  in the graph is of the form  $(T_i \rightarrow T_i \setminus 1 \sim j \geq n, 1 \leq k \leq n)$  where  $T_i$  is the ending node of  $e_i$ . Such an edge is character if the operations in T appears in the schedule left source conflicting operation in  $T_i$ .

Algorithm 17.1: Testing conflict sensitial fity of a schedule S.

- For each manaction T, participating in schedule S, create a rode labeled T in the precedence graph.
- For each case in S where T<sub>i</sub> executes a read iter(X) after T<sub>i</sub> executes a write, iter(X), ensure in edge (T<sub>i</sub> → T<sub>i</sub>) in the tracedence graph.
- For each case in S where T<sub>i</sub> executes a write\_iten(X) ofter T<sub>i</sub> executes a read\_ iten(X), create an edge (T<sub>i</sub> → T<sub>i</sub>) in the precedence graph.
- Let each cow in S where f, executes a write iten(N) after l, execute- a write, iten(N), stellar ended(T → T<sub>i</sub>) in the precedence graph
- 5. The schedule S is sendirable if and only if the precedence graph has no cycles.



FIGURE 17.7 Constructing the precedence graphs for schedules A to D from Eigure 17.5 to test for conflict senalizability. (a) Precedence graph for senal schedule A ibi Precedence graph for senal schedule 8.10. Precedence graph for schedule C trot senalizable). (d) Precedence graph for schedule D (senalizable, equivalent to schedule A).

The precedence graph is constructed as described in Algorithm 17.1. If there is a cycle in the precedence graph, schedule S is not (conflict) schedule; if there is no cycle, it is second able. A cycle it is directed graph is a sequence of edges  $C = ((f_i \rightarrow f_i), (T_k \rightarrow (f_i \rightarrow (T_i)))$  with the property that the starting node of each edge—except the fast edge—is the same as the ending node of the previous edge, and the starting rade of the test edge is the same as the ending node of the last edge (the sequence starts and ends at the same node).

In the precedence graph, an edge from  $T_i$  to  $T_i$  means that ministerion  $T_i$  must contribute transformin  $T_i$  in any seriel schedule that is equivalent to  $S_i$  because two conflicting optations appear in the schedule in that order. If there is no cyclic in the precedence graph, we can create an equivalent serial schedule S' that is equivalent to  $S_i$  by ordering the transactions that participate in S as follows. Whenever, an edge exists in the precedence graph from  $T_i$  to  $T_0$   $T_i$  must appear thefore T in the equivalent serial schedule S' before that equivalent serial schedule S' is follows. Whenever, an edge exists in the precedence graph from  $T_i$  to  $T_0$   $T_i$  must appear before T in the equivalent serial schedule S'. Notice that the edges  $(T_i \rightarrow T_i)$  in a precedence graph can optionally be labeled by the name(S' of the data, tem(S' that led to creating the edge. Figure 17.7 shows such airely on the edges.

In general, several serial schedules can be explicatent to Sut the precedence graph for Shas no cycle. However, if the precedence graph has a cycle, it is ease to show that we cannot create any equivalent serial schedule, so S is not wriatrable. The precedence graphs mared for schedules A to D, respectively, of Figure 17.5 appear in Figure 17.5 at oil. The

<sup>.2.</sup> This process of ordering the nodes of an acodic graph is known as top-logical sorong.

graph for schedule C has a cycle, so it is not serial table. The graph for schedule D has no cycle, so it is schedule C has a cycle, so it is schedule  $\alpha$   $T_1$  followed by  $T_2$ . The graphs for schedule-A and B have no cycles, as expected, because the schedules are sensitivable, and hence schedules.

Another example, in which three transactions participate, is shown in Henre 17.8 Figure 17.85 shows the read free and write itter operations in each transaction. Two-schedules E and F for these transactions are shown in Figure 17.85 and c, respectively, and the procedence graphs for schedules E and F are shown in parts d and e. Schedule E is not sensitizable, because the corresponding precidence graph has cycles. Schedule F is sensitizable, and the senial a bedule equivalent to F is shown in Figure 17.86. Although only one represelent senial a bedule exists for F, in general there may be none these or operations for a sensitizable. From the procedence graph between the correspondence of the senial shown in Figure 17.86. Although only one represelent senial is bedule exists for F, in general, there may be none these or operations are and whether the senial abedule schedule. Figure 17.86 shows a procedence graph teppeopring a schedule that has two equivalent senial schedules.

#### 17.5.3 Uses of Serializability

As we discussed earlier, writig that a schedule S is (conflict) senalizable, what is S is (conflict) equivalent to a social schedule tils randomer to saying that S is correct. Being serials able is distinct from being social, however, A serial schedule represents inefficient processor because no interfeature of operations from different transactions is permitted. This can lead to low CPC influence while a transaction wans for disk (6), or for another transaction is terminate, thus dowing down processing considerable. A senalitable schedule gives the benefits of concorrent execution without group up any correctness. In practice, it is quite difficult to schedule, the interfeatures of operations without group up any correctness. In practice, this quite difficult interfeatures, which are usually executed as processes by the operations store with a difficult of all processes. Factors such as system load, time of transaction submission, and priorities of grocesses can trained and a system load, time of transaction submission, and priorities of grocesses can train the ordering of operations of a schedule will be interfeature difficult to determine by the operating system schedule. Hence, it is difficult to determine how the inperations of a schedule will be interfeatured before and the ordering of operations and priorities of grocesses are transactions of a schedule will be interfeatured before and the ordering of the operation browthe inperations of a schedule will be interfeatured beforehand to ensure sendimination.

If managements are executed at will and three the coulding schedule is tested for semahrability, we must cancel the effect of the schedule if it musts out not to be semahrable. This is a sensus problem that makes this opproach impraction. Hence, the approach taken in most practical systems is to determine methods that ensure semahrability, without having to test the schedules themselves. The approach taken in most commercial DBMS is to design protocols (sets of rules) that—of followed by accountervidual transaction, or of enforced by a DBMS concurrency control subsystem—will ensure senalizability of all schedules in which the transaction.

Another problem appears here: When transactions are submitted continuously to mesystem, it is difficult to determine when a schedule begins and when it ends. Senalitability theory can be adapted to deal with this problem by considering only the committed projection of a schedule S. Recall from Section 17.4.1 that the committed projection C(z)of a schedule S includes only the operations in S that belong to committed transactions We can theoretically define a schedule S to ke senalitable if its continuited projector C(S) is equivalent to some senal schedule, since only committed transactions are guaranteed by the toxis.



| <b>1</b> 01 | Inansaction 7 <sub>1</sub> | tansaction 7,                                    | vaneaction 7 <sub>3</sub> |
|-------------|----------------------------|--------------------------------------------------|---------------------------|
|             |                            | read_tem (Z):<br>read_tem (Y):<br>wrte_item (Y). | <br>!<br>                 |
|             |                            |                                                  | i read dem (Y).           |
|             |                            |                                                  | reed_riem (Z),            |
| I           |                            |                                                  |                           |
| 7000        | NEAD_NEIM (A1):            |                                                  | i                         |
| ing i       | WATELENIN (A.).            |                                                  | WITH NOT (7)              |
| -           |                            |                                                  |                           |
| •           |                            | (X) man_cean                                     |                           |
|             | nast item (V∕k             |                                                  | !                         |
|             | writti_down (Y);           |                                                  | 1                         |
|             |                            | write_herry (X).                                 | 1                         |
|             |                            |                                                  |                           |





Schedule F

**FIGURE 17.8** Another example of socializability testing, (a) The seas and write operations of three transactions  $T_0$ ,  $T_2$ , and  $T_3$ , the Schedule E, is the Schedule E.





In Chapter 18, we discuss a number of different concurrency control protocels that generated senalizability. The most common technique, called two-place locking, is based on locking data items to provent concurrent transactions from interfering with oranother, and entorcing an idditional condition that generated senalizability. This is used in the importity of commercial EDMss. Other protocols have been proposed;<sup>10</sup> these

15 These other previously have not been used much in practice within most systems use some constant of the worphase locking protocol.

include anestandy ordering, where each transaction is assigned a unique transition and the protocol ensures that any conflicting operations are executed in the order of the transaction time-transpip additional protocols, which are based on maintaining multiple versions of data items: and optanistic (also called *conflictation* or reliabilities) protocols, where check for presible vertalizability violations after the transactions terminate but before they are permitted to commut.

# 17.5.4 View Equivalence and View Serializability

In Section 17.5 if, we defined the concepts of conflict equivalence of schedules and conthe serializability. Another less restrictive definition of equivalence of schedules is called two equivalence. This leads to another definition of serializability called new *veralizability*. Too schedules S and S<sup>\*</sup> are said to be **view equivalent** of the following three conditions held:

- The same set of remsactions participates in S and S', and S and S' include the same operations of those transactions.
- For any operation r<sub>i</sub>(X) of T<sub>i</sub> in S<sub>i</sub> if the value of X read by the operation has been written be an operation u<sub>i</sub>(X) of T<sub>i</sub> (or if it is the original value of X before the schedule started), the same condition must hold for the value of X read by operation q(X) of T<sub>i</sub> in S<sup>1</sup>
- 5. If the operation u<sub>k</sub>(Y) of T<sub>k</sub> is the last operation to write item Y in S, then u<sub>k</sub>(Y) of T<sub>k</sub> must also be the last operation to write item Y in S'.

The idea behind mew equivalence is that, as long as each read operation of a transaction reads the result of the same write operation in both schedules, the write operations of each transaction must produce the same results. The read operations are hence said to see the same clear in both schedules. Combiner 3 ensures that the final write operation on each data item is the same in both schedules, so the databast state should be the same at the end of both schedules. A schedule S is said to be view serializable if it is use equivalent to a serial schedule.

The definitions of conflict serializability and dow serializability are similar if a condition known is the constrained write assumption holds on all transactions in the schedule. This condition states that any write operation  $i_{i}(X)$  in  $T_{i}$  is preceded by a  $i_{i}(X)$  in T and that the value written by  $u_{i}(X)$  in  $T_{i}$  depends only on the value of X read by  $i_{i}(X)$ . This assumes that computation of the new value of X is a function i(X) based on the old value of X read from the database. However, the definition is view semiclability is less restrictive than that of conflict semiclability under the unconstrained write assumption, where the value written by an operation w(X) in  $T_{i}$  can be independent of its oblic value from the database. This is called a blind write, and it is illustrated by the following sthedule  $\delta_{i}$  of three transactions  $T_{1}$ :  $\tau_{i}(X)$ ;  $u_{i}(X)$ ;  $T_{2}$ :  $w_{i}(X)$ ; and  $T \in w(X)$ :

 $S_{\mu}(\mathbf{x}_{1}(\mathbf{X}); \mathbf{u}_{2}(\mathbf{X}); \mathbf{y}_{1}(\mathbf{X}); \mathbf{u}_{2}(\mathbf{X}); \mathbf{v}_{1}(\mathbf{x} < \alpha))$ 

In  $S_i$  the operations  $u_i(X)$  and  $u_i(X)$  are blind writes, since  $T_i$  and  $T_i$  do not read the volue of  $X_i$ . The schedule  $S_i$  is view sorial coblections are it is view equivalent to the senal

schedule  $T_1$ ,  $T_2$ ,  $T_3$ . However,  $S_2$  is non-conflict schedule, since it is non-conflict equivalent to any serial schedule. It has here shown that any conflict-schedule schedule is also new schedule. It has here shown that any conflict-schedule schedule is also new schedule but not vice versal as illustrated by the precising example. There is an algorithm to real whether a schedule  $\lambda$  is view sensitiable or ref. However, the problem of jesting for view sensitiability has been shown to be NP hard, meaning that hading an efficient pulynomial rune algorithm for this problem is highly anjhely.

### 17.5.5 Other Types of Equivalence of Schedules

Senalitability of schedules is sometimes considered to be too restrictive as a condition for ensuring the correctness of concurrent esecutions. Some applications can produce schedules that are correct by variations conditions less structeent than either, conflict senalitabiltry or view senalitability. An example is the type of transactions known as debit-credit transactions—tot example, those that apply deposits and withdrawals to a data trem wave value is the current bilance of a bank account. The semantics of debit-credit transactions where the value of a bank account. The semantics of debit-credit operations is that they update the value of a data area. X by either subtracting from or adding to the value of the data item. Pecaese addition and subtractices operations are commutative that is, they can be applied in any order—it is possible to produce correct schedules that are not smalleable. For example, consider the following two transactions, each of which may be used to transfer an amount of more gloctwern two bank accounts:

 $T_{i}: \pi_{i}(X); X := X = IQ; w_{i}(X); v_{i}(Y); Y := Y = IQ; w_{i}(Y);$ 

 $T_{S}(\tau_{S}(Y); Y) = Y - 20; w_{1}(Y); \tau_{S}(X); X := X + 20; w_{2}(X);$ 

Consider the following nonsertializable schedule S<sub>a</sub> for the two transactions

 $S_{0}(r_{1}(X); u_{1}(X); r_{2}(Y); u_{2}(Y); r_{1}(Y); u_{1}(Y); u_{2}(X); u_{2}(X);$ 

With the additional knowledge, or **semantics**, that the operations between each r(i) and w(i) are commutative, we know that the order of executing the sequences consisting of (read-update, write) is not important as long as each (read-update, write) is not important as long as each (read-update, write) sequence by a porticular transaction  $T_i$  on a particular transition in transaction  $T_i$  on a particular transition in the context particular transaction  $T_i$  on a particular transition operations. Hence, the schedule  $S_i$  is considered to the correct even though it is not senalitable. Researchers have been working on extending concurrences control theory to deal with cases where senalitability is considered to be not restrictive as a condition for correctness of schedules.

# 17.6 TRANSACTION SUPPORT IN SQL

The definition of an SQL-transaction is similar to our already defined concept of a transaction. That is, it is a logical and of work and is guaranteed to be atomic. A single SQL statement is fiving considered to be atomic—either, it completes execution without error or it fails and lowes the database unchanged. With SQL, there is no exploit Begin\_Transaction statement. Transaction initiation - done implicitly when particular SQL statements are encountered. However, every transaction must have an explicit end statement, which is either a covert of a particular SQL statement, which is either a covert of a particular SQL transaction has certain characteristics astrobused to it. These characteristics are specified by a stransaction statement in SQL. The characteristics are the access model, the dognosic state state state state level.

The access mode can be specified as 8000 primor step astrt. The definit is 8000 astrt, index the solation level of 8000 accessive is specified (see below), in which case 8000 bevits assumed. A mode of 8000 astrt allows update, insent, delete and create commands to be executed. A mode of \$000 astrt, as the runne in phose is simply for data retrieval.

The **diagnustic** area size option, **D14200710**, **S127** (1, specifies in unreger value 1), indicating the number of conditions that can be held simultaneously in the diagnostic area. These conditions supply feedback information ferrors or exception-) to the user or program on the most recently executed SQL statement.

The isolation level option is specified using the spacetry (astrony constants) as a processing exactly constant option, where the value for scincilations can be as proceeding, and constrain operation, and constants exactly as near the transition level is specified, it is based endowing exclusions that couse dure read, unrepeatable read, and phontons, band in structure identical to the way serializability was defined earlier in Section 17.5. If a transaction executes at a lower isolation level than structures, then one of the following the spectrum of the term structure of the term of the term of the section 17.5. If a transaction executes at a lower isolation level than structures, then one of the following three violations may occut.

- 1. Dirty read: A manuaction T<sub>1</sub> may read the opdate of a manuaction T<sub>2</sub>, which has not ver committed. If T<sub>2</sub> fails and is aborted, then T<sub>1</sub> would have read a value that does not exist and is incorrect.
- *I*. Nonrepeatable read: A transaction  $T_1$  may read a given value from a table. It mother transaction  $T_1$  later updates that value and  $T_1$  reads that value seam, T will see a different value.
- 3. Phantoms: A transaction T<sub>1</sub> may read a set of most from a table, perhaps based on some condition specified in the Stationer-clause. New suppose that a transaction T<sub>1</sub> miseries a new row that also softlikes the offer-clause condition used in T<sub>1</sub>, into the table used by T<sub>1</sub>. If T<sub>1</sub> is repeated, then T<sub>1</sub> will see a phantom, a now that previously did not exist.

Table 17.4 summarizes the possible violations for the different isolation levely. An entry of "yes" indicates that a violation is possible and an entry of "no.1 indicates that it is not possible.

<sup>14</sup> These are similar to the volume leaves does see brook on the end of Section 17.3.

The dury cool and unrepeatable read problems over discussed in Sceneri 17.1.3 Thannoms are discussed in Section 19.0.1.

| The state of the s |                   |                       |         |  |  |
|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------|-----------------------|---------|--|--|
| Levels as Defined in SQL                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                   |                       |         |  |  |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | Type of Violation |                       |         |  |  |
| isolaton<br>level                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | Úkity<br>read     | Nonreposiabla<br>read | Phantom |  |  |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | _                 |                       |         |  |  |
| READ UNCOMMITTED                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | <b>785</b>        | yes -                 | Yes     |  |  |
| READ COMMITTED                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | 40                | yee.                  | yes     |  |  |
| REPEATABLE READ                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 110               | no                    | yes     |  |  |
| SERIALIZABLE                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | PD                | r0                    | no      |  |  |

Provide the Court of them. Research and the data case

A sample soft transaction meht look like the following:

```
EXEC SDL WHENEVER SQLERROR GOTO UNDO:
EXEC SOL SET TRANSACTION
 READ WRITE
 DIACNOSTIC SIZE S
 ISOLATION LEVEL SERIALIZABLE:
EXEC SQL INSERT INTO EMPLOYEE (FNAME, LNAME, SSN, ONC. SALARY)
 VALUES ("Rosent": "SMOTH": "991004321": 2, 3500D);
EXEC SQL UPDATE EMPLOYCE
 SET SALARY - SALARY - 1.1 WHERE DNO - 2;
EXEC SOL COMMIT:
GOTO THE END:
UNDO: EXEC SQL ROLLBACK;
THE END:
```

The above transaction consists of first inserting a new rote in the test cast rable and then updatice the salary of all employees who work in department 2. It an error occurs on any of the SQL statements, the errore transferior is toked back. This implies that an updated salary the this transaction bound to restored to its previous value and that the newly inserted now would be removed

As we have seen, 501 mondes a number of transaction sourcesticatores. The DPA or database enserationers can take advantage of these options to ity improving transaction performance by relaxing senalizability if that is acceptable for their applications

# 17.7 SUMMARY

In this chapter we discussed 06865 concepts for transaction pricessing. We introduced the concept of a database transaction and the operations relevant to transaction processing We compared single-user vesterns to multiuser systems and then presented examples of how uncontrolled execution of concurrent transactions in a multiuser system can lead or incorrect results and database values. We also discussed the varian upper of failures that may occur during transaction execution.

We then introduced the typical states that a transaction passes through during execution, and discussed several concepts that are used in recovery and concurrency control methods. The system log keeps track of database accesses and the system oses this information to recover from failures. A transaction either socceeds and teaches its commit point or it fails and has to be rolled back. A committed transaction has its thonges permatently recorded in the database. We presented an overview of the destrable properties of transactions—namely, atomicity, consistency preservation, isolation, and durability—which are often referred to as the AUD properties.

We then defined a schedule for history) as an execution sequence of the operations at several arabisactions with possible arterleaving. We characterized schedules an series of their recoverability. Recoverable schedules ensure that, once a transaction commiss in never needs to be undone. Classideless schedules add an additional condition to ensure that no aborred transaction requires the containing abort of other transactions. Strict schedules provide an over stronger condition that allows a simple recovery schenic consisting of restoring the old values of items that have been changed by an aborted statisaction.

We then defined equivalence of schedules and say that a senalizable schedule is equivalent to some senal schedule. We defined the concepts of conflict equivalence and vice equivalence, which led to definitions for conflict senalizability and view we alreadulity. A schedule schedule is considered correct. We then presented algorithms for testing the (conflict) senalizable schedule is considered correct. We then presented algorithms tor testing the (conflict) senalizable view of a schedule. We discussed active resting for senalizability is suppractical in a real systems although it can be used to define and verify concorrency control protocols, and we briefly mentioned less restrictive definitions of schedule equivalence. Finally, we give a brief overview of how transaction concepts are used in practice within SQL.

We will discuss concorrency control protocols in Chapter 18, and recovery portocols in Chapter 19.

## Review Questions

- 17.1 What is meaningly the concurrent execution of database transactions in a mulnuser system? Discuss why concurrency control is needed, and give informal examples.
- 17.2 Discuss the different types of follows. What is meant by catostrophic follow?
- 17.3 Discuss the actions taken by the read (item and write atem operations on a database)
- 17.4. Draw a state diagram, and discuss the typical states that a transaction goes alwangly during execution.
- 17.5. What is the system log used for? What are the typical kinds of records in a system log: What are transaction coronar points, and why are they important?
- Discuss the oronicity, durability, isolation, and consistency preservation properties of a database transaction.
- 12.7. What is a schedule (history)? Define the concepts of recoverable, coscideless, and struct schedules, and compare them in terms of their recoverability.

- (1) S. Oberes the different measures of transaction equivalence. What is the difference between conflict equivalence and more equivalence?
- 17.9 What is a serial schedule? What is a serial pable schedule? Why is a serial schedule considered correct? Why is a serial pable schedule considered correct?
- 17.10 What is the difference between the constrained write and the amounstrained write assumptions' Which is more traffistic?
- 17.33 Discuss how senalitability is used to enforce concorrency control in a database system. Why is senalitability sometimes considered too restrictive as a measure of correctness for schedules?
- 17.12 Describe the true levels of solation in SQL
- 17.13 Demo the violations caused by each of the following: duty read, nonrepeatable read, and phantonis.

#### Exercises

17-14. Change transaction T J in Figure 17-26 to read

read item(%): %:= X+%; if X > 90 then exit else write\_item(%):

Discuss the final result of the different schedules in Figure 17.3(a) and (b), where M = 2 and N = 2, with respect to the following questions. Does adding the above condition change the final outcome? Does the outcome obey the implied consistency role (that the capacity of X is 90)?

- 17.15. Repeat Exercise 17.14, adding a check in T<sub>1</sub> so that Y does for exceed 90.
- 17.16. Add the operation commutat the end of each of the transactions T<sub>1</sub> and T<sub>2</sub> from Figure 17.2; then list all possible schedules for the moduled mansactions. Determine which of the schedules are recoverable, which are coscadeless, and which are strict.
- 17.17. List all possible schedules for transactions T<sub>1</sub> and T<sub>2</sub> from Figure 17.2, and determine which are conflict serialitable (correct) and which are not.
- 17.15. How many send schedules exist for the three transactions in Figure 17.868? What are there? What is the total number of possible schedules?
- 17.19. Write a program to create all possible schedules for the three transactions in Figure 17.8(a), and to determine wheels of those schedules are conflict semilitable and which are not. For each conflict semilitable schedule, your program should print the schedule and first all equivalent senial schedules.
- 17.20 Why is are explicit transaction end statement needed in SQL but not an explicit begin statement?
- 17.21 Describe situations where each of the different isolation levels would be useful to transaction grocessing.
- 17.22 Which of the following schedules is (conflict) senalitable? For each senalitable schedules determine the equivalent senalitables.

- $\alpha_i = r_1(X); r_2(X); a_2(X); r_2(X); a_3(X);$
- $\mathbf{b}_{i} = \mathbf{r}_{i}(\mathbf{X}) (\mathbf{x}_{i}(\mathbf{X}); \mathbf{a}_{i}(\mathbf{X}); \mathbf{a}_{i}(\mathbf{X}); \mathbf{r}_{i}(\mathbf{X});$
- $\alpha_i = r_i(X)_{i \in \mathbb{Z}_2^d}(X)_i \ w_j(X)_{i \in \mathbb{Z}_2^d}(X)_i \ w_j(X)_i$
- $\mathrm{d}_{t} = \tau_1(X); \ \tau_2(X); \ u_1(X); \ u_2(X); \ u_1(X);$
- 17.23. Consider the three transactions T<sub>1</sub>, T<sub>2</sub>, and T<sub>3</sub>, and the schedulos N<sub>1</sub> and S<sub>2</sub> given below. Draw the socializability (precedence) graphs for S<sub>1</sub> and S<sub>2</sub>, and state whether each schedule is seculicable or not. If a schedule is secializable, write down the equivalent secial schedule(s).

$$\begin{split} T_1, v_1_1(X), v_1_2(Z), w_1_2(X); \\ T_2, v_2_2(Z), v_1_1(Y), w_2_2(C), w_2_2(Y); \\ T_3, v_3_2(X), v_1_2(Y), w_1_2(Y), \\ S_1, v_1_1(X), v_2_2(C), v_1_2(Z), v_1_2(Y), v_1_2(Y), w_1_2(Y), v_1_2(Y), v_2_2(Y); w_2_2(Y), \\ S_2, v_1_1(X), v_2_2(Z), v_1_2(X), v_1_2(Z), v_2_2(Y), v_1_2(Y), w_1_2(X), w_2_2(Z), w_2_2(Y), \end{split}$$

17.24 Consider schedules S<sub>3</sub>, S<sub>40</sub> and S<sub>3</sub> below. Determine whether each schedule is strict, cascadeless recoverable, or nonrecoverable. (Netermine the strictest recoverability condition that each schedule satisfies.).

# Selected Bibliography

The concept of transaction is discussed in Grav (1981). Bernstein, Hubblaces, and Gradnon (1967) focus on concurrency control and recovery techniques in both centralized and distributed database systems, or is an excellent reference. Papadimitriou (1986) orders a more theoretical perspective. A large reference book of more that a thorsand pages by Grav and Reuter (1993) offers a more practical perspective of transaction processing can cepts and techniques. Elmagatimal (1992) and Bhatgava (1980) offer collections of te-carch pagets in transaction processing. Transaction support in SQL (a described in 15 to and Darwen (1993). The concepts of senalizability are network ceed in Gravier al. (1975). Year senalizability is defined in Yannakakas (1984). Recoverability of schedules is descused in Hadridaces (1983, 1988).



# Concurrency Control Techniques

In this chapter, we discuss a number of concurrency control techniques that are used to ensure the nonumeriterence or tradition property of concurrently executing transactions. Most of these techniques ensure set illudibility of schedules (see Section 17.5), using pretools (*in it* is sets of role i) that guarantee setulizability. One important set of protocols employs the technique of locking documents to prevent multiple transactions from accessins the items concurrently, a number of locking protocols of described in Section 18.1. Locking protocols are timestamps. A timestamp is a unique identifier for each transaction, princated by the system. Concurrences control protocols that use timestamp cadering to iteate sentalizability are described in Section 18.2. Ite Section 18.3, we docus multiversion concurrency control protocols that use transactions of a data itean. In Section 18.4, we present a protocol based on the concept of validation or certification of a transsation after it executes its operations there are some times called optimistic protocols.

Attothet factor that affects concurrency control is the granularity of the data items that is, what portion of the database a data atom represents. An item can be as small us a angle attribute (field) value or as large as it disk black, or over, a whole lab or the entire catabase. We discuss granularity of items in Section (8.5) In Section (8.6, we discuss construction control issues fractance when indexes or used to process transformed is smally, a Section 18.7 are discuss some additional concurrency control issues. It is sufficient to cover Sections 16.1, 18.5, 18.6, and 18.7, and possibly 18.3.2, if the main emphasis is on articlasing the concurrency control technologies that are used most often in practice. The other technologies are trainly of theoretical interval.

# 18.1 Two-Phase Locking Techniques For Concurrency Control

Some of the main rechniques used to control concurrent execution of transactions are based on the centrept of lacking data items. A lock is a variable associated with a data item that describes the status of the item with respect to possible operations that can be applied to it. Generally, there is one lock for each data item in the database. Locks are used as a means of synchronizing the access by concurrent transactions to the database items. In Section 18.1 There docess the nature and types of locks. There, in Section 18.1 There docess the nature scitalizability of transaction schedules. *Finally, in Section 18.1.5 we doce to graphine as class data data item transaction schedules. Finally, in Section 18.1.5 we doce to graphine as class data data data the use of locks-* namely deallock and starest on—and show how these problems are bandled.

### 18.1.1 Types of Locks and System Lock Tables

Several types of locks are used in containing control. To introduce locking concepts gadinitial we first discuss binary locks, which are snaple but restrictive and so are not used as practice. We then discuss shared/exclusive locks, which provide more general locking capabilities and are used in proceed database locking schemes. In Section 18.3.2, we describe a certify lack and show here a cambe used to improve performance of locking promocile.

Bittary LOCKS. A binary lock can have two states or values: locked and unlocked (or 1 and 0, for simplicity). A distinct lock is associated with each database item X. If the value of the lack in X is 1, item X cannot be accessed by a database operation that requests the mini. If the value of the lack on X is 0, the item can be accessed when requested. We refer to the current value (or store) of the lock issociated with item X as LOCKCXI.

Two operations, lock item and unlick (ten, are used with binary locking, A cransformin respects access to an item X by hist using a lock item(X) operation. If (it is S(X) = 1, the maintenance is locked to wait. If  $I \cup CK(X) = 0$ , it is set to 1 (the transaction locks the (tem)) and the transaction is allowed to access ment X. When the transaction is through using the (tem) it is uses an unlick item(X) operation, which set  $(X \in IX)$  to (unlocks the tren) so that X may be accessed by other transactions. Hence, a binary lock enforces mutual exclusion on the data item: A description of the lock item(X) and anlock item(X) operations is shown in Figure 16.1.

Notice that the Inck\_iten and unlock item operations must be implemented as indivisible opers (known as critical sections in operating systems); that is, no interleaving should be allowed once a likely or onlock operation is started until the operation terminores or the transaction ways. In Figure 1841, the wait command within the lock\_

```
louk_item (X).
```

```
B: ALCCK (X)=0 (* item is unitodaed *)
then LCCK (X)+1 (* took the item *)
été tegen
wat (uniti look (X)=0 and
the look manager wakes up the transaction)
go to 8
and:
unitod_term (X)*
LOCK (X)+ 0, (* unitods the life() *)
```

if any bansactions are waiting from wakeup one of the waiting pansactions,

PIGURE 18.1 Enrol and unlock operations for binary locks

iten(X) operation is usually implemented by putting the transaction on a warring queue or item X utaril X is unlocked and the transaction can be granted access to in. Other inmodentiats that its warr to access X are placed on the same queue. Hence, the pair command is considered to be surside the lock\_item operation.

Notice that it is quite simple to implement a pinary lock: all that is needed is a bitaryvalued variable. ICCK, associated with each data item  $\lambda$  in the database. In its simplest form, each lock can be a record with three fields. Schualitem name, aCCK, locking transaction/s plusa queue for transactions that are waiting to access the item. The system needs to maintain right these records for the items that are currently locked in a lock table, which could be agained as a bash the. Items not in the lock table are considered to be unlocked. The 10MS loss a lock manager subsystem to keep track of and control access to locks.

If the sample binary locking achieve described here is used, every transaction must also the following roles:

- A transaction T must usue the operation lock it en(X) before any read\_item[X] or write item[X] operations are performed on T.
- A transactions T must used the operation an Inck\_item(X) after all cead\_nem(X) and write\_item(X) operations are completed in T.
- A manuaction T will not usue a Tock (remX) operation (for already holds the Tock oreitem X)
- A transaction T will not assue in unfock (tentX) operation infess at already holds the lock on atom X.

These rules can be enforced by the lock manager module of the ODMS. Between the lock iten(X) and unlock iten(X) operations in transaction T, T is such as hold the

1. This rule must be removed at we modify the lock premit X1 operation in Figure 15.1 so that if the tences of remains locked by the mynesing promotion in the lock is primited.

**lock** on item X. At most one transaction can bold the lock on a particular item. Thus no two transactions can access the same item concurrently.

Shared/Exclusive (or Ng ad/Write) Lot ks. The preceding buory locking scheme is too restrictive for database items, because at most one transaction can hold a lock on a given item. We should draw several transactions to necess the same item X if they all access X for reading pressure only. However, if a transaction is to write an item X if they all access X for reading pressure only. However, if a transaction is to write an item X if they all access X for reading pressure only. However, if a transaction is to write an item X, it must have exclusive necess to X. For this tampole, it different type of lock called a multiplemode lock is used. In this schemes -colled shared/exclusive or read/write locks—there are three locking operations: read\_lock(X), writte lock(X), and ur?ack(X). A lock associated with an item X, the K(X), new has three possible states: "read-locked," "write locked," or "unlocked." A read-locked item is also relied share-locked, because other transactions are allowed to read the areas, where is a write-locked, item is called exclusive-locked, lockage a single transaction exclusively holds the lock on the item

One method for implementing the preceding dives operations on a read/state kick is to keep track of the number of maniferions that hold in-bared (read) lock on an item in the lock table. Each record in the lock table will large four fields is dota from name, (a) is init of reads, locking transaction(s1>. Again, to save space, the system nace, mantain lock increases only for locked items in the lock table. The value (state) of the site mantain lock increases only for locked items in the lock table. The value (state) of the site increases read-locked or venter-locked items in the lock table. The value (state) of the site items read-locked items). If  $d > k12^{-1}$  write-locked, the value of locking transaction(s) is a single transaction that holds the exclusive (write) lock on X. If the s(X)-readlin keil, the value of locking transaction(s) is a list of one or more transactions that hold the shared (read) lock on X. The three operations read lock(X), write lock(X), and uniterk(X) are described in Figure 18.27. As before, each of the three operations should be considered indivisible in a interflowing should be allowed once one of the operations is started until either the operation reminates by graning the lock on the transaction is placed on availing queue for the term.

When we use the shared/exclusive locking scheme, the system must enforce the following rules:

- A transaction T must issue the operation read Catk(X) or write Cack(X) before any read in tentX) operation is performed in T.
- 2 A transaction T must issue the operation write lack(X) before any write, item(X) operation is genormed in T.
- A monspectrum T must come the operation unlock(X) ofter all read prem(X) and white interview operations are completed in T<sup>3</sup>.
- 4 A transaction if will not usue o read\_lock(X) operation at a dready holds a real (shared) lock or a write (exclusive) lock on train λ -, his rule may be relaxed as we discuss shortly.

2. These algorithms for non-allowing grading to dimensionly of backs, as described later in this section. The reader can extend the algorithms to allow these additional operations.

3. This may have be relaxed to allow a manuscriptor a surface an areas, there look of again later.

```
ved_lock(X):
 B: FLOCK (20-"undocked")
 then begin LOCK (X) ·· "read-looked"
 no_of_reads(X)= 1
 eod
 Aber #1.OCK/2/=freedHotk97
 then no_of_reads(X') \leftarrow no_of_reads(X') + †
 else begin wat (un#LOCK (X)="unlocked" and
 the lock manager wakes up the variabilition),
 00 to 6
 ant
ante look (X)^{\circ}
 Brd EOCK (X)="integed"
 then LOCK (X)+ Write-locked
 abe tean
 wert jungt LOCK(X)-funitoriked* and
 the look menager wakes up the transaction);
 900 ko Bi
 ent
uncok (X)"
 #LOCK (X)= write-locked:
 then begin LOCK (X)+ "unlocked."
 wakeup one of the werting transactions, if any
 end
 else II _OCK(X)- "read-locked"
 then begin
 no of readed XI+ no_ot readed XI+ 1:
 €no ol readalX)=0
 then begin LOCK (X)="unlocked".
 wakeup one of the watting transactions. If any
 end
 ond:
```

PROURE 18.2 Exching and unlocking operations for two-mode tread-write or shared-exclusive) locks.

- 5. A transaction T will not issue a wrote Tack(X) operation if it already holds a read (Gared) lock of write test losive) lock on item X. This rule may be relaxed, as we discuss shortly.
- A transaction T will not issue an unDock(X) operation unless it already holds a read (shared) lock or a write (exclusive) lock on issue X.

Conversion of Lock5. Sometimes it is desirable to relax conditions 4 and 5 in the preceding list in order to allow lock conversions that is, a transaction that already holds a lock on item X is allowed under certain conditions to convert the lock from one locked.

-fate to another For example, it is possible for a transaction T to issue a read\_lock(X) and then later on to upgrade the lock by oscing a write\_lock(X) operation. If T is the only transaction bolding a read lock on X of the time it issues the write\_lock(X) operation, the lock can be upgraded, otherwise, the transaction must wait b to also possible for a transaction T to issue a write\_lock(X) and then later on to downgrade the lock by using a read\_lock(X) and then later on to downgrade the lock by using a read\_lock(X) operation. When opgrading and downgrading of locks is used, the lock table must methode transaction identifiers in the record structure for each lock (m the locking mansaction(s) field) to store the information which transactions hull locks on the item. The descriptions of the read\_lock(X) and write\_lock(X) operations in Figure 18.2 must be changed uppropriately. We have this as an exercise for the readet.

Using binary locks or read/write books in nanoactions, as described earlier, does not guarantee erraligables of schedules on its own. Figure 18.3 shows an example where the preceding locking rates are followed bin a non-senalitable schedule may result. This is because in Figure 18.3 in the items Y in  $T_1$  and X in  $T_2$  were note lead to carrier. This allows a schedule such as the one shown in Figure 18.5 into occur, which is not a senalitable schedule and better gives incorrect results. To gaurantee senalitability, we must follow an additional proposal concerning the positioning of locking and unlocking operations in every transaction. The best known protocol, two-phase locking is described in the new section.

## 18.1.2 Guaranteeing Serializability by Two-Phase Locking

A transaction is and to follow the two-phase locking protocol if all locking operations (read\_lock, write\_lock) precede the insentileck operation in the transaction.<sup>4</sup> Such v transaction can be divided into two phases an **espanding** or **growing (first) phases** during which new locks on non- can be acquited but note can be released; and a shrinking (acund) **phase**, during which existing lock- can be released but no new locks can be acquired if lack entropisments allowed, then upgrading is locks (from read-locked) to write-locked) must be done during the expanding phase, and dowigrading of locks (from write-locked) read-locked) must be done in the shrinking phase. Hence, a read locks (from write-locked to dowigrades an aheady held write lock on X can appear unly in the shrinking phase.

Transmous **T**- and **T**<sub>2</sub> of Figure 18.3 and not tailow the two-phase backing protocol. This is because the write flock(X) operation follows the unflock(X) operation in **T**<sub>2</sub>, and similarly the write flock(Y) operation follows the unflock(X) operation in **T**<sub>2</sub>. If on enforce two-phase locking, the transactions can be rewritten as **T**<sub>1</sub><sup>+</sup> and **T**<sub>2</sub><sup>+</sup>, as shown in Engine 18.4. Note the schedule shown in Figure 18.3 (c) is not permitted for **T**<sub>1</sub><sup>+</sup> and **T**<sub>2</sub><sup>+</sup> (with their modified order of locking and unflocking operations) under the roles of locking described in Section 18.1.1. This is because **T**<sub>2</sub><sup>+</sup> will usual its write, lock(X) below it

This is anticlared a suballow phase containst protocol for nerovers on distributed doubses (see Chapter 75).



**FIGURE 18.3** Transactions that do not obey two obtase for king, ta. Two transactions  $T_{1}$  and  $T_{2}$  (b) **Results of possible serial schedules of**  $T_{2}$  and  $T_{2}$  (c) A nonservalizable schedules of that user locks.

| 4'                      | l2'               |
|-------------------------|-------------------|
| reed lock (%).          | read lock (XI);   |
| read_rierr ( ).         | read item $(X)$ ; |
| wne lock (X).           | ante kock $(Y)$ ; |
| anlock (9):             | unicak (X)        |
| read item ( <i>X</i> ); | vead dem (Y):     |
| K=X+Y;                  | Y:-X+Y:           |
| write item (X):         | wrie_item (M).    |
| aniode ( X );           | unicek (Y)        |

**FIGURE 18.4** Transactions  $T_1^{(i)}$  and  $T_2^{(i)}$ , which are the same as  $T_1$  and  $T_2$  of Figure (8.5 but which follow the two phase locking protocol. Note that they can produce a deadlock

anlocks item  $\Sigma$  consequently when  $T_{\Sigma}$  issues as read\_lock( $\Sigma$ ), it is forced to wait and  $T_{\Sigma}$  releases the lock by issuing an unlock ( $\lambda$ ) or the schedule.

It can be proved that, if creat transaction in a schedule follows the two-phase locking protocol, the schedule is guotanized to be schedule, obviating the need to test for settahlability of schedule- any more. The locking mechanism, by enforcing two-phase locking rules, also enforces schedulety.

Two-phase locking may built the amount of concurrence that can eccut in schedule. This is because a transaction T may not be able to release in trem X after it's through using it it T must lock an additional item Y later one or conversely. T must lock the additional item Y later one or conversely. T must lock the additional item Y later one or conversely. Timust lock the additional item Y later one or conversely. Timust lock the additional item Y later one or conversely. Timust lock the additional item Y later one or conversely. Timust lock the additional item Y later one or conversely. Timust lock the additional item Y later one or conversely. Timust lock the additional item Y later on or conversely. Timust lock an item to be locked to read or write long between the transaction with the master on seeking to access X may be forced to wat, even through T is done with X; now ciscly, if Y is locked rather than it is incided, mother transaction weeking to access Y is forced to wait even through T is one using Y with This is the price for guaranteeing smallpatishing of all whedules without having to check the schedules themselves.

Basic, Conservative, Strict, and Rigorous Two-Phase Locking. There are a routher effectations of two-phase locking (2ft.). The technique just described is known as have 2ft. A constron known as **conservative** 2ft. (or static 2ft.) requires a transaction of kick all the items in accesses fa fore the measurum fages cronition, by predeclaring its radsecond entropy. Recall from Section 17.1.2 that the readiset of a transaction is the second all terms that the transaction made, and the write-set is the set of all items that it writes fi any of the predeclared items meaded carmot be locked, the transaction does not lock any neury instead, it wars openfull the items are accutable for locking. Conservative 5(1) wa doublock-free transaction of the items are accutable for locking. Conservative 5(1) wa doublock-free transaction we shall see in Section 16.1.3 when we discuss the dealled problem. However, it is difficult to use in procise because of the need to predeclare the read-set and write-set, wheelt is not possible at most structures.

In practice, the new popular variation of 2.1 is strict 201, which gratantees and schedules (see Section 17.4), but this canation, a transaction T does not release any efficience (write) locks until after it commits or aborts. Hence, no other transaction can react or write an item that is written by T unless T b is committed, leading its a strict schedule for recoverability Strict 201 is not deadlock-free. A more restrictive variation of strict 201 is regorous 201, which also guarantees strict is bedules. In this variation of strict 201 is regorous 201, which also guarantees strict is bedules. In this variation of strict 201, is regorous 201, which also guarantees strict is bedules. In this variation a transaction T does not release and of its locks (exclusive or shired) until after it commits or aborts, and so it is easier to implement than strict 201. Notice the difference between conservative and inguisms 201; the former must lock all of strict does not be transaction starts it is in its shrinking phase, whereas the latter does not be been used as an of its index (by committee) to aborting) so the transaction is in its expanding phase until it ends

In many cases, the concurrence control subsystem uselt is responsible to generating the read [lock and write\_] are requests. For example, approach vestering to enforce the strict 21. poince of. Then, whenever transaction T issues a read\_iten(X), the system calls the read\_lock(X) operation on behalf of T. If the state of (Oc K(X)) is write\_locked by some other transaction T<sup>\*</sup>, the system places T on the waring queue for non X;

otherwise, it goants the read.  $\operatorname{Iork}(X)$  request and permits the read\_n tertX1 operation of  $\Gamma$  to evecure. On the other hand, if transaction T respects a matter iten(X), the system calorine north  $\Gamma$  is evecure. On the other hand, if transaction T respects a matter iten(X), the system calorine north  $\Gamma$  is the write  $\operatorname{Iork}(X)$  operation on behalf of T. If the state of  $L \otimes S(X)$  is write  $\operatorname{Iork}(X)$  operation on behalf of T. If the state of  $L \otimes S(X)$  is write  $\operatorname{Iork}(X)$  over other transaction T', the system places T on the writing queue or nem X, if the state of  $L \otimes S(X) \cong$  read\_locked out T itself is the only transaction isolating the read lock on X, the system opgrades the lock to write locked and permits the state network operation by  $T_i$  finally, if the state of  $L \otimes S(X)$  is unlocked, the system grants the write  $\operatorname{Iork}(X)$  is unlocked, the system grants the write  $\operatorname{Iork}(X)$  operation to execute. After each action, the system must update its lock rule appropriately.

Although the two-phase locking protocol guarances serializability (that is, every shedule that is perimited is serializable) in dues not perimit all possible serializable schedules (that is, some serializable schedules will be protobilited by the protocol). In addition, the ase of locks can clause two additional problems deadlock and statisation. We discuss these problems and their solutions in the next section.

## 18.1.3 Dealing with Deadlock and Starvation

**Deadlack** occurs when *oach* transaction **T** in cases of *tasket* now transactions is writing for one frem that is locked by some other transaction **T**<sup>\*</sup> in the set. Hence, each transaction is the set is on a waiting queue, waiting for one of the other transactions in the set to release the lock on an item. A simple example is shown in Figure 18.5a, where the two transactions **T**<sub>1</sub><sup>\*</sup> and **T**<sub>2</sub><sup>\*</sup> are deadlocked in a partial schedule, **T**<sub>1</sub><sup>\*</sup> is on the waiting queue at *X*, which is locked by **T**<sub>1</sub><sup>\*</sup>, while **T**<sub>2</sub><sup>\*</sup> is on the waiting queue to *Y*. Which is locked by **T**<sub>1</sub><sup>\*</sup>, while **T**<sub>2</sub><sup>\*</sup> is on the waiting queue to *Y*. Which is locked by **T**<sub>1</sub><sup>\*</sup> not **T**<sub>1</sub><sup>\*</sup> not any other transaction can access items *X* and *Y*.

**Deadlock Prevention Protocols** One way to prevent deadlock is no use a **deadlock prevention protocol.**<sup>5</sup> One deadlock prevention protocol, which is used in conservative two-phase locking, requires that every transaction lock all the factors it needs in



**FIGURE 18.5** Illustrating the deadlock problem. (a) A partial schedule of  $T_1^{(i)}$  and  $T_2^{(i)}$  that is in a state of deadlock. (b) A wait-for graph for the partial schedule in (a).

<sup>5.</sup> These provocols are not generally used in gractice, either because of note dash, assumptions or bounce of their possible overhead. Deadlock derection and functions (see below) are never procritical.

adrance (which is generally not a gradined assumption) wif any of the items cannot be obtained, none of the items are locked. Bother, the transaction waits and then tries again to lock all the items it needs. This solution obviously further limits concorrency. A second protocol, which also limits concorrency, involves ordering all the items in the database and making sure that it transaction that needs several items will lock them according to that order. This requires that the programmer (of the sector) be aware of the chosen order of the items, which is also not practical in the database context.

A number of other deallock preventions schemes have been proposed that make a decision about what to do with a transaction involved an a possible deallock situation. Should it be blocked and made to wait or should it be aborted, or should the transaction preempt and abort another transaction? These techniques use the concept of transaction timestamp, T5(T), which is a unique identifier assigned to each transaction. The timestamps are typically based on the order in which transactions are started; hence, it transaction T<sub>1</sub> starts before transaction T<sub>2</sub>, then TS(T<sub>1</sub>) < TS(T<sub>2</sub>). Notice that the older transaction to each transaction the older transaction has the singlet timestamp value. Two schemes that prevent deadlock are called wait-die and wound-wait. Suppose that transaction 1, with a conflicting lock. The rules followed by these schemes are as follows.

- Wait-dies If TS(T<sub>i</sub>) < TS (T<sub>i</sub>), then (i) older than T<sub>i</sub>) I<sub>i</sub> is allowed to wait; otherwise (T<sub>i</sub> younger than T) about T<sub>i</sub> (T<sub>i</sub> des) and restart it later with the some torestation.
- Wound-waits It  $TS(T_i) \le TS(T_i)$ , then  $(T_i)$  of der than  $T_i$  above  $T_i$   $(T_i)$  could  $T_i$  and research relater and the same concording otherwise  $(T_i)$  younger than  $T_i \setminus T_i$  is allowed to write

In waited e, an older transaction is allowed to vair on a counger transaction, whereas a counger transaction requesting at from held by an older transaction is abouted and testatted. The wound-wait approach does the opposite. A younger transaction is allowed to wait on an older one, whereas an older transaction requesting an item held by a goinger transaction (see opposite) to iterative be about agree to be body schemes end up aboreing the younger of the two transactions that mity be involved in a deadlock. It can be shown that these two techniques are deadlock-free, and c in wait-die, transactions only wait on polarger transactions so no cycle is created. Similarly, in wound-wait, transactions endy wait on idder transactions is no cycle is created. However, both incliniques macause, since transactions in be about and restarted needlossly, even though those transactions may agree actually crucie a doublesk.

Another group of protocols that provent deadlock do not require timestamps. They arelade the recovering (NW) and corriers activity (CW) algorithms. In the no waiting algorithm, if a transaction is enable to obtain a lock, it is unneducely aborted and then restarted after a certain time delay without checking whether a deadlock will actually occur or non-Because this scheme concurse transactions to abort and restart evedlesse, the cantious waiting algorithm was proposed to ity to reduce the number of needless aborts/restarts. Suppose that transaction 1, trues to lock at item X but is not able to do so because X as locked by some other transaction. To with a conflicting lock. The cantous waiting tules are as follows:

 Cautious waiting: If T<sub>i</sub> is not blocked (nor valuing for some other locked trem), then I<sub>i</sub> is blocked and allowed to wart, otherwise about T

It can be shown that cannots waiting is deadlock-free, by considering the time b(T) if which each blocked transaction T was blocked. If the two transactions T, and T acove bon become blocked, and T, is writing on  $T_{\mu}$  then  $b(T_{\mu}) \leq b(T_{\mu})$  since T can only wait on T, at a time when T, is not blocked. Hence, the blocking times form a total ordering of all blocked transactions, so no cycle that causes deadlock can occur

Deadlock Detection and TimeOUts. A second—more producal- opproach to Jealing with deadlock is deadlock detection, where the system checks of a state of Jealing with deadlock is deadlock detection, where the system checks of a state of Jealing with deadlock is solution is antractive if we know there will be linke institution and the transactions—that is an different transaction swill torch becess the same items at the same time. This can happen if the transactions are short and each transaction locks only a few items, or if the transaction locd is light. On the other based, if transactions are long and each transaction uses many items, or if the transaction load is quite heavy, it may be advantageous to use a deadlock prevention scheme.

A simple way to detect a state of deadlock is for the system to construct and minipui a wait-for graph. One node is created in the wait-for graph for each nan-action that is currently executing. Whenever a transaction T<sub>1</sub> is waiting to leek an iten X that is currently locked by a transaction  $T_{\mu}$  a directed edge  $(T_{\mu} \rightarrow T_{\mu}F)$  created at the wait derigraph. When  $T_i$  releases the lock( $\phi$ ) on the items that  $T_i$  was waiting for she directed edge is drupped from the wair-for graph. We have a state of deadlock if and ock if the wait-for graph bus a cycle. One problem with this approach is the matter of determining when the system should check for a deadlock. Criteria such is the number or currently executing transactions of the period of time second transactions have been waiting to look items may be used. Figure 18.55 shows the wait for graph for the partial) schedule shown in Figure 19.5a. If the system is in a state of dividiock, some of the transactions causing the deadlock must be aborted. Cheeping which transactions in about sknown as victim selection. The algorithm for victim selection should generally acoid selecting transactions that have been tunning for a long time and that have performed mony applaces, and it should us instead to select transactions that have not mape many changes.

Another simple scheme to deal with deadlock is the use of timeouts. This method is protical because of its low overheid and simplicity. In this method, if a transaction wanlist a period, longer than a system-defined timeout period, the system assumes that the transaction may be deadlocked and aborts in—regarilless of whether a deadlock actually system for.

Stativation — Another problem that may occor when we use locking is stativation, which occurs when a transaction cannot proceed for an indefinite period of time while other transactions in the system commute normally. This may second the waiting scheme for kelled atoms is uniford, and generity to some teamsterious over others. One solution to stativation is to have a fair waiting scheme, such as using a first-come-first-served queue transactions are enabled to lock an item in the order in which they originally. requested the bork. Another scheme allows some transactions to have priority over others but increases the provider of a transaction the longer it waits, until it eventually gets the highest priority and proceeds. Statisation can also near because of victim selection if the algorithm selects the same transaction; as victim repeatedly, thus causing it to abort and never finish execution. The algorithm can use higher priorities for transactions that have been aborted multiple ones to avoid this problem. The work-die and wound-wait schemes discussed previously avoid statisation.

# 18.2 CONCURRENCY CONTROL BASED ON TIMESTAMP ORDERING

The use of locks, combined with the 20t protocol guarance's senalizability of schedules. The serializable schedules produced by 2.0t have their equivalent simil schedules base on the order in which executing transactions lock the items they acquire. It a transaction needs an item that is already locked, it may be forred to win until the term is released. A different approach that guarantees scriptizability involves using transaction timestamps to order transaction execution for an equivalent serial schedule. In Section 18.2.1 we discus timestamps and in Section 18.2.2 we discuss how senabrability is enforced by ordering transactions based on their timestamps.

## 18.2.1 Timestamps

Recall that a timestamp is a unique identifier created by the DBMS to identify a transaction. Typically, timestamp collies are assigned in the order in which the transactions are solutified to the system, sub-timestamp can be thought of as the basis action saut time. We will refer to the timestamp of transaction T as TS(T). Concurrency control techniques based on representing othering do not use locks; hence, deallocky canor occur.

Timestamps can be generated in several ways. One possibility is to use a conner that is incremented each nine its value is assigned to a transaction. The transaction rangestamps are numbered [1, 2, 3, ..., un this scheme. A computer counter has a frammissionum value, so the system must periodically used the counter to tero when notransactions are executing for some short periodically used the counter to tero when notransactions are executing for some short period of time. Another way to implementtimestamps is to use the current date/time value of the system clock and ensure that notwo timestamp values are generated during the same tick of the clock.

# 18.2.2 The Timestamp Ordering Algorithm

The idea for this scheme is to order the transactions based on their timestamps. A schedule in which the transactions participate is their schulicable, and the equivalent senal schedule has the transactions in order of their timestamp values. This is called **timestamp** ordering (100). Notice how this differs from 201, where a schedule is scriabioble by being equivalent to some serial schedule allowed by the locking protocols. In timestamp order in Nowever, the schedule is equivalent to the principlus code softwork responding to the order of the transaction intestamps. The algorithm must ensure that, for each item accessed by could ring operations as the schedule, the order in which the trem is accessed less not violate the sentantiability order. To do thus, the algorithm associates with each database item X two intestamp (TS) values

- Read\_TS(X): The read timestamp of item X, this is the largest timestamp among all the timestamps of transactions that have successfully tool, tem X—duat is, read\_ TS(X) = TS(T), where T is the yoregest transaction that has read X succesfully.
- Write\_TS(X): The write timestamp of item λ; this is the largest of all the timestamps of transactions that have successfully written item X—that is, write\_TS(N) + TS(T), where T is the youngest transaction that has written X successfully.

Basic Timestamp Ordering. Whenever some transaction T mes to issue a readreation or write\_rien(X) operation, the basic TO algorithm compares the intertainet I with read TS(X) and write TS(X) coversure that the timestamp order of transaction execution is net violated. If this order is violated, then transaction T is advited and robunited to the system as a receiptransaction with a size transaction T is advited and robunited to the system as a receiptransaction with a size transaction probability of T is abouted and robunited to the system as a receiptransaction with a size transaction by T is abouted and rolled back, any transaction T. that may have used a value written by T must also be railed back. Similarly, any transaction T<sub>1</sub> that may have used a value written by T<sub>1</sub> must also be rolled back, and so on. This effect is known as caseading rollback and is one of the prollems associated with basic TC, since the schedules produced are not guaranteed to be recoverable. An additional generation must be enforced to ensure that the schedules are recoverable. An additional generation must be enforced to ensure that the schedules are recoverable, caseadeless, or struct. We first describe the basic TC algorithm here. The sociation performs real-generation must check whether conflicting operations violate the raisestamp evoluting in the following real cases:

- 1. Transaction T issues a write  $\mathsf{riter}(X)$  operation:
  - a. If read\_TS(X) > TS(T) or of write\_TS(X) > TS(T), then about and roll back T and regret the operation. This should be done because some younger transation with a transition greater than TS(T) —and hence *after* T in the timestion ordering—has already read or written the value of nem X before T had a characteria write X, thus violating the timestamp ordering.
  - b. If the condition in part (a) does not occur, then exercise the write (item[X)) operation of T and set write\_1S(X) to TS(T).
- 2 Transaction () issues a read () rea(20) operation:
  - a. If write\_TS(X) > TS(T), then obert and roll back T and refers the operation This should be done because some counger transaction with randomic groups than TS(T)—and hence often T in the innestoring ordering—has already with near the value of item. X before T had a chance to rood X.
  - Is it write TS(X) ≤ TS(T), then execute the read metri(X) operation of T and set read TS(X) to the larger of TS(T) and the current read\_TS(X).

Hence, whenever the basic TO algorithm detects two conflicting operations that occur in the incorrect order, it rejects the later of the two operations by aborting the transaction that issued it. The schedules produced by basic 30 are hence guaranteed to be conflict senalizable, like the 2PC protocol. However, some schedules are possible under each protocol that are not allowed under the other. Hence, norther protocol allows all possible senalizable subudates. As monitoried earlier, deadlock does not occur with transsaction is ordering. However, cyclic restart (and hence statisation) may occur if a transaction is continually aborted and restarted.

Strict Timestamp Ordering. A variation of base (1) called strict To ensures that the schedules are both strict (for easy recoverability) and (conflict) senablable. In this variation, a transaction T that issues a read\_irran(X) or write\_item(X) such that TS(T) >write TS(X) has its read or write operation delived until the nonsaction T' that increasing value of X (hence  $TS(T') = write_TS(X)$ ) has committed or aborted. To implement this algorithm, it is necessary to simulate the lockate of an item X that has been written by transaction T' until T' is entire committed or aborted. This algorithm does not cause deadleck, since T wats for T' only if TS(T) > TS(T').

Thomas's Write Rufe — A modification of the basic 10 algorithm known as **Thomas's write rule**, does not enforce conflict settablishing; but it rejects fewer write operations by modifying the checks for the write litentX) operations follows:

- 1. Erend, 18(X) > TS(1), then abort and foll back T and reject the operation
- 2. It senter TS(X) > TS(1), then do not execute the write operation but continue processing. This is because some transaction with innestance creater than TS(1)—and hence after T in the timestance ordering—has already written the value of X. Hence, we must import the write premiX) operation of T because his already outdated and obsolete. Nonce that any conflict acising from this situation would be detected by case (1).
- It anthor the conduces report (1) nor the condition in part (2) occurs, then execute the write\_r(ter(X) operation of 1 and set write\_TS(X) to 1S(1).

# 18.3 MULTIVERSION CONCURRENCY CONTROL TECHNIQUES

Other protocols for concurrency control keep the old values of a data neurochen the time is updated. These are known as **multiversion** concurrency control, because several sestors (values) of an item are monitorized. When a transaction acquires access room item an appropriate version is chosen to maintain the scripticability of the currently exerciting schedule at possible. The idea is that some read oper more that would be reported in other rechniques can still be accepted by reading an older covers of the trencto maintain seriitability. When a transaction writes an item, it writes a neurometer to maintain seriitability. When a transaction writes an item, it writes a neurometer and the old version of the item is related. Some multiversion concurrency control, diparithms are the coverpt at view actuality when their conflict senalizability.

An obvious drawbock of truthversion techniques is that increastorage is needed to maintain multiple versions of the database items. However, older versions may have to be montained anyway. For example, for recovery purposes, in addition, some database applications require older versions to be kept to maintain a history of the evolution of diranem values. The extreme case is a temporal database (see Chapter 24), which keeps made of all changes and the times at which they occurred, in such cases, there is no additional orage penalty for multicersion techniques, since older versions are already manuously.

Several multiversion concurrency control schemes have been proposed. We docuse assistences have, one based on timestomp ordering and the other based on 201.

### 18.3.1 Multiversion Technique Based on Timestamp Ordering

In this method, several versions  $X_1, X_2, ..., X_k$  of each durativen X are maintained. For each certain, the value of version X, and the following two timestatures are kept:

- read TS(X<sub>i</sub>): The read timestamp of X<sub>i</sub> is the largest of all the investitings of itansactions that have successfully read version λ.
- write\_TS(X<sub>i</sub>): The write timestamp of X<sub>i</sub> is the timestamp of the transaction that wrote the value of version X<sub>i</sub>.

Whenever a transaction T is allowed to execute a write\_trew(X) operations a new reason  $X_{c,1}$  of item X is created, with roth the write\_TS( $X_{c,1}$ ) and the read\_TS( $X_{c,1}$ ) as its TS(T). Correspondingly, when a transiction T is allowed to read the value of version  $X_{a}$  the value of read [TS(X)] is set to the larger of the turnent read\_TS(X) and TS(T).

To ensure senalitability, the following two rules are used

- 1. If transaction T issues a write\_item(X) operation, and version i of X has the highest write\_TS(X) of all versions of X that is also itsy dam or equal to TS(T), and read\_TS(X)  $\geq$  TS(T), then about and roll-back transaction T, otherwise, treate a new version X, of X with read\_TS(X)  $\geq$  write [TS(X) = TS(T).
- 3. It transaction T issues a read liten(X) operation, head the version c of X that has the highest write\_TS(X) of all censions of X that is also less than or equal to TS(T); then return the value of X<sub>1</sub> to transaction T, and set the value of read 1 S(X) to the larger of TS(T) and the current read\_TS(X).

As we can see in case 2, a **read**, item(X) is always successful, once it finds the appropriate version X, to read based on the series\_TS of the various existing censors of X in case 1, however, regression T may be abound and folled back. This happens if T is mempring to write a version of X that should have been read by another transaction T adopt the variant by another transaction T adopt the variant by another transaction T adopt should have been read by another transaction T above transaction with transmup equal to write\_TS(X). If this conflict occurs, I is colled back, etheracise, a new version of X, written by transaction *X*, which was suffer that, if T is colled back, cascading fullback may occur. Hence, to ensure accountility, a measurement T should ever be a lowed to commit until after all the associations that have written some version that T has read have commuted.

# 18.3.2 Multiversion Two-Phase Locking Using Certify Locks

In this multiple-mode locking scheroe, there are does beking bodes for an item real, write, and centry, instead of pise the two modes (read, write) discussed previously. Hence, the state of  $|(\infty K(X))|$  for an item X can be one of read-locked, write-locked, centry-locked, or unlocked. In the standard locking scheme with only read and write locks (see Section 18.1.1), in write locks is an exclusive lock. We can describe the relationship between read and write locks in the standard scheme by means of the linek compatibility table shown in Figure 18.6a. An entry of yes means that, if a transaction T holds the type of lock specified in the colour of the same item X and it transaction T holds the type of lock specified in the pow loader on the same item X, then T' can abade the type of lock specified in the row loader on the same item X, then T' can abade the table inlegares that the locks are compatible. So T' must wait until T releases the lock.

In the standard locking scheme, once a transaction obtains a write lock on an iron, no other transactions can access that item. The idea behind in driversion  $2^{n}$  is mallow other transactions **T** to read on item X while a angle transaction **T** holds a write back on X. This is accomplished by allowing the removie for each item X one version inter alwas have been written by some committed in insaction. The second version X' is created when a transaction T acquires a write lock on the nert. Other transactions can continue to read the committed coston of X while T holds the write lock. Transaction T can write the value of X' as needed, without offecting the value of the committed version X affects of the committed version X and the transaction T can write the value of X' as needed, without offecting the value of the committed version X affects of the value of the committed version X as needed.



FIGURE 18.6 Lock compatibility tables, at A compatibility table for read/write locking scheme, the A compatibility table for read/write/consty locking scheme. controls holds write focks on before it can commit. The certify lock is not compatible with read locks, so the transaction may have to delay its commit until all its write-locked items are released by any reading transactions in order to obtain the certify locks. Once the certify locks—which are exclusive locks—are acquired, the committed version X of the data item is set to the value of version X<sup>2</sup>, version X<sup>2</sup> is discarded, and the certify locks in their released. The lock comparibility table for this scheme is shown in Figure 18.0b.

In this multiversion 20t scheme, reads can proceed concernently with a sugge write operation —an arrangement not permitted under the standard 2.9t schemes. The cost is that a transaction may have to delay its commit unof it obtains exclusive centry locks or all the news it has updated. It can be shown that this scheme avoids cusculing abaris, once transactions are only allowed to read the version X that was written by a committed transaction. Eleweter, deadlocks may occur it apgrading of a read lock to a write lock is allowed, and these must be bandled by variations of the techniques discussed in Section 18.1.3.

# 18.4 VALIDATION (OPTIMISTIC) CONCURRENCY CONTROL TECHNIQUES

In all the concurrency control techniques we have discussed so for, a certain degree of clocking is done before a database operation can be executed. For example, in locking, a clock is done to determine whether the item being accessed is locked. In timestant ordering, the transaction timestamp is checked against the read and write timestamps of the item. Such checking represents overhead doring transaction execution, with the effect of slowing down the transactions.

In optimistic concurrency control techniques, also known is validation or testification techniques, no checking is done while the transaction is excerning. Several project concurrency control methods use the validation technique. We will describe environe scheme here. In this scheme, updates in the transaction are for applied describ to the database items until the transaction reaches its end. During transaction executions all updates are applied to heat eights of the data items that are kept for the transaction.<sup>2</sup> At the end of transaction execution, in validation phase checks whether any of the transaction's updates violate semilicability. Certain information needed by the validation place must be kept by the system. If semilicability is not violated, the transaction is connicted and the database is updated from the local copies; otherwise, the transaction is shorted and then restored later.

There are three phases for this concurrency cuttrol protocole

 Read phase: A transaction can read values of committed data trens from the database. However, updates are applied only in local copies (versions) of the data joins kept justice (papsaction workspace).

6 Note that this can be considered as keeping analoph, which is a closes?

- Validation phases Checking is performed to ensure that secialitability will not be stolated if the transaction updates are applied to the database.
- 3 Write phases It the conductor phase is successful, the transaction updates are applied to the database; otherwise, the updates are distuiced and the transaction is restarted.

The idea behind optimistic concurrency control is to do all the checks at encehence, transaction execution proceeds with a matiatian of overhead until the validation phase is reached. If there is finite interference among transactions, most will be validated successfully. However, if there is much interference, many transactions that execute to completion; will have their readits discarded and must be restarted later. Under these incompletions will have their readits discarded and must be restarted later. Under these incompletions will have their readits discarded and must be restarted later. Under these incompletions will have their readits discarded and must be restarted later. Under these incompletions will have their readits discarded and must be restarted later. Under these incompletions will have their readits discarded and must be restarted later. Under these incompletions will have their readits discarded and must be restarted later. Under these incompletions are called "optimistic" because they assume that little interference will occur and hence that there is no need to do checking during transaction execution.

The optimistic protocol we describe use-transaction timestamps and also requires that the write sets and readisets or the consactions be kept by the system. In addition, we and end rules for some of the three phases need to be kept for each transaction. Recall that the write\_set of a transaction is the set of items it writes and the read\_set is the set of terms in reads. In the volidation phase for transaction T<sub>a</sub> the protocol checks from T<sub>a</sub> day not interfere with any computed transactions or with any other transactions currently in their volidation phase. The volidation phase for T<sub>a</sub> checks that, for each such transactions T<sub>a</sub> that is either computed or is in its validation phase, one of the following conditions holds.

- Transaction T, completes as write phase before T, starts its read phase.
- F<sub>i</sub> starts its write phase after T completes its write phase, and the near isst of T, has no sterils in common with the write set of T.
- Both the read\_set and wrote (set of T, have nonconsin control) with the write, set of T<sub>0</sub> and T<sub>1</sub> completes us read phase before T<sub>1</sub> completes us read phase.

When validaring transaction  $T_i$  the first condition is checked first for each transaction  $T_i$ , since (1) is the simplest condition (acheck. Only if condition (1) is false is condition (2) checked, and only if (2) is false is condition (3)—the most complex to evaluate—checked. If any one or these three conditions holds, there is no interference and  $T_i$  is validated successfully. If none of these three conditions holds, the validation of transaction  $T_i$  tails and at is abserted and testarted later because interference may have occurred.

# 18.5 GRANULARITY OF DATA ITEMS AND MULTIPLE GRANULARITY LOCKING

All concarrency control rechariques assumed that the database was formed of a number of runned data items. A database irom could be chosen to be one of the following:

- A database record.
- A field value of a database record.

- A disk block
- A whole tile.
- The schole database

The symmilarity can affect the performance of concurrency control and recovery in Section 18.5.1, we decreas some of the maleoffs with regard to choosing the granularity level used for locking, and, in Section 18.5.2, we discuss a multiple granularity acking science, where the granularity level (size of the data stend) may be changed dynamically.

#### 18.5.1 Granularity Level Considerations for Locking

The size of data items is often called the data item granularity. Energymotority refers to small item sizes, whereas course granularity refers to large item sizes. Several tradeoffs must be considered in choosing the data item size. We shall discuss data item size in the context of lockarig, although similar arguments can be made for other concentrative control techniques.

First, notice that the larger the data item site is a the lower the degree of concernercy genated. For example, if the data item site is a disk block, a transaction T that weeks to lock a record D must lock the whole disk block X that contains B because a lock is associated with the whole data item (block). Now, if another transaction S wants to lock a different record C that happens to reside in the same block Z in a costileting lock and it is forced in wait. If the data item sin was a single record, transaction S would be she to proceed, because it would be locking a different data item (record).

On the other hand, the smaller the data item size is the more the number of arous in the database. Because every item is issociated with a lock, the system will have a larger number of acrive locks to be handled by the lock manager. More lock and anlack operations will be performed, causing a higher overhead. In addition, more storage space will be required for the lock table. For timestamps, storage is required for the reed\_T5 and write TS for each data near, and there will be similar overhead for handling a large number of items.

Given the above tradeoffs an obvious question can be asked. What is the best item see? The answer is that it depends on the (gives a) considerable medical. If a typical number of records, it is advantageous to have the data item grandarity be one record. On the other hand, it a tempaction typically accesses many areads in the same life, it may be better to have block of the grandarity so that the prander of the same life, it may be better to have block of the grandarity so that the prangetion will consider all these records as one (or a tew) data items.

### 18.5.2 Multiple Granularity Level Locking

Since the best granulatity size depends on the given transaction, it seems appropriate that a datasets system support multiple levels of granular to where the granularity level can be different for various mixes of transactions. Figure 18.7 shows a simple granulatity hierarthy with a database containing two files, each file containing several pages and each page areasing several records. This can be used to illustrate a **multiple granularity level** 201.



FIGURE 18.7 A granularity hierarchy for illustrating multiple granularity level locking

protocol, where a lock can be requested at any level. However, additional types of locks will be needed to efficiently support such a protocol.

Consider the following scenario, with only drafed and exclusive lock types, that refers to the estimple in figure 18.7. Suppose transaction  $\Gamma_1$  wants to update addie zerods in file  $f_1$ , and  $T_1$  requests and is granted an exclusive lock for  $f_1$ . Then all of  $f_1$ 's pages  $f_{10}$  through  $p_{10}$ )—and the records contained on those pages—are locked in exclusive mode. This is menched for  $T_1$  because setting a single file-level lock is more efficient data setting a page-level locks or having to lock each individual record. Now suppose another transaction  $T_1$  only wants to read record  $r_{10}$  from page  $p_1$ , of file  $f_1$ ; then  $T_2$  would request a shared record-level lock or  $r_0$ . However, the database system (that is, the transaction manager or more specifically the lock manager) must verify the compatibility of the requested lock with already held lock. One way to verify the is to traverse the tree from the leaf  $r_{10}$  to  $f_1$  to  $f_1$  to  $f_2$  to  $f_1$  to  $f_1$  to  $f_2$  to  $f_1$  any time a confidering lock is held or any of these structs the traverse the tree from the lock request for  $r_{10}$  is defined and  $T_2$  is blacked and must war. This traversil would be tarry efficient.

However, which it transaction  $T_2$ 's request came before transaction. This request? In this case, the shared record lock is granted to  $T_2$  for  $\tau_{pot}$  but when  $T_1$ 's file level lock is requested, it is quite difficult for the lock marger to check all nodes (pages indirecords) that are descendants of node  $f_1$  for a lock conflict. This would be very methermicant would defend the purpose of having multiple granularity level locks.

To make multiple granulating level locking practical additional repeatif locks, called intention locks, are needed. The idea behind intendom locks is for a transaction to indicate, along the path from the root to the desired node, what type of lock (shared or exclosive) it will response from one of the node's descendants. There are shree type of intention locks:

- Internion-shored (IS) indicates that a shared lack(s) will be requested on some descendant node(s).
- Interitorisexclusive (IX) indicates that on exclusive lock(s) will be requested on some descendars) node(s).
- Shared-intention-exclusive (SIX) indicates that the current node is locked in shared mode but in exclusive lock(s) will be requested on some descendant node(s).

The compatibility table of the three intention locks, and the shired and exclusive locks, is shown an fagore 18.8. Desides the non-duction of the duree types of amention locks, an appropriate locking particul must be used. The **multiple granularity locking** (MGL) protocol consists of the following rules:

- 1. The lock compatibility (based on Figure 18/8) must be adhered to
- 2. The root of the tree must be locked first, in any mode
- A node N can be locked by a transaction T in S or IS mode only if the parent node N is already locked by cran-action T in either IS or IX node.
- A node N can be locked by a transaction T in X, IX, or SIX mode only if the parent of node N is abeady locked by impaction T in either [N or SIX mode.
- A transaction T can lock a node only if it has not unlucked any node (roemforce the 2.1, protocol).
- A transaction T can unlock a node. M only it none of the children of node N are currently locked by T.

Kule I simply states that conflicting locks cannot be granted. Bules 2, 3, and 4 state the conditions when a transaction may lock a given node in any of the lock modes. Bales 5 and p of the MCL protocol enforce 2PL toiles to produce senalitable schedules. To illustrate the MGL protocol with the database hierarchy in Figure 18.7, consider the following three transactions:

- T<sub>1</sub> wants to update record (1) and record (2):
- 2. T<sub>2</sub> worns to opdate all records on page  $p_{12}$ .
- Tywants to read record c<sub>11</sub> and the entire f<sub>2</sub> (the.)

Figure 18.9 shows a possible sensitioble schedule for these three transactions. Only the lock operations are shown. The notation silick (types) satema) is used to display the locking operations in the schedule.

|    | IS  | 18  | 5    | SIX | X  |
|----|-----|-----|------|-----|----|
| 12 | yos | yes | yes. | уез | nn |
| x  | yes | yes | no   | 70  | no |
| ŝ  | yes | no  | yes  | 00  | no |
| sx | yes | ίω  | no   | סר  | na |
| x  | nO  | ю   | no   | סר  | па |

REURE 18.8 Lock compatibility matrix for multiple grapularity locking
| ÷1                                         | <b>-</b> :                               | r <sub>a</sub>                   |
|--------------------------------------------|------------------------------------------|----------------------------------|
| IX(db)<br>DX(d)                            |                                          |                                  |
| 10100                                      | DX(ab)                                   |                                  |
|                                            |                                          | 75( <i>d</i> 0)                  |
|                                            |                                          | 43(4)                            |
| N/1 & 1                                    |                                          | isib'')                          |
| X(p <sub>11</sub> )<br>X(r <sub>11</sub> ) |                                          |                                  |
| 2010                                       | DS(7.)                                   |                                  |
|                                            | X (Provi                                 |                                  |
|                                            |                                          | $S(r_{12})$                      |
| $00(f_2)$                                  |                                          |                                  |
| $100(\mu_{21})$                            |                                          |                                  |
| $X I_{2} _{2}$                             |                                          |                                  |
| wniock(rgg)                                |                                          |                                  |
| $univ(p_{2i})$                             |                                          |                                  |
| unio¢4/,)                                  |                                          |                                  |
|                                            |                                          | $S(t_2)$                         |
|                                            | uniockijp <sub>iz</sub> )<br>uplockijt ( |                                  |
|                                            | uniock(db)                               |                                  |
| unicipal/ <sub>100</sub> 1                 |                                          |                                  |
| $(mlock)(p_m)$                             |                                          |                                  |
| Uniock(C)                                  |                                          |                                  |
| gnigek(189)                                |                                          | uniocikis i                      |
|                                            |                                          | unova(*n.)<br>Un <b>ios</b> k(a) |
|                                            |                                          | uniock(4)                        |
|                                            |                                          | uniock(f <sub>2</sub> )          |
|                                            |                                          | unkack(db)                       |

FIGURE 18.9 Lock operations to illustrate a senalizable schedule.

The multiple granularity level protocol is especially suited when traceosing a mixor transactions that include: (1) short transactions that access only a few items (records at nelds), and (2) long transactions that access entire files. In this environment, lew transaction blocking and less locking averhead is incurred by such a tratecol when composed to a angle level granularity locking approach.

### 18.6 Using Locks for Concurrency Control in Indexes

Iwo-phase locking can also be applied to indexes (see Charmin 14), when the nodes of an index concespond to disk pages, blowever, holding locks on index pages until the shrinking phase inf 20t could chose an under amount of transaction blocking. This is because carefung an index always (forts of the not), so if a transaction wants to insert a record forte operation), the root would be locked in exclusive mode, so all other conflicting lock requests for the index must wait until the transaction emers its sprinking phase. This blocks all other transactions from accessing the index, so in gradice other approaches to lacking an index must be used.

The tree structure of the index can be taken advantage of when developing a concurrency control scheme. For example, when an index south small operation) is being evenered, a path in the tree is traversed from the root to a leaf. Once a lower level base in the path has been accessed, the higher level nodes in that path will not be used again. So once a read look on a child made is obtained, the lock on the parent can be teleased. Second, when an insertion is being applied to a leaf node (that is, when a key and a pointer are inserted), then a specific leaf node must be locked in each when a lower level however, if that node is not full, the insertion will not cause changes to higher-level index nodes, which implies that they need not be locked excharged.

A conservative approach for in-criticit-would be to lack the root node in exclusive node and then to access the appropriate child node of the root. If the child node is not tall, then the lock on the root node can be released. This approach can be applied all the way down the tree to the leaf, which is typically three or four levels from the mar. Although exclusive locks are held, they are snot released. An alternative, more opinistic approach would be in request and hold shared locks on the nodes leading to the leaf node, with an exclusive lock on the leaf. If the insertion causes the leaf to splitinsertion will propagate to a higher level node(s). Then, the locks on the higher level rode(s) can be approach to exclusive mode.

Another approach to index locking is to use a cotiant of the Bittree, called the **Blink tree**. In a B link tree, sibling nodes on the same level the linked together it every level. This allows shared locks to be used when respective a page and requires that the lock be released before accessing the child node. For on insert operation, the shared lock enancies would be operated to exclusive mode. If a split occurs, the parent node must be telocked an exclusive mode. One complication is for search operations executed concurrently with the update. Suppose that a concurrent update operation follows the sine path as the search, and inserts a new entry into the leaf node. In addition, suppose that the usert entries that leaf each to split. When the unsert is done, the search process reames. Following the pointer to the desired leaf, only to find that the key of selecting for a next present because the split has mored that key into a new feat node, which would be the right attempt of the original leaf node. However, the search process can still varied by the original leaf node. However, the search process can still varied by the original leaf node routs right athing, where the desired key may been moved.

Hondling the deletion case, where two or more nodes from the index free increases also part of the B-link free concurrency protocol. In this case, locks on the nodes to be inorged are held as well as a lock on the parent of the two nodes to be merged.

### **18.7 OTHER CONCURRENCY CONTROL ISSUES**

In this section, we docuss some other asses relevant to concurrency control. In Section 13.7.1, we docuss problems associated with insertion and deletion of records and the socalled phanoin problem, which may occur when records are inserted. This problem sodescribed as a potential problem requiring a concurrency control measure in Section 13.5. Then, in Section 16-7.2, we discuss problems that may occur when a transaction or parsome data to a monitor before it commits, and then the transaction is later aborted.

#### 18.7.1 Insertion, Deletion, and Phantom Records

When a new data item is inserted in the database, it obvicusly cannot be accessed untiatter the item is created and the insert operation is completed. In a locking environment a lock for the item can be created and set to exclusive (write) model the lock can be released at the same time as other write locks would be released, based on the correrency control protocol fremg used. For a timestamp-based protocol, the read and write timestamps of the new ticut are set to the signestamp of the creating trapatition.

Next, consider deletion operation that is applied on an existing data item. For lock up protocols, again an exclusive (write) lock must be obtained before the transaction con-delete the item. For timestamp ordering, the protocol must ensure that to late transaction has read or written the trenched or allowing the neurito be deleted.

A struction known as the **pharton problem** can occur when a new record the sheing inserted by some consistion T satisfies a condition that a set of records accessed platother transaction T must satisfy. For example, suppose that transaction T instructionare we EMPLOYEE record whose  $1 \times 0 = 5$ , while transaction T is accessing all EMPLOYEE records whose  $1 \times 0 = 5$ , while transaction T is accessing all EMPLOYEE records whose  $1 \times 0 = 5$ , while transaction T is accessing all EMPLOYEE records whose  $1 \times 0 = 5$  (so), to add up all then SALACO values to calculate the personnel budget for department 5). If the equivalent serial order is T followed by T', then T must read the new EMPLOYEE record and methods its SALACO in the sum calculation. For the equivalent serial order T' followed by T, the new salary should not be included. Notice that although the transactions logically conflict, in the latter case there is really no access (data atom) in common between the two transactions, since T' may have linked all the records with 1000 = 5 *lefore* T inserted the new record. This is because the record that causes the conflict is a **phantony record** that has sublenly appeared in the database on being inserted. If other updations in the rans relations confirm the rans relations to the rans relations of the ransactions of the rans record all the records with 1000 = 5 *lefore* T inserted the new record. This is because the record that causes the conflict is a **phantony record** that has sublenly appeared in the database on being inserted. If other updations in the rans ratio ransactions control protocol.

One solution to the phantom record problem is rease index locking, as divised in Section 1840. Recall from Chapter 14 that an index includes matries that have an attribute taket plus a set of pointers to all records in the file with the value. For example, an index on 1960 of FS12105 -F would include an entry for each distinct DNO value plus a set of pointers to all EMPLOYFE records with that value. If the index entry is locked before the record itself can be accessed, then the conflict on the phontom received can be detected. This is because transaction  $\mathbf{T}$ , would request a read lock on the index entry for DNO = 5, and  $\mathbf{T}$  would request a write lock on the same entry *before* they could place the locks on the actual records. Since the index locks conflict, the phantom centlet would be detected.

A more general rechangue, called predicate locking, would lock access to all records that satisfy an arbitrary fredicate (condition) in a similar manner, however predicate locks have proved to be difficult to implement efficiently.

### 18.7.2 Interactive Transactions

Another problem occurs when interactive transactions reading of and write origin to an interactive device, such as a monitor screen, before they are committed. The problem is due a user can input a value of a data item to b transaction. Tablet is based on some value writer no die screen by transaction T', which may not have committed. This dependency between T and T' conner be moduled by the system concurrency control method, since it is or ly based on the user interacting with the two transactions.

An approach to dealing with this problem is to poetpone output of transactions to the streen until they have committed

#### 18.7.3 Latches

Locks held for a short duration are typically called latches. Each has do not follow the estal concurrency control protocol such as two-ghase locking. For example, a latch can be used to guarantee the physical integrity of a page when that page is being written from the buber to disk. A latch would be acquired for the page, the page written to disk, and then the latch to released.

### 18.8 SUMMARY

In this chapter we discussed U9948 rechniques for concurrency control. We started by discusing lock-based protocols, which are by far the most commonly used in practice. We cestined the two-phase locking (201) protocol and a number of its variations, basic 201, start 201, conservative 201, and figurous 201. The struct and agoreus variations are note common because of their better recoverability properties. We introduced the concepts of ability when used in conjunction with the two-phase locking rule. We also presented varues techniques for dealing with the deadlock grablem, which can occur with locking fuplatice, it is common to use timeouts and deadlock detection (wint-for graphs)

We then presented other concurrency control protocols that are not used often in protice but are important for the theoretical alternatives they show for solving this problem. These include the timestamp ordering protocol, which ensures serialitability level as the order of transaction transitions. Encourances on unique, spreit-spreitoriance transaction identifiers. We discussed Thomas's write tale, which improves performance buildee not guarantee conflict senalitability. The strict timestamp ordering protocol was also presented. We then discussed two multiversion protocols, which assume that older cersions of data items can be kept as the database. One technique, called multiversion two-phase locking (which has been used in practice), assumes that two versions can exist for an item and attempts to increase concurrency by making write and read fields compatible for the cost of introducing an additional certife lock mode). We also presented a unification protocol based on timestamp ordering. We then presented a example of an optimistic protocol, which is also known as a certification or validation protocol.

We then runned our attention to the important practical issue of data item granulating. We described a multigran larity locking protocol that allows the change of granulative (item stell based on the current transaction they with the goal of improving the performance of concurrence control. An important gracited issue was theo presented, which is to develop locking protocols for indexes so that indexes do not become a hindicance to concurrence access. Finally, we introduced the phantom problem and problems with interactive transactions, and briefly described the concept of latches ad how it differs from locks.

In the next chapter, an gree an overview of recovery techniques.

#### Review Questions

- 18.1. What is the two-phase locking promotol? How does it guarantee senalitability?
- 19.2. What the some variations of the new phase locking protocol? Why is struct or ng orous two-phase locking alien preferred?
- 18.3. Discuss the problems of deadlock and starvarian, and the different approaches to dealing with these problems.
- 48.4. Compare binary locks to exclusive/shared tocks. Why is the latter type of lockpreferable?
- 18.5. Describe the workdre and wound-wair protocols for deadlock prevention.
- 18.6. Describe the cautions waiting, no waiting, and timeout protocols for deadlock prevention.
- 18.7. What is a function p? How does the system generate (unestanged
- E3.8. Discuss the timestamp ordering protocol for concurrency control. How does soor rituestamp ordering differ from basic timestamp ordering?
- 18.9. Discuss two multiversion techniques for concurrency control.
- 13.10. What is a certain lock? What are the advantages and disadvantages of using certain locks:
- 18.11. How do extinuistic concurrency control rechniques differ from other concurrency control techniques? Why are they also child validation or certification rechniques? Discuss the typical phases of an optimi-tic concurrency control method.
- 18 12: They does the granularity of data stems affect the performance of concenency control? What factors affect selection of cumularity size for data items?

- 18.13. What type inflocks are norded for insert and delete operations?
- 18.14. What is multiple granularity locking? Under what encoustances is it used?
- 18.15. What are intention locks?
- 18.16. When one latebas used?
- 18-17 What is a phantom record? Discuss the problem that a phantom record can cause for concustency control.
- 18-16. How does index locking resolve the phantom problem?
- 18-19 What is a predicate lock?

### Exercises

- 20.20 Prove that the basic two-phase locking protocol guarantees conflict serial cability of schedules. (Ffor: Show that, if a serializability graph for a schedule has a cycle, then at least one of the transactions participating in the schedule does not obey the two-phase locking protocol.)
- 13.21. Medify the data structures for multiple-mode locks and the algorithms for read\_ lock(X), white\_lock(X), and unlock(X) so that upgrading and descrerating of locks are possible. (East: The lock needs to check the transaction id(s) that hold the lock, it any.)
- (8.22) Prove that artist two phase to king guarant-example schedules.
- 18.23. Prove that the work die and wound wait protocols aword deadlock and statisation.
- (\$24). Prove that campous watting avoids deadlock.
- 15.25. Apply the timestump ordering obsorption to the schedules of Figure 17.8(b) and (c), and determine whether the algorithm will allow the execution of the schedules.
- 18/26 Repeat Exercise 16/25, but use the multiversion timestamp ordering method
- 18.27. Why is two-phase locking net used as a concurrency control method for indexes such as B\*-trees?
- 18/28 The compatibility matrix of Figure 18.8 shows that iS and IX locks are compatible. Blo. Explain why this is valid.
- 18/29. The assumption collections that a transaction T cursual ock a node N, only if none of the children of node N are still locked by transaction T. Show that without this condition, the VGI protocol would be incorrect.

### Selected Bibliography

The two-phase locking protocol, and the concept of predicare locks was first projected by Eswaran et al. (1976). Bernstein et al. (1987). Guy and Rumer (1993), and Papadinamon (1986) focus on concentrate control and recovery. Kumir (1996) focuses on perforbance of concentrates, control methods. Locking is discussed in Gray et al. (1975). Len and Weinberger (1978). Kedem and Silbershatz (1980), and Kotth (1983). Deadlocks and wairder graphs were formalized by Holi (1972), and the wair-wourd and wound-the schemes are presented in Rosenkranti et al. (1978). Cautions waiting is discussed in Silversed in the schemes are presented in Rosenkranti et al. (1978). Cautions waiting is discussed in the schemes are presented in Rosenkranti et al. (1978). Cautions waiting is discussed in Hon et al. (1992). Helaf et al. (1993) compares various locking approaches. Timestrang-based cucuriency control techniques are discussed in Peneticin and Goodman (1980), and Keed (1983). Optimistic commercity control is discussed in Kimit and Rosensin (1981). ind Bassicom (1988). Papadmutriou and Kanellakis (1979) and Beristein and Goodnam (1983) discuss multiversion techniques. Multiversion timestamp endering was proposed in Reed (1978, 1983), and multiversion two-phase locking is discussed in Ear and Witkinson (1984). A method for multiple locking granularities was proposed in Gray (1 al. (1976), and the effects of locking granularities are analyted in Rics and Stonebraker (1977). Bharpina and Reidl (1988) presents an approach for dynamically choising among various concurrency control and recovery methods. Concurrency control methods for indexes are presented in Lehingh and Yau (1981), and in Shasha and Goodman (1988). A performance analyte of various B+ species concurrency control and Goodman (1988). A performance analyte of various B+ species concurrency control and reserved in Stinisoon and Carey (1991).

Other recent work on concurrency control includes semiatric-based concurrence control (Bademath and Baniant tham, 1992), transaction models for long terming activities (Daval et al., 1991), and multilevel transaction minagement (Hosse and Weakum, 1991)





## Database Recovery Techniques

In this chapter we discuss some of the rectiniques that can be used for database recovery term failures. We have already discussed the different causes of failure (such as system cashes and transaction cross, in Section 17.1.4. We have also covered many of the contepts that are used by receivery processes, such as the system log and commit points, in Section 17.2.

We start Section 19.1 with an arthuse of a typical perovery procedures and a coregorinten of recovery algorithms, and then discuss several recovery covcepts, including writeshead longing, in place versus shadow inclutes, and the process of rolling back fundoing) the effect of an incomplete or field transaction. In Section 19.2, we present recovery techniques based on defeoral applice, also known as the NO-ONE-NO-EFFEC technique. In Section 19.3, we discuss recovery techniques based on transchate righter, these reclude the exist (REDO) and ENDO/NE-SETEC algorithms. We discuss the technique known as shalowing or shadow paging, which can be categorized as a NO-ONE-NO-SETEC algorithm in Section 19.4. An example of a practical DWBs receivery scheme, called AterEs is presented in Section 19.5. Recovery in multidatalwases is briefly discussed in Section 19.6. Finally, availables for secovery from qatastrophic tailing are discussed in Section 19.7.

Our emphasis is on conceptually describing several different approaches to recovery. For descriptions of recovery fratures in specific systems, the reader should consult the hibliographic noises and the user manuals for these systems. Recovery techniques are often increased with the concurrency control mechanisms. Certain recovery techniques are been seed with specific concurrency control mechanisms. We will attempt to docuss measure concepts independently of concurrency control mechanisms, but we will discuss the concurstances under which a particular recovery incohorism is best used with a cestair concurrency control protocol.

### **19.1 RECOVERY CONCEPTS**

### 19.1.1 Recovery Outline and Categorization of Recovery Algorithms

Becovery train transaction failures usually means that the database is recorder to the matrecent consistent state just before the sime of tailure. To do this, the system must keep information about the changes that were applied to data atoms by the various strustotions. This information is reproally kept on the system log, as we discussed in Section 1/2/2. A typical strategy for recovery may be summarized informable as follows.

- 1. If there is extensive domage to a wide portion of the database due to catastrophal failure, such as a disk crash, the recovery method testores a past copy of the database that was backed up to archival storage (repeatly tape) and reconstructs a more current store by reapplying or redoing the operations of committed transactions from the facilities do log up to the time of fadure.
- 2. Where the database is not physically damaged but has become incensistent due to non-catastrophic failures of types I through 4 of Section 17.7.4, the strategy is to reverse any changes that caused the meanistency by analoing some operations is may also be necessary to redu some operations in order to restore a consistent state of the database, as we shall see, in this case we do not need a complete archival copy of the database. Eather, the entries kept in the online system lag are em-sulted during recovery.

Conceptually, we can distinguish rate main rechraques for recovery from recordastraphic transaction failures: (1) deferred update and (2) anuaction opdate. The deferred update techniques do not physically update the database on disk until after a transaction reaches as commit point; then the updates are recorded in the duabase. Before reaching commit all transaction updates are recorded in the duabase. Before reaching commit all transaction updates are recorded in the log ars/ then written to the database. If a transaction fails before reaching its commit point, it will not have charged the database. If a transaction fails before reaching its commit point, it will not have charged the database in any way so UNDO is not needed. It may be necessary to REDO the effect of the operations of a committed transaction from the log, because their effect may not yet loss been recorded in the database. Hence, deferred update is also known as the SO-UNDO/ REDO algorithm. We discuss this technique in Section 19.2.

In the immediate update techniques, the database may be updated by some operations of a transaction before the transaction reaches its commit point. However, these operations are typically recorded in the log or disk by force writing folice they are applied to the database, making recovery still possible. If a transaction tails after recording some thanges in the database has before reaching its commit point, the direct of its

operations on the database must be undone, that is, the transaction must be reduced back. In the general case of immediate opdate, both undo and rate new be required ducitar terowery. This rechnique, known as the UNDO/REDO algorithm, requires both operations, and is used most often in practice. A voriation of the algorithm where all apdates are recorded in the database before a transaction commute requires melo only, so it is known as the UNDO/NO-REDO algorithm. We discuss these techniques in Section 19.3.

### 19.1.2 Caching (Buffering) of Disk Blocks

The necessary process is often closely intertwined with operating system functions—in particular, the buffering and eaching of disk pages in main memory. Typically, one or more lisk pages that include the data items to be updated are cached into main memory buffers and their updated in memory before being written back to disk. The eaching of disk pages is traditionally an operating system function, but because of its importance to the efficentry of recovery procedures, it is handled by the DIMEs by colling low-level operating systems routines.

In general, it is convenient to consider recovery in terms of the database disk pages (alarks). Typically a collection of minimum putfers, called the D888 cache, is kept under the control of the 0986 run the purpose of holding these buffers. A directory for the coche is used to keep track of which database items are in the buffers. A directory for the coche is used to keep track of which database items are in the buffers. This can be a table of edisk page address, huffer locations entries. When the D888 requests action on some item, at first checks the cache directory to determine whether the disk page containing the item is in the cache. If it is not, then the item must be located on disk, and the appropriate disk pages are copied into the cache. It may be necessary to replace (or flush) some of the cache buffers to make space available for the new item. Some page-replacement strategy from operating systems, such as least recently used (LRO) entration-thrstion (FEO), can be used to select the buffers for replacement.

Associated with each Soffer in the cache is a dirty bit, which can be included in the directory entry, to indicate whether or not the buffer has been modified. When a gage is first read from the database disk many cache buffer, the cache directory is epdated with the new disk page address, and the dirty bit is set in C (2000). As soon as the buffer is modified the dirty bit for the corresponding directory entry is set to 1 (one). When the buffer is modified in corresponding disk page only if is directory entry is set to 1 (one). When the buffer ordering disk page only if is dirty bit if 1. Another bit, called the **pin-anpin** bir, is also needed—a page in the dach, is **pinned** fort value 1 (one)) if it cannot be written back indick as yet.

Two mean strategies can be employed when the burge a modified burger back to disk. The first strategy, known as in-place applating, writes the buffer back to the sone original disk lacation - thus overwriting the old value of any changed data items on disk.<sup>2</sup> Hence, a single copy of each database disk block is maintained. The second strategy, known as shadowing, writes an updated buffer at a different disk location, so multiple versions of

<sup>1.</sup> They is some effort straight to the concept of page under used by the operating system.

<sup>2</sup> To-place updatang wookd in most systems to practice.

data items can be manutamed. In general, the old value of the data item before a plaine, is called the **before image (BFIM)**, and the new value after opdatine is called the **after image (AFIM)**. In seadowing, both the WDI and the ATDI can be kept on disk, bence, it is not strictly necessary to maintain a log for recovering. We briefly discuss recovery based on shadowing in Section 19.4.

### 19.1.3 Write-Ahead Logging, Steal/No-Steal, and Force/No-Force

As mentioned, the DBMS cache holds the cached database disk blocks, which include not only data flacks but also index blocks and by blocks from the disk. When is log record is written, it is strend in the current log block in the DBMS cache. The log is simplify sequential (appendiordy) disk file and the DBMS cache most contain several log blocks flor example, the last nolog blocks) that will be written to disk. When an update to a data block —stored in the DBMS cache —is made, an associated log record is written to the last hig block in the DBMS cache —is made, an associated log record, is written to the last log block in the DBMS cache —is made, an associated log record, the log blocks that content the non-intext log records for a genticular data block update must and be reciting to disk before the data block itself can be written back to disk.

Standard 196Ms recovery terminishipy includes the terms steal/no-steal and force/noforce, which specify when a page from the database can be written to disk from the cache:

- 1. If a cache page opdated by a transaction coerce benefitters to disk before the transaction commits, this is called a nu-steal approach. The procurput ha indicates if a page connecte eventeen back to disk. Utherwise, if the protocol allows writing on spoking hadren before the transaction originals, it is called steal. Steal is used when the UBMS coche (bother) managet needs a butter frame for another transaction and the bother manager replaces on existing page that had been opdated but whose transaction has not committed.
- If all page- apdated by a transaction are immediately written to disk when the transaction commits, it is a called a force approach. Otherwise, n is called no-force-

The deferred update recovery scheme in Section 19.2 follows a no-sead approach However, typical database systems employ a steal/no-joice strategy. The advantage of steal is that it is old- the need for a very longe buffer space to -tore all updated pages in memory. The advantage of no-torce is that an updated page of a committeel transaction may still be in the buffer when another transaction needs to update it, thus eliminating the 30 cost to read that page again from disk. This may provide a substantial saving in the number of 90 operations when a specific page is updated heavily by multiple transactions.

To permit recovery when in-place updating is used, the uppropriate entries required for recovery must be permanently recorded in the logon disk before changes are applied to the database. For example, consider the following write-ahead logging (WAL) protocol for a recovery algorithm that requires both ONEC and RECE.

- 2. The before image of an item connor be overwritten by its after image in the database on disk ontil all ONEX-type leg records for the updating transferror,—op to this point in time—have been force-written to-disk.
- The commit operation of a transaction connor be completed until all the RENS-type and UNIO-type log records for that transaction have been force-written to disk.

In facilitate the recovery process, the PBMs recovery subsystem near need to mathtam a mudier of lists related to the transactions being processed in the system. These include a five for **active transactions** that have started but not committed as yet, and it may also include has of all **committed** and **aburted transactions** since the list the depoint (see nex) second. Maintaining these lists makes the recovery process more efficient.

### 19.1.4 Checkpoints in the System Log and Fuzzy Checkpointing

Another type of entry in the log is called a **theokpoint**.<sup>1</sup> A [checkpoint] record is written into the log periodically at that point when the system writes out is the database on disk all DBMS buffers that have been modified. As a consequence of this, all transactions that have doen [commit,T] entries in the log before a [checkpoint] entry do not need to have them WRITE operations related on case of a system crash, since all their updates will be recorded in the database on disk during checkpointing.

The recovery manager of a DOMS must decide at what intervals to take a checkpoint. The interval may be measured in time—say, every in minutes, for in the number ( of communed transactions since the last checkpoint, where the values of 0 or ( are system parameters. Taking a checkpoint consists of the fellowing actions.

- 1. Suspend execution of transactions temporarily.
- 2. Force-write all main memory boffers that have been modified to disk.
- 5. Waite a [checkpoint] record to the log- and force-write the log radisk
- Resume executing transactions

 The term checips for his been used to describe more resonance situations or some sestions, such as 961, it has also been used in the hierarure to describe entirely different concepts. As a consequence of step 2, a checkpoint record in the log may also include additional information, such us a list of active storesortion ids, and the locations (addresses) of the first and most recent (lost) records on the log for each active transaction. This can facilitate undoing transaction operations in the event that a transaction must be rolled back.

The time needed to force-write all modified memory buffers may delay transaction processing because of step E. To reduce this delay, it is common to use a technique called fuery checkpointing in practice. In this technique, the system can tesome transaction processing after the [checkpoint] record is written to the log without having to wait for step 2 to finish. However, until step 2 is completed, the previous [checkpoint] record should remain valid. To accomplish this, the system maintains a pointer to the valid checkpoint, which continues to grow to the previous [checkpoint] record is to great to the previous [checkpoint] record step 2 is concluded, that pointer is changed to point to the new checkpoint in the log.

#### 19.1.5 Transaction Rollback

If a transaction toply for wherever reason after updating the database, it may be necessary to roll back the transaction. If any data item values have been changed by the transaction and written to the database, they must be resumed in their previous values (10(Ms). The undatype log entries are used to restore the old values of data neros that must be rolled back.

If a transaction T is rolled back, any transaction S that has in the nucrini, real the value of some data nem X written by T must also be rolled back. Similarly, once S is rolled back, and so on. This preventence is called caseading rollback, and can occur when the recovery protocol ensures workerable schedules but does not ensure some or consideres schedules (see Section 17.4.2). Caseading rollback, and ensure some are quite complex and time-consuming. That is why almost all recovery mechanisms are designed such that caseading rollback is never worker.

Figure 19.1 shows an example where cosciding reffback is required. The read and vente operations of three individual transactions are shown in Figure 19.1a. Figure 19.1b shows the system for at the point of a system crash for a particular execution schedule of these transactions. The values of data items  $A_1(B_1(C))$  and  $D_2$  which are used by the transactions, are shown to the right of the sevem log entries. We assume that the original item values, shown in the first line, are A = 50, B = 15, C = 42, and D = 20. At the point of system balant, transactions  $T_3$  has not reached its conclusion and must be rolled back. The setter operations of  $T_3$ , marked by a single \* in Figure 19.1b, are the  $T_3$  operations that are undorised in original transactions affective during transactions rollback. Figure 19.1b, are the  $T_3$  operations that are undorised in an area of transactions along the time axis.

We must now expect for calcular follows. From Figure 19.1, we see that transaction  $T_1$  reads the value of item B flott was written by transaction  $T_1$ , this can also be determined by examining the log. Because T<sub>1</sub> is rolled back,  $T_2$  must now be rolled back, use. The same coretanons of  $T_2$ , marked by examining the log breaks are independent. Note that only writter item operations need to be undone during transaction rollback; read item operations are recorded in the log only in determine whether case administry relations is necessary.

| (8) | <b>Γ</b>                                  |                              | T <sub>g</sub>                            |         | T <sub>2</sub>                             | ,                                |
|-----|-------------------------------------------|------------------------------|-------------------------------------------|---------|--------------------------------------------|----------------------------------|
| .,  | read_hem(A)<br>read_kem(D)<br>wite_hem(D) | naad<br>wrto<br>veod<br>wrde | dem(B)<br>item(B)<br>_item(D)<br>_item(D) |         | reac_ite<br>wrte, fu<br>read_fi<br>wr4e_ib | 9m(C)<br>9m(B)<br>9m(A)<br>9m(A) |
| (6) |                                           |                              | А<br>30                                   | 8<br>15 | С<br>43                                    | 0<br>20                          |
|     | [star_transaction, T                      | 3                            |                                           |         |                                            |                                  |
|     | $(read_itern, T_0, C)$                    |                              |                                           |         |                                            |                                  |
| •   | (write jitem, 7 <sub>0</sub> , 8,15,      | 12j                          |                                           | 12      |                                            |                                  |
|     | [start_ironsaction, T <sub>2</sub> ]      |                              |                                           |         |                                            |                                  |
|     | $(read\_sec_7, B)$                        |                              |                                           |         |                                            |                                  |
| ••  | [write_wern, 72,8,12,                     | , <b>1</b> 8!                |                                           | 18      |                                            |                                  |
|     | [stert_transaction, 7. ]                  |                              |                                           |         |                                            |                                  |
|     | [read item, T, A]                         |                              |                                           |         |                                            |                                  |
|     | $[read_dem, T_0, O]$                      |                              |                                           |         |                                            |                                  |
|     | wrte_jtem,7.,Ω.20,                        | .25]                         |                                           |         |                                            | 25                               |
|     | [read_item, $T_{p} D $                    |                              |                                           |         |                                            |                                  |
|     | wr.te_itam, $T_{\tilde{e}}$ , 0,25        | .26                          |                                           |         |                                            | 26                               |
|     | [read_item, T <sub>2</sub> , A]           |                              |                                           |         |                                            |                                  |
|     |                                           |                              | +- system                                 | i drau  | sh                                         |                                  |

Ty is relied back hecause it did not leach its commit point.

 $-T_2$  is rolled back because if reads the value of them  $\hat{S}$  withen by  $T_2$ 





In practice, casading rollback of transactions is nerve acquired because practical innovery methods guarantee casualcloss or strict schedules. Hence, there is also no need its record any read item operations in the loc, because these are needed only for determining casualing rollback.

### 19.2 RECOVERY TECHNIQUES BASED ON DEFERRED UPDATE

The idea behind deterred update techniques is to deter or postpone any actual updates to the database until the transaction completes its execution sociessfully and reaches its commit point.<sup>4</sup> During transaction exocution, the updates are recorded only in the lig and in the cache bullets. After the transaction reaches its commit point and the log is for cowritten to disk, the updates are recorded on the database. If a transaction fails before reaching its commit point, there is no need to endo any operations, because the transaction has not affected the database on disk in any way. Although this may simplify recoery, it cannot be used in practice infless transactions are short and each transaction changes few items. For other types of transactions, there is the potential for transme out of buffer space because transaction changes must be held in the cache buffers until the commit point.

We can state a typical deterior apdate pritodol as follows:

- 1. A immogiliar cannot change the database on disk until it reaches its commit peak.
- A transaction does not reach its commut point initial his update operations or recorded in the log axi the log is force-written to disk.

Notice that step 2 of this protocol is a restatement of the perto-ahead longing (wat) protocol. Because the database is never updated on disk until after the transaction oramits, there is never a need to UNDO any operations. Hence, this is known as the 80-CNDO/REDO recovery algorithm. 0000 is mereded in close the system fails after a transaction commute but before all its changes are recorded in the database on disk. In this case, the transaction operations are redone from the log entries.

Usually, the method of recovery from fulnet is closely related to the concurrency control method in andruises storents. First we discuss focio era in single-size systems where no concourtency convril is needed, so that we can inderstand the recovery process independently of any concurrency control method. We then discuss how concurrency control may affect the recovery process.

### 19.2.1 Recovery Using Deferred Update in a Single-User Environment

In such an environment, the recovery algorithm cursible rather simple. The algorithm fue\_s (Recovery using Deferred Update in a Single-aser environment) uses a REFO procedure, given subsequently for redoing contain write\_r tem operations: it works as follows:

PROCEDURE REF\_S: Use two lists of transactions: the committed transactions since the last checkyonal and the networ transactions (at most one transaction will fall in this category, but use the system is single-use). Apply the REPO operation to all the wore the operations of the communed againstations from the log in the order in which they were written to the log. Restair the active transactions.

The REDC procedure is deputed as follows:

**BREACH (WRITE\_OP):** Redound a mitte\_item operation walls or consists of examining is log entry (write\_item, T, X, new\_value] and setting the value of item X in the Jarabuse to new\_value, which is the after image (ARMA).

The 8E x2 operation is required to be idempotent —that is, executing it over and over a squivalent to executing it just once. In fact, the whole recovery process should be idempotent. This is so because, if the system were to fail during the recovery grocess, the recovery attempt might REFO certain write\_iten operations that had already been refore during the first recovery process. The result of recovery from a system crash during rowery should be the same as the result of recovering role is no crash during recovery.

Notice that the only transaction in the active list will have had no effect on the Jatabase because of the deferred update protocol, and it is ignored completely by the recovery process because none of its operations were reflected in the database on data. However, this transaction must now be responde either automatically by the recovery process or manually by the user.

Figure 19.2 shows an example of recovery in a single-user environment, where the test failure occurs during execution of transaction  $T_0$ , as shown in Figure 19.2b. The recovery process will redo the [write iten, T1, D, 26] entry in the log by reserving the abas of item D to 20 (its new value). The (write, T2, ....] entries in the log are ignored by the recovery process because  $T_0$  is not computed. It is second tailore occurs during recovery from the first failure, the same recovery process is repeated from start to finish, with identical results.

7,  $T_{t}$ la l read, itom/8) read\_liert(A) iead\_item(D) wrde dem(Ø) read\_item(O) write\_ilem(Ø) write form(D)(b) [statt\_transaction.7.[ [write\_item, T<sub>1</sub>, 0, 20] [commit 7,7 [starl\_transaction.7<sub>2</sub>] write dem, 7; 9.10 |write\_ittern,T<sub>2</sub>,*O*,26| ← systerii crash

The (write\_item, , ) operations of  $T_1$  are redone.  $T_2$  tog entries are graded by the recovery process

**EGURE 19.2** An example of recovery using deterted update in a single-user environment, (a) The met and write operations of two transactions. (b) The system log at the point of crash.

### 19.2.2 Deferred Update with Concurrent Execution in a Multiuser Environment

For multiuser systems with concurrency control, the recovery process may be more complex. depending on the protocols used for concurrence control. In many cases, the concurrence control and recovery processes are interrelated. In general, the greater the degree of concurrency we walk to achieve, the more time consuming the task of incovery becomes.

Consider a system in which concurrency control owes struct two-phase locking, so the locks on trens remain in effect until the neuronoon works discound point. After that, the locks can be released. This ensures struct and senalitable schedules. Assume that **(checkpoint)** corries are included in the log, a possible recovery algorithm for this case, which we call toolw (Recovery using Deferred Opdate in a Multiuser environment), is given next. This procedure uses the REEO procedure defined earlier.

**PROCESSING** ROU, 9 (WITH CHECKDOINTS): Use two loss of transactions matrituneal by the system: the committed transactions T since the last checkpoint (conmut list), and the active transactions T' (active list), which all the W8110 operations of the committed transactions from the log; in the order in which may use a unitie time the log. The transactions that are active and did not committate officiency cancelul and must be resubmitted.

Figure 19.3 shows a possible schedule of exclusion transactions. When the check point was taken at time  $t_1$  transaction  $T_1$  had committed, whereas transactions  $T_1$  and  $T_2$ had not. Before the system crash in time  $t_2$ ,  $T_1$  and  $T_2$  were committed but not  $T_4$  and  $T_4$ . According to the écol similarity of there is no need to redo the write, item operations of transaction  $T_1$ —or any transactions committed before the last checkpoint time  $t_1$ , write (temoperations of  $T_2$  and  $T_3$  must be redore, however, because both transactions reached



**EIGURE 19.3** An example of recovery in a multiuser environment.

their commut points after the last checkpoint. Recall that the bay is for cowritten before committing a transaction. Transactions  $T_A$  and  $T_a$  are agreed. They are effectively concelled or colled back because non- of their write\_stem operations were recorded in the database under the deterred update protocol. We will refer to Figure 12.3 Later to illustrate other recovery proports.

We can make the NEXUNEX/BE SO recovery algorithm more efficient by noving that if a data irom X has been updated—as indicated in the log entries—more that ance by committed transactions since the last checkpoint, it is only necessary to REDC the last uplate of X from the log during recovery. The other updates would be concurring the this use REDO in any case. In this case, we start from the end of the log, then, wherever on trem o ashene, it is added to a hor of redome items. Before 05100 is upplied to be item, the last is checked: if the items oppears on the list, it is not redome again, since its last value has already been recovered.

It a transaction is aborted for any reason loay by the deadlock detection methods at is simply resubmitted, since it has not changed the database on disk. A drawback of the method described here is that it limits the concurrent execution of transactions because at terms remain torked tontif the metasochim reactes to contrart point. In addition, it may require excessive butter space to hold all opdated items until the transactions contrart. The method's main bencht is that transaction operations need to be notices, for two reasons:

- A transaction does not record any changes in the dat iFase on disk until after it mothes its conunit point. That is, until a completes its execution successfully. Hence, a transaction is never rolled back because of failure during remisaction execution.
- 2 A transaction will never read the value of an item that is written by an uncommitted transaction, because thems mutan for real ontil a transaction reaches its commit point. Hence, no conclude follback will occur.

Figure 19.4 shows an example of recovery for a multiuser system that utilizes the recovery and concarrency control method just described.

#### 19.2.3 Transaction Actions That Do Not Affect the Database

In general, a transaction will have act one that do not affect the database, such as generating and practicely messages in reports from information retrieved from the database. If a transaction fails before completion, we may not wont the user to get these reports, since the transaction has failed to complete, it such connerns reports are produced, part of the recovery process would have to inform the user that these reports are wrong, since the oser may take an action based on these reports that affects the database. Hence, such reports should be generated only after the broadchos reaches do ensure point. A common method of dealing with such actions is to usual the commands that generate the reports have keep them as batch jobs, which are executed only after the transaction reaches its commit point. If the transition table, the batch jobs are conceled.

|      | τ,                                                 | r <sub>2</sub>         | <i>T</i> 9    | т.            |  |  |
|------|----------------------------------------------------|------------------------|---------------|---------------|--|--|
| (a)  | mad jiam(A)                                        | read_itern( <i>B</i> ) | read_rtem(A)  | read_item(\$) |  |  |
| 1-1  | mad_item(D)                                        | vrrte_dem(B)           | wr4e_itemi,4) | write_nem(Ø)  |  |  |
|      | write, item $(D)$                                  | reac_item( <i>D</i> )  | read_lbern(C) | road_nom(A)   |  |  |
|      |                                                    | wnie_ <b>e</b> em(D)   | write_nem(C)  | (A)meh_ethw   |  |  |
| est. | istart impeacie                                    | n T.I                  |               |               |  |  |
| ••;  | pages da backets a pool                            |                        |               |               |  |  |
|      | iverne_eem, rj.c./.c.vj                            |                        |               |               |  |  |
|      | (container, y -)                                   |                        |               |               |  |  |
|      |                                                    |                        |               |               |  |  |
|      | start_transaction, 7_                              |                        |               |               |  |  |
|      | (write), menn, T <sub>4</sub> , <del>0</del> , 16] |                        |               |               |  |  |
|      | write_item.74,A,20                                 |                        |               |               |  |  |
|      | (commit, T_)                                       |                        |               |               |  |  |
|      | (star_transaction, 7 <sub>2</sub> )                |                        |               |               |  |  |
|      | write_item, T <sub>21</sub> B                      | (12)                   |               |               |  |  |
|      | start_transaction                                  | n, 7 <sub>3</sub> ]    |               |               |  |  |
|      | (write_ritem, T <sub>2</sub> , A                   | ,30]                   |               |               |  |  |
|      | (write_item, $T_2, D$                              | 26 + systom crash      | 1             |               |  |  |

### $T_2$ and $T_3$ are ignored because they did not reach their commution rs. $T_4$ is reform because as commit point is after the last system checkpoint.

**FIGURE 19.4** An example of recovery using deterred update with concurrent transactions, (a) the way and write operations of four transactions. (b) System key at the point of crash.

### 19.3 RECOVERY TECHNIQUES BASED ON IMMEDIATE UPDATE

In those techniques, when a transaction issues in update command, the database can be spelated "immediately," without any need to wait for the transaction to reach its commipoint. In these rechniques, however, an update operation must still be recorded in the logtem digk? *lefter* in is applied to the database—using the write ahead logging protocol—so that we can recorded in failure.

Processorys must be made for analong the effect of update operations that have been applied to the database by a failed transaction. This is accomplished by rolling back the transaction and undoing the effect of the transaction's write litter operations. Theorem ally we can distinguish two analy categories of manufaite update algorithms. If the recovery technique ensures that all updates of a transaction are recorded in the database on disk *before* algorithms, there is rever a need to REUC any operations or committed transactions. This is called the UNIXO/NO-REOO recovery algorithms. On the other hand, of the transaction is allowed to commit before all its chappes are written to the database, we have the most general case, known as the UNDO/REDCI recovery algorithm. This is also the most complex technique. Next, we discuss two examples of UNDO/REDCI algorithms and have it as an exercise for the reader to develop the UNDO/REDCI variation. In Section 19.5, we describe a more practical approach known is the ARDES recovery technique.

### 19.3.1 UNDO/REDO Recovery Based on Immediate Update in a Single-User Environment

In a single-user system, if a failure occurs, the executing (acrive) transaction at the time of failure may have recorded some changes in the database. The effect of all such operations must be undone. The recovery absorbant RIC\_S (Recovery using hyperbolic Update in a Single-user environment) uses the REPO procedure defined wither, as well as the REPO procedure defined wither.

#### PROCEDURE RIV\_S

- Use two lists of transactions maintained by the system: the committed trateactions since the last checkgrant and the artive transactions (at most one transaction will fall in this category, because the so-term is stagle oset).
- Undo all the write liters operations of the active transaction from the log, using the user procedure described below.
- Reducthe written itten operations of the commuted transactions from the log, in the order in which they were written in the log, using the sette procedure described earlier.

The USDO procedure is defined as follows

UNDONWRITE\_OP). Undering a **-rice\_iten** operation write, approvises of examininglits log entry [write\_iten, I.X.old\_value.new value] and sering the value of nem X in the database to old\_value which is the base intege (P-Bit). Undering a nonber of write\_iten operations from one or more transactions from the log must process in the receive order from the order in which the operations were written in the log.

### 19.3.2 UNDO/REDO Recovery Based on Immediate Update with Concurrent Execution

When concurrent execution is permitted, the receivery process again depends on the protocols used for conductency courted. The procedure Rugs (Recovery using Immediate Updates for a Multiuser environment) outlines a sectovery algorithm for concurrent transactions with immediate update. Assume that the log includes checkpoints and that the conductency control protocol produces store wheddet—os, for example, the strict twodose locking protocol does. Recall that a strict schedule does not allow a transaction to read or write an item unless the transaction that last wrote the item has committed (or ilsoned and rolled back). However, desidoels can occur in strict two-phase locking, thus requiring abort and CND/2 of transactions. For a since schedule, CNDO of an operation requires choraging the trem back to its old value (PFDC).

#### PROCEEDURE RIL IN

- Use two lists of transactions maintained by the system: the committed transactions since the last checkpoint and the active transactions.
- Undo all the write it ten operations of the scare (uncommuted) transactions, using the UNDO procedure. The operations should be undone in the reverse of me order in which they were sentrem into the log.
- Redcodl the write\_trew operations of the committed transactions from the log in the order machicle they were written into the log.

As we discussed in Section 19.2.2, step 3 is more efficiently done by starting from the end of the log and redoning only the law update of each near X. Whenever an item is redone, it is udded to a lost of redone items and is not redone upper. A similar procedure can be devised to improve the officiency of step 2.

### **19.4 Shadow Paging**

Dus recovery scheme dow not require the ase of a log in a single-user environment. In a multiuser environment, a log may be needed for the concorrence control method. Shadow paging considers the database to be made up of a number of fixed see disk pages (or Jsk blocks)—sig, n = for recovery purposes. A **directory** with a entries? is constructed, where the t<sup>th</sup> entry parts to the iP database page on disk. The directory is kept in main memory for a not too large, and all references—reads or write—to database pages on disk go the tesh if When a transaction begins exercising, the **current directory**—whose entries point to the most recent or current database pages on disk—is copied into a shadow **directory**. The shadow directory is their systed on disk while the current directory is used by the transaction

During transaction execution, the shadow directory is tere modified. When a writeinter-operation is performed, a new copy of the modified database gave is created, but the old copy of that page is not orevention. Instead, the new gave is written elsewhere—or some previously unused disk block. The current directory entry is modified to point to the new disk block, whereas the shadow directory is not multicultand continues to point to the old unmodified disk block. Engage 19.5 illustrates the current's of shadow and current directories. For pages updated by the transaction, two versions are kept. The old version is referenced by the shadow directory, and the new version by the current directory.

To recover from a bolice during transaction execution, it is sufficient to free the modified database pages and to discard the current directory. The state of the database before transaction execution is available through the shadow directory and that state is recovered by constating the shadow directory. The database thus is returned to its state

<sup>5.</sup> The function is similar to the page table monomousling the operating system for each process



FIGURE 19.5 An example of shadow paging

pior to the transaction that was executing when the crash occurred, and any medited pages are discarded. Committing a transaction corresponds to discarding the previous shadow directory. Since receivery involves neither undoing nor redeing data items, this technique can be categorized as a NO4 NDO/NO46FUV rechnique for recovery.

In a multipler environment with concurrent transactions, logs and checkpoints must be incorporated into the shadow paging to bindee. One disadvantage of shadow traging is that the updated database pages change location on disk. This makes it difficult to keep related database pages close together on disk without complex storage transgement strategies. Furthermore, if the directory is large, the overhead of writing shadow directories to disk as iransactions comments significant. A further complex including shadow directories to disk as iransactions comments significant. A further complexitient is how to handle garbage collection when a transaction commits. The old pages referenced by the shadow directory due have been opdated must be infrased and added to a first of free pages for future use. These pages are not eger needed after the transaction commits. Another issue is that the operation to trajente between current and shadow directories must be implemented as on atomic operation.

### 19.5 THE ARIES RECOVERY ALGORITHM

We now describe the ARIES algor thm as an example of a recovery algorithm used in database systems. ARIES uses a steal/no-force approach for writing, and stars based on three concepts. (1) write-physic logging, (2) repeating history during radio, and (5) logging changes during undo. We already discussed writeraboad logging in Section 19.1.3. The second concept, repeating history, means that ARDS will retrace all actions of the database system prior to the trash to reconstruct the database state when the crash accord. Transactions that were uncoromitted at the time of the crash (active transactions) are undone. The fund concept, **logging during undo**, will prevent ARDS from repeating the completed undo operations of a failure occurs during recovery, which causes a restart of the recovery process.

The ARIES recovery procedure consists of three main steps: (1) analysis, (2) REPO and (3) ONDO. The analysis step identifies the dirty (updated) pages in the buffer," and the set of transactions active at the time of the crash. The appropriate point in the losywhere the RFLXD operation should start is also determined. The REDXD phase actually reapples updates from the log to the database. Generally, the REDO operation is applied to only committed transactions. However, in A305, this is not the case. Cleman infoguation in the ARIES log will provide the start point for RE(3.), from which REOO operations are applied until the end of the logits reached. In addition, information stored by ARIES and in the data pages will allow AR-FS to determine whether the operation to be tedone asactually been applied to the dotatiose and heree need not be reapplied. Thus only file necessary REDO oberations are opplied during recovery. Finally, during the UNDO phase, the log is scanned backwards and the operations of transactions that were active at the time of the crash are undone in reverse order. The information needed for ARES to accomplish its recovery procedure includes the log, the Transaction Table, and the Diri-Page Table. In addition, checkpointing is used. These two tables are maintained by the transaction analoger and written to the log during checkpointing.

In ARIES, every log record has an associated log sequence number (LSNF that is monoronically increasing and indicates the address of the log record on disk. Each LSN corresponds to a specific change (action) of some transaction. In addition, each data pace will store the LSN of the latest log record corresponding to a change for ther page. A log record is written for any of the followine actions: opdating a page (write), committing a transaction (opdate), and ending a transaction (cord). The need to including the first three actions in the log bas been discussed, but the last two nord since explanation. When an update is undime, a compensation log record is written in the log. When a transaction ends, whether by committing or aborting, an ending record is written.

Common fields in all log records include: (1) the previous LSN for that transaction, (2) the transaction (0, and (3) the type of log record. The previous USN is important because it links the log records (in reverse order) (of each transaction. For an update (write) action, additional fields in the log record include. (4) the page (D for the page that includes the item, (5) the letteth of the updated item, (6) its offset from the beginning of the pages (7) the before image of the item, and (8) its after image.

<sup>6.</sup> The actual buffets may be lost during a crash, since they are romain memory. Additional roles sport-d in the signal during the kpointing (Dury Edge Table, Transaction Table) allow Aster resident by the information (see next page).

Besides the log, two rables are needed for efficient recovery: the Transaction Table and the Dirry Page Table, which are majatelined by the transaction manager. When a rash occurs, these rables are rebuilt in the adaptive phase of recovery. The Transaction Table company an entry for each actue consistence, with information such is the manaction (D) manufaction states, and the USN of she most recent log record for the transaction. The Dirty Page Table contains an entry for each dury page in the buffer, which includes the page ID and the USN corresponding to the corbest update to that page.

Checkpointing in ABIES consists of the following: (1) writing a begin\_checkpoint record to the logi (2) writing an end\_checkpoint record to the log, and (3) writing the isN of the begin\_checkpoint record to a special file. This special file is accessed during recovery to locate the last checkpoint information. With the end\_checkpoint record, the contents of both the Transaction Table and Darty Page Table are appended to the end of the log. To reduce the cost, **furzy checkpointing** is used so that the DPMS can continue to execute nansactions during checkpointing (see Section 19,1.4). In addition, the contents of the DPMS cache do not have to be flushed to disk during checkpoint, since the Transaction. Table and Darty Page Table—which are appended to the log on dosk romain the information needed for recovery. Notice that if a crash income during checkpointing, the special the will refer to the previous checkpoint, which is used for iccovery.

After a crash, the AGES recovery manager takes over information from the last checkpoint is first accessed through the special file. The **analysis phase** starts at the **regin meckpoint** recording proceeds rouble end of the log. When the **end checkpoint** record in the Transaction Table and Dirty Page Table are accessed frecall that these tables were written in the log during checkpointing). During analysis, the log record was encountered for a transaction T in the Transaction Table. For instance, if an end log record was encountered for a transaction T in the Transaction Table, the the entry for T is deleted from that table. If some other type of log record is encountered for a transaction T in the Transaction Table, if not already present, and the last LSN field is modified. If the log record corresponds to a charace for page T then an entry would be modified. If the log record corresponds to a charace for page T then an entry would be modified. When the analysis phase is complete, the recessary information for T to C and CNPO has been complete in the tables.

The **REDO phase** follows next. To reduce the amount of unnecessary work, ARIES starts redoing at a point in the log where it knows (for sure) that previous changes to difty pages late already need applied to the database on disk. It can determine this by finding the smallest LSN, M, of all the dirty pages in the Dirty Page Table, which indicates the log pointon where AUES needs to start the REOO phase. Any changes encresponding to a LSN  $\leq$  M. for reduable transactions, must have already been propagated to disk or already been overwritten in the buffer; otherwase, these dirty pages with that LSN would be in the line (and the Dirty Page Table). So, RECO starts at the log record with LSN – *M* and starts forward to the end of the log. For each change recorded in the log, the KEPC algorithm would cent, whether or not the change has to be negatived. For example, if a change recorded in the log pertains to page *P* that is not in the Dirty Page Table, then this change is already on disk and need not be recorded. On the change recorded in the log partains to page *P* and the *P* table. On the change recorded in the log both to the recorded in the log pertains to page *P* and the *P* table. On the change recorded in the log partains to page *P* and the *P* table.

with TSN greater than N, then the change is already present. If norther of these two conditions hold, page *P* is read from disk and the U-N stored on that page, TSN(P), is compared with N. If  $N \leq \text{TSN}(P)$ , then the change has been applied and the page road nor be rewritten to disk.

Once the REXC phase is traished, the dombase is in the exact some that it was in when the crash occurred. The set of active transactions—called the **undo\_set**—Fas been **identified** in the Transaction Table during the analysis phase. Now, the UNDO **phase** proceeds be scanning backward from the end of the log and andoing the appropriate actions. A compensating be record is written for each action that is endone. The UNDO reads backward it the log anal every action of the set of inclusionness in the undo set has been andone. When this is completed, the recovery process is furshed act neutral processing can be grant.

Consider the terrivery example shown in Figure 19.6. There are three transactions:  $T_{12}$ ,  $T_{23}$  and  $T_{33}$ ,  $T_{13}$  updates page C.  $T_{13}$  updates pages B and  $C_{33}$  and  $T_{13}$  updates page A. Figure 19.6 (a) shows the partial contents of the log and  $W_{13}$  shows the partial contents of the log and  $W_{23}$  shows the partial contents of the log and  $W_{23}$  shows the partial contents of the log and  $W_{23}$  shows the partial contents of the log and  $W_{23}$  shows the partial contents of the log and  $W_{23}$  shows the partial contents of the log and  $W_{23}$  shows the partial contents of the log and  $W_{23}$  shows the partial contents of the log and  $W_{23}$  shows the partial contents of the log and  $W_{23}$  shows the partial contents of the log and  $W_{23}$  shows the partial contents of the log and  $W_{23}$  shows the partial contents of the log and  $W_{23}$  shows the partial contents of the log and  $W_{23}$  shows the partial contents of the log and  $W_{23}$  shows the partial contents of the log and  $W_{23}$  shows the partial contents of the log and  $W_{23}$  shows the partial contents of the log and  $W_{23}$  shows the partial contents of the log and  $W_{23}$  shows the partial contents of the log and  $W_{23}$  shows the partial contents of the log and  $W_{23}$  shows the log and  $W_{23}$  shows the partial contents of the log and  $W_{23}$  shows the log and  $W_{$ 

| (B) |                   |              |           |             |                  |                   |  |
|-----|-------------------|--------------|-----------|-------------|------------------|-------------------|--|
|     | LSN               | LAST (LSI    | N TRAN JI | D TYPE      | PAGE ID C        | OTHER INFORMATION |  |
|     | 1                 | э            |           | update      | ·                | <u> </u>          |  |
|     | 2                 | 9            | T2        | update      | e                |                   |  |
|     | 3                 | 1            | TI        | commil      |                  |                   |  |
|     | 4                 | begin chockp | vint      |             |                  |                   |  |
|     | 5                 | end checkpo  | arti      |             |                  |                   |  |
|     | 6                 | 0            | ТЭ        | update      | A                |                   |  |
|     | 7                 | 5            | 72        | opdata      | с                |                   |  |
|     | 8                 | 7            | т2        | commit      |                  |                   |  |
|     |                   |              |           |             |                  |                   |  |
| (6) | TRANSACTION TABLE |              |           |             | DIRTY PAGE TABLE |                   |  |
|     | TPAN              | SACTION ID   | LASTLSN   | STATUS      | PAGE ID          | LSN               |  |
|     |                   | 74           | 3         | commit      | С                | 1                 |  |
|     |                   | 72           | 2         | in progress | B                | 2                 |  |
| (2) | TRANSACTION FABLE |              |           | DIRTY PA    | GEITABLE         |                   |  |
|     | TRAN              | SACTION (D   | LASTLSN   | STATUS      | PAGE 10          | LSN               |  |
|     |                   | <br>T1       | 3         | timmoo      | С                | · ·               |  |
|     |                   | Τ2           | R         | commu       | ÷                | 2                 |  |
|     |                   | тэ           | 6         | ID DYDGRESS | A                | <b>F</b>          |  |

**FIGURE 19.6** An example or recovery in ARIES, (a) The log at point of crash, (b) The Transaction and Dirty Page Tables at time of checkpoint, (c) The Transaction and Dirty Page Tables after the analysis phase.

Since a checkpoint has occurred, the address of the associated hegin, checkpoint record promoved which is location 4. The analysis phase starts from location Huntri it reaches the end. The end pheckpoint record would contain the Transaction Table and Dirty fore table in Figure 19.6b, and the analysis phase will further reconstruct these tables. When the analysis phase encounters log record  $b_{1,0}$  new entry for transaction  $T_{0,0}$  is made to the Transaction Table and a new entry for page Alls made in the Dirty Poge table. After log record 8 as analyted, the starts of transaction  $T_{1,0}$  is changed to committed in the Transaction Table. Ingite 19.5b shows the two tables after the analysis phase.

For the REEX phase, the smallest ESN in the Duty Page rable is 1. Hence the REDX will start at log record 1 and proceed with the REEX; or updates. The ESNs {I, Z, 6, 7} corresponding to the inplates for pages *C*, B, A, and C, respectively, are not less than the ESNs of these pages (as shown in the Duty Page table). So these data pages will be read igan and the updates reapplied from the log (assuming the actual ESNs stored in these insight and the UNDX phase is trushed and the UNDX phase starts. From the Transaction Table (Figure 19.6c), ONDX is applied only to the actual ESNs stored in the Last pages only to the actual table). The UNDX phase is trushed and the UNDX phase starts. From the Transaction Table (Figure 19.6c), ONDX is applied only to the actual proceeds backward in the log. The backward chart of applies for transaction *T*, (only log record)) in this example) is followed and undotes:

### 19.6 RECOVERY IN MULTIDATABASE SYSTEMS

So far, we have implicitly assumed that a number on accesses a single database. In some cases a single transaction, called a multidatabase transaction, may require access to multiple databases. These databases may even be stored on different types of DBMSs for example, some DBMss may be relational, whereas others are object-oriented, literarchical, or intwork DBMSs, hi such a case, each PTMS involved in the multidatabase transaction those of the object DBMSs. This situation is somewhat similar to the case of a distributed database man agricent system (see Chapter 25), where parts of the database reside at different sites that are connected by a communication nervork.

To maintain the atomicity of a multidatabase transformation, it is necessary to have a two-level recovery mechanism. A global recovery manager, or coordinator, is needed to maintain information needed for recovery, in addition to the local recovery managers and the information they maintain (log, tables). The coordinator usually toflows a protocol called the two-phase commit protocol, whose two phases can be stated as follows:

Phase 1: When all participating databases signal the coordinator that the part of the multidatabase transaction involving each has concluded, the coordinator sends a message "prepare for commut" to each participant to get ready for commuting the transaction. Each participant gatabase receiving that message will force write all lig records and received automation for local recovery midisk and then send a "mady to commut" or "OK" signal to the coordinator. If the force senting to disk fulls or the local transaction comput" or "OK" signal to the coordinator. If the force senting database sends a "caybor commut" or "OK" signal to the coordinator. If the condition during a "caybor commut" or "Not OK" signal to the coordinator. If the coordinator does not "caybor commut" or "not OK" signal to the coordinator. If the coordinator does not "caybor commut" or "not OK" signal to the coordinator.

reverve a reply from a database within a certain time out arterval, it assumes a "not OK" response.

• Phase 2: If all paracipating databases reply "OK." and the coordinator's vote is also "OK," the transformatic successful, and the coordinator sends a "commut" signal for the transaction and information needed for local recovery have been recorded in the lass of the participating databases. Because all the local effects of the transaction and information needed for local recovery have been recorded in the lass of the participating databases, recovery from tail or is now possible. Each participating database completes transaction community writing a locarmid entry for the transaction in the lag and permanently opdating the database of meeded. On the other black, if one or more of the participating databases or the condition have a from OSC response, the transaction has tabled, and the coordinator sends a message to full back," or UNEO the local effect of the transaction to each participating database. This is done by updating the transaction operations, using the log.

The net effect of the two-phase commit protocol is that either all participating databases commit the effect of the transaction or none of their do. In case and of the participants—on the coordinator—fails at its always possible to recover to a state where either the transaction is committed on it is rolled back. A failure during or before Prove cloudly requires the transaction to be rolled back, whereas a tarlore during thase 2 means that a successful transaction can be ease and commit.

### 19.7 DATABASE BACKUP AND RECOVERY FROM CATASTROPHIC FAILURES

So far, all the rechtigues we have discussed apply to noncaravtrophic failures. A key assumption has been that the sevent log is maintained on the disk and is not lost as a neight of the failure. Similarly the shadow directory must be stored on disk to allow recoery when shadow paging is used. The recovery rechtigues we have discussed use the entries in the system log of the shadow directory to recover from failure by bringing the database back to a consistent store.

The recovery manager of a 1990s must also be exampled to handle more catistrophic fideres such as desc creates. The main rechorque used to liandle such crashes is that it **database** backup. The whole database and the log are periodically copied onto a cheap storage mechanissich as magnetic tapes. In case of a catistrophae system failure, the lates backup copy can be reliaded from the top, to the disk, and the system can be restarted.

To avoid losing all the effects of transactions that have been executed since the last backup, it is enstomany to back up the system log at more frequent intervals than fall database backup by periodically copying it to magnetic tupe. The system log is usually substantially smaller than the database riself and hence can be backed up none requently. This users do not lose all transactions they have performed since the last database backup. All communed transactions recorded in the perturn of the system log that has been backed up to tape our have their effect on the database redone. A new log is stand ofer each database backup. Hence, to recover from dalk future, the database is nest recreated on disk from its latest backup copy on tape. Following that, the effects of all the continuted transactions whose operations have been recorded in the backed-up copies of the system log are reconstructed.

### 19.8 SUMMARY

in this chapter we discussed the techniques for recovery from transaction failures. The main goal of recovery is to ensure the aroundity property of a transaction. If a transaction tails before completing its execution, the recovery incubanism has to make sure that the transaction has no lasting effects on the database. We first gave an informal outfine for a accovery process and then discussed system concepts for recovery. These uncluded a discussion of caching, mightive inplating versus diadowing, before and after images of a data item, (2003) versus (C-D) recovery operations, again outers, and force/no-force policies, system checkpointing, and the write, thead logging protocol.

Next we docusard two djiferran approaches to recovery: deferred update and numericate update. Deferred update techniques postgone any actual updating of the database on disk until a manuaction reaches its contrait point. The transaction forces writes the log to disk bridge recording the updates in the dutabase. This approach, when used with certain concerning, control therbody, is designed rever to require transaction rollback, and recovery simply consists of reduing the operations of transactional committed after the last checkpoint from the log. The disadvantage is that too much histor space may be needed, since apdates are kept in the buffets and are not opplied to disk until a transaction commits. Deterred and are an lead to a recovery algorithm known as NO-UNDO/REDO. Immediate update rechniques may apply changes to the database on disk before the transportion reaches a successful conclusion. Any changes replied to the shobbe must first be recorded in the log and force-written to disk so that these operations can be undone if necessary. We also give an overview of a recovery algorithm for immediate update known as UNIN/RUN). Another algorithm, known as UNIO/NO-EEOC can also be developed for immediate update if all transaction actions, are recorded in the database before commit-

We discussed the shadow paging technique for tecovery, which keeps mark of oul catabase pages by using a shadow direction. This technique, which is classified as NO-UNDO/NO-REPO, does not require a log in single-user systems but still needs the log for indicator systems. We also presented ARDS's a specific recovery scheme used in some of 150fs relational database products. We then discussed the two-phase commit protocol, which is used for recovery from failures involving fundidatabase transactions. Finally, we discussed recovery from catasteephic failures, which is typically done by bucking up the database and the log to tapp. The log can be backed up more trequently than the database, and the backup log can be used to rado operations starting from the last database backup.

### **Review Questions**

- 19.1. Oscues the different types of transaction failures. What is meant by consett-phic failure?
- 19.2 Discuss the action- raken by the read\_titem and write\_item operations on a database
- 19.5 (Reacto Jwar Chapter 77) What is the system log used for? What are the typical kinds of entries in a system log? What are checkpoints, and why are they important? What are transaction commit points, and why are they important?
- 1974. How are buffering and eaching techniques used by the recovery subsystem?
- 19.5 What are the before image (BEM) and after image (AFBM) of a data item? What is the difference between in-place updating and shadowing, with respect to their headbing of BEIM and AE.M?
- 19.6 What are ONDO-type and REDO-type log entries?
- 19.7 Describe the write-ahead logging protocol.
- 1968. Identity three typical lists of transactious that are maintained by the recovery subsystem.
- 19.9. What is meant by transaction rollback? What is meant by cascading rollback? Why do practical recovery methods are protocols that do not permit cascading rollback? Which recovery techniques do not require any rollback?
- 19.12. Discuss the UNDO and REOCO perations and the recovery techniques that use each.
- 19.11. Discuss the deferred update technique of recovery. What are the advantages and disadvantages of this technique? Why is at called the NOR SDO/RECO method?
- 19.12. How can recovery handle transaction operations that do not affect the database, such as the princing of reports by a transaction?
- 19.13. Oncors the innocluste update recovery rechnique in both single-user and nultiuser environments. What are the advantages and disidvantages of immediate update?
- 19.14. What is the difference between the UNDC/REDU and the UNDC/NC-DEDU algorithms for recovery with immediate update? Develop the outline for in UNDC/NC-REDU algorithm.
- 19.15 Describe the shadow paging recovery technique. Under what circonistances does it not require a log?
- 19.16. Describe the three phases of the ARISS recovery method.
- 19.17. What the log sequence nonibers (CENA) in ARIES' How are they used? What intorneation does the Dirty Page Table and Transaction Table contain? Describe how tury checkpointing is used in ARIES.
- 19/15 What do the terms steal/no-steal and force/no-torce mean with regard to fulter management for monsultion processing.
- 19/19 Describe the two-phase commit protocol for inultidatabase transactions.
- 19.20 Differs how recovery from datastrophic failures is handled

#### Exercises

- 19.21. Suppose shot the system evalues before the (read, iter, T3, A) entry is written to the log in Figure 19.25. Will that make any difference in the recovery process?
- 19.37. Suppose that the system crushes before the [write liter.72.0.25.26] entry is written to the log in Figure 19.16. Will that make any difference in the recovery process?
- 19.23. Figure 19.3 shows the log corresponding to a particular schedule at the point of a system crash for four transactions  $T_1$ ,  $T_2$ ,  $T_3$ , and  $T_4$ . Suppose that we use the innedene applied poince's with checkpointing. Describe the recovery process then the system crash. Specify which transactions are talled back, which operations in the log, we redone and which (if any) are undotte, and whether any cosciding rollback takes place.
- 19.4. Suppose that we use the deterted update protocol for the example in Equite 19.7. Show how the log would be different in the case of defetted update by removing the unnecessary log entries; then describe the recovery process, using your modified log. Assume that only RUNO operations are applied, and specify which operations in the log are recipie and which are ignored.
- 19.25. How does checkpointing in ARIES differ from checkpointing as described in Section 19.1 43
- 19.26. How are log sequence numbers used by AR09 to reduce the ancount of REDO work needed for recovery? Illustrate with an example using the information shown in Figure 19.6. You can make your own assumptions to when a yage is written to disk.

[starl\_transaction, 7, ] (read utern, F., A) head litein, T<sub>1</sub>, D1 write item, **J**.,**D** 20, 25) kommut, Tyl (checkpoint) start transaction, T/1 [read\_)(term, T<sub>2</sub>, *B*] [while\_)tem, T\_2.6.12.18] 'starl\_transection, 741 jead item, J<sub>4</sub>, O (wrns\_ilem.T, D. 25.16) (start transaction,  $T_0$ ) [wrde\_item.T.,C. 20.40] [read\_item,  $T_a$ , A]. (write\_item, T, A, 30, 23) Joomme, T<sub>al</sub>i [read\_item  $T_2[D]$ ] [write item, **F**: D,15, 25] - system crash

BGURE 19.7 An example schedule and its corresponding log

19.27. What implications would a mestical/lorde buffer management police have or checkpointing and recovery?

Chuese the correct answer for each of the following multiple unoice questions:

- 19.26. Incremental lagging with deterred updates implies that the recovery system must necessarily.
  - or store the old value of the updated from in the log-
  - b. store the new value of the updated item in the log-
  - constore both the old and new value of the updated item in the log-
  - d store only the Begin Transaction and Commit Transaction records in the log-
- 19/29. The write ahead logging (WAD) protocol simply means that
  - a. The writing of a data item should be done alread of any logging operation.
  - Is the log record for an operation should be written before the actual data is written.
  - c. all log records should be written before a new transaction begins execution
  - d, the log never needs to be written to disk.
- b) In case of transaction failure under a differred update incremental logging scheme, which of the following will be needed.
  - an undo operation.
  - b. a rodo operation.
  - e an undo and redo operation.
  - d i none of the above.
- 19.51 For incremental logging with inmediate updates, a log record for a transaction would contains.
  - a a transaction nume, dota item names old value of 0 cm, now value of item
  - b. a transaction name, data item name, old value of item.
  - c. a transaction many, dota item name, new value of item.
  - de la transaction manie and a data item name.
- 19.32. For current behavior during recovery, and card redo operations must be
  - a. cammulative.
  - b. associative.
  - e. idempotent.
  - d. distriftative.
- 19.53 When a failure occurs, the log is consulted and each operation is either undone or redone. This is a problem because.
  - a seatchine the entire log is time consuming
  - b many redok are unnecessary.
  - bish (a) and (b)
  - d mone of the above.
- 19.34. When using a log based recovery scheme, it aught improve performance as well as providing a recovery mechanism by
  - all writing the log records to disk when each manaction commits.
  - be writing the appropriate log records to disk during the transaction's execution.
  - warring to write the log records and multiple transiences commit and whiing them as a burdi.
  - d, never writing the log records to disk.

- 1935. There is a possibility of a cuscadine follback when
  - a transaction writes items that have seen written only by a committed transaction
  - a transaction writes on item that is previously written by an uncommitted transaction
  - a transaction reads an item that is previously written by an encommitted transaction
  - d. Forth (b) and (c).
- 19.36. To cope with media (disk) (influeed at is meto-say
  - a. for the DBMS to only execute transactions in a single user environment.
  - h. to keep a redundant copy of the database.
  - remever abore a transaction.
  - all of the above.
- 19.37. If the shadowing approach is used for fluctung a data item back realisk, then
  - a the irem is written roadisk only after the transaction commuts.
  - b. the item is written to a different location on disk-
  - control remains written to disk before the transaction commute-
  - d. the trem is written to the same disk location from which it was read.

#### Selected Bibliography

The books by Bernstein et al. (1957) and Phpadimitrica (1986) are decored to the theory and principles of concurrency control and recovery. The book by Cirty and Benter (1994) is an encyclopedic work on concurrency control, recovery, and other transaction-processing reces.

Verbeistud (1978) presents a rutorial and survey of recovery techniques in database systems. Categorizing algorithms based on their ONDO/GEDO characteristics is discussed in Haerder and Reuter (1983) and in Burnstein et al. (1983). Gray (1978) discusses recovery, along with other system aspects of implementing operating systems for databases. The dadow poping technique is discussed in Lonie (1977). Verbofstau (1978), and Reuter (1980). Gray et al. (1981) discuss the recovery inecharities in Statistical Lockeman and Knutsen (1986). Drives (1972), and Pjork (1973) are early papers that discuss the recovery. Chardy et al. (1975) discuss transaction collback. Liben and Bhargma (1985) docuss the corresponding et al. (1985) discuss the recovery integrity block and its use to improve the efficiency of recovery.

Recovery using write-tahead logging is analyted in Hungam and Kheilkar (1992) and is used in the ABIES system (Mohan et al. 1992a). More meant work on measury includes compensating transactions (Korth et al. 1992) and main memory database recovery (Kuniar 1991). The ABIES recovery algorithms (Mohan et al. 1992) have been quite successful in practice. Franklup et al. (1992) discusses recovery in the ESODES system. Two recent books by Kinuar and Flat (1998) and Kumar and Son (1998) discuss recovery in database products.



# OBJECT AND OBJECT-RELATIONAL DATABASES



# Concepts for Object Databases

2()

In this chapter and the next, we discuss object-oriented data models and database systens. Traditional data models and systems, such as relational, circwork, and hierarchical, have been quire successful in developing the database rechroalogy required for many traditional business database applications. However, they have remain shortcomines when more complex database applications must be designed and implemented—for example, databases for engineering design and upon database (CADCAS) and CAST, scientific systemments, telecommunications, scientific information systems, and multimedia.<sup>4</sup> These newer applications have requirements and characteristics that differ from those of rightmonal business applications, such as more complex structures for objects, longerdatation it instructions, new data repes for storing images of large textual items, and the next to define nonstandard application-specific operations. Object-oriented databases were proposed to meet the based- of these more complex applications. The objectoriented approachs offers the flexibility to handle some of these requirements without

<sup>1.</sup> These databases are often referred to us Object Databases and the systems are referred to as Object Database Management Systems (OD5815). However, recease this chapter discusses many sheral object oriented concepts, we will use the term object-oriented instead of just object.

Computer Added TrasguÇC: inputer Added Manufacturing and Computer-Integrated Manufacturing.

<sup>&</sup>lt;sup>3</sup> Malamodia databases must store various types of multimedia objects, such as valso, audzymotos, graphics, and documents (see Compter 24).

being limited by the data types and query languages available in traditional database as term. A key feature of object oriented databases is the power they give the designer to specify both the structure of complex objects and the operations that can be applied to these objects.

Another renson for the creation of object-oriented databases is the increasing use of object-oriented programming languages in developing software applications. Databases are now becoming fundamental components in more software systems, and traditional databases were difficult to use with object-oriented software applications that are developed in or object-oriented programming language such as 0.44. Object-oriented databases are designed so they can be directly—or semilarly—integrated with software that is developed using object-oriented programming languages.

The need for additional data modeling features has used been recognized by relational DRMS vendors, and newer versions of relational systems are incorporating many of the features that were proposed for object oriented databases. This has led to systems that see characterized as object oriented reference DBMSs (see Chapter 22). The latest version of the e.g. semidard for reference DBMSs includes some of these features.

Although many experimental protocopes and commercial object-oriented disabassystems have been created, they have not found widespread use because of the populativ of relational and abject-relational systems. The experimental protocypes included the OREN system developed at Vice <sup>4</sup> OP-NEODE at Texas Inventionary, the PRS system a Hewlert-Packind, aboratories, the CDE system of VIGC Pell Labs<sup>3</sup> and the ENCORY ObServer project of Brown University. Commercially available systems included CENSTUNE/OPAL of Combine Systems, ONION of Ontow, Objectivity of Objectivity Inc. Versitie of Versant Object Technology. OfficerStore of Object Design, ARDENT eff ARDENT Software,<sup>2</sup> and tOET of POET Software. These represent only a partial list of the experimental prototypes and commercial object-outputed database systems that wave created.

As connected object-opented object metadable, the need for a standard model and language was recognized. Because the formal procedure the approval of standards normally takes a multibler of years, a consortium of object-opented 00% vendors and users, called 00% of proposed it standard that is known as the 0.5% (93 standard, which has since been revised. We will describe some forgues of the 0.5% standard in Chapter 21.

Object-oriented databases have adopted many of the concepts that were developed originally for object operated programming languages.<sup>6</sup> In Section 2011, we examine the origins of the object-oriented approach and discuss have a applies to database system. Then, in Sections 2013 through 20.6, we describe the key concepts induced in many object-

<sup>4</sup> Microelectronics and Computer Technology Corporation. Austin. Texas-

Now called Luconi Technologies.

<sup>6.</sup> Entmerty (2) of 62 Technology

<sup>7.</sup> Object Database Managament Oroap.

Similar concepts were also developed as the fields of contrast, data tradeling and knowledge representation.

incrined database systems. Section 20.3 discusses about elevaty, object structure, and type outrus time. Section 20.3 presents the norm pre-of-exceptional of operations and defaultion 4 methods as part of class declarations, and also discusses the mechanisms for storage objects in a database by making them persistent. Section 22.4 describes type and class matches and inheritance in object-oriented databases, and Section 20.5 procedes an overview of the issues that arise when complex objects need to be represented and stored section 20.6 discusses additional concepts, including polycorplusia, quector, aschooling, busine bin flag, multiple and selecting internance, and versioning and congignation of objects.

This chapter presents the general concepts of objectionented databases, whereas (hapter 22 will present the s0000 standard. The reader may skip Sections 20.5 and 20.6 afthis chapter if a less detailed introduction to the topic is desired.

### 20.1 OVERVIEW OF OBJECT-ORIENTED CONCEPTS

The section gives a quick overview of the history and main concepts of object-oriented databases, or COURS for doors. The COUS concepts are then exploined in more detail in Sections 20-2 through 20-6. The term object-oriented sub-securited by O(10) (0) (5) has itorgans in QC programming languages, or COPLs. Today CO concepts are applied in the areas of databases, software engineering, knowledge lose-, artificial intelligence, and conpiter systems in general, COLEs have then mosts in the SAMUA Linguage, which was prepred in the late 1960s. In SIMULA, the concept of a class groups together the internal data structure of an object in a class declaration. Subsequently, researchers proposed the concept of abunch data type, which hides the informal data structures and specifies all posshe external operations that can be applied to an object, leading to the concept of electivsalation. The programming language SMALLTATK, developed at Xerox PARC" in the 1970s, was one of the first languages to explicitly incorporate additional CO concepts. such as message paging and inheritance. It is known as a pize UV programming language, meaning that it was explicitly designed to be object oriented. This contrasts with hybrid octprogramming languages, which incorporate 300 concepts into an already existing langauge. An example of the latter is  $C \rightarrow +$ , which incorporates OOC success into the popular Optogramming language.

An object optically has two components store (value) and behavior (option on all flences or is somewhan similar to a program workle in a programming language except that it will typically have a complex data senience as well as specific optimizers defined be the programmer <sup>16</sup> Objects in an OOTL exist only during program excernion and the bance colled transferriobjects. An OO database can extend the existence of objects without they are stored permanently, and hence the objects perior beyond program termination and on be remeved ther and shared by other programs. In other words, OO database some

<sup>9</sup> Pike Alto Research Center, Palo Alto, California.

<sup>10.</sup> Objects have many other characteristics, as we discuss in the rest of this chapter.
personnal objects permanently on secondary scorage, and allow the sharing of these objects smooth molople programs and applications. This requires the mecoperation of other wellknown features of database management systems, such as indexing mechanisms, concarrency control, and recovery. An exciduations system interfaces with one or more CO programming languages to provide persistent and shared object capabilities.

One goal of OC databases is to maintain a direct correspondence between teolowidd and database objects so that objects do not lose their integrity and identity and car easily be identified and operated upon Trence. OC databases provide a unique system ecterated object identifier (OD) for each object. We can compare this with the relational model where each relation must have a primary key or induce whose value identifies each tuple uniquely in the relational model, if the value of the primary key is changed, the tuple will have a new identity, even though it may still represent the same real-world object. Alternatively, a real-world object may have different manus for key attributes in different relations, making it during to ascertain that the keys represent the same object (for example, the object identifier may by represented as 550 m arother).

Another forture of OC databases is that objects may have an object structure of acting to complexive in order to cont un all of the mecessary information that describes the object. In contrast, in traditional database systems, information about a complex object is earnered over many relations or records, leading to loss of direct correspondence between a real-world object and its database representation.

The internal structure of an object in COD's includes the specification of instance variables, which field the values that define the internal state of the object. Hence, an instance variable is similar to the concept of an attribute in the relational model, except that instance variables may be encapsulated within the object and thus are not increasable visible to external users. Instance variables may also be of actitarily complex data types. Object-oriented systems allow definition of the operations in functions (behaviori that can be applied to objects of a particular type, for fact, some Co-mulels user that all operations a user can apple to an object must be predefined. This forces a complex encopsidance of objects. This rigid approach has been relaxed in most we data models for accept reasons. First, the database user eiten needs to know the intribute names so they can specify selection conditions on the intributes to retrieve specific objects. Second, complete encapsolation implies that any simple retrieval requires a predefined operation, thus making ad bloc direction data the objective on the fly

To encourage encopsolation, an operation is defined in two parts. The first parcalled the signedate or interface of the operation, specifies the operation name and arguments (or parameters). The second part, colled the method or body, specifies the implementation of the operation. Operations can be invoked by passing a message to an object, which includes the operation name and the parameters. The object their excents the method for that operation. This encopsulation permits muchination of the internal structure of an object, as well as the implementation of its operations, without the need to disord, the external programs that invoke these operations. Hence, encopsulation provides a form of data and operation independence (see Chapter 2).

Another key conceptury (X) systems is that of type and cluss hierarchies and unternance. This permits specific ition of new types of closes that inherit much of their structure and/or operations from previously defined types of closes. Hence, specification of object types can proceed systematically. This makes in easier to develop the data types of a system acconentally, and to near existing type detunitions when creating new types of dependent.

One problem in early 30 detabase systems involved representing relationships among abjects. The insistence on complete encapsulation in early 60 data fundels led to the argument that relationships should not be explicitly represented, but should instead be escabed by defining appropriate methods that locate related objects. However, this approach does not work very well for complex databases with many relationships, because it to useful to identify these relationships and make them visible to them. The ODMO standard has recognized this need and it explicitly represents binary relationships via a pair of interest references—that is, by placing the ODMO of related objects within the objects themselves, and nonntaining referential integrity, as we shall describe in Chapter 21.

Some CO-systems private capal three for dealing with and the costons of the same abject—a feature that is essential in design and engineering toplections. For example, an all sension of an object that represents a tested and verified design should be retained intil the new version is tested and verified. A new version of a complex object may include only a few new versions of its component objects, whereas other components include only a few new version to permitting versioning, cO databases should also allow to scheme evolution, which occurs when type decisi ittees are the meed or when new types or relationships are created. These two relations are not specific to CNCPs and should dually be included in all types of Disses.<sup>11</sup>

Another of concept is operator solutioning which refers to an operation's ability to be applied to different types or objects; in such a situation at operation more may refer to several distinct implementations, depending on the type of objects it is applied to. This teature is also called approximate polycophysic. For example, an operation to calculate the creat of a geometric object may differ in its method (implementation), depending on whether the object is of type triangle, circle, or rectangle. This may require the use of the basis of the operation name to the appropriate method at run-time, when the type of coject its which the operation is applied becomes known.

This section provided an overview of the main concepts of 30 databases. In Sections &2 through 20m we discuss these concepts in more doubt.

# 20.2 OBJECT IDENTITY, OBJECT STRUCTURE, AND TYPE CONSTRUCTORS

In this section we first discuss the concept of object identity, and then we present the typwal structuring operations for defining the structure of the state of an object. These studioring operations are notice called **type constructors**. They define head data structuring operations that can be combined to form complex object structures.

To Several schema evolution operations, such as AUT is TABUE, an altendy defond on the relational QU standard (see Section 8.3).

## 20.2.1 Object Identity

An oc) database system provides a unique identity to each independent object stoted in the database. This unique identity is typically implemented via a unique, system-generated object identifier, or OID. The value of an off-its not visible to the external user, but it is used internally by the system to identify cach object implemented to create and manage meeting entropy. The OID can be assigned to program variables of the appropriate typically when needed.

The main property required of an O(1) is that if by **immutable**; that is, the O(1) value of a particular object should not a barge. This preserves the identity of the real world ubject being represented. Hence, an O(1) database system must have some mechanism to genericing O(1) and preserving the immutability property. It is also desirable that each O(1) be used only once: that is, even if an object is removed from the database, its O(1) should not be assigned to morther object. These two properties imply that the O(1) should not depend on and arribure values of the object, since the value of an arribure may be changed or corrected, it is also generally paradited inappropriate to base the O(1) on the physical address of the object in corage, since the physical address of the object in corage, since the physical address of the object in corage, since the physical address of the object home? However, some systems do use the physical address as O(1) to increase the efficiency of object former oddress, which gives the new physical location of the object. It is more common to use long integers as O(2) and then to use some form of high rabit to may the O(1) value to the object in storage.

Some only QC data models repaired that everything if from a simple value to a complex object, the represented as an object bences every basic value, such as an integer, string, or Boolean value, has on QD. This allows two basic values to have different QOs, which can be useful in some cases. For example, the integer value 50 can be used sometimes to mean a weight in knograms and at other times to mean the age of a person. Then, two basic value 50, and to the astrony of a other times to mean the age of a person. Then, two basic value 50, Although useful as a theoretical model, this is not very practical, since it may lead to the generation of two many QUs. Hence, must CO database systems allow for the representation of two many QUs. Hence, must CO database systems allow for the representation of two many QUs. Hence, must CO database systems allow for the representation of two many QUs. Hence, a value is typically stored withes an object and converted from other objects. In some systems, reinplex structured values can also be meased without having a conception QU of the received.

#### 20.2.2 Object Structure

In OO databases, the state (current value) of a complex object may be constructed from other objects (or other values) by using certain type constructors. One formal way of representing such objects is to view each object as a imple (u, v, v), where it is a unique above alconifier (the SO), v is a case constructor if (that is, an indication of how the object state is

(2) This is different read the element operation that is used in C++ and other OCTEs increase new diperts. conventened), and east the object state (in carried mana). The data model will typically include several type constructors. The three most loss constructors are **atom**, **tuple**, and set. Other commonly used constructors include list, **bag**, and **array**. The atom constructor is used to represent all basic atomic values, such as integers, real numbers, character strings, Booleans, and any other basic data types that the system supports directly.

The object state i of an object (i, i, n) is interpreted based on the constructor of the aion, the state (value) are an atomic value from the domain of basic values supported by the state if c = set, the state it is a set of edge t elementers  $\{i_1, i_2, ..., i_n\}$ , which are the OFS for a set of objects that are represely of the same type. If c = ruple, the state a is a tuple of the form  $||s_{i_1}i_{i_1}, a_{i_2}a_{i_2}, ..., a_{i_n}a_{i_n}^{*}||_{s_i}$  where each  $i_i$  is an attribute name<sup>13</sup> and each  $i_i$  is an OFC if c = her, the value a is an indexed by  $[|i_i, i_2, ..., i_n]$  of  $i_i$  Disorbipers of the state  $v_i$  is an OFC if c is similar to a set except that the  $i_i$  is in a list the ordered, and hence we can refer to the first, second, or  $j^{(0)}$  object in a list. For c = array, the state of the object is a singledimensional array of object identifiers. The main difference between array and list is that  $i_i$  is that an arbitrary number of elements whereas an array typically has a maximum  $i_i$ . The difference between set and  $bag^{(1)}$  is that all elements in a set and the distinct whereas a bag can have duplicate elements.

This model of objects allows arbitrary nesting of the set, list, tuple, and other constructors. The state of accobject that is not of type arom wall refer to other objects by this object identificas, blence, the only case where an actual value appears is in *the state after object* of type atom<sup>15</sup>.

The type constructors set, list, array, and bag are called collection types (or bulk types), to distruguish them from basic types and tuple types. The matty characteristic of a collection type is that the state of the object will be a collection of object that may be incodered (such as a set or a bag) or ordered (such as a list or an attract. The **tuple type** constructor is often called a **structured type**, since it corresponds to the struct constructor in the C and CO C programming languages.

EXAMPLE 1: A Complex Object

We now represent some objects from the robotional database shown in Figure 5.6, using the preceding model, where an object is defined by a triple (200), type constructor, state) and the available type consuctors are atom, set, and tuple. We use  $t_1, t_2, t_3, \ldots$  to stand for imple system-generated object identifiers. Consider the following objects:

 $v_1 = (v_1, mont, Houston))$   $v_n = (v_2, mont, Belloire)$  $v_n = (v_2, mont, Sugarland))$ 

Absorblish an assistive network summer on OC rentities wegging

14. Also suffed a moltiser.

15. As we noted outlier in as not practical to generate a unique system identifier for every value, sewal systems allow for both 500% and structural value, which can be structured twitting the same type constances as of perty, except that a value does not face an OD. 
$$\begin{split} & \alpha_{4} = (\alpha_{0}, \operatorname{arom}, \beta) \\ & \alpha_{5} = (\alpha_{0}, \operatorname{arom}, (\operatorname{Restand})) \\ & \alpha_{6} = (\alpha_{6}, \operatorname{arom}, (\operatorname{Restand})) \\ & \alpha_{6} = (\alpha_{6}, \operatorname{arom}, (\operatorname{Restand})) \\ & \alpha_{7} = (\beta_{7}, \operatorname{set}, [i_{1}, i_{2}, i_{3}]) \\ & \alpha_{8} = (\beta_{8}, \operatorname{tople}, < 18.233 \pm i_{5}) \\ & \alpha_{8} = (\beta_{8}, \operatorname{tople}, < 18.233 \pm i_{5}) \\ & \alpha_{8} = (\beta_{5}, \operatorname{tople}, < 18.233 \pm i_{5}) \\ & \alpha_{8} = (\beta_{5}, \operatorname{tople}, < 31.848 \pm i_{5}) \\ & (11.333 \pm i_{5}) \\ & \alpha_{8} = (\beta_{5}, \operatorname{tople}, < 31.848 \pm i_{5}) \\ & \alpha_{8} = (\beta_{1}, \operatorname{set}, [i_{1}, i_{1}, i_{1}]) \\ & \alpha_{1} = (i_{1}, \operatorname{set} \{i_{1}, i_{1}, i_{2}\}) \\ & \alpha_{1} = (i_{1}, \operatorname{set} \{i_{1}, i_{1}, i_{2}\}) \\ & \alpha_{8} = (i_{12}, \operatorname{tople}, < 28.843 \pm i_{5}) \\ & \alpha_{9} = (i_{12}, \operatorname{tople}, < 37.843 \pm i_{5}) \\ & \alpha_{9} = (i_{12}, \operatorname{tople}, < 37.843 \pm i_{5}) \\ \end{array}$$

The test set objects  $\{a_1, a_2\}$  listed here represent atomic values. There will be transsimilar objects, one for each distinct constant atomic value in the database <sup>10</sup> Objective is a set-valued object that represents the set of locations for department 5; the set  $[_{0,1}]$ ed refets to the atomic objects with values [Houston], 'Bell me', 'Sucorland', Objectois a tuple-valued object that represents department 5 itself, and has the attributes toget. DWAMEER, MOST, LOCATIONS, and so OD. The instruction attributes EMAME and EXCABES have atomic objects  $a_3$  and  $a_4$  as then values. The  $\infty$  attribute has a table object  $a_5$  as its value. which inclum has two attributes. The value of the wawZEP attribute is the object whose offers (1), which represents the employee 'Jerm P. Smith' who manages the department. whereas the value of MWAGER\_START EXTENSION of the arcsine object whose value is a date. The value of the (Mamile attribute of a list set object with OL) = a p whose value is the set of object identifiers for the employees who work for the presentation (physers (p. plus)), and opwhich are not shown). Similarly, the value of the brown to attribute of o<sub>2</sub> is a set object with  $000 = t_{00}$  whose value is the set of object identifiers for the projects that are controlled by department number 5 (objects (p. 4)), and growhich are not shown). The object whose GP τ<sub>1</sub>, represents the entrylogue John B. Smith' with all its atornic artributies (fase), site, seer. 559, . . . . Server, that are referencing the originic objects its, its, its, its, its, respect (respectively) also survey to subther references the employee expectively  $(-\infty)$  (i.e. represents 'Lanes F. Borg' who supervises 'John B. Sundy' but as nor shown tand are which references the department object with OD = 1. (this represents department number 5 where 'John B. Snuth' works).

In this nextel, an object can be represented us a graph structure that can be constructed by recursively applying the type constructors. The graph representing an object of can be constructed by first creating a node for the object of itself. The node for ous labeled with the OID and the object constructor of We also create a node in the graph for each basic atomic

<sup>16</sup> These around objects are the cries that any cause a problem ment the use of too many object identitions, if this model is implemented theority.

value, if an object of has an atomic value, we draw a directed are from the node representing  $\gamma$  or the mode representing its basic value. If the object value is constructed, we draw directed area from the object node to a node that represents the constructed value. Figure 10 I shows the graph for the example terms reading to given ratio.

The preceding model permut two types of definitions in a comparison of the seas of on objects for equility. Two objects are said to have identical states (deep equality) of the graphs representing their states are identical in every respect, including the OPS at every well. Another, weaker definition of equality is other two objects have equal states (shallow equality). In this case, the graph structures must be the same, and all the corresponding atomic values in the graphs should also be the same. However, some ourseponding intern d nodes in the two graphs may have abjects with difference OPS.

EXAMPLE 2: Identical Versus Equal Objects

A example can illustrate the difference between the two definitions for comparing object sates for equality. Consider the following objects  $p_1, p_2, p_3, p_4, p_6$  and  $p_6$ 

$$\begin{split} & o_1 = \{i_1, \text{ tuple}_1 \leq a_1 i_2, a_2 i_0 > \} \\ & o_2 = \{i_1, \text{ tuple}_1 \leq a_1 i_2, a_2 i_0 > \} \\ & v_n = \{i_n, \text{ tuple}_1 \leq a_1 i_2, a_2 i_n > \} \\ & v_n = \{i_n, \text{ stom}_1 | 2\} \\ & v_n = \{i_n, \text{ stom}_1 | 2\} \\ & v_n = \{i_n, \text{ stom}_1 | 2\} \\ & v_n = \{i_n, \text{ stom}_1 | 2\} \end{split}$$

The objects  $\sigma_1$  and  $\sigma_2$  have equal states, since their states at the attribut level are the same but the values are reached through distinct objects  $\sigma_1$  and  $\sigma_2$ . However, the states of dejects  $\sigma_1$  and  $\sigma_2$  are identical, even through the objects themselves are not because they have distinct outs. Similarly eithough the states of  $\sigma_1$  and  $\sigma_2$  are identical, the actual objects  $\sigma_1$  and  $\sigma_2$  are identical, even through the states of  $\sigma_2$  and  $\sigma_3$  are identical, the actual objects  $\sigma_1$  and  $\sigma_2$  are identical, the actual objects  $\sigma_1$  and  $\sigma_2$  are identical, the actual objects  $\sigma_1$  and  $\sigma_2$  are equal but not identical. Because they have distinct CHs.

#### 20.2.3 Type Constructors

An object definition language (ODE) fithin incorporates the preording type constructors can be used to define the object types for a particular database application. In Obspice 21, we shall describe the standard CDE of ODMES, but we first introduce the concepts gradually in this section using a simpler catation. The type constructors can be used to define the auto structures for an CO doublase schema. In Section 20.5 we will see how to incorporate the definition of operations for methods) into the CO schema. Figure 20.2 shows how we may declare Employee and Department types corresponding to the object instances shown.

 $<sup>\</sup>widetilde{\Gamma}$  . This would correspond to the 100 (Data Definition lunguage) of the database system (see Output 2)



FIGURE 20.1 Representation of a newsream complex object as a graph

in Figure 20.1. In Figure 20.2, the Date type is defined by a tuple rather than an atomic value as in Figure 20.1. We use the keywords tuple, set, and list for the type construction and the available standard data types (integer, string, float, and so on) for iterate types.

| define type E  | mployee:    |                            |
|----------------|-------------|----------------------------|
| 1ugpte ;       | Iname:      | string,                    |
|                | mana)       | cher:                      |
|                | hame:       | shing;                     |
|                | 384         | stnog;                     |
|                | birthdate:  | Date;                      |
|                | address.    | 61/IP-0.                   |
|                | SARX'       | char;                      |
|                | salary.     | 1080.                      |
|                | sapervisor: | Employee;                  |
|                | depi.       | Dapariment. ).             |
| define type D  | late        |                            |
| luple !        | year:       | integer.                   |
|                | mpolh;      | integer.                   |
|                | day:        | inleger. ):                |
| define type [3 | eparment    | ÷ .                        |
| tuple (        | driame.     | ebing.                     |
|                | dnumber:    | integer                    |
|                | .IIQI :     | tuple ; manager: Employee. |
|                |             | stancare, Data; ).         |
|                | ocations:   | set(string):               |
|                | employees;  | set(Employee).             |
|                | projects    | set(Project) );            |

**FIGURE 20.2** Specifying the object types Employee, Date, and Department using type constructors.

Attributes that refer to other objects—such as dept of Employee or projects of Department—are basically references to other objects and hence serve to represent clarachips among the object types. For example, the attribute dept of Employee is of type Department, and hence is used to refer to a specific Department object (where the Employee works). The value of such an attribute would be an OD for a specific Department object. A binary relationship can be represented in one direction, or it can have an increasingle and example, the attribute employees of Department has as us value a set of operative. The cample, the attribute employees of Department has as us value a set of operates (that is, a set of OTN) to objects of type Employees these are the employees whe work for the department. The inverse is the reference attribute dept of Employee. We will set in Chapter 21 how the ONIG standard allows inverses to be explicitly declared as relationship attributes to ensure that inverse references are consistent.

# 20.3 ENCAPSULATION OF OPERATIONS, METHODS, AND PERSISTENCE

The concept of campulation is one of the main characteristics of CC languages and systens. It is also related to the concepts of abstract data types and aparmation halog in programming languages. In traditional database models and systems, this concept way not appared since it is customary to make the structure of database objects stable to users and external programs. In these traditional models, a number of standard database operations are applicable to objects of all types. For example, in the relational model, the operations for selecting, inserting, deleting, and modifying toples are generic and may be applied to any relation in the database. The relation and its articibates are visible to users and to external programs that access the relation by using these operations.

#### 20.3.1 Specifying Object Behavior via Class Operations

The concepts of information biding and encaptulation can be opplied to database objects. The main idea is to denie the **behavior** of a type of object based on the **operations** that can be externally applied to objects of that type. The internal structure of the object is bidden, and the object is accessible only through a number of predefined operations form operations may be used to create (insert) or destroy (delete) objects of the object state or to apply some calculations. Still other operations may perform a combination of retrieval calculation, and update. In general, the **implementation** of an operation on be specified in a general-particle programming tangenge that provides flexibility and power in denning the operations.

The external acers of the object are only made as no of the interface of the object type, which defines the name and arguments (parameters) of each operation. The implementation is hidden from the external acers, it includes the definition of the internal caro structures of the object and the implementation of each operations for access these structures. In two terminology, the interface part of each operation is called the signature, and the operation implementation is called a method. Typically, a method is invoked by sending a message to the object to execute the corresponding method. Notice that, as part of executing a method, a subsequent message to another object mabe sent, and this mechanism may be used to return values from the object- to the external environment or to when object is.

For database applications, the requirement that all objects be completely encapsulated is too stringent. One way of relaxing this requirement is to divide the structure of an object into visible and hidden attributes (instance variables). Viable autilities pay be directly accessed for reading by external operators, or by a high-level query language. The hidden applicates of an object are completely encapsulated and car be accessed only through predefined operations. Most computes enroley high-level query languages for accessing visible arributes. In Chapter 21, we will describe the OQL query language that is proposed as a standard query language for (ODIs).

In most coses operations that induce the same of its object are encapsilated. This is a way of defining the update sensitives of the objects, given that in many OO data models, few integrity constraints are predefined in the schema. Each type of object has its integrity constraints programmed only the methods that create, define, and update the objects by explicitly writing code to check for constraint violations and to handle exceptions. In such cases, all update operations are implemented by encapsolated operations. More recently, the OPC for the OPSOF standard, dlows the specification of some common constraints such as keys and inverse relationships (referential integrity) so that the system on automatically enforce these constraints (see Chapter 21).

The term class is often used to refer to an object type definition, along with the Johnmons of the operations for that type.<sup>14</sup> Figure 20.3 shows how the type definitions of Figure 20.2 may be extended with operations to define classes. A number of operations as declated for each class, and the signatere (attenface) of each operation is excluded in the class definition. A method (implementation) for each operations include the object constructor operation, which is used to create a new object, and the destructor operations of electron operations are destruction operation, which is used to create a new object, and the destructor operations can

| cerine class En | nployee.                |                            |    |
|-----------------|-------------------------|----------------------------|----|
| type tuple;     | Iname:                  | string.                    |    |
|                 | mnil.                   | char:                      |    |
|                 | Iname:                  | string:                    |    |
|                 | \$\$0.                  | string;                    |    |
|                 | birthdater              | Uete.                      |    |
|                 | address.                | sinng;                     |    |
|                 | sex                     | char                       |    |
|                 | salary:                 | fice                       |    |
|                 | Supervisor:             | Employee                   |    |
|                 | dept                    | Department ::              |    |
| operations      | ape.                    | Integer.                   |    |
|                 | create_emp.             | Employee;                  |    |
|                 | destray emp             | boolean:                   |    |
| end Employee;   |                         |                            |    |
|                 |                         |                            |    |
| deline cleas De | padmeni                 |                            |    |
| type tuple(     | driame.                 | etring.                    |    |
|                 | dnumber:                | integer                    |    |
|                 | nıgr:                   | tuple ( manager, Employee, |    |
|                 |                         | staridate: Date.           | Į. |
|                 | la:::ihar5.             | setisbring).               |    |
|                 | employees:              | eet(Employee);             |    |
|                 | projects!               | set(Project): );           |    |
| operatione      | ne of emps:             | Integer:                   |    |
|                 | creale_dépi.            | Dopartment;                |    |
|                 | desyoy depl:            | boolean:                   |    |
|                 | assign_eirple. Employ   | ree): boolean;             |    |
| (* edi          | ds an employee to the d | lepanment ")               |    |
|                 | remove_emple. Emplo     | yee): booleen              |    |
| ('rer           | huves an employee from  | n me department ")         |    |
| end Department  |                         |                            |    |

FIGURE 20.3 Adding operations to the definitions of Employee and Department

15. This definition of each is similar to how it to used in the popular C++ programming language. The efford standard two- the word integrate in fuldrion review (see Chapter 21). In the PPR model, he form Aco, was used to refer to an object type, along with the set of all objects of that type face (hapter 4). also be declared to modify the states (values) of various arteiburgs of an object. Additional operations can retrieve information about the object.

An operation is replically applied room object by using the dot instation. For example, if d is a reference to a department object, we can invoke an operation such as no\_of\_empty by writing diso\_of\_empty. Similarly, by writing didestrov\_depty the object referenced by d is destroyed (deleted). The only exception is the constructor operation, which returns a reference to a new Department object. Hence, it is constructor operation, which returns a reference to a new Department object. Hence, it is constructor operation this was not used on Pigure 20  $\lambda^{10}$ . The dot notation, is also used to roter to attributes of an object—tor example, by writing didnamber or diagnetartidate.

## 20.3.2 Specifying Object Persistence via Naming and Reachability

Are CODMMS is often closely coupled with an COPL. The COPL is used to specify the method implementations as well as other application ende. An object is typically accord by some executing application program, by invoking the object constructor operation. Not all objects are meant to be stored permanently in the database. **Transient objects** exist in the executing program and disappear once the program terminates. **Persistent objects** are stored in the database and persist after program termination. The typical mechanisms for making an object persistent are meaning and object persistent are meaning and objecting the database.

The naming mechanism involves giving an object a unique pressnent name through which at can be retrieved by this and other programs. This persistent object name can be given via a specific statement or operation in the program, as illustrated in Figure 20.4. All such names given to objects must be unique within a particular database. Hence, the named persistent objects are used as entry points to the database through which users and applications can start their database access. Obviously, it is not practical to give names to alobjects in a large database that includes thousands of objects, so most objects are race persistent by using the second mechanism, colled reachability. The reachability mechanism works by making the object reachable from some persistent object. An object B is soll tole reachable from an object A if a sequence of references in the object graph lead from object of hence, diog is made persistent, all the objects in Figure 20.1 are reachable from object to hence, diog is made persistent, all the other objects in Figure 20.1 also become persistent

If we first create a named persistent object N, whose state is a set or list of objects of some class C, we can make objects of C persistent by adding them to the set or list, and thus making them reachable from N. Hence, N defines a persistent collection of objects of class C. For example, we can define a class DepartmentSet (see Figure 20.4) whose objects are of type set(Department).<sup>55</sup> Suppose that an object of type DepartmentSet is

<sup>19.</sup> Details names for the constructor and destructor operations exist in the C++ programming baguage. For example, the class Employee, the default constructor many is Employee, and the default distructor many is s Employee. It is also example to use the reserver peration to create new objects.

As we shall see in Chapter 21, the OUVER OFL writes uses sets/Department2 unweal of set(Department).

```
define class DepartmentSelf
 sel/Department).
 0.00
 operations, add deaNd: Department/: boolean,
 (" adds a department to the DepartmentSet object "I
 remove_ceptid: Ospamment): boolean;
 If removes a department from the DepartmentSet object "it
 LieparmentSet:
 create dept set
 destroy_depl_sel.
 boolean.
end GepartmentSer;
erralationt name AllDecargoperts, DepartmentSet:
" AlBepartments is a persistent named object of type DepartmentSet ")
dis treate idept,
" create a new Department Object in the variable d ")
```

, state a few papariment officers, no variable a 1

```
a= AlDepartments.add_dept(c).
("makaid persistent by adding it to the persistent set AllDepartments ").
```

FIGURE 20.4 Creating persistent objects by naming and reachability

ateated, and suppose that it is maned AllDepartments and thus made persistent, as illustrated in Figure 20.4. Any Department object that is added to the set of AllDepartments by using the add dept operation becomes persistent by virtue of its being newhable from AllDepartments. The AllDepartments object is often called the extent of the class Department, as it will hold all persistent objects of type Department. As we shall so in Chapter 21, the CDBB conduct grees the schema designer the option of harming an extent as part of class definition.

Notice the difference between traditional database models and  $\infty$  databases in this respect. In traditional database models, such as the relational model or the FFR model, *all* objects are assumed in the persistent. Hence, when an entry type or class such as FFR model, *all* objects are assumed in the FFR model, in represents both the type decloration for EVECOPE and a proton or of all reduct? ubjects. In the  $\infty$  approach, a class declaration of GPP are average spectrum on all persistent requires for a class of objects. The user must separately define a persistent object of type set(SPPCOVE) or list(LPPCOVE) whose value is the collection of GPP are  $\Sigma^{(1)}$ . This allows transient and persistent objects in this is desired, as illustrated in higher  $\Sigma^{(1)}$ . This allows transient and persistent objects to follow the same type and class declarations of the OPE and the  $\infty$  of  $\Sigma$ . In general, in specific to define several persistent objects in the same class definition, it desired

Some systems, such as (OFF, and quarterly quarter the extent for a class

# 20.4 Type and Class Hierarchies and Inheritance

Another main characteristic of CO database systems is that they allow type hierarchies and inheritance. Type forearchies in databases usually imply a constraint on the extencorresponding to the types in the hierarchy. We first discuss type hierarchies (in Section 20.4.1), and then the constraints on the externs (in Section 20.4.2). We use a deferen-CO model in this section—a model in which ortributes and operations are treated uniformly—since both autobates and operations can be inherited. In Chapter 21, we will discuss the inheritance model of the COMO standard, which differs from the model discussed here.

## 20.4.1 Type Hierarchies and Inheritance

In most database applications, there are numerous objects of the same type or class Hence, OC databases must prevale a capability for classifying objects based on their type as do other database systems. But up OC databases, a further requirement is that the spire permit the domition of new types based on other predemical types, leading to a type (ar class) hierarchy.

Typically, a type is defined by assigning it a type name and then defining a number of attributes (instance variables) and operations (methods) for the type  $^{22}$  in some cases, the attributes and operations are together called *jancaines*, since attributes resemble functions with zero attributes. A function name can be used to refer to the value of an attribute of to refer to the resulting value of an operation (method). In this section, we use the term function to refer to both attributes and operations of an object type, since they are teated similarly in a basic introduction to independent in independence.<sup>23</sup>

A type in its subplext form can be defined by giving it a type name and then bring the names of its visible (paids ) functions. When specifying a type in this section, we see the following format, which does not specify arguments of functions, to simplify the discussion:

```
TYPE_NAME: function, function, ..., function
```

For example, a type that describes characteristics of a PERSON may be defined as follows:

PERSON: Name, Address, Brothdate, Age, SSN

In the PERSON type, the Name, Address, 550, and Birthshite functions can be implemented as stored attributes, whereas the Age function can be implemented as a method that calculate the Age from the value of the Birthshite and the current date.

<sup>22.</sup> In this section, we will use the terms type and class as meaning the same thing—nomely the attributes and operations of some type of object.

We will see in Chapter 21 that types with functions are similar to the interfaces used in Otoxi CO ...

The concept of subtype is useful when the designer or user must create a new type that is sumfar but not identical to an already defined type. The subtype then inherits all the functions of the gredefined type, which we shall call the supertype. For example, appear that we want to define two new types **EPEOPT** and **STODY** as follows.

```
EMPLOYFE: Name, Address, Birthdate, Age, SSN, Salary, HireDate, Seniority
STUDENT: Name, Address, Birthdate, Age, SSN, Major, CPA
```

Since both student and tentovic include all the functions defined for PERSON plus some additional functions of their own, we can declare them to be subtypes or PERSON Each will reherit the previously defined functions of PERSON—mainley. Name, Address, Birthdare, Age, and SSR. For Student, it is only necessary to define the new (local) functions Major and PER, which are not inherited. Presumably, Major can be defined as a stored attribute, whereas use may be implemented as a method that culculates the student's grade point worage by accossing the Grade values that are juternally stored (hidden) within each yound object as private arrelation. For excaver, the Salary and HireDate functions may be stored attributes, whereas Seniority may be a method that colculates Seniority from the value of clireDate.

The idea of defining a type involves defining all of its functions and implementing mention either as articlates or as methods. When a subtype is defined, it can then inherit all of these functions and their implementations. Only functions that are stevific or local to the subtype, and hence are not specified in the supertype, need to be defined and implemented. Therefore, we can declare essure and support as follows:

```
EMPLOYEE subtype-of PERSON: Salary, HireOate, Seniority
STUDENT subtype-of PERSON: Major, GPA
```

In general, a subtype includes all of the functions that are defined for its supertype plus some additional functions that are specific only to the subtype. Hence, it is possible to generate a **type hierarchy** to show the supertype/subtype actationships among all the opes declared in the system.

As another example, consider a type that describes objects in plane geometry, which may be defined as follows:

```
GEOMETRY_DBJECT. Shape, Area, ReferencePoint
```

For the store the store three stores in the store of the

```
RECTANGLE subtype-of GEOMETRY_OBJECT: Width. Height
TRIANGLE subtype-of GEOMETRY_OBJECT: Side), SideJ, Angle
CIRCLE subtype-of GEOMETRY_OBJECT. Rad1us
```

Notice that the Area operation may be implemented by a different method for each subtype, since the procedure for area calculation is different for rectangles.

triangles, and circles. Similarly, the attribute ReferencePoint may have a differenmeaning for each subrype: it might be the center point for stational and circle objects, and the vertex point between the two given sides for a treasure object. Some of database systems allow the **renaming** of innerited functions in different subtypes to reflect the meaning more closely.

An alternative way of declaring these three subtypes is to specify the value of the Shape attribute as a condition that must be obtivited for objects of each subtype:

RECTANGLE subtype-of GEOMETRY OBJECT (Shape='rectangle'): Width, Height TRIANGLE subtype-of GE(METRY\_OBJECT (Shape='triangle'): Side1, Side2, Angle CIRCLE subtype-of GEOMETRY OBJECT (Shape='circ'e'): Radius

Here, only cooking\_cosect objects whose Shape='rectangle' are of the subtype accreate, and -initiary for the other two subtypes, for this case, all functions of the accestrac\_costcr supertype are inherited by each of the three subtypes, but the value of the Shape attribute is restricted to a specific value for each.

Notice that type definitions describe objects but do not generate objects evident own. They are just declarations of certain types; and as part of that declaration, the implementation of the functions of each type is specified. In a database application, these are many objects of each type. When an object is created, it typically belongs to one or none of these types that have been declated. For example, a circle object is of type crack and upstrangepter (by inheritance). Each object also becomes a member of one or more gensistent collections of objects (or extents), which are used to group together collections of objects that are meaningful to the database application.

#### 20.4.2 Constraints on Extents Corresponding to a Type Hierarchy<sup>24</sup>

In most C61 databases, the collection of objects in an extent has the same type or class. However, this is not a nonessate condition. For example, S65411164 K, a sescalled repeled to language, allows a collection of objects to contain objects of different types. This can also be the case when other non-object-oriented typeless languages, such as USS, are extended with CO concepts. However, since the majority of OO databases support types, we will assume that extents are collections of objects of the same type for the remainder of this section.

It is common in database applications that each type of subtype will have an extent associated with it, which holds the collectron of all per-intent objects of that type of subtype. In this case, the constraint is that every object main extent that corresponds to a subtype must also be a member of the extent that corresponds to its supertype. Some 'O database systems have a predictioned system type feathed the foot class or the **CBFCT** class.

<sup>24.</sup> In the second edition of this book, we used the rule Class Lizzandée to describe these exemiconstraints. Because the word class has non-inductive in meanings, error is used in this edition. This is also more consistent with a PAn transpology (see Cospiler 21).

where extern contains all the objects in the system.<sup>25</sup> Classification then proceeds by assigning objects into additional solvapes that are instantingful to the application, creating a type hierarchy or class hierarchy for the system. All extents for systems and operdefined classes are subsets of the extent corresponding to the class report, directly or indirectly. In the OPAC model (see Chapter 21), the user may or may not specify an extent for each class (type), depending on the application.

In most CO systems, a distinction is made between persistent and manierit objects and collections. A persistent collection holds it collection of objects that is stored termanently in the database and hence can be accessed and shared by multiple programs. A transient collection exists temperarily during the execution of a program but is not kept when the program terminates. For example, a transient collection may be created in a program to hold the result of a query that selects some cojects from a persistent collection, and copies these objects into the transient collection. The program collection can then nampulate the objects in the transient collection, and once the program terminates, the nampulate the objects in the transient collection, and once the program terminates, the nampulate the objects in the transient collection, and once the program terminates, the nampulate collection coases to exist. In general, numerous collections—transient or persistent—may contain objects of the same type.

Notice that the type constructors discussed in Section 20.2 permit the state of one object to be a collection of objects. Hence, collections ubjects whose types are based on the set constructor can define a number of collections—one corresponding to each object. The set-valued objects themselves are members of another collection. This allows for multilevel classification schemes, where an object in one collection has as its state a collection of objects of a different class.

As we shall see in Chapter 21, the GWOO model discugardies between type inheritance—called interface inheritance and denoted by the "" symbol—and the extent inferitance constraint—denoted by the keyword ES (ESO).

# 20.5 COMPLEX OBJECTS

A principal notivation flow led to the development of (X)-systems was the desire to represent complex objects. There are two main types of complex objects: structured and unstructured. A structured complex object is made up of components and is defined by applying the available type constructors recarsively at various levels. An unstructured complex object typically is a data type that requires a large autount of sturage, such as a data type that represents an image of a large textual object.

## 20.5.1 Unstructured Complex Objects and Type Extensibility

An unstructured complex object facility provided by a OBMS permits the storage and remeval of large objects that are needed by the database application. Typical examples of

<sup>(5</sup> This is called using an the ODAR model (see Chapter 21).

such objects are bitmaji (mage and forg text orings (such as documents); they are also known as binary large objects, or BLOBs for short. Character strings are also known as character large objects, or CLOBs for short. These objects are unstructured in the serve that the DPMS does not know what their structure weekenly the application that uses them can interpret their meaning. For example, the application may have functions to deplay an ingage or to search for certain keyboords in a long text string. The objects are considered complex because they require a large area of storage and are not part of the standard data types provided by maditional DPMSs. Because the object size is quite large, a DBMS may also use buffering and caching techniques to whole object is retrieved. The DBMS may also use buffering and caching techniques to prefer be pottions of the object the application program needs to access them.

The DBMS software does not have the exploiting to directly process selection conditions and other operations based on values of these objects, unless the application provides the code to do the comparison operations acceled for the selection, in an OCTENS, this can be accomplished by defining a new abstract data type for the number-preted objects and by providing the nucleods for selecting, comparing, and displaying such objects. For example, consider objects that are two-dimensional bitmap images. Suppose that the application needs to select from a collection of such objects only those that include a corrain pattern. In this case, the user must provide the pattern recognition program as a method on objects of the bitmap type. The OCODE's then retrieves an object from the database and tures the method for pattern recognition on the object includes the required pattern.

Because an OODBMS allows users to croate new types, and because a type includes both structure and operations, we can view an OODBMS os having on extensible type system. We can create libraries of new types by defining their structure and operations including owaplex types. Applications can then use or modely these types, its the latter case by creating subtypes of the types provided in the libraries. However, the 1868 internals must provide the underlying storage and retrieval capabilities for objects that require large amounts of storage so that the operations may be applied efficiently. Man OCROBIES provide for the storage and retrieval of large unstructured objects such as character strings or bit strings, which can be passed "as is" to the opplication program for interpretation. Recently, relational and extended relational DRMSs have also been able to provide such capabilities. Special indexing rechniques are also being developed.

#### 20.5.2 Structured Complex Objects

A structured complex object differs from an unstructured complex object in that the object's structure is defined by repeated application of the type constructors provided by the ORD/08/9. Hence, the object structure is defined and known to the ORD/08/9. As in example, consider the istestment object shown in Figure 20.1. At the first level, the object has a maple structure with six amplitutes: (MAPE, D9 08/9, 90.0). At the first level, the object has a maple structure with six amplitutes: (MAPE, D9 08/9, 90.0) (COSTORS, FOR COTES, and PROFES, However, only two of these amplitutes-mannly, COSPE, and Sourges, For cortex, and PROFES, the other four have complex structure and hence build the second level of the complex object structure. One of these four (96.0) has a topic structure, and the other three (records, protocos, portexus) have set structures. At the third level, for a rest taple value, we have one

basic anticidente (www.comprosted) and one attribute (www.com) that refers to an employee object, which has a tuple structure from a towners set, we have a set of basic values, but for both the evelopies and the products sets, we have sets of tuple-structured objects.

Two types of reference semartnes exist herween a complex object and its comportents ar each level. The first type, which we can call ownership semantics, applies when the subobjects of a complex object are encopediated within the complex object and are hence considered pair of the complex object. The second type, which we can call reference semantics, applies when the components of the complex chiral are theoriely seindependent objects but may be retroduced from the complex object. For example, we may consider the Lowe, DowafR, 968, and IGCATIONS of thebutes to be owned by a (CEARTHEAT, whereas uncortes and executions are references because they reference independent objects. The first upens also referred notis the is /ups of or is containent of relationship, and the second upp is cilled the is-secontared-with relationship, since it describes an equal association between two orderendent objects. The is-rom of relationship (ownership semantics) for constructing complex objects has the property that the component objects are one psolated within the complex inject and are considered run of the internal ideor state. They need not have object identifies and can only be accessed by methods of that object. They are deleted if the pheer itself is defined. On the other hand, referenced components are emisidered as independent objects that can have their own identity and methods. When a complex abject models to access its referenced components, it must do so by invoking the appropriate methods of the components, since they are not encapsulated within the complex object Hence, reference semantics represents relationships among independent objects. In addition, a referenced component object may be referenced by more than one complex object and hence is not automatically deleted when the contriber object is deleted

An OCODESTS should provide storage options for clustering the component objects of complex object together on secondary storage in order to increase the efficiency of operations that access the complex object. In many cases, the object structure is stored on disk pages in an immutipated fashion. When a disk page that includes an object is intraved into memory, the COOPMS can build up the structured complex object from the information on the disk pages, which may refer to additional disk pages that must be terroved. This is known as **complex object assembly.** 

# 20.6 OTHER OBJECTED-ORIENTED CONCEPTS

In this section we give an overview of some additional CO concepts, including polymorplism (operator overloading), tailityle reheritative, selective informance, versioning, and configurations.

## 20.6.1 Polymorphism (Operator Overloading)

Another characteristic of OC systems is that they provide for polymorphism of operanons, which is also known as operator overfinading. This concept allows the same operaas come or symbol to be bound to two or more different holdersectances of the operator. depending on the type of objects to which the operator is applied. A sample example from programming languages can illustrate this concept. In some languages, the operator some bol "+" can mean different things when applied to operatods (objects) of different types. If the operatods of "+" are of type imager, the operation invoked is integer addition. If the operatods of "+" are of type (know port, the operation invoked is loating punt) addition, if the operatods of "+" are of type set, the operation invoked is set union. The compiler can determine which operation to execute based on the types of operators supplied.

In CO databases, a similar situation may occur. We can use the GOVETER OF ST example discussed in Section 20.4 to idlastrate polymorphism<sup>16</sup> in CO databases. Support that we declare concurr\_concur and its subtypes as follows:

```
GEOMETRY_OBJECT. Shape, Area. ReferencePoint
RECTANGLE subtype-of GEOMETRY_OBJECT (Shape='rectangle'): Width,
Height
TRIANGLE subtype-of GEOMETRY_OBJECT (Shape='triangle'): Sidel, Sidel,
Angle
CIRCLE subtype-of GEOMETRY_OBJECT (Shape='tirle'): Radius
```

Here, the function Area is declated for all objects or type coversy\_covert. however, the implementation of the method for Area may differ for each subtype of covertor\_posit. One possibility is to have a general implementation for calculating the area of a generalized covertsy\_aster (for example, by writing a general algorithm to calculate the area of a polygin) and then to rewrite more efficient algorithms to calculate the areas of specific types at geometric objects, such as a circle, a rectangle, a triangle, and so un lattice case, the Area function is overlaaded by different implementations.

The OCT980S must now select the appropriate method for the Area function basil on the type of geometric object to which it is applied. In strongh typed systems, this can be done at compile time, since the object types just by known. This is termed early (cestatic) binding. However, in systems with weak typing or no typing (such as SMALTALK and 1922), the type of the object to which a function is applied may not be known and tomime. In this case, the function must check the type of object at maxime and then invoke the appropriate method. This is often referred to as late for dynamic) binding.

## 20.6.2 Multiple Inheritance and Selective Inheritance

**Multiple inheritance** in a type hier orchy occurs when a certain subtype T is a subtype of away (or more) types and here e inheres the functions fartributes and methods) of home supertypes. For example, we may create a subtype evolveerance\_wwwals that is a subtype of both wwweek and excuses. This leads to the creation of a type lattice rather than a type farenchy. One problem that can accur with multiple inheritance is that the supertypes from which the subtype inherits may have distinct functions of the same name creating.

26. In programming languages, there are several knowly of polynosphasm. The interested reader size referred to the habitographic notes we weeks that include a more thorough Escussion. an ambiguiry. For example, both source and sources may have a function called Salary If the Salary function is implemented by different methods in the sources and excision supertypes, an ambiguity exists as to which of the two is inherited by the subtype encisions we waves. It is possible, however, that both unusees and sources inherit Salary from the some supertype (such as exproses) higher up in the fartice. The general rule is that if a function is inherited from some common supertype, then it is inherited only once. It such a case, there is no ambiguity; the problem only acises it the functions are distinct in the two separitypes.

There are soveral techniques for dealing with ambiguity in multiple inheritance. One solution is to have the system check for unbiguity when the subtype is created, and to let the user explicitly choose which function is to be inherited at this time. Another solution is to use some system default. A third addition is to disallow multiple inheritance aloge ther if name analoguity occurs, custead forcing the user to charge the more of one of the functions judget of the supertypes. Indeed, some (82 systems do not permit multiple inheritance and the functions judget of the supertypes. Indeed, some (82 systems do not permit multiple inheritance at all.

Selective inheritance occurs when a subtype tuberity only some of the functions of a supertype. Other functions are not inherited. In this case, an EVCEPT clause may be used to be the functions in a supertype that are too to be inherited by the subtype. The mechanism of selective inheritance is not repically provided in 100 database systems, but it is used more frequently in artificial intelligence applications.<sup>25</sup>

#### 20.6.3 Versions and Configurations

Many database applications that use CO sections require the existence of several versions of the same object.<sup>24</sup> For example, consider a database application for a software engineering environment that stores various software artifacts, such as design monthes, soarce ease modaley, and conjugation information to describe which modules should be linked together to ionical complex program, and test cases for testing the system. Commandy, montaneous activities are applied to a software system as its requirements evolve. Maintenance usually involves changing some of the design and implementation modules. If the system is already operational, and if one or more of the modules must be changed, the designer should create a new version of each of these modules to nuplement the changes. Similarly, new versions of the rest questions should not be generated to test the new versions of the modules. However, the existing versions should not be descarded and the new versions have been thoroughly involved and approved only then should the new versions replace the older ones.

Notice that there may be more than two versions of an object. For example, consider two programmers working to update the same software module concurrently. In this case, two versions, in addition to the original module, are needed. The programmers can update their own versions of the same software module concurrently. This is often

<sup>(3)</sup> In the OIWAC model, type inherstance refers to inheritance of operations only, not attributes (see Chapter 215).

Versioning is not a problem that is imagine to (NODIS and it can be applied to relational or other types or LBMSs.

referred to as **concurrent engineering**. However, it eventually occurres necessary to marge these two versions together so that the new (hybrid) version can include the changes made by both programmers. During marging, it is also necessary to make site that their changes are comparable. This necessatares creating yer another version of the object; are that is the result of merging the two independently opdated versions.

As can be seen from the preceding discussion, an COORES should be able to store and manage multiple versions of the sone conceptual object. Several systems do provide the capability, by allowing the application to maintain multiple versions of an object and to refer explicitly to particular versions as merided. However, the problem of merging and reconciling changes anade to two difference versions is typically left to the application developers, who know the semantics of the application. Some 1680-s have certain facilities that can compare the two versions with the original object and determine whether any changes made are incompatible, in online to assist with the merging process. Other systems maintain a version graph that shows the relationships among versions. Whenever a version  $r_1$  originates by copying another version  $r_1$  a directed are table or  $r_1$ , and  $r_2$  to  $r_2$ . The version graph can help uses understand the relationships among the various versions and can be used internally by the system to manage the relationships and  $r_2$  and  $r_3$  to  $r_2$ . The version graph can help uses understand the relationships among the various versions and can be used internally by the system to manage the creation and deletion of versions.

When versioning is applied to complex objects, further issues are that must be resolved. A complex object, such as a software system, may consist of many no chies. When versioning is allowed, each of these meduces may have a number of different versions and a version graph. A configuration of the complex object is a collection consisting of one version of each module arranged to such a way that the module versions in the configuration are compatible and together form a while version of the complex object. A new version or configuration of the complex object does not have to unclude new versions for every module. Hence, remain module versions that have not have to unclude new versions for every module. Hence, remain module versions that have not here to unclude new versions for every module. Hence, remain module versions that have not here to unclude new versions for every module. Hence, remain module versions that have not here to unclude they version is a collection of explanation of the complex object. Notice that a configuration is a collection of explanation of the complex object. Notice that a configuration is a collection af explanation of the complex object. A configuration should follow the type structure of a complex object multiple configurations of the same complex object aractilogeus to multiple versions of a component object.

# 20.7 SUMMARY

In this chapter we discussed the concepts of the object concented approach to database setends, which was proposed to neer the mode of complex database applications and to add database functionality to object streamed programming longuages such as C++. We first discussed the main concepts used in OO databases, which include the following

- Object identity: Objects have imagic identities that are independent of their attribute values.
- Type constructors: Complex object structures can be constructed by recursively applying a set of basic constructors, such as tuple, set, list, and bag.

- Encopsidation of operations: Both the object structure and the operations that can be applied to objects are included in the object class definitions.
- Programming longuage compatibility. Both persistent and transient objects are handled seamlessly. Objects are made persistent by being attached to a persistent collection or by explicit naming.
- Type hierarchies and inheritance: Object types can be specified by using a type hierarchy, which allows the inheritance of both attributes and methods of previously defined types. Multiple inheritance is allowed in some models.
- Eatents: All persistent objects of a particular type can be stored in an extent. Extents
  corresponding to a type literarchy have set/subset constraints enforced on them.
- Support for complex objects: Porth structured and unstructured complex objects can be stored and manipulated
- Polymorphism and operator asynfolding: Operations and method names can be overloaded to apply to different object types with different implementations.
- Versioning: Some (x) -v-torns passible support for maintaining arveral versions of the same object.

In the next chapter, we show how some of these concepts are realized in the ODMOsundard

## **Review Questions**

- 2011. What are the origins of the object-oriented approach?
- 20.2. What primary characteristics should an Offoposess?
- 20.3. Discuss the various type constructors. How are they used to create complex object structures?
- 20.4. Discuss the concept of encapsulation, and rell how it is used to create abstract data types.
- 20.5. Explain what the following terms mean in object oriented database terminology: method, signature message, collection, extent
- 10.6. What is the relationship between a type and its subtype its a type hierarchy? What is the constraint that is enforced on extents corresponding to types its the type hierarchy?
- 32.7 What is the difference between persistent and transient objects? How is persistence handled in typical OO database systems?
- 20.8. How do regular adhesistance, multiple inheritance, and selective inheritance. differ?
- 20.9. Diseases the concept of pulymorphism/operator overloading
- 20.10. What is the difference between structured and instructured complex objects?
- 20.11. What is the difference between ownership semantics and reference semantics in structured complex objects?

20.12. What is versioning? Why is it important? What is the difference between versions and configurations?

#### Exercises

- 20.13. Convert the example of sequences given in Section 20.4.1 from the functional notation to the notation given in Figure 20.3 that distinguishes between attributes and operations. Use the keyword INTILRIT to show that one class inherits from another class.
- Compare inheritance in the 608 model (see Chapter 4) to inheritance to the 0.0 insidel described in Section 20.4.
- 20.15. Consider the Gavees in EER schema of Figure 4.10. Think of what operations are needed for the entry types/classes in the schema. Do not consider constructor and destructor operations.
- 20.16 Consider the convex ER schema of Figure 3.2. Think of what operations are needed for the entity types/classes in the schema. Do not consider constructor and destructor operations.

#### Selected Bibliography

Object-oriented database concepts are an amalgam of concepts from GP programming lategaaces and from database systems and conceptual data models. A number of textbooks describe OO programming languages—Kor example. Strockstrop (1986) and Pohl (1991) for C++, and Goldberg (1989) for SMALLTALK. Recent looks by Cattell (1994) and Lausen and Vossen (1997) describe OO database concepts.

There is a vast hibliography in CO databases. So we can only provide a representative sample here. The October 1991 essne of CAUM and the December 1990 essne of /EEF Compater describe object oriented database concepts and systems. Deterds (1986) as/ Zaniolo et al. (1986) survey the basic concepts of object oriented data models. An early paper on object oriented databases is Baroody and DoWirr (1981). So et al. (1988) presents an object-oriented database is Baroody and DoWirr (1981). So et al. (1988) presents an object-oriented data model that is being used in CAD/CAV opplications. Mitschang (1989) extends the relational algebra re-cover complex objects. Query languages and graphical user interfaces for CO are described in Gyssens et al. (1980). Kin (1989), Alashqui et al. (1989), Bertino et al. (1992). Agrawal et al. (1990), and Cru: (1992).

Polynorgham in databases and object-priented programming languages is discussed in Osborn (1989), Atkinson and Bureman (1987), and Danforth and Toulinson (1989). Object identity is discussed in Abitebook and Kaneffales (1989). Co-programming languages for databases are discussed in Kent (1991). Object constraints are discussed in Delcambre et al. (1991) and Elinewi et al. (1993). Authorization and scenity in Codarabases are examined in Rabitti et al. (1991) and Bertino (1992).

Additional references will be given at the end of Chapter 21



# Object Database Standards, Languages, and Design

As we discussed at the buginning of Chapter 8, having a standard for a particular type of Jarabase system is very important, because it provides surjoint for purtability of database applications. Portability is generally defined as the capability to execute a particular application program on different systems with minimal inclutions to the program uself. In the object database field, portability would allow a program written to access one Objort Database Management System (ODBAS) package to access another ODBAS package as long as both packages support the stundard (athtally, "]his is important to Japhase users because they are generally wary of investing in a new technology if the different vendurs do nor adhere to a standard. To illustrate why portability is suportant, suppose that a particular over invests thousands of dollars in crysting on application that runon a particular vendor's product and is then dissortisfied with that product for some reason —say the performance does not meet their requirements. If the per-blation was write ion using the standard language construction is possible for the user to convert the application to a different vendor's product which adheres to the same language stardards 50 may have better performance for that user's applications, without having to do major modifications that require time and a major momenter investment.

I for this chapter, on will use object database ensured of object original database (as in the previous chapter), since this is non-increase commonly, accepted commology.

A second potential advantage of having and adhering to standards is that it helps in achieving interoperability, which generally refers to the ability of an application to access multiple distinct systems. In database terms, this means that the same application program may access some data stored under one ODIVCS package, and other data stored under another package. There are different levels of interoperability. For example, the DWFs could be two distinct DBMS packages of the same type. For example, the DWFs could be two distinct DBMS packages of the same type. For example, two object database systems on they could be two DWFs packages of different types—say one relational DBMs and one object DBMs. A shard advantage of standards is that it glows costonees to wangere expansional products more easily by determining which parts of the standard are supported by each product.

As we discussed in the introduction to Chapter 8, one of the reasons for the success of commercial relational DEMSs is the SQL standard. The lack of a standard for OEBUSS for several years may have caused some protential users to shy away from converting to this new technology Subsequently, a consertion of ODBAS vendors, called ODBO (Object Data Management Usrouph proposed a standard that is known as the CORR-93 or ODREAD standord. This was revised into OPAC 2.0, which we will describe in this chapter. The standard is made up of several parts: the object model, the object definition language (ODL), the object query language (OQL), and the bindings to object-oriented programming languages. Language bindimes have been specified for several object-oriented programming languages including C++, SMALLTAIK, and J WA. Some vendors only other specific language buildings, without offering the full capabilities of ODL and OQL. We will describe the OD95 abject model in Section 21.1, ODL in Section 21.2, OQL in Section 21.3, and the C++ language binding in Section 21.4. Examples of how to use O(3, OO), and the C+1 language building will use the UNIVERSITY database example introduced in Chapter 4. In our description, we will follow the CONSO 2.0 object model as described in Cartell et al. (1997). has important to note that many of the ideas emissibled up the ODSIG object model are based on two decades of research into conceptual modeling and object orjepted doubses by many asseap hers.

Following the description of the OBAG model, we will describe a technique for object database conceptual design in Section 21.5. We will discuss how object-oriented databases differ from relational databases and show how to map a conceptual database design in the EER model to the ODL statements of the ODAG model.

The reader may skip Sections 21.3 through 21.7 if a less detailed introduction to the topic is desired.

## 21.1 OVERVIEW OF THE OBJECT MODEL OF ODMG

The ODMG object model is the data model upon which the object definition language (ODL) and object query language (OQL) are based. In fact, this object model provides the data types, type constructors, and other concepts that can be indiced in the ODL to specify object database schemas. Since, it is meant to privide a standard data model for object-oriented databases, just as SQL describes a standard data model for relational databases. It

2. The earlier version of the object model was published in 1993.

also provides a standard terminology in a field where the same terms were sometimes used to describe different concepts. We will try to adhere in the COMC terminology in this chapter. Many of the concepts in the OBMC model have already been discussed in Chapter 20, and we assume the reader has already gone through Sections 20.1 through 20.5. We will proport our whenever the COMC terminology differs from that used on Chapter 20.

#### 21.1.1 Objects and Literals

Objects and literals are the basic building blocks of the object model. The main difference between the two is that an object has both an object identifier and a state (or current value), whereas a literal has only a value but no object identifier <sup>3</sup> In either case, the value can have a complex structure. The object state can change over time by modifying the object value. A literal is basically a constant value, possibly having a complex structure, that does not change.

An object is described by four characteristics (1) identifier, (2) name, (5) lifetime, and (4) structure. The object identifier is a unique system-wide identifier (or Osnet) [15].\* Every object must have an object identifier, in addition to the GOECL\_LD, serie objects may optionally be given a mique name within a particular database—this name can be used to refer to the object in a program, and the system should be able to locate the raisect given that name.3 Obviously, not all individual objects will have unique names. Typically, a few ubjects, mainly those that hold collections of objects ut a particular object type such as extents— will have a name. These names are used as entry points to the database; that is, by locaring these objects by their unique name, the user can then locare other objects that are referenced from these objects. Other numeration objects in the application may also have upaque names. All such names within a partecular database must be optique The lifetime of an object specifies whether it is a personal object (that is a database object) or travaient object (that is, an object in an executing program that disappears after the program terminates). Finally, the structure of an object specifies how the object is constructed by using the type constructors. The structure specifies whether an object is arome or a callection object? The term atomic object is different than the way we defined the atom construction in Section 20.2.2, and it is quite different from on atomic literal (see below). In the ODMG model, an atomic object is any object that contra collection, so this also covers smemori objects created using the smer constructor.) We will discuscollection objects in Section 21.1.2 and atomic objects in Section 21.1.3. First, we define the concept of a lateral.

In the object model, a literal is a value that does not have an object identifier. However, the value may have a simple or complex structure. There are three types of literals: (1)

<sup>!</sup> We will use the terms is due and quite interchange slife here.

<sup>1</sup> Corresponds to the OD of Chapter 20.

<sup>5</sup> This corresponds to the reasoning mechanism described in Section 20.3

<sup>6</sup> In the Oblics model, submer electric desired correspond to objects whose values are basic data types. All basic values (integers, reals, etc.) are considered to be usingly.

The struct construct corresponds to the tuple constructor of Chapter 23.

atomic, (2) collection, and (3) structured. Aromic literals" correspond to the values of bate data types and are predefined. The basic data types of the object model include long, short, and unstanded integer numbers (these are specified by the keywords long, Short, Unsymed Long, Unsigned Short in CDL), regular and double precisive floaring point numbers (Floar Double), boolean values (Boolean), single characters (Char), character strings (String), are enumeration types (Enum), among others. Structured literals conceptend toughly to values that are constructed using the tuple constructor described in Section 20.3.2. They include Date, Interval, Time, and Timestamp as built in structures (see Figure 21.1b), as well as an additional user-defined type structures as needed by cluch application." User-defined structures are created using the Struct keyword in ODL as in the C and C++ programming languages. Collection literals specify a value that is a collection of objects or values but the collection itself does not have an OFISCT. IS. The collections in the object mudel are SC (T) Beasts, LISTATS, and According, where  $\tau$  is the type of objects or values in the collection.<sup>15</sup> Another collection type is Dictionary estivation which is a collection of associations scive when eights is a key (a unique search value) associated with a value v; this can be used to create on index on a collection of values.

Figure 21.1 gives a simplified new of the basic components of the object model. The notation of 30500 uses the 'keyword interface where we had used the keywords ope and class in Chapter 20. In fact, interface is a more appropriate term, since in describes the interface of types of objects—naturely their disible attributes, relationships, and operations.<sup>10</sup> These interfaces are typically noninstantiable (that is, no objects are created for an interface) but they serve to define operations that can be infertiably the usidefined objects for a potential application. The keyword class in the object model is reserved for user-specified class declarations that form obtabase schema and are need for the full specifications, see Cartell et al. (1997). We will describe the constructs show in Figure 21.1 as we describe the object model.

```
interface Object (
```

ì

```
baolean sams_as(in Object other_object).
Object copy();
void delet().
```

FIGURE 21.1A Overview of the interface definitions for part of the ODMG object model. The basic Object interface, interfa

The structures for Date, Interval, Time, and Timestong can be used to create either lateral values on objects with identifiers.

These are smaller to the corresponding type constructors described in Section 2012.

 Interface is ofter the keyword used in the constant standard (see Section 21.5) and the (AVA pregramming language)

<sup>8.</sup> The use of the word atomic is down: here! does correspond to the way we used atom continuor in Section 22.7.2.

```
interface Date | Object /
 ອດມາກ
 Weekday
 (Sunday, Monday, Tuesday, Wednesday, Thursday, Friday, Saturday);
 eaura
 Month
 (January, February March April May June, July, August, September, October November December)
 unsigned short
 vearu:
 unsigned short
 monih():
 unsigned short.
 day():
 boolean
 is equal/in Date other Date/c
 boolean.
 is preaten in Date other, Datel,
I
iverface Trine | Object {
 unsigned short.
 houry
 unsiened short
 minute!::
 unsigned short
 SECONCO.
 unsigned short.
 infilisecond)).
 bardean
 is equation Time other. Time);
 buclean
 is greater(in Time other Time):
 Trme
 add_interval/in Interval some Intervaly.
 ίπe
 sobhaci_intervation/merval.some_intervati;;
 Interval
 subjract hme(in Time other Tima);
¢
niartade Timestamo - Object (
 unsigned short.
 vear():
 unsigned short.
 intenth();
 unsigned sho1
 day().
 unsigned short
 houri).
 unsigned short
 minufatt
 unsigned short
 second():
 unspred short
 minsecond().
 f mestamp
 plusion interval some_interval).
 Timestamp
 minus(in Interval some_interval),
 boolean
 is equal/in Timeslamp other Timeslamp's
 auolean.
 is greater(in Timestamp other Timestan'p)
۰.
implace Interval : Object :
 unsigned short
 dayi}.
 unsigned short
 neuri),
 unsigned short
 minugeri
 unsigned short
 sacandi).
 uneigned short.
 milisecond().
 plus[in Interval some_Interval)
 interva.
 Interva
 minus(in interval some_interval);
 luterva.
 product(in long some livalue);
 loterys:
 quotient(in long some_value):
 300lean
 is_equation toterval other_imerval).
 rcoman.
 is greater (in Interval other Imercal)
 . . .
÷
```

**FIGURE 21.1B** Overview of the interface definitions for part of the obset: object model. Some standard interfaces for structured literals.

```
interlace Collection : Object {
 exception.
 ElamentNotFound(any element.).
 unsigned long - cardinality().
 boolsan
 -$_9770Y{})
 sociean
 contains_element(in any element);;
 vais.
 //ised_element(in any element);
 vaid.
 removal elemant/in any elemanti-
 raises(Elemen(NotFound):
 create literator(in boolean slable).
 herasor
х
j reterell socialeri
 exception.
 NoMoroElements();
 ntee ood
 is_stablet).
 N66-03d
 al endi).
 how
 reset()
 gel_element() (alses(NoMoreElements);
 алу.
 next_position() raises(NoMoreElements);
 vord.
١.
intertabe Set : Collaction (
 create_vinion(in Set other_set);
 Set
 baolean.
 is_subset_ef(in Set ather_set) -
1.
interface Bag : Collection (
 unsigned long occurrences of (in any element);
 Ban
 create_union(in Bag othe:_bag):
۱.
interface Ust : Collection (
 exception
 lovald_ndex{unsigned_long.mdex;};
 remove_element_attic unsigned long position)
 şeny.
 raises(invalid index)
 any
 retrieve, element, atjus unsigned long position).
 rases(InvaliaIndex).
 vcið
 replace, element, at(in any element, in unsigned long position).
 rarses(invalid/index)
 void
 insert_element_effer(in any element, in unsigned long poshon).
 'asses) ovalidiodes n
 YCID
 insert_element_brst(in any element)
 remove_first_etement() raises(invatidIndex);
 алу
 retneve_trst_exeme(vit) raises(invalid)/idex(;;
 BOW
 ceneal(in List other_list)
 LISI
 VQIE
 appandun Lişt olher Tisti
ì
```

FIGURE 21.1C Overview of the interface definitions for part of the ODMC object model. Interface definitions for collection ubjects.

```
interface Array : Collection I
 exception.
 Invalid_Index(unsigned_long index_)
 алу
 remove_element_at(in unsigned long_index).
 raises(InvandIndex);
 any
 retrieve_etement_at(in unsigned long index).
 raises(InvalidIndex).
 vod
 raplace element, attin unsigned long index, in any element,
 raises[hyaicfncer].
 void
 resizetiin unsigned long new_size).
I:
struct Association (any key; any value;).
meriace Dictionary | Collection {
 exception
 KeyNofFoundtary Vey 1
 v9+0i
 bind(in any key, in any value)
 vaid
 unbind(in any key) raises(KeyNotFound).
 anv
 lookuppin any key (raises) KeyNotFound);
 boolean
 contains_key)in any key):
I.
```

#### FIGURE 21.1C (CONTINUED)

In the object model, all objects inherit the basic interface or Object, shown in logore 21.1a. Hence, the basic operations that are inherited by all objects (from the Object interface) are copy (creates a new copy of the object), deliver (deletes the object), and **same\_as** (compares the object's identity to another object).<sup>15</sup> In general, operations are applied to object using the dot mutation. For example, given an object of the compare it with another object p, we write

o.same\_as(p)

The result returned by this expression is Boolean and would be true if the identity of  $\epsilon$  is the same as that of  $\epsilon$ , and false otherwise. Similarly, to create a copy p of object  $\epsilon$ , we write

p = 0.copy()

An alternative to the dot notation is the arrow notation: o->same\_as(p) or o->copy().

Type inheritance, which is used to define type/kohype relationships, is specified in the object model using the colors (.) notation, as in the C++ programming language. Hence, in Figure 21.1, we can see that all interfaces, such as Collect on, Dute, and Trace, when the basic Object interface. In the object model, there are two main types of objects. (1) collection objects, described in Section 21.1.2, and (2) atomic tand structured) objects, described at Section 21.1.3.

<sup>12.</sup> Additional operations are defined on objects for busing purposes, which are not shown in Figure 21.1. We discuss locking concepts for databases in Chapter 20.

٩

#### 21.1.2 Built-in Interfaces for Collection Objects

Any collection object inherits the basic Collection interface shown in Figure 2116, which shows the operations for all collection objects. Given a collection object 5, the 5, carcinality() operation cours the number of elements in the collection. The operation 5 is sempty() returns true if the collection 5 is empty, and false otherwise. The operations 0, insert\_element(e) and 0, remove\_element(e) insert of remove an element e from the collection 5 includes element e, and returns take otherwise. The operation 5 includes element e, and returns take otherwise. The operation 5 includes element e, and returns take otherwise. The operation 5 includes element e, and returns take otherwise. The operation 5 includes element in the collection. The interface for iterator object 5 is also shown in Figure 21.1c. The inference() operation sets the iterator at the first element() returns () sets the iterator to the most element. The most remove), and innext position() sets the iterator to the element at which the iterator is currently positioned.

The ODEWS object model uses **exceptions** for reporting errors or particular conditions. For example, the ElementNotFound exception in the Collection interface would be raised by the observe element (P) operation if element in the collection of The NoMoreElements exception in the iterator interface would be raised by the innext position() operation if the iterator is currently positioned at the last element in the collection, and hence no more elements exist for the iterator to point to

Collection objects are further specialized into Set. List. Bag. Array. and Dictionary, which where the operations of the Collection interface. A Set «t» object type can be used to create objects such that the value of object **u** is a second one demonstrate of type  $\mathbf{E}$ . The Set interface includes the additional orderion  $\mathbf{p} = \mathbf{p}$ , create unitor(s). (see Figure 21.1c), which returns a new object p of type Set star institution of the tion sets a and s. Other operations similar to create\_union (not shown in Figure 21.1.) are create intersection(s) and create\_difference(s). Operators for set comparison include the n, is\_subset\_uf(s) operation, which terms the d de set object a is a subset of some other set object s, and returns false otherwise. Similar operations (not shown in Figure 24.1c) are is proper subset of (s), is superset. of(s), and is proper superset of(s). The Bagers object type allows during relements in the collection and also inherits the Collection interface. It has three oversuon create\_union(b), create\_intersection(b), and create\_difference(b)—the all remark a new objectivel type Bagkty. For example, p. = p. create union(b) returns a Bas object a that is the union of a and b (keeping duplicates). The procedurences\_af(e) operation returns the number of duplicate occurrences of element e in bag of

A LOSTATA object type inherits the Collection operations and can be used to create collections where the order of the elements is important. The value of each such object a is an ordered fixt idease elements are of type to Hence, we can refer to the first last, and if "element in the list. Also, when we add an element to the list, see must specify the position in the list where the element is inserted. Some of the List operations are shown in Figure 21.1c. If o is an object of type Listata, the operation

a.irsert\_element first(e) (see Figure 21.1c) inserts the element element is before the first element in the list a, so that element last(e). The operation a linsert element\_ inter shown) is a sinsert\_element last(e). The operation a linsert element\_ after(c, i) in Figure 21.1c inserts the element element evists in a A similar operation intershown) is a insert\_element\_before(e, i). To remove elements from the list, the operations in element\_before(e, i). To remove elements from the list, the operations in element\_at(i); these operations remove the indicated element from the list and return the element as the operation's result. Other operations retrieve an element without removing it from the list. These are ellement(), elle others for element\_at(i) Ensert(), and element(), and ellement(), elle other element from the list and return the element as the operation's result. Other operations retrieve an element without removing it from the list. These are ellement(), ellement(), element(), elle others to manipulate lists are defined. These are  $\rho = 0$  contat(1), which creates a new list p that is the concatenation of lists c and 0 (the elements in list c followed by these in list i), and plaqued(1), which appends the elements of his 1 to the end of list o (without creating a new list object).

The Arrayets object type also inherits the Collection operations. It is similar too her except that an array has a fixed number of elements. The specific operations for an Array object a are a replace element at(i,e), which replaces the array element ar position i with element e; e = 0, remove element at(i), which retrieves the i<sup>th</sup> element and replaces it with a null value; and e = 0, retrieve element at(i), which simply retrieves the u<sup>th</sup> element of the array. Any of these operations can raise the exception. InvalidIngent if u is greater than the array's site. The operation 0, resize(r) changes the number of array elements to p.

The last type of collection objects are of type Dictionarysk. Two This allows the creation of profection of association pairs sky, vs, where all k (key) values are unique. This allows for associative retrieval of a pairicular pair given its key value (similar to an index). If a walcollection object of type Dictionarysky, vs, then  $p_{1}hind(k,v)$  binds value v to the key k as an association  $ek_{1}vs$  in the collection, whereas  $p_{1}unbid(k,v)$  binds value v to the key k as an association  $ek_{1}vs$  in the collection, whereas  $p_{1}unbid(k)$  removes the association with key k formoriand  $v = a_{1} lookup(k)$  returns the value v associated with key k in  $p_{2}$ . The latter two operations can have the exception KeyNotFound. Finally, alcoritains\_key(k) returns that if key k exists in a and returns take otherwise.

Figure 21.2 is a diagram that illustrates the inheritance hierarchy of the built in constructs of the object model. Operations are inherited from the supersyste to the subcept. The collection object interfaces described above are not decity astantable; that is, one cannot directly create objects based on these interfaces. Rather, the interfaces can be used to specify user-defined collection objects—of type Ser. Bog. Lot, Array, or Dictionary—for a particular database application. When a user designs a database schema, they will declate their own object interfaces and classes that are relevant to the database application. If an interface or class is one of the collection objects, say a Set, then it will inherer the operations of the Set interface. For example, in a set objects would be sets of Student objects. The programmer can then use the operations for Setsets to manipulate an object of type SetsStudents. Greating application classes is typically done by utilizing the object definition language ODI (see Section 21.2).



FIGURE 21.2 Inheritance hierarchy for the built-in interfaces of the object model.

It is important to note that all objects in a particular collection mail *be of the* sine type. Hence, although the keyword any appears in the specifications of collection interfaces in Figure 21.1c, this does not mean that objects of any type can be intermixed within the same collection. Rather, it means that any type can be used when specifications the type of elements for a particular collection (including other collection types).

#### 21,1.3 Atomic (User-Defined) Objects

The previous section described the built-in collection types of the object model. We now discuss how object types for meanic object can be constructed. These are specific using the keyword class in 0.4. In the object model, any user-defined object that is as a collection object is called an **atomic object**.<sup>17</sup> For example, in a **universit**'s database application, the user can specify an object type (class) for Student objects. Most sub-objects will be structured objects; for example, a Student object will have a complex structure, with many ourribures, relationships, and operations, but it is still considered atomic because it is not a collection. Such a user defined atomic object type is defined atomic object rule is still considered atomic because it is not a collection. Such a user defined atomic object type is defined as a class by specifying its **properties** and **operations**. The properties define the state of the object and are further during owhed into **attributes** and **relationships**. In this sub-section, we elaborate on the three types of components—attributes, telationships, and operations—ibutes and because attributes can include. We dilustrate our discussion with the two classes Employee and Department shown in Figure 21.3.

<sup>13</sup> As mentioned evolves this detentions if atomic object in the COMO object model is different from the detention of atom constructor given to Chapter 20, which is the definition used in much of the object-oriented database laterature.

```
class Employée
 extent all_employees
 ssn ì
 kαγ
L
 attribute
 alring
 пате;
 atirBuie
 atrino
 $59.
 attribule
 date
 birchdata;
 alitibule
 enum Gender(M, F)
 Sex:
 ettribute
 short
 3080
 relational-up Department
 works, for
 inverse Department; hes_emps;
 void
 reassign_einu(in string new_dname)
 relace/dheme not validi-
);
class Department

 estent al_departments

 dneme, drumper 🔅
 kev
I
 eth-lbute
 etring
 dnáme
 eltribute
 short
 drumber.
 ethribule
 etruct Dept_Mgr (Employee manager, date sla@date)
 mar.
 attribute
 $61<91/102>
 lucations
 attribute
 struct Prois (string proname, lime weekly, hours)
 projs;
 relationship set<Employee>
 has_emps invariae Employee works_for.
 rold
 add_empjin string new enamej reises(ename_not_vaid).
 vold
 change_manager(in siring new_mar_name; in date stancale);
Ŀ
```



An attribute is a property that describes some aspect of an object. Attributes have values, which are cypically literals having a simple or complex structure, that are stored within the object. However, attribute values can also be Object. Ids of other objects. Attribute values can even be specified via methods that are used to calculate the attribute value. In Figure 24.1<sup>16</sup> the attributes for Employee are name, assist birthdate, see, and age, and those for Department are drame, downter, mgr. locations, and profs. The ngr and profs attributes of Department have complex structure and are defined via struct, which corresponds to the aple constructor of Chapter 20. Hence, the value of mgr in each Department object will have two components banager, whose value is an Object\_Id that references the Employee object that manages the Department, and startdate, whose value is a date. The locations attribute of Department object can have a set of locations.

<sup>14</sup> We are using the Object Definition Language 1911 Fnotution in Figure 21.5, which will be disclosed in more fetted in Section 21.2.

A relationship is a property that specifies that two objects in the database are related together. In the object model of (20MG, only binary relationships (see Chapter 3) are explicitly represented, and cock binary relationship is represented by a path of mease reference specified vip the keyword relationship. In Figure 21.3, one relationship exists that relates each Employee to the Department in which he or she works. The works for relationship of Employees that work in the Department ---the has\_emps relationship of Department. The keyword inverse directions each Department is related in the set of Employees that work in the Department—the has\_emps relationship of Department. The keyword inverse directions<sup>15</sup> By specifying inverses, the database system or maintain the referential integrits of the relationship automatically. That is, if the value of works\_for for a particular Employee is relationship to be represented in only one direction, then it has to be modeled as an attribute (or operation). An example is the matager component of the age attribute in Department.

In addition to arrithmest and relationships, the designer can include operations in object type (class) specifications. Each object type can have a number of operation signatures, which specify the operation (time, its argument types, and its retarned cable, if applicable. Operation names are oblique within each object type, but they can be overloaded by having the same operation names of exceptions that can occur during operation execution. The implementation of the operation will include the code to raise these exceptions. In Figure 21.3, the Employee class has one operation, reasonant erg, and the Department class has two operations, add\_emp and change\_manager.

#### 21.1.4 Interfaces, Classes, and Inheritance

In the ODW/ object model, two concepts exist for specifying object types, interfaces and classes. In addition, two types of inheritance relationships exist. In this section, we discuss the differences and similarities among these concepts. Following the OB300 retrinships, we use the word **behavior** to refer to operations, and state to refer to properties (attributes and relationships).

An interface is a specification of the abstract behavior of an object type, which specifies the operation signatures. Although an interface may have state properties fattributes and relationships) as part of 0s specifications, these causes be inherited hera the interface, as we shall see. An interface also is noninstantiable—that is, one cannot create objects that correspond to an interface definition.<sup>16</sup>

A class is a specification of both the abstract behavior and abstract state of an object type, and is **instantiable**—that is, one can greate individual object instances corresponding

<sup>15.</sup> Chapter <sup>1</sup> docussed how a relationship can be represented by two cutodates in inverse directions

<sup>16.</sup> This is somewhat similar to the concern of absizion class in the C++ programming language

to a class definition. Because interfaces are noninstantiable, they are mainly used to specify abstract operations that can be inherited by classes or by other interfaces. This is called **behavior inheritance** and is specified by the "2" symbol <sup>11</sup> Elence, in the ODMO object **model**, behavior inheritance requires the supertype to be an interface, whereas the subtype could be either a class or another interface.

Another inheritance relationship, called EXTENDS and specified by the extends keyword, is used to inherit both state and behavior strictly among classes. In an EXTENDS inheritance, both the supertype and the subtype must be classes. Multiple inheritance via EXTENDS is not permitted. However, multiple inheritance is allowed for behavior inheritance via "". Hence, an interface may inherit behavior from several other interfaces. A class may also inherit behavior from several interfaces via EXTENDS. We will give examples in Section 21.2 of how these two inheritance relationships—"?" and "XTENDS" may be used.

#### 21.1.5 Extents, Keys, and Factory Objects

In the CDMC object modul, the database designer can declare an extent for any object type that is defined via a class declaration. The extent is given a name, and it will contern all persistent objects of that class. Hence, the extent behaves as a verofight that holds all persistent objects of the class. In Figure 21.3, the Enployee and Department classes have extents called all\_enployees and all departments, respectively. This is similar to creating two objects—one of type. Set comployees and the second of type Set departments - and making them persistent by noming them all enployees and all\_departments. Extents are also used to suromatically enforce the set/subset relationship between the extents of a -upertype and its subtype. If two classes A and B base extents all\_A and all\_B, and class B is a subtype of class A (that is, class 6 (XTENTS class 4), then the collection of objects in all\_B must be a subset of those in all\_A at any point in time. This constraint is automatically enforced by the database system.

A class with an extent can have one of more keys. A key consists of one or more properties lattributes or relationships) whose values are constrained to be unique for each object in the extent. For example, in Figure 21 4, the Employee class has the same attribute as key teach Employee object in the extent must have in unique same value), and the Department class has two distinct keys: drame and dourbeet (each Department must have a unique dname and a unique dname). For a composite keys<sup>3</sup> that is made of several properties, the properties that form the key are contained an parentheses. For example, at a class Vehicle with an event all vehicles has they are doubled up of a combination of two

<sup>17.</sup> The OF#90 report also calls interface inheretarize as type/subtype, is-a, and its reconstruction/speculturation relationships, idenoight in the literature, these terms have been used to describe inheretance of both state, and operations (see Chapters 4 and 26).

<sup>15.</sup> A composite key is called a compound key in the ODMO report
attributes state and litense\_number, they would be placed in parentheses as (state, litense\_number) in the key declaration.

Next, we present the concept of factory object—an object that can be used as generate in create individual objects via its operations. Some of the interfaces of factory objects rhat are part of the ODMO object model are shown in Figure 21.4. The interface ObjectFactory has a single operation, newO, which returns a new object with an Object\_ide by inheriting this interface, users can create their own factory interfaces for each operation and differently for each reprior of object. Figure 21.4 also shows a OateFactory interface, which has additional operations for creating a new calendar date, and for creating on object whose value is the carternet\_state, among other operations (not shown in Figure 21.4). As we can see, a factory object basically provides the constructor operations for new objects.

Finally, we discuss the concept of a database. Secare a OB2MS can create name efferent databases, each with its own schema, the OLANA object model has interfaces for **BatabaseFactory** and **Database** objects, its shown in Figure 21.4. Each database has its own database name, and the bind operations can be used to assign individual endpies name to personent objects in a particular database. The **linkup** operation returns an object from the database that has the specified object name, and the **unbind** operation returns the name of a persistent primed object from the database.

```
interface ObjectFactory {
 Object.
 newill:
n.
interface DateFactory : ObjectFactory (
 InvalidDate();
 exception.
 Jare
 calendar dater
 in unsigned short year.
 in unsigned short mantly.
 in unsigned short days
 rakes/invalidDate:
 (Date
 OUMBROKE
j.,
interface DatabaseFactory (
 Oatabase.
 new())
1
interlace Database (
 void
 upen(in strins database (name).
 dose().
 vod.
 void
 bind(in any some lobject, in string object, name).
 Object
 unbindtin shing name);
 Chjart
 Inosupen stong object_hafve)
 raises/Elementho(Found).
```

```
ì.
```

FIGURE 21.4 Interfaces to illustrate factory objects and database objects.

# 21.2 THE OBJECT DEFINITION LANGUAGE ODL

After our overview of the CONRE object model in the previous section, we new show how these concepts can be utilized to create an object database schema using the object definition language OFF.<sup>19</sup> The OFF is designed to support the semantic constructs of the OFVG abject model and is independent of any particular programming language. Its neurin use is to show object specifications—that its classes and interfaces. Hence, OFF is not a full programming language. A user can specify a database schema in OFF, independently of any pregramming language, then use the specific actuations to specify how OFF constructs for being anguage, then use the specific programming language, with a constructs of specific programming languages, such as CF++, SMALLIALS, and JAVA. We will give an overview of the CF++ binding in Section 71.4

Figure 21-5b shows a possible object schema for part of the average in database, which was presented in Chapter 4. We will describe the concepts of OX using this example, and the one in Figure 21-7. The graphical notation for Figure 21-5b is shown in higher 21-5a and can be considered as a variation of EFR diagrams (see Chapter 4) with the added concept of interface robertance but without several EER conserpts, such as categories (interface) and attributes of relationships.

Figure 21.6 shows one possible set of ODL class definitions for the processive Jatabase. In general, there in  $\eta$  be several possible mappings from an object schema Jacom for FFD schema diagram into ODL classes. We will discuss these options further in Section 21.5

Figure 21.6 shows the straightforward way of mapping part of the university database from Chapter 4. Entity types are mapped intra COL classes, and inheritance is done using EVENDY. However, there is no direct way to map categories (union types) or to do nalliple inheritance. In Figure 21.6, the classes Person, Faculty, Student, and GradStudent have the extents persons, faculty, students, and grad\_students, respectively. Both Faculty and Student EXTENDS Person, and GradStudent EXTENDS Student. Hence, the collection of students (and the collection of faculty) will be constrained to be a subset of the collection of persons at any point in time. Similarly the collection of grad\_students will be a subset of students. At the same time, individual Student and Faculty objects will inherit the properties (students and relationships) and astraines of Person, and individual GradStudent objects will inherit these of Student.

The classes Qepartment, Course, Section, and CurrSection in Figure 21.6 are straightforward mappings of the corresponding errory sypta in Figure 21.6b. However, the dass Grade requires some explanation. The Grade class corresponds to the MN relationship between Student and Section in Figure 21.6b. The reason it was made into a separate class (rather than as a pair of inverse relationships) is because it includes the relationship attribute grade <sup>22</sup> Hence, the MN relationship is imapped to she class Grade, and a pair of LN relationship is imapped to she class Grade, and a pair of UN relationship is imapped to she class Grade, and a pair of UN relationship is mapped to she class Grade.

<sup>(9)</sup> The OOT syntax and data types are meant to be compatible with the Insertace Denniton Language (IOD) of CORRA (Common Object Request Birsker Architecture), with extensions for relationships and other database concepts.

<sup>22</sup> We will discuss alternative mappings for griphotes or relationships in Section 21.5.





```
class Person
 extent persons
 Luine -
 sster)
 etaribuse
 all'uct Phame (etning thame, etning motime, string iname).
 Planting:
 attribute
 atrino.
 $550
 ettribute
 dale.
 berthdate:
 ettribute
 enum GendertM, F1 sex;
 ettribute
 BIRICE Addresse
 (short no, string street, short apino, string city, string state, short zip i
 address:
 short soull:
];
class Faculty extends Person
 extent (acuity -)
 attribule
 etrina
 rank;
 attribute.
 Read
 salary:
 attribute.
 etring
 office:
 attribute
 string
 phone;
 relationable Department
 works. In Inverse Department::has jacuity;
 relationarity set<OradStudent> advises inverse GradStudent.advisor.
 relationship set. GradStudents on committee of
 Inverse GradStudem::commuse;
 void
 give_raise(in float raise);
 void
 promotecies string new marks:
ł.
classe Grade
 extent grades ()
٢
ſ
 eth Bude
 enum GradeValues/A.B.C.D.F.I.P2
 grade.
 milational/lip Section section inverse Section (students;
 relationship Student student Inverse Student.completed_sections.
1:
class Student extends Person
 eorient students i
4
ł
 attribute
 string
 class:
 ett Baste
 Cercartment
 minora in:
 relationship Department majors_in inverse Department::has_majors;
 relationship set<Grade> completed_sections inverse Grade::student.
 relationship set-CurrSection> registered in
 inverse CurrSection::registered_students,
 vold
 change_majonin string doeme) releasedname not valid);
 ficel.
 apa():
 n old
 register(in their secre) rejest (sector_not_valid);
 vold
 assign_grade(in short secno, in GradeValue grade)
 refees(section_not_valid,grade_not_valid).
```

FIGURE 21.6 Possible ODU schema for the wavevary database of Figure 21.5(b).

```
class Degree
 string
 attribute.
 college;
 etti Hibuibe
 etring
 degree:
 attribute
 string
 YOUR:
1:
class Grad9tudent antends Student
 extent cred students ()
ί
¢
 stiribute
 set-Degree>
 depress:
 relationship Faculty advisor invense Faculty::advises:
 relationship set-Faculty> committee inverse Faculty::on_committee_ol;
 void
 assign_edvicor(in string iname; in string mame)
 refees(faculty_not_valid);
 eseign_committee_member(in string iname; in etring iname)
 molet
 minestfaculty not velici);
5:
class Department
 extent decamments way drame 5
ŕ.
í
 dname:
 attribute
 string
 attribute
 string
 Dichone:
 all division in the
 string
 outline.
 attribute
 atring
 college:
 attribute.
 Faculty
 Chair,
 relationship set-Faculty> has_locatly inverse Faculty:works in;
 relationable set-Student> has majors inverse Student mejors in;
 relationable set-Course- offers invesse Course::offered_by;
Ŀ.
class Course
 Anteni courses Juny coo 🕴
L
ł
 etribute.
 CPerne:
 etrina.
 distante.
 string
 ćno;
 alification
 etring
 description:
 relationship set-(Section> has_sections inverse Section::of_course;
 relationship Department offered by inverse Department offers.
Ŀ
cless Section
 extent sections ()
ſ
ł
 short
 allitibute
 BROND:
 attribute
 turing
 yên,
 enum Quartert Fall, Winter, Spring, Summer) ptr.
 sitribute
 relationship set<Grade> students inverse Grade;;section;
 relationship Course of_course Inverse Course::has_sections;
ŀ.
class CurrSection actends Section
 extent current eactions.
 3
ι
I
 relationship set-cStudents registered, students immede Studentcregestered_in
 void register studentilin siring san)
 releve/aluxient_not_vaild, section_luli),
Ŀ
```

```
FIGURE 21.6 (CONTINUED)
```



FIGURE 21.7A An illustration of interface inheritance via ":". Graphical schema representation.

Section and Grade <sup>(1)</sup> These two relationships are represented by the tollowing relationship properties: completed\_sections of Student; Section and Student of Grade: and students of Section (see Figure 21.6). Finally, the class Degree is used to represent the composite, multivalued attribute degrees of GradStudent (see Figure 4.10).

Because the previous example did nor include any interfaces, only classes, we now index a different example to illustrate interfaces and interface (behavior) inheritance. Figure 21.7 is part of a database schema for storing geometric objects. An interface GeometryObject is specified, with operations to calculate the perimeter and area of a geometric object, plus operations to translate (move) and rotate an object. Several classes (Rectangle, Triangle, Circle, ...) inherit the GeometryObject numface. Since GeometryObject is an interface or is nonoscentiable—that is, no objects can be created based on this interface directly Elowever, objects of type Rectangle, Triangle, Circle.

. can be created, and these objects inherit all the operations of the **GeometryObject** interface. Note that with interface inheritance, only operations are inherited, not properties (attributes, relationships). Hence, if a property is needed in the inheritang class, it must be repeated in the class definition, as with the reference\_point attribute in Figure 21.7. Notice that the inherited operations can have different implementations in each class. For example, the implementations of the area and pertineter operations may be different for Rectangle, Triangle, and Circle.

Multiple informance of intenfaces by a class is allowed, as is multiple inheritance of interfaces by another intenface. However, with the ENTENDS (class) inheritance, multiple inheritance is 600 (crowned). Hence, a class can inherit via EXTENDS from at most one class (in addition, to inheriting from zero extinuite interfaces).

<sup>21.</sup> This is similar to how an MPN relationship is mapped in the relational model (see Chapter 7) and in the legacy network model (see Appendix C).

```
Interface GeometryObject
 attribule
 entum Shape{Rectangle.Triangle.Circle, ...} shape,
 attribute
 struct Port (short x, short y)
 referance point:
 fioal
 perimete();
 float -
 aree();
 void
 translate(in short x_translation, is short y_translation);
 void
 rotare(in floal angle of_rotation);
I:
class Reclangie : SeometryObject

 extent 'eclangies'

 ı
I
 struct Point (short <. short ()
 releience point.
 attribule
 attribute
 short
 enailt:
 ettribute
 shari
 he aht:
 attribute
 float.
 or engabon_angle;
1:
class Trangla : GeometryObject
; extent triangles (
ł
 attribute
 struct Point (short), short y)
 référence point.
 stiribule
 ahort
 side I:
 ettribule
 short
 order 2:
 swibute
 TIC 61
 sidet side2 and/a.
 ettribute
 floet.
 side1_onentation_angle;
3.
class Crolo : GeometryObject
 extent ordes
¢
 reference_point:
 attribute
 struct Paint [short x, short y].
 attribute
 short
 radius
4
....
```

FIGURE 21.78 An illustration of interface inheritance via Y". Corresponding interface and class definitions in ODC.

# 21.3 THE OBJECT QUERY LANGUAGE OQL

The object query language (0.8.1) is the query language proposed for the ODMG object model. It is designed to work closely with the programming languages for which an ODMG handing is defined, such as  $C + \epsilon_1$  SMALUTALK, and J W & Hence, on OQU query embedded into one of these programming languages can return objects that motch the type system of this language. In addition, the implementations of class operations in an ODMG schema can have their code written in these programming languages. The OQU syntax for queries is similar to the system of the relational standard query language SQL, with additional features for OTMG concepts, such as object identity, complex objects, operations, inhetiatice, polymorphism, and relationships. We will first discuss the syntax of simple (3Q), quotes and the concept of using named objects or extents as database entry points in Section 21.3.1. Then in Section 21.4.2, we discuss the structure of query texults and the use of path expressions to traverse te ationships among objects. Other OQI features for bandling object identity, inheritance, pdynorphism, and other object oriented concepts are discussed in Section 21.4.3. The examples to illustrate (3Q) queries are based on the using or database schema given in Sigure 21.6.

# 21.3.1 Simple OQL Queries, Database Entry Points, and Iterator Variables

The basic 0.83, syntax is a select z = 1 from z = z, where z = structure, as for S(0). For example, the query to retrieve the numes of all departments in the college of 'Engineering' can be written as dollows:

```
QO: SELECT d.dnawe
FROM d in departments
WHERE d.college = 'Engineering';
```

In general, an entry point to the database is needed for ouch query which can be any named besistant object. For many queries, the entry point is the name of the extent of a class. Recall that the extent name is considered to be the name of a persistent object whose type is a collection (in most cases, a set) of objects from the class. Looking at the extent names in Figure 21.6, the named object departments is of type set-(Departments) persons is of type set-(Departments) persons is of type set-(Departments).

The use of or event name—departments in Q0—as in entry point refers to a persistent collection of objects. Whenever a collection is referenced in an Q0, query, we should define an iterator variable  $^{22}$ —d in Q0—that tanges over each object in the collection. In namy cases, as in Q0, the query will select certain objects from the collection, based on the conditions specified in the where-chose. In Q0, only persistent objects d in the collection of departments that satisfy the condition d, college = "Engineering" are selected to the query result. For each selected object d, the value of durame is retrieved in the query result. Hence, the type of the result for Q0 is bigestrings, because the type of cuch drame value is string (even though the actual result is a set because dname is a key auribure). In general, the result of a query would be of type lag for select ..., from ..., as in S(4) (adding the keyword distinct eliminates dup) gates).

Using the example in QC, there are three septractic options for specifying iterator variables.

d in deportments departments d departments as d

<sup>2.</sup> This is smaller to the tuple somebles that range over tuples of solt-pictures.

We will use the first construct in our examples.<sup>23</sup>

The named objects used as database entry points for OQL queries are not limited to the names of extents. Any named persistent object, whether it refers to an atomic (singlet object or to a collection object can be used as a database entry point.

### 21.3.2 Query Results and Path Expressions

The result of a query can in general be of any type that can be expressed in the COMobject model. A query does not have no follow the select ... from ... where structure, in the simplest case, any persenent name on its own is a query, whose result is preference to that persistent object. For example, the query

Ul: departments;

returns a reference to the collection of all persistent department objects, whose type is set (Departments, Similarly, suppose we had given (via the database bind operation, see Figure 21.4) a poisisient name disdepartment to a single department object (the computer science department): then, the query:

Qla: csdepartment,

returns a reference to that individual object of type Department. Once an entry point is specified, the concrept of a **path expression** can be used to specify a *path* to related attributes and objects. A path expression reputally starts at a termstent object name, or a the iterator variable that ranges over individual objects in a collection. This name will be followed by zero or more relationship names or attribute names connected using the data normanor. For example, referring to the investor the database of Figure 21.6, the following an examples of path expressions, which are also valid queries in OQL.

Q2: csdepartment.chair: Q2a: csdepartment.chair.rank; Q2o: csdepartment.has faculty;

The first expression Q2 returns an object of type Faculty, because that is the type of the artificate chain of the Department class. This will be a reference to the Faculty object that is related to the department object whose persistent name is esdepartment via the attribute chain; that is, a reference to the Faculty object who is chainperstore the computer science department. The second expression Q2a is similar, except that a returns the rank of this Faculty object (the computer science department) is the second expression Q2a is similar, except that a returns the rank of this Faculty object (the computer science chain) is the time the object reference; hence, the type returned by Q2a is string, which is the data type for the rank attribute of the Faculty class.

Path expressions Q2 and Q2a return single values, because the attributes chair (of Department) and rank (of Faculty) are both single-valued and they are applied to a single object. The third expression Q2b is different: it returns on object of the set cFaculty's even when applied to a single object, because that is the type of the relationship has\_faculty of the Department class. The collection returned will include

23 Note that the latter two options are similar to the sympax for specifying tuple variables in 83 quoties.

references to all Faculty objects that are related to the department object whose persistent name is codepartment similar to inform the relationship has\_faculty; that is, references to all Faculty objects who are working in the computer science department. Now, to tenim the ranks of computer science faculty, we cannot write

```
Q3': csdepartment.has_faculty.rank;
```

This is because it is not clear whether the object returned would be of type set estimate or bages trings (the latter being more likely, since multiple faculty may share the same rank). Because of this type of ambiguity problem, 0.3, does not allow expressions such as Q3. Rather, one muy me an iteration variable over these collections, as in Q34 or Q3b below.

```
Q3a: Solect f.rank
from f in CSdepartment.has faculty;
Q3b: select distinct f.rank
from f in CSdepartment.has_faculty;
```

Here, QBa returns bagsstrings (duplicate tank values appear in the result), whereas QB returns setsstrings (duplicates are climitated via the distinct keyword). Both QBa and QBs illustrate how on iterator variable can be defined in the frameliause to range over a returneted collection specified in the query. The variable fillingBa and QB ranges over the elements of the collection csdepartment, has\_faculty, which is of type set of acultys, and includes only these faculty that are members of the computer science department.

In general, an CQU query can return a result with a complex structure specified in the query itself by utilizing the struct keyword. Consider the following two examples:

```
Q4: csdepartment.chair.advises:

Q4: select struct (name:struct()ast_name: s.name.lname.

first_name: s.name.fname),

degrees:(select struct (deg: d.degree.

yr: d.year,

College: d.college)

from d in s.degrees)

from 5 in csdepartment.chair.advises;
```

Here, Q4 is straightforward, returning an object of type setkGradStudents as its results this is the collection of graduate students that are advised by the chair of the contractor science department. Now, suppose that a query is needed to retrieve the last and first names of these graduate students, plus the list of previous degrees of each. This can be written as in Q4a, where the variable **s** ranges over the collection of graduate students advised by the collection of graduate students advised by the chairperson, and the variable **d** tanges over the degrees of each auch student **s**. The type of the result of Q4a is a collection of (first-level) straigts where each struct has two components; name and degrees.<sup>24</sup> The name component is a further struct made up of **Tast\_name** and **first\_name**, each being a angle string. The degrees component is defined by an embedded query and is the ling a

<sup>24.</sup> As incremental earlier, STRUCT corresponds to the tupic constructor discussed in Chapter 20.

collection of further (second level) structs, each with three string components, deg. yr, and college

Note that O.S. is enhanced with respect to specifying path expressions. That is attributes, relationships, and operation names (methods) can be used interchargeable, within the path expressions, as long as the type system of O.S. is not compressived. For example, one can write the following queries to retrieve the grade point average or all senior students majorine in computer science, with the result ordered by gra, and within that by last and list name:

Q5a ased the named entry point esdepartment to directly locate the reference to the computer science department and their locate the stadents via the relationship has, majors, whereas Q5b searches the students extent to locate all students inajoring in that department. Notice how attribute names, relationship names, and operation (method) names are all used interchangeably (in an orthogonal manner) in the path expressions gaa is an operation; majors\_in and has\_majors are relationships; and class name, dname, and fname are attributes. The implementation of the gpa operation computes the grade point average and returns its value as a float type for each selected students.

The order by classe is similar to the corresponding sQL construct, and specifies in which order the query result is to be displayed. Hence, the collection returned by a query with an order by classe is of type for

### 21.3.3 Other Features of OQL

Specifying Views as Named Queries. The view mechanism in OQL uses the concept of a named query. The Joffne keyword is used to specify an identifier of the named query, which must be a unique name among all named objects, class names, method names, or function names in the schema. If the identifier has the same name as an existing named query, then the new definition replaces the previous definition. Once defined, a query definition is persistent ontil it is redefined or deleted. A view can also have parameters (arguments) in its definition.

For example, the following view V1 defines a named query bas\_minors to remove the set of objects for students minoring to a given deportment:

```
V1: define has_minors(deptname) as
select s
from s in students
where s.minors_in.mname = deptname;
```

Because the ODL schema in Figure 21.6 only provided a unidirectional mnons\_inittribute for a Student, we can use the above view to represent its inverse without having to explicitly define a relationship. This type of view can be used to represent inverse relationships that are not expected to be used frequently. The user can now utilize the above view to write queries such as

has\_minors('Computer Science');

which would return a bag of students minoring in the Computer Science department. Note that in Figure 21.6, we did define has majors as an explicit relationship, presumably because it is expected to be used more often.

Extracting Single Elements from Singleton Collections. An  $\log_1$  query will, ingeneral, return a collection as its result, such as a bag, set (if distinct is specified), set list (if the order by clause is used). If the user nequires that a query only return a single element, there is an element operates in OQL that is guaranteed to return a single element e from a singleton collection a that contains only one element. If a contains more that are element of a singleton collection a that contains only one element. If a contains more than one element of a singleton for example, Q6 seturns the single object seference to the computer science department:

Since a department name is unique across all departments, the result should be one department. The type of the result is de**Department**.

Collection Operators (Aggregate Functions, Quantitiers) – Because many query expressions specify collections as their result, a number of operators have been defined that are applied to such collections. These include aggregate operators as well as membership and quantification (universal and existential) over a collection.

The aggregate operators (min. max. count, sun, and avg) operate over a collection.<sup>15</sup> The operator count returns on integer type. The romanning aggregate operators (min. max, sun, avg) inturn the same type as the rope of the operand collection. Two examples follow. The query QT returns the number of sindents minoring in "Computer Science," while Q8 returns the average gpa of all seniors majoring in computer science.

Notice that aggregate operations can be applied to any collection of the appropriate type and can be used in any part of a query. For example, the query to retrieve all department names that have more that 100 majors can be written as in Q9:

```
U9: select d.dname
from d in departments
where count (d.bas_majors) > 100;
```

The membrosity and quantification expressions terring a boolean type—that is, true or false. Let v be a variable, c a collection expression, b an expression of type boolean (that is, a boolean condition), and e an element of the type of elements in collection c. Then

(e in c) returns true if clement e is a member of collection c.

(for all v in c(b) returns true it all the elements of collection c satisfy b.

(exists v in c) b) returns true if there is at least one element in c satisfying b.

To illustrate the membership condition, suppose we want to retrieve the names of all students whit completed the course called 'Database Systems I'. This can be written as in QIC, where the nexted query returns the collection of course names that each student's has completed, and the membership condition returns true if 'Database Systems I' is in the collection for a particular student's:

```
Q10: select s.wawe.lname, s.wawe.fname
from s in scudents
where 'Database Systems I' in
(select c.cname
from c in s.completed_sections.sectmon.of_course);
```

Q12 also illustrates a surpler way to specify the select clause of queries that reform a collection of structs; the type returned by Q12 is bagestruct(string, string)».

One can also write queries that return true/false results. As an example, let us assume that there is a named object called Deneny of type Stutent. Then, query Q11 answers the following question: "Is Jereny a computer science numer?" Similarly, Q12 answers the question. "Are all computer science graduate andonis advised by computer science facility?". Both Q11 and Q12 return true or false, which are interpreted as yes or na answers to the above question's

```
QL1: Jenemy in has_nimons('Computer Stience'):
QL2. for all g in
 (select s
 from s in grac_Students
 where simajors in.dname = 'Computer Science''
 ; g.advisor in codepartment.has faculty:
```

Note that query Q12 also illustrates how out bute, relationship, and operation infentance applies to queries. Although s is an iterator that ranges over the extent grad\_stucents, we can write simajors\_in because the najors\_in relationship is inherited by GradStudeot from Studeot via EXTENDS (see Figure 21.6). Finally, to illustrate the exists quantitier, query Q13 answers the following question: 'Docs any graduate compares science major have w 4.0 gpa<sup>10</sup>. Here, again, the operation gua is inherited by GradStudent from Studeot via EXTENDS.

```
Ql:' exists g in
(select s
from s in grad_students
where simajons_in.drame = 'Computer Science')
: g.gpa = 4:
```

Ordered (Indexed) Collection Expressions. As we discussed in Section 21.12, rafections that are less and attaxs have additional operations, such as recreasing the \*\*\*, list and last elements. In addition, operations exist for extracting a sub-officering and annotenating two lists. Hence, query expressions that involve lists or arrays can invoke these operations. We will illustrate a low of these operations using example queries. Q14 retrieves the last name of the faculty member who earlies the highest salary:

```
Q14: First (select struct(faculty: f.name.lhame, salary:
f.salary)
from f זה faculty
order by f səlary desc);
```

Q14 (Instrates the use of the *fixe* operator on a list collection that contains the solares of foculty members suited in disconding order on solary. Thus the first element of factory of the contains the faculty member with the highest solary. This query assumes that only one faculty member came the maximum solary. The next query, Q15, removes the too three computer science analysis based on gpa.

The select-fron-order-by query returns a lost of computer science students ordered by gpa in descending order. The first element of an ordered collection has an index position of 0, so the expression [0:2] returns a list containing the first second and third elements of the select-fron-order-by result.

The Grouping Operator. The group, by clause in 582, olthough similar to the conceptuating clause in 521, provides explicit reference to the collection of objects within each group or promote First we give an estimple, then describe the general form of these queries.

©16 perfectes the number of majors in each department. In this query, the students are grouped into the same partition (group) if they have the same major; that is, the same value for sumajors lin, doars.

```
Ql6: select struct(depiname, number_of_majors:
count (partition))
from s in students
group by depinane: s.majors, in.dname;
```

The result of the grouping specification is of type setestruct(deptname: string, partition: bagestruct(s Student)>1>, which contains a struct for each group (partition: the two components the grouping attribute value (deptname) and ne bag of the student objects in the group (partition). The select clause retains the grouping attribute (name of the department), and a court of the number of elements in each partition (that is, the number of students in each department) where partition is the keyword used is refer to each partition. The result type of the select clause is set\*struct(deptname: string, number\_of\_Majors: integer)>. In general, the syntax for the group by clause is

group by  $f_1=e_1,\ f_2;\ e_3,\ \dots$  .  $f_4;\ e_4$ 

where fill e1, fill e2, ..., the ek is a list of partitioning (grouping) attributes and each partitioning attribute specification filler defines an attribute (field) name fill and at expression e1. The result of applying the grouping (specified in the group by clause) is a set of attributes:

setestruct(f\_: t., f<sub>i</sub>: t<sub>i</sub>, ..., f<sub>i</sub>: t<sub>i</sub>, partition: bag<8>)>

where this the type returned by the expression  $e_{1}$ , partition is a distinguished hele name (a keyword), and 0 is a structure whose fields are the iterator variables (s in Q16) declared in the from classe burging the appropriate type.

Just as in SQL, a having clause can be used to filter the partitioned sets (that is select only some of the groups haved on group conditions). In Q17, the previous garry is modified to illustrate the having clause (and also shows the simplified sentex for the select clause). Q17 retrieves for each department having more than 100 majors, the average groups of its majors. The having clause in Q17 selects only those partitions (groups) that have more than 100 elements (that is, departments with more than 100 students).

```
Q17: select deptname, avg_god: avg (select p.s.god from p
in partition)
from s in students
group by deptname: s.majors in.dname
having coupt (partition) > 200;
```

Note that the select clause of Q17 returns the average gpa of the students in the purtition. The expression

#### select p.s.gpa from p in partition

returns a long of student gravitor that purction. The from clause declares an iterator variable grover the guaration collection, which is of type bagestruct(s). Studenty-

Then the path expression p(s, qpa) is used to access the ground each student in the pathtum.

# 21.4 OVERVIEW OF THE C++ LANGUAGE BINDING

The C++ language binding specifies how OC, constructs are mapped to C++ constructs. This is done via a  $i_i$ ++ class fibrary that provides classes and operations that implement the C14 constructs. An Object Manipulation Language (Ovt.) is received to specify how database objects are retrieved and manipulated within a C++ program, and this is based on the C++ programming language systax and semantics. In addition to the O10/OML bindings, a set of constructs called physical program are defined to allow the programmer sime control over physical storage issues, such as clustering of objects, uniform indices, and memory management.

The class library added to (000 for the (3.30), standard uses the prefix d for class declarations that deal with database concepts.<sup>4</sup> The goal is that the programmer should taink that only one longuage is being used, not two separate longuages. For the programmer to refer to database objects in a program, a class d Ref <1> is defined for each database class T in the schema. Hence, program variables of type d\_Ref <1> can refer to Not persistent and mansient objects of class 1.

In order to other the various built-in rights in the CDMG Object Model such as collection types, various remplate classes are specified in the library. For example, an abstract class d\_Object\*1> specifies the operations to be inherited by all objects. Similarly, an abstract class d\_Collection\*1> specifies the operations of collections. These classes are not instantiable, but only specify the operations that can be inherited by all objects and by collection objects, respectively. A template class is specified for each type of collections there include d\_Set\*T>, d\_Ost\*T>, d\_BageT>, d\_Varray\*T>, and d\_Dictionary\*T>, and correspond to the collection types in the Object Mudel (see Section 31.1). Hence, the programmer can create classes of types such as d\_Set\*d\_RefeStudent>> whose instances would be sets of references to Student objects, or d\_Set\*Strings whose instances would be sets of Stimps. In addition, a class d\_Iterator corresponds to the Ubject Model.

The C++ OD allows a user to specify the cluster of a database schema using the constructs of C++ as well as the constructs provided by the object database library but specifying the data types of attributes.<sup>(3)</sup> basic types such as d\_Short (short integer), d\_LShort (constructs) hor integer), d\_Long (long integer), and d Float (floating point number) are provided in addition to the basic data types, several structured literal types are provided to correspond to the structured literal types of the ODUG Object Model. These include d\_String. d Interval, d Date, d Tine, and c Tinestamp (see Figure 21.1b).

<sup>26.</sup> Prostanyal In di serantis for tértelense classes.

<sup>27.</sup> That is monitor conduction object-oriented programming removalega-

To specify relation-lups, the keyword ReD\_ is used within the prefix of type names, for example, by writing

c Rel\_RefeDepartment, \_has\_majors> majors\_in;

in the Student class, and

d\_kel Set<Studert, majors in> has majors;

in the Department class, we are declicing that mayors\_in and has\_majors are relations ship properties that are inverses of one another and hence represent a 1 N binary relationship between Department and Student.

For the Ox1, the binding overloads the operation actese that a can be used to create either persistent or transient objects. To create persistent objects, one must provide the database name and the persistent name of the object. For example, by writing

d\_Ref<Student> s = new(DB1, 'lnhn\_Smith') Student;

the programmer creates a manual persistent object of type Statest in database 061 with persistent runne John, Smith, Abother operation, delete\_object() can be used to delete objects. Object modulution is done by the operations (methods) defined in each class by the programmer.

The C++ binding also allows the creation of extents by using the library class d Extent. For example, by writing

```
d Extent<Person> AllPersons(081);
```

the triogrammer would create a natioal collection object **A31Persons**—whose type would be it Set «Persons— in the database DB1 that would hold persistent objects of type Person. However, key constraints are not supported in the C++ binding, and any key checks must be programmed in the class methods.<sup>12</sup> Alan, the C++ binding does not support persistence was reachability; the object possible be statically declared to be persistent at the time trip created.

# 21.5 OBJECT DATABASE CONCEPTUAL DESIGN

Section 21.5.1 diseases how Object Dotobase (#081 design differe from Relational Database (RDB) design. Section 21.5.2 outbries a mopping algorithm that can be used to create an OFP schema, made of OTMC 058, closs definitions, from a conceptual HER schema.

# 21.5.1 Differences Between Conceptual Design of ODB and RDB

One of the main differences between 005 and 819 design is how relationships are haniffed. In 059, relationships are typically handled by buying relationship properties or ref-

25. We have only provided a bird overview of the C++ birding. For full details, see Cattell et al (1997), Ch. 5.

erence attributes that include OD(s) of the actance objects. These can be considered as .91 references to the related objects. Both single references and collections of references are allowed. References for a binary relationship can be declared in a single direction, or in both directions, depending on the types of access expected. If declared in both directions, they may be specified as inverses of one another, this enforcing the vittle quivalence of the relational referencial integraty constraint.

In RDB, relationships among raples (records) are specified by artributes with reaching values. These care be considered as value references and are specified on foregriless, which are values of primary key artributes repeated in tup es of the referencing relation. These are limited to being single-valued in each record because multivalued antiferres are not permitted in the basic relationsk taskief. Thus, MoN relationships now be opresented not directly but as a separate relation (table), as cased on Section 7.1.

Marking binary relationships that contoin attributes is not strongluforward in 0.785, since the designer must choose in which direction the attributes should be included. If the attributes are included in both directices, then redonding in storage will exist acd any lead to inconsistent data. Hence, it is sometimes preferable to use the relational approach of creating a separate table by creating a separate class to represent the relationship. This approach can also be used for *n*-ary relationships, with degree  $n \ge 2$ .

Another major area of difference between ODB and RDB design is how inheritance is handled. In ODB, these structures are built into the model, so the mapping is achieved by using the inheritance constructs, such as deveed (1) and ENTENDV. In relational design, as discussed in Section 7.2, there are several options to choose from ance no built-in construct exists for inheritance on the basic relational model. It is important to note, though, that object relational and extended relational systems are adding features to directly model these constructs as well as to include operation specifications in destruct directly model these constructs as well as to include operation specifications in destruct directly and these terms are directly construct.

The third imagor difference is that in OW design, in is necessary to specify the operations cath on jn the design since they are part of the class specifications. Although it is juppertant to specify operations during the design phase for all types of databases, it may be delived in RDS design as it is not structly required antil the juppercentation phase.

### 21.5.2 Mapping an EER Schema to an ODB Schema

to is relatively straightforward to design the type doctatations of object classes for an 2009/15/1500 an FER scheme that contains nother categories not more relationships with n > 2. However, the operations of classes are not specified in the EED diagram and must be idded to the class doctations after the @roctoral mapping is completed. The outline of the mapping from EER is ODU is as follows.

Step 15 Create an ODL class for each EEX entity type or subclass. The type of the ODL class should include oil the attributes of the HER class." Multivaleet attributes are

(9) This implicitly uses a tople constructor at the top level of the type declaration, but ungeneral, the tople constructions of a coplicitly shown in the OFF classifications. declared by using the set, hag, or list constructors <sup>36</sup> If the values of the multivalued attribute for an object should be ordered, the list constructor is chosen, if duplicates are allowed, the bac constructor should be chosen; otherwise, the set constructor is chosen. Composite *constructor* is mapped into a tuple constructor (by using a struct declaration in ODD).

Declare an extent for each class, and specify ony key attributes as keys of the extent (This is possible only if an extent facility and key constraint declarations are available in the ODPMS.)

Step 2: Add relationship properties of reference articlutes for each bicary relationship into the COL classes that participate in the relationship. These may be created in one or both directions, if a binary relationship is represented by references in *both* directions, declare the references to be relationship properties that are inverses of era another, it such a facility exists.<sup>31</sup> If a binary relationship is represented by a reference in any non-dimensional declare the reference to be an attribute in the reference many class where type is the referenced class name.

Depending on the condinality ratio of the binary relationship, the relationship projections or reference attributes may be single-valued or collection types. They will be single-valued for binary rol monships in the 1-2 or N-1 directions: they are collection types (set-valued or list-valued<sup>10</sup>) for relationships in the 1-N or MeN direction. An alternative way for mapping binary MeN relationships is discussed in Step 7 below.

If referenceship attributes exist, a tiple constructor (struct) can be used to recate a singram of the form creference, relationship attributess, which may be included uppend of the reference attribute. However, this does not allow the use of the inverse construint. In addition, at this choice is represented in both direction, the attribute values will be represented invice, are sting redundancy.

Step 3: Include appropriate operations for each class. These are not available from the SE8 scheme and must be added to the database design by referring to the original requirements. A constructor method should include program code that checks are constronts that must hold when a new object is see real. A destructor method should check any constraints that may be violated when an object is defered. Other methods should include only further constraint checks that are relevant.

Step 4): An ODE class that corresponds to a solution in the FEB schema influence (ca EXTENES) the type and methods of its superclass in the OOE schema. Its specific linear influenced) attributes, relationship references, and operations are securified, as discussed in Step-1, 2, and 3.

30. Further analysis of the application domain is needed to decide on which construction to use because this information is not available trivin the EDS oberta.

31. The OD, studing provides for the explore definition of inverse relationships. Some CD90s products may nev provide this support in such to ease, the pregnationers must an anomal even telestionship explicitly breaching the methods that update the objects appropriately.

 $3\omega$ . The detriction whether to use set or list is used to of the fourt the H10 schemol and must be determined from the regarizements.

Step 5: Weak entity types can be mapped in the same way as regular entity types. An alternative mapping is possible for weak entity types that do not participate in any relationships except their identifying relationships these can be mapped as though they were composite innorabled annihules of the context entity type, by using the set-structs, ..., so or the set as constructs. The attributes of the weak entity are included in the structs, ..., so constructors. The attributes of the construction. Attributes are mapped as decreased in Steps 1 and 2.

**Step 6**: Categories (union types) in an EER schema are difficult to map to ODU It is possible to create a mapping similar to the EER-to-relational mapping bee Section 7.2) by declaring a class to represent the category and defining 1.1 relationships between the category and each of its superclasses. Another option is to use a noise type, if it is available.

Step 7: An many relationship with degree  $\kappa > 2$  can be mapped into a separate class, with appropriate references to each participating class. These references are based on mapping a 1.N relationship from each class that represents a participating cutty type to the class that represents the mary relationship. An MaN biaary relationship, especially if it contains relationship attributes, may also use this mapping option, it desired.

The anapping has been applied to a subase of the interaction database schema of Figure 4.10 in the context of the COMC object database standard. The mapped object schema wang the COL notation is shown in Figure 21.6.

# 21.6 SUMMARY

In this chapter we discussed the proposed standard for object oriented dualdass. We started by describing the various constructs of the CDMC object model. The various hull in types, such as Object, Collection, Iterator, Ser, List, and so on were described by their interfaces, which specify the built-in operations of each type. These built-in types are the foundation upon which the object definition language (ONL) and object query language vOQL) are lased. We else described the difference between objects, which have at ObjectId, and Iterals, which are values with no O(1). Users can declare classes for their opplication that inhear operations from the appropriate built-in interfaces. Two types of properties can be specified in a user-defined class- carributes and relationships of in addition to the operaties that can be applied to objects of the class. The ODE allows users to specify both interfaces and classes, and permits two different types of inheritance—interface inheritance that "2" and class inheritance vie IATENTS. A class can have an extent and keys

A description of QCL then followed, and an example database schema for the UNURGIT) database was used to illustrate the UNU constructs. We then presented an overview of the object query language (QQUE. The UQL follows the concept of orthogonality in constructing queries, meaning that an operation can be applied to the result of another operation as long as the type of the result is of the concept uppt type for the operation. The QQL spring follows many of the constructs of QU but includes

additional concepts such as path expressions, inheritance, methods, relationships, and collections. Examples of how couse (K8) over the UNIVERSI or database were given

We then gave an overview of the t ++ language binding, which extends ()++ class declarations with the O(2) type constructors but permits scandess integration of ()++ with the ODAMS.

Kollivering the description of the OD815 model, we described a general technique to designing object oriented database schemes. We descussed how object-sciented database thier from relational databases in three main areas intercares to represent relationships inclusion of operations, and inheritance. We showed how to map a conceptual database design in the LBC model to the recommets of object databases.

#### Review Questions

- 214. What are the differences and surfamiles between objects and literals in the object Model?
- List the basic operations of the following built-re-interfaces of the 00000 Oper-Model: Object, Ceffection, Iterator, Set, List, Bag, Array, and Dictionary
- Consistent the built on structured literals of the COMO Object Model and the operations of each
- 21.4. What are the differences and similarities of attribute and relationship properties of inuser defined (atomic) class?
- 21.5. What are the differences and similarities of FX70S105 and interface " " inheritance?
- Discuss have persistence is specified in the effect Object Model in the C\*+ binding.
- 27.7. Wity me the concepts of extents and keep nepotiant in database applications?
- 21.5. Describe the following site incepts: docators entry points, just copy words, and recordedes, manoi questes turness, suggegne fourness, grouping, and quantifies.
- 21.9. What is pisanely the type orthogonality of sQC
- 21.10. Discuss the general principles behind the COM handing of the cOMG standard.
- 21.11. What are the main differences henceer designing a relational database and an object database?
- Describe the steps of the algorithm for object database design by 600 (700 morphics)

#### Exercises

- 21.13. Design an CO schema for a database application that you are interested in first construction (EE) schema for the application, then ensure the corresponding classes in CO. Specify a number of methods for each class, and then specify on new proQU for your database application.
- 21.14 Contaken the attraction database described in Exercise 4.21. Specify a monologial operation directions that you think should be applied by that application. Specify the CC1 classes and methods for the database.

- 21.15. Map the compare US schema of Figure 3.2 into ODU classes. Include appropriate methods for each class.
- Specify in OQ, the queries in the exercises to Chapters 7 and 8 that apply to the convex database.

# Selected Bibliography

Carrell et al. (1997) describes the city(of 2.2 story) ind and Cartell et al. (1993) describes the earlier versions of the standard. Several books describe the CD363, inclured time-tor example. Baker (1996), Other general references to object oriented databases were given in the bibliographic notes to Chapter 11.

The OZ system is described in Deux et al. (1991) and Bancilhon et al. (1992) includes a last of references to other publications describing various aspects of OC. The OZ model seas formalized in Volet et al. (1989). The ObjectShore system is described in Lamb et al. (1991). Inshitts in et al. (1987) and Wilkinson et al. (1994) discussives an object-oriented (1988) described at cleoclett-Packard laboratories. Moret et al. (1986) and Darters other et al. (1986) and Darters other et al. (1986) describe the design of (CMST/180). Aut 52 system supporting open achieve for developed at Texas Instruments is described in Thompson et al. (1988). The OEE system developed at ATT Bell Eabs is described in Agrow d and Gehoni (1989). The OEE system developed at MOC is described in Krist et al. (1992). Morst et al. (1992) described at MOC is described in Krist et al. (1992). Morst et al. (1992) described at MOC is described in Krist et al. (1992).



# Object-Relational and Extended-Relational Systems

In the proceeding eduptors we have primorily discussed three data models—the Fritte-Relationship (FR) model and us enhanced version, the EFR model, in Chopters 3 and 4: the relational data model and its larg toges and systems in Chapters 5 through 9, and the adjust-oriented data model and object database bagaages and standards in Chapters 20 and 21. We discussed how all these data models have been theorighty developed in terms of the following features.

- Modeling constructs for developing schemas for database applications.
- Constraints for the 6 is expressing certain ropes of relationships and constraints can the data as determined by application semantics.
- Operations and language facilities to manipulate the database.

Out of these three models, the LR model and its variations, has been primarily employed in CASE tools that are used for database and software design, whereas the other two models have been used as the basis for commercial DPMos. This chapter discusses the emerging class of commercial DPMos that are called object relational or *inhum ed* relational systems, and some of the conceptual foundations for these systems. Torse systems is which are effect objects relational DPMos (REDMos/), emerged as a way of enhancing the capitalities of relational DPMos (REDMos/) with some of the fractory that appeared in algoritheses for relational DPMos (REDMos/) with some of the fractory that appeared in algoritheses for the topologies.)

We start in Section 22.1 by groung an overview of the set standard, which provides extended and object explodities to the set standard for 0.0005. In Section 22.2 we give a Instanced perspective of database technology evolution and current trends to understand why these systems emerged. Section 22.3 gives an overview of the Informite database server as an example of a commercial extended OUTData. Section 22.4 discusses the object relational and extended features of Oracle. Section 22.5 discusses some since related to the implementation of extended relational systems and Section 22.6 presents are overview of the nested relational model, which provides some of the theoretical foundations behind extending the relational model with complex objects. Section 22.7 is a summery

Readers interested in typical features of ORU/MIS may read Sections 22.1 through 22.4 Other sections may be skipped in an introductory course.

# 22.1 OVERVIEW OF SQL AND ITS OBJECT-RELATIONAL FEATURES

We introduced S.3. as the standard language for RDRMss in Chapter 8. As we discussed, S.3. was first specified in the 1970s and anderward enhancements in 1989 and 1992. The language continued its evolution roward a new standard called SQL3, which adds objectoriented and other features. A subset of the SQL3 standard, now known as SQL99, was approved. This section highlights some of the features of SQL3 and SQL99 with a particular emphasis on the object relational concepts.

# 22.1.1 The sqL Standard and Its Components

We will briefly point our what each part of the SQL standard deals with, then describsome SQL tearares that are relevant to the object extensions to SQL. The SQL standard now includes the following parts<sup>1</sup>.

- SQL/Framework, SQL/Foundation, SQL/Bindings, SQL/Object.
- New parts addressing temporal, transaction aspects of SQL.
- SQU(CLI (Call Level materiace)
- SQL(PSM (Persistent Stored Mediales)

SQL/Foundation deals with new data types, new predicates, relational operations, corsexs, rules and trigger-cosm-defined types, non-active capabilities, and stored notices SQL(1). (Call Level Interface) (see Chapter 9) provides rules that allow execution of application rule without providing source cade and avoids the need for preprocessing. It contains about 52 maximums for rules such as connection to the set server, allocaring and deallocaring resources, obtaining diagnostic and implementation information, and controlling remaination of interfaces (SQL/SM (Persistent Steried Modules) specifies.

1. The discussion about the standard is longely Fased on Melion, and Merios (1998).

facilities for partitioning an opplication between a chert and a server. The avol is to enhance performance by monimizing network traffic, SQL/Bredings includes Embedded 80L and Direct Invocation. Embedded SQL has been enhanced to metude additional exception declarations. SQL/Temporal deals with historical data, time series data, and other temporal extensions, and it is being proposed by the 7SQL2 committee.<sup>2</sup> SQL/ Transaction specification formalizes the SA interface for use by SQL implementars.

# 22.1.2 Object-Relational Support in 5QL-99

The 500 (Object specification extends 5:01-92 to include object-one need rapidalities. We will discuss some of these features by referring to the curresponding object-oriented concepts that we discussed in Chapter 20. The following are some of the features that have been included in 3:31-99:

- Some type constructors have been added to specify complex objects. These archide the zose type, which corresponds to the tuple (or struct) constructor of Chapter 12. An array cybe for specifying collections is also provided. Other collection type constructors, such as set, list, and bag constructors, are not yet part of the SQL-99 specifications, although some sesters include them and they are espected to be in fourte versions of the standard.
- A mechanism for specifying object identity through the use of reference type is included.
- Encapsulation of operations is provided through the mechanism of user-lenned types that may include operations as part of their declaration.
- Inheritance succhanisms are provided

We now discuss each of these concepts in more detail

Type Constructors. The type constructors out and isotry are used to specify complex types. These are also known as **user-defined types**, or **UDTs**, since the user defines them as a patternar application. A **row type** may be specified using the following syntax:

CREATE TYPE row\_type name AS [ RON ] («component declarations»);

The keyword ROW is optional. An example for specifying a row type for addresses and employees may be core as follows:

```
CREATE TYPE Addr type AS f
street VARCHAR (45),
city VARCHAR (25),
zip CHAR (5)
);
CREATE TYPE Emp_type AS i
```

2. The full proposition providents and Jensen (1996). We discuss temperation deling and manifuse hsqu2 in Chapter 23.

```
name VARCHAR (55).
addr Addr_type.
age INTEGER
);
```

Notice that we can use a previously defined type os a type for an attribute is illustrated by the addit attribute above. An **array type** may be specified for an attribute whose value will be a collection. For example, suppose that a company has up to up locations. There a row type for company may be defined as follows:

```
CREATE TYPE (omp type AS (
rompname VARCHAR (20),
location VARCHAR (20) ARRAY [10]
):
```

Fixed-length array types have their elements referenced using the common notation of square brackets. For example, location[1] refers to the first location value in a location attribute. For nive types, the common dot notation is used to refer to components. For example, addriceity refers the the city component of an addrartribute. Corrently, analy elements cannot be arrays themselves, thus limiting the complex ty of the object structures that can be created.

Object Identifiers Using References. A user defined type can be used either as rype for an arribute, as illustrated by the addr attrabute of Erg. type, out can be used to specify the row types of jubles. For example, we can create two tables based on the row type declarations given earlier as follows:

```
CREATE TABLE Employee OF Employee REF IS emplied SYSTEM GENERATED.
CREATE TABLE (company OF Compleyee (
REF IS complet SYSTEM GENERATED,
PRIMARY KEY (compname)):
```

The above examples also dilustrate how the user can specify that system-generated object identifiers for the individual row- in a table should be created. By using the spiritive

REF IS word\_attributes wealve generation\_methods .

the user doplates that the argubate balance of ideater ibutes will be used to allocate midvideal tuples in the table. The options for svalue generation nethods are SYSTEV SON ERADED of CERENT, in the former case, the system will anomatically generate a usique identifier for each copie, in the larger case, the traditional method of using the user provided primary key color to identify tuples is applied.

A component attribute of this typle may be a reference (specified using the key and 8F?) to a taple of another (or possibly the same) table. For example, we can define the following additional now type and correponding table to relate an employee to a company.

```
CREATE TYPE Employment_type AS (
 employee REF (Emp_type) SCOPE (Employee),
 comsany REF (Comp_type) SCOPE (Company)
);
CREATE TABLE Employment OF Employment type;
```

The keypoint score-specific specific sphericane of the table whose papers can be referenced by the reference applicate. Notice that this is similar to a foreign key, incorporated the system generated value is used rather than the primary key value.

SQL uses a dot nutation to build path expressions that refer to the component antibures of tuples and row types. However, for an attribute whose type is REF, the dereferencing symbol (>) is used. For example, the query below retrieves employees working in the company named 'ABCXYZ' by querying the Employment table.

```
$ELECT c.employee->name
FROM Employment AS e
WMERE e.company->compname = 'ABCXYZ';
```

In SQL, -> is used for dereferencing and has the same meaning assumed to it in the programming language. Thus if ( is a reference to a table and a is a component attribute in that tuple, ( -> a is the value of attribute a in that tuple.

Object identifiers can also be explicitly declared in the type definition rather than in the table declaration. For example, the definition of Energy upper may be changed as follows:

```
CREATE TYPE Emp_type AS (
name CHAR (15),
addr Addr type,
age INTEGER,
emp_>d REF (Emp_type)
):
```

In the above example, the **onplace** values may be specified to be system generated by sing the command.

#### CREATE TABLE Employee OF Employee VALUES FOR emplid ARE SYSTEM GENERATED;

If several relations of the same row repelesion, 302, provides the 30.492 keyword by which a reference artribute may be made to point to a specific ralle of that type by using

SCOPE FOR kattributes IS krelations

Encapsulation of Operations in SQL. In SQL construct smills to closs definition is provided whereby the user can obside (manied user-defined repe with its own behavioral specification by specifying methods for operations) in addition to the artributes. The general form of an ODT specification with methods is

```
CREATE TYPE «type-name» (

list of component attributes with individual types

declaration of EQUAL and LESS THAN functions

declaration of other functions (methods)

);
```

For example, suppose we would like to extract the apartment number of given life a a string that forms the street attribute component of the Addr. type now type doclared previously. We can specify a method for Addr. type as follows:

```
CREATE TYPE Addr_type AS (
 street VAR(HAR (45),
 city VAR(HAR (25),
 zip (HAR (5)
)
METHOD apt_pc() RETURMS CHAR (8):
```

The code for implementing the method still has to be written. We can refer to the method implementation by specifying the file that contains the code for the method as follows:

#### METHOD

```
CREATE FUNCTION apt_np() RETURNS CHAR (6) FOR Addr_type AS
EXTERNAL NAME f/x/y/aptno.class' LANGUAGE fjava':
```

In this example, the implementation is in the JOCA language, and the code is stored in she specified file path name

Soft provides certain built in functions for user defined types. For a 1901 called type 1, the constructor function type 1() returns a new object of that type 1 in the new ODT object, every attribute is initialized to us default value. An observer function A is implicitly created for each attribute A to read its value. Hence A(X) or X:A retorns the value of articlate A of Type 1 of X is of type (type\_1). A mutator function for optimation and attribute sets the value of the attribute to a new value. SQL allows these functions to be blocked from public user an EXECUTE privilege is needed to have access to these toxetone.

In general, a  $00\,\sigma$  can have a number of user-defined functions associated with it. The syntax is

#### METHOD <=ame> (<arqument\_list>) RETURNS <type>;

Two repestor materions can be defined: internal SQL and external. Internal functions are written in the extended PSM language of sQL (see Chapter 9). External functions are written in a host language, with only their signature (interface) appearing in the 1.0° definition. An external function definition can be declared as follows:

#### DECLARE EXTERNAL <function\_name> <signature> LANGUAGE <language name>;

Many ORDMSs have raken the approach of doining a package of Abstract Data Types (ADTs) and associated functions for specific application domains. These could be purchased separately from the basic system. For example, the Data Blades in Informa Universal Server, the Data Cottridges in Oracle, and the Extenders in 302 can be considered as such packages or libratics of ADTs for specific application domains.

OUTS can be used as the types for attributes an SQU and the parameter types of a function of protocolate, and as a source type in a distinct type. Type Equivalence is defined in SQL at two levels. Evolutions are name equivalent if and only if they have the same name. Evolvypes are structurally equivalent if and only if they have the same number of components and the components are partwise type equivalent.

Attributes and functions in GOTs are douded into three categories

- PUBLIC (visible at the Or'T interface).
- PRIVATE (not visible at the UD1 interface).
- PROTECTED (visible only to subtypes).

It is also possible to dolino virtual attribures as part of CDTs, which are computed and updated using functions.

Inheritance and Overloading of Functions in SQL. Recall that we dready discussed many of the principles of internance in Charlet 22, SQL has rules for dealing officientance (specified on the UN/EU keyword). Associated with inheritance are the rules for overloading of function implementations, and for resolution of function ranges. These rules can be summarized as follows:

- All attributes are inherited.
- The order of supertypes in the UNUEC clause determines the informatic hierarchy.
- An instance of a subtype can be used in every context in which a signifype instance is used.
- A subtype can redefine any function that is defined in its supertype, with the restriction that the signature by the same.
- When a function is called, the best match is selected based on the types of all asymmetry.
- For dynamic buking, the metane types of populators is considered.

Consider the following example to illustrate type inheritance. Suppose that we want to create a subtype Manager, type that inherits all the attributes tand methods) of Emptype but has an additional attribute dept managed. Then we can write

```
CREATE TYPE Manager type UNDER Emp type AS (
dept managed CHAR (20)
):
```

This inherits all the attributes and methods of the supertype fmp\_type, and has an additional specific authority dept\_managed. We could also specify additional specific inerhads for the subtype

Another facility in SQL is the sope noble/subtable facility, which is similar to the class of extends inheritance discussed in Chapter 20. Here, it subtable inherits every column from its supertable; every row of a subtable corresponds to one and only one zow in the supertable; every row in the supertable corresponds to at most one too in a subtable. ISOTE: DELETE: and UTDATE operations are uppropriately propagated. For example, consider the real\_estate\_into table defined as follows:

```
CREATE TABLE real_estate info (
 property real_estate,
 owner CHAR(25),
 price MONEY,
):
```

The following subtables can be defined:

```
CREATE TABLE amenicaning) estate UNDER real_estate_info;
CREATE TABLE georgia_real_estate UNDER amenican real_estate.
CREATE TABLE atlanta real_estate UNDER georgia_real_estate;
```

In this example, every rople in the subtable american\_real\_estate must exist only superrable real\_estate\_info; every tuple in the subtable georgia\_real\_estate const exist in its superrable american real\_estate, and so can. However, tuples can exist in a superrable workout being in the subtable.

Unstructured Complex Objects in SQL – SQL has new data types for binary large objects (1008), and huge object locators. Two variations exist for binary large objects (8008) and character large objects (8008). SQL proposes 106 manipulation within the 0688 without having rature external tiles. Certain operators do not apply to LON valued artribairs – for example, enthmetic compatisons, group by, and order by. On the other hand, retrieval of partial value, affect comparisons, group by, and order by. On the other hand, retrieval of partial value, affect comparisons, we will see how large objects are used in Oracle S.

We have given an overview of the proposed object oriented buchties in SQL Arthutime, both the SQL/Foundations and SQLA-bject specification have been standardized in sevident that the facilities that make SQLA-bject specification have been standardized in implemented in commercial CROPARS (SQLAM) (nucleared closely follow what has been implemented in commercial CROPARS (SQLAM) (nucleared)) is being proposed as a separate standard for multimedia database non-opement with multiple parts framework, full text, system central purpose for https://and.sull.im/ige\_We will discuss the use of the two-dimensional data types and the unique and text\_Datablades in Informat Universit Server.

# 22.1.3 Some New Operations and Features in SQL

A major new operation is **linear recursion** for specifying tecorsive queries. To illustrate this, suppose we have a table called **fort\_vase** (**Pare1**). **Pare2**), which contains a tuple spl, p2> whenever part p1 contains part p2 as a component. A query to produce the **bill** of **materials** for some part p1 (that is, all component parts needed to produce p1) wint ten as a recursive query as follows:

```
WITH RELLESIVE

BILL_MATERIAL (Part1, Part2) AS

(SELECT Part1, Part2

FROM PART_TABLE

WHERE Part1 = 'p1'

UNION ALL

SELECT PART TABLE(Part1), PART_TABLE(Part2)

FROM BILL MATERIAL, PART_TABLE

WHERE PART TABLE.Part1 = BILL MATERIAL(Part2))

SFLECT * FROM RI.1 MATERIAL

ORDER BY Part1, Part2;
```

The final result is contained in Bire\_SCREEXER(Part1) Part2). The UNION ALL operation is evaluated by taking a union of all tuples generated by the inner block until no new ruples can be generated. Because SQL2 lacks recursion, it was left to the programmer to accomptish a by appropriate iteration.

For scenary in SQL3, the concept of rule is introduced, which is similar to a "jobdescription" and is subject to authorization of privileges. The artical persons (user accounts) that are assigned to a tole may change, but the rule authorization itself does not have usible changed. SQL3 also includes syntax for the specification of use of triggers (see Chapter 24) as across rules. Triggering events include the USERT, DELEE, and USDATE operations on a table. The trigger can be specified to be considered 66-OKE or AFTER the triggering events. The concept of trigger granularity is included in SQL3, which allows the specification of work now-level triggers (the trigger is considered for each affected row) or statement-level trigger (the trigger is considered for each triggering event). For distributed (cheat-server) databases (see Chapter 25), the concept of a cheat module is included in SQL3. A cheat and the account externally invoked procedures, convers, and temporary tables, which can be specified using SQL3 which and temporary tables, which can be specified using SQL3 which and temporary tables, which can be specified using SQL3 syntax.

SQL3 also is being extended with programming language facilities. Routines written in SQL8.0, with full matching of data types and an integrated environment are referred to as SQL routines. The make the language computationally complete, the following programming control structures are included in the SQL3 syntax CAUT/RETORS, BEARN END, SOP/END\_ROF, IF/THEN/FUS-/END IF, CASE/END\_GASL, FOOP/END FOOP, WHIEF END\_WEILE, REPSAT/ONET/FND\_REPERT, and ESAVE. Variables are declared using EDCLARE, and assignments are specified using SET External routines refer to programs written in a host language (ADAC), COOPOL, PASCIM, etc.), possibly containing embedded SQL and passing possible type nusmerches. The advantage of external routines is that there are existing librates of such postness that we broadly used, which can car down a lot of implementation effort for applications. On the other hand, solit routines are more "pine," but they have not been in when use, SQL routines can be used for server routines (schema-level contines or modules) or as client modules, and they may be procedures of functions that return values. SQL0:11 is described in Chapter 9

# 22.2 EVOLUTION AND CURRENT TRENDS OF DATABASE TECHNOLOGY

In the commercial world today, there are several furnities of FUMs products available. Two important ones are RCPMS and ODIMS, which subscribe to the relational and the object data models respectively. Two other imagor types of FDMS products—hierarchical and network over new being referred to as legacy DBMSs; these are based on the hierarchical and the recover data models, both of which were introduced in the mid-1960s. The bierarchical tanity patiently has one down neutroproduct—RCS of 089, whereas the network

These concepts are discussed in more detail in Chapter 24.

ramily includes a large number of E080ss, such as 0.8 0 (Honeywell), 0.935 (Computer Associates), 0.9363; (Herelett Packar, b, 9.385) 0.985 (Digital), and TOTAT/SURGA (Unicom), to nume a test. The Interarclucul and network data models are summarized in Appendixes E and E<sup>2</sup>

As database technology ecologis, the logicly D008s will be gradually replaced by newer offerings, hi the interim, we must face the import publicit of interoperability – the interoperation of a number of databases belonging to all of the disparate families of D008s – as well as to legacy the management systems. A whole series of new system and tools no deal with this problem are emerging as well. More recently, Mathias emerged as a new standard for data exchange on the Wob (see Chapter 26).

The main target behind the development of extended \$30,000.66 stein from the malelity of the legace (2006) and the basic relational data model as well as the earlier RDIVES to meet the challenges of new applications. These are primarily in area that involve a variety of rypes of data -for example, text in componentialed desktop publishing, mages in setellite imaging or weather forecasting, complex nonconvertined data in engineering designs, in the biological genome information, and in architectural drawings, time series data in bisory of stock market transactions of sales bisories and spatial and geographic data in images indicate pollution data, and mattic data. Hence the examplex of exception data in engineering design databases that can develop, national matter data. Hence there is a clear need to design databases that can develop, national matter data. Hence there objects aroung from such applications. Forthermore, or is becoming necessary to handle digitized information that represents audio and video shows from the original data frames) requiring the storage of 810.68 (for available subjects) is 0.085%.

The populatity of the relational model is helped by a very robost influstructure in terms of the conducteral CRVes that have been designed to support it. However, the basic relational model and earlier versions of its SQL language proved inadequate to meet the above challenges Legacy data models like the network data model have a facility to model inflationships explicitly, but they suffer from a heavy use of pointers to the implementation and have no denicipts like object identity, inheritance, encapsulation, or the support for multiple data types and complex objects. The literarchical model has well with same naturally occurring hierarchical paths in the data. Hence, intend was started to conduct the lost terms of bulk-th bierarchical paths in the data. Hence, intend was started to conduct the lost features of the object data model and languages into the net from data models in the lost features of the object data model and languages into the net from data models to the lost features of the object data model and languages into the net from data models to that it can be extended to deal with the challenging applications of today.

In the remainder of this chapter we highlight the features of two representative 1980s, that exemptify the OROBES approache latornix. Universal Server and Oracle 8. We detailed by briefly discussing the nested relational model, which has its origin in a series of research proposals and precotype implementations; this provides a theoretical momentations; this provides a theoretical momentations; the induction detailed in the relational framework of embedding intermediately structured complex objects within the relational framework.

<sup>4.</sup> These chapters devoted to the Network Data Medel and the Energical Data Nodel ste available at the Web size for this book.

# 22.3 THE INFORMIX UNIVERSAL SERVER<sup>5</sup>

The Informus Universal Server is an ODDNS that combines relational and object database rechnologies from two previously existing products. Informax and Illustral, the latter astrom originated from the DISD 30FS DBAS, which was a research project at the University of Coulombia at Berkeley, that was commercialized as the Montage DBMs and went through the name Mino before being named Illustra. Illustra was then acquired by inforative integrated into its RDBMS, and introduced as the Informax Universal Servers-an opposite.

To see why ORDBMSs emerged, we start by recusing on one way of classifying OBMS applications according to two dimensions or axes. (2) complexity of data—the X-dimension—and (2) complexity of querying—the Y-dimension. We can arrange drive oxes into a simple 2-1 space having four quadrants:

Quadrant 1 (X = 2, Y = 2): Simple data, simple querying Quadrant 2 (X = 0, Y = 1): Simple data, complex querying Quadrant 3 (X = 1, Y = 2): Complex data, simple querying Quadrant 4 (X = 1, Y = 1): Complex data, complex querying

Traditional RDR014s belong to Quadrant 2. Although they support complex id hoc queries and opdates (as well as transaction processing), they can deal only with simple data that can be modeled as a set of nots in a table. Many object databases (opR018s) fall in Quadrant 3, since they concentrate on managing complex data bar have somewhat louired querying capabilities based on travigation." In order to move onto the fourth pudrant to support both complex data and querying, RDR018s have been incorporating more complex data objects while objects have been incorporating more complex pretung (for example, the CQL high-level query logenage, discorporating more complex protons). Universal deriver belongs to Quadrant 4 because it has extended its base of constanted by acceptorating a variety of learnes that make it object relational

Other convest (6008068 that evolved from Ro8058 neclule Oracle from Oracle (hyporotion, Universal DP (0.19) from 1951. Objeter by Hewlett Packard (112) (which extends Oracle's DBVS) and Open OPB from PF (which extends PPS over Allbase/SQI gisduct). The more successful products seem to be those that manufaut the option of working as an R0605 while introducing the additional time transition. Our interval here is not to provide a comparative analysis of these products but only to give an overview of two representative systems.

<sup>5</sup> The Jacobion matheway non-reprint and boost can the work (algorized endowing ways) by Michael Storebooker and Datentry Master (1908), and on the unput provided by Magdi Morst at Information for the value of the messaer may refer to confue versions of Information (1909).

<sup>6</sup> Quadrant 1 includes one software packages that dual with data bandling with softwophasticated data retrieval and in inipiliarion features. These include a read-liceus like EXCEL, word processors use Microsoft Word, or any file instrugement software.

How Informix Universal Server Extends the Relational Data Model. The extensions to the relational data model provided by Elustra and incorporated non-Informix Universal Server fall into the following categories:

- Support for additional or extensible data types.
- Support for user-defined routities (procedures or functions).
- Implicit notion or inheritance.
- Support for indexing extensions.
- Data Blades Application Programming Interface (API).<sup>1</sup>

We give an overview of each of these features in the following sections. We leavialready introduced in a general way the concepts of data types, type constructors, complex objects, and inhoritance in the context of object-oriented models (see Chapter 20).

### 22.3.1 Extensible Data Types

The architecture of Informix Universal Server comprises the basic 1986s plus a number of Data Blade modules. The idea is to treat the DBMS as a rate times which a particular blade is inserted for the support of a specific data type. A number of data types have been previded, including two-dimensional grometric objects (such as points, hees, ritcles, and ellipses), images, time series, text, and Web pages. When Informix announced the Unversal Server, 29 Data Blades were already available fit is also rescable for an upplication to create its own types, thus making the data type norton fully extendible. In addition to the built in types, informis Universal Server provides the aser sorth the following fouconstructs to declare additional types:

- Opaque type
- 2. Distinct type.
- 3 Rose type:
- Collection type.

When creating a type based on one of the first three options, the user has in provide functions and contract for manipulation and conversion, including built in, aggriphe, and operator tenctions as well as any additional user defined functions and mutates. The details of these four types are presented in the following sections.

Opaque Type: The opaque type has its internal representation halden, so it is used for encapsulating a type. The user has to provide casting functions to convert an opaque object between its hilden representation in the server (database) and its visible representation is

<sup>7.</sup> Data Blades, error des extensions to the basic system, as we shall insure later in Section 22. Ifi-

for more information on the Dura Blades for Information Universal Segret, consult the Web site http://www.information/informations/.

seen by the chern (calling program). The user functions waddweave are needed to convert refrom the server attendal representation from/to the chern representation. Similarly, ingva/extent functions are used to convert refram an external representation for bulk copy fund/to the internal representation. Several other functions may be denied for processing the optique types, including assign(), destruct), and compare().

The specification of an opaque type includes its name, internal length if fixedmaximum internal length if it is cariable length, alignment (which is the byte boundary), as well as whether or not it is bashable (for creating a bash access structure). If we write

```
CREATE OPAQUE TYPE fixed phaque uds (INTERNALLENGTH = 9,
ALIGNMENT = 4, CANNOTEASH);
CREATE OPAQUE TYPE var opaque_udt (INTERNALLENGTH = variable,
NAXLEN=1024, ALIGNMENT = 8);
```

then the test statement creates a fixed-length user-defined optique type, named fixed\_ apague, udt, and the second statement creates a variable length one, named var\_ apague, udt. Both me described in an implementation with internal parameters that are not visible to the cleant.

Distinct Type. The distinct data type is used to extend an existing type through inheritance. The newly defined type inherits the functions/routines of its have type, if they are not overridden. For example, the statement

CREATE DISTINCT TYPE hiring\_care AS ()ATE:

creates a new user defined type, mining\_date, which can be used like any other built in type

ROW Type. The row type, which represents a composite offibere, a analogous to a situat type in the C programming language." It is a composite type that contains one of more fields. Row type is also used to support inheritance by using the keyword UNDER, but the type system supports single inheritance only. By creating tables where tuples are of a patientar row type, it is possible to treat a relation as part of an object-formated schema and establish inheritance relationships among the relations. In the following tow type declarations, employee, t and student\_s unlear to declarations have a declarations of personance.

```
CREATE ROW TYPE person t(name VARLFAR(60), social security
NLMERIC(9), birth date DATE):
CREATE ROW TYPE employee t(salary NUMERIC(10,2), hired_um
hiring_date) UNDER person_t;
CREATE ROW TYPE student_t(gua NUMERIC(4,2), address
VARCHAR(200)) UNDER person_t;
```

Collection Type: Informers Universal Server collections include fists, sets, and inclusers (bags) of built-in types as well as n-endefined types.<sup>2</sup> A collection can be the

This is simply in the next conversion discussed in Chapter 34.

10. These are signal into the collector right chicknessed in Chapters 20 and 71.

type of other a hold in a row type or a colorum to a colde. The elements of a set collection common contain dopheats values, and have no specific order. The list may exclude duplacete elements and order is significant. Finally, the multiset may include duplacets and has no specific order. Consider the following example:

CREATE TABLE employee (name VARCHAR(50) NOT NULL, commission MULTISET (MONEY));

Here, the employee table contains the commission colorum which is of type multiset-

# 22.3.2 Support for User-Defined Routines

Informax Universal Server supports ascr-defined functions and tearines to manipulate the user denned types. The implementation of these functions can be in other Stored Procedure Language (SPL), or in the U of LAVA programming languages. User-denned functions enable the user to define operator functions such as (NaSO) (office)), thus(), deale(), paster(), and regime). Both on functions such as (OaSO) (office), and regime). Both on functions such as (OaSO) (office), aggregate functions such as (OaSO) and larg(), and (periodefined requires. This enables futures allowing complexies of both on type whenever the required functions are defined. The following example specifies an equal function to compare two objects of the fixed, opaque, udt type declared eather.

Informas Universal Server also supports cast—a function that converts objects from a source type to a target type. There are two types of iser-defined class. (1) implicit and (2) explicit familiate casts are invoked automatically, whereas explicit casts are invoked only when the cast operator is specified explicitly by using "a" or CAST. AS. If the source and target types have the same internal structure (such as when using the donnet types specification), no user-defined functions, are needed.

Consider the following example to illustrate explicit custing, where the employee rable has a collicolomic of type vaw\_opaque\_udt and a collicolomic of type travel\_opaque\_udt.

```
SELECT coll FROM employee WHERE fixed_opaque odt(:coll = col2;
```

In order to compare cold with cold, the cast operator is applied to cold to convert it from var japaque udt to fixed\_opaque udt.

### 22.3.3 Support for Inheritance

Inhernance is addressed at two levels in Informix Universit Server: (1) data (antibuo) inhernance and (2) function (operation) inhernance.
Data Inheritance. To create subtypes under existing row types, we use the UNDER keyword as discussed earlier. Consider the following example:

```
CREATE ROW TYPE employee_type (
ename VARCHAR(2S),
ssn (HAR(9),
salary INT) ;
CREATE ROW TYPE engineer_type (
degree VARCHAR(10),
hcense VARCHAR(20))
UNDER employee_type;
CREATE ROW TYPE engr_mgr type (
manager_start_date VARCHAR(10),
dept_managed VARCHAR(20))
UMDER engineer type;
```

The above statements create an employee type and a subtype called engineer type, which represents employees who are engineers and hence inherits all attributes of employees and has additional properties of degree and license. Another type called ergn\_wgn\_type is a subtype under ergnween\_type, and hence inherits from engineer\_type and implicitly from engloyee\_type as well, informs. Universit Server does not support multiple inheritance. We can now create tables called employee, engineer, and engr\_mgr\_based on these row types.

Note that storage options for storing type Ensurchies in tables vary. Information Universal Server provides the option, in store instances in different combinations – for example, one instance (record) at each level or one instance that consolidates all levels – show correspond to the mapping options in Section 7.2. The inherited attributes are either represented represented with a reference to the object of the supertype. The processing of SQL commands is appropriately modified based on the type hierarchy. For example, the query

```
SELECT *
FROM employee
WHERE salary > 100000;
```

n terms the employee intrimutation from off-tables where each selected mightyen is represented. Thus the scope of the employee rable extends to all tuples under employee. As a default, queries on the soperrable return columns from the superrable as well as those from the solvables that inherit from that supertable. In contrast, the query

```
SELECT =
FRON ONLY (employee)
WHERE salary > 300000;
```

returns instances from only the employee table because of the keyword ONLY

It is possible to query a supertable using a correlation carfable so that the result contains not only supertable\_type common of the subtables but also subtype-specific columns of the subtables. Such a query returns rows of different sites: the result is called a jagged row result. Remeving all information about an employed from all levels in a "tagged form" is accomplished by

SFIECT # FROM employee e ;

For each employee, depending on whether he or she is an engineer or some other subtype(s), it will return additional sets of ortributes from the oppropriate subtype tables.

Views defined over supertables cannot be induced because placement of inserted rows is ambiguous.

Function Inheritance. In the same way that data is inherited among tables along a type hierarchy, functions can also be inherited in an URDBAS. For example, a function overpaid may be defined on enployee type to select these employees making a signal salary than Bill Brown as fullows:

```
CREATE FUNCTION overpaid (employee_type)
RETURNS BODLEAN AS
RETURN $1.salary > (SELECT salary
FROM employee
WHERE ename = 18:01 Brown');
```

The tables under the employee table our-granically inherit this function. However, the same function may be redenced for the **engringrit type** as those employees making a higher solars than lack jones as follows:

```
CREATE FUNCTION overpaid (engr mgr type)
RETURNS BOOLEAN AS
RETURN $1.salary > (SELECT kalary
FROM employee
WHERE ename = 'Jack Jones');
```

For example, consider the query

```
SELECT elenare
FROM ONLY (employee) e
NHEAE overpaid (e);
```

which is evaluated with the near definition of overpaul. The query

```
SELECT g.ename
FROM engineer g
WHERE overpaid (g);
```

also uses the first definition of averpaid (because it was not redefined for engineer), whereas

```
SELECT gm.enane
FROM engr_mgr gm
WHERE overcard (gm):
```

uses the second definition of accepted, which averages the test. This is called operation (or function) overloading, as was discussed in Section 20.6 under polynorphism. Not, that overpaid—and other functions—contaise by treated as varied attributes, hence over polynosis by reterenced as employee, overpaid or engright, overpaid in a spore

# 22.3.4 Support for Indexing Extensions

Informix Universal Server supports indexing on user-defined routmes ex either a single table or a rable hierarchy. For example,

CREATE INDEX empl\_city ON employee (city (address));

creates an index on the rable employee using the value of the my function.

In order to support user-defined indexes. Informity Universal Server supports operator classes, which are used to support user-defined data types in the generic B tree as well as other secondary access methods such as R-trees.

# 22.3.5 Support for External Data Source

Informax Universal Server supports external data scores (such is data stored in a file system) that are mapped to a table in the database called the virtual table interface. This interface combles the user to define operations that can be used as proved for the other operations, which are needed to access and manipulate the row or rows associated with the interface data source. These operations include open, close, fetch, ansert, and defete, informax Universed Server also supports a set of functions that enables calling SCL statements within a user defined routine withing the overhead of going through a client interface.

# 22.3.6 Support for Data Blades Application Programming Interface

The Data Blades Application Programming Interface (API) of Informus Universal Server provides new data types and functions for specific types of applications. We will review the extensible data types for two-dimensional operations (required in Gis or 740 applications),<sup>11</sup> the data types related to image storage and monogeneous the time series data type, and a few features of the text data type. The strength of OROBMSs to deal with the new unconventional applications is largely attributed to these special data types and the tolored functionality that they provide.

Two-Dimensional (Spatial) Data Types. For a two-dimensional application, the relevant data types would include the following

- A point defined by (X, Y) coordinates.
- A line defined sents two end points
- A polygon defined by an ordered list of a points that form its vertices.
- A path defined by a sequence (ordered hst) of points
- A circle defined by its center point and radius

11. Recall that GP stands for Geographic Intomation Systems and C4D for Computer Aided. Design Grown the above as data types, a function such as distorted nay be defined between two points, a prioritagid a bine, a bine and a circle, and so on, by implementing the appropriate mathematical expressions for distance in a programming language. Similarly, a Biolean gross function—which returns time or false depending on whether two geometric objects errors (or junnsect)—can be defined between a line and a polygon, a joth and a polygon a line and a circle, and wy on. Other relevant Boulean functions for (a) applications would be workel (polygon), concerns (polygon), concerns (polygon), and so on. Note that the concept of overloading (operation polygon) applies when the some function name is used with different argument types.

Image Data Types. Images are stored in a variety of standard tornats—such is T.R., 1916, J.F.G., photoCD, OSGUE 4, and SAX—so one may define a data type for each of their formats and use appropriate library functions to usput images from other media or in render images for display. Alternately, INTAGE can be regarded as a single data type with a large number of options for storage of data. The latter option world allow a column in a table to be of type 105403 and ver accept images in a variety of different formats. The following are some possible functions forterations) on images:

rotate (image, angle) returns image. crop (image, polygon) returns image. enhance (image) returns image.

The cop function extracts the portion of an image that natersects with a polygon-The *calence* function improves the quaity of an image by performing contast enforcement. Multiple images may be supplied as parameters to the following functions

```
common (image1, image2) returns image.
union (image1, image2) returns image.
similarity (image1, image2) returns number.
```

The sindarus function type ally takes into account the distance by were two vectors with components conton, shape, texture, edges that describe the content of the two junges. The Vik Data Blade in Informix Universal Server can be used to accomplish a search on images by content hand on the above similarity measure.

Time Series Data Type. Informs: Universal Server supports a time series data type that makes the handling of time series data much more simplified than storing (i.i. multiple tables. For example, consider storing the closing stock price on the New York Stock Exchange for more than 3.000 stocks for each workday when the marker is open Such a table can be defined as follows:

```
CREATE TABLE stockarices (
company-name VARCHAR(10),
symbol VARCHAR(3),
prices TIME SCRIES OF FLOAT):
```

Begarding the stock price data for all 2000 companies over an entire remotiof sinseveral years, only one relation is adequate thanks to the time series data type for the prices attribute. Without this data type, each compare would need one table. For example, a table for the coca\_cola compary (symbol KO) may be declated as follows

```
CREATE TABLE code cole (
recording_date DATE.
price FLOAT);
```

In this table, there would be approximately 200 tuples per year - one for each bosiness Jay. The time series data type takes (no account the calendar, starting time, recording interval (for example, daily, weekly, monthly), and so on. Functions such as extracting a subset of the time series (for example, closing prices during January 1999), summation gat a coarser granularity (for example, average weekly closing price from the daily closing prices), and constructing moving averages are appropriate.

A query on the stockprices table that gives the moving overage for 40 days storing at June 1, 1999 for the coca\_col a stock can use the MOVING-AVC function us follows:

```
SELECT NGVING-AVU(prices, 30. 1995-06-01')
FROM Slockprices
MHIRE symbol = "KO":
```

The same query in SQL on the table coca\_cola would be much more complicated to write and would access numerous roples, whereas the above query on the stockprices table deals with a single row in the table corresponding to this company. It is closed that using the time series data type provides an order of magnotude performance gain in processing such queries.

Text Data Type. The text DataBlade supports storage, search, and terrieval for text objects. It defines a single data type called **doc**, whose instances are stored as large objects that belong to the built-in data type large-text. We will builty discuss a tew important teatures of this data type.

The underlying storage for large-text is the stine as that for the large-baject data type. References to a single large object are recorded in the 'refcound' system table, which stores information such as number of news referring to the large object, its OF, its surage manager, its Los modification time, and its an investorage manager. Automotion conversion between large-text and text due, types enables any numerions with four arguments to be applied to large-text objects. Thus concatenation of large-text objects as strings as well as extraction of sub-trings from a large-text object are possible.

The Text DataBlade parameters include format for which the dobult is Ascill, with other posabilities such as noses cript, dv ipests cript, nroff, troff, and text. A Text Conversion DataBlade, which is separate from the Text DataBlade, is needed to convert document- among the various formats. An External File parameter instructs the interval representation of doc to store a pointer to an external file rather than unpying it to a large object.

For manipulation of dat objects, functions such as the following arc used:

```
Import_doc (duc. Text) returns duc.
Export_doc (duc. Lext) returns text.
Assign (doc) returns doc.
Destroy (doc) returns void.
```

The Assign and Destroy functions already exist for the built in Targe-object and Targe-text data types, but they must be redefined by the user for objects of type doc. The toflowing statement creates a table called legaldocurrents, where each now has a tule of the document in use column and the document itself as the other column:

```
CREATE TABLE legaldocuments:
title TEXT,
document COC).
```

In insert a new pow into this table of a document called 'Tease.contract,' for following statement catche used

```
INSERT INTO legaldocuments (Litle, document)
VALUES (')cons.contract', 'format (truff):/user/local/
documents/lease');
```

The second value in the values clause is the path name specifong the file location of the document; the format -predication signifies that **u** is a **treff** document. To searly the text, an index must be charted, as in the following statement:

```
CREATE INDEX `egalindex
ON legaldocuments
USING diree(document lext_ops).
```

In the above, text\_ops is an operface (operior close) upthoddle to an access structure called i dtree index, which is a special index structure for doc memo. When a document of the doc date type is inserted into a table, the text is period into individual words. The Text DataBlade is case inserted into a table, the text is period into individual words. The Text DataBlade is case inserted into a table, the text is period into individual words. The Text DataBlade is case inserted into a table, the text is period into individual words. The Text DataBlade is case inserted word: Words are domined according to the WORDNET thesauras. For example, bouses of boasing would be stemmed to house, quickly to quick, and talked to talk. A stopword file is kept, which contains insertheant words such as articles or preportions that are ignored in the searches Examples of stepwords include is, not, a, the, out, for, and if, and so in

Informs Universal Server provides two sets of routines—the contains routines and text-string functions —to enable applications to determine which documents contain a certain word or words and which documents are used in a starch condition, the data is returned in descending order of how well the rotation matches, the documents, with the best match, showing first. There is weight Contains(index to use, tople, id of the document, input string) function and a smillar WeightContainsWords function that returns a precision number between 0 and 1 indicating the closeness of the match between the input string or input words and the specific document for that ruple-id. To illustrate the use of these functions, consider the following queue Find the other of legal documents that contain the top feit fermion in the observations that contain the top feit fermion in the document integration and a specific document for the totles of legal documents that contain the top feit fermion in the observation integration and a specific document integration of the other of legal documents that contain the top feit fermion in the observation in the operation and a specific document integration of the other of legal documents that contain the top feit fermion in the document integration and a specific document integration of legal documents that contain the top feit fermion in the document integration and its contained in the other of legal documents that contain the top feit fermion in the document integration and its contained in the document integration and its contained in the other of legal documents that contain the top feit fermion in the document integration and its contained in the other of legal documents that contain the top feit fermion in the document integration and its contained in the other of legal documents that contain the top feit fermion in the document integration and its contained in the other of legal documents that contain the top feit fermion in the document in the document in the other of legal documents in the

```
SELECT d.title
FROM "egaldocuments d. legalcocuments)
MHERE contains (d.document, AndTerns (TopMTerns(D.document.10)))
AMD L.title - Tlease.contract' AND d.title <> Thease.contract':
```

This quere fluctures how soft can be enhanced with these data type specific functionto yield a very powerful capability of banding texts related functions. In this quere, variable it refers to the entire legal corpus whereas 1 refers to the specific document whose ritle is "lease.commact'. TopNTerms extracts the top ten terms from the "lease.commact' document (1): AndTerms combines these terms into a list: and contains compares the terms on that list with the stemwords in every other document yd) in the table legaldocuments.

Summary of Data Blades. As we can see, Data Blades enhance an RD565 by providing various constructors for abstract data types (AD18) that allow a user to operate on the data as if it were stored in an v10868 using the AD18 as choses. This makes the relational system federate as an OD5055, and drivingable cuts down the programming effort needed when compared with a bieving the same functionality with just SQL endexided in a programming longuage.

# 22.4 OBJECT-RELATIONAL FEATURES OF ORACLE 8

In this section we will review a number of teatures related to the version of the Oracle 00005 product called Release 8.X. which has been enhanced to incorporate object relational features. Additional teatures may have been incorporated into subsequent versons of Oracle. A number of additional data types with related nonipolation facilities called **carreidges** have been added.<sup>12</sup> For example, the sparad corredge allows mapboard and go graphic information to be handled. Management of moleculed a data has been facilitated with new data types. Here we highlight the differences between the release 8 X of Oracle to acadable at the time of this arisingly from the preceding veraces in terms of the tew object-strenged features and data types is well as some storage options. Fortions of the longitude SQL-99, which we discussed in Section 22.1, will be applicable to Oracle. We do not docuss these features here

## 22.4.1 Some Examples of Object-Relational Features of Oracle

As an OSDIME, Or icle 8 continues to prevale the capabilities of an ICTAM and additionally supports object/oriented concepts. This provides higher levels of abstraction so that application developers can manipulate application objects as uppeed to constructing the objects from relational data. The complex information about an object can be hidden, but the properties (attributes, relationships) and methods (operations) of the object can be identified in the data model. Moreover, object type declarations can be reused to inheritance, rhereby reducing application development rune and effort. To facilitate object modeling. Operle introduced the following formers (as well as some of the soft 39 teatures in Section 23.1).

<sup>2.</sup> Christofges as O to leave section an similar to Data Bodes in Informa-

Representing Multivalued Attributes Using VARRAY. Some armbutes of an object/entity could be multivalued. In the relational model, the multivalued attributes would have to be handled by forming a new table (see Section 7.4) and Section 10.1.2 or first normal form). If ten attributes of a large table were meltivalued, we would have eleven tables generated from a single table after normalization. To get the data back, the developer would have to do ten joins across these tables. This does not happen in an object-model since all the auributes of an object-meliding multivalued one---are encapationed within the object. Oracle S achieves this by using a varying length anal (VARICAY) data type, which has the following properties:

1. COUST: Current number of elements.

2. LIMIT: Moximum number of elements the VAREAN can contain. This is user defined

Consider the example of a customer VAREAN entry with attributes name and phone numbers, where anone\_numbers is multivalual. First, we need to define an object type representing a phone number as follows:

CREATE TYPE phone\_num\_type AS OBJECT (phone number CH4R(10));

Then we define a VARRAT whose elements would be objects of type phone inumitype:

CREATE TYPE phone\_list\_type as VARMAY (5) OF phone\_num\_type.

Now us can create the customer\_type data type as an object with attributes customer\_ name and phone\_numbers:

```
CREATE TYPE customer_type AS
OBJECT (customer name VARCHAR(20),
phone numbers phane_list_type);
```

It is now possible to create the customer softle as

CREATE TABLE customer OF customer type:

To remeve a list of all easterners and their phone nonabers, we can issue a simple query without any jours:

```
SELECT customer_name, phone numbers
FROM customers;
```

Using Nesled Tables to Represent Complex Objects. In object modeling, some attributes of an object could be objects themselves. Oracle 8 accomplishes this by baring nested tables (see Section 20.6). Here, columns (equivalent to object attributes) can be declared as tables. In the above example let us assume that we have a discreption atteched to every phone number (for example, hous, office, cellular). This could be madeled using a nested table by first redefining phone run type as follows:

```
CREATE TYPE ahone rum type AS
OBJECT (phone_number CHAR(10), description CHAR(30));
```

We next redefine phone\_Dist\_type as a rable of phone\_number\_type as follows:

CREATE TYPE phone\_list\_type AS TABLE OF phone number type;

We can then create the type **customer** type and the **customer** table as before. The only difference is that **abone\_list\_type** is now a nested table instead of a VABRAN. Both structures have similar functions with a few differences. Nested tables do not have an upper bound on the number of atoms whereas VABRANS do have a limit. Index dual items can be retrieved from the nested tables, but this is not possible with VABRANS. Additional indexes can also be built on nested tables for foster data access.

Object Views. Object views can be used to build virtual objects from relational data, thereby enabling programmers to evolve existing schemas to support objects. This allows relational and object opplications to cover on the same database. In our example, let us say that we had modeled our customer database using a relational model, but management decided to do all forme applications in the object model. Moving over to the object view of the same existing relational data would thus facilitate the transition.

# 22.4.2 Managing Large Objects and Other Storage Features

Oracle can now store extremely large objects like video, audio, and test documents. New dura types have been introduced for this purpose. These include the following:

- BLOB (binary large ubject)
- CLOB (character Linge object)
- BRILE (Finany file stored outside (he database).
- NULOB (modewidth and trbyte 0109).

All of the above except for BFILE, which is stored outside the database, are stored inside the database along with other data. Only the directory name for a PSUE is stored in the database.

Index Only Tables. Standard Oracle 7.X involves keeping indixes as a B<sup>+</sup>-inee that contains pointers to data blocks (see Chapter 14). This gives good perturmance in most situations. However, both the index and the data block must be accessed to read the data. Moreover, key values are stored twice—in the table and in the index—increasing the storage costs. Oracle 8 supports both the standard indexing scheme and also index only tables, where the data records and index are kept regether in a Betree structure (see Chapter 14). This allows faster data retrieval and requires less storage space for smalls to medium sized the where the record size is not true large.

Partitioned Tables and Indexes. Large tables and indexes can be broken down into smaller partitions. The table new becomes a logical which and the partitions become the actual physical structures that hold the data. This gives the following advantages:

- · Continued data availability in the event of partial failures of some partitions.
- Scalable performance allowing substantial growth in data volumes.
- Overall performance improvement in query and transaction processing.

# 22.5 IMPLEMENTATION AND RELATED ISSUES FOR EXTENDED TYPE SYSTEMS

There are carrons jurplementarion resiles regarding the support of an extended type system with associated rightions (operations). We briefly summarize them here,<sup>11</sup>

- The ORTPARS prior dynomically link a user defined function in its address space only when stars required. As we saw in the case of the two ORTPARS, nonterous functions recrequired to operate on two or three dimensional spatial data images, text, and so on. With a storic linking of all function libraries, the 1,948 address space may increase by an order of magnitude. Dynamic linking is available in the two ORTPARS that we studied.
- Chent-server issues dual with the placement and activation of functions. If the server needs to perform a function, it is best to do so in the DBAs accress space other than remotely, due to the large another of overhead. If the function demands computation that is too intensive or if the server is attending to a very large number of cherns, the server may show the function to a separate chem much nell For security reasons, it is better to run functions at the chern using the user 1.5 of the chern. In the future functions are likely to be written in interpreted languages like (AVA).
- It should be possible to run queries inside functions. A function mast operate the same way whether it is used from an application using the application program interlace (3.09, or whether it is invoked by the ODMS as a pert of excurring SQL with the function embedded in applicy, statement. Systems should support a resting of their "callbacks,"
- Because of the variety in the data types in an ORDERS and associated operators, chicient storage and access of the data is important. For spatial data or including isomadata, new storage structures such as Retries, quad trees, or Cind files may be used. The ORDERS must allow new types to be defined with new access structures. De dug with aree text struggs or binary files also opens up o number of storage and search options. It should be possible to explore such new options by defining new data types within the ORDERS.

Other Issues Concerning Object-Relational Systems. In the above discussion of Informix knowers? Server and Oracle 8, we have concentrated on how an opposite extends the relational model. We discussed the features and furthers in provides to operate on relational dation area discubles as if it were an object database. There are other obvious providents to consider an the context of an CROPAGE.

Object-relational database description. We described a procedure for destening object schemas in Section 21.5. Object-relational desen is more complicated because we have to consider not only the underlying design considerations of application semantics and dependencies in the relational data model (which we discussed in Objects 19.10).

<sup>&</sup>lt;sup>15</sup> The distance of enced length transition dynamical Means (1980).

and 11) but also the object-oriented nature of the extended features that we have just discussed

- Query processing and opnosphere. By extending SQL with functions and rules, this
  problem is further compounded beyond the query optimization overview that we discuts for the relational model in Chapter 15.
- Interaction of rides with transmissions: Rule processing as implied in SQL curvers more than just the update-apdate rules (see Section 24.1), which are implemented in RU9Mss as triggers. Moreover, RU9RUSs currently implement only immediate execution of triggers. A deferred execution of triggers involves additional processing.

# 22.6 THE NESTED RELATIONAL MODEL

To complete this discussion, we summarize in this section on approach that proposes the use of rested tables, also known as connormal form relations. No commercial 00%s has chosen to implement this concept incits original form. The nested relational model consists the testriction of first normal form (1N), see Chapter 111 from the basic relational model, and thus is also known as the Non-INE or Non-First Normal Form (NPNF) or NP<sup>2</sup> relational model. In the basic relational model—attrabutes are required to be single-valued and to have atomic domains. The nested relational model model model in the basic relational model—attrabutes are required to be single-valued and to have atomic domains. The nested relational model allows composite and multivalued attrabutes, this leading to complex tuples with a Eiterarchical structure. This is institution for representing objects that are naturally hierarchically structured. In Figure 22.1, part (a) shows a restrict relation alignment of the factories database, and part (b) gives an example of a Non-INF tuple motor.

To define the 012t schema as a nested structure, we can write the following

```
dept = (duo, dname, manager, employees, projects, 'dcathors)
employees = (ename, dependents)
projects = (ename, ploc)
locations = (dloc)
dependents = (dname, age)
```

First, all attributes of the DPF relation are defined. Next, any nested attributes of DPF—namely, DPFDPEES, PROJECTS, and DECATEON—are themselves defined. Next, any second-level nested attributes, such as DPPDOEN s of DPCDPPES, are defined, and so on. All attribute names must be defined in the nested relation definition. Notice that a rested attribute is tepically a **multivalued composite attribute**, thus leading to a "nested relation" unbit each table. For example, the value of the PPCLED attribute within each case tuple is a relation with two attributes (PPONE, PLOCE in the DPO tuple of Pignie 22 Tb, the WOISC'S attribute contains three tuples as its value. Other nested attributes that he **multivalued simple attributes**, such as DECATEON of DDPO. It is also possible to have a acted attribute that is **single-valued and composite**, although thost nested relational accels treat such in attribute as though it were multivalued.



726 Chapter 22 Object Relational and Extended Relational Systems

FIGURE 22.1 Illustrating a dested relation, fatic-relischema, thi Example of a Non-TNE tuple of any to Theo representation of neuroclassical.

When a rested relational database schema is defined it consists of a muci const external relation schemas these define the top level of the indevidual nested relations. In addition, nested attributes are called internal relation schemas, since they define relational structures that are nested incide another relation. In our example, approximation only external relation. All the others, precises, precise is, nutwities, and of renswises are internal relations. Finally, simple apprintutes appear or the leaf level and are not mericid. We can represent each relation schema by means of a tree structure, as shown in Figure 22 fe, where the wor is an external relation schema), the leaves are simple attributes and the internal nodes are internal telation, schemas. Notice the surplatory between datappresentation and a hierarchical schema (see Appendix E) and VMU (see Chapter 20).

It is important to be aware that the three instrieved bested relations in DEPT representindependent injurnation. Hence, EPPCEPER represents the employees tooking for the department. PROTECTS represents the projects controlled by the department, and cookings represents the various department locations. The relationship between CPPCPTES and PROTECTS is not represented in the schema, this is on M-N relationship, which is difficult to represent in a hierarchical structure.

Extensions to the relational algebra and to the relational calculas, as well as to SQL have been proposed for nested relations. The innere-ted ceater is referred to the selected bibliography at the end of this chapter for details. Here, we illustrate two operations. **NEST** and **DNNEST**, that can be used to angine a standard relational algebra operations for converting between nested and that relations. Calculate the flat event **POD** relation of Figure 11.4, and suppose that we project it over the articlates **SN**. **Sumption 6055**, **EVEN** is follows:

EMP\_PROJ\_FLATE DAYS, LHAMS, PROVERS, REDAS (EMP\_DROC)

To creme a nested version of this relation, where can taple exists to each employee and the Likeware, itsues), its nested, we use the NEST operation as follow-

LMP FREE WESTEDS RESTANCES (PROPERLY DOCUMENT)

The effect of this operation is to create an internal nested relation pairs = (#60981+, +6063) within the external relation on two iscure. Hence, XEST groups together the operations rate rate the second relation on the architect that are not specified in the MeST operation; these are the second relation the architect that are not specified in the MeST operation; these are the second relation of example, is one example. For each such group, which represents one employee in some sample, a single nested replaces done has minimal period relation peaks = 0-900000, -0000000, the for each such group looks like the second relation peaks = 0-900000, -0000000, -00000, -00000, -00000, -00000, -00000, -00000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0000, -0

Nonce the sum any between nesting and georphic for aggregate functions. In the former, each group of tuples becomes a stagle nested tuple, or the latter, each group becomes a single summary tuple after an aggregate function is applied to the group.

The ONNEST operation is the inverse of NAST. We can reconvert use, stor Associations suppress, cost as follows:

Here, the posts misted attribute is flattened into its components 9096151, 9005

# 22.7 SUMMARY

In this chapter, we first give an overview of the objection and effectures in SQL<sup>APE</sup>, which are applied ble to object relational systems. Then we discussed the fustory and current trends in dirabase management systems that left to the development of object relational 09468 (OP/16958). We then focused on some or the features of Informix Universal Server and of Oracle 8 in order to allostrate have commercial RDMMS are Keing extended with object features. Other commercial RDBMSs are providing similar extensions. We saw that these systems also provide Data Blades difformed or Catridges (Oracle) that provide spectfic type extensions for newer application domains, such as spatial, time series or text/document databases. Because of the extendibility of vR14MSs, these packages can be included as abstract data  $r_{\rm TP} + (MD7)$  libraries whenever the asers need to implement the types of applications they support. Over an also implement their own extensions as receded by using the ADT facilities of these systems. We briefly discussed some implement tation issues for ADTs. Finally, we give an overview of the nested relational model with hierarchically sufficience complex objects.

### Selected Bibliography

The references provided for the object oriented database opproach in Chapters 11 and 12 are also relevant for object-relational systems. Stonebroket and Moore (1990) provides a comprehensive reference for object-relational DWLss. The discussion about concepts related to Illustra in that book are mostly applicable to the cortent Informat Universe. Server, Kim (1995) docusses miny using related to undern database systems that include object orientation. For the most cortent information on Information and Oracle, consider their Web area wave information and waveversely control to the rest cortent information on Information.

The SQL5 standard is described in corners publications of the ISC With (Working Group 3) reports for example, see Kulkarni et al. (1995) and Melron et al. (1991). As excellent toronal on SQL5 was given at the Very Large Data Bases Conference by Melon and Martos (1996). Ullimon and Widom (1997) have a good discussion of SQL5 with examples

For issues related to rules and triggers, Wildon and Cen (1995) have a collection st chapters on active database. Some comparative studies—for example, Ketabelin et al. (1990)—compare relational D8055 with object D8055; their conclusion shows the superonty of the object-oriented approach for runconventional applications. The nested relational model is discussed in Schek and Schell (1985). Jacshae and Schek (1982), ethan and Kambavashi (1991), and Makiraonchi (1977), moong others. Algebras and query languages for nested relations are presented in Paredaens and VarQuelin (1982), Pistor and Aralersen (1986), Roth et al. (1988), and Ossoyogla et al. (1987), among others, high memorian of prepaying nested relational systems is described in Datam et al. (1986). Deshpande and VarGuebi (1988), and Schek and Scholl (1989).







# Database Security and Authorization

This chapter discusses the techniques used for protecting the database against persons into are not authorized to access either certain pairs of a database or the whole database. Socion 23.1 provides an introduction to security issues and the threats to databases and an overview of the confirminations from no covered on the rest of this chapter. Section 23.2 discusses the mechanisms used to grant and recoke privileges in relational database systems and in SQL incohanisms that are often referred to as discretionary access control. Section 23.3 offers an overview of the mechanisms for enforcing multiple levels of security—a more recent concern in database system security that is known as mandatory access control. It also introduces the more recently developed strategy of role-based access control. Section 23.4 briefly discusses the security problem is statistical databases. Section 23.5 introduces flow control and mentions problems associated with covert channels. Section 23.6 is a brief sommary of encryption and public key infrastructure schemes. Section 23.6 is a brief sommary of encryption and public key infrastructure schemes. Section 23.6 is a brief sommary of encryption and public key infrastructure schemes. Section 23.6 is a brief sommary of encryption and public key infrastructure schemes. Section 23.6 is a brief sommary of encryption and public key infrastructure schemes. Section 23.7 summaries the chapter. Readers who are interested only in basic database security mechanisms will find it sufficient to cover the material in Sections 23.1 and 23.2.

# 23.1 INTRODUCTION TO DATABASE SECURITY ISSUES

# 23.1.1 Types of Security

Database security is a very bread area that addresses many issues, including the following:

- Legal and efficial issues regarding the right to access contain information. Some information may be deemed to be private and contrast be accessed legally by or authorized persons. In the United States, there are nanorosis lows governing privacy of information.
- Folicy issues at the governmental matigarional, or engrence level as mewhat kinds a minimation should not be made publicly available – for example, credit ratings and personal medical records.
- bystemized nod issues such as the system knetch, which various security functions should be entireed. For example, whether a security function should be bondled as the physical hardware level, the operating system level, or the DBAF level.
- The need nessone organizations to recently multiple secondly forels and to categorize the data and users based on these classifications—for example, top secret, secret, our fidential, and anelosified. The security policy of the organization with respect to permitting access to corrow classifications of data must be enforced.

Threads to Databases. Threats to databases result in the loss of degradation of sime or all of the following security goths integrity, availability, and confidentiality.

- Los of (negaty: Database integraty refers to the requirement that information be posteded from improper modification. Modifications of data includes creation, insertion, modification, changing the status of data, and deletion. Integrity is lost if anothorized charges are made to the data by either intentional or accidential acts. If the loss of system or data integrity is not corrected, continued use of the contanimated system or compted data could result in machinesy, fraud, or errorecus decisions.
- Low of autidality. Detabase availability refers to making objects available to a human user or a program to which they have a leptmente right.
- Low of conjudentiality. Database confidentiality refers to the protection of data how unsutherized disclosure. The impact of upportherized disclosure of confidential information can range from confuriou of the Durp Providy Act to the popardigation of national security. Unauthorized, unauticipated, or an informational disclosure could reach its loss of public confidence, embarrossment, or legal action against the organization.

To protect databases against these types of threats four kinds of counter detautes can be nuplements a access counted interface control, flow control, and energy ion. We discuss each of these in this chapter.

In a multisser database system, the 0886s must provide rechniques to enable certain overs of user groups to access selected particles of a database without gaining access of the rest of the database. This is porticularly important when a large integrated database is to be used by many different users within the same organization. For example, sensitive information such as cooployee salaries of period nance reviews should be kept confidented from most of the database system's users. A 180018 typically includes a **database security** and **authorization subsystem** that as responsible for ensuring the security of portions of a database against anarchisticed access of a new customary to refer to two types of database accurity mechanisms:

- Discretionery sciency mechanisms. These are used to grant privileges to users, including the capability to access specific data files, records, or fields in a specified mode (such as read, insert, delete, or update).
- Mendalozy secondy mechanisms: These are used to enforce multilevel seconcy by classifying the data and users into various security classes (or levels) and then implementing the appropriate security policy of the organization. For example, a typical security policy is to permit users in a certain classification level to see only the data memory classified at the users own (in lower) classification level. An extension of this is tokylored security, which enforces pulsees and privalenes based on the concept of roles.

We discuss discretionary scentre in Section 23.7 and non-dulory and rele-based accuration Section 23.5.

A second scenity problem common to all computes secteors is that of presenting arouthorized persons from accosing the system itself, either to obtain intermation or to make molecous changes in a portrox of the database. The seconty mechanism of a DPMS most include provisions for restricting access to the database system as a whole. This function is called access control and is builded by creating accounts and passwords to centrol the zero process by the DRMS. We discuss access control techniques in Section 23.1.3.

A third scenary problem associated with databases is that of controlling the access to a statistical database, which is used to provide statistical information or summaries of values based on various criteria. For example, a dotabase for population statistics may provide statistical chirabase for age groups, incent levels, size of household, education levels, and other criteria. Suitistical database users such as government statisticans or nucleit research time are illeneed to access the database to retrieve statistical information about a population but not to access the database to retrieve statistical information about a population but not to access the database to retrieve statistical information about a population but not to access the database to retrieve statistical information about a population but not to access the database to retrieve statistical information about a population but not to access the database to retrieve statistical information access the database to retrieve statistical information about a population but not to access the database to retrieve statistical information about a population but not to access the database to retrieve statistical information about a population but not to access the database to retrieve statistical on andordalistic must be accessed. It is sometimes possible to deduce or infer certain facts concepting individuals transqueries that are obvious only summary statistical database accurity, is docased by define to be permitted either. This problem, called statistical database accurity, is docased by define more statistical content of access are called inference control measures.

Another security oscents that of flow control, which prevents intermation from this ing in such a way that it reaches unauthorized users. It is doctored in Section 21.5, Ovarnels that are pathways for information to flow implicitly in ways that violate the security policy of an organization are called covert channels. We briefly docuss some re-nes related to covert channels in Section 23.5.1

A final seconicy issue is data encryption, which is used to protect sensitive data (such as used), and number-1 that is being transmitted to a some type of communications network. Encryption can be used to provide additional protection for sensitive portions of a database as well. The data is encoded using some coding olg matin. An unanthomet user who accesses encoded data will have difficulty deephering it, but authorized users are given deceding or decrypting algorithms (or keys) to decipher the dotal Encrypting techniques that are verdifficult to decode without is key have been developed for military applications. Section 35.6 briefly discusses encryption rechniques, including popular techniques such 52 public key encryption, which is heavily used to support. Web-based transactions against databases and digital signatures, which are used in personal communications.

A complete discussion of security in computer systems and databases is outside the scope of this textback. We give only a brief overciew of database security techniques here. The interested reader can refer to several of the references discussed in the solecied bibliography in the end of this chapter for a more comprehensive discussion.

### 23.1.2 Database Security and the DBA

As we discussed in Chapter 1, the database administrator (0.05) is the central authomy for managing a database system. The 2005's responsibilities include granting priodeges to users who need to ase the system and classifying users and data in accordance with the policy of the argumanton. The PPA has a **DBA** account in the P900's sometimes called a system or **superinser account**, which provides powerful capabilities that are not male available to regular database accounts and users.<sup>1</sup> DBA-privileged commands include commands for granting and revolving privileges to individual accounts, or user groups and for performing the following types of actions.

- Account measure This action creates a new account and passwerd for a user or a group of users in smaller access to the DBMS.
- Prindege granting. This action permits the 1988 to grant certain priodeges to contain accounts.
- Printegenees attain This action permits the 1988 to revoke (concel) certain provideges that were previousle given to certain accounts.
- Security ferral assignment. This action consists of assigning user accounts to the appropriate security classification level.

The DBA is responsible for the overall access to the database system. Action 1 in the preceding list is used to control access to the DBMS as a whole, whereas actions 3 and 3 are used to control discretionary database authorization, and action. 4 is used to control manifolding authorization.

### 23.1.3 Access Protection, User Accounts, and Database Audits

Whenever a pursiance a group of persons needs to access a database system, the individual or group must first apply for a user account. The DBA will then create a new account

 $<sup>\</sup>sim$  This account is subfor to the out-on-sign as counts that are given to computer systems shows intersections, allowing access to restricted oper duty system counters.

muchee and password for the user of there is a legitimate riced to acress the database. The user must lug in to the DBMS by entering the account number and password whenever database access is needed. The DBMS breaches that the account number and password are solid; if they are, the user is perimited to use the DBMS and rouccess the database. Application programs can also be considered as users and can be required to supply posswords.

It is straightforward to keep track of database users and their regions, and passwords by creating an energited table or file with the two fields. AccountNumber and Password. This table can easily be maintained by the DBMS. Whenever a new account is created, a new record is inserted after the table. When an account is canceled, the corresponding record must be deleted from the table.

The database system must also keep track of all operations on the database that are applied by a contain user throughout each login session, which consists of the sequence of Jatabase interactions that a user performit from the time of logging active the time of logging off. When a user logs an the COD's can record the user's account tumber and assessment with the terminal from which the user logged in. All operations applied from that terminal are attributed to the user's account tart the user logs off. It is particularly important to keep track of addate operations that are applied to the database so that, if the database is tampeted with, the DOC can find out which user did the tampeting.

To keep a record of all updates applied to the database and of the particular user who sophed each update, we can modify the system kg. Recall from Chapters 17 and 19 that the **system** log includes an entry for each operation applied to the database that may be required for recovery from a transaction failute or system crash. We can expend the log entries so that they also include the account number of the user and the induce terminal 00 that applied each operation recorded at the log. If are transpring with the database is suspected, a database audit is performed, which consists of reviewing the log to examine all accesses and operations applied to the database during a certain time period. When an illegal or anotherized operation is found, the Disk can determine the account number used to perform the operation. Database much are particularly important for sensitive databases that are updated by many transactions and users, such as a bunking database that is updated by many transactions and users, such as a bunking database that is updated by many transactions and users, such as a bunking database that is updated by many transactions and users, such as a bunking database that is updated by many bank tellers. A database log that is used many for security purposes is sumetimes called an **audit trail**.

# 23.2 DISCRETIONARY ACCESS CONTROL BASED ON GRANTING AND REVOKING PRIVILEGES

The typical method of erforcing discretionary access control in a database system is based on the granting and recoking of privileges. Let us consider privileges in the context of a relational 1984s. In particular, we will discuss a system of privileges somewhat similar to the one only a nally developed for the SQL language (see Chapter 8). More correct relational (1986s) as some contation of this technique. The main idea is to include statements in the query language that adove the Dis V and selected users to grant and reviewe privileges.

### 23.2.1 Types of Discretionary Privileges

In SQL2, the concept of an authorization identifier is used to refer, roughly speaking, to a user account (or group of user accounts). For simplicity, we will use the words asce or account interchangeably in place of authorization identifier. The DBMS must private selective access to each relation in the database based on specific accounts. Operations may also be controlled: thus, having an account does not necessarily untille the account holder to all the functionality privated by the 19MS, bitornally, there are two levels for assigning privateges to use the database system.

- The account feeds As this level, the DBA specifies the particular privileges that each account holds independently of the relations in the database.
- The relation (or juble) level: At this level, the D6A run control the provlege to access each individual relation of view in the database.

The privileges at the **account** level apply to the copabilities provided to the account itself and can include the CREATE SCHEWY or CREATE 1400 e privilege, to create a schema or base relations the CREATE SCHEWY or CREATE 1400 e privilege, to create a schema changes such as adding or removing attributes from relations: the F00 ° privilege, to delete relations or views; the MODULY privilege, to insert, delete, or update impless and the SELECT privilege, to retrieve information from the database by using a SLENT query. Notice that these account privileges apply to the account in general. If a certain account does not have the CREATE 7.08EE privilege, no relations can be created from that account Account level privileges are not defined as part of SQL2; they are left to the 1605 implementers to define. In carlier versions of SQL is CREATETAR privilege costed to great an account the privilege to create tables (relations).

The second level of provideges applies to the relation level, whether they are base relations or virtual (view) relations. These provideges are defined for SQL2, he the following disconjoist, the term relation may return other to a base relation level specify for each user the individual relations on which each uppe of command can be applied. Some provideges also refer to individual columns (artributed of relations, SQL2 commands provide provideges of the rolation modernities and with the relation provideges are the rolation and artribute level only. Although this is quite general, a makes in difficult to create accounts with limited provideges. The growing and revoking or provideges generally tollow an authorization model for discretionary provideges known is the access matrix model, where the rows of a matrix M represent subjects (asers, accounts, programs) and the columns represent objects (relations, records, columns, trees, operations), both position M0(, 0) in the matrix (relations, the types of provideges (read, write, update) that subject i holds on object *j*.

To control the granting and revoking of relation produces, cach relation *R* in a database is a-signed on **conner account**, which is typically the occurnit that was used when the relation was created in the first place. The corner of a relation is given all privileges on that relation. In 8002, the D6X can assign an owner to a whole schema by creating the schema and associating the appropriate authorization identifier with that schema, using the CREATE SUBSY A command (see Section 5.1-1). The owner account hold rical pass privileges on any of the owned relations to other users by granting privileges to their

schunts. In sQt the following types of privileges can be granned on each individual relation  $R_{\rm c}$ 

- SELECT (retrieval or read) providege on R: Gives the screent corrieval providege. In SQL this gives the account the providege to use the SELECT statement to retrieve highestrom R.
- MODE provideges on R: This provide account the capability to modely riples of R. In -QUTH's privilege is further divided into 100551-, DELETE, and INSER - privileges to apply the corresponding SQL command to K. In addition, Forb the INSERT and UPDATE privileges can specify that only certain armbures of R can be updated by the account.
- REFERENCES privilege on R: This gives the account the capability to reference relation R when specifying integrity constraints. This privilege can also be assured to specific attributes of R.

Notice that to create a view, the account must have "ELECT privilege on all relations involved in the view definitions.

# 23.2.2 Specifying Privileges Using Views

The mechanism of views is an important discretionary authorization mechanism in its iown right. For example, if the owner A of a relation *R* wants another account B to be able to refrieve only some fields of *R*, then A can create a view V of *R* that mefudes only these officiency and their grant SELECT on V to B. The same applies to finiting B to refressing only certain tuples of *R*: a view V<sup>2</sup> can be created by defining the view by means of a query that selects only those tuples from *R* that A worits to allow B to access. We shall illustrate this discussion with the example given in Section 23-2.5.

# 23.2.3 Revoking Privileges

In some cases it is desirable to grant a privilege to a user temporarily. For example, the owner of a relation may want to grant the SEUS 7 privilege to a user for respective task and then tevoke that privilege once the task is completed. Hence, a mechanism for revoking privileges is needed. In SQL a REVORE command is used in the purpose of careching privileges. We will see how the REVORE command is used in the example in Section 23.2.5.

# 23.2.4 Propagation of Privileges Using the GRANT OPTION

Whenever the owner A of a relation K grants a privilege on K to another account B, the privilege can be given to B with or tollow the GRANT OPTION. If the GRANT OPTION is given, this means that B can also grant that privilege on R to other accounts. Suppose that B is given the GRANT OPTION by A and that B then grants the privilege on R to a

third account (d) also with CEANT CETION. In this way, provideges of: R can propagate to ether accounts without the knowledge of the mener of *R*. If the evener account **A** moviewides the privilege granted to B, all the privileges that B propagated based on that privilege should automatically be revealed by the system.

It is possible for a fiscilito receive a dortatin providege from two or more sources, for example: A4 may receive a cottain CHEATER providege from both A2 and A3. In such a case, if A2 revokes this providege from A4, A4 will still continue to have the priordege bound of having been erasted in from A3. If A3 later revokes the priordege from A4. A4 totally loses the providege. Hence, a 990ts that allows provagition of priordeges from K4. A4 totally loses the priordege. Hence, a 990ts that allows provagition of priordeges from K4. A4 totally loses the priordege. Hence, a 990ts that allows provagition of priordeges from K4. A4 totally and completely.

### 23.2.5 An Example

Suppose that the Covereston accounts I A1, A2, A3, and A4. sand wants only A1 to be able to create have relations then the 1994 must issue the following ORANT communitin 5.2.

#### GRANT CREATETAB TO AL;

The CREATED VS (create rable) privilege gives account A1 the capability to create new database rables (base relations) and is *bence* an *account privilege*. This privilege was part of earlier versions at SQL bar is now left to each individual system implement must to define

In sqt2, the same officer can be accomplished by having the 191A issue a C00ATF SCTEMA command, as follows.

#### CREATE SCHEMA EXAMPLE AUTHORIZATION A1;

Now user account All car create tables under the scheme called (2009). To continue on example, suppose that All creates the two base relations (2009), and interest shown in Figure 23-1. All is then the owner of these two relations and hence has all its relation prosloger on each of them.

Next, suppose that account AT salets to grow to account A2 the prvilege to josen and delete tuples in both of these relations. However, A1 sloes not want A2 to be able to propagate these privileges to addit on a account. AT can issue the following command-

```
GRANT INSERT, DELETE ON EMPLOYEE, DEPARTMENT TO A2.
```

```
EMPLOYEE.
```

NAME SSN ODATE ADDRESS SEX SALARY DAO

#### DEPARTMENT

CNUMBER DNAME MGRSSN

FIGURE 23.1. Schemas for the two relations income and organises .

Notice that the owner account A4 of a relation automatically has the oRANT OPTION, allowing it to grant provideges on the relation to other accounts. However, account A2 cannot grant INMERT and CLULTE provideges on the EMPROSE and EMPROSES tables, because A2 was not given the obANT contents in the preceding communed.

Next, suppose that AT wants to allow account A5 to retrieve information from either is the two tables and also to be able to prepagate the SLILET providege to other accounts. A1 can assue the following command:

#### GRANT SELECT ON EMPLOYEE, DEPARTMENT TO AS WITH GRANT OPTION;

The clouse WITH CRANT OF HON means that A 5 can now propagate the privilege to other accounts twasing GRANT for example. A 5 can graft the SELSCE privilege on the exact of column to A4 by assumption following commands.

```
GRANT SELECT ON EMPLOYEE TO A4;
```

Notice that A4 connect propagate the SELFUT privilegy to other accounts because the GRANT OFTION was not given to A4.

Now suppose that A1 decides to revoke the SELECT privilege on the energy relation from A3, A1 then can issue this command:

#### REWOKE SELECT ON EMPLOYEE FROM A3;

The DDMS must now outomatically revoke the SELICT produce on EMPLONE from A4, too, because A3 granted that privilege to A4, and A3 does not have the privilege any more.

Next, suppose that A1 wants to give back to A3 a limited capability to skiller from the  $\mathbf{E}^{*}$  core relation and wants to allow A3 to be able to propagate the privilege. The limitation is to retrieve only the warf, navie, and 305525 attributes and only for the tuples with 1680 = 5. A1 then can create the following view:

```
CREATE VIEW AREHPLOYEE AS
SELECT NAME, BUATE, AUDRESS
FROM EMPLOYEE
WHERE DNO - S;
```

After the view is created, All can grapt SELECT on the view A30000000 to A313 follows:

GRANE SELECT ON AREMPLOYEE TO AR WITH GRANT OPTION;

Finally, suggesse that AT wants to allow A4 to update only the SMARY attribute of 6990567; A1 tau then issue the following command:

#### GRANT UPDATE ON EMPLOYEE (SALARY) TO 44;

The OPI-WTL or INSERT privilege can specify particular attributes that may be optimized or inserted in a relation. Other privileges (SELECT: DEFETFF are not attribute specific, because this specificity can easily be controlled by creating the appropriate views that include only the desired attributes and granting the corresponding privileges on the views. However, because updating views is not always possible (see Creater 9), the OFPATE and INSUET privileges are given the option to specify particular attributes of a base relation that may be updated.

### 23.2.6 Specifying Limits on Propagation of Privileges

Techniques to limit the propagation of privileges have been developed, although the base not verbeen implemented in most 29:08 and are not a pair of 363. Linuxing homzontal propagation to an integer number random schedular, account B given the OSBN OFTON can grant the privilege to at most rucher accounts. Vertical propagation is more complicated: it limits the depth of the granting of provideges. Granting a privilege with an observation propagation of zero is equivalent to granting the privilege with an OSBN OPTON. If account A grants a privilege to account B with the vertical propagation servation integer number of account B has the OSANT OPTON. If account A grants a privilege rot account B with the vertical propagation servation integer number j > 0, this means that the account B has the OSANT OPTON on that privileges has B can grant the privilege rot other accounts only with a vertical propagation that provideges has B can grant the privilege rot other accounts only with a vertical propagation between the privilege rot other accounts only with a vertical propagation between privileges to other accounts only with a vertical propagation between the given from one account to the next based on a ongle original grant of the privilege.

We now briefly illustrate horizontal and vertical propagation finites—which are net available currently in SQL or other relational systems—with an example. Suppose that Algrants SERCL to AZ on the oscial relation with har coural propagation equal to L and vertical propagation equal to 2. All can then grant SERCL a current one account because the horizontal propagation limitation is set to 1. In addition, A2 cannot grant the privilege to another account except with vertical propagation by at least 1 when possing the providege to others. As this example shows borizontal and vertical propagation techniques are designed to limit the prepagation of provideges.

# 23.3 MANDATORY ACCESS CONTROL AND ROLE-BASED ACCESS CONTROL FOR MULTILEVEL SECURITY<sup>2</sup>

The discretionary access control technique of granting and revoking privileges in relations has traditionally been the main security mechanism for relational database systems. This is an albor-nothing method: A user either has or does not have a certain privilege, in many applications, an additional security policy is needed that classifies data and users based on security classes. This approach, known as **mandatory access** control, would typ neally be combined with the discretion are needed incoharmanic described in Section 23.2. It is important to more that most commercial towers currently provide mechanismerity for discretionary access control. However, the need for multilevel security, exists in

<sup>2.</sup> The somethation of Furlying Farahatond to this and sub-relation sections is oppressived.

government, military and multigence applications, as well as at many industrial and corporce applications.

Expected vectority classes are top secret (TS), where (S), confidential (C), and inclassified (U), where TS is the highest level and U the lowest. Other more complex second, classification schemes exist, in which the second classes are organized in a lattice. For simplicity, we will use the system with four security classification levels, where  $1S \ge S \ge C \ge U$ , to illustrate our discussion. The commonly used inside for multilevel security, knower as the Bell-LaPadula mixtel, classifies each subject (user, account program) and object (relation, tuple, column, view, operation) into one of the security classification TS, S, C, or C). We will refer to the clearance (classification) of a subject  $\Delta$ as class(**S**) and to the classification of an object O as class(**O**). Two testrictions are intered on data accessionsil on the subject/dependence:

- A subject S is not allowed read access to an object O tailen class(S) ≥ class(O). This is known as the simple security property.
- A subject S is not allowed to write an object O onless class(S) ≤ class(O). This is known as the star property (or \*-property).

The first restriction is intoitive and enforces the obvious rule that no subject on read an object whose second galaxification is higher than the subject's security electance. The second restriction is less arrantive. It prohibits a subject from writing an object at a lower actuity classification than the subject's security electance. Violation of this rule would allow information to flow from higher to lower classifications, which violates a basic tener of multificate security. For example, a user (subject) with TS clearance may make a copy of an object with classification TS and their write it back as a new object with classification U, thus making it violate throughout the systems.

To incorporate inclutivel security narrors into the relational database model, it is community consider attribute values and tiples as data objects. Hence, each attribute A passociates with a classification attribute C on the schema, and car's attribute value in a ruple is associated with a corresponding security classification. In addition, in some inside, a tuple classification attribute TC is added to the relation attributes to provide a classification for each tople as a whole. Hence, a multilevel relation schema R with a surface swould be represented as

 $R(A_1, C_1, A_2, C_1, \dots, A_m, C_m, TC)$ 

where each C, represents the desoficiation application 
The value of the WU attribute in each tuple to-which is the toghest of all attribute classification values within t--provides a general classification for the tuple itself, whereas each C, provides a finer security classification are each attribute value within the tuple. The **apparent key** of a mathilevel relation is the set of attributes that would have torned the primary key in a regular (single-level) relation. A multilevel relation will appear to contain different data to subjects (ascis) with different classification levels, in some cases, it is possible responding toples at a lower-level classification through a process known as filtering. In other cases, it is necessary to store two or more tuples of different classification levels with the same value for the appearation through a process known as filtering. In other cases, it is necessary to store two or more tuples of different classification levels with the same value for the appearation level tuples of different classification levels with the same value for the approximation.

6-21

EMPLOYEE

polyinstantiation,<sup>1</sup> where several uples can have the same apparent key value but have different attribute values for users at different classification levels.

We illustrate show concepts with the sample example of a multilevel relation shown in Figure 23.2a, where we display the classification attribute values next to each attribute's value. Assume that the Name attribute is the apparent key, and consider the query sILELT? \* HOOM employee. A user with security charance S would see the same relation shown in Figure 23.2a, since all tuple classifications are less than or equal to S Provever, a user with security classifications are less than or equal to S Provever, a user with security classification. The tuple of Brown and JobPerformance of Smith, since they have higher classification. The tuplewould be filtered to appear as shown in Figure 23.2b, with Salary and JobPerformance.

| Name                                                                     | Salary                                                          | JobPadomianda                                                                 | тс          |
|--------------------------------------------------------------------------|-----------------------------------------------------------------|-------------------------------------------------------------------------------|-------------|
| Smith II                                                                 | 40000 C                                                         | Fair S                                                                        | S           |
| Brown C                                                                  | 80000 S                                                         | Good C                                                                        | s           |
| EMPLOYEE                                                                 |                                                                 |                                                                               |             |
| Name                                                                     | Salary                                                          | JobFerformance                                                                | , ïc        |
| Smith L                                                                  | 40000 C                                                         | nul G                                                                         | c           |
|                                                                          |                                                                 |                                                                               |             |
| Brown C<br>EMPLOYEE                                                      | nul C                                                           | Good C                                                                        | Ċ           |
| Biown C<br>EMPLOYEE                                                      | nul C<br>Sələry                                                 | Goad C<br>JotPeriormance                                                      | с<br>то     |
| Biown C<br>EMPLOYEE<br>Name<br>South U                                   | nul C<br>Sələry<br>nul U                                        | Goad C<br>JotPerformance<br>wull J                                            | с<br>т:<br> |
| Brown C<br>EMPLOYEE<br>Name<br>Sauth U<br>EMFLOYEE                       | nul C<br>Sələry<br>nul U                                        | Gnad C<br>JotPeriormance                                                      | с<br>тс<br> |
| Biown C<br>EMPLOYEE<br>Name<br>South U<br>EMFLOYEE<br>Name               | nul C<br>Sələry<br>nul U<br>Sələry                              | Gnad C<br>JobPerformance<br>rull J<br>JobPerformance                          |             |
| Biown C<br>EMPLOYEE<br>Name<br>South U<br>EMFLOYEE<br>Name<br>Smith U    | nul C<br>Salary<br>nul U<br><u>Salary</u><br>40000 C            | Gnad C<br>JotPerformance<br>rull J<br>JobPerformance<br>Fair S                |             |
| Brown C<br>EMPLOYEE<br>South U<br>EMFLOYEE<br>Name<br>Smith U<br>Smith U | Nul C<br>Selary<br>Nul U<br><u>Salary</u><br>40900 C<br>40900 C | Good C<br>JotPerformance<br>wull J<br>JobPerformance<br>Fair S<br>Excellent C |             |

**FIGURE 23.2** A multilevel relation to illustrate multilevel security, (a) the original secure toples, dot Appearance of exercise after filtering for classification Clusters, (c) Appearance of exercise after filtering for classification Clusters, (d) Polyanstantiation of the Smith tople

k This is stipped to the notice of Lexing colliple versions in the database that a present the serie is d-world object.

appending as null. For a user with secondy doctrance UI, the thering allows only the Bane authore of Smith rotation, with all the other artichutes appearing as null (Figure 23.20). Thus, filtering artification null values for attribute values whose security classification is higher than the overs secondly clearance.

In general, the entity integrity rule for multilevel relations stores that all articloses that are members of the apparent key must not be null and must have the some secon to classification within each individual tuple. In addition, all other attribute values in the tuple must have a security classification greater than or equal to that of the apparent key. This constraint ensures that a user can see the key of the over its permitted to see any parts of the tuple at all. Other integrity rules, called **null integrity** and **interinstance integrity**, miorimally ensure that it is policient to store security level can be filtered to be all to the all apple to be some security level can be filtered tuple in the individuel tuple, then it is sufficient to store the higher-classified tuple in the individuel tuple, then it is sufficient to store the higher-classified tuple in the individuel tuple.

To illustrate polynistration further, suppose that mose with secondly clearance C mes to update the value of JonPerformance of Smith in Figure 23.2 to "Excellent": this formsponds to the following SQL update being issued:

```
UPDATE EMPLOYEE
SET JobPerformance - "Excellent"
MMERE Aane = "Sm)th"
```

Since the crow principal to users with securicy charance C (see Figure 23.2b) permits such an update, the system should not reject it inductions, the user could inter that some normall value exists for the TobPerformance attribute of Smith rather than the null value dust appears. This is on example of inferring unformation through what is known as a covert channel, which should not be permitted in highly secure systems (see Section 23.5.1). However, the user should not be allowed to incurvate the existing value of JabPerformance in the higher classification level. The solution is to create a polyinstantiation for the Smith ruple as the lower classification level C, as shown in Figure 23.2d This is necessary since the new ruple curpor be filtered from the existing tople in classification S.

The basic update operations of the refer and model (users defere apdate) must be moduled to handle this and surdar structure but this aspect of the problem is auside the scope of our presentation. We refer the interested reader to the end-of-chapter sublicesphyrite duratis

### 23.3.1 Comparing Discretionary Access Control and Mandatory Access Control

Discretionary Access Control (DAC) policies are characterized by a high degree of flexibility, which makes them suitable for a large variety of application domains. The main drawbrick of DAC models is their vulnerability to incherous attacks, such as from horses encedded in application programs. The reason is that discretionary authomation models do nor impose any control on how information is propagated and used once it has been received by users authorized to do so. By contrast, mandatory policies ensure a high degree of protections, in a way, they prevent any illegal flow of information. They are therefore sprable for inditary types of applications, which require a high degree is protection. However, mandatory policies have the drawback of being too rigid in that they require a strict classification of subjects and objects into security levels, and therefore they are applicable to very few environments. In noisy practical situations, discretionary policies are preferred because they often a better trade off between security and applicables.

### 23.3.2 Role-Based Access Control

Rele-based access control (RBAC) emerged ispicily in the 1860s as a proven technology for managing und entoring security in large-scale enterprisewide systems. Its basic netion is that permissions are associated with relex, and users are assigned to appropriate roles. Roles can be created using the CREATE BOLE and DESTROY ROLE commands. The ORANT and REVOKE commands discussed under DAC can then be used to assign and twoke produces from rules.

RBAC appears to be a vijable abrimative to traditional discreminary and mandatory access controls in ensures that only authorized users are given access to currain data in resources. Users create sessions during which they may activate a subject of roles to which they belong. Each session can be assigned to many roles, but it maps to only one user or a single subject. Many DBMess have allowed the concept of roles, where proclease can be assigned to roles.

Role hierarchy in RPAC is a normal way of organizing roles to reflect the organization's lines of authomy and responsibility. By concentions jumps mass at the bottom are connected to progressively senior roles as one moves up the linerarchy. The hierarchic diagrams are partial orders, so they are reflexive, transitive, and antisymmetric

Another important consideration in RBAC sectors is the possible temporal constraints that may exist on roles, such as the time and duration of role activations, and timed integering of a role by an activation of another role. Using an RBAC model is a highly desirable goal for addressing the key security requirements at Web basid applications. Enles can be usingted to workflore tasks so that a user with and all the roles rolated to a visk may be ontherized to execute it and may planta certain role for a certain duration only.

RBAC models have several desirable features, such as flexibility, policy neutralitybetter support for security management and idumnistration, and other aspects that make them attractive candidates for developing source. Web-based applications. In contrast DAC and mandatory access control UMAC3 models lack capabilities needed to support the security requirements of emerging enterprises and Web-based applications. In addition, RBAC models can represent modificational DAC and MAC publicies as well as userdefined its organization operator produces. Thus, RBAC becomes a superset model that can in turn immis, the behavior of DACI and MAC systems. Surthermore, an RBAC model provides a manufal mechanism for addressing the security issues related to the execution of tasks and workflows. Easiep deployment over the Internet has been another reason for me success of RBAC models.

## 23.3.3 Access Control Policies for E-Commerce and the Web

Electronic commerce (E-commerce) environments are characterized by invitransactions that are done electronically. They require elaborate access control policies that go boyond todational DBMSs. In conventional database environments, access control is usually performed usive a set of authorizations stated by scennety officers of users occording to some security policies. Such a simple paradigm is not well suited for a dynamic environment like occuminerce. Furthermore, in an eccommerce environment the resources to be protected are not only traditional data but also knowledge and experience. Such pendianties call for more flexibility in specifying increas control policies. The access control mechanism must be flexible enough to support a write spectrum of heterogeneous protection objects.

A second related requirement is the support for content-based scress control. Content-based acress control allows one to express across control policies that take the protection object content into acroant. In order to support content-based access control, across control policies must allow inclusion of conditions based on the object content

A third requirement is related to the herersgeneity of subjects, which requires access control policies based on user characteristics and qualifications rather than on very specific and individual characteristics (e.e., user IDN). A possible solution, to better take into account eser profiles in the formulation of access control policies, is to support the notion of credentials. A credential is a set of properties concerning a user that are relevant for security purposes (for estateple, lage, position within an organization). For instance, by using credentials, one can simply formulate colleres such as "Colly permatent staff with 5 or more with of service can caces documents related to the internals of the system."

It is believed that the XML language can plot a key role in access control for ecommune applications.<sup>4</sup> The reason is that XML is becoming the common representation language for document interchange over the Web, and is also becoming the language for e-commerce. Thus, on the one hand there is the need to make XML representations secure, by previding needs control incluments specifically railored to the protection of XML documents. On the other hand, access control information (that is access control policies and user credenticals) can be expressed using XML useff. The Directory Service Markup funguage provides a number for this, a standard for communicating with the directory services that will be responsible for providing and authenticating over credentials. The uniform presentation of both protection objects and access control policies can be applied to policies and credentials themselves. For instance, some credential properties tsuch as the user named may be accessible to everyone, whereas other properties may be visible only to a restricted class of osers. Additionally, the use of an XML-based language for specifying credentials and access control policies secure credential submission and expert of access control policies.

# 23.4 INTRODUCTION TO STATISTICAL DATABASE SECURITY

A population is a set of tiples of a relation (rabbel that satisfy some selective condition. Hence each selection condition on the PEESSA relation will specify a particular population at PEESSA toples. For example, the condition PEE = -PEE specifies the male population; the $conductor ((NEX = -PEE) PART (LAST_DENDER = PEESE) or rest DENDER = PEESE()) specifies$ the temale copulation that has an NES, or PEECE degree as their highest degree; and thecondition : <math>PE = -PRE = PRE = PR

Surfaced queries involve applying statistical functions to a population of raples for example, we may work to remew the number of individuals in a population of the average income in the population. However, statistical osers are not allowed to remeve individual data, such as the measure of a specific person. Statistical database security techniques must prohibit the remeve of a specific person. Statistical database security techniques must prohibit the remeve of a specific person. Statistical database security techniques must prohibit the remeve of individual data. This can be achieved by prohibiting queries that remeve ormibute values and by allowing only queries that involve statistical aggregate functions such as COUNT, SUM, MN, MAN, AVERAGE, and STANDAGO DECIATION. Such queries are sometimes called statistical queries.

It is the responsibility of a database management system to ensure the confidentiality of information about redividuals, while still providing useful statistical summaries of data about those individuals to users. Provision of **privacy protection** of users in a statistical database is paramation its variation is illustrated in the following example.

In some cases it is possible to infer the values of individual tuples from a sequence of statistical queries. This is particularly true when the conditions result in a population consisting of a small number of taples. As on illustration, consider the following two statistical queries:

```
Q1: SELECT COUNT (1) FROM PERSON
MHERE ACTAGE (INCOME) FROM PERSON
Q2: SELECT AVG (INCOME) FROM PERSON
```

```
WHERE RECEIPTIONS (
```

PERSON

|                 | ;               |                | ·  |          |        |
|-----------------|-----------------|----------------|----|----------|--------|
| NAME <u>ŞSN</u> | NCOME   ADDRESS | CITY ! STATE ! | ZP | SEX LAST | DEGREE |

FIGURE 23.3. The work relation scheme for illustrating statistical database security

Now suppose that we are interested in finding the second of "Jane" Smith", and we know that the has a TH D, degree and that she lives in the city of Belliare. Texas, We assue the statistical query Q1 with the following condition:

```
(LAST_DEGREE="PH.D." AND SEX="F" AND Clive"Rellative" AND STATE="Texas")
```

If we get a result of 1 for this query, we can usue Q2 with the same condition and find the income of Jone Simito. Even if the result of Q1 on the preceding condition is not 1 bet is a small number  $-s_{22}$ , 2 or 3 —bec can issue statistical queries using the functions MAX, MX, and AVERADE to identify the possible range of values for the more of Jane Smith.

The possibility of inferring individual information from statistical queries is reduced if no statistical queries are permitted whenever the number of toples in the population specified by the selection condition falls below some threshold. Another technique for prohibiting retrieval of individual information is no probabilit sequences of queries that refer repeatedly to the same population of toples. It is also existile to introduce slight inaccurate s or "noise" into the results of statistical queries deliberately, to make it difficult to deduce individual information from the results. Another technique is partitioning of the database. Partitioning implies that records are stored in groups of since maximum size; queries can refer to any complete group or set of groups, but reser to subsets at records within a group. The interested reader is referred to the bibliography for adiscussion in these techniques.

# 23.5 INTRODUCTION TO FLOW CONTROL

How control regulates the distribution of flow of information among accessible objects. A flow between object N and object Y occurs when a program reads values from X and writes values into Y. Flow controls check that information contained in some objects does not two explicitly of implicitly into less pionected objects. Thus, a user caetion get indirectly in Y what he or she connot get directly from X. Accive flow control began in the early 1970s. Most flow controls employ some concept of security class; the transfer of information is write to a receiver is allowed only if the recurver's sounty class is at least as privileged as the sender's. Ecomples of a flow control include preventing a service program from leaking a constroner's confidential data, and blocking the transmission of sector rule cary data to an unknown classing loser.

A flow policy specifies the channels along which information is allowed to move. The amplest flow policy specifies just two classes of anformation, confidential (C) and conconfidential (N), and allows all flows except those from class C to class N. This policy can solve the confinement problem that arises when a service program bundles data such as destoner information, some or when may be confidential. For example, an income-tax computing service might be allowed to retain the customer's address and the bill for services rendered, but not the customer's income or deductions.

Access control mechanisms are responsible for checking users' authorizations for resource access: Only granted operations are executed. Flow controls can be enforced by an extended access control mechanism, which involves assigning a security class (usuallcalled the elements) to each mining program. The program is allowed to read a particular memory segment only it is security class is as high as that of the segment. It is allowed to write in a segment only if its class is as low as that of the segment. This automatically ensures that no information transmitted by the person can move from a higher to a lower class. For example, a military program with a secret clearance can read only from offsets that are unclassified and confidential and it can only write into objects that are secret or toy secret.

Two types of flow can be distinguished: explicit flows, occurring as a consequence of assemment instructions, such as  $Y = f(X_n, X_n)$ ; and implicit flows generated by conditional instructions, such as if  $f(X_{n+1}, \dots, X_n)$  then  $y \neq f(X_n, X_n)$ .

How control mechanisms most verify that only authorized flows, both explicit and implicit, are executed. A set of trues must be satisfied to ensure secure information flows. Rules can be expressed using flow relations among classes and assigned to information, stating the authorized flows within a system. (An information flow from A to B occurs when information associated with A affects the value of information associated with B. The flow results from operations that cause information transfer from one object to another.) These relations can define, for a class, the set of classes where information (classified in that class) can flow, or can state the specific relations to be centred between two classes to allow information flow from one to the other. In general, flow cortral unschanging implement the controls by assigning a label result object and by specifying the security class of the object. Labels are then used to verify the flow relations defined in the model.

### 23.5.1 Covert Channels

A covert channel allows a transfer of information that violates the second or the police Specifically, a covert channel allows information to pass from a higher classification level to be lower classification level through improper means. Covert channels can be classified into two broad categories: storage and triung channels. The distinguishing frome between the two is that in a tinding channel the introduction is converyed by the through events or processes, whereas storage channels do not require any temporal synchronization, in that information is converged by accessing system information or what is other wise inaccessible to the user.

In a simple example of a covert channel, consider a distributed database system in which two nodes have user security levels of sector (5) and unclassified (11). In order for a transaction to commit, both nodes must agree to constit. They mutually can only de operations that are consistent with the 2-property which states that in any transaction, the S site cannot write or pass information to the U-ate. However, if these two sites collide to set up a covert channel, between them, a consider involving sector data may be committed unconditionally by the U-site, but the S site may do so in some predefined agreed-upon way so that certain information may be possed on from the S site to the U-site, molating the "grouperty. This may be achieved where the transaction runs repeatedly but the amicos taken by the S site implicitly convex information to the U-site. Measures such as

ording that we docussed in Chapters 17 and 18 precent concurrent writing of the information by users with different security levels into the same objects, preventing the storage-type covert clounnels. Operating systems and distributed durbases provide control over the multiplegromming of operations that allow a shoring of resources without the possibility of encisonchment of one program or process into another's memory or other resources in the system, this preventing timing-oriented covert channels. In general, covert channels are not a major problem in welf-implemented colust database implementations. However, certain achieves may be controved by clever users that implicitly transfer information.

Some security experts believe that one way to avoid covert channels is for programmers to not actually gain access to sensince data that a program is supposed to process after the program has been por into operation. For example, a programmer for a bank has no need to access the names or balances in depoators' accounts. Programmers for brokerage firms do not need to know what hav and sel, orders exist for clients. During program testing, access to a form of real data or some sample test data may be insuffable, but you after the program has been accepted for regular use.

# 23.6 ENCRYPTION AND PUBLIC KEY INFRASTRUCTURES

The previous methods of access and thor control, despite heing strong countermeasures, may not be able to protect databases from some threats. Suppose we contribute databut out data talls into the hands of some usinlegitimate user. In this attraition, by using encryption we can disguise the message so that even if the tratismission is diverted, the message will not be revealed. Encryption is a means of maintaining secure data in at onsecure environment. Encryption consists of applying an encryption algorithm to data using some prespectical encryption key. The resulting data has to be decrypted using a decryption key to recover the original data.

# 23.6.1 The Data and Advanced Encryption Standards

The Data Encryption Standard (DUS) is a system devideed by the U.S. government for use by the general public. It has been widely accepted coversprographic standard both in the United States and abroad. DES can provide end-to-end encryption on the channel between the sender A and vectorer B. The DUS algorithm is a coreful and complex combination of two of the fundamental building blocks of encryption: substitution and permutation (transposition). The algorithm derives its strength from repeated application of these two reclumptes for a total of 16 cycles. Plantext (the original form of the message) is encrypted as blocks of 64 bits. Although the key is 64 bits long, in effect the key can be any 56 bit number. After questioning the adequacy of DES, the National Institute of Standards (NIST) introduced the Advanced Encryption Standards (AES). This algoinformations a block size of 128 bits, compared with DESS 56 block size, and can use keys of 128, 192, or 256 bits, compared with DES's 56-bit key. AES introduces more possible keys compared with DES, and thus takes a much longer time to crack.

# 23.6.2 Public Key Encryption

In 1976 D fhe and Hellman proposed a new kind of cryptosystem, which they colled public key encryption. Public key algorithms are based on motivementical tonchosis rather than operations on bit patterns. They also involve the use of two separate keys, in contrast to convertional encryption, which uses only one key. The use of two keys can have profound consequences in the areas of confidentiality, key distribution, and authentication. The two keys used for public key encryption are referred to as the public key and the private keys invariable, the private key is kept sector, but it is referred to as a *private key* rather than a secret key (the key used or conventional encryption) to avoid continion with conventional encryption.

A public key encryption scheme, or infrastructure, has six ingredients:

- Platment: This is the data or readable message that is fed into the algorithm at impot.
- Encoption algorithm: The encryption algorithm performs various transformations on the planetext.
- 3 and 4. Public and provide keys: These are a pair of keys that have been selected so that more its used for encryption, the other is used for decryption. The exact transformations performed by the encreption algorithm depend on the public or presteries that is provided as input.
  - 5.3 Signetter: This is the serunibled message produced is output. It repeads on the planatest and the key. For a given message, two different keys will produce two different ciphertexts.
  - Decryption algorithm: This dependence accepts the caphertext and the matching key and produces the original plantext.

As the name suggests, the public key of the part is made public for others to use whereas the private key is known only to its usence. A general-purpose public key cryptographic algorithm relies on one key for energyption and a different but related one for decryption. The essential steps are as follows:

- Each user generates a pair of keys to be used for the energition and desreption of these ges.
- Each user places one of the two keys in a public register or other accessible fie. This is the public key. The compation key is kept private.
- 5. It a sender wishes to send a private message to a receiver, the sender energies the message using the receiver's public key.
- 4 When the receiver receives the message, he or she decrypts it using the receiver's private key. No other receiptern can decrypt the message because only the receiver knows his or her private key.

The RSA Public Key Encryption Algorithm. One of the first public key schenges was introduced in 1978 by Ron Rivest. Adi Shamir, and Len Adleman at MIT and is named after them as the RSA scheme. The RSA scheme has since then reigned supreme as the most which y accepted and implemented approach to public key encryption. The RSA encryption algorithm incorporates results from number theory, combined with the difficulty of determining the prime tas tors of a target. The RSA algorithm also operates with mechaiar anthreetic—multi.

Two keys, d and  $e_i$  are used for decryption and encryption. An important property is that they can be interchanged to is chosen as a large integer that is a possible of two large distinct prime numbers,  $\mu$  and b. The encryption key  $e_i$  is a tandouble chosen number between 1 and  $\mu$  that is relatively prime to  $(a - 1) \times (b - 1)$ . The plaintext block P is encrypted as P mod n. Because the exponentiation as performed mod  $\mu$ , factoring P to encrypted as P mod n = P. The decryption key d can be computed to the condition that  $d \times e - 1$  and  $d(a - 1) \times (b - 1)$ . The decryption key d can be computed from the condition that  $d \times e - 1$  and  $d(a - 1) \times (b - 1)$ . Thus, the legitimate receiver who knows d simply examples  $(P^*)^4 \mod \mu - P$  and recovers P without having to factor P.

### 23.6.3 Digital Signatures

A digital signature is an example of using encryption techniques to provide authentication services in electronic commerce applications. Like a handwritten signature, a digital signature is a means of associating a mark unique to an individual with a body of text. The mark should be unforgettable, meaning that others should be able to check that the signature does come from the originates.

A digital signature consists of a string of symbols ff a person's digital signature were always the same for each message, then one could easily constrained to by simply copying the string of symbols. Thus, signatures must be different for each use. This can be achieved by making each digital signature a function of the message that it is signing, logerher with a time string. To be unique to each signer and counterfurption, each digital signature must also depend on some sector number that is similar to the signer. Thus, in general, a counterficience? digital sign state must depend on the message and a unique sector number of the signer. The verifier of the signature, however, should not need to know any sector number. Public key techniques are the best means of creating digital signatures softly these properties.

# 23.7 SUMMARY

This chapter discussed several techniques for enforcing security in database systems if proceived the different threars to databases in terms of less of integrity, availability, and confidentiality. The four types of confirmenciaties to deal with these problems are access control, inference control, flow control, and encryption. We discussed all of these measures in this chapter.
Security enforcement deals with controlling access to the database system as a whole and controlling authoritation to access specific portions of a database. The former is usually done by assigning accounts with passwords to users. The latter can be accomplished by using a system of granting and recoking privileges to individual accounts for accessing specific parts of the database. This approach is generally referred to as discretionary access control. We presented scale SQ inclumends for granting aid revolving privileges, and we allow averated their use with examples. Then, we gave an overview of mandatory access control nucleonisms that enforce multilevel security. These require the classifications of asors and data ordues provised on classes and enforce the toles that prohibit flow of information from higher to lower security levels. Some of the ises concepts underlying the moltrievel relational model, itseluding filtering and polyinstantianan, were presented. Role-based access control was introduced, which assigns privileges based on roles that users play. We briefly discussed the problem a controlling access to statistical databases to protect the privacy of individual information while concurrently providing statistical access to populations of records. The issues related to flow control and the problems associated with covert channels were discussed next. Finally, we covered the mea of encryption of Jata, including the public key infrastructure and digital signatures.

#### Review Questions

- 23.1. Discuss what is mount by each of the fellowing terms database authoritation acress control, data corryption, printinged (system) account, database andit, andit earling
  - Discuss the types of provileges at the account level and those at the relation level.
- 25.2 Which account is designated as the owner of a relation? What privileges does the owner of a relation have?
- 23.3. How is the view mechanism used as an authorization mechanism?
- 23.4. What is mean by granting a privilege?
- 23.5. What is meant by revoking a privilege?
- 25.0. Discuss the system of propagation of privileges and the restraints imposed by horcontal and vertical propagation limits.
- 23.7. List the types of privileges available in sQL
- 23.5. What is the difference between discremonary and mandoory access control?
- 73.9. What are the typical scentify classifications? Discoss the simple security property and the "optoperty, and explain the justification behind these tides for enforcing multiplevel accurity.
- 23.1.3. Describe the multilecel relational data model. Define the following terms aparent key, polyinsaturation, filtering.
- 25-11. What are the relative metits of using DAC or MAUL
- 23-12. What is role-based access control? In what ways is it superior to DAC and MAC?
- 25 13. What is a statistical database? Discuss the problem of statistical database security
- 25.14 How is privacy related to statistical database security "What meanies can be token to ensure some degree of privacy in statistical databases?
- 23-15. What is flow control us a security measure? What types of flow control exist?

- 3516. What are covers channels? Give an esample of a rovers channel
- 3.17 What is the goal of energytion? What process is involved in energyting data and they recovering it at the other end?
- 13.18 Give an example of an encryption algorithm and explain how it works.
- 23.19 Repeat the previous question for the populat RSA algorithms
- 23.20. What is the public key infrastructure scheme? How does it provide security?
- 24.21. What are digital signatures? How do they work?

## Exercises

- 23.22. Consider the relational database schema of Figure 5.5. Suppose that all the relations-were created by (and hence are owned by) user X, who scores to grant the following prodeges to user accounts A, B, C, D, and E.
  - Account A concretions or modely any relation except dependent and congrant any of these privileges to other users.
  - Account B can retrieve all the attributes of employee and department except for solary, nyrssn, and nyrstartdate.
  - c. Account C can retrieve termodify works\_to but each only retrieve the FRANT HINTLINANT, and son atembutes of swearer and the FRANT and PROPAGE atembutes of PROLECT.
  - Account Dieux revneve any attribute of second or dependent and can modify preserver.
  - Account E can retrieve any attribute of transfer but only for ENVINCE tuples that have use = 3.
  - f. Write SQL statements to grant these privileges. Use views where appropriate
- 23.23. Suppose that provide follow Exercise 23.1 is to be given with GRANT OFTION but each state account A can grant, it to at most live accounts, and each of these accounts can propagate the prodege to other accounts but walkaw the GRANT OPTION providege. What would the horizontal and vertical propagation limits be in this case?
- 23.24 Consider the relation shows in Figure 23.2d. How would in appear to a user with classification, U? Suppose a classification. U user tries to update the salary of "Sirifun" to \$53,000; whist would be the result of this action?

# Selected Bibliography

Authorization based on granting and revolving privileges was proposed for the SYSTEM R experimental DBMS and is presented in Griffiths and Wade (1976). Several basics discuss security in databases and computer systems in general, including the books by Lers (1982a) and Fernander et al. (1981). Deening and Denning (1979) is a tutorial paper on Zura security.

Marx papers discuss different techniques for the design and protection of statistical databases. These include McLeish (1969), Chin and Crossvorlu (1981), Less (1952), Wong (1964), and Denning (1982). Ghesh (1964) discusses the use of statistical databases for

quality control. There are also more gapers discussing cryptography and data energyption, including Diffic and Heliman (1979), Rivest et al. (1978), Akl (1983), Pfleeger (1997), Omena et al. (1990), and Stalling (2300).

Multilevel security is discussed in Jajodia and Sandhu (1981), Demning et al. (1987). South and Winsleyt (1982), Stacheor and Thomasingham (1982), Lunt et al. (1990), and Bertino et al. (2001). Overviews of research issues in database security are given by Lunt and Fernander (1990). Jajodia and Sandha (1991). Bertino et al. (1998), Castana et al. (1995), and Thomasingham et al. (2001). The officers of multilevel security on construments control are discussed in Atlan et al. (1997). Security in next-generation, semantic, and object-oriented databases is discussed in Rabbin et al. (1991). Jajodia and Kogan (1990), and Smith (1990). Oh (1999) presents a model for both discretionary and mandatory security. Security models for Web-based applications and role-based access control are discussed in Joshi et al. (2001). Security is-uss for managers in the context of e-commence applications, and the need for risk assessment models for selection at appropriate -county countermeasures are discussed in Farabanand et al. (2002).



# Enhanced Data Models for Advanced Applications

As the use of database systems has grown, users have demanded additional functionality from these subvace packages, with the purpose of making it easier to implement more advanced and complex user applications. Object oriented databases and object reclamonal systems do provide features that allow users to optend their systems by specifying additional distribut data types for each application. However, it is quite useful reclamately certain common features for some of these advanced opplications and to create models that can represent these common features. In addition, specialized storing structures and indexing methods on the implemented to improve the performance of these common teatures. These features can then be implemented as abstract data type or class libraties and separately perchased with the base DBKs software package. The term **database** has been used in Informix and caetridge in Oracle (see Chapter 22) to refer to such optional submachines that can be included in a DBMs package. Users can utilize these features directly if they are suitable for their applications, without having to tension, reinformation to program such common relatives.

This chapter introduces doublese concepts for some of the common features that are needed by advanced applicances and that are starting to have wadespread use. The features we will rever are extremely that are used in active database applications, reaperal concepts that are used in temporal database applications, and briefly some of the issues involving instances databases. We will also discuss defactive databases. It is toportant to note that each of these topos is very bread, and see rangive only a briefly some to coeach area. In fair, each of these topos is very bread, and see rangive only a brief introductive coeach area. In fair, each of these areas can serve as the sole copy for a complete book. In Section 24.1, we will introduce the topic of active databases, which provide additional functionality for specifying **active rules**. These rules can be ouromatically triggered by events that occur, such as a database update or a certain time being reached, and can initiate certain actions that have been specified in the tule deflaration if certain conditions are met. Many commercial packages already have some of the functionality provided by active databases in the form of triggers. Triggers are now part of the Squitey standard.

In Section 24.2 we will introduce the concepts of temporal databases, which permit the database system to store a history of changes, and allow users to query both current and past states of the database. Senie temporal database models also allow users to store future expected information, such as planned schedule. It is important to note that many database applications are arready temporal, but are often implemented actionar having much remporal support from the DEVS package. that is, the temporal concepts were implemented in the application programs that access the database

Sectors 24.3 sell give a biter overview of spatial and multimedia databases. Spatial databases provide concepts for databases that keep track of objects in a multidimensional spatial positions of their objects, which include countries, states, rowers, enter, words, seas and so on. Other databases, such as measerological databases for weather intermation, are three-dimensional spatial positis. Multimedia databases provide features that along the objects in a multidimensional spatial positis. Multimedia databases provide features that along users to strue and query different types of multimedia information, which melodes images (such as provide features or drawings), video clips (such as movies, news rects, or house video), and documents (such as books or arricles).

In Section 34.4, we discuss dolts (we databases,<sup>1</sup> in area that is at the intersection of databases, lugic, and orniforal intelligence or knowledge bases. A **deductive database** system is a database system that includes capabilities to define (**deductive**) rules, which can deduce or infer additional information from the facts that are stored to a database. Because part of the theoretical foundation for some deductive databases. Other types of systems, referred to as expert database systems or knowledge-based systems, about ourperate reasoning and information probabilities such systems use techniques that we developed in the held of artificial intelligence, including semantic metworks, transs purduction systems, or rules for taptung demanospecific knowledge.

Readers may choose to prose the particular topics they are interested in, as the sections in this chapter are practically independent of one another.

Section 24.4 is a summary of a hageer 20 from the third station. The foll a hapter will be actually on the book Web site.

# 24.1 ACTIVE DATABASE CONCEPTS AND TRIGGERS

Rules that specify actions that are intromatically triggered by certain events have been core slered as important enhancements to a durabase system for quite some time. In fact, the concept of triggers—a technique for specifying corrant types of active rules—has existed in early versions of the SSE specification for relational disabases and ringgers are nose per of the 401-99 standard. Commercial relational 1983/8---such as Oracle, DBA, and SYRASE—base had various versions of treggers available. However, much research into which a general model for active databases should look like has been done since the early models of targets were proposed. In Section, 24 1-1, we will present the general concepts that have been proposed for specifying rules for active databases. We will use the syntax of the Ocacle commercial relational DPMS to olliest are these concepts with specific examples, since Oracle trageers are close to the way rules are specified in the SOI standard Section 24-1.2 will discuss some general design and implementation resides to ractive databases. We then give examples of how active databases are implemented in the STAR 568ST experimental 1994S in Section 24.1 3, since STARBUNG provides for many of the concepts of generalized active databases within its framework. Section 24.14 discusses possible applications of active databases. Fundly, Section 24-1.5 describes how treggers are declared in the SQL-99 standard.

## 24.1.1 Generalized Model for Active Databases and Oracle Triggers

The model that has been used for specifying active database rules is referred to as the **Event-Cundition-Action**, or **ECA** model. A rule in the FCA model has three components:

- The event (or econes) that imggers the rule: These events are usually database opdate operations that are explicitly applied to the database. However, in the general model, they could also be temporal events' or other kinds of external events.
- 2. The condition that determines whether the rule action should be executed: Once the triggering event has occurred, an optional condition may be evaluated. If no condition is specified, the action will be executed once the event occurs. If a condition is specified, it is first evaluated, and only if a evaluates to the will the tale action be executed.
- 1 The action to be taken. The action is usually a sequence of %.9 statements, but n could also be a database transaction of an external program that will be automatically executed.

Let us consider some examples to illustrate these concepts. The examples are based on a much simplified variation of the costs of database application from Lenie 5.7, which

 $\Delta$ . An example would be a reproductive specified as a zero did time, such as, frequentias rais every day at 5.50 MM

to shown in Figure 24.1, with each employee birring a name 1969E1, social secondy number (598), salary (59088), department to which they are currently assigned (680, a foreign key to persentent), and a direct supervisor (5090770508\_558, a (seconds)) locking key to the this example, we assume that null is allowed for two indicating that ac employee may be temporarily massigned to any department. Each department has a name (5869E), number (680), the total solary of all employees assigned to the department (75782\_54), and a manager (886858\_558, 586, a foreign key to the total solary of all employees assigned to the department (75782\_54), and a manager (886858\_558, 586, a foreign key to the total.

Notice that the forthe (see attribute is really a derived attribute, whose value should be the sam of the salaries of all employees who are assigned to the particular department. Maintaining the correct value of sizely inderived attribute can be done via an active rule. We first have to determine the events that may cause a charge in the value of 10% (Su, which are as follows:

- Inserting (one or more) new employee toples.
- 2.3 Changing the solary of terre or more) existing employees
- Changing the assignment of existing employees from one department to unother.
- Deleting (one of more) employee toples.

In the case of event 1, we only need to recompose of a last if the new employee is inmediately assigned to a department—that is if the value of the ost annihute for the new employee ruple () cot null (assuming null is allowed for tae). Hence, this would be the condition to be checked. A similar condition could be checked for event 2 (and 3) to determine whether the employee whole solary is changed (or who is being deleted) is currently assigned to a department, but event 3, we will outly a secure an action to monitary the value of (cross-solar) (second) to no condition to be checked (the action is alway-executed).

The action for events 1, 2, and 4 is to automatically update the value of "etac\_34 for the employee's department to reflect the newly inserted, updated, or defeted employee's salary. In the case of event 3, a twofold action is needed; one to update the intra\_salet the employee's old department and the other to update the table sale of the employee's new department.

The four active rules (or triggers) R1, R2, R3, and R4+ corresponding to the above situation — can be specified in the notation of the Oracle 10046 as shown in Figure 14.2a. Let us consider rule R1 to illustrate the syntax of creating triggers in Oracle. The 000441-

EMPLOYEE MANE SSM SALARY DNO SUPERVISOF 55N DEPARTMENT DNAME DNC TOTAL SAL MANAGER SSN

FIGURE 24.1 A simplified convex database used for active rule examples.

- (8) RI: CREATE TRIGGER TOTALSAL1 AFTER INSERT ON EMPLOYEE FOR EACH ROW WHEN INEW DNO IS NOT NULL) (UPDATE DEPARTMENT BET TOTAL SAL-TOTAL SAL ( NEW SALARY WHERE ONO-NEW.DNO;
  - R2; CREATE TRIGGER YOTALSAL2 AFTER UPDATE OF SALARY ON EMPLOYEE FOR EACH ROW WHEN INEW. DAO IS NOT MULL) UPDATE DEPARTMENT SET TOTAL\_SAL-TOTAL\_SAL - NEW.SALARY - OLD.SALARY WHERE DNO+NEW.DAO;
  - R3: CREATE TRIGGER TOTALSAL3 AFTER UPDATE OF DND ON EMPLOYEE FOR CACH ROW BEGIN

UPDATE DEPARTMENT SET TOTAL SALETOTAL\_SALENEW SALARY WHERE DNO-INEW.DNO: UPDATE DEPARTMENT SET TOTAL\_SALETOTAL\_SALE OLD.SALARY WHERE DNO=OLD.DNO, END.

R4: CREATE TRIGGER TOTALSALA AFTER DELETE ON EMPLOYEE FOR EACH ROW WHEN JOLD DNO IS NOT NULL) UPDATE DEPARTMENT SET TOTAL SAL TOTAL SAL - OLD.SALARY WHERE ONO=OLD.DND

(6)

R5: CREATE TRIGGER INFORM [SUPERVISOR] BEFORE INSERT OF UPDATE OF \$ALARY, SUPERVISOR [\$SN ON ÉMPLOYEE FOR EACH ROW WHEN (NEW.SALARY > (SELECT SALARY FROM EMPLOYEE WHERE SSN=NEW.SUPERV(SOR\_SSN)) INFORM SUPERVISOR[NEW SUPERVISOR\_SSN, NEW.SSN);

FROURE 24.2 Specifying an five index as triggers in Oracle notations. (a) Triggers for automatically maintaining the consistency of total second treasurer. (b) Trigger for comparing an employee's salary with that of his or her sopervisor.

TREGOUR statement specifies a mager (or active rule) name—toteGAL1 for RL. The AFTER-clause specifies that the rule will be triggered after the events that mager the rule occur. The triggering events—an insert of a new employee in this example—are specified inflowing the AFTER keyword.<sup>3</sup> The ON-clause specifies the inflation on which the rule is specified—explores for RL. The optional keywords fold EACLEROW specify that the rule will be triggering event.<sup>4</sup> The optional keywords fold EACLEROW specify that the rule will be triggering event.<sup>4</sup> The optional keywords fold EACLEROW specify that the rule will be triggering event.<sup>4</sup> The optional WEN-clause is used to specify any conditions that need to be checked after the rule is original but before the action is executed. Finally, the action(s) to be taken are specified as a figure block, which typically contains one optimore SQL stategreins or calls to execute executed.

The four triggers (acrive rules) R1, R2, R3, and R4 illustrate a number of features of acrive rules. First, the basic events that can be specified for triggering the rules are the standard SQL update commands. (NSER), (PEE1), and UPDATE. These are specified by the keywords INSERT, DELETE, and UPDATE in Obacle notation. In the case of UPDATE one may specify the autobutes to be updated. For example, by writing UPDATE OF SALAR-, particulated by the triggering event. The suplex that have been inserted, deleted, or modified by the triggering event. The keywords NEW and OLD are used in Oracle notation. NEW is used to refer to a newly inserted or newly updated tuple, whereas OLD is used to refer to a deleted or newly updated tuple.

Thus rule R4 is triggered ofter an (N=68T experiation is opplied to the evel-tots relation in R1, the condition (NEW, DV) is NOT NOLLY is checked, and due evaluates to true, meaning that the newly inserted employee suple screlated to a department, then the action is executed. The action inputtes the oseNETMENT triple(s) related to the newly inserted employee by induling their salary (NEW, SALARY) to the TOTAL\_SAL attribute of their related department.

Rule R2 is similar to R1, but iclis triggered by an OPEATE operation that updates the savay of an employee rather than by an INSBIT. Rule R5 is triggered by an update to the tot intribute of ENEOSE, which arguites charging an employee's assignment from one department to another. There is no condition to check in R3, so the action is executed whenever the triggering event occurs. The action updates both the old department and solution to check by adding their salare to total. So of their area department and solution there value of total way from total. So of their department, Note that this should work even of the value of two way only, because in this gase no department will be selected for the rule action.

It is important to note the effect of the optional BOR EACH ROW close, which signifies that the rule is triggered separately for each caple. This is known as a row-level trigger. If this close, was left out, the rigger would be known as a statement-level trigger.

As we shall see later, it is also possible to specify Rd-363 instead or AFTER, which indicates that the rule is triggered follow the mojernic energy is evoluted.

Again, we shall see later than an alternative is to trager nucltifie only once even if multiple news simpler) are affected by the imgraning event.

R1, R2, and R4 can also be wratten without a condition. Photocopy they much here to encount to execute with the condition server the article is fait, needed (relevants) reprive).

and would be imagered once for each imagering statement. To see the difference, consider the following update operation, which grock a 10 percent raise to all employees assigned to department 5. This operation would be an event that triggers rule R2:

| UPDARE | 1944107 | 65    |   |        |
|--------|---------|-------|---|--------|
| SET    | SAUARY  | - 1.1 | ٥ | SALAGY |
| HICHE  | 040 -   | 5;    |   |        |

Because the above statement could update multiple records, a rule using row-level semantics, such as R2 in Figure 24.2, would be triggered once for each nact whereas a rule using statement-level semantics is triggered only more. The Oracle system allows the user to choose which of the above own options is to be used for each rule. Including the optional BAV FACIA IGAV clause creates a row-level trigger, and leaving in our course a statement sevel trigger. Note that the keywords NEW and OFF can only be used with reacheed triggers.

As a second example, appose we construction check whenever or employee's solare to greater than the solary of his or her direct sopervisor. Several events can trigger this rule; inserting a new employee, changing an employee's solary, or changing an employee's solary is r. Suppose that the action to take would be to call on external procedure twose screavisor," which will notify the supervisor. The rule could then be written as in R5 (see Figure 24.2b).

Figure 24.3 shows the syntax for specifying some of the nanu optices available in Oracle impacts. We will describe the syntax for toggers in the SqL99 standard in Section 24.1.5.

#### 24.1.2 Design and Implementation Issues for Active Databases

The previous section give on overview of some of the main concepts for specifying ocrive index. In this section, we discuss some additional issues concerning how rules are destened and implemented. The first osne concerns activation, deactivation, and grouping of rules.

```
«ungger» ::= CREATE TRIGGER <ingger name»
(AFTER | BEFORE) <inggering eventa» ON <table name»
| FOR EACH ROW |
| WHEN <condition»]
<ingger actions» ;
<taggering events» ::~ <ingger events (OR <ingger events)
<ingger events:: - INSERT 1 DELETE | UPDATE | OF <column name» | <column name» } }.</pre>
```

```
strigger accords to =<PL/SQL blocks</p>
```

FIGURE 24.3 A syntax summary for specifying triggers in the Oracle system (main options only)

to Assuming that or appropriate external protection biodisers despired. This is a feature that is now ovallable in SQL.

In addition to creating rules, in active database system should allow user- to activate detectivity and shop rules by reforming to their tule causes. A detectivated rule will not be triggered by the triggering event. This feature allows users to selectively detectivate cules for certain periods of time when they are not needed. The activate cummand will make the rule active again. The drop command deletes the rule troin the system. Another option is to great rules into named rule sets, so the whele of rules could be acrossed, dearrow reds or dropped. It is also useful to have a command that can trigger a rule of rule set via ap explicit PROCESS RULES command issued by the user.

The second issue concerns whether the triggeted action should be executed by second or concurrently with the triggeting event. A related issue is whether the action being executed should be considered as a sejecate protection or whether it should be part of the same transaction that triggered the rule. We will first try to categorize the various optime, it is important to rate that nex all options may be obtained for a particular active database system. In fact, must commercial systems are bridger to one or two of the options that we will now discuss.

Let us assume that the magening event occurs as part of a transaction exerution. We should first consider the contains options for how the triggening event is related to the evaluation of the rule's condition. The rule conductor conduction is also known as rule **consideration**, since the action is to be executed only after considering whether the condition evaluates to indee or folse. There are three mont possibilities for rule consideration.

- Inmediate consideration: The condition is evaluated as port of the same transaction as the triggering event, and is evaluated or mediately. This case can be further catissurged into three options.
  - Evaluate the condition by/or executing the triggening enemy.
  - Evaluate the condition after excenting the triggering event.
  - Evaluate the condition instead of executing the triggering event.
- Defended consideration. The condition is evaluated at the end of the transcript rhor included the megering events by this case, there could be many imgendrule-waiting to have their conditions evaluated.
- Detached consideration. The condition is evaluated as a separate transaction, spawned from the inggenite transaction.

The next set of options rencorns the relationship between evaluating the role condition and eventiang the role action. Here, again, three options are possible: **immediate**, **deferred**, and **detached** execution. However, must active systems use the first option. That is, as soon as the condition is evaluated, if a returns true, the action is monedately executed

The Drack system (see Section 24.1.1) uses the provefair consideration model, but a allows the user to specify for each rule whether the before or oper option is to be used with immediate condition evaluation. It also uses the immediate exclusion model. The STARBURST system (see Section 24.1.3) uses the defended consideration option, metrony that all rules triggered by a transaction with until the triggering transaction reaches us end and issues its COMMIT WORK command before the rule conditions are evaluated <sup>7</sup>.

<sup>7 (</sup>STARO, 30) also allows the user to explicitly startifule consideration via a LROX LSORULES or minere

Another usiac concerning action database rules is the distinction between rouslend takes versus stationeral-local takes. Because 5(0) update statements (which oct as triggering events) can specify a set of teplex, one has to distanguish between whether the rule should be considered once for the whole duration or whether it should be considered separately (or each non-(dual is, tople) affected by the statement. The SQL99 standard (see Section 24.1.5) and the Oracle system (see Section 24.0.1) allow the overt to choose which of the above two options is to be used for each rule, whereas STARIR/RST uses statement-level semantics only. We will give examples of how statement-level inggers can be specified in Section 24.1.3.

One of the difficulties that may have limited the widespread use of active rules, in spite of their potential to simplify database and software development, is that there are no easy-trouse techniques for designifie, writing, and certifying rules. For example, any quite difficult to verify that a set of rules is **consistents** meaning that two or more rules in the set do not contradict one another. It is also difficult to guarantice **termination** or it set of rules index all conconstances. To briefly albetrate the termination problem, consider the rules in Figure 24.4. Here, rule R1 is triggered by an INSERT event on twarf1 and its action includes an update event on structure1 of takes2. However, rule R2V ruggering event is an UFDATE event on structure1 of takes2, and its action includes an USERT event on twarf1. It is easy to see up this example that these two rules can trigger one another indefinitely, leading to noncommution. However, if does in trigger one another indefinitely determing whether retransition is guaranteed or not.

Factive rules are to reach their potential, it is necessary to develop reals for the design, debugging, and non-toring of terms rules that can help users in designing and debugging their rules.

#### 24.1.3 Examples of Statement-Level Active Rules in STARBURST

We now give some examples to dilustrate how toles can be specified in the FTARPOSET experimental DBMS. This will allow us to demonstrate how statement-level toles can be written, since these are the only types of toles allowed in STARPORST.

RE CREATE TRIGGER T1 AFTER INSERT ON TABLE? FOR EACH ROW UPDATE TABLE? SET ATTRIBUTE1 -....

R2: CREATE TRIGGER T2 AFTER UPDATE OF ATTRIBUTE1 ON TABLE2 FOR EACH ROW INSERT INTO TABLE1 VALUES (....);

FIGURE 24.4 An example to illustrate the termination problem for active rules.

The three active rules R1S, R2S, and R3S in Figure 24.5 correspond to the first three rales in Figure 24.2, but use S7A9.008°T cororation and statement-slevel semantics. We can explain the rule structure using rule R1S. The CREATE ICHE statement specifies a rule name-constant for R1S. The CREATE is she relation on which the rule is specified—instance, for R1S. The CREATE associates are relation on which the rule is specified—instance, for R1S. The WHEN-clause is used to specify the events that trager the rule.<sup>8</sup> The opposed it-clause is used to specify any conditions that need to be checked.

ReS: CREATE AULE TOTALSALL ON EMPLOYEE. WHEN INSERTED. HF. EXISTS (SELECT \* FROM (INSERTED WHERE DND) IS NOT NULL) THEN UPDATE DEPARTMENT AS D SET D TOTAL\_SAL=D TOTAL\_SAL + (SELECT SUM(I SALAPY) FROM INSERTED AS I WHERE D DNO = 10N0) WHERE D.DNO IN (SELECT DNO FROM INSERTED) **925: CREATE AULE TOTALSAL2 ON EMPLOYEE** WHEN UPDATED (SALARY) IF EXISTS/SELECT \* FROM NEW-UPDATED WHERE DNO IS NOT NULLI OR EXISTS(GELECT \* FROM OLD-UPDATED WHERE DNO IS NOT NULL) THEN UPDATE DEPARTMENT AS D 6FY DIOVAL\_SAL-D.TOTAL SAL + (SELECT SUM(N SALARY) FROM NEW-UPDATED AS N WHERE 2 DNO HN DND) -(SELECT SUM(O SALARY) FROM OLD-UPDATED AS O WHERE 0.0N0=0.0N0) WHERE D OND IN (SELECT DNO FROM NEW-UPDATED) OR D ONO IN (SELECT ONO FROM OLD-UPDATED): R3S: CREATE BULE TOTALSALS ON EMPLOYEE WHEN UPDATED(ONO) THEN UPDATE DEPARTMENT AS D SET O TOTAL\_SAL-D.TOTAL SAL+ (SELECT SUMIN.SALARY) FROM NEW-UPDATED AS N WHERE D.DND-N RNOL D ONO IN (SELECT UNO FROM NEW-UPDATED). WHERE UPDATE DEPARTMENT AS D SET O TOTAL\_SAL≠D TOTAL\_SAL -(SELECT SUM(O, SALARY) FROM OLD-UPDATED AS O WHERE D.ONO+0.DNOL

#### WHERE O.DNO IN ISELECT DNO FROM OLD-UPDATED);

FIGURE 24.5 Active rules using statement-level semantics in suscaulas, notation.

 Note that the WHEN kernetic specifies are not at \$1A880.851 bit is used to specify the tale one for an SQU and Oracle traggers. Finally, the THENOCLAUSE is used to specify the action (or actional rathe taken, which are typically one or more SQL statements.

In STAR00 BST, the lister oversis that can be specified for ungering the rules are the standard styllopdate commands: INSERT, PHETE, and OPDATE. These are specified by the keywords tescerter, extent, and updates on STAR00881 neuroist. Second, the rule designer needs to have a way to refer to the tuples that have been unalified. The keywords usanten, and updaten or used in STAR00881 neuron to refer to loar transition tables (relations) that unclude the newly inserted toples, the deleted tuples, the updated tuples before they were updated, and the updated toples after they were updated, and the updated toples when writing the condition and action parts of the rule. Transition tables centain toples of the same type as those in the relation specified in the oS-clause of the rule—for RTS, R2S, and R3S, this is the PHEDMETE to ittem.

It statement-level semantics, the rule designer can only refer to the transition tables as a whole and the rule is triggered only once, so the rules must be written differently than for row-level semantics. Because multiple employee toples may be inserted in a smell insert statement, we have to check if at least one of the newly ansatted employee toples is related to a department. In RTS, the condition

#### EXISTS(SELECT > FROM INSERTED WHERE DVD IS NOT NULL)

is checked, and if it evaluates to true, then the action is exercised. The action updates in a single statement the offectment tople(s) related to the newly upserted employee(s) by adding their submes to the fator, we arrichard of each related department. Because more than one newly inserted employee may belong to the some department, we use the SUM aggregate function to ensure that all their submesters are added.

Rule R2S is similar to R1S, but is triggered by an CFDATE operation that updates the solary of one of more employees rather than by an "NSERT. Rule R4S is triggered by an update to the explorithear of explored, which signifies changing one of more employees' assignment from one determinent to atother. There is no condition in R3S, so the action is executed whenever the triggering event occurs." The action updates for the old department(s) and new department(s) of the reassigned employees by adding their solar department(s) of the reassigned employees by adding their solar to the old department(s) of each new department and solarise the reassigned employees by adding their solar to the old department.

In our example, it is more complex to write the statement-level rules than the rowlevel rules, as can be dilustrated by comparing Figure- 24.2 and 24.5. However, this is not cigeneral rule, and other types of active rules may be ensure to specify using statementlevel notation than when using now-level contation.

The execution model for active rules of STARD00057 new deterred considerations. That is all the rules that are traggered within a transaction are placed in a set — colled the conflict

As no the Chacle examples, roles K1S and R2S can be warren without a condition. However above may be more efficient to essence with the condition since the zerion is net invoked ordess any required.

set—which is not considered for evaluation of conditions and execution antil the transaction ends (by issuing its COMMENDARK command). STARPORST also allows the user to explicitly start rule consideration in the middle of a transaction via an explicit PROCESS RefEs conservable. Decause multiple rules must be evaluated, it is necessary to speedy an order atmosp the rules. The syntax for rule declaration on STARPORST allows the specification of ordering among the rules to instruct the system about the order in which a set of rules should be considered.<sup>16</sup> In addition, the transaction tables—tweater, protection that affected and onunward contain the referent all the operations within the transaction that affected order upward contain the referent all the operations within the transaction that affected order upber, since multiple operations may have been applied to each table during the transaction.

#### 24.1.4 Potential Applications for Active Databases

We now briefly discussione of the permittal applications of active rules. Obviously, one important application is to allow **notification** of certain conditions that occur. For example, an active database may be used to monitor, say, the temperature of an industrial fornace. The application can periodically insert in the database the temperature reading records directly from temperature sensors, and across rules can be written that are triggered whenever a temperature record is inserted, with a conductor that checks it the temperature exceeds the datager level, and the action to raise in slocu.

Active rules can also be used to enforce integrity constraints by specifying the types of events that may cause the constraints to be codated and then evolutione appropriate conditions that check whether the constraints are actually colated by the event of not lience, complex application constraints, often known as business rules may be entered that way. For example, in the outvesserv database application, one taken memory the event of a specification, one taken and it may alert the advesser database ender a constraint average of students whenever a new ender is entered, and it may alert the advesser database precision of the event taken advesser database endered and it may alert the advesser database precision of the event allows a certain theshold another rule may check that course precisions are satisfied before allowing a student to entrol in a course; and so an

Other applications include the autoticitic matricentatice of derived data, such as the examples of rules R1 through R4 that maintain the derived attribute 1014. So, whenever individual employee tuples are changed. A similar application is to use nurice ride to maintain the consistency of materialized views (see Chapter 0) whenever the base relationary modified. This application is also relevant to the new data workbousing rechnologies (see Chapter 28). A related application is to maintain replicated tables consistently specifying rules that modified.

#### 24.1.5 Triggers in sQL-99

Triggers of the SqL-99 standard are quite similar to the examples we discussed in Section 24.1.1, with some many symptotic differences. The basic events that can be specified for inggering the rules are the standard SQL update commands. IN FERT, CELETE, and COATE

10. If no order is specified between a point 4 these the system default order is base base placing the rule declared tost obtails of the other rule.

In the case of UPDATE one may specify the attributes to be updated. Poth resolved and statement-level triggers are allowed, indicated in the trigger by the clauses F38 EACH REW and SOR EACH STATEMENT, respectively. One syntactic difference is that the trigger may specify particular tuple variable names for the old and new pipes instead of using the keywords NEW and ODD as in Figure 24.1. Trigger T1 in Figure 24.6 shows how the tow-level trigger R2 from Figure 24.1 (a) movies pecified in SQL-99. In-ide the REPERSECING clause we named tuple variables (classes) O and Net refer to the ODD tuple (lafter modification), respectively. Trigger E2 in Figure 24.6 shows how the tow-level and NEW tuple (lafter modification), respectively. Trigger E2 in Figure 24.6 shows how the statement-level trigger R2S from Figure 24.5 may be specified in SQL-99. For a statement-level trigger, the REPERENTIAN clause is used to refer to the table of all new tuples (arealy unserted or newly updated) as N, whereas the table of all old tuples (deleted tuples or tuples before they were updated) is reversed rains O.

# 24.2 TEMPORAL DATABASE CONCEPTS

Temporal databases, in the broadest sense, encompass all database atablications that require some aspect of time when organizing their information. Hence, they previde a good example to illustrate the need the developing a set of mulying concepts for application developers to use. Temporal database applications have been developed since the early days of database usage. However, in creating these applications, it was manify for to

```
T1:
 CREATE TRIGCER TOTAL SALL
 AFTER UPDATE OF SALARY ON EMPLOYEE
 REFERENCING OLD ROW AS U. NEW ROW AS N
 FOR EACH ROW
 WHEN (N.DND IS NOT MULL)
 UPDATE DEPARTMENT
 SET TOTAL_SAL = TOTAL_SAL + N.SALARY - D.SALARY
 WHERE DNO - N.DAG;
12:
 CREATE TRIGGER TOTALSAL2
 AFTER UPDATE OF SALARY ON EMPLOYEE
 REFFRENCING OLD TABLE AS (), NEW TABLE AS N
 FOR EACH STATEMENT
 WHEN EXISTS(SELECT ? FROM N WHERE N.GNU IS NOT NULL) OR
 EXISTS(SELECT * FROM O WHERE O.(INU IS NOT NULL)
 UPDATE DEPARTMENT AS D
 SET D. TUTAL_SAL = D. TOTAL_SAL
 + (SELECT SUM(N SALARY) FROM N WHERE O DNO=N.DND)

 (SELECT SUM(Q, SALARY) FROM Q WHERE D, DND=Q, DND)

 WHERE CNO IN ((SELECT OND FROM N) UNION (SELECT DWO FROM (0)):
```

FIGURE 24.6 Ingger TT illustrating the syntax for defining inggers in SQL-99.

the application designers and the colores to discover, design, program, and miglement die temporal concepts they need. There are many examples of applications where some ispect of time is needed to maintain the intermation in a database. These meliade headone, where parjoint histories need to be maintained; maintaine, where chains and accident histories are respired as well as information on the times where insorance policies are in effect: reservation systems in general (bote), aithree car restal, train, etc.), where informs to the dates and times when reservations are in effect are required; scientific dowhases, where data collected from experiments, includes the time when each data is measured, an so on Even the two examples used in this book may be easily exploded into temporal applications. In the tonewer database, we may wish to keep sacks, the and reservhistories on each employee. In the parsestry database, time is already included in the sensestry, and reach exercise of carse; the grade history of a viceous and the internation on the each employee. In the parsestry database, time is already included in the sensestry, and reach exercise of carse; the grade history of a viceous and the internation on taken it each exercise of a carse; the grade history of a viceous and the internation on the each start to database to conclude that the magnity of database applications have some temporal information. Users often attempted to simplify originatemporal aspects because of the complexity that they add to their applications.

In this section, we will introduze some of the concepts that have been developed to deal with the complexity of remporal database applications. Section 24.2.1 gives an average of how time as represented in databases, the different types of remporal information, and some of the different dimensions of time that may be needed. Section 24.2.2 discusses how time can be incorporated into relational databases. Section 24.2.4 spaces some additional options for representing time that are possible in database model that allow complex-structured objects, such as object databases. Section 24.2.4 introduces operations for querying temporal databases, and gives a brief overview of the TSO12 language, which extends soft with temporal concepts. Section 24.2.3 tocuses on the series data which is a type of remporal data that is very important in gracines.

#### 24.2.1 Time Representation, Calendars, and Time Dimensions

For remporal databases, time is considered to be an ordered sequence of points in some granularity that is determined by the application, for example, suppose that some temps rail application reject requires time on its that are less than one second. Then, each time good represents one second in time using this granularity. In reality, each second is a (short) teac doution, our a point, since it can be further divided into calliseconds, nucleonation, and a point, since it can be further divided into calliseconds, nucleonation, and a point, since it can be further divided into calliseconds, nucleonation, and a point, since it can be further divided into calliseconds, nucleonation, and so can Temporal database researchers have used the term chromon rise shut point to describe this manimal granularity for a particular application. The main consequence of choosing a minimum granularity for a particular application. The main consequence of choosing a minimum granularity to be simultaneous events occuring within the same second will be considered to be simultaneous events even though in reality they may not be

Because there is no known beginning or ending of time, one needs a reference point from which to measure specific rane points. Various calendars are used by vortous cultures (such as Gregorian (Western), Chiuese, Isbany, Ehndo, Jewish, Copridone), with different reference points. A calendae organizes time involutiferent rune outs for convenience. Nost calendars group 60 seconds into a minute, 60 minutes into an hour, 24 hours into a day (fiscil on the physical time of early ratation around its axis), and 7 cays into a week. Farther grouping of days into months and months into yours either follow solar or lunar natural phenomena, and are generally aregular, in the Gregorian calendar, which is used in most Western countries, days are grouped into anoralis that are either 28, 29, 30, or 31 days, and 12 months are grouped into a year. Camplex formulas are used to map the different time units roome another.

In sQ 2, the temporal data types (see Chapter S) include DATE (specifying Year, Month, and Day ACAMAMATOD), 110-1 (specifying Hear, Monare, and Second as BEAMASS). TALESTAMP (specifying a Date/Time combinet on, with options for and oding sub-second divisions of they are needed). INTERVAL to relative time duration, such as 10 dates or 2n0 mitures), and TERIOD (an antioned time duration with a need statistic point, such as the Tokday period from January 1, 1999, to January 10, 1999, inclusive). <sup>1</sup>

Event Information Versus Duration for State) Information. A remporal disabase will store information concerning when certain events occor, or when certain fibers are considered to be true. There are several different types of remporal information. Point events or facts are typically associated in the database with a single time point in some granulating. For example, a bank deposit event may be associated with the timestation when the deposit was made, or the total monthly sales of a product (fact) may be associated with a particular month (say, February 1989). Note that even though such events or thats may have different etanalattices, each is still associated with a single time value in the database. This type of indefination is often represented as time series data as we shall discuss in Section 24.2.5. Duration events or facts, on the other hand, are associated with a specific time period in the database.<sup>17</sup> For example, an implayee may have worked in a company from August 15, 1893. Lik November 20, 1998.

A time period is represented by its start and end time points [5 an - On ], each ore]. For example, the above period is represented as [1993-08-15, 1998-11-20]. Such a rigat period is often interpreted to mean the set of all time (which from sumations to end-tone, inclusive, in the specified granularity flexes, assuming day granularity, the period [1993-08-15, 1998-11-20] represents the set of all days from August [5, 1993, unit November 20, 1998, inclusive <sup>15</sup>

<sup>11</sup> Unfortunately, the remainscopy has not been used consistently. For example, the term method is the unserval is the unserval to an anchored duration, for consistency, for shall use the sign terminology.

<sup>12</sup> This is the same as an antibored datamate it has also been frequency called a time interval, but is avoid econormic we will use period to be consistent with sign terminology.

<sup>13.</sup> The representation [1993-68-15], 1998-11, 20] is called a closed piteroit representation. One can also use an open microit, denoted [1993-08-15], 1998-11-21], where the set of points does not marke the end point. Although the carter representation is sometimes more convergent, we shall use elsewed intervals through out to avoid confuse a

Valid Time and Transaction Time Dimensions. Given a particular event or tact that is associated with a particular time point or time period in the database, the as-octation may be interpreted to mean different things. The most thought interpretation is that the associated time is the time that the event occurred, or the period during which the fact was considered to be time in the real world. If this interpretation is used, the associated time is often referred to as the valid time. A temporal durabase using the interpretations is alled a valid time database.

However, a different interpretation can be used, where the associated time refers to the time when the information was actually stored in the database; that is, at is the value of the system type clock when the information is valid or the system. <sup>4</sup> In this case, the associated nine is called, the transaction times A temperal database using this interpretation is called a transaction time database.

Other interpretations can also be intended, but these two are considered to be the most comption ones, and they are referred to as **time dimensions**. In some applications, only one of the dimensions is needed and an other cases both time dimensions are required, in which case the temporal database is called a bitemporal database. If other interpretations are attended for time, the user can define the senior rescard program the applications appropriately, and it is called a user-defined time.

The next section shows with exemples how these concepts can be incorporated into relational databases, and Section 24.2.3 shows an approach to incorporate temporal concepts into object databases.

#### 24.2.2 Incorporating Time in Relational Databases Using Tuple Versioning

Valid Time Relations. There is now see how the different types of temporal databases may be represented in the relational model. First, suppose that we would like to include the history of changes as they occur in the real would. Consider again the database in Figure 24.1, and let as assume that, for this applie ition, the examplify of data type is day. Then, we could convert the two relations breacter and DF##Figure 24.1, and let as assume that, for this applie ition, the examplify of data type is 0.75 in ender the two relations breacter and DF##Figure 24.7a, where the relations by aching the attributes vst (Valid Start Time) and vet (Valid Erd Time), whose data type is 0.75 in ender to provide day granularity. This is shown in Figure 24.7a, where the relations have been been been been been been used by a construction.

Consider how the eve\_vi relation differs from the non-convergent function relation (Figure 24.1). The evel of, cach imple variations a version of an amplique's information that is called (in the real world) only during the time period (v.vs7, ...v.vf1), whereas a encored evelop imple represents only the current state or current version of each employee by each opposite value, non-assis in the current version of each employee typically has a special value, non-assis

<sup>14.</sup> Line explanation is more involved, its weishild see in Section 24.2.3.

<sup>15.</sup> A number point relation is also called a snapshot relation as it shows only the entropy snapshot relation can be the database.



FIGURE 24.7 Different types of temporal relational databases, (a) Valid time database schema, dat Transaction time database schema. (c) Bitemporal database schema.

valid end time. This special value, **none**, is a **temporal variable** that implicitly represents the correct time as time progresses. The nontemporal **BOLATEE** relation would only unclude those tuples from the EVELAT relation where VET is *none*.

Figure 24.8 shows a few tuple versions in the valid-time relation- tor\_st and ster\_vt. There are two versions of Smith, three versions of Wong, one version of Brown, and one version of Narayon. We can movile bow a valid time relation should behave when information is changed. Whenever one or more attributes of an employee are updated, rather that, actually overwriting the old values as would happen in a nonremporal relation, the system should create a new version and close the current version by thang ng its zero sche end time. Hence, when the user used the command to update the salary of Smith effective on June 1, 2003, to SBCCCC, the second version of Smith was the current version June 1, 2003, to SBCCCC, the second version of Smith was the current version, with use as its structure after the apdate, the first version of Smith was the current version, with use as its structure after the apdate size close changed to May 31, 2003 (one less than June 1, 2004, in day granulants), to indicate that the version has become a closed or history version and that the new (second) version of Smith is now the current one.

| ENP_VI         | ſ                     |          |     |            |              |            |            |
|----------------|-----------------------|----------|-----|------------|--------------|------------|------------|
| NAME           | SSN                   | SALADY   | DNO | SUPERVI    | SOR_SSN      | <u>VST</u> | VCT_       |
| 30: <b>3</b> 1 | 123456789             | 25000    | s   | 2004       | +5555        | 2002-08-16 | 2000-05 01 |
| 30.99          | 123466759             | 399333   | ę   | 2004       | 45555        | 2003-06-01 | 976        |
| Wong           | 333445555             | 25000    | -1  | 9999       | 97777        | 1589-08.20 | 2001-01-01 |
| Weng           | 333445666             | 00000    | e   | 9999       | 9777         | 2001-02-01 | 2002-02-31 |
| Wong           | 333445555             | 49005    |     | 2880       | 55055        | 8002-04-01 | 00/4       |
| Brown          | 792447777             | 29000    | J   | 3038       | 87777        | 2031-65-01 | 2012-09 (0 |
| Narayan        | <del>1770064444</del> | 39000    | 5   | 3334       | 45036        | 2003 dB-C) | 09%        |
|                |                       |          |     |            |              |            |            |
| CEP7 V         | τ                     |          |     |            |              |            |            |
| DNAME          | DNC)                  | MANAGER  | SSN | VST        | VET          | _          |            |
| Pesearch       | 5                     | 88866555 | 5   | 2001-08-20 | 2016-03-8    | 1          |            |
| Resourch       | . <u>.</u>            | 33344555 | 5   | 2002/04-01 | 11Ç <b>A</b> |            |            |

FIGURE 24.8 Some tuple versions in the valid time relations prejor and analysis.

It is important to take that in a colid time relation, the user must generally provide the valid time of an update. For example, the valiey update of Smith may have been entered in the database on May 15, 2003, at 652/12 A M, say, even though the salary change in the real world is effective on June 1, 2003. For is called a protective update, since it is applied to the database often it becomes effective in the real world. If the implate was applied to the database often it become effective in the real world. If the inplate was applied to the database often it become effective in the real world. If the inplate was applied to the database often it become effective in the real world, it is called a retrustive update. As update that is applied in the same time when it becomes effective is called a simultaneous update.

The action that corresponds to deleting an employed in a nontremporal database would typically be applied to a solid time database by doing the current retrient of the employer being deleted. For example, if Smith haves the company effective January 19, 2004, there this would be applied by changing with the current version of Smith from non-tri-2004.01-19. In Figure 34-5, there is no current version for Brown. Lecause he presumably left the company on 2002-05-10 and was legically deleted. However, because the database is remported, the old information on Prown is still there.

The operation to insert a new employee would correspond to creating the just topic reaches for that employee, and making it the current version, with the ver being the effective (real world) time when the simployee stars work. In Figure 24.7, the tuple of Narayan illustrates this, since the first version has not been updated yet.

Notice that in a valid functional ion nearing and less, such as swarn success, is no longer unique to each tiple (version). The new relation key for **every** was a combination of the nontemporal key and the valid start time attribute vers<sup>16</sup> as we asso (SW, 1991) as

the A combination of the montempositikey and the colid end time at talente with constraine being

primary key. This is because, at any point in time, three should be at most one value consult of each entity. Hence, the constraint that any two tuple versions representing the same entity should have nonmeroceing valid dote primes should hold on valid time relations. Notice that if the nontemporal primers key value may change over time, it is introduct to have a unique surrogate key attribute, whose value never changes for each roal world entity in order to relate rogether all versions of the same real world entity.

Valid time relations basically keep track of the history of changes as they became effective in the real world. Hence, if all real-world changes are applied, the database keeps a fusitory of the real-world states that are represented. However, because updates, insertions, and deletions may be applied retroactively or proversiesh, there is no record of the actual database state at any point in time. If the actual database states are more important to an application, then ever should use represented one reflations.

**Transaction Time Relations**. In a number on time durings, whenever a change is applied to the database, the actual timestamp of the transaction that applied the change (insert, delete, or opdate) is recorded. Such a database is most useful when changes are applied sound measuring in the majority of cases—for example, real-time stock trading or banking transactions. If we convert the nontemporal database of Figure 24.1 into a transaction time database, then the two relations tostever one transactions time relations by adding the attrabutes of (Transaction Start Time), whose data type is topically TMS\*TAMP. This is shown in Figure 24.7b, where the relations have been terrained target and retry to respectively.

In GPC\_TT, each tuple v represents a sension of an employee's information, that was created at actual tupe v. TST and was (logically) removed at actual time v. TET (because the information was no longer correct). In EPS TT, the correct revious of each employee typically has a special value, for (Until Changed), as its transaction end time, which indicates that the tuple represents correct information until it withinged by some other transaction. <sup>1</sup> A manuaction rune database has also been valled a **collback database**,<sup>18</sup> because a user can logically roll back to the actual database state at any past point in time T by introving all tuple considery whose transaction time period [9, 19] - v. TET] includes time point 7.

Bitemporal Relations. Some applications require both cold time and transaction time, leading to **bitemporal relations**. In our example, Figure 24.7c shows now the FNR.046 and NEAR FERT non-temporal relations in Figure 24.1 would appear as bitemporal relations for start and NEAR FERT non-temporal relations in Figure 24.1 would appear as bitemporal relations in Figure 24.2 would appear as bitemporal relations in Figure 24.2 would appear as bitemporal relations in these rables, uples whose transaction end time TET is at out the ones representing currently valid information, whereas ruples whose tet is in absolute timestamp are topics that were called until (just before) that timestamp. Hence, the tuples with its in Figure 24.9 correspond to the valid time tuples in Figure 24.7. The transaction stort and attribute test in each tuple is the timestamp of the transaction that created that ruple

<sup>17.</sup> The subsequences are subsequent to a relation second statistic the statistic structure relations. The second to see sughtly defined through

<sup>16</sup> The term rollback here does not have the same measuring as romaxtash unifold two (Proptor 19) Jumpered overviewhere the normal tion updates are throughly out me. Rother, here the updates or by logically arefore, all wing the user to examine the datakser as it uppeared at a provious true, point

| NAME     | <u>88N</u> | SALARY | ÚNKÚ | SUPERVISOP_SSN | vsr        | VET        | <u>TSI</u>           | TET                  |
|----------|------------|--------|------|----------------|------------|------------|----------------------|----------------------|
| Smith    | 123455785  | 25000  | ÷    | 333445555      | 2002-06-15 | now        | 2002-06-08.13:06:58  | 2003 08-04.08 56:12  |
| Smith    | 123455759  | 25800  | £    | 333445555      | 2002/06-15 | 1998-05-31 | 2003-05-04 08.56:12  | June 1               |
| Smith    | 123455781  | 30000  | £    | 333445555      | 2003-08-01 | 00w        | 3003-06-04 06:56:12  | цс.                  |
| Wong     | 3334495655 | 25000  | 4    | 999687777      | 1999-08-20 | 000wr      | 1999-08-20, 11 18:23 | 2001-01-07, 14 39 02 |
| Wong     | 3334456655 | 25000  | 4    | 999887777      | 1777-08-20 | 1226-01-31 | 2001-01-07, (4:31/02 | 112                  |
| Wang     | 3334455555 | 30000  | 5    | 999667777      | 200002-02  | IEM.       | 2001-01-07,14 33:02  | 2003-03-28,00.29 57  |
| Wang     | 321445555  | 30000  | 5    | 999887777      | 706 H22-06 | 1997-03-01 | 1002-09-28.09 23:57  | 16                   |
| Wang     | 353445555  | 40000  | 5    | 8996600000     | 1002-04-01 | pow .      | 1902-00-26.09 23:57  | UG                   |
| Brown    | 222447771  | 28000  |      | 999807777      | 2001-00-01 | <b>N74</b> | 2001-04-27.16 22:05  | 2002-06-12-19-11 01  |
| Brown    | 222447777  | 28000  |      | 999687777      | 2001-09-07 | 1997-09-10 | 2002-00-12.1011-07   | 10                   |
| Nonityph | 655684444  | 38000  | 5    | 313443655      | 2003-05-01 | angen i    | 2003-07-28.09-25:37  | 40                   |
|          | <b>.</b> . |        |      |                |            |            |                      |                      |

EMP\_BT

DEPT\_V7

| ONAME    | <u>DNO</u> - | MANAGE9_SSN | VST        | VET        | <u><u><u></u><u>т</u><u></u><u></u><u>т</u><u></u><u></u><u></u><u>т</u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u></u></u> | TET                 |
|----------|--------------|-------------|------------|------------|------------------------------------------------------------------------------------------------------------------------------------------|---------------------|
| Regen en | 5            | 02000088    | 2001-09-20 | nger       | 7001-09-15,14 52 12                                                                                                                      | 3001-03-29,09-23.57 |
| Repearch | 5            | 888666555   | 20or 00 20 | 1997-03-31 | 2002-00-20.09/20-57                                                                                                                      | UC                  |
| Besserch | 5            | 333445556   | 2002-04-01 | PO-V       | 2002-03-28.09-23:57                                                                                                                      | 60                  |

FIGURE 24.9 Some tuple versions in the bitemporal relations be\_ur and service

Now consider how an update operation would be upplemented on a bitemperal relation in this model of 0 temporal databases?" (wattributes the physically cherged in any tuple except for the transaction end time attribute  $\tau t \tau$  with a value of  $t c \tau^{-1}$  To illustrate how tuples are created, consider the every relation. The connect terrory viol an employee has no in its  $t \tau$ attribute and tota in its ver attribute. If some attribute—eay, SEAV—is updated, then the transaction  $\tau$  that performs the next salary becomes effective (in the real world). Assume that will be solid time within the new salary becomes effective (in the real world). Assume that  $v^{-1}$  is the time grout belove  $v\tau$  is the giver solid time graniburry and that transaction  $\tau$  have timestamp  $\tau_{\pi}(\tau)$ . Then, the following physical changes would be applied to the form  $B^{-1}$  table:

- Make a copy v2 of the current version v; set v2.vt\* (o v\* , v2. (s) to t5(t), v2. t(t) to set, and user v2 in the july v2 is a copy of the provides current version value of is closed at valid time vt = .
- Make a copy e3 of the content version to set 93, v50 root, 93, e61 roomar, 93, such to the tank salary calarity1. Stata (55(1), 93, 54 to 66, and insert 93 moves (0); c3 represents the new current version.

20. Some bitemportal models allow the withermolate robe schenged also, but the interpretations of the riples are different in these models.

<sup>19.</sup> There have been many proposed remporal statistics models. We are describing question and ets bare as examples to diastring the concepts.

 Ser X, (1) (1) since the current version is no longer representing correct information.

As an illustration, consider the first three tuples v1, v2, and v5 in 98, 60 in Figure 24.9. Before the update of Simth's salary from 25000 to 30060, only v1 was in 89,00 and it was the current version and its 315 was all. There a transaction T whose timestamp 13(7) is 2003-06-04,08:36:12 updates the valary to 30000 with the effective valid time of 2003-06-04, 08:36:12 updates the valary to 30000 with the effective valid time of 2003-06-04. The tuple v2 is created, which is a copy of v1 except that its v51 is set to 2003-06-01. The tuple v2 is created, which is a copy of v1 except that its v51 is set to 2003-06-01, and its than the rise vialed time and its 157 is the timestamp of the updating transaction. The tuple v3 is also created, which has the mix salary, its v51 is set to 2003-06-01, and its two is also the timestamp of the updating transaction. Finally, the st of v1 is set to the timestamp of the updating transaction to 010, 08:56:12. Nore that this is a consecutive update, since the updating transaction time of J003, 08:56:12. Nore that this is a consecutive update, since the updating transaction time of J004, 08:56:12. Nore that this is a consecutive update, since the updating transaction time of J003.

Similarly, when Wong's solary and department are updated to the same time) to 30000 and 5, the updating transaction's tenestarys is 2001-01-07, 14:33:02 and the effective valid time for the apdate is 2001-02-01. Hence, this is a posicity tradate because the transaction tan on fanatary 7, 2001, but the effective date was bebruary 1, 2001. In this case, tuple v4 is legically replaced by 95 and 96.

Next, let us illustrate how a **delete operation** would be implemented on a bitemporal relation by considering the tuples v9 and v10 in the 190 [6] relation of Figure 24.9. Here, employee Brown left the company effective August 10, 2002, and the legical delete is carried out by a massaction T with 15(2) = 2002-08-12, 10:11:07. Before this 99 was the current version of Brown, and us 64 we bet, The logical delete is implemented by setting 99, (1 - 6)/2002/08-12, 10; 11:07 we provide the involution of provide the figure 24.9. Finally, (1 - 6)/2002/08-12, 10; 11:07 we provide the and involves the figure 24.9. Finally, (1 - 6)/2002/08-12, 10; 11:07 we provide the strength operation is implemented by strengthe first region as illustrated by v14 in the tage (1 - 6)/2002/08-10 (see Figure 24.9). Finally, (1 - 6)/2002/08-10 (see Figure 24.9). Finally, (1 - 6)/2002/08-10 (see Figure 24.9).

Implementation. Considerations. There are various options for storing the tuples in a temporal relation. One is to store all the tuples in the same table, as in Figures 23.8 and 23.9. Another option is to create two rables, one for the currently valid information and the other for the test of the tuples. For example, in the bitcinporal field or relation, tuples with actor their (1) and o an for their set would be in one relation, the current wide, since they are the ones currently valid (doot its represent the current snaps) such ad all other tuples would be manother relation. This allows the database administration to have different access paths, such as indexes for each relation, and keeps the size of the current table reasonable. Another possibility is to create a third table for corrected tuples where the current table reasonable. Another possibility is to create a third table for corrected tuples where the current actions at

Another option that is available is to reacody (sofficial the attributes of the temporal relation into separate relations. The master for this is that, if a relation has many attributes, a whole new tuple version is created whenever any one of the attributes is updated. If the intributes are updated asynchromously each new version may differ in only one of the attributes, this meetlessly repeating the other numbure values. If a separate relation is created to contain only the autibury that affects exploring synchromously, with the primary key replicated in each relation, the database is said to be in **temporal normal**  form- However, to combine the information, a variation of join known as temporal intersection join would be needed, which is generally expensive to implement.

It is important to note that biteingestal disbases allow a complete record of changes hven a second of corrections is possible. For example, it is possible that two tuple versions of the same employee may have the same valid time but different attribute values as long as then transaction times are disjoint. In the case, the tuple with the later transaction time is a correction of the other tuple version. Even incorrectly entered valid times may be corrected this way. The metorecit state of the database will still be available as a processis database wave for querying purposes. A database that keeps such a complete record of changes and corrections has been called an append only database.

#### 24.2.3 Incorporating Time in Object-Oriented Databases Using Attribute Versioning

The previous section discussed the tuple versioning approach to implementing tropool databases. In this approach, whenever one attribute value without a shale new tuple version is created, even though all the other attribute value will be identical to the previous tuple version. An alternative approach can be used in database systems that support **complex structured objects**, such as object databases (see Chapters 20 and 21) or object relational systems (see Chapter 22). This approach is called attribute versioning.<sup>21</sup>

In attribute versioning, a single complex object is used to store all the temporal charges or the object. Each attribute that changes over time is called a time-varying attribute, and it has its values versioned over time by adding temporal periods to the attribute. The temporal periods new represent valid time, it its saction time, or Precipical, depending on the application requirements. Attributes that do not change are called **non-time-sarying** and are not associated with the temporal periods. For flostrate this, consider the example in Sigure 24,10, which is an attribute versioned valid time representation of PPLORE using the O.4 notation for object databases (see Chapter 21). Here, we assumed that mane and social security number are non-time-vorying attributes (free do not change over time), whereas salary, department, and supervisor are time-varying attributes (they now classify the particle), whereas salary, department, and supervisor are time-varying attributes (they now classify the particle), whereas the time-varying attribute is represented as a list of imples occurd\_start\_times, while or times, builds top, time.

Whenever an antibute is changed in this model, the current arribute corsion is closed and a **new attribute version** for this arribute only is appended to the list. This allows anothers is change asynchronously. The current value for each antibute has can tor its value for the When using antibute versioning, it is useful to include a lifespar temporal attribute associated with the phole object where cohe is one or more value time periods that indicate the valid time of existence for the whole object. Logical deletion of the object is implemented by closing the hission. The constraint that an time period of an attribute within an object should be a subset of the object's despair should be enforced.

21. Attribute versioner, can also be over in the rusted relational model (see Chapter 22)

```
class Temporal Salary
 valid_start_time;
 attribute
 Örde
 attribute
 Date
 valid_and_hms;
 atinbute
 HC-B1
 salarvi
I.
class Temporal_Dept
÷
 attribute
 Date
 valid_stan_time.
 Miribule
 Date
 valid_end_time:
 attribute
 Department_VT_dept
5
ctass Temporal Supervisor
t
 attribute
 Date
 valid start time;
 Date
 shribute
 valid end time,
 ettripute
 Empiryae VT
 SUDARAGOT)
ł
class Temporal (Lifespan
I
 athibute Date
 valid_ star_time.
 attribute Date
 valid, end time.
I.
class Employee VT
 setent employees)
;
 attribute
 list. Temporal Lifespan>
 tifescan
 attrabute
 string
 пате:
 attrabule.
 sining.
 5571'
 InstalTemporal_Salary >
 attribute
 sel_history;
 list-Temporal_Dept>
 attribute
 dept_history;
 ItateTemporal_Supervisore
 supervisor_history.
 attribute.
I.
```

FIGURE 24.30 Possible ODUschema for a temporal valid time Exployee\_VT object class using attribute versioning.

For higenpoint databases, each attribute version would have a tuple with two components:

<valid\_start\_time, valid\_end\_time, mans\_start\_time, mans\_end\_time, value>

The object lifespan would also include both could and transaction time dimensions. The full copabilities of bitemporal databases can hence be available with arribute versioning. Mechanisms similar to those discussed earlier for updating tople versions can be applied to updating attribute versions.

### 24.2.4 Temporal Querying Constructs and the ISQL2 Language

So far, we have discussed how dat randels may be extended with temporal constructs. We now give a biref overview of how query operations need to be extended for temporal querying. Then we briefly discuss the ToQL2 language, which extends SQL for querying valid time, transaction time, and bitemporal relational databases.

In nontemporal relational dotabases, the typical selection conditions involve attribute conditions, and topics that satisfy these conditions are selected from the set of carrier tagler Following that, the attributes of interest to the query are specified by a projection operation for Chapter 5). For example, in the query to retrieve the names of all employees working in department 5 whose salary is greater than 10000, the selection condition would be:

((salesy > 30000) and (nev = 5))

The projected attribute would be seef. In a temporal database, the conditions may involve time in addition to attributes. A **pure time condition** involves only times for example, to select all employee tuple versions that were valid on a certain time point for that were valid during a content tote period [11, [2]]. In this case, the specified time period is compared with the valid time period of each tuple version (1.95), [1, 197], and only those tuples that satisfy the condition are selected. In these operations, a period is considered to be equivalent to the set of time points from TI to (2) inclusives as the standard set comparison operations can be used. Additional operations, such as whether one nume period ends *logate* another starts are also greaded <sup>14</sup>. Some of the more common operations used in queries are additions.

| [OVSTREATED] INCLUDES [CLO2]          | Equivalent to (1 \$ 105 FAND G \$ 104)    |
|---------------------------------------|-------------------------------------------|
| [rays), rayer] included [in h1, r2] - | Equivalent for the nAST AND CHEMICAE      |
| [AVST FAVET] OVERLAPS [11, 12]        | Equivalent to fit = GVET AND G = 138007   |
| [LVST/GVET] BEFORE [11, r]]           | Equivalent for the tWFT                   |
| [TAYST, CVE1] AFTER [t1, t2]          | Equivalent to $t^2 \simeq t  VST$         |
| [LVST. CVET] MEETS BEFORE [r1, c2]    | Equivalent to $r1 = r \sqrt{30} + 1^{21}$ |
| [rast.care] meets_aeter [cl. c2]      | Equivalent to $t^2 + 1 = t VS$            |

In addition, operations are needed to manipulate time periods, such as computing the unson, or intersection, of twos tion, periods. The results of these operations may notheniselyes be periods, but rather temporal elements—a collection of one or more dojout time periods such that no two time periods in a temporal element are directly adjacent.

<sup>32.</sup> A complete set of overations, larger as Allen's algebra, has been defined for comparing time periods.

<sup>24.</sup> This operation oscillation tracial the intersection of the two periods, short empty, in bacabacheric colled DOLERSEC SINCTR.

<sup>24.</sup> Here, 1.3 (no.) refers to row time point in the specifical erandarity. The 96.75 operations has eithespecify if own period starts included under the other period onds.

That is, for any two time periods [(1, -(2)] and [(3, -(4)] in a temporal element, the following three conditions must hold:

- [(1, (2) intersection [13, 74] is empty
- 13 is not the time point fellowing t2 in the given granulativi-
- (1)is not the time point following (4) in the given granularity.

The latter conditions are necessary to ensure unique representations of temporal elements. If two time periods [11, 12] and [13, 14] are adjacent, they are combined into a single time period [11, 14]. This is called coalescing of time periods. Coalescing also combines intersecting time periods.

To illustrate how pure time conditions can be used, suppose a user wants to select all supply eevenings that were valid at any primit during 2002. The appropriate selection ronalition applied to the relation of Figure 24.8 would be

```
[T.VST, T.VST] OVERLAPS [2022/0] [01, 2002 12, 31]
```

Typically, most temporal selections are applied to the valid time dimension. For a hierarporal database, one usually applies the conditions to the currently correct tuples with us as their transaction end times. However, if the query needs to be applied to a previous database more, an solar it cleare is asympted to the query, which means that the query is applied to the valid time tuples that were correct in the database at time it.

In addition to pure time conditions other selections involve attribute and time conditions. For example, suppose we wish to remove all us\_st ruple versions i for employees who worked in department 5 at any time during 2002. In this case, the condition is

```
([T.VST. T.VET] OVERLAPS [2002-01-01, 2002-12-31]) AND (T 1080 - 5).
```

Finally, we give a brief overview of the Tstyl.2 query language, which extends SQL with constructs for temporal databases. The main idea behind 15QL2 is to allow users to specify which it a relation is incatenized (that is, a standard sQL relation) or temporal. The CREATE TABLE statement is extended with an optimed Associatise to allow users to declare different temporal options. The following options are available:

- as verto state, antegrating field time relation with calid time period;
- as write even scene attrix (valid time relation with calid time provid
- AS TRANSACLUM (management time relation with management time period).
- AS verify state equivalences use measuring (bromparal relation, valid time period)
- as each form accommantes one massifies (bitemporal relation, valid time point).

The keywords STATE and EVENT are used to specify whether trune provider time provides associated with the valid time dimension. In 19202, rather than have the user actually see how the temporal tables are implemented (as we discussed in the process sections), the TAG2 language adds query language constructs to specify various types of temporal selections, temporal projections, temporal oggregations, trunsformation onlong granulations, and many other concepts. The back by Stradglass et al. (1995) describes the language.

#### 24.2.5 Time Series Data

Time series data is used very otten in financial, siles, and economies applications. They involve data values that are recorded according to a specific predefined sequence of time points. They are hence a special type of valid event data, where the event time points are predefinitized according to a track calendar. Consider the example of closing daily stock prices of a particular comparty on the New York Stock Exchange. The granularity here is day, but the days that the stock market is open are known (confidinday weekdays). Hence, it has been common to specify a comparisonal procedure that calculates the particular calendar associated with a time series. Typical queries on time series involve temporal aggregation over higher eranularity intercol-—for example, finding the average or maxmum weekly closing stock price or the maximum and imminium monithy closing stock price from the days information.

As another example, consider the daily sales childr amount at each store of a charvof stores owned by a particular company. Again, typical temporal aggregates would be terreving the weekly, insurthly, or yearly sales from the daily sales information (using me sum aggregate hardon), or comparing same store monthly sites with previous monthly sales, and so on.

Pecause of the specialized nature of time series data, and the lack of support in older DWFs, it has been common to use specialized **time series management** systems, it has been general purpose DFMss for managing such information. In such systems, it has been common to store time series values in sequential order in a tile, and apply specialized time series procedures to analyze the information. The problem with this approach is that the full power of bight level querying in languages such as SQL will not be available in such systems.

More recently, some conducted 0906 packages are offering time same extensions, such as the rune series datablade of Informa Universal Server (see Unipter 22). In addition, the ToQC2 language provides same separation rime series in the form of generic tables.

## 24.3 MULTIMEDIA DATABASES

Because the two topics discussed in this section are very broad, we can ered on via very brief introduction to these fields. Section 24.3.1 introduces spatial databases, and Section 24.3.2 briefly discusses multimedia databases.

#### 24.3.1 Introduction to Spatial Database Concepts

Spatial databases provide concepts for databases that keep track of objects in a millidimensional space. For example, corregipping databases that store maps include toodimensional spatial descriptions of their objects—from countries and states to neers entres, roads, seas, and so on. These applications are also known as Geographical Informtion Systems (GIS), and are used in areas such as environmental, energency, and both management. Other databases, such as meteorelogical databases for weather information, are three-dimensional, since temperatures and other meteorological information are related to three dimensional sputial points. In growtal, a spatial database stores objects that have sputial characteristics that describe them. The spatial relationships among the objects are important, and they are often needed when querying the database. Although a spatial database can in general refer to an a-dimensional space for any relivatively four our discussion to two dimensions as an illustration.

The num extensions that are needed for sparal databases are models that can interpret spatial characteristics, its add too, special indexing and storage structures are often needed to improve perferitionce. Let us first docuss some of the model extensions for two-dimensional spatial databases. The basic extensions needed are to include twodimensional geometric concepts, such as points, bites and the segments, coules, polygons, and area, in order to specify the spatial characteristics of objects. In addition, spatial operations are needed to operate on the objects' spatial characteristics—for example, to compute the distance between two objects spatial characteristics—for example, to compute the distance between two objects spatial postilare. To illustrate, conditions—for example, to check whether two objects spatially overlap. To illustrate, conditions—for example, to check whether two objects spatially overlap. To illustrate, conditions—for example, to check whether two objects spatially overlap. To illustrate, conditions—for example, to check whether two objects apartially overlap. To illustrate, conditions—for example, to check whether two objects and lingloways, water prints generally have storic spatial characteristics, such as structs and highways, water pumps (for free control), pulse startons, are startons, and hospitals. Other objects have dynamic spatial rharacteristics that charge over time, such as policy vehicles, imbulances, or tratricks.

The following caregories illustrate three repical types of spatial queries:

- Range query: Ends the objects of a particular type that are written a given sportal area or within a particular distance from a given location. Hor example, finds all hospitals within the Daillos cary area, or finds ad a obubmees within five finds of an accident location.)
- Nearest neighbor query. Finds an object of a particular type that is closest to a given location. (For example, finds the police car that is closest to a particular location.)
- Special poiss or overlaps: Typically joins the objects of two types based on some search conditions such as the objects intersecting or swerlapping spatially of being within a certain distance of one another. (For example, tools all errors that fall on a popuhighway or fitted all bonnes that are within two on every lake 1.

For these and other types of spatial queries to be answered efficiently, special techniques to spatial indexing are needed. One of the best known techniques is the rise of **Rotrees** and their variations. Rotrees group together objects that are in close spatial physical proximity on the same leaf nodes of a tree-structured index. Since a leaf node of in point theory a certain number of objects, algorithms for dividing the space into rectangular subspaces that include the objects are meaded. Typical errority for dividing the space include include in white space the rectangle areas, since this would lead to a quarker nurrowing of the search space. Problems such as having objects with availapping spatial areas are handled in different variations of Rotrees. The internal nodes of Rotrees are associated with rectangles white area covers all the rectangles in its subtree. Hence, Rotrees on easily arrawer queries, such as find all objects in a given area by lumming the tree search to those subtrees whose terna covers all the rectangles in its subtree. Hence, Rotrees on easily arrawer queries, such as find all objects in a given area by lumming the tree search to those subtrees whose ternangles intersect with the area given in the query. Other spatial storage structures include quadrates and their cariations. Quadrates generally divide each space or subspace introqually storal areas, and proceed with the subdivisions of each subspace to identify the positions of various objects. Recently, many newer spatial access structures have been proposed, and this area is still an acrive research area.

#### 24.3.2 Introduction to Multimedia Database Concepts

**Multimedia databases** provide features that allow users to store and query different type of multimedia information, which includes orages which as photos or drawings), when dips (such as navires, newarcels, or home videos), *mafarabje* (such as sorigs, phone messages or speeches), and documents (such as buoks or articles). The main types of database queries that are needed involve locating informedia sources that contain certain objects of interest. For example, one may want to locate all video clips in a video database that include a certain person in them, say Bill Clinton. One may also want to retrieve video clips based on certain activities included in them, such as a video clips where a giol is scoted in a soccer game by a certain player or team.

The above types of queries are referred in as content-based retrieval, because the indimiedia source is being refrieved based on its containing certain objects or activities. Hence, a multimedia database must use same model to organize and index the multimedia sources based on their contents. Identlying the contents of multimedia sources is a difficult and unne-consuming risk. There are non-main approaches. The tist is based on automatic analysis of the multimedia sources to identify certain insthematical characteristics of their contents. This approach uses different rechniques depending on the rype of multimedia source (image, test, video or audito). The second approach depends on manual identification of the objects and activities of interest it, each multimedia source and on using the information to index the sources. This approach can be applied to all the different multimedia sources, but it requires a manual greprocessing place where a person has to scan each autimedia sources to identify and catalog the objects and activities in contains se that they can be used to index these sources.

In the requireder of this section, we will very briefly discuss some of the characteristics of each 1770 of nultimedia source. Integes, video, and/o, and text sources, in that order.

An image is typically solid either in row form as a set of pixel of cell values, or a compressed form to save space. The image state descriptor describes the geometric shape of the raw image, which is typically a rectangle of cells of a certain width and height llence, each image can be represented by at or by a grid of cells. Each cell contains a pixel value that describes the cell content. In black/white images, pixels can be one built in gray scale or color images, a pixel is multiple bits. Because images may require large anisonauts of space, they are often stored in compressed form. Compression standards, such as off in 1750, use various mathematical transformations to reduce the number of cells stored but still maintain the main image characteristics. The number of cells stored but still maintain the main image characteristics.

that can be used include Discrete Fourier Transform (DIT). Discrete Cosine Hansform (.XT), and waveler transforms

In identity objects of interest in an image, the image is typically divided into homogeneous segments using a homogeneity predicate. For example, in a color image, cells that are adjacent to one another and whose pixel values are close are grouped into a segment. The homogeneity predicate defines the conditions for how to automatically group those cells. Segmentation and compression can hence identity the main characteristics of an image.

A typical image database query would be to find images in the database that are similar to a given image. The given image could be an isolated segment that contains, say, a partern of interest, and the quere is to becate other images that contain that some pattern. There are too main rechniques for this type of search. The first approach uses a distance function to compare the given image with the stored images and their segments. If the distance value returned is small, the probability of a march is highindexes can be created to group together stored images that are close in the distance instances to limit the search space. The second approach, called the transformation approach, measures image similarity by having a small number of transformations that can transform one image's cells to march the other image. Transformations include rotations, transformations, and storing. Although the latter approach is more general, it is also more time consuming and dificult.

A video source is typically represented as a sequence of humes, where each frame is a still image. However, rather than identifying the objects and activities in every individual frame, the order is divided into video segments, where each segment is made up of a sequence of contiguous tranes that includes the same objects/activities. Each segment is identified by its starting and ending frames. The objects and activities identified in each video segment can be used to index the segments. An indexing technique called frame segment root has been proposed for video indexing. The index includes hore objects, such as periors houses, cars, and activities, such as periors dones are also often compressed using standards such as MCBC.

A text/document source is basically the full text of some article, book, or magazine. These sources are typically indexed by identifying the keywords that appear in the text and their relative frequencies. However, filler words one eliminated from that process. Because there could be too many keywords when artempting to index a collection of documents, techniques have been developed to reduce the number of keywords to those that are thost relevant to the collection. A technique called sugalar value decouplesitions (switch is based on matrix transformations can be used for this purpose. An indexing technique called telescolog rector order, or TV-trees, can then be used to group similar documents trajether.

Authorisonness include stored recorded messages, such as speeches, class generitations, or even servedlance recording of ghome messages or conversitions by law enforcement. Here, discrete transforms can be used to identify the main characteristics of a certain grison's voice in order to have similarity based melosing and retrieval. Authoritations to framess include loadness, intensity, prich, and clarics.

# 24.4 INTRODUCTION TO DEDUCTIVE DATABASES

#### 24.4.1 Overview of Deductive Databases

In a deductive database system, we repically specify roles through a declarative language – a language in which we specify what to achieve to their than how to achieve it. An inference engine (or deduction mechanism) within the system can deduce new facts from the database by interpreting these index. The model used for deductive databases is closely related to the relational data model and particularly to the domain relational calculus formalism (see Section 6.6). It is also related to the field of legic programming and the Prolog language. The deductive database work based on logic has used Prolog as a starting point. A variation of Prolog called Datalog is used to define rules declaratively in conjunction with an existing set of relations, which are there allows treated as hierals in the language. Although the language structure of Datalog resembles that of Prolog, its operational semantics—that is, how a Datalog program is to be executed—is still different.

A deductive database uses two main types of specifications: facts and rules. Facts are specified in a moment similar to the way relations are specified, except that it is not necessary to include the attribute names. Recall that a uple in a relation describes some real-world fort whose meaning is parify determined by the attribute names. In a deductive database, the meaning of an attribute value in a tuple is determined scilely by its pointer working the ruple. Rules are somewhat similar to relational views. They specify curried relations that are not actually sored but that can be formed from the facts by applying inference mechanisms looked on the rule specifications. The mean difference between ruleand views is that rules new involve recursion and hence may yield virtual relations the current by defined in terms of bost relational views.

The evaluation of Prolog programs is based on a rechnique colled backward eizanig, which involves a tog-alown evaluation of goals. In the deductive databases that use Datalogattention has seen devoted to handling latee voluenes of data stored as a relational database. Hence, evaluation rechniques have been devised that resemble those for a bottom-up evaluation. Prolog suffers from the limitation that the order of specification of facts and rules is significant in evaluation, moreover, the order of literals telefined later in Section 24.4.31 within a rule is significant. The execution techniques for Datalog programs attempt to circumvent these woldcars.

### 24.4.2 Prolog/Datalog Notation

The notation used in Prolog/Datalog is based on providing predicates with another names. A predicate has in implicit moming, which is suggested by the predicate name, and a fixed number of arguments. If the arguments are all constant values, the predicate simply states that a certain ract is true. If, on the other hand, the predicate has variables as arguments, it is either considered as a query or as part of a rule or constraint. Throughout this chapter, we adopt the Prolog concention that all **constant values** in a producte are other numeric or character strings, they are represented as identities (or names) starting with bacemase know only, whereas variable names allows start with an ignetions fear-

Consider the example shown in Figure 24.11, which is based on the relational database of Figure 5.6, but in a much simplified form. There are three predicate nones: sapersize superior and value/duate. The supervise predicate is defined via a set of facts, each of which has two arguments, a supervisor name, followed by the name of a direct supervisee (subordinate) of that supervisor. These facts correspond to the actual data that is stored in the database, and there can be considered as constituting a set of ruples in a relation Societies with two attributes obose schema is

#### SuPERvIAE(Supervisor.Supervisee)

Thus, supervise(X, V) states the fact that "X apprvises Y." Notice the ortiston of the attribute names in the Probe notation. Attribute names are only represented by vistue of the position of each argument in a predicate, the first argument represents the supervisor, and the second argument represents a direct subordinate.

The other two predicate names are defined by roles. The main contribution of deducinve databases is the ability to specify recursive inles, and to provide a framework for interring new information based on the specified rules. A tule is of the form head is body, where size read as "it and only if." A rule usually has a single predicate to the left of the symbol—called the head or left-hand side (1915) or conclusion of the tule—and one or more predicates to the right of the signified method we arguments is said to be grounds or premise(s) of the rule. A predicate with constants is arguments is said to be grounds we also refer to it as an instantiated predicate. The arguments is said to be grounds we also refer to it as an instantiated predicate. The arguments of the predicates that appear in a rule typically include a number of cariable sputiols, although predicates can also contain constants as arguments. A rule specifies that, if a patterial assignment or hinding of curstant values to the variables in the body (RIEs predicates) makes all the RIEs predicates true, it also makes the head (TIEs predicate) true by using the same assignment of curstant values to variables. Hence, a rule provide size with a way of generating new facts that are instantiations if the head of the rule. These new facts are based on facts that

ίa). (b) Facts iames supervise(franklin.john) supervise(franklin.ramesn). supervise)(ranklin,jayce), IrandiA jennifer supervisit(jermiler alicia). supervise(jenniler ahmad). supervise(james, franklin) supervise(james,jennifer). iaha. ramasin elinia ahmad KT/OB ... Bulas. superior(X,Y) : supervise(X,Y). supercrix(Y) > supervise(X,Z), supercriz(Y). subordinate(X,Y) > suprimor(Y X) Quaries superior ames, YO7 superior(james.joyce)?



already exist, corresponding to the instantiations (or bindings) of predicates in the bady of the rule. Notice that by listing multiple predicates in the bady of a role we implicule apply the **logical and** operator to these predicates. Hence, the commos between the SPS predicates may be read as meaning "and "

Consider the defaution of the predicate supernor in Figure 24.11, whose first areament is an employee name and whose second argument is an employee who is either a deter or an indexet subordinate of the first employee. By itsidete stationhitate, we mean the subordinate of some subordinate down to any number of levels. Thus supervor(X, Y)stands for the rate that "X is a superior of Y" through direct or indirect supervision. We can write two tubes that regeriter specify the meaning of the new producate. The that rule under Roles in the figure states that, for every value of X and Y, if supervise (X,Y), the rule body  $-\infty$  rule, then superior (X,Y) – the rule head  $-\infty$  also rule, since Y would be a direct subordinare at X (at one level down). This rule can be used to geneture all direct superior/subordinate retarion dups from the focts that define the supervise predicate. The second recursive rule states that, if supervise(X, Z) and superior(Z, Y) are both intetheir supernor (X,Y) is also true. This is an example of a recursive rule, where over  $\delta$ the rule body predicates in the 8018 is the same as the rule bead predicate in the 11% in general, the rule body defines a number of premises such that, if they are all role, we can deduce that the conclusion in the cide head is also true. Notice that, if we have two for more) only with the same bead (1)18 producated, it is equivalent to saving that the realdate is true tithat as, that it can be instantiated) if either one of the bodies is true, hence, it is equivalent to a logical ne operation. For example, if we have two rules X :- Y and X := Z, they are equivalent to a rule X := Y  $\omega \epsilon$  Z. The latter form is not used in deducave systems, however, because it is not in the sumdard form of rule, called a been clause. as we disease in Section 24.4.4.

A Prolog system contains a number of builts in predicates that the system can interpret directly. These typically include the equality comparison operator  $\neg(X,Y)$ , which returns true if X and Y are identical and can also be written as X+Y by using the standard infix paration.<sup>25</sup> Other comparison operators for numbers, such as  $<_{1}<_{2}<_{2}>_{3}$  and >=con-be treated as privity predicates. Arithmetic functions such as  $<_{1}<_{2}<_{3}>_{3}$  and ( can be used as areaments in predicates in Profog. In contrast, (Madog (in its base form) does an allow functions such as an indicates in Profog. In contrast, (Madog (in its base form) does an allow functions such as an indicates in Profog. In contrast, (Madog (in its base form) does an allow functions such as an indicates in Profog. In contrast, (Madog base of the main differences between Profog and Datalog, However, later extensions to Datalog have been proposed to include functions.

A query typically involves a predicate symbol with some variable arguments and its measure (or "answer") is to deduce all the different constant combinations that, when bound tassigned) to the variables, can make the predicate time. For example, the first query in Figure 24.11 requests the names of all subordinates of "james" at any level. A different type of query, which has only constant symbols as arguments, returns rober arms or a fable result, depending on whether the signments provided can be deduced free

A Prelygousiem repeatly any number of different equation predicates from have different mer pretations.

the twos and toles, for example, the second query in Figure 24.11 means true, since superficit(janes, joyce) can be deduced.

#### 24.4.3 Datalog Notation

In Doulog, as in other logic-based languages, a program is built from base objects called **stomic formulas**. It is costomary to define the syntax of logic-based languages by describing the syntax of atomic formulas and identifying how rbsy can be combined to form a program. In Patolog, atomic formulas are **laterals** of the form  $p(a_1, a_2, \dots, a_n)$ , where p is the predicate name and n is the number of arguments for predicate p. Different predicate symbols can have different numbers or arguments and the number of arguments p of the other symposities of arguments p is sometimes called the **arity** or **degree** of p. The arguments q in be either constant values or variable names. As mentioned earlier, we use the convention that constant values either are numeric or start with a *logicities* character, whereas cancele names always start with an agencies character.

A number of built-in predicates are included in Datalog, which can also be need to construct atoms formulas. The built-inspectates are of two main types the binary comparison predicates  $\kappa$  (less),  $\kappa_{\pm}$  (less or  $\_$ equal),  $\kappa$  (greater), and  $\nu_{\pm}$  (greater), and  $\nu_{\pm}$  (greater) over ordered domains; and the comparison predicates  $\pm$  (equal) and /= (not equal) over ordered domains; and the comparison predicates  $\pm$  (equal) and /= (not equal) over ordered domains; and the comparison predicates  $\pm$  (equal) and /= (not equal) over ordered or mordered domains. These can be used as binary predicates with the same functional syntax is other predicates—for example by writing less(X, -1) for they can be specified by none the customary offs notation X<3. Notice that, because the domains of these predicates are potentially infinite, they should be used with care in rule definitions. For example, the predicate greater (X, -2), it used alone, generates an infinite set of values for X that satisfy the predicate (all integer numbers greater than 3).

A literal is other an on the formula as defined contern called a positive literal for an atomic forgula preceded be not. The litter is a negated atomic formula, called a negative literal. Thralog programs can be considered to be a subscript the predicate calculus formulas, which are somewhat similar to the formulas of the domain relational calculus free Series 6.7%. In Datalog, however, these formulas are first converted into what is known as clausal form before they are expressed to Datalog, and only formulas given in a restricted clausal form colled from clauses?<sup>6</sup> can be used in United.

#### 24.4.4 Clausal Form and Horn Clauses

Recall from Section 0.6 that a formula in the relational obvious is a condition data includes predicates called along (iss-edition ratios). In addition, a formula conbote quantitiers—mainch, the noiscense quantitier (for all) and the existential quantitiers.
(there exists). In clausal form, a formula must be transformed into another formula with the following characteristics:

- All variables or the formula are universally guarantied. Hence, it is not necessary as include the universal quaranties (for all) explicitly: the quarantiers are removed, and all variables in the formula are anjskudy quartified by the universal quartitier.
- In claused form, the formula is spade up of a matther of clauses, where each clause is composed of a number of filewik connected by a 8 logical connectives only. Hence, each clause is a displacement literals.
- The charges characters are connected by ASD logical connectives only, to form a for mula. Hence, the closed frame of a formalia is a comparer at of clauses.

It can be shown that any formula case be concerned func-classed form. For our purpose, we are mainly interested in the form of the individual clouses, each of which is a dopinetion of Interals. Recall that laterals can be positive laterals or negative laterals. Consolar a clause of the form:

$$\operatorname{aot}(\mathbb{P}_{+}) \circ \otimes \operatorname{not}(\mathbb{P}_{+}) \circ \otimes \operatorname{not}(\mathbb{P}_{p}) \circ \otimes \mathbb{Q}_{1} \circ \otimes \mathbb{Q}_{2} \circ \otimes \mathbb{Q}_{2} \circ \otimes \mathbb{Q}_{p}$$
 (1)

This clause has n negative hierals and or positive laterals. Such a clause can be transtorated into the following equivalent logical formula:

$$P_1$$
 AND  $P_1$  AND  $P_2$  AND  $P_3 = 9 Q_1 \text{ OR } Q_2 \text{ OR } P_2 \oplus Q_2$  (2)

where -> is the **muplice** symbol. The terminas (1) and (2) are repredent, meaning that then truth values are always the same. This is the case because, it all the P, hterals (i = 1, 2, ..., n) are rate, the formula (2) is true only it is least one of the Q's is more which is the meaning of the -> (implies) worked. For formula (1), it all the P, hterals (i = 1, 2, ..., n) are true, ther negations are all take; so in this case formula (1) is true only it at least one of the Q's is true in Datalog, rules are expressed as a restricted form of choices called **Horn** charses, in which a choice can compare all only one positive hereal. Hence, a Horn charse is on the form

$$\operatorname{nor}(P_1) \oplus \operatorname{nor}(P_2) : \mathfrak{R} = \operatorname{or}(\operatorname{nor}(P_2) \otimes \mathbb{Q})$$
 (1)

or of the form

$$\operatorname{mor}(\mathbf{P}_1 \operatorname{EOR}\operatorname{mor}(\mathbf{P}_2) \operatorname{CR}) = \operatorname{OR}\operatorname{mor}(\mathbf{P}_n)$$
 (4)

The Hony chose in (5) can be transformed auto the chose

$$P_1$$
 AND  $P_2$  AND  $P_3$  AND  $P_6 \approx 2.0$  (5)

which is statten in Datalog is the following rule

$$Q_{2}(P_{1}, P_{2}, ..., P_{n})$$
 (6)

The Hom clause in (4) can be transformed into

$$P_1 \text{ and } P_2 \text{ and } P_3 \text{ and } P_4 \neq 0$$
 (2)

which is written in Datalog as follows:

$$\Gamma_{\mu} \Gamma_{\mu} \dots \Gamma_{\mu}$$
 (3)

A Datalog rule, as in (6), is hence a Horn classes, and its meaning, based on formula (3), is that if the predicates  $P_1$  and  $P_2$  and  $z_1$  and  $P_n$  are all true for a particular binding to their canable arguments, then Q is also true and can hence be inferred. The Datalog expression (8) can be considered as an integrity constraint, where all the predicates true; be intert to satisfy the query.

In general, a query in Datalog consists of two components

- A Dutillog program, which is a finite set of rules.
- 8 literal P(X<sub>1</sub>, X<sub>2</sub>, ..., X<sub>n</sub>), where each X<sub>n</sub> is a variable or a constant.

A Prolog or Datalog system has an internal interence engine that can be used to procesand compute the teachy of such queries. Prolog interence engines regically return one result to the query (that is, one set of values teache variables or the query) of a time and mast be prompted to return additional results. On the contrary, Datalog returns results set-at-w-time.

#### 24.4.5 Interpretations of Rules

There are two main alternatives for interpreting the theoretical meaning of tubes projtheoretic and codet-theoretic. In gractical systems, the interence mechanism within a system defines the exact interpretation, which may not confinde with either of the two theoretical interpretations. The interpretation of the meaning of rules, he fus action, we first provides a computational interpretation of the meaning of rules. In this action, we first discuss the two theoretical interpretations. Inference mechanisms are then discussed briefly as a way of defining the meaning of rules.

In the **proof-theoretic interpretation** of roles, we consider the facts and roles to be true statements, or axioms. Ground axioms contain no variables. The facts are ground axioms that are given to be true. Bales are called **deductive axioms**, since they can be used in deduce new facts. The deductive axioms can be used in construct proofs that derive new facts from existing facts. For example, Figure 24.12 shows how to prove the fact superior (james, ahmad) from the roles and facts given in Figure 24.11. The prooftheoretic interpretation gives us a procedural or computational approach for computing an answer to the Datalog squery. The process of proving whether a certain fact (theorem) holds is known as theorem proving

1. superior(X,Y) :- supervise(X,Y). 2. superior(X,Y) :- supervise(X,Z), superior(Z,Y)

3. superviseijerimler, ahmadi.

- 4 superviceljames, ennieri
- 5 superiorijelimfel,ahmadi.

(rule 1) (rule 2)

(ground sxism, given) (ground sxion, given) (apply rule 1 on 3) (apply rule 2 on 4 and 5)

HGURE 24.12 Proving a new fact.

<sup>6</sup> superiorijames ahmaa).

The second type of interpretation is called the **model-theoretic** interpretation. Here, given a finite or an infinite domain of constant values.<sup>27</sup> we assign to a peedic or every possible combination of values as arguments. We must then determine whether the predicate is true or take, in general, it is sufficient to specify the combinations of arguments that make the producte true, and to state that all other combinations make the predicate false. If this is done for every predicate it is called an interpretation of the set of predicates. For example, consider the interpretation shows in Figure 24.15 for the predicates supervise and super or. This interpretation assigns a truth value (true or take) to every possible combination of ingument values (from a finite domain) for the two predicates.

An interpretation is called a **model** for a specific set of rides it those intervalues, much interpretations that is, for any values assured to the variables in the rides the head of the rules is true when we substitut. The truth values assigned to the predicate

Rules

 $\begin{array}{l} \mbox{superior}(X,Y) > \mbox{superior}(X,Y), \\ \mbox{superior}(X,Y) > \mbox{superior}(X,Z), \mbox{superior}(Z,Y), \end{array}$ 

Interpretation

Kouwet Facts:

supervisė(hanki n. joim) is true supervisė(hanki n. ramosti) is true supervisė(hankin joyde) is true supervisė(jennier akciej is true supervisė(jennier akciej) is true supervisė(james frankin) is true supervisė(james,jennie) is true supervisė(james,jennie) is true

Derivent Facts;

supenor(Irandin john) is true supenor(Irandin ramesh) is true supenor(Irandin ramesh) is true supenor(Jennile: alcla) is true supenor(Jennile: alcla) is true superior(James.Frankin) is true superior(James.Jenniler) is true superior(James.John) is true

FIGURE 24.13 An interpretation that is a minimal model.

27. The most composedy chosen door in is time and scalled the Merhana Course.

In the body of the role by that interpretation, blence, whenever a particular substruction (binding) to the variables in the roles is applied, if all the predicates in the body of a role are true under the interpretation, the predicate in the body of the role must also be true. The interpretation shown in Figure 24.13 is a model for the two roles shown, since it can never cause the roles to be violated. Notice that a role is violated if a particular birsing of constants to the variables makes all the predicates at the role body true but makes the predicate in the role body for both the role body for some interpretation, but supervise(a, c) is not true, the interpretation cannot be a model for the requirements:

#### superior(X,Y) :- supervise(X,Z), superior(Z,Y)

In the model-theoretic approach, the meaning of the rules is established by princing a model for these rules. A model is called a **minimal model** for a set of rules if we cannot change any fort from true to false and still get a model for these rules. For example, consider the interpretation in Figure 24.13, and assume that the supervise predicate is defined by a set of known thers, whereas the superior predicate is defined as an interpretation (needed) for the rules. Suppose that we add the predicate supernor (tanes), both to the track the rules. This temants a model for the rules shown, but it is not a functional model, since changing the truth value of superfiel (tanes), bob) how true to take still provides us with a model for the rules. The model shown in Figure 24.13 is the minimal model for the supervise predicate.

In general, the imminal model that corresponds to a given set of facts in the mudelthemetic interpretation should be the same as the facts generated by the proof theorem interpretation for the same original set of ground and deductive axioms. However, this is generally true only for rules with a simple structure. Once we allow negation in the specification of rules, the correspondence between interpretations does not hold, hi foculture negation, numerous minimal models are possible for a given set of facts.

A third approach to interpreting the meaning of rules involves defining an interence mechanism that is used by the system to deduce facts from the rules. This inference mechanism would define a **computational interpretation** to the meaning of the rules. The Proleg logic programming language uses its inference mechanism to define the meaning of the rules and facts in a Prolog program. Not all Prolog programs correspond to the prosttheorem or model-theorem interpretations; it depends on the type of rules in the program. However, for many simple Prolog programs, the Prolog inference mechanism interv the facts that correspond either to the proof theorem; interpretation or to a nominal model under the model-theorem interpretation.

#### 24.4.6 Datalog Programs and Their Safety

There are two main methods of defining the trath values of predicates to actual Datalog programs. Bact-defined predicates (or relations) are defined by listing all the combinations of values (the tuples) that make the predicate must These correspond to base relations whose contents are stored in a database system. Figure 24.14 shows the fact-defined productes employee, male, female, department, supervise, project, and workson,

```
employee(john)
 male(john)
 male(iranklin)
employee(Nen din).
empkyee(siicia).
 male(remesh).
amolovae/jerviller).
 male(ahmad)
 maleijamest
employee(raineell).
employee(joyce).
 lemale(elidia)
employee(ahinad).
employeeijantesj.
 famalejjann fer),
 famale((ovce))
salary(juhn,30000).
salary(frant/tin.40000).
 project(products).
$8,8vy(alicas,25000)
 project(producty).
salary(jennifer,40000).
 project(produciz).
sale-y(ramesh/38000).
 project/computerization).
salary(joyce,25003).
 project(rearganization)
$8(2:yiahmad,250(0)).
 projectificewberrelins).
salary(james,55000).
 worksonijohn products,32}.
cosarment(phn,research)
 workson(john.producty,8).
deuarrypenh(tranklin,research)
 workson(ramesh,productz.40).
department(alicia,administration)
 worksen(jeyce.products.20).
deuzrimeni) jarn fer administration).
 workson(joyce.producty,20)
 workscriftanklin, croducty, 10).
popartment(remetin,research)
 workschiftanklin,productz, 10).
orparbment(loyce.research)
department(ahmed.administration)
 worksen(franklin/computerization 10).
 workson/trankin,reorganization,10)
departmentijames,headguartersi
 workson(alicia,newbenefils,(+)).
 workson(alicia.computerization, 10)
supervise(franklin.john)
 workson(ahmad.computanzation,35)
supervise(tranktinursmesh)
supervise(hankingovce).
 warkcon(ahmari newbenehis 5)
superwae(jennder akcia).
 warkson(jennder newtoenehts,20)
 workson/jengiler reorganization (15).
supervise(jennile; ahmad)
supervise(james.tranklin).
 workson/jamas reorganization, 10).
supervise(james.jenniler)
```

FIGURE 24.14 Fact predicates for part of the database from Figure 5.6

which correspond to part of the retational database shown in Figure 5.6. Rule-defined predicates (or views) are defined by being the head (10-1) of one or more Datalog takes they correspond to cornal relations whose contents can be interred by the interence engine. Figure 24.15 shows a comber of tole-defined predicates.

A program or a rule is said to be safe if it generates a bone set of facts. The general theoretical problem of determining whether a set of rules is site is undecidable. However, one can determine the safety of restricted forms of rules, bot example, the index shown in Figure 24.18 are safe. One situation where we get unsafe toles that can generate an intermedent of facts areas when one of the variables in the rule can range over an infinite domain of values, and that variable is not builted to ranging over a mitter clarier. For example, consider the rule can ende

big salary(Y) > Y >80000

Here, we can get an infinite result at Y ranges even all possible integers. But suppose that we change the rule as follows:

```
\log_salary(Y) > coupleyer(X), solary(X,Y), Y > 60202
```

```
superior(X,Y) -- superiorse(X,Y),
superior(X,Y) -- superior(Y,X),
subordmate(X,Y) -- superior(Y,X),
superivsor(X) -- amployee(X) -- superivse(X,Y),
over_40K, emp(X) -- employee(X), satarv(X,Y), Y >= 40000,
under (40K, superivsor(X) -- supervisor(X), ro%over_40_K -- emp(X),
main products (amp(X) -- employee(X), workson(X,products Y) Y >= 20
```

 $presiden V(X) \sim emp(a_{y}ee(X), not(supervise(Y|X))$ 

FIGURE 24.15 Rule-defined predicates.

In the second rule, the result is not infinite, since the values that Y can be bound to are new restricted to values that are the salary of some employee in the database - presion ably, a finite set of values. We can also rewrite the rule as follows:

big\_salary(Y) > Y > 6XXX, employee(X), salary(X,Y)

In this case, the role is still theoretically safe. However, in Prolog or any other system that uses a top-down, depth-first inference mechanism, the rule creates an infinite loop, ance we first search for a value for Y and then check whether at is a salary of an employee. The result is generation of an infinite number of Y values, even though these, after a certain point, cannot lead to a set of role SHS predicates. One definition of Datalog considers both rules to be sufe, since it does not depend on a particular interence inechanism. Nonetheless, it is generally advisable to write such a tale in the salest form, with the predicates that restrict possible bindings of variables placed first. As another example of an onsale rule, consider the following rule.

las, something(X,YE - employee(X).

Here, an infinite number of Y values can again be generated, since the variable Y appears only in the head of the rule and hence is not innited to a birte set of values. To define sate rules more formable, we use the concept of a braited variable. A variable X is **limited** in a rule of (1) it appears in a regular (not britten) predicant in the body of the rule, (2) it appears in a predicate of the torm X-x or c-X or (c t < x) and X < c < x) in where c < t, and t < 2 are constant values; or (3) if appears in a predicate of the torm X-x or (c t < x) and X < c < x) in where b = t, and t < 2 are constant values; or (3) if appears in a predicate of the form X-Y or Y=X an the rule body, where Y is a limited variable. A rule is said to be safe of all jis cariables are limited.

#### 24.4.7 Use of Relational Operations

It is straightforward to specify many operations of the relational algebra in the form of Datalog roles that define the result of applying these operations on the database relations (fact predicates). This means that relational queries and views can easily be specified in Datalog. The additional power that Ukralog provides is in the specification of recursive. queries, and views based on recutsive queries. In this section, we show how some of the standard relational operations can be specified as Datalog rules. Our examples will use the base relations (fact defined predicates) religinet religivations and religible whose schemas are shown in Figure 24.16. In Datalog, we do not need to specify the attribute nameas in Figure 24.16 mithers the arity (deprected each predicate is the important aspect, heap practical system, the domain (data type) of each attribute is also important for operators are has UNION, INTERSFER COV, and NON, and we assume that the attribute types are comparable for the parrous operations, as discussed in Chapter 5.

Figure 24.16 dilustrates a paraller of basic relational operations. Notice that, if the Divalog model is based on the relational model and hence assumes that predicates (fair relations and query results) specify sets of ruples, duplicate tuples in the same predicate are automatically eliminated. This may or may not be true, depending on the Davalog inference engine. However, it is definitely not the case in Prolog, so and of the rules in Figure 24.16 that involve duplicate elimination are non-correct for Prolog. For example, if we want to specify Prolog rules for the UNION operation with duplicate elimination, we must rewrite them as follows.

```
union_one_two(X,Y,Z) > rel_one(X,Y,Z),
autor_one_two(X,Y,Z) > rel_two(X,Y,Z), not(rel_one(X,Y,Z)).
```

However, the rules shown in Figure 24.06 should work for Datafor, if displicates are automonically elitinitated. Similarly, the rules for the TROJECT operation shown in Figure

```
rel une(A.B.C)
rel. two(D.E.F).
rel_three(G,A,f.J).
select_one_A_eg_c(X,Y,Z) > rat_one(p,Y,Z).
select_one_B_less_5(X,Y,Z) > rel_one(X,Y,Z)_Y<5
select one A equic and B tess 5(X,Y,Z) - ret_one(c,Y,Z), YkS.
select one A eq a or B less f_1(X,X,Z) > ret pred <math>Y_1(Z)
select one A log of or B lass 5(X,Y,Z) :- religone(X,Y,Z), Yk5.
project three on G_H(WIX) > rel_(three(WX,X,Z)).
sinion one two(X,YZ) - rel_one(X,YZ)
vmion_one_twp(X,Y,Z) > rel_twp(X,Y,Z)
intersect_one_two(X,Y,Z) ~ re(_one(X,Y,Z), ref_two(X,Y,Z)
difference_two_one(X,Y,Z) - rel_two(X,Y,Z), nrc(rel_one(X,Y,Z)).
carl_prod_one_lhres(T.U.V.W.X.Y.Z) -
 rel_one(T,J,V), rel_three(W(X,Y,Z))
palural join one librea C et G(U,V,W,X,YZ) ~
 wel_one(U,V,W), rel_threa(W,X,Y,Z)
```

FIGURE 24.16 Predicates for illustrating relational operations.

24.16 should work for Donalog in this case, but they are not correct for Diolog, since diplicates would appear in the latter clise.

#### 24.4.8 Evaluation of Nonrecursive Datalog Queries

In order to use Datalog as a deductive database system, it is appropriate to define an interence mechanism based on relational database query protessing concepts. The inherent strategy involves a bottora-up evaluation, startung with base relations, the order of operations is kept flexible and subject to query optimization. In this section, we discuss an inference mechanism based on relational operations that can be applied to **nonrecursive** Datalog queries. We use the foct and role base shown in Fagures 24.14 and 24.15 to illustrate out discussion.

If a query involves only fact-defined predicates, the interence becomes one of searching anyong the facts for the query result. For example, a query such as

department(X) escarch)?

is a selection of all couplinger names X who work for the research department. In relational algebra, it is the query:

 $\pi_{S1}$  ( $\sigma_{S2}$ ,  $\pi_{Recenter}$  (department))

which can be unswered by searching through the foctalefined coedicate depart nerv(X, Y). The query involves relational PECECT and DROJECE operations on a base relation, and it can be bandled by the database classy processing and optimization techniques discussed in Chapter 15

When a query involves title-defined predicates, the inference incubation must courpute the result based on the title definitions. It a query is nonrecursive and involves a predicate p that appears as the head of a tule  $p := p_1, p_2, \dots, p_n$  the strategy is first to compute the relations corresponding to  $\mathbf{p}_1$ ,  $\mathbf{p}_2$ ,  $\dots$ ,  $\mathbf{p}_n$  and then to compute the relation corresponding to p. It is useful to keep track of the dependency among the predicates of a deductive database in a predicate dependency graph, Figure 24,17 shows the graph for the fact and rale predicates shown in Figures 14.14 and 24.15. The dependency graph contains a node for each preducate. Whenever a predicate A is specified in the body (3Hs) of a rule, and the basel (LHS) of that rule is the predicate B, we say that B depends on A. and we draw a directed edge from A to B. This indicates that, in order to compute the facts for the productive B (the rule head), we must first compute the facts for all the predicates A in the rule body. If the dependency graph has no cycles, we call the rule set nonrecursive. If there is at least one cycle, the rule set is called recursive. In Figure 24-17, there is one recursively defined predicate---namely, superior -- which has a recursive edge pointing back to itself. In addition, because the predicate subordinate depends on supenon at also requires recursion as computing its result

A query that includes only numerousive predicates is called a **numedursive query**. In dus section, we discuss only inference mechanisms for nonrecursor, queries. In Figure 24.17, any query that does not involve the predicates subordinate or superior is nonrecursive. In the predicate dependency graph, the nodes corresponding to fact defined



FIGURE 24,17 Predicate dependency graph for Figures 24.14 and 24.15.

predicates do not have any incoming edges, since all fact-defined predicates have that facts stored in a database relation. The contents of a fact-defined predicate can be computed by directly removing the taples in the corresponding database relations.

The main function of an inference mechanism is to compute the facts that correspond to query producates. This can be accomplished by generating a relational expression meniying relational operators as SELECT, FROJECT, JOIN, UNION, and SET DIFFERENCE (with appropriate provision for dearing with safety oscies) that, when executed, provides the query result. The query car, then be executed by utilizing the internal query processing and optimization operations of a relational database multipgement system. Whenever the interence mechanism needs to compute the fact set corresponding to a nonrecursive rule sletmed predicate p, it test locares all the rules that have p is then head. The idea is in comparin the four set for each such rule and then to apply the UNION operation to facresults, since Oxtox corresponds to a logical CB operation. The dependence graph indicates all predicates gion which each pidepends, and since we assume that the predicate is nongeourswe, we can always determine a partial order among such predicates g. Bebre computing the fact set for p, we first compute the fact sets for all predmates gran which p depends, based on their partial order. For example, if a query involves the predicanunder 40K supervisor, we must first compute both vipervisor and over 40K ero. Since the latter two depend only on the fact-defined predicares employee, salary, and supervise, they can be computed directly from the stored database relations.

This concludes our introduction to deductive databases. Additional material non-asfound at the book Web site, where the complete Chapter 15 from the third edition is available. This includes a discussion on aborithms for recursive query processing.

# 24.5 SUMMARY

In this chapter, we miniduced database concepts for some of the common features that are needed by advanced applications, acrive databases, remporal databases, and spatial and multimedia databases. It is important to note that each of these topics is very broad and warrants a complete resubook.

We first norodisced the topic of active databases, which provide additional functionality for specifying active inles. We introduced the event-condition-action of ICA model for active databases. The rules can be automatically triggered by events that occur--such us a database update—and they can initiate certain actions that have been specified in the rule declaration of certain conditions are true. Many commercial packages aheady have some of the functionality provided by active databases in the form of triggers. We discussed the different options for specifying rules, such as row-level versus statement level, before versus after, and imprediate versus deterred. We gave examples of row-level triggers in the Oracle commercial system, and statement-level rules in the sTARIORST experimental system. The syntax for inggers in the SQL 59 standard was also discussed. We briefly discussed some design issues and some possible applications for verice databases.

We then introduced some of the concepts of temporal databases, which permit the database system to store a history of changes and allow users to query both current and past aptes of the database. We discussed how time is represented and distinguished between the odd time and transaction time dimensions. We then discussed how valid time, transaction time, and bitempotal relations can be implemented using tuple versioning in the relational model, with examples to illustrate how updates, inserts, and deletes are implemented. We also showed how complex inductive can be used to implement temporal databases using attribute versioning. We then linked at some of the query fine operations for temporal relational databases and gave a very biof introduction to the TSQL2 language.

We then turned to spatial and multimedia databases. Spatial databases provide concepts for databases that keep track of objects that have spatial characteristics, and they require models for representing these spatial characteristics and operators for comparing and manipulating them. Multimedia databases provide features that allow asces to store and query different types of multimedia information, which multides images both as pictures or drawingst, video clips (such as movies, news reds, or home indexs), orductlips (such as songs, phone ane-sages, or speeches), and documents (such as broke or crucles). We give a very brief overview of the various types of media sources and how multimedia sources may be indexed.

We concluded the chapter with an introduction to deductive databases and Datalog-

#### Review Questions

- 24.1. What are the differences between row-level and statement-level active rules?
- 74.2. When the slittlepences among immediate, deterred, and detached consideration of series rule conditions?
- 24.5. What are the differences among immediate, deterred, and deteched execution of active rule actions?

- 24.4. Briefly discuss the consistency and remnination problems when designing a set or active rules.
- 24.5. Discussisome applications of active databases.
- 24.6. Discuss how time is represented in temporal databases and compare the different time dimensions.
- 24.7. What are the differences between valid rink, transferiori time, and bitemporal relations?
- 24.8. Describe how the insert, delete, and update commands should be implemented in a valid time relation.
- 24.9 Describe now the unsert, delete, and update commands should be nuplemented on a bitemporal relation.
- 24.10 Describe how the insert, delete, and update commonsl- should be implemented on a transaction once relation.
- 24.11 What are the main differences between tuple versioning and attribute versioning?
- 24.22. How do spatial databases differ from regular databases?
- 24.13. What are the different types of multimedia sources?
- 24.14. How are unifimedia sources indexed for content-based retrieval?

#### Exercises

- 24.15 Consider the covers database described in Figure 5.6. Using the syntax of Oracle suggers, write acrive rules to do the following:
  - a. Whenever an employee's project assignments are changed, check if the total bours per week spent on the employee's projects are less than 30 or greater than 40; if so, notify the employee's dizen supervisor.
  - b. Whenever an event-set is deferred, deferre the evolution ruples and observation ruples related to that employees, and if the employee is managing a department of supervising any employees, sort the eventse for that department to null and set the supervises for those employees to null.
- 24.16. Repear 24.05 but use the syntax of ST WWWSF across rules.
- 24.17. Consider the relational schema shown itt Figure 24.18. Write active rules for Reeping the surgeowissions attribute of souts (result qual to the sum of the covission attribute in sous for each sales persona Your rules should also check if the

SALES

<u>S ID</u> V <u>ID</u> COMMISSION

SALES\_PERSON

SALESPERSON\_IN NAME TITLE PHONE SUM\_COMMISSIONS

FIGURE 24.18 Database schema for sales and salespersume ommissions in Evenise 24.17.

sourcessurves exceeds 100000; if it clock, call a proceedure source.events(s\_st) Wrate Forth statement-level rules on STAGPURST notation and row-level rules in Oracle.

- 24.18 Consider the astysistiv this schema of Figure 4.10. Write some rules (in English) that could be implemented via active rules to enforce some common integrity constraints that you think are relevant to this application.
- 24.49. Discuss which at the opdates that created each of the tuples shown in lingure 24.9 were applied removarized and which were applied preventiely.
- 24.20. Show how the following updates: if applied in sequence, would change the contents of the bitemporal as an relation in Figure 24.9. For each update, state whether it is a symptotype or proactive update.
  - .u. On 2004-03-10, 17:30:20, the value of Naessa is updated to 40000, effective on 2004-03-01.
  - F. On 2003-07-30,08:31:00, the salary of 59(1) was corrected to show that it should have been entered as \$1060 (instead of 30000 as shown), effective on 2003-06-01.
  - c. Un 2004-03-18.08:31:00, the database was changed to indicate that Measure was leaving the company (i.e., logically defered) effective 2004-03-31.
  - d. On 2004-04-20, 14:07, 23, the database was changed to indicate the hiring of a new employee called 199988, with the topic <"3946999". "354455657", 1. Joint officiarie on 2004-04-20.
  - Ciri 2004-04-28, 12:54: 02, the database was changed to incident that Woo was leaving the company (i.e., logically deleted) effective 2004-06-01.
  - On 2004-05-05, 13:07:33, the database was changed to indicate the relating of Books, with the same department and supervisor but with salary 35000 effective on 2004-05-01.
- 24.21. Show how the updates given in Exercise 24.20. if applied in sequence, would change the contents of the valid time EMP\_VF relation in Figure 24.8.
- 24.22 Add the following facts to the example database in Figure 24.3:

Supervise (ahmad, bob), supervise (franklin,gmen).

First modify the supervisory tree in Figure 24.15 to reflect aluschange. Then mulify the diagram in Figure 74.4 showing the top down evaluation of the query supervisor(games, Y).

24.25 Consider the following set of foces for the relation parent (X, Y), where Y is the parent of X:

parent(a,aa), parent(a.ab), parent(aa,aaa), parent(aa,aab), parent(aaa,aaaa), parent(aaa,aaab).

Consider the rules

rl: ancestor(X,Y) :- parenr(X,Y)
r2: ancestor(X,Y) : parent(X,Z), ancestor(Z,Y)

which define uncestor Y of X as above.

as Show how to solve the Datalog query

anlestor(aalX)?

using the naive strategy. Show your work meach step,

 b) Show the same query by consputing only the changes in the oncestor relation and using that in rule 2 each time

[This question is derived from Bancilhon and Bamakrishnan (1986)].

24-24. Consider a dedoctive database with the following tiles.

```
ancestor(X,Y) := father(X,Y)
ancestor(X,Y) := father(X,Z), ancestor(Z,Y)
```

Notice that "FatNer(X,Y)" means that Y is the father of X; "ancestor(X,Y)" means that Y is the ancestor of X. Consider the but line

```
father(Marry,Issac), tather(Issac.John), father(John.Kurt)
```

- Construct a model theoretic interpretation of the above roles using the green facts.
- b. Consider that a database contonis the above relations father(X,Y), enother relation brother(X,Y), and a third relation birth(X, R), where R is the embeddet of person X. Stare a rule that computes the first constra of the following context, their forbers must be brothers:
- c. Show a complete Datalog program with fact-based and rule-based literals that compares the following relation: list of parts of courses, where the first person is born after 1960 and the second after 1970. You may use "greater that has built-in predicates 1N-sec Sample facts for brocher, birth, and person most also be shown 3.
- 24-25. Consider the following miles

```
reacnable(X,Y) := flight(X,Y)
reacnable(X,Y) := flight(X,Z), reachable(Z.Y)
```

where reachable(X, Y) means that city Y can be reached from city X, and flight(X, Y) means that there is a tlight to city Y from city X.

- a Construct fact predicates that describe the following:
  - Los Angeles, New York, Chicago, Arlanta, Frankfurt, Paris, Sugapore Sydney are erries.
  - ii. The following flights exist: LA to NY, NY to Atlanta, Atlanta to Frankfort Frankfort to Atlanta. Frankfort to Singapore, and Singapore to Sylnes (Net: No flight in reverse direction can be sutemarizedly assumed.).
- b. Is the given data cyclic? If so, in what sense?
- c. Construct a model theorem interpretation (that is, an interpretation similar to the one shown in Figure 25.3) of the above focts and piles.
- d. Consider the query.

```
reachable(Atlanta,Sydney)7
```

How will this query be executed using nurse and seminorys evoluation? Usr the series of steps in will go through Consider the following rule defined predicates

round-trip-reachable(X,Y) := reachable(X,Y), reachable(Y,X) duration(X,Y,Z)

Draw o predicate dependency graph for the above predicates. (Note: defactor(X, Y, Z) means that you can take a flight from X to Y in Z hours.)

- 6 Consider the following query: What cities are reachable in 12 hours from Atlanta? Show how to express it in Datalog. Assume built-in predicates like greater-than(X,Y). Can this be converted into a relational algebra strucment in a straightforward way? Why or why not?
- g. Consider the predicate paperlattion(X,Y) where Y is the population of city X. Consider the following query: List all possible bindings of the predicate park (X,Y), where Y is a city that can be reached in two fights treat city X, which has over 1 million people. Show this query in Datalog. Draw a corresponding query tree or relational algebraic terms.

#### Selected Bibliography

The book by Zamolo et al. (1997) consists of several parts, each describing an advanced database concept such as active, temporal, and spatial/text/puthimodia databases. Widom and Cen (1996) and Cen and Fratemali (1997) focus on active database concepts and systems. Snedgrass et al. (1995) describe the TS012 language and data model. Khewhatian and Baker (1996), Faloutses (1996), and Subrahimmian (1998) describe multimedia database concepts. Tansel et al. (1992) is a cullection of chapters on temporal databases.

STABBORST rules are described in Wolum and Finkelstein (1990). Early work on active databases includes the HilfAC project, discussed in Chaktawarthy et al. (1989) and Chaktawarthy (1990). A glossary for remporal databases is given in Jensen et al. (1994). Snodgrass (1987) focuses in TQuel, an early temporal speet language.

Temporal normalization is defined in Nacruhe and Admed (1989), Paron (1999) and Paten and Diat (1999) sorvey acrive databases. Chakravarrhy et al. (1994) describe SENTINEL and object-based acrive systems. Lee et al. (1998) discuss time series inanagement.

The cally developments of the logic and database approach are surveyed by Gallare et al. (1984). Reiter (1984) provides a reconstruction of relational database theory, while hereagine (1984) provides a discussion of incomplete knowledge in light of logic. Gallare and Minker (1976) provide an early book on this ropic. A detailed relation of logic and databases appears in Ullinan (1989, vol. 2), and there is a related chapter in Volume 1 (1988). Cerit Gorduly, and Tanca (1992) present a comprehensive yet concise treatment of logic and databases. Das (1982) is a comprehensive back on deductive databases and light programming. The early history of Patalog is reversed in Mater and Warren (1986). Clarksin and Mellish (1994) is an excellent reference on Prolog brogona.

Abo and Ullman (1979) provide an early algorithm for dealing with recursive queries, using the least faced paint operator. Bace illion and Ramakrishnan (1986) give an excellent and defailed description of the approaches to recursive query processing, with detailed examples of the naive and seminance approaches. Excellent survey articles on

deductive databases and meansive query processing include Warren (1992) and Bamalerishnan and Ullman (1993). A complete description of the seminarve approach based on relational algebra is given in Bancillion (1985). Other approaches to recursive query processing include the recursive query/subquery strategy of Vicille (1986), which is a rop-down interprinted strategy, and the Henschen Naqvi (1984) top-down complet iterative strategy. Balbin and Rob (1985), documents are extension of the approach differential approach for multiple predicates.

The original paper on matic sets is by Borrollion et al. (1986). Beeri and Ramakrishnan (1987) estend in Munick et al. (1996) show the applicability of matic sets to nonrecursive nested SQL queries. Other approaches to optimiting rules orthout reveating them appear its Vieille (1986, 1987). Kilor and Lourisen (1976) propose a different rechnique. Bry (1992) discusses how the topolosin and betton-up approaches can be reconciled. Whang and Naviathe (1992) describe an extended disjunctive normal form bechnique to deal with recursion in relational algebra expressions for providing an experimation unerface over a relational DBMS.

Chang (1981) describes an early system for combining deductive rules with relational databases. The 101 system prototype is described in Chimenti or al. (1992) Krishelamurthy and Naqvi (1989) introduce the "choice" notion in 101 Zamula (1983) discusses the language assists for the 110 system. A language overview of CORAL is provided in Ramakrishnan et al. (1992), and the implementation is described in Ramakrishnan et al. (1993). An extension to support object-oriented features, called CORAL++, is described in Seriestavi et al. (1994). Ullintic (1985) provides the base for the NALL'system, which is described in Merris et al. (1987). Phapps et al. (1991) cescile the GUO-NALE deductive database system.

Zamialo (1990) reviews the theoretical background and the practical importance of disfugrice databases. Nicolas (1997) gives an excellent fustury of the developments buding up to DOUDS. Falcine et al. (1997) survey the DOUD landscape. References on the VACIDATS system include Friesen et al. (1995). Vieillo (1997), and Thetrich et al. (1999).



# Distributed Databases and Client–Server Architectures

In this chapter we turn our agonging to deephured databases (DDW), distributed database management systems 1.3 ( 0.085), and how the chemiserver architecture is used as a platform for database apple around evelopment. The DD8 (righthology emerged as a merger of two technologies: (1) database technology, and (2) network and data communication technology. The latter has made tremendous studes in terms of which and wireless rechnologies—from satellate and cellular communications and Metropolitan Area Networks (MANA) to the standard zation of protocols like Etherner, 2020, and the Asynchromous Transfer Mode (ATV) as well as the explosion of the Internet. Whose carly databases moved reward controlization and resulted in nionolithic graphic databases in the seventies and early solution the trend reversed roward move decentralization and autonomy of processing in the late eightics. With advances multistributed processing and distributed computing that occurred in the operating systems arena, the datasase research community did considerable work to address the tissues of data distribution, dismbuted query and transaction processing, distributed combase metadata management. and other jopies, and developed many research projectives. However, a full-scale comprehensive DDBMS that implements the furn tionality and techniques proposed in DDP research never entered as a conninercially visible product. Most major vonders redireered their efforts from developing a "pare" 713906 product into developing systems based on elient-server, or toward developing technologies for accessing distributed heterogeneous data sources.

Organizations, however, nave been very interested in the decentraligation of processing (in the system level) while achieving an integration of the information resources (at the logical level) within their geographically distributed systems of databases, applications, and users. Coupled with the advances in communications, there is now a general endorsement of the chemisserver approach to application development which assumes many of the 1309 issues.

In this chapter we discuss both distributed databases and chent-server architectures,<sup>1</sup> in the development of database rechnology that is closely ried to advances in communications and network technology. Details of the latter are earside our scope; the reader is referred to a zeries of tests on data communications, and networking (see the Secored Biblioteraphy of the end of this situater).

Section 25.1 introduces distributed database management and related concepts Detailed issues of distributed database deservations ing fragmenting of data and distributing it over multiple sites with possible replication, are discussed in Section 25.2. Section 25.3 introduces different types of distributed database systems, including federated and multidatabase systems and highlights the problems of heterogeneity and the needs of autonomy in federated database systems, which will dominate for years to come. Sections 25.4 and 25.5 introduce distributed database query and transaction processing techniques respectively. Section 25.6 discusses how the chemi-server architectural concepts are related to distributed databases, section 25.7 elaborates on future listness in chemi-server architectures. Section 25.8 discusses thermbored database features of the Oracle R0005.

For a short introduction (1) the topic, only sections 25.1, 25.3, and 25.6 may be covered.

## 25.1 DISTRIBUTED DATABASE CONCEPTS

Distributed databases bring the advantages of distributed computing to the database management domain. A distributed computing system consists of a number of processing dements, nor necessarily homogeneous, that are interconnected by a computer network and that cooperate in performing cortain assigned tasks. As a general goal, distributed compuing systems partition a big, immanageable problem into smaller pieces and solve it effciently in a coordinated matter. The economic clability of this approach stems from noreasons (1) more compared power is homosyscholar solve a complex task, and (2) each autoromous processing element can be in mageal independently and develop its on applicators.

We can define a distributed database (DDB) as a collection of multiple logically interrelated databases do-tributed over a computer network, and a distributed database management system (DDBMS) as a software system that manages a distributed database while making the distribution transporeur to the user? A collection of files stated at different nodes of a network and the manufauture of interrelationships among them via hyperlinks has become a common organization on the Internet, with files of Web pages.

<sup>1.</sup> The reader should review the introduction to chemiscryce arenetics rate in Section 2.5.

This domain or optimate of the discussion in this section are bread of Qisor and Valdaris: (1999).

The common functions of database management, including uniform query processing and transaction processing. *As not* apply to this scenario yet. The technology is however, moving in a direction such that distributed World Wide Web (WWW) databases will become a reality in the near future. We shall discuss issues of necessing databases on the Web in Chapter 20. None of those quadras as DOR by the definition given earlier.

# 25.1.1 Parallel Versus Distributed Technology

Turning our attention to parallel system architectures, there are two main types of multiprocessor system architectures that are commonplace:

- Shared memory (aghtly coupled) architecture. Multiple processors share secondary (disk) storage and also share primary memory.
- Shared disk (loosely coupled) architecture: Multiple processors share secondary (disk) stronge but each has then own primary memory.

These inclinectures enable processors to communicate without the overhead of exchanging messages over a network.<sup>5</sup> Database management systems developed using the above types of architectures are termed parallel database management systems rather than DDBoS, since they utilize parallel processor technology. Another type of multiprocessor architecture is called shared nothing architecture. In this architecture, every processor has its own primary and sciondary tdisk) memory, no common memory exists, and the processors communicate over a high-speed interconnetion memory (has or switch). Although the shared nothing architecture resembles a distributed database computing economicate over a high-speed interconnetion memory (has or switch). Although the shared nothing architecture resembles a distributed database computing economics systems, there is symmetry and homogeneity of nodes: this is not true of the distributed database environment where heretogeneity of hardware and operating system at each node is very common. Shared database inductive is also considered is in environment for parallel databases. Figure 25.1 contrasts these different architectures.

# 25.1.2 Advantages of Distributed Databases

Distributed database management has been proposed for various reasons tanging from organizational decentralization and economical processing to greater autonomy. We highlight some of these advantages here.

 Management of distributed data with different lettek of transporency. Really a DPMS should be distribution transparent in the sense of heding the details of where each file (table, relation) is physically stored within the system. Consider the company database in Figure 5.5 that we have been discussing throughout the

 It both promoty and secondary memories are shared, the architecture is also known as shared everything architecture.



**FIGURE 25.1** Some different database system are blex tures. (a) Shared nothing authorecture, (b) A networked are hitecture with a contralized database at one of the sites. (c) A truly distributed database an bitecture.

basik. The encoder, nearest, and wars low rables may be tragmented horizontally (that is, into sets of rows, as we shall discuss in Section 25.2) and stored with possible replication as drawn in Urgare 25.2. The tallowing types of transparencies are possible:

- Distribution or network encorporately: This seters to freedom for the user from the operational details of the network. It may be divided into location transparency and noming transparency. Location transparency receives to the fact that the command used to perform a task is independent of the location of data and the location of the system where the command was issued. Naming transparency unplies that once a name is specified, the named objects can be accessed unambiguously without additional specification.
- Repletation more parency. As we show to Deare 25.2, copies of data may be stored at multiple sites for better availability, performance, and reliability. Replication transparency makes the user unaware of the existence of copies.
- Fragmentation transferency: Two repeated fragmentation are possible. Herizontal tragmentation distributes a relation into sets of tuples (nows). Vertical fragmentation distributes a relation into subrelations where each subrelation is defined by a subset of the columns of the original relation. A global query by the user point he transformed into several fragment species. Fragmentation transparency gaskes the user unaware of the constance of fragments.



FIGURE 25.2 Data distribution and replication among distributed databases.

- 2. Increased reliability and acadability: These are two of the most common potential advantages eited for distributed databases. Reliability is broadly defined as the probability that a system is tanning (nor down) at a certain time polint, cherous availability is the probability that the system is continuously available during a time interval. When the data and 1990S software are distributed over several size, one size may full while other sizes continue to operate. Only the data and software that exist at the failed size control be accessed. This improves both reliability and availability. Further improvement is achieved by judiciously replicating data and software in more than one site. In a centralized system, fullute at a single site makes the work system unavailable to all users. In a distributed database, some of the data may be unreachable, but every nay still be able to access other parcof, the database.
- 3. Improved performance: A distributed DBMS fragments the detabase by keeping the data closer to where it is needed most. Data localization reduces the contentation of (f) and (f) services and simultaneously technes access delays arealed in order area networks. When a large database is distributed over multiple stessing data at a single site have better performance because of the smaller heal database. In addition, each site has a smaller number of transactions executing rhurs of all transactions are admitted in a single centralized database. Moreover, interquery and intraquery parallelism can be achieved by executing multiple queries and intraquery parallelism can be achieved by executing multiple queries are different orea, or by breaking up a query area a multiple of successing in prodlet. This contributes to improved performance.
- Easier extension. In a distributed environment, expansion of the system in terms of adding more data, increasing database sizes, or adding more processors is useb cased.

The transparencies we discussed in (1) above lead to a compromise between case of use and the overhead cost of providing transparency. Total transparency provides the global user with a view of the entire DDPs as if it is a single contralued system. Transparency as provided as a complement to autonomy, which gives the users righter control over their own local databases. Transparency features may be implemented as a part of the user language, which inw translate the required services into appropriate operations. In addition, tempparency impacts the features that must be provided by the operating system and the DBMS.

### 25.1.3 Additional Functions of Distributed Databases

Distribution leads to increased complexity in the system design and implementation. To achieve the parential advantages listed previously, the 1906/85 suftware must be acle to provide the tofflowing functions in addition to those of a centralized 06685.

 Keeping track of data: The ability to keep track of the data distribution, tagaretation, and replacition for expanding the DDBMS cotalog.

- Oscobiated gaters processing: The ability to access remote sites and transmit queries and data anyong the various sites via a communication network.
- Distributed remaction menagement: The ability to devise execution strategies for queries and transactions that occess data from more than one site and to synchronize the access to distributed data and monitorn integrity of the overall database.
- Replicated data involtigement: The ability to decide which copy of a replicated data item to access and to maintain, the consistency of copies of a replicated data item
- Distributed database recovery: The ability to recover from industrial site crushes and from new types of fadores such as the fadore of a cognition context.
- Security: Distributed transactions must be executed outly the proper management of the security of the data and the authorization/access privileges of users.
- Distributed directory (randog) management. A directory contains information (menadata) about data in the database. The directory may be global for the entire DDB, or local for each site. The placement and distribution of the directory are design and policy issues.

These functions themselves increase the complexity of a DDBAS over a centralized DBMS. Before we can realize the full potential salvantages of distribution, we must find satisfactory solutions to these design issues and problems. Including all this additional functionality is hard to accomplish, and including optimal solutions is a step beyond that.

At the physical hardware level, the following main factors distinguish a 000048 from a centralized system:

- There are multiple comparers, called sites or nodes.
- These sites must be connected by some type of communication network to transmit data and commands among sites, as shown in logare 25-1c.

The sites may all be located in physical proximity—so, within the same building or group of adaptent buildings—and connected via a local area network, or they may be generaphically distributed over large distances and connected via a long-haul or wide area network. Local mea networks typically use calibles, whereas long-haul networks use relephone lines or satellites. It is also possible to use a combination of the two types of networks.

Nerworks may have different topologies that define the direct communication paths among sites. The type and topology of the network ased may have a significant effect on performance and hence on the strategies for distributed query processing and distributed database design, has high-level architectural issues, however, it does not matter which type of network is used, it only thatters that each site is able to communicate, directly or indirectly, with every other site. For the remainder of this chapter, we assume that some type of communication network exists among sites, regardless of the particular topology. We will not address any network specific issues, although it is important to understand that for an efficient operation of a DDPS network design and performance assues are very entreal.

# 25.2 DATA FRAGMENTATION, REPLICATION, AND ALLOCATION TECHNIQUES FOR DISTRIBUTED DATABASE DESIGN

In this section we discuss techniques that are used to break up the database into logical units, called **fragments**, which may be assigned for storage at the various sites. We also discuss the use of **data replication**, which periods certain data to be stored in more than und one, and the process of **allocating** fragments. For replicas of fragments—for storage at the various sites. These techniques are used during the process of **distributed database design**. The information concerning data fragmentation, allocation, and replication is vored in a **global directory** that is accessed by the DDPs applications as needed.

#### 25.2.1 Data Fragmentation

In a DDB, decisions must be made regarding which size should be used to store which pertions of the database. For new, we will assume that there is no regiverion; shorts, each relation—or portion of a relation—is to be stored at only one site. We discuss replication and its effects later in this section. We also use the terminology of relational database similar concepts apply to other data models. We assume that we are starting with a selational database schema and must decide on how to distribute the relations over the varous sites. To illustrate our discussion, we use the relational database schema in Figure 5.5.

Peters we decide on how to distribute the data, we must determine the logical anis of the database that are to be distributed. The simplest logical units are the relations themselves; that is, each schole relation is to be stored at a particular site. In our example, we must decide unit site in store each of the relations everyses, bever, severage, and consistent of Figure 5.5. In many cases, however, a relation can be divided into sneller logical units for distribution. For example, consider the company database shows in Figure 5.6, and assume there are three computer sites if one for each department in the company.<sup>1</sup> We may want to store the database information relating to each department as the company.<sup>1</sup> We may want to store the database information relating to each department as the company.<sup>1</sup> We may want to store the database information relating to each department as the company.<sup>1</sup> We may want to store the database information relating to each department as the company.<sup>1</sup> We may want to store the database information relating to each department as the company.<sup>1</sup> We may want to store the database information relating to each department as the company.<sup>1</sup> We may want to store the database information relating to each department as the company.<sup>1</sup> We may want to store the database information relating to each department as the company.<sup>1</sup> We may want to store the database information relating to each department as the company.<sup>1</sup> We may want to store the database information relating to each department as the company.<sup>1</sup> We may want to store the database information relating to each department as the company.<sup>1</sup> We may want to store the database information relating to each department as the company.<sup>1</sup> We may want to store the database information was the database information was the database information as the company.<sup>1</sup> He may want to store the database information was the database

**Horizontal Lingmentation**. A horizontal fragment of a relation is a subset of the implex in that relation. The implex that belong to the horizontal fragment are specified by a condition, on one of more applied that belong to the horizontal fragment are specified by a condition, on one of more applied of the relation. Often, only a single autobase involved. For example, we may define three horizontal fragments on the relation of the relation of the sectore relation of Figure 5.6 with the following conditions: favore 5). (*more* 4), and (*more* 1)—each fragment combines the encoder department. Sumfath, we may define three horizontal fragments for a particular department. Sumfath, we may define three horizontal fragments for the only of relation, with the conditions (*more* 5), four = 4).

<sup>4.</sup> Of estimation an actual oriention, there will be many more ruples in the relations than these shown in agents pip.

and ( $\infty < 1$ )—each fragment contains the PROFET tuples controlled by a particular department Horizontal fragmentation divides a relation "horizontally" wegoinging rows to create subsets of tuples, where each subset has a certain liquid meaning. These fragments can then be assigned to different sites in the distributed system. **Derived horizontal** fragmentation applies the partitioning of a primary relation (DEPAROE) in our example) to other secondary relations (DEPAROE and PROFET in our example) models are plated to the primary way a foreign key. This way related data between the primary and the secondary relations gets fragmented in the same way.

Vertical Fragmentation. Each site more not need all the armbures of a relation, which would indicate the need for a different type of fragmentation. Vertical fragmentation divides a relation vertically" by columns A vertical fragment of a relation keeps only certain attributes of the relation. For example, we may want to fragment the tention more relation into two vertical fragments. The first fragment includes personal information  $\rightarrow$  were solved relation attributes, and see  $\rightarrow$  and the second includes work related and mattributes for the vertical fragments. The first fragment includes work related and mattributes and see  $\rightarrow$  and the second includes work related and matter  $\rightarrow$  were solved work of the vertical fragments in a solve proper because, if the two fragments in stored separately, we cannot put the original employee topics back trigether, since there is no constant antibiate between the two fragments. It is necessary to include the primary key or since candidate key attribute in every vertical fragment so due to the tull relation can be reconstructed from the fragments. Hence, we prove add the solve attribute to the personal information fragments.

Notice that each horizontal tragment on a relation R can be specified by a  $\sigma_{G}(R)$  operation in the relational algebra. A set of horizontal fragments whose conditions CL C2, ..., Cn include all the tuples in R—that is, every tuple in R satisfies (C1 OR C2 OR ... OR Cn)—is claded a complete horizontal fragmentation of R. In many cases a complete horizontal tragmentation of R. In many cases a complete horizontal fragmentation of R. In many cases a complete horizontal tragmentation is the property of each examples of horizontal fragmentation for the property and worker relations were testic complete and disjoint. To reconstruct the relation R from a complete horizontal tragmentation, we need to apply the UNION operation to the tragments.

A vertical fragment on a relation R can be specified by a  $\pi_0^-$  (R) operation in the relational algebra. A set of vertical fragments whose projection here L1, L2, ..., Ln include all the otherbases on R but share only the primary key attribute of R is called a **complete vertical fragmentation** of R. In this case the projection lists satisfy the following two conditions:

- L10120 ULn = ATTRS(R).
- L(P) L) = PK(R) for any point j, where ACTRs(R) is the set of artributes of R and PK(R) is the pumary keysof R.

To reconstruct the relation R from a complete vertical fragmentation, we apply the OUTER USION operation to the vertical fragments (assuming no horizontal fragmentation) is used). Notice that we could also apply a FCLUCALTER JOIN operation and get the same result for a complete certical fragmentation, even when some horizontal fragmentation may also have been applied. The two vertical fragments of the SPEOREF relation with projection lists 1.1 = [556, 567, 6085, 562] and 1.2 = [556, 56368, 56785, 660] constitute a complete vertical fragmentation of FR2085. Two homportal fragments that are wether complete nur disjoint are thuse defined on the sourcest relation of Figure 5.5 by the conditions (salasy  $\geq$  50000) and (oss = 4); they may not include all sourcest toples, and they may not load common toples. Two vertical fragments that are not complete are those defined by the attribute lists 1.1 = {sant. success} and 1.2 = {sw, second solars} these lists walkes both conditions of a complete vertical fragmentation.

**Mixed** (Hybrid) Fragmentation. We can act must the two types of trapmentation, yielding a **mixed** fragmentation. For example, we now combine the homeostal and vertical fragmentations of the FFOME relation gives earlier into a mixed fragmentation of the FFOME relation gives earlier into a mixed fragmentation of the FFOME relation gives earlier into a mixed fragmentation of the FFOME relation gives earlier into a mixed fragmentation of the FFOME relation gives earlier into a mixed fragmentation of the FFOME relation gives earlier into a mixed fragmentation of the FFOME relation gives earlier into a mixed fragmentation of the FFOME relation gives earlier into a mixed fragmentation of the FFOME relation gives earlier into a mixed fragment of a relation R can be specified by a SEE CHARGED and L  $\neq$  AT (relation of operations  $\pi_1(r_{\rm C}(R))$ )  $1/C \approx 130$  F (that is, all replex are selected) and L  $\neq$  AT (restR) we get a vertical fragment, and if  $C \neq 180$  F and L  $\approx AT(Rs(R),$  we get a horizonial fragment. Finally if  $C \neq 100$  fragment with C = 100 fragment. Nonce that relation can uself be considered a fragment with C = 100 fragment. Nonce that following discussion, the term fragment is used to refer to a relation or to any of the preceding types of fragments.

A tragmentation schema of a database wild definition of a set of fragments that includes all outplotes and tuples in the database and satisfies the condition that the whole database can be reconstructed from the hagments by applying some sequence of CUTFS UNION (or OUTER CONT and ONON operations. It is also sometimes includ—although not necessary—to have all the tragments be disjoint except for the repetition of primars less among vertical (or mixed) tragments, by the latter case, all replication and distribution of tragments is clearly specified at a subsequent stage, separately from fragmentation

An allocation schema describes the allocation of fragments to sites of the POFs: hence, it is a mapping that specifics for each fragment the site(s) at which it is stored, it a tragment is stored, at more than one site, it is said to be replicated. We discuss data replication and obscarion next.

#### 25.2.2 Data Replication and Allocation

Replication is useful in improving the availability of data. The most extreme case is replication of the olide database at every site in the distributed extern, thus creating a fully replicated distributed database. This can improve availability rentarkably because the system can continue to operate as long as at least one site is up. It also improves performance of retrieval to global queries, because the result of such a query can be obtained locarly from any one site hence, a retrieval query can be processed or the local site where it is submitted if that site includes a server module. The disadvantage of full replication is that it can slow down aplate operations disancally, since a single logical update must be performed on every copy of the database to keep the copies consistent. This is especially tour it many copies of the database exist. Full replication makes the concurrency control and secovery techniques more espensive than they would be it there were no replication, as we shall see in Section 25.5.

The other extreme from full replication involves having no replication—that is, each inspirent is stored at exactly one site, in this case all fragments must for disjoint. except for the repetition of primary keys among certical (or mixed) fragments. This is also called **nonredundant allocation**.

Boween above two extremes, we have a wide spectrum of **partial replication** of the data – that is, some fragments of the database may be replicated whereas others may not. The number of corties of each fragment can targe from one up to the total number of sizes in the distributed system. A special case of partial replication is occurring heavily in applications where mobile workers—such as sides forces, finance it planners, and claims adjustors—carry partially replicated databases with them on laptops and personal digital assistants and synchronize them periodically with the server database.<sup>5</sup> A description of the replication of fragments is sometimes called a **replication schema**.

Each fragment—or each copy of a tragment—must be assigned to a porticular site in the distributed system. This process is called **data distribution** (or **data allocation**). The choice of sites and the degree of replication depend on the performance and availability goals of the system and on the types and frequencies of transactions submitted at each site. For example, it high availability is required and transactions can be submitted at each site. For example, it high availability is required and transactions can be submitted at any ate and if most transactions are remeval only, a fully replicated database is a goal choice. However, if certain transactions that access tarticular parts of the database is a goal choice. However, if certain transactions that access tarticular parts of the database is a located at that site only. Data that is accessed at multiple sites can be replicated at those sites. If many agdates are performed, it may be useful to limit replication. Finding an optimal of events goad solutions to distributed data allocation is a complex optimization problem.

#### 25.2.3 Example of Fragmentation, Allocation, and Replication

We now consider an example of fragmenting and distributing the company database of Figures 5.5 and 5.6. Suppose that the company has three computer sites —one for each current department. Sites 2 and 3 on for departments 5 and 4, respectively. At each of these sites, we expect frequent access to the encourt and more transformation for the employees also work in that department and the projects compiled by that department. Further, we assume that these sites mainly access the wave, saw, salary and supersay attributes of encourse. Site 1 is used by company headquarters and accesses all employee and project information regularly, in addition to keeping track of EFFENDENT information for more purposes.

According to these requirements, the whole database of Figure 5.6 can be stored in site 1. To determine the fragments to be replicated at sites 2 and 3, we can first borcontrally fragment theorem by its key boostic. We then apply derived fragmentation to the relations recently, oracle, and free resetures relations based on their foreign keys for department number – called out, base, and fragments to include only the jarribuses (non-vertically fragment the resulting carbost fragments to include only the jarribuses (non-vertically fragment the resulting carbost fragments to include only the jarribuses (non-vertically fragment the resulting carbost fragments to include only the jarribuses (non-vertically fragment the resulting carbost fragments to include only the jarribuses (non-vertically fragment the resulting carbost  $\frac{1}{2}$  is shown the module only the jarribuses (non-vertically fragment the resulting carbost  $\frac{1}{2}$  is shown the module only the jarribuses (non-vertically fragment the resulting carbost  $\frac{1}{2}$  is shown the module only the jarribuses (non-vertically fragment the resulting carbost  $\frac{1}{2}$  is shown the module only the jarribuses (non-vertically fragment the resulting carbost  $\frac{1}{2}$  is shown the module only the jarribuses (non-vertically fragments the resulting carbost  $\frac{1}{2}$  is shown the module only the jarribuses (non-vertically fragments the resulting carbost  $\frac{1}{2}$  is shown the module only the jarribuses (non-vertically fragments) and even  $\frac{1}{2}$  is shown the module only the large  $\frac{1}{2}$  and  $\frac{1}{2}$ 

Kata stalable approach to synchrymite partially reply area databases, see Mahapatier al. (1968).

#### 814 Chapter 25 Distributed Databases and Chent–Server Architectures

| <b>(</b> ₽}         | EN       | EMP05              |                | WE                 | MINIT    | LNAM          | E, <u>SSN</u>  | SALARY         |           | SUPERSON  |                  | DN        | <u>о</u> , |          |         |  |
|---------------------|----------|--------------------|----------------|--------------------|----------|---------------|----------------|----------------|-----------|-----------|------------------|-----------|------------|----------|---------|--|
|                     |          |                    | John           |                    | в        | Smath         | 123455739      | 30007/         |           | 33445655  |                  | 5         | -          |          |         |  |
|                     |          |                    | Hankler        |                    | I        | Wang          | 333445855      | . 40           | 000       | B         | <b>100</b> 00000 | 5         | i          |          |         |  |
|                     |          |                    | Angreen .      |                    | ĸ        | Nersyst       | 600584444      | <u>*</u>       | <u>60</u> | - J       | 10445000         | † 'S      | 1          |          |         |  |
|                     |          |                    | ,kyca          |                    | A        | English       | 453453460      | <u>.</u> 200   |           | 33943235  |                  | - 5       |            |          |         |  |
|                     |          |                    |                |                    |          |               |                |                |           |           |                  |           | _          |          |         |  |
| OEP5                | DNU      | DNAME              |                | ONUMBER            |          | Şrəşn         | MORSTART       |                |           | DEPS_LOCS |                  |           |            | LOCATION |         |  |
|                     | Research |                    | 5              |                    | 33       | 3445656       | 1968-05-22     | <u> </u>       |           | -         |                  | 5         |            | Ballaire |         |  |
|                     |          |                    |                |                    |          |               |                |                |           |           |                  |           | 5          | <u> </u> | eteri i |  |
|                     |          |                    |                |                    |          |               |                |                |           |           | -                |           | _ 110      | uetor    |         |  |
| WORKS_ONS <u>ES</u> |          | <u> 655</u>        | N PNO          |                    | HOL      | <b>F</b> RI j | PR0.85         | POLAM          |           |           | PNUMBE           | PLOCAT    |            | γc       | CP4.M   |  |
| 12945               |          | 129456             | 7890 1         |                    | +        | 15            |                | Product X      |           |           | L 1 .            | Balanc    |            |          | 5       |  |
| 69989               |          | 1000 Z<br>1411 0 7 |                | 1 .                | <u> </u> |               | Produc         |                | 2         |           | Sugerland        |           | 1          | 5        |         |  |
| 453458              |          | <u>s</u> :         |                | 20                 | 5        |               | Product 2      |                |           | 3         |                  | Num       |            | ં        |         |  |
| 463453              |          | 453                | 2              | 20                 | 0        |               |                |                |           |           |                  |           |            |          |         |  |
| 383445              |          | <u>666</u>         | 2              | ( 10               | ω        |               |                |                |           |           |                  |           |            |          |         |  |
| 329445              |          | 98                 | 8 0 N          |                    | 6        |               |                |                |           |           |                  |           |            |          |         |  |
| 303445              |          |                    | щ ļ            | <u>66   10   N</u> |          | ω             |                |                |           |           |                  |           |            |          |         |  |
| 329445              |          |                    | 596            | 86 20 I 10         |          | ω             | Qate at Site 2 |                |           |           |                  |           |            |          |         |  |
|                     |          |                    |                |                    |          |               |                |                |           |           |                  |           |            |          |         |  |
| <b>~</b> ``         |          |                    |                |                    |          |               |                |                |           |           |                  |           |            |          |         |  |
| (0)                 | EMP(04   |                    | FN <b>AM</b> € |                    | MINH     |               | e <u>ssn</u>   | SALARY S       |           | ġ,        | SUPERSON         |           | Þ          |          |         |  |
|                     |          |                    | Adria          |                    |          | 2             | WEIH 777       | 25000          |           | - Al      | 76500            | 4         |            |          |         |  |
|                     |          |                    | Jamita 🛛       |                    | s        | Water         | 967854521      | <0000          |           | 639665555 |                  | 4         |            |          |         |  |
|                     |          |                    | Armod          |                    | Υ.       | Jachar        | 997967987      | 25000          |           | 56765432: |                  | ٩         |            |          |         |  |
|                     |          |                    |                |                    |          |               |                |                | _         |           |                  |           |            |          |         |  |
| CEPH                | DNAME    |                    | DWMBE          |                    | R, M     | GR\$5N        | манатантр      | ATE            | C         | DEP4_LOCS |                  | DNI.ANEER |            | ιœ       | ATTON   |  |
|                     |          |                    |                |                    | 98766432 |               | 1995-01-01     |                |           | +         |                  | 4 34      |            | Stat     | Rond    |  |
|                     |          |                    |                |                    |          |               |                |                |           |           |                  |           |            |          |         |  |
| WORKS_ON4           |          | <u>685</u>         |                |                    | HOU      | RS            | PROJ54         | E PT           |           | FINAME    |                  |           |            |          | DNUM    |  |
|                     |          | 333445565          |                | 10                 | +        | <u> </u>      |                | Computerzation |           | 00        | 10               |           | Stattord   |          | 4       |  |
|                     |          | 96969777           |                | 30                 | 10       | 2             |                | Newbornetty    |           | _         | 30               |           | Stationd   |          | •       |  |
|                     |          | 997987987          |                | 10                 | , 38     | č l           |                | <u> </u>       |           |           |                  | -         |            |          |         |  |
|                     |          | 967967             | 987            | 30                 | 1 3      | ē.            |                |                |           |           |                  |           |            |          |         |  |
|                     |          | 987654             | 92H            | 31                 | 20       | c             |                |                |           |           |                  |           |            |          |         |  |
|                     |          | 067654321 20       |                | 20                 | 15 0     |               | Data at Sile 3 |                |           |           |                  |           |            |          |         |  |

**FIGURE 25.3** Allocation of fragments to sites, (a) Relation fragments at site 2 corresponding to department 5. (b) Relation fragments at site 3 corresponding to department 4.

respectively. The horizontal tragments of sector, presentent, and energy-matrices are similarly frequented by department number. All these tragments – stored at sites 2 and 3 – ore replicated because they are also stored at the headquarters site 1.

We must now fragment the weeks\_or relation and decide which fragments of week or to store of sites 2 and 3. We are continued with the problem that no articlate of weeks  $\infty$ 

directly indicates the department to which each tuple belongs. In fact, each tuple newses\_ ov relates an employee ento a project p. We could fragment solvages based on the department d in which envolves or based on the department d' that controls p. Fragmentation becomes easy if we have a constraint stating that d = d' for all variables tuples—that is, if employees can work only on projects controlled by the department they work to: However, there is no such constraint in our database of Figure 5.6. For example, the statistic uple <335445555, 10, 10,05 reflects an employee when works for department 5 with a project controlled by department 4. In this case we chald fragment statistic to based on the department in which the employee works (which is expressed by the maximum C) and then fragment forther based on the department that controls the projects that employee is working on, as shown in Figure 25.4.

In Figure 25.4, the union of itaginents GL G2, and G3 gives all works\_two ples for employees who work for department 5. Similarly, the union of fragments G4, G5, and G6 gives all work\_to ruples for employees who work for department 4. On the other hand, the union of fragments G1, G4, and G7 gives all works\_tait toples for projects controlled by department 5. The condition for each of the fragments G1 through G9 is shown in Figure 25.4. The relations that represent M5N relationships, such as works\_or, often have several possible logical fragmentations. In our distribution of Figure 25.5, we choose to include all fragments that can be joined to either an two work fulle or a works\_or, often have several possible logical fragmentations. In our distribution of Figure 25.5, we choose to include all fragments that can be joined to either an two work fulle or a works\_or, often have several possible logical fragmentations. In our distribution of Figure 25.5, we choose to include all fragments that can be joined to either an two work fulle or a works\_or, often have several possible logical fragmentations. In our distribution of Figure 25.5, we choose to include all fragments G4, G5, G6, G2, and G8 at set 3. Notice that fragments O2 and G4 are replicated at both ates. This allocation strategy permits the join between the local explore completely fulling. This clearly demonstrates have complex, the problem of directors fragmentation and allocation is for large databases. The Selected Bibliography at the end of this chapter discusses some of the work done in this area.

# 25.3 TYPES OF DISTRIBUTED DATABASE SYSTEMS

The term distributed database management system can describe vortous systems that differ from one another in many respects. The main thing that all such systems have in commora is the fact that data and software are distributed over andriple sites concerned by some form of commanication network. In this section, we discuss a number of types of DEMISS and the criteria and factors that make some of these systems different.

The first factor we consider is the degree of homogeneity of the DD905 software. If all servers (or individual local P6058) use identical software and all users (cherns) use identical software, the DP605 is called homogeneous) otherwise, it is called hemogeneous. Another factor related to the degree of homogeneous; otherwise, it is called hemogeneous. Another factor related to the degree of homogeneous; otherwise, it is called hemogeneous. If there is no provision for the local rate to function as a stand-alone DF055, then the system has no local **autonomy**. On the other hand, if decet access by local transactions to a server is permitted, the system has going degree of head autonomy.

At one extreme of the anonomy sportrum, we have a DDMMS that "looks like" a centralia d 18915 to the user. A single conceptual schema extent, and all process to the system is obtained through a site that is part of the U2BMS—which means that no local

autonomy exosts. At the other extreme we encounter a type of PDIMS called a federated 30506 (or a midtalatakee system). In such a system, each server is an independent and autonomous centralized 00005 (hat has its own local users, local transactions, and DAA and hence has a very high degree of local matorizity. The term **federated database system (FDBS**) is used when there is some global view of schema of the federation of database system (FDBS) is used when there is some global view of schema of the federation of database system (FDBS) is used by the applications. On the other hand, a **multidatabase system** does not have a global schema and interactively rememory one as precised by the application. Both systems are hybrids between distributed and centralized sparents and the distinction we made between them is not stractly followed. We will refer to them as FDBs in a generic sense.



**FIGURE 25.4** Complete and disjoint fragments of the weaks\_as relation. (a) **Fragments of works\_as for emoky** reas working in department 5 (i=[i:souris (struct issue renovation and issue=5)]). (b) Fragments of works\_ on for employies working in department 4 (c=[i:souris (struct issue on innexet one=5)]). (b) Fragments of works\_ membring of works\_as for employies working in department 1 (to [cose up (struct issue one renovation are needs of works\_as for employies working in department 1 (to [cose up (struct issue one renovation are needs)]).

In a heterogeneous EDBS, one server may be a relational DBMS, another a nerwork (1984), and a third on object of hierarchical DBMS. In such a case in to necessary to have a caronical system language and to include longuage transform to transform subqueries from the canonical longuage to the language of each server. We briefly discuss the issues affecting the design of EDBs below.

Foderated Database Management Systems Issues. The type of heterogeneity prosent in EDRs may area from several warres. We discuss these scatters first and then point our how the different types of outonomics contribute to a semantic bettrogeneity that must be tosolved in a beterogeneous EDRs.

- (b)//exercise in data models. Databases in an organization come from a variety of data models including the so-called legocy models (network and hierarchical, see Appendixes E and F), the relational data model, the object data model, and even files. The modeling copubilities of the models vary. Hence, to deal with them uniformly vara single global schema or to process them in a single language is challenging. Even it two databases are both from the RDBMS environment, the same information may be represented as an attribute name, as a relation name, or as a value in different databases. The calls for an intelligent query-processing mechanism that can relate information eased en intelligent query-processing mechanism.
- Differences in constraints. Constraint facilities for specification and implementation eary from system to system. There are comparable learnes that must be reconciled in the construction of a global schema. For example, the relationships from FR models are represented as referential integrity constraints in the relational model. Triggers may have to be used to implement certain constraints in the relational model. The global schema must also deal with potential conthets among constraints.
- Differences or query longuages: fiven with the same data model, the languages and their versions vary. For example, SQL has multiple versions like SQL-69, SQL-92, and SQL-99, and each system has its own set of data types, comparison operators, string manipulation features, and syon.

Semantic: Helerogeneary — Scoundocheterogeneary occurs when there are differences in the meaning, interpretation, and intended use of the same occretated data. Sciencific heterogeneary among comparison database systems (1388) means the biggost hundle in designing global schemas of hererogeneous databases. The design autonomy of comparison DBS\* refers to their freedom of choosing the following design parameters, which in annafted the eventual complexity of the 50-18.

- The intervie of discourse from infect the data is drawn, for example, two customer accounts, databases in the federation may be from United States and Jaron with entitely different sets of attributes about customer accounts required by the accountang processes d'uniency rate fluctuations would also present a problem. Hence, relations in these two databases which have identical names—customer or scores —may have some common and some entitely distinct information.
- Representation and staming. The representation and maming of data elements and the structure of the data model may be prespectived for each local database.

- The toxic@apiding, meaning, and subjective interpretation of data. This is a chief contralsation to semantic heterogeneity.
- Transaction and pully constrainty. These deal with serializability criterial compensating transactions, and other transaction policies.
- Derivation of monomers: Aggregation, summarization, and other data processing for tures and operations supported by the system.

**Communication autonomy** of a component 1998 refers to its ability to decide whether to communicate with another component 1998. **Execution autonomy** refers to the ability of a component 1998 is essential operations without interference from external operations by other component 1998s and 16-ability to decide the order in which to execute them. The association autonomy of a component 1995 implies that it has the ability to decide whether and how much to share its functionality (operations it support-) and resources (data it managed) with other component 1995s. The major challenge of designing E2088 is to let component 1095s interpretate while still providing the above types of autonomies to them.

A typical incidevel schema architectore to support global applications in the File environment is shown in Figure 25.5. In this architecture, the **local schema** is the conceptual schema (full dualyse definition) of a component database, and the component schema is derived by translating the local schema into a canonical data model or common data model (1051) for the FDBS. Schema translation from the local schema to the component schema is accomponed by constitution of mappings to transform commands on a component schema into commands on the corresponding local schema. The **export schema** represents the subset of a component schema that is available to the FDBS. The **bederated schema** is the global schema or view, which is the result of integrating all the shareable export schema. The **external schemas** define the schema to a user group or an application, as in the three-local schema architectore.<sup>3</sup>

All the problems related to query processing, transaction processing, and dust tare and meradata numagement and recovery apply to -DRS: with additional considerations his not within our scope to discuss them in det of here.

# 25.4 QUERY PROCESSING IN DISTRIBUTED DATABASES

We now give incoverview of how a PCOMs processes and optimizes a query. We first discase the communication costs of processing a distributed query; we then discuss a special operation, called a semigrar, that is used in optimizing some types of chernes in a Obbass.

For a detailed document of the automotions and the free-local architecture, of (0.985), we Shuh and Earstra (1.995).



FIGURE 25.5 The five-level schema architecture in a federated database system TOBSI. Source: Adapted from Sheth and Larson, Federated Database Systems for Managing Distributed Heterogeneous Autonomous Databases. ACM Computing Surveys (Vol. 22, No. 3, September 1990).

#### 25.4.1 Data Transfer Costs of Distributed Query Processing

We discussed the issues involved in processing and optimizing a query in a rematiced .8848 in Chapter 15. In a distributed system, several additional factors further complicate query processing. The first is the cost of transferring data over the network. This data includes intermediate files that are transferred to other sites for further processing, as well as the final result files that may have to be transferred to the site where the query result is needed. Although these costs may not be very lagh if the sites are connected via a high-genomence local area network, they become quite significant in other types of networks. Hence, 208MS query optimization algorithms consider the youl of reducing the arriver of data transfer as an optimization algorithms consider the youl of reducing the arriver of data transfer as an optimization conterior in choosing a distributed query execution strategy.

We allocated that with two simple example queries. Suppose that the END we and REEK E-ET relations of Figure 5.5 are distributed as shown in Figure 25.6. We will assume in this example that neutron relation is fragmented. According to Figure 25.6, the are of the EVENTEE indicates  $100 \times 100000 = 10^{\circ}$  hyres, and the size of the incretification is  $100 \times 100000 = 10^{\circ}$  hyres, and the size of the incretification is  $35 \times 100 = 3500$  hyres. Consider the query Q: "For each employee, retrieve the employee.



FIGURE 25.6 Example to illustrate volume of data transferred.

name and the name of the department for which the employee works." This can be stated as follows in the relational algebra.

```
Q: TERNAUE, INAVE, ON ANE (EMPLOYEE M LINC-DW, MITHE CEPARTMENT)
```

The result of this query will include 10,000 recerds, assuming that every employee is related to a department. Suppose that each record in the query result is 40 bytes long. The query is submitted at a distance site 3, which is called the result site because the query result is needed thery. Neither the respective non-the **DEPARTMENT** relations reside at site 3. There are three simple strategies for executing these distributed query:

- Ironster both the seconse and the GENERMENT relations to the result size, and perform the join at size 3. In this case a total of 1,000,000 (13500 = 1,003,500 bytes must be transferred.
- 2 Transfer the PFLATE relation to site 2, exercise the join at site 2, and send the result to site 3. The site of the query result is 40 × 10,000 = 400,000 bytes so 400,000 + 1,000,000 = 1,400,000 bytes must be transferred.
- Transfer the DEPARTMENT relation to size 1, execute the joint of size 1, and send the result to size 3. In this case 400,000 + 3500 = 403,500 bytes must be remisferred.

If muturizing the amount of data transfer is our optimization cruction, we should choose strategy 3. Now consider another query Q's "For each department, connect the department number " This can be stated as follows in the relational algebras

Q': TRANANE INANELCHANE (DEPARTMENT MAGESCHARGEN EMPLOYEE)

Again, suppose that the query is submitted at site 3. The same three strategies for executing query Q apply to Q', except that the result of Q' includes only 100 records, assuming that each department has a manager.

- Transfer both the integration of the integration relations to the result site, and perform the join at site 3. In this case a total of 1,000,000 + 3500 = 1,003,500 bytes must be transferred.
- Transfer the operating relation to site 2, execute the join at site 2, and send the result to site 3. The size of the query result is 40 × 100 - 4000 bytes, so 4000 + 1.000.000 = 1.004.000 bytes must be transferred.
- 3 Transfer the reporter relation to site 1, execute the join at site 1, and send the result to site 3, in this case 4000 + 3500 = 7500 bytes must be transferred.

Again, we would choose strategy 3—in this case by an overwhelming margin over strategies 1 and 2. The preceding three strategies are the most obvious ones for the case where the result site (site 3) is different from all the sites that contain files involved in the query (sites 1 and 2). However, suppose that the result site is site 2; then we have two simple strategies:

- Transfer the EPROPSE relation to site 2, excente the query, and present the result to the user at site 2. Here, the same number of bytes = 1,000,000, smoot be transferred for both Q and Q<sup>2</sup>.
- Transfer the presentent relation to site 1, execute the query as site 1, and send the result back to site 2. In this case 400,200 + 3500 = 403,500 bytes must be transferred for Q and 4000 + 3500 = 7500 bytes for Q'.

A more complex strategy, which constrimes works lienter than these souple strategies, uses an operation called **semijoin**. We introduce this operation and discuss distributed execution using semijoins next.

### 25.4.2 Distributed Query Processing Using Semijoin

The idea behind distributed query processing using the scoupoit operation is in reduce the number of tuples in a relation before transferring it to another site. Infinitively, the idea is to seed the policy cohord, of one relation R to the site where the other solution S is located; this column is then joined with S. Following that, the jum attributes, along with the attributes required in the result, are projected out and shipped back to the original site and guined with R. Hence, only the juming column et R is transferred in our direction, and a subset of S with no extraneous tuples in attributes is nansteried in the other direction. If only a small fraction of the tuples in S participate in the join, this can be quite an efficient solution to minimizing data transfer.

To illustrate this, consider the following strategy for executing Q or Q'.

Project the join attributes of effectives: at site 2, and transfer them to still. For Q, we transfer F = π<sub>matrix</sub>(π //affective), whose site is 4 \* 100 - 400 bytes, whereas for Q<sup>2</sup>, we transfer F<sup>2</sup> = π<sub>matrix</sub>(creatives), whose site is 9 \* 100 = 900 bytes.

- Join the transferred file with the sectoric relation at site 1, and transfer the required attributes from the resulting file to site 2. For Q<sub>2</sub> we transfer R = m<sub>M<sub>2</sub></sub> and file M<sub>1</sub> and file site is 34 + 12,000 = 540,000 bytes, whereas, for Q<sup>2</sup>, we transfer R<sup>3</sup> = m<sub>M<sub>2</sub></sub> and file M<sub>1</sub> and fil
- Execute the query by joining the transferred file R-or R' with present, and present the result to the user at site 2.

Using this strategy, we transfer 340,000 bytes for Q and 4600 bytes for Q'. We limited the two we attributes and tuples transmitted to site 2 in step 2 to only those that will actually be joined with a prostruct tuple in step 3. For query Q, this turned out to include all contour tuples, so little inprovement was achieved. However, for Q' only 100 out of the 12.000 overout tuples were needed.

The semijour operation was devised to formalize this strategy. A semijoin operation  $R \gg_{1.16} S$ , where A and B are contain-compatible attributes of R and S, respectively, produces the same result as the relational algebra expression  $\pi$  (RM<sub>2.1</sub>S). In a distributed environment where R and S reade at different attes, the semijour is typically implemented by first transferring F =  $\pi_A S$ ) to the one where R reades and then puring F with R, thus leading to the strategy discussed here.

Notice that the semijoin operation is not conductative; that is,

R >< 5 7 5 🖂 R

#### 25.4.3 Query and Update Decomposition

In a DDRMS with no domination numbers are the user physics a query directly internated specific frigments. For example, consider another query  $Q_{i}$  "Retrieve the names and hours per week for each employee who works on some project controlled by department 5," which is specified on the distributed database where the relations at site 2 and 3 ac shown in Figure 25.3, and those of site 3 are shown in Figure 5.6, as in carbon example. A user which submits such a query mass specify whether it references the enough at site 2 (figure 25.5) or the isotech and works\_ow relations at site 1 (Figure 5.6). The user must also maintain consistency of replicated data items when updeting a 142005 with no repletation transference.

On the other hand, a 1009MS that supports fail disobation. Engineeriation, and soplication transported y allows the user to specify a query or update request on the schema of Ergone 5.5 just as through the 1000S were centralized. For opdates, the 306MS is responsible for maintaining consistency anong repletated dense by using one of the distributed concorrency control algorithms to be discussed in Soution 25.5. For queries, a query decomposition unclude mast break up or decompose a query into subgremes that can be exercised as the individual sites. In addition, a strategy for conforming the results of the subgremes that an iterative result must be generated. Whenever, the 1009Ms determines that an itera referenced in the query is replicated, it must choose or materialize a particular replication and a generating energy are output.

To determine which replicas its Jude the data preas referenced in a query, the 13980s refers to the fragmentation, replication, and distribution information stored in the 142608 cotalog. For we used fragmentation, the ormbute lot for each fragment is kept in the cotalog. For formontal fragmentation, a condition, sometimes called a guard, is kept for each fragment. This is basically a selection condition that specifies which tuples exist in the fragment, it is called a guard because only taples that stassly this condition are permitted to be stored at the fragment. For anxed fragments, both the attribute list and the guard tendered fragments, both the attribute list and the guard condition are kept in the statalog.

In our earlier example, the goold conditions for fragments at site 1 (Figure 5.6) ore TRUE (all toples), and the attribute lists are \* (all attributes). For the fragments shown in Figure 25.3, we have the grand conditions and attribute lists shown in Figure 25.7. When the 2006ts decompose- are update request, at can determine which fragments must be updated by exampling their guard conditions. For example, a user request to insert a new twelvice tuple <[Alex1, 101, 102], "Colleman", 13456712391, 122 APR-641, 13306 Sandstone, Houston, TX1, M, 53600, 19876543211, 45 would be decomposed by the PODMS minimum two insert requests the first inserts the proceeding tuple in the SPR-666 fragment

```
(a) EMPD5
```

```
attribute tist: FINAME_MINIT_UNAME,SSN,SA_AFY_SUPERSSN, DNO
guard condition: DND+5
DEP5
attribute tist: • (all attributes DNAME DNLMRER MGRSSN,MGRSTAHT DATE)
guare condition: UNUMBER=5
THEP5_LOUS
attribute tist: • (all attributes DNUMBER,LOCATION)
guard condition: UNUMBER=5
PROUSS
attribute tist: • (all attributes PNAME,PNUMBER PLOCATRON,DNUM)
guard condition, DNUM=5
WORKS_ONS
attribute tist: • (all attributes ESSNIPMO,HOURS)
outra condition, ESSNIPT, (Answer (PROUSS))
```

EMPD4

```
 (b) attribute list: FNAME,MINITLINAME.SSN.SALARY.SUPERSSN. DNO

puard consition: DNO+4

DEP4

attribute list + (all attributes DNAME.ONUMBER,MGRSSN.MGRSTARTDATE)

guard condition: DNUMBER-4

DEP4_LOCS

attribute list + (all attributes: DNUMBER.LCCATION)

guard condition: DNUMBER-4

PR().JS4

attribute list + (all attributes: PNAME PNLMBER,PLOCATION,ONITM)

guard condition: DNUMBER-4

PR().JS4

attribute list + (all attributes: PNAME PNLMBER,PLOCATION,ONITM)

guard condition: CNUM-4

WORKS_(IN4

attribute list + (all attributes: ESSN.PNO,HOURS)

guard condition: LSSN IN (5554 (EMPD4))

CRI PNO IN (Strangel + (PRO:S4())
```

FIGURE 25.7 Curard conditions and attributes lists for fragments can Sile 2 fragments, the Sile 3 fragments
at site 1, and the second inserts the projected ruple k'Alex'. (B'. (Coleman', 1345671239', 33000, 1987654321', 45 in the swood tragment at site 5.

For query decomposition, the OPRMS can determine which fragments may contain the required toples by comparing the docty condition with the guard conditions. For example, consider the query Q. "Betrieve the names and hours per week for each employee who works on some project controlled by department 5") this can be specified in SQL on the schema of lagure 5.5 as follows.

Q: SELECT FRANE, LHANE, HOURS FROM EMPLOYES, PROJECT, WORKS ON MINERE UNUM-S AND PHUMBER-PRO AND ESSN-554;

Suppose that the query is submitted at site 2, which is where the query result will be needed. The DOWNs can detergime from the guard condition on 26055 and 26855 as5 that all tuples satisfying the conditions (0609 = 5, AND examples e-root reside at site 2. Hence, o may decompose the query into the following relational algebra -aliquenes:

This decomposition can be used to exercise the query by using a seminon strategy. The 194Wis knows from the guard conditions that usings contains exactly those inplesatisfying (0004 – 5) and that weak\_005 contains all tuples to be joined with (80155) beake subquery 11 can be executed at site 2, and the projected column 55% can be sent to site 1. Subquery T2 can then be executed at site 2, and the projected column 55% can be sent to site 1. Subquery T2 can then be executed at site 1, and the result can be sent back to site 2 where the final query result is calculated and displayed to the user. An alternative strategwould be to send the query Q itself to site 1, which includes all the database tuples, where it would be executed be ally and from which the result would be sent back to site 2. The query optimizer would estimate the class of both strategies and would choose the role with the lower cost compare.

# 25.5 OVERVIEW OF CONCURRENCY CONTROL AND RECOVERY IN DISTRIBUTED DATABASES

For concurrency control and recovery purposes, numerous problems arise in a distributed (PMS) environment that are not encountered in a centralized PMMS environment. These include the following:

- Dealing with multiple copies of the data items: The concurrency control marked is responsible for munitaring consistency among their copies. The recovery method is responsible for making a copy consistent with other copies if the site on which the copy is stored fails and recovers later.
- Failure of activition spect The DDBMs should commune to operate with neutrining actes, it possible, when one or more individual sites (all, When a site recovers, its local database must be brought op accilite with the test of the sites before it reconsthe system.

- Falses of communication toky: Two system must be able to deal with failure of one or more of the communication links that connect the sites. An extreme case of this problem is that network partitioning may occur. This breaks up the sites into two or more partitions, where the sites within each partition can communicate only with one another and not with sites in other partitions.
- Destributed contacts. Problems cap tarks with commutting a transaction that is accessing databases spored on indriple sites of some sites fail during the contact process. The two-phase commit protocol (see Chapter 19) is often used to deal with this problem.
- Distributed deadlock. Deadlock may occur among several stres, so techniques for dealing with deadlocks must be extended to take this into account.

Distributed concurrency control and recovery techniques must deal with these and other problems. In the following subsections, we review some of the rechniques that have been suggested to deal with recovery and concurrency control in DDIMEs.

### 25.5.1 Distributed Concurrency Control Based on a Distinguished Copy of a Data Item

To deal with replicated data, teny in a distributed database, a number of concurrency control methods have been proposed that extend the concurrency control techniques for centralized databases. We discuss these techniques in the context of extending controlited focking. Similar extensions apply to other concurrency control techniques. The idea is to designate a particular copy of each data term as a **distinguished copy**. The tocks for this data item are associated with the disarguished copy, and all cocking and unlecking requests are sent to the site that contains that copy.

A number of different methods are based on the data they different themmethod of choosing the distinguished copies. In the primary site technique, all distinguished copies are kept at the same site. A modification of this approach is the primary site with a backup site. Another approach is the primary copy method, where the dotinguished copies of the various data items can be stored in different sites. A site that includes a distinguished copy of a data item basically acts as the coordinator site for concurrency control evident from We discuss these techniques next.

Primary Site Technique. In this method a single primary site is designated to be the coordinator site for all doublese leave. Hence, all locks are kept or that site, and all requests for locking or unlocking are sent there. This method is thus an extension of the centralized locking approach. For example, if all transactions follow the two-phase locking protocol, semilicability is guaranteed. The advantage of this approach is that it is a simple extension of the centralized approach and hence is not overly complex. However, it has certain inherent disadvantages. One is that all locking requests are sent to a single site, possibly overloading that site and causing a system bottleneck. A second disadvantage is that failure of the primary site paralytes the sector, once all focking information is kept at that site. This can limit system reliability and availability. Although all locks are accessed at the primary site, the items themselves can be accessed at my site at which they reside. For example, once a transaction obtains ( REM\_EOC on a data item from the primary site, it can access any copy of that data item. However, once a transaction obtains a 8117 LOC and updates a data item, the 100405 is responsible for opdating all copies of the data item before releasing the lock.

Primary Sile with Backup Site. This approach addresses the second disadvantage of the primary site method by disagrating a second site to be a backup site. All locking information is maintained at both the primary and the backup site. In case of primary site failure, the backup site takes over as primary site, and a new backup site is chosen. This simplifies the process of recovery from balance of the primary site, since the backup site takes over and processing can resume after a new backup site is chosen and the lock status information is copied to that site. It slows down the process of acquiring locks, however, because all lock requests and granting of locks must be recorded at bols da primary and the backup site a response is sent to the requesting miniaterial the problem of the primary and backup sites becoming overloaded with requests and dowing down the system remains undiminished.

Primary Copy Technique. This method attempts to distribute the local of lock conditiation among various sites by lorging the distinguished copies of different data items stored at different view. Failure of one site afferts any transactions that are accessing links on items whose primary copies reside at that site, but other transactions are not affected. This method can also use backup sites to enhance reliability and availability.

Choosing a New Coordinator Site in Case of Failure. Whenever a coordinate site tails in any of the preceding techniques, the sites that are still running must choose a new coordinator. In the case of the primary site approach with no backup site, all executing transactions must be abouted and restained in a technis recovery process. Part of the recovery process immives choosing a new primary site and creating a look manager process and a record of all lock internation at that site. For methods that use backup sites, transaction processing is inspended while the bit kap site is designated as the new minury site and a new backup site is chosen and is sent copies of all the locking information from the new primary site.

If a backup site X is about to become the new princip site, X can choose the new backup site from imong the system's running sites. However, if no backup site existed, or if both the primary and the backup sites are down, a process colled election can be used to choose the new coordinator site. In this process, any site Y that attempts to commonicate with the coordinator site repeatedly and fails to do so can assume that the coordinator is down and can start the election process by sending a message to all running sites proposing that Y become the new coordinator. As soon as Y become itself is quite complex, but this is the main idea behind the election method. The algorithm itself is quite complex, but this is the main idea behind the election method. The algorithm faces resolves are attempt by two or more sites to become coordinator of the same time. The references no the Selected Bibliourophy at the end of this chapter discuss the process in detail.

### 25.5.2 Distributed Concurrency Control Based on Voting

The concurrency control methods for replicated items docussed earlier all use the idea of a distinguished copy that nonritions the locks for that item. In the voting method, there is no distinguished copy rather, a lock request is sent to all sites that includes a copy of the data item. Each copy maintains its own lock and can grant or deny the request for it. It is transaction that requests a lock is granted that lock by a bianary of the copies, it holds the lock and informs all copies that it has been granted the lock. It is manuation does not receive a majority of votes granting it a lock within a certain time out period, it cancels its request and informs all sites of the cancellations.

The voting method is considered a tridy distributed concurrency control method, since the responsibility for a decision resides with all the sites involved. Sumdation studies have shown that voting has higher message traffic among sites that do the distinguished copy methods. If the algorithm takes into account possible site failures during the voting process of because extremely complex.

### 25.5.3 Distributed Recovery

The recovery process in distributed databases is quite involved. We give only a very brief idea of some of the issues here. In some cases it is quite difficult even to determine whether a site is down without exchanging numerous messages with other sites. For example, suppose that site X sends a message to site Y and expects a response from Y but does not receive in. There are several possible explanations:

- The message was not delivered to Y because of communication follores
- Site Y is down and could not respond.
- Site Y is running and sent a response, but the response was not delivered

Without additional infimations or the sending of additional messages, it is difficult to determine what actually happened.

Another problem with distributed recovery is distributed commit. When a transaction is updating data at several sites, it commot content until it is solverbat, the effect of the transaction on cosy site cannot be lost. This means that every site inust that have recorded the local effects of the transactions permanently in the local site log on disk. The two chose commit protocol, discussed in Section 19.6, is often used to ensure the correctness of distributed commit

# 25.6 AN OVERVIEW OF 3-TIER CLIENT-SERVER ARCHITECTURE

As we pointed one in the chapter introduction, full-scale (2008) so have not been developed resupport all the types of functional uses that we discussed so far, instead, distributed database applications are being developed in the context of the client server architecmics. We already introduced the two-for chent-server architecture in Section 2.5. It is new more remained rouse a directive architecture, particular in Web applications. This architecture is illustrated in Figure 25.8.

In the three tier chent-server architecture, the following three layers exist:

- 1. Presentation layer (client): This provides the user interface and interacts with the user. The programs of this layer present Web interfaces or forms to the client in order to interface with the application. Web browsers are often utilized, and the languages used include MIML, AVA, JavaScopt, PERL Visual Basic, and so on. This layer handles user imput, ourput, and normation, by accepting user commonds and displaying the needed information, estably in the form of static or dynamic Web pages. The latter are employed when the interaction involves dual base access. When a Web interface is used, this layer repically communicates with the application layer via the RITP protocol.
- 2. Application layer (business logic). This layer programs the application logic, for example, queries can be formulated based on user input from the cherit, or query results can be formatted and sent to the client for presentation. Additional applications functionality can be handled at this layer, such as security checks, identity verification, and other functions. The application layer can interact with one or more databases in data sources as needed by consecuring to the database using ODBC, 3DBC, SQU(C) or other database access techniques.
- 3. Database server: This layer handles query and update requests from the appletation layer, processes the requests, and send the results. Usually stg. is used in across the database if it is relational in object relational and stored database pro-



FIGURE 25.8 The threatier chent-server architecture.

codute- may also be invoked. Query results (and queries) may be formatted into XML (see Chapter 26) when transmitted between the application server and the Jatabase server.

Exactly how to divide the DOWs functionality between them, application server, and database server may cary. The common approach is to meltide the functionality of a centralized DOWs at the database server level. A complex infredmional DOWs products have taken this approach, where an SQL server is provided. The application server must then formulate the appropriate SQL queries and contract to the database server when needed. The chent provides the processing for user interface interactions. Since SQL is a relational standard, various SQL servers, possibly provided by different sendors, can accept SQL conmonds through standards such as ODIN 1008, SQL/C11 (see Chapter 9).

In this arch tecture, the application server may also refer to a data dictionary that includes information on the distribution of data unlong the various 500 servers, as well as nodules for decomposing a global query into a number of local queries that can be executed at the various sites. Interaction between application server and database server inght proceed as follows during the processing of an 500 query:

- The application server formulates a user query based on input from the cherit layer and decomposes it into a number of independent site queries. Each site query is sent to the appropriate database server site.
- Each database server processes the lurai query and sends the results to the application server site. Increasingly, 2001 is freing routed as the standard for data exchange (see Chapter 20) so the database server may format the openy result into XMU before sending it to the upplication server.
- 3 The application server combines the results of the subqueries to produce the result of the originally required query formats in into 10781, or some other form accepted by the chern, and sends into the client site for display.

The application server is responsible for generating a distributed execution plon for a multisite query or transaction and for supervising distributed execution by sending commands to servers. These commands include local queries and transactions to be executed, as well as commands to transmit data to other chents or servers. Another function controlled by the application server (or coordinater) is that of ensuring consistency of replicated copies of a data item by employing distributed (or global) concurrency control techniques. The application server anost also ensure the atomicity of plobal transactions by performing global recovery when certain sites forl. We discussed distributed recovery and concurrency control moderner control in Section 25.5.

It the DDRAS has the capability to hale the details of data distribution from the application server, then it coubles the application server to execute global queries and transactions as though the database were contralized, without having to specify the sites at which the data referenced to the query or transaction resides. This property is called distribution transparency. Some DDPRASs do not provide distribution transparency, unstead requiring that applications be owner of the details of data distribution.

## 25.7 DISTRIBUTED DATABASES IN ORACLE

In the cherit-server architecture, the Oracle database system is divided into two parts (1) a frontisend as the client portion, and (2) a back-end as the server portion. The client portion is the front-end database application that interacts with the user. The client has no database exponsibility and inerely bundles the requesting, processing, and presentation of data managed by the server. The server portion rons Oracle and bundles the functions related to concurrent shored access. It accepts 8.20 and 0./8.20 statements originating from cherit-envert applications processes them, and serves the results back to the cherit. Oracle cherit-cover applications provide location non-parency by making location of data transparent to users; so cial features like views, synonytics, and procedures contribute to thes. Olabal matring is achieved by using staticas?. 0.0414645566962 to refer to tables imaginely.

Othele uses a trencphase communiprotocol to deal with concurrent distributed transactions. The CONVERT statement triggers the two phase communiprotonism. The SECO (necocerer) background generies canonicarically resolves the autoime of those distributed transactions in which the communication was interrupted. The RSOO of each local Crucle Server accommittally commits or table back any "in-doubt" distributed transactions consistently commuted involved nodes. For long-term failures, Cracle allows each local CRA to manually committor roll back any in-doubt transactions and free up testories. Clobal consistency can be many-anneal by restoring the database at each site to a predetermined fixed point in the past

Oracle's distributed database architecture is shown in Figure 25.9. A node in a distributed database system can act as a cherit, as a server, to both, depending on the situation. The figure shows two sites where database called HQ (headquarters) and Szles are kept. For example, in the application shown running at the headquarters, for an equistructure issued against local data (for example, with the local data (for example, used a superior acts as a cherit.

All Oracle databases in a distributed database system (DD95) use Oracle's retroorking software Net5 for interdatabase communication. Net8 allows databases to communicate across networks to suggest remote and distributed transactions. It packages S2 statements into one of the many communication protocols to facilitate cherit to server communication and then packages the results back similarly to the chern. Each database has a image global name provided by a hierarchical arrangement of network domain names that is profixed to the database mane to make it unique.

Oracle supports database links that define a one-way communication path from one Oracle database to prother. For example,

CREATE DATABASE LINK sales.us americas:

establishes a connection to the sales database in Figure 25.9 under the network domain us that comes under domain anen i cas.

Data in an Oracle 19176 can be replicated using stugshow or replicated moster tables Replication is provided at the following levels

 Basic replication. Replicos or tables are managed for rend-only access. For appales data must be accessed at a single primary site.



**FIGURE 25.9** Oracle distributed database systems. *Source:* From Oracle (1997a). Copyright © Oracle Corporation 1997. All rights reserved.

 Advanced (connerne) replication. This extends beyond basic replication by allowing applications to update table replicits throughout a replicated DDBS. Data can be read and opdated of any site. This requires additional software called Oracle's advanced replication option. A snapshot generates a copy of a part of the table by means of a query colled the suspition defining query. A simple snapshot definition looks like this:

CREATE SNAPSHOT sales.orders AS SELECT & FROM sales.orders@hg.us.americas; Oracle groots snopshots anto refresh groups. By specifying a refresh interval, the snopshot is automatically refreshed periodically at that interval by ay to ten Snapshot Refresh Processes (SNPs). If the defining query of a snapshot contains a distinct or aggregate function, a GOOOP PY or CONNECT BY clause, or join or set operations the snapshot is termed a complex snapshot and sequeres additional processing. Oracle (up to version 7 f) also supports 600000 snapshots that are lossed on physical row identifies of rows in the ma-ter table.

Heterogeneous Databases in Oracle – for a heterogeneous optis, or least ondatabase is a non-Oracle system. Oracle Open Gateways provides access to a non-Oracle database from an Oracle server, which uses a database link to access data or to execute temote procedures in the non-Oracle system. The Open Gateways teature includes the following

- Distributed consections. Under the two-phase commit mechanism, transactions may span Oracle and non-Oracle systems.
- Transparent 502 access: 5QL statements issued by an application are transparently transformed into 5QL statements understood by the non-Oracle system.
- Pais-through SQL and storal practileres: An application can directly access a non-Uracle system using that system's version of SQL. Stored procedures in a non-Oracle SQL-based system are treated as if they were PL/SQL remote procedures.
- Global query optimization. Cardinality information, indexes, etc., at the non-Olacle system are accounted for by the Oracle Server query optimizer to perform global query optimization.
- Procedural access: Procedural systems like messaging or questing systems are accessed by the Oracle server using PL/SQB remote procedure calls.

In addition to the obove, data diet on any references are translated to make the not-Oracle data dictionary appear as a part of the Oracle Server's dictionary. Character set translations are done between national language character sets to connect multidingail databases.

## 25.8 SUMMARY

In this chapter we provided an introduction to distributed databases. This is a very head togat, and we discussed only seen, of the lysic techniques used with distributed databases. We first discussed the reasons for distribution and the potential advantages of distributed databases over controlled systems. We also defined the concept of distribution transporency and the related concepts of fragmentation transporency and replication transporency. We discussed the design issues related to data fragmentation, replication, and distribution, and we distinguished between homeintal and vertical fragments of relations. We discussed the response to improve system reliability and availability. We categorized (009003) where is of data replications and the improve system reliability and availability. We categorized (009003) where internation and the rescale of head autonomy. We discuss the availability of software methodical of head autonomy. We discussed to a degree of head autonomy. We discuss the rescale of the distribution of the degree of head autonomy.

cussed the usues of federated database management in some detail fedaving on the needs of supporting various types of autonomics and dealing with semantic factorogeneity.

We dilustrated some of the techniques used in distributed query processing, and discussed the test of communication among sites, which is considered a major factor an distributed query optimization. We compared different techniques for executing joins and presented the senargoin technique for joining relations that reside on different sites. We briefly discussed the concursency control and accovery techniques used in DDRSs. We reviewed some of the additional problems that must be dealt with in a distributed environment that do not appear in a centralized environment.

We then discussed the client-server architecture concepts and related them radistributed databases, and we described some of the facilities in Oracle to support distributed databases.

### **Review Questions**

- 25.1. What are the main reasons for and potential advantages of distributed databases?
- 25.2 What additional functions does a PDBMS have over a centralized 209483
- 25.3. What are the main software modules of a 10-03(s) Discuss the main functions of each of these modules in the context of the cheric server architecture.
- 25.4. What is a fragment of a relation? What are the mum types of fragments? Why is fragmentation a useful concept in distributed database design?
- 25.5. Why is data replacition aseful in DDBMSS? What rypical units of data are replicated?
- 25.6. What is meaningly data allocation on distributed database design? What replical units of data are distributed over sites?
- 25.7. How is a horizontal partitioning of a relation sponfied? How can a relation be put back regeriter from a complete horizontal partitioning?
- 25.8. How is a vertical partitioning of a relation specified? How can a relation be parback together from a complete vertical partitioning?
- 25.9 Discuss what is meant by the following terms: degree of homogeneity of a 2088/8, degree of local autonomy of a 10.08518, federated 109518, discobilition transferency, fragmentation transportery replication transparency multifatative system.
- 25-10. Discuss the minime problem in distributed databases.
- 25.11. Discuss the different techniques for executing an equipoit of two files located at different sites. What main factors affect the cost of data transfer?
- 25.12. Discuss the semijous method for executing an equiptin of two tiles located at difletent sites. Under what conditions is an equiption strategy efficient?
- 25.13. Discuss the factors that affect query decomposition. How are goard conditions and attribute lists of fragments used during the query decomposition process?
- 25.14. How is the decomposition of an update request different from the decomposition of a query! How are goard conditions and articlute lists of fragments, and during the decomposition of an update request?
- 25.15. Discuss the factors that do not appear in renealized systems that affect concurrency control and recovery in distributed systems.

- 25.16. Compare the primary size method with the primary copy method for distributed concurrency control. How does the use of backup area affect each?
- 25.17. When are coting and elections used in distributed databases?
- 25.15. When are the software components in a chenreserver 0.98883 Usingare the nontier and three-tier thent-server architectures.

#### Exercises

- 25.19. Consider the data distribution of the COMPANY database, where the fragments at sites 2 and 3 are as shown in Figure 25.3 and the fragments at site 1 are as shown in Figure 5.6. For each of the following quoties, show at least two strategies of decomposing and executing the query. Under what conditions would each of your strategies work we'l?
  - For each employee in department 5, retrieve the employee name and the names of the employee's dependents.
  - b. Front the names of all employees who work in department 5 but who work on some project not controlled by department 5.
- 25/20. Consider the following relations:

eccks (Buck#, Primary\_author, Topic, Total\_stock, Sprice) socksist (Store#, City, State, Zip, Inventory\_value) store (Store#, Book#, Qty)

Te14.\_stock is the total number of books to stock, and "twest685\_vaule is the total inventory value for the store in dollars.

- Give an example of two simple predicates that would be meaningful for the ecosions relation for horizontal partitioning.
- How would a derived horizontal partitioning of store be defined based on the gravitusining of pecksioke?
- Show predicates by which tooks may be horizontally partitioned by topic
- d. Show how the store may be further partitioned from the partitions in (NFb) adding the predicates in (c).
- 25.21. Consider a distributed database for a bookstore chain colled National Books with 5 sites called EAST, MIGOLE, and mEST. The relation schemas are given in question 24.20. Consider that Books are fragmented by \$280.3 amongsts into:

B<sub>1</sub>: 800×1, 117 το \$20 B<sub>2</sub>: 800×2, έτομι \$20 01 το \$50, B<sub>2</sub>: 800×3, έτομι \$50 01 το \$100.

B<sub>2</sub>: BOOK4: S1CO CL and above.

Similarly, consisters are divided by Zipcodes into

S<sub>1</sub>: EAST: Ziplades up to 35000.

Sy: MIDDLE: Zipcodes 35001 to 70000.

Sig WEST: Zipcodes 70001 to 000044.

Assume that STOCK is a derived fragment based on accestors only.

a. Consider the query

```
SELFCT Books, Intal stock
SROM Books
NHERE Sprice > 15 and Sprice < 55;
```

Assume that tragments is **500KSTORE** are non-implicated and assigned based on region. Assume further that **B00X5** are allocated as:

```
\begin{array}{l} \text{EAST}: \ B_1, \ P_2 \\ \text{MIDDLE}: \ B_1, \ B_2 \\ \text{WEST}: \ B_1, \ B_2, \ B_3, \ B_4 \end{array}
```

Assuming the query was submitted in EAST, what remote subqueries does it generate? (write in SqL).

- b. If the bookprice of BOOK#+ 1234 is updated from \$45 to \$55 at site HIDDLF, what opdates does that generate? Write in English and then in SQL.
- Covers an example query issued at MEST that will generate a subquery for MTDDLE.
- d. Write a query involving selection and projection on the above relations and show two possible query trees that denote different sears of execution.
- 25.22. Consider their you have been asked to propose a database architecture in a large organization. General Motors, as an example, to consolidate all databased from Hierarchital and Network models, which are explained in Appendices C and D, no specific knowledge of these models is trended) as well as relational databases, which are geographically distributed so that global applications can be supported. Assume that alternative one is to keep all databases as they are, while alternative two is to first convert them to relational and then support the applications over a distributed integrated database.
  - a Draw two seltematic diagrams for the above internatives showing the hirkness among appropriate schemas, For alternative one, choose the approach of providing export schemas for each database and constructing unified schemas for each application.
  - List the steps one has ro go through under each alternative from the present situation until global applications are visible.
  - Compare mese from the is-ues of: (i) design time considerations, and (ii) iuntime considerations.

# Selected Bibliography

The restbooks by Ceri and Polagarii (1984a) and Ossi and Valdariez (1999) are devoted to distributed databases. Balsial (1996), Tarmenbaum (1996), and Stallings (1997) are textbooks on data communications and computer networks. Conner (1997) discusses networks and interprets. Device (1993) is a textbook on cheat-scorer computing. Ossi et al. (1994) has a collection of papers on distributed object management.

Distributed database design has been addressed in terms of horizontal and vertical tragmentation, allocation, and replication. Ceri et al. (1982) defined the concept of interent horizontal tragments. Cere et al. (1982) developed an integer programma based optimization model for horizontal tragmentation have on allocation. Navathe et al. (1984) developed algorithms for vertical fragmentation based on attribute att

Distributed query pisce-sine, optimitation, and decomposition are discussed in Heyner and Yao (1979). Kerschberg et al. (1982). Apers et al. (1983). Ceri and Pelaeatti (1984), and Podenck et al. (1992). Betristent, and Goodmar. (1981) discuss the theory behind schipom processing. Wong (1983) discusses the nee of relation-haps thatelaten hapmentation. Committein (1981) and recovery schemes are discussed in Betristen and Goudman. (1981a). Kuman and Hise (1998) have some articles related to recovery in distributed distabases. Elections in distributed systems are discussed in Garia-Molina (1982). Lamport (1975) discusses problems with generating unique unicatanity in a distributed system.

A concurrency control technique for repleated due that is based on varing is presented by Thomas (1979). Gittoid (1979) propose the use of weighted earing, and Paris (1986) describes a method callod coting with witnesses. Japana and Mutchler (1990) discuss denamic voting. A rechmique called *available copy* is proposed by Bernson and Cocolinan (1984), and one that uses the isless of calcoop is presented in LLA-badraud Todeg (1988). Other recent work that discusses replicated data acclude. Gladney (1989), Agrawal and ELAbbadr (1980), ELAbbadr and Tooleg (1990), Kumor and Seges (1980), Mukkamala (1989), and Wolfson and Milo (1991). Baseduri (1988) discusses optimistic protocols. For 1008 concurrency control. Gateta-Molana (1983) and Kumar and Stonebraket (1987), declass techniques that use the scitarities of the transactices. Distributed concurrency control. Careta-Molana (1983) and Kumar and Stonebraket (1987), declass techniques that use the scitarities of the transactices. Distributed concurrency control (collignes based on looking and distinguished copies ne presented by Menase et al. (1982) and Minioura and Wiederhold (1982). Obernark (1982) presents algorithms for distributed deadlock detection.

A survey of recovery rechniques in distributed systems is given by Kohler (1981) Reed (1983) discusses atomic actions on distributed data. A book edited be Bhargae (1985) presents corious approaches and techniques for concurrency and reliability in distributed systems.

Lederated database systems were first defined in McLood and Heinibigner (1985). Techniques for schema integration in federated databases are presented by Elmastrieral (1986). Patienterial. (1986). Hayne and Roin (1992), and Motto (1987). Elmastrieral Helpf (1988) and Gamal-Eldar et al. (1985) discuss the update problem in heterogenera-PDBSs. Heterogeneous distributed database issues are discussed in Histori and Kanel (1989). Sherh and Larson (1993) present on exhaustori survey of federated database management. Recently, nullidotabase systems and interpretability have become important ropes. Techniques for dealing with seminic incompatibilities among padrizle databases are examined in DeMichiel (1939). Siegel and Madruck (1991). Krishnanerthy et al. (1991), and Wang and Madrick (1989). Castano et al. (1998) present an excellent survey of techniques for analysis of schemos. Poouta et al. (1995) discuss object orientation in multidotabase systems.

Transaction processing in multidoublases is discussed in Mehrotra et al. (1992), Georgakopoulos et al. (1991), Elmagarmid et al. (1990), and Brietbart et al. (1990), anong others, Elmagarmid et al. (1992) decuss transaction processing for advanced applications, including engineering applications decussed in sleder et al. (1992).

The workflow systems, which are becoming popular to manage information in complex organizations, use multilevel and nested transactions in comparation with distributed databases. Weakon (1991) discusses multilevel transaction management. Alonswet al. (1997) discuss limitations of current workflow systems.

A number of experimental distributed D566s have been nuplemented. Those melude distributed (NGROS (Epstein et al., 1975), CD7S (Devor and Weeldreyer, 1980), SOO-(Rothnic et al., 1980). System R\* (Endsay et al., 1984), SROO-CEETA (Ferrier and Stangret, 1982). and SOUTHASE (South et al., 1981). The OMNRASSE system (Rusinkiewicz et al., 1988) and the Federated Information Rase developed using the Condide data model (Navathe et al., 1994) are examples of tederated (D0808, Prouta et al. (1995) present a comparative survey of the federated database system prizoryges. Most commercial P608 vendors have produces using the client-server approach and offer distributed versions of their systems. Some system issues concerning client-server D808 architectures are discussed in Clarec et al. (1992). DeWirr et al. (1990), and Wing and Rowe (1991). Khoshation et al. (1992) discuss design issues for relational (600-s) in the client-server environment. Client-server management issues are discussed in many books such as Zawange and Adriants (1996).



# EMERGING TECHNOLOGIES







We now turn our attention to how databases are used and accessed from the fatemet Many electronic commerce (e-commerce) and other internet applications provide Web interfaces to access information stored in one or incre databases. These databases are often referred to as **data** sources. It is common to use two-rier and three-rier chertserver architectures for Internet applications (see Section 2.5). In some cases, other variations of the chernserver model are used. E-commerce and other Internet database applications are destended to interact with the user through Web interfaces that display Web pages. The common method of specifying the contents and fornatting of Wish pages is through the use of hyperlink documents. There are various languages for writing these documents, the most common being HTML (Hypertext Markup Language). Although HTML is vedely used for formatting and structuring Web dataments, it as not suitable for specifying strucand data that is extracted from dambases. Recently, a new language-mannely, AMI (Extended Markop Language)--has emerged as the standard for structuring and exchange ing data over the Web XAU can be used to provide information about the structure and meaning of the data in the Web pages rather than just specifying how the Web pages are tormatted for display on the screen. The formatting papers are specified separately --for example, by using a furnaring Emgange such as ASE (Extended Stylesheer Language).

This chapter describes the basics of accessing and each inging information over the Internet. We start in Section 20.1 by discussing how traditional Web pages differ from structured databases, and discuss the differences between structured, semistructured, and instructured data. Then in Section 26.2 we turn out interstom to the XML standard and its receiveructored (hierarchical) data model. Section 20.3 discusses XML documents and the languages for specifying the structure of these doruments, namely, XML 91.5 (Excurrent Type Definition) and XML schema. Section 20.4 presents the various approaches for storage XML documents, whether in their native (text) format, in a compressed form, or in relational and other types of databases. Section 26.5 gives an everynew of the languages proposed for querymic XML data. Section 26.0 summaries the chapter.

# 26.1 STRUCTURED, SEMISTRUCTURED, AND UNSTRUCTURED DATA

The information stored in distabases is known as structured data because in is represented in a sumit termat. For example, each record in a relational database table is such as the rot tree table in Figure 5.0. To how the same formuli as the other records in that table For structured data, it is common to carefully design the database using techniques such as those described in Chapters 5,  $\pm$ , 7. If and 11 in order to errors the database scheme. The Distance her clarks to ensure that all data follows the structures and constraints specined in the schema.

Heavier, nor all data is collected and inserted into controlly designed structured databases. In some applications, data is collected in an address manner before it is known how it will be stored and mannee. This data near how a certain structure, but not all the information collected will have identical structure. Some introduces may be shared among the various entities, but other introduces may exist only in a few entities. Moreover, additional attributes can be introduced in some of the perior data nears at any million address in no predefined scheme. This type of data is known as semistructured data. A number of data models have been introduced for representing semistructured data, orten based on ostor tree of graph data structures rather than the flat relational model structure.

A key difference between structured and semistimetized data concerns how the schema constructs (such as the manes of arributes, relationships, and entite types) achandled. In semistructured data, the schema information is much or with the data values since each data object can have different attributes that are not known in advance Ecoce, this type of data is sometimes referred to as self-describing data. Consider the following example. We want to collect a list of bibliographic references related to a certain research project. Some of these may be books or technical reports, other may be research articles in journals or conference proceedings, and still others may refer to complete journal issues or conference proceedings. Clearly, each of these may how different attributes and different types of information. Even for the same type of references say, conference articles, we may have different information. But example, one article citation may be dure complete, with full information about author names, rule, proceedings, page muniters, and sis on whereas appring clustery may nor have all the information available. New types of bibliographic sources may appear in the haute for example, references to Web pages of to conference torozial - and these may base per arreibates that describe them.



FIGURE 26.1 Representing semistructured data as a graph

Semistractured data may be displayed as a directed graph, as shown in Figure 26.1. The information shown in Figure 26.1 corresponds to some of the structured data shown in Figure 5.6. As we can see, this model somewhat resembles the object model (see Figure 20.1) in its ability to represent complex objects and nested structures. In Figure 26.1, the labels or tags on the directed edges represent the schema names: the rames of armitication object types for entry types or closesty, and relationships. The internal nodes represent individual objects or composite attributes. The leaf nodes represent actual data colors of simple (aromae) numbutes.

There are two main differences between the sensimicruped model and the object model that we discussed in Chapter 20:

- The subcrue information manys or attributes, relationships, and classes (object types) in the somistructured model is intermixed with the objects and their data volues in the some data structure.
- In the semistructured model, there is no requirement for a precisitivel schema to which the data objects most conform.

In addition to structured and sentistroctured data, a third category exists, known as unstructured data because there is very limited indication of the type of data. A typical example is a text document that contains information embedded within a . We's pages in HTML that contains some data are considered to be unstructured data. Consider part of an HTML file, shown in Figure 20.2. Text that appears between nigled brackets,  $\leq ... \geq$  is an HTML tag. A tag with a Unckslush,  $\leq ... \geq$ , indicates an end tag, which represents the

```
<html>
<head>
. . .
</head>
coudys
Al>List of company projects and the employees in each project
 kH2>The ProductX project:</H2>
 <1 No
 <TQ_winth=`$0%"><font_size="2"_face="Arial">)ahn_Smith:</TQ>
 kTD>32.5 hours per weekk/TD>
 </18>
 <18>
 <TD width="$0%%">loyce English:</TD>
 <TD>20.0 hours per week</TD>
 </TR>
</rable>
«II2»The ProductY project: </H2»</pre>
etka
 <TD width="SC%">Dahn Smith:</TD>
 <TD>7.5 hours per weekk/TD>
 ≪/TR>
 < IR>
 <TD width="$0%%">>font size="2" face="Artal">loyce English:</TD>
 <TD>20.0 hours per weekk/TO>
 e∕‴fi>
 «TR»
 <TD width="50%%">Franklin Wong:</TD>
 <TCo10.0 hours per week</TCo
 e∕‴Re
</tables
. . .
</bodys
</html>
```

FIGURE 26.2. Part of an UKM document representing unstructured data.

cooling of the effort of a matching start tag. The rugs mark up the documents in order to instruct an HTML processor how to display the text between a start tag and a nurching end rag. Hence, the tags specify document formating rather than the meaning of the various data elements in the document. HTML tags specify information, such as four as and style (buddlice, indice, and so on), color, heading levels in documents, and so on Some rags provide text structuring in documents, such as specifying a numbers for

1. That is why it is known as Hyperrex: Markup Language

enumbered list or a table. Even these structuring tags specify that the embedded textual data is to be displayed in a certain ansiner, rather than andicating the type of data represented in the table.

187ML uses a large number of predefined tags, which are used to specify a variety of commands for formatting Web documents for display. The start and end tags specify the range of react to be formatted by each command. A few examples of the tags shown in Figure 26.2 follow:

- The <html> ... </html> tags specify the boundaries of the document.
- The document header information- within the shrads ... s/heads tags --sporines various commands that will be used elsewhere in the document. For example, it may specify various script functions in a longoage such as JAVA Script or 0502, or certain formatting styles (funct paragraph styles, header soles, and so on) that can be used in the document. It can also specify a rule to indicate what the 0.201 file is for, and other similar information that will not be displayed as part of the document.
- The body of the document -specified within the <body> ... </body> rags -includes
  the document text and the numbup rags that specify how the text is to be formatted
  and displayed. It can use include references to other objects, such as images, videos,
  voice messages, and other documents.
- The <H1 < ... <011 > tags specify that the text is to be displayed as a level 1 heading. There are many heading levels (<112>, <145>, and so only each displaying text in a less prominent heading format.
- The <rulde> ... 
   Table> tags specify that the following text is to be displayed as a table. Each row in the roble is enclosed within <TR> ... </TR> tags, and the actual text dots in a row is displayed within <TD> ... </TD> tags?
- Some rags may have artributes, which appear within the start tag and describe additional properties of the rag.<sup>2</sup> In Figure 20.2, the stables start rag has four attributes describing various characteristics of the table. The following <TD> and <Suit> start rags have one and two attributes, respectively.

HTML fais a very large number of predefined tags, and whole books are devoted to describing how to use these tags. If designed properly, HTML documents can be formatted so that humans are able to easily understand the document contents, and are able to compare through the resulting. Web documents. However, the source OTML text documents are corry difficult to interpret automatically by compare projects because they do not include schema information about the type of data in the documents. As ecommence and other intermet accharacter because increasingly automated, it is becoming emitted to be able to exchange. Web documents among various computer sites and to interpret their contents automatically. This need was one of the reasons that led to the development of XML, which we document increase section.

<sup>2.</sup> ATRA stands for table test, and ATDA for table data.

<sup>3</sup> This eshow the acmost of the is used in documentation key languages, which differs from boso at used in database models.

# 26.2 XML HIERARCHICAL (TREE) DATA MODEL

We now introduce the data model used in SML. The basic object is SML in the SML deciment. Two main structuring concerns are used to construct an SML document, elements and attributes. It is important to note right away that the term actibute in SML is not used in the same manner as is customory in database terminology, but rather as it is used in document description languages such as SITML and SOML? Attributes in SML provide additional information that describes elements, as we shall see. There are additional cocepts in SML, such as entrices, identifies, and references, but we first concentrate on describing elements and attributes to show the essence of the SML model.

Figure 26.5 shows an example of ite XML element called <projects2. As in (ITML elements are identified in a document by their start tag and end tag. The tag names are enclosed between angled brackets < 1.2, and end tags are further identified by a backslash,  $<f_{1,1}>$ .<sup>3</sup> Complex elements are constructed from other elements hierarchically, whereas simple elements contain data values. A major difference between XML and (ITML as that XML tag names are defined to describe the meaning of the data elements in the document is to be displayed. This makes it pass for ropposes the data elements in the XML document automatically by computer programs.

In general, it is possible to characterize three main types of XML documents:

- Data-centric XM2 doctoments. These documents have multy small data items that x<sup>3</sup>low a specific structure and hence may be extracted from a structured database. These are formatted as XML documents in order to exchange them or display them over the Web.
- Document centric XM2 disconcerts. These are documents with large amounts of text, such as news innotes or books. There are tow or nu structured data clonents in these documents.
- Hybrid XAQ do synguty: These documents muy have party that contrain structured bia and other party that are predominantly instant or unstructured.

It is important to note that datas entric XML documents can be considered either as semistructured data or as sumetored data. If an XML document conforms to a predefined

<sup>4)</sup> SCM, IStandard Generalized Markor Loognage) is a more general language for deverding deconcents and provides capabilities for specificing new type However, or is more complex three trait and Mar.

<sup>5.</sup> The left and right angled bracket characters (< and >) as re-enved characters, as are the supersmall (<), appartophe (1), and single quotes structures (1). To include them on him the text of ( docnments they must be encoded as &int, > &camp: &cap we and &cquot, respectively.

```
<?xml version="1.0" standalone="yes"?v
<projects>
 cprc)ect>
 <Name>ProductX</Name>
 <Number>l</Numbers
 kLocation>Bellanrek/Location>
 <DeptNo>5</DeptN:>>
 «Worker»
 <$$$N$123456789a/$$N5
 kLastNamepSmithk/LastName>
 vhourss32.5k/hourss
 </⊫orker>
 ckorkers
 <$$%>453453453a/$$No
 <Forsthane>Jovres/ForstName>
 <hours>20_0</hours>
 </Workers
 </project>
 </project>
 <Name>ProductYk/Name>
 <Number>2</Aamber>
 <cotionsSugarlancs/Locations</pre>
 <DeptNo >5</DeptNo >
 dior ters
 <55N>123455789</55N>
 <hours>7.5</hours>
 </Workers
 worker's
 ~55N>453453453</55N>
 «hours>20.0«/hours»
 </www.ker>
 vh0/kero
 <55N>333#455555</55N>
 chourss10.0k/hourss
 </Workers
</projects
. . .
```

</projects>

FIGURE 26.3 A complex xwi element called oprojectss.

XML schemation (PTP) (see Section 26.3), then the document can be considered as structured data. On the other hand, ZVL allows documents that do not conform to any schemat and these would be considered as sentents and data. The latter are also known as schemalass XML dataments. When the value of the substation attribute in an XML document is "ets", as in the first line of Ergore 26.5, the document is statishing and schemaless.

XMU attributes are generally used in a manner similar to how they are used in HTML tree Figure 20.21, nonzely, to describe properties and characteristics of the elements (tags) within which they appear. It is also possible to use XML attributes to hold the values of

simple data elements; however this is definitely not recommended. We discuss XML argubates forther in Section 26.3 when we discuss XML scheme and 1001/

# 26.3 XML DOCUMENTS, DTD, AND XML SCHEMA

### 26.3.1 Well-Formed and Valid XML Documents and XML DTD

In Figure 26.3, we save what a sample XML document may look like. An XML document is well formed if it follows a few conditions. Expatticular, it must start with an XML declaration or undicate the version of XML being used as well as any other televant articlases, as shown in the first line of Figure 26.3.1: must also follow the syntactic guidelines of the tree model. This means that there should be a sogle not efforted, and every element and include a matching puriod start and end tags within the source and end tags of the pure document. This ensures that the effected elements specify a well-formal tree variations.

A well-formed XMI document is syntactically correct. This allows it to be processed by generic processors that inverse the document and create an internal tree representation. A standard set of MP (upplication programming interface) function called DOM (Document Object Medel) allows programs to transpulate the resulting tree representation corresponding to a well-formed XML document. However, the whole document must be parsed beforehand when using 1000. Another API called S W aflows processing at XML documents on the fly by notifying the processing program whenever a start or end rag is encountered. This makes it easier to process have documents and allows for processing of sourcefled streaming XML documents, where the processing program carprocess the togs as they are encountered.

A well-formed AML document con-how any tog names for the elements within the document. There is no predefined set of elements (tag names) that a program processing the document knows to expect. This gives the document creator the freedom to specific new elements, but limits the possibilities for automatically interpreting the elements within the document.

FIGURE 26.4 AD AML DTD file called projects.

A stronger enterior is for an XML document to be valid. In this case, the document must be well formed, and in addition the element names used in the start and end tag pairs must follow the structure specified in a separate XML DTD (Document Type Definition) file or XML schema file. We first discuss XML (0TD here, then give an overview of XML schema in Section 26.3.2. Figure 26.4 shows a simple XML 011) file, which specified there aested structures. Any valid documents continuing to thes (0ff) should follow the specified structures. Any valid documents for specifying DTD: files, as illustrated in Figure 26.4. First, a mapper is given to the root tag of the document, which is called projects in the first bare of Figure 26.4. Then the elements and their networks are the figure 26.4. Then the elements and their networks for a project and the first bare of Figure 26.4. Then the elements and their networks are provided structures and figure 26.4.

When specifying elements, the following notation is used:

- A \* following the element name means that the element can be repeated tero or more times in the document. This kind of element is known as an aptional individual (repeating) element.
- A + following the element name means that the element can be repeated one or undertimes in the document. This kind of element is interpret multicalized (repeating) element.
- A l'following the element noise means that the element can be repeated zero of one times. This kind is an optimizal single valued (nonreporting) clanent.
- An element appearing without any of the preceding three symbols most appear exactly once in the document. This kind is a sequend single-tabled monoperingly element.
- The type of the element is specified via patentheses following the element. If the
  patentheses include names of other elements, these latter elements are the children of
  true element in the tree structure. If the parentheses include the keywood #PCDATA or
  include the other data types available in XML DTD, the element is a leat node. PCDATA
  sounds for generic children data, which is roughly similar to a string data type.
- Parentheses can be neared when specifying elements.
- A bar symbol (c<sub>1</sub> 1 e<sub>2</sub>) specifies that other contraction appear in the document.

We can see that the tree structure in Figure 2n 1 and the SVI socument in Figure 26.5 conform to the XML DFU in Figure 26.4. To require that an XML document be checked for conformance to a 17TD, we must specify due in the declaration of the document. For example, we could change the first line in Figure 26.3 to the following:

```
<?xml version="1.0" standalone="no"?>
<!DOCTYPE projects SYSTEM "proj.dtd">
```

When the value of the standature attribute in an XML document is "nu", the document needs in he checked against a separate 1500 document. The DTV file shown in Figure 26.4 should be stored in the same file system as the XML document, and should be group the file name "projudid". Alernaticely, we could include the DTP document rest at the beginning of the XML document itself to allow the checking.

Although XM, DHU's chite adequate for specifying tree structures with required optional, and repeating elements, it has several limitations. First, the data types in DTD are not very general. Second, DTD has its own special seniors and thus requires specialized processors. It would be advantagenes to specify XMU schema documents using the spots rules of XML itself so that the same processors used for XML documents could process XMU schema descriptions. Third, all DUD elements are always forced to follow the specified ordering of the document, so unordered elements are not permitted. These drawbacks kill to the document of XML schema, a more accord language for specifying the structure and elements of XML documents.

### 26.3.2 XML Schema

The XXII schema language is a standard for specifying the storewre of XXII documents, le uses the same syntax rules is regular XXII documents, so that the same processors can be used on both. To disrigginsh the two types of documents, we will use the term XXII rougher document or XXII document for a regular XXII document, and XXII schema document for a document that specifies an XXII schema. Figure 26.5 shows an XXII schema document there would want to doctave shown in Figures 1.2 and 5.5. Although it is unlikely that we would want to doctlar the whole database as a single document, there have been proposals to store data in native XXII format as at alternative to storing the data in relational databases. The schema in Figure 26.5 mostld serve the purpose of specfying the structure of the courses database of it were stored in a native XXII system. We cascues this topic further in Section 26.4

As with XML 10TD, XML schema is based on the rise data model, with elements and attributes as the main structuring concepts. However, it bottows edditional concepts from

```
«?xml version="1.0" encoding="UTF-8" ?>
<xsd:schema_xmlns:ssd="http://www.w3.org/2001/XMLSchema">
 xxsd:arnotation>
 <xsd:documentation xml:lang="en">Company Schema (Element Approach) -
 Propared by Babak Hojabrie/xsd:documentations
 </ksd:annotations
<ssd:element name="company">
 xsd:complexTypes
 casd: sequence>
 sasdrelement name="department" type="Department" nanOccurs="0"
 naxOccuris="unbounded" />
 k#sd.element name="employee" type="Employee" minOccurs="0"
 max0ccurs="unbounded">
 <xsd:unique_same="dependentManeUnique">;
 <xsd:selector xpath="employeeDependent" />
 vxsd:field xpath="dependentName" />
 </xsd:un:cue>
 </ssd:element>
 exsd:@lement name="project" type="Project" minOdours="0"
 maxOccurs="unbounded" />
 </xsd: sequences
 </xsd:complexTypes
```

HOURE 26.5 An XML schema file called conpany.

```
<xsd unique name="departmentNameUnique">
 vksdcselector xpath="department" />
 kxsd:field xpath="departmentName" />
 </xsd:unique>
 xxsd:uvique_wamew"projectNameUn1que">
 «xsd:selector xpath="project" />
 <xsd:field xpath="projectName" />
 </and tuniques.
 <xsd:key_name="projectNumberKey">
 kxsd:selector xpath="project" />
 exscifield soath="projectNumber" /s
 </asd:keys
 kxsd:kcy_name="departmentWunberKey">
 <xsd:selector xpath="department" />
 case;field spath="departmentMumber" />
 </wsdckey>
 xxsd:kev_name="employeeSSMKey">;
 xxsd;selector xpath*"employee" />
 sdifieid knath="employeeSSN" />
 </asd:kev>
 cvsd.keyref name="departmentManagerSSNKeyRef" refer="employeeSSNKey">
 setselector *pach="department" />
 <xsd:field_xpath="departmentManager5SN" />
 </usd:keyref>
 sxsd:keyref_wave="employeeDepartmentNumberKeyRef"
 refer-"departmentNumberKey">
 «xsd:selector »path="employee" />
 sasd:field xpath="employeeDepartmentNumber" />
 </xsd:keyrefa
 <xsd:keyref pume="employeeSupervisorSSNKeyRef" refor="employeeSSNKey">
 kxsd:selector xpath="employee" />
 kase:field apath="employmeSupervisorSSN" />
 </asd:keynets
 kasd:keyref_name="projectDepartmentMunbarkeyRef"
 refer="departmentNumberKey">
 kxsc:selector spath="project" />
 <xsd:field xpath="projectDepartmentNumber" />
 </xsd:keyref>
 <xsd;koyref hame="projectNorkerSSNKeyRef" refer="gmployagSSNKey">
 <xsd:selector xpath="project/projectMorker" />
 xxsd:field xpath="SSA" />
 </www.sd:keyrefa
 <ssd:kevref name="employeeWorksOpProjectNumberKevRef"</pre>
 refer="projectNumberKey">
 «*sd:selector xpath="employee/employeeNorksOn" />
 xsd:field spath="oronectNumber" />
 </asd.keyref>
</usd:element>
```

FIGURE 26.5(CONTINUED) An XML schema file Called, company.

```
xxsd:complexType name="Department">
 casd: secuences.
 cxsd:element mane-"departmentName" Eype-"xsd:String" />
 casd:clement_mame="departmentNumber" type="aod:string" />
 <>sd:@lenent mane-"departmentManagerSSN" type="wsd:string" />
 «xsdtelement name="departmentManagerStartDate" type="xsd:date" />
 <xsdtelement name="departmentLocation" type="xsd:string"</pre>
 minOccurs-"0" naxOccurs-"unbounded" />
 </xsd:sequences
</www.complexTypes
<xsd:comp?exType name="Employee">
 <xsd; sequence>
 «xsd:element name="employeeName" type="Name" />
 exsd:element name="employeeS5N" type="kad;string" /*
 exsdie ement name="employeeSex" type="xsdistring" />
 sksd:element_name="employee5a]ary" type="xsd:unsignedimt" />
 «xsd:element_name="employeeBirthDate" type="xsd:date" />
 <ssd:e!ement name="employeeDepartmentNumber" type="ksd-string" />
 <sdrelement name="employeeSupervisorSSN" type="xsd;string" />
 <xsd:element_name="employeeAddress"_type="Address" />
 exsd.e?ement name="employeeWorksOn" type="WorksOn" minUccurs="1"
 maxUccurs-Tunbounded" />
 <*sdielement make="employeeDependent" type="Dependent" minUccurs="0"</pre>
 maxOccurs="unbounded" />
 </xsd:sequence>
</wsd.comdlexType>
<ssd complexType name="Protect">
 <0.53; Sequence>
 kxsd:element name="projectName" Lype="xsd:string" />
 kxsd:element mane="projectNumber" type="Xsd:strimg" />
 kxsd:element mane="projectLocation" type="xsd:string" />
 <xsd:element name="projectDepartmentNumber" type="asd:string" />
 cusd:element wane="projectWorker" type="Norker" winOccurs="1"
 naxOddurs="unbounded" />
 </wsd:Sequence>
</xid: complexType>
kksd:complexType name="Dependent"s
 kxsd:sequences
 cxsd:clement mane="dependentName" type="sad:String" />
 cosd:element name="dependentSex" type="asd:string" />
 cosd:clement name-"dependentBirthDate" type="asd:date" />
 <xsd:glemony mane-"dependentRelationship" type-"xsd:string" />
 </xsdisequences
</ksd;complexType>
kasd.complexType name="Address">
 conditionquences.
 «vsd;element name="number" type="vsd:string" />
 <ksd;element mawe="street" type="#sd:string" />;
 swsh:element name="city" type="wsd:string" />
 «ksd:element name="state" type="wsd:string" />
 />sdisenuence>
```

FIGURE 26.5(CONTINUED) An XML schemal file s alled conpany.

```
</xsd:complexType>
 <xsd:complexType name="Name">
 xxsd:snguences
 cxsd:element name="firstName" type="xsd;string" />
 kxsd:element name="middleName" type="xsd:string" />
 <xsd:element name="lastName" type="xsd:string" />
 </wsd;sequencex
 </xsd:complexType>
 <xsd;complexType_name="Worker">
 xxsd:sequence>
 <ksd:element name="SSN" type="xsd:string" />
 «xsd element name="hours" cype="xsd:float" />
 </xsd.sequence>
 </r></rd>complexType>
 «xsd:complexType game#"WorksOn"»
 xsd;sequence>
 «xsd:element name="projectNumber" type="xsd:string" />
 <vsd:element name="hours" type="xsd:float" />
 </xsd. sequence>
 </wsd:complexType>
</usd:schemax
```



database and object models, such as keys references, and identifiers. We here describe the features of XMU schema in a step-by step manner, referring to the example XMU schema document of Figure 26.5 for illustration. We musclose and describe some of the schema concepts in the order in which they are used in Figure 26.5.

- 1 Schema descriptions and XM2 remeasures: It is necessary to identify the specific set of XM1 schema bragoage cleatents (togs) being used by specifying a file stored of a Web site location. The second bine in Figure 26.5 specifies the file used in this example, which is "http://www.w3.org/2001/XMLSchema". This is the most commonly used standard for XM1 schema communds. Each such definition is called an XML namespace. Seconds to defines the set of commands (names) that can be used. The file name is assigned to the variable xsd (XM1 schema description) using the attribute onlits (XM1 schema commands (associate is as prefix to al. XM2 schema commands (tag names), and the variable is used as a prefix to al. XM2 schema commands (tag names). For example, in Figure 26.5, when we write xsdielement or xsdisequence, we are reforming to the definitions of the element and sequence tags as defined in the file "http://www.w3.org/ 2001/XMLSchema".
- 2. Accessions: documentation, and largence used: The next couple of lines in Figure 26.5 illustrate the XML schema elegients (Figs) xsd:annotation and esd:documentation, which are used for providing compensions and other descriptions in the XML documents. The attribute xml large of the systementation element specifies the language being used, where "en" couple for the language being used. Scheme "en" couple for the language being used.

- 5 Element and types. Next, we specify the tour element of our XML schema. In XML schema, the name intribute of the anticelement rog specifies the element name, which is called conpany for the non-element in our example (see Figure 26.5). The structure of the company nor element can then be specified, which is our example is associately proved to be a sectence of departments, employees, and projects using the xsd; sequence structure of XML schema Ir is important to note here that the is not the only way to specify an XML schema for the company database. We will discuss other options in Section 26.4.
- 4. Fast-deceretenents in the corrows database. Next, we specify the three first-lever elements under the company cost element in Figure 20.5. These elements are mixed employee, department, and project, and each is specified in an x8dbellement the Notice that it a tag has only attributes and no further solvelements or data within it, it can be ended with the back-lash symbol. (78) directly instead of having a separate matching end tag. These are called empty elements examples are the x8dbellement clements matched department, and project in Figure 20.5.
- 5. Specifying element reperturbation and maximum accurrences: In XML schemel, the attributest type, minOccurs, and maxOccurs in the associatement tog specify daily period multiplicity of each element in any document that contains to the schemelistic of the element must be described separately, typically using the associatement, and project elements in Figure 26.5. On the other location type attribute is specified, the element is tracture can be demonially following the tag, as illustrated by the company non-element in Figure 20.5. The minOccurs and maxOccurs tags are used for specifying lower and upper bounds on the number of occurrences of an element in any document that soutoms to the scheme specified, the accurent that soutoms to the scheme specified of a specifying lower and upper bounds on the number of occurrences of an element in any document that soutoms to the scheme specified, the document that soutoms to the scheme specified are not specifying lower and upper bounds on the number of occurrences of an element in any document that soutoms to the scheme specified. The variable of symbols of XML PTD, and to the (non-max) constraints of the F8 model (see Section X74).
- to Specifying keys. In XML schema, it is possible to specify constraints that correspond to unique and primary key constraints in a relational database (see Section 5.2.2). as well as foreign keys for referenced integrityl constraints (see Section 5.2.4). The usdrumique my specifies elements that correspond to imagic attributes in a relational database that are not primary keys. We can give each such anguaresconstraint a name, and we must specify x\$1:5e16Ctor, and xx0: field tags to it to identify the element type that contains the numpic element and the element many within at that is amout on the spath attribute. This is clustrated by the departmentNameUnique and projectNameUnique clouents in Figure 26.5. For spontying primary keys, the sugastickey is used instead of astronomical solution traced by the projectNumberKey, departmentNumberKey, and employeeSShKey elements in Figure 26.5. For spontying foreign keys, the rag ostickey reflix and, as illustrated by the ascentickey reficience in Figure 20.5. When specifying t torrigh key, the attribute referil of the and segref rag spontics the addressed primary kee, whereas the may as disePertur and as difficill specify the reference ing element type and foreign key (see Figure 30.5).

- 7. Speedyag) the structures of contrient elements we complex type: The next part of our estatisple specifies the structures of the complex elements Department, Employee, Project, and Dependent, using the tag xisticomplexType (see Figure 26.5). We specify each of these as a sequence of subelements corresponding to the database attributes of each entity type (see Figure 3.2 and 5.7) by using the xisticemente and xis celement tags of XMI schemes. Each element is given a name and type of xisticement, Wilcom also specify minOccurs and nasOccurs attributes of we need to change the default of exactly one occurrence. For (optional) database attributes where null is allowed, we need to specify minOccurs = 6, whereas for unditrobaed database attributes we need to specify mixOccurs. = 6, whereas for unditrobaed database attributes we need to specify not point of subelements where null is allowed, we need to specify not point of years for unditrobaed database attributes we need to specify not point of years for unditrobaed database attributes we need to specify not point of the corresponding element. Notice that if we were not point if years the unditrobaed database attributes using the subelements within the parent element definitions directly without having to specify complex types. However, when unique, primary key, and foreign key constraints need to be specified, we must define complex types to specify the element structures.
- 8 Compound towapound) armbure: Composite intributes from Figure 3.2 are also specified as complex specific figure 26.5, as if usually the Address. Name, worker, and worksOn complex types. These could have been directly embedded within their parent elements.

This example a lastrates some of the near features of XML schemo. There are other teatures, but they are beyond the scope of our presentation. In the next section, we discuss the different approaches to creating XML documents from relational databases and storing XML documents.

# 26.4 XML DOCUMENTS AND DATABASES

We now discuss how various types of XML documents can be stered and retrieved. Section 26.4.1 gives an everynew of the various approxithes for storing XML documents. Section 26.4.2 discusses one of these approaches, or which data-contrict XML documents are extracted from existing databases, in more detail, by curricular, we show how true structured documents can be crosted from graph structure to red databases. Section 26.4.3 documents the problem of cycles and how it can be dealt with

### 26.4.1 Approaches to Storing XML Documents

Several approaches to organizing the contents of Wit documents to fac littue their subsequent querying and retrieval have been groposed. The following are the most common approaches

 Using a DWGS to store the doctments as fewer A relational or object DWGS can be used to store whole XML documents as text fields within the DBWS records or objects. This approach could be used if the OBMS has a special module for document processing, and would work for storing schemaless and document-centric XML. documents. The keyword indexing functions of the document processing module (see Chapter 22) can be used to index and speed up search and retrieval of the document-

- 2. Using a DBMS to some the document contents as data elements. This approach would work for storing a collection of documents that follow a specific XML 07D or XML achiever. Because all the documents have the same structure, one can design a relational (or object) database to store the lead level data elements within the XML documents. This approach would require mapping algorithms to design a database schema that is compatible with the XML documents from the same structure as questied in the XML schema or 1000 and to recreate the VML documents from the sone data. These algorithms can be implemented either us an internal DBMS module or as separate middleware that is not part of the 2005.
- V Designing a specialized system for soring radius NML data: A new type of database system based on the literarchical (tree) model could be designed and implemented. The sestem would include specialized indexing and querying techniques, and would work for all types of XML documents. It could also include data compression techniques to reduce the size of the documents for atorage.
- 4 Creating or publishing customated XML decrements from precisiting relational database Because there are enformed a mounts of data already stored in relational databases, parts of this data may need to be tormatted as documents for exchanging or displaying over the Web. This approach would use a separate middlescare service layer to handle the convertions provided between the XML documents and the relational database.

All four of these approaches have received considerable attention over the past for years. We focus on approach 4 in the next subsection, because it gives a good conceptual understanding of the differences between the X60, tree data model and the madrianal database models based on that files (relational model) and graph, representations (66 model).

### 26.4.2 Extracting XML Documents from Relational Databases

This section discusses the representational usages that arise when genverting data from a database system into XMI documents. As we have discussed XMI uses a hierarchical (rree) model to represent documents. The database systems with the most widestread as follow the flat relational data model. When we add referential integrity constraints a relational schema can be considered to be a graph structure (for example, see Figure 3.7). Similarly, the ER model represents data using etaphlike structures (for example, see Figure 3.7). We saw in Chapter 7 that there are straightforward mappings between the 3.8 and relational models, so we can conceptually represent a relational database schema mag the corresponding ER schema. Although we will use the ER model in our decision and examples to clarify the conceptual data references between the BR models, the same issues apply to converting relational data to XML.

We will use the singlifued interestion 56 schema shown in Figure 26.0 to illustrate our discussion. Suppose that on application needs to extract 2005 documents for student, course, and grade information from the interestivation of the entry lapped that needed for these documents is contained in the database attributes of the entry types (00857, 557-106, and 55.0667 from Figure 26.6, and the relationships 5-5 and (-5 between them, 26 general, most documents extracted from a dotabase will only use a subart of the attributes, entry types, and relationships in the database. In this example, the subset of the database that is needed is shown in Figure 26.7.



FIGURE 26.6 An CR schema diagram for a simplified assessmentatabase.



FIGURE 26.7 Subset of the care extra database schema needed for XML document extraction.

At least three possible document hierarchies can be extracted from the datases subset to hence 25.7. Fost, we can choose cross as the toot, as illustrated in Figure 26.8. Here, each course entity has the set of its sections as subelements, and each section has its students as subelements. We can see one consequence of modeling the information in a hierarchical tree structure. If a student has taken multiple sections, that student's automation will appear multiple times in the document—once under each soution. A possible simplified XML schema for this view is shown in Figure 26.9. The Grade database attribute in the \$-5 relationship is anguared to the \$1000\* element. This is because \$1.60\* becomes a child of section in this hierarchy, so each strucelement inder a specific section clement can have a specific grade in that section, la this document hierarchy, a student taking innire than one section will have several replices, one under each section, and each replica will have the specific grade given in that your plant section.

In the second hierarchical document may, we can choose scotter as root (Figure 2017). In this hierarchical view, each student has a set of sections as its child elements and each section is related to one course as its child, because the relationship between statics and course is N(1). We can hence merge the course and section elements in this



FIGURE 26.8 Hierarchical (tree) view with course as the mot.

```
systimized energy names "moot">
axsd: sequences
<asc:elenent names"course" minOccurs="0" makOccurs="unbounded">
 xxsd.sequence>
 vxsd.element_name="cname" type="xsd:string" />
 cxsd:element_name="cmumber"_type="xsd:unsignedInt" />
 <xsd:elevent_name="section" ninCccurs="0" naxOccurs="unbounded">
 xxsd:sequences
 <xsd:element name="secondeber" type="xsd:unsignedInt" />
 xxsd:element name="year" type="asd:string" />
 exsd:element name-"quarter" type-"ssd:string" />
 <xsd:element mame="student" minOccurs="D" matOccurs="unbounded">
 kxsd:sequences
 <xsp:element name="ssn" type="xsd:string" />
 kxsd:element_mane="sname" type="xsd:string" />
 osu:element name="class" type="xsd:string" />
 xxsc:clement name="grade" type="xxd:string" />
 </xsc:sequences
 </xsd:element>
 </xsd:sequence>
 «/ksd:element»
 </xsd:sequence>
</asd:clements
</xsd;senuence>
```

FIGURE 26.9 IAM schema document with coase as the mot-

view, as shown in Figure 26.10. In addition, the conductive attribute can be inigrated to the sector element. In this hierarchy, the combined coast/section information is replicated under each student who completed the section. A possible samplified SML schema for this view is shown in Figure 26.11.

The third possible way is to choose section as the root, as shown in Figure 20.12. Similar to the second hierarchical view, the coose information can be morged into the section element. The same database attribute can be migrated to the student element. As we can see, even in this simple example, there can be minimum bierarchical document views, each corresponding to a different mut and a different XML document structure.

### 26.4.3 Breaking Cycles to Convert Graphs into Trees

fre the previous examples, the subset of the database of interest bad no cycles. It is pooble to have a more complex other with one or more cycles, indicating indicide relationships among the entries. In this case, it is more complex to decide how no create the document hierarchies. Additional duplication of entries may be needed to represent the multiple relationships. We shall illustrate this with an example using the FR schema in Figure 26.6.



PIGURE 26.10 Hierarchical Proves view with subject as the mot.

Suppose that we need the miorination in all the entity types and relation-hips of Figure 26.6 for a particular XM2 document, with stocket as the zoot element. Figure 26.13 illustrates how a possible literarchical free structure can be created for this document. First, we get a lattice with student as the more as shown in part (1) of Figure 26.13. This is not a free structure because of the cycles. One way to break the cycles is no replicate the entity types mendeed in the cycles. First, we replicate tastraction as shown in part (2) of Figure 26.13. Unling the replica to the replicate tastraction as shown in part (2) of Figure 26.13. Unling the replica to the replicate tastraction as shown in part (2) of Figure 26.13. Unling the replica to the replication and the sections they reach, whereas the instruction instructions and the sections they reach, whereas the instruction on the right represents the relationship between instructures and the occle involving (006), so we can replicate count in a similar manner, leading to the heaterche shown in part (3) of Figure 20.13. The Gregical replication the left represents the relationship between courses and their sections, whereas the course in part (3) of Figure 20.13. The Gregical replication the left represents the relationship between courses and their sections, whereas the course replicates the course and their sections whereas the course and their sections whereas the course and their sections whereas the course and the optication between courses and their sections.

In part (3) of Figure 26-13, we have converted the annial graph-to a biefarthy. We can do further merging if desired this in our previous example) before creating the final hierarchy and the corresponding Matschemp structure.

```
«xsd:element name="ront"»
«xsd:seguence»
exsd:eiement name="student" minOccurs="0" maxOccurs="unbounded">
 xxsd: sequences
 <xsd:element name="ssn" type="xsd:string" />
 **sd.element name="sname" type="#sd:string" />
 <xsd:element name="class" type="xsd:string" />
 <sd:element name="section" monOccurs="0" manOccurs="unbounded">
 exsd.sequence>
 sd:element name="secnuaber" type="xsd:unsignedInt" />
 xsd:element mame="year" type="xsd:string" />
 «xsd:element name="quarter" type="asd:string" />
 xsd:element mane="chumber" type="asd:unsignedInt" />
 xxd:element wawe="cnawe" type="xsd:string" />
 <xsd:element mame="grade" type="xsd:string" />
 </ASB: sequence>
 </xsd:e)ements
 </ksdcsequences
</xsduelenents
</ksdcsequence>
</xsd:element>
```

```
FIGURE 26.11 was schema document with some as the root.
```



FIGURE 26.12 Frierarchinal prentyview with secrets as the root


FIGURE 26.13 Converting a graph with cycles into a bierarchical (tree) structure.

### 26.4.4 Other Steps for Extracting XML Documents from Databases

In addition to creating the appropriate XML hierarchy and corresponding XML scherics donment, several other steps are needed to extract a particular XML dominient from a database

- It is necessary to create the correct query in SQL to extract the desired information for the SQL documents.
- Grace the query is executed, us result must be structured from the flue relational form to the SM , tree structure.
- 3 The query can be customized to select other a single object or nulriple objects into the document. For example, in the view of Figure 26.73, the query can select a single student entity and create a document corresponding to that single student, or it may select several—or even all of—the students and create a document with multiple students.

# 26.5 XML QUERYING

There have been several proposals for XME query longuages, but two standards have emerged. The first is XFath, which provides language constructs for specifying yall expressions to identify certain nodes (elements) within an XME document that match specirc patterns. The second is XQuery, which is a more general query language. XQuery uses XPath expressions but has additional constructs. We give an overview of each of these languages in this section.

### 26.5.1 XPath: Specifying Path Expressions in XML

An XPush expression returns a collection of element codes that satisfy certain porterns specified in the expression. The names in the XPath expression are node names in the XVI document tree that are other tag (element) names or articlate names, possibly with addictional qualifier conditions to further restrict the nodes that satisfy the pattern. Two main separators are used when specifying a path single slash (f) and double slash (f). A single slash before a tap specifies that the tag must appear as a descendant of the previous (gas an appear as a descendant of the previous tag at an above level. Let us look at some examples of XPath as shown in Figure 20.14.

The first XPath expression in Figure 26.14 remarks the company root node and all its descendant nodes, which means that it returns the whole XMI document. We should care that it is customary to include the file name in the XPath query. This allows us to specify any local file nome or even any path name that specifies a file on the Web. For example, it the custom XMI document is stored at the location.

www.company.com/info.xml

then the first XPath expression at Fighte 26-14 can be written as

doc(www.company.com/info.kml)/company

This prefix would also be included in the other examples.

The accord comple in Figure 26.14 in turns all department nodes (elements) and their descendant subtrees. None that the nodes (elements) in an Xet, document are ordered, so the XPath result that returns multiple nodes will do so in the same order in which the nodes are ordered in the document tree.

The third XPath expression in Figure 20.14 illustrates the ase of  $K_{\rm c}$  which is convenient to use if we do not know the full path name we are searching for the do know the name of some tass of interest within the XML document. This is particularly useful for schemaless XML documents or for documents with many nested levels of nodes.<sup>6</sup> The

- 1. /company
- /company/department
- 3. //employee [employeeSalary gt 70000]/employeeName
- 4. /company/employee [employeeSalary gt 70000]/employeeName
- /company/project/projectWorker (hours ge 20.0]

FIGURE 26.14 Some examples of XPath expressions on XML documents that follow: the XML schema file convexy in Figure 26.5.

6. We are using the terms made tog, and element interchangeably here.

expression returns all enployeeName nodes that are direct children of an enployee node, such that the enployee node has another child element enployeeSalaxy whose value is greater than 73032. This illustrates the use of qualitier conditions, which restrict the nodes selected by the XPath expression to those that satisfy the condition. XPath has a number of enorganison operations for use in qualifier conditions, including standard arithmetic, string, and sets onights on operations.

The fourth XDath expression should return the same result as the previous one, except that we specified the full path came in this example. The fifth expression in Figure 26.19 netures all project Monker nodes and their descendant nodes that are children under a path /company/project and have which node hours with a value greater than 20 0 hours

### 26.5.2 XQuery: Specifying Queries in XML

XPath allows us to write expressions that select nodes from a free-structured XML document. XQuere permuts the specification of more general queries on one or more XML docments. The typical form of a query in XQuery is known as a H.WR expression, which stands for the four many choices of XQuery and has the following form:

```
FOR «variable bindings to individual modes (elements)»
LET «variable bindings (o collections of modes (elements)»
WHERE «coalifier conditions»
&FTURN «query result specification»
```

Figure 26.15 includes some examples of queries in XQuery that can be specified on AML network documents that follow the XMI schema document in Figure 26.5. The first query retrieves the first and last names of employees who carrindoc than \$72,000. The

```
1. FUR %x IN
 doc(www.company.com/info.xml)
 //employee (employeeSalary gt 70000]/employeeName
 RETURN knes> 3x/linstName, 3x/lastName k/res>
FOR $x IN
 doc(www.company.com/infe.xml)/company/employee
 WHERE $x/employeeSalary gt 70000
 RETURN knesk 3x/employeeName/firstName,
 $x/rmoloyenName/lastName </resa
3. FOR $x IN
 doc(www.company.com/info.xml)/company
 /project[projectNumber = 5]/projectWorker.
 γγ 1N
 doc(www.company.com/info.wml)/company/employee
 WHERE $x/hours gt 20.0 AND $y.ssn = $x.ssn
 RETURN kres> Sy/employeeName/firstName,
 Sy/employeeKane/lastName, Sk/hours k/resv
```

**FIGURE 26.15** Some examples of XQuery queries on XML documents that follow the XML scheme tile corever in Figure 26.5.

vaniable **5**c is bound to each exployeeName clement that is a child of an employee element, but only for employee clements that satisfy the qualitar that then employeeSalary value is greater than \$70,000. The result retrieves the Firstmane and lastmane child elements of the selected employeeName elements. The second energy is an alternative way of a tracking the same elements retrieved by the first query.

The third query illustrates how a join operation can be performed by having more than one variable. Here, the **b**s variable is bound to each project worker element that is a child of project number 5, whereas the **s**y variable is bound to each **exployee** element. The poin condition matches SSN values in order to retrieve the employee names.

This concludes our brief annodaction to XQuery. The interested reader is referred to the Web site www.w3.org, which contours documents describing the latest standards related to NML.

### 26.6 SUMMARY

This chapter gave an overview of the standard for representing and exchanging data over the Internet. We started by discussing the differences between structured, sensitive and and unstructured data, then discussed why there was a need for a specification language such as XML. We described the XML standard and its mes-structured (hierarchical) data insidel, and discussed XML documents and the languages for specifying the structure of these documents, namely, XML (XC) (Decoment, Type Definition) and XML scheme. We then gave an overview of the various approaches for storing XML documents, whether in their native flexification in a compressed form, or in relational indicate types of databases, and discussed the mapping issues that arise when there is need to convert databases, and discussed the mapping issues that arise when there is need to convert databases, and discussed the mapping issues that arise when there is need to convert databases, and discussed the mapping issues that arise when there is need to convert databases, and discussed the mapping issues that arise when there is need to convert datastrend in traditional databases into XML documents. Finally, we gave an overview of the XFath and XQuery languages proposed for querying XML data.

### **Review Questions**

- 26.1. What are the differences between structured, semistractured, and unstructured data?
- 26.2. Under which of the above categories do XMP documents fall? What about selfdescribing data?
- 26.5. What are the differences between the use of tags in XML versus #CML\*
- 26.4 What is the difference between doto-centric and document-centric XML documents:
- 26.5 What is the difference between autobutes and elements in MML/List some of the important attributes used at specifying elements in MML schema.
- 26.6 What is the difference between XML schema and XML DTD<sup>3</sup>

### Exercises

- 26.7 Create an XML instance document to correspond to the data stored in the relational database shown in Figure 5.6 such that the XML document conforms to the XML schema document in Figure 26.5.
- Cruate XML schemes documents to correspond to the hierarchies shown in Figures 26.12 and 26.13 pair (3).
- 26.9. Consider the creater relational database scheme of Figure 5.20. Create an 800, science document that corresponds to this database scheme.
- 26.10. Specify the following move as queries in XQuery on the Ordam XML schema shown in Figure 26.5.
  - A view that has the department name, manager name, and manager solars for every department.
  - b. A view that has the employee name, appendisor name, and employee solars for each employee who works in the Research department.
  - c) A view that has the project name, controlling deparament name, number of employees, and total hours worked per week on the project for each project.
  - d. A view that has the project name, controlline deportment name, puraber of employee-, and total hours worked per week on the project for each project with more than one employee working on it.

### Selected Bibliography

There are so many articles and books on various aspects of XML that it would be impossible to make even a modest lost. We will menuion one book. Choudbro Rushid, and Zicaneds (2003). This book discusses carrous aspects of XML and contains a list of some recent references to XML research and practice.



# Data Mining Concepts

Over the last three decades, many organizations have generated a hitge analogic of nucline-readable data in the term of files and databases. To process this data, we have the database technology available that supports query languages like 500. The problem with 500 is that at is a structured language that assumes the user is aware of the database schema. SQL supports operations of relational algebra that allow a user to select most and columns of data from tables or join related information from tables based on common fields. In the next chapter, we shall see that data networketing technology attends several types of functionality: that of copyolidation, aggregation, and summarization of data. Data watchenses let us view the some information along multiple dimensions. In this chapter, we will fix to our attention on another very popular area of interest known as data min inter As the term contrates, data mining refers to the mining or docuvery of new information in terms of putterns or rules from vast amounts of databases. To date, it is not wellinterest be carried out efficiently or, large files and databases. To date, it is not wellinterested with database management systems.

We will briefly review the state of the art of this rather extensive field of data mining, which uses rechinques from such areas as machine learning, statistics, neural networks, and generic algorithms. We will highlight the nature of the information that as discovered, the types of problems fixed when trying to mine databases, and the types of applications of data mining. We also survey, the state of the art of a large number of commercial tools available (see Section 20.2.5) and describe a number of research advances that are needed to make this area viable.

## 27.1 OVERVIEW OF DATA MINING TECHNOLOGY

In reports such as the very popular Gartner Report.<sup>1</sup> data annung has been harled as one of the rop reclinicle gies for the near luture. In this section we colore data mining to the breader area called knowledge discovery and contrast the two by means of an illipitative example.

Data Mining versus Data Warehousing. The goal of a data wirehouse (see Usayrer 28) is to support decision traking with data. Data mining can be used in comportion with a data warehouse to hely with certain types of decisions. Data mining can be applied to operational databases with individual transactions. To make data mining more efficient, the data warehouse should have an aggregated or summaried cofficient of data. Data mining helps in extracting normaniful new patterns that cannot be found necessarily by merely spectrum or processing data or metadata in the data wirehouse. Data mining applications should therefore be strongly considered early duting the design of a data warehouse. Also, data mining tools should be designed to facilitate their use in comparction with data warehouses. In fact, for wety large databases mining mito terabytes of data, successful use of data mining applications will depend that or the construction of a data is accessful use of data mining applications will depend that or the construction of a data warehouse.

Data Mining as a Part of the Knowledge Discovery Process. Knowledge Discovery in **Databases**, frequently abbreviated as **NDD**, typically encompasses more than data mining. The knowledge discovery process comprises six phases. Data selection, data cleanong, enrichment, data transformation or encoding, data mining, and the reporting and display of the discovered information.

As an example, consider a transaction database manifolded by a specialty consumer goods for older. Suppose the client data includes a customer name, up code, phore nameer, date of purchase, item code, price, quartity and total inform. A variety of row knowledge can be decovered by KDD processing on this client database. During data adjustant, data about specific items or categories of items, in from stores in a specific region or area of the country, may be selected. The data chorong process them may correct invalid tip codes or eliminate records with incorrect phone proteets them may correct invalid tip codes or eliminate records with incorrect phone protects. Environce: up callenhances the data with additional sources of information. For example, gives the cherr names and phone nombers, the store may putchase other data about age, income, and credit rating and append them to each tecord. Data reoisformation and encoding may be slone to reduce the amount of data for instance, item codes may be enoughd in terms of product categories into and/o, video, supplies, electronic gadgets, coment, accession and so on. Zip codes may be aggregated into geographic regions, maximum accession to the data store. In Figure 28.1, we will show a step colled clining as a precursor to the

The Output Report is one example of the many rechtrology sorvey publications (b3) response managers toly on to make the ritechnology schemen decisions.

This discussion is largely pased on Admann and Zurange (1996).

data watchouse creation. If data minime is based on an existing watchouse for this retail store chain, we would expect that the cleaning has already been applied. It is end, after such preprocessing that data moting techniques are used to time different tules and patterns.

The result of mining may be to discover the following type of "new" information:

- Association rules—for example, whenever a customer buys video equipment, he or she also buys another electronic gadger.
- b. Sequential patterns—for example, suppose a customer bays a camera, and within three months be or she bays photographic supplies, then within six months he is highly to have an accessery tion. This defines a sequential pattern of transactions. A customer who have more than twice in the loop periods may be likely to buy at least once during the Christmas period.
- c. Classification recommendates customers may be classified by frequency of visits, by types of financing used, by auto introf purchase, or by athinty, for types of trends, and some revealing statistics may be generated for such classes.

We can see that many possibilities exist for discovering new knowledge about buying patterns, relating factors such as age, meonic group, place of residence, to what such hey much the customers purchase. This information can then be utilized to plan additional store forations based on demographics, to run store promotions, to combine items in advertisements, or to plan wascoal marketing strategies. As this retail store example shows, data mining must be preceded by significant data preparation before it can yield useful information that can directly influence biarros decisiers.

The results of data mining may be reported in a variety of formats, such as listings, graphic outputs, summary tables, or visualizations.

Graals of Data Mining and Knowledge Discovery. Due mining is typically carried out with some end goals or applications. Broadle speaking, these goals fall unit the following classes, prediction, identification, classification, and optimization.

- Prediction—Data mining can show how certain artributes within the data will behave in the future. Examples of productive data mining include the analysis of boyang transactions to predict what consumers will buy order certain discounts, how much sales volume a store would generate in a given period, and whether deleting a product line would yield more profits. In such applications, business logic is used coupled with data timining. In a scientific context, certain setsing wave potterns may preduct an earthquake with high probability.
- Identification—Data patterns can be used to identify the existence of an item, an event, or an activity. For example, intruders trying to break a system may be identified by the programs executed, files accessed, and CTU true per session. In biological applications, existence of a gene may be identified by certain sequences of nucleonde symbols in the DNA sequence. The area known as godientification, is a form of identification, in ascertains whether a user is indeed a specific user or one from an uthorized class, and involves a comparison of parameters or jugges or signals against a database.

- Classification—Data mining can partition the data so that different classes or categories can be identified based on conducations of parameters. For example, customers in a supermarket can be categoried into discount-seeking shoppers, shoppers in a rosh, local regular shoppers, shoppers attached to name brands, and intrequent shoppers. This classification may be used in different analyses of customer breng transcettors as a post priming activity. Somerimes classification based on common domain knowledge is used as an input to decompose the mining problem and make it simpler. For instance, health foods, porty finds, er school bandt foods are distinct categories in the supermatiket biopess. It makes serve to analyze relationships within and across categories as separate problems. Each categorization may be used to reacted in education appropriately before subjecting in to further short categories.
- Optimization—One eventual goal of data juming may be to optimize the use of limited resources such as time, space, money or materials and to maganize extra ables such as sales or profits under a given set of constraints. As such, this goal of cata moning resembles the objective binction used in operations research profilems that deals with optimizations under constraints.

The term data mining is popularly being used in a very broad serve. In while structions it includes staristical analysis and constrained optimization as well as nachine learning. There is no sharp line separating data numing from these disciplines. It is beyond our scope, therefore, to discuss an detail the struct targe of applications that make op the case body of work. For a detailed understanding of the area, readers are referred to specialized books devoted its data mining.

Types of Knowledge Discovered During Data Mining — The rem "knowledge" is very broudly interpreted is involving some degree of intelligence. There is a progression from taw data to intermation to knowledge as we go through additional processing Knowledge is eiten clustified as inductive versus deductive. Deductive knowledge deduces new information based on applying (respectfee) legis at rates of deduction on the groundate. Data mining addresses inductive knowledge, which docrovers new tubes and patterns from the supplied data. Knowledge can be represented in mark forms. In an instructured sense, it can be represented by tules or propositional logic. In a structured form, it may be represented in decision frees, semantic metworks, including therarchies of classes or frames. It is common to describe the knowledge discovered during data mining inframes, as follows.

- Association rules—These rules conclute the presence of a second neuros with another range of values for another second variables. Examples: (1) When a female result shopper basis a haralbag, she is Ekely to buy shoes, (2) An Xzray image comming charactensistics a and bus likely ratalsa exhibit characteristics.
- Classification hierarchies—The goal is to work from an existing set of complex transactions to create a factority of classes, beampless (1) A population may be divided into five range- of credit worthiness based on a history of previous credit transactions (2) A model may be developed for the factors that determine the desiability of location of a store on a 1-10 scale. (1) Matcol funds may be classified based on performance data using characterizers such as growth, income, and stability

- Sequential patterns—A sequence of activity or events is straight. Example: If a patient underwent cardinal hypots suggers for blocked arteries and an aneurysin and later developed high block inter within a year of suggery, he or she is likely to safer from kidney tailore within the next 15 months. Detection of sequential patterns is equivalent to detecting associations among events with certain temporal relationships.
- Patterns within time series. -Similarities can be detected within positions of a time series of data, which is a sequence of data taken at resular intervals such as daily sales or daily closing stock prices. Examples: (1) Stocks of a unibity company. ABC Power, and a financial company. XYZ Securities, showed the same pattern, during 2002 in terms of closing stock price. (2) Two products show the same selling pattern in sumnier bout different one in a uncer. (3) A pattern in solar magnetic wind may be used to predict changes in earth atmospheric conditions.
- Clustering—A given population of events or stems can be partitioned (segmented) into-sets of "autilat" elements. Examples (1) An entire population of treatment data or a disease may be divided into groups based on the similarity of side effects produced. (2) The talult population in the United States may be caregorized into five groups from "most likely to buy" to "least likely to buy" a new product, (3) The web accesses incide by a collection of users against a set of documents (say, in a digital library) may be analyted in terms of the keywords of documents to reveal clusters or untegories of users.

For most applications, the desired knowledge is a combination of the above types. We expand on each of the above knowledge types in the following sections.

### 27.2 Association Rules

#### 27.2.1 Market-Basket Model, Support, and Confidence

One of the major technologies in data turning involves the discovery of association rules. The database is reparted as a collection of transactions, each mealwing a set of iteries. A communic completies that of **market-basket data**. Here the market basket corresponds to the sets of iteries a constituer bask in a supermarket during one visit. Consider that such transactions in a random sample shown in Figure 25.1.

An association rule is of the form  $X = Y_x$  where  $X = \{x_1, x_2, ..., x_n\}$ , and  $Y = i x_1$ ,  $y_1, ..., y_n$  have sets of items, with  $x_i$  and  $x_i$  being distinct items for all *i* and all *j*. This association stores that if a customer bays  $X_x$  be or she is also likely to bay  $Y_i$  by general, any association rule has the form LHS (left hand side)  $\rightarrow$  RHS (right-hand side), where LHS and RHS are sets of items. The set  $z_i$ HS  $\cup$  RHS is called an itemset, the set of items purchased by customers, for an association rule to be of interest to a data inner, the rule should satisfy some interest mansure. Two common interest measures are suggest and confidence.

The support for a rule LHS +> RHS is with respect to the incluse() it refers to how iroquantly a specific iteraset occurs in the database. That is, the support is the precentage

| Transaction-id | Тітте | Items-Bought                |
|----------------|-------|-----------------------------|
| 101            | 6:35  | milk, bread, cookies, juice |
| 792            | 7:38  | mik, juice                  |
| 7130           | 8.05  | milk eggs                   |
| 1735           | B:40  | bread, cook es colfee       |

FIGURE 27.1 Example transactions in market-basket model

of transactions that contain all of the items in the itemset, UHS  $\cup$  RHS. If the support is low, it implies that there is no overschelming evidence that items in LHS  $\cup$  RHS occur together, because the itemset occurs in only a small fraction of transactions. Another items for support is precalence of the rule.

The confidence is with regard to the implication shows in the rule. The confidence of the rule LHS  $\rightarrow$  RHS is computed as the support LHS  $\cup$  RHS/support (LHS). We can thus of it as the probability that the terms in RHS will be proclassed given that the items in LHS are putchased by a customer. Acsolver term for confidence is strength of the rule.

As an example of support and confidence, consider the following two rules. Milk  $\rightarrow$  Jace and Bread  $\rightarrow$  Jace. Looking at our iour sample transactions in Figure 27.1, we we that the support of {Milk, junc} is 50% and the support of (Bread Junc) is only 25%. The confidence of Milk  $\rightarrow$  Jace is 66.7% (meaning that, et three transactions in which milk access (we contain junc) and the confidence of Dread  $\rightarrow$  Jace is 50% (meaning that, et three transactions in which milk access (we contain junc) and the confidence of Dread  $\rightarrow$  Jace is 50% (meaning that or transactions contains) provide the confidence of the support of the support of the access (we contain junc) and the confidence of Dread  $\rightarrow$  Jace is 50% (meaning the support of the super of the support of the support

As we can see, support and confidence do not necessarily go hand in hand. The god of mixing association rules, then, is to generate all possible rules that exceed some minimum inser-specified support and confidence thresholds. The problem is thus decompased into two subproblems:

- a. Generate all iterators that have a support that exceeds the threshold. These sets of items are called large (or frequence) itemsets. Note that large here incurs large support.
- 5: For each large itemser, all the rules that have a numbrum confidence are generated as follows: For a large itemser X and Y ⊂ X, let Z ≤ X = Y; then if support(X)/support(Z) > nummum confidence, the rule Z => Y (that :-, X = Y => Y) is a solid rule.

Occertatize rules by using all large itemsets and their supports is relatively stronghtforward However, discovering all large itemsets together with the value for their support is a major problem of the cardinality of the set of items is very high. A typical supermarket has thousands of items. The number of distinct itemsets is 2% where wils the number of items, and counting support for all possible itemsets becomes very computation-interview. To reduce the combinatorial search space, algorithms for finding association rules utilize the following properties:

- A subset of a large itemset must also be large (that is each subset of a large itemset exceeds the minumum required support).
- Conversely, a superset of a small itemset is also small fimplying that of does not have enough support).

The first property is referred to as **downward** closure. The second property, called the antimenotonicity property, helps in reducing the search space of possible relations. Thus is, once an iteraset is found to be small (not a large iteraset), then any extension to that iteraset, formed by adding one or more iteras to the set, will also yield a small iteraset.

### 27.2.2 Apriori Algorithm

The first algorithm in use the downward closing and autonomously properties was the Apriori algorithm, shown as Algorithm 27.1.

Algorithm 27.4: Aprior, algorithm for finding frequent (large) nerisets

Input: detabase of a transactions, D, and a functional support, most represented as a fraction of m

Output: frequent, industry,  $L_1, L_2, \dots, L_k$ 

Begin

compute supports j = count(i)/m for each individual item,  $i_1, i_2, ..., i_n$  by scanning the database since and counting the number of transactions that item  $i_i$  appears in (that is, count(i)).

the condidate frequent 1-itemset,  $C_{10}$  will be the set of items  $i_1, i_2, \dots, i_n$ 

the subset of items containing it from  $C_1$  where support  $f_0$  >= mins, becomes the frequent

```
Coremset, Lp.
```

```
k = 1;
```

```
termination = false;
```

repeat

4<sub>00</sub> = 113

create the condidate import. (k+1) itemset,  $U_{k+1}$  by combining members of  $L_k$  that have k-1 items in common; (thus forms condidate frequent: (k+1) accusets by selectively extending frequent: k-network by two items).

in addition, only consider as elements of  $C_{k+1}$  those k+1 items such that every subset of size k appears on  $L_k$ .

scan the database once and compute the support for each member of  $C_{con}$  if the support for a member of  $C_{con} >=$  norm, then odd that member to  $L_{con}$ .

if  $\Sigma_{i,i,1}$  is empty then terminotion = true

 $e_{i}^{k} = k + 1;$ 

until termination;

End:

We different Algorithm 27.1 using the transaction data in Figure 27.1 using as mithinum support of 0.5. The candidate lancinsets are [milk, bread, junce, cookies, eggs coffee] and their respective supports are 0.75. 0.5, 0.5, 0.5, 0.25 and 0.25. The first four items spatial for  $L_1$  since each support is greater than or equal to 0.5. In the first heration of the repeat-loop, we except the frequent determines to create the candidate frequent  $\lambda$  itemsets  $U_{2}$ .  $U_{2}$  contains [milk], bread, [lined, or equal to 0.5. In the first heration of the repeat-loop, we except the frequent determines to create the candidate frequent  $\lambda$  itemsets  $U_{2}$ .  $U_{2}$  contains [milk], bread}, [milk, juice], [bread, nuce], [milk, coskies], thread, coskies] and [juice, coskies]. Notice, for example that [milk, eggs] does not appear in  $U_3$  since (eggs) is small (b) the antimovorenceity property) and does not appear in  $U_3$  since (eggs) is small (b) the antimovorenceity property) and does not appear in  $L_1$ . The supports for the set of transactions. Only the second 2-stender (milk, juice) and ite field coskies] base support greater than or equal to 0.5. These two 2-stenders (bread, cookies] base support greater than or equal to 0.5. These two 2-stenders torm the frequent 2-stenders,  $L_1$ .

In the next iteration of the repeat-loop, we construct candidate frequent 3-iterasets for adding additional iteras to server 12. However, for no extension of iterasets in Ly will all 2itera subsets be contained in Ly. For example, consider build, juser, bread); the 2-iteraset (injfy, bread) is not in Ly, hence (milk, june, bread) cannot be a frequent 3-iteraset by the desenword closer property. At this point the algorithm commutes with L<sub>1</sub> equal to ( bridd), (bread), (price), (cookies) ( and L<sub>2</sub> equal to ( (milk, price), (bread, cookies) )

Several other algorithms have been proposed to mine ossiciation tules. They carring internet of how the condidate itemsets are genericed, and how the supports for the condidate itemsets are counted. Some algorithms use such data structures as bitmaps and bashrices to keep information about itemsets. Several algorithms have been proposed that use multiple scars of the database because the parential number of reinsets. 21, can be too large to set up conners during a single scan. We will examine three improved algorithms (compared to the Agricic algorithm) for association rule mumiciple scanging a single scan.

### 27.2.3 Sampling Algorithm

The main idea for the **Sampling algorithm** is to select a small sample, one that first in main memory, of the database of transactions and to determine the frequent itemsets from that simple. It those frequent itemsets form a superset of the frequent itemsets for the real frequent itemsets for the remainder of the database in order to compute the exact support values for the superset of the frequent itemsets can usually be found from the supple human, for example, the Asner electricity, with a lowered transaction support.

In some rare cases, some frequent iteracts may be missed and a second scan of the database is non-ded. To decide whether any frequent trensiers have been missed, the concept of the negative border is used. The targative border with respect to a frequent iterast. S, and set of trans, I, is the minimal iterastic contained in PowerSet(I) and not in S. The basic idea is that the negative border of a set of frequent trensets contains the closest iterastic that could also be frequent. Consider the case where a set X is not contained in the set of frequent iterastic. If all subsets of X are contained in the set of frequent stensets, then X would be in the negative border.

We dihistrate this with the following example. Consider the set of items  $1 - \{A, B, C, D, E\}$  and let the combined frequent itemsets of size 1 to 3 be S - - 1 [A], 40, 40, [C], {D], [AB], (AC], {DC], {AD}, {CD}, {ABC1}. The negative burder is {1E], [BD], {ACD}, {CD}}. The set {2] is the only 1-itemset not contained in S, [BD] is the only 2-itemset not in S but whose 1 itemset subsets are, and {ACD} is the only 3-itemset whose 2-itemset subsets are all in S. The negative border is important since in its necessary to determine the support for those remisers in the negative border to ensure that the large tensors are mixed from analyting the sample data.

Support for the negative border w determined when the remainder of the database is scanned. If we find that an itemset, X, in the negative border belongs in the set of all frequent itemsets, then there is a potential for a superset of X to also be frequent. If this bugpens, then a second pass over the database is needed to make some that all frequent itemsets are found.

#### 27.2.4 Frequent-Pattern Tree Algorithm

The **Frequent-pattern tree algorithm** is notimated by the fact that Apriori based algorithms may generate and test a very large number of condidate incovery for example,

with 1000 frequent 1-itensets, the Aption algorithm would have to generate  $\begin{pmatrix} 1020\\ 2 \end{pmatrix}$  or

499.502 candidate 2-itemset\*. The IP-growth algorithm is one approach that climinates the generation of a large number of candidate itemsets.

The algorithm first produces a compressed cersion of the darabase in terms of an FPtree (frequent pattern tree). The FF-tree stores to exact memory information and allows for the efficient discovery of tequent densets. The arread numming process independented and conquer strategy where the mining process is decomposed into a set of smaller tasks that each operate on a conditional FP tree, a subset (projection) of the original tree. To stort with, we examine how the FP-tree is constructed. The database is fast scanned and the frequent 1-itemsets along with their support are computed. With this algorithm, the support is the count of transactions containing the item rather than the fraction of transactions containing the item. The irrequent 1-itemsets are then solved in nonitterees ingorder of their support. Next, the pool of the FP-tree is cleated with a "null" label. The database is scanned a second time and for each itemset on T in the database, the frequent 1-itemsets in T are placed in order as was done with the frequent 1-itemset in We can designate this sected list for T as consisting of a first new, the head, and the remaining items, the tisl. The itemsecuntorination (head, tisk) is inserted into the H2-tice reconsively, starting at the mot node, as follows:

- of the current usele. No of the EP-free has a child with an item name # head then increment the count associated with node N by 1 else create a new node. Novith in court of 1, bitk N to it's parent and link N with the item header table (used for efficient iree traversal).
- it tail is nonempty, then repeat step (1) using as the started list only the tail, i.e., the old broad is removed and the new head is the first item from the tail, and the remaining items become the new tail.

The item header table, created drang the process of building the Efsitee, contains three fields per entry for each frequent item, which are item identifier, support courn, and node lipk. The item identifier and support courn are self-explanatory. The node link is a parter to be occurrence of that, repuising the Efsites. Since multiple occurrences of a angle item may appear in the Efsites, these items are linked regeriter as a bit where the start of the list is pointed to by the node link of the item header table. We illustrate the building of the Efsites or ing the transactions yields the following frequent forements with associated is point. (10m/k, 30), [[bread, 20]] {(cookies, 2)}, (quice, 21). The database is scatted it second to no and each transaction will be processed again.

For the first transaction, we create the sorted hst, T = [milk, bread, cockies, joite]. The items in T are the 'acquein' 1 itemsers from the first transaction. The items are indered based on the nonincreasing ordering of the count of the 1-itemsets fund in pass**i**, (i.e., malk first, bread second, etc.). We create a null root node for the FP-tree and insent "finilk" as a child of the more "Fread" as a child of "cookies" as a child of "bread" and "junc" as a child of "cookies". We adjust the entries for the frequent items in the item header table.

For the second transaction, we have the sorted har build, poincel. Sturning at the now, we see that a child node with label "mark" every, so we move to that made and update is count (to account for the second manageron that constraints mult). We see that these is no child of the current node with label "parce," so we greate a new node with label "parce," for we greate a new node with label "parce," for we greate a new node with label "parce," for we greate a new node with label "parce," for the term header table is adjusted.

The third transaction only has 1-frequent tent {milk]. Again, starting at the non-we see that the node with label "milk" exists, so we move to that node, increment its court, and adjust the neur header while. The final transaction contains frequent items, fitted, twokies! At the root node, we see that there does not exist a child with label "bred." Thus, we create a new child of the noot, initializents counter, and then insert "cookies" as a child of the node and initializents count. After the item header table is optiated, we can up with the 1P-tree and item header table as shown in Figure 27.2. If we examine this Hytree, we see that it indeed represents the original transactions at a compressed format (that is, only showing the terms from each transaction that are large 1-itemsets).

Algorithm 27.2) FF-growth Algorithm for hiding frequent itensets

Input: lp-tree and a merimum support, mins. Output: frequent patterns (memory)



FIGURE 27.2 19-tree and dem header table

```
procedure FP-growth (rice, alpha);
Begin
it free contains a single path P then
tot each combination, between the nodes in the path-
 generate pattern. (beta O alpha)
 with support = communication of randes in leta.
else
for each item, it in the header of the trie de-
 begin
 generate partern bein = () U alphal work support = usurport.
 reinstruct beenk conditional pattern base
 construct beta's conditional IP-mee bera_mee.
 if bera tree is not empty then
 FP-gnoeth(heta_may, heta);
 end:
End:
```

Algorithm 27.2 is used for mining the FP-tree for frequent patterns. With the FP-tree, it is possible to find all frequent patterns that contain a given frequent item by starting from the item header table for that stendard traversing the node links in the FP-tree. The algorithm starts with a frequent 1-itemset (soffix pattern), censtructs its conditional pattern base and then its conditional 2P-tree. The conditional pattern base is made up of a set of prefix patter inclusiver the frequent frequent frequent is a softwe for example, if we consider the item patter, we see from Figure 27.2 that there are two paths in the IP-tree.

that end with price. (rails, bread, cookies, juice) and (milk, price). The two associated prefix paths are (milk, bread, cockies) and (milk). The conditional FP tree is constructed from the patterns in the conditional pattern base. The running is recursively performed on this FP-tree. The frequence patterns are formed by concarenating the softx pattern with the frequence patterns produced from a conditional FP tree.

We illustrate the algorithm using the data in Figure 27.1 and the tree in Figure 27.5. The procedure FP-growth is called with the two parameters the original FP-tree and null for the variable alpha. Since the original FP-tree lass more than a single path, we execute the else part of the first if statement. We start with the frequent item, juice. We will examine the frequent items in order of clowest support bluat is, from the last entry in the table to the next). The variable beta is set to juice with support equal to 2.

Following the node link in the item header table, we construct the conditional pattern base consisting of two gaths (with ince as suffix). These are (trulk, bread, cockies: 1) and (milk: 1). The conditional FP tree consists of only a single node, milk 2. This is due to a suggest of only 1 for node bread and cookies, which is below the minimal suggest of 2. The algorithm is called recursively with on EP-tree of only a single node (i.e., angle 2) and a 2ota calce of pince. Since this EP-tree only has one path, all conditionations of best and mides in the path are generated. (that is, [milk.junce]) with support of 2.

Next, the trequest item, cockies, is used. The variable beta is set to cockies with support = 2. Following the mide link in the item beader table, we construct the conditional patiern have consisting of two paths. These are stuffly, breads 11 and (breads 1). The conditional EP tree is only a single mode, breads2. The algorithm is called incorrigively with an EP tree of only a single node (that is, breads2) and a bera value of cookies. Since this EP tree only has one path, all combinations of beta and nodes in the path are generated, that is, diread, cookies? with support of 2. The request item, bread, is considered next. The variable beta is set to bread with support = 2. Following the table, we construct the conditional pattern have consisting of ore path, which is findly 1). The conditional FP tree is empty in the term based to be beauties of the term breads table, we construct the conditional pattern have consisting of ore path, which is findly 1). The conditional FP tree is empty in frequent patterns will be generated.

The last focurent item to consider is malk. This is the top item in the neutheader table and as such has an empty conditional pattern rate and empty conditional Febture. As a result, no frequent patterns are added. The result of executing the algorithm is the following frequent patterns (or itemsets) with their support: { {mdk/3}, {bread 2}, {cookies/2}, {junce/2}, {mdk/3k, {bread 2}, {cookies/2}, {junce/2}, {mdk/ak, junce/2}, {bread/cookies/2}}.

### 27.2.5 Partition Algorithm

Another algorithm, called the **Partition algorithm**, "is summarized below. If we are given a database with a small murber of potential bigg itemsets, say a lew thousand, then the support for all of them can be rested in one some by using a partitioning technique. Pot-

<sup>3.</sup> Size Soussere (\* al. (1995) for details of the algorithm, the data structures used to implement it and its performance comparisons.

tioning doudes the database into nonoverlapping subsets; there are individually considneed as separate databases and all large itemsets for that partition, called kard pequent itemsets are generated in one pass. The Apricin algorithm can then be used efficiently exeach partition of it fits entricity in main memory. Partitions are cluster in such a way that each partition of it fits entricity in main memory. Partitions are cluster in such a way that each partition of it fits entricity in main memory. As such, a partition is read only once an each pass. The only caveat with the partition memory. As such, a partition is read only once an each partition has a slightly different meaning from the original value. The immunous apport is based on the size of the partition in the there there the size of the database for determining local frequent (large) (transets. The actual support threshold value is the same as given earlier, but the support is computed only for a partition.

At the end of pass one, we take the union of all frequent itensets from each partition. These form the global candidate frequent itensets for the entire database. When these lists are merged, they may contain some false positives. That is, some of the removes that are frequent (large) in one partition may not qualify in several other partitions and hence may not exceed the minimum support when the original database is considered. Note that there are no false negatives; etc. Lage itensets will be mosed. The global candidate large itensets identified in pass one are verified in pass two that is, then actual support is measured for the entire database. At the end of phase two, all global large stensets are identified. The Partner algorithm lends itself naturally to a parallel or distributed implementation for better efficiency. Further improvements to this algorithm have been suggested.<sup>4</sup>

#### 27.2.6 Other Types of Association Rules

Association Rules among Hierarchies. There are certain types of associations that are particularly interesting for a special reason. These associations occur among hierarchies of items. Typically, it is possible to decide items among disjoint hierarchies based en the nature of the domain. For example, tools in a supermarket, items in a department store, or articles in a sports shop can be caregorized into classes and subclasses that give rise to hierarchies. Consuler Figure 27.3, which shows the taxonomy of items in a supermarket. The figure shows two hierarchies—beverages and deserts, respectively. The entire groups may now produce associations of the form beverages => deserts, or deserts => heverages. However, associations of the type Heidthy-brand friden yogun =>bortfed water, or Richerteam-brand ice crean => wing couler non-produce enough confidence and support to be valid association rules of interest.

Therefore, if the application area has a notional classification of the itemsets into hierarchies, discovering associations within the hierarchies is of no particular interest. The ones of specific interest are associations across hierarchies. They may accur among item grouppings at different levels.

• •



FIGURE 27.3 Taxonomy of items in a supermarket.

Multirimmensional Associations. Discovering association rules involves searching for patterns in a ble. At the Ecginimity of the data mining section, we have an example of a file of commer transactions with three dimensions. Transactionedd, Time and Itens Bought, However, our data mining tasks and algorithms incrudiced up to this point only involve one dimension, the items/bought. The following tale is an example, where we include the label of the single dimension: Items-Bought(nilk) => Items Bought(jure), h may be of interest to find association rules that involve multiple dimensions, e.g., Time( $\hbar/20.45,00$ ) => Items-Bought(milk), Rules like these are called multidimensional association rules. The dimensions represent arributes of records of a file or, in terms of relations, columns of toxes of a relation, and can be categorical or submittative. Categorical attributes have a time set of values that display no ordering relationship, e.g., s, Items-Dought is an example of a categorical outflotte and Transactionship, e.g., s, Items-Dought is an example of a categorical outflotte and Transactionship.

One approach to handling a quantitative attribute is to partition its values ratio is investagoing intravals that are assigned labels. This can be done in a state matter back in domain specific knowledge. For example, a concept hierarchy may group values to salary into three distance classes low income (0 < solars < 29,090), middle accore (32,000 < solary < 74,900) and high theorem (32,000 < solary < 75,000). From here, the typical Aprinei type algorithm or one of its variants cars be used for the rule matting size the quantitative attributes size (now high like coregeneral attributes. Another approach to partitioning is in group a tribute values together based on data distribution, for example, requiredepth partitioning, and to assign integer values to each partition. The partitioning at this stage may be relatively fine, that is, a larger number of intervals. Then during the

mining process, these partitions may combine with other adjacent partitions of their support is less than some predefined maximum value. An Apriodi-type algorithm can be used here as well for the data mining.

Negative Associations. The problem of discovering a negative association is barder than that of discovering a positive association. A negative association is of the following type: "10%, of enstorners who buy portion chips do not buy Sorted corer." (Here, she 60% refers to the controlence for the negative association rule 1 in a database with 10,200 items, there are 2<sup>15,600</sup> possible combinations of items, a mijority of which do not appear even once in the database. If the absence of a certain term combination is taken to mean a negative association, then we potentially have millions and millions of negative association, then we potentially have millions and millions of negative association, then we potentially have millions and millions of negative association, then we potentially have millions and millions of negative association, then we potentially have millions and millions of negative association, then we potentially have millions and millions of negative association, then we potentially have millions and millions of negative association, then we potentially have millions and millions of negative association. The problem there is no find only interving negative rules. In general, we are intervised in cases in which two specific sets of items appear very tarely in the same transaction. This poses two problems.

- For a conditions invention of 10,000 means, the probability of ant two being bought together is (1/10,000) \* (1/10,000) = 10<sup>-7</sup>. If we find the actual support for these two occurring together to be zero, that does not represent a significant departure from expectation and hence is not an interesting (negative) association.
- The other problem is more serious. We are looking for item combinations with yory low support, and there are influent and millions with low or over anti-opport. For example, what set of 10 million transactions has most of the 2.5 billion pairwise combinations of 12,000 items traising. This would generate billions of useless rules.

Therefore, to make negative association fulles interesting, we must use prior knowledge about the itemsets. One approach is to use hierarchies. Suppose we use the hierarchies of soft drucks and chips shown in Figure 27.4.

A strong positive association has been shown hoween solid miles and chips, if we and a large support for the fact that when customers buy Days chips they predomasantly ruy Tapsy and not Joke and not Wakeup, that would be interesting. This is so because we would normally expect that if there is a strong association between Days and Topsy there should also be such a strong association between Days and toke or Days and Wakeup.<sup>5</sup>

In the frozen yogurt and bottled stater groupings in Figure 27.3, suppose the feedbace versus Healthy-brand division is 82–20 and the Plays and Clear brands division is 62–40 among respective categories. This would give a joint probability of Reduce fricen yogurt



FIGURE 27.4 Simple hierarchy m soft drinks and chips

5 For simplicity according courts in distribution of transactions in pagmenticus of chiermich-

being purchased with Clain bottled water as 4M a anong the transactions containing a frazen yngan and a bottled water. It dus support, huwe een is found to be only 2Cm, that would indicate a significant negative association atomig Reduce yogur, and Plain borded water; ogain, that would be increasing.

The problem of finding negative association is important in the above situations given the domain knowledge in the form of intro-generalization literarchies (that is, the beverage prom and deserts hierarchies shown in Figure 27.3), the existing positive associations (such as between the fracen (ogen) and burded other groups), and the distribution of terms (such as the name brands within refined groups). Work has been reported by the database group at Georgia Tech in this context (see bibliographic mores). The scope of discovery of negative isociations is dimined in terms of knowing the trein hierarchies and distributions. Exponential groups of negative associations remains a challenge.

### 27.2.7 Additional Considerations for Association Rules

Mining association tales in real-life databases is complicated by the following factors

- The cardinality of itemsets in most situations is extremely large, and the volume of itemsections is very large as well. Some operational databases in retailing and commumention industries collect tension nullions of transactions per day.
- Transactions show variability in such factors as peographic location and seasons, making simpling difficult.
- frem classifications exist along multiple dimensions. Hence, driving the decovery process with domain knowledge, particularly for negative fulles, is extremely difficult.
- Quality of data is variable; significant problems exist with innsing enroleous, conthering, as well as redundant data memory industries.

### 27.3 CLASSIFICATION

Classification is the process of learning a model that describes different classes of data. The classes are predeterimmed. For example, in a banking application, costemets who apply for a credit card may be classified as a "poor risk," a "fair risk," or a "good risk." Hence this type of articity is also called supervised learning. Once the model is built, then it can be used to classify new data. The first step, of learning the model, is accomplished by using a training sit of data that has already be each learning. Once the model, is accomplished by using a training sit of data that has already be each learning the model, is accomplished by using a training an attribute, called the class label, that inducates which class the record belongs to The model that is produced to usually in the four of a decision in site of foles. Some of the important issues with regard to the model and the algorithm that produces the model include the model's ability to predict the concett class of new data, the computuonal cost associated with the algorithm, and the scalability of the algorithm.

We will examine the approach where out model is in the form of a decision tree. A decision tree is simply a graphical representation of the decoption of each class or a

other words, a representation of the classification rules. An example decision free is pretired in Figure 27.5. We set from Figure 27.5 that it a customer is "manuel" and their salary  $\geq$ -50K, then they are a good risk for a credit card from the bank. This is one of the rules that describe the class "good risk." Other rules for this class and the two other classes are formed by traversing the decision free from the root to each leaf node. Algorithm 27.5 shows the precedure for constructing a decision free from a training data set. Initially all forming samples are at the root of the tree. The samples are partitioned recursively based on selected attributes. The attribute used at a node to partition the samples is the one with the best splitting enterior for complet the one that maximizes the information gain measure.

Algorithm 27.3: Algorithm for decision free increation

**Input:** we of training data Records  $R_0, R_0, \dots, R_n$  and set of Attributes:  $A_1, A_0, \dots, A_n$ . **Output:** decision tree

procedure Build pree (Records, Attribures): Begin critore « nod+ N.

it all Records belong to the same class. Orbon Return Nus a teaf node with class labe? Ci-

nt Artributes is empty then Return Nos a leaf node with class label C, such that the majority of Records belong rour;

select attribute A, (with the fighest information goot) from Attributes;

label node N with A :



FIGURE 27.5 Example decision free for right Card applications.

tor each knimen value, V<sub>μ</sub> et A<sub>μ</sub> do hegin hald a branch truin node N for the condition: A = V<sub>μ</sub> S<sub>μ</sub> = subset of Records where A<sub>μ</sub> = V<sub>μ</sub> if S<sub>μ</sub> is empty then hald a leat, L, with class label C, such that the majority of Records belong four and Return L else add the node returned by Build\_two (S<sub>μ</sub> Attributes - A<sub>μ</sub>): end: End;

Before we alloctrize Algorithm 27.5, we will explore in more detail the information gain measure. The use of entropy is the information gain measure is moreored by the geal of minimizing the information needed to classify the sample data in the relating partitions and thus minimizing the expected number of conditional tests needed to classify a new second. The expected information needed to classify maning data of s samples, where the Class attribute has invalues  $(x_1,...,x_n)$  and s, is the number of samples belonging to Class information by

$$\{(S_1,S_2,...,S_n)\} = -\sum_{n=1}^n p_n \log_2 p_n$$

where p, is the probability that a random sample belongs to class with label  $v_i$ . An estimate for  $p_i$  is  $\delta \in Consider an attribute A with values <math>\{v_1, \dots, v_n\}$  used as the set attribute for splitting in the decision tree. Antribute A partitions the samples into the subsets  $S_1, \dots, S_n$  where samples in each  $S_i$  have a value of  $v_i$  for attribute A. Each  $S_i$  may contrain samples that belong to class of the classes. The number of samples in  $S_i$  that belong to class i can be denoted as  $v_n$ . The entropy associated with using strubute A as the test attribute i defined as

$$\mathbf{E}(\mathbf{A}) = \sum_{j=1}^{n} \frac{S_{j1}}{s} \frac{y_{j-1} - S_{j2}}{s} \bullet \mathbf{I}(S_{j1}, S_{j2}, ..., S_{jn})$$

 $l(s_1,...,s_n)$  can be defined using the formulation for  $l(s_1,...,s_n)$  with  $p_i$  being replaced by  $p_n$  where  $p_n = s_p/s_n$ . Now the subsequations gain by partitioning on netribute  $A_i$ . Const.A), is defined as  $H_{s_1}...,s_n = l(A)$ . We can use the sample training data from Figure 26.6 to talks trate. Algorithms

The attribute RW represents the record identifier used for identifying an individual record and is an internal attribute. We use it to identify a particular record in core example. First, we compare the expected information needed to closify the traiting data of 6 records as  $1_{V_1,V_2}$ ) where the first class label value corresponds to "ves" and the second to "au". So

$$I(3,3) = -0.5 \log(0.5 + 0.5 \log(0.5) + 1)$$

Now, we compute the entropy for each of the 4 attributes as shown below. For Married  $\pi$  yes, we have  $\|s\|_2 = 2$ ,  $s_{21} = 1$  and  $\|(s_{11}, s_{22}) = 0.92$ . For Married = no we have

| BID | Married | Salary         | Acct Balance | Age  | Loanworlby |
|-----|---------|----------------|--------------|------|------------|
| 1   | no      | >=50k          | <5k          | >=25 | yes        |
| 5   | yes     | > <b>≈</b> 50k | t≽≓5k        | >=25 | yes        |
| 3   | yes     | 20k - 50k      | ~ 5k         | <25  | 00         |
| 4   | rio     | <20k           | ≻=5k         | <25  | rio        |
| 5   | по      | <20≪           | <5k          | >=25 | no         |
| 6   | yes     | 20k50k         | ≈=5k         | >=25 | yes        |

FIGURE 27.6 Sample training data for classification algorithm.

 $s_{12} = 1$ ,  $s_{22} = 2$  and  $l(s_{12}, s_{22}) = 2.92$ . So, the expected intermation needed to classify a sample using attribute married as the partitioning attribute is

 $E(Married) = 3/6 I(s_1, s_2) + 3/6 I(s_2, s_2) = 0.92.$ 

The gon in information, Gam(MarnesI), would be 4 - 0.92 = 0.08. If we follow sturthsteps for computing the gain with respect to the other three attributes we end op with

Since the greatest gain occurs for attribute Salary, it is chosen as the partitioning attribute. The cost of the tree is created with label Salary and has three bunches, one for each value of Salary. For two of the three values, i.e.,  $\leq 20k$  and  $\geq -50k$ , all the samples that are portitioned accordingly factorils with RIDs 4 and 5 for  $\geq 20k$  and records with RIDs 1 and 2 for  $\geq -50k$ ) tall within the same class "loanweithy no" and "bonworthy yes," respectively for those two values. So we create a frat node for each. The only branch that needs to be expanded is for the value 20k...50k with two samples, records with RIDs 3 and 6 in the training data. Continuing the process using these two records, we find that Gain(Married) is 0. Gain(A or Balance) is 1 and Gain(Age) is 1.

We can choose either Age or Acci Belance since they both have the largest gain. Eet us choose Age as the partitioning attribute. We add a node with label Age that has two branches, less than 25, and greater or equal to 25. Each branch partitions the reasoning sample data such that one sample record belongs to each branch and beace one class. Two leaf nodes are created and we are **finished**. The final decision tree is pictured in Figure 27.7.

# 27.4 CLUSTERING

The previous dota imming task of classification decla with participant data based on using a pre-classified training sample. However, it is aften useful to partition data without backing a training sample, this is also known as **unsupervised learning**, for example, in business, it may be important to determine groups of costoners who have similar boying patterns, or in medicine, it may be important to determine groups of rotients who show



**FIGURE 27.7** Decision free based on sample training data where the leaf nodes are represented by a set of RIDs of the partitioned records.

similar references to prescribed drugs. The goal of clustering is to place records into groups, such that records in a group are similar to each other and dissimilar to records in other groups. The groups are usually disjoint.

An important facet of clustering is the summary function that is used. When the data is numeric, a similarity function based on distance is typically used. For example, the Euclidean distance can be used to measure similarity. Consider two **n**-dimensional data points (records)) and  $r_1$ . We can consider the value for the  $i^m$  dimension as  $r_1$  and  $r_2$  or the two records. The Euclidean distance between points  $r_1$  and  $r_2$  or a-dimensional space is calculated a-

Distance
$$(\mathbf{x}_{i}, \mathbf{r}_{k}) = \left[\sqrt{|\mathbf{x}_{jk} - \mathbf{r}_{kj}|^{2} - |\mathbf{r}_{kj}|^{2} - |\mathbf{r}_{kj}|^{2} + \dots + \frac{b}{|\mathbf{r}_{jk}|} - |\mathbf{r}_{kj}|^{2}}\right]$$

The smaller the distance between two points, the greaterns the similarity as we think of them. A classic clustering algorithm is the k-Means algorithm. Algorithm 27.4.

Algorithm 27.45 K-means clustering algorithm

**Input**: a database D, at invectorily,  $r_1, \ldots, r_n$ , and a desired number of clusters k **Output**: set of k clusters that minimizes the squared error criterice.

#### Regin

randomly choose k records as the contraids for the k clusters;

repear

assign each record, r<sub>e</sub> to a charer such that the distance between r<sub>e</sub> and the cluster centroid (mean) is the smallest among the k charters.

recalculate the centroid (mean) for each cluster based on the records assured to the cluster.

```
word notch tages
```

End:

The algorithm begins by randomly choosing k records to represent the centroids (means),  $m_1, \ldots, m_k$ , of the clusters,  $C_1, \ldots, C_k$ . All the records are placed in a given cluster based on the distance between the record and the cluster mean. If the distance between in and record  $r_i$  is the smallest among all cluster means, then record  $r_i$  is ploted in cluster  $C_i$ . Once all records have been initially placed in a cluster, the mean for each cluster is recomputed. Then the process repeats, by examining each record again and placing it in the cluster whose mean is closest. Several iterations may be needed, but the algorithm will converge, although it may terminate at a local optimum. The terminating condition is usually the squared-error criterion. For clusters  $C_1, \ldots, C_k$  with means  $m_1, \ldots, m_k$ , the error is defined as.

Error = 
$$\sum_{i=1}^{k} \sum_{i=1}^{k} \sum_{i \in \mathcal{N}_{i}} \text{Distance}(r_{p}, m_{i})^{2}$$

We will examine how Algorithm 26.4 works with the (2-dimensional) records in Figure 27.5. Assume that the number of desired clusters k is 2. Let the algorithm choose records, with RID 3 for cluster  $C_1$  and RID 6 for cluster  $C_2$  as the initial cluster centroids. The remaining tecords will be assigned to one of those clusters during the first treation of the repeat loop. The record with RID 1 has a distance from  $C_1$  of 22.4 and a distance from  $C_2$  of 32.0, so it joins cluster  $C_1$ . The record with RID 2 has a distance from  $C_1$  of 5.0, so it joins cluster  $C_1$ . The record with RID 4 has a distance from  $C_1$  of 25.5 and a distance from  $C_2$  of 36.6, so it joins cluster  $C_1$ . The record with RID 5 has a distance from  $C_1$  of 20.6 and a distance from  $C_2$  of 30.7 and a distance from  $C_2$  of 36.6, so it joins cluster  $C_1$ . The record with RID 5 has a distance from  $C_1$  of 20.6 and a distance from  $C_2$  of 29.2, so it joins cluster  $C_1$ . New, the new means (centroids) for the two clusters are computed. The mean for a cluster,  $C_2$ , with a records of m dimensions is the vector.

$$\tilde{C}_{j} = \left( \frac{1}{u} \sum_{\mathbf{V}_{T} \in [C_{i}]^{T_{j} \times \cdots \times T_{n}}} \frac{1}{u} \sum_{\mathbf{V}_{T} \in [C_{i}]^{T_{j} \times \cdots \times T_{n}}} \frac{1}{v_{T} \in [C_{i}]^{T_{j} \times \cdots \times T_{n}}} \right)$$

The new mean for  $C_1$  is (33.75, 8.75) and the new mean for  $C_2$  is (52.5, 25). A second iteration proceeds and the six records are placed into the two clusters as follows: records with RIDs 1, 4, 5 are placed in  $C_1$  and records with RIDs 2, 3, 6 are placed in  $C_2$ . The mean for  $C_1$  and  $C_2$  is recomputed as (28.3, 6.7) and (5), 7, 21.7), respectively in the next iteration, all records stay in their previous clusters and the algorithm terminates.

|     |     | Years      |  |  |
|-----|-----|------------|--|--|
| 6ID | Age | of Service |  |  |
| 1   | 30  | 5          |  |  |
| 2   | 50  | 25         |  |  |
| 3   | 50  | 15         |  |  |
| 4   | 25  | 5          |  |  |
| 5   | 30  | 10         |  |  |
| 6   | 55  | 25         |  |  |

FIGURE 27.8 Sample 2-dimensional records for clustering example like RID colunin is not considered). Tradmonally, clostering algorithms assume that the entire data set firs in main memory. More recently, researchers have been developing algorithms that are efficient and are scalable for very large databases. One such algorithm is called BIRCH. BIRCH is a hybrid approach that uses both a hierarchical clustering approach, which builds a tree representation of the data, as well as additional clustering methods, which are applied to the leaf nodes of the tree. Evolution parameters are used by the BIRCH algorithm. One specifies the amount of available main memory and the other is an initial threshold for the radius of any cluster. Main memory is used to store descriptive cluster information such as the center (mean) of a cluster and the radius of the cluster (clusters are assumed to be spherical in shape). The radius threshold affects the number of clusters that are produced. For example, if the radius threshold value is large, then few clusters of many records will be formed. The algorithm must to maintain the number of clusters such that their radius is below the radius threshold, lif available memory is multicient, then the radius threshold, for an increased.

The BIRCH algorithm reads the data records sequentially and inserts them into an immemory tree structure, which this to preserve the clustering structure of the data. The records are inserted into the appropriate leaf nodes (potential clusters) based on the distance between the record and the cluster center. The leaf node where the insertion happens may have to split, depending upon the updated center and tadus of the elester and the radius threshold parameter. In addition, when splitting, extra cluster information is stored and if memory becomes matheticing, then the radius threshold will be mereased. Increasing the radius threshold may actually produce a side effect of reducing the number of clusters since some modes may be merged.

Overall, BIRCH (s an efficient clustering morbod with a linear computational complexity in terms of the number of records to be clustered.

### 27.5 APPROACHES TO OTHER DATA MINING PROBLEMS

#### 27.5.1 Discovery of Sequential Patterns

the discovery of sequential patterns is based on the concept of a sequence of itemsets. We assume that transactions such as the supermarket-basket transactions we discussed previously are ordered by time of purchase. That ordering yields a sequence of itemsets for example, (milk, bread, juice), (bread, eggs), (cookies, nalk, coffee) may be such a sequence of itemsets based on three visits of the same customer to the store. The support for a sequence, by this example, (milk, bread, juice), (bread, eggs), (cookies, nalk, coffee) may be such a sequence of itemsets based on three visits of the same customer to the store. The support for a sequence, by this example, (milk, bread, juice) (bread, eggs) and (bread, eggs) (cookies, nalk, coffee) are considered subsequences. The problem of identifying sequented patterns, then, is to find all subsequences from the given sets of sequences that have a user-defined imminum support. The sequence  $S_1, S_2, S_3, \dots$  is a predictor of the tat that a customer who based on the frequency (support) of this sequence in the pase. Various algorithms have been investigated for sequence different.

### 27.5.2 Discovery of Patterns in Time Series

Time series are sequences of events; each event may be a given fixed type of a transaction. For example, the closing price of a stock or a fund is an event that occurs every weekday tot each stock and fund. The sequence of these values per stock is find constitutes a time series. For a time series, one may look for a variety of patterns by analyzing wequences and subsequences as we did above. For example, we might find the period during which the stock rose or held steady for it days, or we might find the longest period over which the stock had a fluctuation of no more than 1% over previous clusing price, or we might find the quarter during which the stock had a fluctuation of no more than 1% over previous clusing price, or we might find the quarter during which the stock had the next percentage gum or percentage loss. Time series may be compared by establishing measures of similarity to identify comparies whose stocks heldwe in a similar fashion. Analysis and mining of time series is an extended functionality of temporal data management (see Chapter 74).

### 27.3.3 Regression

Regression is a special application of the classification rule. If a classification rule is regarded as a function over the variables that maps these variables into a rarger class variable, the rule is called a regression rule. A general application of regression occurs when, instead of mapping a tuple of data from a relation to a specific class, the value of a variable is predoted based on that tuple. For example, consider a relation

LAB\_TESTS (parient ID, rest 1, test 2, ....., test a)

which contains values that are results from a series of a tests for one patient. The target variable that we wish to product is *P*, the probability of survival of the patient. Then the rule for regression takes the form:

```
(test 1 in range.) and (test 2 in range,) and . . . (test n in range,) => P = x, er x < P < y
```

The choice depends on whether we can predict a unique value of P or a range of values for P. B we regard P as a function:

 $P = f(\text{test } 1, \text{test } 2, \dots, \text{test } n)$ 

the function is called a regression function to predict P. In general, if the function appears as

 $\boldsymbol{X} = f(\boldsymbol{x}_1, \boldsymbol{x}_2, \dots, \boldsymbol{x}_n),$ 

and *f* is linear in the domain variables **x**, the process of deriving *f* from a given set of tuples for  $\leq x_1, x_2, \ldots, x_n, x \geq v$  called linear regression. Linear regression is a commonly used statistical technique for fitting a set of observations or points in 6 dimensions with the target variable *x*.

Regression analysis is a very common tool for analysis of data in many research domains. The discovery of the function to predict the target variable is equivalent to a data muture operation.

### 27.5.4 Neural Networks

Neural network is a technique derived from artificial intelligence research that uses generalized regression and provides on aterative method to carry et out. Neural networks use the carve-lifting approach to unfer a function from a set of samples. This technique provides a "learning approach": it is driven by a test sample that is used for the (initial inference and learning. With this kind of learning method, responses to new inputs may be able to be interpolated from the known samples. This interpolation however, depends on the eard model (internal representation of the problem domain) developed by the learning method.

Neural networks can be broadly classified into two coregories: supervised and unsupervised networks. Adaptive methods that attempt to reduce the output error are supervised learning methods, whereas those that develop internal representations without sample outputs are called unsupervised learning methods.

Neural networks self-adapt; that is, they learn from information on a specific problem. They perform well un classification tasks and are therefore useful in data mining. Yet, they are not without problems. Although they learn, they do not provide a good representation of usual they have learned. Their outputs are highly quantitative and not easy to understand. As another limitation, the program representations developed by neural networks are not unique. Also, in general, neural networks have trouble modeling time series data. Despite these shorecomings, they are popular and frequently used by several commercial vendors.

### 27.5.5 Genetic Algorithms

**Genetic algorithms** (GAs) are a class of randomized search procedures capable of adaptive and rebust search over a wide range of search space topologies. Modeled after the adaptive raneigence of biological species from evolutionary mechanisms, and incruduced by Holland,<sup>4</sup> CAs have been successfully applied in such diverse fields such as mage analysis, scheduling, and engineering design.

Genetic algorithms extend the idea from human genetics of the functional phalot thased on the A.C.T.O nucleutides) of the human DNA code. The construction of a genetic algorithm involves deviang an alphabet that encodes the solutions to the decision problem in terms of strings of that alphabet. Strings are equivalent to individuals. A fitness function defines which solutions can survive and which cannot. The ways in which solutions can be combatted are patterned after the costs-over operation of catting and combating strings from a father and a reother. An insteal population of well-varied population is provided, and a game of evolution is played in which mutations occur among strings. They combine to produce a new generation of individuals; the fittest individuals survive and mutate unsil a family or successful solutions develops.

The solutions produced by genetic algorithms (GAs) are distinguished from oust other search techniques by the following characteristics:

<sup>6.</sup> Hollandy scannal work (1975) enricled "Adaptation in Natural and Archeval Systems" acroduced the idea of genetic algorithms.

- A GA search uses a set of solutions during each generation rather than a single solution.
- The search in the string-space represents a much larger patallel search in the space of encoded solutions.
- The memory of the search done is represented solely by the set of solutions available for a generation.
- A genetic algorithm is a randomized algorithm since search mechanisms use probabibatic operators.
- While progressing from one generation to the next, a GA linds near-optimal balance between knowledge acquisition and exploration by manipulating encoded solutions.

Genetic algorithms are used for problem solving and clustering problems. Their ability to solve problems in parallel provides a powerful tool for data mining. The drawbacks of OAs include the large overproduction of individual solutions, the random character of the searching process, and the high demand on computer processing. In general, substantial computing power is required to achieve anything of significance with genetic algorithms.

# 27.6 APPLICATIONS OF DATA MINING

Data numing technologies can be applied to a large variety of decision-making contexts in humans. In particular, areas of significant payoffs are expected to include the following:

- Marketang—Applications include analysis of consumer behavior based on buying parrents; derenonation of marketing stategics including advertising, store location, and targeted mailing, segmentation of customers, stores, or products; and design of catalogs, store layouts, and advertising campaigns.
- Finance—Applications include analysis of creditworthiness of thems, segmentation
  of account acceivables, performance analysis of finance investments like stocks,
  bonds, and mutual funds, evaluation of financing options; and found detection
- Manufacturing—Applications involve optimization of resources like machines, manpower, and materials; optimal design of manufacturing processes, shop-floor layouts, and product design, such as fin automobiles lasted on customer requirements.
- Health: Care—Applications include discovering patterns in radiological images, analysis of microarray (gene-chip) experimental data to relate to diseases, analyzing side effects of drups, and effectiveness of certain treatments; optimization of processes within a hospital, relating patient wellness data with doctor qualifications.

# 27.7 COMMERCIAL DATA MINING TOOLS

At the present time, commercial data mining tools use several common techniques to extract knowledge. These include association tules, clustering, neural networks, sequencing, and statistical analysis. We have discussed these earlier. Also used ate decision trees.

which are a representation of the inles used in classification of clustering, and statistical analyses, which may include regression and many other techniques. Other communal products use advanced techniques such as genetic algorithms, case-based reasoning, Bayesian networks, nonlinear regression, combinational optimization, pattern matching, and fuzzy logic. In this chapter we have already discussed some of these.

Most data mining tools use the CDBC (Open Database Connectivity) interface. CDRC is an industry standard that works with database; it enables access to data in most of the gopalar database programs with as Access, dRASE, Informus, Oracle, and SQL Server. Some of these software packages provide interfaces to specific database programs; the must common are Oracle, Access, and SQL Server. Most of the tools work in the Microsoft Windows environment and a few work in the UNIX operating system. The trend is for all products to operate under the Microsoft Windows environment. One tool Data Surveyor, mentions ODMG compliance; see Chapter 21 where we discuss the COXE, object-oriented standard.

In general, these programs perform sequential processing in a single machine. Many of these products work in the client server mode. Some products incorporate parallel processing in parallel computer architectures and work as a part of online analytical processing (GLAP) tools.

User Interface. Most of the tools run in a graphical tact factorface (0:1) environment. Some products include suphisticated visualization techniques in view data and rules (e.g., MineSet of \$00), and are oven able to manipulate data this way interactively. Text interfaces are rare and are more concorn in cools available for UNIS, such as DMA Intelligent Miner.

Application Programming Interface. Usually, the application programming interface. Usually, the application programming their internal functions. However, some of them allow the application programmer to reuse their code. The must common interfaces are C libraries and Dynamic Link Libraries (DLLs). Some tools include proprietary database command languages.

In Table 27.1 we list 11 representative data mining tools. To date there are almost a hundred commercial data mining products available worldwide. Non-U.S. products include Data Superyor from the Netherlands and Folyanalyst from Rossia.

Future Directions. Data mining tools are commually evolving, building on ideas have the latest scientific research. Many of these tools incorporate the latest algorithms taken frien artificial meelligence (AI), statistics, and optimization.

At present, fast princessing is done using modern database techniques—such as distributed processing—in client-server architectutes, in parallel databases, and in data warehousing. For the future, the trend is toward developing laternet capabilities more fully. In addition, hybrid approaches will become commonplace, and princessing will be done using all resources available. Processing will take advantage of both parallel and distributed computing environments. This shift is especially important because modern databases contain very large amounts of information. Not only are undifined atabases growing, but image storage and retrieval are both slow operations. Also, the cost of

| COMPANY         | PRODUCT     | TECHNIQUE                  | PLATFORM        | INTERFACE            |
|-----------------|-------------|----------------------------|-----------------|----------------------|
| Acknessfi       | Кате        | Decision trees,            | Win NT          | Matteria Access      |
|                 |             | Clase-based                | UNIX            |                      |
|                 |             | 0.9800.009                 |                 |                      |
| Anness          | Knowledge   | Decision trees.            | Win NT          | ODBC                 |
|                 | Seeker      | Statistics                 |                 |                      |
| Business        | Bosanesa    | Neural nets.               | Win NT          | 12090                |
| Objects         | Miner       | Machine learning           |                 |                      |
| CrossZ          | QueryObject | Statistical Analysis       | Win NT          | CODEC .              |
|                 |             | Optimization               | <b>N/VS</b>     |                      |
|                 |             | sigoruhin                  | UNIX            |                      |
| Daxa            | Data        | Comprehensive.             | UNIX            | ODBC                 |
| Distillence     | Surveyor    | Can time DM                |                 | ODKG-compliant       |
| DBMmer          | DBMiner     | OLAP analysis.             | Wm NT           | Microsoft 7.0        |
| Technology Inc. |             | Associations, Classificati | i.: <b>:.</b> , | OLAP MG <sub>1</sub> |
|                 |             | Clustering                 |                 |                      |
|                 |             | algorithms                 |                 |                      |
| JAM             | Intelligent | Classification,            | ONIX            | IOM .                |
|                 | Miger       | Association roles.         | (AIX)           | 1902                 |
|                 |             | Fredictive models          |                 |                      |
| Megapater       | Polyanalyst | Symbolic                   | Win NT          | OPBC                 |
| Intelligence    |             | knowledge                  | 08/2            | Oracle               |
| -               |             | acquisition                |                 | 1902                 |
|                 |             | Evolutionary               |                 |                      |
|                 |             | programming                |                 |                      |
| NCR             | Management  | Association rules          | Win NT          | OD80                 |
|                 | Discovery   |                            |                 |                      |
|                 | Tool (MDT)  |                            |                 |                      |
| 5.4.5           | Enterprise  | Decision trees.            | UNIX            | OD8C                 |
|                 | Minier      | Association rules.         | (Solaras)       | Oracle               |
|                 |             | Neural nets,               | Wm NT           | AS/400               |
|                 |             | Regression,                | Macrotosh       |                      |
|                 |             | Clustering                 |                 |                      |
| Sihuon          | MineSet     | Decision trees,            | UN:X            | Oracle               |
| Oraphics        |             | Association rules          | (Irix)          | Sybase               |
|                 |             |                            |                 | Informix             |

TABLE 27.1 SOME REPRESENTATIVE DATA MINING TOOLS

CDN : Open Data Base Connectority:

OPMG, Object Data Management Group

secondary storage is decreasing, so massive information storage will be feasible, even for small companies. Thus, data mining programs will have to deal with larger sets of data of more companies.

In the near fature it seems that Microsoft Windows NL and UNIX will be the standard platforms, with NT being dominant. Most of data mining software will use the CDSC standard to extract data from business databases; proprietary input formats can be expected to disappear. There is a definite need to include nonstandard data, including images and other multimedia data, as source data for data mining. However, the algorithmic developments for nonstandard data mining have not seached a maturity level sufficient for commercialization.

### 27.8 SUMMARY

In this chapter we surveyed the important discipline of data mining, which uses database technology to discover additional knowledge or patterns in the data. We gave an illustrative example of knowledge discovery in databases, which has a wider stope than data training. For data mining, among the various rechniques, we focused on the details of association rule mining, classification and clustering. We presented algorithms in each of these areas and illustrated how chose algorithms work with the aid of examples.

A voriety of other techniques, including the Al-based zonral networks and genetal algorithms, were also briefly docussed. Active research is ongoing in data mining and we have outlined some of the expected research directions. In the future database technology products market, a great deal of data mining activity is expected. We summatized 11 out of nearly a hundred data mining tools available today, future research is expected to extend the number and functionality significantly.

#### Review Questions

- 27.1 What are the different phases of the knowledge discovery from databased Describe a complete application scenar o in which new knowledge may be mined from an existing database of transactions.
- 27.2 What are the goals or tasks that data mitting attempts to for little?
- 27.3 What are the five types of knowledge produced from data mining?
- 27.4. What are association rules as a type of knowledge? Unvertable definition of support and confidence and use them to define an association rule.
- 27.5. What is the downward closure property? How does it aid in developing an efficient algorithm for finding association rules, i.e., with regard to finding large itemsets?
- 27.6. What was the motivating factor for the development of the FP-tree algorithm for association rule mining?
- 27.7. Describe an association rule among hierarchies with an example.
- 27.8. What is a negative association rule in the context of the literarchy of Figure 27.9.
- 27.9. What are the difficulties of mining association rules from large databases?

- 27.10. What are classification rules and how are decision trees related to them?
- 27.11. What is enjropy and how is it used in building decision trees?
- 27.12. How does clustering differ from classification?
- 27.13. Describe neural networks and generic algorithms as techniques for data muning. What are the main difficulties in using these techniques?

### Exercises

27.14. Apply the Aprior algorithm to the following data set:

| Trans ID 👘 Iten | ns Putchased |
|-----------------|--------------|
|-----------------|--------------|

| 121 | nuk, | bicad, | eggs |
|-----|------|--------|------|
|     |      |        |      |

- 102 milk, junc
- t03 juice, butter
- 104 milk, bread, eggs
- 105 collee, eggs
- 106 coffee
- 107 coffee, juice
- 108 milk, bread, cookies, eggs
- 109 cookies butter
- 110 milk, bread

The set of items is {milk, bread, cookies, eggs, butter, coffee, juice}. Use 0.2 for the minimum support value.

- 27.15 Show two rules that have a confidence of 0.7 or greater for an itemset containing three items from Exercise 23.
- 27.16. For the Partition algorithm, prove that any frequent itemset in the database must appear as a local frequent itemset in at least one partition.
- Show the FP tree that would be made for the data from Exercise 23.
- Apply the FP-growth algorithm to the FP tree from Exercise 26 and show the frequent itenvets.
- Apply the classification algorithm, to the following set of data records. The class artificate is Repeat Customer.

| RID   | Age   | City | Gender | Education   | Repeat Customer |
|-------|-------|------|--------|-------------|-----------------|
| 101   | 20.30 | NY   |        | cullege     | YES             |
| 102   | 20.30 | SF   | M      | graduate    | YES             |
| 103   | 31,42 | NY   | 7      | college     | YES             |
| 104   | 51.60 | NY   | F      | college     | NO              |
| 105   | 31.40 | 1A   | M      | high school | . NO            |
| 105   | 41.52 | NY   | ¥      | college     | YES             |
| 107 - | 4150  | NY   | F      | graduare    | YES             |
| 108   | 20.30 | LΑ   | M      | college     | YES             |
| 109   | 2030  | NΥ   | F      | high school | i NO            |
| 110   | 20.32 | NY   | F      | college     | YES             |

| RID | Dimension1 | Dimension2 |
|-----|------------|------------|
| I I | 8          | 4          |
| 2   | 5          | 4          |
| 3   | 2          | 4          |
| 4   | 2          | 6          |
| 5   | 2          | х          |
| é   | ŝ          | 6          |

27.20. Consider the following set of two-dimensional records:

Also consider two different clustering schemes: (1) where Cluster, contains records  $\{1,2,3\}$  and Cluster<sub>2</sub> contains records  $\{4,5,6\}$  and  $\{2,3,4,5\}$ . Where Cluster<sub>1</sub> contains seconds  $\{1,6\}$  and Cluster<sub>2</sub> contains records  $\{2,3,4,5\}$ . Which scheme is better and why?

- 27.21. Use the K-means algorithm to cluster the data from Exercise 29. We can use a value of 3 for K and can assume that the records with RIDs 1, 3 and 5 are used for the initial cluster centroids (means).
- 27.22. The K-means algorithm uses a similarity metric of distance between a record and a cluster centroid. If the attributes of the records are not quantitative, but categorical in mature, such as income Level with values (low, medium, high) or Marned with values (Yes, No) or State of Residence with values (Alabana, Alaska, ..., Wyoming) then the distance metric is not meaningful. Define a more suitable similarity metric that can be used for clustering data records that contain categorical data.

### Selected Bibliography

Literature on data mining comes from several fields, including statistics, mathematical optimization, machine learning, and arringial intelligence. Data mining has only recently become a topic in the database literature. We, therefore, mention only a few databaserelated works. Chen et al. (1996) give a good similarith of the database persidentie on data mirring. The book by Han and Kamber (2001) is an excellent text, describing in detail the different algorithms and tochniques used in the data imming area. Work at IBM Almaden research has produced a large number of early concepts and algorithms as well as results from some performance studies. Agrawal et al. (1995) report the first major study on association rules. Their Apriori algorithm for market stoket dara in Agrawal and Stikant (1994) is improved by using partitioning in Sovasere et al. (1995); Towonen (1996) proposes sampling as a way to reduce the processing effort. Cheung et al. (1996) extends the partitioning to distributed environments; Lin and Dynhom (1995) propose techniques to overcome problems with data skews Agraval et al. (1993b) discuss the performance perspective on association rules. Mannila et al. (1994). Park et al. (1995), and Amir et al. (1997) present additional officient algorithms related to association piles. Han et al. (2000) present the FP size algorithm discussed in this chapter. Stikant (1995) preposes mining generalized rules. Savasere et al. (1998) present the first approach to mining negative associations. Agravul et al. (1996) describe the Ocean system at IBM. Saravagi et al. (1996) describe an implementation where association tales are integrated with a

relational database management system. Pistesky-Shapiro and Frawley (1992) have contributed papers from a wide range of topics related to knowledge discovery. Zhang et al. (1996) present the BIRCH algorithm for clustering large databases. Information about discovery tree learning and the classification algorithm presented in this chapter can be found in Mitchell (1997).

Adriaans and Zantinge (1996) and Woos and Indurkhya (1998) are two recent books devoted to the different aspects of data mining and its use in prediction. The idea of genetic algorithms was proposed by Holland (1975); a good survey of genetic algorithms appears in Srinivas and Patnatk (1974). Neural networks have a vast literature; a comprehensive nurodaction is available in Cippman (1987).






# Overview of Data Warehousing and OLAP

The increasing processing power and sophistication of analytical tools and techniques have resulted in the development of what are known as data watehouses. These data warehouses provide storage, lumitionality, and tesponsiveness to queries beyond the capahilities of transaction-priented databases. Accompanying this even increasing power has come a great demand to imprive the data access performance of databases. As we have seen throughout the book, traditional databases before the requirement of data access with the need to ensure integrity of data. In modern organizations, users of data are often completely removed from the data sources. Many people unly need read access to data, but still need a very rapid access to a larger volume of data than can conveniently be downloaded to the desktop. Often such data romes from multiple databases. Because many of the analyses rectorned are recurrent and predictable, software vendors and systems support staff have begin to design systems to support these functions. At present there is a great need to provide decision makers from middle management upward with information at the correct level or detail to support decision making. Data machousing, online solutrical processing (CO,A,I), and data mining provide this functionality. We already gave an introduction to data mining techniques in Chapter 27. In this chapter we give a broad overview of data warehousing and OLAP technologies

# 28.1 INTRODUCTION, DEFINITIONS, AND TERMINOLOGY

In Chapter 1 we defined database as a collection of related data and a database system as a database and database software together. A data warehouse is also a collection of information as well as a supporting system. I lowever, a clear distinction exists. Traditional databases are transactional federated object-oriented, network, or hierarchical). Data warehouses have the distinguishing characteristic that they are mainly intended for decision-support applications. They are optimized for data retrieval, not routine transaction processing

Because data watchouses have been developed in numerous organizations to meet particular needs, there is no single, canonical definition of the term data watchouse. Professional magazine articles and books in the popular press have elaborated on the meaning its a variety of ways. Vendors have capitalized on the popularity of the term to help mader a variety of related products, and constituants have provided a large variety of services, all under the data watchousing bonnet. However, data watchouses are quite distinct from tradmonal databases in their structure, functioning, justormance, and purpose.

W. 11 Introot characterized a data warehouse as "a subject-oriented, integrated, nonvolatile, time-variant collection of data in support of management's decisions." Data warehouses provide access to data for complex analysis, knowledge discovery, and decision making. They support high-performance demands on an oreanisation's data and information. Several types of applications—OLAP, USS, and data mining applications—are supported. We define each of chose next

OUAP (online analytical processing) is a term used to describe the analysis of complex data from the data warehouse. In the hands of stalled knowledge workers, OLAP tools use distributed computing capabilities for analyses that require more storage and processing power than can be computedly and efficiently located on an individual desktop.

D53 Idecision-support systems) also known as EIS (excentive information quiens) (not to be confused with emergence integration systems) support an organization's leading decision makers with higher level data for complex and important decisions. Data mining (which we discussed in decail in Chapter 27) is used for knowledge discovery, the process of searching data for imparticipated new knowledge.

Traditional databases support online transaction processing (ODP), which includes insertions, updates, and deletions, while also supporting information query requirements. Traditional relational databases are optimized to process queries that may traich a small part of the database and transactions that deal with insertions or updates of a few rupleper relation to process. This, they cannot be optimized for C0, OP, 1985, or data mining Ps contrast, data warehouses, are designed precisely to support efficient estimation processing, and presentation for analytic and decision-making purposes. In comparison to traditional databases, data watehouses generally contain, very large amounts of data from multiple sources that may include databases from different data models and sometimes nles acquired from independent systems and platforms.

<sup>6</sup> known (1982) has been credired with manafly using the term data warehouse.

# 28.2 CHARACTERISTICS OF DATA WAREHOUSES

To discuss data warehouses and distinguish them from transactional databases calls for an appropriate data model. The multidimensional data model texplained in more detail in Section 28.3) is a good fit for CLAP and decision-suggest technologies. In contrast to multidatabases, which provide access to disjoint and usually beterogeneous databases, a data warehouse is troquently a store of integrated data from nultiple sources, processed for storage in a multidimensional model. Unlike most transactional databases, data warehouses typically support time series and strend analysis, both of which require more historical data than is generally maintained in transactional databases.

Compared with transactional databases, data warehouses are nonvolatile. That means that information in the data warehouse changes fat less often and may be regarded as non-real-time with periodic updating. In transactional systems, transactions are the unit and are the agent of change to the database, by contrast, data warehouse information is much more coarse grained and is refreshed according to a careful choice of refresh policy, usually incremental. Warehouse updates are handled by the warehouse's acquisition component that provides all required preprocessing.

We can also describe data warehousing more generally as "a collection of decision support technologies, aimed at enabling the knowledge worker (executive, manager, nuclyst) to make better and faster decisions."<sup>3</sup> Figure 28.1 gives an overview of the conceptual structure of a data warehouse. It shows the entire data warehousing process. This process includes possible cleaning and reformating of data before its warehousing. At the back end of the process, OLAP, data mining, and 1%% may generate new relevant information such as rules; this information is shown in the figure going back into the warehouse. The figure also shows that data sources may include files.



FIGURE 28.1 Example transactions in market-basket model.

Croudhum and Daya! (1997) provide an excellent natorial on the topic, with this as a starting definition. Data watchouses have the following distinctive characteristics.<sup>1</sup>

- multidimensional conceptual view
- generic dunensionality
- unlimited dimensions and sugregation levels.
- unrestricted cross-dimensional operations
- dynamic sparse matrix handling
- client-server architecture
- malm-user support
- accessibility
- transparency
- intuitive data manipulation;
- consistent reporting performance
- flexible reporting

Because they encompass large volumes of data, data watebroacs are generally in order of magnitude (sometimes two orders of magnitude) larger than the source databases. The sheer volume of data (likely in be in terabytes) is an issue that has been dealy with through enterprise-wide data watebroacs, viatual data watebroacs, and data matter

- Enterprise-wide data warehouses are huge projects requiring massive investment of time and resources.
- Virtual data warehouses provide views of operational databases that are materialized for efficient access.
- Data marts generally are targeted to a subset of the organization, such as a department, and are more tightly locused.

# 28.3 DATA MODELING FOR DATA WAREHOUSES

Multidimensional models take advantage of inherent relationships in data to popular, data in multidimensional matrices called data cubes. (These may be called hypercubes if they have more than three dimensions.) For data that lends itself to dimensional formating, query performance in multidimensional matrices can be much better than in the relational data model. Three examples of dimensions in a corporate data warehouse would be the corporation's fiscal periods, products, and regions

A standard spreadsheet is a two-dimensional matrix. One example would be a spreadsheet of regional sales by product for a particular time period. Dioducts could be

Codd (1993) couled the term OLAP and metricored these character sites. We have reardered Codd's original list.

shown as tows, with sales revenues for each region comprising the columns. (Figure 26.2 shows this two-dimensional organization.) Adding a time dimension, such as an organization's fiscal quarters, would produce a three-dimensional matrix, which could be represented using a data cube

In Figure 28.3 there is a direc-dimensional data cube that organizes product sales data by fiscal quarters and sales regions. Each cell could contain data for a specific product,



FIGURE 28.2 Two-dimensional matrix pipilel.



FIGURE 28.3 A three-dimensional data cube model.

specific fiscal quarter, and specific region. By including additional dimensions, a data hypercube could be produced, although more than three dimensions cannot be easily visualized at all or presented graphically. The data can be queried directly in an corphination of dimensions, hypersong complex database queries. Tools exist for viewing data according to the over's choice of dimensions.

Changing from one dimensional hierarchy (orientation) to another is easily accomplished in a data tube by a technique called **pivoting** (also called rotation). In this technique the data cube can be thought of as rotating to show a different orientation of the axes. For example, you might given the data cube to show regional sales revenues as rows, the fiscal quarter revenue totals as columns, and the company's products in the third dimension (Figure 28.4). Hence, this technique is equivalent to having a regional sales table for each product separately, where each table shows quarterly sales for that product region by region.

Multidimensional models load themselves readily to hierarchical views in what is known as roll-up display and drill-down display. **Roll-up display** moves up the hierarchy, grouping into larger units along a dimension to ge summing weekly data by quarter, or by yeark. Figure 28.5 shows a roll-up display that moves from individual products to a croaser grain of product categories. Shown in Figure 28.6, a drill-down display provides the opposite capability, furnishing a finer-grained view, perhaps disaggreeating country sake by region and then regional sales by subregion and also breaking up products by styles.

The multidimensional storage model involves two types of tables, dimension tables and fact tables. A dimension table consists of tuples of attributes of the dimension. A fact table can be thought of as having tuples, one per a recorded fact. This fact contains some measured or observed variable(s) and identifies at (theos) with pointers to dimension



FIGURE 28.4 Pivoted version of the data cube from Figure 26.3.



FIGURE 28.5 The roll-up operation.



FIGURE 28.6 The drill-down operation.

tables. The fact table contains the data, and the dimensions identify each tople in that data. Figure 28.7 contains an example of a fact table that can be viewed from the perspective of multiple dimension tables.

Two common multidimensional schemas are the star schema and the snowllake schema. The star schema consists of a fact table with a single table for each dimension (Figure 28.7). The snowflake schema is a variation on the star schema is which the dimensional tables from a star schema are organized into a hierarchy by normalizing them (Figure 28.8). Some installations are normalizing data warehouses up to the third normal form so that they can access the data warehouse to the fixest level of detail. A fact constellation is a set of fact tables that share some dimension tables. Figure 28.9 shows a fact constellation with two fact tables, business results and mainess forecast. These share the dimension table called product. Fact constellations limit the possible queries for the warehouse.



HOURE 28.7 A star schema with fact and dimensional tables



DIMENSION TOBLES

FIGURE 28.8 A snowilake schema.

Data watchouse statage also unlizes indexing techniques to support high performance access (see Chapter 5 for a discussion of indexing). A technique called bitmap indexing constructs a bit vector for each value in a domain (column) being indexed. It works very well for domains of low cardinality. There is a 1 bit placed in the



FIGURE 28.9 A fact constellation.

ith position in the vector if the *j*th row contains the value heing indexed. For example, internet an inventory of 100,000 cars with a hittinap index on car size. If there are four car sizes—economy, compact, midsize, and follstize—there will be four bat vectors, each containing 100,000 bits (12.5 K) for a total index size of 50K. Bitmap indexing can provide considerable impar/output and storage space advantages in low-cardinality domains. With bit vectors a bitmap index can provide dramatist improvements in comparison, aggregation, and join performance.

In a star schema, dimensional data can be indexed to tuples in the fact table by join indexing. Join indexes are traditional indexes to maintain relationships between primary key and foreign key values. They relate the values of a dimension of a star schema to rows in the fact table. For example, consider a sales fact table that has city and facal quarter as dimensions. If there is a join index on city, for each city the join index maintains the tuple fDs of tuples containing that city. Join indexes may involve multiple dimensions.

Data warehouse storage can facilitate access to summary data by taking lutther advantage of the runavolatility of data warehouses and a degree of predictability of the analyses that will be performed using them. Two approaches have been used: (1) smaller tables including summary data such as quarterly sales or revenue by product line, and (2) encoding of level (e.g., weekly, quarterly, annual) into existing tables. By comparison, the overticeal of creating and maintaining such aggregations would likely be excessive in a volatile, transaction-oriented database.

# 28.4 BUILDING A DATA WAREHOUSE

In constructing a data watchouse, buildets should take a broad view of the anticipated use of the watchouse. There is no way to anticipate all possible queries or analyses during the design phase. However, the design should specifically support **ad-bot querying**, that is, accessing data with any meaningful combination of values for the attributes in the dimension or fact tables. For example, a marketing-intensive consumer-products company would require different ways of organizing the data warehouse that would a nonprofit chapty focused or, fund raising. An appropriate schema should be chosen that reflects anticipated usage.

Acquisition of data for the warehouse involves the following steps:

- The data must be extracted from incluiple, heterogeneous sources, for example, databases or other data feeds such as those containing financial market data or environmental data.
- Data must be formatted for consistency within the watchouse. Nomes, meanings, and domains of data from unrelated sources must be reconciled. For instance, subsidiary companies of a large corporation may have different fiscal calendars with quarters ending on different dates, making it difficult to aggregate financial data by quarter. Various credit cards may report their transactions differently, making it difficult to compute all credit sales. These format inconsistencies must be resolved.
- The data must be cleaned to ensure validity. Data cleaning is an involved and complex process that has been identified as the largest labor-domanding component of data wateleouse construction. For input data, cleaning must occur before the data is loaded into the warehouse. There is nothing about cleaning data that is specific to data warehousing and that could not be applied to a host database. However, since input data must be examined and formatted consistently, data watebouse puddes should take this opportunity in check for validity and quality. Recognizing enuneous and muching lefe data is difficult to automate, and cleaning that requires automatic errar correction can be even tougher. Some especial such as domain checking, are easily coded into data cleaning commes, but automatic renegration of other data problems can be more challenging. (For example, one might require that City - 'San Francisco' together with State = 'CT' be recognized as an incorrect combination.) Atter such problems have been taken care of, similar data from different sources must be contributed for loading into the watehouse. As data managers in the organization discover that their data is being cleaged for joing through watchause, they will likely want to upgrade their data with the cleaned data. The process of returning cleaned data to the source is called backflushing (see Figure 28.1).
- The data must be fitted into the data model of the watchouse. Data from the various sources must be installed in the data model of the watchouse. Data may have to be converted from relational, object-oriented, or legacy databases (network and/or hierarchical) to a multidimensional model.
- The data must be loaded into the warehouse. The sheer volume of data in the warehouse makes loading the data a significant task. Monitoring tools for loads as well as methods to recover from incomplete or memorical loads are required. With the hoge volume of data in the warehouse, incremental updating is usually the only feasible approach. The refresh policy will probably emerge as a compromise that takes into account the answers to the following questions:
  - How ap-to-date must the data be?
  - Can the warehouse go off-hind, and for how hing?

- What are the data interdependencies?
- What is the storage availability?
- What are the distribution requirements (such as for replication and partitioning)?
- What is the loading time (including cleaning, lormating, copying, transmitting, and overhead such as index rebuilding)?

As we have said, databases must strike a balance between efficiency or transaction processing and supporting query requirements (ad hoc oser requests), but a data warehouse is typically optimized for access both a decision maker's needs. Data storage in a data warehouse reflects this specialization and involves the following processes:

- Storing the data according to the data model of the warehouse
- · Creating and maintaining required data structures
- Creating and maintaining appropriate access paths
- · Providing for time variant data as new data are added
- Supporting the updating of warehouse data
- Refreshing the data
- Purging dara

Although adequate time can be devoted initially to constructing the watehouse, the sheer volume of data in the watehouse generally makes it impossible to simply teload the watehouse in its entirety later on. Alternatives include selective (partial) refreshing of data and separate watehouse vetsions (requiring double storage capacity for the watehouse!) When the watehouse uses an incremental data refreshing mechanism, data may need to be periodically puged; for example, a watehouse that maintains data on the previous (webye business quarters may periodically puge as data each year.

Data warehouses must also be designed with full consideration of the environment in which they will reside. Important design considerations include the following:

- Usage projections
- The lat of the data model
- Characteristics of available sources
- Design of the metodata component
- Modular component design
- Design for manageability and change
- Considerations of distributed and parallel architecture

We discuss each of these in rum. Warehouse design is initially driven by usage projections; that is, by expectations about who will use the watehouse and in what way. Choice of a data model to support this usage is a key initial decision. Usage projections and the characteristics of the watehouse's data sources are both taken into account. Modular design is a practical necessity to allow the watehouse to evolve with the organization and its information environment. In addition, a well-built data watehouse

must be designed for maintainability, enabling the warehouse managers to effectively plan for and manage change while providing optimal support to users.

You may recall the term meradata from Chapter 2, metadata was defined as the description of a database including its schema definition. The metadata repository is a key data watchouse component. The metadata repository includes both rechnical and husiness metadata. The first rechnical metadata, covers details of acquisition processing, storage structures, data descriptions, watchouse operations and maintenance, and access support functionality. The second, business metadata, includes the relevant business rules and organizational details supporting the watchouse.

The architecture of the organization's distributed computing environment is a major determining characteristic for the design of the warehouse.

There are two basic distributed architectures: the distributed wirehouse and the federated warehouse. For a distributed warehouse, all the issues of distributed databases are relevant, for example, replication, partitioning, communications, and consistence concerns. A distributed architecture can provide benefits particularly important to warehouse performance, such as improved load balancing, scalability of performance, and higher availability. A single replicated metadata repository would reside at each distribution site. The idea of the federated warehouse is like that of the federated database: a decentralized confederation of autonomous data warehouses, each with its riven metadata repository. Given the magnitude of the challenge inherent to data warehouses, it is likely that such federations will consist of smaller scale components, such as data marts. Large ingravations may choose to federate data marts rather than build hinge data warehouses.

# 28.5 Typical Functionality of a Data Warehouse

Data watchouses exist to facilitate tomplex, data-intensive, and frequent ad hoc queries. Accordingly, data watchouses must provide far greater and more efficient query support than is demanded of transactional databases. The data watchouse access component supports enhanced spreadsheet functionality, efficient query processing, structured queries, ad hoc queries, data initing, and materialited views. In particular, enhanced spreadsheet functionality includes support for state-of-the-art spreadsheet applications (e.g., MS Excel) as well as for OLAP applications programs. These offer preprogrammed functionalities such as the fullowing:

- Roll-up: Data is summarized with increasing generalization (e.g., weekly to quarterly to annually).
- Drill-down: Increasing levels of detail are revealed (the complement of roll-up).
- Prvot: Cross tabulation (also referred as totation) is performed.
- Slice and dice: Ferfining projection operations on the dimensions.
- Surring: Data is sorted by ordinal value.

- Selection: Data is available by value or range.
- Derived (computed) attributes: Attributes are computed by operations on stored and derived values.

Because data watchnuses are free from the restrictions of the transactionuenvironment, there is an increased efficiency in query processing. Anyong the tools and techniques used are query transformation, index intersection and innon, special ROLAP (relatingued CCAP) and MOLAP (inultidimensional OLAF) functions, SQL extensions, advanced jum methods, and intelligent scatting (as in piggy-backing multiple queries).

Improved performance has also been attained with parallel processing. Parallel server architectures include symmetric multiprocessor (SMP), cluster, and massively garallel processing (MPP), and combinations of these.

Knowledge workers and decision makers use rools ranging from parametric queries to ad hot queries to data mining. Thus, the access component of the data watchouse must provide support for structured queries (both parametric and ad hot). These cogether make up a managed query environment. Data mining itself uses rechniques from statistical analysis and artificial intelligence. Statistical analysis can be performed by advanced spreadsheets, by sophisticated statistical analysis software, or by custom-written programs. Techniques such as lagging, moving averages, and regression analysis are also commanify employed. Artificial intelligence techniques, which may include genetic algorithms and neural networks, are used for classification and are employed to discover knowledge from the data watehouse that may be unexpected or difficult to specify in queries. (We meat data mining in detail in Cloquer 27.)

# 28.6 DATA WAREHOUSE VERSUS VIEWS

Some people have considered dota warehouses to be an extension of database views. Earher we mentioned materialized views as one way of neeting requirements for improved access to data (see Chapter 8 for a discussion of views). Materialized views have been explored for their performance enhancement. Views, however, provide only a subset of the functions and capabilities of data warehouses. Views and data warehouses are alike in that they both have read-only extracts from databases and subject-orientation. However, data warehouses are different from views in the following ways:

- Data warehouses exist as persistent storage instead of bring materialized on demand.
- Data warehouses are not usually relational, but rather multidimensional. Views of a relational database are relational.
- Data warehouses can be indexed to optimize performance. Views cannot be indexed independent of the underlying databases.
- Data watchouses characteristically provide specific support of functionality, views cannot.
- Data watchouses provide large amounts of integrated and often temporal data, generally more than is contained in one database, whereas views are an extract of a database.

# 28.7 PROBLEMS AND OPEN ISSUES IN DATA WAREHOUSES

### 28.7.1 Difficulties of Implementing Data Warehouses

Some significant operational issues arise with data watchensing: construction, administration, and quality control. Project management—the design, construction, and implementation of the watchense—is an important and challenging crassideration that should nut be inderestimated. The building of an enterprise-wide watchense in a large regarization is a major undertaking, potentially taking years from conceptualization to implementation. Because of the difficulty and amount of lead time required for such an undertaking the widespread development and deployment of data marts may provide an attractive alternative, especially to those organizations with urgent needs for (30AP, DSS, and/or data mining support.

The administration of a data watchouse is an intensive entriprise, proportional to the size and complexity of the watchouse. An organization that attempts to administer a data watchouse must realistically understand the complex nature of its administration. Although designed for read-access, a data watchouse is no more a static structure then any 1st its information sources. Source databases can be expected to evolve. The watchouse's schema and acquisition component must be expected to be updated in foundle these evolutions.

A significant issue in data watehousing is the quality control of data. Both quality and consistency of data are major concerns. Although the data passes through a cleaning function during acquisition, quality and consistency remain significant issues for the database administrator. Melding duta from heterogeneous and disparate sources is a major challenge given differences in moving, domain definitions, identification numbers, and the like. Every time a source database changes, the data watehouse administrator must consider the possible interactions with other elements of the watehouse.

Usage projections should be estimated conservatively prior to construction of the data watchause and should be revised continually to reflect current requirements. As utilization patterns become clear and change over time, storage and access paths can be tuned to remain optimized for suggest of the organization's use of its warehouse. This activity should continue throughout the life of the warehouse in order to remain applicate should also be designed to accommodate addition and attriviou of data sources without major redesign. Sources and source data will evolve, and the warehouse must accommodate such change. Fitting the available source data into the data model of the warehouse will be a continual challenge, a task that is as much an as science. Because there is continual rapid change considerably over time. Additionally data warehousing technology itself will continue to evolve for some time so that component structures and functionalities will continuely be apgraded. This certain change is excellent metivation for having fully medular design of components.

Administration of a data watchouse will require far broader skills than are needed for traditional database administration. A team of highly skilled reclimital experts with overlapping areas of expertise will likely be needed, rather than a single individual. Like database administration, data warehouse administration is only partly technical; a large part of the responsibility requires working effectively with all the members of the organization, with an interest in the data warehouse. However difficult that can be at times for database administrators, it is that much more challenging for data warehouse administrators, as the scope of their responsibilities is considerably broader.

Design of the management function and selection of the management team for a database warehouse are tructial. Managing the data warehouse in a large organization will surely be a major task. Many commercial tools are obready available to support management functions. Effective data warehouse management will certainly be a team function, requiring a wide set of rechnical skills, careful coordination, and effective leadership. Just as we must prepare for the evolution of the warehouse, we must also recognize that the skills of the management team will, of necessary, evolve with it.

### 28.7.2 Open Issues in Data Warehousing

There has been much marketing hyperbole surrounding the term "data warehouse": the exaggerated expectations will probably subade, but the concept of imegrated data collections to support sophisticated analysis and decision support will undoubserly endure.

Data warehousing as an active research area is likely to see increased research arrivity in the near future as warehouses and data marts proliferare. Old problems will receive new emphasis; for example, data cleaning, indexing, partitioning, and views could receive received attention.

Academic research into data warehousing technologies will likely focus on automating aspects of the warehouse data contently requise significant manual intervention, such as the data acquisition, data quality management, selection and construction of appropriate access paths and structures, self-maintainability, functionality, and performance optimization. Application of active database functionality (see Section 23.1) into the warehouse is likely also to receive considerable ottention. Incorporation of domain and business aules appropriately into the warehouse creation and maintenance process may make it mote intelligent, relevant, and self-governing.

Commercial software for data watchousing is already available from a number of vendors, focusing principally on management of the data watchouse and OLAP/PES applications. Other expects of data watchousing, such as design and data acquisition tespecially cleaning), are being addressed principally by teams of in-house IT managers and consultants.

# 28.8 SUMMARY

In this chapter we surveyed the field known as data watchousing. Data watchousing can be seen as a process that requires a variety of activities to precede it. In contrast, data mining (see Chapter 27) may be thought of as an activity that draws knowledge from an existing data watchouse. We instruduced key concepts related to data watchousing and we

discussed the special functionality associated with a multidimensional vow of data. We also discussed the ways in which data warehouses supply decision makers with information at the correct level of detail, based on an appropriate organization and perspective

### **Review Questions**

- 28.1. What is a data warehouse? How does it differ from a database?
- Define the terms: OLAP (Online Analytical Processing), ROLAP (Relational OLAP), and MOLAP (Multidimensional USAP), DSS (Decision Support Systems)
- 28.3. Describe the characteristics of a data watchouse. Divide them into functionality of a watchouse and advantages users derive from it.
- 28.4. What is the includimensional data model? How is it used in data watchousing?
- 28.5. Define these terms: Star Schema, Snoveflake Schema, Fact Constellation, Data Marts.
- 28.6. What types of indexes are built for a watchouse? Illustrate the uses for each with an example.
- 28.7. Describe the steps of building a wardiouse.
- 28.8. What considerations play a major role in the design of a watehouse!
- 28.9. Describe the functions a user can perform on a data warehouse and illustrate the results of these functions on a sample multidimensional data warehouse.
- 26.10. How is the concept of a relational view related to data watchwas and dota mats.' In what way are they different'
- 28.11. List the difficulties in implementing a data watchoose.
- 28.12. List the open issues and research problems in data watchousing.

## Selected Bibliography

Data warehousing has been in a very populat topic and has appeared in many publications in the last tew years. Inmen (1992) is credited for giving this term wide acceptince (field (1993) popularized the term online analytical processing (OLAP) and defined a set of characteristics for data warehouses to support OLAP. Mattison (1996) is one of the severic linuks on data warehousing rhat gives a comprehensive analysis of techniques available in data warehouses and the strategies companies should use in deploying them. Bischoff and Alexander (1997) is a compilation of advice from experts. Chaudhari and Daval (1997) give an excellent turorial on the ropin, while Widom (1995) points to a number of outstanding research problems.



# 29

Emerging Database Technologies and Applications

Throughout this block we have discussed a variety of asces related to the modeling. design, and functions of databases as well as to the internal structure and performance issues related to database management systems. In Chapter 26 we covered the internet dapabases that provide universal access to data and discussed the use of XML that will facilitate development of applications involving non-y distributed databases and different DBMS platforms. In the previous two chapters we considered variations of database management technology, such as data mining and data warehouses, that provide very large databases and tools for decision support. We now turn our attention in this chapter to two categories of continuously evolving developments in the database field: (1) emerging database technologies, and (2) the major amplication domains. We do not claim to do su exhaustively and only address some prominent rechnological and application advances. The first deals with creating new functionality in OBMss so that a variety of new applicaruous can be supported, including mobile databases to allow users widespread and flexible access to data while being mobile, and multimedia databases providing surport for storage and processing of matrimedia information. Sections 29.1 and 29.2 will briefly introduce and discuss the issues and approaches to solving the specific problems that arise in the mobile and multimedia database rechnologies.

We next consider two application domants that have instancally relied upon manual pricessing of file systems, or calored system valutions. Section 29.3 discusses geographic information systems, which deal with geographic data alone or spatial data combined with propagatial data, such as census counts. Section 29.4 discusses biological datalises

and their applications, particularly containing genetic data on different organisms, including the human genome data. A common characteristic of all these applications is the domain-specific nature of data in each specific application domain. Furthermore, they are all characterised by their "static" nature— a situation where the end user can only retrieve from the database; updatang with new information is limited to database domain experts who supervise and analyze the new data heing entered.

# 29.1 MOBILE DATABASES<sup>1</sup>

Becent advances in portable and wireless technology have led to mobile computing, a new dimension in data communication and processing. Portable computing devices coupled with wireless communications allow clients to access duto from virtually onywhere and at any time. This feature is especially useful to geographically dispersed organizations. Typical examples might include electromic valets, news reporting, broketage services, and automated salesforces. However, there are a number of hardwate and software problems that must be resolved before the capabilities of mobile computing can be fully utilized.

Some of the software problems—which may involve data management, transaction management, and database recovery—have then origins in distributed database systems. In mobile computing, however, these problems are more difficult, mainly because of the limited and intermittent connectivity allorded by wareless communications, the limited life of the power supply (battery) of mobile units, and the changing topology of the network. In additions, mobile computing introduces new architectural possibilities and chailenges.

### 29.1.1 Mobile Computing Architecture

The general architecture of a mobile platform is illustrated in Figure 29.1. It is a distributed architecture where a number of computers, generally referred to as Fixed Hosts and Base Stations, are interconnected through a high-speed wired network. Fixed hosts are general purpose computers that are not typically equipped to manage mobile units but can be configured to do so. Base stations function as gateways to the fixed network for the Mobile Units. They are equipped with wireless interfaces and offer network access services of which mobile units are chents.

Wireless Communications. The wireless medium on which mobile units and base stations communicate have bandwidths significantly lower than those of a wired network. The current generation of wireless technology has data rates that range from the tens to hundreds of kilobits per second (2G cellular telephony) to tens of megabus per second (wireless Ethernet, popularly known as W(Fi). Modern (wired) Ethernet, by comparison, provides data rates on the order of hundreds of megabus per second

<sup>1.</sup> The contribution of Waigen Yee and Wanxia Xie to this section is appreciated.



FIGURE 29.1 A general architecture of a mobile platform (Adapted from Donham) and Helal (1995)).

Besides data rates, other characteristics also distinguish wireless connectivity options. Some of these characteristics include range, interference, locality of access, and support for packet switching. Some wireless access options allow scattless training thoughout a geographical region (e.g., collular networks), whereas WiFi networks are localized around a base station. Some wireless networks, such as WiFi and Bluetooth, use indicensed aroas of the frequency spectrum, which may cause interference with other opplications, such is confless relephones. Finally, modern wireless networks can transfer data in units called packets, that are commonly used in wired networks or order to conserve bundwidth. Wireless applications must consider these characteristics when choising a conneuroration option. For example, physical objects block infrared frequencies. While inconvenient for wine upplications, such blockage allows for service wireless communications within a closed norm.

Client/Network Relationships. Mobile anits can nowe heely in a geographic mobility domain, an area that is circumscribed by wireless network coverage. To manage the mobility of units, the critic geographic mobility domain is divided into one or more smaller domains, called cells, each of which is supported by at least one base station. The rindide discipling requires that the movement of mobile units by intrestricted throughout the cells of a generaphic mobility domain, white maintaining information access contiguity—i.e., movement, especially intercell movement, does not negatively affect the data retrieval process.

The communication architecture just described is designed to give the nubile unit the supression that it is arrached in a lixed network, coulating a multitonal client-server architecture. Witeless communications, however, make other architectures possible. One phremative is a mobile ad-hoc network (MANET), illustrated in Figure 29.2.<sup>3</sup> In a MANET, co-located mobile anits do not need to communicate via a fixed network, but instead, form their own using cost-effective technologies such as Bluerooth. In a MANET, mobile units are responsible for noting their own data, effectively acting a base stations as well as chearts. Moreover, they must be robust enough to handle changes in the network togology, such as the arrival or departure of other mobile units.

MANEL upplications can be considered as peci-to-peer, meaning that a mobile onit is simultaneously a client and a server. Transaction processing and data consistency control become areas oblicoit since there is no central control in this architecture. Resource decovery and data moting by mobile units make computing in a MANET even more complicated. Sample MANET applications are multi-user games, shared whiteboards, distributed calendars, and bartic information sharing. The expectation is that these networks and related applications will become dominant in a few years. Currently MANETs are an active research area in both academia and industry. This research is still in its infancy, so the following discussion will forms on the basic mobile computing architecture described previously.



FIGURE 29.2 The architecture of a mobile ad hoc network.

 This architecture is losed on the IETF proposal in IETF(1969) with comments by Carson and Market (1996)

### 29.1.2 Characteristics of Mobile Environments

As we discussed in the previous section, the characteristics of anoble computing include high communication latency, intermittent wireless connectivity, limited battery life, and, of course, changing client location. Latency is caused by the processes unique to the wireless medium, such as coding data for wireless transfer, and tracking and filtering wireless signals at the receiver. Battery life is directly related to battery size, and indirectly related to the mobile device's capabilities. International connectivity can be intentional or unintentional. Unintentional disconnections happen in areas intreless signals cannot reach, e.g., elevator shafts or subway termels. International disconnections near by user intente.g., during an anglane takeoff, or when the mobile device is provered down. Finally, clients are expected to move, which afters the network topology and may cause their data requirements to change. All of these characteristics impact data management, and robust mobile applications must consider them in their design.<sup>4</sup>

To compensate for high latences and oureholde connectivity, clients cache replicas of important, frequently accessed data, and work offline, if necessary. Besides mereasing data availability and response time, eaching can also reduce cherit power consumption by eliminating the ored to make energy-consuming wireless data transmissions for each data access.

On the other hand, the server may not be able to reach a client. A client may be unreachable because it is during—in an energy-conserving state in which many subsystems are shot down—or because it is out of range of a base station. In either case, neither client nor server can reach the other, and modifications must be made to the architecture in order to compensate for this case. Proxies for unreachable components are added to the architecture. For a client (and symmetrically for a server), the proxy can cathe updates intended for the server. When a connection becomes available, the proxy puromatically forwards these eached updates to their ultimate destination

As suggested above, mobile computing poses challenges for servers as well as chemis. The latency involved in wireless communication makes scalability a problem. Because latency due to wireless communications increases the time to service each client request, the server can bandle fewer chemis. One way servers relieve this problem is by broadcasting data whenever possible. Provideast takes advantage of a natural characteristic of radio communications, and is scalable because a single broadcast of a data item can satisfy all outstanding requests for it. For example, instead of sending weather infimitation to all chemits in a cell individually, a server can singly broadcast it periodically. Broadcast also reduces the load on the server, as chemits do not have to maintain active connections to it.

Client mobility also proce many data management challenges. First, servers must keep track of chent locations in order to efficiently mute messages to them. Second, client data should be stored in the network location that primitizes the traffit processary to access it. Keeping data in a fixed location increases access latency if the client moves "far away" from it. Finally, as stated above, the act of groving between rells must be

This architecture is based on the IETF proposal an IETF(1999), with comments by Carson and Macker (1998).

transparent to the client. The server must be able to gracefully divert the shipment of data from one base station to another, without the client notiong.

Chent mobility also allows new applications that are licenten-based. For example, consider an electronic value application that can tell a user the location of the nearest restaurant. Clearly, "nearest" is relative to the chent's cubent position, and movement can unvalidate any previously cached responses. Upon movement, the chent must efficiently invalidate parts of its cache and request updated data from the database.

### 29.1.3 Data Management Issues

From a data management standpoint, multile computing may be considered a variation of distributed computing. Mobile databases can be distributed under two possible scenarios

- The entire database is distributed manify among the wired components, possibly with full or partial replecation. A base station or fixed bost manages its own datahase with a DBMS-like functionality, with additional functionality for locating mobile units and additional query and transaction management features to meet the requirements of mobile onvironments.
- The database is distributed among wired and wireless components. Data management responsibility is shared among base statuous or fixed bosts and mobile inco.

Hence, the distributed data management usnes we discussed in Chapter 24 can also be applied to mobile databases with the following additional considerations and variations:

- Data distribution and replication: Data is unevently distributed among the base stations and mobile units. The consistency constraints compound the problem of eache management. Caches attempt to provide the most frequently accessed and updated Jaro to mobile units that process their own transactions and may be disconnected over long periods.
- Transaction models, Issues of fault rolerance and correctness of transactions are aggravared in the mobile environment. A mobile transaction is executed sequentially through several base stations and possibly on multiple data sets depending upon the movement of the mobile unst. Central coordination of transaction execution is lacking, particularly in scenario (2) above. Moreover, a mobile transaction is expected to be long-lived because of discontraction in mobile units. Hence, traditional ACID properties of transactions (see Chapter 19) may need to be modified and new transaction models must be defined.
- Query powersing: Awareness of where data is located is important and affects the cost/ benefit analysis of query processing. Query optimization is more complicated because of mobility and rapid resource changes of mobile muts. The query tesponse needs to be returned to mobile units that may be in transit or new closs cell boundaries yet must receive complete and correct query results.
- Recovery and fould tederative: The multile database environment must deal with site, media, transaction, and communication failures. Site failure of a mobile unit is fre-

quent due to funited battery power. A voluntary shordown of a mobile unit should not be treated as a failure. Transaction failures are routine during handoff when a mobile unit crosses cells. The transaction manager should be able to deal with such frequent failures.

- Mobile database design. The global name resolution problem for handling queries is compounded because of mobility and frequent shutdown. Mobile database design must consider many issues of metadata management—for example, the constant updating of location information.
- Location-based service: As clients move, location-dependent cache information may become stale. Evention techniques are important in this case. Furthermore, frequently updating location dependent queries, then applying these (spatial) queries in order to refresh the cache poses a problem.
- Dresson of labor: Certain characteristics of the molele environment force a change in the division of labor in query processing. In some cases, the effect must function independent of the server. However, what are the consequences of allowing full independent acress to replicated data? The relationship between client responsibilities and their consequences has yet to be developed.
- Scentry: Mobile data is less secure than that which is left at the fixed location. Proper rechniques for managing and authorizing access to critical data become more imporrant in this environment. Data is also more vulatile, and techniques must be able to compensate for its loss.

### 29.1.4 Application: Intermittently Synchronized Databases

One mobile computing scenario is becoming increasingly commonplace is people conduct their work away from their offices and homes and perform a wide range of activities and functions, all kinds of sales, particularly in pharmaceuticals, consumer goeds, and industrial parts; law enforcement; insurance and financial consulting and planning; real estate in property management activities; council and transportation services, and so on In these applications, a server or a group of servers manages the central database and the cherits carry laptops or paintups with a resident DBMS software to do "local" transaction activity for most of the time. The clients connect via a network or a dial-up connection (or possibly even through the Internet) with the server, typically for a short session—say, 30 to 62 minutes. They send their updates to the server, and the server must in turn enter them in its central database, which myst maintain up-to-date data and prepare appropriate comes for all clients on the system. Thus, whenever clients connect—through a process known in the industry as synchronization of a client with a server—they receive a Fatch of undates to be installed on their local database. The primary characteristic of this scenario is that the clients are mostly disconnected; the server is not necessarily able to reach them. This environment has problems similar to those in distributed and chentserver databases, and some from mobile databases, but presents several additional research problems for investigation. We refer to this environment as Intermittently Synchronized

Database Environment (ISD56), and the corresponding databases as Intermittently Synchronized Databases (ISD86).

Together, the following characteristics of ISDB's make them district from the mobile databases we have discussed thus fat:

- A chent connects to the server when it wants to exchange updates. This communication may be meant one-on-one communication between the server and the olight—or malmass—one sender or server may periodically communicate to a set of receivers or ngdate a group of clients.
- A server cannot connerr to a client at will.
- bases of wireless versus wired chent contractions and power conservation are genorally immaterial.
- 4 A chent is free to manage its own data and transactions while it is disconnected. It can also perform its own recovery to some extent.
- A client has multiple ways of connecting to a server and, in case of many servers, may choose a particular server to connect to based on proximity, communication nodes available, resources available, etc.

Because of such differences, there is a need to address a number of problems related to 1803s that are different from those optically involving mobile database systems. These include server database design for server databases, consistency and synchronization management among client and server databases, transaction and update processing, efficient use of the server bandwidth, and achieving scalability in the ISDB environments.

### 29.1.5 Selected Bibliography for Mobile Databases

There has been a sudden surge of interest in mobile computing, and research on mobile databases has had a significant growth for the last five to six years. The June 1995 issue of Byte magazine discusses many aspects of mobile computing. Among brocks written on this topic, Dhawari (1997) is an excellent source on minhule computing. Wireless networks and their future are diversed in Holizman and Goodman (1993). Inuclinski and Bademath (1994) provide a good survey of multile database reases and also discuss in huselinski and Badringth (1992) data and metadata allocation in a mobile architecture. Darigam and Helpl (1995) discuss problems of query processing, data distribution, and transaction management for mohile databases. Foreman and Zahorjan (1994) destribe the capabilities and the problems of mobile computing and make a convinting argument in its favor as a viable solution for many information system applications of the future. Piotesta and Semara-(1995) describe all assocrs of mobile database problems and solutions. Chrosophis (1993) describes a transaction model that is designed to operate in an environment with mobile chents. In particular, this model allows a chert to share the transaction processing load with provaes in order to facilitate mobility. Bertino et al. (1999) discuss approaches to fault rolerance and techwerv in mobile databases. Acharya et al. (1995) consider broadcast schedules that minimize average query latency, and explore the impact of such schedules on optimal chent caching strategies. Miligicio et al. (2002) present a tutorial on peer-ro-peer computing. Corson and Macker (1999) is a response to IETF(1999) report that discusses the mobile ad-hoc networking protocol performance issues. Broadcasting (or pushing) data as a means of scalably disseminating information to clients is covered in Yee et al. (2002). Chintalapari et al. (1997) provide an adaptive location management algorithm. Jensen et al. (2001) discuss data management issues as they pertain to location-based services. Wolfson (2001) describes a novel way of efficiently modeling object mobility by describing position using trajectories instead of points. For an initial discussion of the ISO6 scalability issues and an approach by aggregation of data and grouping of clients, see Mahajari et al. (1995). Specific aggregation algorithms for grouping data at the server in ISOB applications are described in Yee et al. (2001). Gray et al. (1993) discuss ISOB update conflicts and resolution techniques under various ISDB architectures. Brothart et al. (1999) go into forther detail about deferred synchronication algorithms for replicated data.

# 29.2 MULTIMEDIA DATABASES

In the years ahead multimedia information systems are expected to dominate our daily loves. Our houses will be wired for bandwidth to harville interactive multimedia applications. Our high-definition TV/computer workstations will have access to a large number of databases, including digital Horaries, image and video databases that will distribute visit amounts of multisource multimedia content.

### 29.2.1 The Nature of Multimedia Data and Applications

In Section 24.3 we decussed the advanced modeling issues related to multimedia data. We also examined the processing of multiple types of data in Chapter 22 in the context of object relational DBMSs (OROBAISs). DBMSs have been constantly adding to the types of data they support. Today the following types of inultimedia data are available in current systems:

- Text: May be formatted or unformatted. For ease of parsing structured documents, standards like SCML and variations such as HEML are being used.
- Graphics: Examples include drawings and illustrations that are encoded using some descriptive standards (e.g., COM PRT, postscript).
- Images: Includes drawings, photographs, and so forth, encoded in standard formus such as hitmap, (PEG, and MPEG. Compression is built into (PEG) and MPEG. These images are not subdivided into compenents. Hence querying them by content (e.g., and all images containing circles is nontrivial.
- Anfractions: Temperal sequences of image of graphic data.
- Video: A set of temporally sequenced photographic data for presentation at specified rates—for example, 30 frames per second
- Structured and/or A sequence of audio components comprising note, tune, duration, and so forth.

- Andio: Sample data generated from autol recordings in a string of bits in digitized form. Analog recordings are typically converted into digital form before storage.
- Conjourse or most instituted data: A combination of multimedia data types such as audio and viden which may be physically mixed to yield a new storage format or logically mixed while retaining original types and formats. Composite data also contains additional control information describing how the information should be rendered.

Nature of Multimedia Applications. Multimedia data may be stored, delivered, and artified in many different ways. Applications may be entegorized based on their data management characteristics as follows.

- Rejustory optications: A large atcount of multimedia data as well as metadata is striced for retrieval purposes. A central repositivy containing multimedia data may be maintenned by a D908 and may be organized into a hierarchy of storage levels - Intal disks, termary disks and tapes, optical disks, and si on. Examples include repositories or vorellive images, engineering drawings and designs, space photographs, and tablology scanned pictures.
- Presentation applications: A large number of applications involve delivery of nultimedia data subject to temporal constraints. Audio and video data are delivered this way in these applications optimal viewing or listening conditions require the DRMS to deliver data at certain rates offering "quality of service" above a certain threshold. Data is censured as it is delivered, and the in repository applications, where it may be precessed later (e.g., indimedia electronic mail). Simple mathematics, where it may be precessed later (e.g., indimedia electronic mail). Simple mathematics, complex and interactive environments in system to simulate VCR-like functionality. Complex and interactive environments are a series of in parallel. Interactive environments must support capabilities such as real-time editing acalysis or anizotating of video and audio data.
- Colleburation work using indicatedia togormation: This is a new category of applications in which engineers may execute a complex design task by merging drawings, htting subjects to design constraints, and generating new documentation, change notifications, and so forth, incolligent healthcare networks as well as telemedicine will involve dia tors collaborating among themselves, analyzing multimedia patient data and information in real time as it is generated.

All of these application areas present insjot challenges for the design of multimedial database systems

### 29.2.2 Data Management Issues

Multimedia applications dealing with thousands of images, documents, and one to deal wideo segments, and free text data depend critically on appropriate modeling of the structure and contents of data and then designing appropriate database schemas for scoring and retrieving multimedia information. Multimedia information systems are very complex and embrace a large set of issues, including the following:

- Modeling This area has the potential for applying database versus information retrieval techniques to the problem. There are problems of dealing with complex objects (see Chapter 20) made up of a wide range of types of data: numeric, text, graphic (computer-generated unage), animated graphic image, audio stream, and video sequence. Documents constitute a specialized area and deserve special consideration.
- Design: The conceptual, logical, and physical design of multimedia databases has not heen addressed fully, and it remains an area of active research. The design process can be based on the general methodology described in Chapter 12, but the performance and tuning issues at each level are far more complex.
- Storage: Storage of multimedia data on standard disklike divices presents problems of representation, compression, mapping to device hierarchies, archiving, and buffering during the input/output operation. Adhening to standards such as IPEG or SUEG is one way most vendors of multimedia products are likely to deal with this issue. In DPMSs, a "PEGP" (Binary Large Object) facility allows untyped bitmaps to be stored and remeved. Standardized software will be required to deal with synchronization and compression/decompression, and will be coupled with indexing problems, which are still in the research domain.
- Querses and retrieval: The "database" way of retrieving information is based on query languages and internal index structures. The "information retrieval" way relics strictly on keywords or prolefined index terms. For images, video data, and autor data, this opens up many issues, among them efficient query formulation, query execution, and optimization. The standard optimization rechniques we discussed in Chapter to need to be modified to work with multimedia data types.
- Performance: For inclume dia applications involving only documents and text, performance constraints are subjectively determined by the user. For applications involving video playback or audio-video synchronization, physical limitations dominate. For instance, video must be delivered at a steady rate of 60 frames per second. Techniques for query optimization may compute expected response time before evaluating the query. The use of parallel processing of data may allevrate some problems, but such efforts are correctly subject to further experimentation.

Such issues have given rise to a variety of open research problems. We look at a few representative problems now

### 29.2.3 Open Research Problems

Information Retrieval Perspective in Querying Mutimedia Databases. Modeling data content has not been an assic in database models and systems because the data has a rigid structure and the meaning of a data instance can be inferred from the schema. In contrast, information retrieval (IR) is mainly concerned with modeling the content of text documents (through the use of keywords, phraval indexes, semantic networks, word frequencies, soundex encoding, and su on) for which structure is generally neglected. By modeling content, the system can determine whether a document is relevant to a query by examining the content descriptors of the document. Consider, for instance, an insurance company's accident claim report as a multimedia object intincludes images of the accidem, structured insurance forms, and/o recordings of the parties involved in the arcident, the rest report of the insurance company's representative, and other information. Which data model should be used to represent multimedia information such as this? How should duertes be formulated against this data? Efficient execution thus becomes a complex issue, and the semantic heterogeneity and representational complexity of multimedia information gives rise as many new problems.

Requirements of Multimedia/Lippermedia Data Modeling and Retrieval. To capture the full expressive power of multimedia data modeling, the system should have a general construct that lets the user specify unks between any two arbitrary nodes. Plypermedia links, or hyperlenks, have a number of different characteristics:

- Links can be specified with or without associated information, and they may have large descriptions associated with them.
- Links can start from a specific point within a node or from the whole node.
- Links can be directional or mondirectional when they can be traversed in either direction.

The link capability of the data model should take into account all of these variations. When content-based retrieval of multimetha data is needed, the query mechanism should have access to the links and the link-associated information. The system should provide facilities for defining views over all links—private and public. Valuable contextual information can be obtained from the structural information. Automatically generated hypermedia links do not reveal anything new about the two nodes, and in contrast to manually generated hypermedia links, would have different significance. Facilities for creating and unified links, as well as developing and using navigational query languages to utilize the links, are important features of any system permitting effective use of multimedia unformation. This area is important to interlinked databases on the wWW.

The World Wide Web prevents an opportunity to access a vast amount of information via an array of unstructured and structured databases that are interlinked. The phenomenal success and growth of the web has made the problem of finding, accessing, and maintaming this information extremely challenging. For the last few years several projects are attempting to define frameworks and languages that will allow us to define the semantic content of the web that will be machine processable. The effort is collectively known by the term semantic web. The RDF (resource description framework), XHTML (Extensible Hypertext Markup Language), DAML (DARPA Agent Markup Language), and Oil (Ontology Inference Laver) are among some of its major components.<sup>4</sup> Further details are outside the scope of our discussion.

Indexing of Images. There are two approaches to indexing amages: (1) identifying objects automatically through image-processing techniques and (2) assigning index terms

4 See Fensel (2000) for an overview of these terms.

and phoses through manual indexing. An important problem in using image-processing techniques to index partness relates to scalability. The current state of the art allows the indexing of only simple patterns in images. Complexity increases with the number of recognitable features. Another important problem relates to the complexity of the query. Rules and inference mechanisms can be used to derive higher-level facts from simply possible to define in terms of a set of <a href="https://www.sulessing.com">according.com</a> the used to derive higher-level facts from simply possible to define in terms of a set of <a href="https://www.sulessing.com">active complexity of the query. Rules and inference mechanisms can be used to derive higher-level facts from simply possible to define in terms of a set of <a href="https://www.sulessing.com">active complexity active complexity possible to define in terms of a set of <a href="https://www.sulessing.com">active complexity active complexity possible to define in terms of a set of <a href="https://www.sulessing.com">active complexity active complexity possible to define in terms of a set of <a href="https://www.sulessing.com">active complexity active complexity possible to define in terms of a set of <a href="https://www.sulessing.com">active complexity active complexity possible to define in terms of a set of <a href="https://www.sulessing.com">active complexity active complexity possible to define in terms of a set of <a href="https://www.sulessing.com">active complexity active complexity possible to define in terms of a set of <a href="https://www.sulessing.com">active complexity active complexity possible to define in terms of a set of <a href="https://www.sulessing.com">active complexity active complexity possible to define in terms of a set of <a href="https://www.sulessing.com">active complexity active complexity possible to define in terms of a set of <a href="https://www.sulessing.com">active complexity active complexity possile to define in terms of a set of <a hre

The information-retrieval approach to image indexing is based on one of three indexing schemes:

- Classificatory systems: Classifies integes hierarchically into predetermined categories. In this approach, the indexet and the user should have a good knowledge of the available categories. Finer detuits of a complex image and relationships among objects in an image cannot be captured.
- 2. Keyword based sequence: Uses an indexing vocabulary similar to that used in the indexing of rextual documents. Simple facts represented in the image (like "ico-capped region") and facts derived as a result of high-level interpretation by humans (like permanent ice, recent showfall, and polar ice) can be captured.
- Entity-cut/bite-relationship systems: All objects in the picture and the relationships between objects and the attributes of the objects are identified.

In the case of text documents, an indexer can choose the keywords from the pool of words available in the document to be indexed. This is not possible in the case of visual and video data.

**Problems in Text Retrieval.** Text retrieval has always been the key feature in bioness applications and bhrary systems, and although much work has gone into some of the following problems, there remains an ongoing need for improvement, especially regarding the following issues:

- Phrase indexists: Substantial improvements can be realized if phrase descriptors (as
  opposed to single-word index terms) are assigned to documents and used in queries,
  growided that these descriptors are good indicators of document content and information need.
- Use of thestaters. One reason for the poor recall of current systems is that the vocabulary of the user differs from the vocabulary used to aidex the documents. One solution is to use a thestatrois to expand the user's query with related terms. The problem then becames one of finding a thestatrois for the domain of interest. Another resource in this context is instologies. An ontology necessarily entails or embodies some sort of world view with respect to a given domain. The world view is often conceived as a set of concepts deig, entitud, artributes, process), their definitions and their interrelationships which describe a targer world. An ontology can be constructed in two ways, domain dependent and generic. The purpose of generic ontologies is to make a ways, domain dependent and generic. The purpose of generic ontologies is to make a set of some of the set of some set.

general framework for all ( or most) categories encountered by human existence. A variety of domain ontologies such as gene antidogy (see Section 29.4) or outdogy for electronic components have been constructed.<sup>2</sup>

Revolving ambiguity. One of the reasons for low precision (the ratio of the number of relevant items retrieved to the tutal number of retrieved items) in sost information, retrieval systems is that words have multiple meanings. One way to resolve ambiguity is to use an online distionary or ontology: another is to compare the contexts in which the two words occur.

In the first three decades of DBMS development—roughly from 1965 to 1995—the pumary focus had been on the management of mostly numeric business and industrial data in the next few decades, nonnumeric textual information will probably dominate database content. The text retrieval problem is becoming very relevant in the context of HTML and XML documents. The web currently contains several fullion of these pages. Search engines find relevant documents given lists of words which is a case of fire form natural language query. Obtaining the instruct result that meets the requirements of both precision (% of remeved documents that are relevant) and recall (% of total relevant documents that are retrieved), which are standard metrics in information retrieval, remains a challenge. As a consequence, a variety of functionabites involving compation, conceptualization, understanding, indexing, and summarization of documents will be added to CBMEs. Moltimedia information systems promise to bring about a joining of disciplines that have Instructably been separate areas: information retrieval and database management.

### 29.2.4 Multimedia Database Applications

Large-scale applications of multimedia databases can be expected to encompass a large number of disciplines and enhance existing capabilities. Some important applications will be involved:

- Distances and records indiagement. A large number of industries and businesses keep very detailed seconds and a variety of documents. The data may include engineering design and manufacturing data, medical records of patients, publishing material, and insurance claim, records.
- Knowledge dimension: The pultimedia mode, a very effective means of knowledge dissemigration, will encompase a phenomenal growth in electronic books, catalogs, manuals, encyclopedias and repusitories of information on many impres.
- Education and training: Teaching materials for different audiencess from kindergarten students to equipment operators to professionals - can be designed from multimedia sources. Dirital libraries are expected to have a major influence on the way future students and researchers as well as other users will access vasi repositories of educational material.

<sup>5.</sup> A grad-discussion of oncologies is given in Usefaeld and Gruninger (1996).

- Markening, advertising, retailing, entertainment, and travel: There are virtually no limits to using multimedia information in these applications—from effective sales presentations to virtual tours of ciries and art galleries. The film industry has already shown the power of special effects in according animetions and synthetically designed onimals, aliens, and special effects. The use of predesigned stored objects in multimedia databases will expand the range of these applications.
- Real-sine control and monitoring. Coupled with active database technology (see Chapter 24), multimedia presentation of information can be a very effective means for monitoring and controlling complex tasks such as manufacturing operations, nuclear power plants, patients in intensive care units, and transportation systems.

Commercial Systems for Multimedia Information Management — There are no DBMSs designed for the sole purpose of multimedia data management, and therefore there are none that have the range of functionality required to tally support all of the multimedia information management applications that we discussed above. However, several DBMSs today support multimedia data types; these include Informix Dynamic Server, DD2 Universal database (DD5) of DM, Oracle 9 and 10, CA- (ASMINF, Sylvase, OD5 0. All of these DIMEs have support for objects, which is essential for modeling a carety of complex multimedia objects. One major problem with these systems is that the "blacks captinges, and extenders" for bandlarg multimedia data are designed in a very of box manner. The functionality is provided without much apparent attention to scalability and performance. There are produces available that operate either stand-alone of in comunction with other vendors' systems to allow retrieval of image data by content. They include Virage, Excalibur, and IBM's QBC1. Operations on multimedia need to be standardized. The MPEC-7 and other standards are addressing some of these issues.

### 29.2.5 Selected Bibliography on Multimedia Databases

Multimedia database management is becoming a very heavily researched area with several industrial projects on the way. Gresky (1994, 1997) provides two excellent tatenals on the topic. Parandak and Srivastava (1995) provide an evoluation of database systems related to the requirements of multimedia databases. Grocky et al. (1997) conrains contributed arriches including a survey on content-based indexing and remeval by Jagadish (1997). Faloutous et al. (1994) also discuss a system for image querying by content. Li et al. (1998) introduce image modeling in which an image is viewed as a hierarchical structured complex object with both semantics and visual properties. Novou et al. (1996) and Subramanian and Jajodia (1997) have written books on the topic. Lassila (1998) discusses the need for meradato for accessing mutimedia information ov the web; the semantic web effort is summaticed in Fensel (2000). Khuw (2000) and a discritation on ontology-based information retrieval. Usefold and Gruninge: (1996) is a good resource on ontologies Corcho et al. (2003) compare ontology languages and discuss methadologies to build outploges. Multimedia content analysis, acdexing, and filtering are discussed in Duritrova (1999). A sugvey of content-based multimedia retrieval is provided by Yoslotaka and Izbakawa (1999). The following WWW references may be consulted for additional information:

CA+TASMINE (Multimedia ODDMS): http://www.caccom/products/jasminse.htm Excalibur techniclogicy.http://www.excaldocom Vitage, Inc. (Content based image retrieval): http://www.vitage.com (BM's QBM: (Query by Image Content) product:

# 29.3 GEOGRAPHIC INFORMATION SYSTEMS

Geographic information systems (145) are used to collect, model, store, and analyze information describing physical properties of the geographical world. The scope of the breadly encomposes two types of data: (1) spatial data, originating from maps, digital images, administrative and political boundaries, roads, transportation networks: physical data such as rivers, soil characteristics, climatic regions, bind elevations, and (2) nanspatial data, such as social encourant data (like ceasus counts), encountie data, and sales or marketing information. GP is a rapidly developing domain that offers highly innovative opproaches to most some challenging technical demands.

### 29.3.1 GIS Applications

It is possible to double GBS into duree caregories (1) carrographic applications, (2) digital remain modeling applications, and (3) geographic objects applications. Figure 29.3 summarizes these categories.

In corrographic and terrain modeling applications, variations in sparial attributes are captured—for example, soil clustocretaries, envirolements, and air quality. In geographic objects applications, objects of interest are identified from a physical domain—for example, power plants, electoral districts, property parcels, product distribution districts, and city landmarks. These objects are related with pertinent application data—which may be, for this specific example, power consumption, voting patterns, property siles volumes, product siles volume, and traffic density.

The first two categories of GN applications require a field-based representation, whereas the third category requires an object-based one. The corresponde approach according to the third category requires an object-based one. The corresponded approach according to the the overlapping of layers of maps to combine attribute data that will allow for coample, the measuring of distances in three-dunchsional space and the reclassification of data on the map. Digital terrain modeling requires a digital representation of parts of earth's anface using latid elevations at sample points that are connected to yield a sofface model such as a three-duncesional net (connected lines in 3D) showing the surface terrain. It requires longtoes of interpolation between observed points as well as visualization. In object-based geographic applications, additional spatial functions are needed to deal with data related to roads, physical pipelines, communication cables, power lines, and such. For example, for a given region.



**FIGURE 29.3** A possible classification of GIS applications (Adapted from Adam and Gangopadhyay (1997)).

comparable maps can be used for comparison at various points of time to show changes in certain data such as locations of roads, cables, buildings, and streams.

### 29.3.2 Data Management Requirements of CIS

The functional requirements of the CIS applications above translate into the following database requirements.

Data Modeling and Representation. (1) data can be broadly represented in run formats: (1) vector and  $\{z\}$  raster. Vector data represents geometric objects such as points, lines, and polygons. Thus a lake may be represented as a polygon, a river by a series of line segments. Raster data is characterized as an urray of points, where each point represents the value of an attribute for a ceal-world location. Informally, raster images are n-dimensional arrays where each entry is a unit of the image and represents an attribute. Two-dimensional units are called peeds, while three-dimensional units are called (1606) format. Another raster format callest triangular irregular network (108) is a topological vector-based approach that models surfaces by compacting sample primes as vertices of triangles and has a point density that may vary with the roughness of the terrain. Rectangular grids (or elevation

matrices) are two-dimensional array structures. In **digital termin modeling** (0163), the model also may be used by substanting the elevation with some attribute of interest such as population density or air temperature. GIS data often includes a temporal structure in addition to a spatial structure. For example, troffic flow or average vehicular speeds in traffic may be measured every 60 seconds at a set of points in a meadway nework.

Data Analysis. Oils data undergoes various types of analysis. For example, in applications such as soil crosion studies, environmental impact studies, or hydrological ranoff simulations. DTM data may undergo various types of geomorphometric analysis—measurementsuch as slope values, gradients (the rate of change in abitude), uspect (the compass direction of the gradient), profile convexity (the rate of change of gradient), plan convexity (the convexity of contours and other parameters). When (105 data is used for decision support applications, it may undergo aggregation and expansion operations using data watchousing as we discussed in Section 28.3. In addition, geometric operations (to compute distances areas, volumes), topological operations (to compute overlaps, intersections, shortest para) and temporal operations (to compute internal-based or event-based queries) are involved Analysis involves a number of temporal and spatial operations, which were discussed in Chapter 24.

Data Integration. Give must integrate both vector and raster data from a variety of sources. Sometimes edges and regions are inferred from a taster image to form a vector model, or conversely, roster images such as aerial photographs are used to apdate vector models. Several coordinate systems such as Universal Transverse Mercator (UTN), latitude/lengitude, and local codastral systems are used to identify locations. Data originating from different coordinate systems acputes appropriate transformations. Major public sources of gaugraphic data, including the TIGER files maintained by U.S. Department of Commerce, are used for road maps by many Web-based map drawing tools (e.g., http:// maps/athro.com). Often there are high-accuracy, enrolate sport maps that have to be merged with low-accuracy, attribute-rich maps. This is done with a process called "nurber-bandling" where the user defines a set of comtent of the low accuracy map is accomplished by licing up the control points. A major integration issue is to create and maintain attribute micromation (such as an epidety or traffic flow), which can be related to and maintain attribute micromation (such as an epidety or traffic flow), which can be related to and maintain attribute and appropriate gaugraphical information over time as both evolve.

Data Capture. The first step in developing a spatial database for contregraphic modeling is to capture the two-dimensional or three-dimensional geographical information in digital Joan—a process that is sometimes impeded by source map characteristics such as resolution, type of projection, map scales, cattographic licensing, diversity of measurement techniques, and coordinate system differences. Spatial data can also be captured from remote sensors in scatellines such as Landsat, NORA, and Advanced Very High Resolution Radiometer (AVHRE) as well as SDOT TIRV (High Resolution Visible Range Instrument), which is free of interpretive bias and very accurate. For digital terrain incololing, data capture methods range from manual to fully automated. Or and surveys are the traditional approach and the most accurate, but they are very time consuming. Other techniques include photogrammetric sampling and digitizing cattographic documents.

# 29.3.3 Specific GIS Data Operations

GIS applications are conducted shrough the use of special operators such as the following:

- Interputation. This process derives elevation data for points at which no samples have been taken. It includes computation at single points, computation for a rectangular grid or along a consour, and so furth. Most interpolation methods are based on triangulation shar uses the TIN method for interpolating elevations made the triangle based on those of its vertices.
- 2. Interpretation: Digital tension modeling involves the interpretation of operations on terrain data such as ediring, smoothing, reducing details, and enhancing Additional operations involve patching in apping the borders of triangles (in TIN data), and interging, which implies combining overlapping models and resolving comflicts among attribute data. Conversions among grid models, contour models, and TIN data are involved in the interpretation of the terrain.
- 5. Proximity analysis: Several classes of proximity analysis include computations of "somes of interest" around objects, such as the determination of a hoffer around a cut on a highway. Shortest path algorithms using 2D or 3D information is an important class of proximity analysis.
- 4 Raster image processing. This process can be divided into two categories: (1) map algebra, which is used to integrate geographic features on different map layers to produce new maps algebraically: and (2) digital image analysis, which deals with analysis of a digital image for features such as edge detection and object detection. Deructing mads in a satellite image of a city is an example of the latter.
- 5 Analysis of networks: Networks occur in GIS in many contexts that must be analyzed and may be subjected to segmentations, overlays, and so on. Network overlay refers to a type of spatial join where a given network—for example, a highway network—is joined with a point database. —for example, incident focations—to yield, an this case, a profile of high-incident toodways.

Other Database Functionality. The functionality of a Ots database is also subject to other considerations.

- Extensibility 1088 are required to be extensible to accommodate a variety of constantly evolving opplications and corresponding data types. If a standard DBM(S is used, it must allow a core set of data types with a provision for defining additional types and methods for those types.
- Data quality centrol: As in many other applications, quality of source data is of paramount importance for providing accurate results to queries. This problem is particelarly significant in the OIS context because of the variety of data, sources, and measurement techniques involved and the obsolute accuracy expected by applications users.
- Vasadigation: A crucial function in OIS is related to visualization—the graphical display of terrain information and the appropriate representation of application.

attributes to go with it. Major visualization techniques include (1) containing through the use of isolines, spatial units of lines or units of equal attribute values; (2) fullshading, an illipinination method used for qualitative relief depiction using varied light intensities for individual facets of the terrain model; and (3) perspective displays three dimensional images of terrain model facets using perspective projection methods from computer graphics. These techniques impose carrigraphic data and other three-dimensional objects on terrain data providing animated scepe renderings such as those in flight simulations and animated movies.

Such requirements clearly illustrate that standard RDBMSs or COBMSs do nor meet the special needs of sits. It is therefore necessary to design systems that support the vector and tester representations and the spatial functionality as well as the recorred DBMS<sup>4</sup> features. A popular CBS software called ARCOMBCA which is nor a OBMS but integrates RDBMS functionality in the INPO part of the system is briefly discussed on the subsection that follows. More systems are likely to be designed in the future to work with relational or object databases that will contain some of the spatial and invest of the nonspatial information.

### 29.3.4 An Example of a GIS Software: ARC-INFO

ARC/DFC—a popular OE software launched in 1961 by Environmental System Research Institute (ESRI)—uses the arc node model to store spatial data. A geographic layer—called coverage in ARC/DFC—decreases of three primitives (1) nodes (points), (2) are 6-ander to lines), and (3) polygons. The arc is the most important of the three and stores a large amount of repulsional information. An arc has a start node and an end node (and it Gerefore has direction too). Is addition, the polygons to the left and the right of the arc are also stored along with each arc. As there is no restriction on the shape of the arc, shape points that have no impological information are also stored along with each arc. The database rights are applied by the  $\beta$ FD/B( $\beta$ MS thus consists of three required tables (1) rule antibute table (NAT), (2) are applied value (AAT), and (3) polygon artifiare table (PAT). Additional information can be stored in separate tables and joined with any of these three tables.

The NAT contrains an internal ID for the node, a user-specified ID, the recordinates of the node, and any other information associated with that node (e.g., numes of the intersecting roads at the node). The AAT contains an internal ID for the arc, a user specified iD, the internal ID of the sour and end nodes, the internal ID of the polygons to the left and the right, a series of coordinates of shape points (if any), the length of the sour and end nodes, the internal ID of the polygons to the left and the right, a series of coordinates of shape points (if any), the length of the sour and any other data associated with the arc (e.g., the name of the road the arc represents). The PAT contains an internal ID for the polygon, a user-specified ID, the area of the polygon, the perimeter of the polygon, and any other associated data (e.g., name of the county the polygon represents).

Typical spatial queries are related to adjacency, contoinment, and connectivity. The aic node model has enough information to sorisfy all three types of queries, but the RDBMS is not ideally sorted for this type of querying. A simple example will highlight the number of times a relational database has to be queried to extract adjacency information. Assume that we are trying to determine whether two polygons. A and B, are adjacent to each other. We would have to extract to extract to determine whether there is an edge that has A.
on one side and Bon the other. The search cannot be limited to the edges of earlier polygon as we do not explicitly store all the ares that make a polygon in the CAT. Storing all the ares on the PAT would be redundent because all the information is already there in the VAT.

E46 has released Arc/Storm (Arc Store Manager) which allows inultiple users to use the same 50-5 handles distributed databases, and anregistes with other commercial RPRMSs like ORACLE. INFORMES, and SPRASE. While it offers many reformance and functional advantages over ARC/INPO. It is essentially an RDBMs embedded within a GS-

## 29.3.5 Problems and Future Issues in GIS

OIS is an expanding application area of databases, reflecting an explosion in the number of end n-cis using digitized maps, terrain data, space images, we then data, and traffic infommation support data. As a consequence, an increasing number of problems related to CIS applications has been generated and will need to be solved:

- 1. New aschierenets, OIS applications will need a new cherit-server aschierenet that will benefit from existing advances in RD/MS and OP2015 technology. One possible solution is 80 separate spatial from nonspatial data and to manage the latter entirely by a D906. Such a process calls for appropriate modeling and integration as both types of data evolve. Commercial vendors find that it is more visible to keep a small number of independent databases with an automatic pristing of insideres across them. Appropriate tools for data transfer, change management, and workflow management will be required.
- 2. Veryoning and algent ble work appracely Because of constantly evolving prographic call tearures, GSs must maintain elaborate cartographic and terrain data to munagement problem that might be cased by incremental updating coupled with update authorization schemes for different levels of users. Under the object lifecycle approach, which covers the activities of cruating, destroying, and modifying objects as well as promoting versions into permanent objects, a complete set of anotheds may be predefined to control these activities for OS objects.
- 3. Data sandards. Because of the diversity of representation schemes and prociets, formalitation of data transfer standards is crucial for the success of OS. The attentional standards attom body (180, 70211) and the Europeon standards body (188, 70278) are now us the process of debating relevant assess among them conversion between vector and taster data for fost query performance.
- 4. Matching applications and dota structures. Looking again at Figure 37.5, we see that a classification of GS applications is based on the nature and organization of data. In the future, systems covering a wide trace of functions—trom market analysis and influties to car masigation—will need boundary-oriented data are functionality. On the other hand, applications in environmental science. Individual, and agriculture will require more area-oriented and tencan model data. It is not clear that all this function raty can be supported by a single general-purpose (45. The specialized needs of OIS) will require that general purpose 1000% must be

enhanced with additional data types and functionality before full-fledged and applications can be supported.

 Eack of reporter in dust synchrony. This is evident especially in maps. Information such as highway, and road crossings may be difficult to determine based on the stored data. One way streets are also hard to represent in the present GPS. Transportation CAD systems have incorporated such semantics into GPS.

### 29.3.6 Selected Bibliography for GIS

There are a number of books written on GIS. Adam and Gung padhyav (1997) and Laumi and Thompson (1992) focus on GIS database and information management problems Kemp (1993) gives an overview of GIS issues and data sources. Huxhold (1991) gives an intraduction to Othan (TS. Mogure et al. (1991) have a very good collection of GIS-related popers. Antenocci (1995) presents a discussion of the GIS technologies. Shekhar and Chaoda (2002) discusses as test and approaches to spatial data management which is at the core of all (TS. Demets (2002)) is another recent book on the fundamentars of GS. Bostomater and Green (2002) is a primer on GIS operations, languages, metadata paradigms and standards. Penp and Toou (2003) discusses internet GIS which metades a sinte of emergine new technologies aimed at making GIS more recenter GIS which metades a sinte of emergine new technologies aimed at making GIS more recenter. The TIGER files for exist data in the United States are managed by the U.S. Department of Commerce (1993). Laser-Scan's Web site (http://www.bl.co.id/pape.rs) is a grand source of internation.

Environmental System Research Invitinte (ISR.) has an excellent library of Gisbuoks for all levels at http://www.e-ri.com. The Gis-terminology is defined at http:// www.esri.com/abrary/glossary/glossary/istuil. The university of Edinburgh maintains a GIS-WWW resource last at http://www.geo.ediac.uk/home/giswww.brud

# 29.4 GENOME DATA MANAGEMENT

### 29.4.1 Biological Sciences and Genetics

The biological sciences encompass an enormous carety of information. Environmental seence gives us a view of how spones like and interact in a world filled with national phenomena. Biology and ecology study particular species. Anotomy focuses on the overall structure of an organism documenting the physical opects of indivadual bodies. Traditional medicine and physiology break the organism into systems and tissues and strive to collect information on the workings of these systems and the organism as a whole. Lastology and cell biologidelve into the tissue and cellular levels and provide knowledge about the inner structure and function of the cell. This wealth of information that has been generated, classified, and steed for centuries basionly recently become a major application of database technology.

Genetics has emerged as an ideal field for the application of information technology In a broad sense, it can be thought of as the construction of models based on information about genes---which can be defined as basic units of herediw -- and populations and the seeking out of relationships in that information. The study of genetics can be divided into three Sumchess (1). Mendelian genetics, (2) indecidar genetics, and (3) population genetics. Mendelian genetics is the study of the transmission of traits between generations. Molecular genetics is the study of the chemical structure and functions of genes at the molecular level. Population genetics is the study of her study of how genetic molecular information varies across populations of eigenvaries.

Molecular genetics provides a more detailed look at genetic information by allowing researchers to examine the composition, structure, and function of genes. The origins of molecular genetics can be traced to two important docaveries. The first occarred in 1969 when Friedrich Miesener discovered nuclein and its primary component, deaxynlycaucleic and (18 A). In subsequent research DNA and a related compound, rehomicleic azid (RSA). were found to be commoned of nucleotides (a sugar, a phosphate, and a base, which combined to form mechae and linked into long polymers via the sugar and phosphate. The second discovery was the demonstration in 1944 by Owold Avery that DNA was indeed the understar substance carrying genetic informatica. Genes were this shown to be composed of chajns of public ands arranged linearly on chromosomes and to some three primary functions: (1) repleasing genetic information between generations. (2) providing Elugerities for the areation of polynoprides, and (3) accumulating changes, thereby allowing evolution to occur. Wasten and Crick found the double-helix structure of the DNA up 1953, which gave molecular genetics research a new direction." Discovery of the DNA and its structure is hailed as probably the most important biological work of the last 100 years, and the field it opened may be the scientific frontier for the next 100. In 1982, Watson, Circk, and Wilking won the Nobel Price for physiology/molicine for this breakthrough <sup>1</sup>

## 29.4.2 Characteristics of Biological Data

Biological data exhibits many special characteristics that make management of biological information a particularly challenging problem. We will thus begin by summation the characteristics related to biological information, and focusing on a multidisciplinary field called bioinformatics that has emerged, with graduate degree programs now in place in several universities. Bioinformatics addresses information management of genetic information with special emphasis in 105A sequence analysis. It needs to be broadened into a wider scope to harness all types is biological information—its nondeling, storage, retrieval, and management. Morrover, applications of bioinformatics spin design of targets for dogs, study of mutations and related diverses, anthropological investigations on migration patterns of tribes, and therapeatic frestingerts.

**Characteristic** 1: Biological data is highly complex when compared with most other domains or applications. Definitions of such data must thus be able to represent a complex substructure of data as well as relationships and to ensure that no information is lost during biological data mindeling. The structure of biological data often provides an additional context for interpretation of the information. Biological information systems must be able to represent any level of complexity in any data schema, relationship, or whema substructures nor pat hierarchical, binary, or table data. As an example, SUTOVAP is a database documenting the human introductional genome.<sup>5</sup> This single genome is a small, circular piece of DNA encompassing information about 10.569 micheoride bases, 52 gene loci encoding messenger RNA, information about 10.569 micheoride bases, 52 gene loci encoding messenger RNA, information about 10.569 micheoride bases, 52 gene loci encoding messenger RNA, information about 10.569 micheoride bases, 52 gene loci encoding messenger RNA, information about 10.569 micheoride bases, 52 gene loci encoding messenger RNA, information about 10.569 micheoride bases, 52 gene loci encoding messenger RNA, information about 10.569 micheoride bases, 52 gene loci encoding messenger RNA, information about 10.569 micheoride bases, 52 gene loci encoding messenger RNA, information about 10.569 micheoride bases, 52 gene loci encoding messenger RNA, information about 10.569 micheoride bases, 52 gene loci encoding messenger RNA, information about 10.569 micheoride bases, 52 gene loci encoding messenger RNA, information about 10.569 micheoride bases, 52 gene loci encoding messenger RNA, information about 10.569 micheoride bases, 52 gene loci encoding messenger RNA, information and a limited set of known base of the biochemical energy producing pathway of ordanize phosphorylation. As might be expected, its interagement has encountered a large member of problems, we have been onable to use the readition d ROPAIS or OPAIS or OPAIS or OPAIS or OPAIS or OPAIS or opais to compute all aspects of the bata.

**Characteristic** 2: The anomer and some of considery rulear is high. Hence, biological systems must be flexible in handling data types and values. With such a sole range of possible data values, placing constraints on data types must be limited since this may exclude unexpected values—e.g. conflict values—that are possible formation in the biological domain. Exclusion of such values results in a loss of information. In addition, frequent exceptions to biological data structures may require a choice of data types to be available for a given piece of data.

**Characteristic 3**: Scheres in niskignal slatables change at a rapid pare release, for improved information flow between generations or releases of databases, scheres evolution and data object imparton must be supported. The ability to extend the scheres, a frequent occurrence in the biological setting, is unsupported in most relational and object database systems. Presently systems such as GenPortic reference the entire database with new schemas once or twite a cear rather than incrementally changing the system as changes become necessary. Such an evolutionary database would provide a timely and orderly mechanism for following changes to included data entities in biological databases over time. This surf of tracking is important for biological researchers to be able to access and reproduce provides results.

**Characteristic** 4: Representations of the some data by *different indopols* and likely be *different (even obser using the same system)*. Hence, moleanisms for "different," different feating of schemas or different versions of schemas should be supported. Given the complexity of biological data, there are a multimide at ways of modeling any given emity, with the results after reflecting the particular factors of the scientist. While two individuals may produce different data models if asked in interpret the same entity, these models will likely have numerous points in common. In such situations, it would be oscial to biological investigators to be able to pro-squeries across these common, points. By linking data elements in a network of schemas, this could be accomplished.

**Characteristic 5** Most users of biological data do not require tonic access to the database reachards access is adequate. Write access is limited to provideged users called controls for example, the database created as part of the MITOWAN project has no average more than

Details of 2017.0040 and or antorin mon-complexity can be seen in Kogelrak et al. (1997, 2005) and a fungely away analysing.

15.000 users per month on the Internet. There are lever than twenty notestrator generated submissions to MHOMA2 every month. In other words, the number of users requiring write access is small. Users generate a wide variety of read-access patterns into the databases, but these patterns are not the same as those seen in traditional relational databases. User requested ad bac searches demand indexing of often unexpected combinations of data instance classes.

**Characteristic** 6. Most biologies are not likely to have any knowledge of the internal structure of the database or about scheme doign. Diological database interfaces should deplay information to users in a manner that is appreable to the problem they are trying to address and that reflects the orderlying data structure. Diological users anally know which data they require, but they have no technical knowledge of the data structure of law a DBMs represents the data. They tely on technical users to provide them with views into the data structure of the user require, but they have no technical users to provide them with views into the data structure of the data structure of the user requires the data. They tely on technical users to provide them with views into the data Belational schemas fail to provide cues or any intuitive information to the user regarding the meaning of their schema. Web interfaces in particular often provide preset search interfaces, which new limit access into the database. However, if these interfaces are generated directly from database structures, they are likely in produce a wider possible range of access although they may not guarantee usability.

**Characteristic** 7: The context of data gees added accurate for its new in biological applications. Hence, context must be maintained and conveyed to the user when appropriate. In addition, it should be possible to integrate as many contexts as possible to maximize the interpretation of a biological data value. Isolated values are of less use in biological systems. For example, the sequence of a 7NA strand is not particularly useful without idditional information describing its organization, function, and such. A single successful on a DNA strand, for example, seen an context with nondisease-causing .'NA strands, could be seen its a causative element for siddle cell around.

Characteristic 8: Defining and segretientatig complex queries is extrainely superstant to the follogical systems must support complex queries. Without any knowledge of the data structure (see Characturistic fit) average users carnot construct a complex query across data serie on their own. Thus, it order to be truly useful, systems must provide some tools for building these queries. As mentioned previously, many systems provide previously, many systems provide previously, many systems provide previously.

Characteristic 9:4 how of bashgind information often require access to 'out' rahes of the data—periodicity a fain averying providely reproved woulds. Hence, changes to the values of data in the database must be supported through a system of archives. Access to both the most recent to resolve of > data value and its previous version, are important in the biological domain. Investigators consistently want to query the most op-ro-date data, but they must also be able to reconstruct previous work and reevaluate prior and current information. Consequently, values than are about to be updated in a biological database connor simply be thrown away.

All of these coaracteristics clearly point to the fact that roday's DWFS do not fully core to the requirements of complex biological data. A new direction in database inattagement systems is necessary?

<sup>9.</sup> See Koyemik et al. (1997, 2008) f. muriller details.

## 29.4.3 The Human Genome Project and Existing Biological Databases

The term genome is defined as the total genetic information that can be obtained about an entity. The human genome, for example, generally refers to the complete set of genos required to create a human being —ostimated to be more than 30,000 genes spread over 33 pairs of chromosomes, with an estimated 5 to 4 billion nucleondes. The goal of the Human Genome Project (EOF) has been to obtain the complete sequence—the ordering of the bases—of those nucleondes. A rough draft of entire human genome sequence was announced in June 2000 and the D-year effort will end in year 2004 with the completion of the human genetic sequence. In isolation, the human UNA sequence is not particularly useful. The sequence can however be combined with other data and used as a powerful root is help address questions in genetics, biochemistry, medicine, anthropology, and acticulate in the sequence databases, the torus has been on "curating" (or collecting with some initial seturity and quality check) and classifying information about genome sequence data in addition to the human genome, runnerious organistics such as  $E \cos i$ . Descepting and Classifying information about genome sequence data in addition to the human genome, runnerious organistics such as  $E \cos i$ . Descepting and Classifying information about genome sequence data in addition to the human genome, runnerious organistics such as  $E \cos i$ . Descepting and Classifying information about genome sequence systems that are supporting or have grown out of the Human Genome Project.

GenBonk. The prominent DNA sequence dotabase in the world today is GenBank, numinitized by the National Center for Brotechnology Information (S030) of the National Library of Medicine (NEM). It was established in 1978 as a central repository for DNA sequence data. Since then it has expanded somewhat to scope to include expressed sequence that data protein sequence data, three-dimensional protein structure, toxonom, and links to the biomedical Interature (MEPLINE). As of release 135.0 in April 2024, GenBank contains over 31 ballion nucleotide bases of more than 24 million sequences from over 1005000 species with roughly 1400 new organisms being added each month. The database size to flat file formatics over 100 GIV incompressed and has been doubling every 15 months. Through international collaboration with the European Molecular Biology Laboratory(EMBE) in the U.K. and the DNA Data Bark of Japan (DDBJ), data are excitatinged among the three sites on a darly basis. The matroing of acquence data at the three area alfords fast access to this data to scientists in visions generaphical parts of the world.

While it is a complex, comprehensive database, the scope of its coverage is focused on human sequences and littles to the literature. Other limited data sources (e.g. threedimensioned structure and OMM, discussed below), have been added recently by reformating the existing OMB3 and PD5 databases and redesigning the structure of the GerBank system to accommodate these new data sets.

The system is maintained as a combination of that files, relational databases, and files containing Abstract Syntax Nutation One (ASN.<sup>3</sup>) – a syntax for detaying data structures developed for the relacaninumications industry. Each GenBank entry is assigned a unique identifier by the NGM. Updates are assigned a new identifier, with the identifier of the original entry remaining unchanged for archival purposes. Older references to an entry thus do not indetermently inducate a new and possibly inappropriate value. The most current concepts also receive a second set of unique identifiers (0.008), which mark the

most up-ra-date form of a concept while allowing older versions to be accessed via their original identifier.

The average user of the database is not able to access the structure of the data directly for querying or other functions, although complete stapshots of the database are available for export in a number of tor nats, including ASN h. The query inechation provided is via the Entree application for its World Wide Web versionly, which allows keyword, sequence, and Genbark QID searching through a static interface.

The Genome Database (GOB) — Created in 1989, the Genome Database (GOB) is a catalog of burnan gene mapping data, a process that associates a prece of information with a particular location on the human economy. The degree of precision of this location on the map depends upon the source of the data, but it is usually not at the level of individual nucleotide bases. GDB data includes data describing primarily map information (distance and confidence humas), and Polymenuse Chain Reaction (2008) public data (experimental conditions, PCR primers, and reagents used). More recently efforts have been include to add data or mutations linked to genetic include the sized in experiments, LNA public formation data.

The GDU spaten is built around SDUGER, a commercial relational DBUG, and its data are modeled using standard Entity-Relationship techniques (see Chapters 3 and 4). The implementors of GDG have noted difficulties in using this model to capture more than simple map and probe data. In order to improve data integrity and to simplify the programming for application writers, GDB distributes a Database Acress Toolkit. However, most assets use a Web interface to search the remainterlinked data managers. Each manager keeps mack of the links (relationships) for one of the remainded within the CDB system. As with GenBank, users are given only a very high-level view of the data at the time of searching and thus carnot work make use of any knowledge gloaned from the structure of the GDB tables, ee inch methods are most useful when users are simply looking for an index atto map or probe data. Exploratory ad bio searching of the database is not encouraged by present interface. Integration of the database structures of GDB and CMBM (see below) was never fully established.

Online Mendelian Inheritance in Man. Online Mendelian Inheritance in Man (2008) is an electronic comparation of information on the generic basis of human disease. Begun in Bard-copy form by Vietor McCusick in 1966 with 1500 entries, it was converted to a full-text electronic form between 1987 and 1989 by the 0218. In 1991 its administration was transferred from Johns Hopkuts University to the Net9, and the entrie database was converted to Net805 GenBark format. Today it contains more than 14.000 entries.

ONIN covers material on five disease areas based loosely on organs and systems. Any morphological, blochemical, behavioral, or other properties under study are referred to as **phenotype** of an incrvidual (or a cell). Mendel realized that genes can exist in numerous different forms known as alleles. A **genotype** refers to the initial offender of an individual.

The structure of the phenotype and cenotype entries contains textual data loosely structured as general descriptions, contendature, mades of inhumance, variations, gene structure, mapping, and explements lesser transportes. The full root entries were converted to an ASN-1 structured format when OMIN was transferred to the NOB. This greatly improved the ability to link OMIN data to other databases and it also provided a riportial structure for the data. However, the basic form of the database remained difficult to mosbity

ECOCyc. The Encyclopedia of Evolution of Geres and Motabolism (EcoCyc) is a recent experiment in combining internation about the genome and the motabolism of *E*, cull K-12. The database was created in 1996 as a collaboration between Stanford Research Institution and the Marine Biological Laborational it catalogs and describes the known genes of *E*, cult the enzymes encoded by those genes, and the biochemical reactions taralyzed by each enzyme and their organization into metabolic pathways. In so doing, EcoCyc stars the sequence and function domains of genomic information. It contains 1263 comparision with 965 structures as well as lists of bonds and atoms, molecular weights, and empirical formulas. It contains 4038 biochemical reactions described using 269 data closes.

An object-onceated data model was first used to implement the system, with data stored on Ocelet, a frame knowledge representation system. Ecolect data was alranged in a literarchy of object classes based on the observations that (1) the properties of a reaction are independent of an enquire that catalyzes it, and (2) an ensyme has a number of properties that are "logically distinct" from its reactions.

EroCive provides two methods or querying: (1) direct (via predefined queries) and (2) indirect (via hypertext navigation). Direct queries are performed using menus and dialege that can initiate a large but finite set or queries. No navigation of the actual data structures is supported. In addition, no juochamism for evolving the schema is documented.

Table 29.1 summarizes the features of the major genome related databases, as well as FROMON and ACLON databases. Sume additional primeric databases exist; they cultarn information about protein structures. Provingerst protein databases include sweet-PROT at the University of Genevia, Protein Data Bank 1(200) at Brookhaven National Laboratory, and Protein Identification Resource (208) at National Biomedical Research Foundation.

Over the past (or past, due that been an increasing interest in the applications of databases in biology and medicine. GenBank, GDB, and GDB, have been created as central repositories of certain types of biological data but, while extremely useful, they do not vercover the complete spectrum of the Human Genome Project data. However, efforts are under way around the world to design new tools and rechtiques that will allevate the datamatagement problem for the biological scientists and medical researchers.

Gene Ontology – We already explained the concept of ortrologies in Section 29.2.5 in the context (4 modeling of multimedia information. Gene Ontolegy (OO) Consertion, was formed in 1998 as a collaboration anothe three model organism durabases. HoBase, Mouse Genome informatics (MGO) and Saccharomyces or yeast Genome Database (SOD). Its goal is to produce a structured, precisely defined, common, controlled vocabulary for describing the roles of genes and gene products in any organism. With the completion of genome sequencing of nearly species, it has been observed that a large fraction of genes among organisms display similarity in biological roles and large fraction.

biologists have acknowledge that there is likely to be a single himited universe of genesand proteins that are conserved in most or all living cells. On the other hand, genotic data is increasing exponentially and there is no uniform way to interpret and conceptualize the shored biological elements. Gene Ontology makes possible the annuration of price products using a common violabulary based on their shored biological attributes and interpretiability between gradient databases.

The GO Consortion has developed three ontologies: Mulecular function, bological process, and cellular component, to describe attributes of genes gene products or gene products function is defined as the buchemical activity of a gene product. Biological process refers to a biological objective to which the gene or gene product contributes. Cellular component refers to the place in the cell where a gene product is across. Each consology comprises a set of well defined weak-bucheries of terms and relationships. The terms are organized in the form of directed acyclic graphs (DAGs), in

| DATABASE<br>NAME | MAKOR<br>CONTENT                                              | INDIAI<br>TECHNOLOGY                 | CURRENT<br>TECHNOLOGY                 | DB PROBLEM<br>Areas                                                            | PRIMARY DATA<br>TYPES                     |  |
|------------------|---------------------------------------------------------------|--------------------------------------|---------------------------------------|--------------------------------------------------------------------------------|-------------------------------------------|--|
| Genbank          | DN MRNA<br>Sequences<br>Protein                               | Testtile≤                            | Flat.dde/ASN.1                        | Schema brows-<br>ing. schema<br>evolution. lipk-<br>ing scother dbs            | Text, nonortic,<br>some complex<br>repres |  |
| 0904             | Disease<br>phenotypes and<br>gettorypes, etc.                 | Index cards/resr<br>files            | Flor-file/388.4                       | Unstructured,<br>free text entries<br>linking to other<br>dbs                  | Техт                                      |  |
| GDB              | Genetie mags<br>Tirikary: data                                | Ela: nle                             | Relation i.                           | Schemalexpan-<br>4 on/evolution,<br>complex<br>objects lunking<br>to other dis | fest, numerie                             |  |
| ACHOK            | Cenetie may<br>lankage data,<br>sequence data<br>(non-bactan) | ×0                                   | СС:                                   | Schenas expan-<br>sion/evolution.<br>huking to other<br>dbs                    | Text. insiderate                          |  |
| Heavi is         | Sequence and<br>sequence<br>varia <b>n</b> s                  | Flot file<br>application<br>specific | Flat-file-<br>application<br>specific | Schema expon-<br>sion/evolution,<br>lanking to other<br>dbs                    | Text                                      |  |
| Eo Cive          | Biochemical<br>reactions and<br>pathways                      | 0.5<br>                              | ero<br>Alta                           | Locked into<br>class hierarchy<br>schema<br>evolution                          | Complex types,<br>rext, numeric           |  |

TABLE 29.1. SUMMARY OF THE MAJOR GENOME-RELATED DATABASES

which a term mode may have moltiple parents and multiple children. A child term can be arconstance of fis all or a part of its parent. In the latest release of the GO shitakose, there are over 13,020 terms and more than 18,000 relationships between terms. The annotation of gene products is operated independently by each of the collaborating databases. A subset of the annotations is included in GO database, which concerns over 1.356,000 geres products and 5.244,000 associations between gene products and GO terms.

The Oute Ontology was implemented using MySQL, an open source relational DBMS and a monthly database release is available in SQL and XML formats. A set of mulk and hbranes, written in C. Java, Perl and XML ere, is available for database access and development of applications. Web lessed and stand alone OO bransers are available from the OO consortium.

### 29.4.4 Selected Bibliography for Genome Databases

Disanformatics has become a popular area of research in recent years and more workshops and conferences are being organized around this topic. Robbins (1993) gives a good overview while Frenkel (1991) servers the human genome project with its special role in bioinformatics at large. Controchia et al. (1993), Benson et al. (2002), and Pear-on et al. (1994) are references on GDP, GenRank, and Okitst. In an international collaboration among GenePoink (108A). TNA Data Bank of Japan (DDBJ) (http://www.skilg-me.uc.jp/E-mail/ koncology.html), and Equeran Molecular Biology Laboratory (EBML) (Stoesser G, 2003), data are exchanged amongst the collaborating databases on a daily losis to achieve optimal synchronization. Wheeler et al. (2000) discuss the various tools that currently allow users access and analysis of the data available in the databases.

Wallace (1995) has been a protect in the numb hendrial perionic research, which deals with a specific part of the human generic; the sequence and organizational detail of this area appear in Audorson et al. (1981) Recent work in Kogelnik et al. (1997, 1998) and Kogelnik (1998) addresses the development of a generic solution to the data management problem in biological sciences by developing a protoxy periodotion. Apwellet et al. (2023) review the core Biopformatics resources maintained at the Fox pean Biomformatics Institute (EBH) (stick as Storssprot + TrEMB1) and summatice important (Steps Sequence Database such as DDB]/EMBL/ GENEBANK Nucleotide Sequence Database such as DDB]/EMBL/ GENEBANK Nucleotide Sequence (Point Discuss such as PROSITE, PRINTS and Diam and Interorial Point Point and Interorial Point Point (Point, PRINTS, POIDOR, PROSITE, SMAR 7, and FIGREAMS).

The European Biendomatica institute Micromolecular Structure Database (E-MSD), which is a relational database (http://www.conactok/msd) (Boutselakos et al. 2003) is designed to builly alongle access point for protein and nucleic and structures and related intomation. The database is derived from Protein Data Bank (FDB) entries. The search database contains an extensive set of derived properties, goodness of databases and links to other EBI databases including InterPro, GO, and SWISS-PROT, together with links to SCOP, CATH, CFAM and PROSITE, Karp (1990) docusses the problems of interlinking the variety of databases mentioned in this action. He defines near types of links, those that integrate the data and those that relate the data between databases. These were used to design the Ecocyc database.

Some of the important web links include the following. The Human Genome sequence information can be found an brtp://www.ncbi.nl.n.mh.gov/genome/seq/-

The MITOMAP database developed in Kogeloik (1995) can be accessed at http://www.mitomap.org/ The Euglest protein database SW155-PSOV can be accessed from http://expass.heuge.ch/spirit/ The ACC258 database information is available at http://probe-nalusda.gov/5080/acedocs/





# Alternative Diagrammatic Notations for ER Models

Figure A.1 shows a number of different diagrammatic notations for representing E8 and EER model concepts. Unfortunately, there is no standard instances different database design practitioners prefer different norations. Similarly, various CASE (computer acted software engineering) mask and OOA (object sonenred analysis) methodologies are cancels notations. Some nutations are associated with models that bace additional concepts and constraints beyond thuse of the E8 and VE8 models described in Chapters 5 and 24, while other models pave fewer concepts and constraints. The noration we used in Chapter 3 is space close to the original non-tiers for E8 diagrams, which is still widely used. We discuss some alternaty not more here.

Figure A 1(a) shows different notations for displaying entry types/classes, attributes, and relation-hips. In Chapters 4 and 24, we used the symbols marked (i) in Figure A-1(a)—namely, rectonely, oval, and diamond. Notice that symbol (ii) for entry types/classes, symbol, (ii) for attributes, and symbol (ii) for relationships are similar, but they are used by different methodologies to represent torce different concepts. The stronght line symbol (iii) for representing relationships is used by several tools and methodologies.

Figure A. 1(b) shows some autotums for attaching attributes to unory types. We used notation (it). Notation (ii) uses the third notation (iii) for attributes from Figure A.1(a). The last two neutrons in Figure A.3(b)—(ai) and (iv)—arc popular in COS, neithedologies and an some CASE tools. In particular, the last notation displays both the articlutes and the methods of pickas, separated by a horizontal line.

#### 948 Appendix A. Alternative Diagrammatic Notations for Ek Models



**FIGURE A.1** Alternative notations. (a) Symbols for entity type/class, attribute, and relationship. (b) Displaying attributes. (c) Displaying cardinality ratios. (d) Various (min. max) notations. (e) Notations for displaying specialization/generalization.

Figure A 1(c) shows various non-arous for representing the cordinality form of binary relationships. We used nonation (i) on Chapters 3 and 24. Notation (ii)—known as the chicken free paration – is quite popular. Notation (iv) uses the arrow as a functional reference (from the N to the 1 side) and resembles our notation for foreign keys in the relational model (see Figure 7.7); notation (v)—used in Bachman Stigattes—use- the

arrow in dx receive direction (from the 1 to the N side). For a 1-1 relationship, (ii) uses a straight line without any chicken feets (iii) makes both halves of the diamond white, and (iv) places arrowheads on both (ides, For an M:N relationship, (ii) uses chicken feet at both ends of the line; (iii) makes both halves of the diamond black; and (iv) does not display any arrowheads.

Figure A I(d) shows several furnations for deplaying (num max) constraints, which are used to display both cardinality ratio and total/partial participation. Notation (n) is the alternative notation we used in Figure 3.15 and discussed in Section 3.7.4. Recall that our notation specifies the constraint that each entity must participate in at least min and at most max relationship instances. Hence, for a 1:1 relationship, both max values are 1: and for MiN, both max values are in. A min value greater than C (zero) specifies total participation (existence dependence). In nother discussed in section straight line for displaying relationships, it is common to reactive the tendentiage of the finite, most constraints, as shown an (m). Attacher popular rechrisple , which follows the same positioning as first i is to display the num as  $\alpha$  ("ob" or circle, which stands for D) or as 1 (vertical dash, which stands for 1), and to display the mox as 1 (vertical dash, which stands for 1), as shown in (jv).

Figure Added shows some notations for displaying specialization/generalization. We used notation (d) in Charter 14, where a d in the circle specifies that the subclasses (S1, S2, and S3) are disjoint and an o specifies overlapping subclasses. Notation (ii) uses G there generalization) to specify disjoint, and Gs to specify overlapping; some notations use the solid atrow while others use the empty arrow (shown at the side). Notation (iii) uses a triangle pointing roward the superclass, and notation (v) uses a triangle pointing roward the superclass, and notations in the some methodology, web (ai) indicating generalization and (v) indicating specialization. Notation (iv) places the bases representing subclasses within the box representing the apportance. Of the notations based on (vi), some use a single-fined arrow and others use a clouble-lined arrow (shown at the state).

The rotations shown in Figure A.1 show only some of the diagrammatic symbols that have been used or suggested for displaying database conceptual schemes. Other cotations as well as various combinations of the preceding, have also been used. It would be useful to establish a standard that everyone would adhere to, in order to prevent anisonderstandings and reduce confusion.



The most important disk parameter is the time required to locate an arbitrary disk block, given its block address, and then to transfer the block between the disk and a main memory butter. This is the random access time for accessing a disk block. There are three time comparamets to consider:

- 1. Seek time (v): This is the time model? to incohereably position the road/write head on the conference mark for movable-head disks. (For fixed-head disks, it is the rime model to electronic ills writely to the appropriate read/write head.) For movable head disks this time varies, depending on the distance between the current mark under the read/pence head and the track specified in the block address. Usually, the disk miniatorrore provides an overage seek time in gibbseconds. The typical range of overage seek time is 10 to 60 msec. This is the main "oulprit" for the delay involved in transferring blocks between disk and memory.
- 2 Rotational delay (rd): Once the read/write head is at the corner track, the user must writ for the beginning of the required block to rotate into position and/it the read/write head. On the average, this takes about the time for half a resolution of the disk, but it actually ninees from immediate access (if the statt of the required block is in position under the read/write head right after the seek) to a fall disk resolution (if the statt of the required block just passed the read/write head after the state of the

the seek!. If the speed of disk population is prevolutions per minute (rpm), then the average recational delay rd is given by

rd = (1/2)\*(1/p) min = (60\*1000)/(2\*p) msec

A typical value for p is 10.000 rpm, which gives a rotational delay of rd = 3 msec. For fixed-head disks, where the seek time is negligible, this component causes the greatest delay in transforming a disk Nock.

3 Block transfer time (bit): Once the readjointe head is at the beginning of the required block, some time is needed to transfer the data in the block. This block transfer time depends on the block size, the track size, and the obtained speed. If the transfer rate for the disk is tribytes/msec and the block size is B bytes, then

bot = B/tr msec

If we have a track size of 50 Kbytes and p is 5600 rpm, the transfer fate in Tytes' inset is

```
tr = ($0~1000)/(60×1000/3600) = 3000 bytes/msec
```

In this case, but = B/3000 insect, where B is the block are in bytes.

The overage time needed to find and transfer a block, given its block address is estimated by

(s + rd + btl) msec

This holds for either reading or writing a block. The principal method of reducing this time is to transfer several blocks that are stored on one or more tracks of the same cylinder; then the seek time is required only for the first block. To transfer consecutively k nonconfiguous blocks that are on the same cylinder, we need approximately

s + (k = (rd + btt)) msec

In this case, we need two or more buffers in main storage, because we are continuously reading or writing the k blocks, as we discussed in Section 4.3. The transfer time per block is reduced even further when consequing blocks on the same track or cylinder are transferred. This eliminates the rotational delay for all but the first block, so the estimate for transferring k consecutive blocks is

s + rd + (k + btt) msec

A more accurate estimate for transferring consecutive blocks takes into account the interblock gap (see Section 5.2.1), which includes the information that enables the read/write head to determine which block it is about to read. Usually, the disk manufacturer provides a **bulk transfer rate** (brt) that takes the gap size into account when reading consecutively stored blocks. If the gap size is G bytes, then

btr =  $(B/(B + G)) \leq \tau r$  bytes/msec

The bulk transfer rate is the rate of transferring mefid bytes in the data blocks. The disk read/write head must go over all bytes on a track as the disk rotates, including the bytes in the interblock gaps, which store cosmol momination bet not real data. When the bulk transfer tate is used, the time needed to transfer the useful data mone block out

or several consecutive blocks is B/htr. Here e, the estimated time to read k blocks consecutively stored on the same cylinder becomes

> + rd + (k < (8/btr)) msec</pre>

Another parameter of disks is the **rewrite time**. This is useful in cases when we read a block from the disk into a main memory buffer, update the buffer, and then write the buffer back to the cases disk block on which it was stored. In many cases, the rune received to opdate the buffer in main memory so less than the time required for one disk revolution. If see know that the buffer is ready for rewriting, the system can keep the disk holds on the same macks and during the next disk revolution the stated buffer is rewrite the case disk revolution the stated buffer is rewrited to be some macks and during the case disk revolution the stated buffer is rewrite time  $T_{\rm es}$  is usually estimated to be the time needed for one disk revolution.

 $\mathbf{T}_{r,\mathbf{x}} = 2 + rd$  msec

To summarize, here is a list of the parameters we have discussed and the symbols we use for them:

seek time: s more intational delay: rd maec block transfer time, bit insec transfer rote:  $T_{i,k}$  mass transfer rote: tr bytes/more bolk (proofer pire) bit hytes/more block size. B hytes unterblock cap size. O bytes



# Overview of the QBE Language

The Query-By-Example (QPE) language is important because it is one of the first graphical query languages with minimum syntax developed for database systems. It was developed at 38M Research and is available as an 18M commercial product as part of the QNF (Query Management Facility) materface option to DP2. The language was also implemented in the TARALON DBMS, and is related to a point-and-click type interface in the ACLESS (BMS (see Chapter 12), it differs from SQL is that the user does not have to specity a structured query explicitly; rather, the query is formulated by filling or templates of relations that are displayed intra-momentor acreent. Figure 9.5 shows how these templates may link for the database of Figure 7.6. The user does not have to include out attributes or rolations, because they are displayed as part of these templates. In addition, the user does not have to fullow any rigid syntax rules for query specification; rather, constants and variables are entered in the columns of the templates to construct an example related in the permetor of equals (QN- is related in the domain relational calculus, as we shall see, and its original specification has been shown to be relationally complete.

# D.1 BASIC RETRIEVALS IN QBE

In Q66, retrieval queries are specified by filling in one or more rows in the remplates of the rubles. For a single relation spery, we enter either constants or **example elements** (a Q46 rem) in the cohoms of the remplate of that relation. Are example element stards for a





domain variable and is specified as an example value preceded by the underscore character  $t_{\rm e}$  L Additionally, a P, prefix (called the P dut operator) is entered in certain volumes to indicate that we would like to print (or display) volues in these columns for our read-The constants specify volues that insist be exactly matched in thus columns.

For example, consider the query QO. "Retrieve the Entiblate and address of John B. Smith." We show in Figures 9.6(a) through 9.6(d) how this query can be specified in a progressively more interform in QPE. In Figure 9.6(a) an example of an employee is presented as the type of row that we are interested in . By leaving John B. Smith as constants in the POPC PDD, and DOPC columns, we are specifying an exact match in those columns. All the rest of the columns are preceded by an underscore indicating that they are domain.

| la) | EMPLOYFE | FN4k <b>E</b> | MNIT       | INAME   | SSN       | 5DAIT  | ADERESS              | 389                                          | SALABY   | SUPERSSN  | LINO  |
|-----|----------|---------------|------------|---------|-----------|--------|----------------------|----------------------------------------------|----------|-----------|-------|
|     |          | 367           | e          | 5.0m    | 120456785 | R 5160 | P 100 Van Helston TV | ч                                            | 25000    | 123450789 |       |
|     |          |               |            |         |           |        |                      |                                              |          |           |       |
| ю   | EMPLOYES | FNAME         | WINLT      | [usavē] | SSN       | 60ATE  | ADCITESS             | 3EX                                          | NINTAR I | SUPERSSN  | [0v0] |
|     |          | kr o          | э          | Sht     |           | P 0160 | 2 (0015 e 10,5 e 17  |                                              |          |           |       |
|     |          |               |            |         |           |        |                      |                                              |          |           |       |
| Ľ;  | EMPLOYEE | FNAME         | MNI        | LANAVE  | SEN       | BEATS. | ALCH5S3              | SEX                                          | SOLAHY   | SUPERSSIN | UND   |
|     |          | 100           | Э          | 5-0     |           | 2.     | P.2                  |                                              |          |           |       |
|     |          |               |            |         |           |        |                      |                                              |          |           |       |
| lai | EMPLOYEE | ENAME         | <b>MNI</b> | DAMAE   | GaN       | BD915  | ALCHESS              | 96 X                                         | SALAHY   | SUPERSIN  | DND   |
|     |          | kan.          | A          | Save    |           | μ      | P P                  | <u>†                                    </u> |          |           |       |

FIGURE D.2 Hour ways of specifying the query QU in Q61.

variables (example elements). The  $\beta$  prefix is placed in the stars and sames, columns to indicate that we would like to output value(s) or those columns.

Q0 can be abbreviated as shown in Figure 9.6(b). There is no need to specify example values for columns in which we are not interested. Moreover, because example values are completely arbitrary, we can just specify variable names for them, as shown in Figure 9.6(c). Finally, we can also leave can the example values entirely, as shown in Figure 9.6(d), and just specify a E under the columns to be retrieved.

To see how remeal queries in QBE are similar to the domain relational calculus, compare Figure 9.6(d) such QC (simplified) in domain releases which is as follows:

#### QD. ( or 1 EMPLOYEE(granowexst) and getJohn' and retBi and setSmith')

We can think of each column in a Q88 reinplate as an implicit domain comble, hence, **FORE** corresponds to the domain variable question corresponds to respondences and two correspondences in the Q88 query, the columns with P, correspond to variables specified to the left of the bar in domain columns, whereas the columns with constant values correspond to traple variables with equality selection conditions on them. The condition tweat(query access) and the existential quantifiers are implicit in the Q88 query because the template corresponding to the instantial quantifiers are implicit in the Q89 query because the template corresponding to the instant relation is used.

In QBL, the user interface first allows the user to choose the tables (relations) needed to formulate a query by displaying a list of all relation manas. The templates for the chosen relations are then displayed. The user moves to the appropriate columns in the templaner and specifies the query. Special function keys were provided to move among templates and perform certain functions.

We now give examples to illustrate basic facilities of QeF. Comparison operators other than = (such as  $\geq$  or  $\geq$ ) may be emerged in a column before typing a constant value. For example, the query QCA: "List the social scenary numbers of employees who work more than 20 hours per week on project number 1," can be specified as shown in Figure 9.7(a). For more complex conditions, the user can ask for a condition hox, which is every ded by pressing a particular function key. The user can then type the correlex conditional for example, the query QCB—"List the social scenary munders of employees who work more than 20 hours per week on enter project 1 or project 2° —con be specified as shown in Figure 9.7(b).

Some complex conditions can be specified with at a condition box. The rule is that all conditions specified on the same row of a relation template are connected by the and logical connective tail must be satisfied by a selected table), whereas conditions specified on distinct rows are connected by or far idast one must be satisfied). Hence, Q8B can also be specified, as shown in Figure 9.7(c), by enformed two distinct rows in the reioplate

Now consider query QCC: "List the social security numbers of employees who work in both project 1 and project 21; this current be specified as in Figure 9.8(a), which lists these who work on *advar* project 1 or project 2. The example variable 145 will build (self to each values in < -, 2, -> toples as usal as in those in < -, 2, -> toples. Figure 9.8(b)

| FSSN<br>P     | PNO  | HOURS                                                               |
|---------------|------|---------------------------------------------------------------------|
| E             |      |                                                                     |
| -             |      | >20                                                                 |
|               |      |                                                                     |
| )/ PX_109 PX. | .21  | -<br>-                                                              |
| Aline da int  |      | _                                                                   |
| E\$\$N_       | PNO  | HOURS                                                               |
| D             |      | - 22                                                                |
|               | ESSN | ESSN PNO<br>. P. PX<br>. PX PX<br>. PX - 109 (PX - 2)<br>. ESSN PNO |

**FIGURE D.3** Specifying complex conditions in Q8: (a) the same query Q0A. (b) The query Q0B with a condition box, (c) The query Q0B without a condition box.

| (a) | WORKS_ON   | ESSN           | PNO      | HOURS |
|-----|------------|----------------|----------|-------|
|     | I          | P. ES<br>P. ES | 1 2      | 1     |
| 264 |            | -              |          |       |
| (6) | WORKS_ON   | EŞSN           |          | HOURS |
|     |            | R_EX           |          |       |
|     | I          | P_EY           | <u>2</u> |       |
|     | CONDITIONS |                |          |       |
|     | _EX # _EY  |                |          |       |

FIGURE D.4. Specifying analowers who work on both projects. (a) Incorrect specification of an AND condition. (b) Correct specification.

shows how to specify QOC correctly, where the condition (, EX = (EY) in the box nodes the (EX and - EY) variables bind only to identical sist values.

In general, once a query is specified, the resulting values are displayed in the template under the appropriate columns. If the result contains more tows than can be deplayed on the screen, most GWL nuplementations have function keys to allow screelling up and down the rows. Similarly, if a template is several templates are too wide to appear in the screen, it is possible to screll adeways to examine all the templates.



FIGURE D.5 Illustrating (OIN and result relations in QBE, (a) The query Q1, (b) The query Q8.

A join operation is specified in Q6E by using the same antiable in the columns to be joined. For example, the query Q1: "List the name and address of all employees who work for the 'Research' deputiment," can be specified as shown in Figure 9.9(a). Any number of joins can be specified as shown in Figure 9.9(a). Any number of joins can be specified as shown in Figure 9.9(a). Any number of joins can be specified as shown in Figure 9.9(a). Any number of joins can be specified in a single query. We can also specify a result table its display the result of the join query to shown in Figure 9.9(a); this is needed if the result includes attributes from two or more relations. If no result table is specified, the system provides the query result in the calumns of the various relations, which may make it difficult to interpret. Figure 9.9(a) also illustrates the feature of Q9E for specifying dot, all authors of a relation should be particled, be placing the R operator under the relation more in the relation template.

To join a table with itself, we specify different variables to regression the different references to the table. For example, query Q8—"for each employee retrieve the employee's first and last name as well as the first and last name of his or her unneedone supervisor! can be specified as shown in Figure 9.9(b), where the variables starting with **B** refer to an employee and those starting with S refer to a supervisor.

# D.2 GROUPING, AGGREGATION, AND DATABASE MODIFICATION IN QBE

Next, consider the types of queries that require grouping or aggregate functions. A grouping operator G (can be specified in a column to indicate that suples should be geoged by the value of that column. Common functions can be specified, such as AVO, SUV, CNL (court), MAX a and MiN. In QRE the functions AVO, SUV, and CNL are applied to distinct values within a group in the default case. If we want these functions to apply to all values, we must use the prefix AEC.<sup>3</sup> This convention as *fd*/evolutin SQL, where the default is to apple a function to all values.

Figure 9.10(a) shows query Q23, which counts the number of distant salary values in the F991083 (elation, Query Q25A (Figure 9.10b) counts all salary values, which is the same as counting the number of employees. Figure 9.10(c) shows Q24, which is the each department rundler and the number of employees and average salary within each department; hence, the fast column is used for grouping as indicated by the O. function. Several of the operators O., E., and A11 cars by specified an a single column. Figure 9.10(d) shows query Q26, which displays each project name and the number of employees working on it for projects on which more than two employees work.

QRF has a negation symbol,  $\neg$ , which is used in a matter similar to the NOT EXISTS function in  $\leq 0$ . Figure 9.11 shows query Q6, which lists the names of simplicities who have not dependents. The negation symbol  $\neg$  says that we will select values of the  $\_s\lambda$  variable from the formation relation only if they do not occur in the 05000 for relation. The same effect can be produced by placing a  $\neg \_s\lambda$  in the 6800 m.

| (a) | EMPLOYEE     | FNOME   | MINE               | UNAME    | SŚŃ     | i slare i | ADDRESS      | SEX SALAR   | 1 SUPPASSN                            |      |
|-----|--------------|---------|--------------------|----------|---------|-----------|--------------|-------------|---------------------------------------|------|
|     |              | Ι.      | I                  |          |         | ' I       |              | I peve      |                                       | I '  |
|     |              |         |                    |          |         |           |              |             |                                       |      |
| (4) | EMPLOYFE     | FNAME 1 | MINIT <sup>1</sup> | LNAME    | SSN     | BEATE     | ADDRESS      | SEX / SALAR | V SUPERSEN                            | DNO  |
|     |              | · ·     | 1                  |          |         | ' '       |              | DOM AL      | L .                                   |      |
|     |              |         |                    |          |         |           |              |             |                                       |      |
| ¦c) | EMPLOYEE     | FNAME   | MINT               | INAME    | 55N     | BCATE     | ADORESS      | SFX SALAP   | U SUPERSSN                            | CheO |
|     |              | ·· ·· · | • •                | -        | PONTALL | ; - 1     | · ··· · ···· | TEANSAL     | · · · · · · · · · · · · · · · · · · · | TegT |
|     |              |         |                    |          |         |           |              |             |                                       |      |
| {0} | PROFECT      | PNALE   | PHA                |          |         |           |              |             |                                       |      |
|     |              | P       |                    | PX       |         |           |              |             |                                       |      |
|     |              |         |                    | -        |         |           |              |             |                                       |      |
|     | WEEKS ON     | = 585N  |                    | Пюсії    | 8 I     |           |              |             |                                       |      |
|     |              | PCAT E  | 1 3 10             | <u>.</u> | $\neg$  |           |              |             |                                       |      |
|     |              |         |                    |          |         |           |              |             |                                       |      |
|     | CONDITIONS   | :       |                    |          |         |           |              |             |                                       |      |
|     | L 1041 (HX+2 | _       |                    |          |         |           |              |             |                                       |      |

**FIGURE D.6** Functions and grouping in Q81. (a) The query Q23. (b) The query Q23A. (c) The query Q24. (d) The query Q26.

5. M.C. in QBE is unrelated to the unscensal quantifier.



FIGURE D.7. Illustrating negation by the query Q6.

Although the QBL language as originally proposed was shown to appoint the equivalent of the EXISTS and NOT EXISTS functions of SQL, the QBE implementation in QMF (index the DBZ system) dues not provide this support. Hence, the QMF version of QaE, which we discuss here, is not relationally complete. Queries such as Q3—"Find muplicyces when work en all projects currentled by department 5"—cannot be specified.

There are three Q65 operators for modifying the database: I, for insert, D, for delate, and U, for update. The coster and delete operators are specified in the template columns under the relation name, whereas the update operator is specified on the template columns to be updated. Figure 9.12(a) shows how to insert a new consist update, for deletion, we first enter the D, operator and their specify the tuples to be deleted by a condition (Figure 9.12b). To update a tuple, we specify the U, operator under the attribute name, followed by the new value of the jurribute. We should also select the tuple of tuples to be updated in the usual way. Figure 9.12(c) shows an update request to increase the solary of (John Smith' by 10 percent and also to reassign him to deputinent number 4.

Q8E also has data demation capabilities. The tables of a database can be specified interactively, and a table definition can also be updated by adding, renaming, or removing a column. We can also specify various characteristics for each column, such as whether it is a key of the relation, what its data rype is, and whether an index should be created on that field, Q8E also has facilities for view definition, authorization, storing query definitions for later use, and so end.

QR, does not use the "Imean" style is seen rather, it is a "two-dimensional" language, because users specify a query moving around the full area of the screen. Tests on users

| (a) | EMPLOYEE   | FNAME         | MINIT | LNAME        | SSN        | REATE     | ADDRESS                     | SEX. | FIGLARY | SUPPASSN   | DNO   |
|-----|------------|---------------|-------|--------------|------------|-----------|-----------------------------|------|---------|------------|-------|
|     | · · ·      | dutant        | A.    | 9au          | HALLANDER  | includes. | "On-Claic reasoning any The | . H. | 2,08,01 | 96 1654(21 | 4     |
|     |            |               |       |              |            |           |                             |      |         |            |       |
|     |            |               |       |              |            |           |                             |      |         |            |       |
| (D) | EMPLOYEE   | <b>FINAME</b> | NIM I | <b>CNAME</b> | SSN        | BOWE      | A004ES8                     |      | SALAHY  | SUPERSSN   | CANO  |
|     | D 10       | '             |       | ·            | :511-98658 | -         |                             | . –  | -       |            | · '   |
|     |            |               |       |              |            |           |                             |      |         |            |       |
|     |            |               |       |              |            |           |                             |      |         |            |       |
| (C) | F MOLOH'SH | FNAME         | Mihil | INAME        | 555        | BUALE     | A JAHE SS                   | SH X | GALAHY  | Add+168N   | LINES |
|     |            | ,John 1       |       | tamelu       |            | I         |                             |      | 0.800   |            | 0.1   |

FIGURE D.8 Monitying the database in QBT (a) Insertion, (b) Deletion, (c) Update in QBT.

have shown that QPE is cover to learn than SQI, especially for non-prevalets. In this sense, QRE was the first user-friendly "visual" relational database barguage.

More morently, numerics other inter-friendly interfaces have been developed for coninternal database systems. The use of memor, graphics, and forms is now becoming quite complete. Visual every languages, which are still not so commun, are likely to be effected with completeial relational databases in the lutture.

# Selected Bibliography

Abbreviations Used in the Dibinoeraphy ACM: Association for Computing Machinery AFIPS - American Federation of Information Processing Societies CACM: Communications of the ACM (coursal). CIKM: Proceedings of the International Conference on Information and Knowledge Management EDS. Proceedings of the International Conference on Experi Database Systems ER Conference: Proceedings of the International Conference on Entity-Relationship Approach (now called International Conference on Conceptual Modeling) 10DE: Proceedings of the IEEE International Conference on Data Engineering IEEE: Institute of Electrical and Electronics Engineers IEEE Compaters Computer magazine (pornal) of the IEEE CS IEEE CS. IEEE Computer Society IEIP: International Rederation for Information Processing TACM: Journal of the ACM KDD: Knowledge Discovery in Databases **UNCS: Lecture Notes in Computer Science** NCC: Proceedings of the National Computer Conference (published by AFIPS)

OOPSLA: Proceedings of the ACM Contenence on Object-Oriented Programming Systems, Languages, and Appacations.
PODS: Proceedings of the ACM Symposium on Principles of Database Systems.
SICMOD: Proceedings of the ACM SIGMOD International Conference on Management of Data.
TKDE: IEEE Transactions on Knowledge and Data Engineering (journal).
PODS: ACM Transactions on Database Systems (journal).
TODS: ACM Transactions on Information Systems (journal).
TODS: ACM Transactions on Information Systems (journal).
TODS: ACM Transactions on Office Information Systems (journal).
TODS: ACM Transactions on Office Information Systems (journal).
TODS: ACM Transactions on Office Information Systems (journal).
TSE: IEEE Transactions on Software Engineering (journal).
VIDB: Proceedings of the International Conference on Very Large Data Bases (iscestate, 1981 available from Morean Kaufmann, Menlo Park, Colifornal).

## Format for Bibliographic Citations

Book titles are in boldface—for example, **Database Computers**. Conference proceedings names are in status—for example, ACM Pacific Conference, Journal names are in bulk bac—for example, TODS or **Information Systems**. For journal citations, we give the volinne number and usine number (within, the volume, if any) and date of issue. For example "TODS, 164, December 1978" refers to the December 1978 issue of ACM Transmission on Database Systems, which is Volume 3. Number 4. Articles that appear in bunks or conterence proceedings that are themselvers cised in the bibliography are referenced as "in" these references—for example, "in 20196 [1978]" or "in Rustin [1974]." Page numbers (abbreviated "pp.") are provided with pp. of the end of the citation whenever available. For citations with more than tour outbors, we call give the first author only followed by et al. In the selected bibliography at the end of each chapter, we use et al. of there are more than two authors.

## **BIBLIOGRAPHIC REFERENCES**

- Abborr, R., and Garcus-Mohna, H. [1989] "Scheduling Real-Time Transactions with Disk Resident Data," in VID8 [1989]
- Abitebour, S., and Kaneflakis, D (1989) "Object Identity as a Query Language Frantise," in SIGMOD [1989].
- Abitchind, S. Hull, R., and Visino, V. (1995) Foundations of Databases, Addison-Wesley, 1995.

Abrial, J. [1974] "Data Scinantics," in Klimbie and Kotteman [1974].

Adam, N., and Compopedhyay, A. [1993] "Integrating Forestional and Data Modeling ma-Computer Integrated Manufacturing System "in KDS [1993].

- Adrigans, P. and Zaminge, D. [1996] Data Mining, Addison-Wesley, 1996.
- Afsarmanesh, H., McLeud, D., Knapp, D., and Parker, A. [1985] 'An Evrensible Object-Oriented Approach to Databases for V150(DAD," in Vi 88 [1985])
- Agraval, D., and ElAbbadi, A. (1990) "Storage Efficient Replacated Databases," TKDE, 2(3, September 1990).
- Agraval, R., and Gehani, N. [1989] "ODE: The Language and the Data Model," at SIX-MOD [1989].
- Agrawal, R., Gehani, N., and Sunivasin, J. [1990] "OdeView: "The Graphical Interface ro-Cole," in SIGVA 3: [1990]
- Agrawol, R., Jimehnski, T., and Swami A. [1993] "Mining Association Bale- Between Serie of Items in Databases," in 503400 [1993].
- Agrawal, R., Imrelinski, T. and Swami, A. (19936) "Database Mining: A Performance Perspective," IEEE TRDE 5-6 [December 1993]
- Agraval, R., Mehta, M., and Skater, J. and Srikant, R. [1996] "The Quest Data Mining System," in KDD [1996].
- Agrawal, R., and Sukant, R. [1994] "Fast Algorithus for Mining Association Rules in Large Databases," in VED8 [1994].
- Alcal, R., and Basu, A. (1991) "ESQL: A Query Language for the Relational Model Supporting heap: Domains," in EDE [1991]
- Aho, A., Been, C., and Uihnan, J. [1979] "The Theury of Jones in Relational Databases," TODS, 4:3, September 1979.
- Aho, A., Sagw, Y., and Ullman, J. [1979a] "Efficient Optimization of a Class of Relational Expressions," (DDs, 4:4, December 1979.
- Abo, A. and Ollman, J. [1979] "Curversality of Dara Retrieval Unigeoges," Proceedings of the POPL Conference, San Automic TX, ACM, 1979.
- Akl, S. [1983] "Digital Signatures: A Totorial Survey," (EVE Computer, 16:2, February (1983).
- Alashqur, A., Su, S., and Lom, H. [1989] "OQL: A Query Longuage for Manipulating Object-Otiented Databases," in VLDB [1989].
- Albano, A., Cardella, L., and Orani, R. [1985] "GALICES: A Strongly Typed Interactive Conceptual Language," TODS, 10:2, [une 1985.
- Allen, E. Leonis, M., and Matmiros, M. [1982] "The Integrated Dationary/Directory System," ACM Computing Surveys, 14-2, June 1982.
- Alonso, G., Agrawal, O., El Abbadi, A., and Mohao, C. [1997] "Functionalities and Lumitations of Correct Workflow Management Systems," IEEE Expert. 1997.
- Amu, A., Feldman, R., and Kashi, R. [1997] "A New and Versatile Method for Association Generation," Information Systems, 22:6, September 1887.
- Anderson, S., Bankier, A., Barrell, B., deBruijn, M., Coulsen, A., Duman, J., Eperon, L., Nietheb, O., Bose, B., Sanger, L., Schreier, D., Smith, A., Staden, R., Yeune, I. [1951] "Sequence and Organization of the Human Mitochondrial Genome." Nature, 250:457–465, 1951.

- Andrews, T., and Harris, C. [1987] "Combining Language and Database Advances in an Object-Oriented Development Environment," (OPSLA, 1987)
- [0881]1975] American National Standards In-Oture Study Group on Data Base Marogement Systems: Interim Report, (DT, 7:2.1803), 1975.
- ANSI [1986] American National Standards Institute: The Database Language 50t. Document ANSI X3.135, 1986.
- ANSI [1986a] American National Standards Institute: The Database Language NDL Ducoment ASSI X3 133, 1986
- ANS, [1989] American National Standards Institute: Information Resource Deciency, Systems, Discument 48851X3,138, 1989.
- Anwar, T., Beck, H., and Navarhe, S. [1992] "Knowledge Maturg by Imprecise Querying: A Classification Based Approach," in 1006 [1992].
- Apers, P., Heyner, A., and Yao, S. [1983] "Optimization Algorithms for Distributed Quenes," 18E, 801, January 1983.
- Armstrong, W. [1974] "Opendence Structures of Data Base Relationships," Proceedings of the (E) Congress, 1974.
- Astraham, M., et al. (1976) "System R. A Relational Approach to Data Base Management," TODs, I 2, June 1976.
- Arkinson, M., and Boneman, P. [1987] "Types and Persistence in Dotabase Programming Languages" in ACM Computing Surveys, 19-2, June 1987.
- Arlon, V. Jajoda, S., Keefe, J.F., McColloni, C., and Mukkanada, R. [1997] "Multilevel Secure Transaction Processing: Status and Prospects," in Database Security: Status and Prospects, Chapman and Hall (1997, pp. 79–98).
- Attern, P., and De Antonellis, V. [1993] Relational Database Theory, Denjamars/Junimings, 1993.
- Attent, P., Mecca, C., and Menaldo, P (1997) "To Weave the Web," in VLDP [1997].
- Bachman, C. [1909] 'Data Structure Diograms," Data Base (Bulletin of ACM SIG-104) F. 1(2), March 1969.
- Bachman, C. [1973] "The Programmer as a Navigator," CACM, 16:1, Novimber 1973.
- Bachman, C. [1974] "The Data Structure Set Model," in Rustin [1974].
- Bachman, C., and Williams, S. [1964] "A General Purpose Diogramming System for Random Access Memories," Proceedings of the Falt June Computer Conference: AFPS, 26 [964].
- Budal, D. and Popek, G. [1979] "Clost and Performance Analysis of Semantic Progray Validation Methods" in SatMop [1979].
- Bodrinath, B. and Bonaterriham, K. (1992) "Semantics Based Concorrency Control Beyond Commutativity" TODS, 17-1, March 1992
- Baera-Yares, R., and Larson, P. A. (1989) "Performance of 60-trees with Parcial Expansions," TKDE, 1-2, June 1989.
- Boeza-Yares, R., and Ribero-Nero, B. [1999] Modern Information Remeval. Addison-Wesley, 1999.

- Balbin, L. and Romanobunico, K. [1937] "A Generalization of the Different Approach to Recursive Query Evaluation," Journal of Logic Programming, 15:4, 1987.
- Boracilson, E., and Bonanan, P. eds. [1990] Advances in Database Programming Languages. ACM Press, 1990.
- Duncilhon, E. Delobel, C., and Kattellakis, P., eds. [1992] Building an Object-Oriented Database System: The Story of O2, Morean Kaumann, 1992.
- Panellion, E. Maier, D., Sagiy, Y., and Ullman, J. [1986] "Magn. sets and other strange ways to mij lement logic programs," DODS [1986].
- Benerlhon, E., and Ramakushnan, R. [1956] "Ap Amateur's Introduction to Recursive Query Processing Strategies," in Station (1986).
- Banerjee, J., et al. [1957] "Data Model Issues for Object/Oriented Applications," TOOIS, 5:1, January 1957.
- Banerjee, J., Kim, W., Kim, H., and Karth, H. [1957a] "Semantics and Implementation of Schema Evolution in Object-Coreneed Databases," in *Academy* [1987].
- Barcody, A., and DeWitt, D. (1981) 'An Object/Original Approach to Database System Implementation." TODS, 6:4, December 1981.
- Barsalou, T., Siambela, N., Keller, A., and Wiederbold, G. [1991] "Updrving Relational Databases Through Object Based Views," in a GMOD [1991].
- Bassioent, M. [1988] "Single Site and Distributed Opmaistic Protocols for Concurrency Council," 188, 1488, August 1988.
- Barm, C., Cerr, S., and Navarhe, S. [1992] Database Design: An Entity-Relationship Approach, Benjamm/Commings, 1992.
- Batim, C., Lenterm, M., and Navarbe, S. [1957] "A Comporative Analysis of Methodologies for Database Schema Integration," ACM Computing Surveys, 18(4, December 1987)
- Barory, D., and Buchmann, A. [1954] "Molecolar Objects: Abstract Data Types, and Data Models: A framework," in VEDV[1984].
- Barory, D., et al. [1988] 'GENESIS' An Extensible Database Management System," TSE, 14(1), November 1988
- Bayer, R., Grahara, M., and Seeginuffer, G., eds. [1978] Operating Systems: An Advanced Course, Springer-Veilag, 1978
- Bayer, R., and McCreight, E. [1972] "Organization and Montemance of Large Ordered Indexes," Acta Informatica, 1:3. February 1972.
- Beck, H., Amear, T. and Navathe, S. (1995) 'A Conceptual Chatering Algorithm for Database Schema Design. TRDB, to appear.
- Brela, H., Oola, S., and Navathe, S. [1980] "Classification as a Query Processing Technique in the CONDUE Semanne Data Model," jp. a (pt. [1989]).
- Beern C., Eagin, R., and Howard, J. [1977] "A Complete Assoniatization for Endemonaland Multivalued Dependencies," in SteMOD [1977].
- Beeri, C., and Ramakrishnan, R. [1967] "On the Power of Magic" in 1008 [1987].
- Benson, D., Bogiski, M., Lipmus, D., and Okrell, J. "GenBank," Nucleic Acids Research, 24 1, 1996.

- Ben-Zvi, J. [1982] "The Time Relational Model " Ph.D. dissertation: University of Calitornia, Los Angeles, 1982.
- Berg, B. and Roth, J. (1989) Sultware for Optical Disk, Meckler, 1989.
- Berners-Lee, T., Chillian, R., Groutt, J., Pullemann, F. (1992) "World-Wille Web: The Information Universe," Electronic Networking: Research, Applications and Policy, 1:2, 1992.
- Berners-Lee, T. Caillian, E., Lautonen, A., Nielsen, H., and Secut. A. [1994] "The World Wide Web." CACM, 132, August 1994.
- Berastein, P. [1976] "Synthesizing Third Normal Youn Relations from Functional Dependencies," TODS, 1:4, December 1975.
- Bernstein, P., Blaustein, B., and Clarke, E. [1980] "East Majorenance of Semantic Integtity Assertions Using Redundant Aggregate Data," in VLOB (1980].
- Bertstein, P. and Goodman, N. [1980] "Timestaup-Based Algorithms for Concutrency Control in Distributed Database Systems," in V206 [1980].
- Bernstein, P., and Goodman, N. [1981] "The Power of Natural Semijoins," SIAM Journal of Computing, 10:4, December 1981.
- Bernstein, P., and Goodman, N. [1989a] "Concurrency Control in D studented Database Systems," ACM Computing Surveys, 13 2, June 1981
- Beinstein, D. and Goodman, N. [1964] "An Algorithm for Concurrence Coercel and Recovery in Replicated Distributed Databases," TODS, 964, December 1984.
- Bernstein, P., Hadrilacos, V., and Goodman, N. [1985] Concurrency Control and Recovery in Database Systems, Addison-Wesley, 1988.
- Bertino, E. (1992) "Data Hiding and Security in Object-Oriented Databases," in 8 OE (1992).
- Burtono, E., Catanna, B., and Fernari, E. [2001] "A Nested Transactions Model for Multilevel Science Database Management Systems," ACM Transactions on Information and System Security, 4:4, November 2001, pp. 321–370.
- Bertino, E., and Ferrari, E. [1998] "Data Security," Taxiar Second Annual International Conference on Computer Software and Application. August 1998, pp. 228–237.
- Bertine, E., and Kim, W. [1989] "Indexing Techniques for Queries on Nesred Objects," TRDE, 102, June 1989.
- Bernno, E., Negri, M., Pelaguru, G., and Shattella, L. [1992] "Object-Oriented Query Languages: The Norton and the sates," TRDF, 4(3) [one 1997]
- Berrino, E., Pagoni, E., and Rosa, C. [1992] "hadt federance and Recovery in Mobile Computing Systems in Komputind Hap [1992].
- Bernno, F., Rabhani and Gibbs, S. [1988] "Query Processing in a Makimedia Environment," TOIS, 6, 1988
- Bhargava, B., ed. [1967] Concurrency and Reliability in Distributed Systems, Van Nostrond-Rembold, 1987
- Bhargava, B., and Helaí, A. [1995] "Efficient Reliability Mechanisms in Distributed Database Systems,", 1889; Novembur 1993.

- Bhargaya, B., and Beidl, J. [1955] "A Model for Adaptable Systems for Transaction Processing," in 10 4 [1988]
- Bilins, A. '1992] "The Performance of Three Database Storage Structures for Managing Large Objects," in 465001 [1992].
- Biller, H. [1979] "On the Eccivalence of Data Base Schemas—A Schumtic Approach to Data Translation," Information Systems, 4:1, 1979.
- Bischoff, J., and T. Alexander, eds., Data Warehouses Practical Advice from the Experts, Prentice 4140, 1997.
- Biskup, J., Daval, U., and Bernstein, E [1979] "Synthesizing Independent Database Schemas," in 51040031979].
- Biork, A. [1973] "Recurrent Scenario for a 20/DC System," Proceedings of the 30.36 National Conference, 1973
- Borner, D., and Lovengren, H. [1982] "Formalization of Database Systems and a Formal Definition of DIST in VLIS" [1982].
- Piaña, M., Fremerlant, W. [1998] Object-Oriented Modeling and Design for Database Applications, Prentice-Hall, 1998.
- Blakeley, J., Coburn, N., and Larson, F. (1989) "Updated Derived Relations: Detecting Irrelevant and Autonomously Computable Updates," TODS, 14(3). September 1989.
- Blakeloj, J., and Martin, N. [1990] "Join Index, Materialized View, and Hybrid-Hash Join: A Performance Analysis," in tech. [1990].
- Blayern, M., and Eswaran, K. [1976] "On the Evaluation of Quenes in a Relational Database System," JBM Systems Journal, 16:1, January 1976.
- Blasgen, M., et al. [1986] "System R: An Architectural Overview," IBM Systems Journal, 20(1, January 1981.
- Bleter, R., and Vorbaus, A. [1966] "File Organization in the SOUTOMS," Proceedings of the 1819 Congress.
- Beeca, J. [3986] "ED16 "F—A: Marcage of Convenience: Prolog and J. Relational DBMS," Proceedings of the Third International Conference on Logic Programming: Springer-Verlae, 1966.
- Bocca, J. [198ba]. 'On the Evaluation Strategy of 2000b," in SIGMOD [1986].
- Bodora k, P., Riardon, J., and Pyra, J. (1992) "Dociding on Correct Distributed Query Processing," TKDF, 4:3, June 1992.
- Booch, G., Rumbaugh, J., and Jacobson, L. Unified Modeling Language User Guide, Addison-Wesley, 1999.
- Borguda, A., Brachman, R., McCinjnness, D., and Resnick, L. [1989] "U.J.A.SSU," A Structural Data Model for Objects," in - (5000 [1989].
- Borkin, S. [1978] "Data Model Equivalence," in VLVP [1976].
- Boazeghoub, M., and Merais, E. [1991] "Semantic Modelling of Object-Oriented Databases," in VLDP [1991].
- Poyce, R., Chamberlin, D., King, W., and Hammer, M. [1975] "Specifying Quenes as Relational Expressions," CACM, 18:11, November 1975.

- Bracchi, G., Paolim, P., and Pelagatti, G. [1976] "Binary Logical Associations in Data Modelling," in Nussen [1976].
- Brachman, R., and Levesque, H. [1984] "What Makes a Knowledge Base Knowledgeable? A View of Databases from the Knowledge Leves" in FPS [1984].
- Brothengenger, K. (1984) "Hashing Methods and Relational Algebra Operators," in VUDB [1984].
- Bray, O. [1986] Computer Integrated Manufacturing—The Data Management Strategy, Digital Press, 1988.
- Breithart, Y., Silberschutt, A., and Thompson, G. [1990] "Reliable Transaction Managemean macMelinLatobase System," in Stitution (1990).
- Brodie, M., and Mylopoulos, J., eds. [1985] On Knowledge Base Management Systems, Springer-Verlag, 1985.
- Brodie, M., Mylopoulos J., and Schnidts J., eds. [1984] On Conceptual Modeling, Springer-Verlag, 1984.
- Brosey, M., and Shneulerman, B. [1978] "Two Experimental Companions of Relational and Therarchical Database Models," International Journal of Man-Machine Studies, 1978.
- Biy, E [1990] "Query Evaluation in Recursive Databases: Portom-up and Top-down Reconeded." TRDE, 2, 1980.
- Bukhres, O. [1992] "Performance Comparison of Distributed Deadlock Detection Aleombins," in 1010 [1992].
- Baneman, P., and Frankel, R. [1979] "Peter A Functional Query Language," in SIGMOD [1979].
- Barkhurd, W. [1976] 'Hashing and The Algorithus for Partial Match Retrieval, "TODS, 1(2) June 1976, pp. 175–87.
- Barkhard, W. [1979] "Parrial match Hash Codarg: Perions of Redamancy," TODS, 4.2, June 1979, pp. 228–59.
- Bash, V. [1945] "As We May Think." Advance Monthly, 15(5), January 1945. Reprinted in Kochen, M., ed., The Growth of Knowledge, Wiley, 1967.
- Byre [1995] Sperial Issue on Mobile Computing, June 1995
- (20.3) [1995] Special uses of the Communications of the ACM, on Digital Library, 38.5, May 1995.
- CAUM [1998] Special issue of the Communications of the ACM on Utertal Libraries. Global Socyce and Undimited Access. 41:4, April 1998.
- Connectora, S., Remachandra, P. and Shone, D. [1989] "Extending a Relation (Dorabase with Defended Referencial Integrity Checking and Intelligent Joins," in 583609 [1989].
- Campbell, D., Embley, D., and Czeido, B. [1985] "A Relationally Complete Query Language for the Entry-Relationship Model," in ER Conference [1985]
- Cardenss, A. [1985] Data Base Management Systems, 2nd ed., Allyn and Bacan, 1985.

- Carey, M., et al. [1956] "The Architecture of the EXOLOF Extensible DBMF." in Damehand Poyal [1986].
- Carev, M., DeWarr, D., Richardson, J. and Shekira, E. [1986a] "Object and File Management in the EVODUS Extensible Dotabase System," in VLCP [1986].
- Carew M., DeWitt, D., and Vandenberg, S. [1988] "A Data Model and Query Language for Evodus," in SIGNOD [1988]
- Catey, M., Frankhir, M., Livny, M., and Shekata, E. [1991] "Data Carling Tradeotis in Chem. Server 19945 Architectures," in SIGMAD [1991].
- Carlis, J. [1986] TTAS, a Relational Algebra Operator or Divide Is Nov Enough to Conquer," in ICPE [1986]
- Carlos, J., and March, S. [1984] "A Descriptive Model of Physical Database Design Problens and Solutions," in 1001 [1984].
- Carroll, J. M., [1995] Scenario Based Design: Envisioning Work and Technology in System Development, Wiley, 1995.
- Casanova, M., Fagin, R., and Papadimitriou, C. [1981] "Inclusion Expendencies and Their Interaction with Emotional Dependencies," in (2005 [1981]).
- Casanova, M., Funado, A., and Tuchermann, L. [1993] "A Softwore Tool for Modular Database Design," roots, 16.2, June 1993.
- Czsanova, M., Tuchermann, L., Fursado, A., and Braga, A. (1989] "Optimization of Relational Schemas Containing Inclusion Dependencies," in VLOB [1989]
- Casurova, M., and Vidal, V. [1982] "Toward & Sound View Integration Method." in PODS [1982].
- Cattell, R., and Skeen, J. [1992] "Object Operations Perichmark," TODS, 17-1, March 1992.
- Castano, S., DeAriterellio, V., Fugini, M.G., and Perna i, B. [1998] "Conceptual Schema Analysis: Techniques and Applications," TODS, 23:3. September 1998, pp. 286-332.
- Castano, S., Fogun, M., Marvello G., and Samaratj, P. [1995] Database Security, ACM Press and AdMison Wesley, 1995.
- Catarei, T., Costabile, M. E. Saratuera, G., and Taraprigga L., eds. [1993] Proceedings of the French International Workshop on Advanced Visual Interfaces, ACM Press, 1998.
- Cararer, T., Costalule, M. E. Levioldi, S., and Barini, C. [1997] "Visual Query Systems for Databases: A Survey," Journal of Visual Languages and Computing, 5(2) June 1997, pp. 215–62.
- Cattell, R., ed. [1993] The Object Database Standard: ODMG-93, Release 1-2, Morgan Kaufmann, 1993.
- Catrell, R., ed. [1997] The Object Database Standard: ODMG, Release 2.0, Morgan Kautmann, 1997
- Cert, S., and Fraternoli, P. [1997] Designing Database Applications with Objects and Rules: The IDEA Methodology, Addison-Wesley, 1997.
- Cert, S., Gottlob, G., Tanca, L. [1990], Logic Programming and Databases, Springer-Verlag, 1992.

- Corn S., Navarhe, S., and Wiederhold, G. (1983) "Distribution Dosign of English Data Even Schemats," TSE, 9:4, July 1983.
- Cert, S., Negri, M., and Petagarn, G. [1982] "Horizonial Data Partitioning in Database Design," in StoMetr [1982].
- Cert, S., and Owacki, S. [1983] "On the Use of Optimistic Methods for Concorring Control in Distributed Databases," *Proceedings of the Neub Reckeles Workshop on Disrelated Data Management and Computer Neusonks*, February 1983.
- Cera S., and Felgatti, G. [1984] "Correctness of Query Execution Strategies in Dutrisuted Databases," TODS, 8-4, Eccember 1984
- Cer., S., and Pelakatu, G. [1984a] Distributed Databases: Principles and Systems, McGraw-Hull, 1984.
- Cen, S., and Tanca, L. [1987] "Optimization of Systems of Algebraic Equations for Evaluating Database Queries," in VD9 [1987].
- Cesarina, E. and Soda, G. [1991] "A Dynamic Hash Method with Signature," TODS, 16-2, June 1994.
- Chakravarthy, S. [1990] "Aurice Database Management Systems: Requirements, State-ofthe-Art, and an Ecahartjon," in FR Conterence, 1990].
- Chakracarthy, S. [1991] "Decide and Conquer: A Basis for Augmenting a Conventional Query Optimizer with Multiple Query Processing Capabilities," in 6:05 [1991].
- Chakroverby, S., Anway, E., Manges, I., and Mishra, D. [1994] Design of Senrinel: An Object-oriented (1998) with Event based Rules, Information and Software Technology, 36 9, 1994.
- Chaktavardw, F. et al. (1989) "HiPAC: A Research Project in Active, Time Constrained Database Manageraem," Final Technical Report. XAI(1-69-02, Nerox Advanced Information Technology, August 1969.
- Chakravarthy, S., Karlapalene, K., Navarhe, S., and Junaka, A. [1995] "Database Supported Co-operative Problem Solving," in International Journal of Intelligent Cooperative Information Systems, 2(3), September 1995.
- Chakravarthy, O., Grant, J., and Minker, J. [1990] "Logic-Pased Approach to Semantic Query Optimization," TODS, 15:2, June 1990.
- Chalmers, M., and Christon, P. (1992) "Bead: Explorations in Information Visualization," Proceedings of the ACM 2018 International Conference, June 1992.
- Chamberlin, D. and Boyce, R. [1974] "SEQ 41. A Structured English Query Language," in FERMOR [1984].
- Chamberlin, D., et al. [1976] "Solgart 2: A Unified Approach to Para Definition, Manapulation, and Control " **BM Journal of Research and Development**, 20:6, Newcouler 1976.
- Chamberlin, D. et al. [1981] "A Horon and Evaluation of System R." CACM, 24 10, October 1981.
- Chan, C., Oo, B., and Lu, H. [1992] "Extensible Buffer Management of Indexes," in verse [1993].

- Chande, K., Browne, J., Dosley, C., and Uhng, W. [1975] "Analytical Models for Roll back and Recovery Strategies in Database Systems, "TSE, 10: March 1975.
- Chang, C. [1981] "On the Evaluation of Queries Community Derived Relations in a Relational Database" in Ci-Plane et al. [1984].
- Chang, C., and Walker, A. [1984] "BOSQUEA Prolog Programming Interface with SQL" US," in 505 [1984].
- Chang, E., and Kora, R. [1989] "Exploriting Inheritance and Substate Semantics for Effective Clustering and Baffering in Object-Oriented Databases," in ScottO1 [1989].
- Charae, N., and Fu, K. [1981] "Picture Query Languages for Electronal Databases," IEEE Computer, 14:11, November 1981.
- Chang, F. and Myre, W [1988] "0811 Ef Database Manager: Overview and Technical Highlights," IBM Systems Journal, 27:2, 1985.
- Chilag, S., Lui, B., and Walser, R. [1979] "Generalized Zowining Techniques for Picturial Database Systems," Nucl. AIRS, 48, 1979.
- Chen, M., and Yu, E [1994] "Determining Beneficial Semijains for a Join Sequence in Distributed Query Processing," in ICPE [1994].
- Chatcoglu, P. D., and McCharley, L. A. [1997] "Requirements Capture and Analysis: A Survey of Current Fraction," Requirements Engineering, 1997, pp. 75–88.
- Chaodhan, S., and Dayd, D. [1997] "An Overview of Data Warehousing and OUAP Technology," SIGNOD Record, Vol. 26, No. 1, March 1997.
- Chen, M., Han, J., Yu, P.S., [1996] " Data Mining: An Overview from a Database Perspectnye," IEEE TKDE, 8(6), December 1996.
- Chen, P.(1976) "The Entry Relationship Modes" Toward a Unified View of Data," TODS, 141, March 1976.
- Chen, P. Lee E. Oilson G., Kott, R., and Patterson, D. [1994] RAID High Performance. Reliable Secondary Stomeet ACM Computing Surveys, 26(2) 1994.
- Chen, P., and Potterson, D. [1993]. "Maximuting performance in a striped disk array," in Proceedings of Spoipsing on Computer Architecture, 1812, New York, 1990.
- Chen, Q., and Kambujushi, Y. [1991] "Ne-ted Relation Based Database Knewledge Representation," in StoMOP [1991].
- Cheng, J. [1991] "Effective Clustering of Complex Objects in Object-Oriented Databases," in SlOMCU [1991].
- Uheang, D., Han, J., Ng, V., Fu, A.W., and Fu, A.Y., "A Fast and Distributed Algorithm for Mining Association Rules," in *Proceedings of International Conference on Parallel*and Distributed Information Systems. PDIS [1996].
- Chikls, D. [1968] "Ecolubility of a Sci Theoremical Data Structure ---A Central Science Based on a Reconstituted Detruition of Relation," Proceedings of the 16th Contrasts, 1968.
- Uhimony, D., et al. [1987] "An Overview of the 153 System " MUU Technic P Report #AUA STE370-87. Austin, TX, November 1987.
- Chimeru, D., et al. [1990] "The OO System Pretotype," TKDE, 2.1, M arch 1990.
- Chin, F. [1978] "Security in Statistical Dotabases for Queries with Small Counts," TODS, 34, March 1978.
- Chin, F. and Otsoyoglu, G. [1981] "Statistical Patabase Design," TODS, 6-1, March 1951
- Chartalapati, R. Kumat, V. and Datta, A. [1997] "An Adaptive Location Management Alconthin for Mobile Computing," *Proceedings of 22nd Annual Conference on Local Computer Networks (LCN '97)*, Minnequells, 1997.
- Chou HJ, and Kim, W. [1986] 'A Unitying Framework for Version Control in a CAD Environment," in VLDB [1986].
- Christodoulakos, S., et al. [1934] "Development of a Multimedia Information System for an Office Environment," in Verbs (1984).
- Christindinutakis, S., and Faloutses, C. [1986] "Design and Ferturinance Considerations for an Optical Disk Based Multimedia Object Server," IEEE Computer, 19:12. December 1986.
- Chu, W., and Hurley, P. [1982] "Optimal Query Processing for Distributed Database Systerus," (EEE Transactions on Computers, 31:9), September 1952.
- Cuborra, C., Micliorese, E., and Romano, P. [1984] 'A Methodological Inquiry of Organizaronal Noise in Socio Technical Systems," Human Relations, 37:5, 1984.
- Claybrook, B. [1953] File Management Techniques, Wiley, 1963
- Claybrock, B. [1992] OLTP: OnLine Transaction Processing Systems, Wiley, 1982.
- Clifford, J., and Tansel, A. [1985] "On an Algebra for Horomed Relational Databases Two Views, Jin s165000 [1985]
- Clocksin, W. F. and Mellish, C. S. [1984] Programming in Prolog. 2nd ed., Springer-Verlag, 1984.
- CODASYL [1978] Data Description Language Journal of Development, Canadian Government Publishing Centre, 1978
- Codd. E. [1970] "A Relational Model for Large Shared Data Banks," CACM, 13-6, June 1970.
- Codd, S. [1971] "A Data Base Sublanguage Founded on the Relational Calculus," Proceedings of the ACM SIGFIDET Workshop on Data Description. Access. and Convol. November 1971.
- Codd. E. [1972] "Relational Completeness of Data Base Sublanguages," in Rustin [1972].
- Codd, E. [1972a] "Further Normalization of the Data Base Relational Model," in Rustin [1972].
- Codd. E. [1974] "Recent Investigations in Relational Database Systems," Proceedings of the 9512 Congress, 1974.
- Codd. 5. [1978] "How Abour Recently? (English During with Relational Data Bases Using Recidentions Version 1)," in Structbergan [1978].
- Codd. F. [1979] "Extending the Database Relational Model to Capture More Meaning," TODS, 4:4, December 1979.
- Cadd, F. [1982] "Relational Database: A Priorical Foundation for Productivity" CACM, 25:2, December 1982.

- Codd. E. [1985] "Is Your DBMS Ready Relational" and "Does Your DBMS Rule By the Rules?," COMPUTER WORLD, October 14 and October 21, 1985
- Codd, E. [1986] "An Evaluation Scheme for Database Management Systems That Are Clauned in Br Relational," in ICOS (1986).
- Codd, E. [1990] Relational Model for Data Management-Version 2. Addison Wesley, 1990.
- Codd, E. E. Codd, S. B. and Salley, C. T [1993] "Prividing (2) AP (On-Line Analytical Processing) to User Analysis An IT Mandate," a white paper at http://www.orborsett.com/CLAEhtral, 1993.
- Comer, D. [1979] "The Ubiquirous 8-tree," ACM Computing Surveys, 11-2, June 1979.
- Corner, D. [1997] Computer Networks and Internets, Prentice-Hall, 1997.
- Cornelio, A. and Navarhe, S. [1995] "Applying Acrive Database Models for Simulation," in Proceedings of 1993 Wiener Simulation Conference, IFFR, Los Angeles, December 1993.
- Covinadakos, S., Konellakis, P. C., and Vardi, M. (1996) "Polynomial-time Implication Problems for Unary Inclusion Dependencies," JACM, 37:1, 1996, pp. 15–46.
- Cine, I. §1992] "Deedle: A Visual Language for Object-Oriented Databases," in SIGW201 [1992]
- Cartice, 70 [1981] "Data Dictionaries: An Assessment of Corrent Practice and Problems," in v108 [1981]
- Cerrechia, A., Fasmon, K., Kotesbury, D., Roboins, R., and Pearson, P. [1993] "The GDB Human Genome Database Anno 1993 "Nucleic Acids Research, 21:13, 1993.
- Crejdo, B., Ehnasm, R., Rusmknewner, M., and Embley, D. (1987) "An Algebraic Language for Graphical Query Formulation Using in Extended Entity Relationship Model." *Proceedings of the ACM Chargener Science Conference*, 1987.
- Dahl, R., and Bubenko, J. [1982] "1009: An Interactive Design Tool for (200AS)1/00TO Type [Databases," in VUD8 [1982].
- Dahl, V. [1984] "Logic Programming for Constructive Database Systems," in ED- [1984].
- Das, S. [1992] Deductive Databases and Logic Programming, Addison-Wesley, 1992.
- Date, C. [1983] An Introduction to Database Systems, Vol. 2, Addison-Wesley, 1983.
- Date, C. [1983a] "The Outer Join," Proceedings of the Second International Conference on Entabases (ICOD-2), 1983.
- Date, C. [1984] "A Critique of the SQL Database Language," ACM SIGMOD Record, 24:3, Nuveraber 1984
- Date, C. [1995] An Introduction Database Systems, 6th ed., Addison-Wesley, 1995.
- Dare, C. J., and Darwen, H. [1993] A Guide to the SQL Standard. 3rd ed., Addison-Wesley
- Dare, C., and White, C. [1989] A Guide to D82, 3rd ed., Addren, Wrsley, 1989.
- Pate, C., and White, C. [1988a] A Guide to SQL/DS, Addison Workey, 1988.
- Davies, C. (1973) "Recovery Semantics for a DRUG System," Proceedings of the 2004 National Conference, 1973.

- [Dayal, U., and Fernstein, P. [1978] "On the Updatability of Relational Views," in VLOS [1978].
- Dayal, U., Hsu, M., and Luday, R. [1991] "A Transaction Model for Long-Running Activities," in VLID [1991].
- Dayal, U., et al. [1987] "2000Z Final Report," Technical Report 12(3):687-02, Computer Corporation of America, December 1987.
- DB7G [1976] Report of the CODASTE Data Base Eask Group (Ac3), April 1971.
- Delcandere, L., Lun, R., and Urban, S. [1991] "Object-Centered Constraints," in (CDF [1991].
- Di March, T. [1979] Structured Analysis and System Specification, Prentice (10), 1979.
- DeMichael, L. [1989] "Ferforming Operations Over Misparched Domains, "in tran-[1989].
- Dentring, D. [1980] "Secure Statistical Databases with Random Sample Quenes," 1005, 5.3, September 1980.
- Detuning, D., and Detuning, D [1979] "Data Security," ACM Computing Surveys, 114, September 1979, pp. 227–249.
- Deshpande, A. [1989] "Att Implementation for Nested Relational Databases," Technical Report, Ph.D. desciration. Indiana University, 1989.
- Dever, C., and Weeldrever, J. [1980, 201075] A Testbed for Distributed Database Research," Proceedings of the Asim Pacific Conference, 1980.
- Deserte, D. [1993] Client Server Computing, McGraw-Hill, 1993.
- DeWitt, D., et al. [1984] "Implementation: fermiques for Main Memory Databases," in 5408400 [1984].
- DeWitt, D., et al. [1942] "The Camma Database Machine Project," "KDL, 2:1. March 1990.
- DeWitt, D., Eutensack, P., Maier, D., and Veler, F. [1990] "A Stude of Three Alternative Workstation Server Architectures for Object Opicited Database Systems," in V115 [1990].
- Dhawan, C. [1997, Muhile Computing, McGraw-Hill, 1997.
- Dietrich, S., Friesen, O., W. Calliss [1998] On Deductive and Object Oriented Databases The VALIDITY Experience," Technical Report. Arizona State University, 1999.
- Duffe, W., and Helbnan, M. [1979] "Providy and Authentication," Proceedings of the IEEE, 67(3), March 1979.
- Diport, B., and Lavy M. (1993) Designing with Flash Memory. Ann thoras 1993.
- Dittrich, K. [1986] "Object-Oriented Database Systems: The Notion and the Issues," in Dittrich and Dayal [1986].
- Ditrijch, K., and Dayak, U., eds. [1986] Proceedings. J the International Workshop on Object-Oriented Database Systems. "EEE CS: Pacific Grave, CA, September 1986.
- Dirinch, K., Korr, A., and Mille, J. (1986) "An Event/Trigger Mechanism to Enforce Complex Consistency Constraints in Design Databases," in SIGNOT Record, 153, 1986.

- Dodd, C. (1809, "API+: A Language for Associative Data Handling in 117.," Proceedings of the Foll hear Computer Conference, AEDS, 29, 1969.
- Dodd. O. [1969] "Elements of Data Management Systems," ACM Computing Surveys, 1:2, huse 1969.
- Dogae, A., Oran, M. T., Bilins, A., Sellis, T., eds. [1994] Advances in Object-oriented Databases Systems, Springer Verlag, 1994.
- Dogie, A., [1995] Special Section on Electronic Commerce, ACM Signod Record 27:4, December 1998.
- Dis Sorros, C., Neubold, E., and Furtado, A. [1979] "A Data Type Approach to the Entity Relationship Model," in 58 Conference [1979].
- Da. D., and Tong, S. [1991] "Multilevel Extendible Pay-hung: A life Structure for Very Large Databases," TRDE, 3(5, September 1991.
- Du, H., and Ghanra, S. [1987] "A Framework for Efficient & July (CAD Futabases," in JCDT [1947].
- Dumas, P., et al. [1982] "SOBILE-Burotrque: Prospects for the Future," in Naffah [1982]
- Dumpala, S., and Arona, S. [1983] "Schema Translation Using the Entity-Relationship Approach," in US Conference (1983).
- Duninam, M., and Helai, A. [1995] "Mobile Comparing and Databases: Anything New " SIGMOD Record, 24:4, December 1995.
- Dwyer, S., et al. (1982) "A Diagnostic Digital Imaging System," Proceedings of the 414 CS Conference on Pattern Recognition and Image Processing, June 1982.
- Eastman, C. [1987] "Database Facilities for Engineering Design," Proceedings of the IEEE, 69:10, October 1981.
- EDS [1984] Expert Database Systems, Keischberg, L., ed. (Proceedings of the Fast breathtonicl Workshop on Expert Database Systems, Kinoch Island, SU, October 1984), Benjamin/Commungs, 1986.
- FOS [1986] Expert Database Systems, Kerschberg, U., ed. (Proceedings of the East Internaaonal Conference on Expert Database Systems, Charleston, SC, April 1986), Benjamin/Commungs, 1987.
- E188 (1988) Expert Database Systems, Kerschherg, L., ed. (Proceedings. J. the Second Internarional Conference on Expert Database Systems, Tysons Corner, VA, April 1988), Benjamm/Cummings (forthcoming).
- Eick, C. [1991] "A Methodology for the Design and Transformation of Conception Schemas," in VLDS [1991].
- EIAMade, A., and Touce, S. [1985] "The Group Paradigm for Concurrency Control," in SICSOD [1988].
- EIAHsadi, A., and Toneg, S. [1959] "Maintaining Availability in Partitional Replicated Databases," TODS, 14:2, June 1989.
- Ellis, C., and Nutt, G. (1950) "Office Informations Systems and Computer Science," ACM Computing Surveys, 12:1, March 1980

- Floragarmid A. K., ed. [1992] Database Transaction Models for Advanced Applications, Morgan Kautmann, 1992.
- Elmaganuid, A., Leu, Y., Litwin, W., and Rosenkiewicz, M. (1990) "A Multiclarabase Transaction Model for Interbase," in v108 [1990].
- Elmosti, R., James, S., and Kouarnapan, V [1994] "Automatic Class and Method Generarias for Object-Oriented Databases," *Encertings of the Third International Confer*ence on Edulating and Object-Oriented Databases (ES X9D-931, Physicis, A7, December 1994).
- Lineisti, R., Kootataonan, M. and Fernando, S. (1993). "Jemposal Database Modeling: An Object-Oriented Approach." Citoxi. November 1993.
- Elmann, R., and Larson, J. [1985] "A Oraphical Query Excility for E8 Databases," in E8 Conference [1985]
- Elmeszt, R., Larson, J., and Navarhe, S. [1956] "Schema Integration Algorithms for Federated Dirabases and Logical Database Desiry," Honeywell (SPD), Technical Report US, 50-9, 8212, January 1980.
- Elmasri, R., Sirrovas, P., and Thomas, G. [1987] "Fragmentation and Query Decomposition in the LCR Model," in ICLN. (1987).
- Ehmasn, R., Weeldreyer, J., and Heyner, A. [1965] "The Category Concept: An Extension to the Emity-Relationship Model," International Journal on Data and Knowledge Engineering, 1:1, May 1985.
- Flanasri, R., and Wiederhold, G. [1959] "Data Midel Integration Using the Structural Midel," in State 37 [1979].
- Elmant, R., and Wiederhold, G. [1967] "Structural Properties of Relationships and Their Representation," 2001, Astro-49, 1960
- Elmasn, R., and Wiederhold, G. [1981] "Complete A Forma", High Level Query Language for the Enviro-Belarianship Model," in F8 Conference [1981].
- Elmasti, R., and Wun, G. (1990) "Al feingerof Model and Query Longuage for ER Databases," which is 1996(), in value [1996].
- Emistry R., and Woo. C. [1990] "The Time Index: An Access Structure for Temporal Data," in VUDB [1990].
- Encelbact, O., and Loglish, W. [1968] "A Research Center for Augmenting Human Interlect," Proceedings of the Full Joint Computer Conference, AFDS, December 1968.
- Epstein, R., Sionebiaker, M., and Wong, E. [5978] "Distributed Query Processing in a Relational Database System," in SIGMOD [1978].
- FR Conference [1979] Entity-Relationship Approach to Systems Analysis and Design, Cheu, P., ed. (Proceedings of the First International Conference on Endity-Relationship Approach, Los Angeles, December 1979), North Holland, 1980.
- Us Conference [1981] Entity-Relationship Approach to Information Modeling and Analysis. Chem. P. eds. (Proceedings of the Second International Conference on Entity-Relationship Approach, Washington, October 1981], Elsevier Science, 1981.

- E3 Conference [1983] Entity-Relationship Approach to Software Engineering, Dwiss, U., Loodia, S., Ng, P. and Yeh, R., eds. (Proceedings of the Third International Confesence on Entity-Relationship Approach, Analysian, CA, October 1983), North-Helland, 1983.
- 66 Conterence [1985] Proceedings of the Fourth international Conference on Entiry Releasanship Approach. Lin, J., ed., Chicagos October 1985, 1041 CS.
- En Conference (1986) Proceedings of the Fifth International Conference on Enter-Relationship Approach, Spacesprotta, S., ed., Dipor, France, November 1986, Express Trages.
- ER Conference [1987] Prix coding- of dz Synh International Conference on Europel(clamorship Approach, March, S., ed., New York, November 1987.
- E8 Conference [1998] Proceedings of the Several International Conference on Divity-Relationdup Approach, Borring Cased, Rome, November 1955.
- ES Conference [1989] Proceedings of the Eighth International Conference on Entity-Rolationship Approach. Lochowsky, E. ed., To onto, Cecober 1989.
- ER Conference [1992] Proceedings of the North International Conference on Entity-Relationship Approach, Kangassalo, H., ed., Lansanne, Switzerland, September 1980.
- its Conference [1991] Proceedings of the Tenth International Conference on Energy-Relationship Approach. Teorey. T. ed., San Matrio, CA, October 1991.
- At Construction [1992] Proceedings of the Electrich International Conference on Futing-Relationship Approach, Period, G., and Tjun, A., eds., Karlsmite, Germany, October 1992.
- -n Contensare [1993] Proceedings of the Twelfth International Conference on Frank Relationship Approach, Elmosti, R., and Komanomara, V., eds. Athraton, TX, December 1993.
- 58 Conference [1994] Proceedings of the Three with International Conference on Lattry-Relationship Approach, Loucopoulos, P., and Theodoudicis, B., eds., Manchester, England, [December 1994]
- Contesence [1995] Proceedings of the Fotoxeenth International Conference on ER-OO. Modeline, Physicologica, M., and Jan. Z. eds., Busbane, Australia, December 1995.
- LR Condetence [1996] Proceedings of the Efficient's International Conference on Conceptual Modeling, Thalleeins, B., ed., Cetthus, Germany, October 1996.
- ER Conference [1997] Proceedings of the Statewith International Conference on Conceptual Modeling, Embley, D., ed., Los Angeles, October 1997.
- FR Conference [1998] Proceedings of the Scienteenth International Conference on Conceptual Modeling, Ling, T.-K., ed., Singapore, November 1998.
- Envaran, K., and Chamberlin, D. [1975] "Functional Specifications of a Subsystem for Database Integrity," in NUBS [1975].
- E-waran, K., Gray, E., Lorie, R., and Trager, J. [1976] "The Notions of Consistency and Predicate Locks in a Data Base System," CACM, 19:11, November 1976
- Everett, G., Dissly, C., and Haidgiave, W [1971] \$956 User Manual, TRSI-10, Computing Center, University of Texas at Austin, 1981.

- Fagur, R. [1977] "Multivalued Dependencies and a New Normal Form for Relational Databases," TOPS, 2-3, September 1977.
- Engin, R. [1979] "Normal Forms and Relational Database Operators," in stastop [1979].
- Fagur, R. [1981] "A Normal Form for Relational Databases That Is Based on To analysis and Keys," TODS, 6:3, September 1981.
- Fagir, R., Niecorgelt, J., Pippenger, N., and Strong, A. [1979] "Evrendible Hisbing—A Fast Access Method for Dynamic Files," TODS, 4-4, September 1979.
- Falconet S., and Paton, N. [1997]. "Deductive Object-Oriented Database Systems: A Survey," Proceedings of the 3rd International Worksitap Rides in Database Systems (JRDS'97). Skowdet Sweden, June 1997.
- Falcursos, C. [1996] Searching Multimedia Databases by Content, Klower, 1996
- Falcorsos, G., and Jagadish, H. [1992] "On B-Tree Induces for Skewed Distributions," in VLDB [1992].
- Faloutsus, G., Barber, R., Fackner, M., Elsiner, J., Niklack, W., Ferkovie, D., and Equite, W. [1964] Efficient and effective querying by image content. Int Journal of Intelligent Information Systems, 5:4, 1994.
- Earap, W., and Teorey, J. [1993] 'FunBase: A Function-Essed Information Management System," (2004, November 1993.
- Earahmand, E., Navathe, S. B., and Enskow, P. H. (2002) "Electronic Commerce and Security—Management Perspective," INFORMS 7th Actual Conference on Informatons Systems and Technology, CIST 2002, November 2002 (http://www.sba.ucommedu/OPDM/CIST/).
- Fernander, E., Summers, R., and Wood, C. [1981] Database Security and Integrity, Addison Wosley, 1981.
- Ferrier, A., and Stangrou, C. [1982] "referencements in the Distributed Database Management System S186 (SAC) in V109 [1982].
- Fishman, D., et al. [1986] "BUS, An Object-Oriented 13MS," TOOIS, 4(2, April 1986)
- Folk, M. J., Zoellick, B., and Raccardi, G. [1998] File Structures: An Object Oriented Approach with C++, 3rd ed., Addison-Wesley, 1998.
- Ford, D., Blakelen, J., and Ponnoz, J. (1994) "Open oceas: A Modalar Object-Oriented (1998)," in Science (1993).
- Feed, D. and Christodoulakis, S. [1991] "Optimizing Rendom Retrievals from CLV Format Optical Disks," in VIOB [1991].
- Foreman, G., and Zahorjan, J. [1994] "The Challenges of Mobile Computing" (EEE Camputer, April 1994)
- Fischer, M., and Scott, K. [1997] UMI, distilled, Addison-Wesley, 1997.
- Franasæk, P., Robinson, J., and Thomasian, A. [1992] Concurrency Control for High Contention Environments," (Ops, 17:2, Jung 1992.
- Franklin, F. et al. [1992] 'Crash Recovery in Chent Server EXOD(28," in SIGMED (1992).

- Fratemali, F. [1995] Tools and Approaches for Data Intensive Web Applications: A Survey, ACM Computing Surveys, Mid. September 1999.
- Frenkel, K. [1991] "The Human Genome Project and Informatics." CACM, November 1991.
- Friesen, O., Gaurhier-Villars, G., Lefelorne, A., and Meille, L., "Applications of Deductive Object-Oriented Databases Using DEL," in Ramakrishnan (1995).
- Furndo, A. [1978] "Formal Aspects of the Relational Model," Information Systems, 32, 1978.
- Gadar, S. [1988] "A Homogeneous Relational Model and Query Language for Temporal Databases," TODS, 13:4, December 1988
- Gan, J. [1988] "The Optical File Cabinet: A Random-Accuss File System for Write-Once-Optical Disks," IEEE Computer, 21:6, June 1988.
- Oallone, H., and Murker, J., eds. [1978] Logic and Databases, Plenom Press, 1978.
- Gallare, H., Minker, J., and Nucolas, J. [1984] "Logic and Databases: A Evolucitive Approach," ACM Computing Surveys, 16:2, June 1984.
- Callaire, H., Minker, J., and Nicolas, J., eds. [1981]. Advances in Database Theory, vol. 1, Plenum Press, 1981.
- Conal-Eldin, M., Thomas, G., and Elmasri, R. [1988] "Integrating Relational Databases with Support for Updates," Proceedings of the International Symposium on Databases in Parallel and Distributed Systems, (EEE CS, December 1988).
- Gaue, C., and Sarson, T. [1977] Structured Systems Analysis: Tools and Techniques, Improved Systems Technologies, 1977.
- Gangopadhyay, A., and Adam, N. (1997). Database Issues in Geographic Information Systems, Klower Academic Publishers, 1997.
- Garcia-Molina, H. [1982] "Elections in Distributed Computing Systems," IEEE Transactions on Computers, 31(1) January 1982.
- Garcia-Molina, H. [1985] "Using Seminatic Knowledge for Transaction Processing in a Distributed Database," (ODS, 8.2, June 1983.
- Cohoni, N., Jagdish, H., and Shumeli, O. [1992] "Composite Event Specification in Active Databases Model and Intelementation," in VDD8 [1992].
- Georgakopoulos, D., Rusinkiewicz, M., and Sheth, A. (1991) "On Senalitability of Multidatabase Transactions Through Forced Local Conflicts," in: RDE [1991].
- Gerritsen, R. [1975] "A Preliminary System for the Design of D816 Data Structures," CACM, 18 (C, October 1975)
- Ghrsh, S. [1984] "An Application of Statistical Databases in Manufacturing Testing," in ICOC(1984).
- Ohish, S. [1986] "Statistical Data Reduction for Manufacturing Testing," in (COC [1986].
- Gifford, D. [1979] "Weighted Voting for Replicated Data," Proceedings of the Seventh ACM Symposium on Operating Systems Principles, 1979.
- Gladney, H. [1989] "Data Replicas in Distributed Information Services," (ODS, 144, March 1989).

- Oogolla, M., and Hulenstein, U. [1991] "Towards a Semannic View of an Extended Energy-Relationship Model," TODS, 15:3. September 1991.
- Coldberg, A., and Robson, D. [1983] Smalltalk-80: The Language and Its Implementation, Addison-Wesley, 1963.
- Coldine, A., and Konig, P [1988] A Technical Overrien of the Information Review: Diens nerv System (1803), 2nd ed., 868 (6-38) 3700, National Bore in of Standards.
- Cotheb, L. [1975] 'Computing Juns of Relations," in SIGKAR [1975].
- Graefe, G. [1993] "Query Bealuation Techniques for Longe Databases," ACM Computing Surveys, 25.2, Jone 1993.
- Graefe, G., and DeWut, D. [1987] "The EXODUS Optimizer Generator," in SICMOP [1987].
- Gravano, L., and Garcas-Molina, H. [1997] "Merging Ranks from Hereogeneous Sources," in VL08[1997].
- Grav. J. [1978] "Note- on Data Base Operating Systems," in Payer, Graham, and Seegmuller [1978].
- Gray J. [1981] "The Transaction Concept: Virtues and Limitations," in virtue [1981]
- Oray, J., Lorie, R., and Patroln, G. [1975] "Oranidarity of Locks and Degrees of Consistency in a Shared Data Base," in Nijssen (1975).
- Gray, J., MirJones, P., and Bhoger, M. [1981] "The Kellovery Manager of the System R Database Manager," ACM Computing Surveys, 13:2, June 1981.
- Oray J., and Reuter, A. (1993) Transaction Processing: Concepts and Techniques, Morgan Kautmann, 1993.
- Orifubs, P. and Wade, B. [1976] "An Anthonianon Mechanism for a February Data base System," (ODS, 1:3, September 1976.
- Opechowski, E., and Payr, R. F. H 9961 "Future Trands in Hard Disk Drives," IEEE Transactions on Magnetics, 32(5, May 1996).
- Cipasky, W. [1994] "Mutimedia Information Systems," in IEFF Multimedia, 141 Sering 1994.
- Onsky, W. [1997] "Managing Multimedia Information in Database Systems," in CACM, 40.12 December 1987.
- Grosky, W. Jain, R., and Mehrorra, R., eds. [1997]. The Handbook of Multimedia Intormation Management, Prennee-Fiall PTR, 1997.
- Guttman, A. [1954] "R-frees: A Dynamic Index Structure for Spatial Searching," in Sto-MOD [1984]
- Costayer, M. [1996] Oracle Designer/2000 Web Server Generator Technical Overview Oversion 1 3(2), Technical Report, Oracle Corporation, September 1990.
- Halstal, F. [1996] Data Communications, Computer Networks and Open Systems, 4th ed., Addison-Wesley, 1996.
- Haas, P., Naughten, J., Seshadri, S. and Stokes, L. (1995) Sampling-based Estimation of the Number of Distinct Values of an Attribute," in V116 (1995).

- Haas, P., and Swann, A.[1995] "Sampling-based Selectivity Estimation for Joins Using Augmented Frequence Value Statistics," in ICPE [1995].
- Hachent, N. and Borra, P. [1992] "New Order Preserving Access Methods for Very Large Fales Derived from Linear Hashing," TKOE, 4.1, February 1992.
- Had; Jacos, V. [1983] "An Operational Model for Database System Reliability," in Proceedings of Micro Testion 50 Conference, March 1983.
- Hadrilacus, V. [1986] "A Theory of Relaibility in Database Systems," 1986.
- Haerder, T., and Rothermel, K. [1987] "Concepts for Transaction Recuvery in Nested Transactions," in SIGRERS [1987].
- Haerder, T., and Reuter, A. [1983] "Principles of Transaction Oriented Database Recovery –A. Taxonomy," ACM Computing Surveys, 15.4, September 1963, pp. 257–318.
- Hall, P. [1976] "Optimization of a Single Relational Expression in a Relational Data Base System," IBM Journal of Research and Development, 20-3, May 1976.
- Homilton, G., Catteli, R., and Fisher, M. [1997] JOBC Database Access with Java-A. Tutorial and Annotated Reference, Addison Wesley, 1997
- Hammer, M., and McLeod, D. [1975] "Semantic Integrity in a Relational Data Base System," in VIL:8 [1975].
- Hammer, M., and McLeod, D. [1981] "Database Description with SDM: A Somantic Data Model," TOD5, 6-3, September 1981.
- Hammer, M., and Sann, S. [1978] "Efficient Monatering of Database Assertions," in SIGMOD [1978].
- J. Harr and M. Kamber, Data Muong: Concepts and Technapacs, Mongan Kaufmann, San-Francisco, 2001.
- J. Han, J. Per and Y. Ym. "Mining Frequent Patterns without Candidate Generation," Proc. ACM, SIGMOD Conference, 2000.
- Hanson, E. [1992] "Rule Condition Testing and Action Execution in Actel," in SERGO-[1992].
- Hardgrave, W. [1984] "6.0 Y: A Retrieval Language for Tree-Structured Database Systems," in YOC [1984].
- Hardgrave, W. [1980] "Arphysicity in Proceeding Bodean Queries on 11008 Tree Structures: A 5rudy of Four Different Philosophers," (SE, 6.4, July 1980.
- Harrington, J. [1987] Relational Database Management for Microcomputer: Design and Implementation. Holt, Binehart, and Winstow, 1987.
- Harris, L. [1978] "The ROBO" System: Natural Language Processing Applied to Data Base Query," Proceedings of the ACM National Conference, December 1978.
- Hoskin, R., and Lorie, R. [1982] "On Extending the Euronoms of a Relational Database System," in SIGMOD [1982]
- Hasse, C., and Weikum, G. [1991] "A Performance Evaluation of Multi-Level Transaction Matagement," in VLDB [1991].

- Hayes-Roth, F. Waterman, D., and Lenat, D., eds. (1983) Building Expert Systems. Addison-Wesley, 1983.
- Hayne, S., and Ram, S. [1990] "Multi-User View Integration System: An Expert System for View Integration," in ICLE (1990).
- Heder, S., and Zdonack, S. [1990] "Object Views Extending the Vision," in 1019 [1990]
- Heiler, S., Haidhvalal, S., Zdonik, S., Blanstein, B., and Rosenthali, A. [1992] "A Flexible Framework for Transaction Management in Engineering Environment," in Elizagamid [1992].
- Helal, A., Hu, T., Ehnami, R., and Mukherjee, S. [1993] 'Adaptive Transaction Scheduling," CCKM. November 1993.
- Held, O., and Stouchraker, M. [1978] "B-Trees Reexamined," CACM, 21(2, February 1978.
- Henschers, L., and Naqua S. [1954]. "On Compiling Querres in Recursive First-Onler Databases." JACM, 31:1, January 1984.
- Hernandez, H., and Chan, E. [1991] "Constraint-Time-Mulintainable BCNS Database Schemes," TODS, 16:4, December 1993
- Heror, C. [1980] "Spatial Management of Data " (2008, 54, Diversiter 1950)
- Hevner, A., and Yao, S. (1979) "Query Processing in Distributed Database Systems," T86, 5-3, May 1979.
- Hoffer, J. (1982) "An Empirical Investigation with Individual Differences in Database Models," Proceedings of the Third International Information Systems Conference, December 1982.
- Holland, I. (1975) Adaptation in Natural and Artificial Systems, University of Michigan Press, 1975.
- Holsapple, C., and Whinston, A., eds. [1987] Decisions Support Systems Theory and Application, Springer-Verlag, 1987.
- Holtanan J. M., and Goodman D. J., eds. [1943] Wireless Communications: Future Directions, Klover, 1993
- Hstael, D., and Kames, M. (1989) "Heterogeneous Databases: Prohibitation: Issues, and Solutions," TKDE, 1:1, March 1989.
- Hsu, A., and Inuclinsky, T. [1935] "Integrity Checking for Multiple Updates," in SCROP [1985].
- Hull, R., and King, R. [1987] "Semantic Database Modeling: Survey, Applications, and Research Issues," ACM Computing Surveys, 193, September 1987.
- 1994 [1978] QBE Terrumal Users Guide, Form Number 5H20-2078 C.
- 1081 (1992) Systems Application Architecture Common Programming Interface Database Lovel 2 Reference, Document Number Sci26 4795-01.
- (L1) F [1984] Proceedings of the IEEE CN International Confinence on Data Engineering, Shiney, Ruled, Los Angeles, CA, April 1984.
- ICDE [1986] Proceedings of the IEPE CS International Conference on Data Engineering, Wiedethold, G., ed., Los Angeles, February 1986.

- ROP [1967] Proceedings of the traff of International Conference on Data Engineering, Wah, B., ed., Les Angeles, February 1987.
- 0.00 [1988] Proceedings of the 1996 Conference on Data Engineering, Carlis, J., ed., Low Angeley, February 1985.
- 6:DE [1989] Proceedings of the tEEF CS International Conference on Data Engineering, Shuey, R., ed., Los Aragoles, February 1989.
- ICDE [1990] Proceedings of the IEEE CS [International Conference on Data Engineering, [ international Conference on Data Engineering, ] international conference on Data Engineering, [ international Conference on Data Engineering, ] international Conference on Data En
- [CDF [1991] Proceedings of the 1855 CS International Conference on Data Engineering, Cercone, N., and Tsuchiyo, M., eds., Kobe, Japan, April 1991.
- (CIW [1992] Proceedings of the IEEE CS International Conference on Data Engineering, Collbuot, F. ed. Physical, AZ, February 1992.
- ICDE [1993] Proceedings of the IEEE CS International Conference on Dota Engineering, Elmagammal, A., and Neuhold, E., eds., Vienna, Austria, April 1993.
- ICDF [1994] Proceedings of the (EEU/CS International Conference on Data Engineering
- (CDE [1995] Proceedings of the JEEE CN international Conference on Data Engineering, Yu. P. S., and Chen, A. L. A., eds., Taiper, Tarwan, 1995.
- (CDE [1996] Proceedings of the IEEE US International Conference on Data Engineering, Su, S., Y. W., ed., New Orleans, 1996.
- (LDE [1997] Proceedings of the IEEE US International Conference on Data Engineering, Gray, A., and Larson, P. A., eds., Bermingham, England, 1997.
- 30:DE [1998] Proceedings of the IELE CS International Conference on Data Engineering, Orlando, FL, 1998.
- WD6 [1900] Proceedings of the JEFF US International Conference on Data Engineering, Sydney, Australia, 1999.
- 635 [1983] International Graphers Exchange Specification Version 2, Natural Barran of Standards, U.S. Department of Commerce, January 1983.
- Junelinski, T., and S.abrinath, B. [1994] "Mobile Writeless Comparing: Challenges in Data Management," CACM, 37:10, October 1994.
- Janelinska, T., and Lipski, W. [1981] "On Representing Incomplete Information jula Relational Database," in VLDP [1981].
- Informes (1998) "Web Integration Option for Informix Dynamic Server," available at http://www.informix.com.
- Inmon, W 11 (1992) Building the Data Warehouse, Wiley, 1992.
- Ioannidis, Y., and Kong, Y. [1990] "Randomized Algorithms for Optimizing Large Join Queries," in SIOMOD [1990].
- Ioannidis, Y., and Kung, Y. [1991] "Left-Deep vs. Bushy Trees: An Analysis of Strategy Spaces and us Implications for Query Optimization," in SIGMOD [1991]
- Iconmidis, Y., and Wong, E. [1988] "Transforming Non-Linear Recursion to Enear Recursion," in EDS (1988).

- Josophidis, J. [1979] "A Translator to Convert the DDL of EBM to the DDL of System 2000," in E8 Conference [1979].
- Itani, K., Purkavastha, S., and Teorey, J. [1979] "A Designer for D697-Processable Logical Database Structures," in VLD8 [1979].
- Jacobson, J., Christerson, M., Jonsson, P., Overgaard, G. (1992) Object Oriented Software Engineering: A Use Case Driven Approach. Addison-Wesley, 1992.
- Jagadish, H. [1989] "Incorporating Hierarchy in a Relational Model of Data?" in \$65800 [1989]
- Jagadish, Ju [1997] "Content-based Indexing and Retrieval," in Civisly et al. [1997]
- Jajiaha, S., and Kogan, B. [1992] "Integrating as Object-oriented Data Model with Multilevel Security," IEEE Symposium on Security and Principy, May 1980, pp. 76–85.
- Jajodva, S., and Mutchlet, D. [1990] "Dynamic Voting Algorithms for Maintaining the Consistency of a Replicated Database," TODS, 15:2, June, 1990
- Jajudia, S., Ng, P., and Springsteel, E [1983] "The Froblem of Equivalence for Entity-Relationship Diagrams," TSE, 9:5, September, 1983.
- Japadra, S., and Sandha, R. [1991] "Toward a Multilevel Score Relational Data Maxlel," in stometry [1001].
- Jankne, D., ed. [1977] The ANSI/SPARC DBMS Model, North-Holland, 1977.
- Jarke, M., and Koch, J. [1984] "Query Optimization in Database Systems," ACM Computing Surveys, 16:2, June 1984.
- Jepsen, C., and Snodgrass, R. [1992] "Temporal Specialization," in (209 [1992].
- Jensen, C., et al. [1994, "A Glowary of Temporal Database Concepts," ACM SIGMOD Record, 23(1, March 1994.
- Johnson, T., and Shasha, D. [1993] 'The Performance of Current & Tree Algorithms," (2008, 13-1, March 1993).
- Joshi, J. B. D., Arel, W. G., Chatser, A., and Spatrord, E. H. (2001) "Security Models for Web-Based Applications," Communications of the ACM, Job Lary 2001, pp. 38–44.
- Kaefer, W., and Schoenong, H. [1992] "Realizing a Temporal Complex-Object Data Model," in stoward [1992]
- Kamel, L. and Faloatsos, C. [1993] "On Packing R-trees," CIEM, November 1995.
- Kamel, N., and King, R. [1985] "A Model of Data Distribution Posed on Texture Analysis," in SIGKOD [1985].
- Kupp, D., and Leron, J. (1978) IMS Programming Techniques. Van Nostrand-Reinhold, 1978.
- Kappel, G., and Schreft, M. [1991] "Object/Hehavior Diagrams," in (116/1991).
- Karlapaleni, K., Navathe, S. B., and Amman M. [1996] "Optimal Redesign Policies to Support Dynamic Processing of Applications on a Distributed Relational Database System," Information Systems, 21:4, (1996), pp. 353–67.
- Kate, R. [1985] Information Management for Engineering Design: Surveys in Computer Science, Springer-Verlag, 1985.
- Kate, R., and Wong, E. [1982] "Decompiling CONVEY: DW" into Relational Quezies," TODS, 7.1, March 1982.

- KDD [1996] Proceedings of the Second International Conference on Knowledge Discovery as Databases and Data Mining. PortLand, Oregon, August 1996.
- Kedem, Z., and Suberschatz, A. [1950] "Non-Two Plasse Locking Protocols with Shared and Exclusive Locks," in VIPP [1950].
- Keller, A. [1952] "Opdates to Relational Database Through Views Involving Joins," in Scheuermann [1982].
- Kemp, K. [1993] "Spatial Databases: Sources and Issnes," in Environmental Modeling with GIS, Oxford University Press, New York, 1993.
- Kemper, A., Luckemann, F., and Wallrath, M. [1987] "An Object-Oriented Database System for Engineering Applications," in S158005 [1987].
- Kemper, A., Moerkutte, G., and Steinbrunn, M. [1992] "Optimizing Boolean Expressions in Object Bases," in VLDB [1992].
- Kemper, A., and Walliath, M. [1987] "An Acalysis of Geometric Modeling in Database Systems," ACM Computing Surveys, 19:1, March 1987.
- Kern, W. [1978] Data and Reality, North-Holland, 1978.
- Kent, W. [1979] "Limitations of Record Based Information Models," (1978, 4)t, March 1979.
- Kenr, W. [1991] "Object-Oriented Database Programming Languages," in VLDB (1994].
- Kerschberg, L., Ting, P., and Yao, S. [1982] "Query Optimization in Star Computer Networks," PODS, 7 4, December 1982.
- Kerabchi, M. A., Mathur, S., Risch, T., and Chen, J. [1990] "Comparative Analysis of RDBMS and COPENIS: A Case Study," IEEE International Conference on Manufacturing, 1990.
- Khoshailan, S. and Baker A., [1996] Multimedia and Imaging Databases, Morgan Kaufmann, 1996
- Khoshañary, S., Chan, A., Wong, A., and Wong, H. K. T. (1992) Developing Client. Server Applications, Morgan Koutmann, 1992.
- Kifer, M., and Lorinskir, E. [1956] "A Framework for an Efficient Implementation of Deductive Databases," Proceedings of the Sixth Advanced Database Symptoxican. Tokyo, August 1986.
- Kom, P. [1996] "A Taxonomy on the Architecture of Database Guroways for the Web," Working Paper TR-96-U-10, Changman National University, Taejon, Karoa favailable from http://grigg.changman.sc.kt/projects/UniWeb).
- Kim, W. (1982) "On Optimizing an sQL-like Nested Query," TODS, 3(3, September 1982).
- Kim, W. [1989] "A Model at Queries for Object-Oriented Databases," qr v119 [1989].
- Kim, W. [1992] "Object-Oriented Databases: Definition and Research Directions," TRUE, 2:3, September 1990.
- Kjin W. [1995] Modern Database Systems: The Object Model, Interoperability, and Beyond, ACM Press, Addison Wesley, 1995.

- Kim, W., Reiner, D., and Barory, D., eds. [1985] Query Processing in Database Systems. Springer-Verlag, 1985.
- Kum, W. et al. [1987] "Features of the CRACK Object Directed Database System," Microelectronics and Computer Technology Corporation, Technical Report ACA-87-308-57, September 1987.
- Konhali, R. [1996] The Data Warehouse Toolkii, Wiley, Inc. 1996.
- King, J. [1981] "Quils 1: A System for Semantic Query Optimization in Relational Databases," in VLVP [1981].
- Kitouregowa, M., Nakayamo, M., and Takogi, M. [1989] "The Effect of Bucket Size Toming in the Dynamic Hybrid GRACE Flash [on: Method." in MEDB [1989].

Slimble, J., and Koffenian, K., eds. [1974] Data Base Management, North-Holland, 1974.

- Klug, A. [1982] "Equivalence of Relational Algebra and Relational Calculus Query Languages Having Aggregate Functions," JACM, 29-4, July 1982.
- Knoth, D. [1973] The Art of Computer Programming, Vol. 3: Sorting and Searching, Addson-Wesley, 1973.
- Kegelnik, A. [1998] "Biological Information Management with Application to Human Genome Data," Ph.D. dissertation, Georgia Institute of Technology and Emory University, 1998.
- Kogelnik, A., Lott, M., Brown, M., Navathe, S., Wallace, D. [1995] "MITOMAP: A human mitochondrial genome database: -1995 update," Nucleic Acids Research, 26:1, January 1998
- Kegelnik, A., Navache, S., Wallace, D. [1997] "GENOME: A system for managing Human Genome Project Data," Proceedings of Genome Informatics '97, Eighth Workshop on Genome Informatics, Tokyo, Japan. Sponsor: Human Genome Center, University of Tokyo, December 1997.
- Kohler, W. (1981) "A Survey of Techniques for Synchronization and Recovery in Decentradiced Computer Systems," ACM Computing Surveys, 13:2, June 1981.
- Kunsynski, B., Bracker, L., and Bracker, W. [1982] "A Model for Specification of Office Communications, 30:1, January 1982.
- Kudhage, R. [1991] "To Sec. or Not to See: Is that the Query?" in Proceedings of the Acta MOR International Conference, June 1991.
- Kouth, H. [1983] "Locking Primitaves in a Database System," JACM, 30(1, January 1983.
- Korth, H., Levy, E., and Silberschatz, A. [1990] "A Formal Approach to Recovery by Compensating Transactions," (n VLOB [1990].
- Kurz, A., Dutrich, K., Mulle, J. (1986) "Supporting Semantic Rules by a Generalized Event/Trigger Mechanism," in VLIB (1988)
- Krishnatnuriliv, R., Lirwin, W., and Kenr, W [1991] "Language Features for Interoperability of Databases with Semantic Discrepancies," in SIGMOD [1991]
- Krishnamurthy R., and Naqvi, S., [1988] "Database Updates in Logic Programming, Rev. 1." MCC Technical Report #ACA-ST/010-88, Rev. 1, September 1988

- Krishnonnarthy, R., and Naqya, S. [1989] "Non-Determinantic Choice in Datalog." Proceedings of the 3rd International Conference on Data and Knowledge Bases. Jerusalem, June 1989.
- Krovett, R., and Crett B. [1992] "Lexical Analogoity and Information Retrieval" in TOIS, 10, April 1992.
- Kulkarm K., Carey, M., DeMichiel, L., Mattos, N., Hong, W., and Ubell M., "Introducing Reference Types and Clearing Up SQLW Object Model," ISO WUB Report X0H2-95 (456), November 1995.
- Kumar, A. [1991] "Performance Measurement of Some Main Memory Recovery Algorithms," in ICOE [1991].
- Kumar, A., and Segev, A. [1993] "Over and Availability Tradeoffs in Replicated Concurrency Constol," TODS, 18-1, March 1993.
- Kuntat, A., and Stonebraker, M. [1987] "Semantics Based Transaction Management Techniques for Replicated Data," in SIGNOD [1987]
- Kuntor, V., and Han, M., eds. [1992] Recovery Mechanisms in Database Systems, Prentice-Hall, 1992.
- Kumat, V., and Hsu, M. [1995] Recovery Mechanisms in Database Systems, Frencisco-Hall (PTR), 1998.
- Kumar, M., and Song, H. S. [1998] Database Recovery, Klower Academic, 1998.
- Kung, H., and Robasson, J. [1961] Optimistic Concurrency Control." TODS, 6:2, June 1981.
- Lacroix, M., and Pirntte, A. [1977] "Domain-Oriented Relational Languages," in VIET [1977].
- Lacroix, M., and Piratre, A. [1977a] "IU: An English Structured Query Language for Relational Para Bases," in Nijsen [1977].
- Lauport, L. [1978] "Time, Clocks, and the Ordering of Events in a Distributed System," CAUM, 21-7, July 1978
- Langerak, R. (1990) "View Updates in Relational Databases with an Independent Scheme," TODS, 154, March 1990.
- Lanka, S., and Mays. P. [1991] "Fully Persistem B1-Trees,7 in SCMOD [1991].
- Larson, J. [1983] "Bridging the Cisp Detween Network and Relational Database Management Systems," IEEE Computer, 16:9. September 1983.
- Laisni, J., Navathe, S., and Elmasni R. [1939] "Attribute Equivalence and its Use in Schema Integration," 18E, 15:2, April 1989.
- Larson, P. (1978) "Dynamic Hashing," BIT, 18, 1978.
- Larson, P. [1981] "Analysis of Index Sequential Files with Overflow Chairing," TODS, 6:4, Dynamber 1981.
- Laurini, R., and Thompson, D. [1062] Fundamentals of Spatial Information Systems, Academic Press, 1992.
- Lefman, E. and Yao, S. (1981) "Efficient Locking for Concurrent Operations on B Trees," TODS, 64, December 1981.

- Lee, J., Ehnassi, R., and Won, J. (1995) " An Integrated Temporal Data Midel Incorponating Time Series Concepts," Data and Knowledge Engineering, 24, 1998, pp. 257–276.
- Lehman, T., and Lundsay, B. [1989] "The Starburst Long Field Manager," in VIUS [1989].
- Leiss, E. [2982] "Randomicing: A Priorical Method for Promoting Statistical Excelases Against Compromae," in VIDB [1982].
- Leiss, E. [1982a] Principles of Data Security, Plenam Press, 1982
- Lenzermi, M., and Santucci, C. [1983] "Cardinality Constraints in the Entity Relationship Model," in ER Conference [1985].
- Leung, C., Hibler, B., and Mwara, N. [1992] "Picture Removal by Context Description," in Journal of Information Science, 1992, pp. 111–19.
- Levesque, H. [1984] "The Legic of Incomplete Knowledge Bases," in Brudie et al., cb. 7 [1984].
- Li, W., Selek Candan, K., Hirata, K., and Hara, Y. [1996] Therarchical Image Mudeling for Object-based Media Retrieval in 181, 27:2, September 1998, pp. 139–76.
- Lien, E., and Weinberger, F. [1978] "Consistency, Concurrency, and Crash Recovery," in SIGMOD (1978).
- Lieuwen, L., and DeWitr, D. [1992] "A Transformation Based Approach to Optimizing Loops in Database Programming Languages," in SIGMOD [1992].
- Lilion, L., and Bhargava, B. [1985] "Database Integrity Block Construct. Concepts and Design Issues," TSE, 11:9, September 1985.
- Lin, J., and Dunham, M. H. [1998] "Mining Association Rules" in 1009 [1998].
- Lindsay, B., et al. [1984] "Computation and Communication in R\*: A Distributed Database Manager," (DOCS, 24), January 1984.
- Lippman R. [1987] "An Introduction to Computing with Neural Nets." IEEE ASSP Magazine, April 1987.
- Lipski, W. [1979] "On Semantic Issues Connected with Incomplete Information," TODS, 4:3, September 1979.
- Lipton, R., Naughton, J., and Schneider, D. [1990] "Practical Selectivity Estimation through Adaptive Sampling," in SIGMON [1990].
- Liskov, B., and Zilles, S. [1975] "Specification Techniques for Data Abstractions," TSE, 1:1, March 1975.
- Litwin, W. [1980] "Linear Hashing: A New Tool for File and Table Addressing," in v170 (1980).
- Lau, K., and Sundernaman, R. [1988] "On Representing Indennite and Maybe Information in Relational Databases," in ICDE [1986].
- Liu, L. and Morrsman, R. [1992] "Acrivity Model: A Declarative Approach for Carpaining Communication Boltawior in Object-Oriented Databases," in V406 [1992].
- Livadas, P. [1989] File Structures: Theory and Practice, Prentice/Hall, 1989.

- Lockemann, P. and Knutsen, W. [1968] "Recovery of Disk Contents After System Isilure," CACM, 11-8, August 1968.
- Lorie, R. (1977) "Physical Integrity in a Large Segmented Database," TODS, 2.1. Match 1977.
- Lorie, R., and Plouffe, W [1983] "Complex Objects and Their Use in Design Transacrons," in SCMOD [1983].
- LOARAKE, E. (1986) "A Problem-Oriented Interential Database System," (2005, 11)3. September 1986.
- Lu, Fi., Mikkihnem, K., and Richardsen, J. [1967] "Design and Evaluation of Algorithms to Compute the Transitive Closure of a Database Relation," in JCDI [1987]
- Luhars, M., Potts, C., and Richter, C. (1993) " A Review of the State of Practice in Requirements Modeling," IEEE International Symposium on Requirements Engineering, San Diego, CA, 1993.
- Lucyk, B. [1993] Advanced Topics in DB2, Addison Wesley, 1993.
- Magure, D., Oondchild, M. and Righd D., eds. [1997] Geographical Information Systems: Principles and Applications. vols. 1 and 2. Longing: Scientific and Technical, New York.
- Mahajan, S., Donahoo, M. J., Navorhe, S. B., Ammar, M., Malik, S. (1998) "Crouping Techinques for Update Propagation in Intermittently Connected Databases," in ICCP [1998].
- Majer, D. [1983] The Theory of Relational Databases, Computer Science Press, 1983.
- Maier, D., Stein, J., Otis, A., and Zardy, A. [1986] "Development of an Object-Oriented DBMS," Oxi752A, 1986.
- Malley, C. and Zdonick, S. [1986] "A Knowledge-Based Approach to Query Optimizatain," in EDS [1986].
- Mater, D. and Warren, D. S. [1985] Computing with Logic, Penjamin Commungs, 1986.
- Mannila, H., Toeronen, H., and Verkaines A. (1994) "Efficient Algorithms for Discovering Association Rules," in KDD-94, AAAI Workshop on Knoteledge Discovery in Databases, Scattle, 1994.
- Manola, F. [1998] "Towards a Richer Web Object Model," in SIGMOD Record, 27.1, March 1998.
- March, S., and Severance, D. [1977] "The Determination of Efficient Record Segmentamons and Blocking Encross for Shared Files," TODS, 2:5, September 1977.
- Mork, U., Resissopoulos, N., Newsome, T., and Lindopaparrana, P. [1992] "Incrementally Maintained Network to Relational Mappings," Software Practice & Experience, 22:12, December 1992.
- Markowitz, V. and Raz, Y. [1983] "ERROL: An Entry-Relationship. Role Oriented, Query Language," in ER Conference [1953]
- Martin, J., Chapman, K., and Leben, J. [1989] DB2-Concepts, Design, and Programming, Prentice-Hall 1989.
- Martin, J., and Odell, J. [1992] Object Oriented Analysis and Design. Prentice Holl. 1992.

- Maryanski, F. [1980] "Backend Database Machines," ACM Computing Surveys, 12:1, March 1980.
- Masmaga, Y. [1987] "Multimedia Databases: A Formal Tranework," Proceedings of the IEEE Office Automation Symposium, April 1987.
- Martison, R., Data Warehousing: Strategies, Technologies, and Techniques. McGraw-Hull. 1996.
- McFadden, F and Hoffer, J. [1985] Database Management, 2nd ed.: Benjamin/Commings, 1988.
- McFadden, F. R., and Hotter, J. A. [1994] Modern Database Management, 4th ed., Resjacon Commings, 1994.
- McGee, W. [1977] "The Information Management System (808) S. Part I: General Structure and Operation," IBM Systems Journal, 16:2, June 1977.
- McLeish, M. [1989] "Further Resolts on the Security of Parrithmed Dynamic Statistical Databases," (2005, 14)1, Match 1989.
- McLeosf, D., and Heimbigner, D. [1985] "A believant's Architecture for Information Systerns," 10015, 3:3, July 1985.
- McBrotra, S., et al. [1992] 'The Concurrency Control Problem in Multi-Jatabases Characteristics and Bolutions," in SIGMOD [1992].
- Melton, J., Bauer, J., and Kulkarra, K. [1991] "Object A1/05 (with improvements for value ADTs)," 190 (WG3) Report X3712-97-083, April 1991.
- Melton, J., and Matros, N. (1996) An Orenview of SQL3 The Lorenging New Generation of the SQL Standard, Tatorial No. T5, VLDB, Rombay, September 1996.
- Melton, J., and Simon, A. R. [1993] Understanding the New SQL: A Complete Guides Mergan Kanhoatm.
- Menaste, D., Popek, G., and Mutte, R. (1980) "A Locking Protocol to: Resource Coerdonation in Distributed Databases," TODS, 5(2) June 1980.
- Mendelton, A., and Mater, D. [1979] "Generalized Mutual Dependencies and the Decomperitors of Database Relations," in ALCO [1979].
- Mendelson, A., Milaila, G., Milo, T. (1997) "Querying the World Wide Web." Journal of Digital Libraries, 1:1, April 1997
- Metais, E., Kedad, Z., Comyn-Watuau, C., Bouregheab, M., "Using Linguistic Knowlindge in View Integration: Toward a Third Constrainty of Tools," in UKE 23:1, June 1977.
- Mikkilmeni, K., and Su, S. [1988] "An Evaluation of Belational Join Algorithms in a Popelined Query Processing Environment," 199, 1460, June 1988.
- Miller 'N [1987] File Structures Using PASCAL, Beniamin Commungs, 1987.
- Minours, T., and Wiederhold, G. [1981] "Resilient Extended True-Copy Token Scheme for a Discriminal Discrimination," TSE, 3-3, May 1981.
- Missikoff, M., and Wiederhold, G. [1964] "Toward a Unified Approach for Experience Database Systems," in SDS [1964].
- Metchell, Machine Learning, McGraw Hill, New York, 1997.

- Mirschung, B. [1989] "listending the Relational Algebra to Capture Complex Objects," in VLDB [1989].
- Mohon, C. [1993] "IBM's Relational Disabase Produces Features and Technologies," in SIGMOD [1993].
- Mohan, C., Haderle, D., Luidsay, B., Piraheski, H. and Schwart, P. [1992] "ARIES: A Transaction Recovery Method Supporting Fine-Granularity Locking and Parrial Rollbacks using Write-Ahead Logging," TODs, 17-1, Match 1992.
- Mohan, C., and Levine, F. (1992] "AREL9M: An Efficient and Eligh-Concurrency Index Management Method Using Write-Ahead Logging," in 500400 [1992]
- Mohan, C. and Narang, J. [1992] "Algorithms for Creating Indexes for Very Large Tables without Quassing Updates," in SIGMOD [1992].
- Mohan, C. et al. (1992) "AKEL: A Transaction Recovery Method Supporting Fine-Granalarity Lucking and Fartial Rollbacks Using Write-Ahead Loggrag," TOD8, 17:1, March 1992.
- Morris, K., Ullman, J., and VonGelden, A. [1986] "Design Overview of the NAIL Sysrem." Proceedings of the Third International Conference on Logic Programming, Springer-Verlag, 1986.
- Morns, K., et al. [1987] "YAWN! (Yet Another Window or NAIL!), or ROF [1987].
- Morris R. [1968] "Statter Storage Techniques," CACM, 11-1, January 1968.
- Morsi, M., Navathe, S., and Kim, H. [1992] "An Extensible Object-Oriented Database Testhed," in JODE [1992].
- Moss, J. [1982] "Nested Transactions and Reliable Distributed Computing," Proceedings of the Symposium on Reliability in Distributed Software and Database Systems, IEEE CS, July 1982.
- Motro, A. [1987] "Superviews: Virtual Integration of Multiple Databases." TSE, 13:7, July 1987.
- Mukkamala, R. [1989] "Measuring the Effort of Data Dotribution and Replication Modelsier, Performance Evaluation of Distributed Systems," in ICDE [1989].
- Manuck, I., Finkelstein, S., Pitahosh, H., and Ramakrishnan, R. [1990] 'Magic Is Reicmant," in SIGSOD [1990].
- Munuck, J., Pirabesh, H., and Kamakrishnon, R. (1990) "The Magic of Duplatates and Aggregates," in V105 [1990].
- Muralikrishna, M. [1992] "Improved Unnesting Algorithms for Join and Aggregate SQL Querics," in VLDB [1992].
- Muralikrishna, M., and DeWirt, D. [1986] "Equildepth Histograms for Estimating Scient twity Factors for Multi-dimensional Queries," in SIGMOD [1988].
- Mylopolous, L. Berrotein, P., and Wong, H. [1990] "A Language Facility for Designitial Database-Intensive Applications," TODS, 5(2, June 1980)
- Narsh, I., and Thom, J. [1983] "The M. 570 (2003) Dedictive Database." Technical Report 83(10, Separtment of Computer Science, University of Melbourne, 1983.

- Navarbe, S. [1980] "An Intuitive View to Normalite Network-Structured Edits," in VLDB [0980].
- Navathe, S., and Ahmed, R. [1989] "A Temporal Relational Model and Query Langaoge," Information Sciences, 47.2, March 1989, pp. 147–75
- Novathe, S., Cero, S., Worderhold, G., and Dou, J. (1984) "Vertical Partitioning Algorithms for Database Design," TODS, 9:4, December 1984.
- Navathe, S., Elmasru R., and Larson, J. [1966] "Integrating User Views in Database Design," IEEE Computer, 19(1), January 1956.
- Navathe, S., and Gadgil, S. [1982] "A Methodology for View Integration in Logical Database Design." in VL23 [1982]
- Navathe, S. B. Karlapalem, K., and Ba, M.Y. [1996] "A Mixed Fragmentation Methodology for the Initial Distributed Database Design," Journal of Computers and Softwate Engineering, 3:4, 1996.
- Navathe, S., and Kerschberg, L. (1986) "Rule of Data Dictinitation in Database Design." Information and Management, 10:1, January 1986.
- Navathe, S., and Pillalamatri, M. [1988] "Toward Making the ER Approach Object-Onented," in ER Conference [1988].
- Navathe, S., Sashadhar, T., and Elmasri, R. [1984a] "Relationship Merging in Schema Integration," in ViDb [1984].
- Navathe, S., and Savasere. A. [1996] "A Practical Schema Integration Facility using an Object Oriented Approach." in Multiplatahave Systems (A. Elmagarmid and O. Bukhres, eds.), Prentice-Hall, 1996.
- Navathe, S. B., Savasere, A., Anwar, T. M., Beck, H., and Gala, S. (1994) "Object Modeling Using Classification in CANDOE and Its Application," in Dogae et al. (1994).
- Navathe, S., and Schkolnick, M. [1978] "View Representation in Logical Database Design," in SIGMAD [1978].
- Negri, M., Pelugatta, S., and Shatella, L. [1991] "Formal Semantics of SQL Queries," TODS, 16(3, September 1991.
- NC. P. [1981] "Further Analysis of the Entity-Relationship Agyrouch to Database Design." TSE, 7 1, Junuary 1981
- Nicolas, J. [1978] "Motual Dependencies and Some Results on Undecomposable Relations," in VLDB [1978]
- Nicolas, J. [1997] "Deductive Object-oriented Databases: Technology, Products, and Applications: Where Are We?" Proceedings of the Symposium on Digital Media Information Base (DMIB'97). Nata, Japan, November 1997.
- Nicolas, J., Phipps, G., Dett. M., and Ross, K. [1991] "Glue-NAIL': A Deductive Database System," in SIOMOD [1991].
- Nievergelt, J. [1974] "Binary Search Trees and File Organization," ACM Computing Surveys, 6:3, September 1974.
- Nievergelt, J., Hinterberger, H., and Seveik, K. [1984]. "The Grid Ede: An Adaptable Symmetric Multikey Ede Structure," TODS, 9(1, March 1984, pp. 38–7).

- Nussen, G., ed. [1976] Modelling in Data Base Management Systems, North-Flolland, 1976.
- Nussen, G., ed. (1977) Architecture and Models in Data Base Management Systems, North-Holland, 1977.
- NWOM, K., Berra, P., Thurstoingham, B., eds. [1996], Design and Implementation of Multimedia Database Management Systems, Klower Academic, 1996.
- Obermarck, R. [1982] "Distributed Deadlock Detection Algorithms," 1008, 7:2, June 1982.
- Dh. Y.C., [1999] "Secure Establish Modeling and Design." Ph.D. dissertation. College of Comparing Ceorgia Institute of Technology, Nucl. 1999.
- Obsiga, S. [1982] "Knowledge Based Systems as a New Interactive Computer System of the Next Generation," in Computer Science and Technologies, North-Holland, 1982.
- Ollo, [1] [1978] The CODASYL Approach to Data Base Management, Wiley, 1975.
- Olle, J., Sol. H., and Verrijn-Stuart, A., eds. [1982] Information System Design Methodology, North-Holland, 1982.
- Omecniski, E., and Scheuenmann, P. [1990] "A Parallel Algorithm for Record Clustering," TODS, 15:4, December 1990.
- Ornom, J. K. (1990) "Nocel Applications of Cryptography in Digital Communications," IEEE Communications 28.5, May 1990, pp. 21–29.
- O'Neill, P. [1994] Database: Principles, Programming, Performance, Morgan Kaufmann, 1994.
- Oracle (1992a) RUBMS Database Administrator's Guide, OR WIE, 1992.
- Oracle (1992 b] Performance Tuning Guide, Version 7.0, ORACLE, 1992.
- Oracle [1997a] Oracle 8 Server Concepts, vols [1] and 2. Release 3-3. Oracle Corporation, 1997.
- Otacle [19970] Oracle 8 Server Distributed Database Systems, Release 8.0, 1997.
- Oracle [1997c] FL/SQL User's Guide and Reference, Release 8.0, 1997.
- Oracle [1997d] Oracle 8 Server SQL Reference, Release S.C. 1997.
- Oracle (1997e) Oracle & Parallel Server, Concepts and Administration, Release 8.0, 1997.
- Opach [19970] Oracle 8 Server Spatial Cartridge, User's Guide and Reference, Release 8.0.3, 1997.
- Orborn, S. [1977] Normal Forms for Relational Databases, J'h D. dissertation, University of Waterloo, 1977.
- Othom, S. [1979] "Towards a Universal Relation Interface," in VLDB [1979].
- Oshorn, S. [1989] "The Role of Polynophismon Scheme Lyolution in an Object-Oriented Database," TKDE, 1-5, September 1969.
- Orseyndu, G., Ocsoyoghi, Z., and Matos, V. [1985] "Extending Relational Algebra and Relational Calculus with Set Valued Attributes and Aggrogate Functions," TODS, 12:4, December 1987.

- Otsoyoglu, Z., and Yuan, L. [1987] "A New Normal Form for Nested Relations," TODS, 12-1, March 1987.
- Olsu, M. T., and Valdutter, F. [1999] Principles of Distributed Database Systems, 2nd ed., Ptentice-Hall, 1999.
- Papadomition, C. [1979] "The Socializability of Concurrent Database Updates," JACM, 26:4, October 1979.
- Papadimitriou, C. [1986] The Theory of Database Concorrency Control, Computer Science Press, 1986.
- Papadiuntrion, C., and Kanellakis, P. [1979] "On Concurrency Control by Multiple Versums," (1)08, 9(1) March 1974.
- Paparaghu, M., and Valder, W. [1989] Relational Database Management: A Systems Programming Approach, Prentice Hall, 1989.
- Paredaens, J., and Van Goeht, D. [1992] "Conversing Nested Algebra Expressions into Flat Algebra Expressions," TOP8, 17(1) March 1992.
- Parent, C., and Spaccapietra, S. [1985] "An Algebra for a Ceneral Entity Relationship Model," 186, 11:7, July 1985.
- Paris, J. [1986] "Voring with Witnesses: A Consistency Scheme for Replicated Files 1 in 1006 [1986].
- Park, I., Chen, M., and Ya, P. [1995] "An Effective Hash Based Algorithm for Mining Association Rules," in SIGMOD [1995].
- Paron, A. W. ed. [1999] Active Rules in Database Systems, Springer Verlag, 1999.
- Puron, N. W. and Diat, O. [1999] Survey of Acros e Database Sesteins. ACM Computing Surveys, to appear.
- Patterson, D., Gilson, G., and Katt, R. [1985]. "A Case for Redundant Arrays of Inexpansive Disks (RAID)." in SIG90D [1988].
- Paul, D., et al. [1987] "Architecture and Implementation of the Damistadr Database Kernel System," in SIGMOD [1987].
- Parandak, P., and Suvastava, J., "Evaluating Object DBMSs for Multimedia," IEEE Multimedia, 4(3, pp. 34–49).
- PDES [1991] "A High-Lead Architecture for Implementing a PDES/STEP Data Sharing Environment," Publication Number PT 1017.23.00, 1008 Inc., May 1991.
- Pearson, P., Francomaan, C., Fester, P., Boerhun, C., Li, P., and M. Kusick, V. [1994] "The Sease of Online Mendelion adversaries in Man (CMIM) Media 1994" Nucleic Acids Research 22:17, 1994.
- Peckham, J., and Maryanski, F. [1985] "Sensitive Data Models," ACM Computing Surveys, 20:3, September 1988, pp. 153–89.
- Pfloeger, C. P. [1997] Security in Computing, Prencice Hall, 1997.
- Phipps, G., Derr, M., Ross, K. [1991] "Glue-NALE: A Deductive Database System," in Spaxiop [1991].
- Piatesky-Shapiro, G., and Frayley, W., eds. [1991] Knowledge Discovery in Databases, NAM Press/MIT Press, 1991.

- Pistor P. and Anderson, E [1986] "Designing a Generalized NF Model with an Sigurype Language Interface," in V110 [1986], pp. 278–85.
- Picoura, E., Bickhres, O., and Elmagarmid, A. [1995] "Object Orientation: in Multidatabase Systems," ACM Computing Surveys, 27:2, June 1995.

Pitoura, E., and Samaras, G. [1998] Data Management for Mobile Computing, Kluwer, 1998.

- Poosala, V., Ioannidis, Y., Haas, P., and Shekuta, F. [1996] "Improved Histograms for Solectivity Estimation of Range Predicates," in SIGVOD [1996]
- Potter, B., Sinclair, J., Till, D. [1991] An Introduction to Formal Specification and Z, Prenuce-Hall, 1991.
- Rabitti, F., Bertino, E., Kim, W., and Woelk, D. (1991) "A Model of Authorization for Next-Generation Database Systems," TODS, 16:1, March 1991.

Romokrishnan, R., AJ [1995] Applications of Logic Databases. Klower Academic, 1995.

Ramakrishnon, R. [1997] Database Management Systems, McGraw-Hill, 1997.

- Ramakushnan, R., Suvastava, D. and Sudarshan, S. [1992] "ICORAL: [C]ontrol, [R]elations and [L]opte," in VLDB [1992]
- Ramakrishnan, R., Stivastava, D., Sudaishan, S. and Sheshada, P. (1993) "Implementation of the KORAL) deductive database system." in SIGMOD [1993].
- Ramakrishnan, R., and Ullman, J. [1995] "Survey of Research in Deductive Database Systerms," Journal Of Logic Programming, 23:2, 1995, pp. 125-49.
- Ramanasouthy, C., and Wals, B. [1979] "The Placement of Relations in a Distributed Relational Database," Proceedings of the First International Conference on Distributed Compating Systems, (RFLCS, 1979).
- Ramesh, V., and Ram, S. [1997] "Integrity Constraint Integration in Heterogeneous Databases on Enhanced Methodology for Schoma Integration," Information Systems, 22:8, December 1997, pp. 423–46.
- Reed, D. [1981] "Implementing Atomic Actions on Decembrahied Data," TOCS, 1:1. Febmary 1983.
- Reisner, P. [1977] "Use of Psychological Experimentation as an Aid to Development of a Query Language," 186, 3(3, May 1977)
- Reisner, P. [1951] "Human Factors Studies of Dotabase Query Languaged A Souvey and Assessment," ACM Computing Surveys, 13-1, March 1981.
- Reiter, R. [1984] "Towards a Logical Reconstruction of Relational Database Theory," in Brodie et al., ch. 5, [1984].
- Ries, D., and Stonebraker, M. [1977] "Effects of Locking Granularity in a Database Management System," TODS, 2-3, September 1977.
- Rissanen, J. [1977] "Independent Components of Relations," TODS, 2:4. December 1977 -
- Robbins, R. (1993) "Genome Informatics: Requirements and Challenges," Proceedings of the Second International Conference on Bioinformatics, Supercomparing and Complex Genome Analysis, World Scientific Publishing, 1993.
- Roth, M., and Korth, H. [1987] "The Design of Non-TNF Relational Databases into Nested Normal Form," in 505000 [1987].

- Both, M. A., Korth, H. E. and Silberschutz, A. [1988] Extended Algebra and Colculus for non-INF relational Databases," TOD3, 13:4, 1988, pp. 389–417.
- Rothmu, J., et al. [1990] "Introduction in a System for Distributed Databases (SOV 1)," TODS, 5.1, March 1980.
- Rocssogoulos, N. [1991] 'An Incremental Access Method for View-Carbo Concept. Algorithms, and Cost Analysis," TODS, 10(3, September 1991.
- Roten, S., and Shasha, D. [1991] "A Framework for Automating Physical Database Design," in V108 [2091].
- Rudensteiner, E. [1992] "Multiview: A. Methodulogy for Supporting Multiple Views in Object/Oriented Databases," in VOB [1992].
- Rueminder, C., and Wilkes, J. [1994] "An Introduction to Dick Drive Modeling," 954 Computer, 27:5, March 1994, pp. 17–27.
- Rombaogh, J., Blaha, M., Premerlani, W., Eddy, E., and Lorensen, W. [1991] Object Oriented Modeling and Design, Prentice Hall, 1991.
- Rusinkiewicz, M., et al. [1955] "OVANIB/CF—A Loosely Coupled. Design and Implemenration of a Multidurabase System," IEEE Distributed Processing Newsletter, 10(2) November 1968.
- Rustin, R., ed. [1972] Data Base Systems, Prentice-Holl, 1972.
- Restin, R., ed. [1974] Proceedings of the BJN/AV2.
- Socca, M, and Zaniolo, C. [1987] "Intelementation of Recursive Queries for a Data Langrage Based on Pure Horn Clauses," Proceedings of the Frank International Conference on Logic Programming, MIT Press, 1986.
- Sodri, F. and Ullman, J. [1952] "Templote Dependencies: A Large Class of Dependenciein Relational Databases and Its Complete Accomptization," JACM, 29:2, April 1982.
- Saery, Y., and Yannakakas, M. [1981] "Equivalence among Relational Expressions with the Unior, and Difference Operators," JACM, 27 4, November 1981.
- Sakar H. [1950] "Entry-Relationship Approach to Conceptual Schema Pesign" in SIC MON [1950].
- Salaberg, B. [1988] File Structures: An Analytic Approach, Prentice-Hall, 1988.
- Salzberg, B., et al. [1990] "FastSort: A Distributed Single-Input Single-Output External Sort," in Stemoti [1990].
- Salton, G., and Buckley, C. [1991] "Global Text Matching for Information Retrieval" in Science, 253, August 1991.
- Samer, H. [1990] The Design and Analysis of Spatial Data Structures, Addison-Wesley, 1990.
- Samet, H. [1990a] Applications of Spatial Data Structures: Computer Graphics, Image Processing and US, Addocm Wesley, 1990.
- Sammur, C., and Sammur, R. [1983] "The Implementation of UN-20-PRODOC," The Australian Computer Journal, May 1985.
- Satistia, W., and O'Neill, W. [1999]. GIS in Transportation, in Taylor and Francis [1999].

- Sarawagi, S., Thomas, S., Agrawal, R. [1998] "Integrating Association Rules Mining with Relational Database systems: Alternatives and http://joarions/" in SIC28400 [1998].
- Savasere, A., Omiecraski, E., and Navarke, S. [1995] "An Efficient Algorithm for Mining Association Roles," in VI16 [1995]
- Savasere, A., Omeerinski, E., and Navathe, S. [1998] "Mining for Strong Negative Association in a Large Database of Customer Transactions," in 1005 [1998].
- Schotz, B. [1995] "Information Analysis in the Net-The Inter-pace of the Twenty-First Century" Keymon Pienary Learner or American Science for Information Science (ASIS) Annual Meeting, Chicago, October 11, 1995.
- Schatz, B. [1997] "Information Retrieval in Digital Libraries: Bringing Search to the Net," Science, vol. 275, 17 January 1997.
- Schek, H. J. and Scholl, M. H. [1986] "The Relational Model with Relation-valued Attributes," Information Systems, 11:2, 1986.
- Schek, H. J., Paul, H. B., Scholl, M. H., and Weikiam, O. (1990) "The DASDRS Project, Objects, Experioricus, and Future Projects," IEEE TRDE, 2(1) 1990.
- Schouermann, P., Schuffner, G., and Webur, H. [1979] "Abstraction Capabilities and Invariant Properties Modeling within the Entity-Relationship Approach," in EP. Conference [1979].
- Schlimmer, L. Mirchell, T. McDermetr, J. [1991] "Justification Based Refinement of Expert Knowledge" in Platesky-Shapiro and Frawley [1991].
- Schmidt, J., and Swenson, J. (1975) "On the Semantics of the Retational Model," in Sig-MOD [1975].
- Sciore, E. [1982] "A Complete Axiomatization for Full Juin Dependencies," JACM, 29:2, April 1982.
- Sehnger, P., et al. [1979] "Access Path Selection in a Relational Database Management System," in SIGMOD [1979].
- Senko, M. [1975] "Specification of Stored Data Structures and Desired Output in DIAM II. with ECRAL," in VILLE [1975].
- Senka, M. [1980] "A Query Maintenance Language for the Data Independent Accessing Model II." Information Systems, 5:4, 1980.
- Shapiro, L. [1966] "Join Processing in Database Systems with Large Main Memories," (COS, 11:3, 1986.
- Shasha, O. [1992] Database Tuning: A Principled Approach, Prentice Holl, 1992.
- Shosha, D., and Goodman, N. [1958] "Concurrent Search Structure Algorithms," TODS, 13(1), Match 1948.
- Shekita, E., and Carey, M. [1989] "Performance Enhancement Through Replication in an Object-Oriented DBvs," in SIGMOF [1989].
- Shenoy, S., and Osoyaglu, Z. [1989] "Design and Implementation of a Semarite Query Optimizer," TKDE, 1:3. September 1989.

- Sherh, A., Golo, S., Novathe, S. [1993]" On Automatic Reasoning for Schema Integrarion." in International Journal of Intelligent Co-operative Information Systems, 2.1, March 1993.
- Sheih, A. D. and Larson, J. A. [1990] "Federated Database Systems for Managine Distributed, Heterogeneous, and Autonomous Databases," ACM Computing Surveys, 22-3, September 1990, pp. 183–236.
- Sherh, A., Lurson, J., Cornelio, A., and Navothe, S. [1988] "A 'Tool for Integrating Conceptual Schemas and User Views "in 1018 (1988).
- Shipman, D. [1981] "The Functional Data Model and the Data Language DAPLEX," TODS, 6-1, March 1981.
- Shlaer, S., Mellor, S. (1988) Object-Oriented System Analysis: Modeling the World In Data, Yourder, Press, 1988.
- Shneiderman, B., ed. [1978] Databases: Improving Usability and Responsiveness, Acadenne Press, 1978.
- Subley, E., and Kerschberg, L. [1977] "Data Architecture and Data Model Considerations," ACC, ArIN, 46, 1977.
- Sregel, M., and Madnick, S. [1991] "A Metadata Approach to Resolving Semantic Conllicits," in vito6 [1991].
- Stegel, M., Smore, E., and Salveter, S. [1992] "A Method for Automatic Rule Derivation to Support Semantic Query Optimization," TODS, 17:4. December 1992.
- SIGMOD [1974] Proceedings of the ACM SIGMOD-AUALIST Conference on Data Descriptions. Access, and Control, Russin, R., ed., May 1974.
- SICHARD [1975] Proceedings of the 1975 ACM SPECIAL International Conference on Management of Data. King, E. ed., San Josef, CA, May 1975.
- SIGMAD [1976] Proceedings of the 1976 ACM SN MOD International Conference on Management of Units, Rothmer, J., ed., Washington, June 1976.
- SICVOD [1977] Proceedings of the 1977 ACV SEDARCE International Conference on Management of Linux, Structure Direct. Toronto, August 1977.
- SICMOD [1978], Proceedings of the 1978 ACM SIGMORT International Conference on Management of Data. Lowenthal, L. and Dale, N., eds., Austin, TX, May/June 1978.
- SIGMOD [1979] Proceedings of the 1979 Acts SIGMOD Discretarional Conference on Management of Data. Bernstein, P. ed., Boston, MA, May/June 1979.
- SIGMOD [1950] Proceedings of the 1950 ACM SIGM (D International Conference on Management of Data, Chen, F and Sprowls, R., eds., Santa Monica, CA, May 1980.
- SIGMOD [1981] Proceedings of the 1984 (MitM MitMed) International Conference on Management of Data. Liess, Y., ed., Arm Arbor, MI, April/May 1981.
- SIGMOD [1982] Proceedings of the 1982 ACM SIGMOD International Conference on Management of Data. Schkolnick, M., ed., Orlando, FL, June 1982.
- SEANOD [1963] Proceedings of the 1983 ACM STOMOD International Conference on Management of Data, DeWitt, D. and Oardarin, G., eds., San Jose, UA: May 1983.

- SIGMER) [1984] Proceedings of the 1984 ACM SIGMERD International Conference on Management of Data, Yormark, E., ed., Boston, MA, June 1984.
- SIGMOD [1985] Proceedings of the 1985 ACM SIGMOD International Conference on Management of Data, Navarhe, S., ed., Austin, TX, May 1985.
- 5.03000 [1986] Proceedings of the 1986 (6094 (200400) International Conference on Management of Data, Zaniolo, C., ed., Washington, May 1986.
- SUMOD [1987] Proceedings of the 1987 ACM SIGMOD International Conference on Management of Data: Data1, U. and Traiger, L. eds., San Francisco, CA, May 1987.
- SLOMOLE [1958] Proceedings of the 1988 ACM SIGMOD International Conference on Management of Data. Bond, H., and Larson, P., eds., Chicago, June 1988.
- 5.0MOD [1939] Proceedings of the 1999 ACM SIGMOD International Conference on Management of Data, Chitroid, J., Luidsay, B., and Mater, D. eds., Portland, OR, June 1989.
- SIOMOD [1980] Proceedings of the 1990 ACM SIGMOD International Conference on Management of Data, Garcia-Molina, H., and Jagadiah, H., eds., Atlantic City, NJ, June 1990.
- SIGMOD [1991] Proceedings of the 1991 ACM MOMON International Conference on Managermeter of Data, Chifford, J. and King, R., eds., Deriver, CO, June 1991.
- SKOMOD [1992] Proceedings of the (1992 ACM SIGMOD International Conference on Management of Data, Stonebraker, Maced., San Diego, CA, June 1992.
- SIGMOD [1993] Proceedings of the 1993 ACM SIGMOD International "Conference on Management of Data, Punchan, P. and Jajodia, S., eds., Witshington, Jone 1993.
- SICMCO [1994] Proceedings of 1994 ACM MGMOD International Conference on Managemon of Data, Snodgrass, R. T., and Winslett, M., eds., Minneapolis, MN, June 1994.
- SIGMOD [1995] Proceedings of 1995. SCM SIGMOD International Conference on Management of Data, Carey, M., and Schneider, D. A., eds., Minneapolis, MN, June 1995.
- SIGMOD [1996] Proceedings of 1996 ACM SIGNOG International Conference on Management of Data, Jagadish, H. V., and Murnick, I. P. eds., Montreal, Jurie 1996.
- SICMOD [1997] Proceedings of 1997 ACM SIGMOD International Conference on Management of Data, Peekham, J., ed., Teeson, AZ, May 1997.
- SIGMOD [1998] Exceedings of 1998 As St MCMOF: International Conference on Management of Data, Haus, L., and Tiwary, A., eds., Scattle, WA, June 1998.
- SUMOD [1999] Proceedings of 1999 ACM SUBSOD International Conference on Management of Data, Faloutsos, C., ed., Philadeiphia, PA, May 1999.
- Silberschure, A., Stonebraker, M., and Ullman, J. [1990] "Database Systems: Achievements and Opportunities," in ACM SIGMOD Record, 19:4, December 1990.
- Sillerschare, A., Korth, H., and Sudarshan, S. (2001) Database System Concepts. 4th ed., McGraw-Hill, 2001.
- Smith, G. [1990] "The Semantic Data Model for Sciurity Representing the Security Semantics of an Application," in ICDE [1990].
- Smith, J., and Chang, P. [1975] "Optimizing the Performance of a Relational Algebra. Interface," CACM, 18-10, October 1975.

- Smith, J., and Smith, D. (1977) "Database Abstractions: Aggregation and Generalization "TODS, 2-2, june 1977.
- Smith, J., et al. [1951] "MULTIBASI: Integrating Distributed Heterogeneous Database Systems," NuCl. ATRS, 50, 1981.
- South, K., and Worslett, M. [1992] "Entry Mudeling in the Mrs Relational Model," in Virthe [1992].
- Smith, E. and Barnes, G. [1987] Files and Databases: An Introduction, Addiant-Wesley, 1987.
- Shudgrass, R. [1987] "The Temporal Query Language TQuel," (ODS, 12:2, June 1987).
- Studgrass, R., ed. [1995] The TSQL2 Temporal Query Language, Klower, 1995.
- Spedgrass, R., and Ahn, J. [1985] "A Taxonomy of Time in Databases," in SIGMOD [1985].
- Souton, G. [1998] "Analysis of Constraints for Norty Relationships," in FR98.
- Spaccapierra, S., and Jain, R., eds. [1995] Proceedings of the Visual Database Wirekshap, Lass nine, Switzerland, Occober 1995.
- Spooner D., Michael, A., and Desaid, B. (1986) "Medeling CND Data with Data Abstraction and Object Oriented Technique," in JCDE [1986].
- Stokant, R., and Agrawal, R. [1995] "Mining Generalized Association Roles," in VL09 [1995].
- Soutvas, M., and Fatnark, L. [1994] "Genetic Algorithms: A Survey." (EEE Computer, June 1994)
- Suntvasan, V., and Carey, M. (1991) "Performance of D-Tree Conductorics y Control Algorithms," in Stokico [1991].
- Suvastavo, D., Ramakushnan, R., Sudarshan, S. and Sheshadu, P. (1994) "Cural+to Adding Object-orientation to a Logic Dambase Language," in V109 [1993].
- Stachour, P., and Thurasongham, B. (1992) "The Design and Implementation of INCRES," TKDE, 2:2, June 199C.
- Stallings, W. [1997] Dava and Computer Communications, 5th ed., Prontice Hall, 1997.
- Stallings, W. (2000) Network Security Essentials: Applications and Standards, Promise Hall, 2000.
- Stonebtaker, M. [1975] "Implementation of Integrity Constraints and Views by Query Medineation," in FRANDE [1975].
- Stonebraker, M. [1993] "The Miro PPMS" in SIGMOD [1995].
- Stenebroker, M., ed. [1994] Readings in Database Systems, 2nd ed., Morgan Kaufmann, 1994.
- Stonewaker, M., Hanson, et and Hong, C. [1987] "The Design of the reisticizes Rates System," in ICTW [1987]
- Stonebraker, M., with Moore, J. (1996). Object-Relational DBMSs: The Next Great Wave, Motgan Kauhnan, 1996.
- Stonebraker, M., and Rove, L. [1980] "The Design at NetGRES," in SICKOD (1980].

- Stunebraker, M., Wung, E., Kreps, P., and Held, G. [1976] "The Design and Implementation of INSIRES," TOOS, 1:3, September 1976.
- Su, S. [1985] "A Semantic Association Model for Corpurate and Sciencific Statistical Databases," Information Science, 29, 1985
- Su. S. [1988] Database Computers, McGraw-Hill, 1988.
- Su, S., Kushnamerthy, W. and Lun, H. [1988] "An Object-Oriented Semantic Association Model (OSAM\*)," in AL in Industrial Engineering and Manufacturing: Theoretical Issues and Applications, American Institute of Industrial Engineers, 1988.
- Subrahmanian, V. [1998] Principles of Multimedia Databases Systems, Morgan Kaufmann. 1998.
- Solimmanian V. S., and Jajudia, S., eds. [1996] Multimedia Database Systems: Issues and Research Directions, Springer Verlag, 1996.
- Sunderramon, R. [1999] ORACLE Programming: A Primer, Addison Wesley Longman, 1999.
- Swami, A., and Gapta, A. [1989] "Optimization of Large Join Queries: Combining Hearistics and Combinatorial Techniques." in SIGMOD (1989)
- Tanenbaum, A. [1996] Computer Networks, Prentice Holl PTR, 1996.
- Tansel, A., et al., eds. [1993] Temporal Databases: Theory, Design, and Implementation, Benjamin Commings, 1993.
- Teorey, T. (1994) Database Modeling and Designs The Fundamental Principles, 2nd ed., Murgan Kaufmann, 1994.
- Teorey, T., Yang, D., and Fry, J. [1986] "A Legical Design Methodology for Relational Databases Using the Extended Enrity-Relationship Model," ACM Computing Surveys, 18:2, June 1986.
- Thomas, J., and Gould, J. (1975) "A Psychological Study of Quark by Example," NOC APJ75, 44, 1975.
- Thomas, R. [1979] "A Majority Consensus Approach to Concurrency Control for Moltiple Copy Data Bases," TOD5, 4(2, June 1979).
- Thomastan, A. [1991] "Performance Limits of Two-Phase Locking," in ICDE [1991].
- Thuraisingham, B., et al. [2001] "Directions for Web and E-Commerce Applications Scienty," Tenth IEEE International Workshops on Enabling Technologies. Infrastructure for Collaborative Enterprises, 2001, pp. 200–204.
- [Jodd, S. [1976] "The Peterlee Relational Test Vehicle—A System Overview," IBM Systems Journal, 15 4, Excember 1976.
- Toivoxien, H., "Sampling Large Databases for Association Rules," in V108 [1996].
- Too, J., ed. [1984] Information Systems COINS-IV, Plenum Press, 1984.
- Tsangaris, M., and Naughton, J. [1992] "On the Performance of Object Clustering Techinques," in SIGMOD [1992]
- Tsichritais, D. [1952] "Forms Management," CACM, 25-7, july 1982
- Tsichmens, D., and Klug, A., eds. [1978] The ANSI/XJ/SPARC DBMS Framework, AFIPS Press, 1978.

- Tsichritze, D., and Lochowsky, F. [1976] "Hierarchical Data-base Managemeter: A Survey," ACM Computing Surveys, 8:1, March 1976.
- Tsichrings, D., and Lochovsky, F [1982] Data Models, Prentice Hull, 1982.
- Totras, V., and Gopinath, B. [1992] "Oprinod Versinging of Object Classes," in EQP (7992].
- Tou, D. M., and Fischer, P. C. [1982] "Decomposition of a Relation Scheme into Poyce CoM/Normal Form," SIGACT News, 14.3, 1982, pp. 23–29.
- Ullman, J. (1982) Principles of Database Systems, 2nd ed., Computer Science Press, 1982.
- Ullman, J. [1985] "Implementation of Logical Query Languages for Databases," 1008, 10(3) September 1985.
- Ullman, J. [1966] Principles of Database and Knowledge-Base Systems, vol. 1, Computer Science Press, 1986.
- Ullman, J. [1959] Principles of Database and Knowledge-Base Systems, vol. 2, Computer Science Press, 1989.
- Ullman, J. D. and Widom, J. [1997] A First Course in Database Systems, Frence-Hall, 1997.
- U.S. Congress [1988] "Office of Technology Report, Appendix D: Databases, Repestones, and Informatics," in Mapping Our Genest Genome Projects: How Big. How Fast? John Hopkins University Press, 1988.
- U.S. Department of Connucree [1993]. TIGER/Line Files, Barcan of Census, Wishington, 1993.
- Valdanes, P., and Gardann, G. [1989] Analysis and Comparison of Relational Database Systems, Addison-Wesley, 1989.
- Vasalion, Y. [1980] "Functional Dependencies and Incomplete Information," in VDB-[1980].
- Verheijen, O., and VanBekkum, J. (1982) "NYAM: An Information Analysis Method," in Officiental, [1982].
- Verhofstadt, J. [1978] "Recovery Techniques for Distubase Systems," ACM Computing Surveys, 10.2, June 1978.
- Vielle, L. [1986] "Recursive Axioms in Deductive Databases: The Query-Subquery Approach," in EDS [1966].
- Vielle, L. [1987] "Database Complete Proof Production Based on SLD-resolution." in Proceedings of the Fourth International Conference on Logic Programming, 1987.
- Vielle, L. [1986] "From QSQ Towards Q/SaQ: Global Optimization of Recursive Queries," in the [1988]
- Vieiller, L. [1998] "VALIDITY: Knowledge Independence for Electronic Mediation," invited paper in Practical Appleations of ProtogPractical Applications of Constraint Technology (PAPIPACT: 981, London, March 1995, available from Interfle@computersorg.

- Vin, H., Zellweger, P., Swineharr, D., and Venkar Ranean, B [1991] "Multimedia Conferencing in the Frherphone Environment," IEEE Computer, Special issue on Multime-Bia Information Systems, 24(10) October 1991.
- VUDB [1975] Proceedings of the Eirst International Conference on Very Large Data Bases, Kerr, D., eds. Framinghams MA: September 1975.
- VLDB [1976] Systems for Large Databases. Lockemann, P. and Neuhold, E., eds., in Proceedings of the Scound International Conference on Very Large Data Bases. Brussels, Belgram, July 1976. North-Holland, 1976.
- VLIM [1977] Proceedings of the Third International Conference on Very Large Data Bases, Morton, A., ed. Sokyo, Japan, October 1977.
- VLTW [1975] Proceedings of the boarth International Conference on Very Large Data Bases, Bubenko, J., and Yao, S., eds., West Berlin, Germany, September 1978.
- VLPB [1979] Proceedings of the P\$16 International Conference on Very Large Data Bases, Fretodo, A., and Morgan, H., eds., Rio de Janeiro, Brand, October 1979.
- XLDB (1980) Proceedings of the Sixth International Conference on Very Large Data Bases, Lochovsky, E. and Taylor, R., eds., Montreal, Canada, October 1980.
- VLOB [1951] Proceedings of the Secondi International Conference on Very Large Data Bases, Zamichi, C., and Delobel, C., eds., Carmes, France, September 1981.
- vi D8 [1982] Evaluatings of the Fighth International Conference on Very Large Data Bases, McLevel, D., and Villaseport Y., eds., Mexiro U(iv) September 1972.
- VLDB [1983] Proceedings of the Ninth International Conference on Very Large Data Bases, Schoolnick, Maland Thanas, Calenda, Eurence, Italy, October/November 1983.
- VLDB [1984] Proceedings of the Tenris International Conference on Very Large Data Bases, Davab. U., Schlageter, G., and Song, L., eds., Singapore, August 1984.
- VEDB [1986] Proceedings of the Eleventh International Conference on Very Large Data Bases, Pirester, A., and Vassihou, Yu. eds., 5sockbolm, Sweden, August 1985.
- VLDP [1956] Proceedings of the Tweifth International Conference on Very Large Data Bases, Chu, W., Gardacev G., and Ohsuga, S., eds., Kystu, Japan, August 1986.
- VIT/B [1987] Proceedings of the Throteonth International Conference on Very Large Data Bases, Stocker, P., Kerst, W., and Hammersley, P., eds., Brighton, England, September 1987.
- VLFR (1988) Proceedings of the Fourteenth International Conference on Very Large Data Bases, Bancolhon, E, and DeWitt, D., eds., Los Angeles, August/September 1988.
- VLDB [1989] Proceedings of the Fifteenth International Conference on Very Large Data Bases, Aprils, E., and Wiederhold, G., eds., Amsterdam, Angasr 1989.
- VLOB [1990] Proceedings of the Sortsenth International Conference on Vers Large Data Bases, McLeeol. D., Sacks-Davis, R., and Schek, H., eds., Brisbane, Australia, August 1940.
- VILM [1991] Proceedings of the Seventeenth Intervisional Conference on Very Large Data Bases, Lohman, G., Sernadas, A., and Camps, R., eds., Barcelons, Carabania, Spinn, September 1991.

- VDB [1992] Proceedings of the Eighteenth International Conference on Very Large Data Bown Yoan, L., ed., Vancouver, Canada, August 1992.
- VLDB [1993] Proceedings of the Nineteenth International Conference on Very Large Data Bases, Agrawal, R., Baker, S., and Bell, D. A., eds., Dubhn, Ireland, August 1993.
- VLDP [1994] Proceedings of the 20th International Conference on Very Large Data Bases, Bocca, J., Jacke, M., and Zanioles, C., eds., Santiago, Chile, Soptember 1994.
- VEDP [1995] Proceedings of the 21st International Conference on Very Large Data Bases, Dayal, U., Gray, P.M.D., and Nishio, S., eds., Zurich, Switzerland, September 1995.
- VLDP [1996] Proceedings of the 22nd international Conference in Very Large Data Bases, Vijavaratnans, T. M., Buchmans, A. P., Mohan, C., and Sardo, N. L., eds., Bombay, India, September 1996.
- VLDP [1997] Proceedings of the 23rd International Conference on Very Large Data Bases, Jarke, M., Carey, M. J., Dirtrich, K. R., Lochowsky, P. Fil, and Loncepoulos, P.(edutors), Zurich, Switzerland, September 1997.
- VLDB [1998] Proceedings of the 24th International Conference on Very Large Data Bases, Gupta, A., Shimuch, O., and Wildom, J., eds., New York, September 1998.
- VLOB [1999] Proceedings of the 25th International Conference on Very Large Data Bases, Zdornk, S. B., Valdutiez, P., and Orlowska, M., eds., Edinburgh, Scotland, September 1999.
- Virhaus, A., and Mills, R. (1967) "The Time-Shared Data Management System: A New Approach to Data Management," System Development Corporation, Report SP-2634, 1967.
- Wallace, D. (1995) "1994 William Allan Award Address: Mitochondual DNA Variation in Human Evolution, Degenerative Disease, and Aging," American Journal of Human Genetics, 57:201–223, 1995.
- Walton, C., Dale, A., and Jenevere, R. [1991] "A Taxonomy and Performance Model of Data Skew Effects in Parallel Joins," in VI 00 [1991].
- Wang, K. [1990] "Polynomial Time Designs Toward Both (SINE and Efficient Data Manipulation," in E-GMOD [1990].
- Wang, Y., and Madnick, S. [1989] "The Inter-Database Instance Identity Problem in Integrating Autonomous Systems," in ICLE [1989].
- Wang, Y. and Rowe, L. [1991] "Usable Consistency and Concurrency Control in a Chent/ Server DBMS Aschirecture," in SIGMOD [1991]
- Warren, D. [1992] "Memoing for Logic Programs," CACM, 35(3), ACM, March 1992.
- Weddell, G. [1992] "Ressoning About Functional Dependencies Constantiated for Semanine Data Models," TODs, 17,1, March 1992.
- Weikom, G. [1991] "Principles and Realization Strategies of Multilevel Transaction Maragement," TODS, 16:1. March 1991.
- Weiss, S. and Indurkhya, N. [1998] Predictive Data Mining: A Practical Guide, Morgan-Kautmator, 1998

- Whang, K. [1985] "Query Optimization in Office By Example," IBM Research Report RC: 11971. December 1985
- Whang, K., Mallantra, A., Sockut, G., and Butny, L. [1992] "Supporting Universal Quaninframmerine a Two-Dimensional Database Query Language," in K 306 [1990].
- Whang, K., and Navathe, S. [1987] "An Extended Disjunctive Normal Form Approach for Processing Recursive Logic Queries in Loosely Coupled Environments," in VLD8 [1987].
- Whang, K., and Navathe, S. [1992] "Integrating Expert Systems with Database Magagement Systems—an Extended Disjonetive Normal Form Approach," Information Sciences, 64, March 1992.
- Whang, K., Wiederhold, G., and Sagalowicz, Li [1982] "Physical Design of Network Model Databases Using the Property of Separability," in VLDB [1982].
- Widom, J., "Research Problems in Data Watehousing," (38M, November 1995).
- Widom, J., and Ceri, S. [1996] Active Database Systems, Morgan Kaufmann, 1996.
- Widom, J., and Finkelstein, S. [1990] "Set Opented Production Bules in Relational Database Systems" in 500MOD [1990].
- Wiederhold, G. [1983] Database Design, 2nd ed., McGraw-Hall, 1983.
- Wiederheld, G. [1954] "Knowledge and Database Management," IEEE Software, January 1984.
- Wiederheld, G. [1995] "Digital Libraries, Value, and Productivity," CACM, April 1995.
- Wiederhold, G., Beetem, A., and Short, G. [1982] "A Database Approach to Communication in VCSI Design," IEEE Transactions on Computer-Aided Design of Integrated Circuits and Systems, 1-2, April 1982.
- Wiederhold, G., and Elmaste, R. (1979) "The Structural Model for Database Design," in 18 Contention (1979).
- Wilkmson, K., Lynghack, P., and Hasan, W. [1990] "The IRIS Architecture and Implementation." TKDE, 2:1. March 1990.
- Willshire, M. [1991] "How Spacey Can They Get? Space Overhead for Storage and Indexing with Object-Oriented Databases," in JCCE [1991].
- Wilson, B., and Navathe, S. [1986] "An Analytical Framework for Lincked Redexign of Distributed Databases," *Proceedings of the South Advanced Database Symposium*, Tokyo, August 1986.
- Wiorkowski, G., and Kull, D. [1992] DB2-Design and Development Guide, 3rd ed., Addison-Wesley, 1997.
- Worth, N. [197] Algorithms + Data Structures = Programs, Prentice-Hall, 1972.
- Wood, J. and Scher, D [1989]] Joint Application Design: How to Design Quality Systems in 40%: Less Time, Wiley, 1989
- Wong, E. [1963] "Dynamic Rematerialization-Processing Distributed Queries Using Redundant Data," TSE, 9:3, May 1983.
- Wong, E., and Youssefi, K. [1976] "Decomposition—A Strategy for Query Processing," TODS, 1-3, September 1976

- Wong, H. [1984] "Murri and Macro Statistical/Scientific Database Management," in 2006 [1984].
- Wu, X., and Juhakawa, T. [1992] "KDA: A Knowledge-based Database Absistant with a Query Guiding Facility," TRDE 4:5, October 1992.
- Yaunakakas, Y. [1984] "Serializability by Locking," JACM, 11:2, 1984.
- Yan, S. [1979] "Opministion of Query Evaluation Algorithms," TODS, 4:2, June 1979.
- You, S., ed. [1985] Principles of Database Design. vol. 1: Logical Organizations, Prentice Hall, 1985.
- Yousseh, K., and Wong, E. [1979] "Query Processing in a Relational Database Maragement System," in VIDB [1979].
- Zadeh, L. [1983] "The Role of Fuzzy Logic in the Monogenieur of Uncertainty in Expert Systems," Fuzzy Sets and Systems, 11, North Holland, 1983.
- Zaniolo, C. [1976] "Analysis and Drsign of Relational Schemata for Durabase Systems," Ph.D. discruttion, University of California, Los Angeles, 1976.
- Zamelo, C. [1988] "Design and Implementation of a Logic Based Language for Data Intensive Applications," MOC Technical Report #ACA-ST-199-86, June 1988.
- Zanielo, C., et al. [1986] "Object-Oriented Database Systems and Knowledge Systems," in E28 [1984].
- Zamolo, C., et al. [1997] Advanced Database Systems, Morgan Karfmann, 1997.
- Zave, P. (1997) "Classifications of Research Efforts in Requirements Engineering," ACM Computing Surveys, 29:4 December 1997
- T. Zhang, R. Ramakrishnan and M. LIVIN, "Brich: An Efficient Data Clustering Method for Very Large Databases," Proc. ACM SIGMOD Conference, 1996.
- Zican, R. (1991) "A Francework for Schema Updates in an Object-Oriented Database System," in ICOM (1991).
- Zhoof, M. [1975] "Query by Example," NCC: AE05, 44, 1975.
- Zleof, M. [1952] "Office By Example: A Business Language That Onfice Data. Word Processing, and Electronic Mail," IBM Systems Journal, 21:3, 1982.
- Zohek, J., Meffan, A., and Sacks-Davia, R. (1992) "An Efficient Indexing Technique for Bull-Text Database Systems," in VLDB [1992].
- Zook, W., et al. [1977] INGRES Reference Manual, Department of EE:S. University of California at Berkeley, 1977.
- Zvieli, A. [1966] "A Fulty Belational Calculus," in 603 [1986].

## Index

## A

abstract operation, 11 Abstract Synax Notation (Dne (488-1), 940 abara, trep concepts, 110 at Cens access pathod, 429 19A0 (domesionary access control) (743-744 data access, 42 discretaneous 735-740 E-connectue policies (14) tile 429 maridatory access control, 740-743 protection, 734-735 RRAU (role based access control) 744 secuential access dovines, 420-421 uniorbories, restricting, 16 accounts, superiser, 759 ACM Comparing Surveys 24 actions, 257 activate contoand, 362 semiation, expense disgrams, 389-390 active database systems, 19design and unprementation issues [761-761 generalized model Kie, 757-761

potential applications for, 369 technology, 3 active state. Transactions, 559. activity desgrams, 392 acyclic graphs, 44 ad-line querying, 907 addition (+) operator, 227 administrators. Sovidarabase administrations Advance | Energytzan Standards (208), 749 advanced replacation, 831 appressive functions, 165-166, 238-242, 539-511 aggregation, 75-1 (2-114 algebra. See relationed algebra, alsontine normalization, 345-047 relational database design, 340-147 abases, 222 all-key relations, 350 officeasilon, 812–815 contrigueus, 426 indexed, 427 lanked, 426 A0168 command, 217-218
We operated \$70, \$79 Animacans, multimedia Jasa, 923 soon diss detetion 300 insertion, 249-300 moduleation, 300 1.7.4are, 29%, iOD-iC2 Att (application programming interface), 41, 262, 275. aposnephe (2, 22) application based constrainty, 133 application development environments, 37 applicature layer (three nor chent-server acchatemore). ትረክ application programmers, 14 application programming interface (329), 41, 262, 275 application programs, 49, 262 application servers, 36, 42 applications data mong, 22 database, 49, 57-53, 255, 262 v [8, 930-93] pan'ny sala datahasasi 929–929 strengther, 22 spot al. 22 time series 22 Aprove algorithms 873-874 Also Justic software, 934–915 archived target 421 AFIES recovery algorithm, 425-629 suthmetric operations, 226-228 Amistrone's inference rules, \$99 atnow monation, 671 ASCREWWORD, 225 ASNU (Abstract Syntax Noration One), 940 assertions, 140 constraints as, 256, 257 declimative, 250 association caronomy, 613 assessments on many data mining urning hierarchics, 579 Apriori algorithm, 874-874 confidence, 472 trepent-patient tree deserther #75-\$78 marker-backer data, 374 and characteristical associations, \$80-881 negative associations, 381–682 paration algorishine, 876-879 sampling a)g. Auf (m, 474–875) ឈ្មោះលោក ស៊ីរ៉ាំ served patricity oppreparately 112-113 Inducertonal, 76 huony 75. defined 75

spadrhed, 76 reflexive, 76 uniderectional, 76 asterick (\*), 76, 224 ACM (Asynchroniws Transfer Mode), 805 atumic zriribates, 55-56 atomic literals, 668 aromic objects, 674, 676 arome value, 130 arons, 175, 153 arrithme-defined specialization, 92, 104 aurdanes, 75 uromur, 55, 56 Redean type 200 complex, 56-57 composite, 55, 56, 5d Johnsd, 27 Jenved, 56 discriminations, 201 londain st. 75 entities and, 53-57 entity types of \$7-58. gravitatione, 240 nuoge, 201 inforutance, 96 ken: 57. lunk, 70 Jocal, 59 multivalued, 56 isull values, 50 prime, 314 of relationship types, 67-68 relationships as 63-64 retaining, 246 supple, 55, 56 single-valued, 56 specific, 39 stored, 56 rags 544 value service, 59, 60 angles, marinny ha slata, 924 atality security, 735 autorestion identified 209 authorization subsystem, security and, 16automated database design tests, 401-405 autonomy in federated OBMS, 616-817 availabhra 807 AVED VGE 100/10-m. 165 456 Junction, 248-240

## p

B) -trees, 474, 481 B-trees, 443, 475-474

Postupan diamons, 948 hacklighting, 909 buckur and recovery systems, 17, 37, 630-641 base class, 104 Pase Stations, 916 taise rables, 210 Faste replacement 830 pegen (junsa) (jugs, 551 behavior inhermatice, 677 bidites (jonal association), 70 binary ages of sms [75, 105–108 binary decompositions, 336–340 Bmary Large Othes is (DLOPs), 423-658 Jamary Jocks, 594-585 binory relational operations, 158-162 Innary relationships, 03 hinary search, 431 bind operation, 678 binding, 264 homogeneous 997 biological sciences and generics, 956-939 NRCP algorithm, 888 hat. 415 Ist-level data struping, 446 hot-string data types, 212 htmap indexing, 906-907 PLOBS (Binary Large Objects), 423, 658 Mouk-level strinang, 440 Mock transfer times, 419, 952 blocking lactor, 425 hlucks Fulrences of, 421 sportures, 495-490 Joulean data types, 717 Boolean type attodates, 200 horean-up concertral synthesis, 98 bottom-up-design methodology, relation schema, 294 brund columns, 274 Bower-Clodd trormal form (BCN1), 524-326 broadcastine, 91% branising unerfaces, 34 bre (block analyfer (ime), disk parameters, 952 buffer inanager modules, 50 buriering modules, 17 balk transfer rates, 422 byres, M.S.

## С

1500-+, 255 C++ language bunding (93-694 cache mentory, 412 caching, of Jisk blocks, 613-614 calculus. Secretationed calculus

Call Language Interface 0.010, 248 CALL statement: 285 randidate keys, 105, 514 canned transactions, 202 cardanality taxio, 65-66, 129 ARTE-TAN PRODUCT operation, 11d causeling follback, 565, 616 CASE (computer-associal activate implementing), 585, 407-405 casual end overs, 13-14 ratalog Jabwa, 9 9.5., 209-213 category, 96-100, 202-201 centralated (00855, 35) character-sterne data types, 212 CHECK clause, 216 checkpaints, recovery, 615-616 child nisks 469 classic agrams, 74-76, 586-587 ritas libraries 230 class nume, 75 (last properties [1]) classical class relationships 36 (Losep. 103, 280 Bos. 104 defined, 75 eriver manager, 290 (adependent, (1)) la sí, 104 meta-class, 111 classification data intring, 870-882-885 Jehned, LLE dauser FROM, 219–220 INTC: 207 chuck, 216 writer chiffers o principal 261 KOREIGN, 215 GINULP B2, 240-243 HAV:NG, 240-243 PRIMARY, 214 SELECT, 219, 223, 498, 501 UNIOUR, 715 BOR UPDATE OF, 264 WHERE, 2.9-220, 223-224 C[1] [Call Language [prof.sch] [2]6 chent computers, & Gent mochines, 19 chentaix deles, 25 Chent programs 36, 263 chengseiver architecture 38 clients, 3cfined, 40

**REOSE CORSOR LONDEARD**, 269 clustering data mening, 585–585 indexes, 459-462 clusters, 419, 426-427, 444 CORCE, 255 collaboration degrams, 390 collection data types, 715-714 collection brerab, 668 coffection objects, 672-674 cullision, bashing and, 436 columns, bound, 27d commands. See also functions; operations activate, 762 ALTER, 217-218 TOOME CORSORE 268 CREATE SCHEMA, 269 CREATE TAISE 210-215 1 REATE VIEW, 258-259 161518.247 DRCH, 217, 262 DRCP VIEW, 769 FFIX - , 268 OD NOT: 717-715 INSERT, 245-247 OPEN A UTIMON, 267 REVOKE, 737 UPDATE 247-248 commercial to dy. data mining, 894-894 commit point, transactions, 561-562 containing state. Dortsactions, 563 communication actonomy, 516 communications variables, 260 Communications of the ACM, 24 communications software, 38 commutative operations, \$20 compaundus, 17 complete lastamul fragmentary of 811 completeness constraint, 93 complex attributes, 56-57 complex objects, 657-659 component diagrams, 567-488 composite attributes, 55-56, 55 homptome Lets, 485 computer as stell software engineering (CASE), 583 402, 403 conceptual data models, 26 conceptual durabase design, 52, 371-380 conceptual representation, [C Conceptual when a, 30, 52, 97–96 conceptualization, [15 concurrence control Jeadloviks, 591-594 district or all databases systems, 625-827

in indexes, 603-606 mulessension, 596-599 optimistic technoques, 599-800 phantom problem, 606 software 12 system look tubles, 584-588 renessampeng, 594, 596-597 validation techniques, 599-400 remourrent engineering, 602 condition-defined subclasses, 92 conditions, 175, 192, 257 confidence, data intuing, d/2 esomething fields, #42 connection chieves, 281 connection reasofs, 277 connections duratione servers, 205 re dambaas, 206 CONSTRAINTS application-based, 133 Jo associations, 256-257 completences, 95 constraint operation tanguage, 140 disjointness, 93 doman, 133 entity integroty, 114 infrerencinosiel-based, 139 integroy, 135 tomaing 216 referential integrity, 355 satisfied, 256 schema based, 133 SOL 213-217 state 140 transition, 140 tople-based, 216 violated, 250 configurus allocation, 426 controlled redundancy, 15, 16 conversion nutrines, 385 conversion cools, 37 correlated nested queries, 232-254 row based overy oparatization, 524-552 COUNT markets, 238-240 r ocean columnels, 746, 746, 749 (TREATE ASSERTION STOREMORY 1350 CREATS & HEMA command, 209 USPATE TABLE commapsk [2](-2)] CREAT- VIEW command, 758-739 credeouply, 745 CROSS PRODUCT's persition, 162 content state, 29 CORNER, 2011, 267 Chindens, disks, 416

#### D

DAC (discretionary access comps?), 743-744 CAMUIDARIA Agent Markor Language), 926 Jangling tuples, 442, 445 DARRA Agent Markup Lunguage (114 SIL), 926 Jara abriaction, 10 access, 42 bir level dara str. p. (ig. 446 complex relationships around 14 elements, 6 encoded, 743 flow diagrams, 12 tesementarion, 810-812 independence, 3 = 52 loc algoritori, 807 marker basker, 873 wifedescribing, 842 semistore torio1, 842 structurasil, 842 sublanguage, 13, 255. unstructured, 843 vietnal, 14 Data Blade modeles, 712 data definition language (100..), 52, 147 data die nonary systems, 17, 364 data driven design, 367 Data Energyption Standard (DES), 749-750 dota instrugentent issues. mobile databases, 920-921 multimoder Jatabases, 924–975 open, research problems, 925-924 Coto manipolocion language (1991), 42-54 data marte 202 data minung 22 applications of 891 association rules, 871-582 el Essécation, 852-585 clustering, 655, 888 commercial races, 891-894 discovery of prineriss in type scilles, 849 Jacovery of sequential potterns, \$88 generic algorithms, 890-891 neural newworks, 890 regression rule, 989 ischnology overview, 868–874 dara model, 43 consister of 26-27 dara wazehonses, 905-907 defines1, 10, 26 dara model mapping, 52 dara polorers, 476 data nyy kitore system, 37 data regultements, 50-32

énia serves, 41 dane sources, 150, 841 Zara types, 39, 423 bir-string, 212 Koolean, 212 character string 212 Jana, 212, 214 dare, 425 defined, 7 domains, 127 extonsible, 712, 714 smage, 718 interes (213 numeric, 212 \$01.209-213 rext, 719-520 rime, 212, 213, 423 rime scries (715-719 transformed 215 tasa-Juneosasaal, 717-718 dara watchooses, 3 laukhyg, 937-910 characteristic and 901-902 data morts, 507 data modeling for 902-907. defined, 900 distributed, 910 enterense-data, 902 Exterated, 910 functionality of 910-911 problems and open issues in 1917-914 views cervas, 911 vertual Japa, 202 database administrations (FIGP, 12 unterlaces for -34 security and, 734 database applications, 49, 52-54, 255, 267 database deneraris, 13 database management systems. Sov providotabase programmine approaches to, 261+263 unpedance misinarch, 263 Linignages, 262 sequence interaction, 265–264 Jatab ise programming languages, 255 Janabase schema, 27–28, 115 database servers, 36, 264 database state, 28 database sestems across 19 as use drubble reclinities, 3 characteristics, 8–11 deputrice, 19 environment, 35-18

multimedia dituloses, 3 obsects rented, 10, 16 object-relational, 10 overview, 1-4 real-time database technology, 3 single example 3,764 three-schema architecture, 29-31 maditional applications, 3 induties, i6-37 database utilities, 36-37 dataiouses connections to: 206 constructing, 3 defined, 4 Jećnims, 5 large, 362 loaded, 24, 335 nonemenariae. 7 inebile, 916-925 muhamedia, 780-782, 924-940 personal, 365 pupulare2, 29 sharing, 5 spanal, 760-762 storage of, 414-415 UNIVERSITY database example: 101-103 Dytalog notation, 787 date data type, 212–214, 423 LBA [cardose administrators], 12 interfaces for 34 secondy and, 714 19508 (Juliahawa ngapagangen) systemas), 43 advantages of 15-20 catalog, 9 contrained and chemissover as hitertures for 38-42 classification, 43-45 communent modules, 15-36 database design, 190-193 TYPOME (discributed (WVI9), 43 defined, 5 disadvantages, 23 general purpose, 41 interfaces, 9-34 languages, 52-55 legacy, 209-710 mistriuser, 11–12 personnel required for, 12-14 platino, 562 procedural program code, 19 RUSSAF (relational database in monogeneeps asserva). (2) special purpose, 5, 44 1978915 Idistributed (19805), 43 P14 Trhats delivation bargange's 52, 157 deadlocks, 592–594

decision superior systems (1963), 900 Leclarative assertions, 156 écolarative expressions, 173 declare section, shared variables, 265 decompositions. See relational decomposition deduction roles, inferencing, 19 deducing database systems, 19 Datalog notation, 387 Errorn clouses, 787-789 interpretations of rules, 789-791 werview, 784 Prolog/Datalog notation, 134–167 returional operations, use of (193–193 default context, 271 deterred update rechniques, recovery concepts, 612. 64b-621 Jeppee or homogeneary, 515 sectoral automotors 815 of policing, 127 relationship, 105 "PERTERIORMERAL 247 Deletesperation, 147-143 deletion anonulies, 300 deletion matkens, 150 deletson spectrum, 606 DEVI (digital elevation model), 911 skus maliatura, 540 dense underses 457 dependencies functional, 324-312 anclusion, 354-355 JCHC 353-354 muleyolaed, 547-353 template, 455-457 dependency-preservation, 313, 315-336 3. rlayment diagrams, 398 Jenved armbures, 56 derived horizontal (assuremention, of th Jenved tables, 255 1365 IDate Energytion Standard1, 749-750 PESC keyword, 228 description records, 277 descriptors, 204 design, darahase design active database systems, 761–763 automatical sociation 401-405 ceptralice3 scherna design approach, 572 conceptual design, 52, 571–580. slata-driven, 367 data medel manageros, 383 database designees, 13 slatabase tuning, 369 [388] S. Charless, 380–381

design methodologe 301 19. design, 71-71 local Jesien, 52 logical database design, 365, 393 physical, 52, 369, 343-384 process-driven, 367 Relation Rose design tool, 395-399 requirements collection and analysis, 369-371 system designets, 14 system implementation and running, 954, 355. 10000g, 543-544 CMa diagrams as aid to, 385–395 L'inversay durabase design example, 493–494 energy integration opproach, 372 design autonomy, 617 dugningmatic notificing, 60 mixtely, 947, 949 ർധ്യാതം data (kw. 52) secondor, 57 dictomory, 115 Jugital electricity model (1989), 931 Juntal suggetutes, 751 dagu di serosan ayadehna (1938), 932 Junenson tables, 924 Jarected graphs, 570 discretionary access control (04c)1,735-740,743-744 discrotonating mathema, 201 Lisermonator net ML terranology, 76 disjonations constraint, 93 Juska evlunders, 410 devices, hardware descriptions of, 415-420 slock blocks, 417 Josk controllers, ADJ disk drives (419) Jisk packs, 415 double-saled, 415 ble records on: 422-427 hasd-bend, 419 connections, 417 iomalicanico, 417 magnetic rape storage devices, 470-471 parameters, 4, 951-953 read commonds 417 seading to Jurial, 414 June 1, 805 orighesiskyl, 415 tracks, 410 white commond, 412 distinct data types, 713 distributed that thase systems advamages of 805-805 allocation # 2-815 concurrency control, 825-827

Jaca traementation, 810-813 dara replication, d12-915 Innettens, SC8-8C9 in Orield, 830-833 overview, 804-805 parallel rerves distributed to tradegy, 805 query processing in, 818–824 recovery, NZr rhree-nor cherr server architectury, \$27, 529. rypestal, 815-818 distributed DBMS (DOBMS), 43 distributed watchease, 912 distribution transponence, 329 devision ()) operator, 227 OPTHON operation, [63-165 DMc (dam manipulation language) 32-33 Document Type Dominism (1979), 849 documents, 113 availers, 645 XML: 855-859 doment-key normal for no GoSNE), 557 demand of knowledge, 110 doarane of attributes, 75 constraints, 131 Desical definitions of 127 situatured, 75 det notation, 852, 671 deubse butfering, 421 Jouble-sided disks 415 downward closure property, 873. doning, mobile cuveronments, 919 Juli alown display, 904 driver manager, 180 1900P command, 217, 762 1360° VIEW command, 259 1999 (decision support systems), 920 15719 (Destangent Type Definition), 849 INTEE Edigonal Gerrain, novidelling), 982 doplicate el initiation, 151 dynamic files, 429 Jygamia SQL, 256, 770–271

# F

e-continent, telectricate conducerse), 21, 745 control version, 19 Fr. 5 model, active database systems, 757–760 LEPROM (Electrically Erisidele Fregosine shie Bend-Only Memory), 415 LLR (Enhanced-ER) model, 50, 36, 600 model concepts, 103–104 model concepts, 103–104 model constructs, napping tradations, 199–202, 208 electronic commerce (accommerce), 21, 745

embedded -cpt, 256, 362, 264-369 empty state, 29 encapsulate set (49-653 encoded data (713) categorie di 733 AIS (Advanced Encryption Standards), 749 DDs (Data Encryption Standard), 749–750 public key, 750-751 rud mes. 643 cial transactions, 553 and users. 13-14 cognocis, software, 14 unbaneed UR 85 Embanded-ER (EBR) model, 50, 86 medel concepts, 103-104 model constructs, matiping to relations, 199-202, 206 Emorprise Resource Planning (ak.), 447 enterprise wide data warehouses 902 ellinism attributes and, 54, 51 defined, 27 earning cases in constraint, 148 entiry sets, 57 entory types, 57 generational, 103 key arrebutes of 57,38 mapping of regular, 194 owner optits, 68 regular, 65 strong, 65 subsclass of 46 weak, 58, 58-69, 194 entropy, Mi4 environment records 277 TQUIJUN operation, 161–162 equivalence, of sets of functoread dependencies, MO-MD 10 database schutiwe, 70, 947-343 18 design, 49-73 LR deservation 50 sternolive neormetics for, 73-74 summary of notation tor, AP-71 TR (Entity-Relationship) model John J. 49 FR-to relational mapping, 192–199 ER.: (Enterprise Resource Planning), 447 LVenss, 257, 768 exception observed 111 execution autonomy, 315 existence dependency, 60 exterential quart het, 176 sports examples, 177-178 transformations, 178-179 XXATS for contraction 234-236 explicit sets, 736

w30ressions declarative 173 F.WR, 864 rsth. 656 -afc, [8] urisale, 181 expressive power, 174 «Monded Markap Language, See RNL extended relational systems, 44 extendible hashing 439-441 ENTENIS Revealed 677 extensible data types (712-714 extension, wheinas, 29 external bodying, 437–439 external schemas, 50 external soming, queries, 496–495

#### F

two-debried preducines, 791 taan tables, 924 tacieny objects, 575 tailed state, transactions, 260 talares, maps charp, 558-559 PALSS enhies 229 FDPS (Indefated database system), 8[6] tederated warehouse, 910 tex-flood karps, Autobase design, 367 FETCH command, 255 letch speciation, 709 hells clustering, 459 connectine, 442 optional, 424 erdening, 431 repeating, 423 50th in smool form (586), 353–454 file processing, b tile servers. 3d áles. access, 429 Hurcks, 426–427 dynamic, 429 expansion, 439-442 gral files, 484–485 h.ish 434 headers, 427 Scup, #40 ralesco, 17 up.p. 435 master, 435 mixed, 424 operations on A27-A29 opdered, 437-434

organication, 429, 442, 443 overfam, 433 rate, 440 record-stransmole operations, 425 reorganization, 37, 430 acards 498 segments, 427 service or intractive rations, 428-429 2011eJ. 431-444 sorring, 430 static, 429 transaction, 453 to once applications, data numing, 891 test level induced indexes, 465 hist recondly (m. (1997), 141, 315, 315 freed-bear disks 419 fased leasts, 916 weed-length accords, 473 flash memory, 413 flat relacional model 131 flow control (733, 747-749) sEWR expression, 464 EOR OPENATE OF clause, 269 force/no force approach, recovery techniques, 614 force writting, 562 BOREICK CLUSE, 215 foreign key, 148 formary disks, 487 douvigns, 127 6 cm-tring styles, \$45 Kainsi 34 forms-based interfaces, 54 forms specification languages, 54 framul*i*is, 175, 182 fourth normal form (461), 351-355 fragmentarios data, 810-812 handomal, 813-817 named, 512 ситыратетер, А.И. certical ALL to spient-partern pre-algorithm, 8/5, 8/6 100% (Lause, 219-220) TELLOS TELEJOIN OPPORTUDE, 170 time transfishependencies definitions of 304-306 equivalence of sets (4, 310-311 interence rules, 106-310 numinal sets of 311-012 rimerional requirements, 52 functions. See also commands: operations upping are, 248-240, 509-511 AVERAGE, 165

AVG, 234, 240 ( CLN ), 239, 240 defined, 59 ( CNN ), 253, 250 MAX, 244, 240 ( MAXIAI, M, 165 MAXIAI, 238-240 ( NICCH, 238

#### G

generalization. See also specialization, 90-91, 103. constraints on, 92-94 Jefinee, 50, 112 hierarchies, 97 hierarchies and lartices, 94-97 Luckey, 97 mapping, 199-201 in returning conceptual whemas, 97-98 generalized eprint regel 104 generalized superclass, 90 genetic algorithms, 690-691 generate data approximent. hernformatics, 937 hological sciences and genetics \$10-\$16 human genome project and biological databases. 94(-943 resources for, 944-945 geographic information systems. Ber Obgeographic insibility domain, 917-918 GIS (geographic internetion systems) -3, 35 appleanons, 940-941 ARE VINRO SOLDELIN, 954-935 data management populationents of \$11,932 data openations, 931-934 problems and future issues (0, 935-936 resources for, 936. closure 115 100 ANT COMMOND, 737-738 grantine privileges, 730-735 graphical user interfaces (GUS), 18, 34, 181 graphics, multimedia Jaca, 92J graphy acoche, <del>14</del> directed, 570° precedence, 570 predicate dependency, 795 gaum, 512-513 senalgation 570 eersoon, 662 errd (dss, 494-465

...201 p. 65 of a set 240, 243 group ang ang ang ang 240 groups 825 set is (graphical area interfaces), 16, 34, 521

# Н

inandles, 277 hastung techniques dynamic file expansion, 439-442 extendible, 434, 441 external, 457–439 internal, 434–437 Lipsun 441-442 paritaged, 885, 484 static, 458. HAVENG CLAUSE, 240-243 heaters Jocuments, 845 63es. 427 boolth one opplications, data mening, SML hearailes, 430 hierarchical and network systems, 20 Sherarchic al data models, 27, **9**3 hierarchies acyclic graphs, 44 association tiles dataing, 679 generalization, 94-97 spectalization, 94-97 high-level data involutes, 26 Jagh-Jevel Dk(U, 35 higher-degree relationships. Secretions relationships bemogeneous (Daby)5, 915 herponym-, 578 lorosental tragmentarissa, ECT, 810-911 log means partitioning, 544 horrschen propagation, 740 Hom clauses, 187-789 lust languages, 34, 264 hoses, toxal 516 HTS3 (ThereText Marking Language 5, 21) human genome project and biological dorshages, 940-943 hyperlinks denned, 21 Accuments, 841 TyperTest Markup Language (11081), 21

# 1

ulentnication data manage 209 defated, 111 identifier authorstation, 209 defined, 111

identifying entity type, 6d mage arritoine, 201 mage data types, 715 dia adriana rapisha daju, 923 uster metas, 933 storage and territy of of, 22 unincluite update techniques, recovery concepts, 612, 622-624 meedance manatch, 17, 263 implementation active database systems, 701-764 Burthase design, 354-385 views, 259, 261 implementation data models, 27 implementation level, relation schema, 294 maplementaria. DBMS environment, 14 inclusion dependencies, 554-355 incremental opdates, 260 independent (Lesses 115 indexed allocation, 427 uslears brung indexing, 936-907 clostering, 459-462 concurrence control in, 60% 1976 defined, 17 jour, indexing, 907 lognial, 485-496 multilevel, 464, 469 physical, 485–486 ppmag, 457-459 secondary, 462, 464 ton ng 540-545 types of, 456. interences, HC, 550 information repusitories, 37, 364 information resource management (1884), 462 information systems database application life evole, 565-566 information system life cycle, 164–165 role in equilibriums, 362 -364 information technology IIT3, 352 Internet Universal Server, 711-712 inherance, ules (mensual dependencies, 306-348 inheren model-based constraints, 133 (aberitance arrebore, 86helptviet, 677 multiple, 92, 98, 202, 660-661 relationship, 86 wlends of 601 single, 92 support for, 714-718. ivre, Bo wpe bienirchy and, 65**4**–656

mmal states 29 ing tah Astrony, disks, 📲 🗸 innerture), nested querses, 212 Jaseri operation, 141-142 mettion anothelies, 299-350 ansOnces defined 28 relation, 128 variables, 642 instant able interfaces, 676 instantation defined. 111 polyinstantiation, 742 integrity constraints, 18, 135 intention, schemas, 29 pressrive interfaces, 201 interactive transactions, 607 micrbls ck gaps, 417 instantances 10850S, 13-14 defined, 10 unstantiable, 676 interactive, 261 multiple user, 15 noninstatiable, 676 user frien.lly, 33 Work, 262 Internationally Synchronized Database Environment 4150PF5, 921-922 roternal hashing, 434, 430 internal schemas, 29 interreciptions 710 interrolations 933 INTERSECTION Operation, 155-157 entervol data types, 213 INTER clause, 207 insvalid state, 136 (RM Onformation resource management), 362 18 A relationship, 112 (SDB) Unremirrently Synchronized Database Environment), 921-922 isslation property, 12 IT Inforgation (Appeloge), 362 der Gunergerkers (390) terator variables, 263 -terators, 273

#### ]

AVA 255 Doubler (Borland), 37 (180) class libraries, 250 DBC driver, 260 (con dependencies, 353–354) John mäexing, 907
 JON operation, 159–101, 501–508
 Johns di tables, 217–219
 Johns auchiway, 501
 Semijoin, 818, 823–622

#### к

KDO (Knowledge Dawczery in Darzhases), 858key artificites, 57
key condidate key 135
condidate key 135
composite, 493
composite, 493
composite, 493
primary, 138
primary, 138, 314
surgeney, 134, 314
surgeney, 202
Showledge Discovery in Fundases (K00), 868
kpasikolge representation, 85, 112

## 1.

Jabels, 843 Inneooges, 32-33 data subbanguage, 255 database programming, 155, 253 hust, 264 LANS (Local area networks), 38, 869 large clauseases, 302 Inches, 609 Larrices peneralization, 94-47 multiple informance, 92 specialization, 94, 97 leaf nude, 95 les ming stancered, 482 titisopervises, Mit left-deep trees, 529 "FLET AN THE RANGE OF COMMON. 710 lectacy mostes, 709-740 legal relation states, 305 linear Justang, 441–442 Linear recarston, 708 linear regression, 659 link attributes, 75 inked alƙasiman, 426 Jugks, 75 hterals atomic, iXiN collection, 668 structured, 668

Icaded databases, 29, 485 Lading and av. 42 local area nerovsky (LANS), 58, 409 local armbures, 89 le calidesiga, 52 location transparency, 807 lories. bmar., 184 (185 conversion of \$57-588 (ar. bas, 607 includes gravalarity level (607-604 nor Ima No. 586 read, write, 546 should be closely a 596 two-phase, 588-591 loa records, reprochans, 560-561 log seconde nordser (USN), 626. Logical Nock Address II FAU 417 logical dots independence, 31 logical Jacabase design, 368, 383 logical indexes, 495, 496 logical level, relation schemas, 293 ingreal theory, 115 login sessions, 735 Deslession property, 313, 355-337, 141-542 Loo level dara models, 26

## М

macro life cycle, antorination sostems, 164 munifles 434 multimemory 417-463 his numarice personnel. 14 autodatory second courted, 747-743 MANNET (mobile address network), 215 mental alcohikation, 782 complectoring applications, data instrugg, 891 mappings categories (oncorrespes), 202-203 Standal, 52 3choed, M EER model constructs to relations, 199-203, 206 SP-to-relational, 192-199 shared subclasses, 203 mark op. 344 marker-based data, 871 marketing epplications, data mining, 891 mass storage, 412 massively parallel processing (MDP), 911 master blos, 493 machematical relation, 125-129 SPAX mine (10n, 298-240 MAXIMOM function, 165 IDEBOTS. rache 412

FEPROM (Electrically Erasable Programmable Read Only Memory), 414 H.c.h. 414 mam, 412, 413 6 531 (Random Access Memory 5, 412 shared, 8C5 inenu-based intenfaces, 31-34 mennessichneit, 13-34 iner as biss. 111 ineta-Juni, 9, 39 metadata repearinty, 9[0 morbats, 44 micro-life exclet (information systems, 365) auddle tora three-rier client pervs rorchreceure, 42 909 fonction, 238-240 minutal cover functional dependencies, 311 nomination cardinolity constraint, 67 MINIMUM function, 165 minneerid, 4 MINUS optimition, 155–157 mirroring, 445 mixed files, 424 mixed fragmentation, 812 mixed transactions, 380. Mos relationship type (67-66) mobile ad hoc network (MASET), 918 mobile databases characteristics of, 919-920 computing architecture, 916-915 dara manugement issues, 920-931 18DEF Entermittently Synchronized Database Envir-CONDECT1 921 922 reference cluster; ds for, 927-925 Produktation anomalies, 500 modules Inflet approach, 50 Luffering, L7 client, 35 Data Disde, 712 Jefined, 14 persistencecored, 284 server, 15 stored disk manager, JS MOLAP (multidimensional OLAPI, 911) MPP (massively particle) processing), 914 milindimensional associations, 980-981 initridimensional OLAP (MCCAP), 911 multilevel indexes, 464-469 malrimedia databawa, 3 arplications, 928-929 concepts, 782-783 dara and applicant no. 923-924 dara madagement issues, 924-925 readdress for 1929, 930 spatial darabutes, 780 (782

multiple groundonty level backing, 601–604
multiple inheritoway, 92,95, 202, 663–661
multiple relation options, 200
multiplications (\*) operation, 227
multiplications, 36
multiplications, 36
multiplications, 224
multiplications, 224
multiplications, 40
multiplications, arethogen, 56
multiplications, arethogen, 56
multiplications, 201
multiplications, 501

## N

Nacy relationship types, 108, 199–197 s relativezhip type, 67 name end overs, 14 paired (teratory, 275 particul que curs. 638 names, construitis, 716 participe schenge constructs, 71 nationage intersportering, SQT National Inspirate of Spiralayds (SIST), 742 NoTCIAL KIN operation, 161-162 notural language interfaces, 34 regative association's PSI-SSI prested quipties, 230-231 nested relational model, 728-727 nested relations, 249-316 network data models, 27, 43-44 network partitioning, 825 networks LANS (local area networks), 35, 869 neural, 590 5888 (Storage Area Networks), 445 WANS (wide area nerworks), 509. neural activation 390 MIST (National Instante or Standards), 744 nonaldisave join payony, 915 accomposition, 341–342 Kasa 840, 540 resting binary decompositions for, 336–540 restorstantial le interlages, 676 compresessional language, 173 nonpopulsace sportes, 795 presedopslast albertion, M3 potentatile stronge, 414 normal forais Bence Cashi (N (at ), 524-526 demode 312 Jonnarskey (1968F) 357 nul, (5NP), (51-154

first (1834), 115-118 fourth (45FL )51-353 practical use of 119-314 projectopin (PINE), 354 relation decompositions and inserficiency of \$34, 135 sec.md (284), 318-319, 321-323, ikird (381), 319, 520, 323, 324 กมะเพลในระบบท algorithms, 345-347 denominalization, 540 process, 512 NOT operator, 176, 179 OPTIMUM ames 671 .bar 652, 671 relational data models, 152 out values, 56, 131, 229 py-51-up with 143-145 or tuples, ICI pumeric data types, 212

## ø

Object Data Management Group Securitions shipecedata models, 27, 43 Object Definition Language (0.01), 647–679–684 object diagrams, 387 object identifiers, 249 Obser Management: Group (OMG) (385 Object Manipulation Language 10801, 691 object twodelung, 85 object-oriented database systems, 10–16 concepts, 641-643 ark spulation, 646-660 abject behavior, was class operations, 650–652 object (Jennay, 644) object persi-torice, 652–653 object structure, 644, 647 overcicis, 649-641 polycourphose, 659-60C sype constructors, 647-649 Object (Duery Language 10031), 654-693 object-polations, database systems, 12, 43–44 SOL standards and components, 702-704 superat, 701–708 algerts atomac. 674-676 Stows (Davary Large Clipcus), 424, 658 collection &72--674 complex, 657-659 connection, 281 exception, 111 f istory, 678. pensistente, 652

statement, 251 гылысыг, 652 ore-sletuel, 674-676 us concess, 18 CD8: (Open Database Contactivity), 41–248, 256, 275. OD, (Dhert Dehistory Linguage), 679-684 CDWs (Direct Data Management Group) atomic objects 674-676 Collection interface, 872–674 infrared overview, MG-667 objects and literals, 667-671 CIL (Oncology Inference Loyer), 926 CLAP (online imalytical processing), 3, 204–205, 900 CEPT (critice transaction processing), 12, 43, 930 OMU (Ohjeer Management Group), 385 CML (Clocer Manapulation Language), 603. online analytical processing (OLAC), 3, 208-209, 900. online transaction processing (OLTP), 12, 43, 900 onualog, 110. Criticiogy Interence Luyer (OIL), 926 opaque data types, 712, 713. OPTN CLESCE commona, 267 Open Database Connectivity (OPEC), 41, 24d, 25f, 225 operating system (OS), 35 operations, Secular commands, functions, 52, 75 Jamary relational, 153-162 CARTES AN PRODUCT (158 communative, 156 C10365 FIG 90, C7, 162 det us.l. Delete, 142-143 DARKON 163-165 E0000000000161-162 TELLOGIER JOIN 173 Insert 141-142 INTERSICITION, 155-457 JOIN, 158, 161, 501, 508 LEFT CALIFER JOAN, 170 MINUS, 155, 157 NATURAL KON, 161–162 .XITER (CAN, 169-170 COLER UNION, 170–171 PROFECT, 165-164 REIXO, 619 RENAME, 154-165 JOGET OUTIS JOIN 150 SPEED, 151-153 sequence of 154-155 46T 503-509 UNDS 135-157 ordue, 143 operator overloading, tel 3, 659 - Organola 4NO.176. 79

DPMS environment, 14 sor, 176, 179 OIC 176, 179 optimist concurrency control, 594-600 optimization: cost-kozel que pes, 523-542 dara ja kung 1870 optional fields, 424 Oct (Olyce) Query Language), 684–695 Oli operator, 176, 179 Oracle distributed database systems in \$55,852 O-ocle 8, 721 ORDER for clause, 226 order preservation, 438 indered files, 431–434 organizations, information systems, 562-364 OS (operating system), 35. OF JER JOYN Operations, 169-170rolliet que tres, 239 COTER CNAM operation, 172-171 oventus áles, 434 overlapping 103 concentity type 68

# P

proable] database pranagement systems, 505. patallel processing, 553 patameter insde, 285 parameter types, 285 parametries, stritement, 275-281 passmetric end users, 13, 14 partent modes, 469 partially conjuncted state transactions, 559. portial, defined, 163 pareof key, 68, 318 pare al replication, SD partial specialization, 94 participation constraints, 64, 67 paretion algorithm, 878-879 patterione2 bashing, 464, 464 Parturiorang handwriail, 544 network, 825 vertical, 544 path expressions, 656 partein march ng, 216-228 performance monororing, 57 personent objects, 652 provise of end storage, 16 persistent stors building, 7d4 personal databases, 403 phantom problem, conjunctory constral, 60%

physical data independence, M physical data models, 26 plysical dotabase Jessen, 383-384 decisions about, 589-541 influencing factors, 537-539 olosical Assign, 53 prosical indexes, 495, 486 rele filos, 400 pipelining, 511+512 pisonag, 904. peartorins, 382 pointung device, 34 poly, respansion, 742 poly atorphism, 659-660 populated databases, 79 portability, 665 programmal metators, 274 ReverBudder (Schose), 37 precompilets, 36, 262, 264 predicate-defined soliclases, 92, 103 predicates 137 fact-defined, 791 rule-defined, 792 prediction goals, data mining, 869 preprocessors, 362, 264 presentation layer (three-user client server archirezture). 828 FRIMARY clause 214 primary indexes, 457–459 primary key, 135, 514. primary storage, 412 prime aurobates, 314 printer servers, 49 producy protecture, 746 privideged software, 16. privileges granting, 746-748 honocestal propagatory, 742 corerview, 755 ecokang, 737 rypes of, 745-737 vectical propagations 74C procedural DML, 13 procedural language, 173 mocodural program code, 19 procedures, stored, 284, 286 protess-driven design 367 program-data independence, 9 Program Sissed Modules (35M), 248 programmers, 14 programming. approaches to: 262-263 impedance manarch, 263

multiprogramming, 552 sequence micracitus, 263-264 programs opplication, 262 oftent, 263 project-jour normal form (FJSF), 354 (RoBFC) operation, 155-154 Prolog/Datalog normion, 764-757 projecties of av, 111 manageanus, 362-363 protection, Service Modules), 248 public key energy (so), 750-751

# Q

Q51 (Qamy-By-Example), 150 aggregation 960 database modification in: 260-262 jaracin na. 959-960 remeasis in 1955–1954 outlified aggregation, 76 qualified associations, 76 quantifiers existential, 176-179 umiversal, 176, 178-191 quesies, 544-547 ad-hoc guerving, 407 blocks, 498, 496 correlated mesnel, 242–243 cost-based optimization, 524-532 deromposition, 322 in distributed database systems, \$18-824 existential quantizer 177-178 external caring, 496-498 euphy 512–513 operations 732 must hear on 259 oagarda 638 nested, 230-233 nonactursise 795 retrimention, 532–533 outer, 230 relational absolute 171-176 semantic optimization, 513-504 501,218-245 rrees, 512-515 ura.versal quantatier, 179-151 validation, 493 XMI 962-965 Query-By-Example, See QBE quere compiler, 40

query language, 33 query processing, 17 query servers, 41 queration marks (\*1, 227

## R

RAIN (Redundant Arrays of Independent Daks), 441-447 KAM (Random Access Menuey), 412 range relation, 174 ынын талдо роктемпа, 933-KRAC (referbased access control), 744 ef frotational delay). Jek parameters, 950–952 300088 (in brotio) dotabase management systems), 71 x04 (tesource destription famiework), 926. ised command, disks, 417 read-set to reactions, 554 read Intrestantp, 597 read/write locks, 586 cosming mechanisms, 110 record-at-at-mag DML U file operations, 428 record-based data models, 27 record pointers 455 record types (425) records connection, 277 cenned, 422 description, 277 environment, 277 http://dength/473 quagned, 426 statement, 277 ansganned, 476 values and items, 422. wmpble-length, 425. IN ARTS vRife algorithms 625-629 backages, 630--631 caching of Jak blocks, 523-614 co-cohne rollback, 616 checknomits, 615-616 deterred update, 618-621 distributed database systems, 827 force/no-force approach, 614 immediate updates, 622-624 in muhidatahase systems, 629-640 confirmes and categorization, 612-612 shadow piging, 624–625 swallmosteal approach, 614 174nsaction rollback, 616-617 massacrions, 558, 554.

GMEN/REDC algorithm, 613 white-alpead logging, 614 recovery and buckup systems, 17, 37 recursive closure, 168 reconsidentel attoriships, rade names and, 64 Retions 619 edosdoney, controlling, 35–16 Reduction: Arrays of Independent Disks (RAID), 443-447 referencing relation, 136 referencial integrity constraints, 355 referential triggered action, 215 reflexive association, 76 regression tule, 589 regular ecnary cypes, 65 Relation Rose triol, 395-399 relation: schema, 127 barrom up design methodology, 294 implementation level, 294 logical level, 293 semannes, 295-298 top down design methodology, 294 ruples, generation of 401–304 ruples, call values in, 501 ruples, estimating internation in 298–301. ielational algebra segregate functions, 165-168 CARTS AN EROPECT specifion, 158 ) BOSS SKOLAUT OPERATION, 162 defined, 149 20/3008 operation, (65-165 EQUJOIN operation, 161-162 extrement, 149 INTERSECTION Operations, 155–157 MINUS operation, 155–157 IN VTURIAL JOIN operation (161) 162 OUTER (OIN) operations, 164–170 OUGER ONEON operation, 170-171 PROMAT operation, 153-154 cylery disamples, 171-173 recursive closure, 168 RESAME operation, 154–155 SELECT operation, 151, 155 transformation jules, 518-520 UNION (peration, 155–157 relational calculus defined, 149-150, 174 domain colodia, 181–184 esistential quantitiers, 176 sale expressions, UPL un versal quantifiers, 176 onsate expressions, 181 nelational data invidels. constraints, 133–140

flar 131 noration, 1/2 overview, 120; update operations, 140-145 relational Line modules, 27, 43 colarional dutubase design. See also design, 196-197 atterrahma for, 340, 347 EER model constructs, mapping to relations, 194–203 ER to-relational inapping algorithm, 192-199 relational database management systems (RP9049), 21 relational database schema, 135 relational decomposition matrivatured dependencies, 147-353 properties of, 154, 140 queries, NZZ relational DLAP (ROLAP), 521 relations all-key, 150 characteristics, 129-132 extension, 128 instance, 178 approx.128 interpretation, 131–132 mathematical, 175-129 ocstol, 249, 316 tangs: 174 referencing, 138 sigar, 129 vot.ol. 210 to stronghig athendance, 86 relationships instance, 61 relationship sets, 61 ectionship exces attributes of . 67-68 constraints on, 64-67 defined, 61 Server of all specific, 89 rehability, 50% RENAME operation, 194–195 renaming attributes, 236 reorganication, hies, 450 repeating fields, 423 replication, 812-815 advanced, 831 has a dife symmetric, 831 replication transparency, 407 representational data models, 27 responsible to the company and analysis, 50–52, 369 requirements specification technologies, datalese design 370 cossing description transework (IUCE), 976 restrict cost unantificatived access, 16

retries al operations, 427 rentieval aninsierious, 380 aerorevais, 140 coverse organizering, 204, 307 PEVORE command, 737 rewrite time, disk parameters, 953 RIGHT OCTIR ICTN operation, 172 ROLSE Invlational OLAPIE 911 role-based access control (RISAL), (44 role numes, recursive relationships and, 64 roll up Jisplay, 904. ronamonial delige (add. disk guarmeness, 951–952 row dara reges, 713 new level magers, 760 rule-definist predictaes, 792 rutes, 19 rantume database processor, 16

## 8

s (seek totoe), disk jat algebra, 951 and expressions, 185 sampling algorithm 574-575 5 NN (Statuge Area Networks), 447-449 school please transactions conflict car walent, 5fel conflict sensitiability, 570-572 delatoredit gan actions, 576. 1883 YANK 503-504 n coverability 565-566 weal: cauntalent, 564 small fitts scenability uses of \$72-\$75 seruhroble, 568 view schalizable (575–576) schema change structure pote 317-218 conceptual, 30, 52, 45, 98 constructs, project maintne of, 72 Janabase, 115 ER database, 70 extension (x), 29 external, 30 intension of 79 internal, 29 rolmon, 127 relational darabase, 135 snowdake, 905  $\approx 0.209$ star 905 SML 810-815 a jama bised contraints, [13] schema construct, 28 schema diagram, 28

schema evolution, 29 scherna name, 209 scientific applications, 22 serum functions, 545 ses. (Small Compater Storage Interface), 419 SIA. (storage definition Lungiage), 32 search trees, 470 searches, binard sounds [45] record level, multilocel indexes, 4n6 oso nd normal torin (ZNF), 318-319, 321-323 secondary indexed 462,464 secondary storage, \$12, \$15-422 second vectors protection, 734-735 and its 735 authorization subsystem and 16 (080) and 734 digital sensitizes [75] COCOTEMO, 749-751 rism control, 743, 747, 749 Ligin servicis, 755 protection. 5 statistical database, 746-747 rhreats, 734 types of 732,734 seek time (s), disk patana ters, 951 560 Full Clause, 219-020, 496-501 where drops where black, 219-221 SELECT OPERATION, 151-153 when there is addresses 220 relective inheritorice, 661 self-describing Jara, 8-2 semantic data modeling, 55 semantic query optimization, 533-534 Semantic Web (113, 926 semantics, 18, 295, 298 sompoins, 818, 521-622 semistructured data, \$42 sectors. (Structured English Query Junguage), 278 requerce dugrans, 52, 489-490 sequencial accessive views, 420-421 sequences partering 583 serial schedales, 568 semplization graphs, 570 server modules, 25 Servers application, 42 data 41 database, 263 Jeffned, 40 o mais, 59 hler, 35 promer 49 querc +1

specialized, W transaction 41 \$666, 19, 42 solitated subject DML 33 hte operations, 428, 429 SET operations, 509, 509 set of enrod that , 15 set reces. 44 set as multisets, 234 aubles as 124–225 shocksonie, 445, 624, 625 should databases, 5 shated disk, 805 shared/exclusive locks, 588 showed memory 505 shared subclass, 95, 202 shared variables, 204 signanines, 650, 751 single arrthues, 35-36 singer-user systems, 45 single inhermance, 92 single-relation options, 200 single-sided Joks, 415 single-valuest situation, 56 Small Comparen Storage Interface (SUSI), 419 SOP (spin nettic multiprocessin), 914 Snapshot Refresh Processes (5818), 412 snarshuts, 28 snowtlake schema, 905 SNES (Sciapshor Refresh Processes), 832 whiware contrainications, 54 concurrency control, 12 mixileged, 16 software engineers, 14 sophisticated asets, 13 sorred bles, 433-444 spanned rec. adv. 426 sporse indexes, 457 spatial applications, 23 spatial databases, 740-752 specialization. See also generalization, 88–90, 123 attribute-defined, 92, 104 constraints en . 92-94 defined, 86, 112 hierarchies and lattices, 94, 97 mapping, 199-201 parrial, 94 in rehning conceptual schemas, 47-98 total, 63 specialized services, 58 specific attributes, 69

specific relationship types, 83 -receivestion, (15) spannous tuples, generative (s), \$21+304 52,92,703 AU 499, 208, 766-767 SQU/CL1 (Call Level Interface), 256, 275 xxx selacina, 239 SQU (Structured Query Language) constraints, 213-217, 256-257 data isres, 209-213 database programming, 261–264 DELLI E COMMAND 247 discussed, 207 dynamic, 256, 270 - 271 embedded, 256, 264, 264 INSERT command, 245–247 quertes, 218-245 schema enange statements, 217–218. 820 171-175 stored procedures, 284-286 syntax submoory, 350 manuana sippar, 576–578 1 PDA15 comptand: 247-248 views 257-261 5.00000E variable 266 SOLST VEE CADABLE, 266 atan disali yasi mwa sa 13–14 autschema #15 start togs, 644 state constraints, 140 stated bar dog-orgs, 502-542 statement-level (nggers, 760, 763-766) statement objects, 281 statement parameter, 278, 285 statement records 277 stationarity. URLE, 285 CREATE ASSERTION, 256 embedded, 262 statue database programming approach, 275 status files, 429 statu hashing, 438 statistical database security (46-74) steau no sreal approach, accovery techniques, 614 stotest cupating 413 of databases, 414-415 Interarchics, 412-414 magnetic rape devices, 420-421 mass strange, 412 nonvolardo: 414 persetence 16 primary, 412 9 No (Storage Area Networks), 447-449.

505% (Small Compute: Storage Interface), 419 secondary, 412 secondary storage device, 415-420 velanke, 414 Storage Area Nerworks to ANS), 447–449 stopso: channels, 748 starage definition language (SPLI, 32 stored attributes, 56 stated disk manager modules, 15 stated procedures, 284, 286 stream based processing, 511-512 strong ender types, ob-Struet Reyword, 668 structural constraints, h7 structured complex objects, 658-659 structured dam, 842 structured damain, 75 Securitized English (Que y Longituge (HEQ, BL), 358 structured breads, 668 Structured Overv Language, See Sec. subclasses, 86-87, 90, 103 condition-defined, 92 predictre-Jenned, 92, 103 shared, 95, 202 escodeburd, 91, 103. sofestoing pottern matching, 226-228 solution (+) operator, 227 SUM Tradictione, 165, 258-242 superclasses, 86–98, 90 (03) seperkey 58, 154, 314 superuser scorunts, 234 supervised learning, 382 יוופרייו esternol data sources, 717 indexing exicitsions, 717. ipherance, 714–7180 n-er-defined functions, 714 sittogno key, 202 synumetric multiprocessor (58/P), 411 symmetric replication, 841 чиончив, 376 system analysis, 14 system designers, 14 system lock tables, concurrency control, 584-586 system protection, 5

## Т

tables base, 210 deficed 255 cubristin, 204 fact, 204 pured, 217–216

is sets, 404, 224, 226 errnal, 295, 238 tags, 644 unimbutes 845 eral 543 mark or, S+4 start, 5/4 tape drives, 420-421 taps (recb., 422, 431) tapes, archived, #21 taxoromys 1.5 remplate dependencies, 555-357 Temporal Juri Fases, 767-768 guerying constructs, 778-779 nme representation, 768-750 rime series data, 780 remainized states transiemony, 560 remary relationships, 83, 105–115 IC:XF data repest 719-723 multimedia data, 923 thesautus, 115 that a level, multilevel, indexes, 460 the Josephal Katal (564), 319-370, 321-374 directs, 753 three-schema architecture, 39-31 three-tier client/server architectures, 42, 627-829 rune data types, 212-213, 423 time representation data municip. 549 temporal databases, 768-770-782 tune settes applications, 22 tuno series data types, 716-719 time-ramp data types, 213 rinksionplattering, 544, 596 CD4 (intragolar imegalar network), 911 tool developers, 14 iouls. promoted database design, #01-405 concursion tools, 17 Jobs forming, 591-694 Relation Rose, 345, 309 top-down conceptual referement process, 95 top down design methodology, relation schemes, 204 toral participation, 67 ictal specialization, 93 tracks čísky, 410 redutional database applications, 3 transaction nles, 453 transaction follback tos overy, 616-617 its ownition servers, 41 trapsections, 52 Negin, 553 contest, 262

commulperand 56] 562 concurrency control 555-557 defined, 12 end, 555 autors, 558-559. identifying functional lichwayers, 374 interactive 607 mixed, 380 processing systems, 351-552 properties, 562-563 read-ser, 554 relowers, 558, 559 retrieval, 380 schedules. So schedules of transactions single over decas molectory stems, 552-553 ≪X.510 575 system concepts, 559-502 waters log. 563-561 unrepeatable read, 557 optate, 140 white-set, 554 (constructions, 178-178 pariment objects (652 transition constraints, 140 transitive dependency, 219 manscarency distribution, 929 frugioentation, 507 location, 807 naimina, 807 replacation, 307 ffee structures Julence, 512-515 subtrees, 469 XKU, 846-898 Lines-8+-trees, 474-481 Princes, 478, 474 Icri-Jeep, 529 search, 470 mangular cregular network (TIN), 351 rriegors, 140 events, 257 granularity, 769 news seen, 700. in SQL991 766-767 statement level, 560, 763-766 tracial ACO, 549 TRUT Colors, (29) trath values 176-187 tuple-based constraints, 216 turles dimeline 343-345 male.plc, 267-269, 270-279

null colors nr 301 induction intermetion in 298–404 adation (29-13) spirmos, generation of 101-104 two-slupensional data types, 717-718 two-ties sheri/server on Interture, 41 (Writeway preparation) type hierorchy constraints en extents, 656, 666 unkentanov and 654–656 ruse inheutonce, 58 type batter, 660 types lara types, 42 i pataeneter, 285 record types, 425

## υ

CMI, diagrams activity, 992 class diserains, 386-587 collaboration, 390 component, 387-358 us database application design, 486–497 deplocinent, 358 as design specification standard, 385–386 obrea, 387 sequence, 159-390 statechum, 490-392 uso case. Sob USIL (Universal Moneling Language), iC. 74, 76 unary operations, 150 DOMET operation, 153, 154 SELECT operation 151-154 raymithmized access, restricting, 16 UNDOWERO algorithm, recovery techniques, 615, 623 imparter stallassications, 26 narrow compatible, 156 1 Slots operation, 155–157 amon type 94-100, 202-203 NOUE tone (note 215, 243-249) an sparages constraint, attributes, 57 Discretes of Micheling Language (COMP), 5P, 74-76 on versal quantifier, 176 speeves updes 179-181 transloundooris, 176-179 on versil relation, 334-535 answerse of discourse (UoD), 4 on-ole extressions, 191 unspanned records, 426 uner ochred dar 1, 843 i unsepervised learning, 845 COD (intriverse of discourse). A

opslate anomalies, 294, 500–507 GPOATE command, 247–248 opslate operations, 143, 427 opslate transactions, 385 update transactions, 385 updates, stens, 259–261 use doe diagrams, 388 user-lefined (updates, 7, 4 user-lefined objects, 674–676 user-lefined logicates, 674–676 user-lefined logicates, 7, 4 over-transfly root-takes, 33 utilities, 34–37

## v

walid space, 29, 136 validation concurrency control, 599-600 quenes, 493 value wits actributes, 54-60 variable deright records, 423 variables communication, 206 instance, 642 irerator, 203 shared 204 SCEA 755, 266 SQUESTATE, 266 VDL (view committion language), 42 version graphs, tibž vertical tragmentation, 80%, 813 verrical paritien, 153, 544 vertició propagation, (40 rideo sources, 783, 923 view definition language (VDL), 32 VIEWS contenista, 257-258 CREATE VIEW command, 258 dara warehouses consus, 914 DROUVERW command (259 involvement in a singligibility, 254, 261 incremental optimes, 260 specialization (4, 255–259 snew materialization, 259 Critical Jaca, 11, 902 virtual relations, 210 virtual storage access medical (VSAM), 456 curred tables 255-259 volatile storage #14 V55331 (vartual storage access method), 486

## W

W 12s , wide area networks), 839 workcuses, See Zao warehouses weak entity type, 59, 68–69, 194 Web access control patiends, 245 eccommerce and, 21
Web-based over interfaces, 34
Web versery, 39, 42
Web versery, 39, 44
Web versery, 39, 44
Verser, 30, 39, 54

## х

880. (eXtended Markap Language), 22, 45, 441 documents, 546 documents and databases, 855-862 hierarchical data model, 846, 849 querying, 862-866 solvents, 850, 855 well-turned and value documents, 547, 550